# Advanced Concepts in Lumbar Degenerative Disk Disease

João Luiz Pinheiro-Franco Alexander R. Vaccaro Edward C. Benzel H. Michael Mayer *Editors* 



Advanced Concepts in Lumbar Degenerative Disk Disease

João Luiz Pinheiro-Franco Alexander R. Vaccaro Edward C. Benzel H. Michael Mayer Editors

# Advanced Concepts in Lumbar Degenerative Disk Disease



*Editors* João Luiz Pinheiro-Franco Neurosurgery and Spine Surgery Hospital Samaritano – São Paulo Brazil

Alexander R. Vaccaro Rothman Institute Thomas Jefferson University Medical Center Philadelphia, PA USA Edward C. Benzel Cleveland Clinic Neurological Surgery Cleveland, OH USA

H. Michael Mayer Schön Klinik München Harlaching FG Wirbelsäulenchirurgie München Germany

The Work was first published in 2010 by Dilivros Editora Ltda., Rua Dr. Satamini 55, Tijuca, Rio de Janeiro, RJ, Brazil with the following title: Conceitos Avançados em Doença Degenerativa Discal Lombar.

ISBN 978-3-662-47755-7 ISBN 978-3-662-47756-4 (eBook) DOI 10.1007/978-3-662-47756-4

Library of Congress Control Number: 2015952115

Springer Heidelberg New York Dordrecht London

© Springer-Verlag Berlin Heidelberg 2016

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

The publisher, the authors and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, express or implied, with respect to the material contained herein or for any errors or omissions that may have been made.

Printed on acid-free paper

Springer-Verlag GmbH Berlin Heidelberg is part of Springer Science+Business Media (www.springer.com)

This book is dedicated to the neurosurgeon who inspired my career: my father, Luiz Fernando Pinheiro-Franco, who is fond of saying Natura non facit saltus (Latin for "Nature (Life) makes no leap").

João Luiz Pinheiro-Franco

We also dedicate this book to our patients and our colleagues. Without all of them, this book would not have been possible. João Luiz Pinheiro-Franco Alexander R. Vaccaro Edward C. Benzel H. Michael Mayer

## Preface

After my graduation in neurosurgery, I had the unexpected privilege to immerse myself in 4 years of clinical and research fellowships, working with or simply observing the clinical and surgical techniques of several worldrenowned experts in treating spinal diseases. Being far from my native country, I had at times longed for family, friends, and colleagues, yet at the same time, the experience thrust me into the emerging field of globalized spinal surgery. It was during this time that I first came in direct contact with incredibly interesting people, all of us breaking the barriers of language to make sense of many different concepts, with different ideas bubbling forth from so many gifted minds. Although there was tremendous diversity in our avenues of thought, everyone was driven by the common pursuit of great results.

These were my earliest impressions as a "global spinal surgeon." Reflecting on that time, I can say undoubtedly that the single unifying element of that great diversity in talent and ideas was an underlying modus operandi dedicated to the common goal of serving the well-being of patients suffering from lower back ailments. It was during this period when I conceived the notion of a "puzzle theory" of knowledge: different pieces of a puzzle falling into place. At first, only a few scattered pieces seemed to make sense, others less so or not at all. However, gradually, as more pieces came together, a clear image steadily emerged, more tangible, more solid, and more understandable.

I share this experience with you to offer insight into the origin of this book. In some respects, it is a tribute to the people from around the world who contributed with total mind and dedication to what had once seemed the serendipity of solving the "puzzle" of what today we call *lumbar degenerative disk disease*. This is the wellspring we present to you as Advanced Concepts in *Lumbar Degenerative Disk Disease*.

It is a great honor to share with you the insights and rigor of my coeditors, Dr. Alexander R. Vaccaro, Dr. Edward C. Benzel, and Dr. H. Michael Mayer. They are perhaps the three greatest minds I have ever had the opportunity to work with. Above all, they each possess a particular magnetism and charisma that electrify audiences with their knowledge in a way that influences people to make a difference in the world. They themselves have made many great differences, yet they are modest and sincere in their relationships with both colleagues and patients. I thank them dearly for the time invested and the enormous knowledge shared in the chapters of this book.

Similarly, I extend my thanks to everyone who participated in the production of the various chapters in this book. This dedicated group of men and women from all over the world helped gather a variety of concepts into an extensive knowledge base for our field. These contributors include internationally recognized thinkers of new concepts, creators of innovative techniques and novel instruments, and courageous voices of provocative new philosophies—all at the vanguard of lumbar degenerative disk disease.

Advanced Concepts in Lumbar Degenerative Disk Disease was written and designed for spinal surgeons, neurosurgeons, and orthopedic surgeons: those who are new to the field as well as those who are more seasoned professionals.

Part I of the book begins by laying out the foundations of bipedalism and the importance of the verticalization of the spine, that is, the alignment of the intervertebral disks to bear the weight and function of the upper body. The opening chapter was written by the internationally award-winning French paleoanthropologist Yves Coppens, who also gave his name to the asteroid. Then, Gusmão et al., in fine detail, take the reader through the evolution of the concept of sciatica to what is known today as lumbar disk degeneration. Part I continues with an award-winning German scientist's discussion of the pathophysiologic fundamentals of disk degeneration and the degenerative cascade. Epidemiology is treated by the late Pierre Kehrli, with the section on genetics perhaps deserving the most attention. This subject is examined further in the chapter by Cheung et al.

Part II opens with contemporary advancements in spinal imaging, with a subsequent chapter by Dr. Michael Modic. An experienced team of experts then takes on the controversial theme of diskography.

Part III examines the day-to-day issues faced by surgeons in practice: psychosocial aspects in patient care, work-related issues, costs, outcome measures, and conservative treatments. This section also includes a comprehensive chapter dedicated to facet pain.

Part IV deals with lumbar disk herniation, once disregarded by many surgeons as "just an herniated disk." Here, the subject receives the attention it deserves. This section closes with chapters on scientific considerations, technical operation, and revision surgeries.

Part V focuses on the surgeon's decision-making process in providing individualized care. In detail, it examines when to operate, when to fuse or not to fuse, adjacent disease, biomechanics, techniques to increase lordosis, bone substitutes, the osteoporotic spine, and the advantages of different accesses: frontal, posterior, lateral, transpsoas, and oblique.

Part VI consists of several engaging discussions regarding studies on minimally invasive techniques: intradiscal therapy, endoscopy, spinal injections, use of tubes, disk cell transplantation, and robotic spinal surgery, as well as a comprehensive chapter on the use of spinal injections after spine surgery.

Part VII addresses nonfusion technologies such as disk arthroplasty and dynamic techniques based on pedicle screws and interspinous devices.

Part VIII includes discussions on degenerative scoliosis, the modern concept of sagittal balance of the spine, compensatory mechanisms of sagittal imbalance, and osteotomy techniques. Finally, in part IX, "Lessons from a Life," some of the most experienced spine surgeons today share their personal clinical experiences. This is a valuable resource for all surgeons.

This is just a glimpse of what we have included in *Advanced Concepts in Lumbar Degenerative Disk Disease*. I hope you enjoy reading it as much as I have enjoyed bringing it together for you.

São Paulo, Brazil

João Luiz Pinheiro-Franco

Spine Committee - World Federation of Neurosurgical Societies (WFNS)

Assistant Editorial Board - Spine (Phila Pa - 1976)

Editorial Board - European Spine Journal

Spine Section Editor – World Neurosurgery (Official Journal of the WFNS)

Member of the centenary Academia de Medicina de São Paulo Member of the Elected Board of the Brazilian Spine Society

# Contents

Part	I: Essentials in Lumbar Degenerative Disk Disease	
1	We've Been Standing up for 10 Million Years	3
2	A Historical Overview of Sciatica Sebastião Gusmão, Aluizio Arantes, and João Luiz Pinheiro-Franco	13
3	Lumbar Intervertebral Disk Injury, Herniationand DegenerationMichael A. Adams and Patricia Dolan	23
4	Advances in Lumbar Degenerative Disk Disease Pathophysiology Comprehension	41
5	<b>Epidemiology of Lumbar Degenerative Disk Disease</b> Pierre Kehrli, Philippe Esposito, and João Luiz Pinheiro-Franco	61
6	Genetics of Lumbar Disk Degeneration	67
Part	II: Imaging	
7	Imaging of Degenerative Disk Disease	91
8	Diskography Richard Derby, Lee R. Wolfer, and Milton H. Landers	109
9	Modic Changes and Symptomatic Lumbar Degenerative Disk Disease: Is There Any Correlation? João Luiz Pinheiro-Franco and Philippe Esposito	137
10	<b>Degenerative Marrow Changes: Natural History</b> <b>Biomechanics in Relation to Symptoms</b> Michael T. Modic	159

#### Part III: General Important Aspects When Treating Symptomatic Lumbar Degenerative Disk Disease

11	<b>Psychosocial Aspects and Work-Related Issues</b> <b>Regarding Lumbar Degenerative Disk Disease</b> Clayton L. Haldeman, Andrew Bauer, Daniel K. Resnick, and Nathaniel Brooks	171	
12	Legal Aspects in the Surgical Treatment of Lumbar Degenerative Disk Disease	179	
13	<b>Cost-Effective Spinal Surgery: Reality or Oxymoron?</b> Casey Madura, Daniel K. Resnick, and Nathaniel Brooks	185	
14	Outcome Measures for Spinal Surgery Elizabeth P. Norheim, Steven D. Glassman, and Leah Yacat Carreon	195	
15	Lumbar Degenerative Disk Disease: Workup and Conservative Treatment William Jeremy Beckworth, Diana K. Sodiq, Lisa Guo Foster, and Walter I. Sussman	203	
16	Facet Joint Pain: Presentation and Treatment,Is It a Myth?Stephan Klessinger	219	
Par	Part IV: Lumbar Disk Herniations		
17	Advanced Scientific Considerations for Surgery in Patients with Lumbar Disk Herniation	251	
18	Surgery for Lumbar Disk Herniations	263	
19	Lumbar Disk Revision Surgery Christoph Mehren and H. Michael Mayer	279	
Par Disl	t V: Surgical Treatment of Lumbar Degenerative x Disease: Doubts, Decisions, and Techniques		
20	The Decision to Operate: Advanced Concepts James P. Lawrence and Todd J. Albert	289	
21	To Fuse or Not to Fuse: That's the Question	299	
	Ryan P. Ponton, Eric B. Harris, and Alan Hilibrand		

23	Adjacent Segment Disease: Natural History of Lumbar Degeneration or Consequence of Fusion? Mark P. Coseo, Nelson S. Saldua, Eric B. Harris, and Alan Hilibrand	311
24	How to Obtain the Best Lumbar Lordosis Jean-Paul Steib and Yann Philippe Charles	321
25	Bone Substitutes Alem Yacob, Glenn S. Russo, and Jonathan N. Grauer	331
26	<b>Techniques for the Osteoporotic Spine</b> <b>That Needs Fusion</b> Paul F. Heini	343
27	The Choice of TLIF for Lumbar Interbody Fusion David Straus, Ricardo B.V. Fontes, and Vincent C. Traynelis	355
28	Anterior Lumbar Interbody Fusion (ALIF) Seth M. Zeidman and Daniel J. Hoh	365
29	Interbody Fusion Through the Transpsoas Approach Lee A. Tan, Manish K. Kasliwal, and Richard G. Fessler	383
30	<b>Oblique Lumbar Interbody Fusion</b> Clément Silvestre and Pierre Roussouly	393
31	Instrumented PLIF in Lumbar Degenerative Spine: Principles, Indications, Technical Aspects, Results, Complications and Pitfalls Olivier Launay, Gilles Perrin, and Cédric Barrey	407
Par	t VI: Minimally Invasive Techniques	
32	<b>Degenerative Disk Disease: Stages of Degeneration,</b> <b>Low Back Pain, and Insights on Intradiskal Therapies</b> Jean-Louis Husson, Jean Lombard, and Florian Cueff	423
33	Lumbar Spine Injections: An Evidence-Based Review Byron J. Schneider, Neal Varghis, and David J. Kennedy	441
34	Endoscopic Procedures for the Lumbar Spine: A Comprehensive View Sebastian Ruetten, Martin Komp, Semih Oezdemir, and Patrick Hahn	471
35	Minimally Invasive Lumbar Disk Herniation Surgery with Tubular Retractors: Indications and Technical Aspects Carter S. Gerard, Ricardo B.V. Fontes, Laura A. Snyder, Lee A. Tan, and Richard G. Fessler	491

36	Minimally Invasive Transforaminal Lumbar Interbody Fusion (TLIF): Indications and Techniques Richard Assaker, Fahed Zairi, and Mohamed Allaoui	503
37	Minimally Invasive Operation for Lumbar Fusion, Canal Stenosis, Degenerative Scoliosis, and Spondylolisthesis. Is It Possible? Christopher C. Gillis and Richard G. Fessler	513
38	Spine Injections for Persistent Lumbar and Radicular Pain after Lumbar Spine Surgery Lee R. Wolfer, Richard Derby, and Jeong-Eun Lee	525
39	Cell Transplantation for Lumbar Spine Degenerative Disk Disease Christian Hohaus and Hans Jörg Meisel	575
40	<b>Spinal Robotics: Present Indications and Trends</b> Konstantin Hockel, Gottlieb Maier, Marcos Tatagiba, and Florian Roser	591
Par	t VII: Nonfusion Technologies	
41	Disk Arthroplasty: A 30-Year History Klaus John Schnake and Frank Kandziora	603
42	<b>Biomechanics of Lumbar Disk Arthroplasty</b> Fanor Saavedra, Christopher A. Iannotti, Dani Bidros, and Edward C. Benzel	613
43	Pedicle Screw-Based Dynamic StabilizationDevices in the Lumbar Spine: BiomechanicalConcepts, Technologies, Classification,and Clinical ResultsCédric Barrey, Eurico Freitas, and Gilles Perrin	633
44	A Word from the Inventor of Intervertebral Dynamic Fixation: On Interspinous Devices Jacques Sénégas	665
45	Interspinous Process Fixation for Motion Preservation David M. Benglis Jr., Rishi Wadhwa, Praveen V. Mummaneni, and Regis W. Haid Jr.	681
Part VIII: DDD and Spine Deformity and Sagittal Balance		
46	Degenerative Scoliosis: Surgical Treatment	695
47	The Importance of Sagittal Balance for the Treatment of Lumbar Degenerative Disk Disease João Luiz Pinheiro-Franco and Pierre Roussouly	703

<b>48</b>	Compensatory Mechanisms Contributing	
	to the Maintenance of Sagittal Balance	
	in Degenerative Diseases of the Lumbar Spine	725
	Cédric Barrey, João Luiz Pinheiro-Franco,	
	Jean-Charles Le-Huec, Gilles Perrin,	
	and Pierre Roussouly	
49	Posterior Impaction Osteotomy for Correction	
	of Sagittal Imbalance in Iatrogenic Flat Back:	
	Surgical Technique	739
	Christian Mazel, Laurent Balabaud, and Alexandre Cogan	
Par	t IX: Lessons from a Life	
50	Rational Evaluation and Management	
	of the Patient with Spinal Pain	755
	Donlin M. Long	
51	Lessons Learned from a Life	769
•1	H. Michael Mayer	107
52	Lassans from 35 Voors in the Transbos	777
54	Edward C Benzel Kene Ugokwe and Nina 7 Moore	///
	Edward C. Benzer, Kene Ogokwe, and Anna Z. Moore	
53	History of Lumbar Endoscopic Spinal Surgery	
	and the Intradiskal Therapies.	783
	Yong Ahn, Gun Choi, and Sang-Ho Lee	
54	The Importance of a Registry in Spinal Surgery	793
	Emin Aghayev, Christoph Röder,	
	Helton L.A. Defino, Carlos F. Herrero,	
	and Max Aebi	
55	Lessons from a Life: The Journey of Spinal	
	Neurosurgery in the United States	805
	Brian P. Curry, M. Yashar S. Kalani,	
	and Volker K.H. Sonntag	
		017
Ind	<b>ex</b>	817

## Contributors

**E. Emily Abbott, MD** Department of Neurological Surgery, Center for Spine Health, Neurological Institute, Cleveland Clinic, Cleveland, OH, USA

Michael A. Adams, BSc, PhD Centre for Comparative and Clinical Anatomy, University of Bristol, Bristol, UK

Max Aebi, MD, DHC Hirslanden-Salem Hospital, "Das Rückenzentrum", Bern, Switzerland

Institute for Evaluative Research in Medicine, University of Bern, Bern, Switzerland

**Emin Aghayev, MD, MSc** Institute for Evaluative Research in Medicine, University of Bern, Bern, Switzerland

**Yong Ahn, MD, PhD** Department of Neurosurgery, Nanoori Hospital, Seoul, South Korea

Department of Neurosurgery, Wooridul Spine Hospital, Seoul, South Korea

**Todd J. Albert, MD** Department of Orthopedics, Thomas Jefferson University and Hospitals, Philadelphia, PA, USA

**Mohamed Allaoui, MD** Department of Neurosurgery - Hôpital Roger Salengro, Lille, France

Aluísio Arantes, MD Neurosurgery Department, Universidade Federal de Minas Gerais, Belo Horizonte, Minas, Brazil

Hospital Madre Teresa, Belo Horizonte, Brazil

**Richard Assaker, MD, PhD** Department of Neurosurgery - Hôpital Roger Salengro, Lille, France

Laurent Balabaud, MD Department of Orthopedic Surgery, Institut Mutualiste Montsouris, Paris, France

**Cédric Barrey, MD, PhD** Department of Neurosurgery C and Spine Surgery, Neurological Hospital, University Claude Bernard Lyon 1, Lyon, France

Laboratory of Biomechanics, ENSAM, Arts et Metiers ParisTech, Paris, France

Andrew Bauer, MD Department of Neurological Surgery, University of Wisconsin Hospitals and Clinics, Madison, WI, USA

William Jeremy Beckworth, MD Emory Spine Center, Attending Physiatrist, Emory University, Assistant Professor of Orthopaedics and Rehabilitation Medicine, Atlanta, GA, USA

**David M. Benglis Jr., MD** Department of Neurosurgery/Neuroscience, Atlanta Brain and Spine Care, Piedmont Hospital, Atlanta, GA, USA

Edward C. Benzel, MD Department of Neurological Surgery, Center for Spine Health, Neurological Institute, Cleveland Clinic, Cleveland, OH, USA

Department of Neurosurgery, Cleveland Clinic, Neurological Institute, Cleveland, OH, USA

**Dani Bidros, MD** Lafayette Brain and Spine, Neurological Associates of Louisiana, Lafayette, LA, USA

**Guillaume Bierry, MD, PhD** Department of Radiology 2, University Hospital of Strasbourg, Strasbourg, France

**Norbert Boos, MD, MBA** Prodorso Centre for Spinal Medicine, Zurich, Switzerland

Nathaniel Brooks, MD, FAANS Department of Neurological Surgery, University of Wisconsin Hospitals and Clinics, Madison, WI, USA

Jacob M. Buchowski, MD, MS Department of Orthopaedic Surgery, Washington University in St. Louis, St. Louis, MO, USA

Leah Yacat Carreon, MD, MSc Norton Leatherman Spine Center, Louisville, KY, USA

**Danny Chan, PhD** Department of Biochemistry, The University of Hong Kong, Hong Kong, China

Alexandre Cogan Department of Orthopedic Surgery, Institut Mutualiste Montsouris, Paris, France

**Yann Philippe Charles, PhD** Service de Chirurgie du Rachis, Hôpitaux Universitaires de Strasbourg, Strasbourg, France

Kenneth M.C. Cheung, MD Department of Orthopaedics and Traumatology, Professorial Block, Queen Mary Hospital, The University of Hong Kong, Hong Kong, China

**Gun Choi, MD, PhD** Department of Neurosurgery, Wooridul Spine Hospital, Seoul, South Korea

**Yves Coppens, PhD** Prehistory and Paleoanthropology, Collège de France, Paris, France

Mark P. Coseo, MD Naval Medical Center San Diego, San Diego, CA, USA

**Florian Cueff, MD** Department of Orthopedic Surgery, Centre Hospitalier Privé des Côtes d'Armor, Centre Orthopedique Briochin, Plerin, France

Department of Orthopedic Surgery, Centre Hospitalo–Universitaire Pontchaillou, Rennes, France

Brian P. Curry, BS, MA, MD Boston University School of Medicine, Boston, MA, USA

Walter Reed Medical Center, Washington, DC, USA

Helton L.A. Defino, MD Department of Biomechanics, Medicine and Rehabilitation of the Locomotor System, Ribeirão Preto Medical School, Ribeirão Preto, Brazil

**Richard Derby, MD, FIPP, ABIPP** Department of Anesthesiology, Spinal Diagnostics and Treatment Center, Daly City, CA, USA

**Jean-Louis Dietemann, MD** Department of Radiology 2, University Hospital of Strasbourg, Strasbourg, France

**Patricia Dolan, BSc, PhD** Centre for Comparative and Clinical Anatomy, University of Bristol, Bristol, UK

**Philippe Esposito, MD** Clinique des Diaconesses, Spine Surgical Center -Clinique du Diaconat, Strasbourg, France

**Richard G. Fessler, MD, PhD** Department of Neurosurgery, Rush University Medical Center, Chicago, IL, USA

**Ricardo B.V. Fontes, MD, PhD** Department of Neurosurgery, Rush University Medical Center, Chicago, IL, USA

Lisa Guo Foster, MD Department of Physical Medicine and Rehabilitation, Emory University, Atlanta, GA, USA

**Eurico Freitas, MD** Neurochirurgie C Chirurgie du Rachis, Hôpital Neurologique Groupement Hospitalier Est Hospices Civils de Lyon, Lyon, France

**Carter S. Gerard, MD** Department of Neurosurgery, Rush University Medical Center, Chicago, IL, USA

Christopher C. Gillis, MD Department of Neurosurgery, Rush University Medical Center, Chicago, IL, USA

**Steven D. Glassman, MD** Department of Orthopaedic Surgery, University of Louisville School of Medicine and the Norton Leatherman Spine Center, Louisville, KY, USA

**Jonathan N. Grauer, MD** Department of Orthopaedics and Rehabilitation, Yale–New Haven Hospital, Yale University School of Medicine, New Haven, CT, USA

Jeffrey L. Gum, MD Department of Orthopaedics, Norton Leatherman Spine Center, Louisville, KY, USA Sebastião Gusmão, MD, PhD Neurosurgery Department, Universidade Federal de Minas Gerais, Belo Horizonte, Minas, Brazil

Neurosurgery Department, Hospital das Clínicas da Universidade Federal de Minas Gerais, Belo Horizonte, Minas, Brazil

**Patrick Hahn, MD** Center for Spine Surgery and Pain Therapy, Center for Orthopaedics and Traumatology of the St. Elisabeth Group – Catholic Hospitals Rhein-Ruhr, St. Anna Hospital Herne/ Marienhospital Herne University Hospital/Marien Hospital Witten, Herne, Germany

**Regis W. Haid Jr., MD** Piedmont Spine Center and Neuroscience Service Line, Atlanta Brain and Spine Care, Piedmont Atlanta Hospital, Atlanta, GA, USA

**Clayton L. Haldeman, MD, MHS** Department of Neurological Surgery, University of Wisconsin Hospitals and Clinics, Madison, WI, USA

**Eric B. Harris, MD** Department of Orthopaedic Surgery, Naval Medical Center San Diego, San Diego, CA, USA

**Paul F. Heini, MD, PhD** Spine Service, Orthopaedic Department, Sonnenhof Hospital, Bern, Switzerland

**Carlos F. Herrero, MD** Department of Biomechanics, Medicine and Rehabilitation of the Locomotor System, Ribeirão Preto Medical School, Ribeirão Preto, Brazil

**Alan Hilibrand, MD** Professor of Orthopaedic Surgery at Jefferson Medical College, Director of Orthopaedic Medical Education, Professor of Neurosurgery at Jefferson Medical College, PA, USA

Konstantin Hockel, MD Neurosurgical Department, University Hospital Tuebingen, Tuebingen, Germany

**Daniel J. Hoh, MD** Center for Spine Health, Cleveland Clinic Hospitals, Cleveland, OH, USA

**Christian Hohaus, MD** Department of Neurosurgery, Professional Association Hospital "Bergmannstrost", Halle, Germany

Translational Center of Regenerative Medicine, University of Leipzig, Leipzig, Germany

**Jean-Louis Husson, MD** Department of Orthopedic Surgery, Centre Hospitalo–Universitaire Pontchaillou, Rennes, France

Christopher A. Iannotti, MD, PhD Arizona Neurosurgery and Spine Specialists, Phoenix, AZ, USA

**Wilco C.H. Jacobs, PhD** Department of Neurosurgery, Leiden University Medical Center, Leiden, The Netherlands **M. Yashar S. Kalani, MD, PhD** Division of Neurological Surgery, Barrow Neurological Institute, St. Joseph's Hospital and Medical Center, Phoenix, AZ, USA

**Frank Kandziora, MD, PhD** Center for Spinal Surgery and Neurotraumatology, BG-Unfallklinik Frankfurt am Main, Frankfurt am Main, Germany

Manish K. Kasliwal, MD, MCh Department of Neurosurgery, Rush University Medical Center, Chicago, IL, USA

**Pierre Kehrli, MD, PhD** Department of Neurosurgery, Strasbourg University Hospital, Strasbourg, France

**David J. Kennedy, MD** Division of Physical Medicine and Rehabilitation, Department of Orthopaedic Surgery, Stanford University, Redwood City, CA, USA

**Stephan Klessinger, MD** Department of Neurosurgery, University of Ulm, Ulm, Germany

Department of Neurosurgery, Nova Clinic, Biberach, Germany

Martin Komp, MD Center for Spine Surgery and Pain Therapy, Center for Orthopaedics and Traumatology of the St. Elisabeth Group – Catholic Hospitals Rhein-Ruhr, St. Anna Hospital Herne/ Marienhospital Herne University Hospital/Marien Hospital Witten, Herne, Germany

**Milton H. Landers, DO, PhD** Department of Anesthesiology, University of Kansas School of Medicine, Wichita, KS, USA

Kansas Spine Institute, Wichita, KS, USA

**Olivier Launay, MD** Service de Chirurgie du Rachis, Clinique de Fontaine, Fontaine-les-Dijon, France

James P. Lawrence, MD, MBA Division of Orthopedics, Albany Medical College, Albany, NY, USA

**Jeong-Eun Lee, PT** Department of Anesthesiology, Spinal Diagnostics and Treatment Center, Daly City, CA, USA

Graduate School of Medicine, Korea University, Seoul, South Korea

Sang-Ho Lee, MD, PhD Department of Neurosurgery, Wooridul Spine Hospital, Seoul, South Korea

Jean-Charles Le-Huec, MD, PhD Department of Spine Surgery, Pellegrin University Hospital, Bordeaux, France

Yan Li, MSc Department of Psychiatry, Centre for Genomics Sciences, The University of Hong Kong, Hong Kong, China **Bjorn Lobo, MD** Department of Neurological Surgery, Center for Spine Health, Neurological Institute, Cleveland Clinic, Cleveland, OH, USA

**Jean Lombard, MD** Department of Orthopedic Surgery, Centre Hospitalier de Niort, Niort, France

**Donlin M. Long, MD, PhD** Department of Neurosurgery, The Johns Hopkins University School of Medicine, Baltimore, MD, USA

**Casey Madura, MD** Department of Neurological Surgery, University of Wisconsin Hospitals and Clinics, Madison, WI, USA

**Gottlieb Maier, MD** Department of Neurosurgery, Klinikum Stuttgart, Stuttgart, Germany

**H. Michael Mayer, MD, PhD** Department of Spine Surgery, Schön Klinik München Harlaching, Munich, Germany

Spine Center, Orthopädische Klinik München-Harlaching, München, Germany

Christian Mazel, MD, PhD Department of Orthopedic Surgery and Spine, Institut Mutualiste Montsouris, Paris, France

Christoph Mehren, MD Department of Spine Surgery, Schön Klinik München Harlaching, Munich, Germany

Hans Jörg Meisel, MD, PhD Department of Neurosurgery, Professional Association Hospital Bergmannstrost, Halle, Germany

Translational Center of Regenerative Medicine, University of Leipzig, Leipzig, Germany

Michael T. Modic, MD, FACR Cleveland Clinic, Neurological Institute, Cleveland, OH, USA

**Nina Z. Moore, MD, MSE** Department of Neurosurgery, Center for Neurological Restoration, Cleveland Clinic, Neurological Institute, Cleveland, OH, USA

**Praveen V. Mummaneni, MD** Department of Neurosurgery, University of California, San Francisco, CA, USA

Andreas G. Nerlich, MD, PhD, MSc Institute of Pathology, Academic Clinic Munich-Bogenhausen, Klinikum München-Bogenhausen, Munich, Germany

Elizabeth P. Norheim, MD Norton Leatherman Spine Center, Louisville, KY, USA

Semih Oezdemir, MD Center for Spine Surgery and Pain Therapy, Center for Orthopaedics and Traumatology of the St. Elisabeth Group – Catholic Hospitals Rhein-Ruhr, St. Anna Hospital Herne/Marienhospital Herne University Hospital/Marien Hospital Witten, Herne, Germany Luca Papavero, MD Department of Spinal Surgery, Clinic for Spine Surgery, Schön Klinik Hamburg Eilbek, Hamburg, Germany

**Gilles Perrin, MD** Neurochirurgie C Chirurgie du Rachis, Hôpital Neurologique Groupement Hospitalier Est Hospices Civils de Lyon, Lyon, France

Department of Neurosurgery C and Spine Surgery, Neurological Hospital, University Claude Bernard Lyon 1, Lyon, France

**Wilco C. Peul, MD, PhD** Department of Neurosurgery, Leiden University Medical Center, Leiden, The Netherlands

Medical Center Haaglanden, The Hague, The Netherlands

**João Luiz Pinheiro-Franco, MD** Neurological and Spine Surgery, Samaritano Hospital, São Paulo, Brazil

**Ryan P. Ponton, MD** Department of Orthopaedic Surgery, Naval Medical Center San Diego, San Diego, CA, USA

**Daniel K. Resnick, MD, MS** Department of Neurological Surgery, University of Wisconsin Hospitals and Clinics, Madison, WI, USA

**Christoph Röder, MD, MPH** Institute for Evaluative Research in Medicine, University of Bern, Bern, Switzerland

**Florian Roser, MD, PhD** Department of Neurosurgery, Cleveland Clinic Abu Dhabi, Neurological Institute, Abu Dhabi, United Arab Emirates

**Pierre Roussouly, MD** Department of Orthopedic Surgery, Centre Médico-Chirurgical de Réadaptation des Massues, Lyon, France

Chirurgie de la colonne vertébrale, Croix Rouge Française, CMCR des Massues, Lyon, France

Sebastian Ruetten, MD, PhD Center for Spine Surgery and Pain Therapy, Center for Orthopaedics and Traumatology of the St. Elisabeth Group – Catholic Hospitals Rhein-Ruhr, St. Anna Hospital Herne/ Marienhospital Herne University Hospital/Marien Hospital Witten, Herne, Germany

**Glenn S. Russo, MD, MS** Department of Orthopaedics and Rehabilitation, Yale–New Haven Hospital, Yale University School of Medicine, New Haven, CT, USA

**Fanor Saavedra, MD** Department of Neurosurgery, University of Puerto Rico, San Juan, Puerto Rico

Nelson S. Saldua, MD Department of Orthopaedic Surgery, Naval Medical Center San Diego, San Diego, CA, USA

Klaus John Schnake, MD Center for Spinal Therapy, Schön Klinik Nürnberg Fürth, Fürth, Germany **Byron J. Schneider, MD** Division of Physical Medicine and Rehabilitation, Department of Orthopaedic Surgery, Stanford University, Redwood City, CA, USA

Jacques Sénégas, MD Clinique Saint Martin, Centre Aquitain du Dos, Pessac, France

Pak Chung Sham, BM, BCh, MA, MSc, PhD, MRCPsych Department of Psychiatry, Centre for Genomics Sciences, The University of Hong Kong, Hong Kong, China

**Clément Silvestre, MD** Service de Chirurgie Orthopédique, Centre des Massues, Lyon, France

Laura A. Snyder, MD Department of Neurosurgery, Rush University Medical Center, Chicago, IL, USA

**Diana K. Sodiq, DO** Emory Spine Center, Attending Physiatrist, Emory University, Assistant Professor of Orthopaedics and Rehabilitation Medicine, Atlanta, GA, USA

**Volker K.H. Sonntag, MD** Division of Neurological Surgery, Barrow Neurological Institute, St. Joseph's Hospital and Medical Center, Phoenix, AZ, USA

**Jean-Paul Steib, MD** Service de Chirurgie du Rachis, Hôpitaux Universitaires de Strasbourg, Strasbourg, France

**David Straus, MD** Department of Neurosurgery, Rush University Medical Center, Chicago, IL, USA

**Walter I. Sussman, DO** Department of Physical Medicine and Rehabilitation, Emory University, Atlanta, GA, USA

Lee A. Tan, MD Department of Neurosurgery, Rush University Medical Center, Chicago, IL, USA

Marcos Tatagiba, MD Neurosurgical Department, University Hospital Tuebingen, Tuebingen, Germany

Vincent C. Traynelis, MD Department of Neurosurgery, Rush University Medical Center, Chicago, IL, USA

**Kene Ugokwe, MD** Department of Surgery, Mercy Health Youngstown St. Elizabeth's Hospital, Youngstown, OH, USA

**Neal Varghis, MD** Department of Physical Medicine and Rehabilitation, Stanford Hospital and Clinics, Redwood City, CA, USA

Department of Orthopaedic SurgeryStanford University, Palo Alto, CA, USA

**Rishi Wadhwa, MD** Department of Neurosurgery, University of California, San Francisco, CA, USA Lee R. Wolfer, MD, MS Department of Anesthesiology, Spinal Diagnostics and Treatment Center, Daly City, CA, USA

Alem Yacob, MD, MSc Department of Orthopaedics and Rehabilitation, Yale–New Haven Hospital, Yale University School of Medicine, New Haven, CT, USA

**Fahed Zairi, MD** Department of Neurosurgery - Hôpital Roger Salengro, Lille, France

**Seth M. Zeidman, MD** Rochester Brain and Spine Neurosurgery, Rochester, NY, USA

Part I

Essentials in Lumbar Degenerative Disk Disease

# We've Been Standing Up for 10 Million Years

**Yves** Coppens

#### 1.1 Introduction

Editor's Note In this chapter, Professor Yves Coppens provides an enlightening perspective regarding a field of science that he pioneered. His account, in conversation form, is unique from historical and scientific perspectives.

Dr. João Luiz Pinheiro-Franco has invited me to contribute to this important work on advanced concepts of degenerative lumbar disk disease. As this subject is undoubtedly beyond my field of expertise, he proposed that I elaborate on the developmental factors involved during the human transition from locomoting in quadruped position to the biped upright standing position, which comfortably fits within my academic considerations.

Thusly framed, I have decided to address the history of our human-primate "kinship," that period when Homininae separated themselves from the Paninae, probably for environmental reasons, somewhere in tropical Africa, 10 million years ago. Human beings are, obviously, living beings and as such have their place in a taxonomy of their presumed natural relationships: a chronologically ascending and integrative classification, we are all at once a eukaryote, metazoon, chordate, vertebrate, gnathostomata, sarcopterygian, tetrapod, amniote, synapsid, mammal, and primate. And among primates the taxonomy continues: haplorrhine, simiiform, catarrhinian, hominoid, and hominid. At present, in most scientific classifications, Hominidae include Paninae, which are in common terms the pre-chimpanzees and

Y. Coppens

Anthropological Biology – Natural History Museum, Paris, France

Department of Prehistory and Paleoanthropology, Collège de France, 3 rue d'Ulm, 75231 Paris, France e-mail: yves.coppens@college-de-france.fr

Editor's Note Professor Yves Coppens, along with Donald Johanson and Maurice Taïeb, discovered the now famous "Lucy," at that time, the oldest bipedal hominid skeleton. The name Lucy was given as reference to the Beatles song, "Lucy in the sky with diamonds," which was popular at the time of their excavations and research. Prof. Coppens was the Chairman of Anthropological Biology in the Natural History Museum (Paris, France, 1969-1983). For 22 years (1983-2005), he served as Chair of Paleoanthropology and Prehistory at the prestigious Collège de France. He is also member of the "Académie des Sciences de l'Institut de France" (since 1983) and member of the National Academy of Medicine (France) since 1991. From 2005, Professor Coppens serves as Emeritus Professor of Paleoanthropology and Prehistory in the Collège de France. The Collège de France was founded in 1530. Its alumni include renowned scientists such as André-Marie Ampère (1824-1836) and Charles-Édward Brown-Séquard (1878-1894), among others. He has discovered tons of fossils of vertebrates and signed or cosigned six new Hominidae. He was nominated Grand Officier de la Légion D'Honneur of France. His name was given to an asteroid (172850 Coppens).

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_1

chimpanzees, and equally include the Homininae, which are the prehumans and humans of today. This leads to the assumption that Paninae and Homininae share a common ancestry.

As it is known that all primates have tropical origin and Paninae stem from African origin, there is the significant probability that this common ancestry and at least their first descendants were tropical and African.

And, in fact, it is only tropical Africa that provided the necessary conditions.

Furthermore, analyses of the great morphological, anatomical, physiological, genetic, molecular, and ethological distances between our "cousins" Paninae and ourselves have allowed us to calculate our last common ancestry to have happened in the upper Miocene, around 10 million years ago, the birth date of our taxonomic subfamily. The location: tropical Africa.

Today, there are three candidates with such an origin and age for this ancestry: *Chororapithecus abyssinicus*, from Ethiopia, 10.7 to 10.1 million years ago; *Nakalipithecus nakayamai*, from Kenya, 9.88 to 9.89 million years ago; and *Samburupithecus kiptalami*, also from Kenya, 9.6 million years ago.

Fossils of these candidates provide an idea of our common ancestor's appearance, but not clear enough to place them before or after the divergence of Paninae/Homininae, raising the dilemma if they were already Paninae (pre-chimpanzee) or Homininae (prehuman) or existing side by side. What is known is that the location was unequivocably a tropical and forestal biotope and that it was at this time that the separation occurred for environmental reasons. This was the departure point for our evolution as a Homininae, our exclusive developmental path.

Over the course of 10 million years, this path has been recorded by genus and species, primarily prehumans from 10 to 1 million years ago and then humans, from 3 million years ago until today and into the future. This trajectory, therefore, implies that the last prehumans were contemporaries of the first humans.

The prehumans are numerous and differentiate widely into 7 genera and 14 species, discovered

in South Africa, Malawi, Tanzania, Kenya, Ethiopia, and Chad, and all share tropical origins that are solely African. These specimens also all possessed a static, permanent upright position with a biped and arboreal locomotion initially, then transitioning to be exclusively bipedal. These prehumans also demonstrate a brain in mild expansion and facial feature undergoing a mild reduction with teeth at times under reduction and at other periods under expansion.

#### 1.2 How Did We Become Upright?

The acquisition of an upright posture – the underlying contingency which made it possible for early humans to extend the trunk, pelvis, thigh, and legs – combined with the resulting bipedal locomotion, represents the key transformation point in the history of Homininae, one that gradually and mechanically induced other transformations, in particular changes in the hands and brain, which facilitated consecutively the emergence of tools and consciousness, culture, and society.

In successive order, from an anatomic and as functional as possible perspective, I shall lay out the underlying factors concerning the acquisition of an upright posture in a static condition and the ability to walk upon the hind feet. The following considerations are based upon observations gleaned from different parts of the skeleton of the *Australopithecus afarensis* species.

Observations for body size and body displacement movements were made from a fragmentary skeleton excavated from a field in Ethiopia, AL288,<sup>1</sup> a separate group of bones related to the same species from the same excavation field and 34 footprints from a field in Tanzania; indeed AL288 is the most complete archeological sample set known concerning erect posture acquisition and hind feet locomotion.

<sup>&</sup>lt;sup>1</sup>AL288 is "Lucy," discovered in 1974 by Yves Coppens. It was, at that time, the oldest bipedal hominid ever discovered, over more than 3 million years old.

#### 1.2.1 Vertebral Column

There are ten artifacts to describe the vertebral column, all from skeleton AL288: seven thoracic vertebrae, two lumbar vertebrae, and one sacrum.

The seven thoracic vertebrae – T2, T6, T7, T8, T10, T11, and T12 – are very similar to their human homologues. On initial observation, they differ only for two main features, moreover without any relationship between them: *Australopithecus afarensis (AL288)* vertebrae are significantly smaller in all linear dimensions. However, there is one exception: the sagittal diameter is proportionally very large as it is artificially increased due to a bony arch on the ventral surface of vertebral bodies.

The two lumbar vertebrae -L3 and L4 - are also small in size. Their morphology and the orientation of their different parts make one surmise that the thoracic kyphosis had extended until them. Therefore, it had been more akin to a thoracolumbar kyphosis with a large radius curvature.

Finally, the sacrum, formed by its five fused parts, appears strikingly human, albeit obviously smaller in measurement and proportion. Besides being shallow, it is proportionately extended at its frontal dimension.

Although extremely fragmentary as evidence, this spinal column clearly represents an upright and erect being. Cervical lordosis was highly likely, while thoracic kyphosis was clearly undeniable, appearing only slightly more stretched downward into the lumbar region than ours today. It is appropriate to consider the human variability in this matter: the sagittal angle of its curvature can be estimated to be between  $30^{\circ}$  and  $40^{\circ}$ . Lumbosacral lordosis is also present but, however, clearly reduced due to the thoracolumbar stretch. In addition, lordosis is slight, with a curvature of between 40° and 55°. Furthermore, it is quite probable that the spinal cord had had a lower cervical dilation and lower lumbar component (transversal diameter of T2 triangular and large; sagittotransversal index of the L4 vertebral hole quite high).

#### 1.2.2 Pelvis

Half of the pelvis of the AL288 skeleton is very well preserved. The proportions of this pelvis are human-like; however, its anatomy differs in a certain number of particularly interesting features and in their functional consequences, which we shall touch upon briefly.

Firstly, the iliac bones are oriented in a much more frontal coronal plane than in the human pelvis and are clearly wider than human counterparts. Of the ilium, there is a very slight indentation in the internal iliac fossa and is also very wide. The pelvic cavity is broad and also extended thankfully to the important development of the acetabular diameter and the length of the pubis cranial ramus, with its ventrocaudal inclination. The sacrum is, as previously mentioned, also very wide, though short and slightly curved. The sciatic notch is barely marked, and finally, the coxofemoral joints are seemingly undersized. The length of the caudal segment and the short size of the cranial segment confer the ilium-specific longitudinal proportions: a narrow sacral plane and excessively broad iliac plane attenuating to a slender lower extremity.

Furthermore, the ilium presents an iliac crest almost straight in line between the ventrocranial iliac spine (anterior superior iliac spine) and dorsocranial iliac spine (posterior superior iliac spine). The ventrocranial iliac spine is, indeed, distinctively beak shaped, which at this level results in the superposition of the iliac pillar above the iliac spine.

The following discussion involves other segments of the inferior member that cannot be disassociated with the pelvis. The femur is short with a slim and elongated neck. It is obliquely oriented relative to the pelvis, but almost perpendicular to the diaphysis, which is also oblique. The large trochanter is flattened, while the intercondylar fossa is broad and deep. The tibia is short with compact and poorly developed spines possessing asymmetric cavities and a slightly convex external aspect. The foot is short, broad, and flat, with a splayed first radius and stepping onto external support. Vertebrae morphology and the projected curves of the column represent a weight-bearing pelvis, with a distinctively extra large width and small height proportions. Coupled with the obliquity of the femur to the pelvis, all these aspects are evidence that the AL288 skeleton is an incontestable example of a being endowed with an upright erect posture with bipedal locomotion.

Also confirming these conclusions are the extraordinary fossilized footprints found in Tanzania dating back to the time period of 3.7 million years ago.

The reconstitution of balance to accommodate a static upright posture and bipedal gait was evidently accompanied by the adaptation of the muscles, in particular, the gluteal and ischiotibial complex. Consequently, as in many other aspects of anatomy, the Australopithecus appears similar to its ancestors, which are also the common ancestors of the Paninae, and appears similar to their descendants: humans. However, on closer examination of the gluteal muscles of the current representatives of the two lineages, that is, chimpanzees and humans, they are clearly quite different from each other. Schematically, while the chimpanzee possesses, at the superficial level, a thin tensor fasciae latae and a thin gluteus maximus; a thick ischio-femoralis and a thick gluteus medius at the mid-tissue level; and finally, at a deep tissue level, a modest gluteus minimus, human beings, while also endowed with a tensor fasciae latae, gluteus medius, and gluteus minimus, have however a very generous, thick, and fasciculated gluteus maximus, extending from the dorsal part of the ilium and the lateral border of the sacrum to the proximal femoral diaphysis, and do not posses in any form an ischio-femoralis. Another fundamental difference is the diminutive insertion size of the *gluteus maximus* to that of the Australopithecus pelvis. This conjures the hypothesis that the Australopithecus gluteal muscles were more representative of Paninae. This is also supported by the presence of a biacetabular enlargement of the pelvis, the orientation of the iliac wings, the smallness of the coxofemoral joints, as well as the length and orientation of the femoral neck. These clearly indicate a more

efficient Paninae-like biomechanics with a greater importance on the ischio-femoralis: especially a tight external rotator, extensor (with the biceps femoris caput longum), and hip adductor (with *pectineus* and *magnus* adductor); and a powerful gluteus minimus, flexor, which does not occur in humans, and hip abductor, and also internal thigh rotator (with the gluteus medius, which represents the external rotator in humans). According to the fossilized footprints cited above, the surface contact position of the feet was in varus the surface contact position of the feet was in varus and corresponds exactly to the axis of the previous step, the axis of the previous step, sometimes intersecting. From this data, for Australopithecus, it is clear that while walking or in a static upright position, the lower members were extended and adducted, while the pelvis was rotated around the vertebral column.

The sum of these direct and indirect anatomical data, as well as the reconstructions obtained from these data, leads to the academic conclusion that Australopithecus was capable of an upright static balance with inferior quality than that of humans and a bipedal locomotion that considerably consumed more energy. The instability of the hip joint, that was just discussed, and the inherent weakness of the knees and ankles, to be elaborated on below, in all likelihood did not allow Australopithecus to support weightbearing pressure or static motionlessness for extended periods of time. The extensor muscles were, as already noted, efficient, however probably utilized more to rotate the pelvis than to stabilize it during extension. The Australopithecus gait, although aided by powerful flexors and extensors, was most likely to have been relatively unstable, quickly thrown out of balance, and therefore briskly executed, more like a trot, with extended lower limbs (enabled by the small, dorsally oriented ischial tuberosities and farther spaced from femoral diaphysis than in humans); the knee buckled slightly inward causing the tibial plateau to slope outward and downward. In order to ensure some stability to the lower support members and shift the center of gravity toward the knee, Australopithecus would have had to generate an exaggerated rotation of the pelvis and shoulders around the vertebral axis, a rotational movement obtained through an exceptionally demanding balance of the forelimbs while framing a broad and heavy thorax. This rotation has been estimated to be between  $50^{\circ}$  and  $60^{\circ}$  instead of  $4^{\circ}$  found in humans.

Considering parturition, it must be assumed that the skull of the fetus passes sagittally between the farthest point of the sacrum and the lower edge of the pubic symphysis, and the maximum length for Australopithecus afarensis is estimated to be between 85 and 90 mm. Given the particularly flattened shape of the pelvic brim, the orientation of the fetal skull into the narrow pelvic outlet must have been oblique or transverse, and consequently the head would not have been able to pass through the pelvic outlet as easily as that in human parturition. Also, the fetus would have had to rotate just as humans do. Bipedalism, in fact, initiated the processes that significantly reduced the distance between the sacroiliac and coxofemoral joints due to the inherent limitation of ponderal pressure on the pelvic walls and hip. It also determined a better suspension of the abdominal organs and especially the fetus during pregnancy. However, at the same time, bipedalism triggered a reduction in the pelvic cavity giving rise to new proportions between the fetal head dimensions and pelvic outlet and an increase in pressure exerted by the pelvic walls upon the fetus, hence a complex set of obstetric mechanics consisting of a double dorsal flexion movement, deflection, and twist of the baby itself. Despite the biacetabular extension and elongation of the cranial pubis ramus aligned with the ilium ventral edge (whereas in humans the same ramus is perpendicular to it), the shape of the pelvic outlet forced the evolution toward a curved trajectory and parturition forward of ischial tuberosities, as seen in women today.

If, instead, there were a rectilinear trajectory and retrosciatic parturition, as in the great apes, the fetus would have had to exit in a transversal position into the pelvic outlet, which would have forced the ischial tuberosities to exert a dangerous amount of pressure on fetal frontal and occipital skull walls, walls incapable of compressing as the parietals. And the absence of fetal rotation during parturition would have entirely impeded passage of the shoulders.

In all, this is a pelvis and spinal column of a female being with an undeniably upright orientation, able to walk on its two hind limbs and give birth as a woman. It is apparent however that the upright, standing posture was difficult to maintain and that prolonged bipedal locomotion probably proved exceedingly wearisome.

#### 1.2.3 Lower Limbs

For analysis there are a femur, a tibia, and a talus from the AL288 skeleton and also a femur, tibia, three calcanei, a first cuneiform, the first three metatarsals, and some phalanges from the same archeological site and 34 footprints from a site in Tanzania.

The femur, like the tibia, is short, 28 cm for the reconstituted femur and 23 cm for the tibia, also reconstituted. The neck of the femur, as noted previously, is elongated and thin, while the trochlea is wide. This femur clearly demonstrates the three traditional traits of bipedalism: femoral diaphysis obliquity, an anterior prominence, and an exaggerated external trochlea labium height. Associated with the former two, there is an elliptical profile of the external condyle. Although the obliquity of the femoral diaphysis is particularly elevated and the anterior prominence especially small (possessing a slightly sharp trochlear groove), the side profile of the external condyle is undeniably elliptical.

On even deeper examination, it can be observed that the distal femoral epiphysis is predominantly asymmetrical regarding the parasagittal plane, passing through the middle of the trochlea, with an obvious preponderance in size on the internal aspect compared to the external side. This epiphysis can be inscribed schematically inside a rectangle, and it can be noted, finally, that the intercondylar fossa is disproportionately wider than higher. These three characteristics are exclusively shared among the chimpanzee.

The tibial plateau, when examined, reveals an exceptional proximity between the tibial spines, a

convexity of the lateral condyle and a cleft marking the single insertion of an external meniscus, also all homologous traits of Paninae.

Thus, the knee joint differs remarkably from that of human beings. Considering the gap between the very narrow pivot of the tibial spine and the width of the intercondylar notch, this can be assumed a poor fit for articular surfaces to have such a large rotational amplitude in knee motion. The only mode of insertion of the meniscus, which determines the type of mobility, in turn influences rotational amplitude. Mediolateral enlargement of the femur and tibia epiphyses suggests a more measured amplitude in the flexion-extension movement and a shorter stride.

The slight protrusion of the femoral trochlea and the convexity of the external tibial plateau, also associated to femoral obliquity, do not seem to fully allow for the adducted position of the knees, like those present in humans. The asymmetry of tibial epiphysis and the rectangular contour of the intercondylar femoral notch support the assertion, finally, that full extension of the knee was probably not the normal use. Taken together, these data reveal the probable active practice of arboreal locomotion where freedom of the knee joint was necessary than its solidity.

The Australopithecus afarensis foot had an estimated total length of 18.5 cm. The talus, while having the size of a chimpanzee although with human proportions, possesses an original morphology, suggesting a unique relationship between the foot and the leg. The calcaneus has a large curved axis with medial concavity instead of being straight as in humans, and its lateral side is equipped with a large fibular tubercle, which when present in humans has only the appearance of a modest button. Its dorsal surface is egg shaped with its major axis tilted toward the lateral side and has a lone medial tubercle. With this heel bone morphology, one can deduce that movement was unstable, in varus, i.e., rotating over the external side.

The first cuneiform, instead of being flat as in humans, has a highly rounded articulate surface with the corresponding metatarsus. This contact flexibility suggests that the first metatarsus might have rotated around the first cuneiform and as consequence a gap in the first commissure. The first metatarsus does in fact confirm this articular property with a characteristic posterior surface that is a hollow, bilobular set between two distinctive edges. The phalanges are long. They represent up to thirty percent (30 %) of the foot length, while in humans the proportion is just twenty-three percent (23 %). They are wider at the level of the diaphysis, flattened dorsoventrally, and curved with a ventral cavity.

The footprints have each from 17 to 20.5 cm in length and 46 cm distance between three consecutive footprints that track of the smaller feet (G1) and each with a length from 30.5 to 33 cm with the track of the larger feet (G2) and a distance of 47.5 cm between two consecutive footprints (sizes of the steps are measured between points that correspond to the heel).

These footprint sample sets have a number of common characteristics that confirm and compliment the anatomical data. Firstly, all both foot tracks have a pointed narrow heel whose axis is slightly toward the medial side, with a clear medial expansion, rectilinear lateral depression, and separation between the first ray and lateral rays, which are often represented by small round cavities. Secondly, the depression of the heel corresponds to a tuberosity of the calcaneus. The medial prominence indicates the existence of an abductor hallucis longus muscle, as observed in chimpanzees, and the lateral depressions indicate external support and varus; separation between the first ray and the others indicates hallux abduction, and small rounded cavities indicate the toes bent in on themselves. A cross-sectional flexion over the tarsus, observed in some of the footprints, indicates a certain foot flexibility; however, it is difficult to compare it to a true plantar arch. It begs to be noted that the human footprint has broad heel, hallux, and laterally adduced toes; five short, thin, and straight toes; and, usually, a very marked plantar arch. From these footprints, a line can be drawn (for G2) and a thin band of just 1 cm wide (to G1), tangentially to the medial edge of the footprints, which indicates that the steps had been placed consistently one in front of the other, which is reminiscent of pelvis rotation described above. The axis of these footprints is a thirty (30) degree angle from the centerline of the whole track.

#### 1.2.4 Upper Limbs

This sample is comprised of one scapula, two humeri, two ulnas, and a radius bone from the AL288 skeleton and six other humerus fragments and two ulna fragments from the same archeological site.

The glenoid fossa of the scapula has a more cranial than lateral orientation, as observed in the large apes. Moreover, it presents a large supraglenoid tubercle and an elongated infraglenoid tubercle, also as in the large apes. Inserted into the former was, in fact, the biceps brachii caput longum muscle, and into the later, triceps muscle brachii caput longum, both flexors used for suspension. The coracoid process, curved when observed laterally, is strikingly Paninae and developmentally very important and different than the shape of the coracoid process observed in humans. It is particular to Paninae in that it is, literally, triangular, and as it has insertions bored into it by the *coracobrachialis*, the flexor of the humerus over the scapula and by the biceps brachii caput breve, the flexor of the radius over the humerus. From these observations it can be hypothesized that Australopithecines used this structure of the upper limbs for the arboreal part of its locomotor repertoire.

The humerus is not long, measuring about a mere 23 cm. Its head is rounded, like a "half ball," and slightly elongated vertically; its anatomical neck is very pronounced, almost as much as that in a chimpanzee; its very long tuberculum minus is prominent above the bicipital groove, and as it bears an important insertion of the subscapularis muscle, it can be presumed that the internal arm rotational movements and humerus adduction over the scapula were developed. The tuberculum majus, significant here, is similar to its Paninae counterpart, especially with the elongation of one of its two facets, which is nearest to the groove, over which inserts the supraspinatus muscle, the abductor of the arm over the scapula. These muscles, at the same time, reinforce the humeral head joint within the glenoid cavity. The bicipital groove is notoriously narrow and deep as in Homininae, which uses its upper limbs in a certain form of suspension. The two crests, crista tuberculi minoris and crista tuberculi majoris,

are highly differentiated. The insertion on teres major and pectoralis major firmly suggests an ample arm adduction and a significant humerus internal rotation over the scapula. The morphology of distal articular extremity of the humerus is a double trochlea, as in Paninae. Indeed, a bony lateral prominence inserts itself between the trochlea and capitulum, reinforcing elbow joint stability, extremely valuable in suspension movements. Known in great apes, the epicondylus *lateralis* clearly carves out a lateral prominence; however, here its projection is skewed upward in relation to the capitulum humeri in a position somewhere between the great apes and human beings. Yet interposed between these two beings' morphologies is the extension to the back of capitulum humeri. The morphology of the fossa olecrani is, on the other hand, exactly to that of human beings, devoid of the strong lateral crest that exists in Paninae, a crest which reinforces stabilization of the elbow in Paninae. The incisura trochlearis of the proximal extremity of the ulna is equally devoid of a lateral facet, which supports in the existence of this ridge in Paninae. Finally, the extremely well-developed epicondylus medialis lends merit to the assumption that the flexor muscles were also, as a whole, very well developed.

The ulna presents, at its proximal extremity, a very slight development of an *incisura trochlearis* and a very modest expression of the fossa adjacent to the olecranon, as in humans, but with a pronounced reentrance of the *incisura radialis*, both lateral and rounded, as in the chimpanzee. This last incisura suggests the importance of prone and supine movements of the upper extremities. As to radius proximal extremity, the tuberosity and the long and narrow neck, and the joint circumference highlighted by a crest in the middle, all these elements are reminiscent of the morphology observed in chimpanzees, indicating a more stable suspension and better supination.

The wrist joint demonstrates stabilization marks comparable to those observed in the shoulder and elbow joints. These characteristics suggest the usage of the upper limb at least partially for arboreal locomotion.

As for the hand, there are long and curved phalanges, resembling the development as the

foot, probably indicating the same interpretation of the morphology.

Hence, this is the description of a small living being that had an active and arboreal locomotion, not quadrupedal, not even in the use of its arms, but rather a mixture of climbing trees and suspending itself from its arms.

#### 1.3 Final Considerations

The vertebral column, pelvis, and limbs described also belong to the same small being: still arboreal and already biped.

Contrary to what has often been written, bipedalism is not, since the early history of our evolution, the same as today. Bipedal locomotion has its own history, its own evolution. Most importantly, contrary to what has often been claimed, bipedalism was not initiated through the transformation of feet. The transition to walking exclusively upright on two feet undoubtedly began through adaptations within the pelvis, that is, from the pelvis downward to the feet. Australopithecus afarensis is, thus, the first species that clearly demonstrates the adaptations wrought upon a skeleton by the simultaneous use of these two very different locomotion practices. The recent discovery of a young individual skeleton of the same species, a female, 3-4 years old, called Selam, by its discoverer, Alemseghed Zeresenay at a site near Hadar (AL288 site), and of comparable age, brilliantly confirmed the association between static erect position, bipedalism (mainly lower limbs), and tree climbing (using mainly the upper limbs).

Early Homininae, those related to other genera and other species, *Orrorin tugenensis*, 6 million years ago, from Kenya, and *Ardipithecus ramidus*, 4.4 million years old, from Ethiopia, that were discovered after Lucy (AL288), were also endowed with this double locomotive movement, although slightly different from that described here in *Australopithecus afarensis*, which opens the possibility that this locomotive state was a pattern among all first prehumans and perhaps even among "second-generation" prehumans such as Lucy. As for exclusive bipedal locomotion, it has been estimated to have begun 4 million years ago with another Homininae, genera *Australopithecus*, which has been named *Australopithecus anamensis*. This genus was a contemporary of Lucy, *Australopithecus afarensis*, having lived for 4 and 3 million years ago and in the same regions, Kenya and Ethiopia, but most probably not in the same niche. *Australopithecus anamensis* by comparison possessed typically modern lower limbs, i.e., extremely stable, and upper limbs that were much less solid, literally the opposite of *Australopithecus afarensis* with unstable lower limbs and very solid upper limbs.

Worth mention are the ecological changes that were also taking place 4 million years ago, a landscape opening to glades and widening grasslands that may have favored the transition to exclusive bipedalism. Humans, the genus *Homo*, appear precisely after that second "step" of prehumans, around 3 million years ago, ostensibly from a prehuman ancestor that we have not as yet identified. This means that whoever this ancestor may be, mankind also has tropical Africa as its evolutionary cradle, but with a geological age of 3 million years.

What was taking place 3 and 2 million years ago in tropical Africa? The climate was changing and evolving from a humid one to one much drier. As a result, flora and fauna archeology specimens demonstrate grandly this climate change, especially in the generous and extensive sites of the lower Omo Valley in Ethiopia, where these geologic levels are well represented and organic remnants are especially numerous today. Among these are prehuman remnants dating before this change, i.e., Australopithecus afarensis, Australopithecus anamensis, and Kenyanthropus platyops, and human remnants from after the change, Homo habilis and Homo rudolfensis. The emergence of human beings, as with many other mammals of that geological era, arises as the result to the necessary evolution of prehumans which were forced to adapt to a new and pervasive environment. This adaptation was dependent upon a much better performing brain and improved dentition to accommodate an opportunistic feeding regime, becoming omnivore. Homininae, therefore, came to have an upright posture some 10 million years ago while being exclusively bipedal for 4 million years. The new environmental circumstances also offered the emergence of a reflective consciousness.

In my writings, I gave name to this climatic and historical event as the (H)Omo event, combining words Homo and Omo to remind all that it was in the Ethiopian Omo valley where I was able to link, for the first time, the appearance of human beings to climate change (1975).

It was through this change that for the first time a living being, human in its being, would carve stone, as an act through its own will and be used upon its environment in order to form circumstances to its advantage. This is reflective consciousness, for example, deliberately hitting one stone against another in order to change the shape of the first stone and obtain a new form for the function that it was consciously intended.

For 10 million years, that's how this somewhat funny small mammal standing on its hind legs has evolved as a result of a changed environment, the only mammal that 3 million years ago was able to develop a cultural environment like no other. Paradoxically, within this new environment, two somewhat opposite distinguishing human hallmarks have emerged: freedom and responsibility on the one hand and arrogance and guilt on the other hand.
# A Historical Overview of Sciatica

2

Sebastião Gusmão, Aluizio Arantes, and João Luiz Pinheiro-Franco

# 2.1 Introduction

The currently well-established concept of sciatica due to lumbar intervertebral disk degeneration is the result of a long historical evolution whose various stages in thinking and exploration are the focus of this brief historical review.

# 2.2 Sciatica: Historical Overview

The first neurological reference to this ailment is credited to Imhotep, a few thousand years ago in Egypt. Imhotep, astronomer, physician, and architect to Pharaoh Djoser (2686–2613 BC), is thought to be the author of precious medical papyrus, studied by Brested in 1930, which

Neurosurgery Department, Hospital das Clínicas da Universidade Federal de Minas Gerais, Belo Horizonte, Brazil e-mail: sebastiaogusmao@gmail.com

A. Arantes Department of Neurosurgery, Hospital Madre Teresa, Belo Horizonte, Brazil

Universidade Federal de Minas Gerais, Belo Horizonte, Brazil e-mail: aluizio\_arantes@uol.com.br through various case studies differentiated lesions with or without neurological signs. Most striking is Case 48 in which he describes a clinical examination of the leg identical to the Lasègue maneuver commonly used today.

By the time of Jacob, the father of the 12 tribes of Israel, the debilitating condition of sciatica had already reached biblical proportions. In fact, in ancient Hebrew medicine, Jacob lost his wellknown wrestling match against an angel because of an injury to his sciatic nerve, as reported in Genesis 32: 24–32:

And Jacob was left alone. And then a man wrestled with him until daybreak. When the man saw that he could not overcome him, he touched the sinew of his thigh, and forthwith it shrank while they were wrestling ... and he was limping on a leg.

This is the reason why until this day, children of Israel do not eat the sinew of the thigh, remembering the one from Jacob's thigh that was touched and lost the movement.

In fact, with regard to Jacob's injury, the sciatic nerves of animals were declared unsuitable for human consumption. To better regulate this recommendation, the Talmud provided specific instructions for the removal of the sciatic

S. Gusmão (🖂)

Neurosurgery Department, Universidade Federal de Minas Gerais, Belo Horizonte, Brazil

J.L. Pinheiro-Franco Department of Neurosurgery and Spine Surgery, Hospital Samaritano, São Paulo, SP, Brazil e-mail: jlpfranco@uol.com.br

Editor's Note In this chapter, Gusmão et al. retrace, in conversational style, the evolution of medical thought and practice that ultimately led to the field of sciatica in the context of lumbar degenerative disk disease. This fascinating account is unique in both historical and clinical perspectives.

nerve from the flesh of slaughtered animals. This Bible passage is shown in the painting "Jacob wrestling with the angel" (1850–1861, Chapelle des Saints-Anges – the Chapel of the Guardian Angels- at the Church of Saint-Sulpice, Paris, France) by Ferdinand Victor Eugène Delacroix (1798–1863). This painting shows the angel touching the back of Jacob's left thigh.

Not until the Greek and Roman times does sciatica move from iconography to a form of medical description of disease. Hippocrates was the first physician to use the term "sciatica" taken from the Greek ischios meaning hip. The earliest Greek and Roman references are reports of pain in the area of the sciatic nerve, which was always referenced as a disease of the hip joint. The concomitant pain in the pelvis and leg was diagnosed as sciatica and attributed to a subluxated hip or a hip disease. Hippocrates (Fifth Century BC), in his Treatise of Predictions, described the natural history of patients with "cramps and colds at the loin and the legs."

Galen (130–200 AD), another Roman from Pergamum in Asia Minor and one of the greatest physicians of antiquity, reported several cases of sciatica with specific descriptions of abnormal spinal posture. He coined the labels for conditions that still stand today: lordosis, kyphosis, and scoliosis. It was at this same time that other Greek and Roman authors have described sciatica without distinguishing between pain arising in the hip joint and spine.

Despite Andreas Vesalius, the undisputed father of modern anatomy (1514–1564) (Fig. 2.1) and his landmark description of the intervertebral disk, the Medieval and Renaissance periods contributed very little or nothing to the concept of the mechanism and etiology of sciatica. But the word sciatica was widespread. The words "Thou cold sciatica" were placed into the mouth of Timon of Athens by William Shakespeare.

A significant step forward in the field occurred in 1764, when Domenico Cotugno (1736–1822) (Fig. 2.2), anatomist and professor of surgery in Naples, published *De ischiade nervosa commentarius* ("Remarks on nervous ischialgia") (Fig. 2.3), which defined sciatica as a clinical entity and related the pain in the leg to disease of



**Fig. 2.1** Andreas Vesalius (1514–1564). This portrait is from his book *De humani corporis fabrica libri septem*, or *on the fabric of the human body* (Basel: Johannes Oporinus; June 1543), and is attributed to Ján Stephan van Calcar



Fig. 2.2 Domenico Cotugno (1736–1822)



Fig. 2.3 Domenico Cotugno (1736–1822). Cover from *De Ischiade Nervosa Commentarius* (Ed. Typograph. Naples: Simoniaca; 1779)

the sciatic nerve. He distinguished pain of the lower limb of "arthritic" origin, or "arthritic sciatica," identified as hip pain, from pain of "nervous" origin or "nervous sciatica," which was classified as "postica" (posterior) or "antica" (anterior). He accurately differentiated sciatic nerve pain from arthritis of the hip with a precise description of the clinical status and by indicating the relationship of pain to the sciatic nerve [1]:

If the pain indicated by the patient's finger runs from the foot to the sacrum, as competent anatomists we must evaluate this patient by tracing the precise course of the sciatic nerve. The patient's pain is felt in this nerve, and we should look for the cause of their limping in this nerve, and for the origin of paresis and impairment in this disease.

It is thus how the eponym "Cotugno's syndrome" found its way into medical vernacular to indicate a unilateral sciatic neuralgia. Cotugno attributed these symptoms to the accumulation of an acrid matter within the sheath of the sciatic nerve that had derived from the vessels irrigating the sheaths of the nerve or from the brain itself. In order to prove the existence of free circulation between the cranial and spinal dura, he propped cadavers upright on their feet and then decapitated them so as to observe the flow of cerebrospinal fluid. Twenty headless cadavers stood in for the cause which established the existence of this free circulation of the "Liquor Cottunnii," the first known reference to cerebrospinal fluid.

His description of this "Liquor Cottunnii," or cerebrospinal fluid as we know it today, included a precise indication as to its formation and absorption from the vessels. From this, Cotugno postulated its relationship with sciatica [1]:

The cerebrospinal liquor is in *perenni statu renovationis*, through exudation by minimal arteries and reabsorption through minimal veins. It penetrates into the dural sleeves of the nerve roots; hence, it is apt to accumulate in the sheaths of the sciatic nerve and as so promote pain along its course. The sciatic pain, weakness and limping may be cured, if necessary, by the use of vesicants and caustics in order to leech out the hydrops.

Within the folds of the history of sciatica, there were intermittent parallel investigations into the intervertebral disk; however, such findings were not correlated with sciatica. Nearly a hundred years after Cotugno's landmark work, Rudolf Virchow (1821–1902) (Fig. 2.4) described the first case of traumatic rupture of the intervertebral disk. This rupture, known as "Virchow's tumor," was discovered during the necropsy of a trauma victim [2].

In 1896 Swiss surgeon Kocher documented the first case of a disk displacement leading to paraplegia; however, its clinical correlation to sciatica remained moot [3]. The patient developed paraplegia after a fall and subsequently succumbed to injury to the internal organs. During necropsy, Kocher identified dorsally displaced disk at L1–L2. Sciatica and the lumbar region would remain dissociated until the beginning of the twentieth century. Until the late nineteenth century, sciatica was interpreted as a sciatic nerve pain, a neuralgia, or neuritis. The cause remained unknown, and the authors were limited to describing symptoms. French neurologist, Lasègue (1816–1883) (Fig. 2.5) called attention to a pain provoked by lifting the leg in patients with sciatica. He later demonstrated that this phenomenon was due to the stretching of the sciatic nerve roots (1864)

[4]. Although his medical writings do not mention this sign, it was his former pupil, Forst, who published the findings of his master and illustrated the "Lasègue maneuver" for the first time (Fig. 2.6) [6].



**Fig. 2.4** Rudolf Virchow (1821–1902) (Image from the History of Medicine, US National Library of Medicine)



Fig. 2.5 Ernest Lasègue (1816–1883)



Fig. 2.6 Lasègue Forst sign, 1881 (From de Castro et al. [5]; licensed under a Creative Commons Attribution License)

Concurrently, another French physician, François Valleix (1807–1855), identified specific sensitive points within the course of the sciatic nerve. These anatomic points, known today as Valleix's points, are points at which pain is experienced upon pressure in cases of neuralgia. Valleix's points are in fact segments of the sciatic nerve that are accessible to palpation in patients with sciatic pain. Along the sciatic nerve these segments lie in the buttocks, thigh, leg, and foot [7].

It were Charcot and Brissaud who in 1888 described "sciatica scoliosis," abnormal postures that were in fact caused by severe sciatica. They accurately distinguished crossed and ipsilateral scoliosis. In crossed scoliosis, the trunk is inclined laterally toward the side not affected by sciatica. In direct or ipsilateral scoliosis, the lateral inflection of the spine always tilts toward the side affected by sciatica [7].

For sciatica, the early twentieth century opened to the phenomena of "sensorimotor radiculitis." Dejerine demonstrated that sciatica is a root condition and not a truncal condition. He noted that some cases of sciatica are accompanied by areas of hypesthesia or cutaneous anesthesia, and the distribution of the insensitive areas did not correspond to the areas of the sciatic nerve branches but rather to the areas of the nerve roots. These cases of cutaneous anesthesia corresponded more precisely to the areas of the fifth lumbar nerve root and first sacral nerve root. He defined the phenomenon as a sensorimotor radiculitis caused by "partial lumbar meningitis," following the then-current interpretation of the impairment of the nerves and roots by inflammation caused by syphilis [7].

At the transition between the height of inductive reasoning and the dawn of evidence-based medicine, the French medical establishment reigned supreme. Continuing in that fine tradition, Jean-Anselme Sicard (1872–1929), a neurologist, presented sciatica in 1918, as a condition caused by spinal conditions. He advanced the hypothesis that compromise of the roots that form the sciatic nerve does not occur within the dural sac but outside the dura mater, at its exit of the intervertebral foramen. Furthermore, he postulated that the cause of this impairment in the roots could be due to bony and ligamentous elements surrounding the root within the intervertebral foramen. He was of course inferring to the fact that the thick sciatic nerve roots, especially the L5 root, pass within a particularly narrow osteoligamentous canal [7].

So it seems that by the start of the twentieth century, the road map into the scientific understanding had been laid. Social circumstances of the new century such as the progression of modern thought with its exchange of ideas and technology, particularly the invention of anesthesia conspired to favor the advancement of surgical technique and subsequently to what was once allencompassing sciatica, would now become the consequence of lumbar intervertebral disk degeneration.

First of these advancements was the hypothesis that sciatica might be associated with herniated lumbar disk. There was then a small but growing body of surgical reports of uncommon benign tumors, chondromas on patients undergoing surgical treatment for sciatica. Mixter and Barr were at the vanguard of these advancements demonstrating that such lesions corresponded to a herniated disk.

Victor Horsley, in 1887, is credited to be the first surgeon to remove such a tumor; however, the association to sciatica was not recognized. In fact, the first successful removal of a herniated disk took place in 1901 via laminectomy under the hands of Fedor Krause (1857–1937) (Fig. 2.7) and Hermann Oppenheim. They removed what they had thought was an "enchondroma" [8].

And the evidence continued to mount. It wasn't until 1911 when Joel Goldthwaite (1866–1961), building upon Middleton and Teacher's work of the same year, conceived the hypothesis that there existed a precise correlation between herniated disk and sciatic pain. Middleton and Teacher described two cases of intervertebral rupture observed during necropsy. Similar to Virchow and Kocher, they failed to postulate a correlation with lower back pain or sciatic pain [9]. Joel Goldthwaite did however.

Goldthwaite discussed a case of a patient with recurrent sciatica in the absence of any apparent



**Fig. 2.7** Fedor Krause (1857–1937) (Courtesy of the US National Library of Medicine)



**Fig. 2.8** William J. Mixter (1880–1958) (Courtesy of the Mixter Library Collection)

lesion who had been operated on by Harvey Cushing. He concluded that the pain resulted from recurrent disk dislocation into the vertebral canal. He went on to explain the negative exploration by the fact that the disk had consistently slipped back into place. He hypothesized, "Such a condition could produce the symptom of sciatica, low back pain" [10]. At last, the foggy vagueness surrounding the causality of sciatic had begun to lift. And even though it was proved a correlation, however not precise correlation between herniated disk and radiculopathy in Goldthwaite's landmark work, his assertion that herniated disk and sciatic pain were related failed to arouse interest in the medical establishment at that time.

Soon exciting new enabling technologies began to filter into the modern medical armory: radiology. The first gas myelography was performed by Jacobus and Ackerland in 1921, soon to be followed by the first lipiodol myelography in 1922 by Sicard and Forestier [11].

Although not recognized for its value until a review article by Barr, an orthopedist at Massachusetts General Hospital in 1932, Schmörl and Junghanns of the Dresden Pathology Institute in 1925, took on a mammoth radiological and pathological study of 5,000 human spines that conclusively documented the posterior prolapse and degeneration of the nucleus pulposus. However, no correlation to sciatica could ever be made as the study failed to include anatomoclinical correlations [12].

As visualizing and surgical techniques advanced, so too did the understanding of the intervertebral disk. In two cases where patients underwent surgery for cauda equina syndrome, Walter Dandy, in 1929, found cartilaginous fragments within the spinal canal. He described these fragments as being "loose cartilage from the intervertebral disk," simulating in their agglomeration, a spinal tumor [13]. He was the first to consider that intervertebral disk disease could be added to the list of indications for decompressive laminectomy.

Alas! 1932: the Holy Grail sciatica was with reach. The landmark breakthrough was to happen in the United States by William Mixter (1880–1958) (Fig. 2.8) and Joseph S. Barr. These two orthopedists from the Massachusetts General Hospital proved beyond all reason of doubt, the relationship between alterations in the intervertebral disk caused by trauma or degeneration and sciatica. From this point on, sciatica would never be vaguely associated with interstitial neuritis related to emotion, fatigue, or infection.

A patient suffering from sciatic pain came to Barr in 1932. Barr subsequently consulted Mixter who recommended a myelogram which later revealed a defect in the dural sac. Mixter performed the surgery, removing a "tumor" via laminectomy. Barr conducted the postoperative study of the tumor and performed microscopic comparisons with those in Schmörl's study of which Barr had previously published a comprehensive review. He recognized the specimen from the index patient to be of the nucleus pulposus.

Together with Mallory, a pathologist, Mixter and Barr reassessed all the cases at the Massachusetts General Hospital that had previously been "diagnosed" as having chondromas. Retrospectively, they diagnosed nineteen patients with reported sciatic pain due to disk prolapse, more specifically, disk herniation compressing the nerve root. This landmark study was published in 1934 in the New England Journal of Medicine. It heralded in a pioneering era that encouraged intervertebral disk intervention as the standard treatment for sciatic pain.

Soon after its publication, Mixter and Barr's diagnosis and laminectomy technique for lumbar disk herniations became the most frequently surgery performed by neurosurgeons [14]. Widespread acceptance of laminectomies to remove the intervertebral disk spurned the emergence of newer surgical techniques. And entire new body of literature dedicated to the association of sciatic pain due to herniated disk appeared. The scientific discussion moved off the strict analysis of symptoms and fostered new areas of endeavor such as contrast examinations, surgical techniques, and the careful cataloging of postoperative statistics. In less than 10 years after the first laminectomies by Mixter and Barr, the removal of the herniated disk by fenestration, a technique developed by Love from the Mayo Clinic in 1939, had become universally widespread [15, 16].

Subsequently, advances in lumbar intervertebral disk degeneration diagnosis and surgical treatment followed. First in the list of outstanding accomplishments was the "diagnostic puncture of intervertebral discs in sciatica" by Lindblom from the Karolinska Institute in 1948. This first study on diskography used 15 diskographies in 13 patients.

In 1969, Fischgold and Gonsette proposed radiculography with Dimer X. Then in the early 1970s computer tomography and magnetic resonance were introduced and dramatically improved diagnosis of disk injuries.

Then there was the advent of microscopeassisted surgery in 1977. M. Gazi Yasargil reported on 105 cases of lumbar disk surgeries using microscopes [17]. That same year, Caspar published a report on 102 patients undergoing microdiskectomy and then later adding a medial facetectomy to the procedure [18]. By the end of the 1970s, Robert Williams popularized microsurgery, which became the standard procedure for the treatment of herniated lumbar disks [19].

Advancements continued relentlessly until the close of the twentieth century. Advancements came from abled scientists and surgeons from around the world. Microsurgery spawned the quest for progressively less invasive surgical techniques in the treatment of herniated lumbar disks. New techniques were developed: the percutaneous diskectomy and the direct approach to the disk using a percutaneous technique (chemonucleolysis). Hijikata pioneered the development of instrumentation for the percutaneous removal of a herniated lumbar disk [20]. Others perfected upon Hijikata's instrumentation. Lyman Smith introduced chemical nucleolysis, with the injection of an enzyme into the nucleus pulposus to chemically destroy the disk [21].

Spinal fusion had become a reality and even more so recognized as being useful treatment option for lumbar degenerative disk disease associated or not with sciatica. The principles of spinal fusion evolved over the course of several decades during the twentieth century. Its evolution was discreet and separate from that of sciatica-lumbar degenerative disease. So many authors contributed to the surgical innovations of spinal fusion. Among them are Lange [22] for the stabilization of the lumbar spine with plastic bars, followed on by the use of steel bars attached to wires, Campbell [23] pioneering the use of the iliac crest bone graft in spinal fusion, Briggs and Milligan [24] perfecting the posterior lumbar interbody fusion along with Cloward [25], Boucher [26] who advanced the use of pedicle screws, and Roy-Camille [27] who perfected osteosynthesis of the spine using metal plates with pedicle screws: such a rich retrospective tribute and not by any means comprehensive.

A recent chapter in the history of the surgical treatment of lumbar degenerative disk disease has been the search to replace the lumbar disk, starting in the late 1950s. The concepts of lumbar arthroplasty began in the 1950s with isolated attempts to replace the nucleus pulposus with artificial implants to relieve pain and restore function of the degenerated spinal motor segment.

One of the first such artificial implants was the stainless steel ball. In 1964, Fernström implanted steel balls and published the clinical results in a 30-month follow-up. The results suggested that arthroplasty with stainless steel balls was better than results of diskectomy alone or spinal fusion. While primary goal of these implants is preservation of disk height and segment mobility, an unexpected negative outcome, the premature subsidence into the vertebral bodies, led to the failure of the use of the instrumentation [28].

Alternative to the arthroplasty approach was the search for substitutes for the nucleus pulposus. Some substitutes were developed, as the Raymedica Prosthetic Disk Nucleus (PDN) (Raymedica, Minneapolis, MN) and the Aquarelle (Stryker Howmedica Osteonics, Allendale, NJ). Common to both disk substitutes is the use of a hydrogel core.

Total lumbar disk replacement, commonly known as an artificial disk, represents an evolution in lumbar disk arthroplasty. Current study and limited clinical use is tug-of-war between several competing patent holders. There exist different models of these implants on the market now, all with similar implantation techniques with minor differences in design and mechanical function.

And while the industry of our profession vies for market space, the role of lumbar disk

arthroplasty among researchers remains unclear. Two of its major advantages over fusion are preservation of motion and prevention of adjacent segment degeneration.

#### 2.2.1 Final Considerations

From Jacob to lumbar arthroplasty is a story of the evolution of knowledge on the pathophysiology of lumbar degenerative disk disease and sciatica. Even as current research seems pinned in a balance over the uncertainty of arthroplasty technology, the story of sciatica is a testament to the persistence of disease in humankind and the physician's endless commitment to do battle with condemning angels.

#### References

- 1. Cotugno D. De ischiade nervosa commentaries. Vienna: Apud Rudolphum Gräffer; 1770.
- 2. Weller CV. Rudolf Virchow–pathologist. Sci Mon. 1921;13(1):33–9.
- Kocher T. Die verletzungen der wirbelsäule zugleich als beiträg zur physiologie des menschlichen rückenmarks. Mitt Grenzgeb Med Chir. 1896;1:415–80.
- 4. Lasègue C. Considérations sur la sciatique. Arch Genet Med. 1864;24:558.
- Castro I, Paes dos Santos D, de Holanda Christoph D, Landeiro JA. The history of spinal surgery for disc disease: an illustrated timeline. Arq Neuropsiquiatr 2005;63(3-A):701–6.
- 6. Forst JJ. Contribution to the clinical study of sciatica. Arch Neurol. 1969;21(2):220–1.
- 7. de Séze S. History of sciatica [in French]. Rev Neurol (Paris). 1982;138(12):1019–25.
- Oppenheim H, Krause F. Über Einklemmung bzw. Strangulation der cauda equina. Dtsch Med Wochenschr. 1909;35:697–700.
- Middleton GS, Teacher JH. Injury of the spinal cord due to rupture of an intervertebral disc during muscular effort. Glasgow Med J. 1911;76:1–6.
- Goldthwait JE. The lumbosacral articulation. An explanation of many cases of "lumbago", "sciatica" and "paraplegia". Boston Med Surg J. 1911;164: 365–72.
- Sicard JA, Forestier J. Méthode générale d'exploration radiólogique par l'huile iodee (Lipiodol). Bull Mem Soc Med Hôpitaux de Paris. 1922;46:463.
- Schmorl G, Junghanns H. Die gesunde und kranke Wirbelsäule in Röntgenbild und Klinik, 2nd ed. G. Thieme, Stuttgart; 1951. p. 203.

- Dandy WE. Loose cartilage from intervertebral disk simulating tumor of the spinal cord. By Walter E. Dandy, 1929. Clin Orthop Relat Res. 1989;238:4–8.
- Mixter WJ, Barr JS. Rupture of the intervertebral disc with involvement of the spinal canal. N Engl J Med. 1934;211:210–5.
- Love JG. Protruded intervertebral disks with a note regarding hypertrophy of ligamenta flava. JAMA. 1939;113(23):2029–35.
- Camp JD. The roentgenologic diagnosis of intraspinal protrusion of intervertebral disks by means of radiopaque oil. JAMA. 1939;113(23):2024–9.
- Yasargil MG. Microsurgical operation of herniated lumbar disc. In: Wüllenweber R, Brock M, Hamer J, Klinger M, Spoerri O, editors. Lumbar disc adult hydrocephalus. Berlin: Springer; 1977. p. 81.
- Caspar W. A new surgical procedure for lumbar disc herniation causing less tissue damage through a microsurgical approach. Adv Neurosurg. 1977;4:74–80.
- Williams RW. Microlumbar discectomy: a conservative surgical approach to the virgin herniated lumbar disc. Spine. 1978;3:175–82.
- Hijikata S, Yamagishi M, Nakayama T, et al. Percutaneous nucleotomy: a new treatment method for lumbar disc herniation. J Toden Hosp. 1975;5:5–13.

- 21. Smith L. Chemonucleolysis. Clin Orthop. 1969;67: 72–80.
- Lange F. Support for the spondylotic spine by means of buried steel bars attached to the vertebra. Am J Orthop. 1910;8:544–61.
- Campbell WC. Transference of the crest of the ilium for flexion contractures of the hip. South Med J. 1925;16:289.
- Briggs H, Milligan P. Chip fusion of the low back following exploration of the spinal canal. J Bone Joint Surg. 1944;26:125–30.
- Cloward RB. The treatment of ruptured lumbar intervertebral discs by vertebral body fusion. I. Indications, operative technique, after care. J Neurosurg. 1953;10: 154–68.
- Boucher HH. A method of spinal fusion. J Bone Joint Surg. 1959;41B:248–59.
- Roy-Camille R, Roy-Camille M, Demeulenaere C. Osteosynthesis of dorsal, lumbar, and lumbosacral spine with metallic plates screwed into vertebral pedicles and articular apophyses [in French]. Presse Med. 1970;78(32):1447–8.
- Fernström U. Arthroplasty with intercorporal endoprothesis in herniated disc and in painful disc. Acta Chir Scand Suppl. 1966;357:154–9.

# Lumbar Intervertebral Disk Injury, Herniation and Degeneration

Michael A. Adams and Patricia Dolan

# 3.1 Introduction

#### 3.1.1 Intervertebral Disks

Intervertebral disks are pads of fibrocartilage which lie between the vertebral bodies of the spine. A soft nucleus pulposus is surrounded by concentric layers ('lamellae') of the annulus fibrosus (Fig. 3.1). Human lumbar disks are the largest avascular structures in the body, and the consequent metabolite transport problems ensure that cell density is barely sufficient to maintain the disk in health and quite insufficient to repair it when injured or diseased.

The disk nucleus contains a high concentration of proteoglycans, which attract water into the tissue and ensure that it behaves like a pressurised fluid, distributing load evenly on the adjacent vertebral bodies. The annulus is mostly comprised of coarse collagen type I fibres which run obliquely between the vertebrae (Fig. 3.1). Annulus tissue is sufficiently soft to allow small intervertebral movements and yet strong enough to prevent excessive movements and to restrain the nucleus. Disks are too stiff to act as efficient shock absorbers.

# 3.1.2 Disk Injury or Degeneration?

Disks must distribute forces evenly on the vertebral bodies, even when the spine is heavily loaded in flexed or extended postures. This is a physically demanding role, especially in the lower lumbar spine of humans, and disks often show signs of structural damage even by the age of 20 years [1]. The traditional orthopaedic view of intervertebral disk pathology emphasised structural lesions [2] and physical causes [3, 4]. More recently, however, this common-sense approach has been overshadowed by the 'disk degeneration' paradigm, which considers that some biological process related to genetics and disk nutrition can be held responsible for all aspects of disk pathology and pain. Mechanisms of disk injury, either by wear and tear ('fatigue') or by trauma, have been overlooked on the grounds that they represent an outmoded 'injury model', with some advocates of degeneration appearing to deny the possibility that intervertebral disks can be injured at all!

And yet, there is increasing evidence that intervertebral disks often are injured and that injury leads directly to altered biology, including nerve ingrowth and pain. Animal experiments show that

M.A. Adams, BSc, PhD (⊠) • P. Dolan, BSc, PhD Centre for Comparative and Clinical Anatomy, University of Bristol, Southwell Street, Bristol BS2 8EJ, UK e-mail: M.A.Adams@bris.ac.uk

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_3





disk injury leads inevitably to degeneration [5–7], and longitudinal studies on humans confirm that disk degeneration usually follows an injury to a disk or vertebral body [8, 9]. Epidemiological studies show that excessive mechanical loading is associated with disk degeneration and prolapse [10–12], although we now suspect that moderate mechanical loading actually strengthens spinal tissues by the process of adaptive remodelling [13, 14]. All of the structural features of disk degeneration (such as annulus fissures, nucleus herniation, endplate defects and internal disk disruption) can be reproduced by excessive mechanical loading applied to cadaveric spines, and the failure mechanisms have been explained by mathematical models [15]. No 'disk degeneration gene' has been found, only a range of gene variants (alleles) which appear to exert their influence by weakening the extracellular matrix of the disk [16, 17] or bony endplates [18]. Finally, we now know that metabolite transport into the disk increases with advancing age and degeneration, as the vertebral endplates become more porous and permeable [19], suggesting that inadequate disk nutrition is unlikely to be a direct cause of disk degeneration.

# 3.1.3 Purpose and Scope of This Chapter

The purpose of this chapter is not to revive some old 'injury' model of intervertebral disk failure, but to integrate old and new evidence concerning disk injury with that concerning disk biology, pathology and pain. Essentially, the following sections will show how some intervertebral disks can be so weakened by genetic inheritance and by advancing age that they fail mechanically in response to specific types of everyday mechanical loading. Attempts at healing are frustrated by low cell density and the harsh mechanical environment, so that disk cell metabolism becomes increasingly abnormal ('degeneration') and structural disruption increases. Eventually, blood vessels and nerves are able to grow into the disrupted tissue, and the disk becomes painful.

The chapter begins with an account of forces acting on the spine, because many clinicians consider only gravitational forces and overlook the much greater forces that arise from muscle tension and from accelerations. This is followed by a detailed account of how excessive forces can cause specific types of structural damage to lumbar disks and their adjacent vertebrae, including disk herniation. Damage can be caused by injury or by fatigue ('wear and tear') loading, and forces need not be high if the tissues are weak. Subsequent sections compare disk ageing and degeneration, because they are not the same thing: ageing is only one of several risk factors for degeneration, which is not inevitable, not even in old age.

# 3.2 Forces Acting on the Lumbar Spine

# 3.2.1 Compression, Shear, Bending and Torsion

The nature of these forces can be appreciated from Fig. 3.2a. The compressive force denotes the force acting down the long axis of the spine, perpendicular to the mid-plane of each intervertebral



**Fig. 3.2** (a) Human lumbar spine showing the direction of the spinal compressive force (*C*) and shear force (*S*) acting on the L5–S1 disk. A bending moment (*BM*) causes the spine to flex, extend or bend laterally, and an axial torque (*AT*) twists the spine about its long axis. (b) Analysis of forces during manual handling. A high tensile force (*F*) must be generated by the back muscles in order to lift up the weight (*W*), as well as upper body weight (*w*).

The back muscles act on a short lever arm (**d**) relative to the centre of rotation (*O*), whereas *W* and *w* act on much longer lever arms (*D* and  $d_w$ ). Therefore F must be much greater than *W*. The equations show how to calculate the spinal compressive force (**c**) in a static analysis. F and C would increase greatly if the weight was lifted quickly. Reproduced from Adams et al. [15] with permission disk. Note that its direction varies with spinal level and with posture. Compression is resisted mostly by the intervertebral disks and vertebral bodies [20]. The shear force acts parallel to the mid-plane of each disk and is steeply inclined to the horizontal in the lower lumbar spine. Forward shear is resisted by the articular surfaces of the apophyseal joints as well as by the disk [21]. A 'bending moment' (force X distance) causes the spine to bend forwards, backwards or laterally. Forward and backward bending are resisted primarily by spinal ligaments [22] and the neural arch [23], respectively. Torsion (axial rotation) twists the spine about its long axis and in the lumbar spine is resisted primarily by the neural arch [24].

#### 3.2.2 Gravitational Loading

Superincumbent body weight exerts a vertical force of approximately 60 % of body weight on the lumbosacral joint. In upright postures, the gravitational force mostly compresses the spine, but also exerts a forward shear force on the lower lumbar vertebrae (Fig. 3.2a).

# 3.2.3 Inertial Forces During Rapid Movements and Falls

High forces are required to move the body quickly, in accordance with Newton's 2nd Law of motion which states that

Force = mass 
$$\times$$
 acceleration (3.1)

When a pilot ejects from an aircraft, the vertical force on his lumbar spine is equal to his upper body mass multiplied by the vertical acceleration (rate of change of velocity) of his seat. This acceleration can be sufficient to crush one or more vertebrae. More commonly, high inertial forces are generated by a fall on the buttocks, when the deceleration on impact magnifies the effect of upper body weight. Peak deceleration increases with the length of the persons' legs and with the hardness of the landing.

# 3.2.4 Forces Arising from Muscle Tension

Muscle contractions pull their bony attachments towards each other, often compressing the joint that lies between them. Standing or sitting erect requires considerable 'antagonistic' muscle tension to stabilise the spine, in a similar manner to cables that stabilise a radio mast, so that the compressive force on the lumbar spine is typically 80-100 kg [25]. Muscle forces rise to high levels when objects are lifted on outstretched arms (Fig. 3.2b) because the external moment of the weight being lifted must be balanced by an internal moment, generated by muscle tension acting on a short lever arm, close to the centre of rotation. Internal muscle forces increase when movements are performed quickly [26, 27] because fast movements require high accelerations, which in turn require high forces (Eq. 3.1). It follows that maximum muscle forces are required to move the body as quickly as possible. Evidence from people suffering from epilepsy indicates that, when neural inhibitory controls are deficient, muscle tension can be high enough to crush a vertebra [28].

Confusion occasionally arises in medicolegal disputes (see final section, below) when an otherwise expert witness is unaware that forces generated by muscle tension and by falls can be *much* greater than the weight of the external object being handled and that the external object often weighs a good deal less than upper body weight, which must also be lifted against gravity by muscle tension (Fig. 3.2b).

#### 3.3 Mechanisms of Disk Injury and Prolapse

#### 3.3.1 Compressive Injury Causes Endplate Fracture

Excessive compressive loading, in the direction of the spine's long axis, always damages a vertebral endplate before the adjacent disk. This fact has been demonstrated in many experiments on cadaveric spines [29] and probably contributed to the myth that healthy disks can degenerate but not become injured. The problem appears to have arisen historically when calculations first showed how great the compressive force on the spine can be (see Fig. 3.2b). Spinal loading became synonymous with spinal *compressive* loading, and the harmful effects of torsion and bending (see below) were overlooked.

Compressive failure occurs by fracture of the bony endplate, as the incompressible fluidlike disk nucleus causes it to bulge into the vertebral body [30]. Various types of endplate fracture have been described [29, 31], and later work showed how a wider range of vertebral body fracture patterns can occur, depending on posture (flexion or extension) and on the degree of disk degeneration [32]. If compressive loading is applied in a repetitive cyclic manner, in order to simulate manual labour, then similar fracture types are observed to those caused by sudden injury, but the peak compressive force can be up to 50 % lower [33]. Generally, it is the central region of the superior endplate (relative to the vertebra) that is damaged first, because it is the thinnest and supported by less dense trabecular bone [34]. Thickness and strength of the central endplate evidently have a very low margin of safety, probably because thickness is minimised to allow adequate metabolite transport into the disk nucleus [35].

# 3.3.2 Torsion and Bending Injuries Tear the Annulus

Axial rotation ('torsion') of the lumbar spine (Fig. 3.2a) occurs about a centre of rotation in the posterior quadrant of the disk [24]. The movement is resisted primarily by the anterior annulus and by the articular facets of the apophyseal joints. Damage probably occurs first in the apophyseal joint that is compressed, at a rotation angle of  $1-3^{\circ}$ , although this is not certain [24]. If the neural arch is removed, then unprotected lumbar disks can be rotated to much greater angles [36, 37], and failure occurs in the annulus, by a mechanism that involves delamination



**Fig. 3.3** The mechanism of disk prolapse depends on how the lumbar disks are deformed in flexed posture (such as that shown in Fig. 3.2b). The upper diagram shows typical dimensions of a lumbar disk in upright posture (*AF* annulus fibrosus, *NP* nucleus pulposus, *H* anterior height, *h* posterior height, *T* thickness of posterior annulus). Full flexion then deforms the annulus as shown in the lower diagram. Note that the posterior annulus typically is stretched vertically by 50 % and so must be thinned by 37 % to maintain constant volume. The stretched and thinned posterior annulus is then vulnerable to high pressures in the disk nucleus

and rim tears [36]. However, torsion has not been shown to cause radial fissures or disk prolapse.

# 3.3.3 Complex Loading Can Cause Disk Herniation

Bending of the spine causes the intervertebral disks to be compressed on one side and stretched on the other (Fig. 3.3). In the case of forward bending (flexion), the posterior annulus is typically stretched by 50 % in a full range movement [38]. There is a corresponding decrease in annulus thickness in the radial (inner to outer) direction because the wet tissue must maintain constant volume, at least initially. Thinning of the posterior annulus in flexion has been confirmed in a radioactive tracer study [39], and it leaves this region of annulus vulnerable to injury from a high nucleus pressure. Consequently, if a cadaveric spine specimen is first positioned in full flexion or hyperflexion, and then compressed vigorously, the most common mode of failure is for the nucleus to herniate through the stretched posterior annulus [38, 40]. The herniated tissue



**Fig. 3.4** Structural defects in degenerated and herniated lumbar disks. Anterior is on the left in all figures. (a) Microradiograph of a midsagittal slice of a cadaveric vertebral body, showing a large Schmorl's node resulting from an endplate fracture that was followed by vertical disk herniation. (b) Photograph of a cadaveric disk that has herniated vertically through the endplate in response to compressive loading. Lamellae of the inner annulus are

collapsing into the decompressed nucleus. (c) Midsagittal section of a cadaveric intervertebral disk showing a complete radial fissure in the posterior annulus. (d) Similar to (c), except that in this specimen, high loading in bending and compression has caused the nucleus to herniate through the annulus. The disk was intact before loading. Images adapted from Adams et al. [15] with permission

(Figs. 3.4d and 3.5) emerges in a fraction of a second, sometimes with an audible 'pop'. Spinal bending is crucial to the disk herniation mechanism, and no cadaveric experiment has created disk herniation in the absence of a high bending moment.

The mechanism occurs most readily in spines aged 40–49 years [38] at which age there is a demonstrable fluid pressure in the nucleus [41], and yet the annulus is normally showing some signs of age-related weakening, most notably in the proliferation of concentric tears or clefts [1]. It is



**Fig. 3.5** Large disk prolapse created by mechanical loading of a cadaveric spine in bending and compression (bar=5 mm). The herniated tissue is mostly nucleus and inner annulus, but there is also some hyaline cartilage and bone from the adjacent endplate. Adapted from Adams et al. [15] with permission

not easy to create disk herniation in this way if the disk is already degenerated [38] or if its nucleus has been dehydrated artificially by prior creep loading [42], indicating that high nucleus pressure plays an important role. Lower lumbar disks are more likely to herniate in this way [38], presumably because they have a thin posterior annulus which can be stretched substantially when the spine is flexed.

The underlying mechanism of herniation pressurised nucleus pulposus bursting through a stretched and weakened annulus - has been clearly demonstrated by artificially raising the pressure inside animal disks and tracking the pattern of annulus disruption from inner to outer lamellae [43]. Later work emphasised the importance of flexion in causing a radial fissure to track through the posterior annulus, often close to its junction with the vertebral endplate [44]. Anterolateral bending (forwards and to one side) is closely related to disk prolapse in life [10], possibly because it causes maximum stretching of one posterolateral corner of the annulus, which lies further from the axis of bending than does the midline posterior annulus.

If the same combination of bending and compression is applied in a repetitive fashion, to simulate heavy manual labour, then a radial fissure (Fig. 3.4c) can be formed in a gradual progressive manner [45]. Only a small quantity of nucleus pulposus can then be extruded though the annulus, presumably because even a small loss of nucleus volume causes a relatively large drop in nucleus pressure [46] so that no more material is forced down the fissure. Experiments on young porcine disks confirm that this mechanism of progressive radial fissure formation in the posterior annulus does not require any prior degenerative changes in the disk [47, 48].

#### 3.3.4 Endplate Involvement in Disk Prolapse

Disk herniations are mostly comprised of displaced annulus fibrosus and nucleus pulposus tissues [49], with the annulus/nucleus ratio increasing in patients aged over 30 years [50]. In addition, almost half of all herniations also contain some hyaline cartilage from the vertebral endplate [49, 51]. The relatively hard and smooth hyaline cartilage has a three-dimensional network of fine collagen type II fibrils which prevent it from swelling much, even when physically disrupted [52]. Consequently, hyaline cartilage fragments lose little of their proteoglycan content, so that they are relatively resistant to revascularisation, reinnervation and resorption [51]. This probably explains why their presence in a herniation is associated with more severe and prolonged clinical symptoms [50]. The cartilage endplate is mechanically integrated with the collagen network of the disk [53, 54], and yet it can easily be peeled off the underlying bone if the annulus is stretched vertically [51], so spinal bending is likely to play an important role in cartilaginous herniations. They are most frequent in patients aged 50-60 years, and relatively large cartilage fragments are most commonly found in herniations of L5–S1 disks [50].

Bone fragments also occur in some disk herniations [49], including those created in the laboratory [38]: see Fig. 3.5. The reported frequency in patients varies from 6 % [50] to 58 % [55], with the latter figure including any abnormality (including avulsion fracture) of the vertebral rim that could be visualised by CT. Close association between hyaline cartilage and bone fragments (Fig. 3.5) suggests that herniating annulus sometimes pulls away a small fragment of cartilage and bone endplate together. In other cases, displaced outer annulus appears to pull some bone off the posterior vertebral margin, without the inclusion of any hyaline cartilage. Detailed study of cadaveric bony endplates has shown that 'erosions' are common in the posterolateral margins of lower lumbar vertebrae [56]. Their appearance, location in the spine and association with patients' symptoms are all consistent with them representing the stripping of cartilage off the underlying bone during disk herniation.

Defects in the human bony endplate are densely innervated [57], so it is no surprise that disk herniations involving the endplate are of particular clinical significance. As well as the abovementioned link with prolonged sciatica, it appears that they are also associated with inflammatory ('Modic') changes in the endplate and with severe and painful disk infections [58, 59]. One possible explanation for these findings is that focal loss of hyaline cartilage endplate exposes the subchondral bone, which has much greater porosity and permeability than the cartilage [19]. Hence, focal cartilage loss increases focal endplate permeability, allowing anaerobic bacteria to enter the nucleus from the vertebral body and for inflammatory cytokines from disk cells to leave the nucleus and sensitise nerves in the bony endplate.

# 3.3.5 Mechanical Consequences of Disk Injury and Herniation

Endplate fracture creates more space for the disk nucleus (Fig. 3.4a, b), causing the pressure within it to fall immediately, often by more than 50 % [40, 60, 61]. Consequently, compressive load bearing is shifted from the nucleus to annulus and also from the disk overall to the neural arch [60], with effects being the greatest in older

spines and at upper lumbar levels [61]. High stresses in the inner annulus can cause it to collapse into the decompressed nucleus, as shown in Fig. 3.4b [40]. Localised tears in the outer annulus have much less immediate effect on nucleus pressure [62], although animal experiments suggest that peripheral annulus tears can propagate inwards over several months until they reach the nucleus [5].

Another immediate mechanical consequence concerns the tissue displaced in a disk herniation: once removed from the pressurised confines of the disk, it can swell rapidly in tissue fluid. Cadaver experiments suggest that herniated tissue can more than double its weight in just 4 h and then shrink again during the following few days as it loses both proteoglycans and water [63]. These physico-chemical events could explain why some patients report a gradual onset of sciatica several hours after some recalled incident. Disk herniation also reduces nucleus pressure and volume, probably by an amount that is proportional to the herniated mass [64].

Over a longer timescale, nucleus decompression arising from either endplate fracture or disk herniation allows the annulus to bulge radially outwards, and also inwards, so that the disk loses height [65, 66]. Disk narrowing can result in the transfer of more than 50 % of the spinal compressive load on to the neural arch [20], where it appears to lead to osteoarthritis in the apophyseal joints [67–69]. Reduced separation of adjacent vertebrae creates slack in the intervertebral ligaments, so that they resist bending less [42], and the motion segment is then able to 'wobble' freely [70]. This 'segmental instability' can subsequently be reversed by the growth of vertebral body osteophytes [71].

# 3.4 Lumbar Disk Ageing and Degeneration

These processes are considered in detail elsewhere in this book, but they are mentioned briefly here in order to differentiate them from each other and to integrate them with the evidence concerning disk injury.

# 3.4.1 Inevitable Age-Related Changes in Human Lumbar Disks

All old intervertebral disks become dehydrated, discoloured and fibrous. Progressive age-related fragmentation and loss of proteoglycans explain why disk water content falls, especially in the nucleus [72]. Concurrent replacement of collagen type II with type I explains the increased fibrous texture of the nucleus and inner annulus [73], and increased collagen cross-linking involving sugars explains the yellowish discoloration [74]. These biochemical changes can, in turn, be attributed to an increasing proportion of disk cells becoming senescent [75] so that continuing repair ('turnover') of matrix macromolecules becomes slower with advancing age [76]. After growth is complete, age-related changes in disk composition make them stiffer, less able to distribute loading evenly on the adjacent vertebrae and more easily injured. Hence, minor structural defects tend to accumulate in the disk after the second decade [1]. In addition, the bony endplate becomes thinner and more porous with increasing age [77] reflecting systemic osteopaenia in many older people. However, most old disks do not become grossly disrupted or narrowed [78], and their internal mechanical functioning is little affected [41]. Generally, there is no blood vessel or nerve ingrowth in most old disks [79].

# 3.4.2 Features of Intervertebral Disk Degeneration

Disk 'degeneration' has traditionally been graded on numerical scales (e.g. 1–4) according to the presence or absence of specific features [80–82]. Features associated with 'degeneration' include accelerated age-related changes in composition, radial and circumferential fissures in the annulus, rim tears in the outer annulus, bulging or damaged endplates, annulus bulging radially outwards or collapsing inwards, disk space narrowing, marginal osteophytes on adjacent vertebral bodies and ingrowing blood vessels and nerves. Disk degeneration scales are exercises in pattern recognition and do not seek to define or explain what is going on. However, they can be used to establish statistical associations between disk degeneration and back pain [83, 84] and with aspects of abnormal disk function such as a decompressed nucleus [41], high stress gradients in the annulus [85] and decreasing spinal mobility [86].

#### 3.4.3 What <u>Is</u> 'Disk Degeneration'?

If all age-related changes in intervertebral disks are considered to represent 'degeneration', then it becomes difficult for epidemiologists to identify risk factors for the condition (because all disks age) or to assess the efficacy of disease-modifying treatments (reversing ageing may prove difficult!). Evidently, specific age-related changes that are closely related to pain need to be distinguished from 'normal' ageing.

The most widely cited definitions were suggested in 2006 [87]. *Intervertebral disk degeneration* is proposed to be 'an aberrant, cell-mediated response to progressive structural failure', and a *degenerate disk* is the one with 'structural failure combined with accelerated or advanced signs of ageing'. *Degenerative disk disease* is 'a degenerated disk which is also painful'.

The rationale behind these definitions is that some disks degenerate because they are so weakened by genetic inheritance, age and loading history that they can be injured during normal everyday activity. Disk injuries create regions of abnormally high and low matrix compressive stress [40], both of which inhibit matrix synthesis [88, 89] and create abnormal cell signalling [90], elevated enzyme activity [72, 91] and cell clustering [92] as the injured tissue attempts to repair itself. Unfortunately, repair is frustrated by the disk's low cell density, so that degeneration often progresses to complete structural failure, as suggested in Fig. 3.6. The emphasis on structural failure is justified by the close association between structural defects and pain and by the fact that all of the structural features of disk degeneration can be created in cadaveric specimens by mechanical loading.





It is sometimes implied that disks degenerate *because of* altered cell signalling or *because of* impaired regulation of matrix-degrading enzymes, but there is no evidence to support this conjecture, and it merely begs the question: what caused the abnormal signalling or impaired enzyme regulation in the first place?

# 3.4.4 Two Disk Degeneration 'Phenotypes'?

Disk herniation can be considered as one specific feature of degeneration or as a separate pathological entity. But whichever convention is used, it should be realised that disk herniation is much more closely related to mechanical loading, and to pain, than are other features of degeneration such as MRI signal intensity, which is primarily related to water loss. It is instructive to consider distinct 'phenotypes' of lumbar disk degeneration, which have different characteristics, risk factors and consequences. In a recent review, we proposed that there are two such phenotypes [93].

'Endplate-driven' degeneration is initiated by endplate fracture, involves internal annulus disruption, mostly affects the upper lumbar and thoracic disks, has relatively high heritability and can start early in life, but has a lower association with pain. In contrast, 'annulus-driven' degeneration involves a radial fissure and/or a disk herniation, mostly affects the lower lumbar spine, has relatively low heritability, develops progressively in middle age and often leads to severe back pain and sciatica. The structural defects which initiate the two processes both act to decompress the disk nucleus, making it less likely that the other defect could occur subsequently, and in this sense the two disk degeneration phenotypes can be viewed as distinct. It remains to be seen just how useful this concept is in identifying risk factors and in devising treatments [93].

# 3.4.5 A 'Final Common Pathway' for Disk Degeneration?

The annulus of a decompressed degenerating disk tends to bulge like a flat tyre and typically loses 3 % of its height per year [94]. The degeneration process may therefore take many years before the disk space is obliterated and the adjacent vertebrae fuse. Loss of disk height transfers compressive loading on to the neural arch and probably precipitates a degenerative cascade of segmental instability, apophyseal joint osteoarthritis, vertebral osteophytosis and spinal stenosis [15].

#### 3.4.6 Disk Degeneration, Back Pain and Sciatica

Injuries play an important role in the genesis of diskogenic back pain and sciatica. A disk herniation can compress a spinal nerve root or ganglion, and chemicals leaching out from the swollen and displaced tissue [63] may sensitise the affected neurons [95]. Fissures within the annulus encourage ingrowth of blood vessels and nerves [96] because they are focal regions of low pressure [97] in which hollow blood vessels would be less likely to collapse. The disrupted edges of annulus fissures are able to swell and lose proteoglycans, leaving a collagenous scaffold [97] that would not inhibit nerve and blood vessel ingrowth as much as a proteoglycan-rich matrix would [98, 99]. Similar events probably occur in a disrupted endplate, because endplate defects have been shown to have an increased nerve density [100]. In both annulus and endplate, the sequence of events leading to chronic pain is likely to be injury, reinnervation, sensitisation by inflammation or infection and provocation of sensitised nerves by focal stress concentrations within the disrupted tissue.

# 3.5 When Do Disk Injuries Occur In Vivo?

# 3.5.1 Moderate Mechanical Loading Strengthens the Spine

The small cell population of adult human disks does not normally decline after skeletal maturity [101] and appears well adapted to the anaerobic conditions that arise from poor metabolite transport [102]. Slow matrix turnover is measurable [76], and comparisons between the mechanical properties of human disk tissues and their adjacent vertebrae suggest that disks are capable of limited adaptation to mechanical demands [103], at least in the outer annulus where cell density is highest [104]. This could explain why effective healing is observed in the outer annulus of sheep disks following surgical injury [5] and why increased body mass appears to have positive effects on disk hydration and health [14]. Other fibrous connective tissues are known to strengthen in response to moderate mechanical loading [105], as of course is bone [106].

# 3.5.2 'Injury' Occurs When Loading In Excessive

An 'injury' denotes damage to a living tissue. If the cause is mechanical, then injury starts at the elastic limit, when resistance to loading first becomes impaired. Beyond this limit, nonreversible deformation starts and is probably accompanied by pain [107]. In brittle tissues such as bone, injury is accompanied by visual and audible signs. However, in tough fibrous and cartilaginous tissues, injury begins with collagen fibres sliding imperceptibly past each other [108], and 'injury' can be identified in vitro only from subtle changes in a force-deformation graph [109].

In living people, soft tissue injuries must be inferred from patients' symptoms and from MRI scans, and often there is no objective confirmation that an injury has occurred. This disparity in the ability to detect hard and soft tissue injuries can lead to the latter being overlooked or dismissed as unimportant. It is true that some wellvascularised soft tissues heal quickly and are not as serious as bone fractures. But injuries to intervertebral disks, cartilage and tendon can be *more* serious than bone fractures, because their low healing potential often results in progressive and painful degenerative conditions [87].

The word 'injury' should not be used synonymously with 'trauma', which implies very high loading associated with collisions and falls. Injury simply requires the mechanical loading to exceed tissue strength, and if the tissue has been severely weakened by the combined influences of an unfavourable genetic inheritance, ageing and prior 'fatigue' loading, then injury can occur during the activities of everyday living. This is widely accepted in the case of osteoporotic vertebral fracture, which can be caused simply by opening a window [110].

# 3.5.3 Repetitive Loading and 'Fatigue Failure'

Mechanical injury may represent a single excessive loading cycle or a large number of more moderate loading cycles which cause microscopic damage to propagate within the material until gross 'fatigue failure' occurs. This is the process by which metal fatigue can propagate in aeroplane wings. For example, fatigue failure is likely to occur in human vertebrae following 5,000 loading cycles at only 40–50 % of the normal failure load [33]. Similar weakening occurs if annulus fibrosus is subjected to repetitive loading [111]. Intervertebral disks are particularly vulnerable to accumulating fatigue damage because they have so few cells to turnover and repair their matrix.

# 3.5.4 Why Are Some Intervertebral Disks so Weak?

Genetic inheritance explains approximately 30 % of the variance in disk degeneration in the lower lumbar spine and 50 % of variance in the upper lumbar spine [112]. The difference is probably explained by greater environmental (especially mechanical) influences at L4-S1. Heritability can rise as high as 70 % in middle-aged women [113], who are less likely to be 'discordant' for mechanical loading. There is no single 'disk degeneration gene': rather, many gene variants exert small influences on matrix strength and metabolism. Evidently, genes and environmental influences are both important in disk degeneration. Age-related increases in disk degeneration [83, 114] are probably attributable to the matrix becoming weaker and more vulnerable to injury.

#### 3.5.5 Medicolegal Considerations

The following section is an abridged account of a recent review [115].

#### 3.5.5.1 Disk Degeneration vs Herniation

These two terms should not be used synonymously. Disk herniation could be viewed as a distinct pathological entity or as an advanced stage of 'annulus-driven' disk degeneration, as described above. Certainly, both of these disk degeneration phenotypes should be distinguished from disks that are merely growing old, perhaps with a little 'middle-aged spread'. Unfortunately, there is still no scientific consensus on what 'disk degeneration' actually means, with scientists placing varying emphasis on the role of disk nutrition, or genetic inheritance, or abnormal cell signalling or mechanical loading. Disk herniation is better understood than disk degeneration, and there is no doubt that it can be a mechanical injury, but there is still disagreement over the relative influences of age and genetic inheritance.

#### 3.5.5.2 Must a Disk Degenerate Before It Can Herniate?

No, this much is now certain. A high proportion of middle-aged lower lumbar cadaveric disks will herniate if loaded severely enough [38], and most of the 'degenerative' changes found in herniated disk material removed at surgery probably occur after the herniation takes place, as a result of tissue swelling, leaching of proteoglycans and revascularisation [92]. It is important to avoid circular arguments such as 'This disk herniated because it was degenerated. We know it was degenerated because it herniated'.

# 3.5.5.3 Mechanical 'Acceleration' of Disk Degeneration?

It is sometimes proposed that an injury or work practise has 'accelerated' disk degeneration, so that pain and disability develop earlier than might be expected. The problem with this concept is that excessive mechanical loading does not influence the musculoskeletal system by accelerating metabolic ageing effects. Rather, it diverts the disk from its normal 'ageing' pathway to a separate 'degeneration' pathway, which involves structural disruption, altered biomechanics and metabolism, revascularisation and reinnervation. It is the diverging 'degeneration' pathway rather than the ageing pathway that leads to pain and disability.

#### 3.5.5.4 Who Is Prone to Injury?

The evidence presented in this chapter may appear to blame mechanical loading for diskogenic back pain, by diverting an ageing disk on to a 'degeneration pathway'. However, even moderate mechanical loading can disrupt a very weak disk, and tissue weakening depends on genetic inheritance and ageing. If pain and disability arise in the absence of any substantial mechanical provocation, then it can mostly be attributed to constitutional factors (ageing and genetic inheritance) that weaken the disk and *predispose* it to injury and degeneration. Alternatively, if there is a substantial mechanical provocation, then pain and disability can mostly be blamed on the injury or work practise which *precipitated* the disk injury and degeneration. Liability should be apportioned according to the perceived relative importance of these predisposing and precipitating causes.

#### 3.5.5.5 Summary of Recent Scientific Advances

The following statements would be difficult to refute on the basis of current evidence:

- Injury to the annulus or endplate can cause intervertebral disks to degenerate.
- Excessive mechanical loading can cause many lower lumbar intervertebral disks to herniate, even if they appear 'normal' for their age.
- Disk herniations are often injuries, but few are traumatic, and prior tissue weakening (arising from genetic inheritance, age and prior 'wear and tear') will often contribute to the herniation.
- Most degenerative changes in surgically removed disk herniations are consistent with them occurring *after* herniation.
- It should not be assumed that a herniated disk must have been degenerated before it herniated, unless there is independent evidence of this prior degeneration.

#### References

- Haefeli M, Kalberer F, Saegesser D, Nerlich AG, Boos N, Paesold G. The course of macroscopic degeneration in the human lumbar intervertebral disc. Spine. 2006;31:1522–31.
- Hirsch C, Schajowicz F. Studies on structural changes in the lumbar annulus fibrosus. Acta Orthop Scand. 1953;22:184–231.

- Newman PH. Sprung back. J Bone Joint Surg [Br]. 1952;34:30–7.
- 4. Nachemson AL. Lumbar intradiscal pressure. Acta Orthop Scand Suppl. 1960;43:1–104.
- Osti OL, Vernon-Roberts B, Fraser RD. Volvo award in experimental studies. Anulus tears and intervertebral disc degeneration. Spine. 1990;15:762–7.
- Holm S, Holm AK, Ekstrom L, Karladani A, Hansson T. Experimental disc degeneration due to endplate injury. J Spinal Disord Tech. 2004;17: 64–71.
- Ulrich JA, Liebenberg EC, Thuillier DU, Lotz JC. ISSLS prize winner: repeated disc injury causes persistent inflammation. Spine. 2007;32:2812–9.
- Kerttula LI, Serlo WS, Tervonen OA, Paakko EL, Vanharanta HV. Post-traumatic findings of the spine after earlier vertebral fracture in young patients: clinical and MRI study. Spine. 2000;25:1104–8.
- Carragee EJ, Don AS, Hurwitz EL, Cuellar JM, Carrino J, Herzog R. ISSLS prize winner: does discography cause accelerated progression of degeneration changes in the lumbar disc: a ten-year matched cohort study. Spine. 2009;34:2338–45.
- Kelsey JL, Githens PB, White AA, Holford TR, Walter SD, O'Connor T, Ostfeld AM, Weil U, Southwick WO, Calogero JA. An epidemiologic study of lifting and twisting on the job and risk for acute prolapsed lumbar intervertebral disc. J Orthop Res. 1984;2:61–6.
- Sward L, Hellstrom M, Jacobsson B, Nyman R, Peterson L. Disc degeneration and associated abnormalities of the spine in elite gymnasts. A magnetic resonance imaging study. Spine. 1991;16:437–43.
- Seidler A, Bolm-Audorff U, Siol T, Henkel N, Fuchs C, Schug H, Leheta F, Marquardt G, Schmitt E, Ulrich PT, Beck W, Missalla A, Elsner G. Occupational risk factors for symptomatic lumbar disc herniation; a case-control study. Occup Environ Med. 2003;60: 821–30.
- Adams MA, Dolan P. Could sudden increases in physical activity cause degeneration of intervertebral discs? Lancet. 1997;350:734–5.
- Videman T, Gibbons LE, Kaprio J, Battie MC. Challenging the cumulative injury model: positive effects of greater body mass on disc degeneration. Spine J. 2010;10:26–31.
- Adams M, Bogduk N, Burton K, Dolan P. The biomechanics of back pain. 3rd ed. Edinburgh: Churchill Livingstone; 2013.
- Kawaguchi Y, Osada R, Kanamori M, Ishihara H, Ohmori K, Matsui H, Kimura T. Association between an aggrecan gene polymorphism and lumbar disc degeneration. Spine. 1999;24:2456–60.
- 17. Seki S, Kawaguchi Y, Chiba K, Mikami Y, Kizawa H, Oya T, Mio F, Mori M, Miyamoto Y, Masuda I, Tsunoda T, Kamata M, Kubo T, Toyama Y, Kimura T, Nakamura Y, Ikegawa S. A functional SNP in CILP, encoding cartilage intermediate layer protein, is associated with susceptibility to lumbar disc disease. Nat Genet. 2005;37:607–12.

- Videman T, Leppavuori J, Kaprio J, Battie MC, Gibbons LE, Peltonen L, Koskenvuo M. Intragenic polymorphisms of the vitamin D receptor gene associated with intervertebral disc degeneration. Spine. 1998;23:2477–85.
- Rodriguez AG, Slichter CK, Acosta FL, Rodriguez-Soto AE, Burghardt AJ, Majumdar S, Lotz JC. Human disc nucleus properties and vertebral endplate permeability. Spine. 2011;36:512–20.
- Pollintine P, Przybyla AS, Dolan P, Adams MA. Neural arch load-bearing in old and degenerated spines. J Biomech. 2004;37:197–204.
- Skrzypiec DM, Bishop NE, Klein A, Püschel K, Morlock MM, Huber G. Estimation of shear load sharing in moderately degenerated human lumbar spine. J Biomech. 2013;46:651–7.
- Adams MA, Hutton WC, Stott JR. The resistance to flexion of the lumbar intervertebral joint. Spine. 1980;5:245–53.
- Adams MA, Dolan P, Hutton WC. The lumbar spine in backward bending. Spine. 1988;13:1019–26.
- Adams MA, Hutton WC. The relevance of torsion to the mechanical derangement of the lumbar spine. Spine. 1981;6:241–8.
- Sato K, Kikuchi S, Yonezawa T. In vivo intradiscal pressure measurement in healthy individuals and in patients with ongoing back problems. Spine. 1999;24: 2468–74.
- Dolan P, Earley M, Adams MA. Bending and compressive stresses acting on the lumbar spine during lifting activities. J Biomech. 1994;27:1237–48.
- Mannion AF, Adams MA, Dolan P. Sudden and unexpected loading generates high forces on the lumbar spine. Spine. 2000;25:842–52.
- Vascancelos D. Compression fractures of the vertebra during major epileptic seizures. Epilepsia. 1973;14: 323–8.
- Brinckmann P, Biggemann M, Hilweg D. Prediction of the compressive strength of human lumbar vertebrae. Clin Biomech. 1989;4 Suppl 2:1–27.
- Brinckmann P, Frobin W, Hierholzer E, Horst M. Deformation of the vertebral end-plate under axial loading of the spine. Spine. 1983;8:851–6.
- Brinckmann P, Biggemann M, Hilweg D. Prediction of the compressive strength of human lumbar vertebrae. Spine. 1989;14:606–10.
- Adams MA, Dolan P. Biomechanics of vertebral compression fractures and clinical application. Arch Orthop Trauma Surg. 2011;131:1703–10.
- Brinckmann P, Biggemann M, Hilweg D. Fatigue fracture of human lumbar vertebrae. Clin Biomech. 1988;3 Suppl 1:11–8.
- 34. Zhao FD, Pollintine P, Hole BD, Adams MA, Dolan P. Vertebral fractures usually affect the cranial endplate because it is thinner and supported by less-dense trabecular bone. Bone. 2009;44:372–9.
- Ferguson SJ, Ito K, Nolte LP. Fluid flow and convective transport of solutes within the intervertebral disc. J Biomech. 2004;37:213–21.

- Farfan HF, Cossette JW, Robertson GH, Wells RV, Kraus H. The effects of torsion on the lumbar intervertebral joints: the role of torsion in the production of disc degeneration. J Bone Joint Surg Am. 1970;52: 468–97.
- Adams MA, Hutton WC. The mechanical function of the lumbar apophyseal joints. Spine. 1983;8: 327–30.
- Adams MA, Hutton WC. Prolapsed intervertebral disc. A hyperflexion injury 1981 Volvo Award in basic science. Spine. 1982;7:184–91.
- Adams MA, Hutton WC. The effect of posture on diffusion into lumbar intervertebral discs. J Anat. 1986;147:121–34.
- Adams MA, Freeman BJ, Morrison HP, Nelson IW, Dolan P. Mechanical initiation of intervertebral disc degeneration. Spine. 2000;25:1625–36.
- Adams MA, McNally DS, Dolan P. 'Stress' distributions inside intervertebral discs. The effects of age and degeneration. J Bone Joint Surg Br. 1996;78: 965–72.
- Adams MA, Dolan P, Hutton WC. Diurnal variations in the stresses on the lumbar spine. Spine. 1987;12: 130–7.
- 43. Veres SP, Robertson PA, Broom ND. ISSLS prize winner: microstructure and mechanical disruption of the lumbar disc annulus: part II: how the annulus fails under hydrostatic pressure. Spine. 2008;33:2711–20.
- Veres SP, Robertson PA, Broom ND. The morphology of acute disc herniation: a clinically relevant model defining the role of flexion. Spine. 2009;34:2288–96.
- Adams MA, Hutton WC. Gradual disc prolapse. Spine. 1985;10:524–31.
- Adams MA, McMillan DW, Green TP, Dolan P. Sustained loading generates stress concentrations in lumbar intervertebral discs. Spine. 1996;21:434–8.
- 47. Tampier C, Drake JD, Callaghan JP, McGill SM. Progressive disc herniation: an investigation of the mechanism using radiologic, histochemical, and microscopic dissection techniques on a porcine model. Spine. 2007;32:2869–74.
- Yates JP, Giangregorio L, McGill SM. The influence of intervertebral disc shape on the pathway of posterior/posterolateral partial herniation. Spine. 2010;35: 734–9.
- Moore RJ, Vernon-Roberts B, Fraser RD, Osti OL, Schembri M. The origin and fate of herniated lumbar intervertebral disc tissue. Spine. 1996;21:2149–55.
- Willburger RE, Ehiosun UK, Kuhnen C, Kramer J, Schmid G. Clinical symptoms in lumbar disc herniations and their correlation to the histological composition of the extruded disc material. Spine. 2004;29: 1655–61.
- Lama P, Zehra U, Balkovec C, Claireaux H, Flower L, Harding IJ, Dolan P, Adams MA. Significance of cartilage endplate within herniated disc tissue. Eur Spine J (submitted for publication) 2014;23:1869–77.
- Summers GC, Merrill A, Sharif M, Adams MA. Swelling of articular cartilage depends on the integrity

of adjacent cartilage and bone. Biorheology. 2008;45: 365–74.

- Rodrigues SA, Wade KR, Thambyah A, Broom ND. Micromechanics of annulus-end plate integration in the intervertebral disc. Spine J. 2012;12:143–50.
- Wade KR, Robertson PA, Broom ND. On how nucleus-endplate integration is achieved at the fibrillar level in the ovine lumbar disc. J Anat. 2012;221: 39–46.
- Rajasekaran S, Bajaj N, Tubaki V, Kanna RM, Shetty AP. ISSLS prize winner: the anatomy of failure in lumbar disc herniation: an in vivo, multimodal, prospective study of 181 subjects. Spine. 2013;38: 1491–500.
- Wang Y, Videman T, Battie MC. ISSLS prize winner: lumbar vertebral endplate lesions: associations with disc degeneration and back pain history. Spine. 2012;37:1490–6.
- Fields AJ, Liebenberg EC, Lotz JC. Innervation of pathologies in the lumbar vertebral end plate and intervertebral disc. Spine J. 2013;3:153–64.
- 58. Albert HB, Lambert P, Rollason J, Sorensen JS, Worthington T, Pedersen MB, Norgaard HS, Vernallis A, Busch F, Manniche C, Elliott T. Does nuclear tissue infected with bacteria following disc herniations lead to modic changes in the adjacent vertebrae? Eur Spine J. 2013;22:690–6.
- 59. Albert HB, Sorensen JS, Christensen BS, Manniche C. Antibiotic treatment in patients with chronic low back pain and vertebral bone edema (modic type 1 changes): a double-blind randomized clinical controlled trial of efficacy. Eur Spine J. 2013;22: 697–707.
- Luo J, Skrzypiec DM, Pollintine P, Adams MA, Annesley-Williams DJ, Dolan P. Mechanical efficacy of vertebroplasty: influence of cement type, BMD, fracture severity, and disc degeneration. Bone. 2007;40:1110–9.
- Dolan P, Luo J, Pollintine P, Landham PR, Stefanakis M, Adams MA. Intervertebral disc decompression following endplate damage: implications for disc degeneration depend on spinal level and age. Spine. 2013;38:1473–81.
- 62. Przybyla A, Pollintine P, Bedzinski R, Adams MA. Outer annulus tears have less effect than endplate fracture on stress distributions inside intervertebral discs: relevance to disc degeneration. Clin Biomech. 2006;21:1013–9.
- Dolan P, Adams MA, Hutton WC. The short-term effects of chymopapain on intervertebral discs. J Bone Joint Surg [Br]. 1987;69:422–8.
- Ranu HS. Multipoint determination of pressurevolume curves in human intervertebral discs. Ann Rheum Dis. 1993;52:142–6.
- Heuer F, Schmidt H, Wilke HJ. Stepwise reduction of functional spinal structures increase disc bulge and surface strains. J Biomech. 2008;41:1953–60.
- 66. O'Connell GD, Malhotra NR, Vresilovic EJ, Elliott DM. The effect of nucleotomy and the dependence of

degeneration of human intervertebral disc strain in axial compression. Spine. 2011;36:1765–71.

- Butler D, Trafimow JH, Andersson GB, McNeill TW, Huckman MS. Discs degenerate before facets. Spine. 1990;15:111–3.
- Tischer T, Aktas T, Milz S, Putz RV. Detailed pathological changes of human lumbar facet joints L1–L5 in elderly individuals. Eur Spine J. 2006;15: 308–15.
- Robson-Brown K, Pollintine P, Adams MA. Biomechanical implications of degenerative joint disease in the apophyseal joints of human thoracic and lumbar vertebrae. Am J Phys Anthropol. 2008;136: 318–26.
- Zhao F, Pollintine P, Hole BD, Dolan P, Adams MA. Discogenic origins of spinal instability. Spine. 2005;30:2621–30.
- Al-Rawahi M, Luo J, Pollintine P, Dolan P, Adams MA. Mechanical function of vertebral body osteophytes, as revealed by experiments on cadaveric spines. Spine. 2011;36:770–7.
- 72. Antoniou J, Steffen T, Nelson F, Winterbottom N, Hollander AP, Poole RA, Aebi M, Alini M. The human lumbar intervertebral disc: evidence for changes in the biosynthesis and denaturation of the extracellular matrix with growth, maturation, ageing, and degeneration. J Clin Invest. 1996;98:996–1003.
- Schollmeier G, Lahr-Eigen R, Lewandrowski K-U. Observations on fiber-forming collagens in the anulus fibrosus. Spine. 2000;25:2736–41.
- 74. DeGroot J, Verzijl N, Wenting-Van Wijk MJ, Jacobs KM, Van El B, Van Roermund PM, Bank RA, Bijlsma JW, TeKoppele JM, Lafeber FP. Accumulation of advanced glycation end products as a molecular mechanism for aging as a risk factor in osteoarthritis. Arthritis Rheum. 2004;50:1207–15.
- 75. Le Maitre CL, Freemont AJ, Hoyland JA. Accelerated cellular senescence in degenerate intervertebral discs: a possible role in the pathogenesis of intervertebral disc degeneration. Arthritis Res Ther. 2007;9:R45.
- 76. Sivan SS, Wachtel E, Tsitron E, Sakkee N, van der Ham F, Degroot J, Roberts S, Maroudas A. Collagen turnover in normal and degenerate human intervertebral discs as determined by the racemization of aspartic acid. J Biol Chem. 2008;283:8796–801.
- Rodriguez AG, Rodriguez-Soto AE, Burghardt AJ, Berven S, Majumdar S, Lotz JC. Morphology of the human vertebral endplate. J Orthop Res. 2012;30: 280–7.
- Adams MA, Lama P, Zehra U, D P. Why do some intervertebral discs degenerate, when others (in the same spine) do not? J Clin Anat. 2015;28(2):195–204.
- Palmgren T, Gronblad M, Virri J, Kaapa E, Karaharju E. An immunohistochemical study of nerve structures in the anulus fibrosus of human normal lumbar intervertebral discs. Spine. 1999;24:2075–9.
- Adams MA, Dolan P, Hutton WC. The stages of disc degeneration as revealed by discograms. J Bone Joint Surg [Br]. 1986;68:36–41.

- Thompson JP, Pearce RH, Schechter MT, Adams ME, Tsang IK, Bishop PB. Preliminary evaluation of a scheme for grading the gross morphology of the human intervertebral disc. Spine. 1990;15:411–5.
- Pfirrmann CW, Metzdorf A, Zanetti M, Hodler J, Boos N. Magnetic resonance classification of lumbar intervertebral disc degeneration. Spine. 2001;26: 1873–8.
- 83. Cheung KM, Karppinen J, Chan D, Ho DW, Song YQ, Sham P, Cheah KS, Leong JC, Luk KD. Prevalence and pattern of lumbar magnetic resonance imaging changes in a population study of one thousand fortythree individuals. Spine. 2009;34:934–40.
- 84. de Schepper EI, Damen J, van Meurs JB, Ginai AZ, Popham M, Hofman A, Koes BW, Bierma-Zeinstra SM. The association between lumbar disc degeneration and low back pain: the influence of age, gender, and individual radiographic features. Spine. 2010;35: 531–6.
- Stefanakis M, Luo J, Pollintine P, Dolan P, Adams MA. ISSLS prize winner: mechanical influences in progressive intervertebral disc degeneration. Spine. 2014;39(17):1365–72.
- 86. Kettler A, Rohlmann F, Ring C, Mack C, Wilke HJ. Do early stages of lumbar intervertebral disc degeneration really cause instability? Evaluation of an in vitro database. Eur Spine J. 2011;20:578–84.
- Adams MA, Roughley PJ. What is intervertebral disc degeneration, and what causes it? Spine. 2006;31: 2151–61.
- Ishihara H, McNally DS, Urban JP, Hall AC. Effects of hydrostatic pressure on matrix synthesis in different regions of the intervertebral disk. J Appl Physiol. 1996;80:839–46.
- Adams MA, Dolan P, McNally DS. The internal mechanical functioning of intervertebral discs and articular cartilage, and its relevance to matrix biology. Matrix Biol. 2009;28:384–9.
- 90. Shamji MF, Setton LA, Jarvis W, So S, Chen J, Jing L, Bullock R, Isaacs RE, Brown C, Richardson WJ. Proinflammatory cytokine expression profile in degenerated and herniated human intervertebral disc tissues. Arthritis Rheum. 2010;62:1974–82.
- 91. Weiler C, Nerlich AG, Zipperer J, Bachmeier BE, Boos N. SSE Award Competition in Basic Science: expression of major matrix metalloproteinases is associated with intervertebral disc degradation and resorption. Eur Spine J. 2002;11:308–20.
- Lama P, Le Maitre CL, Dolan P, Tarlton JF, Harding IJ, Adams MA. Do intervertebral discs degenerate before they herniate, or after? Bone Joint J. 2013; 95-B:1127–33.
- Adams MA, Dolan P. Intervertebral disc degeneration: evidence for two distinct phenotypes. J Anat. 2012;221:497–506.
- Hassett G, Hart DJ, Manek NJ, Doyle DV, Spector TD. Risk factors for progression of lumbar spine disc degeneration: the Chingford Study. Arthritis Rheum. 2003;48:3112–7.

- Olmarker K. Puncture of a lumbar intervertebral disc induces changes in spontaneous pain behavior: an experimental study in rats. Spine. 2008;33:850–5.
- 96. Peng B, Wu W, Hou S, Li P, Zhang C, Yang Y. The pathogenesis of discogenic low back pain. J Bone Joint Surg Br. 2005;87:62–7.
- 97. Stefanakis M, Al-Abbasi M, Harding I, Pollintine P, Dolan P, Tarlton J, Adams MA. Annulus fissures are mechanically and chemically conducive to the ingrowth of nerves and blood vessels. Spine (Phila Pa 1976). 2012;37:1883–91.
- Johnson WE, Caterson B, Eisenstein SM, Hynds DL, Snow DM, Roberts S. Human intervertebral disc aggrecan inhibits nerve growth in vitro. Arthritis Rheum. 2002;46:2658–64.
- Johnson WE, Caterson B, Eisenstein SM, Roberts S. Human intervertebral disc aggrecan inhibits endothelial cell adhesion and cell migration in vitro. Spine. 2005;30:1139–47.
- 100. Fields AJ, Liebenberg EC, Lotz JC. Innervation of pathologies in the lumbar vertebral end plate and intervertebral disc. Spine J. 2014;14:513–21.
- Liebscher T, Haefeli M, Wuertz K, Nerlich AG, Boos N. Age-related variation in cell density of human lumbar intervertebral disc. Spine. 2011;36: 153–9.
- 102. Bibby SR, Jones DA, Ripley RM, Urban JP. Metabolism of the intervertebral disc: effects of low levels of oxygen, glucose, and pH on rates of energy metabolism of bovine nucleus pulposus cells. Spine. 2005;30:487–96.
- 103. Skrzypiec D, Tarala M, Pollintine P, Dolan P, Adams MA. When are intervertebral discs stronger than their adjacent vertebrae? Spine. 2007;32:2455–61.
- 104. Hastreiter D, Ozuna RM, Spector M. Regional variations in certain cellular characteristics in human lumbar intervertebral discs, including the presence of alpha-smooth muscle actin. J Orthop Res. 2001;19:597–604.
- 105. Rumian AP, Draper ER, Wallace AL, Goodship AE. The influence of the mechanical environment on remodelling of the patellar tendon. J Bone Joint Surg Br. 2009;91:557–64.
- 106. Goodship AE, Lanyon LE, McFie H. Functional adaptation of bone to increased stress. An experimental study. J Bone Joint Surg Am. 1979;61: 539–46.
- 107. Yoganandan N, Ray G, Pintar FA, Myklebust JB, Sances Jr A. Stiffness and strain energy criteria to evaluate the threshold of injury to an intervertebral joint [see comments]. J Biomech. 1989;22:135–42.
- 108. Screen HR, Lee DA, Bader DL, Shelton JC. An investigation into the effects of the hierarchical structure of tendon fascicles on micromechanical properties. Proc Inst Mech Eng H. 2004;218: 109–19.
- Kerin AJ, Wisnom MR, Adams MA. The compressive strength of articular cartilage. Proc Inst Mech Eng H. 1998;212:273–80.

- Melton 3rd LJ. Epidemiology of spinal osteoporosis. Spine. 1997;22:2S–11.
- 111. Green TP, Adams MA, Dolan P. Tensile properties of the annulus fibrosus II. Ultimate tensile strength and fatigue life. Eur Spine J. 1993;2: 209–14.
- 112. Battie MC, Videman T, Levalahti E, Gill K, Kaprio J. Genetic and environmental effects on disc degeneration by phenotype and spinal level: a multivariate twin study. Spine. 2008;33:2801–8.
- 113. MacGregor AJ, Andrew T, Sambrook PN, Spector TD. Structural, psychological, and genetic influences

on low back and neck pain: a study of adult female twins. Arthritis Rheum. 2004;51:160–7.

- 114. Miller JA, Schmatz C, Schultz AB. Lumbar disc degeneration: correlation with age, sex, and spine level in 600 autopsy specimens. Spine. 1988;13: 173–8.
- 115. Adams MA. Mechanical influences in disc degeneration and prolapse: medico-legal relevance. Bone Joint J. 2014;3(2):32–5.
- Marchand F, Ahmed AM. Investigation of the laminate structure of lumbar disc anulus fibrosus. Spine. 1990;15:402–10.

# Advances in Lumbar Degenerative Disk Disease Pathophysiology Comprehension

4

# Andreas G. Nerlich and Norbert Boos

# 4.1 Chronic Back Pain and Disk Degeneration: Physiology and Pathophysiology of the Ageing Spine

Nowadays, low back pain faces two important issues:

- 1. Chronic low back pain is still one of the most frequent diseases in Western industrialised countries. Estimates count approx. 70 % of all adults in those countries to suffer from chronic or recurrent back pain during their lives, and figures are still increasing. Accordingly, this disease is the second most frequent cause for short or long-term absence from work.
- Despite its extensive clinical significance and increasing scientific research efforts, aetiology and pathogenesis of this disease are still a matter of great debate and differing concepts.

A major obstacle in understanding disk degeneration is the problem of defining "degeneration" from (normal) "ageing". Currently, it is

A.G. Nerlich, MD, PhD, MSc ()

Institute of Pathology, Academic Clinic Munich-Bogenhausen, Klinikum München-Bogenhausen, Englschalkingerstr. 77, Munich D-81925, Germany e-mail: Andreas.Nerlich@klinikum-muenchen.de

N. Boos, MD, MBA Prodorso Centre for Spinal Medicine, Zurich, Switzerland the concept to identify disk degeneration as a clinically symptomatic process that frequently, but not exclusively, is linked to premature or pathologic ageing. The concept allows us to combine the most recent findings that initial degenerative alterations even occur in early infantile disks [1, 2] and the observation that various disk levels within one individual are morphologically "degenerated" to a similar level, while there exists significant interindividual variation in occurrence and extent of degeneration [3]. Taking these preliminary issues together, both age and individual (extrinsic and intrinsic) factors seem to affect disk degeneration.

The human spine - and in particular the disk is subjected to an age-related disarrangement which proceeds with increasing age leading to "degeneration". The extent of degenerated disk disease (DDD) depends on age and affects various anatomic structures differently [1, 2]. Therefore, substantial individual differences can be observed in the sense that young individuals exhibit the disk of an elderly person and vice versa. Because of the extensive destructive changes that ultimately lead to an osseous transformed (ankylosed) motion segment, many clinicians and researchers believe that the intervertebral disk is a predominant source of low back pain. From a clinical point of view, differentiating "normal" age-related (i.e. asymptomatic) from "pathologic" degenerative (i.e. painful) changes would be sensible. However,

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016 J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_4

this task is extremely difficult due to the lacking reference standard for painful disk degeneration. So far, the best reference standard is provocative diskography. However, the role and potential benefit of this diagnostic procedure are controversially discussed in the literature.

While it is becoming more and more clear that the degenerated disk suffers from a progressive disease, the cartilaginous end plates that frame the nucleus pulposus and major parts of the annulus fibrosus come into focus for the pathophysiology of DDD. In consequence, the changes in the structure and the biochemical composition of the end plates – in combination with alterations of the anterior/posterior annulus fibrosus (AF) and the nucleus pulposus (NP) – are of utmost significance for the development and understanding of DDD. In order to understand the potential pathomechanism, we will be evaluating both the "normal" age-related changes in the disk with particular reference to the end plate and also the "premature" pathological changes that may end up in degenerated disk structures.

# 4.2 Structure and Function of the Normal Disk

The intervertebral disk is composed of three major components which are intimately linked to each others, but also to several adjacent structures, underlining the complexity of the disk (Fig. 4.1). In order to understand the function – and to evaluate any pathological condition – a concise analysis of the normal disk structure is necessary. Only recently, several extensive studies have been performed on the morphological changes in autopsy and surgical material which define the following disk components (see also Fig. 4.1):



**Fig. 4.1** From foetal to old-age disk – macroscopy of the intervertebral disk. (a) Foetal lumbar spine (36th week of gestation) vs. (b) old-aged spine (86 years) showing the tremendous changes of the disk structure during age

- The nucleus pulposus (NP) is a gelatinous mass filling a central "core" of the disk between the two end plates and the ring-like fibres of the annulus fibrosus (AF). It is made up of a particular type of cartilage which contains relatively few chondrocytic cells and an abundant matrix mainly of proteoglycans (aggrecan) along with moderate amounts of collagen molecules (mainly collagen type II). The high concentration of proteoglycans leads to significant binding of water and swelling of the highly hydrated NP the extension of which is limited by the adjacent structures. Therefore, the NP has the main function of a "shock absorber", a cushion compression, providing both stiffness and flexibility of the motion segment.
- The annulus fibrosus (AF) is comprised of dense sheets of highly oriented collagen in concentric lamellae running fibres obliquely between adjacent vertebral bodies. The angle relative to the vertebral bodies alternates from one lamella to the next resulting in a cross-woven network. The AF can be subdivided into an inner and outer part. The inner AF represents a structure close to that of the NP, while the outer AF is composed of a densely organised structure with thick bundles of collagen as the main constituent. The fibre network allows some relative movement between the bundles, thereby providing a certain degree of disk deformation during flexion and extension of the vertebral column. The main structural component of the AF is collagen (mainly made of collagen types I and III, with some type II collagen in the inner AF, but no collagen II in the outer AF). The content of proteoglycans is significantly lower than in the NP and does not serve mainly as a water adsorbent, but these proteoglycans are important for a proper interaction of the collagen lamellae and therefore for a regular function of the AF fibre network.
- The normal disk is framed by two end plates (EP) forming each a sheet of hyaline cartilage adjacent to the upper and lower vertebral body which is very similar to articular hyaline

cartilage. This layer of cartilage separates both the NP and the AF from the vertebral bones providing a firm attachment of AF fibres to the bone structures. Therefore, the end plate is essential for the rigidity of the motion segment. In addition, the end plate is of significance for the nutrition of discal structures, as the major part of the disk is nourished via diffusion of oxygen and nutrients from the bone marrow spaces of the vertebral bones (in "normal" adult disks, there exists only minor vascularisation of the outer AF from small capillary vessels). Structurally, the EP is composed of a fine collagenous network mainly made up by collagen type II fibrils along with proteoglycans. The orientation of the collagen fibres as horizontal layers provides significant rigidity of the motion segment. At birth, the human cartilage end plates make up approximately 50 % of the intervertebral space (compared with approximately 5 % in the adult) and have large vascular channels running through them. Soon after birth, the vascular channels of the cartilage end plate fill in with extracellular matrix such that no channels remain by the end of the first life decade.

Besides the NP, the AF and the EP, the disk is surrounded by various other structural elements that contribute to the function of the motion segment. This holds particularly true for the anterior and posterior longitudinal ligaments which provide further stiffness, but also some flexibility of the spine and which delineate important adjacent structures such as the spinal canal. The ligament zone is important not only with respect to the vascular supply but also with regard to a significant innervation which may be relevant during the induction of disk pain.

The vertebral bodies are further functional elements of the spine which intimately interact with the disks. These osseous bodies significantly stabilise the vertebral column, but they are also important with respect to the nutrition of the various disk structures since the bone marrow cavity is filled with blood with low perfusion velocity. Under physiological circumstances, these various structures interact properly with each others. There exist, however, significant influences of structure and thus function with different stages of age. Therefore, age-related changes have to be taken into account when differentiated from pathological alterations.

# 4.3 The Morphology of the Disk During Development and Ageing

The disk and its various substructures undergo significant macro- and micromorphological changes with advancing age which are important for the evaluation of DDD. Again, the disk has to be considered as a whole, but alterations of the end plates seem to be particularly important.

# 4.3.1 Embryonal and Foetal Development

The vertebral column is already determined during very early embryogenesis (Fig. 4.2). At about 4 weeks gestational age, the human spine can be identified as a typical series of vertebral bodies. These form under the combined influence of the notochord and neuronal tube. The disk grows initially in an environment with only few blood vessels and is surrounded by a perichondrial layer which forms the future longitudinal ligaments. Between the vertebrae, the notochord expands as local aggregations of cells, the notochordal cells, within a proteoglycan-rich matrix, forming the gelatinous centre which results in the NP (Fig. 4.2). The circularly arranged fibres surrounding the NP finally form the AF which is derived from the perichordal mesenchyme. With ongoing development, the notochordal cells are



**Fig. 4.2** Anatomy of the intervertebral disk. (a) Schematic presentation of the motion segment and the disk. (b) Macroscopically the central nucleus pulposus (NP) is surrounded by the annulus fibrosus (AF). (c) Histologically the very different fibre structure of the various structures is evident

replaced by chondrocytic cells, while the cells of the AF region have a more fibroblastic phenotype. As yet, the exact role and interaction of notochordal, chondrocytic and fibroblastic cells within the developing disk remains unclear. Recent studies suggest, however, that there exists a significant communication between those cells leading to the final disk structure [4]. This synergy may be of importance in maintaining a normal, "non-degenerated" disk. Interestingly, the notochordal cells disappear from the human disk during early infancy. This is in contrast to various species, such as cat, pig, mouse, rat, rabbit and others where notochordal cells persist in the nucleus pulposus into adulthood. As yet, it remains speculative whether those animals with persisting notochordal cells have a much lesser degree of disk degeneration or if other factors may be the cause therefore. In summary, the loss of notochordal cells may be seen as an important, initial step towards ageing of the disk.

A further significant structural peculiarity of foetal (and early infantile; see below) disks is the presence of blood vessels within the disk (Fig. 4.3). These are mainly formed by large vessel "loops" which originate from bone marrow vessels, perforate the EP and extend into the inner AF close to the NP. These vessels have thin walls such as seen in venules. Due to the dimension, it can be assumed that they provide a major blood supply to all disk structures. Only to a significantly fewer part, small vessels of the capillary type extend from the longitudinal ligaments into the AF, but not into more distant structures. Accordingly, they are restricted to the outer AF.

During the postnatal development, three major time periods can be separated with respect to morphological age-related changes. These are infancy and early adolescence (c. 0–17 years), young and medium adulthood (c. 18–60 years) and advanced age (more than 60 years). These time periods are characterised by significant structural changes affecting the various anatomic sub-settings differently but leading to a progressive loss of "normal" structure and thus function (Fig. 4.4). Any distinction between "physiological" ageing and "pathological" degeneration is highly problematic and should be oriented at the clinical situation, i.e. at the presence/absence of pain and/or disabling loss of function.

# 4.3.2 The Disk of Infants and Adolescents

Besides the loss of notochordal cells which gradually is seen between the prenatal period to an age of approx. 4 years, the most important structural change is a complete loss of disk vascularisation (see above, also Fig. 4.3). This is closely related to changes in the EP structure and is seen in infants of few months of age presenting with regressively obliterated vessel residues. In a series of autopsy cases of lumbar spines, we did not find intradiscal vessels originating from the bone marrow space as early as 4 years of age [1]. Coincidently with these two features, even the infantile and early adolescent lumbar disk provided focal, minor and very initial morphological changes which suggest initial "degeneration". Likewise, we observed foci of granular matrix alteration, beginning clonal chondrocyte proliferation and initial small disruptions of the matrix even in this age group, mostly restricted to the NP without alterations of the AF. In the EP mainly those areas with presumably obliterated blood vessels show a disturbance of the EP structure. The extent of all these changes correlated well with the increasing age [1].

#### 4.3.3 The Disks of Young and Medium-Aged Adults

With further proceeding age, the extent and degree of the aforementioned changes increases (Figs. 4.4 and 4.5). The most dramatic rise in morphological signs for tissue degeneration is seen at the end of puberty when the rapid growth process has led to a significant increase of diffusion distances within the disk. This impairment of the disk nutrition has previously been made responsible for most degeneration-associated morphological changes. However, more recent concepts – that will have to be discussed below



**Fig. 4.3** Embryonal vertebra (11th week of gestation). (a) The overview shows the typical segmentation; in the centre (b) islands of notochordal cells are visible (*arrow*)

more in detail – suggest that several other factors seem to play an essential deleterious role during DDD. As the most typical morphological signs for disk degeneration, we previously identified significant tears and clefts of the disk matrix, an increasing amount of clonally growing "proliferating" chondrocytes, granular and mucoid matrix changes and the occurrence of decaying cells

which can selectively be labelled by immunohistochemistry (*brown* reaction product, cytokeratin 8, c)

(Fig. 4.6) [1]. The changes are seen first in the NP, but frequently extend to the inner AF and to lesser extent to the outer AF. Then the outer AF provides focal clefts which – when in connection with similar tears and clefts of the inner AF and the NP – may lead to protrusions and even prolapse of the disk into the spinal canal. Those changes are then often associated with an



Fig. 4.4 Changes in disk vascularisation. The foetal end plate contains large blood vessels (a) which are obliterated during the first postnatal months (b). There, small

islands of irregular cartilage indicate the residues of the previous vasculature

"inflammatory" reaction as seen by the in-growth of capillary blood vessels and histiocytic cells into the affected AF areas.

#### 4.3.4 The Disks of Adults of Advanced Age

Finally, the age-related changes in the old-age group are characterised by even more pronounced morphological signs of disk degeneration. While part of those cases that have been investigated as yet have shown extensive clefts, a loss of disk height, chondrocytic proliferation and extensive mucoid matrix degeneration (Fig. 4.6b, c), we identified individuals with a transition into a scarlike morphology with loss of the chondroid matrix and replacement by a more or less "fibrous" tissue. Those disks frequently reveal a complete loss of the disk tissue structure and can be termed as a "burnt-out" appearance.

# 4.3.5 Particular Morphologic Changes of the End Plate

Since the end plates form a very unique structure, in the context of this chapter, a particular view on its anatomic and developmental structure might be presented.

Within the thin cartilage end plates, collagen fibres run horizontal and parallel to the vertebral bodies along with the fibres continuing into the disk. This separates both the NP and the AF from the vertebral bones and provides a firm attachment of AF fibres to the bone structures. Therefore, the end plate is essential for the rigidity of the motion segment. At birth, the human cartilage end plates make up approximately 50 % of the complete intervertebral space (compared with approximately 5 % in the adult). They have large vascular channels running through them. Soon after birth, the vascular channels of the cartilage end plate fill in with extracellular matrix



Fig. 4.5 Macroscopic aspects of the intervertebral disk with increasing age. This figure shows typical macro-scopic examples of age-related changes of the disk. The

alterations with increasing age are used in the classification by Thompson in five grades: I (a), II (b), III (c), IV (d) and V (e)

such that no channels remain by the end of the first life decade. This change gains even more significance as the disk grows strongly in size during puberty, and therefore, the distances between the vascular supply and the disk cells elongate substantially (see below).

The cartilage end plate in humans functions in early life as a growth plate for the adjacent vertebral body; its structure is typical of that seen in the epiphyseal growth plates of long bones. This structure is lost during skeletal maturity. By adulthood, the cartilage end plate is a layer of hyaline cartilage (approximately 0.6 mm thick) with calcified cartilage adjoining the bone. In adults, the end plate occupies the central 90 % of the interface between the disk



**Fig. 4.6** Histology of complete motion segments of intervertebral disks with various age. (a) The section through the complete disk of a young adult individual shows regular features of disk morphology. However, despite the young age, the disk reveals already some clefts within the nucleus pulposus. (b) The disk of a 62-year-old individual in contrast demonstrates a much more irregular

and the vertebral body, encompassed by a ring of bone that forms via the epiphysis fusing with the vertebral body in the rim region. The end plate is totally avascular and aneural in all healthy adults.

Within the cartilage end plates, typical lacunar cartilage cells are surrounded by a small territorial matrix (thereby forming the typical functional unit of cartilage called "chondron") which in turn is enveloped by the inter-territorial matrix. The latter is somewhat more "compact" than the territorial matrix which is reflected on the molecular level by differences in the collagen type distribution. The biomechanical properties of the end plate are defined by its extracellular matrix composition which includes collagen types II, III, V, VI, IX and X and major proteoglycans and

shape of the disk with extensive clefts and tears of NP and AF. Furthermore, the disk shows reduced height and a lack of anatomical distinction between AF and NP. (c) The disk of a 77-year-old individual with very extensive alterations (when compared to **b**) with highly significant reduction in disk height and loss of disk tissues (**a** Masson's trichrome; **b**, **c** alcian blue-PAS)

glycoproteins. The composition of those constituents alters by age [5].

Functionally, the end plate is involved in two important mechanical functions [6]:

- I. Preventing the nucleus pulposus from bulging into the vertebral bodies
- II. Partially absorbing the hydrostatic pressure dissipated by the nucleus pulposus under loading. Similar to the disk, the ability of the end plate to withstand mechanical forces depends on the structural integrity of the matrix. In addition, the end plate is of significance for the nutrition of discal structures, as the major part of the disk is nourished via diffusion of oxygen and nutrients from the bone marrow spaces of the vertebral bones.



**Fig. 4.7** Typical microscopic alterations of disk degeneration. This figure shows characteristic microscopic features of disk degeneration: (a) Extensive clefting, originating in the NP, extends with ongoing degeneration to the AF. (b) Focal, "cluster"-like cell proliferation. (c)

Small areas with typical mucoid matrix indicate substantial changes in the matrix composition; this can selectively be made visible in the alcian blue-PAS staining. (d) In late-stage degeneration tissue defects occur ( $\mathbf{a}$ ,  $\mathbf{c}$  alcian blue-PAS;  $\mathbf{b}$  H d Masson's trichrome)

Age-related changes of the EP have been identified to comprise [7] fissure formation, fractures of calcified cartilage, horizontal cleft formation, death of chondrocytes (apoptosis), increased vascular penetration, extension of calcification and ossification. These changes start to occur in the third decade of life. A study of cadaveric human vertebrae demonstrated that the number of vascular channels perforating the osseous vertebral end plate diminishes drastically between 6 and 30 months of age [8]. Analyses on the microscopic level revealed that the abundance of obliterated blood vessels in the end plate gradually increases between 1 month and 16 years of age. The decrease in blood vessels is paralleled by an increase of cartilage disorganisation, end plate cell density, cartilage cracks and microfractures (Fig. 4.7).

These changes, especially the loss of blood vessels, can cause nutritional consequences for the intervertebral disk (Fig. 4.8). With advanced degeneration and markedly reduced disk height, further changes of the end plate are induced resulting in complete end plate disarrangement and dense sclerosis of the adjacent vertebral bodies.


**Fig. 4.8** Biomechanical and nutritional conditions of the intervertebral disk. While the nucleus pulpous has high water-binding properties resulting in a cushion-like structure, the annulus fibrosus and the end plates frame the nucleus. However, the avascular disk structure requires long distance diffusion of oxygen and nutritional substances from the bone marrow to the central portions of the disk

# 4.4 Aetiology of Disk Degeneration and Obvious Factors Influencing Disk Degeneration

Although the aetiology of disk degeneration is far from being completely understood, there is consensus that not a single factor can be held responsible for the complex phenomenon of disk degeneration. Extensive studies suggest a complex cascade of events, factors, transmitters, etc. that seem to induce, promote and enhance disk degeneration. Both exogenous and endogenous factors, each contributing individually, might influence the progress of degenerative changes of the disks. In consequence, we can divide the process of DDD into an initiation, propagation, (abortive) healing and burn-out stage.

#### 4.4.1 Genetic Predisposition of DDD

To our knowledge, DDD has its origin very early in life. Likewise, the methodical analyses of infantile disks indicate that as early as 2 years of age, initial morphological evidence disk alteration is present [1] predominantly related to the blood supply of the intervertebral disk through the end plates. These alterations mimic the later typical features of (clinically relevant) degeneration. This suggests that three main causes may have to be associated with this process: genetic predisposition, biomechanical load and altered disk metabolism/nutrition.

The very first evidence for the influence of genetic predisposition came from twin studies that had demonstrated an overall heritability between 52 and 74 % for disk disease of the lumbar spine [9, 10]. These studies furthermore suggested that the influence of genetic factors is much higher than environmental, behavioural and anthropometric factors, such as gender, obesity, height, occupational activities and exogenic influences by food or smoking habits [9].

Association studies of genes encoding for structural and functional disk components have highlighted the participation of polymorphisms in DDD with a major focus on structural proteins, mainly collagen types I, IX and XI; proteoglycans such as aggrecan or the cartilage intermediate layer protein (CILP); and also on matrix-degrading enzymes and their inhibitors, such as MMP-3, MMP-9 and TIMP-2. Finally, a genetic association with inflammatory cytokines, such as interleukin-1 $\alpha$  and interleukin-1 $\beta$  (IL-1 $\alpha/\beta$ ), IL-6 and TNF- $\alpha$ , has been described. Cell receptors, such as the vitamin D-receptor (VDR) were also identified such as thrombos-pondin-2 (THBS2) [11].

Within the focus of the genetic investigation is the role of molecularly modified proteins, such as the aggrecan gene which was demonstrated with shorter variable numbers of tandem repeat length of the gene, leading to a shorter aggrecan core protein [12] which in turn might lead to a lower water content similar to what is seen in degenerated disks. Polymorphisms affecting collagens have mostly been found in collagen type IX genes. The mutations were located in COL9A2 and COL9A3, thus belonging to the so-called cartilage "minor" collagen chain genes. Meanwhile, genetic studies confirmed an association of DDD and other structural proteins, mainly collagen types I [13] and XI [14]. Among the non-collagenous matrix proteins, polymorphisms in the gene encoding

the cartilage intermediate layer protein (CILP) have been recently identified to enhance the susceptibility to lumbar disk disease [15]. All these studies suggest that minor genetic modifications of structural proteins, as in aggrecan, or altered production capacity of the disk cells to produce those proteins may strongly predispose to the development of DDD.

Besides the association between a genetic diversity of disk structural proteins and DDD, similar studies found a comparable association with matrix-degrading enzymes and their inhibitors. Likewise, the major collagen-degrading enzymes MMP-3 and MMP-9 and the metalloproteinase inhibitor TIMP-2 [16] were associated with DDD. Here again, structural changes and/or changes in the amount or activity of those enzymes may be involved in an altered matrix metabolism that may lead more rapidly to degenerative alteration. In particular, polymorphisms in the promoter responsible for the expression of the matrix-degrading enzyme matrix metalloproteinase-3 (MMP-3) were found to accelerate degenerative changes in the lumbar disks in the elderly [16]. As an example, the authors identified a mutation in the MMP-3 promoter that resulted in an increased expression of the matrixdegrading enzyme, which together with environmental conditions might lead to enhanced disk degeneration.

But not only polymorphisms in genes encoding for matrix proteins have been associated with disk degeneration. Recently, mutations in various genes intimately involved in DDD propagation have been identified. Likewise, genes encoding the pro-inflammatory cytokine interleukin-1ß (IL-1 $\beta$ ) have been associated with disk degeneration and low back pain [17]. The authors suggest that the mutation in the IL-1 gene cluster modifies the effect of occupation on disk bulges and joint occurrence of degenerative changes. Similarly, genetic variability of the IL-1 receptor, interleukin-1 $\alpha$  (IL-1 $\alpha$ ) and IL-6 and IL-10 have statistically been linked with the process of developing DDD [18]. These findings suggest that mutations in these gene clusters contribute to the pathogenesis of lumbar disk degeneration possibly via enhancement of inflammation within the disks of predisposed individuals.

#### 4.4.2 Biomechanical Load

For a long period of time, it was the main theory that higher than normal loads during any physical activities were the main contributors to DDD. These observations were mainly nourished from the observations that the lumbar spine is affected more often and to higher degree than any other spinal zone, a region of the vertebral column that has to carry more mechanical load that any other spine region. Meanwhile, more sophisticated studies confirmed an association between mechanical load and DDD [11] with the main speculation that overload-induced alterations mainly affect the end plates, since it has not clearly been shown that the disk itself is directly suffering from abnormal loads [19], but the adjacent system of vertebral bodies and the cartilage frame work of the end plates. Further support for a more "indirect" role of direct mechanical influences on developing DDD comes from the histological observations that indicate early signs of disk degeneration even in early infantile disks when mechanical load normally does not play any major role and the recent systemic analysis of disks of various disk heights (cervical, thoracal and lumbar) that showed major interindividual differences in the extent of morphological disk degeneration, but not within the individuals analysed.

To investigate the influence of mechanical stress on disk degeneration uncoupled from psycho-social factors and work perception and to be able to tightly control the load impacting the intervertebral disk, animal experiments have been carried out. By applying dynamic load forces to the intervertebral disks of various experimental animals, changes indicating the onset of degeneration were found when the disks were analysed macro-morphologically or histologically [20]. Recently it has also been shown that not only compressive forces induce changes in the intervertebral disk related to degenerative changes but also particularly the application of vibrational





forces revealed signs of degenerative changes. Although the majority of the animal studies might suggest that certain forms of mechanical loads suffice to induce disk degeneration, several studies in humans did not provide a strong causal link between occupational exposures and disk degeneration, suggesting a more complex aetiology of disk degeneration.

# 4.4.3 Metabolic and Nutritional Effects

Insufficient nutritional supply of the disk cells is thought to be a major problem contributing to disk degeneration. Since the intervertebral disk is the largest avascular tissue in the human body, its cells are facing the precarious situation of having to maintain a huge extracellular matrix with a "fragile" supply of nutrients that is easily disturbed.

Although some vascularisation of the disk exists during the foetal and early infantile stage of life (see above), there is consensus that the tissue becomes avascular during the juvenile age and that it remains normally avascular throughout further life. The blood vessels originally extending into the disk then terminate before reaching the cartilage end plate. Whereas the cells in the outer annulus fibrosus may be supplied with nutrients from blood vessels in the adjacent longitudinal ligaments, the supply of the nucleus pulposus cells is almost completely dependent on

the capillary network and the sinus of the vertebral bone marrow [21] and the concomitantly long diffusion distances into the disk. With the originally cartilaginous end plates becoming calcified when degeneration progresses, the supply with nutrients becomes even more restricted [22]. Not only the supply with nutrients like glucose and oxygen is restricted due to diffusion distances, also the removal of metabolic waste becomes critical [23]. In vitro experiments have shown that low oxygen concentrations and acidic pH significantly affect the synthetic activity of disk cells (Fig. 4.9). Especially proteoglycan synthesis rates were observed to be particularly sensitive to a decrease of the extracellular pH. The reduced proteoglycan synthesis rates might then lead to a fall in proteoglycan content and therefore to disk degeneration. This combination of restrictive nutrient supply and metabolite environment may lead to increased cell death and therefore reduced cell numbers in the disk.

Beyond the mere physical influence of e.g. diffusion distances, several other indirect factors may also play an important role. Recent studies evaluated the role of a systemic reduction of perfusion, such as by arteriosclerosis (particularly arteriolosclerosis in diabetic microangiopathy), chronic smoking habits and systemic vascular diseases, such as lipidaemia, hyperuricaemia and others, as risk factors to develop more rapidly DDD [24]. Likewise, it has been hypothesised that the low metabolic activity of disk cells hinders adaptation of the disk to abrupt increases in physical activity.

# 4.5 Presumed Molecular Mechanisms of Degeneration

As described in the chapter before, a whole series of endogenous and exogenous factors induce and promote DDD. These factors act via various molecular mechanisms that have meanwhile been identified. Induction of the degeneration process requires the translation of adverse impacts like nutritional restrictions to the disk cells and mechanical stress into molecular mechanisms. None of the adverse conditions seem to be sufficient to induce degeneration on its own, thereby leading to a concomitant induction of degeneration and low-back pain. In this regard the disk cells play a central regulatory role in the translation of the various impacts into molecular mechanisms. This holds, e.g. true for the expression of an array of matrix-degrading enzymes, changes of the disk cell phenotype or initiation of signal transduction cascades.

#### 4.5.1 Matrix-Degrading Enzymes

Disorganisation of the disk matrix is one of the prevalent features of disk degeneration. This is seen in very "early" stages of disk degeneration and is present very consistently until a burnt-out stage has been reached. Several studies indicate an enhanced presence/activity of major matrixdegrading enzymes [25]. The most important and best-characterised group of degrading enzymes are the matrix metalloproteinases (MMPs) which are responsible for degradation of the various types of collagens. According to their specificity, MMPs can be divided into four groups. Activation of MMPs during degenerative disk disease might be of ambiguous benefit. Whereas MMP activation in intact disks might contribute to the degeneration of matrix molecules and therefore be unfavourable, it might be beneficial in herniation by supporting the resorption of the prolapsed/ extruded tissue.

The expression of major MMPs is accordingly enhanced around clefts and tears in the matrix of degenerated disks providing a co-localisation



**Fig. 4.10** Immunohistochemical staining for MMP-1. The immunostainings for MMP-1 shows a highly specific and selective localisation of this matrix-degrading enzyme (*brown* reaction product). This indicates focal enzyme action resulting in matrix disruption

between enzymatic activity of MMPs and formation of clefts and tears. Evidence for an influence of infiltrating macrophages on MMP production by disk cells from herniated disk material was mostly provided by in vitro studies showing that co-incubation of herniated disk material with homologous peripheral blood mononuclear cells induced production of MMP-1 and MMP-3 by the disk cells. Additional strong evidence shows that the disk cells themselves also produce matrix-degrading MMPs during disk degeneration (Figs. 4.10 and 4.11) [26].

Collagens, however, are not the only matrix components that are degraded during degeneration. Aggrecan, a large proteoglycan that forms macromolecular aggregates with hyaluronic acid via link protein, is also degraded during degeneration. Recently, the number of cells expressing aggrecanase-1 (ADAMTS4), a member of the ADAMTS (*a* disintegrin and metalloproteinase with *t*hrombospondin motifs) gene family, has been found to increase with increasing degeneration [27]. This finding suggests that ADAMTS4 might play an important role in tissue degradation during disk degeneration.

#### 4.5.2 Pro-inflammatory Mediators

Disk cells do not only produce matrix-degrading enzymes but also have the ability to initiate or propagate signal transduction cascades to



**Fig. 4.11** Enzymatic degradation of disk tissue by MMPs. In a special detection technique (in situ zymography), the local activity of specific MMPs can be localised. While in degenerated disk tissue (**a**) a loss of staining

around disk cells (*arrows*) indicates enzymatic activity of MMPs, in non-degenerated disk tissue (**b**) cells remain unstained (*arrows*)

manipulate their own environment. Production and secretion of cytokines, growth factors and the respective receptors by disk cells has been investigated extensively. The knowledge gained so far mostly originates from studies on herniated disks (protruded, extruded or sequestrated disks), in vitro experiments or the use of animal models [18].

Studies using protruded but contained disk material show that disk cells do have the possibility to produce an array of pro-inflammatory factors. There, chondrocytes were found to produce significant levels of IL-1 $\alpha$ , IL-1 $\beta$ , IL-6, TNF- $\alpha$  and granulocyte-macrophage colonystimulating factor (GM-CSF). Consistently, strongly increased levels of prostaglandin E2, a key modulator of inflammation, were found in in vitro experiments. Since prostaglandin E2 is also involved in pain induction, the authors suggest that the inflammatory cytokines IL-1a and TNF- $\alpha$  produced by disk cells might contribute to pain induction [16]. This hypothesis has been further evidenced by detection of IL-1 $\alpha$ , TNF- $\beta$ and COX-2, an enzyme essential for the synthesis of prostaglandin E2, in vivo and in vitro. In addition a positive feedback loop of IL-1 $\alpha$  has been identified in vitro, upregulating its own production and also the production of IL-6 and COX-2. Recently, it has been demonstrated that TNF- $\alpha$  is substantially more expressed in disk material from symptomatic patients compared to samples taken at autopsy. Accordingly, one can assume that disk cells do have the potential to produce the inflammatory cytokines necessary to mediate and propagate an inflammatory reaction. In addition, the expression of cytokine receptors suggests that these cells are able to not only initiate signal transduction cascades but also endue the requirements to react to proinflammatory mediators. Several studies provide strong evidence that the chondrocyte-like cells of the nucleus pulposus are the origin of the observed mediators in contained, degenerated disks.

#### 4.5.3 Growth Factors

Growth factors are usually low-molecular-weight proteins that have the potential to increase mitogenesis, cyto-differentiation and also matrix synthesis. This combination of proliferative and biosynthetic effects on target cells and tissues promoted the investigation of growth factors and their effects on disk tissue. The first "research targets" were the transforming growth factor- $\beta$ (TGF- $\beta$ ) and epidermal growth factor (EGF) which revealed a strongly significant increase of collagen and proteoglycan synthesis in disk cells in vitro and in experimental models. Additionally, TGF- $\beta$  has not only positively affected the extracellular matrix by increasing proteoglycan synthesis but also reduced tissue resorption by decreasing the levels of MMP-2. Human disk cells do not only respond to exogenous TGF- $\beta$  but are also able to produce this growth factor. TGF- $\beta$  was generally found in tissue specimens from human herniated disks and was mainly associated with disk cells.

In a series of animal experiments using socalled senescence accelerated mice (SAM), it has been shown that the disks of young mice contain all members of the TGF- $\beta$  family and the respective receptors type I and type II were expressed. During ageing, the expression of TGF- $\beta$ s and receptors was found to decrease consistently in all mice used in this study, indicating a role of TGF- $\beta$ s in this type of rapid disk degeneration [28].

As a further target, insulin-like growth factor-1 (IGF-1) has also been studied in detail. Originally known for its ability to stimulate growth of bone and cartilage, IGF-1 has also been described to increase proteoglycan synthesis in the nucleus pulposus along with a reduction of tissue resorption in disks by lowering the level of active MMP-2 in nucleus pulposus cells. Previous work has suggested that apoptosis substantially contributes to this loss of viable cells. The ability of IGF-1 to significantly reduce the number of apoptotic cells in vitro after serum withdrawal from cultured human intervertebral disk cells might hint the importance of IGF-1. Expression of IGF-1 was found in disk cells from human, cattle, canines and rats and the positive effect on proteoglycan synthesis seen in human disks was verified in vitro using cultured bovine and rat disk cells.

# 4.5.4 The NF-κB Pathway and Disk Degeneration

Recent molecular investigations identified the so-called NF- $\kappa$ B pathway as one of the central control mechanisms that may also be involved in disk degeneration [29]. This family

of transcription factors plays a central role in the mediation of cellular responses to damage, stress and inflammation. Enhanced or chronic activation leads to degenerative diseases of various organs and organ systems and recent data include DDD into this list of diseases.

First evidence for the activation for this transcription factor pathway came from the aforementioned observations of major interleukins and matrix metalloproteinases which are mainly regulated through this pathway. Very recent in vitro experiments provide substantial support to this idea since it has been shown that the application of NF-kB pathway inhibitors blocked major cellular response of nucleus pulposus cells to stress factors in terms of inhibition of cytokine and MMP production and activation. These observations - albeit still in an experimental format - offer new opportunities for a therapeutical intervention on one of the obviously most important regulatory pathways during early disk degeneration. It is important to note that natural regulators of the NF-KB pathway, such as by the natural substance curcumin [30], may provide novel options in the prevention of (preclinical) early stage disk disease [31].

# 4.6 Pathophysiologic Role of the End Plate in DDD

As outlined before, DDD results from various alterations of the complex structures of the intervertebral disk. Within the current concepts of disk pathophysiology, the end plate plays a pivotal role that is, however, mainly restricted to the biomechanical and nutritional/metabolic part of disk degeneration and that does not seem to play a major (direct) role in pain induction/ transmission:

- It seems to control the nutritional influx of metabolic substances, as well as the efflux of waste products from the disk cells.
- The structural and molecular composition of the end plate is, therefore, crucial for the

preservation of the phenotype of disk cells (mainly those of the NP).

• The biomechanical integrity of the disk is based on the structure of the end plates.

Taking the aforementioned pathophysiologic features of DDD into account, the end plates seem to be the main player in the initiation and promotion of disk degeneration, but they do not seem to be involved in the transmission of diskogenic signalling, such as pain induction. The latter may be promoted by the clefting of the NP and AP and the subsequently much more rapid translocation of pain-inducing (pro-inflammatory) cytokines to the peridiscal space with its neuroceptive structures.

In general, the currently most favoured hypothesis for DDD comprises:

- Genetic predisposition renders individuals more or less susceptible to develop disk degeneration.
- Changes in the biomechanical and or nutritional/metabolic conditions of the disk, e.g. by overload, associated with an enhanced "weakness" of the disk tissue by genetic conditions and metabolic "stress" (including alterations of nutrition and metabolism by arteriosclerosis, chronic nicotine and other drug application, etc.).
- Alteration in the composition and structure of the end plate leading to disturbed supply of nutrients and oxygen and accumulation of waste products within the disk
- Change in the disk cell metabolism with the upregulation of pro-inflammatory cytokines and growth factors.
- Altered composition of the pericellular/interstitial matrix of the disks (change in collagen type and proteoglycan production, enhanced synthesis of matrix-degrading enzymes such as MMPs).
- Disruption of the disk structure and cleft and tear formation of NP and subsequently of the AF.
- Liberation of pro-inflammatory messenger substances (such as Tumor-necrosis-factor α,

etc.) from the altered disk cells and rapid transport via the tissue clefts to the peridiscal space.

• Induction of pain sensation in the peridiscal space (i.e. diskogenic low back pain).

Besides this hypothesis, it must be clear that several other mechanisms and pathways may also play a role in the process of disk ageing and degeneration, such as facet joint arthrosis, production of oxygen radicals and NO, etc. The exact role and contribution of those factors to the degeneration process is still being unclear.

# 4.7 Conclusions and Clinical Implications

The past decades have provided us with considerable insights into the mechanisms of DDD. The identification of major pathways that are disturbed and the elucidation of factors such as distinct cytokines and their pathways not only provided us with an understanding of DDD but also may offer us novel therapeutic approaches in the future. The application of specific drugs/substances that interfere with cytokine pathways and their regulation may offer new options to block pain induction mechanisms. The identification of phenotypic changes of the disk cells helps us to design strategies for gene therapy of disk cell cure. The investigation of matrix changes let us understand which environment is needed to successfully implant cells into the disk during disk repair.

Finally, all these studies make clear that the prevention of disk degeneration is one of the most desired (and obviously most effective) therapeutic means. We have also learned that this prevention must start early before secondary changes of the facet joints, ligaments and muscles occur. Early intervention must include not only the control of intrinsic metabolic factors (e.g. diabetes) but also the avoidance of deleterious exogenic influences (e.g. smoking). From a clinical perspective, however, there is still a long way to bring the intriguing research results of the past two decades from the bench to the bedside.

#### References

- Boos N, Weissbach S, Rohrbach H, Weiler C, Spratt KF, Nerlich AG. 2002 Volvo award in basic science: classification of age-related changes in lumbar intervertebral discs. Spine. 2002;27(23):2631–44.
- Weiler C. In situ analysis of pathomechanisms of human intervertebral disc degeneration. Pathologe. 2013;34 Suppl 2:251–9.
- Weiler C, Schietzsch M, Kirchner T, Nerlich AG, Boos N, Wuertz K. Age-related changes inhuman cervical, thoracal and lumbar intervertebral disc exhibit a strong intra-individual correlation. Eur Spine J. 2012;21 Suppl 6:S810–8.
- Vernon-Roberts B. Disc pathology and disease states. In: Gosh P, editor. The biology of the intervertebral disc. Boca Raton: CRC Press; 1988. p. 73–119.
- Nerlich AG, Schleicher ED, Boos N. 1997 Volvo award winner in basic science studies. Immunohistologic markers for age-related changes of human lumbar intervertebral discs. Spine. 1997;22:2781–95.
- Broberg KB. On the mechanical behaviour of intervertebral discs. Spine. 1983;8:151–65.
- Roberts S, Menage J, Duance V, Wotton S, Ayad S. 1991 Volvo award in basic sciences. Collagen types around the cells of the intervertebral disc and cartilage end plate: an immunolocalization study. Spine. 1991;16:1030–8.
- Edelson JG, Nathan H. Stages in the natural history of the vertebral end-plates. Spine. 1988;13:21–6.
- Ala-Kokko L. Genetic risk factors for lumbar disc disease. Ann Med. 2002;34:42–7.
- Battie MC, Videman T, Gibbons LE, Kaprio J, Gibbons LE, Gill K, Manninen H, Saarela J, Peltonen L. The twin spine study: contributions to a changing view of disc degeneration. Spine J. 2009;9:47–59.
- Chan SC, Ferguson SJ, Gantenbein-Ritter B. The effects of dynamic loading on the intervertebral disc. Eur Spine J. 2011;20:1796–812.
- Kawaguchi Y, Osada R, Kanamori M, Ishihara H, Ohmori K, Matsui H, Kimura T. Association between an aggrecan gene polymorphism and lumbar disc degeneration. Spine. 1999;24:2456–60.
- Pluijm SMF, Essen V, Bravenboer N. Collagen type I α1 Sp1 polymorphism, osteoporosis, and intervertebral disc degeneration in older men and women. Ann Rheum Dis. 2004;63:71–7.
- Mio F, Chiba Y, Hirose Y. A functional polymorphism in Col11A1 which encodes the α1 chain of type XI collagen is associated with susceptibility to lumbar disc herniation. Am J Hum Genet. 2007;81:1271–7.
- Seki S, Kawaguchi K, Chiba K. A functional SNP in CILP encoding cartilage intermediate layer protein is associated with susceptibility to lumbar disc disease. Nat Genet. 2005;37:607–12.

- 16. Takahashi M, Haro H, Wakabayashi Y, Kawauchi T, Komori H, Shinomiya K. The association of degeneration of the intervertebral disc with 5a/6a polymorphism in the promoter of the human matrix metalloproteinase-3 gene. J Bone Joint Surg Br. 2001;83:491–5.
- Solovieva S, Kouhia S, Leino-Arjas P, Ala-Kokko L, Luoma K, Raininko R, Saarela J, Riihimaki H. Interleukin 1 polymorphisms and intervertebral disc degeneration. Epidemiology. 2004;15:626–33.
- Wuertz K, Haglund L. Inflammatory mediators in intervertebral disk degeneration and discogenic pain. Glob Spine J. 2013;3:175–84.
- 19. Sandover J. Dynamic loading as a possible source of low-back disorders. Spine. 1983;8:652–8.
- Ching CT, Chow DH, Yao FY, Holmes AD. The effect of cyclic compression on the mechanical properties of the inter-vertebral disc: an in vivo study in a rat tail model. Clin Biomech (Bristol, Avon). 2003;18: 182–9.
- Roberts S, Urban JPG, Evans H, Eisenstein SM. Transport properties of the human cartilage endplate in relation to its composition and calcification. Spine. 1996;21(4):415–20.
- 22. Horner HA, Phil M, Urban JPG. 2001 Volvo award winner in basic science: effect of nutrient supply on the viability of cells from the nucleus pulposus of the intervertebral disc. Spine. 2001;26(23):2543–9.
- Holm S, Maroudas A, Urban JP, Selstam G, Nachemson A. Nutrition of the intervertebral disc: solute transport and metabolism. Connect Tissue Res. 1981;8:101–19.
- Adams MA, Freeman BJC, Morrison HP, Nelson IW, Dolan P. Mechanical initiation of intervertebral disc degeneration. Spine. 2000;25(13):1625–36.
- Bachmeier BE, Bachmeier BE, Nerlich A, Mittermaier N, Weiler C, Lumenta C, Wuertz K, Boos N. Matrix metalloproteinase expression levels suggest distinct enzyme roles during lumbar disc herniation and degeneration. Eur Spine J. 2009;1811: 1573–86.
- Weiler C, Nerlich AG, Zipperer J, Bachmeier BE, Boos N. 2002 SSE award in basic science: expression of major matrix metalloproteinases is associated with intervertebral disc degradation and resorption. Eur Spine J. 2002;11(4):308–20.
- Le Maitre CL, Freemont AJ, Hoyland JA. Localization of degradative enzymes and their inhibitors in the degenerate human intervertebral disc. J Pathol. 2004;204:47–54.
- Matsunaga S, Nagano S, Onishi T, Morimoto N, Suzuki S, Komiya S. Age-related changes in expression of transforming growth factor-beta and receptors in cells of intervertebral discs. J Neurosurg. 2003;98: 63–7.

- Zhongyi S, Sai Z, Chao L, Jiwei T. Effects of NF-kB signalling pathway in human intervertebral disc degeneration. Spine. 2015;40:224–232.
- Bachmeier BE, Killian P, Pfeffer U, Nerlich AG. Novel aspects for the application of Curcumin in chemoprevention of various cancers. Front Biosci (Schol Ed). 2010;2:697–717.
- Klawitter M, Quero L, Klasen J. Curcuma DMSO extracts and curcumin exhibit an anti-inflammatory and anti-catabolic effect on human intervertebral disc cells, possibly influencing TLR2 expression and JNK activity. J Inflamm (London). 2012;9:29.

# Epidemiology of Lumbar Degenerative Disk Disease

5

# Pierre Kehrli<sup>\*</sup>, Philippe Esposito, and João Luiz Pinheiro-Franco

It is of the highest interest for surgeons to study the epidemiology of symptomatic lumbar degenerative disk disease: A basic question to pose is "What is DDD?"

In the recent literature, DDD covers a wide spectrum of symptoms and radiological findings, from low back pain to acquired spinal canal stenosis and symptomatic herniated disk-related symptoms. Symptoms at first glance appear to result from the cumulative exposure of the spinal elements to excessive loading, sports, and principally work. However, this simple theory is

Department of Neurosurgery, Strasbourg University Hospital, Strasbourg, France

P. Esposito, MD Spine Surgery Center, Clinique du Diaconat, Strasbourg, France e-mail: p.espo@wanadoo.fr

J.L. Pinheiro-Franco, MD (⊠) Neurological and Spine Surgery, Samaritano Hospital, São Paulo, Brazil e-mail: jlpfranco@uol.com.br not consistent with our contemporary understanding of developmental biology and genetics. A fundamental question arises too: Are we studying radiological aspects or real illness?

# 5.1 Disk Degeneration: A Spectrum of Definitions

It is not always clear if the term "disk degeneration" is routinely employed for normal or pathological states, reflecting the most difficult challenge in treatment: In a given case, are the symptoms related to what is seen on even the most sophisticated radiological images?

For example, the definition taken from Battié and Videman's paper [1] includes the paradox of simultaneous occurrences of normal and abnormal states: "disk degeneration is a product of lifelong degradation of the intervertebral disk with synchronized remodeling of the disk and neighboring vertebrae, including simultaneous adaptation of the discal structures to changes in physical loading and responses to occasional injury."

In this definition, the word "injury" is referring to trauma, reflecting an acute event, but may also include the concept of cumulative stresses from the environment, i.e., way of life, history of accidents, heavy work, etc.

This cumulative load may reflect chronic injury to the spine, a sort of premature aging.

Author was deceased at the time of publication

Dr. Pierre Kehrli, Chairman of Neurosurgery of the Strasbourg University, France, tragically died before completing this chapter. He died suddenly, in May 2014, at the age of 50, from an aortic aneurysm. His legacy to the privileged people who worked with him included his extreme technical ability as a neurosurgeon and his gentle character. The coauthors pay a small tribute to this fantastic human being with this chapter to the book.

P. Kehrli, MD, PhD

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_5

The term "disk degeneration" is an ambiguous term which covers the spectrum of radiographic images not always correlated to pain. This findings on images such as plain X-ray films, CT, and MRI include disk height narrowing, osteophytes, bulging disk, disk herniation, low-signal appearance on T2-weighted MRI, end plate changes, fatty marrow degeneration, posterior facet changes, etc. [2].

# 5.2 Traditional Epidemiological Studies

### 5.2.1 Age and Sex

"Normal aging" presupposes proportional "normal disk alterations or changes" over time; however, the findings of advanced early disk degeneration are well known in certain individuals [3].

Similar disk degenerative changes occur in women as in men, but with a delay usually of 10 years [4, 5]. Females may report low back pain associated with hormonal changes related to changes in systemic exposure to progesterone treatment common in pregnancy. A confusing clinical scenario is a woman with an ovarian cyst in mid-menstrual cycle with concomitant lumbar DDD and symptomatic low back pain. In this situation, the spinal surgeon has to be very careful in proposing surgery, even if radiological findings demonstrate significant DDD.

It is quite common in the elderly to see impressive DDD on radiological films in the absence of symptoms.

### 5.2.2 Postural Factors

Postural factor has often been cited as a cause for low back pain. This has not been fully supported by the peer review literature. For example, there is no relationship between retrolisthesis in patients with L5-S1 disk degeneration and baseline pain or function [6].

A comparison of lumbar degenerative changes between primates and humans demonstrated similar aging DDD patterns species nonspecific. This suggests "that the bipedal posture may not be the singular, or even the most important, biomechanical factor in the development of human DDD" [7].

#### 5.2.3 Weight

In symptomatic patients with sciatica, excessive weight has often been cited as a causative factor for sciatica [8]. Studies indicated that overweight and obesity could increase the risk of chronic or severe low back pain [9, 10]. However, a definitive association between sciatica and excessive weight is just suggested, not widely proven in the literature.

#### 5.2.4 Loading Efforts

Lifting heavy loads, torsional stress, and motor vehicle driving are among the best-identified environmental risk factors associated with symptomatic DDD. A definitive association between loads carried and symptomatic lumbar degenerative disease has also not been proven. Routine heavy physical loading demands at work or at leisure explain only a minor portion of the overall variance in lumbar disk degeneration in Battié's study of twins [11, 12].

#### 5.2.5 Work/Sports

Excessive loading due to repetitive work activities was shown in one study to be second to inheritance as a causative factor in symptomatic disk degeneration [13]. A study by Ong et al. [14] in athletes found that elite athletes have a greater prevalence and degree of lumbar disk degeneration than the normal population. This may also vary with the type of sport.

# 5.2.6 Vibrations

There was no clear correlation in a study of 45 pairs of monozygotic twins, highly discordant for exposure to motorized vehicles and associated

whole-body vibration, between lumbar disk degeneration on MRI and extensive lifetime driving histories [15].

However, the study of Weber found that after a long duration to whole-body vibration exposure, an increased degree of spondylotic changes was noted at the thoracolumbar junction and mid-lumbar spine [16].

#### 5.2.7 Cigarette Smoking

Results from twins studies are inconclusive on the influence of cigarette use and symptomatic lumbar DDD. An association with cigarette use and heavy work style has been associated with vertebral inflammatory disease [17]. If this finding can be reproduced in other studies, it may have consequences in relation to both primary and secondary prevention measures for low back pain.

#### 5.2.8 Genetic Factors

The recent literature on DDD has been enriched with genetic studies which have given us new insight on the influence of inheritance and symptomatic DDD [18]. In 1991, Varlotta et al. [19] found that 32 % of adolescent patients with a symptomatic disk herniation had a family history.

As Ala-Kokko in 2002 concluded a study, "Even though several environmental and constitutional risk factors have been implicated in this disease, their effects are relatively minor. And recent family and twin studies have suggested that sciatica, disk herniation and disk degeneration may be explained to a large degree by genetic factors" [20].

In 1992, Matsui et al. [21] reported that there is familial clustering of symptomatic lumbar disk herniation among the young (18-year-old or younger). In 1997, Matsui et al. [22] surveyed 3,042 Japanese factory workers regarding a history of acute low back pain. The authors found that the average age of initial symptoms in workers whose parents also suffered from the same condition was significantly younger than that in workers with no familial history.

Sambrook et al. [23] conducted a classic twin study using the Australian and British twin registries to examine the hypothesis that disk degeneration has a major genetic component. Spine MRIs were obtained on 86 pairs of monozygotic twins and 154 pairs of dizygotic twins, 80 % of whom were female. A substantial genetic influence on disk degeneration was found. A summary score of disk degenerative changes were compiled which included disk height, signal intensity changes, bulging of the intervertebral disk, and anterior osteophyte formation. This revealed that heritability estimates were very high, 74 % (95 % confidence interval, 64-81 %) for the lumbar spine and 73 % (95 % confidence interval, 64-80 %) for the cervical spine, after adjusting for age, weight, smoking, occupation, and physical activity. An analysis on individual MRI findings suggested that disk bulging and height were influenced primarily through genetic influences. A genetic influence on intervertebral signal intensity was not apparent. A comprehensive chapter in this book will discuss in great detail genetics and lumbar DDD.

#### 5.2.9 Twin Studies

A series of studies on exposure-discordant monozygotic twins revealed only modest, if any, effects of various environmental exposures on the acceleration of disk degeneration [18].

Results from studies conducted on male monozygotic twin pairs demonstrated substantial familial aggregation in terms of the extent and location of disk degeneration: in 1995, Battie et al. [12] assessed lumbar MRIs of 115 pairs of male monozygotic twins in order to investigate the relative effects of environmental risk factors, age, and familial aggregation on disk degeneration. Disk bulging, height narrowing, and disk desiccation as indicated through signal intensity changes were used as phenotypic clinical markers. In a multivariate analysis of the thoracic 12lumbar 4 vertebral region, physical loading exposures explained 7 % of the variance in disk degeneration scores, with this percentage climbing to 16 % due to age and 77 % when familial aggregation was considered. In the lumbar 4–5 vertebrae and lumbar 5–sacral 1 vertebrae region, physical loading explained only 2 % of the variance in disk degeneration summary scores in multivariate analysis. The variance score in the lower lumbar region due to age was approximately 9 % which increased to 43 % when considering familial aggregation. This study provided the first estimate of the relative importance of specific environmental agents and overall familial influences including genetic factors on the presence of symptomatic DDD.

Chromosomal loci particular to the presence of DDD have been evaluated in several studies: The 21q locus was identified in a finish population (based on clinical and radiological findings) when evaluating for imaging presence of lumbar DDD [24]. Another study found that chromosome 19 (study based on MRI findings), near a locus associated with hand osteoarthritis [25], was associated with imaging evidence of lumbar DDD. In the latter twin series, a significant correlation on gene loci and imaging presence of DDD was noted only in the lumbar spine and not the cervical spine. In this study the majority of the twins did not have pain.

In terms of molecular biology, Zhang noted that collagen IX, a structural component of the nucleus pulposus and annulus fibrosus of the intervertebral disk, is considered to serve as a bridge between collagenous and non-collagenous proteins within the intervertebral disk [26]. Collagen IX is a heterotrimer of three alpha chains, 1(IX), 2(IX), and 3(IX), encoded by the genes *COL9A1*, *COL9A2*, and *COL9A3*, respectively. Tryptophan alleles of COL9A2 and COL9A3 have been shown to be associated with lumbar disk disease in the Finnish population [27].

Sox genes appear to encode transcription factors with diverse roles in differentiation and development. *Sox9* is expressed in mesenchymal condensations prior to and during chondrogenesis and has been shown to activate Col2a1, the gene encoding type II collagen, the major component of cartilage matrix. Gruber et al. considered that the loss of expression of *Sox9* in some of the annulus cell population may play a role in disk aging and degeneration, possibly through decreased

modulation of the expression and production of type II collagen within the disk cells [28].

Polymorphisms in the vitamin D receptor gene, the matrix metalloproteinase-3 gene (MMP-3), and the aggrecan gene (AGC1) VNTR have been reported to be associated with disk degeneration [2].

Inflammatory cytokines have a wellrecognized contribution to the generation of back pain. Interleukin-1, in particular, has been shown to contribute to disk degeneration by inducing enzymes that destroy proteoglycan.

#### 5.2.10 Psychological Factors

Anxiety, depression, and somatization have been suggested as risk factors for chronic low back pain in several prospective studies [29, 30].

#### Conclusion

Epidemiological studies have now shed light on the important influence of molecular biology and genetics on asymptomatic and symptomatic DDD, derailing the long-held beliefs of the strong association of environmental factors and axial spine pain. Multiple comorbid factors including lifestyle and physical, psychological, and social factors influence overall prevalence and outcomes.

#### References

- Battié MC, Videman T. Lumbar disc degeneration: epidemiology and genetics. J Bone Joint Surg Am. 2006;88:3–9.
- Videman T, Battié MC, Gill K, Manninen H, Gibbons LE, Fisher LD. Magnetic resonance imaging findings and their relationships in the thoracic and lumbar spine. Insights into the etiopathogenesis of spinal degeneration. Spine. 1995;20:928–35.
- Beneke R. Zur Lehre von der Spondylitis deformans. Versammlung Dtsch Naturforscher Ärzte. 1897;4:69.
- Heine J. Über die Arthritis deformans. Virchows Arch Pathol Anat. 1926;260:521–663.
- Miller JA, Schmatz C, Schultz AB. Lumbar disc degeneration: correlation with age, sex, and spine level in 600 autopsy specimens. Spine. 1988;13: 173–8.

- Shen M, Razi A, Lurie JD, Hanscom B, Weinstein J. Retrolisthesis and lumbar disc herniation: a preoperative assessment of patient function. Spine J. 2007;7(4):406–13.
- Kramer PA, Newell-Morris LL, Siinkin PA. Spinal degenerative disk disease (DDD) in female macaque monkeys: epidemiology and comparison with women. J Orthop Res. 2002;20:399–408.
- Shiri R, Karppinen J, Leino-Arjas P, Solovieva S, Varonen H, Kalso E, Ukkola O, Viikari-Juntura E. Cardiovascular and lifestyle risk factors in lumbar radicular pain or clinically defined sciatica: a systematic review. Eur Spine J. 2007;16:2043–54.
- Leino-Arjas P, Solovieva S, Kirjonen J, Reunanen A, Riihimaki H. Cardiovascular risk factors and lowback pain in a long-term follow-up of industrial employees. Scand J Work Environ Health. 2006;32: 12–9.
- Shiri R, Karppinen J, Leino-Arjas P, Solovieva S, Viikari-Juntura E. The association between obesity and low back pain: a meta-analysis. Am J Epidemiol. 2010;171(2):135–54.
- Battié MC, Haynor DR, Fisher LD, Gill K, Gibbons LE, Videman T. Similarities in degenerative findings on magnetic resonance images of the lumbar spines of identical twins. J Bone Joint Surg Am. 1995;77: 1662–70.
- Battié MC, Videman T, Gibbons LE, Fisher LD, Manninen H, Gill K. Determinants of lumbar disc degeneration. A study relating lifetime exposures and magnetic resonance imaging findings in identical twins. Spine. 1995;20:2601–12.
- Saftić R, Grgić M, Ebling B, Splavski B. Case-control study of risk factors for lumbar intervertebral disc herniation in Croatian Island populations. Croat Med J. 2006;47(4):593–600.
- 14. Ong A, Anderson J, Roche J. A pilot study of the prevalence of lumbar disc degeneration in elite athletes with lower back pain at the Sydney 2000 Olympic Games. Br J Sports Med. 2003;37(3): 263–6.
- Battié MC, Videman T, Gibbons LE, Manninen H, Gill K, Pope M, Kaprio J. Occupational driving and lumbar disc degeneration: a case-control study. Lancet. 2002;360:1369–74.
- Weber M. Is lumbar disk disease an occupational disease? Scientific background, radiological findings, and medical legal interpretations. Z Orthop Ihre Grenzgeb. 2002;140(5):512–7.
- Leboeuf-Yde C, Kjær P, Bendix T, Manniche C. Selfreported hard physical work combined with heavy smoking or overweight may result in so-called Modic

changes. BMC Musculoskelet Disord. 2008;9:5. Published online 2008 January 14.

- Battié MC, Videman T, Parent E. Lumbar disc degeneration: epidemiology and genetic influences. Spine. 2004;29:2679–90.
- Varlotta GP, Brown MD, Kelsey JL, Golden AL. Familial predisposition for herniation of a lumbar disc in patients who are less than twenty-one years old. J Bone Joint Surg Am. 1991;73:124–8.
- Ala-Kokko L. Genetic risk factors for lumbar disc disease. Ann Med. 2002;34:42–7.
- Matsui H, Terahata N, Tsuji H, Hirano N, Naruse Y. Familial predisposition and clustering for juvenile lumbar disc herniation. Spine. 1992;17:1323–8.
- Matsui H, Maeda A, Tsuji H, Naruse Y. Risk indicators of low back pain among workers in Japan. Association of familial and physical factors with low back pain. Spine. 1997;22:1242–7.
- Sambrook PN, MacGregor AJ, Spector TD. Genetic influences on cervical and lumbar disc degeneration: a magnetic resonance imaging study in twins. Arthritis Rheum. 1999;42:366–72.
- 24. Virtanen IM, Noponen N, Barral S, Karppinen J, Li H, Vuoristo M, et al. A putative susceptibility locus on chromosome 21q for lumbar disc disease (LDD) in the Finnish population. J Bone Miner Res. 2007;22:701–7.
- 25. Williams FMK, Kato BS, Livshits G, Sambrook PN, Spector TD, MacGregor AJ. Lumbar disc disease shows linkage to chromosome 19 overlapping with a QTL for hand OA. Ann Rheum Dis. 2008;67: 117–9.
- Zhang Y, Sun Z, Liu J, Guo X. Advances in susceptibility genetics of intervertebral degenerative disc disease. Int J Biol Sci. 2008;4:283–90.
- 27. Solovieva S, Lohiniva J, Leino-Arjas P, Raininko R, Luoma K, Ala-Kokko L, Riihimaki H. COL9A3 gene polymorphism and obesity in intervertebral disc degeneration of the lumbar spine: evidence of geneenvironment interaction. Spine. 2002;27:2691–6.
- Gruber HE, Norton HJ, Ingram JA, Hanley Jr EN. The SOX9 transcription factor in the human disc: decreased immunolocalization with age and disc degeneration. Spine. 2005;30:625–30.
- Currie SR, Wang J. More data on major depression as an antecedent risk factor for first onset of chronic back pain. Psychol Med. 2005;35:1275–82.
- 30. Shaw WS, Means-Christensen A, Slater MA, Patterson TL, Webster JS, Atkinson JH. Shared and independent associations of psychosocial factors on work status among men with subacute low back pain. Clin J Pain. 2007;23:409–16.

# Genetics of Lumbar Disk Degeneration

# 6

Kenneth M.C. Cheung, Yan Li, Pak Chung Sham, and Danny Chan

# 6.1 Introduction

The field of genetics is playing an increasingly important role in clinical medicine. As the fundamental "blueprint" of any organism, genes now are considered a central component of most common diseases. The aim of genetic research is to determine which genes and environmental factors contribute to traits of interest (called phenotypes) and, of most interest to clinicians, to identify mutations that contribute to diseases. The application of such knowledge might lead to early detection, better treatment, and, ultimately, prevention of disease. Low back pain (LBP) is one of the most common disorders for which patients seek medical consultation. The severity of the pain ranges from mild discomfort to severe, disabling pain. The etiology of LBP is

Y. Li, MSc • P.C. Sham, BM, BCh, MA, MSc, PhD, MRCPsych Department of Psychiatry, Centre for Genomics Sciences, The University of Hong Kong, Hong Kong, China e-mail: yanlea@hku.hk; pcsham@hku.hk

D. Chan, PhD Department of Biochemistry, The University of Hong Kong, Hong Kong, China complex and involves multiple factors, with disk degeneration of the lumbar spine being a major contributor [1]. This chapter aims to equip readers with a basic understanding of genetics and introduce the principles and findings of genetic studies in lumbar disk degeneration (LDD), following a general framework for genetic studies of all complex diseases (Fig. 6.1).

# 6.2 Basic Concept of Genetics

# 6.2.1 Structure and Function of Genes and Chromosomes

The gene is the basic unit of heredity of living organisms. It stores the instructions for diverse protein molecules that control development, survival, and reproduction. The whole human genome encodes 20,000-30,000 genes. Deoxyribonucleic acid (DNA) is the molecular component of a gene. The molecular unit of DNA is the nucleotide, which has three basic components: a pentose sugar, a phosphate group, and a nitrogenous base. There are four types of nitrogenous bases: cytosine, thymine, adenine, and guanine. They commonly are represented by their first letters: C, T, A, and G. Nucleotides attach to one another in a certain order to form a polynucleotide chain. Two complementary polynucleotide chains, G pairing with C and T pairing with G, held together by weak thermodynamic

K.M.C. Cheung, MD (🖂)

Department of Orthopaedics and Traumatology, Professorial Block, Queen Mary Hospital, The University of Hong Kong, Pok Fu Lam, Hong Kong, China e-mail: cheungmc@hku.hk



forces, form a DNA molecule. Different sequences of nucleotides represent different proteins or different regulatory functions. To encode all the information of the human body, each cell contains approximately three billion nucleotide pairs. To package all this DNA into a tiny cell nucleus, DNA coils around histones in an organized manner and loops into a helical line, thereby forming chromosomes (Fig. 6.2).

For diploid organisms, chromosomes exist in pairs. One member of each pair originates from the father and the other originates from the mother. Each human somatic cell contains 23 pairs of chromosomes, including 22 homologous pairs of autosomes and one pair of sex chromosomes. In normal males, the sex chromosomes are a Y chromosome inherited from the father and an X chromosome inherited from the mother. Two X chromosomes are found in normal females, each inherited from one parent. Two homologues have similar sequences; only a small fraction of positions have sequencing variations that can be used to distinguish the chromosomes. The variants on the same position of paired chromosomes are defined as alleles. The two alleles at a certain position may be the same (homozygous) or different (heterozygous).

The functions of genes may be classified into two general categories: protein-coding and nonprotein-coding genes. The coding sequences

of protein-coding genes (exons) are separated from each other by noncoding intervening sequences (introns) (Fig. 6.3). They encode proteins through two major steps: transcription and translation. First, the DNA is transcribed premessenger ribonucleic acid (pre-mRNA). Introns then are spliced out, and exons join to form mature messenger RNAs (mRNAs). Second, mRNAs are translated to protein. Every three nucleotides (codons) in mRNA represent an amino acid. Nonprotein-coding genes are transcribed into noncoding RNAs, which are not translated to protein, but form secondary structures to mediate gene expression or mRNA degradation (microRNAs).

#### 6.2.2 **Genetic Variations** and Genetic Markers

Any two human genomes differ at millions of different positions. There are small variations in the individual nucleotides of the genomes, as well as many larger variations, such as microsatellites, insertions, deletions, and copy number variations. Any of these may cause alterations of protein structure or gene expression profile, altering the risk of disease.

One important type of genetic variant is the single-nucleotide polymorphism (SNP). SNPs

diseases



are a 1 bp substitution of DNA sequences that may be found in any position of the genome. In the human genome, on average, an SNP occurs every 300 nucleotides. Approximately 90 % of the genetic variants in the genome are SNPs, which means there are roughly ten million SNPs in the human genome.

A second major type of variation is the microsatellite, also called a variable number of tandem repeats (VNTR). This term refers to a short nucleotide sequence that occurs as a repeating sequence. The number of repeats differs among different chromosomes in different people, enabling VNTR to be a marker for personal identification.

Another type of genetic variant is the insertion or deletion (indel) of one or more bases in DNA sequences. For protein-coding genes, these changes may result in extra or missing amino acids in a protein if the inserted or deleted sequence is 3 bp long or in a complete change of the protein sequence beyond the indel if the inserted or deleted sequence is not a multiple of 3 (the socalled frameshift). Changes in amino acid sequences may have profound consequences, and many serious genetic diseases are caused by such changes.

On a larger scale, recent studies also focused on many large structural variations in DNA sequences. For example, duplications of large chromosome segments containing one or more genes have been studied widely. The number of duplicates is defined as the *copy number*, and variation in the number of duplicates among people is termed *copy number variation* (CNV). Another type of large structure variation is an inversion, a segment of chromosomes reversed end to end. Inversions also have been associated with complex diseases.



Fig. 6.3 Gene structure, transcript, and translation

A genetic marker is a variable DNA sequence at a known location in the genome. All genetic variants may be used as genetic markers once their locations are confirmed. One of the most important objectives of the HapMap Project is to identify the positions of genetic variants [2]. Genetic markers may be used to study the relationship between an inherited disease and its genetic causes (see details later).

# 6.2.3 Mutations and Polymorphisms

A mutation is a change in DNA sequence caused by unrepaired damage to DNA or replication errors. Mutations lead to new genetic variants, and their effects range from fatal to mildly detrimental or even beneficial. If a mutation is not fatal, the individual carrying it can reproduce, thereby allowing the mutation to be passed to the next generations and to increase in frequency. For a seriously detrimental mutation, the individual is less likely to survive and reproduce, so the mutation very likely would become extinct over a few generations. Therefore, such deleterious mutations usually are rare in the general population. Genetic variants with a minor allele frequency (MAF) of less than 1 % are classified as rare variants. If the MAF is greater than 1 %, the variant is termed a polymorphism. Among polymorphisms, variants with an MAF greater than 5 % are called common variants; variants with an MAF from 1 % to 5 % are called low-frequency variants. In genetic studies, different methods and technologies are used to assay genetic variants with different frequencies (see later).

# 6.2.4 Mendelian Genetics and Complex Disease

A genetic disease is a disorder caused by an abnormality in an individual's DNA. Abnormalities range from a small mutation in a single gene to the addition or subtraction of a subset of chromosomes or even an entire chromosome. There are two major categories of genetic diseases: Mendelian and complex diseases. With regard to Mendelian



disorders, mutations in a single gene are sufficient to cause disease. Mendelian disorders are relatively rare and often are first recognized clinically by their predictable patterns of inheritance in families. Common Mendelian modes of inheritance include dominant inheritance, recessive inheritance, and X-linked and Y-linked dominant or recessive modes. More than 4,000 human diseases are caused by single-gene defects. Several skeletal abnormalities follow Mendelian inheritance; classic examples include osteogenesis imperfecta (OI) and spondyloepiphyseal dysplasia.

Mendelian disorders account for only a small proportion of the total burden of human genetic diseases. A much larger component is composed of congenital malformations and common adult diseases. These diseases have significant genetic components and also are influenced by multiple environmental factors. They commonly are called complex diseases or polygenic diseases. Complex disorders are caused by variant forms of genes and environmental factors; they may act independently or modify the effects of one another through gene-gene and/or gene-environment interactions.

LDD is an example of a complex disease. In earlier studies, age, gender, occupation, cigarette smoking, height, and weight were found to be associated with LDD [3]. Later studies suggested a large degree of genetic influences [4]. Recently, many susceptible genes were found to be associated with LDD [5].

## 6.2.5 Recombination, Linkage, and Linkage Disequilibrium

With regard to the relationship between two genetic variants in a population, some alleles located near each other on the same chromosome are transmitted together, rather than independently during reproduction. Such co-segregated behavior is termed *linkage*. However, not all genes on the same chromosome are linked, because recombination occurs during meiosis. When homologous chromosomes are paired, crossover occurs between non-sister chromosomes that are part of the chromosome exchanged (Fig. 6.4). As a result of exchange, new combinations of alleles can be formed on a chromosome. The probability of crossover events varies among the different regions of a chromosome. The nearer two positions are, the less chance recombination will occur. The recombination rate is an indication of the distance between two genetic markers, because the further away the markers are, the greater the likelihood of recombination. The distance between two markers may be expressed in centimorgans; 1 cM corresponds to 1 % of recombination and to

about one billion bases. In the human population, crossover positions vary; therefore, the allelic combinations of two markers differ among individuals. Normally, the frequency of non-recombined and recombined alleles is not equal to the randomized frequency. This nonrandom association between markers is called *linkage disequilibrium* (LD) and is measured by  $r^2$ . The greater the  $r^2$ , the more likely co-segregation will occur between two markers. Thus, the presence of one allele of a marker provides information indicating the allele of the nearer markers by the degree of LD. Recombination and LD are the theoretical basis of the linkage and association studies introduced later.

# 6.3 Identify the Genetic Cause of Disease

# 6.3.1 Estimate the Genetic Contribution of LDD

The first step in studies to identify disease genes is to estimate the heritability or the genetic contribution to the disease. This estimation involves observation and statistical analysis of the patterns of phenotypes with various genetic or environmental backgrounds in close kin, such as parent– offspring, siblings, and twins [6]. Familial aggregation and twin studies are two widely used methods for estimating heritability. Familial aggregation estimates the likelihood of a phenotype in close relatives compared with that in controls. On the other hand, studies of twins are useful in estimating the contribution to a phenotype by comparing monozygotic (MZ) pairs (in which all genes are shared) with dizygotic (DZ) pairs (in which half the genes are shared). If the similarity between MZ pairs is greater than that between DZ pairs, the greater part of similarity must be caused by genetics.

Traditionally, disk degeneration was thought to be caused by aging and "wear and tear" from mechanical changes and injuries; however, after familial aggregation and twin studies were conducted, there was a dramatic change in our knowledge of what causes disk degeneration. Several familial aggregation studies conducted in LDD found that young patients with disk herniation had a family history of the disease [7, 8].

The first systematic evaluation of lumbar degenerative changes blinded to twinship was conducted in 1995. Twenty MZ twin pairs from Finland were studied in a pilot investigation of disk degeneration, disk height narrowing, and disk bulging or herniation detected by MRI. A high degree of similarity (26-72 %) was observed between identical twin pairs [4]. Subsequently, a larger and more comprehensive investigation was conducted in 115 pairs of MZ twins. The result showed that 61 % of the variance in disk degeneration was explained by familial aggregation; beyond that, age and occupations requiring heavy lifting together explained 16 % [9]. A 1999 study of female twins in the United Kingdom and Australia enrolled 86 pairs of MZ twins and 77 pairs of DZ twins. The investigators reported 74 % heritability of LDD after adjusting for age, body weight, body height, smoking, occupation, and exercise [10].

# 6.3.2 Identify the Specific Genes Involved in the Degeneration of Disks

The high heritability estimates for LDD motivated researchers to identify the specific genes responsible. Accurate mapping of the genetic architecture of LDD provides clues to its etiology and pathogenesis. Moreover, those genes might be targets for drug discovery or markers for diagnosis. For Mendelian and complex diseases, the methods for identifying causal genes are different because of the different genetic architectures of these disorders. In Mendelian diseases, causal genes usually are rare and might be identified by studying affected families. Accurate analysis of the mode of disease inheritance might lead to the direct location of the causal variant on the genome. As for complex diseases, the involvement of multiple genes, as well as interactions between genes and the environment, makes it more difficult to identify the causal genetic factors. Two study designs for mapping disease genes are used commonly, namely, family-based design and population-based design; the statistical methods are linkage and association analysis, respectively. A study approach may be classified further as a candidate gene or genome-wide approach depending on whether prior biologic and etiologic knowledge is applied.

#### 6.3.2.1 Linkage Studies

Linkage analysis determines whether the marker segregates with the disease in families with multiple affected individuals, according to a Mendelian mode of inheritance. If a marker is passed down through generations of a family, it may be used as a surrogate for locating adjacent genes. Based on the characteristics of recombination, a linkage study maps disease-causing genes by tracking how alleles of a genetic marker have combined with different members of a family to identify possible crossover regions, thereby indicating shared regions in cases and controls, respectively. Regions shared in cases but not in controls are the candidate positions of disease association genes (Fig. 6.5). Because the crossover event occurs at different positions in different families, a linkage study in multiple families with the same phenotype will narrow down the candidate regions further by overlapping the candidate regions. Microsatellites and SNPs commonly are used as DNA markers in linkage studies.

Regarding LDD, a two-stage linkage study in 18 families of southern Chinese descent with

early-onset LDD was reported in 2013 [11]. A novel susceptible variant in carbohydrate sulfotransferase 3 (CHST3) was found to be associated with LDD. In the first stage of the study, 400 microsatellites of 89 individuals from 10 families were genotyped. Regions on chromosomes 1, 5, 8, 10, and 20 were identified as candidate regions. In the second stage, 37 individuals from 8 families were added, and another candidate in chromosome 10 was identified. The following association study detected the variant in CHST3 associated with LDD. Additionally, expression of CHST3 mRNA decreased significantly in the intervertebral disk cells of individuals carrying the A allele of SNP rs4148941.

Linkage studies in OI also have been conducted. In one of these studies, an autosomal recessive cause of OI was found in five Turkish families. Linkage mapping demonstrated that all affected individuals shared a 0.83 Mb region on chromosome 17. Further sequencing of this region revealed that the OI phenotype resulted from homozygosity for an in-frame deletion in *FKBP10* [12]. A frameshift mutation in transcription factor Osterix, which causes recessive OI, was detected in an Egyptian child [13]. Also found was a missense change, causing autosomal recessive OI [14].

#### 6.3.2.2 Association Studies

Genetic association studies look for a correlation between disease status and genetic variations to identify candidate genes or genome regions that contribute to a specific disease. A genetic association exists if a particular allele is more frequent in the affected group than in the nonaffected group. The greater frequency of this allele in the affected group may indicate that it increases the risk of the disease in question. Statistical analysis usually is performed to determine the significance of the frequency differences. SNPs are the most widely tested genetic markers in association studies, although microsatellite markers, indels, and CNVs also may be used.

A genetic association study usually is conducted in a population-based sample of affected



Fig. 6.5 Linkage analysis

and unaffected individuals (case–control study). After genotyping is performed (Box 6.1), the genotype distributions of each marker are compared between the case and control groups (Fig. 6.6). Statistical measurements of frequency differences then are calculated. A marker that is significantly more frequent in the case versus the control group is identified as a possible diseasecausing variant.

Association studies may be performed in one of two ways: (i) direct testing of an exposure SNP with known variable phenotypes,



Fig. 6.6 Case-control association studies

such as altered protein structure, and (ii) indirect testing of an SNP, which is a surrogate marker for locating adjacent functional genes that contribute to disease status. The first method requires identification of all variants in the coding and regulatory regions of genes. Although the cost of genotyping continues to decrease, genotyping all ten million SNPs of the whole genome is still expensive. The latter method eliminates the need to catalog all potential susceptibility variants by relying instead on the LD between a genetic marker and susceptible polymorphisms; only a subset of SNPs is genotyped. The genetic constitution of SNPs that are not genotyped may be inferred based on  $r^2$ , which is provided by the HapMap Project [2] (Box 6.1).

#### Box 6.1. Technologies of SNP Genotyping

The International HapMap Project has enabled researchers to identify most of the common variants across the whole genome [2]. The project also generated allele frequencies and correlations ( $r^2$ ) of the SNPs in different populations. From these data, the LD map was generated as a reference, recording the co-inheritance of the SNPs. The recently launched 1000 Genomes Project aims to provide a more profound characterization of genetic variations in multiple populations [15]. The use of high-throughput sequencing technologies has provided an opportunity to detect not only common variants but also rare variants in different populations. Based on the data generated from these two projects, commercial genotyping microarrays were designed. The genotyping microarray is a solid surface on which artificial microscopic DNA spots of known variants may be attached. The attached DNA spots can capture corresponding sequences marked with an optical signal for identifying genotypes. The Illumina Omni array can detect more than 4.3 million markers, whereas the Affymetrix SNP array can assess 1.8 million markers. Furthermore, the multiplexing Sequenom MassARRAY system and Illumina VeraCode technology have enabled us to genotype a specific region with many SNPs efficiently and costeffectively. Thanks to these microarrays, a two-step cost-effective approach has been used widely by researchers. The first step is to select a subset of SNPs as a marker for genotyping and to perform an association scan to identify candidate regions. The second step is to genotype a denser set of SNPs within potential regions to look for the exact functional genetic variants.

#### **Candidate Gene Approach**

In recent years, the cost of genotyping has dropped dramatically, but the cost of customized genotyping platforms remains high, and limited budgets prevent studies from genotyping a dense set of SNPs across the whole genome. The statistical significance and power of a study are directly affected by the number of individuals tested and by the number of SNPs genotyped. The more SNPs that are genotyped, the more information is obtained and the greater the chance a causal SNP will be identified. To balance cost with study power, investigators select and type the number of SNPs that will maximize the power of the association study. Usually, they do this by choosing a subset of SNPs from candidate genes whose number is commonly determined by the available resources.

The candidate gene approach focuses on selecting genes according to the biological knowledge and etiology of the disease. It takes advantage of our biological understanding of the phenotype, tissues, genes, and proteins that likely are involved in the disease. With regard to disk degeneration, for example, collagen and aggrecan, together with other structural proteins, form the basis of the extracellular matrix, which is an integral part of the disk. These proteins are essential for normal disk function in terms of tensile strength and osmotic pressure. Therefore, the extracellular matrix genes have been prime candidates for genetic study. In addition, gene expression studies are another important way to identify candidate genes. Several tissue homeostasis genes, such as matrix metalloproteinases (MMPs), have elevated expression levels, as well as increased enzymatic activity, during disk degeneration [16]. Moreover, genes identified from other skeletal disorders also may be candidates for LDD. For example, growth differentiation factor 5 (GDF5) is one of the promising candidate genes of osteoarthritis (OA), as it has been reported in multiple populations and has shown high significance [17]. GDF5 was studied as a candidate gene of LDD in northern European women. An SNP (rs143383) was found to be significantly associated with the combination of disk space narrowing and osteophytes [18]. One major drawback of the candidate gene approach, however, is that a priori knowledge of the pathogenesis of the disease is required. If the molecular mechanism is poorly understood, the wrong genes might be selected. Therefore, candidate gene studies are more successful when used as a follow-up to linkage studies [11].

# Genome-Wide Association and Meta-Analysis

An advantage of a genome-wide association study (GWAS) is that it requires no prior knowledge of the structure or function of susceptibility genes. A GWAS examines common genetic variants across the whole genome in different individuals to determine whether any variant is associated with disease status. Therefore, this approach makes it possible to identify novel suspected genes. Typically, the power of a GWAS relies on the sample size, definition of the phenotypes, and control of environmental factors. A large sample size, a well-defined phenotype, and proper control of environmental factors might lead to a successful GWAS.

Although GWASs have identified many variants associated with complex diseases, these variants currently explain little about the heritability of most diseases. Normally, the effect sizes of common variants are small, and detection of such small effects requires large sample sizes. Although individual GWASs are underpowered, meta-analyses increase power and reduce falsepositive findings. In a meta-analysis, results from several independent studies are contrasted and combined in the hope of identifying the same patterns among study results, sources of disagreement among those results, or other interesting relationships that might come to light in the context of multiple studies.

#### 6.3.2.3 Interpreting the Results

A significant genetic association may be interpreted as (1) a direct association, in which the genotyped SNP is the true causal variant conferring disease susceptibility; (2) an indirect association, in which an SNP in LD with the true causal variant is genotyped; or (3) a false-positive result, in which there is either chance or systematic confounding, such as population stratification. Population stratification is the presence of a systematic difference in allele frequency between subpopulations of the same population, possibly as the result of different ancestry, especially in the context of association studies.

In recent years, the number of genetic studies of LDD has been increasing. It is important to interpret and integrate the results from these studies to improve our overall understanding of LDD, especially with regard to phenotype definitions, statistical significance, and the effects of suspected genes.

#### Phenotype Definitions

A precise phenotype definition is essential for genetic studies in that the phenotype should be a distinguishable trait and preferably quantifiable. Generally, a trait may be classified as qualitative or quantitative; a qualitative trait can fit into distinct phenotypic categories (case or control), whereas a quantitative trait is measurable as a continuous variable.

With regard to LDD, for example, current assessments of disk degeneration rely on imaging, including radiography and magnetic resonance imaging (MRI). Radiographs may provide information on disk height and osteophyte formation, whereas MRI may show hydration status and disk bulging and herniation, as well as endplate irregularities. From these images, the presence or absence, or even the severity, of disk degeneration can be defined.

There are several ways to evaluate the degenerative changes in the intervertebral disk. The first is to make a diagnosis based on the presence or absence of disk degeneration, which is an example of a qualitative trait; it is simple and commonly used clinically. However, a disadvantage is that it provides no information about the progressive changes that take place during the degenerative process. Another method is to classify the severity of degeneration based on some well-defined criteria. This method is the one used most widely in genetic studies, and several scoring systems have been developed. For radiographic studies, the Kellgren scale combines the features of osteophytes and joint space narrowing to generate a score ranging from 1, indicating no or very small osteophytes, to 4, representing large osteophytes and pronounced disk space narrowing [19]. For MRI, two scoring systems were developed: the Schneiderman and Pfirrmann scales. Schneiderman's system focuses on the signal intensity of the nucleus pulposus on MRI, classifying it into four grades, with 0 indicating normal disk with a hyperintense signal (bright disk) and 3 illustrating a hypointense signal with disk space narrowing (black disk) [20]. Pfirrmann's classification uses MRI images to evaluate the homogeneity of disk structure, signal intensity, and distinction between the nucleus pulposus and annulus fibrosus, as well as disk height. This information is converted into one of five grades, the lowest of which is applied to homogeneous disk structure,

	Disea	se status
Risk	Present	Absent
Allele 1	a	b
Allele 2	С	d

b

 $RR = \frac{\boxed{a + b}}{c} = \frac{a * (c + d)}{c * (a + b)}$ 

 $OR = \frac{\frac{a}{a+b}}{1-\frac{a}{a+b}} = \frac{a*d}{b*c}$  $\frac{\frac{c}{c+d}}{1-\frac{c}{c+d}}$ 

Fig. 6.7 Relative risk and odds ratio

hyperintense signal, and normal disk height and the highest to inhomogeneous disk structure, hypointense signal, and loss of distinction between the nucleus pulposus and annulus fibrosus, as well as collapsed disk space [21]. These grading systems provide a semiquantitative evaluation of degenerative status, reflecting the severity of disk degeneration. Interpreting the images from MRI is subjective and thus requires multiple experienced radiologists to perform the grading. The third method for assessing disk degeneration is computational evaluation, which avoids personal errors and saves human resources. Both semiautomated [22] and automated [23] frameworks have been developed for the diagnosis of degenerative disks. The output of computational evaluation is quantitative measurement.

#### Statistical Significance

The most conspicuous information from a genetic study is whether a genetic variant or gene is associated with a certain disease and may be derived from the test statistics and their corresponding P values. The P value represents the possibility of no association, indicating there is no difference in genotypic distribution between cases and controls. Generally, if P < 0.05, the hypothesis of no association is rejected, meaning there is an association between a genotype and the disease

status. In candidate gene studies or GWASs, every genetic variant is tested. If the number of hypotheses being tested increases, the false-positive rate also might increase; therefore, several methods were developed to address this possibility, with the Bonferroni correction being one of the most widely used approaches [24]. The corrected statistical significance level is 1/n times what it would be if only one hypothesis were tested. Thus, the significance threshold of association studies should be 0.05/number of markers being tested.

#### **Effects of Suspect Genes**

In epidemiology, relative risk (RR) is the ratio of the probability of a disease occurring in an exposed group to the probability of it occurring in a comparative, nonexposed group (Fig. 6.7a). An RR of N means that the affected group has a risk N times greater than that of the nonaffected group. In genetic association studies, the fundamental unit for reporting effect sizes of suspect genes is the odds ratio (OR). The OR represents the ratio between two proportions: the proportion of individuals in the case group with a specific allele and the proportion of individuals in the control group having the same allele (Fig. 6.7b). An OR >1 demonstrates that individuals carrying the allele are more susceptible to the disease; an OR <1 indicates less susceptibility. The OR is

а

similar to the RR when the disease prevalence is low (in Fig. 6.7, when *a* and *c* are small). ORs usually are reported together with 95 % confidence intervals, which show the possible range of a gene's effect.

#### 6.3.3 Genes Associated with LDD

So far, more than 20 genes have been associated with LDD (Table 6.1), although only a few of them can be replicated, and the results of these studies should be scrutinized closely. In 2008, a Human Genome Epidemiology Network (HuGENet) working group developed a scoring system to provide a reasonable assessment based on three criteria: amount of evidence, replication, and protection from bias. For each criterion, a classification of strong (A), moderate (B), or weak (C) is assigned to the gene study or studies [53]. Eskola et al. [5] used the HuGENet criteria in their systematic review of genetic association studies in LDD. According to their results, most of the associations presented with a weak level of evidence; only five genes showed moderate evidence. None of the studies of disk degeneration genes reached the level of strong credibility. In this chapter, some specific genes are introduced to enhance our understanding of the genetics of LDD.

Vitamin D receptor (VDR) is the best replicated gene and has been verified in three different populations. Identified in a Finnish twin study, it was the first gene reported to be associated with LDD [25]. In this study, a reduction in signal intensity on MRI was associated with the TaqI and FokI polymorphisms. The association of *TaqI* was replicated later in a Japanese study [26], with genotype Tt occurring more frequently in individuals with disk degeneration and severe disk degeneration. TaqI was replicated further in a Chinese population-based study [27]. The replication of the TaqI polymorphism of VDR in three different ethnic populations indicates that VDR is the most robust gene associated with LDD. Although no functional validation has been performed, it is hypothesized that the variation in VDR leads to changes in the structural characteristic of the extracellular matrix in the vertebral disks [54].

Aggrecan, encoded by the gene ACAN, is the major proteoglycan component of cartilage and the nucleus pulpous of the intervertebral disk and is responsible for maintaining hydration of the disk structure. Thus, ACAN is considered a good candidate for genetic association studies. A variable number of tandem repeats (VNTR) in ACAN were first associated with LDD in a group of 64 young Japanese women aged 20–29 years [28]. The study found that the shorter allele (<25 repeats) carried a higher risk of disk degeneration. This association was replicated in Han Chinese [29], Korean [30], and Turkish [31] populations. The Han Chinese study also showed that the shorter allele carried an even higher risk (OR, 4.5) for symptomatic disk degeneration in smokers, suggesting there might be a link between the risk allele and smoking. However, a Finnish study found that the allele with 26 repeats was significantly associated with a dark nucleus pulposus [32]. These conflicting findings might be the result of ethnic differences. Nevertheless, the Finnish study still showed that ACAN is a risk factor for LDD.

An association between the asporin gene (ASPN) and LDD was found in two independent Asian cohorts, one Japanese (N=1,353) and the other Chinese (N=1,055) [33]. ASPN became a candidate gene of LDD susceptibility because asporin is an extracellular matrix protein shown to be associated with OA of the knees. In the Japanese cases, the presence of at least one D14 allele was found to be significantly associated with lumbar disk herniation (LDH) characterized by sciatica, whereas in the Chinese population, the presence of at least one D14 repeat was associated with LDD. A meta-analysis using the aforementioned phenotypes showed that individuals carrying the D14 allele had a higher risk of LDH or disk degeneration, with an OR of 1.58.

Several types of collagen have been studied widely in LDD, with type XI collagen, encoded by *COL11A1*, providing the most reliable evidence. A Japanese study found an association between the *COL11A1* rs1676486 T allele and LDH characterized by sciatica. This study consisted of three stages (N=367, N=645, N=710), each of which showed a significant association. When the populations were combined for meta-analysis (823 cases and 838 controls), the individuals with minor

Table 6.1 G	Jenetic risk factors of l	lumbar dis	sk degeneration			
Gene	Cohort	z	Variant	OR	Phenotype	Reference
VDR	Finnish	85 MZ pairs	TagI and FokI		LDD	Videman et al. [25]
	Japanese	205	TaqI		LDD, LDH	Kawaguchi et al. [26]
	Southern Chinese	804	TaqI	2.61	LDD, LDH	Cheung et al. [27]
ACAN	Japanese	64	VNTR		Aggrecan protein size	Doege et al. [28]
	Han Chinese	132	VNTR	1.03-4.5	LDD	Cong et al. [29]
	Korean	104	VNTR		LDD	Kim et al. [30]
	Turkish	100	VNTR		LDD radiographic	Eser et al. [31]
	Finnish	132	VNTR		LDD	Solovieva et al. [32]
ASPN	Japanese	1,353	D14 allele	1.69	LDD	Song et al. [33]
	Chinese	1,055	D14 allele	1.49	LDD	Song et al. [33]
COL11A1	Japanese	1,722	rs1676486	1.34-1.55	LDD, LDH	Mio et al. [34]
COL9A2	Finnish	157	TRP2		Lumbar spinal stenosis, sciatica	Annunen et al. [35]
	Southern Chinese	804	TRP2	2.4, 4	Annular tears, DDD, and end-plate herniations	Jim et al. [36]
COL9A3	Finnish	164	TRP3	3	DDD	Paassilta et al. [37]
COLIA1, COLIA2	Dutch	517	rs1800012	3.6	Osteophytes and articular joint space narrowing radiographs	Pluijm et al. [38]
	Greek	40	rs1800012		LDD	Tilkeridis et al. [39]
GDF5	Northern European	5,259	rs143383	1.72	Disk space narrowing and osteophytes	Williams et al. [18]
SKT	Japanese	1,758	rs16924573	1.34	LDD	Karasugi et al. [40]
	Finnish	506	rs16924573	1.34	LDD	Karasugi et al. [40]
	Finnish	538	rs16924573	0.27	LDD	Kelempisioti et al. [41]
<b>MMP1</b>	Southern Chinese	691	Indel in promoter, -1607G/D	1.5	LDD	Song et al. [42]
MMP2	Chinese	480	SNP in promoter, -1306C->T	3.08	LDD	Dong et al. [43]
MMP3	Japanese	49	SNP in promoter, 5A/6A		LDD	Takahashi et al. [44]
MMP9	Han Chinese	859	SNP in promoter, -1562C->T	1.29	LDD	Sun et al. [45]
THBS2	Japanese	1,743	rs9406328	1.38	LDH	Hirose et al. [46]
PARK2	Northern European	4,600	rs926849		LDD	Williams et al. [47]
CASP9	Han Chinese	799	SNP Ex5+32 G/A	1.91	LDH	Sun et al. [48]

		-				
CILP	Japanese	1,121	SNP 1184 T ->C	1.61	LDD	Seki et al. [49]
FAS	Chinese	563	rs2234767	1.88	LDD	Zhu et al. [50]
FASL	Chinese	563	rs763110	2.34	LDD	Zhu et al. [50]
ILIRN	Korean	281	VNTR A1, A3	0.45, 3.86	LDH	Kim et al. [51]
IL10	Chinese	589	-592 A/C		LDD	Lin et al. [52]

allele T had a higher risk of LDH. In addition, the study suggested the SNP rs167486 affected mRNA stability [34].

A rare mutation (Trp2 allele) of the type IX collagen gene (COL9A2) was first identified in 6 of 157 Finnish patients with sciatica, with no evidence of this mutation in any of the controls. Furthermore, a family linkage study indicated that this mutation causes disease [35]. The allele frequency of this mutation is much higher in the Chinese population, and this association was found to be age related, with the greatest effect observed in the 40-49 age group [36]. Another variant (Trp3 allele) also was found in the Finnish study [37] but not in the Chinese population [36], and it had no replications in southern Europeans [55]. A subsequent functional study suggested that the Trp2 and Trp3 variants in type IX collagen are associated with degenerative lumbar spinal stenosis.

Type I collagen, encoded by *COL1A1* and *COL1A2*, was first studied in patients with reduced bone mineral density [56]; an SNP (rs1800012) was reported to be associated. Later, a study in 517 Dutch cohorts reported that those with genotype TT of this SNP have a higher risk of disk degeneration; however, the disk degeneration phenotype was defined by the presence of osteophytes and articular joint space narrowing based on radiographs [38]. The only investigation using an MRI definition of disk degeneration was a Greek study with 40 young army recruits, which was not a large enough sample [39]. Therefore, how this SNP increases the risk of LDD remains unknown.

A recent collaborative study with northern European subjects investigated the SNP (rs143383) of the growth differentiation factor (GDF5) gene. Of the total population (N=5,259), one cohort (N=613) underwent MRI. Among women, the meta-analysis revealed a significant association between rs143383 and the combined phenotype of disk space narrowing and osteophytes. When only the MRI cohort was investigated, however, the association was not statistically significant, thus generating some inconsistency regarding this association [18]. GDF5 is required for joint formation [57] but also associated with OA [17].

The polymorphisms of the sickle tail (SKT) gene were analyzed in a Japanese population, in which an SNP (rs16924573) was strongly associated with LDH (N=1,758), and this finding was replicated among Finnish subjects (N=506) [40]. Although the allele frequencies were different between the Finnish and Japanese populations, the meta-analysis of more than 2,200 subjects supported the association. A replication study between decreased disk signal intensity and SKT rs16924573 in a Finnish population has been published (OR, 0.27; 95 % CI, 0.07-0.96; P=0.024) [41]. The G allele was more frequent in the case group in both studies, thus indicating an increased risk. The function of SKT is unknown and further studies are needed.

Matrix metalloproteinases (MMPs) are a large protein family expressed in disks, and the genes MMP1, MMP2, MMP3, and MMP9 are reported to be associated with disk degeneration. Interestingly, all the susceptible polymorphisms of *MMP*s are located in the promoter region. In a study of MMP1 in a southern Chinese cohort of 691 individuals, significant signal was found only among individuals aged 40 or older [42]. An SNP on the promoter region of MMP2 was found to be associated with severe disk degeneration in another Chinese cohort [43]. MMP3 was reported to have a polymorphism on its promoter associated with both the onset and progression of disk degeneration in 49 elderly Japanese patients [44]; however, this association was not replicated in young subjects. Regarding MMP9, an SNP in the promoter region was associated with disk degeneration in a northern Chinese cohort (N=859) [45].

The thrombospondin 2 (*THBS2*) gene was examined in two independent Japanese populations (N=1,089 and N=654) as a candidate gene for LDH. The SNP (rs9406328) showed significant association in each population independently as well as in the meta-analysis [46]. These studies also suggest that *THBS2* might be involved in regulating *MMP* expression in the disk, in turn participating in the pathogenesis of disk herniation. Moreover, the authors also examined the combined effects of *THBS2* and *MMP9*; the OR of 3.3 for the combination indicates a potential gene–gene interaction [46]. The first GWAS of LDD, a meta-analysis of a northern European cohort, involved 4,600 samples from four populations. This study identified an association between an SNP (rs926849) in the parkin gene *PARK2* ( $P=2.8 \times 10-8$ ) and LDD [47].

# 6.3.4 Moving from Common Variants to Rare Variants

Disk degeneration provides an overall, subjective impression of the spine's condition. It may include signal intensity loss, bulging, herniation, end-plate irregularities, osteophytes, and narrowing of the disk space. A wide range of symptoms and degrees of severity are associated with LDD. Nearly everyone shows some signs of wear and tear on the lumbar disks as they age; however, not everyone will have symptoms such as back pain or sciatica. Severe disk degeneration sometimes is observed in the very young, whereas relatively normal disks may be found in older individuals, although these situations are rare and different from commonly seen cases. On the other hand, according to estimates from twin studies, LDD has up to 74 % heritability, although very few common variants for LDD have been functionally validated. Considering the genetic variants currently associated with LDD, as well as their effect size, a substantial number of cases remain unexplained, suggesting that there are several genetic factors waiting to be identified.

So far, most genetic studies have focused on common SNPs in candidate genes, whereas the human genome contains many more variations, including rare SNPs, indels, CNVs, inversions, and others. Evidence is accumulating that rare variants do play a role in common complex disease; for example, three rare deletions were found to be associated with schizophrenia, one of which had an OR of 14.8 [58]. Four rare SNPs in *IFIH1* independently reduced the risk of type 1 diabetes, with one of the SNPs having an OR of 0.5 [59], and 36 very rare non-synonymous variants were associated with type 2 diabetes with an OR of 3.3 [60]. These findings suggest that rare variants contributing to susceptibility of common diseases is not an unusual event.

Therefore, a more complete understanding of LDD requires genome-wide studies that fully examine both common and noncommon variants in populations.

To identify rare variants, it will be necessary to sequence entire genomes of cases, instead of genotyping a catalog of variants. Although the best method is to sequence the whole genome, this approach is expensive, especially if there is a large cohort to be sequenced for a complex disease. To reduce the cost, whole-exome sequencing has been used widely to detect variants located in exons. A targeted sequencing method also has been used to sequence regions where potential associations have been indicated through linkage or GWASs (Box 6.2).

#### Box 6.2. DNA Sequencing Technologies

DNA sequencing is the process used to determine the precise order of nucleotides within a DNA molecule. The first DNA sequencing methods were developed in the 1970s. Frederick Sanger and colleagues [61] developed rapid DNA sequencing methods with chain-terminating inhibitors, whereas Walter Gilbert and Allan Maxam [62] developed sequencing methods by using chemical degradation. These two methods commonly are referred to as firstgeneration sequencing. Since then, the demand for reducing cost and increasing throughput has driven the development of DNA sequencing technology. By the 1990s, new technologies paralleled the sequencing process, greatly improving throughput; thus, DNA sequencing entered the next generation and several secondgeneration sequencing technologies were developed. The major commercial entities that came into existence include 454 Life Sciences, Solexa/Illumina, SOLiD, and the Polonator. With the availability of a multitude of platforms, dramatically lower costs, and a higher throughput of sequencing, second-generation sequencing technologies have been used widely in studying genetic diseases.

#### 6.4 Epigenetics

*Epigenetics* refers to heritable, functionally relevant changes to the genome that do not involve a change in the nucleotide sequence. Packaging genes into chromatin and dynamic chromatin remodeling processes are required for the initial step in gene expression (transcription). Epigenetic factors are responsible for this regulatory process, the major components of which are DNA methylation, histone modifications, and the action of noncoding RNAs. Unlike the DNA sequence, which largely is fixed throughout a person's lifetime, epigenetic patterns not only vary from tissue to tissue but become altered with advancing age and are sensitive to environmental exposures.

Recent studies provide evidence that DNA methylation, a biochemical process whereby a methyl group is added to the cytosine or adenine DNA nucleotides, is involved in the development of complex disease. Hypermethylation at one CpG island of the PARK2 promoter was associated with MRI-determined LDD [47]. DNA methylation changes also were reported in other complex diseases: in schizophrenia and bipolar disorder, substantial differences in genome-wide DNA methylation patterns were found between monozygotic twins discordant for these diseases [63]. Epigenetic variation also might help explain the late onset and progressive nature of most common diseases, the quantitative nature of complex traits, and the role of environment in disease development [64]. Evidence exists that epigenetic patterns are altered by environmental factors known to be associated with disease risk (e.g., diet, smoking, alcohol intake, environmental toxicants, and stress) [65]. These observations suggest that complex diseases are the product of the integration of epigenetic and genetic factors.

Currently, arrays and sequencing technologies both are available for identifying genome-wide methylation variations [66]. Similar to the GWAS, an association scan that uses methylation variations is referred to as an *epigenome-wide association study* [66].

Besides DNA methylation, noncoding RNAs, functional RNA molecules that are not translated

into a protein, are integral components of biological networks with a fundamental role in regulating gene expression. Significantly different expressions of microRNAs have been detected between patients with intervertebral disk degeneration and those with an intervertebral disk injury [67]. Differential expressions of microRNA also have been reported in OA [68].

### 6.5 The Way Forward

Genetic studies of LDD have been carried out for nearly two decades. With the continuous advancement of genotyping and DNA sequencing technologies, as well as improvements in imaging and statistical methods, our knowledge about the natural history of degenerative disk disease is constantly improving [69]. However, the complexity of the degenerative process still is not fully understood [5]. One serious problem is that very few susceptible genes can be replicated [5]. In addition, there still is a large proportion of missing heritability that cannot be explained by current findings. According to our experience, there are several approaches that might be able to maximize the chance of identifying new LDD risk factors. For a complex trait, a larger sample size may provide a more accurate evaluation of variants. In OA, a recent GWAS from the United Kingdom enrolled up to 70,000 subjects and found several promising loci [70]. For LDD, which is more prevalent than OA, the largest association study enrolled only 4,600 subjects [47]. In a study of body height and lipids, a very large sample size enabled researchers to find many reliable genetic factors [71]. Therefore, to identify reliable and relevant variants associated with LDD, larger-scale multiethnic genetic studies and interethnic comparisons through international collaborations inevitably are necessary. Nevertheless, there is a wide variation in phenotype definitions among research groups, with differences in sample selection strategies and image features used to represent disk degeneration. Therefore, to ensure the same group of genetic factors is identified, a unified phenotypic definition is required. Moreover, based on the

hypothesis that disk degeneration is the product of genetic, epigenetic, and environmental factors, as well as their interactions, more information should be considered in future genetic studies, such as gene–environment interaction, gene regulation, and so on. Also, instead of looking at one genetic variant at a time, all factors should be studied systematically, because the overall picture may provide a better understanding of the genetic architecture and etiology of degenerative disk disease.

#### References

- Cheung KM, Karppinen J, Chan D, Ho DW, Song YQ, Sham P, Cheah KS, Leong JC, Luk KD. Prevalence and pattern of lumbar magnetic resonance imaging changes in a population study of one thousand fortythree individuals. Spine (Phila Pa 1976). 2009;34: 934–40.
- International Hapmap, C. The international HapMap project. Nature. 2003;426:789–96.
- Hassett G, Hart DJ, Manek NJ, Doyle DV, Spector TD. Risk factors for progression of lumbar spine disc degeneration: the Chingford Study. Arthritis Rheumatol. 2003;48:3112–7.
- Battie MC, Haynor DR, Fisher LD, Gill K, Gibbons LE, Videman T. Similarities in degenerative findings on magnetic resonance images of the lumbar spines of identical twins. J Bone Joint Surg Am. 1995;77:1662–70.
- Eskola PJ, Lemmela S, Kjaer P, Solovieva S, Mannikko M, Tommerup N, Lind-Thomsen A, Husgafvel-Pursiainen K, Cheung KM, Chan D, Samartzis D, Karppinen J. Genetic association studies in lumbar disc degeneration: a systematic review. PLoS One. 2012;7:e49995.
- Visscher PM, Hill WG, Wray NR. Heritability in the genomics era–concepts and misconceptions. Nat Rev Genet. 2008;9:255–66.
- Varlotta GP, Brown MD, Kelsey JL, Golden AL. Familial predisposition for herniation of a lumbar disc in patients who are less than twenty-one years old. J Bone Joint Surg (Am Vol). 1991;73:124–8.
- Matsui H, Terahata N, Tsuji H, Hirano N, Naruse Y. Familial predisposition and clustering for juvenile lumbar disc herniation. Spine. 1992;17:1323–8.
- Battie MC, Videman T, Gibbons LE, Fisher LD, Manninen H, Gill K. 1995 Volvo Award in clinical sciences. Determinants of lumbar disc degeneration. A study relating lifetime exposures and magnetic resonance imaging findings in identical twins. Spine. 1995;20:2601–12.
- Sambrook PN, Macgregor AJ, Spector TD. Genetic influences on cervical and lumbar disc degeneration: a

magnetic resonance imaging study in twins. Arthritis Rheum. 1999;42:366–72.

- 11. Song YQ, Karasugi T, Cheung KM, Chiba K, Ho DW, Miyake A, Kao PY, Sze KL, Yee A, Takahashi A, Kawaguchi Y, Mikami Y, Matsumoto M, Togawa D, Kanayama M, Shi D, Dai J, Jiang Q, Wu C, Tian W, Wang N, Leong JC, Luk KD, Yip SP, Cherny SS, Wang J, Mundlos S, Kelempisioti A, Eskola PJ, Mannikko M, Makela P, Karppinen J, Jarvelin MR, O'Reilly PF, Kubo M, Kimura T, Kubo T, Toyama Y, Mizuta H, Cheah KS, Tsunoda T, Sham PC, Ikegawa S, Chan D. Lumbar disc degeneration is linked to a carbohydrate sulfotransferase 3 variant. J Clin Invest. 2013;123:4909–17.
- 12. Alanay Y, Avaygan H, Camacho N, Utine GE, Boduroglu K, Aktas D, Alikasifoglu M, Tuncbilek E, Orhan D, Bakar FT, Zabel B, Superti-Furga A, Bruckner-Tuderman L, Curry CJ, Pyott S, Byers PH, Eyre DR, Baldridge D, Lee B, Merrill AE, Davis EC, Cohn DH, Akarsu N, Krakow D. Mutations in the gene encoding the RER protein FKBP65 cause autosomal-recessive osteogenesis imperfecta. Am J Hum Genet. 2010;86:551–9.
- Lapunzina P, Aglan M, Temtamy S, Caparros-Martin JA, Valencia M, Leton R, Martinez-Glez V, Elhossini R, Amr K, Vilaboa N, Ruiz-Perez VL. Identification of a frameshift mutation in Osterix in a patient with recessive osteogenesis imperfecta. Am J Hum Genet. 2010;87:110–4.
- Martinez-Glez V, Valencia M, Caparros-Martin JA, Aglan M, Temtamy S, Tenorio J, Pulido V, Lindert U, Rohrbach M, Eyre D, Giunta C, Lapunzina P, Ruiz-Perez VL. Identification of a mutation causing deficient BMP1/mTLD proteolytic activity in autosomal recessive osteogenesis imperfecta. Hum Mutat. 2012;33:343–50.
- Genomes Project, C, Abecasis GR, Altshuler D, Auton A, Brooks LD, Durbin RM, Gibbs RA, Hurles ME, Mcvean GA. A map of human genome variation from population-scale sequencing. Nature. 2010;467: 1061–73.
- Bachmeier BE, Nerlich A, Mittermaier N, Weiler C, Lumenta C, Wuertz K, Boos N. Matrix metalloproteinase expression levels suggest distinct enzyme roles during lumbar disc herniation and degeneration. Eur Spine J. 2009;18:1573–86.
- Miyamoto Y, Mabuchi A, Shi D, Kubo T, Takatori Y, Saito S, Fujioka M, Sudo A, Uchida A, Yamamoto S, Ozaki K, Takigawa M, Tanaka T, Nakamura Y, Jiang Q, Ikegawa S. A functional polymorphism in the 5' UTR of GDF5 is associated with susceptibility to osteoarthritis. Nat Genet. 2007;39: 529–33.
- 18. Williams FM, Popham M, Hart DJ, de Schepper E, Bierma-Zeinstra S, Hofman A, Uitterlinden AG, Arden NK, Cooper C, Spector TD, Valdes AM, van Meurs J. GDF5 single-nucleotide polymorphism rs143383 is associated with lumbar disc degeneration in Northern European women. Arthritis Rheum. 2011;63:708–12.

- Kellgren JH, Lawrence JS. Radiological assessment of osteo-arthrosis. Ann Rheum Dis. 1957;16: 494–502.
- Schneiderman G, Flannigan B, Kingston S, Thomas J, Dillin WH, Watkins RG. Magnetic resonance imaging in the diagnosis of disc degeneration: correlation with discography. Spine (Phila Pa 1976). 1987;12:276–81.
- Pfirrmann CW, Metzdorf A, Zanetti M, Hodler J, Boos N. Magnetic resonance classification of lumbar intervertebral disc degeneration. Spine (Phila Pa 1976). 2001;26:1873–8.
- 22. Videman T, Nummi P, Battie MC, Gill K. Digital assessment of MRI for lumbar disc desiccation. A comparison of digital versus subjective assessments and digital intensity profiles versus discogram and macroanatomic findings. Spine (Phila Pa 1976). 1994;19:192–8.
- Auerbach JD, Johannessen W, Borthakur A, Wheaton AJ, Dolinskas CA, Balderston RA, Reddy R, Elliott DM. In vivo quantification of human lumbar disc degeneration using T(1rho)-weighted magnetic resonance imaging. Eur Spine J. 2006;15 Suppl 3:S338–44.
- Bland JM, Altman DG. Multiple significance tests: the Bonferroni method. BMJ. 1995;310:170.
- Videman T, Leppavuori J, Kaprio J, Battie MC, Gibbons LE, Peltonen L, Koskenvuo M. Intragenic polymorphisms of the vitamin D receptor gene associated with intervertebral disc degeneration. Spine. 1998;23:2477–85.
- Kawaguchi Y, Kanamori M, Ishihara H, Ohmori K, Matsui H, Kimura T. The association of lumbar disc disease with vitamin-D receptor gene polymorphism. J Bone Joint Surg Am. 2002;84-A:2022–8.
- 27. Cheung KM, Chan D, Karppinen J, Chen Y, Jim JJ, Yip SP, Ott J, Wong KK, Sham P, Luk KD, Cheah KS, Leong JC, Song YQ. Association of the Taq I allele in vitamin D receptor with degenerative disc disease and disc bulge in a Chinese population. Spine (Phila Pa 1976). 2006;31:1143–8.
- Doege KJ, Coulter SN, Meek LM, Maslen K, Wood JG. A human-specific polymorphism in the coding region of the aggrecan gene. Variable number of tandem repeats produce a range of core protein sizes in the general population. J Biol Chem. 1997;272:13974–9.
- 29. Cong L, Pang H, Xuan D, Tu G. The interaction between aggrecan gene VNTR polymorphism and cigarette smoking in predicting incident symptomatic intervertebral disc degeneration. Connect Tissue Res. 2010;51:397–403.
- Kim NK, Shin DA, Han IB, Yoo EH, Kim SH, Chung SS. The association of aggrecan gene polymorphism with the risk of intervertebral disc degeneration. Acta Neurochir (Wien). 2011;153:129–33.
- Eser O, Eser B, Cosar M, Erdogan MO, Aslan A, Yildiz H, Solak M, Haktanir A. Short aggrecan gene repetitive alleles associated with lumbar degenerative disc disease in Turkish patients. Genet Mol Res. 2011;10:1923–30.

- 32. Solovieva S, Noponen N, Mannikko M, Leino-Arjas P, Luoma K, Raininko R, Ala-Kokko L, Riihimaki H. Association between the aggrecan gene variable number of tandem repeats polymorphism and intervertebral disc degeneration. Spine (Phila Pa 1976). 2007;32:1700–5.
- 33. Song YQ, Cheung KM, Ho DW, Poon SC, Chiba K, Kawaguchi Y, Hirose Y, Alini M, Grad S, Yee AF, Leong JC, Luk KD, Yip SP, Karppinen J, Cheah KS, Sham P, Ikegawa S, Chan D. Association of the asporin D14 allele with lumbar-disc degeneration in Asians. Am J Hum Genet. 2008;82:744–7.
- 34. Mio F, Chiba K, Hirose Y, Kawaguchi Y, Mikami Y, Oya T, Mori M, Kamata M, Matsumoto M, Ozaki K, Tanaka T, Takahashi A, Kubo T, Kimura T, Toyama Y, Ikegawa S. A functional polymorphism in COL11A1, which encodes the alpha 1 chain of type XI collagen, is associated with susceptibility to lumbar disc herniation. Am J Hum Genet. 2007;81:1271–7.
- 35. Annunen S, Paassilta P, Lohiniva J, Perala M, Pihlajamaa T, Karppinen J, Tervonen O, Kroger H, Lahde S, Vanharanta H, Ryhanen L, Goring HH, Ott J, Prockop DJ, Ala-Kokko L. An allele of COL9A2 associated with intervertebral disc disease. Science. 1999;285:409–12.
- 36. Jim JJ, Noponen-Hietala N, Cheung KM, Ott J, Karppinen J, Sahraravand A, Luk KD, Yip SP, Sham PC, Song YQ, Leong JC, Cheah KS, Ala-Kokko L, Chan D. The TRP2 allele of COL9A2 is an agedependent risk factor for the development and severity of intervertebral disc degeneration. Spine (Phila Pa 1976). 2005;30:2735–42.
- 37. Paassilta P, Lohiniva J, Goring HH, Perala M, Raina SS, Karppinen J, Hakala M, Palm T, Kroger H, Kaitila I, Vanharanta H, Ott J, Ala-Kokko L. Identification of a novel common genetic risk factor for lumbar disk disease. JAMA. 2001;285:1843–9.
- 38. Pluijm SM, van Essen HW, Bravenboer N, Uitterlinden AG, Smit JH, Pols HA, Lips P. Collagen type I alpha1 Sp1 polymorphism, osteoporosis, and intervertebral disc degeneration in older men and women. Ann Rheum Dis. 2004;63:71–7.
- Tilkeridis C, Bei T, Garantziotis S, Stratakis CA. Association of a COL1A1 polymorphism with lumbar disc disease in young military recruits. J Med Genet. 2005;42, e44.
- 40. Karasugi T, Semba K, Hirose Y, Kelempisioti A, Nakajima M, Miyake A, Furuichi T, Kawaguchi Y, Mikami Y, Chiba K, Kamata M, Ozaki K, Takahashi A, Makela P, Karppinen J, Kimura T, Kubo T, Toyama Y, Yamamura K, Mannikko M, Mizuta H, Ikegawa S. Association of the tag SNPs in the human SKT gene (KIAA1217) with lumbar disc herniation. J Bone Miner Res. 2009;24:1537–43.
- 41. Kelempisioti A, Eskola PJ, Okuloff A, Karjalainen U, Takatalo J, Daavittila I, Niinimaki J, Sequeiros RB, Tervonen O, Solovieva S, Kao PY, Song YQ, Cheung KM, Chan D, Ala-Kokko L, Jarvelin MR, Karppinen J, Mannikko M. Genetic susceptibility of intervertebral

disc degeneration among young Finnish adults. BMC Med Genet. 2011;12:153.

- 42. Song YQ, Ho DW, Karppinen J, Kao PY, Fan BJ, Luk KD, Yip SP, Leong JC, Cheah KS, Sham P, Chan D, Cheung KM. Association between promoter –1607 polymorphism of MMP1 and lumbar disc disease in Southern Chinese. BMC Med Genet. 2008;9:38.
- 43. Dong DM, Yao M, Liu B, Sun CY, Jiang YQ, Wang YS. Association between the -1306C/T polymorphism of matrix metalloproteinase-2 gene and lumbar disc disease in Chinese young adults. Eur Spine J. 2007;16:1958–61.
- 44. Takahashi M, Haro H, Wakabayashi Y, Kawa-Uchi T, Komori H, Shinomiya K. The association of degeneration of the intervertebral disc with 5a/6a polymorphism in the promoter of the human matrix metalloproteinase-3 gene. J Bone Joint Surg (Br). 2001;83:491–5.
- 45. Sun ZM, Miao L, Zhang YG, Ming L. Association between the -1562 C/T polymorphism of matrix metalloproteinase-9 gene and lumbar disc disease in the young adult population in North China. Connect Tissue Res. 2009;50:181–5.
- 46. Hirose Y, Chiba K, Karasugi T, Nakajima M, Kawaguchi Y, Mikami Y, Furuichi T, Mio F, Miyake A, Miyamoto T, Ozaki K, Takahashi A, Mizuta H, Kubo T, Kimura T, Tanaka T, Toyama Y, Ikegawa S. A functional polymorphism in THBS2 that affects alternative splicing and MMP binding is associated with lumbardisc herniation. Am J Hum Genet. 2008;82:1122–9.
- 47. Williams FM, Bansal AT, van Meurs JB, Bell JT, Meulenbelt I, Suri P, Rivadeneira F, Sambrook PN, Hofman A, Bierma-Zeinstra S, Menni C, Kloppenburg M, Slagboom PE, Hunter DJ, Macgregor AJ, Uitterlinden AG, Spector TD. Novel genetic variants associated with lumbar disc degeneration in northern Europeans: a meta-analysis of 4600 subjects. Ann Rheum Dis. 2013;72:1141–8.
- 48. Sun ZM, Ling M, Huo Y, Chang Y, Li Y, Qin H, Yang G, Lucas R. Caspase 9 gene polymorphism and susceptibility to lumbar disc disease in the Han population in northern China. Connect Tissue Res 52(3):198–202.
- 49. Seki S, Kawaguchi Y, Chiba K, Mikami Y, Kizawa H, Oya T, Mio F, Mori M, Miyamoto Y, Masuda I, Tsunoda T, Kamata M, Kubo T, Toyama Y, Kimura T, Nakamura Y, Ikegawa S. A functional SNP in CILP, encoding cartilage intermediate layer protein, is associated with susceptibility to lumbar disc disease. Nat Genet 37(6):607–612.
- 50. Zhu GB, Jiang XR, Xia CL, Sun YJ, Zeng QS, Wu XM, Li XC. Association of FAS and FAS ligand polymorphisms with the susceptibility and severity of lumbar disc degeneration in Chinese Han population. Biomarkers 16(6):485–490.
- 51. Kim DH, Lee SH, Kim KT, Yu SD. Association of interleukin-1 receptor antagonist gene polymorphism with response to conservative treatment of lumbar herniated nucleus pulposus. Spine (Phila Pa 1976) 35(16):1527–1531.

- 52. Lin WP, Lin JH, Chen XW, Wu CY, Zhang LQ, Huang ZD, Lai JM. Interleukin-10 promoter polymorphisms associated with susceptibility to lumbar disc degeneration in a Chinese cohort. Genet Mol Res 10(3):1719–1727.
- 53. Ioannidis JP, Boffetta P, Little J, O'Brien TR, Uitterlinden AG, Vineis P, Balding DJ, Chokkalingam A, Dolan SM, Flanders WD, Higgins JP, Mccarthy MI, Mcdermott DH, Page GP, Rebbeck TR, Seminara D, Khoury MJ. Assessment of cumulative evidence on genetic associations: interim guidelines. Int J Epidemiol. 2008;37:120–32.
- 54. Fernandes I, Hampson G, Cahours X, Morin P, Coureau C, Couette S, Prie D, Biber J, Murer H, Friedlander G, Silve C. Abnormal sulfate metabolism in vitamin D-deficient rats. J Clin Invest. 1997;100: 2196–203.
- 55. Kales SN, Linos A, Chatzis C, Sai Y, Halla M, Nasioulas G, Christiani DC. The role of collagen IX tryptophan polymorphisms in symptomatic intervertebral disc disease in Southern European patients. Spine (Phila Pa 1976). 2004;29:1266–70.
- 56. Grant SF, Reid DM, Blake G, Herd R, Fogelman I, Ralston SH. Reduced bone density and osteoporosis associated with a polymorphic Sp1 binding site in the collagen type I alpha 1 gene. Nat Genet. 1996;14: 203–5.
- 57. Francis-West PH, Abdelfattah A, Chen P, Allen C, Parish J, Ladher R, Allen S, Macpherson S, Luyten FP, Archer CW. Mechanisms of GDF-5 action during skeletal development. Development. 1999;126: 1305–15.
- 58. Stefansson H, Rujescu D, Cichon S, Pietilainen OP, Ingason A, Steinberg S, Fossdal R, Sigurdsson E, Sigmundsson T, Buizer-Voskamp JE, Hansen T, Jakobsen KD, Muglia P, Francks C, Matthews PM, Gylfason A, Halldorsson BV, Gudbjartsson D, Thorgeirsson TE, Sigurdsson A, Jonasdottir A, Jonasdottir A, Bjornsson A, Mattiasdottir S, Blondal T, Haraldsson M, Magnusdottir BB, Giegling I, Moller HJ, Hartmann A, Shianna KV, Ge D, Need AC, Crombie C, Fraser G, Walker N, Lonnqvist J, Suvisaari J, Tuulio-Henriksson A, Paunio T, Toulopoulou T, Bramon E, Di Forti M, Murray R, Ruggeri M, Vassos E, Tosato S, Walshe M, Li T, Vasilescu C, Muhleisen TW, Wang AG, Ullum H, Djurovic S, Melle I, Olesen J, Kiemeney LA, Franke B, Group, Sabatti C, Freimer NB, Gulcher JR, Thorsteinsdottir U, Kong A, Andreassen OA, Ophoff RA, Georgi A, Rietschel M, Werge T, Petursson H, Goldstein DB, Nothen MM, Peltonen L, Collier DA, St Clair D, Stefansson K. Large recurrent microdeletions associated with schizophrenia. Nature. 2008;455:232-6.
- Nejentsev S, Walker N, Riches D, Egholm M, Todd JA. Rare variants of IFIH1, a gene implicated in antiviral responses, protect against type 1 diabetes. Science. 2009;324:387–9.
- 60. Bonnefond A, Clement N, Fawcett K, Yengo L, Vaillant E, Guillaume JL, Dechaume A, Payne F,

Roussel R, Czernichow S, Hercberg S, Hadjadj S, Balkau B, Marre M, Lantieri O, Langenberg C, Bouatia-Naji N, Meta-Analysis of, G, Insulin-Related Traits, C, Charpentier G, Vaxillaire M, Rocheleau G, Wareham NJ, Sladek R, McCarthy MI, Dina C, Barroso I, Jockers R, Froguel P. Rare MTNR1B variants impairing melatonin receptor 1B function contribute to type 2 diabetes. Nat Genet. 2012;44: 297–301.

- Sanger F, Nicklen S, Coulson AR. DNA sequencing with chain-terminating inhibitors. Proc Natl Acad Sci U S A. 1977;74:5463–7.
- Maxam AM, Gilbert W. A new method for sequencing DNA. Proc Natl Acad Sci U S A. 1977;74: 560–4.
- 63. Dempster EL, Pidsley R, Schalkwyk LC, Owens S, Georgiades A, Kane F, Kalidindi S, Picchioni M, Kravariti E, Toulopoulou T, Murray RM, Mill J. Disease-associated epigenetic changes in monozygotic twins discordant for schizophrenia and bipolar disorder. Hum Mol Genet. 2011;20:4786–96.
- 64. Meaney MJ. Epigenetics and the biological definition of gene x environment interactions. Child Dev. 2010;81:41–79.
- Nicholls RD. The impact of genomic imprinting for neurobehavioral and developmental disorders. J Clin Invest. 2000;105:413–8.
- 66. Rakyan VK, Down TA, Balding DJ, Beck S. Epigenome-wide association studies for common human diseases. Nat Rev Genet. 2011;12:529–41.
- Zhao B, Yu Q, Li H, Guo X, He X. Characterization of microRNA expression profiles in patients with intervertebral disc degeneration. Int J Mol Med. 2014;33:43–50.

- 68. Jones SW, Watkins G, Le Good N, Roberts S, Murphy CL, Brockbank SM, Needham MR, Read SJ, Newham P. The identification of differentially expressed microRNA in osteoarthritic tissue that modulate the production of TNF-alpha and MMP13. Osteoarthr Cartil. 2009;17:464–72.
- Chan WC, Sze KL, Samartzis D, Leung VY, Chan D. Structure and biology of the intervertebral disk in health and disease. Orthop Clin N Am. 2011; 42(447–64):vii.
- 70. Arc OC, Arc OC, Zeggini E, Panoutsopoulou K, Southam L, Rayner NW, Day-Williams AG, Lopes MC, Boraska V, Esko T, Evangelou E, Hoffman A, Houwing-Duistermaat JJ, Ingvarsson T, Jonsdottir I, Jonnson H, Kerkhof HJ, Kloppenburg M, Bos SD, Mangino M, Metrustry S, Slagboom PE, Thorleifsson G, Raine EV, Ratnayake M, Ricketts M, Beazley C, Blackburn H, Bumpstead S, Elliott KS, Hunt SE, Potter SC, Shin SY, Yadav VK, Zhai G, Sherburn K, Dixon K, Arden E, Aslam N, Battley PK, Carluke I, Doherty S, Gordon A, Joseph J, Keen R, Koller NC, Mitchell S, O'Neill F, Paling E, Reed MR, Rivadeneira F, Swift D, Walker K, Watkins B, Wheeler M, Birrell F, Ioannidis JP, Meulenbelt I, Metspalu A, Rai A, Salter D, Stefansson K, Stykarsdottir U, Uitterlinden AG, van Meurs JB, Chapman K, Deloukas P, Ollier WE, Wallis GA, Arden N, Carr A, Doherty M, Mccaskie A, Willkinson JM, Ralston SH, Valdes AM, Spector TD, Loughlin J. Identification of new susceptibility loci for osteoarthritis (arcOGEN): a genomewide association study. Lancet. 2012;380:815-23.
- Hegele RA. Genome-wide association studies of plasma lipids: have we reached the limit? Arterioscler Thromb Vasc Biol. 2010;30:2084–6.
Part II

Imaging

# Imaging of Degenerative Disk Disease

# Guillaume Bierry and Jean-Louis Dietemann

# 7.1 Introduction

Degenerative disk disease (DDD) results in alterations of both the intervertebral disk and vertebral end plates and has three common sequelae: disk herniation, stenosis, and instability [1]. The progressive dehydration of the nucleus pulposus secondary to degradation of proteoglycans and the development of clefts within the anulus fibrosus lead to loss of height of the intervertebral space and bulging of anulus outer fibers. Besides, cartilaginous end plate thinning and focal anular rupture occur, associated with alterations of subchondral bone properties. DDD is usually associated with apophyseal joint osteoarthritis which contributes to central, radicular, and/or foraminal stenosis. Instability secondary to the loss of normal spinal biomechanics may be identified on static or dynamic radiographs and manifest potentially as spondylolisthesis or retrolisthesis.

Plain radiographs are able to demonstrate disk space loss of height; however, CT is more accurate in the morphological evaluation of spinal bony structures, and MRI is the modality of choice for the evaluation of soft tissues including

Department of Radiology 2,

University Hospital of Strasbourg,

10 Avenue Molière, 67098 Strasbourg, France e-mail: guillaume.bierry@chru-strasbourg.fr; jean-louis.dietemann@chru-strasbourg.fr disks, ligaments, neural structures, and bone marrow. In postsurgical conditions with spinal hardware, myelography might be useful to detect nerve root compression. In rare cases, DDD may be difficult to differentiate from infectious diskitis or rheumatoid arthritis; PET-CT or biopsy under fluoroscopic or CT control may be required to assist in the diagnosis.

# 7.2 Radiographs

The major signs of DDD on radiographs are disk space narrowing, sclerosis of the vertebral end plates, and osteophytes.

In nondegenerative conditions, the disk progressively increases in height from T12/L1 to L4/ L5 with the L5/S1 level being generally equal in height to L4/L5 [2] (Fig. 7.1). Decreased disk height is correlated with dehydration on T2-weighted MRI [3].

Osteophytes are bony projections that develop along vertebral end plates, classically in the axial plane, and are mostly anterior and lateral in the lumbar and thoracic spine and anterior and posterior with uncovertebral osteoarthritis in the cervical spine (Figs. 7.2, 7.3, and 7.4). Osteophytes differ from traction spurs located 2–4 mm above or below the anterior vertebral body edge, indicative of instability [4].

As DDD progresses, a vacuum disk phenomenon which represents the accumulation of gas

G. Bierry (🖂) • J.-L. Dietemann

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_7



**Fig. 7.1** Focal degenerative disk disease at L5-S1 level. Lateral radiograph shows isolated height loss of L5-S1 disk with anterior osteophytes (*arrow*)

within disk can appear (Fig. 7.2). Intradiscal gas is identified in 20 % of elderly patients and is related to negative intradiscal pressure and may be accentuated during spinal extension and decreased during spinal flexion [5]. The vacuum phenomenon may be related to lower back pain in the morning and when standing up and may be influenced by changes in weather and barometric pressure (atmosphere depression) [6, 7]. Identification of a vacuum phenomenon may be difficult on conventional spin-echo MR sequences; however, gradient-echo MR appears more sensitive [8, 9].

As in peripheral skeleton osteoarthrosis, subchondral sclerosis of the end plates may be seen with advanced disk space loss (Fig. 7.4).

Facet arthrosis is present in 50 % of adults younger than 30, the most common level involved being L4-L5 [10, 11]. Facet joint osteophytes with foraminal stenosis are often seen on lateral radiographs (Fig. 7.5).



**Fig. 7.2** Degenerative disk disease at multiple levels. Lateral radiograph shows disks narrowing with osteo-phytes (*arrows*) and vacuum phenomenon (*arrowhead*)

Other plain radiographic signs include Schmorl nodes and disk calcification which is most frequent in the lower thoracic and upper lumbar spine [12]. Weakening of the end plate and subchondral bone related to osteochondrosis may predispose to cartilaginous Schmorl nodes, which appear as a round radiolucent lesion that could have various depths within the vertebral body. Intervertebral disk calcifications involving the anulus fibrosus are common in elderly patients in the lower thoracic spine (60 %) [13].



**Fig. 7.3** Lateral radiograph shows asymmetrical disk space narrowing and massive osteophytes with typical "parrot beak" aspect (*arrow*)

# 7.3 Computed Tomography

High-resolution multislice CT is more sensitive for demonstrating a vacuum phenomenon, disk calcification, subchondral osteosclerosis, and associated bony changes, particularly on sagittal, para-axial, and coronal reformations (Fig. 7.6). Herniations including diffuse bulging disk are clearly demonstrated on axial scans and sagittal reformations (Fig. 7.7).

Anterior and lateral osteophytes do not compress nerve roots in the lumbar spine; large lateral osteophytes may be associated with lumbar scoliosis (Fig. 7.6). Axial reformats demonstrate accurately facet joint osteoarthrosis with disk space narrowing, sclerosis, and osteophyte formation (Fig. 7.8).

Spondylosis deformans with degenerative lumbar scoliosis leads to transverse vertebral



**Fig. 7.4** Radiograph (AP view) shows L4/L5 disk space narrowing with osteophytes and subchondral sclerosis of end plates (*arrow*)

displacement, spondylolisthesis, retrolisthesis, and anterior and anterolateral protrusion of disk material and osteophytes (Fig. 7.9).

Disk calcifications of degenerative origin are mainly located within the anulus fibrosus. Lumbar intervertebral disk calcifications are noted in 50 % of elderly patients [14].

Schmorl nodes appear as a round radiolucent lesion, sometimes containing gas, with a rim of bony sclerosis on CT (Fig. 7.10).

# 7.4 Magnetic Resonance Imaging (MRI)

MRI is the modality of choice for the evaluation of the degenerative spine, as it allows an analysis of the disk, bone marrow, and facet changes as well as structures that might be injured secondary to degenerative changes such as nerve roots and muscles. Routine MRI of the spine includes sagittal spin-echo T1 and fast-spin-echo T2-weighted

**Fig. 7.5** Posterior articulation degenerative disease. Lateral radiograph shows posterior articulation arthrosis with osteophytes narrowing intervertebral foramen (*arrow*)

imaging. T2 (sagittal and/or coronal) sequences with fat saturation (STIR, fatsat) are useful for the visualization of bone marrow edema. Axial T1- and T2-weighted images are useful for visualization of nerve root compression in patients with radicular pain. The use of intravenous gadolinium may demonstrate enhancement related to neovascularization involving the intervertebral disk and/or subchondral bone.

On T1- and T2-weighted MRI, the signal of the normal intervertebral disk is, respectively, lower and higher than that of the vertebral body; the high signal is related to bounded water by the proteoglycans of the nucleus pulposus. According to Pfirrmann et al. [15], five grades can be described for lumbar disk degeneration on T2-weighted MRI (Fig. 7.11). In normal young patients, the high signal on T2-weighted images



**Fig. 7.6** Multilevel degenerative disk disease. CT sagittal reformat shows multiple disk space narrowing with intradiscal gas (*arrowheads*) and osteophytes (*arrow*)

appears homogeneous in the central area of the disk; the peripheral anulus may demonstrate a low signal (grade 1) (Figs. 7.11, 7.12, 7.13, and 7.14). Loss of signal intensity in the nucleus pulposus on T2-weighted MRI closely correlates with disk dehydration related to alteration of proteoglycans. During the second decade of life, a horizontal linear hypointense band appears within the nucleus pulposus on T2-weighted images as a result of the development of collagen fibers within the nucleus pulposus (grade 2).



**Fig. 7.7** Disk herniations. CT sagittal reformat shows posterior disk bulge on L3/L4 level and larger herniation on L4/L5 level



**Fig. 7.8** Posterior articulation degenerative disease. Axial CT reformat shows sclerosis of articular facets with osteophytes (*arrow*)



**Fig. 7.9** Spondylolisthesis. Sagittal CT reformat shows L4/L5 degenerative spondylolisthesis with major disk space narrowing and intradiscal gas (*arrow*)

Later, a diffuse signal loss is noted on T2-weighted images and is associated with mild narrowing of the intervertebral space (grade 3) (Figs. 7.11, 7.12, 7.13, and 7.14). At this stage, posterior radial tears may be detected as an area of high signal intensity on T2-weighted images (sometimes described as a HIZ or high-intensity zone lesion) in the posterior and peripheral anulus; enhancement is possible after intravenous administration of gadolinium (Fig. 7.13). A black disk with significant narrowing of the space (grade 4) or with a collapsed disk space (grade 5) corresponds to severe disk degeneration.

Neovascularization is often observed within the degenerative intervertebral disk and leads to a band-like enhancement parallel to the end plates or, less commonly, in the center of the disk. Such findings are associated with local back pain [16] (Fig. 7.14).



**Fig. 7.10** Schmorl node. Sagittal CT reformat shows hypodense lesion within the end plate with peripheral sclerosis (*arrow*)

Intradiscal gas appears hypointense on all sequences, and the vacuum phenomenon is therefore more effectively detected on radiographs and CT. Intradiscal calcifications can have various presentations on MRI: they may appear hypointense on T1- and T2-weighted images or hyperintense on T1-weighted images due to the presence of fatty marrow within ossification of the disk [17, 18].

End plate bone marrow is frequently abnormal on MRI in the setting of DDD. According to Modic et al. [19, 20], two types of signal intensity changes involving the bone marrow of the adjacent vertebral body may be associated with DDD. Type 1 changes are visualized as low signal intensity on T1-weighted images and high signal intensity on T2-weighted images with enhancement after administration of gadolinium



**Fig. 7.11** Different grades of disk degeneration on sagittal MR T2w images with disk bulge at L4/L5 level and herniation at L5/S1 level

(Fig. 7.15). This results from the replacement of normal bone marrow with fibrovascular marrow with an increase in free water and hypervascularity responsible for enhancement after gadolinium injection [19]. Modic type 1 changes are noted in 4 % of patients with back pain and are closely correlated with painful disk derangement [15, 21, 22]. A positive pain provocation test is not clearly correlated with Modic type 1 end plate changes [21, 22]. Similar changes are observed after surgery or percutaneous treatment of a disk herniation.

Modic type 1 end plate changes may at times simulate infectious diskitis and should be correlated with patients' symptoms, clinical history, and laboratory data. In difficult cases, differentiation of degenerative and infectious end plate abnormalities may require a biopsy, MRI follow-



**Fig.7.12** Isolated degeneration of L4/L5 disk on sagittal T2w MR image

**Fig. 7.13** Multilevel disk degeneration with L2/L3 posterior hyperintense zone (HIZ) on sagittal T2w MR image (*arrow*). L3/L4 level demonstrates posterior disk bulge

up, or fluorodeoxyglucose (FDG) positron emission tomography (PET) [23]. Modic type 1 end plate changes may also simulate erosive diskitis associated with rheumatoid arthritis, gout, or chronic hemodialysis [24–27].

Modic type 2 end plate changes are visualized as high-signal-intensity lesions on T1- and T2-weighted images without enhancement; type 2 changes represent fatty marrow and are observed in 16 % of patients presenting with back pain [28] (Figs. 7.16 and 7.17).

Modic type 1 changes represent a dynamic process that often converts to Modic type 2 changes over time (Fig. 7.13). Modic type 2 end plate changes rarely progress [29]. Conversion of Modic type 1 changes to Modic type 2 changes is a very slow process (only 50 % conversion over an observation interval ranging from 1 to 6 years);

regression of back pain is noted in two thirds of patients that fully convert from Modic type 1 changes to Modic type 2 changes; associated factors (apophyseal joint osteoarthritis, instability) may explain the absence or partial regression of symptoms [29]. Rapid conversion from Modic 1 to 2 changes is seen after a lumbar posterior arthrodesis [30]. Reverse transformation of Modic type 2 changes to Modic type 1 changes occurs rarely and is probably related to superimposed disease [31, 32].

Modic type 3 changes represent sclerosis of the end plates at the end stage of DDD and appear as a low signal intensity on T1- and T2-weighted images without enhancement (Fig. 7.18).

The signal appearance of recent cartilaginous Schmorl node formation is similar to that of the corresponding intervertebral disk on T1-weighted images and appears slightly hyperintense on T2-weighted images; enhancement is possible



**Fig. 7.14** Canal stenosis (*arrows*) secondary to posterior disk bulge and posterior facet hypertrophy on sagittal T2w MR image (**a**); degenerative disk demonstrates linear

enhancement on sagittal fat-saturated gadoliniumenhanced T1w MR image (**b**) (*arrow*)

and may remain in the chronic stages. During acute and subacute stages, edema and inflammatory changes involving the surrounding bone marrow are noted as a low signal intensity on T1-weighted images, high signal intensity on T2-weighted images, and postcontrast enhancement and may be correlated with acute or subacute back pain (Fig. 7.19).

Reactive edematous changes can be observed in osteoarthritis involving apophyseal joints (facet joint effusion, inflammation involving articular processes) especially on axial and coronal STIR or T2-weighted images with fat saturation. Facet joint edema (seen in 14 % of patients with low back pain) and intra-articular fluid can also be seen [10, 33].

MRI is the modality of choice for the evaluation of ligament morphology. DDD is associated with hypertrophy and/or degeneration of the ligamentum flavum that, associated with discal bulging and osseous changes, might be responsible for central or foraminal stenosis (Figs. 7.11, 7.13, 7.14, 7.19, 7.20, 7.21, 7.22, and 7.23). Axial T1or T2-weighted images can clearly depict ligament hypertrophy and neural compression. In addition, modifications in interspinous changes such as bursitis (Baastrup disease) are easily identified on sagittal T2 or STIR images [33–36] (Fig. 7.24).

Paraspinal muscles can be involved in DDD: progressive atrophy can be observed due to reduced activity of the patient. Acute myositis can be seen in cases of nerve root injuries secondary to spinal instability. While axial T1 images are best suited for the evaluation of muscular atrophy, muscular inflammation is best explored on coronal STIR images (Fig. 7.25) [37–39].



**Fig.7.15** Modic type 1 L4-L5 end plate changes, hypointense on T1 (**a**) and hyperintense on T2 (**b**) sagittal MR scans. Signal abnormalities are located on the portion of the end plates next to the disk

# 7.5 Postoperative Imaging

In postoperative follow-up, imaging is used to check the integrity of the neurologic structures (foramina, spinal cord, nerve roots) and adjacent elements (vessels, muscles, soft tissues) [40].

If instrumentation is present, CT is the best modality to evaluate hardware-related complications. Correct position and integrity of implants, screws, and rods as well as progress of bone fusion can be assessed (Figs. 7.26 and 7.27).

Complications involving nerves and soft tissue are best visualized by MRI. The most sensitive finding is a collection along the surgical bed with intense rim enhancement. The identification of disk hernia recurrence requires the injection of contrast media in order to discriminate recurrence versus epidural fibrosis: fibrosis enhances, while the absence of enhancement confirms hernia recurrence (Figs. 7.28 and 7.29). Muscles can be injured during spinal surgery, either directly during rod and screw implantation or secondary to posterior nerve root sectioning. Involved muscles present with diffuse high signal imaging. The extent of injury is best assessed using coronal fat saturation T2-weighted images (Fig. 7.30).

# 7.6 Perspectives in Spinal Imaging

Nerves can be morphologically evaluated using routine MR protocols. More specific sequences might nevertheless be used such as MR neurography that allows for more precise evaluation of nerve entrapment.

Apparent diffusion coefficient (ADC) has been widely evaluated in nondegenerative and degenerative disk disease. Most studies report



**Fig. 7.16** L4-L5 and L5-S1 Modic type 2 end plate changes hyperintense on T1 (**a**) and hyperintense on T2 (**b**) sagittal MR images

that ADC tends to decrease with disk degeneration [41, 42]. Sequencing with ultrashort TE (UTE) time allows the identification of the different components (calcified and uncalcified) of the vertebral end plate and might enable the identification of early degenerative changes [43, 44]. Nevertheless, the role of diffusion and UTE imaging in routine practice remains to be defined. Similar sensitivity is being investigated with glycosaminoglycan-dependent chemical exchange saturation transfer (gagCEST) imaging, 3-T MRI for quantification of glycosaminoglycan (GAG) content in the intervertebral disk [45], and in vivo magnetic resonance spectroscopy [46].

Besides static MR examination, dynamic evaluation with upright positional MRI of the lumbar spine may enhance the presence of a mobile disk herniation or vertebral translation. Supine MRI with the legs straightened seems to be comparable to vertical MRI for evaluation of spinal canal morphology [47, 48]. In addition, imaging of lumbar spine in the prone position with axial loading may reveal significant changes in canal dimension as compared to non-loaded examinations [49].

EOS imaging is increasingly used in the evaluation of spinal disorders, notably in scoliosisrelated degenerative changes. EOS imaging is based on the simultaneous detection of X-rays by two orthogonal innovative Nobel-awarded detectors, resulting in the acquisition of biplanar images with the possibility of generating 3D reformat images [50]. EOS imaging enables the production of high-quality full-body images in the weight-bearing position, using low irradiation. The potential use of EOS imaging includes scoliosis evaluation in children or in the evaluation of spinal balance in elderly with degenerative disk disease [51].



**Fig. 7.17** Modic type 1 L5-S1 end plate changes with partial conversion Modic type 2. End plates appear mainly hypointense on T1 (*black arrow* on **a**) and hyperintense on

T2 (**b**) sagittal MR images, but area of high signal appears next to the disk on T1w images (*white arrow* in **a**)



Fig. 7.18 L5-S1 type III end plate changes hypointense on T1 (a) and T2 (b) in keeping with end plate sclerosing



**Fig. 7.19** Cartilaginous Schmorl node (*arrow*) with hyperintensity on T2 areas (*arrow*) involving the surrounding bone marrow indicating edema related to inflammatory changes. Posterior disk bulge is present (*arrowhead*)



**Fig. 7.21** Hemorrhagic facet joint cyst with canal mass effect. Sagittal T1w MR image shows hyperintense lobulated mass before facet joints (*arrow*)





**Fig. 7.20** Degenerative changes involving L4-L5 facet joints. Hyperintense signal consistent with facet joint effusion (*arrow*) and hypertrophy of ligamentum flavum (*arrowhead*) are seen on axial T2w MR image

**Fig. 7.22** Canal stenosis secondary to degenerative changes. Image T2w MR image shows canal narrowing due to posterior disk bulge (*arrow*) and facet joint and *ligamentum flavum* hypertrophy (*arrowhead*)



**Fig. 7.23** Intervertebral foramen narrowing. Sagittal T1w MR images show mild reduction of fat around the nerve root (*arrow*) secondary to posterior disk bulge and facet hypertrophy



**Fig. 7.24** Baastrup disease involving the L4-L5 interspinous space. Fluid within the interspinous space appears hyperintense on sagittal fat-saturated T2 MR images (*arrow*). Note severe L5-S1 disk space narrowing



**Fig. 7.25** L4-L5 facet joint osteoarthrosis associated with paraspinal muscle atrophy with fatty replacement (*arrows*) visualized on axial T1 MR image. Severe canal stenosis secondary to posterior disk bulge and facet hypertrophy is present



Fig. 7.26 Aseptic loosening of S1 screw after L5/S1 arthrodesis. CT sagittal reformat shows lucency around screw (*arrowheads*)



**Fig. 7.28** Persistent radicular pain after diskal surgery. Axial gadolinium-enhanced T1w MR image shows enhancement of periradicular space consistent with fibrosis (*arrow*) as well as moderate nerve root enhancement (*arrowhead*)



**Fig. 7.27** Radicular pain after lumbar arthrodesis. CT oblique reformat demonstrates low and medial position of S1 screw (*arrow*) that displaces S1 root (*arrowheads*)



**Fig. 7.29** Persistent radicular pain after diskal surgery. Axial gadolinium-enhanced T1w MR image shows a nonenhancing mass within foramina consistent with hernia recurrence (*arrow*)



**Fig.7.30** Back pain after diskal surgery. Axial gadolinium-enhanced T1w MR image (**a**) and coronal fatsat T2w image (**b**) demonstrate multifidus edema consistent with postoperative denervation (*arrow*)

# References

- Ikegawa S. The genetics of common degenerative skeletal disorders: osteoarthritis and degenerative disc disease. Annu Rev Genomics Hum Genet. 2013;14: 245–56.
- Roberts N, Gratin C, Whitehouse G. MRI analysis of lumbar intervertebral disc height in young and older populations. J Magn Reson Imaging. 2005;7: 880–6.
- Frobin W, Brinckmann P, Kramer M, Hartwig E. Height of lumbar discs measured from radiographs compared with degeneration and height classified from MR images. Eur Radiol. 2001;11:263–9.
- 4. McCulloch JA, Transfeldt EE. Macnab's backache. Baltimore: Williams & Wilkins; 1997.
- Goobar JE, Pate D, Resnick D, Sartoris DJ. Radiography of the hyperextended lumbar spine: an effective technique for the demonstration of discal vacuum phenomena. Can Assoc Radiol J. 1987;38: 271–4.
- Kasai Y, Takegami K, Uchida A. Change of barometric pressure influences low back pain in patients with vacuum phenomenon within lumbar intervertebral disc. J Spinal Disord Tech. 2002;15:290–3.
- Morishita K, Kasai Y, Uchida A. Clinical symptoms of patients with intervertebral vacuum phenomenon. Neurologist. 2008;14:37–9.
- Berns DH, Ross JS, Kormos D, Modic MT. The spinal vacuum phenomenon: evaluation by gradient echo MR imaging. J Comput Assist Tomogr. 1991;15: 233–6.
- Grenier N, Grossman RI, Schiebler ML, Yeager BA, Goldberg HI, Kressel HY. Degenerative lumbar disk disease: pitfalls and usefulness of MR imaging in

detection of vacuum phenomenon. Radiology. 1987; 164:861–5.

- Eubanks JD, Lee MJ, Cassinelli E, Ahn NU. Prevalence of lumbar facet arthrosis and its relationship to age, sex, and race: an anatomic study of cadaveric specimens. Spine. 2007;32:2058–62.
- Friedrich KM, Nemec S, Peloschek P, Pinker K, Weber M, Trattnig S. The prevalence of lumbar facet joint edema in patients with low back pain. Skelet Radiol. 2007;36:755–60.
- 12. Pfirrmann CW, Resnick D. Schmorl nodes of the thoracic and lumbar spine: radiographic-pathologic study of prevalence, characterization, and correlation with degenerative changes of 1,650 spinal levels in 100 cadavers. Radiology. 2001;219:368–74.
- Chanchairujira K, Chung CB, Kim JY, et al. Intervertebral disk calcification of the spine in an elderly population: radiographic prevalence, location, and distribution and correlation with spinal degeneration. Radiology. 2004;230:499–503.
- Cheng XG, Brys P, Nijs J, et al. Radiological prevalence of lumbar intervertebral disc calcification in the elderly: an autopsy study. Skelet Radiol. 1996;25: 231–5.
- Pfirrmann CW, Metzdorf A, Zanetti M, Hodler J, Boos N. Magnetic resonance classification of lumbar intervertebral disc degeneration. Spine. 2001;26: 1873–8.
- Stabler A, Weiss M, Scheidler J, Krodel A, Seiderer M, Reiser M. Degenerative disk vascularization on MRI: correlation with clinical and histopathologic findings. Skelet Radiol. 1996;25:119–26.
- Malghem J, Lecouvet FE, Francois R, et al. High signal intensity of intervertebral calcified disks on T1-weighted MR images resulting from fat content. Skelet Radiol. 2005;34:80–6.

- Bangert BA, Modic MT, Ross JS, et al. Hyperintense disks on T1-weighted MR images: correlation with calcification. Radiology. 1995;195:437–43.
- Modic MT, Masaryk TJ, Ross JS, Carter JR. Imaging of degenerative disk disease. Radiology. 1988;168: 177–86.
- Modic MT, Ross JS. Lumbar degenerative disk disease. Radiology. 2007;245:43–61.
- Sandhu HS, Sanchez-Caso LP, Parvataneni HK, Cammisa Jr FP, Girardi FP, Ghelman B. Association between findings of provocative discography and vertebral endplate signal changes as seen on MRI. J Spinal Disord. 2000;13:438–43.
- Weishaupt D, Zanetti M, Hodler J, et al. Painful lumbar disk derangement: relevance of endplate abnormalities at MR imaging. Radiology. 2001;218: 420–7.
- 23. Stumpe KD, Zanetti M, Weishaupt D, Hodler J, Boos N, Von Schulthess GK. FDG positron emission tomography for differentiation of degenerative and infectious endplate abnormalities in the lumbar spine detected on MR imaging. AJR Am J Roentgenol. 2002;179:1151–7.
- 24. Kawaguchi Y, Matsuno H, Kanamori M, Ishihara H, Ohmori K, Kimura T. Radiologic findings of the lumbar spine in patients with rheumatoid arthritis, and a review of pathologic mechanisms. J Spinal Disord Tech. 2003;16:38–43.
- Khalfallah M, Faure A, Hamel O, et al. Dialysisassociated spondyloarthropathy. Case report and literature review. Neurochirurgie. 2005;51:165–72.
- Lagier R, Mac GW. Spondylodiscal erosions due to gout: anatomico-radiological study of a case. Ann Rheum Dis. 1983;42:350–3.
- Runge M, Bui P, Bonneville JF. Pseudospondylodiscitis in chronic hemodialysis. Apropos of 2 cases and review of the literature. J Radiol. 1987;68:511–8.
- Modic MT, Steinberg PM, Ross JS, Masaryk TJ, Carter JR. Degenerative disk disease: assessment of changes in vertebral body marrow with MR imaging. Radiology. 1988;166:193–9.
- Mitra D, Cassar-Pullicino VN, McCall IW. Longitudinal study of vertebral type-1 end-plate changes on MR of the lumbar spine. Eur Radiol. 2004;14:1574–81.
- Vital JM, Gille O, Pointillart V, et al. Course of Modic 1 six months after lumbar posterior osteosynthesis. Spine. 2003;28:715–20. discussion 721.
- 31. Marshman LA, Trewhella M, Friesem T, Bhatia CK, Krishna M. Reverse transformation of Modic type 2 changes to Modic type 1 changes during sustained chronic low-back pain severity. Report of two cases and review of the literature. J Neurosurg Spine. 2007;6:152–5.
- 32. Modic MT. Modic type 1 and type 2 changes. J Neurosurg Spine. 2007;6:150–1. discussion 151.
- 33. Lakadamyali H, Tarhan NC, Ergun T, Cakir B, Agildere AM. STIR sequence for depiction of degenerative changes in posterior stabilizing elements in

patients with lower back pain. AJR Am J Roentgenol. 2008;191:973–9.

- 34. D'Aprile P, Tarantino A, Jinkins JR, Brindicci D. The value of fat saturation sequences and contrast medium administration in MRI of degenerative disease of the posterior/perispinal elements of the lumbosacral spine. Eur Radiol. 2007;17:523–31.
- Doyle AJ, Merrilees M. Synovial cysts of the lumbar facet joints in a symptomatic population: prevalence on magnetic resonance imaging. Spine. 2004;29: 874–8.
- 36. Maes R, Morrison WB, Parker L, Schweitzer ME, Carrino JA. Lumbar interspinous bursitis (Baastrup disease) in a symptomatic population: prevalence on magnetic resonance imaging. Spine. 2008;33: E211–5.
- Kader DF, Wardlaw D, Smith FW. Correlation between the MRI changes in the lumbar multifidus muscles and leg pain. Clin Radiol. 2000;55: 145–9.
- Bierry G, Kremer S, Kellner F, Abu Eid M, Bogorin A, Dietemann JL. Disorders of paravertebral lumbar muscles: from pathology to cross-sectional imaging. Skelet Radiol. 2008;37:967–77.
- 39. Jinkins JR. Acquired degenerative changes of the intervertebral segments at and suprajacent to the lumbosacral junction. A radioanatomic analysis of the nondiscal structures of the spinal column and perispinal soft tissues. Eur J Radiol. 2004;50:134–58.
- Thakkar RS, Malloy JP, Thakkar SC, Carrino JA, Khanna AJ. Imaging the postoperative spine. Radiol Clin N Am. 2012;50:731–47.
- Antoniou J, Dmers C, Beaudouin G. Apparent diffusion coefficient of intervertebral discs related to matrix composition and integrity. Magn Reson Imaging. 2004;22:963–72.
- Kealey S, Aho T, Delong D, Barboriak D, Provenzale J, Eastwood J. Assessment of apparent diffusion coefficient in normal and degenerated intervertebral lumbar disks: initial experience. Radiology. 2005;235: 569–74.
- 43. Gatehouse PD, He T, Hughes SP, Bydder GM. MR imaging of degenerative disc disease in the lumbar spine with ultrashort TE pulse sequences. MAGMA. 2004;16:160–6.
- 44. Bae WC, Statum S, Zhang Z, et al. Morphology of the cartilaginous endplates in human intervertebral disks with ultrashort echo time MR imaging. Radiology. 2013;266:564–74.
- 45. Haneder S, Apprich S, Schmitt B, et al. Assessment of glycosaminoglycan content in intervertebral discs using chemical exchange saturation transfer at 3.0 Tesla: preliminary results in patients with low-back pain. Eur Radiol. 2013;23:861–8.
- 46. Zuo J, Joseph G, Li X, et al. In vivo intervertebral disc characterization using magnetic resonance spectroscopy and T1ρ imaging: association with discography and Oswestry Disability Index and Short Form-36 Health Survey. Spine. 2012;37:214–21.

- Alyas F, Connell D, Saifuddin A. Upright positional MRI of the lumbar spine. Clin Radiol. 2008;63: 1035–48.
- 48. Madsen R, Jensen TS, Pope M, Sorensen JS, Bendix T. The effect of body position and axial load on spinal canal morphology: an MRI study of central spinal stenosis. Spine. 2008;33:61–7.
- 49. Kinder A, Filho F, Ribeiro E, et al. Magnetic resonance imaging of the lumbar spine with axial loading: a review of 120 cases. Eur J Radiol. 2012;81:561–4.
- Illes T, Somoskeoy S. The EOS imaging system and its uses in daily orthopaedic practice. Int Orthop. 2012;36:1325–31.
- Ilharreborde B, Steffen JS, Nectoux E, et al. Angle measurement reproducibility using EOS threedimensional reconstructions in adolescent idiopathic scoliosis treated by posterior instrumentation. Spine. 2011;36:E1306–13.

# Diskography

Richard Derby, Lee R. Wolfer, and Milton H. Landers

# 8.1 Introduction

Until the introduction of diskography by Lindblom in the 1940s, oil-contrast myelography was utilized to diagnose disk herniations [1-4]. During the era of the "herniated disk" in the 1930s [5], both axial pain and referred pain were thought to be due to the disk compressing neural elements. Myelography was limited in that it only visualized the thecal sac and dural root sleeves. Diskography allowed visualization of the disk itself, including internal morphology and lateral protrusions which were missed on myelography. Diskography was initially utilized to diagnose disk herniations prior to surgery in patients with radicular pain [1-3]. Interestingly however, most of the disks examined by diskography exhibited annular disruption, but no frank herniation or protrusion. More importantly, upon injection of

M.H. Landers, DO, PhD Department of Anesthesiology, University of Kansas School of Medicine, Wichita, KS USA

Kansas Spine Institute, 825 N Hillside, Wichita, KS 67214, USA e-mail: mlandersdophd@gmail.com contrast media into these internally disrupted disks, some patients experienced reproduction of their familiar pain [6, 7]. These observations led surgeons to use provocation diskography not only to reveal structural abnormalities but also to identify and treat painfully internally disrupted disks. Diskography became a sophisticated extension of the physical exam, a means of "palpating" the disk to elicit pain [8]. Previously, diskography was only used as a presurgical planning test; however, with the introduction of new "regenerative medicine" techniques such as platelet-rich plasma, there is a renewed interest in understanding and treating the disk as a pain generator.

Over the most recent decades, histochemical and anatomical studies have provided evidence that the disk is an innervated structure capable of transmitting pain. There is no doubt that the disk can be a source of pain. Typically, in a normal disk, innervation is limited to the outer annulus; however, we know that pathologically painful disks (based on positive diskogram pre-fusion) show neo-innervation to the inner annulus as well as nucleus pulposus [9-11]. It is believed that injection of contrast dye into the nucleus stimulates nerve endings [12] via two mechanisms: a chemical stimulus from contact between contrast dye and sensitized tissues and a mechanical stimulus resulting from fluid-distending stress. What is controversial is not whether the disk is a source of pain, but whether disk pain can be reliably diagnosed?

R. Derby, MD (⊠) • L.R. Wolfer, MD, MS Department of Anesthesiology, Spinal Diagnostics and Treatment Center, 901 Campus Drive, Suite 310, Daly City, CA 94015, USA e-mail: rderby@spinaldiagnostics.com; drwolfer@gmail.com

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_8

The validity and accuracy of diskography have been challenged, particularly in the last 10 years, with reports of unacceptably high false-positive rates in asymptomatic subjects, particularly in patients with chronic pain or evidence of psychologic pathology. However, these negative studies have been refuted after close review and a performance of a meta-analysis of false-positive rates of lumbar diskography. Combining all the major studies on diskography in asymptomatic subjects, a false-positive rate of less than 10 % can be achieved if the diskographer adheres to strict operational standards and interpretation criteria [13]. Although MRI and CT are the most commonly used advanced imaging modalities for low back pain, diskography still occupies a critical place in the diagnostic algorithm. We know that abnormal disk morphology alone is not diagnostic of diskogenic pain, as many individuals with abnormal CT scans or MRIs are asymptomatic of low back pain [14, 15]; however, MRI cannot distinguish between a painful and painless disk.

In contemporary practice, the criterion standard for diagnosis of a painful internally disrupted disk by provocation diskography is pain  $\geq$ 7/10, pressure <50 psi a.o. (above opening pressure), concordant pain,  $\geq$ grade 3 annular tear, volume  $\leq$ 3.5 ml, and the presence of a negative control disk [16].

In this chapter, the indications for diskography, technical considerations, as well as complications of lumbar diskography are discussed. Procedural descriptions have also been extensively described in our prior publications [17, 18]. The technical section of the chapter is followed by a brief updated literature review in regard to the role of diskography as a diagnostic test (versus MRI and CT), the false-positive controversy, predictive value, as well as an update on analgesic diskography.

# 8.2 Indications and Contraindications

Diskography is not an initial screening examination. Disk stimulation follows failed conservative treatment modalities and is only used when other less invasive diagnostic tests are inconclusive. Diskography is invasive, and irreversible surgical

procedures may be chosen based on the results. The principal indication for provocation diskography is to determine whether or not a disk is pathologically painful and to determine the extent of annular or end plate disruption. Results of diskography may then used to guide the surgeon or spine interventionalist. Pertinent clinical information is then used to establish a diagnosis of diskogenic pain and consideration of targeted disk therapies. Only diskography followed by CT scanning can define the internal anatomy of the disk [19]. Postdiskography CT scanning is also particularly useful in post-diskectomy patients with suspected residual or recurrent disk herniations. Diskography is useful in problematic cases unresolved by MRI or myelography and in patients for whom surgery is contemplated [20]. Diskography can offer a potential solution to the diagnostic dilemma concerning which patients to treat surgically and at what segmental level. When a single disk is found to be symptomatic in the presence of adjacent asymptomatic disks, focused intradiscal or surgical therapy can be entertained. A no less important application is to identify asymptomatic disks which do not need intervention. Patients with symptomatic or abnormal disks at multiple levels  $(\geq 3)$  constitute a greater surgical challenge.

If the patient is not a surgical candidate, diskography is useful to provide a diagnosis that can be used to bring the workup to closure and direct the patient to non-interventional pain management treatment options. The position statement of the North American Spine Society on diskography is as follows [19]:

Diskography is indicated for the evaluation of patients with unremitting spinal pain, with or without extremity pain, of greater than four months' duration, when the pain has been unresponsive to all appropriate methods of conservative therapy. Before diskography, the patient should have undergone investigation with other modalities which have failed to explain the source of pain; such modalities should include, but not be limited to, either computed tomography (CT) scanning, magnetic resonance imaging (MRI) scanning and/ or myelography.

#### Inclusion criteria:

- 1. Failed conservative treatment for low back pain of probable spinal origin.
- 2. Pain has been ongoing for greater than 3 months.

- Symptoms are clinically consistent with disk pain. Ideally the zygapophyseal joints and sacroiliac joints have been ruled out as pain generators using appropriate dual diagnostic blocks with local anesthetic.
- 4. Symptoms are severe enough to consider surgery or percutaneous interventions.
- 5. Surgery is planned and the surgeon desires an assessment of the adjacent disk levels
- 6. The patient is capable of understanding the nature of the technique and can participate in the subjective interpretation.
- Both the patient and physician have a need to know of the source of pain in order to guide further treatments.

Contraindications:

- 1. Unable or unwilling to consent to the procedure
- Inability to assess patient response during the procedure
- 3. Inability of patient to cooperate
- 4. Known localized or systemic infection
- 5. Pregnancy
- 6. Anticoagulants or bleeding diathesis

Relative contraindications:

- 1. Allergy to contrast medium, antibiotics, or local anesthetics
- 2. Significant psychological overlay
- Any other condition, medical, congenital, postsurgical, anatomical, or psychological which would increase the risk of the performance of the procedure to an acceptable level

# 8.3 Preprocedure Evaluation

Prior to diskography, one should obtain a complete history and perform a physical examination to reveal any procedural contraindications. Red flags such as fever, night pain, history of malignancy, or unexplained weight loss should alert the physician to an alternate diagnosis. Prior to diskography, obtain informed consent from the patient. The purpose and nature of the procedure; its risks, benefits, alternatives, and complications; and what to expect post-procedure are explained. Patients are instructed in the use of a 0–10 pain scale. Specifically, the patient should understand that diskography is commonly painful and that they will need to describe the location, intensity, and concordance of any provoked pain in respect to their ongoing complaints. A trained observer can independently monitor patient pain responses during the procedure. Some diskographers have their patients fill out a brief psychometric test such as the Distress and Risk Assessment Method (DRAM) to assess if the patient has a normal, at-risk, distressed depressive, or distressed somatic profile [21].

In patients with a history of allergy to nonionic water-soluble contrast media (iohexol or iopamidol) or other related agents, the risks versus benefits of the procedure must be weighed and discussed with the patient. For patients with iodine allergies, pretreat patients with corticosteroids and H1 and H2 blockers prior to the procedure. If the risk of allergic reaction to contrast is significant, use saline instead of contrast or add a very small volume of gadolinium to the saline and obtain an immediate post-procedure MRI [22, 23].

In all cases of lumbar diskography, an MRI or CT scan should be reviewed. The majority of diskographers select test levels according to the appearance of the MRI T2-weighted images. Most test disks with decreased T2-weighted signal intensity; an adjacent, less degenerated disk is usually selected as a control. Rarely is it necessary to inject greater than three levels.

#### 8.4 Patient Preparation

#### 8.4.1 Antibiotic Prophylaxis

Intravenous access is standard. Diskitis is the most common serious (albeit rare) complication. Prophylactic antibiotic (cefazolin 1 g, gentamicin 80 mg, clindamycin 900 mg, or ciprofloxacin 400 mg) is administered prior to the procedure and within 30 min of needle insertion. Sheep studies confirmed optimal antibiotic levels in the annulus 30 min post-IV administration; no antibiotics were present at 60 min [24, 25]. Post-procedure, aminoglycosides are not required for prophylaxis [26]. Along with IV antibiotics, many diskographers mix intradiscal antibiotics

with the contrast dye (between 1 and 6 mg per ml of cefazolin or an equivalent dose of another antibiotic) [27–30]. Klessig et al. [30] reported that cefazolin and gentamicin 1 mg/cc and clindamycin 7.5 mg/cc exceed the minimum inhibitory concentrations (MICs) for the three most common organisms causing diskitis: *Escherichia coli, Staphylococcus aureus*, and *Staphylococcus epidermidis*. All procedures should be performed under sterile conditions with double gloves.

# 8.4.2 Sedation

As a provocative test, diskography is at best uncomfortable and at worst very painful. How much and which drug to use for preoperative sedation varies according to the diskographer's skill and training. Intravenous sedation should be titrated to maintain patient comfort during needle insertion while minimizing masking of the evoked pain response. The patient must be awake and conversant during disk stimulation. Doses of midazolam between 2.0 and 5.0 mg provide effective anxiolysis and sedation during diskography but cause retrograde amnesia. Many spine interventionalists, particularly those with an anesthesia background, use propofol (an ultrashort-acting hypnotic) in intermittent boluses of 10-30 mg. Propofol causes rapid sedation and amnesia during needle insertion, but because of its short half-life, the patient will be awake when during disk stimulation.

Some diskographers believe that opioids should not be utilized prior to or during diskography [31–35]. They assert that diskography is a provocational test; therefore, pain intensity needs to be compared and quantified in relation to the patient's usual pain intensity, and opioid analgesics could decrease the pain response and increase the chance of a false-negative response. Alternatively, others [36] believe that a small dose of analgesics (meperidine 50 mg, fentanyl 50 mcg, or morphine 5 mg) prior to the procedure helps decrease the rate of false positives in patients with clinically insignificant diskogenic pain. Most diskographers agree that chronically opiate-tolerant patients (who have also been NPO) should be given a reasonable dose of IV opiate to avoid false-positive responses in the setting of early opiate withdrawal and possible heightening pain responses or, at a minimum, the patients should take their usual morning pain medications with small sips of water. During diskography, patients must be monitored appropriately. Although respiratory depression is uncommon with this protocol, subjects should have a pulse oximeter and blood pressure cuff. Supplemental oxygen is supplied by nasal cannula. Personnel competent in airway management and resuscitation should be present during the procedure.

#### 8.4.3 Sterile Technique

The skin and draping technique for diskography is similar to the sterile technique used for surgery. Standard draping to provide a sterile field may include the use of sterile towels or fenestrated drapes per the injectionist's preference. Povidoneiodine 10 % (*Betadine* solution) and/or *DuraPrep* (iodophor 0.7 % and isopropyl alcohol 74 %) are the preparations of choice. Chlorhexidine and alcohol may be substituted if the patient has allergies to the aforementioned solutions. The procedure room staff should be dressed in clean clothes (scrub suites). Surgical caps and masks are mandatory for any personnel in close contact to the sterile field. Many injectionists scrub, gown, and glove as for an open surgical procedure. The C-arm image intensifier should also be draped.

# 8.5 Lumbar Diskography: Technique

Diskography can be performed in a procedure room appropriate for aseptic procedures. Fluoroscopy is required for safe visualization of spinal anatomy in anterior-posterior (AP), lateral, and oblique projections. Biplanar fluoroscopy can be utilized, but most diskographers use C-arm units which allow excellent visualization without repositioning the patient. An adjustable, radiolucent procedure table is useful.

# 8.6 Patient Position

The patient lies in a prone position on a fluoroscopy table. Most diskographers place a pillow or bolster under the patient's abdomen to slightly flex the spine and decrease the lumbar lordosis. Elevating the target side approximately 15° allows the fluoroscopy tube to remain in a more AP projection and reduces radiation scatter. If needed, a folded towel or soft wedge can be placed under the patient's flank to prevent side bending of the lumbar spine. Monitoring and light sedation are initiated. On the side selected for puncture, a wide area of the skin of the back is prepped and draped from the costal margin to the mid-buttock and from the midline to the flanks.

# 8.7 Disk Puncture

Until the 1960s, diskography was performed with a posterior interpedicular or transdural approach; however, this technique is seldom utilized today because it requires dual punctures of the dura. Currently, a lateral, or extra-pedicular, approach [37, 38] is used except in rare situations where anatomical variation or postsurgical changes prevent disk access by a lateral approach.

Prior to injection, a fluoroscopic examination of the spine is performed to confirm segmentation and determine the appropriate level for needle placement. The target disk is identified on AP view. The image intensifier of the C-arm is tilted in a cephalad or caudad direction until the end plate of the vertebral body, caudad to the target disk, is parallel to the X-ray beam (Fig. 8.1). The end plate is visualized as a line rather than an oval. After selecting the target disk on AP view, the fluoroscopic beam is obliquely rotated until the superior articular process of the adjacent caudal vertebrae appears to lie under the midpoint of the inferior vertebral end plate of the level above. In this view, the insertion point is 1 mm lateral to the lateral margin of the superior articular process (SAP) (Fig. 8.2). This positioning of the fluoroscope allows needles to be passed using "tunnel vision" (i.e., parallel to the beam when



**Fig. 8.1** AP view of the lumbar spine. Target disk is at L2–L3. *Closed white arrow* indicates superior end plate of L3 parallel to the X-ray beam



**Fig. 8.2** Right oblique view. Tip of the SAP of L3 is at approximate midpoint of the inferior end plate of the L2 vertebral body (\* target point, *SAP* superior articular process, *P* pedicle)

the skin puncture site is aligned with the target structure) just lateral to the SAP.

The disk is preferentially approached from the side opposite the patient's usual pain to avoid the patient mistaking discomfort secondary to needle placement with provoked pain secondary by disk The insertion point is marked on the skin. The distance between the opposite superior articular processes increases at the lower lumbar levels. At T12–L1, the needle insertion point is about 3–4 cm lateral to the midline; at L5–S1 it is approximately 6–7 cm lateral. At L5–S1, because of the iliac crest and increased interfacetal distance, ideal access to the disk may not be possible. The fluoroscope is therefore rotated only far enough to bring the superior articular process approximately 25 % of the distance between the anterior and posterior vertebral margins.

Prior to needle placement, a skin wheal is made with lidocaine 1 % (~1 cc) using a 25 gauge 1.5 in. needle. To anesthetize the needle track, one can use a 25 gauge 3.5 in. needle advanced under "tunnel vision," i.e., parallel to the X-ray beam, to the level of the SAP. Exercise caution so as not to anesthetize the dorsal root ganglion within the foramen. Overenthusiastic anesthetization may obscure nerve root impalement and could potentially anesthetize the sinuvertebral and ramus communicans nerves, thus altering the evoked pain response during disk stimulation and creating a false-negative response.

A one- or two-needle technique may be used; however, most diskographers currently use a twoneedle technique. Prior to the routine use of prophylactic antibiotics, Fraser et al. [24] reported a rate of diskitis with single non-styletted versus double needles of 2.7 % versus 0.7 %, respectively. Both the North American Spine Society and the International Spinal Injection Society recommend a two-needle approach [16, 19].

The two-needle technique utilizes a shorter, larger gauge introducer needle through which a longer, smaller gauge needle is advanced past the tip of the introducer needle into the targeted intervertebral disk, theoretically avoiding picking up any skin flora. The trend is to use a 20 gauge 3.5 in. introducer with the 25 gauge 6 in. disk puncture needle. The 25 gauge needle theoretically minimizes any trauma to the disk. Less experienced operators may start with the 18/22

**Fig. 8.3** Right oblique view. Introducer needle in lateral to SAP at L2–L3 target disk (see *closed white arrow*) (*SAP* superior articular process, *P* pedicle)

gauge needle combination, particularly because the curve of the disk puncture needle is easier to maintain. The body habitus of the patient will dictate if longer needles are necessary, i.e., 5 in. introducer and an 8 in. disk puncture needle. A slight bend, opposite the bevel, is typically made at the tip of the disk puncture needle to allow the operator to "steer" the needle during extra- and intradiscal insertion [39–42]. At times a larger curve, at the distal third of the disk puncture needle, must be utilized to compensate for less than ideal anatomy or postsurgical changes.

The introducer needle is passed through the skin wheal at the skin puncture point, using a "down the beam" or "tunnel vision" technique on the oblique fluoroscopic view and is often felt to enter the foramen (Fig. 8.3). To protect the diskographer's hand from radiation exposure, forceps may be used to grasp the introducing needle. To avoid potential neural injury, direct the needle into the region below the segmental nerve, just lateral to the superior articular process and above the end plate (Fig. 8.4). The disk puncture needle must travel under the segmental nerve coursing medial to lateral, and dorsal to ventral, to puncture the annulus fibrosus of the disk at the midpoint of the disk when seen in lateral and AP views. To minimize nerve trauma, one might





**Fig. 8.4** Location for diskogram needle insertion. *NR* nerve root outlined by contrast media, *SAP* superior articular process, *EP* end plate



**Fig. 8.5** Lateral fluoroscopic view of the lumbar spine. All introducer needles in place, at or just ventral to the posterior elements, for L1–L2 through L5–S1 disks

consider use of a needle with a short, non-cutting bevel. However, a Quinke tip spinal needle is appropriate in that contact with the ventral ramus should be a rare occurrence if a modicum of care is utilized. Forward advancement is stopped at the approximate level of the SAP, although placement within the foramen, ventral to the intervertebral disk annulus, is acceptable. Use a lateral fluoroscopic view to check needle depth (Fig. 8.5).

The stylette is then removed from the introducer, and the longer, smaller gauge disk puncture needle is advanced slowly under real-time lateral fluoroscopy. The needle will be seen to transverse the intervertebral foramen; then, a firm but distinct change in resistance will be noted as the needle touches and punctures the annulus fibrosus. On the lateral view, the needle will typically contact the posterior disk margin 1-3 mm posterior to the vertebral margin. On AP projection, the diskogram needle ideally contacts the disk margin on a line drawn between the midpoints of the pedicles above and below (Fig. 8.6). The patient may experience a brief sharp "pinch" or sudden aching sensation when the needle pierces the innervated outer annulus fibrosus. In no case should one advance the introducer or diskogram needle medial to the inner pedicle margins before contacting the intervertebral disk. Using lateral fluoroscopy, the needle is then advanced to the center of the disk as seen on both lateral and AP projections (Figs. 8.7 and 8.8). AP and lateral projections are used to assure good needle placement, and spot films saved for documentation prior to injection of contrast.

If bony obstruction is encountered, the physician should use fluoroscopy to determine whether the needle has contacted the superior articular process or the vertebral body. If the SAP is contacted, the introducer needle can be withdrawn slightly and its trajectory modified. The introducer needle can be advanced to just over the lateral edge of the superior articular process or advanced to the dorsal margin of the disk. If the vertebral body is contacted, the introducer needle is withdrawn to a point where manipulation of the slightly bent disk puncture needle can compensate for the nonoptimal placement of the introducer needle.

If the patient experiences any radicular pain or dysesthesia during needle advancement, insertion of the needle must halt. The ventral ramus may be encountered because it crosses the posterior-lateral aspect of the disk in close proximity to the disk entry site. In such a case, the needle is partially withdrawn to alter its course and redirected toward the disk. A slight



Fig. 8.6 The diskogram needle should contact the disk between the midpoint of the posterior vertebral margins on the lateral view (**a**, *white vertical line*) and at the line

between the midpoint of the pedicles on the AP view (**b**, *white vertical line*) (*P* pedicle)



Fig. 8.7 Lateral fluoroscopic view. Diskogram needles placed into center of intervertebral disks

bend in the needle tip facilitates small directional adjustments. Typically, redirection of the needle more medially and caudally will avoid the segmental nerve. If greater direction changes



**Fig. 8.8** AP view. Disk puncture needles in the center of the nucleus pulposus of each intervertebral disk

are needed, withdraw and redirect the introducer needle as well.

The above technique can be utilized for disk puncture in greater than 95 % of lumbar disk



**Fig. 8.9** (a) Oblique view of L5–S1 needle position. The tip of the diskogram needle can be seen just beyond the introducer just lateral to the S1 superior articular process and medial to the iliac crest. (b) Lateral view. The needle

is advanced slowly under direct fluoroscopic vision, and the guide needle is simultaneously retracted. The inner needle should contact the disk 2–3 mm posterior to the vertebral margin

levels; however, occasionally, due to anatomical variations (i.e., overriding iliac crest, osteophytes) or postsurgical changes (i.e., posterior intertransverse fusion mass or fusion hardware), variations in the procedure must be utilized. A detailed description of the myriad modifications with which a diskographer might be faced is beyond the scope of this chapter; however, most involve either a more lateral or more medial needle insertion with the disk puncture needle bent or curved to varying degrees. Rarely, the posterior interpedicular, transdural, approach must be used for disk puncture. This technique increases the chance for morbidity since the dura is punctured twice. Risks and benefits of this technique must be weighed. At the L2-L3 level and above, the posterior approach should never be used due to the real risk of impaling the spinal cord.

Disk puncture at L5–S1 can be technically more challenging than the L1–L4 levels when due to the increased inter-facet distance and the proximity of the iliac crest which obscures direct access to the disk nucleus. When optimal obliquity is not possible due to bone, less rotation is required. After the L5–S1 intervertebral disk has been identified and the superior end plate of the S1 vertebral body is aligned with the X-ray beam, the image intensifier is ipsilaterally obliqued until the ilium or sacral ala obscures the disk target. Counter rotation then is used to evidence a clear path to the lateral disk access. At this point, the S1 superior articular process (SAP) may be only 25 % of the distance between lateral vertebral borders. Less than 2 cm of the L5–S1 disk is visualized between the superior articular process of S1 and the sacral ala (Fig. 8.9). Unlike the direct approach at the levels above, a slight to marked curve or "hockey-stick" bend is required for the diskogram needle insertion at this level. Under the oblique fluoroscopic view, the introducer needle is advanced toward the bony notch between the S1 superior articular process (SAP) and sacral ala. The needle tip should be immediately adjacent to the anterolateral aspect of the S1 SAP (Fig. 8.9). Next, sterile gauze is used to curve the distal 2-3 cm of the diskogram needle into a smooth arc in the direction opposite the bevel (Fig. 8.10). The degree of curve is operator dependent, based on the amount of medial deflection required to reach the center of the disk. Under a live lateral fluoroscopic view, the curved diskogram needle is advanced through the guide



**Fig. 8.10** Operator uses sterile gauze to bend the distal tip of the diskogram needle (bevel facing out). This technique is often used at the L5–S1 disk space (Photo courtesy: Richard Derby, MD)

needle until the tip emerges and is felt to gain purchase in the outer annulus. The needle is then directed in a medial and slightly posterior course around the SAP to stay within the safe region (Fig. 8.4). For some diskographers, the 18/22 gauge needle combination may be easier to direct versus the 20/25 gauge needle combination used at upper levels. Obese or muscular patients may require longer needles. Once the diskogram needle reaches the annulus fibrosus, its position is checked in both AP and lateral views. In the lateral view, the needle should contact the disk 2-3 mm posterior to the vertebral margin (Fig. 8.9b), and in the AP view, the needle should ideally be on a line bisecting the midpoint of the L5 and S1 pedicles. The needle course must be closely monitored; if the needle does not curve sufficiently in the medial direction, it will not reach the center of the disk; moreover, it may strike the ventral ramus. If the needle does not track medially and posteriorly, it must be removed and its curvature accentuated. Once the disk puncture needle is within the outer annulus and is slowly advanced, the introducer needle can be withdrawn to reinstitute and accentuate the bend on the disk puncture needle facilitating medial deviation. As mentioned previously, if the needle contacts bone, determine whether the superior articular process or the vertebral body has been encountered and make appropriate adjustments. Ideally, the final needle position is the center of the disk; however, there is leeway. In severely

degenerated disks, the needle position is not as crucial because the contrast spreads throughout the disk due to loss of the intrinsic disks architectural integrity. Ideally, the needle should be within 4–5 mm of the disk center on AP and lateral fluoroscopy.

#### 8.7.1 Disk Provocation

Diskography is a provocational test which attempts to mimic physiologic disk loads and evoke the patient's pain by increasing intradiscal pressure with an injection of contrast medium. Increased intradiscal pressure is thought to stimulate annular nerve endings, sensitized nociceptors, and/or pathologically innervated annular fissures. Historically, pressure standards have been lacking, no doubt leading to erroneous conclusions. This approach is taxonomically unsound; emerging standards require unambiguous operational criteria that establish a threshold intensity for both pain response and stimulation intensity. Both require a precise method to apply the stimulus and strict criteria for interpretation. The intensity of the provocation stimulus must be carefully controlled through the skilled operation of a manometer syringe or an automated manometer. A 3 cc syringe with manual thumb pressure is still utilized by some operators, but this does not reflect current standards. Stimulus intensity can also be quantified with a controlled inflation syringe and digital pressure readout, permitting more precise comparisons between patient disks and between diskographers.

Most abnormal disks will be painful between 15 and 50 psi a.o. [43] and are termed "mechanically sensitive" based on a four-type classification introduced in the 1990s by Derby et al. in respect to annular sensitivity [44]. Disks which are painful at pressures <15 psi a.o. are termed low-pressure positive or "chemically sensitive" disks [44]; if painful between 15 and 50 psi a.o., they are termed "mechanically sensitive" disks. Indeterminate disks are painful between 51 and 90 psi a.o. and normal disks had no pain provocation. An important caveat is that a normal disk can hurt if pressurized too high with uncontrolled, manual "thumb" pressurization. Much of the recent research reporting a high false-positive rate for lumbar diskography in asymptomatic subjects used uncontrolled, manual thumb pressurization to 100 psi a.o. [45, 46]. If a disk is painful at >50 psi, the response must be reported as indeterminate, because it is difficult to distinguish between a pathologically painful disk and the pain evoked from simply mechanically stimulating a normal or subclinically symptomatic disk [47]. To limit false-positive responses, the most up-todate diskography standards set a pressure criteria of <50 psi a.o. to define a positive response [16].

Injection speed is also a confounding factor and may account inter-operator variability in results and increased false-positive responses. At high injection speeds, the true intradiscal pressure (dynamic pressure) is higher than the recorded static pressure [48]. The dynamic pressure, measured only in research settings, is the actual pressure which would be recorded with an intradiscal pressure sensor. Currently, we measure pressure indirectly via a manometric syringe which records plateau static pressures postinjection. The pain during activities of daily living is more closely correlated to dynamic peak pressure [44]. Static pressure is reflective of dynamic pressure when recorded by needle sensor and manometer only at slower injection speeds (<0.08 ml/s) [48]. Currently, injection speed can be standardized with an automated manometer or manually by a skilled operator.

When all needles are positioned in the nuclei pulposi of the target disks, injection can commence. The patient should be awake and able to describe sensations produced by disk stimulation. The patient should be blinded to both the initiation of stimulation and the level of injection. Nonionic contrast medium combined with antibiotic is injected into each disk at a slow velocity using a calibrated injection syringe or automated manometer with digital pressure readout. The total volume injected should probably be limited to  $\leq 3.5$  ml. A standardized procedure form is recommended to record the stimulation parameters, patient response, and observations concerning internal disk morphology.

Opening pressure is reported first, with typical values from 5 to 25 psi, varying with the degree of disk degeneration. If the opening pressure is greater than 30 psi, this usually indicates that the needle tip is in the inner annulus and therefore must be repositioned. At each 0.5 ml aliquot, the following data is collected: total volume, static and dynamic pressures, pain response (intensity and concordance), pain behaviors (vocal or physical), and contrast pattern. Injection continues until one of the following end points is reached: pain response  $\geq$ 7/10, intradiscal pressure  $\geq$ 50 psi above opening in a disk with a grade 3 or greater annular tear or 80-100 psi in a disk with normal-appearing nucleogram, epidural or vascular pattern is evident, or a total of 3.5 ml of contrast medium has been injected.

Some severely degenerated disks may accept greater volume; however, the incidence of falsepositive pain responses may increase. If the disk cannot be pressurized in a slow sustained manner, to greater than 50 psi above opening pressure at  $\leq 3.5$  ml volume (due to an annular or end plate leak or severe disk degeneration), one can use a rapid manual injection of a small volume to elicit a dynamic pressure of 50 psi above opening. However, the diskographer should be aware that in the setting of a leak, stimulation of structures adjacent to the disk (e.g., posterior longitudinal ligament, DRG, nerve roots, etc.) could provoke back pain or referred pain. Furthermore, with injection into the disk nucleus, the height of the intervertebral disk can increase causing motion of the contiguous vertebral bodies and possibly

stimulating an adjacent disk with pain provocation. This would be expected to a greater extent when high volumes of injectate are utilized or high pressures obtained.

# 8.8 Imaging

Anterior-posterior (AP) and lateral images of all injected disks are saved as part of the permanent record. A descriptive classification [49] is used for the fluoroscopic images: cotton ball, lobular, irregular, fissured, and ruptured (Fig. 8.11). A variety of patterns may occur in abnormal disks [49]. The appearance of the normal nucleus following the injection of contrast medium is classic: the contrast medium assumes either a lobular pattern or a bilobed "hamburger" pattern (Fig. 8.11). Contrast medium may extend into radial fissures of various lengths but remain contained within the disk (Figs. 8.11 and 8.12). Contrast may escape into the epidural spaces through a torn annulus (Figs. 8.11 and 8.13). In Fig. 8.13, note how epidural contrast outlines the location of the left S1 nerve root as it passes under the pedicle. This might explain how a



**Fig. 8.11** Lateral fluoroscopic view after disk injection. Large *closed white arrows* point to disks. L3–L4 disk, classic bilobed, "hamburger" pattern; L4–L5 disk, posterior annular fissure with contrast dye outlining disk protrusion; L5–S1, posterior annular tear extends into disk protrusion with very small leak visible (*thin vertical white arrow*)

patient could experience both axial pain and pseudo radicular pain. In some cases the contrast medium may escape through a defect in the vertebral end plate [8]. In other cases, the disk is completely fissured and disrupted (Fig. 8.14). However, none of these patterns alone is indicative of whether the disk is painful; that can be



**Fig. 8.12** AP fluoroscopic view of same patient in Fig. 8.11. *Closed white arrows* point to disk spaces. L3–L4 disk, bilobed dye pattern; L4–L5 disk, left-sided annular fissure extending into lateral protrusion. L5–S1 disk, marked annular disruption with small leak visible inferiorly\*



**Fig 8.13** AP fluoroscopic view of lumbar spine. \*, epidural leak at *left* L5–S1 level; note how contrast outlines the location of the S1 nerve root (*thin white arrow*). Large *white arrow* shows right lateral disk protrusion below L3 osteophyte



**Fig. 8.14** AP fluoroscopic view of the lumbar spine. Multilevel severe degenerative disk disease from L1–L2 to L5–S1

ascertained only by the patient's subjective pain response to disk injection.

Post-diskography axial CT scanning provides the most accurate depiction of internal disk architecture. The degree of degeneration is described by dividing the disk into four quadrants [50]. If the contrast is confined to the nucleus, then no quadrant disruption is present; if the contrast is dispersed, then its location is described (e.g., single-quadrant disruption, right posterior; twoquadrant disruption, left anterolateral and right posterior, etc.). The degree of radial and annular disruption is most commonly described [50, 51] using the Modified Dallas Discogram Scale (Fig. 8.15) [16, 52, 53]: grade 0 indicates contrast is contained within the nucleus; grades 1-3 describe degree of fissuring extending to the inner, middle, and outer annulus, respectively; grade 4 describes a grade 3 annular fissure with a



**Fig. 8.15** Modified Dallas Discogram Scale. Grade 0, no annular disruption; grade 1, radial disruption into the inner third of the annulus; grade 2, contrast spread into the middle third of the annulus; grade 3, contrast into the

innervated outer third of the annulus; grade 4, grade 3 with >30° circumferential tear; grade 5, spread of contrast into the epidural space (Adapted from Endres and Bogduk [29], p 23)



**Fig. 8.16** Axial post-diskography CT scan with grade 0 annular tear. Contrast is contained in the nucleus pulposus



**Fig. 8.18** Axial post-diskography CT scan with grade 1 and 2 annular tears. Contrast extends slightly into the inner annulus in the right posterior quadrant and into the middle annulus in the left posterior quadrant



**Fig. 8.17** Axial post-diskography CT scan with grade 1 annular tear. Contrast extends slightly into the inner annulus in the right posterior quadrant

greater than  $30^{\circ}$  circumferential arc of contrast (Figs. 8.16, 8.17, 8.18, 8.19, and 8.20), and a grade 5 annular tear indicates rupture or spread of contrast beyond the outer annulus into the epidural space or foramen (Fig. 8.13).



**Fig. 8.19** Axial post-diskography CT scan with grade 3 annular tear. Contrast extends posteriorly to the outer annulus within a contained protrusion

# 8.9 Diskography Standards

Both the techniques for performing diskography and the criteria for interpreting the findings have been in a constant state of evolution since their



**Fig. 8.20** Axial post-diskography CT scan with grade 4 annular tear. Contrast extends into the left posterior quadrant and into a circumferential tear

introduction in the 1940s. Until very recently, diskography has been performed without strict operational standards with respect to pressure limits, injection speed, volume, and validated clinical end points. The current standard for determining a positive response to diskography with the use of pressure-controlled diskography is pain  $\geq$ 7/10, pressure <50 psi a.o., concordant pain, grade 3 or greater annular tear,  $\leq 3.5$  ml volume, and at least one negative control disk [16]. One can refine the criteria by adding the Walsh criteria, which stipulate that a positive response includes  $\geq 2/5$  pain behaviors (guard/brace/withdraw, rubbing, sighing, verbalizing, and grimacing) [54]. Provocation diskography is the best diagnostic test we have to diagnose diskogenic pain. However, if performed without consistent operational and interpretation standards, diskographers can obtain inaccurate results.

# 8.10 Caveats

The following are techniques employed by experienced diskographers to optimize performance of the test, as well as limit false-positive and false-negative responses:

- The diskographer must be skilled in needle placement; otherwise, further pain provocation will be hard to interpret. Inexperienced diskographers often impale the adjacent segmental nerve or create significant tissue trauma from multiple needle insertion attempts.
- Carefully evaluate pain produced ipsilaterally to the needle insertion site. Referred pain may be caused by the diskogram needle impinging upon the dorsal root ganglion. Gently "jiggle" the needle to distinguish needle pain from diskogenic pain.
- 3. Transient pain may be provoked if an asymptomatic fissure or previously healed annular tear with a fibrous cap is abruptly opened during pressurization. A true positive pain response is ≥7/10 and sustained for greater than 30–60 s; true diskogenic pain is less likely to decrease rapidly. Pain which resolves within 10 s should be discounted. Clinically, patients with diskogenic pain tend to have increased pain postoperatively and an exacerbation of symptoms for 3–7 days.
- Confirm all positive responses with manual re-pressurization with a small volume. If repressurization does not provoke concordant ≥7/10 pain at <50 psi a.o., then the response is considered indeterminate.
- 5. If the patient has significant pain in a disk without a grade 3 tear (adjacent to a positive response disk), consider injecting 1 ml of 4 % Xylocaine into the painful adjacent disk and retest the normal-appearing disk in 10 min. One may find that the disk is no longer painful. There is likely segmental overlap with innervation or stimulation of the adjacent disk due to vertebral body movement.
- 6. Injected volume should be limited to 3.5 ml. Painless, morphologically severely degenerated disks can be made painful if excessive volume is injected.
- 7. When diskography is performed on disks status post prior diskectomy, the false-positive response is likely higher. The results should be reported as indeterminate unless the disk is painful at low volume and low pressure.

# 8.11 Post-procedure Care

After the procedure, patients are taken to the recovery room for vital sign and clinical status monitoring by nurses trained in spine injection management. The patient is checked immediately post-transfer and 30 min post-procedure for any subcutaneous bleeding. Analgesic medications (oral, IV, or IM) is provided as needed. The patient is advised to that they may experience an exacerbation of their typical symptoms for 2-7 days. The patient is instructed to contact the office if he or she develops fever, chills, or severe (or delayed) onset of pain. Patients are observed and discharged according to institutional protocol. Typically, the patient is discharged to the care of a responsible adult and instructed not to drive for the remainder of the day. All patients should be contacted by phone 2-4 days postprocedure to screen to complications or adverse side effects.

# 8.12 Potential Risks and Complications

Post-diskography complications are well described [25, 55]. Complications can occur secondary to the disk puncture itself and to misadventures during needle placement or can be related to medications utilized. Complications vary from minor (e.g., increased low back pain, nausea, headache) to major (diskitis, seizure, permanent neurologic injury, and death) [56].

Concern over prolonged pain after diskography has been overblown. In the 1960s, Holt reported prolonged low back pain after diskography [57]; however, serious criticisms were raised about this author's patient population (prisoners), technique, and the use of noxious Hypaque dye. More recent studies also have serious shortcomings. The claim that diskography causes prolonged pain at 1 year in subjects asymptomatic of low back pain who underwent diskography is based on six patients [58]. However, a closer look must be taken at these six patients, as they are not "psychologically" representative of patients undergoing PD. All patients had been diagnosed

distressed somatics or with somatization as disorder. Two of the six patients were chronic pain patients status post failed cervical fusion, on daily opiate medications with active worker's compensation claims. Psychometric testing of these two patients also revealed that they were distressed somatics. The other four patients had a primary diagnosis of somatization disorder; two of these four patients were unable to tolerate the initial needle placement at more than one or two levels and were excluded from further study on diskography, yet they were included in the 1-year follow-up publication reporting that diskography caused prolonged pain. Such a small sample size limits generalizability of the conclusions; moreover, it is well recognized that persons with somatization disorder commonly complain of recurrent pain and conversion phenomena (pseudoneurologic symptoms) and are at risk for iatrogenic illness [59]. Furthermore, somatization disorder patients are hospitalized or undergo surgery three times as often as depressed patients [**60**].

Diskitis is the most common serious complication of diskography, reported to be less than 0.15 % per patient and 0.08 % per disk [19]. The incidence of diskitis has been clearly diminished with the double- vs. single-needle technique [24]. Also, with careful preprocedure screening for infection (e.g., UTI or skin), aseptic skin preparation, styletted needles, and intravenous and intradiscal antibiotics, diskitis is now very rare. Over a 10-year period, in our clinic (RD), only one case of diskitis per over 2,000 patients has been recorded, while in the practice of the other author (ML), in over 5,000 disks injected, no cases of diskitis are known. In these practices using standard prophylactic measures, the combined rate of diskitis is less than 1/11,000 or <0.009 %. To prevent diskitis, the authors recommend a surgical skin preparation and draping, a double-needle technique, and intravenous prophylaxis with antibiotics before the procedure as well as intradiscal antibiotics. However, even with prophylactic antibiotics, an epidural abscess after diskography has been reported [61, 62].

The most common causative organisms for diskitis after lumbar diskography are *S. aureus*, *S.* 

epidermis, and E. coli [25, 63, 64] suggesting inoculation with surface flora or inadvertent bowel perforation. Clinically, the patient with diskitis presents with severe, unremitting, disabling pain in the days to weeks after the procedure. The patient may report a change in the quality of their pain as well as typical relieving factors. Some patients have a fever, although this is not a universal symptom. The workup should include a physical examination, laboratory, and imaging studies. Laboratory tests include complete blood count (CBC) with differential, C-reactive protein (CRP), sedimentation rate erythrocyte sedimentation rate (ESR), and blood cultures. The CRP usually increases within days of onset of infection; the ESR is not as sensitive and may not be elevated for a month. If the end plates have not been breached, the blood cultures and CBC will be normal. MRI is the preferred imaging modality [65–67]. Technectium-99 bone scan is less sensitive and specific than MRI [68]. MRI within 3–4 days of symptoms shows increased T2 signal in the disk and end plate hyperemia. Biopsy in the acute phase, before end plate breach, is more likely to be positive. After end plate breach, sanguineous spread creates a sterile environment and activation of the immune system [69]. Treatment of diskitis typically requires prolonged antibiotic therapy, although some mild self-limiting cases have been reported [69]. Empyema or abscess formation requires CT-guided drainage or surgical intervention [70–72].

Striking a ventral ramus is a potential hazard but may be avoided by careful attention to correct technique. The needle should be prevented from straying beyond its required and intended course. In a conscious patient, contact with the ventral ramus will be obviously indicated by a severe, sharp lancinating pain. Other complications include spinal cord or nerve root injury, cord compression or myelopathy, urticaria, retroperitoneal hemorrhage, nausea, convulsions, headache, and, most commonly, increased pain [19]. Disk herniation following diskography is very rare [73], and there is little evidence that diskography damages the disk [74]. Freeman et al. recently reported no histological damage following needle insertion into sheep disks [75]. Death

after diskography was reported in a patient who inadvertently received contrast dye mixed with cefazolin (12.5 mg/ml) intrathecally. She developed intractable seizures and coma and ultimately died [76].

# 8.13 Diskography Controversies: Brief Literature Review

# 8.13.1 Utility of Provocation Diskography

While numerous papers have examined the usefulness of diskography, some physicians still question its reliability [58, 77]. Critics point toward mismatches between morphological features, clinical complaints, and false-positive rates. Lumbar provocation diskography for the diagnosis of abnormalities involving intervertebral disks has been used extensively as a diagnostic tool for evaluating low back pain (LBP) since the 1950s. Diskography has always been controversial, with both staunch proponents and opponents. Diskography is not meant to be a stand-alone test. Its proper use is in the diagnostic algorithm for chronic, intractable low back pain unresponsive to conservative care and when the patient is considering intradiscal or surgical treatments. Perhaps unappreciated is the value of diskography in ruling out adjacent segment diskogenic pain and limiting the number of levels treated surgically. Diskography remains the criterion standard for visualizing internal disk architecture. As a provocation test, despite its liabilities and limitations, diskography is also our best means of diagnosing diskogenic pain. Provocation diskography has an 81 % sensitivity and 64 % specificity for pain [78].

Provocation diskography (PD) is commonly compared to CT or MRI of the spine, yet PD is distinctive in that it is both an imaging and provocation test. MRI is unable to discern a painless from a painful disk. We are well aware that morphologic abnormalities of the disk are common in both patients with and without chronic low back pain. In one series, 78 % of patients with chronic low back pain had abnormal imaging findings on
diskography combined with post-CT scanning [73, 79]. Morphologic abnormalities also increase with age [55, 80]. Admittedly, abnormal morphological structures revealed by diskography without provocation are too nonspecific to be clinically useful; therefore, positive results should be limited to those eliciting concordant pain [54, 77]. In 1982, Millete and Melanson [81] reported retrospectively that concordant pain was evoked by injection in 37 % of patients with a diskographic morphologic abnormality. Studies in the 1990s and 2000s on the prevalence of painful internal disk disruption in patients presenting to tertiary referral centers with chronic low back pain range from 26 % to 39 % [82, 83].

For most cases, MRI is sufficient for advanced imaging; however, provocation diskography adds additional useful diagnostic information to confirm or refute the hypothesis that a particular disk is the source of the patient's pain. Neither highresolution CT nor MRI demonstrates annular morphology with the same detail as diskography [84, 85]. CT-diskography was found more sensitive than MRI in detecting early annular disruption. In 18/177 disks with a normal T2-weighted MRIs, annular and radial fissures were revealed on CT-diskography [86]. When compared to surgical findings, in 94 % of patients, CT-diskography correctly diagnosed the type of disk herniation, including in the previously operated spine [86]. Diskography has also been correlated to pain drawings [87].

The high-intensity zone (HIZ) has been proposed as a marker for painful disks [52, 88], yet while highly specific, the sensitivity of this finding is only 26 %, limiting the usefulness of the HIZ in selecting patients for surgery [89]. However, if an HIZ is contained within a protrusion, there is a correlation with a positive diskogram response [90]. Modic changes are a highly specific marker pain at diskography [91]. These authors have suggested that diskography is not needed if the index disk level has Modic type I changes. Surgical decision-making requires a high level of diagnostic confidence, and research has shown that MRI cannot always reliably predict which disks will be symptomatic with

diskography [92, 93]. Diskography has fallen out of favor in many countries due to the studies reporting a high false-positive rate and potential disk injury after PD, in spite of subsequent studies disputing these findings.

#### 8.13.2 False-Positive Rates

Diskography's greatest liability at this time is the series of studies reporting high false-positive rates in asymptomatic subjects [46, 57]. These recent negative studies also reported a correlation between the presence of chronic pain and abnormal psychometric scores with positive diskogram responses. Early work by Holt in 1968 reported a 36 % rate of positive diskography in asymptomatic subjects, although this study contained serious methodological flaws [57]. Reanalysis of the Holt data actually reveals a false-positive rate of 3.7 % [13]. Holt's findings were also refuted by Walsh et al. [54], who demonstrated a 0 % rate of positive diskography in ten asymptomatic volunteers and established reproducible criteria for positive diskography (criteria:  $\geq 3/5$  pain (pain thermometer), concordant pain,  $\geq 2/5$  pain behaviors, pressure limited to approximately 60-70 psi a.o. (manual pressurization by a highly skilled diskographer), and abnormal disk morphology). Walsh's study was critiqued due to his asymptomatic subject choice of primarily young men in their 20s. Using strict pressure criteria and pressure manometry, Derby et al. [94] studied 13 asymptomatic volunteers and also reported a 0 % false-positive rate.

Carragee [46] performed diskography on several subject populations asymptomatic of low back pain and reported the following falsepositive rates: residual pain after iliac crest bone harvest, 50 %; no low back pain and no chronic pain, 10 %; chronic cervical pain, 40 %; postdiskectomy, 35 %; and somatization disorder, 75 %. He also studied symptomatic subjects with "benign" or "mild persistent low back pain" and reported a false-positive rate of 36 %.

More recent literature has reached different conclusions. A critical examination of all the

studies since the 1960s including a systematic review and a meta-analyses of false-positive rates [13] shows that an acceptably low false-positive rate can be achieved using PD utilizing the ISIS/ IASP standards for a positive diskogram: pain  $\geq$ 7/10, concordant pain, pressure <50 psi a.o.,  $\geq$ grade 3 annular tear, volume limit  $\leq$  3.5 ml, and presence of a negative control disk. A recent meta-analysis of all studies of asymptomatic subjects undergoing diskography obtained a specificity of 0.94 (95 % CI 0.89–0.98) or a false-positive rate of 6 % per disk and 9.3 % per patient [13]. Among subjects asymptomatic of any confounding factors, one can obtain a false-positive rate of 3 % per patient and 2.1 % per disk; for subjects with chronic pain, the rate is 5.6 % per patient and 3.85 % per disk. Taken alone as a group, post-diskectomy subjects appear to have a slightly higher false-positive rate of 15 % per patient and 9.1 % per disk. Given our limited knowledge of diskography in post-diskectomy patients and the possibility that provocation may open previously healed granulation tissue along surgical planes, diskographers may consider pressure- and speedcontrolled manometry and low pressure and volume limits for defining a positive value. Among subjects with somatization disorder, the falsepositive rate is 50 % per patient (95 % CI 0-100 %) and 22 % per disk. Concern has also been raised regarding chronic pain and psychologic comorbidity as significant confounding factors in patients undergoing diskography. Is chronic pain a significant confounding factor? Evidence indicates that patients with chronic or chronic intermittent low back pain respond similarly to disk stimulation as do asymptomatic volunteers undergoing diskography. Derby studied the effect of chronic low back pain on negative and positive disks versus asymptomatic controls [94]. For example, comparing disk stimulation in disks with grade three annular tears, there was no difference in pain scores (1.6/10 versus 1.1/10)reported by asymptomatic volunteer disks compared to negative patient disks. Patients undergoing diskography can readily distinguish between a negative and positive disk level [94, 95]. Shin et al. also recently reported that a majority of patients with grade 4 patients could distinguish between positive and negative disks by magnitude of pain response [95]. The argument that a majority of patients with chronic pain who undergo diskography will overreport pain is not supportable.

Are psychologic comorbidities confounding factors? Perhaps, but the data is conflicting. Carragee [46] studied six subjects with somatization disorder. Only 4/6 subjects were able to complete their diskogram because of pain (the cause of the pain is not reported, i.e., secondary to placement of diskogram needles versus disk stimulation). From this small sample size, a 75 % false-positive rate with a 95 % confidence interval from 0 % to 100 % was reported. Given the type of patients studied and the statistical shortcomings of the analysis, the generalizability of these findings is limited. Furthermore, contrary to these previous findings, a larger, randomizedcontrolled trial comparing diskography results of 25 patients with and without somatization disorder found no significant difference in positive responses between groups [96]. There was also no difference in positive responses in patients with depression and/or general anxiety disorder. Derby et al. [21] reported Distress and Risk Assessment Method (DRAM) scores of 81 patients undergoing diskography: 15 % (12/81) were normal, 52 % (42/81) were at risk, and 33 %(27/81) were abnormal (distressed depressive or somatic). The positive rates of diskography were not statistically significant by subgroup (p > 0.05). In patients with chronic low back pain, no correlation was found between presenting DRAM score and diskography result.

Concern has also been raised that diskography is falsely positive in patients with "benign" or chronic persistent mild low back pain [97]. To be considered a subject with "benign" low back pain, subjects reported that they did not restrict their activity or seek medical care for their back pain. They reported their pain as mild to moderate or 2–4/10. Provocation diskography was reported as positive in 9/25 patients; thus, the author reported obtaining a 36 % "false-positive" rate. Of note, 72 % (18/25)

of the patients had chronic pain due to failed cervical surgeries and were on medications, including opiates for their pain, which may have masked their pain. In addition, patients taking narcotic medications for as little as 1 month might be expected to evidence hyperalgesia and allodynia resulting in an inaccurate rate of false-positive results. Moreover, such subjects may have been understandably reluctant to seek medical care for their low back pain after a failed cervical surgery. The 36 % reported false positives are arguably true positives. One could argue that these chronic low back pain volunteers are no different from patients undergoing diskography who often have varying degrees of and duration of pain flare-ups. Given the history of neck pain in these patients, the fact that some have painfully internally disrupted disks in their lumbar spine is not surprising. In fact, the reported prevalence of positive diskograms in patients referred for chronic low back pain ranges from 26 % to 39 % [83, 98]. The argument that these positive responses represent "false-positive responses" is not supportable. Diskography was not developed and should not be used to determine the clinical significance of a patient's perceived suffering and disability related to chronic low back pain.

In fact, one of the reasons researchers [45, 46]obtained so many false-positive responses may have been because of the high pretest probability of diskogenic disease in the majority of the research subjects. All subjects, except for the somatization disorder and iliac crest pain patients, had a history of known diskogenic pain severe enough to require surgery. The subjects may have had asymptomatic or minimally symptomatic disease provoked with the use of high pressurization (up to 100 psi a.o.) or high dynamic pressures in the setting of manual diskography. Also, we know that the co-occurrence of cervical and lumbar diskogenic disease is commonplace. MRI studies of twins showed 79 % and 64 % heritability for "severe" disease in the cervical and lumbar spine, respectively [99]. In one follow-up study of 200 patients who underwent cervical surgery, 100 % had significant low back pain episodes (suggestive of disk herniation) and/or underwent back surgery [100]. Another reason for reported high false-positive rates may have been the use of uncontrolled manual pressurization (to 100 psi a.o.) with measurement of only static pressures [48]. Lastly, in some prior studies, false-positive rates are reported per patient instead of per disk (subjects status post iliac crest harvesting, status post-diskectomy, or with somatization disorder), leading to a significantly higher absolute number. Provocation diskography is a test designed to confirm or refute the hypothesis that a particular disk is a source of a patient's pain; however, a positive diskogram does not rule out other significant sources of pain.

Diskography continues to be controversial. In 2009, researchers reported [101] that modern diskography techniques utilizing a small-gauge needle and limited pressurization resulted in accelerated disk degeneration, disk herniation, loss of disk height and signal, and the development of reactive end plate changes compared to match controls [101]. In 2013, Ohtori et al. [102] obtained follow-up MRIs 5 years after diskography, utilizing contrast and bupivacaine ("diskoblock" or analgesic diskography). One concern raised by opponents of diskography in addition to the effects of disk puncture is a possible toxic effect to disk cells due to anesthetic. Ohtori found neither evidence of accelerated disk degeneration in patients who underwent modern diskography with contrast nor accelerated degeneration with the use of bupivacaine versus controls.

An additional concern raised by the recent study [101] was the effects of disk puncture as a cause of disk injury. Martin et al. [103] compared the effects of 29 gauge versus 26 gauge needles placed into mouse disks. The larger needle led to increased disk degeneration at 8 weeks; the smaller needle did not initiate degenerative changes. If this model is extrapolated to human disks, a 29, 26, or 25 gauge needle placed into the disk of a 70 kg male versus that of a mouse weighing an average of 1 oz. would not be likely to cause any radiologically or clinically significant disk degeneration. Continued long-term follow-up MRI of subjects undergoing diskography is needed to clarify long-term effects.

#### 8.14 Analgesic Diskography

Due to the controversies over provocation diskography and conflicting studies regarding high false-positive rates, new diagnostic methods have arisen. Just as facet pain is diagnosed by blocking a medial branch nerve with local anesthetic, the concept of blocking intradiscal nociceptors arose. Previously, anatomists reported that only the outer annulus of the disk was innervated; however, in painful disks, sensory nerves extend to the inner annulus and nucleus pulposus [10, 104, 105]. Painful internally disrupted lumbar disks also have greater concentrations of sensory fibers in their nucleus and end plates than normal disks and along annular fissures [9, 11, 106]. In disks sampled from surgery, researchers find higher levels of pro-inflammatory cytokines, including IL-8 and PGE2, thought to create hyperalgesia, thus "sensitizing" the disk mechanoreceptors [107, 108].

The first reported use of analgesic diskography (AD) was in 1948 by Hirsch [109]. He injected 0.5 cc of 1 % Novocain into the disks of patients who reported pain during disk puncture or needle movement. For the ensuing 2-4 h, after local anesthetic injection, these subjects were essentially free of back pain with normal mobility and negative straight-leg-raise testing. Subsequently, various surgeons have used the injection of local anesthetic to confirm the results of PD [11, 109, 110]. In the 1970s, Roth [110] studied AD in the cervical spine and reported that analgesic diskography was diagnostically superior to PD. He reported 93 % good to excellent surgical outcomes in 71 patients undergoing cervical fusion. Other descriptions of AD have largely been "embedded" in the methods sections of various diskography studies. Coppes et al. [11] used 0.5–1.0 ml of bupivacaine as a confirmatory test after provocative diskography and reported 1-4 h pain relief.

Alamin [111–113] was the first to formally study analgesic diskography using a balloon-tipped catheter (Functional Analgesic Diskography, FAD, Discyphor<sup>TM</sup> Kyphon, Sunnyvale, CA, USA) which he codeveloped. The catheter can be anchored in the disk for pre- and post-functional positional testing after instilling the disk with local anesthetic. Alamin et al. [113] compared the results of standard pressure-controlled PD to functional anesthetic diskography (FAD) in a study with 52 chronic low back pain. PD was performed first; FAD was performed in positive cases or in patients with clinical features and imaging studies highly suggestive of symptomatic disk degeneration. A positive FAD response was defined as a 2 or greater decrease in VAS. Discordant results were found in 46 % of the patients; 26 % of patients with positive PD had negative findings on the FAD test; 16 % had positive findings at a single level where the PD had found 2 or more positive levels; 4 % of patients had new positive findings on FAD. He reported that using FAD immediately following PD, the false-positive rate of PD could be significantly reduced.

DePalma [114] investigated whether outer annulus fissures stimulated during PD are a source of diskogenic pain. He found that 80 % of painful disks diagnosed by PD were also diagnosed with FAD, demonstrating >50 % pain reduction. Approximately 10 % demonstrated partial pain Unlike Alamin's reduction. results [113], DePalma found no correlation between psychiatric history and positive PD/FAD. The study did not report the percentage of discordant findings between PD and AD. DePalma found a similar percentage of patients with a positive PD and negative FAD as did Alamin et al. equal to 20 % in the second published study using the FAD system.

Subsequently, Derby et al. [115] correlated data from different analgesic protocols (including the DePalma study with FAD) where compared to obtain the incidence of pain relief following injection of local anesthetic (LA) into lumbar disks that caused concordant pain during provocation testing. Subjective pain relief was compared at three separate facilities: 23 patients undergoing routine provocation diskography (PD), 47 patients undergoing combined provocative diskography (CPD) with equal volumes of LA and contrast, 120 patients injected with LA following routine PD (AD/PD, same session), 33 patients undergoing stand-alone analgesic diskography (SAAD, separate sessions), and 28 patients injected with LA through a catheter (FAD) placed during provocative diskography testing. If the criterion standard to confirm annular tears is concordant pain provocation and 80 % or greater pain relief following LA injected into lumbar disks, the SAAD, AD/PD, and FAD protocols show statistically similar 20–30 % prevalences of diskogenic pain. As yet, there is no single validated technique for analgesic diskography.

At this time the evidence is insufficient to recommend that AD or FAD replace PD. However, these tests may be useful confirmatory tests. Further research is needed to compare PD to AD, to refine technical performance of the test, to determine if AD has a lower false-positive rate than PD, and to determine if surgical outcomes are superior with AD.

## 8.15 Predictive Value

One of the obvious challenges with diskography, whether using provocative or analgesic diskography, is the lack of a clear "gold standard" to which the outcomes of testing for diskogenic pain can be compared. The best proxy we have for a gold standard is the outcome from the surgical intervention. Various intradiscal procedures remain poorly tested; therefore, surgical fusion is the current standard treatment for diskogenic LBP unresponsive to conservative therapy. However, this proxy also has limitations, as there are many different surgical fusion techniques used to treat diskogenic pain. Carragee et al. [116] compared circumferential fusion for singlelevel diskogenic pain determined by PD with the results of the same surgery for single-level isthmic spondylolisthesis. High-grade clinical success was achieved in 72 % (23 of 32 patients in this control group) of the patients with isthmic spondylolisthesis versus 27 % (8 of 30 patients) with presumed diskogenic pain. The authors of this study attributed the poor response of the "diskogenic pain" patients to the high falsepositive rate of PD; however, other authors also point out that isthmic spondylolisthesis and painful internal disk disruption are fundamentally different clinical entities in terms of their predictable response to fusion.

Several other studies have reported on surgical outcomes in patients undergoing diskography.

Colhoun et al. [117] studied 195 patients with axial pain and reported that of 137 patients with diskogram positive for disk disease and provoked concordant pain, 89 % derived significant, sustained clinical benefit from operation. Twenty-five patients showed morphologic disk abnormalities, but no provocation of concordant pain on diskography. Among this group, only 52 % had clinical success. Blumenthal et al. (1988) [118] reported that 74 % of patients with internal disk disruption returned to work following anterior lumbar fusion performed based upon diskography. In a multicenter surgical and nonsurgical outcome study after pressure-controlled diskography, Derby et al. [44] stated that precise prospective categorization of positive diskographic diagnoses may predict treatment outcomes, surgical or otherwise, thereby greatly facilitating therapeutic decision-making. In addition, patients with highly sensitive disks at low pressure appear to achieve significantly better long-term outcomes with interbody/combined fusion than with intertransverse fusion. Lettice et al. [119] used pressure-controlled manometric PD with strict criteria (pressure <50 psi a.o. with grade III annular tear) to compare results of short-segment (1-2) vs. long-segment (3-5) fusions. Typically long-segment fusions show progressively poorer clinical outcomes. In an SF-36-based outcome study of total joint arthroplasty, PCS score increments of 10 and 12 have been reported. Similar PCS score improvements (11.35 and 10.06 for the shortand long-segment groups during the 2-year follow-up, respectively) [120] were obtained in this study. The authors attributed this success to the use of criterion-based pressure-controlled, manometric PD.

Recently, Cooper et al. [121] studied the prognostic value to lumbar disk stimulation and to validate the ability of disk stimulation to predict treatment response, using the ISIS guidelines for positive (>70 points), negative (40-60 points), and indeterminate (<40 points) responses [16] and physician determination of diskography: positive, indeterminate, and negative. He performed an opportunistic audit of patients who underwent diskography and their surgical outcomes; the data was collected retrospectively; thus, no interference occurred with respect to diskographer or surgeon. Patients agreed to be followed up regarding response to treatment, pain scores, the use of health care, and functional status. Eightynine patients were included in the study. The results demonstrated that diskography was predictive of treatment response. A cutoff score of >50 and physician interpretation both showed statistical significance. A patient undergoing fusion with a score of >50 had the following results:  $5 \times$  more likely to return to >25 % ADLs, 3.4 times more likely to return to >50 % ADLs, and 3.3 times more likely to have less pain than patients who did not choose fusion. With an ISIS score of >50, fusion outcome was superior to IDET. With a score of <50, conservative treatment was superior to fusion. Cooper et al.'s [121] final recommendation was to utilize an ISIS score of 50 as a cutoff with clinical judgment to obtain the best patient outcome.

Two authors have reported on surgical fusion outcomes using FAD or AD. Alamin [111, 112] has followed 16 patients out to 6 months: the mean Oswestry score decreased from 55 to 25; mean back pain VAS decreased from 6.9 to 2.6. Ohtori et al. [122] performed an RCT comparing surgical outcomes of 15 patients undergoing PD versus AD. At 3 years, the VAS score, Japanese Orthopedic Association Score, Oswestry Disability Index, and patient satisfaction score were superior (p < 0.05) in the AD group.

Finally, although imperfect, diskography is safe in experienced hands, shows substantial sensitivity for identifying painful disks, and may predict surgery-related outcomes. Analgesic diskography may give additional value to confirm a positive finding on PD.

#### 8.16 Summary

For over 50 years, lumbar diskography has been widely used for evaluating diskogenic low back pain. Since these earlier negative studies, diskography techniques and diagnostic criteria have also advanced. Complications are minimal in skilled hands. Most importantly, pressure- and speed-controlled manometric diskography has been adopted. Previous diskogram studies assessing false-positive rates had limitations including the use of excessive pressurization, manual injections, uncontrolled injection rates, unrecorded and/or unreported opening, and dynamic pressures, volumes, and maximal volumes. Walsh et al.'s [54] study performed by a skilled operator with strict criteria reported a 0 % false-positive rate. Derby et al. [123] used pressure-controlled manometry and precise criteria for positive diskography and also obtained a 0 % false-positive rate. We recommend positive criteria for lumbar pressure-controlled manometric diskography as follows per ISIS/IASP standards: pain  $\geq$ 7/10, concordant pain, <50 psi a.o., at least grade 3 annular tear,  $\leq 3.5$  ml total volume, and at least one negative control disk [16]. Based on the use of these standards, our best estimate of the falsepositive rate for lumbar diskography is acceptable. A recent meta-analysis combining all the recent diskography studies in asymptomatic subjects (including data from Carragee) obtained a 6 % false-positive rate per disk [13]. Analgesic diskography is an emerging test in which local anesthetic is used to attempt to block intradiscal nociceptors. Proponents of AD report that it has a lower false-positive rate than PD; other researchers have found various AD protocols reporting 20-30 % prevalence. AD is currently being used as a confirmatory test to PD. Recently, researchers reported an increased incidence of disk degeneration on follow-up MRI in patients undergoing diskography [101]. However, more recent study [102] has not replicated these findings. Clearly, continued research is needed. Diskography remains an important diagnostic test for diskogenic pain.

#### References

- Lindblom K. Diagnostic puncture of intervertebral disks in sciatica. Acta Orthop Scand. 1948;17(3–4): 231–9.
- Lindblom K. Technique and results in myelography and disc puncture. Acta Radiol. 1950;34(4–5): 321–30.
- Lindblom K. Technique and results of diagnostic disc puncture and injection (discography) in the lumbar region. Acta Orthop Scand. 1951;20(4):315–26.
- Dandy WE. Loose cartilage from intervertebral disk simulating tumor of the spinal cord. By Walter E. Dandy, 1929. Clin Orthop Relat Res. 1989;238:4–8.
- Mixter W, Barr J. Rupture of the intervertebral disc with involvement of the spinal canal. N Engl J Med. 1934;211:210.
- Gardner WJ, et al. X-ray visualization of the intervertebral disk; with a consideration of the morbidity of disk puncture. AMA Arch Surg. 1952;64(3):355–64.
- Wise RE, Gardner WJ, Hosier RB. X-ray visualization of the intervertebral disk. N Engl J Med. 1957;257(1):6–10.
- Bogduk N, Aprill C, Derby R. Discography. In: White A, Schofferman J, editors. Spine care: diagnosis and conservative treatment. St. Louis: Mosby; 1995. p. 219–36.
- 9. Peng B, et al. The pathogenesis of discogenic low back pain. J Bone Joint Surg. 2005;87(1):62–7.
- Freemont AJ, et al. Nerve ingrowth into diseased intervertebral disc in chronic back pain. Lancet. 1997;350(9072):178–81.
- Coppes MH, et al. Innervation of "painful" lumbar discs. Spine. 1997;22(20):2342–9.
- O'Neill C, Derby R. Percutaneous discectomy using nucleoplasty. In: International 21st course for percutaneous endoscopic spinal surgery and complementary techniques. Zurich: Spital Zolikerberg; 2003.
- Wolfer LR, et al. Systematic review of lumbar provocation discography in asymptomatic subjects with a meta-analysis of false-positive rates. Pain Physician. 2008;11(4):513–38.
- Boden SD, et al. Abnormal magnetic-resonance scans of the cervical spine in asymptomatic subjects. A prospective investigation. J Bone Joint Surg Am. 1990;72(8):1178–84.
- Jensen MC, et al. Magnetic resonance imaging of the lumbar spine in people without back pain. N Engl J Med. 1994;331(2):69–73.
- Bogduk N. Practice guidelines for spinal diagnostic and treatment procedures. In: Bogduk N, editor. Lumbar disc stimulation. San Francisco: International Spine Intervention Society; 2004.
- Derby R, Lee SH, Kim BJ. Discography. In: Slipman CW et al., editors. Interventional spine: an algorithmic approach. Philadelphia: Saunders Elsevier; 2008. p. 291–302.
- Landers MH, et al. Lumbar spinal neuroaxial procedures. In: Raj PP et al., editors. Interventional pain

management: image-guided procedures. Philadelphia: Saunders Elsevier; 2008. p. 322–67.

- Guyer RD, Ohnmeiss DD. Lumbar discography. Position statement from the North American Spine Society Diagnostic and Therapeutic Committee. Spine. 1995;20(18):2048–59.
- 20. Greenspan A, et al. Is there a role for diskography in the era of magnetic resonance imaging? Prospective correlation and quantitative analysis of computed tomography-diskography, magnetic resonance imaging, and surgical findings. J Spinal Disord. 1992;5(1): 26–31.
- Derby R, et al. The influence of psychologic factors on diskography in patients with chronic axial low back pain. Arch Phys Med Rehabil. 2008;89(7): 1300–4.
- Huang TS, et al. Gadopentetate dimeglumine as an intradiscal contrast agent. Spine. 2002;27(8):839–43.
- 23. Falco FJ, Moran JG. Lumbar discography using gadolinium in patients with iodine contrast allergy followed by postdiscography computed tomography scan. Spine. 2003;28(1):E1–4.
- Fraser RD, Osti OL, Vernon-Roberts B. Discitis after discography. J Bone Joint Surg Br. 1987;69(1): 26–35.
- Fraser RD, Osti OL, Vernon-Roberts B. Iatrogenic discitis: the role of intravenous antibiotics in prevention and treatment. An experimental study. Spine. 1989;14(9):1025–32.
- Polk Jr HC, Christmas AB. Prophylactic antibiotics in surgery and surgical wound infections. Am Surg. 2000;66(2):105–11.
- Osti OL, Fraser RD, Vernon-Roberts B. Discitis after discography. The role of prophylactic antibiotics. J Bone Joint Surg Br. 1990;72(2):271–4.
- Aprill C. Diagnostic disc injections: II. Diagnostic lumbar disc injection. In: The adult spine. Principles and practice. 2nd ed. Philadelphia: Lippincott-Raven Publisher; 1997. p. 539–62.
- Endres S, Bogduk N. Lumbar disc stimulation. In: Bogduk N, editor. International Spine Intervention Society: practice guidelines and protocols. San Francisco: International Spine Intervention Society; 2005.
- Klessig HT, Showsh SA, Sekorski A. The use of intradiscal antibiotics for discography: an in vitro study of gentamicin, cefazolin, and clindamycin. Spine. 2003;28(15):1735–8.
- Carrino JA, Morrison WB. Discography: current concepts and techniques. Appl Radiol. 2002;31:32–40.
- Aprill C. Diagnostic disc injections: I. Cervical disc injection. In: The adult spine: principles and practice. 2nd ed. Philadelphia: Lippincott-Raven Publishers; 1997. p. 523–38.
- Fenton DS, Czervionke. Discography. In: Fenton DS, Czervionke, editors. Image guided spine intervention. Philadelphia: Saunders; 2003. p. 227–55.
- Landers MH. Discography. In: Waldman SD, editor. Pain Management, 2nd Edition. Philadelphia, PA: WB Saunders; 2011. p. 116–38.

- 35. Vivian D, Landers MH. Discography: Intervertebral Disc Access and Stimulation: Lumbar, Thoracic, and Cervical. In: Lennard TA, editor. Pain Procedures in Clinical Practice, 3rd edition. Philadelphia, PA: Elsevier; 2011. p. 418–40.
- Endres S, Bogduk N. Lumbar disc stimulation. In: Bugdok N, editor. Practice guidelines for spinal diagnostic and treatment procedures. San Francisco: International Spine Intervention Society; 2004. p. 20–46.
- Day PL. Lateral approach for lumbar diskogram and chemonucleolysis. Clin Orthop Relat Res. 1969;67: 90–3.
- Edholm P, Fernstrom I, Lindblom K. Extradural lumbar disk puncture. Acta Radiol Diagn (Stockh). 1967;6(4):322–8.
- Drummond GB, Scott DH. Deflection of spinal needles by the bevel. Anaesthesia. 1980;35(9):854–7.
- 40. Sitzman BT, Uncles DR. The effects of needle type, gauge, and tip bend on spinal needle deflection. Anesth Analg. 1996;82(2):297–301.
- 41. Dreyfuss P. The power of bevel control. In: International Spinal Injection Society Scientific (ISIS) Newsletter. San Francisco: International Spine Intervention Society; 1998. p. 16.
- Kumar N, Agorastides ID. The curved needle technique for accessing the L5/S1 disc space. Br J Radiol. 2000;73(870):655–7.
- Derby R. Lumbar discometry. Sci Newsl Int Spine Injection Soc Newsl. 1993;1(5):8–17.
- 44. Derby R, et al. The ability of pressure-controlled discography to predict surgical and nonsurgical outcomes. Spine. 1999;24(4):364–71.
- Carragee EJ, et al. False-positive findings on lumbar discography. Reliability of subjective concordance assessment during provocative disc injection. Spine. 1999;24(23):2542–7.
- 46. Carragee EJ, et al. The rates of false-positive lumbar discography in select patients without low back symptoms. Spine. 2000;25(11):1373–80.
- Endres S, Bogduk N. Lumbar disc stimulation. In: International Spine Intervention Society (ISIS) practice standards and protocols. San Francisco: International Spine Intervention Society; 2003.
- Seo K-S, et al. In vitro measurement of pressure differences using manometry at various injection speeds during discography. Spine J. 2007;7(1):68–73.
- Adams MA, Dolan P, Hutton WC. The stages of disc degeneration as revealed by discograms. J Bone Joint Surg Br. 1986;68(1):36–41.
- Vanharanta H, et al. The relationship of pain provocation to lumbar disc deterioration as seen by CT/discography. Spine. 1987;12(3):295–8.
- Derby R, et al. The relation between annular disruption on computed tomography scan and pressurecontrolled diskography. Arch Phys Med Rehabil. 2005;86(8):1534–8.
- Aprill C, Bogduk N. High-intensity zone: a diagnostic sign of painful lumbar disc on magnetic resonance imaging. Br J Radiol. 1992;65(773):361–9.

- Sachs BL, et al. Dallas discogram description. A new classification of CT/discography in low-back disorders. Spine. 1987;12(3):287–94.
- Walsh TR, et al. Lumbar discography in normal subjects. A controlled, prospective study. J Bone Joint Surg. 1990;72(7):1081–8.
- Vanharanta H, et al. Pain provocation and disc deterioration by age. A CT/discography study in a low-back pain population. Spine. 1989;14(4):420–3.
- Thomas PS. Image-guided pain management. Philadelphia: Lippincott-Raven; 1997.
- Holt EP. The question of lumbar discography. J Bone Joint Surg Am. 1968;50(4):720–6.
- Carragee EJ, et al. Provocative discography in patients after limited lumbar discectomy: a controlled, randomized study of pain response in symptomatic and asymptomatic subjects. Spine. 2000;25(23):3065–71.
- Ketterer MW, Buckholtz CD. Somatization disorder. J Am Osteopath Assoc. 1989;89(4):489–90. 495.
- Zoccolillo MS, Cloninger CR. Excess medical care of women with somatization disorder. South Med J. 1986;79(5):532–5.
- Tsuji N, Igarashi S, Koyama T. Spinal epidural abscess – report of 5 cases. No Shinkei Geka. 1987; 15(10):1079–85.
- Junila J, Niinimaki T, Tervonen O. Epidural abscess after lumbar discography. A case report. Spine. 1997;22(18):2191–3.
- Agre K, et al. Chymodiactin postmarketing surveillance. Demographic and adverse experience data in 29,075 patients. Spine. 1984;9(5):479–85.
- 64. Guyer RD, et al. Discitis after discography. Spine. 1988;13(12):1352–4.
- 65. Arrington JA, et al. Magnetic resonance imaging of postdiscogram discitis and osteomyelitis in the lumbar spine: case report. J Fla Med Assoc. 1986;73(3): 192–4.
- Modic MT, et al. Vertebral osteomyelitis: assessment using MR. Radiology. 1985;157(1):157–66.
- Ledermann HP, et al. MR imaging findings in spinal infections: rules or myths? Radiology. 2003;228(2): 506–14.
- 68. Szypryt EP, et al. A comparison between magnetic resonance imaging and scintigraphic bone imaging in the diagnosis of disc space infection in an animal model. Spine. 1988;13(9):1042–8.
- Aprill C. Diagnostic disc injections. In: Aprill C, editor. The adult spine: principles and practice. Philadelphia: Lippincott-Raven; 1997. p. 523–38.
- Baker AS, et al. Spinal epidural abscess. N Engl J Med. 1975;293(10):463–8.
- Ravicovitch MA, Spallone A. Spinal epidural abscesses. Surgical and parasurgical management. Eur Neurol. 1982;21(5):347–57.
- Lownie SP, Ferguson GG. Spinal subdural empyema complicating cervical discography. Spine. 1989; 14(12):1415–7.
- 73. Grubb SA, Lipscomb HJ, Guilford WB. The relative value of lumbar roentgenograms, metrizamide myelography, and discography in the assessment of

patients with chronic low-back syndrome. Spine. 1987;12(3):282-6.

- Johnson RG. Does discography injure normal discs? An analysis of repeat discograms. Spine. 1989;14(4): 424–6.
- 75. Freeman BJC, et al. Does intradiscal electrothermal therapy denervate and repair experimentally induced posterolateral annular tears in an animal model? Spine. 2003;28(23):2602–8.
- Boswell MV, Wolfe JR. Intrathecal cefazolin-induced seizures following attempted discography. Pain Physician. 2004;7(1):103–6.
- Resnick DK, Malone DG, Ryken TC. Guidelines for the use of discography for the diagnosis of painful degenerative lumbar disc disease. Neurosurg Focus. 2002;13(2):1–9.
- Antti-Poika I, et al. Clinical relevance of discography combined with CT scanning. A study of 100 patients. J Bone Joint Surg Br. 1990;72(3):480–5.
- Slipman CW, et al. Side of symptomatic annular tear and site of low back pain: is there a correlation? Spine. 2001;26(8):E165–9.
- Smith SE, et al. Outcome of unoperated discogrampositive low back pain. Spine. 1995;20(18):1997– 2000. discussion 2000-1.
- Milette PC, Melanson D. A reappraisal of lumbar discography. J Can Assoc Radiol. 1982;33(3):176–82.
- Manchikanti L, et al. Influence of psychological factors on the ability to diagnose chronic low back pain of facet joint origin. Pain Physician. 2001;4(4): 349–57.
- Schwarzer AC, et al. The prevalence and clinical features of internal disc disruption in patients with chronic low back pain. Spine. 1995;20(17):1878–83.
- Osti OL, Fraser RD. MRI and discography of annular tears and intervertebral disc degeneration. A prospective clinical comparison. J Bone Joint Surg Br. 1992;74(3):431–5.
- Milette PC, Raymond J, Fontaine S. Comparison of high-resolution computed tomography with discography in the evaluation of lumbar disc herniations. Spine. 1990;15(6):525–33.
- Bernard TN. Lumbar discography followed by computed tomography. Refining the diagnosis of low-back pain. Spine. 1990;15(7):690–707.
- Ohnmeiss DD, Vanharanta H, Guyer RD. The association between pain drawings and computed tomographic/discographic pain responses. Spine. 1995;20(6):729–33.
- Schellhas KP, et al. Lumbar disc high-intensity zone. Correlation of magnetic resonance imaging and discography. Spine. 1996;21(1):79–86.
- Saifuddin A, et al. The value of lumbar spine magnetic resonance imaging in the demonstration of annular tears. Spine. 1998;23(4):453–7.
- Kang CH, et al. Can magnetic resonance imaging accurately predict concordant pain provocation during provocative disc injection? Skeletal Radiol. 2009;38(9):877–85.

- Braithwaite I, et al. Vertebral end-plate (Modic) changes on lumbar spine MRI: correlation with pain reproduction at lumbar discography. Eur Spine J. 1998;7(5):363–8.
- Simmons JW, et al. Awake discography. A comparison study with magnetic resonance imaging. Spine. 1991;16(6 Suppl):S216–21.
- Zucherman J, et al. Normal magnetic resonance imaging with abnormal discography. Spine. 1988; 13(12):1355–9.
- 94. Derby R, et al. Comparison of discographic findings in asymptomatic subject discs and the negative discs of chronic LBP patients: can discography distinguish asymptomatic discs among morphologically abnormal discs? Spine J. 2005;5(4):389–94.
- Shin D, et al. Diagnostic relevance of pressure-controlled discography. J Korean Med Sci. 2006;21:911–6.
- 96. Manchikanti L, et al. Provocative discography in low back pain patients with or without somatization disorder: a randomized prospective evaluation. Pain Physician. 2001;4(3):227–39.
- Carragee EJ, et al. Provocative discography in volunteer subjects with mild persistent low back pain. Spine J. 2002;2(1):25–34.
- Manchikanti L, et al. Evaluation of the relative contributions of various structures in chronic low back pain. Pain Physician. 2001;4(4):308–16.
- Sambrook PN, MacGregor AJ, Spector TD. Genetic influences on cervical and lumbar disc degeneration: a magnetic resonance imaging study in twins. Arthritis Rheum. 1999;42(2):366–72.
- Jacobs B, Ghelman B, Marchisello P. Coexistence of cervical and lumbar disc disease. Spine. 1990;15(12): 1261–4.
- 101. Carragee EJ, et al. Does discography cause accelerated progression of degeneration changes in the lumbar disc: a ten-year matched cohort study. Spine (Phila Pa 1976). 2009;13(21):2338–45.
- 102. Ohtori S, et al. No acceleration of intervertebral disc degeneration after a single injection of bupivacaine in young age group with follow-up of 5 years. Asian Spine J. 2013;7(3):212–7.
- 103. Martin JT, et al. Needle puncture injury causes acute and long-term mechanical deficiency in a mouse model of intervertebral disc degeneration. J Orthop Res. 2013;31(8):1276–82.
- 104. Bogduk N. The innervation of the lumbar spine. Spine (Phila Pa 1976). 1983;8(3):286–93.
- 105. Freemont AJ, et al. Nerve growth factor expression and innervation of the painful intervertebral disc. J Pathol. 2002;197(3):286–92.
- 106. Brown MF, et al. Sensory and sympathetic innervation of the vertebral endplate in patients with degenerative disc disease. J Bone Joint Surg Br. 1997;79(1):147–53.
- 107. Burke JG, et al. Intervertebral discs which cause low back pain secrete high levels of proinflammatory mediators. J Bone Joint Surg Br. 2002;84(2): 196–201.

- Cunha JM, et al. Cytokine-mediated inflammatory hyperalgesia limited by interleukin-1 receptor antagonist. Br J Pharmacol. 2000;130(6):1418–24.
- 109. Hirsch C. An attempt to diagnose the level of a disc lesion clinically by disc puncture. Acta Orthop Scand. 1948;18:132–40.
- 110. Roth DA. Cervical analgesic discography. A new test for the definitive diagnosis of the painful-disk syndrome. JAMA. 1976;235(16):1713–4.
- 111. Alamin T. Discography versus functional analgesic discography: comparative results and post-operative outcomes. In: International Society for the Study of the Lumbar Spine. Bergen: International Society for the Study of the Lumbar Spine; 2006.
- 112. Alamin T, Arawal V, Carragee EJ. FAD versus provocative discography: comparative results and postoperative clinical outcomes. Proceedings of the NASS 22nd Annual Meeting. Spine J. 2007;7:39S–40.
- 113. Alamin TF, Kim MJ, Agarwal V. Provocative lumbar discography versus functional anesthetic discography: a comparison of the results of two different diagnostic techniques in 52 patients with chronic low back pain. Spine J. 2011;11(8):756–65.
- 114. DePalma M. Are outer annular fissures stimulated during diskography the source of diskogenic lowback pain? An analysis of analgesic diskography data. Pain Med. 2009;10(3):488–94.
- 115. Derby R, et al. Comparison of four different analgesic discogram protocols comparing the incidence of reported pain relief following local anesthetic injec-

tion into concordantly painful lumbar intervertebral discs. Pain Med. 2012;13(12):1547–53.

- 116. Carragee EJ, et al. A gold standard evaluation of the "discogenic pain" diagnosis as determined by provocative discography. Spine. 2006;31(18):2115–23.
- 117. Colhoun E, et al. Provocation discography as a guide to planning operations on the spine. J Bone Joint Surg Br. 1988;70(2):267–71.
- Blumenthal SL, et al. The role of anterior lumbar fusion for internal disc disruption. Spine. 1988;13(5): 566–9.
- 119. Lettice JJ, et al. Does the number of levels affect lumbar fusion outcome? Spine. 2005;30(6):675–81.
- 120. Mahomed NN, et al. The importance of patient expectations in predicting functional outcomes after total joint arthroplasty. J Rheumatol. 2002;29(6): 1273–9.
- 121. Cooper G, Kahn S, Lutz G. Predictive value of provocative lumbar disc stimulation. In: International Spine Intervention Society. Las Vegas: International Spine Intervention Society; 2008. p. 174–9.
- 122. Ohtori S, Kinoshita T, Nakamura S. Surgical results for discogenic low back pain randomized study using discography versus discoblock. In: Proceedings of International Society for Study of the Lumbar Spine. Geneva: International Society for Study of the Lumbar Spine; 2008. p. 59.
- 123. Derby R, et al. Pressure-controlled lumbar discography in volunteers without low back symptoms. Pain Med. 2005;6(3):213–21.

# Modic Changes and Symptomatic Lumbar Degenerative Disk Disease: Is There Any Correlation?

## João Luiz Pinheiro-Franco and Philippe Esposito

## 9.1 Introduction

Low back pain (LBP) is the most common health problem for individuals between the ages of 20 and 50 years [1]. The lifetime prevalence of LBP is approximately 80 % [2]. According to the United Nations, the population is estimated to increase from 13 % of those older than 60 years now to 20 % by 2050. One out of every ten persons is now 60 years or older; by 2050, one out of five will be 60 years or older [3]. This exponential increase in age carries direct correlation with the increase of LBP. Consequently, chronic physical disability originating from LBP will become a public health priority [4]. Chronic LBP carries tremendous socioeconomic and psychological impact in modern societies.

Diagnosis of LBP is challenging. Diverse structures in the lumbar and adjacent regions have nociceptors and are, therefore, potential pain generators. Only about 20 % of LBP diagnoses can directly be linked to a specific pathoanatomical entity [5]. A primary source of LBP,

Samaritano Hospital, Sao Paulo, Brazil e-mail: jlpfranco@uol.com.br

P. Esposito, MD

lumbar degenerative disk disease (DDD), is intimately related to aging and oftentimes may not be symptomatic. LBP due to DDD is often referred in the nomenclature as "nonspecific LBP" differentiating it from specific, e.g., tumoral, infectious, inflammatory, and traumatic, causes of LBP. The term "nonspecific LBP," which accounts for 80 % of diagnoses, however, is not a satisfying diagnosis for both the patient and physician [6].

Lumbar pain in the setting of DDD may be a symptom of many complex and intricate underlying processes with multiple variables involved. The identification and diagnosis of relevant subgroups of patients with persistent, chronic LBP, preferably with a sound patho-anatomical basis, is strongly needed. Attempts to subclassify patients with LBP may result in a better targeting of treatment [7].

The aging of the functional spinal unit (intervertebral disk + facet joints) is related to LBP. The disk has its nourishment supplied by the cartilaginous end plates. Working as a double way crossroad and permitting disk nutrition and elimination of metabolites, the vertebral end plates have a crucial place in maintaining disk health. So far, the exact place of vertebral end plate degeneration in the degenerative cascade is not fully understood. Recently awarded research demonstrated that the histological degenerative changes found in the vertebral end plates preceded the degenerative changes of the nucleus pulposus [8].

J.L. Pinheiro-Franco, MD (🖂)

Neurological Surgery and Spine,

Clinique des Diaconesses, Spine Surgical Center -Clinique du Diaconat, Strasbourg, France e-mail: p.espo@wanadoo.fr

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_9

This chapter attempts to (1) understand how the vertebral end plate signal changes (VESC) on magnetic resonance imaging (MRI) – named Modic changes (MC) – especially the inflammatory type 1 MC (M1) may reflect disturbances in the dynamics of the unit intervertebral diskcartilaginous end plate-vertebral bone marrow and (2) to analyze correlations between LBP and MC.

Whether vertebral end plate degeneration, in association with mechanical and genetic factors, precedes nuclear degeneration, it remains to be definitively proved. We indeed emphasize the necessity of understanding vertebral end plate function and dysfunction and the need to comprehend eventual associations between their assumed mirror MRI images (Modic changes) and clinical manifestations. Despite conflicting evidence in the literature that MC may or may not be associated with clinical findings, a wide review of the facts is necessary.

As clinicians, we have a duty to apply basic scientific principles in our practice. In physical science, just one validated exception is sufficient to invalidate any theory. In biological science, however (where individual "natural" variation prevails), less rigor must frequently be tolerated [9]. This may be the case of MC and LBP.

## 9.2 Lumbar DDD: A Physiopathological Review

A detailed analysis of the physiopathological processes of DDD is beyond the scope of this chapter and is discussed elsewhere in this book. A brief overview, however, is necessary.

Disk degeneration is a natural phenomenon of the aging spine, and the distinction between physiological and pathological DDD is not obvious. DDD is deemed to be induced mechanically and mediated by biochemical responses, often concurrent with aging and probably influenced by genetic particularities [4]. Identification of apoptotic cells in intervertebral disks indicates that programmed cell death may also play a non-negligible role in the pathophysiology of disk degeneration [10–13]. The intervertebral disk is a highly specialized structure working as a fixed cushion that resists loads and allows controlled movement between vertebrae. The end plates are considered to be a dynamic barrier. The movement of molecules through osmosis allows the passage of nutrients and the elimination of metabolites [14], in such a manner that local metabolism acts in equilibrium. Disk degeneration may result from an imbalance between the anabolic and catabolic processes or the loss of steady-state metabolism. The dynamics of such end plate barrier is thought to play a key role in the onset and progression of degeneration, but the biochemical progression that regulates these changes remains poorly understood.

Disk nutrition disturbance has therefore a definitive importance in DDD process. The normal disk is the largest non-vascularized structure in the adult human body [15], and its nutrition is supplied through the vertebral end plates and thin cartilaginous plates between the vertebrae surface and the disk. The mineralized portion of the end plate is penetrated by marrow contact channels (MCC), through which capillary buds emerge. These capillary buds connect the trabecular spaces to the cartilaginous end plate, but do not penetrate into it [16]. For small solutes, diffusion may predominate, but transport of larger solutes such as proteins are believed to be significantly facilitated by the bulk flow of fluid [17] through MCC. End plate blood vessels have been observed to diminish after the first decade of life when the first signs of disk degeneration are evident [18]. Also, reduced permeability of end plates is generally seen in association with DDD and age-related changes [16]. Both of these changes may be due to calcification of the end plates and occlusion of the MCCs observed with disease and age [19–21].

The biochemical composition of the end plates, from normality through the spectrum of degenerative conditions, has been documented extensively [22]. Benneker et al. [15] demonstrated an indirect correlation between the density of openings in the osseous end plate and the morphologic degeneration grade of the disk. These results support the hypothesis that occlusion of these openings may deprive the cells of nutrients, leading to insufficient maintenance of the extracellular matrix and disk degeneration.

All disk degeneration processes start early in life [8, 23]. A recent extensive study on histologic degeneration noted unequivocal signs of degeneration of disk tissue as early as the first half of the second decade (11–16 years of age) [8, 24]. In small children's disks there is a net of fine capillaries that promotes disk nutrition. During early childhood this blood supply to the vertebral end plate begins to decrease and remains as a small nutrition supporting network in the outer boundaries of the posterior anulus fibrosus. With aging, loss of proteoglycans in the nucleus pulposus occurs which is believed to be a critical factor in DDD [23, 24]. The nucleus becomes dehydrated, and the anulus fibrosus loses its organization. The load bearing in the disk structures changes and, per se, contributes to a continuum of disk degeneration. Progressive DDD may impair the disk's ability to defend against mechanical loads. A recent study demonstrated the sequential histological degenerative changes across nine decades [8]. Generally, these changes affect the end plate first, then the nucleus, and, finally, the anulus. Different spine levels behave likewise, the same sequence [8].

Tremendous changes in cellular phenotype and extracellular matrix composition are associated to DDD [23, 25] and are intimately related. Collagenous disk matrix modifies itself during aging and disk degeneration. The huge phenotype modifications may ultimately lead to disk's biomechanical failure.

Chemical events have a remarkable role in symptomatic DDD. Proinflammatory cytokines such as interleukin-1 (IL-1), interleukin-6 (IL-6), and tumor necrosis factor (TNF) are known mediators of the peripheral inflammatory response [26, 27]. It has been demonstrated that painful disks secrete higher levels of proinflammatory mediators, such as TNF, IL-6, and nerve growth factor (NGF) compared with asymptomatic disks [28, 29]. The nucleus pulposus can produce a formidable range of proinflammatory cytokines [30–35]. The impact of these mediators

in the nucleus pulposus has been confirmed in both animal and human studies of lumbar disk herniation [36]. Although inflammatory mediators seem to be native to the disk, their presence, such as cytokines and proteinases, is also conspicuous in granulation tissues [31, 37]. The influx of cytokines (TNF, IL-6, IL-1) and proteinases (stromelysin) is a common phenomenon in degenerated disks [33, 38, 39]. Degenerative disks also contain significantly upregulated activity of major matrix-degrading enzymes, the matrix metalloproteinases [40, 41]. Despite all scientific advances, the exact pathomechanism leading to increased production of inflammatory mediators by the nucleus pulposus in patients with diskogenic LBP is not fully known [28]. It has been proposed that as some disks degenerate the cells of the nucleus may be exposed to a proinflammatory stimulus leading to the formation of inflammatory degeneration which gives rise to LBP [28].

Intranuclear innervation has been suggested to be involved in pain generation. Painful disks have higher concentrations of sensory nerves than those which are asymptomatic [42]. The sensory nerves in painful disks were found in the nucleus and in the end plates. Nerve ingrowth into degenerated symptomatic disks may be mediated by chemotactic substances released by the degenerating disks [43]. Therefore, a combination of increased production of proinflammatory mediators and innervation of the nucleus pulposus may constitute the basis for diskogenic LBP. Hyperalgesia, which is known to be induced by IL-8 and PGE2 [40], may be responsible in part for diskogenic LBP.

Pain provocation studies have correlated severe LBP with relatively innocuous mechanical stimulation of the outer posterior anulus and end plate [44]. In normal disks, the posterior anulus and its adhering longitudinal ligament are innervated by the sinuvertebral nerve, a mixed autonomic and somatic nerve believed capable of nociception. Nociceptive nerve fibers normally penetrate only the outermost 1–3 mm of the annulus [45, 46]. In painful and disrupted disks, the ingrowth progresses anteriorly into the nucleus [45]. The vertebral end plate has a similar density of innervation [47]. It has been shown that in patients with severe LBP and disk height collapse, there is an increase in the density of sensory nerve fibers into the end plates with associated defects in the end plate cartilage, strongly suggesting that the end plates and vertebral bodies are sources of pain [48]. Hence, the ingrowth of nerves and blood vessels is a significant feature of structurally disrupted disks and appears to be directly, though variably, associated with pain [45].

Other events that may contribute to nerve ingrowth are (1) the loss of hydrostatic pressure of an intact disk followed by collapse of hollow capillaries [49] and (2) the reduction in proteoglycan content [50] as aggrecan can inhibit nerve ingrowth in vitro [51].

Despite the increasing data on the pathogenesis of disk degeneration, very little is as yet known on any putative markers that identify the early and "active" stages of disk matrix alterations. Advances in imaging techniques may intend to associate histological markers with imaging findings and provide a foundation for an earlier treatment of lumbar DDD.

## 9.3 History

In 1984, 4 years before the classic manuscript, Modic et al. [52] noted a high MR signal intensity (suggesting inflammation or scarring) in the vertebral end plates adjacent to a marked degenerated disk in a patient who underwent chymopapain (CP) injection.

Three years later Hajek et al. [53] studied retrospectively spine MRIs from 120 patients and observed that 60 % of cases depicted localized or spotty bone marrow alterations (usually affecting vertebral bodies adjacent to the end plates) with a characteristic fat signal intensity. The study group was quite heterogeneous including proved metastasis and operated patients. The authors concluded that focal fatty replacement of hematopoietic bone marrow is a common phenomenon at all ages in both sexes, hence a physiologic process.

De Roos et al. [54] published in 1987 a review of lumbar MRIs and observed band-like sheets adjacent to lumbar vertebral end plates in 50 % of 203 degenerated disks. These signs didn't relate to infection or tumor, but were deemed to represent inflammation or ischemic necrosis, as well as fat replacement and fibrosis/sclerosis. The authors noted that 100 % of these findings were associated with DDD [54-57]. Furthermore, patients with lumbar DDD and these signal changes were usually older than patients with lumbar DDD without such changes. Signal changes compatible with fatty replacement were found in 83 % of degenerated disks. These findings support the premise that MC represent natural evolutions inside DDD, often associated with the aging process.

In 1988, Modic et al. [55] developed a classification system of vertebral bone marrow changes based on representative MR T1- and T2-weighted images. Modic type 1 changes (M1) demonstrated hypointense signal findings on T1-weighted images and hyperintense signal changes on T2-weighted images. Modic type 2 changes were characterized by hyperintense signal on T1-weighted images and isointense or slightly hyperintense signal changes on T2-weighted images. Rare type 3 (M3) changes (hypointense signal on both T1- and T2-weighted images) were added to the classification at a later time.

Modic et al. hypothesized that these signal changes reflect a spectrum of marrow changes related to DDD that could be distinguished from vertebral osteomyelitis and other vertebral body abnormalities [55].

## 9.4 Physiopathology of Modic Changes

The pathophysiology of MC remains obscure. The factors that lead to the onset of signal changes in the vertebral bone marrow adjacent to degenerative disks in MRIs are not well understood. It is of general agreement, however, that the physiologic association between the end plates, subchondral bone, and intervertebral disk is of paramount importance to the understanding of DDD [58].

MC and the dual complex of aging/DDD are intimately, but not specifically, related. Mechanical loads, inflammation, genetics, and apoptosis have all been implicated in the genesis of MC and will be briefly discussed.

The cartilaginous end plate degenerates with aging. It is known to decrease in thickness and tear and eventually disappear with aging. Matrix production and accumulation within the cartilaginous end plate (CEP) decrease with aging [19, 22, 59], possibly because of decreased cell viability and reactivity of viable cells within the CEP [60]. The cause of these age-related changes remains unclear.

Programmed cell death has been implicated in this age-related end plate regression. Using a mouse spondylosis model, Ariga et al. [60] investigated the role of apoptosis in DDD. They analyzed operated and non-operated disks and controlled for age. The operated group was considered as having been exposed to marked mechanical stress. Apoptosis, particularly noticeable in the cartilaginous end plates, increased with age and resulted in a marked decrease in cell density. Interestingly, the extensive occurrence of apoptosis seen in the CEP was not observed in the nucleus pulposus. The authors speculated that CEP changes may possibly occur first and may cause the changes in the nucleus pulposus [60].

Ariga et al. [60] concluded that apoptosis was more evident in the surgically treated group than in the naturally aging group. The structure of the end plate began to disappear more rapidly in the surgically treated group supporting the role of mechanical stress in the degenerative process. Their conclusion was in parallel to reports demonstrating that cells within the CEP may be induced into apoptosis by the application of mechanical stress [13]. These observations suggest that acceleration of apoptosis in the CEP is involved in the cascade of events constituting disk degeneration and that mechanical stress may be one of the factors that induce apoptosis within the CEP [60].

Mechanical factors are supposed to induce end plate degeneration. The vertebral end plates were considered the spine's "weak link" in compression. Nachemson [61] affirmed that the degenerative process within disks results in greater axial loading and increased stress on the vertebral end plates. In patients with nonspecific LBP, it was hypothesized that disk degeneration leads to increased loading and shear forces on the end plates, which might lead to fissures of the end plate [62]. Animal models have demonstrated that injury to the disk induces changes in the adjacent vertebrae with subsequent bone marrow depletion and degeneration followed by bone regeneration [58, 63, 64].

It is speculated that MC may represent vertebral bone marrow reactions occurring after end plate damage or a bone marrow response to an acute or chronic inflammatory reaction in the disk.

Whatever might be the origin of MC, the importance of the end plate and subchondral bone [65] on the integrity of the intervertebral disk has been firmly established. The normal nucleus pulposus is designed to distribute weightbearing forces evenly onto the adjacent vertebral bodies. When the normal weightbearing properties of the nucleus pulposus are altered, weightbearing forces on the spine become distributed unevenly on the adjacent vertebrae, resulting in microfractures and bone necrosis [58]. These events result in inflammation of the adjacent bone marrow. Microtrauma to the end plate would permit contact between chemical agents in the degenerative disk and cells in the bone marrow [62, 66] and therefore may be considered the initiating inflammatory or autoimmune event in a progressive deteriorating process.

End plate damage decompresses the adjacent nucleus and transfers load onto the anulus, causing it to bulge into the nucleus cavity [67, 68]. Accumulating trabecular microdamage [69] probably explains why the nucleus increasingly bulges into the vertebral bodies in later life [70]. If the nucleus pulposus herniates through a damaged end plate, then subsequent calcification may create a "Schmorl's node."

MC are more frequent in lower lumbar levels [71]. This provides indirect evidence that there is a mechanical component to MC [71, 72]. A

mechanical cause is also supported by the higher prevalence of DDD in the lumbar rather than the cervical or thoracic spine [73]. This is also supported by a report associating MC changes with physical loading [74].

## 9.4.1 Surgical Rupture of the End Plate

Animal models demonstrate that surgical disruptions of the end plate or anulus will inexorably provoke degenerative changes in all disk structures [58, 68, 75]. Perforation of the end plate from the side of the vertebral body causes nucleus decompression, proteoglycan loss, and internal disruption of the annulus [68]. The anulus disruption model, which simulates a peripheral rim tear, causes subsequent changes in the nucleus and end plate [63, 75]. An injury to the end plate has been found to provide a potent mechanical stimulus for disk degeneration [66] and end plate trauma following fracture. It has been suggested to cause accelerated intervertebral disk degeneration [76].

## 9.4.2 End Plate Fracture

A noteworthy hypothesis regarding MC origin concerns end plate fractures. Histopathologic studies have confirmed the existence of end plate cartilage in herniated disks. This kind of herniation, described as "avulsion of end plate cartilage," represents the predominant histopathologic type of disk herniation in the elderly population [77]. These avulsion fractures of the cartilaginous end plate take place in the middle third of the sagittal diameter of the end plate [77].

Clinical relevance of cartilage material in disk herniation remains unclear. Schmid et al. [78] demonstrated that MC are associated with cartilage material in the extruded disk herniation. They compared preoperative MC and corresponding herniated disk tissue (n=51, mean age=40, range 17–62) and found 59 % of preoperative MC adjacent to surgically treated disks (M1 in 11 % and M2 in 47 % of patients). There was a significant statistical correlation between MC extending over 33 % of the vertebral end plate and the presence of cartilaginous end plate material in the extruded disk. End plate cartilage avulsion would theoretically provoke a reaction in the adjacent bone marrow, resulting in MC.

It has been hypothesized that if hyaline cartilage from the vertebral end plate was found in herniated disk material, the end plate avulsion may have preceded the process of disk herniation [78]. Therefore, vertebral end plate signal intensity changes may be regarded as osteocartilaginous fracture signs similar to other skeletal manifestations. This data supplemented the knowledge that signal intensity changes along the end plates may be caused not only by edema, fissuring of the end plates, and formation of granulation tissue but also in part by end plate avulsion. The end plate cracks/fractures/avulsions may cause inflammation in the adjacent bone marrow with corresponding M1 signs on MRI.

Histopathologic studies [79-81] demonstrated that hyaline cartilage from the end plate present in an extruded disk material may suppress the process of neovascularization. Neovascularization is important for extruded disk material absorption. Therefore, the presence of hyaline cartilage from the end plate present in an extruded disk material would suppress disk absorption. M1 changes have been associated with the presence of hyaline cartilage in an extruded disk; therefore, an extruded disk is less likely to disappear when associated with adjacent M1 changes. It may be hypothesized that patients with M1 changes and an extruded disk would have less benefit from conservative treatment since the disk would probably not be reabsorbed. Patients without M1 changes and an extruded disk may respond better to conservative treatment [78].

## 9.4.3 Chymopapain and the Brutal Acceleration of Disk Degeneration

The theory exposed above may explain some consequences of chymopapain (CP) injection into intervertebral disks. CP injection for the treatment of disk herniations can result in M1 inflammatory signs [55]. An accepted model of accelerated disk degeneration, intradiscal CP, provokes acute disk desiccation. This "shrinking" process may result in the acute detachment of the vertebral end plate. This causes an abrupt interruption of disk nutrition through the acute loss of end plate vascularization. Disk metabolism has adapted itself to chronic slow degenerative changes but does not seem adapted for sudden changes in its local environment. Sudden changes to the disk architecture may provoke acute inflammation.

Meanwhile, a normal nucleus is supposed to distribute weightbearing forces evenly onto the adjacent vertebral bodies. When the normal weightbearing properties of the nucleus pulposus are altered, weightbearing forces on the end plates become distributed unevenly, resulting in microfractures and bone necrosis [58]. These events would result in inflammation of the adjacent bone marrow.

When the process is chronic, a process of end plate healing ensues followed by stabilization of disk function [82]. This theory would in part explain why M2 anomalies (chronic, fatty replacement) are less seldom associated with pain production. Bone remodeling is thought to be induced by changes in the mechanical stress on vertebral bodies caused by DDD. Such stresses may result in pathological changes in the vertebral bone marrow. An insufficient healing process, with impaired bone remodeling, beginning with bone marrow edema and necrosis [65, 71] may result in an end plate defect. Inflammation within the end plates with M1 subchondral signal changes has been suggested to be induced by TNF [83].

#### 9.4.4 Genetics and MC

Genetic evidence of MC prevalence was observed by Karppinen et al. [84]. These authors studied whether genetic factors were associated with MC in vertebral lumbar end plates and noted an association between a combination of IL1 and MMP3 gene variations and M2.

#### 9.4.5 Inflammation

Inflammation is suggested to play a formidable role in MC as it does in DDD. The inflammatory nature of M1 was recently confirmed. Ohtori et al. [28, 83] presented immunohistological evidence that M1 is associated with inflammation of the vertebral end plates. The study suggested that TNF expression in chondrocytes gives rise to inflammation in the end plate of patients with M1. Furthermore, they noted that an increase in nerve fibers within the vertebral end plates in patients with M1 and M2 changes may be a cause of LBP, suggesting that the pain may originate from abnormal end plates [83].

Burke et al. [85] demonstrated that MC in patients with symptomatic (painful) disks are associated with high levels of production of proinflammatory mediators. These observations support their part as an objective marker of diskogenic LBP [85]. These authors studied the levels of IL-6, IL-8, and prostaglandin E2 (PGE2) in the disks of a surgically treated cohort of patients with sciatica (n=40, with 12% of M1) and a surgically treated cohort of patients with (positive discography) diskogenic LBP (n=12, with 40 % of M1). There was a statistically significant difference between levels of IL-6, IL-8, and PGE2 production by M1 and M2 groups compared with the group without MC. The authors speculated that these findings were firm arguments favoring the link between MC and pain. M1 was more common in patients with diskogenic LBP, whereas M2 changes occurred in 50 % of patients who underwent surgery for sciatica. The cause of this high prevalence of M2 in the sciatic group is unknown.

MC may be considered in at least some cases (probably M1) a secondary sign of "internal disk disruption" [72].

## 9.4.6 Are Modic Changes a Result of Infection?

Albert et al. performed a double-blind RCT with 162 patients whose only known illness was chronic LBP of greater than a 6-month duration occurring after a previous disk herniation and who also had M1 changes in the vertebrae adjacent to the previous herniation [86].

Patients were randomized to either 100 days of antibiotic treatment (Bioclavid) or placebo and were blindly evaluated at baseline, at end of treatment, and at 1-year follow-up. The antibiotic protocol in their study was significantly more effective for the LBP group (chronic LBP associated with M1) than the placebo in all the primary and secondary outcomes.

This subject remains a matter of debate. Recently, several issues concerning the previous study were pointed out [87]: very high treatment effect vs no benefit at all in the placebo arm; arguments about placebo masking (patient can detect whether or not they are receiving antibiotics); probably underpowered as two different antibiotic doses tested; no evidence of culturepositive infection at baseline, so a Koch postulate (the microorganism must be isolated from a diseased organism and grown in pure culture) fails if no known means of culture organisms (e.g., prions) are not fulfilled; is this optimum antibiotic regime, should we focus on P. acnes or other microflora as well?; precisely equal numbers in each group weakens the case for complete blinding, no description of skin preparation regime of culture from wound edges. Finally there is a concern relative to use of 100 days of antibiotics, that it would be maybe acceptable to treat diskitis but not if treating unconfirmed infection: with many public health/safety issues attending extended courses of antibiotics.

## 9.5 Epidemiology of Modic Changes

The prevalence of MC varies greatly between studies, from 19 to 60 % [54–56, 72, 74, 82, 85, 88]. This wide range in reported prevalence rates may be due to the high variation in study design including patient selection, age and sex distribution, clinical (surgical vs nonsurgical) or nonclinical (asymptomatic) series, clinical series of patients with LBP and/or sciatica, or occupational particularities. The real MC prevalence in

asymptomatic and symptomatic populations is yet to be determined.

Jensen et al. [73] performed a systematic literature review to investigate the relation of the prevalence of vertebral end plate sign changes (VESC) (including MC) and the association with nonspecific LBP. The median prevalence of the reported prevalence rates for any type of VESC (M1 + M2) was 43 % in patients with nonspecific LBP and/or sciatica and 6 % in nonsymptomatic populations. The prevalence was positively associated with age and was negatively associated with the overall quality of the studies. A positive association between VESC and nonspecific LBP was found in seven of ten studies describing general, working clinical populations. VESC was a common MRI finding in patients with nonspecific LBP and was associated with pain. However, it may be present in individuals without LBP.

Weishaupt et al. [88] confirmed this finding demonstrating that MC were uncommon in 60 asymptomatic volunteers aged 20–50 years (mean age 35). According to the authors, the low prevalence of MC supported the hypothesis that MC may be predictive of LBP. Kjaer et al. [57] did a well-conducted study of 412 Danes older than 40 years old divided into three groups: MC and DDD, DDD but no MC, and the absence of DDD and MC. They observed a 15 % incidence of M1 and 7 % incidence of M2 for a total MC prevalence of 22 %. They concluded that there is an association between MC and LBP.

Most studies have demonstrated an association between the presence of MC and DDD [53–55, 72]. However, MC may also be found in the absence of DDD. Kjaer et al. [57] observed MC in 9 % of 198 non degenerated disks. Modic et al. [55] retrospectively reviewed lumbar MRIs of 474 consecutive patients referred for LBP or sciatica. They observed 4 % of M1 and 16 % of M2. Every MC was adjacent to a degenerated disk. The reverse, however, is not true, as only a minority of degenerated disks present with MC [57]. These relations are therefore not understood. The fact that MC have great specificity but a relative small sensitivity for the presence of DDD remains unclear. MC have great specificity but a relative small sensitivity for the presence of DDD is unclear. It is not known why some degenerative disks are associated with MC, while others are not [72]. MC (especially M2) may correspond or represent at least part of the natural evolution of DDD. The association between MC and LBP seems to be stronger than that between DDD and LBP [57].

In the same line of previous studies, there is general agreement that vertebral marrow signal changes occur with increasing frequency with age [55, 89, 90]. This seems plausible as VESC has been correlated with disk degeneration [55, 57], which in turn is correlated with age [91].

Other variables were also analyzed. There was no apparent difference regarding gender. Kjaer et al. [74], in turn, found no significant gender difference among those with MC. They noted a 23 % prevalence of MC among 40-year-old men and women. A recent study, however, observed that MC were significantly associated with the male gender [90].

Most studies have demonstrated a predominance of M2 changes [54–56, 72, 85, 89, 92]. Hajek et al. [53] noted an age-related prevalence of "focal fat deposition" (M2). They observed these changes in 52 % of patients aged 31–40 years, 93 % in those aged 51–60 years and 100 % of those older than 60 years of age. It is known that there is a physiologic conversion of red marrow to fat marrow throughout life. A few studies have noted a predominance of M1 [57, 72, 93, 94]. Mixed forms and M3 were more rarely reported in the literature. The prevalence of mixed M1/M2 anomalies is estimated at approximately 8 % [93].

#### 9.6 Reproducibility

Weighted kappa statistics are generally performed to analyze both intraobserver reliability and interobserver reproducibility. The kappa coefficient represents the percentage of instances of agreement, while the likelihood of agreement based on chance alone is taken into account. A kappa coefficient of 1.00 indicates perfect agreement, whereas a kappa coefficient of 0.00 implies no more agreement than would be expected by chance alone. Recent studies have demonstrated that the Modic classification is both reliable and reproducible. High kappa values were demonstrated in the identification of MC [95]. These MRI signal changes were also comparable between specialties [95].

Jensen et al. [93] evaluated MRI scans of 50 individuals representative of the general Danish population aged approximately 40 years of age. Intra- and interobserver agreement of the detailed evaluation of variables describing vertebral signal changes (Modic type, location, volume, maximum height, and end plate area) were all found to have substantial to almost perfect agreement.

VESC is an MRI finding easy to evaluate [93] and is supported by studies that have reported the interobserver reproducibility of a detailed evaluation of VESC with kappa values ranging from 0.64 to 0.91 [89, 90, 93, 96, 97].

## 9.7 Bone Marrow Anatomy, Histology, and Physiology

Being MC a reflection of vertebral bone narrow structure, a brief review of its anatomy, histology, and physiology is needed.

## 9.7.1 Bone Marrow Structure and Its Aging Natural History

Vertebral bone marrow consists of red and yellow marrow. "Red marrow" is considered hematopoietically active, being responsible for the production of red cells, white cells, and platelets. "Yellow marrow" is hematopoietically inactive. Red marrow contains approximately 40 % water, 40 % fat, and 20 % protein, while yellow marrow presents 15 % water, 80 % fat, and 5 % protein. There are also structural differences. Conversion of red to yellow marrow occurs during growth and development and has a predictable and orderly pattern [98].

Modic MR changes are MR imaging mirrors of vertebral bone marrow histological structure. While portions of the end plate can be harvested at the time of surgery, the in vivo removal of unadulterated vertebral bone marrow for 146

histopathological study in the setting of lumbar DDD cannot be easily obtained. Thus histological data concerning vertebral bone marrow changes are scarce. To our knowledge there is one cadaver study that correlated MC and vertebral bone marrow histological changes. Hajek et al. [53] performed a macroscopic evaluation of sectioned cadaveric spine specimens and observed well-defined, yellow areas within the bone marrow that corresponded precisely with the localized zones of characteristic fat signal intensity on MR images (M2). Histologic evaluation of these regions demonstrated typical focal fatty replacement of normal hematopoietic marrow. The authors argued that fat in the bone marrow would exclude the presence of inflammation. Later on it would be proved that both fat and inflammation can coexist.

## 9.8 Vertebral End Plate MR Signal Changes (Modic Types)

#### Modic 1 Changes: Inflammation

M1 consists basically of inflammatory changes within the cartilaginous end plate and vertebral bone marrow. Modic et al. [55] examined histologically end plate material in specimens demonstrating M1 and M2 changes. End plate specimens were obtained during lumbar surgery for DDD in three patients with M1 and three patients with M2. M1 histology revealed disruption and fissuring of the end plates with regions of degeneration and regeneration and vascular granulation tissues adjacent to the end plates. There was an increased amount of reactive woven bone with thickened trabeculae and prominent osteoclasts and osteoblasts [55].

In M1 histologic tissue, the adjacent marrow of the vertebral body is replaced by loosely textured fibrous tissue with multiple small blood vessels. A fairly abrupt transition of these changes to normal cellular marrow at a variable distance from the end plate occurred [55]. Vande Berg et al. [99] observed granulation tissue with marked immature fibrosis, vessels, and few areas of subchondral necrosis along with thickened trabeculae. All these changes are compatible with M1 findings on MRI.

Among Modic types, M1 has been suggested mostly to correlate with pain. Kuisma et al. [56] affirmed that the suggestion of pain related to M1 findings is not surprising as this type is thought to have an inflammatory component. Crock [100] suggested that repeated trauma to the intervertebral disk results in the production of inflammatory mediators in the nucleus pulposus and that the diffusion of such toxic chemicals through vertebral end plates could result in a local inflammatory reaction resulting in LBP.

The inflammatory nature of M1 was confirmed in a recent observational immunohistological study on patients with diskogenic LBP [83]. Ohtori et al. [83] evaluated TNF expression and the existence of nerve fibers in the end plate of patients with diskogenic LBP. The results of immunohistochemistry were compared with end plate changes observed by MRI. There were two cohorts: a cohort of 14 patients who had LBP and disk degeneration on MRI and a control group of 4 patients. Vertebral end plates from patients suffering LBP with M1 and M2 changes had significantly more PGP 9.5-immunoreactive nerve fibers and TNFimmunoreactive cells in comparison with patients with normal end plates on MRI (P < 0.01). The number of TNF-immunoreactive cells in end plates exhibiting M1 was significantly higher than in end plates exhibiting M2 (P < 0.05) [83].

The results of their study suggested that TNF expression in chondrocytes results in end plate inflammation in patients with M1. TNF and the resulting inflammation affect sensory nerve fibers causing pain. The authors didn't specifically examine TNF expression and PGP 9.5-immunoreactive nerve fibers in patients with degenerated disks, but they speculated that the presence of nerve ingrowth in the inner disk layers caused diskogenic LBP in patients without MC. In the same line, an additional significant positive correlation was identified for nuclear TNF expression, disk degeneration, and age. TNF expression in the setting of M1 and M2 changes may also induce nerve ingrowth into the end plates [83]. Sensory nerve fibers innervating the vertebral end plate have previously been described [48].

Rannou et al. [101] performed a pilot prospective study of patients with chronic LBP and "no Modic" M1 and M2 changes and their relationship to high-sensitivity C-reactive protein (hsCRP) levels. hsCRP is a sensitive systemic marker of low-grade inflammation (IL-6 is the major upregulator of CRP gene expression). Their conclusion was that hsCRP is increased in patients with chronic LBP and M1, which supports a local inflammation phenomenon occurring at the vertebral end plate level.

### Differential Diagnosis of Inflammatory M1 Changes

Vertebral end plate signal changes consistent with marrow edema may be seen in infectious diskitis [102], accompanying Schmorl's nodes [103] and within 3 months of chemonucleolysis [104]. However, in the absence of such predisposing causes, VESC have only been identified adjacent to degenerated disks [72]. In spondylodiskitis the T2 signal intensity of the disk and end plate is typically increased [105-107], and the end plates are eroded infection [105, 107]. The presence of paraspinal or epidural inflammation and/or a collection [105] as well as the clinical presentation and erythrocyte sedimentation rate and C-reactive protein (CRP) may help guide the diagnosis of an infectious process [105]. CRP is a very reliable indicator of disk infection, being elevated in up to 100 % of patients at the time of diagnosis. Diffusion-weighted imaging (DWI) is a sensitive and fast sequence that offers the possibility of quantifying diffusion coefficients of the lesions, which could help to discriminate between spondyloarthritis axial active inflammation and type 1 Modic changes [108].

#### Modic 2: Fatty Replacement

Histological specimens [55] demonstrated disruption of the end plates with evidence of chronic repetitive trauma (increased reactive bone and granulation tissue) similar to that seen with M1. However, the adjacent marrow was devoid of hematopoietic elements and was replaced by abundant fat (yellow marrow). These regions also showed a return to normal hematopoietic marrow at a variable distance from the end plate. Toyone et al. [82] stated that these changes are common in patients with late-stage DDD, which may represent the "restabilization phase" described by Kirkaldy-Willis and Farfan [109]. Thus, M2 may simply accompany aging or may coexist with the natural healing/restabilization phase of the chronic degenerative process.

## **Modic 3 Changes**

M3 means a sclerotic state, corresponding to sclerosis on radiographs [110]. The presence of sclerosis is better appreciated with plain radiographs than with MRI, as it is a reflection of dense woven bone within the vertebral body. MR imaging reflects or displays the presence of marrow elements such as normal hematopoietic tissue, fibrovascular tissue, and lipid content between the bony trabeculae.

#### **Mixed Forms**

Mixed forms of MC consist of different histological substrates (e.g., M1 inflammatory changes + M2 fat replacement, often referred to as M1/ M2) concurrently in the same vertebral body marrow adjacent to a degenerated disk [72]. Marrow content over time may and will often progress from one to another type. Many studies have confirmed these transformations [55, 72, 111, 112]. The most common type of mixed MC is M1/M2. Longitudinal studies confirmed that often there is a progression from M1 to M2 or from M1/M2 to M2 [112]. This may correspond to a transition from an (less stable) inflammatory state to a more stable (chronic, not meaning pathology) state (M2) [55].

The distinction between M1 and M2 is very specific as inflammation and fat replacement are distinct histopathological configurations. It is therefore obvious to refer to M1 or M2 and not the "generic" "Modic changes" expression, as M1 and M2 represent different histologies. These more specific M1 and M2 definitions are requested when studying the presence of LBP or MC natural history [113].

Inflammatory changes (M1) may be masked if they are superimposed by fatty replacement tissue (M2) [88]. Traditional T1 and T2 sequences may not be precise enough to distinguish between various Modic types [72]. MR accuracy can be improved through supplementary fat suppression techniques followed by contrast injection or STIR (short TI inversion recovery) sequence. Using the first protocol, the sensitivity of detecting M1 changes improves as gadolinium enhances bone marrow inflammation in sites where fat is suppressed. STIR is a sequence with specific timing so as to suppress the signal from fat. These techniques improve the detection of inflammation.

## 9.9 Natural History: Modic Changes

Little is known about the natural history of MC. There are no studies analyzing the natural course of MC in asymptomatic populations. Existing data basically concerns clinical and surgical series [6, 55, 89, 112, 114]. Kuisma et al. [89] performed a longitudinal study in patients with symptomatic sciatica and noted a 6 % incidence over 3 years of new MC. All new cases of MC were primarily M1 and adjacent to herniated disks.

As previously stated, MC may convert from one type to another [55, 72]. The identification of several mixed forms is consistent with this idea and may represent different stages of the same pathological process [72]. It has been suggested that M1 represents an acute stage and may later transform to M2 [55, 112], as the subchondral bone heals. While several clinical and surgical series demonstrated this type of evolution [55, 111, 112, 114], it is not known whether all M2 are preceded by at least a discrete M1.

The mixed forms M1/M2 are thought to be transitory states during this transformation [89, 111]. Modic et al. [55] followed longitudinally 16 patients (6 M1 and 10 M2) in a series of 412 patients referred for MRI due to LBP/and sciatica. Five out of six M1 had at least partial conversion to M2 over a time period of 14 months to 3 years. All six patients with M2 changes remained unchanged.

Mitra et al. [112] observed M1 in 18 % of 670 patients referred for MRI due to LBP or sciatica. These patients were rescanned 12–72 months

later. M1 transformed accordingly: 37 % M1 converted fully to M2, 15 % converted partially to M2, and just 8 % underwent no change. Forty percent of M1 changes increased in size. The authors correlated the evolution of M1 with the patients' symptoms and noted that in patients whose symptoms had improved, M1 transformed into M2. In patients reporting a worsening of their symptoms, M1 changes had progressively worsened. The conditions that govern the transit of a theoretically less stable M1 to a more stable M2 are unknown.

Albert and Manniche [6] observed that M1 changes were closely related to a previous disk herniation. The authors performed a longitudinal study where 181 patients were recruited from a randomized controlled study (RCT) comparing two active conservative treatments. The patients, who at baseline had radicular pain ending or radiating below the knee, had follow-up MRIs 14 months later. The prevalence of M1 increased from 9 to 29 %. M2 and M3 remained unchanged.

Modic et al. [55] believed that M1 didn't seem to invariably transform into M2. M1 could eventually regress to M0 or convert to M3 [71]. In a longitudinal series of sciatic patients, it was observed that no patients evolved from M1 to M2 over a 3-year period, and only two patients with M1/M2 changes converted to M2 [89]. The same study observed that several patients with M1 inflammatory changes regress to M0.

Considering that M2 represents a chronic state and M1 an acute inflammatory state, it is reasonable to think that the change from M2 to M1 would represent a sudden decompensation of the chronic degenerative state that M2 characterizes [115]. A new ongoing acute process may aggravate the degenerative process and provoke M1 inflammatory changes [89]. Marshman et al. [9] described two cases of M2 conversion to M1. Follow-up MRI of one of these patients demonstrated a vertebral slippage appearing concurrently with the transformation of M2 to M1. Luoma et al. [71] observed three cases in which M2 regressed (disappeared or decreased), concomitant with enlargement of M1 changes (conversion of M2 to M1). Kuisma et al. [89] noted eight individuals who had M2 changes which transformed into the intermediate form M1/M2 (six cases) and to M1 (two cases). Some patients evolved from the theoretically more stable M2 state to the less stable M1 state [89]. In contrast to previous findings, these authors suggested that M2 may be less stable than previously assumed. Most transformations occurred at L5–S1 and were associated with a symptomatic disk herniation.

M1, M2, and M3 size may remain stable or increase in size [6, 71, 112]. One series demonstrated that 41 % of M2 changes enlarged over time [71]. The increase in M2 size was more prevalent in those that did not convert to another type [89].

The factors that determine the fate of M1 inflammatory changes are not fully understood.

Luoma et al. [71] studied the natural course of M1 changes. They grouped together M1/M2 and pure M1 findings but labeled them all as M1. Moreover no clinical correlation was made with MC evolution. The study included 1015 consecutive nonspecific chronic LBP patients, and 24 (2,4 %) of the patients had 54 M1 (28 M1 and 26 M1/M2). These 24 patients underwent a followup MRI study within 18-72 months. Almost 100 % of new M2 (22 from 23) evolved from M1, supporting the M1 to M2 evolution hypothesis. However, most (67 %) of the 54 M1 (28 M1 and 26 M1/M2 grouped together) regressed over time, disappearing or undergoing size reduction possibly suggesting a total or partial resolution of the inflammatory process. The longer the interval, the more likely M1 tended to decrease or disappear. It is possible to conclude from this study that most M1 regress with healing and that most M2 evolve from M1. However when considering a general population, M2 would not be expected to always originate from M1 as fatty conversion of the marrow is part of a physiologic process. Another study showed that 34 % of M1 persisted or enlarged over time, and therefore subchondral edema may last for years [111].

Chymopapain injection, due to its mechanism of action of provoking sudden nucleolysis, is considered a model of accelerated disk degeneration. In this scenario M1 inflammatory changes may appear over a time period of 6–12 weeks [55]. M1 changes may therefore be viewed, at least in this clinical situation, as the result of an aggressive acceleration of the physiological degenerative cascade or an abrupt decompensation of the very slow degenerative disk process.

A disk herniation, when aggressively surgically removed (aggressive diskectomy), would theoretically provoke a similar "sudden loss of disk height" effect, resulting in a loss of vertebral body support. In a series [6] of 181 patients with sciatica in which the majority were treated conservatively, 12 patients underwent surgery for a lumbar disk herniation (LDH) during the 1-year follow-up period. Follow-up MRI at 1 year demonstrated a trend (though not statistically significant) for the operated patients to develop MC (especially M1).

Kerttula et al. [116] stated that M1 is a sign of a fast progressing and deforming "pathologic" disk degeneration. These authors affirmed that the accelerated process of degeneration in disks with an adjacent M1 may lead to deforming changes in the diskovertebral unit in a much shorter time than would be expected with agedependent degeneration. They performed a prospective MRI study in chronic LBP patients evaluating the natural course of degenerative lumbar spine changes in relation to M1 (or a mixed Modic change M1/M2 or M1/M3) within 1 year. From 3,811 consecutive chronic LBP patients referred to lumbar spine, MRI 54 patients had M1. Follow-up MRI was obtained within 11-18 months. In follow-up, an unstable M1 was associated both with an increase of end plate lesions, decrease of disk height, and change in disk signal intensity, most found at L4/L5 or L5/ S1. In disk spaces without M1, progression of degenerative changes was rare [116].

#### 9.10 Clinical Correlation

There are no precise clinical, diagnostic, or imaging pathognomonic patterns that define the exact source of LBP. Radiological lumbar degenerative findings do not imply clinical symptoms. Several studies have demonstrated a high prevalence of morphologic abnormalities in both symptomatic and asymptomatic individuals [117, 118].

Most of the studies concerning MC and LBP have been performed in selected series of patients

J.L. Pinheiro-Franco and P. Esposito

[56]. Only one population-based study has focused on the association of MC and LBP [57]. In a population-based sample of 412 forty-yearold Danes, information including MRI findings, patient questionnaires, and clinical examination was collected. Three subgroups of patients were created: those with both DDD and MC, those with only DDD, and those with neither DDD nor MC. Clinical characteristics of each group were investigated to see if there was a distinct difference between subgroups [57]. The authors observed an association between MC and LBP.

MC are uncommon in an asymptomatic adult (aged 20-50 years) population. The low prevalence of end plate abnormalities in the asymptomatic population supports the hypothesis that end plate changes may be predictive of LBP [88]. However, MC are not specific of LBP. Many authors have suggested that M1 is associated with a higher prevalence of LBP [6, 56, 73, 82, 85, 111, 114, 119–121] thought secondary to DDD [6, 55, 57, 72, 82, 120]. Other authors have noted an association of M2 with LBP [6, 9, 56, 72, 94]. M1 may be more strongly associated with pain compared to M2 [6]. Albert and Manniche [122] observed from their outpatients' clinic that a large proportion of patients with persistent LBP had MC and patients with sciatica who were treated conservatively were three times as likely to report LBP if they had developed MC at 14 months of follow-up as compared to those who had not. These authors [6] conducted an RCT of 181 patients with severe sciatica treated with two active conservative treatments. Nonspecific LBP was more frequent in people with M1 compared to those with M2, but the difference between types was not statistically significant. The prevalence of MC was higher in patients who had undergone surgery for a herniated disk. A lumbar disk herniation was a strong risk factor for developing MC (especially type 1) during the following year, and the development of new MC was closely related to the level of a previous disk herniation [6].

Kuisma et al. [56] studied a population of 228 middle-aged male workers (159 train engineers and 69 sedentary controls) with a mean age of 47 years (range 36–56). MC was present in 56 %

of patients, divided into M1=15 % and M2=32 %. Train engineers had on the average higher sciatic pain scores than the sedentary controls, but the prevalence of MC was similar in both occupational groups. The presence of MC was associated with an increased number of LBP episodes, with higher LBP scores during the past week and past 3 months. At specific disk levels, an association of LBP with MC was seen only at L5–S1, with MC being associated with a higher number of previous LBP and sciatica episodes and with a higher LBP score during the past week. MC at the upper lumbar levels or at L4-L5 were not associated with any pain variables [56]. M1 at any level (all levels combined) was related to a higher number of previous LBP episodes, higher LBP scores during the past week and past 3 months, and higher sciatic pain score during the past 3 months. At L5–S1, the associations were similar. At the upper levels, analyzed separately, the authors found no association between pain symptoms and M1 and M2 changes at any level (all levels combined). At L5-S1, M2 changes were associated with a higher number of previous LBP episodes.

The authors concluded that MC at L5–S1 and M1 are more likely associated with pain than other types of MC or changes located at other lumbar levels. The importance of this study was that it was the first to correlate occupation and MC [56]. The authors speculated that the association of LBP with the L5–S1 level might be due to mechanical factors, but the pathophysiology of this phenomenon needs further investigation.

A systematic review of the literature concerning VESC and nonspecific LBP [73] was performed. The authors defined manuscript quality criteria and reviewed 137 full text articles. They verified a positive association between VESC and nonspecific LBP in the majority of studies regardless of country of origin or whether it included a working or nonworking population [56, 74, 123]. The median prevalence of VESC changes in the nonspecific LBP population was 43 % and only 6 % in the nonclinical population. The authors hypothesized that if VESC is a condition that results in LBP, then the prevalence would be highest in study samples of patients with LBP, lower in study samples of the general and working population, and lowest in individuals without LBP. When all populations of patients were combined into one group, the prevalence of VESC was found to be more than seven times higher among patients with nonspecific LBP than in asymptomatic (nonclinical) patients [73].

A recent 14-month longitudinal cohort study with MRI demonstrated that the presence of M1 at both baseline and follow-up is associated with a poor outcome in patients with persistent LBP [124]. Bailly et al. [125] compared, in a casecontrol study, clinical characteristics of patients with LBP with and without M1. On multivariate analysis, M1 patients were associated with sedentary work, pain with lumbar extension, and inflammatory pain pattern.

Kerttula et al. [116] studied the relation of the sizes of M1 and M2 and the Modic type (mixed M1-M2 or pure M1) with the intensity of LBP and level of perceived disability in 62 patients with large M1 and chronic LBP. They observed that the size of M1 did not directly correlate with clinical symptoms, but the Modic type (M1) was more important. They suggested that pure M1 may exist for a relatively short time after the onset of the process and then turn into M2 gradually. The authors remarked that when the inflammatory process turned to the mixed M1-M2 lesions, clinical symptoms improved [116].

## 9.11 Discography or MC or None to Define Painful Disk?

Provocative discography is not universally accepted as a pathognomonic test for diskogenic pain but remains the only useful functional study used to localize the pain generator in the setting of a degenerated disk. Diverse patient and operator variables may alter and influence patient responses and may lead to false-positive and false-negative findings. A correlation between provocative discography and MC has been evaluated by many authors [72, 94, 120, 126]. A verifiable positive association between MC and a painful disk would obviate the need for provocative discography. The presence of MC in a grouping of degenerative disks would allow a surgeon to target more appropriately treatment after excluding other causes of LBP. Such an association to date has not been verified [126].

Braithwait et al. [72] reviewed MRI studies and diskograms of 58 patients with supposed diskogenic LBP. The presence of MC was correlated with pain reproduction at 152 disks. MC occurred in 48.3 % of patients and 24.2 % of degenerative disks. There were 23 disks with associated MC that underwent discography, and the vast majority (21 out of 23) was associated with pain reproduction. However, pain was also reproduced at 69 levels where no MC was present. The sensitivity, specificity, positive predictive value (PPV), and negative predictive value (NPV) for an MC as a marker of a painful disk was 23.3 %, 96.8 %, 91.3 %, and 46.5 %, respectively. These authors concluded that the presence of MC was a relatively specific but an insensitive sign of a painful lumbar disk in patients with diskogenic LBP.

Weishaupt et al. [94] affirmed that moderate and severe end plate abnormalities appear to be useful in the prediction of a painful disk in patients with symptomatic LBP. The authors performed a retrospective MRI review of 30 consecutive patients aged 20–50 years with long-standing LBP who underwent both MR imaging and discography. MC existed in 53 %. When only moderate and severe M1 and M2 were considered, all injected disks caused eed pain with provocation (sensitivity, 38 %; specificity, 100 %; PPV, 100 %).

Jensen et al. [73], in their literature review, similarly stated that the presence of VESC increases the likelihood of a concordant response during provocative discography. Buttermann et al. [121] noted a positive association between discography and M1 and M2, but the series was quite small. Sandhu et al. [126] retrospectively reviewed 53 consecutive patients who underwent discography and MRI for MC analysis. Their data showed no significant relationship between these distinct diagnostic tools. They concluded that both diagnostic entities had poor specificity for the diskogenic LBP and that each method may identify distinct pathologic entities.

## 9.12 Modic Changes and Surgical Series

## 9.12.1 Fusion

The first correlation of bone marrow changes on MRI and surgical intervention was published by Lang et al. [119]. These authors clearly described M1 changes in 10 out of 14 patients in which a diagnosis of pseudoarthrosis was suspected after segmental fusion. They also found M2 changes in 16 out of 19 patients who had solid fusions. These findings are in agreement with the presumed role of segmental hypermobility [82] in the genesis and/or persistence of degenerative vertebral end plate inflammatory changes.

Chataigner et al. [127] retrospectively reviewed 56 patients with symptomatic DDD surgically treated with an anterior lumbar interbody fusion (ALIF) technique. Improved clinical results were present in patients presenting with M1 changes compared to those with M0 or M2 changes. They concluded that an interbody fusion was an effective treatment method in lumbar DDD associated with M1 changes. Clarke et al. [128] prospectively compared 13 patients without MC with 13 patients with 2 M1 and 11 M2 changes and observed that MC did not predict improved outcome after instrumented posterolateral fusion. However, no statistical comparison between Modic groups was possible due to the very small number of M1 patients. Vital et al. [114] published a consecutive series of 17 surgically managed patients with DDD and M1 changes. All patients experienced improved pain and disability at a follow-up ranging from 12 to 72 months. The authors also reported conversion to M2 (n=13) or M0 (n=4) in all cases within 6 months after arthrodesis. This supports the contention that segmental arthrodesis may accelerate healing in patients with M1 changes by modifying loads exerted on the degenerated disk.

Esposito et al. [111] reported the findings of a prospective study of a cohort of 60 consecutive patients with chronic diskogenic LBP (duration >6 months) refractory to conservative treatment. The population of 30 men and 30 women constituted a clinically homogeneous cohort. All were severely disabled (visual analog scale [VAS] scores

>= 6; Japanese Orthopedic Association [JOA] scores <=10), with advanced disk degeneration (Grades 3–5 according to Pfirrmann classification). When there were more than one degenerated disk on MRI, discography was performed to determine the painful level. Patients underwent either a posterior 1-level instrumented arthrodesis and posterolateral fusion with autograft (38 patients) or an interbody fusion with a carbon fiber composite (polyetheretherketone) cage filled with bone graft (22 patients). Changes were classified as Modic 0 in 15 patients, M1 in 22, M2 in 14, and M1/M2 (defined by them as a transitory state) in 9. The Wilcoxon paired sample test was used to assess significance (p < 0.05 considered significant) in comparing the results of pre- and postoperative evaluations of pain and functional status. Patients with M1 improved much more than others (p < 0.01), with good to excellent results noted in 72.7 % of patients. In the M2 group, the results were generally poor, with good to excellent results observed in only 14.3 % of patients. For the M1/M2 group, clinical outcomes were comparable to patients who had presented with pure M1 (p < 0.01). In the group without MC, there were also significant improvements in both VAS and JOA scores, but in a smaller proportion of patients than in the M1 and M1/M2 groups (p=0.0395). The authors concluded that patients with LBP of presumed discal origin and severe DDD with an M1 may expect an optimal outcome following fusion. They also concluded that an arthrodesis in patients with an M2 lesion may experience less benefit of doubtful clinical significance. Despite the absence of a clinical control cohort, the results of this study provide important contributions to the understanding of the pathomechanism of MC and their relationship to LBP and its response to surgery. Surgery may be considered an artificial means of acceleration, the healing process/ bone remodeling of this inflammatory state leading to symptom improvement especially in patients with M1 changes.

Ohtori et al. [129] examined the change of M1 to M2 after instrumented posterolateral fusion plus decompression surgery for lumbar canal stenosis and observed that M1 changed to M2, but M2 didn't convert to M1. M1 and M2 changed to normal bone marrow signals in four cases. This suggested that M2 is the final stabilized stage; however, the bone marrow may be able to "regenerate" after surgical stabilization [129].

#### 9.12.2 Dynamic Fusion

Dynamic systems for treating MC were described. A dynamic stabilization in theory would avoid drawbacks of fusion, leaving the segment mobile and controlling load-bearing pattern of the motion segment and abnormal motion at the segment [130]. Eser et al. [131] operated 88 patients with chronic LBP and M1 and M2 using dynamic systems Cosmic (Ulrich GmbH & Co. KG, Ulm, Germany) and Safinaz (Medikon AS, Turkey) consisting of dynamic pedicle screws and a rigid rod system. The authors performed microdiskectomy in all patients. All cases of M1 turned into M2 or M3 in the 2-year follow-up.

#### 9.12.3 Disk Arthroplasty

Blondel et al. [132] studied the clinical results of lumbar total disk arthroplasty in accordance with Modic signs, with a 2-year minimum follow-up, and observed superior clinical results in M1 patients compared to M2 and no Modic sign.

Siepe et al. [133] analyzed the effect of MRI end plate changes on outcome after lumbar arthroplasty. They followed 92 patients for a minimum of 24 months. Patients were divided into four groups: isolated DDD, DDD associated with disk herniation, DDD associated with MC, and postdiskectomy syndrome. At final follow-up, a significant clinical improvement was reported for the entire cohort, and the best results were achieved in the group with an associated disk herniation. MC were not associated with significantly better clinical outcomes.

#### 9.12.4 Diskectomy

Chin et al. [134] performed a prospective casecontrol study to assess the outcomes following microdiskectomy in 15 consecutive patients with symptomatic disk herniation and sciatica associated with LBP and M1 and or M2 changes compared with similar patients without MC. Relief of LBP was less predictable than sciatica. The authors suggested that the substantial improvement seen in both groups supports the premise that a diskectomy alone is adequate in patients with MC and sciatica correlating with a disk herniation. There was a trend toward greater improvement of LBP in patients without any MC.

Sørlie et al. [135] found that patients with M1 who were operated for lumbar disk herniation had less improvement of LBP 1 year after surgery, compared to those who had no or other types of MC. However, in the multivariate analyses, this negative association no longer showed statistical significance when adjusted for smoking, which remained the only independent risk factor. Patients with preoperative M1 can expect less but still significant improvement of LBP 1 year after microdiskectomy, but not if they smoke cigarettes. Lurie et al. [136] stated that patients with M1 had worse outcomes after diskectomy for lumbar disk herniation. They affirmed that patients with small disk herniations and M1 may not benefit substantially from diskectomy.

Braithwait et al. [72] affirmed that patients with M1 and severe LBP did well after anterior diskectomy and fusion, but that the outcome was less predictable in M1 absence. Buttermann et al. [121] observed that when diskectomy was followed by the development of M1, patients often experienced continued LBP. Of the 24 patients with M1, 18 had undergone a prior procedure: 17 a diskectomy and 1 a nucleolysis. Another group of 24 patients, most of whom had also undergone a prior diskectomy, had M2. Most of the patients with LBP and M2 also had an inflammatory component manifested as mixed M1/M2. In their series, 19 out of 24 patients with M1 who underwent fusion continued to experience LBP (p < 0.03). Precise data regarding pseudoarthrosis rates were not reported. The authors noted a correlation between clinical failure of a posterior fusion and the presence of M1 at the symptomatic level.

#### 9.12.5 Intradiscal Steroid Injection

Fayad et al. [137] analyzed a series of 74 patients with LBP and found that patients with chronic

LBP and predominantly M1 had better shortterm relief of symptoms following lumbar intradiscal steroid injection (IDIC) than M2. They noted that at 3 and 6 months, IDIC tended to be more effective in patients with M1 and M1/ M2 changes, although this was not statistically significant. Similar findings were reported by Beaudreuil et al. [138] who reported that patients with disabling chronic LBP and M1 have specific acute response to intradiscal injection of methylprednisolone.

#### Conclusion

Much has been written recently concerning Modic changes and their relationship to symptomatic low back pain. Despite anedoctal evidence that Modic changes, especially M1 subchondral inflammatory changes, may be causally related to symptoms of low back pain, well-designed studies are needed to develop a better understanding of their true significance and prognostic value.

## References

- 1. Andersson GB. Epidemiology of low back pain. Acta Orthop Scand Suppl. 1998;281:28–31.
- 2. Robertson JT. The rape of the spine. Surg Neurol. 1993;39:5–12.
- United Nations. The ageing of the world's population. Population Division, Department of Economic and Social Affairs, United Nations Secretariat. http:// www.un.org/esa/socdev/ageing/
- Podichetty VK. The aging spine: the role of inflammatory mediators in intervertebral disc degeneration. Cell Mol Biol (Noisy-le-Grand). 2007;53(5):4–18.
- Waddell G. 1987 Volvo award in clinical sciences. A new clinical model for the treatment of low-back pain. Spine. 1987;12:632–44.
- Albert HB, Manniche C. Modic changes following lumbar disc herniation. Eur Spine J. 2007;16(7): 977–82.
- Brennan GP, Fritz J, Hunter SJ, Thackeray A, Delitto A, Erhard RE. Identifying subgroups of patients with acute/subacute "nonspecific" low back pain results of a randomized clinical trial. Spine. 2006;31(6):623– 31. Brennan GP Fritz J Hunt. SJ Thackeray Delitto Erhard RE 2006 Identifying Subgr. Patients Acute subacute "nonspecific" Low Back Pain Results Randomized Clin. Trial Spine 316623–631.
- Boos N, Weissbach S, Rohrbach H, Weiler C, Spratt KF, Nerlich AG. Classification of age-related changes

in lumbar intervertebral discs: 2002 Volvo Award in basic science. Spine. 2002;27:2631–44.

- Marshman LA, Trewhella M, Friesem T, Bhatia CK, Krishna M. Reverse transformation of Modic type 2 changes to Modic type 1 changes during sustained chronic low-back pain severity. Report of two cases and review of the literature. J Neurosurg Spine. 2007;6(2):152–5.
- Gruber HE, Hanley Jr EN. Analysis of aging and degeneration of the human intervertebral disc: comparison of surgical specimens with normal controls. Spine. 1998;23:751–7.
- Gruber HE, Norton HJ, Hanley Jr EN. Anti-apoptotic effects of IGF-1 and PDGF on human intervertebral disc cells in vitro. Spine. 2000;25:2153–7.
- Lotz JC, Colliou OK, Chin JR, et al. Compressioninduced degeneration of the intervertebral disc: an in vivo mouse model and finite-element study. Spine. 1998;23:2493–506.
- Lotz JC, Chin JR. Intervertebral disc cell death is dependent on the magnitude and duration of spinal loading. Spine. 2000;25:1477–83.
- Urban JP, Holm S, Maroudas A, Nachemson A. Nutrition of the intervertebral disc: effect of fluid flow on solute transport. Clin Orthop Relat Res. 1982;170: 296–302.
- Benneker LM, Heini PF, Alini M, Anderson SE, Ito K. 2004 Young Investigator Award winner: vertebral endplate marrow contact channel occlusions and intervertebral disc degeneration. Spine. 2005;30: 167–73.
- Nachemson A, Lewin T, Maroudas A, Freeman MA. In vitro diffusion of dye through the end-plates and the annulus fibrosus of human lumbar inter-vertebral discs. Acta Orthop Scand. 1970;41:589–607.
- Tomlinson N, Maroudas A. The effect of cyclic and continuous compression on the penetration of large molecules into articular cartilage. J Bone Joint Surg Br. 1980;62B:251.
- Chandraraj S, Briggs CA, Opeskin K. Disc herniations in the young and end-plate vascularity. Clin Anat N Y N. 1998;11:171–6.
- Bernick S, Cailliet R. Vertebral end-plate changes with aging of human vertebrae. Spine. 1982;7: 97–102.
- Roberts S, Menage J, Eisenstein SM. The cartilage end-plate and intervertebral disc in scoliosis: calcification and other sequelae. J Orthop Res Off Publ Orthop Res Soc. 1993;11:747–57.
- Roberts S, Urban JP, Evans H, Eisenstein SM. Transport properties of the human cartilage endplate in relation to its composition and calcification. Spine. 1996;21:415–20.
- 22. Antoniou J, et al. The human lumbar endplate. Evidence of changes in biosynthesis and denaturation of the extracellular matrix with growth, maturation, aging, and degeneration. Spine. 1996;21:1153–61.
- Nerlich AG, Schleicher ED, Boos N. Immunohistologic markers for age-related changes of human lumbar intervertebral discs. Spine. 1997;22:2781–95.

- Urban JP, Roberts S. Degeneration of the intervertebral disc. Arthritis Res Ther. 2003;5:120–30.
- 25. Roberts S, Menage J, Duance V, Wotton S, Ayad S. 1991 Volvo Award in basic sciences. Collagen types around the cells of the intervertebral disc and cartilage end plate: an immunolocalization study. Spine. 1991;16:1030–8.
- Bartholdi D, Schwab ME. Expression of proinflammatory cytokine and chemokine mRNA upon experimental spinal cord injury in mouse: an in situ hybridization study. Eur J Neurosci. 1997;9: 1422–38.
- Botchkina GI, Meistrell 3rd ME, Botchkina IL, Tracey KJ. Expression of TNF and TNF receptors (p55 and p75) in the rat brain after focal cerebral ischemia. Mol Med Camb Mass. 1997;3:765–81.
- Burke JG, et al. Intervertebral discs which cause low back pain secrete high levels of proinflammatory mediators. J Bone Joint Surg Br. 2002;84:196–201.
- 29. Weiler C, Nerlich AG, Bachmeier BE, Boos N. Expression and distribution of tumor necrosis factor alpha in human lumbar intervertebral discs: a study in surgical specimen and autopsy controls. Spine. 2005;30:44–53. discussion 54.
- Roberts S, Menage J, Evans EH, et al. Inflammation of the intervertebral disc: an uncommon finding in discs associated with discogenic back pain. J Bone Joint Surg [Br]. 2000;82-B(Supp II):98.
- Rand N, Reichert F, Floman Y, Rotshenker S. Murine nucleus pulposus-derived cells secrete interleukins-1beta, -6, and -10 and granulocyte-macrophage colonystimulating factor in cell culture. Spine. 1997;22: 2598–601. discussion 2602.
- 32. Kang JD, Stefanovic-Racic M, McIntyre LA, Georgescu HI, Evans CH. Toward a biochemical understanding of human intervertebral disc degeneration and herniation. Contributions of nitric oxide, interleukins, prostaglandin E2, and matrix metalloproteinases. Spine. 1997;22:1065–73.
- 33. Kang JD, et al. Herniated lumbar intervertebral discs spontaneously produce matrix metalloproteinases, nitric oxide, interleukin-6, and prostaglandin E2. Spine. 1996;21:271–7.
- O'Donnell JL, O'Donnell AL. Prostaglandin E2 content in herniated lumbar disc disease. Spine. 1996;21:1653–5. discussion 1655–1656.
- 35. Andrade P, et al. Elevated IL-1β and IL-6 levels in lumbar herniated discs in patients with sciatic pain. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2013;22:714–20.
- Grönblad M, et al. A controlled immunohistochemical study of inflammatory cells in disc herniation tissue. Spine. 1994;19:2744–51.
- Olmarker K, Larsson K. Tumor necrosis factor alpha and nucleus-pulposus-induced nerve root injury. Spine. 1998;23:2538–44.
- Takahashi H, et al. Inflammatory cytokines in the herniated disc of the lumbar spine. Spine. 1996;21: 218–24.

- Fujita K, Nakagawa T, Hirabayashi K, Nagai Y. Neutral proteinases in human intervertebral disc. Role in degeneration and probable origin. Spine. 1993;18: 1766–73.
- Roberts S, et al. Matrix metalloproteinases and aggrecanase: their role in disorders of the human intervertebral disc. Spine. 2000;25:3005–13.
- 41. Weiler C, Nerlich AG, Zipperer J, Bachmeier BE, Boos N. 2002 SSE Award Competition in basic science: expression of major matrix metalloproteinases is associated with intervertebral disc degradation and resorption. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2002;11:308–20.
- Coppes MH, Marani E, Thomeer RT, Groen GJ. Innervation of 'painful' lumbar discs. Spine. 1997;22:2342–9. discussion 2349–2350.
- Tessier-Lavigne M, Placzek M. Target attraction: are developing axons guided by chemotropism? Trends Neurosci. 1991;14:303–10.
- 44. Kuslich SD, Ulstrom CL, Michael CJ. The tissue origin of low back pain and sciatica: a report of pain response to tissue stimulation during operations on the lumbar spine using local anesthesia. Orthop Clin North Am. 1991;22:181–7.
- Freemont AJ, et al. Nerve ingrowth into diseased intervertebral disc in chronic back pain. Lancet. 1997;350:178–81.
- 46. Palmgren T, Grönblad M, Virri J, Kääpä E, Karaharju E. An immunohistochemical study of nerve structures in the anulus fibrosus of human normal lumbar intervertebral discs. Spine. 1999;24:2075–9.
- Fagan A, Moore R, Vernon Roberts B, Blumbergs P, Fraser R. ISSLS prize winner: the innervation of the intervertebral disc: a quantitative analysis. Spine. 2003;28:2570–6.
- Brown MF, et al. Sensory and sympathetic innervation of the vertebral endplate in patients with degenerative disc disease. J Bone Joint Surg Br. 1997;79: 147–53.
- Adams MA, Roughley PJ. What is intervertebral disc degeneration, and what causes it? Spine. 2006;31: 2151–61.
- Melrose J, Roberts S, Smith S, Menage J, Ghosh P. Increased nerve and blood vessel ingrowth associated with proteoglycan depletion in an ovine anular lesion model of experimental disc degeneration. Spine. 2002;27:1278–85.
- Johnson WEB, Caterson B, Eisenstein SM, Roberts S. Human intervertebral disc aggrecan inhibits endothelial cell adhesion and cell migration in vitro. Spine. 2005;30:1139–47.
- Modic MT, et al. Magnetic resonance imaging of intervertebral disk disease. Clinical and pulse sequence considerations. Radiology. 1984;152:103–11.
- Hajek PC, et al. Focal fat deposition in axial bone marrow: MR characteristics. Radiology. 1987;162: 245–9.
- 54. De Roos A, Kressel H, Spritzer C, Dalinka M. MR imaging of marrow changes adjacent to end plates in

degenerative lumbar disk disease. AJR Am J Roentgenol. 1987;149:531-4.

- Modic MT, Steinberg PM, Ross JS, Masaryk TJ, Carter JR. Degenerative disk disease: assessment of changes in vertebral body marrow with MR imaging. Radiology. 1988;166:193–9.
- 56. Kuisma M, et al. Modic changes in endplates of lumbar vertebral bodies: prevalence and association with low back and sciatic pain among middle-aged male workers. Spine. 2007;32:1116–22.
- 57. Kjaer P, Korsholm L, Bendix T, Sorensen JS, Leboeuf-Yde C. Modic changes and their associations with clinical findings. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2006;15:1312–9.
- Malinin T, Brown MD. Changes in vertebral bodies adjacent to acutely narrowed intervertebral discs: observations in baboons. Spine. 2007;32:E603–7.
- Bishop PB, Pearce RH. The proteoglycans of the cartilaginous end-plate of the human intervertebral disc change after maturity. J Orthop Res Off Publ Orthop Res Soc. 1993;11:324–31.
- Ariga K, et al. The relationship between apoptosis of endplate chondrocytes and aging and degeneration of the intervertebral disc. Spine. 2001;26:2414–20.
- Nachemson A. The load on lumbar disks in different positions of the body. Clin Orthop. 1966;45:107–22.
- Albert HB, et al. Modic changes, possible causes and relation to low back pain. Med Hypotheses. 2008;70: 361–8.
- Moore RJ, Vernon-Roberts B, Osti OL, Fraser RD. Remodeling of vertebral bone after outer anular injury in sheep. Spine. 1996;21:936–40.
- Ulrich JA, Liebenberg EC, Thuillier DU, Lotz JC. ISSLS prize winner: repeated disc injury causes persistent inflammation. Spine. 2007;32:2812–9.
- Imhof H, et al. Subchondral bone and cartilage disease: a rediscovered functional unit. Invest Radiol. 2000;35:581–8.
- 66. Przybyla A, Pollintine P, Bedzinski R, Adams MA. Outer annulus tears have less effect than endplate fracture on stress distributions inside intervertebral discs: relevance to disc degeneration. Clin Biomech (Bristol, Avon). 2006;21:1013–9.
- Adams MA, Freeman BJ, Morrison HP, Nelson IW, Dolan P. Mechanical initiation of intervertebral disc degeneration. Spine. 2000;25:1625–36.
- Holm S, Holm AK, Ekström L, Karladani A, Hansson T. Experimental disc degeneration due to endplate injury. J Spinal Disord Tech. 2004;17:64–71.
- Vernon-Roberts B, Pirie CJ. Healing trabecular microfractures in the bodies of lumbar vertebrae. Ann Rheum Dis. 1973;32:406–12.
- Twomey L, Taylor J. Age changes in lumbar intervertebral discs. Acta Orthop Scand. 1985;56:496–9.
- 71. Luoma K, Vehmas T, Grönblad M, Kerttula L, Kääpä E. MRI follow-up of subchondral signal abnormalities in a selected group of chronic low back pain patients. Eur Spine J Off Publ Eur Spine Soc Eur

Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2008;17:1300–8.

- 72. Braithwaite I, White J, Saifuddin A, Renton P, Taylor BA. Vertebral end-plate (Modic) changes on lumbar spine MRI: correlation with pain reproduction at lumbar discography. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 1998;7:363–8.
- 73. Jensen TS, Karppinen J, Sorensen JS, Niinimäki J, Leboeuf-Yde C. Vertebral endplate signal changes (Modic change): a systematic literature review of prevalence and association with non-specific low back pain. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2008;17: 1407–22.
- 74. Kjaer P, Leboeuf-Yde C, Korsholm L, Sorensen JS, Bendix T. Magnetic resonance imaging and low back pain in adults: a diagnostic imaging study of 40-yearold men and women. Spine. 2005;30:1173–80.
- Osti OL, Vernon-Roberts B, Fraser RD. 1990 Volvo Award in experimental studies. Anulus tears and intervertebral disc degeneration. An experimental study using an animal model. Spine. 1990;15:762–7.
- Kerttula LI, Serlo WS, Tervonen OA, Pääkkö EL, Vanharanta HV. Post-traumatic findings of the spine after earlier vertebral fracture in young patients: clinical and MRI study. Spine. 2000;25:1104–8.
- Tanaka M, Nakahara S, Inoue H. A pathologic study of discs in the elderly. Separation between the cartilaginous endplate and the vertebral body. Spine. 1993;18:1456–62.
- Schmid G, et al. Lumbar disk herniation: correlation of histologic findings with marrow signal intensity changes in vertebral endplates at MR imaging. Radiology. 2004;231:352–8.
- Carreon LY, Ito T, Yamada M, Uchiyama S, Takahashi HE. Neovascularization induced by anulus and its inhibition by cartilage endplate. Its role in disc absorption. Spine. 1997;22:1429–34. discussion 1446–1447.
- Moses MA, Sudhalter J, Langer R. Identification of an inhibitor of neovascularization from cartilage. Science. 1990;248:1408–10.
- Ikeda T, et al. Pathomechanism of spontaneous regression of the herniated lumbar disc: histologic and immunohistochemical study. J Spinal Disord. 1996;9: 136–40.
- Toyone T, et al. Vertebral bone-marrow changes in degenerative lumbar disc disease. An MRI study of 74 patients with low back pain. J Bone Joint Surg Br. 1994;76:757–64.
- 83. Ohtori S, et al. Tumor necrosis factor-immunoreactive cells and PGP 9.5-immunoreactive nerve fibers in vertebral endplates of patients with discogenic low back pain and Modic type 1 or type 2 changes on MRI. Spine. 2006;31:1026–31.
- 84. Karppinen J, et al. Genetic factors are associated with modic changes in endplates of lumbar vertebral bodies. Spine. 2008;33:1236–41.

- Burke JG, Watson RWG, McCormack D, Fitzpatrick JM, Stack J, Walsh MG. Modic changes are associated with increased disc inflammatory mediator production. J Bone Joint Surg Br. 2003;85 Suppl 2:164.
- 86. Albert HB, Sorensen JS, Christensen BS, Manniche C. Antibiotic treatment in patients with chronic low back pain and vertebral bone edema (Modic type 1 changes): a double-blind randomized clinical controlled trial of efficacy. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2013;22:697–707.
- Fairbanks J. Britspine 2014. Debate: "Should antibiotics be given to back pain?" Warwick, 4th Apr 2014.
- Weishaupt D, Zanetti M, Hodler J, Boos N. MR imaging of the lumbar spine: prevalence of intervertebral disk extrusion and sequestration, nerve root compression, end plate abnormalities, and osteoarthritis of the facet joints in asymptomatic volunteers. Radiology. 1998;209:661–6.
- Kuisma M, et al. A three-year follow-up of lumbar spine endplate (Modic) changes. Spine. 2006;31: 1714–8.
- Karchevsky M, et al. Reactive endplate marrow changes: a systematic morphologic and epidemiologic evaluation. Skeletal Radiol. 2005;34:125–9.
- Battié MC, Videman T, Parent E. Lumbar disc degeneration: epidemiology and genetic influences. Spine. 2004;29:2679–90.
- 92. Kuisma M, Karppinen J, Niinimaki J, et al. Modic type and lumbar level affects painfulness of Modic changes. Annual Meeting of the International Society for the Study of the Lumbar Spine, New York, May 10–14, 2005.
- 93. Jensen TS, Sorensen JS, Kjaer P. Intra- and interobserver reproducibility of vertebral endplate signal (modic) changes in the lumbar spine: the Nordic Modic Consensus Group classification. Acta Radiol Stockh Swed. 2007;1987(48):748–54.
- Weishaupt D, et al. Painful lumbar disk derangement: relevance of endplate abnormalities at MR imaging. Radiology. 2001;218:420–7.
- 95. Mulconrey DS, Knight RQ, Bramble JD, Paknikar S, Harty PA. Interobserver reliability in the interpretation of diagnostic lumbar MRI and nuclear imaging. Spine J Off J N Am Spine Soc. 2006;6:177–84.
- 96. Chung CB, et al. End plate marrow changes in the asymptomatic lumbosacral spine: frequency, distribution and correlation with age and degenerative changes. Skeletal Radiol. 2004;33:399–404.
- Jones A, Clarke A, Freeman BJC, Lam KS, Grevitt MP. The Modic classification: inter- and intraobserver error in clinical practice. Spine. 2005;30: 1867–9.
- Vogler 3rd JB, Murphy WA. Bone marrow imaging. Radiology. 1988;168:679–93.
- Vande Berg BC, et al. The lumbar vertebral body and diskovertebral junction. Radio MR imaging

anatomic correlations. Radiol Clin North Am. 2000;38:1153–75.

- Crock HV. Internal disc disruption. A challenge to disc prolapse fifty years on. Spine. 1986;11:650–3.
- 101. Rannou F, et al. High-sensitivity C-reactive protein in chronic low back pain with vertebral end-plate Modic signal changes. Arthritis Rheum. 2007;57: 1311–5.
- 102. Dagirmanjian A, Schils J, McHenry M, Modic MT. MR imaging of vertebral osteomyelitis revisited. AJR Am J Roentgenol. 1996;167:1539–43.
- 103. Kato F, Ando T, Kawakami N, Mimatsu K, Iwata H. The increased signal intensity at the vertebral body endplates after chemonucleolysis demonstrated by magnetic resonance imaging. Spine. 1993;18: 2276–81.
- 104. Stäbler A, et al. MR imaging of enhancing intraosseous disk herniation (Schmorl's nodes). AJR Am J Roentgenol. 1997;168:933–8.
- James SLJ, Davies AM. Imaging of infectious spinal disorders in children and adults. Eur J Radiol. 2006;58:27–40.
- Ross JS, Modic MT. Current assessment of spinal degenerative disease with magnetic resonance imaging. Clin Orthop Relat Res. 1992;279:68–81.
- Ledermann HP, Schweitzer ME, Morrison WB, Carrino JA. MR imaging findings in spinal infections: rules or myths? Radiology. 2003;228:506–14.
- 108. Dallaudière B, et al. Comparison of apparent diffusion coefficient in spondylarthritis axial active inflammatory lesions and type 1 modic changes. Eur J Radiol. 2013. doi:10.1016/j.ejrad.2013.10.009.
- Kirkaldy-Willis WH, Farfan HF. Instability of the lumbar spine. Clin Orthop Relat Res. 1982;165: 110–23.
- Modic MT, Masaryk TJ, Ross JS, Carter JR. Imaging of degenerative disk disease. Radiology. 1988;168: 177–86.
- 111. Esposito P, Pinheiro-Franco JL, Froelich S, Maitrot D. Predictive value of MRI vertebral endplate signal changes (Modic) on outcome of surgically treated degenerative disc disease. Results of a cohort study including 60 patients. Neurochirurgie. 2006;52:315–22.
- 112. Mitra D, Cassar-Pullicino VN, McCall IW. Longitudinal study of vertebral type-1 end-plate changes on MR of the lumbar spine. Eur Radiol. 2004;14:1574–81.
- 113. Pinheiro-Franco JL, Franco JLP. Modic changes: 'age, si quid agis'. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2008;17:1766–8. author reply 1769–1770.
- Vital JM, et al. Course of Modic 1 six months after lumbar posterior osteosynthesis. Spine. 2003;28: 715–20. discussion 721.
- 115. Modic MT. Modic type 1 and type 2 changes. J Neurosurg Spine. 2007;6:150–1. discussion 151.
- 116. Kerttula L, Luoma K, Vehmas T, Grönblad M, Kääpä E. Modic type I change may predict rapid

progressive, deforming disc degeneration: a prospective 1-year follow-up study. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2012;21:1135–42.

- 117. Jensen MC, et al. Magnetic resonance imaging of the lumbar spine in people without back pain. N Engl J Med. 1994;331:69–73.
- 118. Boden SD, Davis DO, Dina TS, Patronas NJ, Wiesel SW. Abnormal magnetic-resonance scans of the lumbar spine in asymptomatic subjects. A prospective investigation. J Bone Joint Surg Am. 1990;72: 403–8.
- 119. Lang P, Chafetz N, Genant HK, Morris JM. Lumbar spinal fusion. Assessment of functional stability with magnetic resonance imaging. Spine. 1990; 15:581–8.
- 120. McCall LW, Cessar-Pullicino VN, Tyrell PN. MR vertebral endplate changes and back pain. Presented at the 24th annual meeting of the International Society of the Study of the Lumbar Spine, Singapore, 2–6 Jun 1997.
- 121. Buttermann GR, Heithoff KB, Ogilvie JW, Transfeldt EE, Cohen M. Vertebral body MRI related to lumbar fusion results. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 1997;6:115–20.
- 122. Albert HB, Manniche C. Modic changes the prevalence and relationship to lumbar disc herniation. A possible new pathogenesis of low back pain (abstract, 7th annual meeting of the Spine Society of Europe, Barcelona, Spain). Eur Spine J. 2005;14(suppl 1): S1-S90.
- 123. Schenk P, Läubli T, Hodler J, Klipstein A. Magnetic resonance imaging of the lumbar spine: findings in female subjects from administrative and nursing professions. Spine. 2006;31:2701–6.
- 124. Jensen RK, et al. Is the development of Modic changes associated with clinical symptoms? A 14-month cohort study with MRI. Eur Spine J. 2012;21:2271–9.
- 125. Bailly F, et al. Inflammatory pain pattern and pain with lumbar extension associated with Modic 1 changes on MRI: a prospective case-control study of 120 patients. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2013. doi:10.1007/s00586-013-3036-6
- 126. Sandhu HS, et al. Association between findings of provocative discography and vertebral endplate signal changes as seen on MRI. J Spinal Disord. 2000;13:438–43.

- 127. Chataigner H, Onimus M, Polette A. Surgery for degenerative lumbar disc disease. Should the black disc be grafted? Rev Chir Orthopédique Réparatrice Appar Mot. 1998;84:583–9.
- 128. Clarke A, Lam KS, Freeman BJF. Do vertebral end plate changes predict better outcome following lumbar interbody fusion. A prospective study with minimum 2-year follow-up. Presented at the First Spineweek Meeting, Porto (Portugal) (abstract). 2004.
- 129. Ohtori S, et al. Change in Modic type 1 and 2 signals after posterolateral fusion surgery. Spine. 2010;35: 1231–5.
- Sengupta DK. Dynamic stabilization devices in the treatment of low back pain. Neurol India. 2005;53: 466–74.
- 131. Eser O, et al. Dynamic stabilisation in the treatment of degenerative disc disease with modic changes. Adv Orthop. 2013;2013:1–6.
- 132. Blondel B, Tropiano P, Gaudart J, Huang RC, Marnay T. Clinical results of lumbar total disc arthroplasty in accordance with Modic signs, with a 2-yearminimum follow-up. Spine. 2011;36:2309–15.
- 133. Siepe CJ, Mayer HM, Wiechert K, Korge A. Clinical results of total lumbar disc replacement with ProDisc II: three-year results for different indications. Spine. 2006;31:1923–32.
- 134. Chin KR, Tomlinson DT, Auerbach JD, Shatsky JB, Deirmengian CA. Success of lumbar microdiskectomy in patients with modic changes and low-back pain: a prospective pilot study. J Spinal Disord Tech. 2008;21:139–44.
- 135. Sørlie A, et al. Modic type I changes and recovery of back pain after lumbar microdiskectomy. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2012;21:2252–8.
- Lurie JD, et al. Magnetic resonance imaging predictors of surgical outcome in patients with lumbar intervertebral disc herniation. Spine. 2013;38:1216–25.
- 137. Fayad F, et al. Relation of inflammatory modic changes to intradiscal steroid injection outcome in chronic low back pain. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2007;16:925–31.
- 138. Beaudreuil J, Dieude P, Poiraudeau S, Revel M. Disabling chronic low back pain with Modic type 1 MRI signal: acute reduction in pain with intradiscal corticotherapy. Ann Phys Rehabil Med. 2012;55: 139–47.

## Degenerative Marrow Changes: Natural History Biomechanics in Relation to Symptoms

# 10

## Michael T. Modic

Signal intensity changes of the vertebral body marrow adjacent to the end plates of degenerated disks are a long recognized and common observation on MR images of the lumbar spine [1, 2]. However, despite a growing body of literature on this subject, their clinical importance, etiology, and relationship to symptoms remain unclear [3].

These marrow changes appear to take three main forms on MR imaging. Type I changes demonstrate decreased signal intensity on TI-weighted images and increased signal intensity on T2-weighted images (Fig. 10.1). They have been identified in approximately 4 % of patients scanned for lumbar disease [2], approximately 8 % of patients after diskectomy [4], and in 40-50 % of chymopapain-treated disks, which may be viewed as a model of acute disk degeneration [5]. Histopathologic sections of disks with type I changes show disruption and fissuring of the end plate and vascularized fibrous tissues within the adjacent marrow, prolonging T1 and T2. Enhancement of type I vertebral body marrow changes is seen with administration of gadolinium that at times extends to involve the disk itself and is presumably related to the vascularized fibrous tissue within the adjacent marrow. Type II changes are represented by increased

signal intensity on T1-weighted images and isointense or slightly hyperintense signal on T2-weighted images (Fig. 10.2). They have been identified in approximately 16 % of patients at MR imaging. Disks with type II changes also show evidence of end plate disruption, with yellow (lipid) marrow replacement in the adjacent vertebral body resulting in a shorter T1. Type III changes are represented by decreased signal intensity on both T1- and T2-weighted images and correlate with extensive bony sclerosis on plain radiographs (Fig. 10.3). The lack of signal in the type III change no doubt reflects the relative absence of marrow in areas of advanced sclerosis. Unlike type III, types I and II changes show no definite correlation with sclerosis at radiography [6]. This is not surprising when one considers the histology; the sclerosis seen on plain radiographs is a reflection of dense woven bone within the vertebral body, whereas the MR changes are more a reflection of the intervening marrow elements. While the aforementioned histologic changes appear to describe the underlying anatomic substrate for the MR signal changes, they by no means describe the etiology of the underlying causative process.

M.T. Modic, MD, FACR

Cleveland Clinic, Neurological Institute, 9500 Euclid Ave., NA4, Cleveland, OH 44195, USA e-mail: modicm1@ccf.org; SKOWROL@ccf.org

Editor's Note Dr. Modic gave us the great honor of a chapter written by himself about the radiological findings that were marked in medicine as Modic changes. We are very grateful to him for his current appraisal of the matter.

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_10



**Fig. 10.1** Degenerative type I marrow changes: sagittal midline T1 (**a**) and T2 (**b**) spin-echo images of the lumbar spine. There is decreased signal intensity of the inferior

Similar marrow changes have also been noted in the pedicles. While originally described as being associated with spondylolysis, they have also been noted in patients with degenerative facet disease and pedicle fractures [7, 8] (Fig. 10.4). We do not know the exact mechanism by which these marrow changes occur. Their association with degenerative disk disease, facet changes, and pars and pedicle fractures suggest they are a response to biomechanical stress. This then suggests the first and likely most common etiology – mechanical.

The bone is a dynamic architectural substance that responds to changes imposed upon it. When stressed, the bone behaves according to Wolff's law [9]. Wolff's law of bone states that the architecture of a bone is determined by the mechanical stresses placed upon it and the bone's adaptation to withstand these stresses. Wolff's law is an example of the complementarity of form and function, showing that the form of a bone is

aspect of the L5 vertebral body on T1 ( $\rightarrow$ ) and increased signal intensity on T2 ( $\rightarrow$ ). The L5/S1 disk is degenerated

shaped by its functional experience. The trabeculae of spongy bone often develop along lines of stress. The major trabecular orientation in the vertebral bodies and pedicles is in-line with the principal direction of loading, whereas perpendicular laid-down support elements or "struts" increase the overall strength. As the bone comes under consistently applied stresses, it may react through the development of microfractures as well as osteoblast depositing new osseous tissues. The orderly remodeling of bone depends on a precise balance between deposition and resorption, between osteoblasts and osteoclasts, a process which repairs microfractures. The remodeled bone is known to contain microfractures that can demonstrate abnormal uptake at scintigraphy. It has been suggested that the type I MR signal intensity changes may be a reflection of remodeling trabecular bone with microfractures and associated marrow changes [10–12].



**Fig. 10.2** Degenerative type II marrow changes: sagittal T1 (a) and T2 (b) spin-echo images of the lumbar spine. On the sagittal T1-weighted images, there is increased signal intensity of the inferior aspect of L5 ( $\rightarrow$ ) and supe-

rior aspect of S1. The signal intensity on the T2-weighted images is slightly increased in this same region. The disk space is degenerated and there is evidence of a disk protrusion

Of these three types, type I changes appear to be more fluid and variable, a reflection of some ongoing underlying pathological process such as continuing degeneration with resulting changing biomechanical stresses. Of the three types, type I is most often associated with ongoing low back symptomatology [13–17]. In a longitudinal study, the incidence of new degenerative marrow changes was 6 % over a 3-year period, most of these being of type I [15]. In a study of nonoperated patients with low back pain, Mitra [14] found that 92 % of type I changes converted either wholly or partially into type II (52 %), became more extensive (40 %), or remained unchanged (8%). There was an improvement in symptoms in patients where type I changes converted to type II.

Some studies of diskography in patients with degenerative marrow changes have suggested that type I marrow changes are invariably associated with painful disks [18, 19]. Others [20, 21] have failed to be able to reproduce this association, and thus the relationship of degenerative marrow changes and diskogenic pain remains unproven.

In most cases, type II degenerative changes appear to be associated with a more stable state. Type II changes, however, are not always permanent and conversion between type II and I has been demonstrated. In general, when type II marrow changes convert to type I, there is usually a superimposed process such as continued or accelerated degeneration or vertebral



**Fig. 10.3** Degenerative type III marrow changes: sagittal midline T1 (a) and T2 (b) spin-echo images of the lumbar spine. There is decreased signal intensity of the adjacent

portions of L4 and L5  $(\rightarrow)$  with an intervening degenerated disk on both T1- and T2-weighted sequences

osteomyelitis. Some authors have suggested that mixed lesions are more common than originally thought and indicative of overlap and progression of one type to another [15] (Fig. 10.5).

In most studies of marrow changes, type II are the most prevalent and the prevalence increases with age [15, 22, 23]. Others have suggested that type II changes are less stable and may be as active as type I and equipotent relative to symptomatology [15, 24, 25]. In a study by Määttä et al. [26], there was a 46 % prevalence of marrow changes in patients referred to spinal surgery. In a study by Jensen [25], this prevalence was 43 % in patients with low back pain seeking care. In fact, Marshman et al. [24] reject the contention that type I lesions are more active. They speculate that the bone marrow appearances are merely epiphenomena. As such, they detract one from the more important consideration that the de novo pain afferents have traversed the disk space providing a substrate for diskogenic pain which is a more important consideration than the gross histological appearance and MR signal intensity change of the vertebral body marrow. This last point is valid in that MR changes are likely a con-


**Fig. 10.4** Pars fracture and pedicle hyperintensity: (a) parasagittal T1-, (b) T2-, and (c) STIR-weighted images of the lumbar spine (d, e) are oblique and sagittal multiplanar reformatted (MPR) CT images (respectively) of the lumbar spine. Note the decreased signal intensity on T1

( $\rightarrow$  Fig. 10.4a) and increased signal intensity on T2- and STIR-weighted images ( $\rightarrow$  Fig. 10.4b, c) within the pedicle of the L4 vertebral body on the *right*. A subtle pars fracture is demonstrated on the oblique (**d**) and sagittal (**e**) MPR CT images ( $\rightarrow$ )



**Fig. 10.5** Type II marrow conversion: sagittal midline T1 (**a**) and T2 (**b**) spin-echo images of the lumbar spine in a patient with low back pain and radiculopathy. This patient underwent a diskectomy at L4/L5 and initially did well but 1 year postoperatively developed recurrent low back pain and was reimaged. Figure 10.5c, d is sagittal

sequence of the biomechanical, cellular, and immunological factors that are primarily responsible for symptomatology. The signal

midline T1 and T2 spin-echo images which demonstrate a loss of the lipid marrow signal intensity (Fig. 10.5a) typically seen in type II degenerative marrow changes at L4/5. There is now a more mixed signal intensity of the marrow space ( $\mathbf{c}$ ). There is subtle increase signal intensity on the T2-weighted images ( $\mathbf{d}$ )

intensity changes on MR are a secondary reflection. However, I believe the available data would support that type I marrow changes are more



**Fig. 10.6** Type I marrow conversion following lumbar fusion: sagittal midline T1 spin-echo images of the lumbar spine preoperatively (**a**) and postoperatively (**b**). Figure 10.6c, d is sagittal midline T2-weighted images of the lumbar spine preoperative and postoperative, respec-

strongly associated with symptomatology than type II and more fluid, and their resolution or change is more common and associated with clinical improvement.

Type III degenerative marrow changes are the least common and probably are a reflection of end-stage degenerative disk disease. There is no enough data to make meaningful comments about its relationship to symptomatology or even the preceding two types.

In a study by Toyone [13], 70 % of patients with type I marrow changes had segmental hypermobility versus 16 % with type II. Probably the greatest support for suggesting these marrow changes, particularly type I, is related to biomechanical instability and is based on observations following fusion. Chataigner [27] has suggested that type I marrow changes have much better outcomes with surgery than those with isolated degenerative disk disease and normal or type II

tively. Note the typical type I degenerative marrow changes at L4/5 preoperatively  $(\mathbf{a}, \mathbf{c})$  which covert to type II degenerative marrow changes  $(\mathbf{b}, \mathbf{d})$ , respectively, following lumbar fusion. Note the laminectomy defect and posterior fluid on the postoperative images  $(\mathbf{b}, \mathbf{d})$ 

marrow changes. In addition, resolution of type I marrow changes to either normal or type II was associated with higher fusion rates and better outcomes. Other studies support the contention that persistence of type I changes after fusion suggests pseudoarthrosis and is associated with a greater percentage of patients with persistent symptoms. Conversely, resolution of type I marrow changes to either normal or type II was associated with higher fusion rates and better outcomes [28–30]. The conclusion then would be that fusion produces greater stability, reduces biomechanical stresses, and accelerates the improvement in the course of type I marrow changes (Fig. 10.6).

As further support for these fluid marrow changes reflecting biomechanical stress, we have seen similar marrow conversion in the pedicles of vertebral bodies associated with symptomatic pars and pedicle fractures as well as severe degenerative facet joint disease. Twentytwo patients with type I marrow changes of the pedicles and back pain were followed longitudinally. The type I pedicle marrow changes resolved in 17 patients but persisted in 5. Selfreported pain scores tended to improve over time with concordant resolution of marrow signal intensity, but this was not statistically significant – functional improvement was. Of the 17 patients with resolution of the type I marrow change, 6 converted to type II and 11 turned to normal marrow signal. This result suggests that the pedicle marrow type I conversion to a normal or type II appearance is associated with improved symptoms [31].

While the data is strong that there is a mechanical etiology to many of these marrow changes, there is a growing body of literature that suggests that in some there is a true infectious or inflammatory cause [32]. Multiple authors have observed a variety of inflammatory mediators in association with degenerative marrow changes. Burke et al. [21] observed an increase in proinflammatory mediators such as interleukin-6, interleukin-8, and prostaglandin E-2 in the disks of patients with type I marrow changes and in patients undergoing fusion for LBP. Ohtori et al. [33] found that the cartilaginous end plates of patients with type I marrow changes had more protein gene product (PGP) 9.5 immunoreactive nerve fibers and tumor necrosis factor (TNF) immunoreactive cells than those with normal end plates. PGP 9.5 immunoreactivity was seen exclusively in patients with diskogenic LBP. TNF immunoreactive cells in end plates with type I marrow changes were higher than those with type II marrow changes. These authors concluded that type I marrow changes represented a more active inflammation by mediated proinflammatory cytokines, whereas type II and type III changes appeared to be more quiescent [3]. Korhonen [34] in a study of infliximab, a monoclonal antibody against TNF-alpha, suggests it was most effective when there were degenerative type I marrow changes at the symptomatic level. Nevertheless, the relationship to immunobiologic and cellular response mechanisms, while probably important, remains unclear.

In patients with the low back pain and type I marrow changes, an important differential consideration is vertebral osteomyelitis. Disk space infection typically gives rise to similar vertebral body marrow signal changes which manifested by low signal intensity on T1, high signal intensity on T2, and nonanatomic high signal intensity within the intervertebral disk. Contrast enhancement of the disk and end plates may occur in both disk space infection and degenerative type I marrow changes (Fig. 10.7). The vertebral end plates are usually preserved in degenerative type I marrow changes rather than eroded, destroyed, and indistinct as in disk space infection. There is more often an associated paraspinal or epidural soft tissue mass with disk space infection than degenerative disk and type I marrow changes. C-reactive protein (CRP) and erythrocyte sedimentation rate (ESR) are almost invariably elevated in vertebral osteomyelitis and usually normal in uncomplicated degenerative type I marrow changes [35]. While classic pyogenic and fungal osteomyelitis may, in their earliest stages, overlap in appearance on MR with type I marrow degenerative changes, classic osteomyelitis has a distinctly different clinical and more rapidly changing imaging picture. More recently, it has been proposed that some type I marrow changes which here to fore have been presumed to be degenerative may in fact been secondary to a low virulent anaerobic bacterial process [36].

This hypothesis rests on several observations and studies. Stirling [37] used a newly developed serological test to diagnose deep-seated infections caused by low virulent gram-positive organisms. They studied disk material from 36 patients obtained during microdiskectomy. Nineteen (53 %) had positive cultures after long-term incubation. *Propionibacterium acnes* was the organism isolated in 16 of 19 (84 %). They proposed that these microorganisms colonized degenerated and herniated disks because of access provided by minor trauma and tissue disruption.

Albert reproduced some of these same observations [38]. They acquired disk samples from 61 patients who underwent disk surgery and subjected them to microbiologic analysis. All



**Fig. 10.7** Type I marrow conversion following lumbar fusion: sagittal midline T1 spin-echo images of the lumbar spine preoperatively (**a**) and postoperatively (**b**). Figure 10.7c, d is sagittal midline T2-weighted images of the lumbar spine preoperative and postoperative, respec-

patients then had an MR at baseline prior to surgery and 1-2 years postsurgery and disk sampling. Microbiologic cultures were positive in 28 of 61 (46 %) of which 26 (43 %) grew anaerobic organisms and 4 (7 %) dual microbial organisms. In the disk with anaerobic bacteria, 80 % subsequently developed new type I marrow changes on MR. This study confirmed the original findings of Stirling et al. [37] demonstrating the extruded nuclear material frequently has microorganisms present. As a counterargument of contamination being the cause of infection, a control group of 27 patients who had undergone spinal procedures for scoliosis, trauma, or a malignancy were sampled and studied in a similar fashion. In this control group, no organisms were isolated [39]. None of the aerobic positive cultures (presumably contaminates) developed marrow changes. Additionally, in patients where all cultures were negative, only 44 % developed new marrow changes by MR.

tively. Note the typical type I degenerative marrow changes at L4/5 preoperatively  $(\mathbf{a}, \mathbf{c})$  which covert to type II degenerative marrow changes  $(\mathbf{b}, \mathbf{d})$ , respectively, following lumbar fusion. Note the laminectomy defect and posterior fluid on the postoperative images  $(\mathbf{b}, \mathbf{d})$ 

The theory of infection as an etiology for marrow changes has been tested further. In a subsequent study, Albert and his group proposed the hypothesis that disk herniations provide an environment conducive to low-grade anaerobic infections. If these infections were responsible for symptoms and marrow changes, they should be responsive to antibiotic therapy. The aim then of their study was to test the efficacy of antibiotic treatment in patients with chronic low back pain and MC type I. In order to test this hypothesis, the authors performed a double-blind randomized controlled trial in a patient cohort with chronic low back pain. The initial group was composed of patients with low back or low back and leg pain with a demonstrable disk herniation on MR for back or back and leg pain which demonstrated a disk herniation on MR. To be eligible for randomization, the patients would then subsequently have to have had continued chronic low back pain for 6 months or longer and

on follow-up MR examination have developed type I marrow changes. One hundred sixty-two patients meeting the entry criteria were randomized to either 100 days of antibiotic treatment or placebo. Patients treated with antibiotics showed statistically significant improvements compared to the placebo group in all measured parameters including MR appearance. In addition, the authors observed that the improvement achieved in the antibiotic-treated group was greater than those described with other established conservative treatments.

The authors hypothesize that the marrow changes are a side effect of the cytokine propionic acid production from the bacteria entering the adjacent marrow space, presumably through degenerative changes related to disk herniation and underlying degenerative disk disease. Their hypothesis then is following a tear in the outer fibers of the annulus with disk herniation, there is neovascularization and new capillary formation associated with reparative/inflammatory changes surrounding the herniated disk material. Through these new vascular channels, anaerobic bacteria could enter the anaerobic disk environment and produce a slowly developing low virulent infection. The associated marrow changes could be the visible signs of the inflammatory and low-grade bony destructive changes.

#### Conclusion

Degenerative marrow changes appear to be an age-related process associated with degenerative disease. Type I changes are strongly associated with active low back symptoms and probably some degree of biomechanical instability. Type I changes have been suggested to predict an excellent outcome following stabilization with fusion. Resolution of type I marrow changes has been associated with the reduction of symptoms. There can be conversion between types, most commonly type I to type II or normal. Type II changes appear to be more stable over time and less strongly associated with low back pain. They are most common at L4/5 and L5/S1. They may convert to type I with superimposed changes such as infection or accelerated degeneration.

A biomechanical cause is felt likely because of the strong association with degenerative disk disease, improvement with fusion, and relationship to and resolution with changing symptoms. While the data is strong that there is a mechanical etiology to many of these marrow changes, there is a growing body of literature that suggests that in some, they are a reflection of an inflammatory or infectious process facilitated by the degenerative change. The altered signal intensity detected by MR imaging is not, in and of itself, the causal pathological process, but rather a reflection of the causal process that is probably some type of mechanical stress or instability and superimposed or concomitant immunobiologic, cellular, or possibly even infectious response.

#### References

- DeRoss A, Kressel H, Spritzer C, et al. MR imaging of marrow changes adjacent to end plates in degenerative lumbar disc disease. AJR Am J Roentgenol. 1987;149:531–4.
- Modic MT, Steinbert PM, Ross JS, et al. Degenerative disc disease; assessment of changes in vertebral body marrow with MR imaging. Radiology. 1988;166: 193–9.
- 3. Rahme R, Moussa R. The Modic vertebral endplate and marrow changes: pathologic significance and relation to low back pain and segmental instability of the lumbar spine. AJNR. 2008;29:838–42.
- Ross JS, Obuchowski N, Zepp R. The postoperative lumbar spine: evaluation of epidural scar over a 1-year period. AJNR Am J Neuroradiol. 1998;19:183–6.
- Masaryk TJ, Boumphrey F, Modic MT, Tamborrello C, Ross JS, Brown MD. Effects of chemonucleolysis demonstrated by MR imaging. J Comput Assist Tomogr. 1986;10:917–23.
- Modic MT, Masary TJ, Ross JS, et al. Imaging of degenerative disk disease. Radiology. 1988;168: 177–86.
- Ulmer JL, Elster AD, Mathews VP, Allen AM. Lumbar spondylolysis: reactive marrow changes seen in adjacent pedicles on MR images. AJR Am J Roentgenol. 1995;164:429–33.
- Morrison JL, Kaplan PA, Dussault RG, Anderson MW. Pedicle marrow signal intensity changes in the lumbar spine: a manifestation of facet degenerative joint disease. Skelet Radiol. 2000;29:703–7.
- Chamay A, Tshants P. Mechanical influence in bone remodeling: experimental research on Wolff's law. J Biomech. 1972;5:173–80.

- Schweitzer ME. Does altered biomechanics cause marrow edema? Radiology. 1996;198(3):851–3.
- 11. Carter DR. Mechanical loading histories and cortical bone remodeling. Calcif Tissue Int. 1984;36:19–24.
- Roub LW, Gumerman LW, Hanley Jr EN, Clar MW, Goodman M, Herbert DL. Bone stress: a radionuclide imaging perspective. Radiology. 1979;132:431–8.
- Toyone T, Takahashi K, Kitahara H, et al. Vertebral bone marrow changes in degenerative lumbar disc disease: an MRI study of 74 patients with low back pain. J Bone Joint Surg (Br). 1994;76:757–64.
- Mitra D, Cassar-Pullicino VN, McCall IW. Longitudinal study of vertebral type-1 end-plate changes on MR of the lumbar spine. Eur Radiol. 2004;14:1574–81.
- Kuisma M, et al. A three-year follow up of lumbar spine endplate (Modic) changes. Spine. 2006;31(15): 1714–8.
- Albert HB, Manniche C. Modic changes following lumbar disc herniation. Eur Spine J. 2007;16: 977–82.
- Modic MT. Modic type I and type 2 changes. J Neurosurg Spine. 2007;6:150–1.
- Braithwaite I, White J, Saifuddin A, Renton P, Taylor BA. Vertebral end-plate (Modic) changes on lumbar spine MRI: correlation with pain reproduction at lumbar discography. Eur Spine J. 1998;7:363–8.
- Weishaupt D, Zanetti M, Hodler J, et al. Painful lumbar disk derangement: relevance of endplate abnormalities at MR imaging. Radiology. 2001;218: 420–7.
- Sandhu HS, Sanchez-Caso LP, Parvataneni HK, Cammisa Jr FP, Girardi FP, Ghelman B. Association between findings of provocative discography and vertebral endplate signal changes as seen on MRI. J Spinal Disord. 2000;13:438–43.
- Burke JG, Watson RW, McCormack D, et al. Intervertebral discs which cause low back pain secrete high levels of proinflammatory mediators. J Bone Joint Surg (Br). 2002;84:196–201.
- Karchevsky M, Schweitzer ME, Carrino JA, et al. Reactive endplate marrow changes: a systematic morphologic and epidemiologic evaluation. Skelet Radiol. 2005;34:125–9.
- Boos N, Weissbach S, Rohrbach H, et al. Classification of age-related changes in lumbar intervertebral discs: 2002 Volvo award in basic science. Spine. 2002;27: 2631–44.
- 24. Marshman LA, Trewhella M, Friesem T, et al. Reverse transformation of Modic type II changes to Modic type I changes during sustained chronic low-back pain severity; report of two cases and review of the literature. J Neurosurg Spine. 2007;6:152–5.
- 25. Jensen TS. Quality of like in low back pain patients with MRI-lesions in spinal bone marrow and vertebral endplates (Modic-changes): clinical significant for

outcome of spinal surgery? Scand J Pain. 2014;5: 34-5.

- Määttä J, et al. Association of Modic changes with health-related quality of life among patients referred to spine surgery. Scand J Pain. 2014;5:36–40.
- Chataigner H, Onimus M, Polette A. Surgery for degenerative lumbar disc disease: should the black disc be grafted (in French)? Rev Chir Orthop Reparatrice Appar Mot. 1998;84:583–9.
- Buttermann GR, Heithoff KB, Ogilvie JW, et al. Vertebral body MRI related to lumbar fusion results. Eur Spine J. 1997;6:115–20.
- 29. Esposito P, Pinheiro-Franco JL, Froelich S, et al. Predictive value of MRI vertebral end-plate signal changes (Modic) on outcome of surgically treated degenerative disc disease: results of a cohort study including 60 patients. Neurochirurgie. 2006;52:315–22.
- Vital JM, Gille O, Pointillart V, et al. Course of Modic 1 6 months after lumbar posterior osteosynthesis. Spine. 2003;28:715–20.
- Borg B, Modic M, Ash L, Ross J, Hatem S, Obuchowski N. MR pedicle marrow signal abnormalities. In: RSNA Proceedings, November 22– December 2, Chicago. Oak Brook, IL: RSNA, 2005 (Abstract SSC12–03).
- Albert H, et al. Modic changes, possible causes and relation to low back pain. Med Hypotheses. 2008;70: 361–8.
- 33. Ohtori S, Inoue G, Ito T, et al. Tumor necrosis factorimmunoreactive cells and PGP 9.5-immunoreactive nerve fibers in vertebral endplates of patients with discogenic low back pain and Modic type 1 or type 2 changes on MRI. Spine. 2006;31:1026–31.
- 34. Korhonen T, Karppinen J, Paimela L, et al. The treatment of disc-herniation induced sciatica with infliximab: one year follow-up results of FIRST II, a randomized controlled trial. Spine. 2006;31:2759–66.
- Ross JS, Modic MT. Current assessment of spinal degenerative disc disease with magnetic resonance imaging. Clin Orthop Relat Res. 1992;279: 68–81.
- 36. Albert H, et al. Antibiotic treatment in patients with chronic low back pain and vertebral bone edema (Modic type 1 changes): a double-blind randomized controlled trial of efficacy. Eur Spine J. 2013;22: 697–707.
- Stirling AJ, et al. Association between sciatica and propionibacterium acnes. Lancet. 2001;357:2024–5.
- Albert H, et al. Does nuclear tissue infected with bacteria following disc herniations lead to Modic changes in the adjacent vertebrae? Eur Spine J. 2013;22(4): 690–6.
- Stirling AJ, Jiggins M. Association between sciatica and skin commensals. Cleveland: International Society for the Study of the Lumbar Spine; 2002.

Part III

General Important Aspects When Treating Symptomatic Lumbar Degenerative Disk Disease

# Psychosocial Aspects and Work-Related Issues Regarding Lumbar Degenerative Disk Disease

11

# Clayton L. Haldeman, Andrew Bauer, Daniel K. Resnick, and Nathaniel Brooks

# 11.1 Introduction

The latest Global Burden of Disease Study ranks low back pain as the leading cause of disability worldwide [1]. The one-year incidence has been estimated to be as high as 36 %, while the lifetime prevalence is thought to be close to 80 % [2]. Regardless of treatment most patients enjoy full recovery from their symptoms [3]. However recurrence is common. And while the prevalence is difficult to estimate, some patients go on to develop chronic back pain, typically defined as pain lasting longer than 3 months.

Back pain is not a new ailment. The oldest known surgical text, the Edwin Smith papyrus from 1550 B.C., contains a description of sciatica. However the idea of disability as a result of chronic back pain does seem to be a relatively new concept [4]. There is no reason to think that back pain today is any more severe, frequent, or

D.K. Resnick, MD, MS • N. Brooks, MD (⊠) Department of Neurological Surgery, University of Wisconsin Hospitals and Clinics, Madison, WI, USA

e-mail: n.brooks@neurosurgery.wisc.edu

otherwise different from the pain our ancestors experienced. A more recent book, "The Back Pain Revolution," authored by Gordon Waddell highlights this point [5]. Dr. Waddell discusses his time spent in Oman as it transitioned from an underdeveloped country to become more "westernized." New oil money brought modern medical treatments to this country in the mid-1980s. At the time, patients with back pain flooded into the newly established clinics seeking treatment for their pain. These patients had very similar problems with similar etiology to patients in developed western countries. The interesting part is that nearly none of them were off work or "disabled" from their pain. Waddell's observation was that the patients who were able to escape the confines of their country to have "modern" medical procedures in other countries became disabled after surgery at a much higher rate than those who did not have access to modern medical care. This is an illustration that suggests that low back pain is nothing new, but low back disability is largely a product of modern western medicine. What has changed then? One theory is that as physicians have embraced the scientific method, they have lost touch with the more ancient aspects of medicine, which were equipped to treat the psychological and social aspects of illness.

Over previous centuries, back pain was poorly understood. More recently it was proposed that pain was a direct indication of tissue injury and that repair of the injuring mechanism would

C.L. Haldeman, MD, MHS • A. Bauer, MD

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_11

relieve the pain. The first notion that back pain came from spine and nervous system dysfunction came from Brown in 1828 [6]. This was followed by the discovery of the ruptured disk by Mixter and Barr [7] in 1934. King [8] declared that "pain in the back, as a result of injury, is the most frequent affection for which compensation is demanded from the casualty company." As the twentieth century progressed, physicians imagined that an incomplete understanding of the pathology was the only thing standing in the way of a cure to back pain.

This approach, also known as the disease model of illness, depends upon physical pathology causing symptoms proportional to severity. However, despite the remarkable advances in imaging, surgical technique, antibiotics, and pain medicines that have occurred since the early 1900s, low back pain continues to be one of the most common reasons for loss of work today. It is now understood that some components of low back disability may be a manifestation of actual physical pain, but the vast majority of such may be due to the psychological reaction to pain.

Many clinicians have found the disease model of illness to provide an inadequate understanding of low back pain and have turned to the biopsychosocial model. Psychiatrists pioneered this model in an effort to better understand and treat mental illness; however, many have found it advantageous to think about chronic pain using this model as well [9]. In a review of the model, Borrell-Carrio summarizes it as "... a way of understanding how suffering, disease, and illness are affected by multiple levels of organization, from the societal to the molecular" [10]. Where the disease model assumes disease to be fully accounted for by deviations from the norm of measurable biological variables, the biopsychosocial model "is a way of understanding the patient's subjective experience as an essential contributor to accurate diagnosis, health outcomes, and humane care." In a decade where prescriptions for opioid pain medications have doubled, and low back pain remains the most common cause of disability, the biopsychosocial model provides paradigm shift for understanding chronic pain [11].

# 11.2 Epidemiology and Risk Factors

Studies regarding the epidemiology of low back pain are highly variable. The incidence of developing a new episode of back pain has been estimated to be as low as 4 % and as high as 93 % [12–15]. Larger longitudinal studies indicate that this incidence is much lower, i.e., between 3 and 5 %. The incidence of back pain that did not require professional medical care was much higher at 30 % [13]. Prevalence is difficult to study due to variance among study populations and the varying factors that may affect the development of low back pain. Studies estimate that 15-20 % of adults experience memorable low back pain within 1 year. Up to 80 % experience such pain over a lifetime [16–19].

Back pain varies with age as well. Back problems are more often related to claimed disability during the third to fifth decades. These are the prime working years where low back pain leads to the greatest disability and days off work. Interestingly, the symptoms of low back pain do not worsen with age-related degeneration of intervertebral disks [20-24]. Back pain in the elderly is thought to be one of the most important factors to affecting the state of health [25]. Similar to younger adults, the prevalence of back pain in patients older than 65 has been to be 13-49 % [26], but such pain seems to be more episodic and intermittent with a lesser occurrence of chronic pain [27]. Despite the relatively high prevalence of abnormal curvature of the spine in adolescents, the incidence of low back pain is quite low. Some studies suggest that the peak age for development of back pain in children is 13-14 years. Beyond this age, the risk for developing back pain is similar to that of adults [28-30].

Risk factors for the development of low back pain include demographic, physical, socioeconomic, psychological, and occupational factors. It is typically comorbid with other chronic pain and medical conditions. In one study of chronic spinal pain, 68 % reported some type of other chronic pain, 55 % had chronic illness, and 35 % had a mental disorder [31]. Many studies of these risk factors are small and include only self-reports of the variables. A review by Hildebrandt discusses 55 factors related to the individual and 24 occupational factors that have been linked to low back pain [32]. Many studies have looked at the relationship between socioeconomic status and level of education with the development of back pain. The association seems to be not so much with the incidence of pain, but with the ability to adjust to pain. The incidence of disability from back pain was 22–25 times higher in patients with less than 7 years of education compared with those with college degrees [4].

# 11.3 Observations Regarding Low Back Pain and Disability

Many people live with low back pain without disability. What is the difference between these patients and disabled patients? In order to understand this, it is important to differentiate between pain and disability. Both are related in that they are generally subjectively relayed by the patient and are not viewed the same in any two patients. There is no objective measure for either of these disorders. Pain is an unpleasant feeling often caused by intense or damaging stimuli. Disability is related to the patient's perceptions and attitudes *about* pain [33, 34] and is therefore made up of a host of psychological, social, and cultural issues. It is often based on avoidance, previous painful experiences, and maladaptive coping skills [24, 35, 36].

It is also useful to discuss the difference between acute and chronic pain. Acute pain often bears a close relationship to an inciting event and may be thought to stem directly from tissue injury. Chronic pain, on the other hand, is often due to behavioral adaptations which may have little relationship to the initial physical injury. Therefore it is difficult to treat by medical or surgical means. An example of this is the "failed back syndrome." Chronic pain becomes a syndrome of emotional distress, depression, and disease conviction [24].

Despite low back pain and sciatica taking center stage in many medical circles today, there is

no evidence that the biology of the problem has changed at all over the years. Back pain is the same as it always has been. It is low back disability that is a new concept. Ninety percent of patients with low back pain get better within 6 weeks, in spite of technologically advanced medical and surgical care or interestingly no care at all [3, 21, 24, 37]. This is likely a product of the explosion in the size and complexity of the healthcare systems of western countries and people's perceptions that the abilities of modern health care should be able to "eliminate" pain. Physicians bear some responsibility in this regard. It is they who "certify" patients with pain to be excused from work, thereby amplifying the global problem of low back disability.

## 11.4 Breakdown of the Disease Model of Illness

Recent functional imaging studies have borne out what humans have always intrinsically known pain perception is a multifaceted sensory and emotional experience and thus can be modified by mental, emotional, or sensory mechanisms [38, 39]. It makes most sense, therefore, to attempt to understand primary pain disorders by addressing the entire context of the patient's illness; this is best approached with the biopsychosocial model. This model stresses the integration of the subjective experience of the patient with the objective physical findings of illness, and that both contribute to the patient's perception of disease. Engel, the father of this model, emphasizes the responsibility of the physician to treat the body, but also to assist the patient in understanding and adjusting to their illness, along coping with it mentally.

Just as physicians are often frustrated with the inadequacy of treatments available for back pain, many patients are not satisfied with the office visit either. Without the establishment of a diagnosis based on real pathology, the patient often has difficulties putting their pain in context, which can exacerbate anxiety and illness. Disk disease has become so ubiquitous, and patients may be given the nominal diagnosis of disk prolapse, without any signs of nerve root compression or radiographic evidence. It is not long until this nominal diagnosis is confused with real disease pathology, and the patient receives the label of discogenic low back pain. These patients may eventually be treated with surgery that was not indicated and then "bounce" from clinic to clinic when their "curative operation" failed. Clinics are clogged with these patients, thus making it difficult to care for patients with true pathology. Making matters worse, patients will often go from clinic to clinic until a diagnosis is made, resulting in an incentive for physicians to make nominal diagnoses or risk losing patients. Indeed, a large study of the indications for spinal surgery in the mid-1980s showed that surgical decision-making was often driven by the duration and severity of pain and disability, the patient's illness behavior, and the failure of conservative treatment [40]. As might be expected, the success rate for surgical treatment based on a nominal diagnosis is at best 30-40 %. Interestingly, nearly every study in the last 50 years has shown presence of a psychiatric disease as an extremely poor predictor for good surgical outcome [24, 41–49]. Thus, the responsible surgeon must use the history and physical examination to tease out signs of psychiatric imbalance and consider this carefully prior to proceeding with surgery.

#### 11.5 Work-Related Issues

Since complaints of low back pain peak during productive working years, one must discuss this process as it relates to time off work. First, this problem is most prominent in the group of chronic low back pain patients. In a study by Volinn et al., 2 % of workers eligible for industrial insurance filed a claim for back pain in 1 year. Of those, 12 % were off work for 90 days or more, thus consuming more than 88 % of the wage and medical compensation paid by insurance carriers [50]. This same study found that the complaints of back sprain and pain were closely related to workplace dissatisfaction and monotonous job tasks. The medical costs were largely comprised of surgery and hospital stays for

"medical back problems." A study of Medicare patients found that 71 % of these "medical back" hospitalizations were inappropriate [51]. In their review of low back pain and healthcare utilization, Volinn et al. suggest that the level of both cognitive and economic investment in low back pain drives the therapy [52]. Only when further knowledge and education of outcomes regarding the treatment of low back pain become available and third party payers invoke more stringent guidelines for what will and will not be reimbursed will the trends in surgical and medical management change.

The historical and still common practice of "therapeutic rest" appears to be based on multiple fallacies. First, that pain is related to tissue injury and inflammation in the spine, and that rest will help to reverse or alleviate this process. Second, if the pain does not come from inflammation of the spine, it must come from degenerative disk disease, and the only way to allow the disk to heal is with rest. By the disease model of illness, this seems to be a logical progression, but as previously discussed, the disease model of illness does not translate well into the world of low back pain. Considering the biopsychosocial model and assuming that chronic pain is not due to significant injury or to instability of the spine, this treatment does not make sense at all. It aims to treat a process that likely is not active and fails to treat, and may actually worsen, the psychological aspects of the disease. Therefore, it may encourage the assumption of the "sick role." Indeed, there is only marginal evidence in the literature that suggests that rest improves low back pain or even sciatica. This is a somewhat difficult area to study without a high degree of bias, and as one might expect, the major studies are methodologically flawed. In the majority of these studies, it was found that shorter periods of rest were more beneficial (or less harmful) than longer periods [24]. There have been no studies that suggest that activity worsens pain or tissue injury in the absence of a known pathological correlative lesion. Many of these patients will continue to complain of the same degree of pain, whether they are performing their daily activities or not. It is clear that prolonged rest is harmful to both the

body (bone demineralization [53], cardiac deconditioning [54, 55], loss of muscle strength) and mind with depression and anhedonia [56, 57]. The physician who has prescribed rest to the patient with low back pain has clearly done them no favors in the majority of cases.

Similarly, regarding the notion that a person's back pain can affect their job, the characteristics of their job can affect their back pain. A study by Boos et al. showed that the characteristics of one's job (listlessness, job satisfaction, working in shifts) were more likely than MRI identified disk abnormalities to predict which patients would seek medical treatment [58]. Similarly, these factors also are useful in predicting which patients are likely to be off work at follow-up.

All of these issues have opened up much controversy regarding litigation and workers' compensation in our current healthcare system. In the present system, compensation is largely tied to the presence of physical examination findings and imaging confirmation of disk herniation. Some studies suggest that psychological factors may also be tied in to the selection of patients for workers' compensation benefits and that patients with emotional instability may be less likely to receive compensation [59]. There is decent evidence to suggest that patients who receive timelimited workers' compensation, as opposed to long-term disability, are more likely to return to work and have a good outcome [60]. Another study by Atlas et al. also revealed that patients who were receiving workers' compensation at baseline prior to their low back disability were more likely to be receiving long-term disability benefits than those who were not (27 % vs 7 %)and were also slightly less likely to be working at a 4-year follow-up [61]. This correlates with our earlier assertion that time off work and prolonged compensation benefits allow patients to more easily adopt the sick role.

#### Conclusions

It is clear that low back pain and disability are epidemic in virtually all parts of the industrialized world today [1]. The main differences among countries are manifested by the way back pain is viewed and treated. In the western society, the expectation from patients is generally that they will benefit from surgery, and if not, rest and time off work will be their best treatment options. It is also clear that low back pain does not fit into the classic treatment paradigm of the disease model of illness. In this case, the biopsychosocial factors may be more at play than actual physical tissue injury. These patients place a large burden on the medical system and often bounce from clinic to clinic until they find a physician who will treat them. They then are often the victim of unindicated surgery and fall into the category of failed back syndrome. It is clear that when surgical candidates are chosen carefully and selectively, surgical therapy can lead to the best and most efficient outcomes with early return to work and relief of symptoms. It is also apparent that the traditional method of therapeutic rest is inadequate and may actually lead to a decline in the patient's functional status. The exact role of workers' compensation and disability is still somewhat unclear, but it is likely that these will only reinforce sick behavior. We advocate for a multidisciplinary approach that involves spine surgeons, occupational therapists, physical therapists, mental health professionals, sports medicine specialists, and social workers. Via this method, appropriate surgical candidates can be selected, and the remainder of patients can be funneled into a low back training program that encourages them to become empowered and take control of their "disease" and avoid their "shopping around" for further treatment. Only by addressing all of these issues can patients with low back pain be adequately and efficiently managed.

#### References

- Buchbinder R, Blyth FM, March LM, Brooks P, Woolf AD, Hoy DG. Placing the global burden of low back pain in context. Best Pract Res Clin Rheumatol. 2013;27(5):575–89. doi:10.1016/j.berh.2013.10.007.
- Manchikanti L. Epidemiology of low back pain. Pain Physician. 2000;3(2):167–92.
- Benn RT, Wood PH. Pain in the back: an attempt to estimate the size of the problem. Rheumatol Rehabil. 1975;14(3):121–8.

- Goubert L, Crombez G, De Bourdeaudhuij I. Low back pain, disability and back pain myths in a community sample: prevalence and interrelationships. Eur J Pain. 2004;8(4):385–94. doi:10.1016/j.ejpain.2003.11.004. S1090-3801(03)00143-5 [pii].
- 5. Waddell G. The back pain revolution. 2nd ed. London: Elsevier Limited; 2004.
- Brown T. On irritation of the spinal nerves. Glasgow Med J. 1828;1:131–60.
- Mixter W, Barr J. Rupture of the intervertebral disc with involvement of the spinal canal. N Engl J Med. 1934;211:210–5.
- King H. Injuries of the back from a medical legal standpoint. Tex State J Med. 1915;11:442–5.
- 9. Engel GL. The need for a new medical model: a challenge for biomedicine. Science. 1977;196(4286):129–36.
- Borrell-Carrio F, Suchman AL, Epstein RM. The biopsychosocial model 25 years later: principles, practice, and scientific inquiry. Ann Fam Med. 2004;2(6):576–82. doi:10.1370/afm.245.
- Daubresse M, Chang HY, Yu Y, Viswanathan S, Shah ND, Stafford RS, Kruszewski SP, Alexander GC. Ambulatory diagnosis and treatment of nonmalignant pain in the United States, 2000–2010. Med Care. 2013;51(10):870–8. doi:10.1097/ MLR.0b013e3182a95d86.
- Rubin DI. Epidemiology and risk factors for spine pain. Neurol Clin. 2007;25(2):353–71. doi:10.1016/j. ncl.2007.01.004. S0733-8619(07)00005-9 [pii].
- Papageorgiou AC, Croft PR, Thomas E, Ferry S, Jayson MI, Silman AJ. Influence of previous pain experience on the episode incidence of low back pain: results from the South Manchester Back Pain Study. Pain. 1996;66(2–3):181–5.
- Cassidy JD, Cote P, Carroll LJ, Kristman V. Incidence and course of low back pain episodes in the general population. Spine. 2005;30(24):2817–23. 00007632-200512150-00021 [pii].
- Waxman R, Tennant A, Helliwell P. A prospective follow-up study of low back pain in the community. Spine. 2000;25(16):2085–90.
- Andersson GB. Epidemiology of low back pain. Acta Orthop Scand Suppl. 1998;281:28–31.
- 17. Lawrence RC, Helmick CG, Arnett FC, Deyo RA, Felson DT, Giannini EH, Heyse SP, Hirsch R, Hochberg MC, Hunder GG, Liang MH, Pillemer SR, Steen VD, Wolfe F. Estimates of the prevalence of arthritis and selected musculoskeletal disorders in the United States. Arthritis Rheum. 1998;41(5):778–99. doi:10.1002/1529-0131(199805)41:5<778::AID-ART4>3.0.CO;2-V.
- Hestbaek L, Leboeuf-Yde C, Manniche C. Low back pain: what is the long-term course? A review of studies of general patient populations. Eur Spine J. 2003;12(2):149–65. doi:10.1007/ s00586-002-0508-5.
- Walker BF. The prevalence of low back pain: a systematic review of the literature from 1966 to 1998. J Spinal Disord. 2000;13(3):205–17.

- Fullenlove TM, Williams AJ. Comparative roentgen findings in symptomatic and asymptomatic backs. Radiology. 1957;68(4):572–4.
- Horal J. The clinical appearance of low back disorders in the city of Gothenburg, Sweden. Comparisons of incapacitated probands with matched controls. Acta Orthop Scand Suppl. 1969;118:1–109.
- 22. La Rocca H, Macnab I. Value of pre-employment radiographic assessment of the lumbar spine. Can Med Assoc J. 1969;101(7):49–54.
- Splithoff CA. Lumbosacral junction; roentgenographic comparison of patients with and without backaches. J Am Med Assoc. 1953;152(17):1610–3.
- Waddell G. Low back disability. A syndrome of Western civilization. Neurosurg Clin N Am. 1991;2(4):719–38.
- Cooper JK, Kohlmann T. Factors associated with health status of older Americans. Age Ageing. 2001;30(6):495–501.
- Bressler HB, Keyes WJ, Rochon PA, Badley E. The prevalence of low back pain in the elderly. A systematic review of the literature. Spine. 1999;24(17):1813–9.
- Hartvigsen J, Christensen K, Frederiksen H. Back and neck pain exhibit many common features in old age: a population-based study of 4,486 Danish twins 70–102 years of age. Spine. 2004;29(5):576–80. 00007632-200403010-00019 [pii].
- Salminen JJ, Erkintalo M, Laine M, Pentti J. Low back pain in the young. A prospective three-year follow-up study of subjects with and without low back pain. Spine. 1995;20(19):2101–7; discussion 2108.
- Fairbank JC, Pynsent PB, Van Poortvliet JA, Phillips H. Influence of anthropometric factors and joint laxity in the incidence of adolescent back pain. Spine. 1984;9(5):461–4.
- Burton AK, Clarke RD, McClune TD, Tillotson KM. The natural history of low back pain in adolescents. Spine. 1996;21(20):2323–8.
- 31. Von Korff M, Crane P, Lane M, Miglioretti DL, Simon G, Saunders K, Stang P, Brandenburg N, Kessler R. Chronic spinal pain and physical-mental comorbidity in the United States: results from the national comorbidity survey replication. Pain. 2005;113(3):331–9. doi:10.1016/j.pain.2004.11.010.
- 32. Hildebrandt V. A review of epidemiological risk factors on risk factors of low back pain. In: Buckle P, editor. Musculo-skeletal disorders at work. London: Taylor & Francis; 1987. p. 9–16.
- Melzack R. The McGill pain questionnaire: major properties and scoring methods. Pain. 1975;1(3):277– 99. 0304-3959(75)90044-5 [pii].
- Waddell G, Main CJ, Morris EW, Di Paola M, Gray IC. Chronic low-back pain, psychologic distress, and illness behavior. Spine. 1984;9(2):209–13.
- Lethem J, Slade PD, Troup JD, Bentley G. Outline of a fear-avoidance model of exaggerated pain perception – I. Behav Res Ther. 1983;21(4):401–8. 0005-7967(83)90009-8 [pii].
- Troup JD, Foreman TK, Baxter CE, Brown D. 1987 Volvo award in clinical sciences. The perception of

back pain and the role of psychophysical tests of lifting capacity. Spine. 1987;12(7):645–57.

- 37. Rowe ML. Low back pain in industry. A position paper. J Occup Med. 1969;11(4):161–9.
- Melzack R, Wall PD. Pain mechanisms: a new theory. Science. 1965;150(699):971–9.
- Bushnell MC, Ceko M, Low LA. Cognitive and emotional control of pain and its disruption in chronic pain. Nat Rev Neurosci. 2013;14(7):502–11. doi:10.1038/ nrn3516.
- Waddell G, Morris EW, Di Paola MP, Bircher M, Finlayson D. A concept of illness tested as an improved basis for surgical decisions in low-back disorders. Spine. 1986;11(7):712–9.
- 41. Herron L, Turner J, Weiner P. A comparison of the Millon clinical multiaxial inventory and the Minnesota multiphasic personality inventory as predictors of successful treatment by lumbar laminectomy. Clin Orthop Relat Res. 1986;203:232–8.
- 42. Herron LD, Turner J, Clancy S, Weiner P. The differential utility of the Minnesota multiphasic personality inventory. A predictor of outcome in lumbar laminectomy for disc herniation versus spinal stenosis. Spine. 1986;11(8):847–50.
- Turner JA, Herron L, Weiner P. Utility of the MMPI pain assessment index in predicting outcome after lumbar surgery. J Clin Psychol. 1986;42(5):764–9.
- 44. Wiltse LL, Rocchio PD. Preoperative psychological tests as predictors of success of chemonucleolysis in the treatment of the low-back syndrome. J Bone Joint Surg Am. 1975;57(4):478–83.
- Brown T, Nemiah JC, Barr JS, Barry Jr H. Psychologic factors in lowback pain. N Engl J Med. 1954;251(4):123–8.
- 46. Flynn RJ, Salomone PR. Performance of the MMPI in predicting rehabilitation outcome: a discriminant analysis, double cross-validation assessment. Rehabil Lit. 1977;38(1):12–5.
- Dzioba RB, Doxey NC. A prospective investigation into the orthopaedic and psychologic predictors of outcome of first lumbar surgery following industrial injury. Spine. 1984;9(6):614–23.
- Oostdam EM, Duivenvoorden HJ, Pondaag W. Predictive value of some psychological tests on the outcome of surgical intervention in low back pain patients. J Psychosom Res. 1981;25(3):227–35.
- Pheasant HC, Gilbert D, Goldfarb J, Herron L. The MMPI as a predictor of outcome in low-back surgery. Spine. 1979;4(1):78–84.

- Volinn E, Van Koevering D, Loeser JD. Back sprain in industry. The role of socioeconomic factors in chronicity. Spine. 1991;16(5):542–8.
- Payne SM. Targeting utilization review to diagnostic categories. QRB Qual Rev Bull. 1987;13(12):394–404.
- Volinn E, Turczyn KM, Loeser JD. Theories of back pain and health care utilization. Neurosurg Clin N Am. 1991;2(4):739–48.
- Hansson T, Sandstrom J, Roos B, Jonson R, Andersson GB. The bone mineral content of the lumbar spine in patients with chronic low-back pain. Spine. 1985;10(2):158–60.
- Mayer TG, Gatchel RJ, Kishino N, Keeley J, Capra P, Mayer H, Barnett J, Mooney V. Objective assessment of spine function following industrial injury. A prospective study with comparison group and one-year follow-up. Spine. 1985;10(6):482–93.
- Mayer TG, Gatchel RJ, Kishino N, Keeley J, Mayer H, Capra P, Mooney V. A prospective short-term study of chronic low back pain patients utilizing novel objective functional measurement. Pain. 1986;25(1):53–68. 0304-3959(86)90007-2 [pii].
- McGill CM. Industrial back problems. A control program. J Occup Med. 1968;10(4):174–8.
- Strang J. The chronic disability syndrome. In: Aronoff G, editor. Evaluation and treatment of chronic pain. Baltimore: Urban & Schwarzenberg; 1985.
- 58. Boos N, Semmer N, Elfering A, Schade V, Gal I, Zanetti M, Kissling R, Buchegger N, Hodler J, Main CJ. Natural history of individuals with asymptomatic disc abnormalities in magnetic resonance imaging: predictors of low back pain-related medical consultation and work incapacity. Spine. 2000;25(12):1484–92.
- Gallagher RM, Williams RA, Skelly J, Haugh LD, Rauh V, Milhous R, Frymoyer J. Workers' compensation and return-to-work in low back pain. Pain. 1995;61(2):299–307. 0304-3959(94)00190-P [pii].
- Jamison RN, Matt DA, Parris WC. Effects of timelimited vs unlimited compensation on pain behavior and treatment outcome in low back pain patients. J Psychosom Res. 1988;32(3):277–83. 0022-3999(88)90069-4 [pii].
- Atlas SJ, Chang Y, Kammann E, Keller RB, Deyo RA, Singer DE. Long-term disability and return to work among patients who have a herniated lumbar disc: the effect of disability compensation. J Bone Joint Surg Am. 2000;82(1):4–15.

# Legal Aspects in the Surgical Treatment of Lumbar Degenerative Disk Disease

12

# Vincent C. Traynelis and Ricardo B.V. Fontes

Legal issues and the practice of medicine are intrinsically intertwined. This is especially true in terms of the management of symptomatic lumbar degenerative disk disease (DDD). This chapter will focus on some of the legal aspects as they relate to lumbar DDD with particular emphasis on the surgically treated patient. Medicolegal issues, as they relate to the preoperative, intraoperative, and postoperative periods, will be individually addressed. Additionally perspectives on the management of worker's compensation cases and injuries which are being litigated will be included.

Lumbar surgery is indicated for those patients who experience symptoms due to a surgically treatable lesion despite appropriate nonoperative therapy. Once these conditions are met, a candid discussion should be held with the patient. The purpose of this discussion is to inform the patient of the numerous facts and, in essence, begin the process of obtaining informed consent. Within the USA, the legal concept of informed consent relative to health law began about 50 years ago. The first court records addressing a physician's duty to discuss the details and information concerning a proposed procedure appeared in 1957 [1, 2]. Since then, there have been numerous rulings and judgments related to informed consent, and this concept has been embraced almost universally by patients. Over 90 % of patients want the proper information to allow them to actively participate in their medical decision-making process. This is not always fully appreciated by physicians, and, in fact, many surgeons not only underestimate patients' desire for information but also overestimate how much time they spend obtaining consent [3].

In the USA, informed consent is formed from both civil law (tort law) and constitutional law (codified law). The foundations of US civil law are rooted in decisions regarding injuries inflicted upon one individual by another formulated by the medieval English courts of law. Civil law may be codified into state or federal laws and/or statutes if legislative bodies perceive that such an action is in the best interest of the society. Overall, the law of informed consent is based predominantly on civil law with constitutional laws codifying some essential elements. Informed consent is not a document and does not have to be written but rather a process which begins with a dialogue between the surgeon and the patient. The discussion should establish the expectations of both the patient and surgeon with regard to the procedure. The written documentation serves as a record that the discussion took place, but such notations usually do not completely describe the entire

V.C. Traynelis, MD • R.B.V. Fontes, MD, PhD (⊠) Department of Neurosurgery, Rush University Medical Center, 1725 W Harrison, Suite 855, Chicago, IL 60610, USA e-mail: Vincent\_Traynelis@rush.edu; ricardo\_fontes@rush.edu

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016 J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_12

exchange of information between the patient and physician. It is important to record the key features of the discussion, and certainly the more detailed the written documentation, the greater its benefit should there be some question at a later date as to what transpired during the informed consent process.

The surgeon has many obligations in the informed consent process. The primary duty is to inform the patient and obtain the patient's consent to proceed with a treatment plan in a respectful manner. The information exchange should include the results of pertinent diagnostic studies, explanation of the surgical procedure, and probable outcome of surgery. Alternative management strategies need to be listed, and the risks, benefits, outcomes, and potential complications of these options are reviewed. This portion of the discussion is critical; intentional suppression of such information has been judged by the courts to invalidate consent. Many surgeons do not discuss their specific surgical outcomes but, rather, rely on what is published in selected articles; this practice places the surgeon at great legal risk [4].

Providing reliable information on the outcome and benefits of surgery for degenerative lumbar disk disease is not easy. There are many conflicting opinions regarding the management of symptomatic lumbar spondylosis, and often even experts do not agree on the best treatment. For these reasons, a great effort to present an honest appraisal of the proposed procedure is necessary. It may be reasonable in many instances to explain that surgical treatment of degenerative disk disease is not an exact science, and there are many differing opinions. On the other hand, it is important that the surgeon adheres to the procedure for which the consent was obtained. Failure to do constitutes breach of contract, and battery may also be charged [5].

Spinal surgery dominates neurosurgical malpractice claims in the USA accounting for over 40 % of all claims. A widely publicized study from the New England Journal of Medicine revealed that neurosurgeons are the physicians most prone to face a malpractice claim – the yearly risk may be as high as 19.1 % [6]. Orthopedic surgeons followed closely in fourth place [6]. Most claims faced by neurosurgeons are related to elective lumbar surgery [7, 8]. While the reasons for these claims are many, a few of these complications require mention, particularly the infrequent and catastrophic ones. Introduction of sharp instruments such as rongeurs and curettes into the abdominal cavity may occur if the anterior annulus is violated. These instruments may harm the major blood vessels, ureters, or bowel. Such injuries are potentially life-threatening, and the lumbar surgeon should be aware of their various signs and symptoms and act promptly to diagnosis and treat the problem. Though frequently overlooked when obtaining consent due to its presumed rarity, symptomatic injury of intra-abdominal contents is estimated to occur in around 100 cases per year in the USA [9]. A 1998 study focusing on the medicolegal aspects of this potentially catastrophic injury demonstrates that it can be successfully defended, especially if immediately recognized and treated, but it still resulted in a verdict for the plaintiff or a settlement in 48 % of the time [9].

Postoperative vision loss (POVL) is another infrequent but potentially catastrophic complication of surgery for lumbar degenerative disk disease. POVL can happen due to ischemic optic neuropathy, central retinal artery occlusion, cortical blindness, or another unexplained mechanism [10]. Long associated with prone positioning, other risk factors include elevated blood loss, duration of surgery longer than 6 h, patient comorbidities (particularly diabetes), and systemic hypotension during surgery [11]. Its estimated incidence is 0.2 % of all prone lumbar surgeries, thus making it a very infrequent complication, though with life-altering consequences. Unfortunately, POVL is often not discussed by surgeons during informed consent. The American Society of Anesthesiologists issued an advisory suggesting that a physician "consider" disclosing the risk of POVL to high-risk patients; it does not address who (anesthesiologist versus surgeon), how, or when to do it. A possible reason for surgeons refraining from discussing this complication is a belief that patients would not accept surgery if told of the risk for POVL; however, Corda et al. have recently shown that a vast majority of patients would prefer to be informed of POVL in a face-to-face discussion with the surgeon, and it would still not affect their decision to undergo surgery [12].

Cauda equina syndrome (CES) is also one of the main diagnoses quoted in legal claims related to spine surgery. Despite a low incidence (1-6%)of all disk herniations), it possesses a disproportionately high medicolegal profile [13]. It is a classical neurosurgical teaching that decompression should be achieved on an emergent basis, particularly if the syndrome is incomplete and either residual motor or urinary function is present [14]. Despite a number of meta-analyses published in recent years claiming benefit for decompression within 24 or 48 h, in reality the available evidence is of very low quality. Between 50 % and 70 % of cases have a very quick onset and progress to a complete syndrome rapidly; it may be argued that these patients already have their outcome set by the time they present to the hospital [14]. Ultimately, 75 % of CES patients will recover acceptable urinary function; this number may be higher in patients presenting with an incomplete injury. Most legal claims made over CES cases involve a delay in diagnosis and/ or treatment; therefore, not only physicians of other specialties may be involved as well but actually be the main target of the claim. Gardner et al. reported on 63 claims made in the UK that 48 were directed to a general practitioner or an emergency physician, 2 against radiologists, and 13 to the spine surgeon [14]. Daniels et al. also found a significant correlation between verdict for the plaintiff and time to surgery >48 h [13]. Significant, persistent disability, on the other hand, has not been always associated with increased claims or verdicts for the plaintiff [13–15].

Considerable controversy is found in the recent medical literature over wrong-site and wrong-level surgeries. Wrong-site surgery not only fails to improve the patient's symptoms, but has major medical, legal, social, and emotional implications. An American Academy of Orthopedic Surgeons (AAOS) bulletin report stated, "A successful legal defense to surgery performed on the incorrect limb is almost

impossible" [16]. In 2012, wrong-patient, wrongsite, wrong-procedure-type events were reported to be the second most common sentinel event (12%) by the Joint Commission on Accreditation of Healthcare Organizations (JCAHO) [17]. Preoperative marking of the surgical site, "timeout" verification routines, and intraoperative radiographic verification are some interventions designed to minimize the occurrence of this serious problem [18]. Utilization of intraoperative imaging is believed to be one of the most effective measures to reduce wrong-level surgery -Ammerman et al. demonstrated that incorrect identification of a lumbar disk level by an experienced surgeon without intraoperative imaging may happen in up to 15 % of cases. Despite having been designated a "never event" by the National Quality Forum, there is controversy in grouping wrong-level surgery in the same category as wrong-site, wrong-patient, or wrongprocedure errors [19]. While laterality or identification can be easily determined by anybody without medical training, identification of a specific thoracic or lumbar vertebral level is dependent on variable anatomy and interpretation of radiological studies, and therefore wronglevel errors should not be grouped in the "never event" category [20]. This may explain why over 90 % of spine surgeons report a "close call" of wrong-level error during surgery, and over 50 % admit to having performed a wrong-level procedure at least once; in two thirds of these cases, the error was identified and corrected still during the index procedure [21]. It may also contribute to a perception that while self-verification (i.e., by the patient) and "time-out" procedures have effectively contributed to decrease the number of wrong-site procedures, they may be ineffective to prevent wrong-level errors [20, 21]. In contrast to the fact that wrong-site surgery is virtually indefensible, Goodkin and Laska reviewed 68 cases of wrong-level surgery taken to court up until 2004 and found that in 13 of them a verdict in favor of the surgeon was achieved [9]. Perhaps their most important finding was that there was only one case in which the surgeon identified the wrong-level error during the index procedure and corrected it and was ever taken to court; in that

specific instance, settlement was based on the occurrence of an aortic laceration and death and not the wrong-level surgery [9]. In contrast, failure to identify the error, delays in addressing it, and alteration of medical records have all been associated with higher payments [9, 22].

Degenerative disk disease also has a profound societal impact based on the direct and indirect costs associated with it. Several studies have attempted to estimate the costs associated with LBP utilizing diverse methodology, but a common finding is that the indirect costs resulting from loss of productivity far exceed the direct costs of the treatment itself. In Sweden, Hansson and Hansson calculated that the total costs associated with treatment of degenerative disk disease were equivalent to 1 % of the Swedish gross national product. Direct costs accounted for only 7 % of the total costs; despite these monetary costs, only 28 % of patients placed on disability for DDD ever returned to work. Return-to-work rates after 2 years of disability were virtually nil [23]. Wieser et al. performed a similar study in Switzerland and calculated that the direct costs of treating low back pain in 2005 amounted to 2.6 billion euros or 6.1 % of the national spending in health care. Indirect costs, including social security payments, reached 4.1 billion euros; both figures combined represented 2.3 % of the Swiss GNP during that year [24].

The lumbar surgeon is also confronted with a spectrum of special issues when dealing with the patient who became symptomatic while working. Compensation status has been reported to be associated with poor outcome after surgery since the late nineteenth and early twentieth centuries in conditions compensated through litigation, such as the "railway spine" [25]. This issue has become much more important with the establishment of worker's compensation laws in industrialized countries [26, 27]. Harris et al. performed a meta-analysis on the literature examining the association between compensation status and outcome after surgery [28]. They reviewed 211 publications addressing this topic and found that 175 stated that the presence of compensation (worker's compensation with or without litigation) was associated with a worse outcome, and

35 found no difference or did not describe a difference, and 1 described a benefit associated with compensation [28]. A meta-analysis of 129 studies with available data (n=20,498 patients)revealed the summary odds ratio for an unsatisfactory outcome in compensated patients to be 3.79 (95 % confidence interval, 3.28-4.37 by random effects model). Secondary analyses examining the data by country, procedure, length of follow-up, total follow-up, study type, and type of compensation showed the association to be consistent for all subgroups. This review demonstrated that the higher the compensation incentive, the greater the possibility of a negative outcome [29]. Other potential types of secondary gain such as pending litigation from an alleged accidental nonwork-related injury probably also adversely impact the surgical outcome of patients with lumbar degenerative disk disease.

The current legal system in the USA for adjudication of alleged medical malpractice and determination of damages has shortcomings. Surgeons may be assessed damages for medical malpractice when appropriate care was rendered or forced into a situation where a settlement within the limits of an insurance policy is the most advantageous way out of a lawsuit. Legislative restrictions on medical liability have been helpful in limiting such activity, but a few of these recent legal reforms have been overturned [9, 30]. A myriad of interests are at stake when these discussions are undertaken at the legislative level, and it is uncertain, at this point, if these recent changes will continue or be reverted. Other legal systems allow for a different malpractice environment. In Brazil, a Roman law-based system dictates that civil cases are not decided by a lay jury but by the judge him or herself. While by no means infrequent, the necessity of a strong demonstration of departure from accepted medical standards and harm done to the patient keeps malpractice lawsuits at a minimum, resulting that very few spine surgeons even maintain insurance coverage for legal actions. It has been repeatedly affirmed in Brazilian courts that the contract between the patient and surgeon is one of the means and not the results: as long as documentation demonstrates that the surgeon was available,

had the necessary expertise, and applied every reasonable effort on the patients' behalf, a final, unfavorable verdict is extremely rare [31]. Despite specific regional differences, two very simple measures, however, are universally known to reduce the number of lawsuits. Patients and their families are extremely unlikely to sue a surgeon they respect and perceive to work on their behalf; similarly, maintenance of poor records has been identified as a major complicating factor in a successful legal defense. Addressing these two simple problems through efficient communication and documentation is a prerequisite for a successful spine practice; before being surgeons, we should be staunch patient advocates.

#### References

- Alfidi RJ. Informed consent. A study of patient reaction. JAMA J Am Med Assoc. 1971;216(8):1325–9.
- Scarrow AM, Scarrow MR. Informed consent for the neurosurgeon. Surg Neurol. 2002;57(1):63–8; discussion 68–9.
- Faden RR, Beauchamp TL. Decision-making and informed consent: a study of the impact of disclosed information. Soc Indic Res. 1980;7(1–4):313–36.
- Watts C. Informed consent in neurosurgery: a case study. Surg Neurol. 2008;69(4):428–9. doi:10.1016/j. surneu.2007.09.037.
- 5. Kern SI. Battery redefined. Med Econ. 2007;84(16):30.
- Jena AB, Seabury S, Lakdawalla D, Chandra A. Malpractice risk according to physician specialty. N Engl J Med. 2011;365(7):629–36. doi:10.1056/ NEJMsa1012370.
- Fager CA. Malpractice issues in neurological surgery. Surg Neurol. 2006;65(4):416–21. doi:10.1016/j. surneu.2005.09.026.
- Rovit RL, Simon AS, Drew J, Murali R, Robb J. Neurosurgical experience with malpractice litigation: an analysis of closed claims against neuro-surgeons in New York State, 1999 through 2003. J Neurosurg. 2007;106(6):1108–14. doi:10.3171/jns.2007.106.6.1108.
- Goodkin R, Laska LL. Vascular and visceral injuries associated with lumbar disc surgery: medicolegal implications. Surg Neurol. 1998;49(4):358–70; discussion 370–2.
- Zimmerer S, Koehler M, Turtschi S, Palmowski-Wolfe A, Girard T. Amaurosis after spine surgery: survey of the literature and discussion of one case. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2011;20(2):171–6. doi:10.1007/s00586-010-1557-9.

- Baig MN, Lubow M, Immesoete P, Bergese SD, Hamdy E-A, Mendel E. Vision loss after spine surgery: review of the literature and recommendations. Neurosurg Focus. 2007;23(5):E15. doi:10.3171/ FOC-07/11/15.
- Corda DM, Dexter F, Pasternak JJ, Trentman TL, Nottmeier EW, Brull SJ. Patients' perspective on full disclosure and informed consent regarding postoperative visual loss associated with spinal surgery in the prone position. Mayo Clin Proc Mayo Clin. 2011;86(9):865–8. doi:10.4065/mcp.2011.0279.
- Daniels EW, Gordon Z, French K, Ahn UM, Ahn NU. Review of medicolegal cases for cauda equina syndrome: what factors lead to an adverse outcome for the provider? Orthopedics. 2012;35(3):e414–9. doi:10.3928/01477447-20120222-15.
- Gardner A, Gardner E, Morley T. Cauda equina syndrome: a review of the current clinical and medicolegal position. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2011;20(5):690–7. doi:10.1007/s00586-010-1668-3.
- Lavy C, James A, Wilson-MacDonald J, Fairbank J. Cauda equina syndrome. BMJ. 2009;338:b936.
- Levy DA. No defense for wrong-site surgery. AAOS Bull. 1998;46:18.
- Sentinel Event Data Summary; 2012. Available at: http://www.jointcommission.org/sentinel\_event.aspx. Accessed 12 Sep 2013.
- Longo UG, Loppini M, Romeo G, Maffulli N, Denaro V. Errors of level in spinal surgery: an evidence-based systematic review. J Bone Joint Surg Br. 2012;94(11): 1546–50. doi:10.1302/0301-620X.94B11.29553.
- Michaels RK, Makary MA, Dahab Y, et al. Achieving the National Quality Forum's "Never Events": prevention of wrong site, wrong procedure, and wrong patient operations. Ann Surg. 2007;245(4):526–32. doi:10.1097/01.sla.0000251573.52463.d2.
- McCormick PC. Letter to the editor: site versus level. J Neurosurg Spine. 2012;16(3):320–1. doi:10.3171/2011.11.SPINE11963.
- Groff MW, Heller JE, Potts EA, Mummaneni PV, Shaffrey CI, Smith JS. A survey-based study of wrong-level lumbar spine surgery: the scope of the problem and current practices in place to help avoid these errors. World Neurosurg. 2013;79(3–4):585–92. doi:10.1016/j.wneu.2012.03.017.
- Morgan H. Wrong disc space level surgery: medicolegal implications. Surg Neurol. 2004;62(3):278. doi:10.1016/j.surneu.2004.04.018; author reply 278.
- 23. Hansson EK, Hansson TH. The costs for persons sick-listed more than one month because of low back or neck problems. A two-year prospective study of Swedish patients. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2005;14(4):337–45. doi:10.1007/ s00586-004-0731-3.
- 24. Wieser S, Horisberger B, Schmidhauser S, et al. Cost of low back pain in Switzerland in 2005. Eur J Heal Econ HEPAC Health Econ Prev Care. 2010. doi:10.1007/s10198-010-0258-y.

- 25. Thecostsofrailwaycollisions.Lancet.1860;76(1930):195. doi:10.1016/S0140-6736(02)56112-2.
- Bednar JM, Baesher-Griffith P, Osterman AL. Workers compensation. Effect of state law on treatment cost and work status. Clin Orthop. 1998;351:74–7.
- Parker N. Accident litigants with neurotic symptoms. Med J Aust. 1977;2(10):318–22.
- Harris I, Mulford J, Solomon M, van Gelder JM, Young J. Association between compensation status and outcome after surgery: a meta-analysis. JAMA J Am Med Assoc. 2005;293(13):1644–52. doi:10.1001/ jama.293.13.1644.
- 29. Young JN, Shaffrey CI, Laws Jr ER, Lovell LR. Lumbar disc surgery in a fixed compensation population: a model for influence of secondary gain on surgical outcome. Surg Neurol. 1997;48(6):552–8; discussion 558–9.
- Epstein NE. It is easier to confuse a jury than convince a judge: the crisis in medical malpractice. Spine. 2002;27(22):2425–30. doi:10.1097/01. BRS.0000031260.22750.CA.
- Mirabete JF. Manual de direito penal. 5th ed. São Paulo: Editora Atlas; 1990. rev. e ampliada.

# **Cost-Effective Spinal Surgery: Reality or Oxymoron?**

Casey Madura, Daniel K. Resnick, and Nathaniel Brooks

## 13.1 Introduction

Economic inquiry regarding the cost-effectiveness of medical treatments is necessary but not sufficient for medical decision-making. Nowhere is this more obvious than in spine surgery. Fusion surgery in particular has been singled out as expensive, dangerous, and ineffective [1,2]. Recent congressional inquiries into the relationship between surgeons and industry [3] reflect the depth of concern political leaders, and by proxy the public, have regarding these issues.

This chapter is an overview of costeffectiveness research. A description of the process of cost-effectiveness analysis (CEA) is presented along with several examples of these types of analyses that have been applied to different types of spinal surgery. The goal of this chapter is to provide a working knowledge of the

C. Madura, MD • D.K. Resnick, MD, MS N. Brooks, MD, FAANS (⊠) Department of Neurological Surgery, University of Wisconsin Hospitals and Clinics, Madison, WI, USA e-mail: c.madura@neurosurgery.wisc.edu; resnick@neurosurgery.wisc.edu; elements of a cost-effectiveness analysis and to help you understand the benefits and pitfalls of this type of research. This should aid in determining the quality of cost-effectiveness studies.

# 13.2 Cost-Effectiveness Analysis (CEA)

CEA is a method of comparison of the costs and effects of two competing medical technologies (in this sense, a medical technology is any new therapy, whether medical or surgical). It is a method of study that helps determine the most efficient way to spend resources that have already been committed. CEAs are used to answer the question "What is the best way to spend resources that have already been committed to health care?"

At the most basic level, CEAs are conducted in order to determine the cost of one technology versus another (the comparison technology is often referred to as the *comparator*) relative to the health benefits of each. Cost-utility analysis (CUA) represents a special case of CEA in which the outcome is in terms of cost per QALY.

Health effects (either negative or positive) are often referred to as *health utilities*. By convention, a state of perfect health is given a health utility of 1, while death is given a value of 0. Various levels of health are then assigned values declining from a maximum of 1 with each decrement in health (note that negative utilities are possible,

DOI 10.1007/978-3-662-47756-4\_13

n.brooks@neurosurgery.wisc.edu

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016 J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease,

representing health states worse than death). These utility values are then translated to *quality-adjusted life years* (QALYs).

The difference in cost between the two procedures is then calculated and compared to the difference in benefits. This outcome, the incremental cost-effectiveness ratio (ICER), represents the expenditure necessary to gain one additional unit of the measure of choice (e.g., 1 year of life in perfect health (1 QALY) in a CUA).

The results of these studies can be represented graphically. Figure 13.1 demonstrates the four possible cost-effectiveness scenarios. As you can see the "dominant" situation in the "southeast" corner of the graph is obviously good, i.e., increased benefit with decreased costs. The next situation, the "northeast" corner, represents increased benefit at increased cost. This can be good or bad depending on how costly and how significant the benefit and is by far the most common modern-day scenario. For example, an expensive instrumented fusion following decompression in a patient with a cauda equina syndrome may only slightly improve the QALYs and may not be considered worthwhile the cost. But a simple laminectomy in an incompletely paralyzed patient that allows full recovery would be considered good by almost everyone.

The "northwest" corner of the graph is considered to be bad. This will also give a negative ICER and corresponds to a more expensive worse outcome. This situation does occur from time to time and illustrates the need for high-quality research to investigate "standard" practices. For example, the use of steroids in patients with severe head injury was studied and was found to both worsen outcome and accrue increased costs. The "southwest" corner reflects a less expensive but worse outcome. Again, whether this is a good or bad result can be subjected to interpretation.



**Fig. 13.1** This is a typical graph seen in cost-effectiveness analysis literature comparing incremental costs versus incremental QALYs. The x-axis represents the health benefit in terms of QALYs. The y-axis represents the cost. After a CEA is conducted, if a technology is found to be beneficial from the perspective of QALYs and cost less than its comparator, it is said to be *dominant* (southeast quadrant). Likewise, a technology found to be less beneficial and more costly than its *comparator* is said to be dominated (northwest quadrant). Such results require only common sense to interpret. Most often, however, CEAs

will report that a technology is more beneficial *and* more costly (northeast quadrant) *or* less beneficial and *less costly* (southwest quadrant). The interpretation of such results depends on the perspective of the interpreter. A technology that is marginally better in terms of utility at great cost may not be "cost-effective" if the more important aspect is cost control. On the other hand, a technology that is marginally worse than its comparator but at much less cost may be judged to be cost-effective if cost is the most important aspect of the decision process

At face value, CEA is a relatively simple analysis. There are, after all, only four input categories: (1) cost of the medical technology of interest, (2) cost of the comparator medical technology, (3) health utility associated with the new technology, and (4) health utility associated with the comparator medical technology.

## 13.3 Establishing Health Utilities or What Is a QALY?

QALYs are a unique outcome measure because they represent a composite measure of quality and quantity of life. It was first used to evaluate the effectiveness of hypertension medications [4]. QALYs are derived from health utilities multiplied by time (typically years). This is important to understand because not all QALYs are created equal (see Fig. 13.2). For example, one full year of life in perfect health represents one QALY (Fig. 13.2). Similarly, one full year of life in a health state valued at 0.5 (half as good as perfect health) represents 0.5 QALY (Fig. 13.2). Figure 13.2 provides a demonstration of two very different scenarios in which there are equal QALY values.

Health utilities are assigned using preferencebased health state questionnaires in which populations respond to questionnaires developed to inquire about a number of functional domains. Some commonly used health state questionnaires are the EuroQol-5D, HUI2 and HUI3, etc. The EQ-5D, for example, has 243 possible health states, each with a unique utility value. Utilities for each health state are obtained by asking a general population to rate a sample set of those 243 health states (mathematical modeling was used for the remainder). Instruments such as the HUI2 and HUI3 have even more possible states of health (requiring even more mathematical modeling).



**Fig. 13.2** Time, measured in years, is on the *x*-axis. Health utility is measured on the *y*-axis with a value of 1 representing perfect health and 0 representing death. Note that there is space for utilities valued below 0 reserved for health states valued as worse than death. In graphical format, the number of QALYs is equal to the total shaded area. In the case of patient A, one half-year of life was spent in a state of health valued at 0.5 followed by sudden

death, resulting in a total of 0.5 QALY. Patient B, on the other hand, experiences a gradual decrement in health state utility until death at 1 year. This also translates into 0.5 QALY. While QALYs represent a way to reduce benefits across patients and interventions into a comparable number, it is important to understand that two very different life courses can represent the exact same number of QALYs

There are a number of important questions to consider that affect the generalizability of CEA results. First, the assignment of health state utilities is biased based on the group providing the ratings. In the case of the EQ-5D, the utilities were derived based on the preferences of the general British population. Since its initial inception, there have been a number of other utility value sets derived for American, Danish, German, and many other populations. However, if a study uses a health state questionnaire that has not been validated in a particular study population, then the health utilities and thus QALYs derived from these questionnaires should be viewed critically. Second, many important considerations (economic, social, etc.) are omitted due to complexity, insufficient data, or limitations of the perspective used for the study [5,6]. If the economic benefits of return to work or the ability to maintain a higher level of employment, for example, is not considered, then the reported QALY may be understated. Lastly, it is also important to recognize that much of the available data comes from carefully designed clinical trials designed to answer one or maybe two hypotheses. Very little pragmatic data is included in these trials. Much like the narrowly defined randomized control trial, the generalizability of a CEA study that uses this design is often not generalizable to many clinical situations. Like other aspects of CEAs discussed above, rather than invalidating CEA results, knowledge of such limitations helps inform each reader's final interpretation.

# 13.4 Establishing Cost

At first glance, cost seems like the simplest aspect of a CEA. Currency is much easier to understand than utility. Yet, significant heterogeneity exists from hospital to hospital and clinic to clinic clouding the "true cost" of a good. Additionally, confusion over the origins and accuracy of hospital charges makes determining the true cost even more difficult [7]. Finally, there are often costs that are not included in studies. These can include the cost of missing work, not only to the patient but to the company; the cost of the time family members and friends spend providing "free" or informal care to patients; the cost to society of diverting resources that could have been used elsewhere to health care instead. Most frequently these are the costs of informal care, but may also include many other aspects that go into the "true cost" of health care.

# 13.5 Establishing Comparison Technologies

A CEA involves comparison between two existing technologies. The important detail, however, is that one is not free to choose any existing technology as a baseline. Instead, the comparator (i.e., the technology serving as the baseline) to which the new technology is being compared must be the next best treatment that remains costeffective. For example, in evaluating new methods to promote fusion in spinal surgery, it is not acceptable to compare each new intervention to medical therapy alone; rather, the comparison must be between the new technology and the standard surgical fusion procedure (see example below). This has two implications. First, in designing a CEA, it is important to choose a baseline technology that is not already known to be *dominated* (i.e., more effective at less cost, Fig. 13.3a). Second, inappropriate choice of comparators can lead to calculation of an extremely misleading value known as the average cost-effectiveness ratio (ACER) rather than the ICER (Fig. 13.3c).

Drawing conclusions from the CEA literature requires a true understanding of this concept. For example, in 2012, Virk et al. [8] set out to determine the cost-effectiveness of the various options available for fusion grafts. The question is simple enough: in patients undergoing L4–L5 fusion for degenerative spondylolisthesis, which bone graft option (iliac crest bone graft (ICBG), local bone graft (LBG) only, corticocancellous chips (CCC) plus LBG, recombinant human bone morphogenetic protein (RhBMP) plus LBG, demineralized bone matrix (DBM) plus LBG) is the most costeffective in the treatment of these patients? The



Fig. 13.3 (a) A standard cost-effectiveness curve with health effect, often reported as QALYs, on the x-axis and cost on the y-axis. Four different technologies are represented here representing four different treatments for the same disease. Technology D is both more costly and less effective than technology A. This is the definition of dominance, and therefore, technology D would be referred to as being dominated by technology A (sometimes this concept is reported as a negative ICER). Therefore, there is no sense in conducting a CEA for technology D versus any of the other technologies because there is already an alternative that is better in terms of cost and health effect. (b) Lines connecting technology A to technology B and technology B to technology C have been drawn. These lines are a graphical representation of what is measured in a CEA, i.e., the difference in cost compared to the difference in health effect. The slopes of these lines, referred to as Slope AB and Slope BC, respectively, represent the ICERs of technology A compared to technology B and technology B compared to technology C. Remember that the slope of a line on an xy-axis is the amount of change in vertical units (y-axis units) for an increase of one horizontal unit (x-axis units). In this case, Slope AB and Slope BC represent the cost of A relative to B and B relative to C for each increase of one QALY. This is otherwise known as

the ICER. (c) In this case, technology A is compared directly to technology C. The line connecting technology A to technology C has a slope designated "Slope AC." Note that even though technology B derives some benefit for patients relative to technology C, a comparison such as this ascribes all of the health benefits to technology A. This is despite the fact that technology A adds very little benefit on top of technology B at a significantly higher cost. (d) The effect of inappropriately choosing a comparator that is not the next-best non-dominated alternative. If a CEA analysis were performed comparing technology A to technology C, ignoring technology B, the result would be reported as an ICER equal to the slope of the line connecting technology A to technology C, denoted here as slope AC. This is in contrast to Slope AB referring to the slope of the line connecting technology A to technology B. A more horizontal line represents a lesser slope, and therefore, Slope AC is much less than Slope AB. If only slope AC was reported, technology A would appear to be much more cost-effective than it really is. Therefore, understanding the alternatives available and choosing the appropriate alternative technology for comparison are crucial in conducting CEAs and are critical points that must be understood and examined when interpreting a CEA

paper concludes resoundingly that LBG + RhBMP is the most cost-effective method of fusion with an ICER of \$16,595/QALY.

The paper falls short in its methodological integrity in its designation of "no surgical intervention" as the comparator for each of the other types of fusion (p. E206). In this study, it is clear that the purpose is to determine the most costeffective mode of fusion. With "no surgical intervention" as the comparator, the only possible comparison for this paper is "one type of fusion versus nothing". Rather, LBG + DMB, +CCA, and +BMP should each have been compared ONLY to the LBG group. Likewise, ICBG and LBG should have been compared to each other. In each case, these would have been a comparison between a medical technology and its next-best, non-dominated alternative. This choice of comparator limits cost calculations in this paper to ACERs only, taking away any ability to make true cost-effectiveness comparisons with this data.

## 13.6 How Much Should a QALY Cost? It Depends

An additional question to consider is in regard to how to interpret an ICER. After all, the ICER is simply a value stating the cost of one additional QALY. The question remains, "How much *should* a QALY cost?" Basic economics states that, in a free market, the cost of a QALY should be equal to the opportunity cost of treatment. The opportunity cost is the value of the benefits that would have been gained had another alternative been selected. For example, an employed patient choosing to undergo surgery loses earnings as a result of required time off work. Unfortunately, because health care in the USA is far from a free market, tolerable cost-per-QALY values are often arbitrarily assigned. In the USA, a value of \$50,000 has been the predominant figure and can be traced back to the amount Medicare was willing to pay for patients with end-stage renal disease on renal dialysis in a year in the mid-1980s [9]. Why this value was chosen is not clear. Regardless, the arbitrary nature of this figure is best illustrated by the fact that it has not changed since its inception [10].

Some have suggested that true US societal preferences have upper and lower bounds of \$264,000 and \$183,000, respectively [11]. The UK National Institute for Health and Care Excellence (NICE) has never explicitly stated an appropriate ICER value but in general has always accepted interventions with an ICER less than £20,000, evaluated those with an ICER between £20,000 and £100,000 lb on a case-by-case basis, and rejected those with ICERs greater than £100,000 [11].

The appropriate value for an ICER might also change based on the identity of the decision-maker. A perfectly healthy 87-year-old widower in a nursing home who has lost independence due to financial stress may not be willing to pay much for one additional QALY. A young male in his 30s with children may have a completely different threshold regarding his willingness to pay for one additional QALY. What value is appropriate for the hospital administrator trying to provide the best care at the least cost? What about the insurance executive responsible to shareholders and not the patient?

#### 13.7 Sensitivity Analysis

CEA results are often dependent on values that represent a range rather than a single definite value. Sensitivity analysis represents a method to evaluate the effect of such variability, parameter by parameter. Because CEAs typically depend on some type of model with different events, outcomes, probabilities, costs, and utilities, they are dependent on decisions made by the researchers conducting the study and by the available evidence. These decisions are more frequently than not subject to significant debate and controversy. While discussion of the details of this topic is beyond the scope of this chapter, any true CEA should include such a section.

# 13.8 Examples of CEA in Spinal Surgery

This first study is an example of a retrospective analysis that only takes into account hospital costs. Using CEA Angevine et al. [1] wanted to figure out what the benefit was using allograft versus autograft and/or using a plate following anterior cervical diskectomy [12]. The authors determined the estimated cost by reviewing the hospital bills for 78 patients who had undergone treatment for single-level degenerative cervical disease at one institution. Results are reported as cost per QALY gained. The incremental cost-effectiveness ratio (ICER) was \$496 comparing ACDF with autograft to ACDF with allograft. When a plate was added, the ICER increased to \$32,560.

This marked increase in cost can be misleading. This is not the cost of the surgery and hospitalization, roughly \$11,290 for the ACDF with allograft group and \$12,690 for the ACDF with plate group. This is a ratio of difference in cost of these two groups over incremental change in QALY. In this instance, that is, \$1,400/0.043, QALY=\$32,560 per QALY.

The study showed that short-term costs of anterior cervical diskectomy and fusion with plating might be higher than those associated with ACDF with autograft or allograft and that the 5-year cost-effectiveness of the procedures is similar. This study did not include post discharge costs, lost wages, or return to work into its model. Therefore, although all of the patients seem to do well in the short term and long term from their symptoms, the proposed underlying benefits to using instrumentation, faster return to work and life and no need for external orthosis, are not reflected in the results. If such factors were considered, a lower ICER for plate fixation would be expected.

One of the most common procedures in lumbar spine surgery is the treatment of herniated lumbar disks (HLD). Hansson and Hansson performed a cost-effectiveness analysis in 2006 evaluating operative intervention versus conservative management in patients with HLD [13]. This was a cohort study of 184 individuals in Sweden with a 2-year follow-up. Each person from the surgery group was matched to a similar person in the nonsurgery group with the same age, gender, distribution of pain, intensity, and diagnosis with a total of 92 patients per group. Among other variables the EuroQol-5D questionnaire provided a measure of overall healthrelated quality of life. The data from this questionnaire was used to establish a QALY for the cohort. Costs were estimated using both direct (e.g., surgical costs) and indirect costs (e.g., absenteeism from work). The surgical group cost was \$43,119 and the nonsurgical was \$44,638 for the 2-year follow-up. Patients reported a benefit from surgery of 0.327 QALY. The direct cost for improvement of 1 QALY or incremental cost-effectiveness ratio (ICER) was -\$4,648 for the surgical group.

In this situation the negative ICER is a good thing. In cost-effective speak the surgical treatment is said to be "dominant." This means that in this study surgery increases QALYs at less cost than conservative treatment.

Soegaard et al. used cost-effectiveness analysis to evaluate circumferential lumbar fusion versus posterolateral fusion [14]. This study included a total of 148 patients from a single center, randomized, prospective clinical trial in Denmark with a 9-year follow-up. The study was taken from the perspective of societal viewpoint; therefore, all possible resources and activity occurring in relation to the treatment contribute to the overall cost estimates. This study included the cost of surgery reoperations and rehospitalizations, primary care service utilization, medications, patient costs, and productivity costs. All costs were expressed in 2004 US dollars. The study had the unexpected result of finding that circumferential fusion costs are roughly \$50,000 less per QALY than posterolateral fusion and again "dominates." One explanatory variable for this finding may be the lower rate of reoperations seen in the circumferential group. Also important were the costs of primary care, ancillary care (PT/rehab/chiropractors) and patient out-of-pocket expenses in the follow-up period.

# 13.9 Using CEA to Improve Practice

It comes as no surprise that the majority of CEAs originate in countries with socialized health care where the question is how to allocate resources that have already been appropriated. These studies are used to provide a basis for prioritizing treatments within national healthcare budgets. When applied to health care in the USA, CEAs are of much more questionable value. Despite this, CEAs are increasingly being used by hospitals, payers, and providers to justify or deny coverage of procedures.

Polly et al. [15] provided an example of the potential for CEAs to allow comparison across treatment. In their retrospective study of 1,826 lumbar spine fusion patients, the authors compared lumbar fusion surgery to total knee arthroplasty (TKA), total hip arthroplasty (THA), and coronary artery bypass graft (CABG). The outcome measure was clinically significant improvement on the SF-36 (5.42 points) vs. hospital reimbursement. The cost for lumbar fusion was \$7,300, TKA \$6,600, THR \$4,500, and CABG \$22,000. While this study is flawed because of its retrospective nature, heterogeneous cohorts, and unclear demonstration of meaningful benefit, it does provide some data comparing "costeffective" surgical treatments within our society with lumbar fusion surgery.

Studying cost-effectiveness is of little utility when comparing interventions with different outcomes, but when the outcomes are reduced to a comparable measure, i.e., QALYs, CEAs may be able to act as a tiebreaker to help determine which treatment may be of benefit. For example, does PEEK offer any benefit for ACDF compared to allograft? From the clinician's point of view, it is important to carefully inspect all CEAs. Assumptions made can drastically influence the outcome. Concepts such as sensitivity and dealing with uncertainty are not even discussed here. Sensitivity analysis and other concepts such as discounting and extended dominance are beyond this chapter's scope.

The bottom line is this: perfectly conducted CEAs are subject to many significant problems. Many of these are subjective judgments that are necessitated by a lack of data on the exact topic of investigation. Whether or not a clinician agrees with the assumptions made greatly influences the applicability of the paper's findings to a clinician's own patient. Given this limitation, CEAs can only be recommended as an additional tool for clinicians to use in deciding upon a treatment. In countries such as the USA where the healthcare system is fragmented with significant heterogeneity in treatment decisions, CEAs are only likely to be the deciding factor in treatment decisions where everything else is equal. Such situations are understandably rare.

In other countries, where health care is administered within a predetermined budget, the use of CEAs differs. The National Institute for Health and Care Excellence (NICE) in the UK has an entire program dedicated to the development of comprehensive CEAs which represent a crucial component used by the UK's National Health Service in making determinations of which treatments to cover. The true value of a CEA depends almost entirely on the context in which it is being used.

#### 13.10 Final Word

CEAs can be both extremely valuable and extremely misleading. The above discussion sheds light on many of the issues that go into CEAs. At the end of the day, the value of each study depends on whom one is talking to. Much of the data is objective, but decisions about which data to include and exclude are often subjective. In some cases, even the data has an element of subjectivity to it (expert opinion). As should be obvious, CEAs are objective measures reliant upon subjective data. This chapter provides only a cursory introduction to some of the issues that surround the interpretation of CEAs. It is critical to understand that the ICERs generated by these studies represent only one piece of information that should be considered when making policy decisions.

The purpose of this chapter was twofold. First, it is intended to introduce some of the most basic concepts in CEAs. Second, it underscores the subjectivity inherent in many of these studies. Ultimately, CEAs are only one more piece of information rather than *the* answer. In other words, economic evaluation is necessary but not sufficient for medical decision-making. Consideration of all of these limitations requires the reader to insert his/her own judgments as to whether decisions made by the investigators were appropriate. Interpretation of such studies requires significant subjective judgments on the part of the reader. Just as when reading a spine film or performing a procedure, algorithms are helpful.

#### References

- Angevine PD, Zivin JG, McCormick PC. Costeffectiveness of single-level anterior cervical discectomy and fusion for cervical spondylosis. Spine (Phila Pa 1976). 2005;30:1989–97.
- Abelson R. An operation to ease back pain bolsters the bottom line, too. Arthur Ochs Sulzberger, Jr. New York Times. 2003.
- 3. Physician Payments Sunshine Act of 2009. 111th Congress ed. 2009.
- Weinstein MC, Stason WB. Hypertension: a policy perspective. Cambridge, MA: Harvard University Press; 1976.
- Fryback DG, Dasbach EJ, Klein R, et al. The Beaver Dam Health Outcomes Study: initial catalog of health-state quality factors. Med Dec Making: Int J Soc Med Dec Making. 1993;13:89–102.
- Edwards P, Arango M, Balica L, et al. Final results of MRC CRASH, a randomised placebo-controlled trial of intravenous corticosteroid in adults with head injury-outcomes at 6 months. Lancet. 2005;365:1957–9.

- Gawande A. The cost conundrum. The New Yorker: Conde Nast Digital; 2009.
- Virk S, Sandhu HS, Khan SN. Cost effectiveness analysis of graft options in spinal fusion surgery using a Markov model. J Spinal Disord Tech. 2012;25:E204–10.
- Hirth RA, Chernew ME, Miller E, Fendrick AM, Weissert WG. Willingness to pay for a qualityadjusted life year: in search of a standard. Med Dec Making: Int J Soc Med Dec Making. 2000;20: 332–42.
- Ubel PA, Hirth RA, Chernew ME, Fendrick AM. What is the price of life and why doesn't it increase at the rate of inflation? Arch Intern Med. 2003;163:1637–41.
- Braithwaite RS, Meltzer DO, King Jr JT, Leslie D, Roberts MS. What does the value of modern medicine say about the \$50,000 per quality-adjusted life-year decision rule? Med Care. 2008;46:349–56.
- McLaughlin MR, Purighalla V, Pizzi FJ. Cost advantages of two-level anterior cervical fusion with rigid internal fixation for radiculopathy and degenerative disease. Surg Neurol. 1997;48:560–5.
- Hansson E, Hansson T. The cost-utility of lumbar disc herniation surgery. Eur Spine J. 2007;16:329–37.
- Soegaard R, Bunger C, Christensen T, Hoy K, Eiskjaer S, Christensen FB. Circumferential fusion is dominant over posterolateral fusion: cost-utility evaluation of a randomized controlled trial in severe, chronic low back pain. Spine. 2007;32:2405–14.
- Polly D, Branch Jr C, Burkus K, Shaffrey CI, Glassman SD, Gornet MF, Matthews H, Peloza J, Sandhu H, Schuler T, Schwender J. SF36 PCS benefit/cost ratio of lumbar fusion: comparison to other surgical interventions. Spine J. 2005;5S:S53–4.

# Outcome Measures for Spinal Surgery

# Elizabeth P. Norheim, Steven D. Glassman, and Leah Yacat Carreon

# Abbreviations

CER	Comparative	Effectiveness	
	Research		
Cost/QALY	Cost per Quality-A	djusted Life	
	Year Gained		
EQ-5D	European Quality	of Life-5	
	Dimensions		
HRQOL	Health-Related Quality of Life		
ICER	Incremental Cost-Effectiveness		
	Ratio		
MCID	Minimum Clinicall	y Important	
	Difference		
NDI	Neck Disability Index		
ODI	Oswestry Disability Index		
PRO	Patient Reported Outcome		
QALY	Quality-Adjusted Life Year		
RMDQ	Roland-Morris	Disability	
	Questionnaire		
SCB	Substantial Clinical I	Benefit	

E.P. Norheim, MD • L.Y. Carreon, MD, MSc (⊠) Norton Leatherman Spine Center, 210 East Gray Street, Suite 900, Louisville, KY 40202, USA e-mail: bettynorheim@gmail.com; leah.carreon@nortonhealthcare.org

S.D. Glassman, MD

Department of Orthopaedic Surgery, University of Louisville School of Medicine and the Norton Leatherman Spine Center, Louisville, KY, USA e-mail: tana.allgeyer@nortonhealthcare.org

SF-12	Short Form	-12		
SF-36	Short Form-36			
SF-6D	Short Form-6 Dimensions			
SPORT	Spine Outcomes Research Trial			
SRS-22R	Scoliosis	Research	Society-22	
	Revised			
VAS	Visual Ana	logue Scale		

# 14.1 Introduction

The past two decades have seen major advancements in the use of an evidence-based approach to medical treatment, including in the field of spinal surgery. Historically, there has been a lack of patient-reported outcomes (PRO) data to guide surgical decision making. Prior parameters of success in spinal surgery—such as fusion rates, physician assessment, and complications—did not necessarily correlate with clinical outcomes [1-3].

More recently functional outcomes, patient satisfaction, and healthcare costs have become the major focus of spinal surgery research. Newer PRO measures are at the forefront and provide valid data that better reflects the change in overall health status of an individual after treatment. By quantifying health-related quality of life (HRQOL) changes experienced by the patient, these outcome measures allow for a more relevant assessment of treatment effectiveness.

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_14

## 14.2 Outcome Measures Categories

Health-related quality of life (HRQOL) measures assess, through self-reported means, how a patient's physical and mental health is affected over time by a disease process or disability. They can also quantify response to treatment for the specific disease or disability. There are several types of HRQOL measures. These include generic measures, disease-specific measures, pain scales, and health utility scales.

#### 14.2.1 Generic Measures

Generic measures apply to a variety of disease and treatment groups and seek to evaluate multidimensional aspects of health-related function [4]. Originally described by Ware in 1992 to survey health status in the Medical Outcomes Study (MOS) [5], the Medical Outcomes Short Form-36 (SF-36) is the most well known and widely used of these generic outcomes tools. The SF-36 is a 36-item self-administered questionnaire that explores physical and mental health through eight health concepts or domains. These domains include physical functioning, social functioning, general health, mental health, role emotional, role physical, bodily pain, and vitality. From the SF-36 two summary scores can be measured: a physical composite summary score (PCS) and a mental composite summary score (MCS). Using norm-based scoring, all domain scales have a mean of 50 and a standard deviation of 10 based on the general 1998 US population [5]. The SF-36 has been found to be a valid and reliable measure in different disease states, including patients with low back pain [6, 7]. In contrast to other surveys, the SF-36 presents a considerable respondent burden. There are shorter versions available, such as the SF-12 [8] and SF-8 [9], but all items must be answered in order to calculate the PCS and MCS. In addition, the shorter versions do not provide scores across the eight health domains. Lastly, the SF-36, SF-12, and SF-8 require a license to administer and cannot be easily scored in clinic.

Although generic measures translate the effects of treatments across multiple diseases and populations into a numerical value, they do have some limitations. A major disadvantage is that they can miss important components of health evaluation as related to specific diseases or treatments.

#### 14.2.2 Disease-Specific Measures

Disease-specific measures focus on the effects on HRQOL associated with a specific medical condition. Therefore, these disease-specific measures are more likely to detect the effects of a specific intervention on a disease process. The Oswestry Disability Index (ODI) [10, 11], **Roland-Morris** Disability Questionnaire (RMDQ) [12], Neck Disability Index (NDI) [13], and Scoliosis Research Society-22R (SRS-22R) [14–17] examples of spine-specific are measures.

The ODI is a self-administered survey measuring "low back-specific function" on a ten-item scale with six response categories each [10, 11, 18]. Each item is scored from 0 to 5 and then transformed into a 0-100-point scale with the higher the score signifying a greater disability experienced by the patient. Patients scoring between 0 and 20 have minimal disability, between 21 and 40 have moderate disability, between 41 and 60 have severe disability, between 61 and 80 are crippled, and between 81 and 100 are bed-bound or exaggerating their symptoms. Relative to RMDQ, it has less of a floor effect and is better used in populations with more severe disability [12]. ODI can be easily administered, scored, and interpreted in the clinic. This allows it to help guide treatment decisions.

The RMDQ is a measure of function and daily activity limitations. There are 12 items answerable by yes or no, giving a score ranging from 0 to 24. A higher score reflects greater disability. Relative to ODI, it has less of a ceiling effect and is better used in evaluating populations with lesser disability [12, 19].

The NDI is a ten-item self-administered survey measuring disability in patients with

neck pain. Each item is scored from 0 to 5 for a maximum score of 50 [13]. Similar to ODI and RMDQ, a higher score is associated with a greater disability. Some authors may use a percentage score when one section is missed or is not applicable to what is studied, in which case the range of scores would be from 0 to 100 %.

The SRS-22R is a 22-item questionnaire with five domains, including self-image, pain, activity, mental, and satisfaction, measuring disability specific to scoliosis patients [14, 16, 17]. Each domain is scored from 1 to 5, with higher scores indicating improved outcomes. For patients with either adolescent idiopathic scoliosis or adult spinal deformity, the SRS-22R is the most widely used outcome instrument to measure disease burden and the effect of treatment [20–22].

#### 14.2.3 Pain Scales

The visual analogue scale (VAS) [23–26] and numeric rating scales [27, 28] are commonly used pain scales in which the patient subjectively interprets the pain experience and assigns a value to the measurement scale. The VAS is a 100-mm-long horizontal line anchored on the left with "no pain" and on the right with "worst pain experience." Patients mark on the line the point that they feel best represents their current level of pain. The score is determined by measuring in millimeters from the left-hand side of the line to the point that the patient marks. Numeric rating scales are a variant of the VAS and ask patients to rate their pain levels using numeric values, such as from 0 being no pain at all to 10 being the worst pain imaginable [27]. Advantages of pain scales include a low respondent burden, ease of administration, and a universally accepted tool. Limitations include the difficulty in interpreting objectively the subjective nature of the measurement. Pain scales are also difficult to translate when patients' pain is in flux and changes depending on the time of day, level of activity, or other extrinsic factors.

#### 14.3 Health Utility Scales

Health utilities measure the impact of a disease on society by quantifying health status, or change in health status, weighted for societal preference. Societal impact could theoretically be evaluated by valuing change in SF-36 PCS or ODI score. However, these outcome measures do not give a value that can be directly used in economic analysis.

Utilities or health state values are measured using a single index score for each state of health. Scores range from 0 for death to 1 for perfect health. They are weighted for the relative desirability of the health state. Utility scores are combined with life years for use in economic analysis.

Standards for economic evaluations recommend using societal values (utilities or preferences) [29]. There are two common approaches to obtaining "societal health states values," direct and indirect [30, 31]. Direct measurement of value for health states of a representative sample of the population uses methods such as standard gamble, time-trade-off, and visual analogue scale ratings [30]. In standard gamble, the subject is asked to choose between remaining in a state of ill health for a period of time, or undergoing a medical intervention with the gamble that it will either restore them to normal health or lead to death. As implied by the name, time-trade-off method asks respondents to choose between remaining in a state of ill health for a period of time or being restored to normal health but also having a shorter life expectancy. Visual analogue scale asks respondents to rate a state of ill health on a scale from 0 to 100, with 0 being dead and 100 being perfect health. Indirect measurement uses preference-based measurement systems such as the Quality of Well-Being Scale [32], the EuroQol EQ-5D [33], SF-6D [34], or the Health Utilities Index (HUI) [35].

#### 14.3.1 EQ-5D

Established in 1987 by the EuroQol Group, the EQ-5D has two parts. The first part assesses five

dimensions, including mobility, self-care, usual activities, pain/discomfort, and anxiety/depression. Each dimension receives a score based upon patterns of response. From these responses, a simple descriptive profile is created. In addition, from this data set a health index score is made by assigning a value to the health states from a set of population-based preference weights. The second part of EQ-5D is a 20-cm visual analog scale where patients rate their own health status.

#### 14.3.2 SF-6D

Derived from a selection of SF-36 items as a means of economic evaluation, the SF-6D estimates a preference-based health measure and describes 18,000 health states. It was constructed using a selection of 11 items from six of the eight SF-36 health dimensions. A representative sample of the UK general population (n=611) were asked to value a subset of 249 SF-6D states using the valuation technique of standard gamble. Regression models were used to estimate health states. The resultant algorithm can be used to convert SF-36 data at the individual level to societal health state values or preference scores.

#### 14.3.3 Health Utility Score Uses

The value of health utility scores derives from their ability to be converted to a standard measure across populations and diseases through utility weighting and in turn be used to determine cost utility or value.

The most commonly used measure of costeffectiveness is the cost per QALY gained (cost/ QALY). In general, when an intervention cost/ QALY is in the range of \$50,000–\$100,000, the intervention is considered cost-effective [36]. To take into account the local financial situation, some researchers have suggested a cost/QALY of less than the local GDP as a means to define an intervention as cost-effective in that region [37].

Although health utilities have been widely utilized in Europe, there is limited data from other countries with regard to spinal disorders. Fortunately, it has been shown that standard disease-specific measures for spinal disorders— NDI [38], ODI [39], Cervical Spine Outcomes Questionnaire (CSOQ) [40], and the SRS-22R [41] —have been shown to accurately predict SF-6D scores and therefore can be used to determine cost utility and QALY values. With the use of these data transformations [42, 43], utility scores can be created from prior studies and provide a means for cost-effectiveness research.

Healthcare economists and health outcomes researchers use QALYs as a means of communication. Universally, healthcare costs have skyrocketed. Along with these increases has come the importance of cost in healthcare decision making to both the patient and society.

#### 14.4 Relevant Outcomes Measures

Researchers in spinal disorders have been at the forefront of measuring outcomes based on health and cost utility. In the mid-1990s, clinical researchers studied generic measures such as SF-36 to evaluate surgical intervention for common spinal disorders [44, 45]. With HRQOL measures clearly proving to be important, both generic and disease-specific measures—ODI, SF-36, back, and leg pain scores—became the benchmark of clinical outcomes research in spine surgery. However, for most surgeons the observed changes in HRQOL measures do not clearly convey an intrinsic value of the treatment.

Another limitation of outcomes research has stemmed from interpretation of statistical significance. Previously, clinical relevance came from a change in outcome that reached statistical significance. However, as more study sample sizes have become larger and therefore better powered, small changes in outcome may reach statistical significance. These statistically significant differences identified do not always lead to clinically relevant differences. Therefore, thresholds of clinical relevance must be established. Many studies today only report group means or the mean change for a group. This may have some relevance, but a small number of patients who do very well or very poorly may distort the interpretation. Authors should also report on proportion of patients who improve, remain the same, and deteriorate in order to provide the entire story.

#### 14.5 Current and Future Outcomes

With an unclear interpretation of HRQOL measure changes and reaching statistical significance, there has been an increase in use of thresholds to define clinical relevance; the most widely used being the minimum clinically important difference (MCID) [46, 47]. MCID is defined as the smallest change or level of improvement a patient can reproducibly identify as a clinically relevant change [46-49]. The MCID can be thought of as change above the statistical noise. Unlike thresholds that report the group means, MCID reports a distribution of patients who achieve the MCID. While MCID's strength comes from providing a measureable clinical change, its weakness comes from not necessarily providing a worthwhile clinical improvement. The MCID threshold is less than an optimal surgical result. It is more appropriate as a floor value, rather than a goal for defining clinical success [50].

Substantial clinical benefit (SCB)—a threshold for commonly used HRQOL measures in lumbar spine function correlated to a patient's perception of major clinical improvement in outcome—has been suggested as an alternate for MCID. SCB values typically are between 50 and 100 % higher than MCID values for most commonly used HRQOL measures [50]. SCBs are a more optimal surgical result in comparison to MCIDs. SCB provides an important tool to evaluate clinical outcomes based on our expectation and patients' expectation for substantial benefit with interventional treatment.

The future of spine surgery relies upon the community to generate comprehensive and valuable data that provides the appropriate information for clinical analysis. However, the data collection must not be overly burdensome for patients or physicians. As the cost of healthcare

rises, the evaluation of cost versus benefit for treatments becomes more and more critical. The cost per HRQOL score improvement or more likely cost per QALY gained are ways of translating HRQOL to a financial value. The comprehensive and valuable data must include spine-specific measures along with generic measures or health utilities, which permit the assessment of value to society. The ODI, back/leg pain scores, and EQ-5D would provide a core data set for lumbar disorders while simultaneously minimizing respondent burden. This data in turn would provide information on both HRQOL and costeffectiveness, thereby facilitating societal valuation of clinical outcomes.

#### References

- Djurasovic M, Glassman SD, Dimar JR, et al. Does fusion status correlate with patient outcomes in lumbar spinal fusion? Spine (Phila Pa 1976). 2011;36:404–9.
- Gwilym S, Neen D, Birch N. Clinical outcomes of circumferential spinal fusion do not match radiological results despite rigorous patient selection. Int J Spine Surg. 2005;1(2).
- Park JH, Roh SW. Long-term clinical and radiological outcomes following stand-alone PLIF surgery using expandable cylindrical threaded cages in patients with degenerative lumbar spine disease. Acta Neurochir (Wien). 2011;153:1409–16. discussion 1416.
- Radosevich D, Kane R. Conduction health outcomes research. Sudbury: Jones and Barlett Learning; 2011.
- Ware Jr JE, Sherbourne CD. The MOS 36-item shortform health survey (SF-36). I. Conceptual framework and item selection. Med Care. 1992;30:473–83.
- Walsh TL, Hanscom B, Lurie JD, et al. Is a conditionspecific instrument for patients with low back pain/ leg symptoms really necessary? The responsiveness of the Oswestry Disability Index, MODEMS, and the SF-36. Spine (Phila Pa 1976). 2003;28:607–15.
- Guilfoyle MR, Seeley H, Laing RJ. The Short Form 36 health survey in spine disease–validation against condition-specific measures. Br J Neurosurg. 2009;23:401–5.
- Ware Jr J, Kosinski M, Keller SD. A 12-Item Short-Form Health Survey: construction of scales and preliminary tests of reliability and validity. Med Care. 1996;34:220–33.
- Turner-Bowker DM, Bayliss MS, et al. Usefulness of the SF-8TM health survey for comparing the impact of migraine and other conditions. Qual Life Res. 2003;12:1003–10012.

- Fairbank JC, Couper J, Davies JB, et al. The Oswestry low back pain disability questionnaire. Physiotherapy. 1980;66:271–3.
- Fairbank JC, Pynsent PB. The Oswestry Disability Index. Spine (Phila Pa 1976). 2000;25:2940–52. discussion 2952.
- Roland M, Fairbank J. The Roland-Morris Disability Questionnaire and the Oswestry Disability Questionnaire. Spine (Phila Pa 1976). 2000;25:3115–24.
- Vernon H, Mior S. The Neck Disability Index: a study of reliability and validity. J Manipulative Physiol Ther. 1991;14:409–15.
- Asher MA, Lai SM, Glattes RC, et al. Refinement of the SRS-22 health-related quality of life questionnaire function domain. Spine (Phila Pa 1976). 2006;31:593–7.
- Asher M, Min Lai S, Burton D, et al. Discrimination validity of the scoliosis research society-22 patient questionnaire: relationship to idiopathic scoliosis curve pattern and curve size. Spine (Phila Pa 1976). 2003;28:74–8.
- Asher M, Min Lai S, Burton D, et al. Scoliosis research society-22 patient questionnaire: responsiveness to change associated with surgical treatment. Spine (Phila Pa 1976). 2003;28:70–3.
- Asher M, Min Lai S, Burton D, et al. The reliability and concurrent validity of the scoliosis research society-22 patient questionnaire for idiopathic scoliosis. Spine (Phila Pa 1976). 2003;28:63–9.
- Baker D, Pynsent P, Fairbank J. The Oswestry Disability revisited. In: Jenner R, Manchester, editors. Back pain: new approaches to rehabilitation and education. Manchester, UK: Manchester University Press; 1989. p. 174–6.
- Stratford PW, Binkley J, Solomon P, et al. Defining the minimum level of detectable change for the Roland-Morris questionnaire. Phys Ther. 1996;76:359–65. discussion 366-8.
- Berven S, Deviren V, Demir-Deviren S, et al. Studies in the modified scoliosis research society outcomes instrument in adults: validation, reliability, and discriminatory capacity. Spine (Phila Pa 1976). 2003;28:2164–9. discussion 2169.
- Bridwell KH, Berven S, Glassman S, et al. Is the SRS-22 instrument responsive to change in adult scoliosis patients having primary spinal deformity surgery? Spine (Phila Pa 1976). 2007;32:2220–5.
- 22. Bridwell KH, Cats-Baril W, Harrast J, et al. The validity of the SRS-22 instrument in an adult spinal deformity population compared with the Oswestry and SF-12: a study of response distribution, concurrent validity, internal consistency, and reliability. Spine (Phila Pa 1976). 2005;30:455–61.
- Revill SI, Robinson JO, Rosen M, Hogg MIJ. The reliability of a linear analogue for evaluating pain. Anaesthesia. 1976;31:1191–8.

- 24. Downie WW, Leatham PA, Rhind VM, et al. Studies with pain rating scales. Ann Rheum Dis. 1978;37:378–81.
- Huskisson EC. Measurement of pain. J Rheumatol. 1982;9:768–9.
- Jensen MP, Karoly P, Braver S. The measurement of clinical pain intensity: a comparison of six methods. Pain. 1986;27:117–26.
- Jensen MP, Turner JA, Romano JM. Correlates of improvement in multidisciplinary treatment of chronic pain. J Consult Clin Psychol. 1994;62:172–9.
- McCaffery M, Beebe A. Pain: clinical manual for nursing practice. Baltimore: V.V. Mosby Company; 1993.
- Gold M, Franks P, Erickson P. Assessing the health of the nation. The predictive validity of a preferencebased measure and self-rated health. Med Care. 1996;34:163–77.
- Brazier J, Deverill M, Green C, Harper R, Booth A. A review of the use of health status measures in economics evaluation. Health Technol Assess. 1999;3:1–164.
- Arnold D, Girling A, Stevens A, Lilford R. Comparison of direct and indirect methods of estimating health state utilities for resource allocation: review and empirical analysis. BMJ. 2009;339:b2688.
- Kaplan RM. Health-related quality of life in cardiovascular disease. J Consult Clin Psychol. 1988;56: 382–92.
- The EuroQol Group. EuroQol–a new facility for the measurement of health-related quality of life. Health Policy. 1990;16:199–208.
- 34. Brazier J, Roberts J, Deverill M. The estimation of a preference-based measure of health from the SF-36. J Health Econ. 2002;21:271–92.
- 35. Feeny D, Furlong W, Torrance GW, et al. Multiattribute and single-attribute utility functions for the health utilities index mark 3 system. Med Care. 2002;40:113–28.
- Winkelmayer WC, Weinstein MC, Mittleman MA, et al. Health economic evaluations: the special case of end-stage renal disease treatment. Med Decis Making. 2002;22:417–30.
- 37. World Health Organization. Macroeconomics and health: investing in health for economic development: report of the commission on macroeconomics and health. World Health Organization; 2001, Accessed December 02, 2013.
- Carreon LY, Anderson PA, McDonough CM, et al. Predicting SF-6D utility scores from the neck disability index and numeric rating scales for neck and arm pain. Spine (Phila Pa 1976). 2011;36:490–4.
- Carreon LY, Glassman SD, McDonough CM, et al. Predicting SF-6D utility scores from the Oswestry Disability Index and numeric rating scales for back and leg pain. Spine (Phila Pa 1976). 2009;34:2085–9.
- 40. Skolasky RL, Carreon LY, Anderson PA, et al. Predicting health utility scores from the Cervical

Spine Outcomes Questionnaire in a multicenter nationwide study of anterior cervical spine surgery. Spine (Phila Pa 1976). 2011;36(25):2211–6.

- 41. Richardson S, Berven S, Carreon L, et al. Translation of the scoliosis research society outcomes instrument to utility scores for the cost-effectiveness analysis of spine treatments for adult deformity, in IMAST; 2011. Copenhagen.
- 42. Brox JI, Reikeras O, Nygaard O, et al. Lumbar instrumented fusion compared with cognitive intervention and exercises in patients with chronic back pain after previous surgery for disc herniation: a prospective randomized controlled study. Pain. 2006;122:145–55.
- 43. Brox JI, Sorensen R, Friis A, et al. Randomized clinical trial of lumbar instrumented fusion and cognitive intervention and exercises in patients with chronic low back pain and disc degeneration. Spine (Phila Pa 1976). 2003;28:1913–21.
- 44. Albert TJ, Purtill J, Mesa J, et al. Health outcome assessment before and after adult deformity surgery. A prospective study. Spine (Phila Pa 1976). 1995;20:2002–4. discussion p2005.
- 45. Glassman SD, Minkow RE, Dimar JR, et al. Effect of prior lumbar discectomy on outcome of lumbar

fusion: a prospective analysis using the SF-36 measure. J Spinal Disord. 1998;11:383–8.

- 46. Hagg O, Fritzell P, Nordwall A. The clinical importance of changes in outcome scores after treatment for chronic low back pain. Eur Spine J. 2003;12: 12–20.
- 47. Jaeschke R, Singer J, Guyatt GH. Measurement of health status. Ascertaining the minimal clinically important difference. Control Clin Trials. 1989;10:407–15.
- 48. Copay AG, Glassman SD, Subach BR, et al. Minimum clinically important difference in lumbar spine surgery patients: a choice of methods using the Oswestry Disability Index, Medical Outcomes Study questionnaire Short Form 36, and pain scales. Spine J. 2008;8:968–74.
- Copay AG, Subach BR, Glassman SD, et al. Understanding the minimum clinically important difference: a review of concepts and methods. Spine J. 2007;7:541–6.
- Glassman SD, Copay AG, Berven SH, et al. Defining substantial clinical benefit following lumbar spine arthrodesis. J Bone Joint Surg Am. 2008;90: 1839–47.
# Lumbar Degenerative Disk Disease: Workup and Conservative Treatment

15

# William Jeremy Beckworth, Diana K. Sodiq, Lisa Guo Foster, and Walter I. Sussman

Low back pain (LBP) is a common complaint, with over 80 % of the population experiencing an episode of LBP during their lifetime [1]. Recent data from the Centers for Disease Control and Prevention found that 29 % of interviewees had experienced LBP pain at some point during the previous 3 months [2]. Usually, the clinical course is benign and most patients recover within a few months. It has been reported that over 90 % of patients will recover within 3–6 months [1, 3].

Identifying the pain generator of low back pain is often challenging. Several structures have been implicated as possible pain progenitors in chronic low back pain, including the intervertebral disks, zygapophyseal joints, bone, ligaments, fascia, and muscles of the lumbar spine. Axial pain that is thought to originate from the disk is termed "diskogenic" pain. The etiology of low back pain is often controversial, but multiple studies suggest that the intervertebral disk is the most common source of chronic LBP [4–6].

Lumbar degenerative disk disease is a common imaging finding. This finding increases with age and typically does not correlate with pain. However, studies have shown that a degenerative disk can be symptomatic, specifically those degenerative disks with outer annular tears. These tears are chemically sensitized with annular granulation tissue and have nerve fibers containing nociceptive neurotransmitters such as substance P [7, 8].

Degenerative disk disease (DDD) can be seen as early as the second decade of life and is closely linked to age [9]. With progression of degenerative changes of the spine, intervertebral disks transition from hydrated and pliable to dessicated with internal disruption, narrowing of the disk, annular bulging, and changes of the biochemical composition. Degenerative changes are usually observed in the surrounding structures, resulting in osteophyte formation, zygapophyseal joint arthrosis, vertebral endplate changes, and loss of disk interspace height. After the age of 50, degenerative changes are radiographically evident in nearly all individuals [10]. This chapter explores the pathophysiology of these changes and explores the clinical presentation, radiographic correlations, and conservative management for disk-related low back pain.

# 15.1 The Degenerated Disk: Anatomy and Pathophysiology

The disk morphology is composed of the central nucleus pulposus and the outer annulus fibrosis. The vertebral end plates are the top and bottom portions of the vertebrae that interface with the disk.

W.J. Beckworth, MD (⊠) • D.K. Sodiq, DO Emory Spine Center, Attending Physiatrist, Emory University, Assistant Professor of Orthopaedics and Rehabilitation Medicine, Atlanta, GA, USA e-mail: WBECKWO@emory.edu

L.G. Foster, MD • W.I. Sussman, DO Department of Physical Medicine and Rehabilitation,

Emory University, Atlanta, GA, USA

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_15

# 15.1.1 Nucleus Pulposus

The nucleus pulposus is composed of collagen, proteoglycans, and glycoproteins. Aggrecan, the major structural proteoglycan with the disk, maintains tissue hydration. The strong negative charges of the constituent glycosaminoglycan chains attract and bind water. This water composes 90 % of the composition of the disk and provides the viscoelastic properties of the disk and allows compressive forces [10–13]. The nucleus pulposus is also prevented from deforming while loaded by the cartilaginous end plates, located between the intervertebral disk and the adjacent vertebral body. The end plate is composed primarily of hyaline cartilage [14].

# 15.1.2 Annulus Fibrosis

The integrity of the annulus is critical to maintaining disk structure. A normal annulus is comprised of concentric sheets of fibrocartilage, called lamellae, separated by layers of ground substance. The tensile strength of the peripheral annulus is due to the lamellae of collagen fibers which are predominately type I collagen. Type I collagen is the primary type of collage found in tendons elsewhere in the body [14].

# 15.1.3 Disk Innervation

Degenerative changes that occur with aging are difficult to differentiate from those that are pathological. The lack of a clear understanding of the nociceptive source of low back is partially to blame. Initially histological studies were unable to demonstrate that the intervertebral disk was even innervated [15–17]. Improved histological techniques using immunohistochemistry have demonstrated that the innervation of the human intervertebral disk is by the sinuvertebral nerve [16, 17]. In a healthy disk, only the outer annulus is innervated with nerve fibers penetrating as deep as 3 mm into the superficial annulus [18]. However, as part of the degenerative process, the nerve fibers extend deeper into the inner annulus

and nucleus pulposus. Freemont et al. demonstrated nerve growth into the inner third of the annulus and into the nucleus, 46 and 22 % of cases, respectively, in patients with chronic LBP who underwent spinal fusion [19].

This more extensive innervation with penetration deeper into the disk is a potential source of pain. Disk tissue is biologically active, and a mechanical insult will result in a proinflammatory cascade. Inflammatory cytokines, such as tumor necrosis factor alpha and interleukin-1, perpetuate the inflammatory cascade [20]. The ingrowth of small nonmyelinated nerve fibers can expose the nociceptive fibers to inflammatory cytokines and potentially trigger symptomatic DDD [13, 17, 18, 21–23].

# 15.2 The Degenerated Disk: Clinical Presentation

# 15.2.1 History

The importance of taking a good history cannot be overemphasized. This is the most important part of the outpatient visit and can help a clinician in determining an accurate diagnosis as well as directing treatment. There are specific findings on history that can predict pain from a disk etiology, also known as "diskogenic pain." Certain characteristics in the patient's history, such as age, body mass index, gender, to name a few, can better predict diskogenic pain. These characteristics will be reviewed in detail in the following paragraphs.

#### 15.2.1.1 Screening Questions

General screening questions for LBP include characteristics of pain, such as onset, location, quality, severity, duration, progression, aggravating factors, alleviating factors, and associated symptoms. A pain drawing can be a helpful tool [24]. It is important to screen for the so-called red flag symptoms that may suggest a more serious issue. This can include back pain with unexplained weight loss (malignancy), fevers (infection), and progressive neurological changes including weakness, bladder dysfunction, bowel dysfunction, or saddle anesthesia (cauda equina syndrome). Past and ongoing medical issues are important to include. For example, it would be important to know about prior malignancies, systemic steroid use, history of intravenous drug abuse, or an immunocompromised status [25].

#### 15.2.1.2 Location of Pain

Identifying the location of pain is helpful in determining an etiology. Diskogenic LBP is generally axial. Alternatively a disk herniation can lead to radicular symptoms or a combination of axial and radicular symptoms. Regardless, in this context, we are referring to primarily axial pain. It should be noted that the most common location for diskogenic pain is in the lower lumbar region. DePalma reported the most common levels for diskogenic pain were L5–S1 (40 %) followed by L4–L5 (30 %) [26].

The presence of midline low back pain, defined by pain over the spinous process, has been reported as a good indicator for a diskogenic source of pain. If a patient reports midline LBP, the reported positive predictive value for diskogenic pain is 73 %. A patient who does not report midline LBP has 96 % chance of not having diskogenic LBP. This is in comparison to those who report paramidline pain, which is defined by pain more than one fingerbreadth lateral to the midline. Paramidline pain is more often associated with facet joint pain (FJP) or sacroiliac joint pain (SIJP). Therefore, the presence of midline pain can be used as a tool that increases the probability of diskogenic pain and decreases the probability of a facet or sacroiliac joint pain [26].

It has been reported that the "centralization" phenomenon is a predictor of diskogenic pain and further correlates with the idea of a midline component of pain. This phenomenon is the movement of pain away from radicular and more centrally towards midline in response to repeated lumbar movements (McKenzie assessment).

#### 15.2.1.3 Age

As stated, the radiographic finding of a degenerative disk becomes more common as a person ages. However, a symptomatic painful disk is more common in the younger age group. Thus, increases in age are associated with a decreased likelihood of diskogenic pain but an increased likelihood of facet or sacroiliac joint pain, as seen in Fig. 15.1 [27, 28].

Young adults (age 20–35 years) are reported to be more likely to have diskogenic pain as the source of chronic low back pain (70–98 %) regardless of gender or BMI. By age 50, diskogenic pain was still the most likely source of pain (40–65 %), except for women with low BMIs (average 18.5 kg/m2) in which sacroiliac pain was more likely (49 %). By age 65, men, regardless of BMI, were more likely to have a facet pain (30–54 %), while only women with BMI greater than 30 were more likely to have facet pain (46–64 %) [29].

# 15.2.1.4 Aggravating and Alleviating Factors

The literature suggests that that diskogenic LBP is worse with sitting, flexion, and rotational forces. Additionally, it has been reported that increasing intrathoracic pressure (coughing, sneezing, bearing down for a bowel movement, etc.) may ultimately transfer forces to a sensitized disk and aggravate diskogenic pain. Diskogenic pain may be alleviated with lying or standing, which correlates with Nachemson's classical study looking at disk pressures in various positions (Fig. 15.2) [30–32].

By measuring intradiscal pressures, Nachemson's landmark study has provided clinicians with a fundamental understanding of postural changes and its effect on intradiscal pressure. There is a linear relationship between the applied external load and the measured intradiscal pressure, with the highest strains occurring in the posterolateral region of the annulus fibrosus. Compared with pressures in the upright position, reclining reduces intradiscal pressure by 50-80 %, unsupported sitting increases the pressure by 40 %, forward flexion and weight lifting increases the pressure by 100 %, while forward flexion combined with rotation increases pressure by 400 % [30]. Figure 15.2 further shows the various positions and exercises and their corresponding disk pressures [33]. The greatest risk for annular strain is a combination of flexion, axial rotation, and



**Fig. 15.1** Predicted probabilities and 95 % confidence intervals for internal disk disruption (*IDD*), facet joint pain (*FJP*), sacroiliac joint pain (*SIJP*), and other sources

of low back pain (*LBP*) as a function of age (Adapted from DePalma et al. [27]; with permission)



**Fig. 15.2** (a) Relative change in pressure (or load) in the third lumbar disk in various positions in living subjects. (b) Relative change in pressure (or load) in the third lum-



Positions and exercises

bar disk in various muscle-strengthening exercises in living subjects (Adapted from Nachemson [30])

compression forces, placing the most pressure in the posterolateral inner annular zone. These asymmetrical loads on the spine are believed to be the source of chronic mechanical overload and may explain why lumbar disk herniations tend to occur in the posterolateral region of the annulus fibrosis [31].

Additionally, it has been reported that aggravation of pain when rising from a seated position correlates with a probable positive diskogram and diskogenic pain. With zygapophyseal joint pain, there is no provocation of pain when rising from sitting position. SI joint pain on the other hand was associated with rising from sitting but is more unilateral and lacks a midline pain component [34]. Unfortunately, the side of a symptomatic annular tear on imaging does not correlate with the side of a patient's back pain [35].

# 15.2.1.5 BMI

An elevated body mass index is associated with lumbar degenerative disk disease seen on lumbar MRI [36]. However, as noted previously, radiographic changes are not always associated with clinically painful disks [37, 38]. A recent study has indicated that a higher BMI does not correlate with diskogenic low back pain, but rather facet or sacroiliac joint pain [ [39]].

#### 15.2.1.6 Gender

Gender differences also correlate with the source of LBP. Women, when adjusting for age and BMI, had increased odds of sacroiliac joint pain compared to diskogenic pain or facet joint pain. Younger men were more likely to have diskogenic pain as their source for chronic pain.

#### 15.2.1.7 Surgical History

Unfortunately, a history of prior back surgeries can increase an individual's likelihood for chronic LBP. Diskogenic pain is the most common reason for chronic LBP after a lumbar diskectomy (Table 15.1). Although radicular pain usually improves after a diskectomy, some have residual low back pain. It has been shown that the most common etiology in this situation is diskogenic pain (82 %) [40]. Conversely, if there has been a lumbar fusion especially when fused to the

		D 11	c	1. 1	•	
Table 1	5.1	Predictors	of	disk	ogenic r	naın

Midline pain
Young age
Male gender
Smoker
Pain aggravated with seated position or sit to stand
Pain alleviated with supine or standing
History of prior lumbar diskectomy

sacrum, sacroiliac joint pain is the most likely source of low back pain followed by diskogenic pain, facetogenic pain, and soft tissue irritation due to fusion hardware.

#### 15.2.1.8 Psychosocial Considerations

Since pain is a subjective experience, it is also important to explore the psychosocial issues. Evidence of psychosocial stressors can often amplify or prolong the pain response. Factors including emotional stress, anxiety, depression, occupational factors such as poor job satisfaction, involvement in litigation, number of failed previous treatments, poor sleep, avoidance behaviors, and catastrophic thinking all contribute to a better understanding of the patient's condition, diagnosis, and therapy plan and also provide some prognostic value [41, 42]. These psychosocial factors are valuable tools in discriminating between symptomatic and asymptomatic disk herniation [43]. Additionally, it is important to inquire about health habits such as smoking and alcohol and drug use. Smoking has been associated with LBP and degenerative disk findings on imaging studies [44].

# 15.2.2 Physical Examination

The pertinent physical examination findings generally can be found on the focused exam of the lumbar spine. On general inspection, it is not uncommon to see some loss of lumbar lordosis, especially if there is secondary muscle guarding. Range-of-motion (ROM) testing can be utilized. As mentioned before, the presence of midline low back pain over the spinous process has been reported to be a good indicator for diskogenic pain. Thus, important information can be gathered by palpating over the spinous process.

Signs	Description
Distraction	Inconsistent findings of pain when patient is distracted. An example would be radicular pain with a positive straight leg raise, but when distracted there is no pain on seated straight leg raise
Overreaction	Inappropriate, disproportionate reactions to a request. This may manifest with exaggerated verbalization, facial expression, tremors, or collapsing
Regionalization	Motor or sensory abnormalities without anatomic basis. Diffuse give away weakness could go along with this
Simulation	Lumbar pain with a light axial load on the head. Lumbar pain with simultaneous pelvis and shoulder rotation in unison
Tenderness	Exaggerated sensitivity or dramatic reproduction of pain with light touch of the soft tissue or with skin rolling

Table 15.2 Waddell signs

Conversely, if there is tenderness over the paramidline region, this may suggest facet or sacroiliac pain [26, 34]. In addition, paramidline pain can reflect myofascial pain.

With diskogenic low back pain that is axial in location, there generally should not be any associated neurological deficits. Still, it is always prudent on an initial evaluation to evaluate manual motor testing, sensory testing, and reflex exam. These are primarily used to rule out other more concerning issues.

Lastly, some patients may have secondary gain issues. They may present with symptom magnification or malingering. Waddell signs have been described as a way to screen for such patients (Table 15.2). However, some studies have suggested that Waddell signs are not a reliable method to discriminate from organic pain [45].

# 15.3 The Degenerated Disk: Imaging

The high prevalence of degenerative findings on MRI in asymptomatic individuals has led the clinical relevance of these findings to be questioned [46–48].

Still, imaging for chronic LBP is used and plain radiographs are typically the initial imaging choice. MRI is felt to be more helpful in the evaluation of diskogenic LBP [49]. Diskography has been considered by many to be the gold standard, not without controversy. Imaging findings are eloquently discussed elsewhere in this book; thus, they will only be briefly discussed here.

Some studies have shown a positive association between low back pain and reduced signal intensity and/or reduced disk height [50-52]. The relationship between degenerative findings on imaging and low back pain remains controversial, with conflicting conclusions in the literature. Many studies fail to show any association between structural abnormalities and symptoms [53-55]. In addition, abnormal structural findings are not predictive of the future development of low back pain [51].

In a patient with diskogenic pain, one would expect that there would be at least some disk desiccation on T2-weighted images (T2WI). The likelihood of diskogenic pain is unlikely if the disk appears well hydrated with normal architecture on MRI.

Annular fissures have been associated with diskogenic pain. These appear as high intensity zones (HIZs) on T2WI (Fig. 15.3). These were first described by April and Bogduk in 1992; the presence of an HIZ was found to be highly specific for a concordant reproduction of pain on diskography [56]. Other studies found the correlation between HIZ and reproduction of pain on diskography to be low [57-59]. The discrepancy in the literature has led the accuracy and reliability of HIZ as a sign of diskogenic pain to be questioned. A number of studies have described a high prevalence of HIZ in asymptomatic patients, ranging from 24 to 50 % in different studies [60, 61]. It is clear from the high prevalence of HIZs in the asymptomatic population that not all annular tears are painful. Aprill, who first described HIZs, has suggested that true symptomatic annular tears tend to be larger and not the commonly seen small HIZs.

Vertebral body endplate signal changes on MRI were first described by Modic et al. in 1988, when he correlated endplate changes on MRI with histopathological findings [62]. Modic et al.



Fig. 15.3 L5/S1 posterior disk high-intensity zone

described three distinct subchondral bone marrow changes, or Modic changes. Type I changes appear edematous (dark on T1WI, bright on T2WI), type II have a fatty appearance (bright on both T1WI and T2WI), and type III changes appear sclerotic (dark on both T1WI and T2WI). Modic changes appear dynamic and are likely different stages of the same pathological process.

Modic changes have been associated with low back pain. Type I changes are more commonly associated with pain. There have been studies suggesting that type I Modic correlates with LBP 20-73 % of the time and has been associated with the reproduction of low back pain with diskography [53, 62–66]. More severe Modic changes, extending over 25 % of the vertebral height, had a 100 % concordance with pain on diskography in one study [58]. The presence of Modic changes in addition to other degenerative findings (loss of disk height and a reduction of disk signal) increased the specificity for positive diskography from 79 to 97 % [66]. Other studies have failed to demonstrate a statistically significant association between Modic changes and concordant pain on diskography [67–69].

Diskography is thought by many to be the gold standard for diagnosing diskogenic low back pain, although there is controversy. Diskography is considered positive when there is reproduction of concordant pain, outer annular tear (confirmed on post diskography CT scan), low pressure provocation, and normal control disk(s). Numerous studies have confirmed these criteria, suggesting this is a valid and reliable test [70]. However, intense debate continues. Recently, a published study suggested that diskography may cause accelerated progression of degenerative findings. Many clinicians have questioned these findings. Nevertheless, such findings do raise concerns [71].

# 15.4 The Degenerated Disk: Conservative Treatment

There are a variety of treatment options for diskogenic low back pain. Most treatments are focused on conservative options including physical activity, physical therapy, medications, lifestyle changes, complementary medicine, and injections. These treatments often have limitations in diskogenic low back pain. However, there are other interventions like biological injections that have shown promise for the future.

# 15.4.1 Physical Activity

Physical activity is a frequently used modality to address those with diskogenic low back pain. There is evidence to suggest that activity is better than inactivity in dealing with low back pain. Studies have shown that exercise improves pain, global well-being, and physical function [72]. It also produces multiple other health benefits compared to those who are sedentary and decreases risk of long-term disability [73]. Providers should encourage patients to start slow and advance to regular daily exercise. To remain engaged and continuously challenged, patients should be encouraged to vary the types of activity and set personal goals [73, 74]. Although bed rest had traditionally been the treatment for acute low back pain, this theory has been refuted. Patients with acute low back pain who were asked to avoid bed rest fared better than those who were asked to rest in bed for 48 h. By day 7, those who had remained active had more fully recovered compared to the bed rest group. Furthermore, there were no adverse outcomes in the non-bed rest group [75]. It should be emphasized to the patient that light activity can actually enhance the repair process and, conversely, fear avoidance behaviors can lead to a vicious cycle of chronic pain as an outcome [76].

# 15.4.2 Physical Therapy

Physical therapy has been shown to be helpful for those with nonspecific chronic low back pain, both in adult and pediatric populations [77, 78]. Active exercise is generally preferred to passive modalities. There are a number of exercise programs including activity as usual, aerobic activity (e.g., walking, cycling), aquatic activity (pool rehabilitation), directional preference (McKenzie method), flexibility training (e.g., yoga), proprioception/coordination (stability ball, wobble board), stabilization training (e.g., low load exercise targeting abdominal, pelvic, and spinal truck muscles), and strengthen training. There is some data that suggests lumbar strengthening and proper mechanics can increase stability and decrease stress and strain on the spine. In theory, this should help with pain [76].

# 15.4.3 Medications

Medications are commonly used for the management of low back pain. Acetaminophen (paracetamol), nonsteroidal antiinflammatory agents (NSAIDs), muscle relaxants, antidepressants, and pain medication are some of the more commonly prescribed medications for low back pain. A number of systematic reviews in the literature provide good evidence regarding the efficacy of these medications [79].

A number of studies have assessed the effectiveness of acetaminophen (paracetamol) for nonspecific low back pain. These studies suggest that it is helpful and that it is comparable or slightly inferior to NSAID use [79–84]. Acetaminophen has an excellent safety profile when administered in proper therapeutic doses (less than 4000 mg per day), but hepatotoxicity can occur with misuse and overdose. In the United States, acetaminophen toxicity has replaced viral hepatitis as the most common cause of acute hepatic failure and is the second most common cause of liver failure requiring transplantation. In response to this, the FDA in January 2014 issued a statement that combination prescription pain relievers that contain more than 325 mg of acetaminophen per tablet, capsule, or other dosage unit should no longer be prescribed because of a risk of liver damage.

NSAIDs are among the most commonly used medications for pain. They have been shown to be helpful for acute and chronic low back pain [79–81, 85]. There is currently insufficient evidence for aspirin (acetylsalicylic acid) use for low back pain. The mechanism of action of NSAIDs is the blockage of cyclooxygenase activity which converts arachidonic acid to prostaglandin H2, the precursor of prostanoids. COX exists as two isoforms: COX-1, which I is responsible for the hemostatic prostanoid synthesis, and COX-2, which is responsible for proinflammatory prostanoid production. COX-1 is constitutive within platelets and is associated with the production of thromboxane, which strongly promotes platelet aggregation.

NSAIDs have well-documented GI, renal, and cardiovascular side effects. COX-2 medications have less GI side effects, but both nonselective and COX-2 NSAIDs affect the kidney. Some COX-2 medications have been taken off the market because of increased atherothrombotic vascular events [86, 87]. More recent studies have suggested that all NSAIDs have a cardiovascular risk, with naproxen having the least risk [88]. Thus, it is recommended that NSAIDs be used with caution, especially in those with GI, renal, and cardiovascular risks. Additionally, as a person ages, NSAIDs become more risky. In general, NSAIDs should be used for the shortest time period possible.

The U.S. Food and Drug Administration requires that the summaries of product characteristics of all

NSAIDs carry a boxed warning about the risks of cardiovascular disease, whereas the European Medicines Agency's Committee for Medicinal Products for Human Use decided that coxibs (but not NSAIDs) should be contraindicated in patients with coronary heart disease or stroke and used with caution in patients with risk factors for coronary heart disease [89–91].

Muscle relaxants are occasionally used, as well. A Cochrane Review found that these medications are moderately superior to placebo for short-term relief in acute low back pain [92, 93]. There is a lack of evidence in chronic low back pain [82, 94, 95]. This whole class of medication is well known to cause sedation, although serious complications are quite rare.

Antidepressants are also frequently used for low back pain. Two high-quality systematic reviews found antidepressants to be more effective than placebo for pain relief. However, effects on pain were not consistent across antidepressants. Tricyclic antidepressants (TCAs) were slightly to moderately more effective than placebo [95, 96]. Selective serotonin reuptake inhibitors (SSRIs) have not been shown to be effective for pain relief. But serotonin and norepinephrine reuptake inhibitors (SNRIs) have been used for pain relief. In the United States, duloxetine is the only FDA-approved medication for chronic musculoskeletal pain and chronic low back pain.

Common side effects of TCAs include drowsiness, dry mouth, dizziness, and constipation. The common side effects of SSRIs and SNRIs are nausea, sexual side effects, and depression. There is a black box warning of suicidality with these medications. In addition, the SNRIs affect on norepinephrine can potentially elevate blood pressure.

Opioid medications are also used for low back pain. There is moderate evidence for short-term benefit. There is a lack of evidence for long-term use in chronic low back pain [79]. The most common side effects with these medications include dizziness, drowsiness, nausea, constipation, rash, and at high doses respiratory suppression. Yet other issues plague this class of medication. The risk for abuse and misuse poses a problem. The use of opioids has skyrocketed in many countries. For example, the most commonly prescribed medication in the United States is hydrocodone. Further concerns have been raised concerning opioid-induced hyperalgesia (increased sensitivity to pain) in chronic opioid use especially at higher doses [97, 98]. Additionally, opioids are known to affect the hormone system including testosterone, estrogen, thyroid hormones, growth hormones, ACTH/cortisol, and vasopressin [99– 103]. Again, these have generally been documented at higher doses.

#### 15.4.4 Lifestyle Modifications

Lifestyle modifications including smoking cessation, weight loss, and diet are considered an essential part of the comprehensive treatment for chronic low back pain. Multiple studies have demonstrated the association between smoking with low back pain and disk degeneration [104– 107]. Smoking increases the risk of circulatory proinflammatory cytokines and affects healing. Smoking also compromises blood vessel integrity to spinal structures leading to degeneration of disks [105, 107]. Current and former smokers have a higher prevalence and incidence of low back pain compared to those who have never smoked. This association was stronger in adolescents compared to adults showing the importance of avoiding smoking early in life [105].

Obesity is an independent risk factor for development of low back pain [104, 108]. It is believed to have harmful effects on the lumbar spine by creating a biomechanics disadvantage leading to increased load bearing, excessive wear, and early degeneration. There is a positive relationship between body mass index (BMI) and low back pain. Those with BMIs >29 were 1.7 times more likely to have back pain compared to those with BMIs in the lowest 20 % of the population. Furthermore, a prospective cohort study found that patients with BMIs greater than 30 are at increased risk of developing chronic low back pain after an 11-year period compared to those who had BMIs less than 25 [108]. Having a healthy body weight is key for long-term treatment of chronic low back pain and is an addressable risk factor.

Diet is also a modifiable risk factor for the development of degenerative disk disease. Atherosclerotic disease from a typical western diet has been associated with degenerative disk disease in the spine. Postmortem studies have shown that DDD is seen more common when atherosclerotic disease is present in the arteries that supply the specific intervertebral disk. Specifically, aortic calcification and stenosis of the lumbar arteries were both associated with low back pain. Nutrition to the lumbar intervertebral disks is supplied by the lumbar arteries which originate from the abdominal aorta. Arterial occlusion is believed to decrease the vascular supply leading to the degeneration of disk [109, 110]. Other cardiac modifiable risk factors include hypertension, elevated LDL cholesterol, and hypertriglyceridemia [109]. A healthy diet has been shown to decrease the risk of atherosclerotic disease in a number of studies. Lifestyle modifications are an essential step in the treatment and management of chronic low back pain and must be addressed with all patients for optimal management.

# 15.4.5 Complementary Medicine

There is mixed evidence for the use of complementary and alternative therapies in the treatment of chronic low back pain. These therapies can include traction, osteopathic or chiropractic manipulation, acupuncture, herbs, vitamins, minerals, and homeopathic supplements.

Traction has been used historically to treat spine disorders. Multiple theories have been proposed to explain the benefits of traction including changing the disk nerve interface, decreasing nucleus pulposus pressure, and increasing foraminal area; however, the evidence is not clear. A systematic review on four randomized control trials shows that sustained lumbar traction with 30–50 % body weight is no better than low-dose sham traction, mineral baths, underwater massage, or traditional physical therapy for low back pain of greater than 4 weeks' duration [111].

Spinal manipulation and mobilization also dates as far back as 2700 BC in China for the

treatment of low back pain [112]. Spinal manipulation therapy is the use of high-velocity, lowamplitude manual thrusts to the spinal joints slightly beyond the passive range of joint motion, while spinal mobilization is the use of manual force to the spinal joints within the passive range of joint motion that does not involve a thrust [113]. There continues to be mixed evidence regarding efficacy. In a large meta-analysis of randomized clinical trials, there was no evidence to show that spinal manipulation was superior to standard treatment such as general practitioner care, analgesics, physical therapy, exercises, or back school [114]. Another study showed moderate evidence that spinal manipulation is similar in effect to a combination of NSAIDS and exercise in both the short and long term [112]. Given that they are at least as effective as other methods diskussed above, this may be another treatment option for patients.

Another tool to help alleviate low back pain may be acupuncture. Although the precise mechanisms of acupuncture for the treatment of chronic low back pain are not fully understood, it is hypothesized that it stimulates the production of endorphins and inhibits the central nervous system via the gait control theory. There is some evidence from short-term follow-up that suggests when compared to no treatment at all, there is some benefit in pain relief and functional improvement immediately after a series of treatment sessions [115].

Massage, another historical tool for the treatment of pain, can also help in the treatment of nonspecific low back pain, especially in combination with exercise [116].

Transcutaneous electrical stimulation (TENs) is commonly used in the treatment of nonspecific low back pain. However, a 2010 review of the evidence showed no benefit for these devices. As of 2012, Medicare no longer provides coverage for such TENs units for diagnosis codes of low back pain or degenerative disk disease [117].

There are a vast number of herbs and supplements that are not a part of conventional medicine but may help in the management of chronic low back pain. As an example, in a randomized control trial, turmeric (active ingredient curcumin) was compared to ibuprofen for osteoarthritis pain and did equal to slightly better than ibuprofen [118]. With these alternative treatments, there can be side effects that are not always well appreciated. For example, turmeric at higher doses can increase urinary oxalate levels and increase the risk for kidney stones. In general, these alternative treatments have potential benefit but have not been well validated.

# 15.4.6 Spinal Injections and Interventions

There is limited evidence for the use of epidural steroid injections (ESIs) for the management of diskogenic low back pain. There is better evidence for epidurals with radicular pain. Still, epidurals are sometimes used in the treatment of back pain, and there is some literature present on this practice.

One study noted a reduction in chronic back pain after caudal ESI in patients with chronic diskogenic pain without herniation or radiculitis. This was a randomized, double-blinded, controlled trial, which showed significant reduction in pain (defined as > 50 % improvement) and functional status at 1-year follow-up with caudal epidural injections using fluoroscopic guidance. Fifty-five percent of patients who received caudal epidural injections experienced significant pain relief with local anesthetic compared to 68 % of patients who received a combination of local anesthetic and steroid [119]. A similar study using the lumbar interlaminar approach via fluoroscopic guidance at 2-year follow-up also showed significant pain relief and functional status improvement in 72 % receiving local anesthetic alone compared to 67 % receiving a combination of steroid and local anesthetic [120]. These results have not been replicated, and also these studies were limited by the lack of a true placebo control group.

There are a few intradiscal thermal procedures that have been looked at for treatment of diskogenic low back pain. These procedures include intradiscal electrothermal therapy (IDET), disc-TRODE, and biacuplasty. The evidence is not overly convincing for any of these procedures at this point [121]. Although a recent randomized placebo-controlled trial of 64 subjects with biacuplasty showed statistically significant improvements in physical function, pain, and disability compared to the placebo group at 6-month follow-up [122]. Still, there is no overwhelming evidence in support of these procedures.

The most promising of treatments are biological agents including growth factors, cell therapy, use of gene transfer, and tissue engineering to repair the damaged disk and reduce the nociceptive input. There are many features involved in the interplay of diskogenic pain that continue to be poorly understood including oxygen tension, pH, Modic changes, and disk hydration [123].

# 15.4.7 Other Interventions

Other interventions have shown promise as a potential treatment for the future. Protein factors such as bone morphogenetic protein (BMP) 2, BMP 7 (osteogenic protein-1, OP-1), and BMP 14 (growth differentiation factor 5, GDF-5), transforming growth factor beta (TGF-B), insulinlike growth factor, fibrin sealant, platelet-rich plasma, and gene therapy can all hypothetically stimulate matrix production. There have been preclinical animal studies with many of these that show promise with disk restoration [123].

One human study that has shown promise to date has been a stem cell study looking at the intradiscal injection of mesenchymal precursor cells (MPCs) for diskogenic low back pain. This was a double-blinded randomized control of 100 patients. There were two treatment groups that included high- and low-dose MPCs which received 18 million and 6 million MPCs, respectively. The control groups received either an injection of saline or hyaluronic acid into the disk. At 1 year out, mean VAS decreased from 72 to 32 in the high-dose and 70 to 33 in the low-dose MPCs. The high dose was statistically significant compared to the control groups. At 12 months, the 62 % of the high-dose group and 69 % of the low-dose group had >50 % relief of pain with p < 0.05 for both groups compared to control groups. Finally, 52 % of the low dose and 42 % of the high dose had pain scores <20 on the VAS (p 0.05 and 0.01, respectively, vs. control) [124].

Gene therapy also has the potential to promote endogenous synthesis of therapeutic proteins via viral or nonviral vectors. Currently, studies have only been done in animal models. Direct intradiscal injections of BMP-2 complementary deoxyribonucleic acid (cDNA) carried by recombinant adenovirus vector resulted in restoration of disk height in animal models. Cell-based delivery of genes that promote the growth of proteoglycan and collagen production has been shown to be successful in rat models. The safety concerns regarding these methods are still not fully understood; however, gene therapy may be a potential treatment option in the future to restore internal disk disruptions [123].

Unfortunately, there is no gold standard in the treatment of diskogenic low back pain. Conservative measures include physical activity, physical therapy, medications, lifestyle modifications, and educational strategies. If conservative measures fail, a comprehensive multidisciplinary approach along with interventional procedures can be attempted. Further research is required for a variety of treatment methods; however, there is optimism regarding the future of diskogenic treatment, especially with the advances in biological agents.

# References

- 1. Rubin DI. Epidemiology and risk factors for spine pain. Neurol Clin. 2007;25(2):353–71.
- Schiller JS, Lucas JW, Ward BW, Peregoy JA. Summary health statistics for U.S. adults: National Health Interview Survey, 2010. National Center for Health Statistics. Vital Health Stat. 2012;10(252):1.
- Carey TS, Garrett J, Jackman A, et al. The outcomes and costs of care for acute low back pain among patients seen by primary care practitioners, chiropractors, and orthopedic surgeons: the North Carolina back pain project. N Engl J Med. 1995;333(14):913–7.
- Schwarzer AC, Aprill CN, Derby R, et al. The relative contributions of the disc and zygapophyseal joint in chronic low back pain. Spine. 1994;19(7):801–6.
- 5. Schwarzer AC, Aprill CN, Derby R, et al. The prevalence and clinical features of internal disc disrup-

tion in patients with chronic low back pain. Spine. 1995;20(17):1878–81.

- DePalma M. Biologic treatments for discogenic low back pain. SpineLine. 2012;4(3):19–26.
- Crock HV. A reappraisal of intervertebral disc lesions. Med J Aust. 1970;1(20):983–9.
- Peng B, Wu W, Hou S, et al. The pathogenesis of discogenic low back pain. J Bone Joint Surg. 2005;87:62–7.
- Boos N, Weissbach S, Rohrbach H, et al. Classification of age-related changes in lumbar intervertebral discs. Spine. 2002;27:2631–44.
- Slipman C, Derby R, Simeone FA, et al. Interventional spine: an algorithmic approach. Philadelphia: Saunders Elsevier; 2008.
- Bushell GR, Ghosh P, Taylor TF, et al. Proteoglycan chemistry of the intervertebral disks. Clin Orthop Relat Res. 1997;129:115–23.
- Sivan SS, Hayes AJ, Wachtel E, et al. Biochemical composition and turnover of the extracellular matrix of the normal and degenerate intervertebral disc. Eur Spine J. 2013.
- Urban JPG, Roberts S. Review: degeneration of the intervertebral disc. Arthritis Res Ther. 2003;5:120–30.
- Modic MT, Masaryk TJ, Ross JS, et al. Imaging of degenerative disk disease. Radiology. 1988;168:177–86.
- Hirsch C, Schajowicz F. Studies on structural changes in the lumbar annulus fibrosus. Acta Orthop Scand. 1952;22:184–231.
- Roberts S, Eisenstein SM, Menage J, et al. Mechanoreceptors in intervertebral discs: morphology, distribution, and neuropeptides. Spine. 1995;20(24):2645–51.
- Ashton IK, Roberts S, Jaffray C, et al. Neuropeptides in the human vertebral disc. J Orthop Res. 1994;12: 186–92.
- Palmgren T, Gronblad M, Virri J, et al. An immunohistochemical study of nerve structures in the annulus fibrosis of human normal lumbar intervertebral discs. Spine. 1999;24(20):2075–9.
- Freemont AJ, Peacock TE, Goupille P, et al. Nerve ingrowth into diseased intervertebral disc in chronic back pain. Lancet. 1997;350:178–81.
- Weiler C, Nerlich AG, Bachmeier BE, Boos N. Expression and distribution of tumor necrosis factor alpha in human lumbar intervertebral discs: a study in surgical specimen and autopsy controls. Spine. 2005;30:44–53.
- Modic MT, Ross JS. Lumbar degenerative disk disease. Radiology. 2007;245(1):43–61.
- 22. Takahashi H, Suguro T, Okazima Y, et al. Inflammatory cytokines in the herniated disc of the lumbar spine. Spine. 1996;21(2):218–24.
- Konttinen YT, Gronblad M, Antti-Poika I, et al. Neuroimmunohistochemical analysis of peridiscal nociceptive neural elements. Spine. 1990;15(5):383–6.
- Uden A, Landin LA. Pain drawing and myelography in sciatic pain. Clin Orthop. 1987;216:124–30.
- 25. Nachemson A, Vingard E. Assessment of patients with neck and back pain: a best-evidence synthesis.

In: Nachemson AL, Johnsson B, editors. Neck and back pain: the scientific evidence of cause, diagnosis, and treatment. Philadelphia: Lippincott Williams & Wilkins; 2001. (Braddom 879).

- DePalma MJ, et al. Does the location of low back pain predict its source? PM&R. 2011;3(1):33–9.
- Depalma MJ, Ketchum J. What is the source of chronic low back pain and does age play a role? Pain Med. 2011;12:224–33.
- DePalma, Michael J. Interventional Spine Care Director, and Stryker Biotech. Multivariable analysis of the relationship between pain referral patterns and the source of chronic low back pain. Pain Phys. 2012; 15:171–8.
- DePalma MJ, Ketchum JM, Saullo TR. Multivariable analyses of the relationships between age, gender, and body mass index and the source of chronic low back pain. Pain Med. 2012;13(4):498–506.
- Nachemson AL. Disc pressure measurements. Spine. 1981;6(1):93–7.
- Steffen T, et al. Lumbar intradiscal pressure measured in the anterior and posterolateral annular regions during asymmetrical loading. Clin Biomech. 1998;13(7):495–505.
- Schmidt H, et al. Intradiscal pressure, shear strain, and fiber strain in the intervertebral disc under combined loading. Spine. 2007;32(7):748–55.
- Nachemson AL. The lumbar spine an orthopaedic challenge. Spine. 1976;1(1):59–71.
- Young S, Aprill C, Laslett M. Correlation of clinical examination characteristics with three sources of chronic low back pain. Spine J. 2003;3(6):460–5.
- 35. Slipman CW, et al. Side of symptomatic annular tear and site of low back pain: is there a correlation? Spine. 2001;26(8):E165–8.
- 36. Samartzis D. The association of lumbar intervertebral disc degeneration on magnetic resonance imaging with body mass index in overweight and obese adults. Arthritis Rheum. 2012;64(5):1488–96. doi:10.1002/ art.33462.
- Kang CH, et al. Can magnetic resonance imaging accurately predict concordant pain provocation during provocative disc injection? Skeletal Radiol. 2009;38(9):877–85.
- Zhou Y, Abdi S. Diagnosis and minimally invasive treatment of lumbar discogenic pain-a review of the literature. Clin J Pain. 2006;22(5):468–81.
- DePalma MJ, Ketchum JM, Saullo TR. Etiology of chronic low back pain in patients having undergone lumbar fusion. Pain Med. 2011;12(5):732–9.
- 40. DePalma R, et al. Is the history of a surgical discectomy related to the source of chronic low back pain? Pain Physician. 2012;15:E53–8.
- Waddell G. Illness behavior. In: The back pain revolution. Edinburgh: Churchill Livingstone; 2000, (braddom 235).
- 42. Deyo RA, Phillips WR. Low back pain: a primary care challenge. Spine. 1996;21(24):2826–32.
- Boos N, et al. The diagnostic accuracy of magnetic resonance imaging, work perception, and psychoso-

cial factors in identifying symptomatic disc herniations. Spine. 1995;20(24):2613–25.

- 44. Shiri K, et al. The association between smoking and low back pain: a meta-analysis. Am J Med. 2010;123(1):87.e7–35.
- 45. Fishbain DA, Cole B, Cutler RB, Lewis J, Rosomoff HL, Rosomoff RS. A structured evidence-based review on the meaning of nonorganic physical signs: Waddell signs. Pain Med. 2003;4(2):141–81.
- Boden S, Davis D, Dina T, et al. Abnormal magneticresonance scans of the lumbar spine in asymptomatic patients: a prospective investigation. J Bone Joint Surg. 1990;72A:403–8.
- Jensen MC, Brant-Zawadzki MN, Obuchowski N, et al. Magnetic resonance imaging of the lumbar spine in people without back pain. N Engl J Med. 1994;331:69–73.
- Wiesel SW, Tsourmas N, Feffer HL, et al. A study of computer-assisted tomography: the incidence of positive CAT scans in an asymptomatic group of patients. Spine. 1984;9:549–51.
- Malik KM, Cohen SP, Walega DR, et al. Diagnostic criteria and treatment of discogenic pain: a systematic review of recent clinical literature. Spine J. 2013;13:1675–89.
- Kjaer P, Leboeuf-Yde C, Korsholm L, et al. Magnetic resonance imaging and low back pain in adults: a diagnostic imaging study of 40-year-old men and women. Spine. 2005;30:1173–80.
- 51. Borenstein DG, O'Mara Jr JW, Boden SD, et al. The value of magnetic resonance imaging of the lumbar spine to predict low back pain in asymptomatic subjects: a 7-year follow-up study. J Bone Joint Surg Am. 2001;83-A:1306–11.
- Videman T, Battie MC, Gibbons LE, et al. Associations between back pain history and lumbar MRI findings. Spine. 2003;28:582–8.
- 53. Kleinstuck F, Dvorak J, Mannion AF. Are "structural abnormalities" on magnetic resonance imaging a contraindication to the successful conservative treatment of chronic nonspecific low back pain? Spine. 2006;31:2250–7.
- Jarvik JJ, Hollingworth W, Heagerty P, et al. The longitudinal assessment of imaging and disability of the back (LAIDBack) study: baseline data. Spine. 2001;26:1158–66.
- Horton WC, Daftari TK. Which disc as visualized by magnetic resonance imaging is actually a source of pain? Spine. 1992;17:S164–71.
- Aprill C, Bogduk N. High intensity zone: a diagnostic sign of painful lumbar disc on magnetic resonance imaging. Br J Radiol. 1992;65:361–9.
- Ito M, Incorvaia KM, Yu SF, et al. Predictive signs of discogenic lumbar pain on magnetic resonance imaging with discography correlation. Spine. 1998;23(11):1252–60.
- Weishaupt D, Zanetti M, Hodler J, et al. Painful lumbar disk derangement: relevance of endplate abnormalities at MR imaging. Radiology. 2001;218:420–7.

- Ricketson R, Simmons JW, Hauser BO. The prolapsed intervertebral disc: the high-intensity zone with discography correlation. Spine. 1996;21:2758–62.
- Carragee EJ, Chen Y, Tanner CM, et al. Can discography cause long-term back symptoms in previously asymptomatic subjects? Spine. 2000;25:1803–8.
- Stadnik TW, Lee RR, Coen HL, et al. Annular tears and disk herniation: prevalence and contrast enhancement on MR images in the absence of low back pain or sciatica. Radiology. 1998;206:49–55.
- Modic TM, Steinberg PM, Ross JS, et al. Degenerative disk disease: assessment of changes in vertebral body marrow with MR imaging. Radiology. 1988;166:193–9.
- de Roos A, Kressel K, Spritzer C, et al. MR imaging of marrow changes adjacent to end plates in degenerative lumbar disc disease. AJR. 1987;149:531–4.
- Karchevsky M, Schweitzer ME, Carrino JA, et al. Reactive endplate marrow changes: a systematic morphologic and epidemiologic evaluation. Skeletal Radiol. 2005;34:125–9.
- Toyone T, Takahashi K, Kitahara H, et al. Vertebral bone-marrow changes in degenerative lumbar disc disease. J Bone Joint Surg. 1994;76B:757–63.
- 66. Braithwaite I, White J, Saifuddin A, et al. Vertebral endplate (Modic) changes on lumbar spine MRI: correlation with pain reproduction at lumbar discography. Eur Spine J. 1998;7:363–8.
- Sandhu HS, Sanchez-Caso LP, Parvataneni HK, et al. Association between findings of provocative discography and vertebral endplate signal changes as seen on MRI. J Spinal Disord. 2000;13:438–43.
- 68. Kokkonen SM, Kurunlahti M, Tervonen O, et al. Endplate degeneration observed on magnetic resonance imaging of the lumbar spine: correlation with pain provocation and disc changes observed on computed tomography diskography. Spine. 2002;27:2274–8.
- Vital JM, Gille O, Pointillart V, et al. Course of Modic 1 six months after lumbar posterior osteosynthesis. Spine. 2003;28:715–21.
- 70. Manchikanti L, Benyamin RM, Singh V, Falco FJ, Hameed H, Derby R, Wolfer LR, Helm 2nd S, Calodney AK, Datta S, Snook LT, Caraway DL, Hirsch JA, Cohen SP. An update of the systematic appraisal of the accuracy and utility of lumbar discography in chronic low back pain. Pain Physician. 2013;16(2 Suppl):SE55–95. Review.
- 71. Carragee EJ, Don AS, Hurwitz EL, Cuellar JM, Carrino JA, Herzog R. 2009 ISSLS prize winner: does discography cause accelerated progression of degeneration changes in the lumbar disc: a tenyear matched cohort study. Spine (Phila Pa 1976). 2009;34(21):2338–45.
- Busch AJ, et al. Exercise for fibromyalgia: a systematic review. J Rheumatol. 2008;35(6):1130–44.
- Wai EK, et al. Evidence-informed management of chronic low back pain with physical activity, smoking cessation, and weight loss. Spine J. 2008;8(1):195–202.

- 74. 2008 Physical Activity Guidelines for Americans. US Department of Health and Human Services. http:// www.health.gov/paguidelines/pdf/paguide.pdf.
- Wilkinson MJ. Does 48 hours' bed rest influence the outcome of acute low back pain? Br J Gen Pract. 1995;45(398):481.
- Indahl A, Velund L, Reikeraas O. Good prognosis for low back pain when left untampered: a randomized clinical trial. Spine. 1995;20(4):473–7.
- van Tulder MW, Malmivaara A, Esmail R, Koes BW. Exercise therapy for low back pain. Cochrane Database Syst Rev. 2000;2:CD000335. BMC Musculoskelet Disord. 2013 Feb 2;14:55. doi: 10.1186/1471-2474-14-55.
- Calvo-Muñoz I, Gómez-Conesa A, Sánchez-Meca J. Physical therapy treatments for low back pain in children and adolescents: a meta-analysis. BMC Musculoskelet Disord. 2013;14(1):55.
- 79. Chou R, Huffman LH. Medications for acute and chronic low back pain: a review of the evidence for an American Pain Society/American College of Physicians clinical practice guideline. Ann Intern Med. 2007;147(7):505–14.
- Van Tulder MW, Scholten RJ, Koes BW, Deyo RA. Nonsteroidal anti-inflammatory drugs for low back pain. Cochrane Database Syst Rev. 2000:CD000396.
- 81. van Tulder MW, Scholten RJ, Koes BW, Deyo RA. Nonsteroidal anti-inflammatory drugs for low back pain: a systematic review within the framework of the Cochrane Collaboration Back Review Group. Spine. 2000;25:2501–13.
- Schnitzer TJ, Ferraro A, Hunsche E, Kong SX. A comprehensive review of clinical trials on the efficacy and safety of drugs for the treatment of low back pain. J Pain Symptom Manage. 2004;28:72–95. PMID: 15223086.
- Hickey RF. Chronic low back pain: a comparison of diflunisal with paracetamol. N Z Med J. 1982;95:312–4.
- 84. Lee C, Straus WL, Balshaw R, Barlas S, Vogel S, Schnitzer TJ. A comparison of the efficacy and safety of nonsteroidal antiinflammatory agents versus acetaminophen in the treatment of osteoarthritis: a metaanalysis. Arthritis Rheum. 2004;51:746–54.
- 85. Berry H, Bloom B, Hamilton EB, Swinson DR. Naproxen sodium, diflunisal, and placebo in the treatment of chronic back pain. Ann Rheum Dis. 1982;41:129–32.
- Bresalier RS, et al. Cardiovascular events associated with rofecoxib in a colorectal adenoma chemoprevention trial. N Engl J Med. 2005;352(11):1092–102.
- Solomon SD, et al. Cardiovascular risk associated with celecoxib in a clinical trial for colorectal adenoma prevention. N Engl J Med. 2005;352(11):1071–80.
- Bhala, et al. Coxib and traditional NSAID Trialists' (CTT) Collaboration. Vascular and upper GI effects of NSAIDS: meta analysis. Lancet. 2013;382:769–79.
- US Food and Drug Administration. Analysis and recommendations for agency action regarding non-steroidal anti-inflammatory drugs and cardiovascular risk. http://www.fda.gov/ Drugs/ DrugSafety/ Postmarket

DrugSafetyInformationforPatientsandProviders/ Accessed 20 May 2013.

- 90. European Medicines Agency. Press release: European Medicines Agency review concludes positive benefit-risk balance for non-selective NSAIDs. 24/10/2006 http://www.ema.europa.eu/docs/ en\_GB/document\_library/Press\_release/2009/12/ WC500017362.pdf May 2013.
- 91. European Medicines Agency. Press release: European Medicines Agency concludes action on COX-2 inhibitors. 27/06/2005. http://www.ema. europa.eu/docs/en\_GB/document\_library/Press\_ release/2010/01/WC500059088.pdf . May 2013.
- Van Tulder, MW, et al. Muscle relaxants for nonspecific low back pain. Spine (Phila Pa 1976). 2003;28(17):1978–92.
- Tulder V, Maurits W, et al. Muscle relaxants for nonspecific low back pain: a systematic review within the framework of the cochrane collaboration. Spine. 2003;28(17):1978–92.
- 94. Basmajian JV. Cyclobenzaprine hydrochloride effect on skeletal muscle spasm in the lumbar region and neck: two double-blind controlled clinical and laboratory studies. Arch Phys Med Rehabil. 1978;59(2):58–63.
- Salerno SM, Browning R, Jackson JL. The effect of antidepressant treatment on chronic back pain: a meta-analysis. Arch Intern Med. 2002;162(1):19–24.
- Staiger TO, et al. Systematic review of antidepressants in the treatment of chronic low back pain. Spine. 2003;28(22):2540–5.
- Angst MS, David Clark J. Opioid-induced hyperalgesia: a qualitative systematic review. Anesthesiology. 2006;104(3):570–87.
- Lee SH, et al. Tramadol induced paradoxical hyperalgesia. Pain Physician. 2013;16(1):41–4.
- Rhodin A, Stridsberg M, Gordh T. Opioid endocrinopathy: a clinical problem in patients with chronic pain and long-term oral opioid treatment. Clin J Pain. 2010;26(5):374–80.
- Pake JA, Penn RD, Ryan WG. Altered sexual function and decreased testosterone in patients receiving intraspinal opioids. J Pain Symptom Manage. 1994;9(2):126–31.
- 101. Xenidis M, Pandya N, Hames E. Effects of intrathecal opioid administration on pituitary function. Pain Med. 2013;14(11):1741–4.
- Grossman A. 10-brain opiates and neuroendocrine function. Clin Endocrinol Metab. 1983;12(3):725–46.
- Oltmanns KM, Fehm HL, Peters A. Chronic fentanyl application induces adrenocortical insufficiency. J Intern Med. 2005;257(5):478–80.
- 104. Deyo RA, Bass JE. Lifestyle and low-back pain: the influence of smoking and obesity. Spine. 1989;14(5):501–6.
- 105. Shiri R, et al. The association between smoking and low back pain: a meta-analysis. Am J Med. 2010;123(1):87–e7.
- 106. Leboeuf-Yde C. Smoking and low back pain: a systematic literature review of 41 journal arti-

cles reporting 47 epidemiologic studies. Spine. 1999;24(14):1463.

- 107. Goldberg MS, Scott SC, Mayo NE. A review of the association between cigarette smoking and the development of nonspecific back pain and related outcomes. Spine. 2000;25(8):995–1014.
- 108. Heuch I, et al. Body mass index as a risk factor for developing chronic low back pain: a follow-up in the Nord-Trøndelag Health Study. Spine. 2013;38(2):133–9.
- Kauppila LI. Atherosclerosis and disc degeneration/low-back pain–a systematic review. Eur J Vasc Endovasc Surg. 2009;37(6):661–70.
- 110. Kurunlahti M, et al. Association of atherosclerosis with low back pain and the degree of disc degeneration. Spine. 1999;24(20):2080.
- 111. Gay RE, Brault JS. Evidence-informed management of chronic low back pain with traction therapy. Spine J. 2008;8(1):234–42.
- 112. Bronfort G, et al. Evidence-informed management of chronic low back pain with spinal manipulation and mobilization. Spine J. 2008;8(1):213–25.
- 113. Haldeman S, Phillips RB. Spinal manipulative therapy in the management of low back pain. In: Frymoyer JW, Ducker TB, Hadler NM, Kostuik JP, Weinstein JN, Whitecloud TS, editors. The adult spine: principles and practice. New York: Raven Press, Ltd.; 1991. p. 1581–605.
- 114. Assendelft WJJ, et al. Spinal manipulative therapy for low back pain a meta-analysis of effectiveness relative to other therapies. Ann Intern Med. 2003;138(11):871–81.
- 115. Ammendolia C, et al. Evidence-informed management of chronic low back pain with needle acupuncture. Spine J. 2008;8(1):160–72.
- Imamura M, et al. Evidence-informed management of chronic low back pain with massage. Spine J. 2008;8(1):121–33.
- 117. Becker JA, Stumbo JR. Back pain in adults. Prim Care: Clin Off Pract. 2013;40(2):271–88.
- 118. Kuptniratsaikul V, et al. Efficacy and safety of Curcuma domestica extracts in patients with knee osteoarthritis. J Altern Complement Med. 2009;15(8):891–7.
- 119. Manchikanti L, et al. One-year results of a randomized, double-blind, active controlled trial of fluoroscopic caudal epidural injections with or without steroids in managing chronic discogenic low back pain without disc herniation or radiculitis. Pain Physician. 2011;14(1):25.
- 120. Manchikanti L, et al. The role of fluoroscopic interlaminar epidural injections in managing chronic pain of lumbar disc herniation or radiculitis: a randomized, double-blind trial. Pain Pract. 2013;13(7):547–58.
- 121. Standiford Helm II, St Jude Medtronic. Effectiveness of thermal annular procedures in treating discogenic low back pain. Pain Physician. 2012;15: E279–304.
- 122. Kapural L, et al. A randomized, placebo-controlled trial of transdiscal radiofrequency, biacuplasty for treatment of discogenic lower back pain. Pain Med. 2013;14(3):362–73.

- 123. DePalma, M. Biologic treatments for discogenic low back pain. Spine Line Mag Prof. 2012:19–26. www. spineline-digital.org
- 124. Yin W, et al. Intradiscal injection of fibrin sealant for the treatment of symptomatic lumbar internal disc disruption: results of a prospective multi-

center pilot study with 24-month follow-Up. Pain Med. 2014;15(1):16–31. Mesoblast.com. January 30, 2014 news release. http://www.mesoblast.com/ news-and-media/news-announcements. Accessed 8 Mar 2014.

# Facet Joint Pain: Presentation and Treatment, Is It a Myth?

16

Stephan Klessinger

# 16.1 Introduction

This chapter on facet joints is included in a book that is actually dedicated to the intervertebral disk, because there are some correlations between the disk and the facet joints. Together with the intervertebral disk, the facet joints form a functional spine unit. Indeed, the contemplation of the joints is an important aspect when looking at the disk, so this chapter deserves its inclusion in part III of this book.

The nomenclature of the small joints of the vertebral spine is inconsistent. *Facet joint* is commonly used in North American literature to describe paired synovial joints between the posterior elements of adjacent vertebrae. The joints are also known as *zygapophysial joints*, *zygapophyseal joints*, *apophysial joints*, or *posterior intervertebral joints*. Because a facet is simply a small articular surface and, as such, pertains to any small joint, in this chapter the term *zygapophysial joint* is used. The term *zygapophysial stems* from the Greek roots *zygos*, meaning "yoke"; *apo*, meaning "away"; and *physis*,

Department of Neurosurgery, Nova Clinic, Eichendorffweg 5, Biberach 88400, Germany meaning "process." The facet joints bridge the vertebrae behind the vertebral foramina.

The existence of pain deriving from the zygapophysial joints is discussed controversially. The importance of the zygapophysial joints is often underestimated. The existing literature does not support the existence of a *facet syndrome*. There are no typical examination findings or diagnostic proofs to justify the term *syndrome*. Is zygapophysial joint pain therefore a myth? In the section about anatomy, it is shown that zygapophysial joints are typical synovial joints. We know pain not only from bigger joints like those in the knee or hip but also from the small joints in the fingers. So why should the zygapophysial joints not be a source of pain?

Zygapophysial joint pain is defined as pain originating from any structure integral to both the function and configuration of the lumbar facet joints, including the fibrous capsule, synovial membrane, hyaline cartilage surfaces, and bony articulations [1]. Regardless, some postulates must be satisfied for any structure to be deemed a cause of back pain [2]: (1) The structure should have nerve supply; (2) the structure should be capable of causing pain, ideally demonstrated in normal volunteers; (3) the structure should be susceptible to diseases or injuries that are known to be painful; and (4) the structure should have been shown to be a source of pain in patients.

This chapter investigates if these postulates are met. Zygapophysial joint pain remains a

S. Klessinger, MD

Department of Neurosurgery, University of Ulm, Albert-Einstein-Allee 23, Ulm 89081, Germany e-mail: klessinger@nova-clinic.de; stephan.klessinger@uni-ulm.de

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_16

misunderstood, misdiagnosed, and improperly treated medical condition. Therefore, an overview is provided about the clinical presentation and treatment of zygapophysial joint pain where the significance of zygapophysial joints as a pain source is highlighted.

# 16.1.1 History

The proposition that the lumbar zygapophysial joints might be a source of back pain had initially been articulated more than 100 years ago by Goldthwait in 1911 [3]. Sixteen years later, after anatomical dissections, Putti [4] suggested that local inflammation and degenerative changes in lumbar facet joints could result in sciatica from the irritation of nerve roots. In 1933, the term facet joint syndrome was introduced [5]. Badgley [6] emphasized in 1941 the significance of the zygapophysial joints as a pain source. According to him, 80 % of all back pain and referred pain were caused by the joints. With the implementation of successful operations of herniated disks by Mixter and Barr in 1934 [7], the focus was directed away from the zygapophysial joints and to the intervertebral disks.

The first description of a method to treat particular pain arising from zygapophysial joints was published in 1971 [8]. Rees attempted to sever the articular nerves using a special scalpel to make longitudinal incisions through the back muscles. The procedure was called *rhizolysis* [9, 10]. However, the articular nerves did not run where Rees had depicted [11]. Inspired by the success for the treatment of trigeminal neuralgia with radiofrequency of White and Sweet in 1969 [12, 13], Shealy (1974–1976) modified the intervention using radiofrequency electrodes to coagulate the articular nerve [14–17]. This procedure was known as facet denervation. However, again, the nerves were not located where Shealy described placing his electrodes [18, 19].

More detailed anatomical knowledge was necessary to correct the surgical anatomy. The targets for denervating zygapophysial joints were the medial branches of the dorsal rami. This procedure was named *medial branch neurotomy* to distinguish it from *facet denervation* [8, 18, 19]. For several years, the procedure became controversial, and no further studies appeared. The impetus for new studies and for comparative or placebo-controlled trials for lumbar and cervical radiofrequency came when cervical radiofrequency neurotomy was shown to be efficacious in a randomized placebo-controlled trial in 1996 [20].

A different approach, cryolesioning, was subsequently introduced. The scientific groundwork for this therapeutic option was laid between 30 and 40 years ago, and although the primary target of this technology was the treatment of tumors, it has been used successfully in the treatment of a wide range of peripheral pain syndromes [21– 25]. Lloyd et al. [26] coined the term *cryoanalge*sia for its use in pain management. He proposed that this technique was superior to other methods of peripheral nerve destruction (e.g., alcohol, phenol, or surgical lesions), because it is not followed by neuritis or neuralgia [27]. Barnard and colleagues [28–30] popularized cryoneuroablation in the early 1980s, but relatively little has been written on the technique since.

# 16.2 Anatomy

#### 16.2.1 Joint

The smallest functional motion unit is called *functional spine unit* or *motion segment*. It consists of the vertebrae, all adjoining ligaments between them, and three joints. The most prominent component of this unit is the interbody joint which consists of the intervertebral disk and the vertebral endplates. The other two joints are the paired zygapophysial joints, which are formed by the articulation of the inferior articular processes of one lumbar vertebra with the superior articular processes of the next vertebra.

Although the zygapophysial joints are small, they exhibit the features typical of synovial joints [31]. This means the facets are enclosed by a capsule. The surface of the facets is covered by cartilage, a typical synovium, and even a meniscoid exists. The joint space is capable of accommodating between 1 and 1.5 mL of fluid [32].

#### 16.2.1.1 Articular Facets

The articular facets of the lumbar vertebrae are ovoid in shape, measuring some 16 mm in height and 14 mm in width and having a surface area of about 160 mm<sup>2</sup> [33].

Viewed from behind (Fig. 16.1), the articular facets of the lumbar zygapophysial joints appear as straight surfaces in the longitudinal direction. However, viewed from above, the articular facets vary both in the shape of their articular surfaces and in the general direction they face. In the transverse plane, the articular facets may be flat or planar, or they may be curved to varying extents (Fig. 16.2). In the upper lumbar spine, approximately 80 % of the facet joints are curved, and 20 % are flat. In the lower lumbar spine, these numbers are reversed [34].

The orientation of a lumbar zygapophysial joint is, by convention, defined by the angle made by the average plane of the joint with respect to the sagittal plane (Fig. 16.2) [31]. Smaller angles (less than  $45^{\circ}$ ) are found more often in the upper lumbar spine. The levels L3–4 to L5–S1 usually show angles about  $45-50^{\circ}$  [34]. The extent to which a joint can resist forward displacement or rotatory displacement depends on the shape and orientation of the joint. The smaller the angle and



**Fig. 16.1** Posterior view of the L3–L4 zygapophysial joints. (*Arrow*) straight surface of the right joint, (*AC*) articular cartilage, (*I*) inferior articular process L3, (*S*) superior articular process L4

the closer the joint is orientated toward the sagittal plane, the less the vertebra can resist forward displacement [31].

In the case of joints with curved articular surfaces, particular portions of the surface are involved in resisting different movements. During rotation the entire articular surface is in contact. Therefore, rotation is well resisted [31].

# 16.2.1.2 Capsule

Each lumbar zygapophysial joint is enclosed by a fibrous capsule that is about 1 mm thick. At the superior and inferior ends of the joint, the capsule is long and relatively lax and attaches somewhat away from the articular margin [31, 34]. Its laxity accommodates the superior and inferior displacements of the articular process during flexion of the lumbar spine (Fig. 16.3). In a joint in its neutral position, this lax capsule creates subcapsular recesses that extend over the surface of the articular process, at the superior and inferior poles of the joint. In some patients, the anterior synovial recess may extend into the ligamentum flavum. The posterior synovial recess of the lumbar zygapophysial joint often extends beyond the articulating surfaces of the lumbar facets into the posterior fibrous capsule [35]. In both the superior and inferior parts of the capsule, there is a tiny hole that permits the passage of fat during movements of the joint [31]. Anteriorly, the fibrous capsule is replaced by the ligamentum flavum [36].

# 16.2.1.3 Intra-articular Structures

Articular cartilage assumes the same concave or convex curvature as the underlying facet. It is thickest over the center of each facet, rising to a height of about 2 mm [31, 37]. The articular cartilage rests on a thickened layer of bone known as the subchondral bone. Age changes and degenerative changes affect the articular cartilage and also the subchondral bone.

There are no particular features of the synovium of the lumbar zygapophysial joints that distinguish it from the synovium of any typical synovial joint. It attaches along the entire peripheral margin of the articular cartilage on one facet and extends across the joint to attach to the margin of the opposite articular cartilage [31].



**Fig. 16.2** Varieties of orientation and curvature of the lumbar zygapophysial joints. (a) Flat joint orientated  $60^{\circ}$  (L3–L4), (b) curved joint (C shaped) orientated  $30^{\circ}$  (L2–

Two more intra-articular structures are present. These are fat and a structure referred to as *meniscoid*. The fat fills all leftover space underneath the capsule. It communicates with the fat outside the joint through the foramen in the capsule [31]. From their histology, it is clear that meniscoids are not comparable to a meniscus in the knee. They resemble more the intra-articular structures found in the small joints of the hand [38, 39]. There have been many different interpretations of the meniscoid structures. The most

L3), (c) flat joint orientated 25° (L3–L4), (d) flat joint orientated 75° (L5–S1)

comprehensive study identifies three types [31, 40, 41]. The smallest structure is the connective tissue rim, a thickening of the internal surface of the capsule. The second type of structure is an adipose tissue pad (Fig. 17.3), consisting of a fold of synovium, fat, and blood vessels. The largest structure is fibroadipose meniscoids which also consist of synovium, fat collagen, and blood vessels. The adipose tissue pads and the fibroadipose meniscoids have been interpreted as serving a protective function [31, 40].



**Fig. 16.3** L4–L5 zygapophysial joint in a spine in an intact cadaver and frozen in situ. The extension causes the tip of the inferior articular process to come in contact with the pars interarticularis of L5 (*arrow*). The joint capsule is elongated and severely compressed against the pars interarticularis. A richly vascularized meniscoid (\*) is projecting into the opening of the superior joint space (Courtesy of W. Rauschning)

# 16.2.2 Innervation

The zygapophysial joints of the lumbar spine have a dual nerve supply from the medial branches of the dorsal rami of the spinal nerves at the same level and from the level above. The numbering of the spinal nerve and the bone that it crosses is different. The spinal nerve of a particular segment issues from below the vertebra with the same segmental number as the spinal nerve, but the spinal nerve then crosses the superior articular process of the next vertebra.

The medial branch of the dorsal ramus in the lumbar spine courses over the base of the



**Fig. 16.4** Lumbar medial branch anatomy. Left anterior oblique illustration. (L3-SI) spinous processes, (mal) mamillo-accessory ligament, (nr) nerve root, (I) inferior articular process, (S) superior articular process, (sb) superior branch from medial branch, (ib) inferior branch of medial branch, (dr) dorsal ramus, (mb) medial branch (Reproduced from Klessinger [42])

transverse process at the junction of the superior articulating process (Fig. 16.4) [18]. The lumbar dorsal rami carry the same segmental number as the vertebra from which they originate. In their subsequent course, these nerves cross structures and innervate joints below their segment of origin [43]. The course of the medial branches L1–L4 is similar. Each nerve runs in the groove formed by the junction of the transverse process and the superior articular process. Subsequently, each medial branch runs under the mamillo-accessory ligament [44]. This ligament is responsible for the reliable location. It can be large and sometimes ossified, particularly at lower levels [44]. Beyond the ligament, the medial branch sends branches to innervate the zygapophysial joint, multifidus muscle, interspinal muscles, and the interspinous ligaments [45]. There are three branches of the medial branch. The proximal branch hooks around the articular process to supply the facet above. The medial descending

branch courses in inferomedial manner to innervate the superior and medial portions of the capsule below plus muscle and skin. The ascending branch supplies the joint above [46]. As a consequence, each zygapophysial joint has a dual nerve supply. For example, the L2 and L3 medial branches innervate the L3–L4 joint. In addition to the joints, the medial branch also innervates the multifidus muscle, the interspinous muscle, and the periosteum [45, 47–49].

The L5 dorsal ramus crosses the ala of the sacrum. There is no medial branch for L5. The target is the dorsal ramus itself. The medial branch does not arise until the dorsal ramus reaches the caudal region of the L5–S1 joint [49].

The joint capsules and the surrounding structures are richly innervated by encapsulated, unencapsulated, and free nerve endings. Nociceptors fire when the capsule is stretched or subjected to compressive forces [47, 50]. The presence of lowthreshold, rapidly adapting mechanosensitive neurons suggests that in addition to transmitting nociceptive information, the facet capsule also serves a proprioceptive function [51]. Nerve fibers have also been found in subchondral bone and intra-articular inclusions of zygapophysial joints, signifying that facet-mediated pain may originate in structures other than the joint capsule [52, 53].

# 16.3 Physiology

The zygapophysial joints are involved in all principal movements of the spine. Possible movements are axial compression/distraction, flexion/ extension, axial rotation, and lateral flexion. Horizontal translation does not occur as isolated movement [36].

In the reflection of the joints as a possible pain source, the applied loads and the restriction of movements are particularly important.

# 16.3.1 Axial Compression/Distraction

The interbody joints (i.e., the intervertebral disks) are designed as the principal weight-bearing components of the spine. The importance of the zygapophysial joints is discussed controversially. Studies reported that all the compressive force is resisted by the disk [54]. Others found that the zygapophysial joints can bear 28 % or more of vertically applied load [55].

Three conditions lead to a remarkable load to the zygapophysial joints: First, with axial load in combination with a backward movement, the articular facets are driven into each other, and load can be transmitted through the joints [36]. Second, with severe or sustained axial compression, the inferior articular processes can be lowered until their tips impact the laminae of the vertebrae below [56]. Axial loads can be transmitted through the inferior articular process to the laminae. Third, in prolonged standing with a lordotic spine, the joints at each segmental level bear an average of some 16 % of the axial load [54, 57].

In contrast, in a neutral position, the articular surfaces run parallel to the direction of axially applied load. Thus, in a neutral position, they cannot sustain the load. Also, in the conditions of erect sitting, the zygapophysial joints are not impacted [36].

Axial distraction has been studied far less. The capsules of the zygapophysial joints are remarkably strong when subjected to longitudinal tension [36]. A single capsule can sustain 600 N before failing [58]. Stretching of the capsule is painful because of the free nerve endings.

# 16.3.2 Flexion/Extension

Flexion involves a combination of anterior sagittal rotation and anterior translation. Anterior translation is resisted by the impaction of the superior and inferior facets. In curved joints, the load is concentrated on the anteromedial portions of the facets [59], where commonly age changes are seen. The sagittal rotation component involves tension in the joint capsule. The capsules of the joints contribute about 39 % of the resistance during flexion.

Extension involves posterior sagittal rotation and posterior translation together with the downward movement of the inferior articular process and the spinous process. This movement is limited by bony impaction between the spinous processes [36, 60]. The interspinous ligament buckles and becomes trapped.

# 16.3.3 Axial Rotation

The zygapophysial joints protect the intervertebral disk from excessive torsion. The inferior articular facets of the upper vertebra will be impacted against its opposing superior articular facet. Because the joint space is quite narrow, the range of movement before impaction occurs is quite small. The capsule of the opposite joint is being stretched. Experimental studies have established that the zygapophysial joints contribute between 42 and 54 % of the torsional stiffness of a segment [36, 61]. The zygapophysial joints provide a substantial buffer during the first 3° of rotation. They must be severely compressed before rotation exceeds the critical range of 3° [36].

# 16.3.4 Lateral Flexion

Lateral flexion involves a complex and variable combination of lateral bending and rotatory movements [36].

# 16.3.5 Rotation in Flexion

This is a common movement associated with the onset of back pain. However, studies offer conflicting results and opinions that stem from the complexities of this movement [36].

# 16.4 Pathological Changes

#### 16.4.1 Degeneration

During life, changes occur to the intervertebral disk and to the zygapophysial joints called spondylosis or osteoarthrosis. These changes are not per se a disease but an expression of the morphological consequences of stress applied to the disk and the joints during life. The incidence of osteoarthrosis is just as great in patients with symptoms as in patients without symptoms [62, 63]. Additional factors must be present to make the zygapophysial joints a pain source.

The degenerative changes are more advanced in the concave superior articular process than in the inferior articular process. It is the backwardfacing portion of the facet that resists the forward shear stresses applied to the intervertebral joint during weight-bearing and flexion movements [64]. After the fifth decade, the subchondral bone of the zygapophysial joint gets thinner [65]. The articular cartilage exhibits focal changes. Vertical fibrillation of the cartilage, which reflects the repeated stress, and sclerosis of the subchondral bone plate are common [66]. Severe or repeated pressure may result in erosions and focal thinning of the cartilage (Fig. 16.5). Other regions might exhibit swelling of the cartilage. Where cartilage is lost, fibrofatty intra-articular inclusions may increase in size [66].

Older joints exhibit gross thickening (Fig. 16.6a). The development of osteophytes along the attachment sites of the joint capsule and ligamentum flavum to the superior articular process increases. As a result of repeated stress during rotatory movements, the articular cartilage spreads out to cover and protect the edges of the bony articular process [64].

A progressive decrease of range in movement with age is evident in the entire lumbar spine and in individual intervertebral joints [67, 68].

Zygapophysial joints are frequently affected by osteoarthritis. The arthritis is usually secondary to disk degeneration or spondylosis [69], but in 20 % of cases, it can be totally independent [70]. This condition is believed to be a possible cause of zygapophysial joint pain [71–74].

Inflammatory mediators, such as cytokines, prostaglandins, and neuropeptides, increase within the joint and the dorsal root ganglion in joint inflammation and arthritis [75–77]. Specifically, prostaglandin E2 (PGE2) has been identified as a key mediator of inflammation-induced behavioral sensitivity and increased neuronal excitability [78–80]. Overexpression of



**Fig. 16.5** Sagittal section through the neuroforamina of a severely degenerated lower lumbar spine of a 70-yearold man. The zygapophysial joints are in a subluxated position due to the loss of segmental height. The pars interarticularis of L5 is being eroded superiorly by the inferior articular process of L4 and inferiorly by the superior articular process of S1 (\*). Such pars erosion is a prerequisite for the development of degenerative spondylolisthesis. There is no cartilage in the L5–S1 zygapophysial joint (*arrow heads*) (Courtesy of W. Rauschning)

MMP-1, induced by interleukin-1 $\beta$ , plays an important role in the inflammatory process of lumbar zygapophysial joint degeneration [81].

#### 16.4.2 Degenerative Disk Disease

The intervertebral disk and the paired zygapophysial joints form a functional unit, and therefore, changes in the disk height are relevant for the load of the zygapophysial joints. The pressure between the facets increases significantly with the narrowing of the disk space [59]. Increased pressure may be a source of pain in patients with reduced disk spaces [59]. Maintenance of disk height is the normal feature of aging. Overt disk narrowing invites the consideration of some process other than aging [64].

One possible process is internal disk disruption. It may lead to disk degradation and disk resorption and is independent of degenerative changes [82]. Degradation of the nucleus of the disk is initiated by an endplate fracture that progressively destroys the nucleus pulposus [2]. Less able to bind water, the nucleus is less able to sustain pressure. In time, the annulus buckles under the load, and the disk loses height, which compromises the function of all joints in the affected segment [2]. As a result, reactive changes occur in the form of osteophyte formation in the zygapophysial joints. Disk narrowing may also predispose zygapophysial joint disease. When the disk becomes narrowed, up to 70 % of the compressive force usually applied to the disk is transferred to the zygapophysial joints [54].

Another reason for a loss of disk height is herniation of disk material. The material that is extruded into the spinal canal can no longer contribute to the disk height. In contrast to degenerative changes, the loss of disk material and disk height occurs within a short time. A gradual adaptation of the involved structures to the new situation is hardly possible. In addition, in case of surgery, sometimes not only the extruded disk material but also parts of the annulus and the central portion of the nucleus are removed. Therefore, patients with a herniated disk and compression of the nerve root suffer not only from radicular pain but also often from zygapophysial pain, which can exacerbate after surgery.

Degeneration and loss of structural integrity of the intervertebral disks have been shown to result in concomitant degenerative changes in the zygapophysial joints [83–85]. The reverse is also true. Degeneration and motion abnormalities at the zygapophysial joints can induce and accelerate degeneration of the intervertebral disks [57, 86, 87]. In a magnetic resonance imaging (MRI) study evaluating the relation



**Fig. 16.6** Examples of magnetic resonance imaging findings concerning the zygapophysial joints. (a) Degenerative changes, (b) synovial cyst of the zygapoph-

ysial joint and increased joint volume, (c) asymmetric joint gap, (d) increased joint volume (Reproduced from Klessinger [42])

between facet joint osteoarthritis and degenerative disk disease, facet joint osteoarthritis was rarely found in the absence of disk degeneration but tended to be most pronounced at spinal levels associated with advanced degenerative disk disease [88].

# 16.4.3 Synovial Cysts

The term synovial cyst refers to cysts that arise from the zygapophysial joint capsule of the lumbar spine (Fig. 16.6b) [89]. They can be lined with synovium and contain serous, gelatinous, or hemorrhagic fluid [90]. The development is linked to degenerative spondylosis, segmental instability, and perhaps trauma [90, 91]. They are a cause of back pain and radiculopathy, with zygapophysial joint degeneration being the most common cause for cyst formation [81].

Intrafacetal synovial cysts can be a source of pain because of distension and pressure on adjacent pain-generating structures, calcification, and asymmetrical facet hypertrophy [91–95].

# 16.4.4 Asymmetric Load

A temporary one-sided load is often found in the context of knee or hip problems with appropriate gait disturbance or when walking with crutches. These patients develop often zygapophysial joint pain without structural changes. The reason is unusual strain or overuse of the joint. The treatment prognosis is good.

Facet tropism (asymmetry of the facet angles) may have a relationship to degenerative changes in the spine, either as the cause of degenerative changes or as the result of abnormal forces produced by degeneration [96]. These degenerative changes can be a potential cause of back pain [96]. The clinical significance of facet tropism is not yet well established [96–101]. A difference of facet angles of more than 7° (Fig. 16.6c) is found in 77 % of men and 66 % of women [96]. Facet tropism is a predisposing factor for degenerative changes [102, 103] but does not seem to be associated with zygapophysial joint osteoarthritis [96].

However, there is a positive association between the sagittal orientation of the facets and osteoarthritis [96]. Severe osteoarthritis is associated with back pain, independent of sociodemographics and the narrowing of disk height [104].

Scoliosis is a further condition with asymmetric load. Asymmetric degeneration leads to increased asymmetric load and therefore to a progression of the degeneration and deformity, as either scoliosis or kyphosis. The destruction of zygapophysial joints, joint capsules, disks, and ligaments may create mono- or multisegmental instability and, eventually, spinal canal stenosis [105]. In primary degenerative scoliosis, the degeneration ends up with zygapophysial joint arthritis with hypertrophic capsules, calcification, and osteophytes [105]. The most frequent clinical problem of adult scoliosis is back pain. At the site of the curve, it can be localized either at the apex or in its concavity, and zygapophysial joint pain can be localized in the countercurve from below the curve to above the curve [105].

# 16.4.5 Spondylolisthesis

Arthritis of the zygapophysial joints with loss of their normal structural support is the major local reason that probably leads to the development of degenerative vertebral slippage [106, 107]. It seems to be evident that morphological abnormalities of zygapophysial joints in the lumbar spine are a significant cause of low back pain and segmental instability and a predisposing factor in the development of degenerative spondylolisthesis [108-110]. One of the most probable sources of pain related to degenerative spondylolisthesis is degenerated and subluxated zygapophysial joints and segmental instability that causes tension in the zygapophysial joint capsule and ligaments [106, 109]. Patients with degenerative spondylolisthesis have more sagittally orientated zygapophysial joints and more significant zygapophysial joint tropism than normal control subjects [109]. The cephalad portion of the zygapophysial joints is more sagittally oriented, and the caudad portion of the zygapophysial joints is more coronally oriented in patients with degenerative spondylolisthesis [111]. Often, an increased joint volume indicates spinal instability [112], or synovial cysts associated with degenerative spondylolisthesis and zygapophysial joint osteoarthritis can be found [113]. Exaggerated fluid in the facets seen on axial MRI (Fig. 16.6d) is significantly suggestive of spondylolisthesis [114].

It is well known, though, that patients with degenerative spondylolisthesis might have sources of pain other than the zygapophysial joints [115]. In particular, the often additionally present spinal canal stenosis causes symptoms. The second pathology often interlinked with degenerative spondylolisthesis is disk degeneration [106, 107].

Spondylolisthesis is a characteristic example of concurrent pain sources in the same patient at the same time. The proportion by which the zygapophysial joints are involved in the complex symptoms is often difficult to diagnose [116].

# 16.4.6 Injuries

Extension of the spine is limited by the impaction of the inferior articular process on the lamina below. Under this condition, the continued application of an extension force results in a rotation around the impacted articular process and draws the contralateral zygapophysial joint backward. A rupture of the joint capsule is possible [2]. Rotation is also limited by the impaction of the zygapophysial joint. Further rotation also can result in a rupture of the contralateral capsule.

Zygapophysial joint pain is likely to occur with repetitive, chronic strains as might be seen in the elderly or, less frequently, after an acute event such as tearing the joint capsule by stretching it beyond its physiologic limits. This hypothesis is supported by clinical studies indicating a higher prevalence of facet arthropathy in elderly patients [117–119] and numerous cases of lumbar facet arthropathy after high-energy trauma [120]. There are more than two dozen reported cases of lumbar facet dislocation after rapid deceleration injuries [120–123]. The mechanism of injury in these cases is purported to be a combination of hyperflexion, distraction, and rotation [120, 121, 124].

Both in biomechanical studies and in postmortem studies, capsular tears, capsular avulsion, subchondral fractures, intra-articular hemorrhage, and fractures of the articular process have been found [2, 125–129]. Fractures of the zygapophysial joints cannot be detected on plain radiographs and might be too small to be seen in computer tomography (CT) scans [128, 129]. Lesions such as capsular tears cannot be detected by radiography, CT, or MRI. It may be that these lesions underlie zygapophysial joint pain [2].

# 16.4.7 Other Conditions

These include inflammatory arthritides such as rheumatoid arthritis, ankylosing spondylitis and reactive arthritis [130–132], synovial impingement, meniscoid entrapment, chondromalacia facetae, pseudogout, synovial inflammation, villonodular synovitis, and acute and chronic infection [71, 133–136].

# 16.5 Symptoms

Pain originating from the zygapophysial joints is a *lumbar spinal pain* [137]. This means that the pain is arising in an area between the lateral borders of the erector spinae at any lumbar level. The pain results from noxious stimulation and is therefore a somatic pain. Somatic pain must be distinguished from visceral pain and from neurogenic pain. Neurogenic pain results from damage or irritation of the axons or cell bodies of a peripheral nerve. Radicular pain is a typical example of neurogenic pain. Zygapophysial joint pain is often associated with pain in the buttock or in the leg. However, in this case, it is a somatic referred pain and not a radicular pain. Referred pain is perceived in a region innervated by nerves other than those that innervate the actual source of pain [2, 36]. Referred pain occurs because of a misperception of the region of the signal that reaches the brain by a convergent sensory pathway [2]. Somatic referred pain is perceived deeply. It is diffuse and hard to localize and it is aching in quality [138].

The joint capsule seems to be more likely to generate pain than the synovium or articular cartilage. There is considerable overlap between all lumbar facet joints, with the referral pattern being more widespread and variable in patients with chronic pain than in asymptomatic volunteers [1]. 230

The zygapophysial joints meet all requirements that are necessary to be a possible pain source: They are well innervated by the medial branches, and free nerve endings are found in the joint capsules (see Sect. 16.2.2). In both patients and volunteers, mechanical stimulation or chemical stimulation of the joints with injection of hypertonic saline or with a contrast medium produces back pain and referred pain identical to that commonly seen in patients [47, 70, 139]. Pain can be relieved by anesthetizing one or more of the lumbar zygapophysial joints. Therefore, like other synovial joints in the human body, the zygapophysial joints represent a potential pain generator in patients with chronic low back pain.

Often, the patient can localize the center of zygapophysial joint pain at a specific level, unilateral or bilateral. Sometimes tenderness is evident over the affected joint [140]. Additionally, referred pain with diffuse borders is present. It occurs predominantly in the buttock and in the thigh in a nondermatomal distribution. Radiation below the knee can occur, even as far as the foot [70, 141]. All of the lumbar facet joints are capable of producing pain that can be referred into the groin, although this is more common with lower facet joint pathology [1]. Pain emanating from upper facet joints tends to extend into the flank, hip, and upper lateral thigh, whereas pain from the lower facet joints is likely to penetrate deeper into the thigh, usually laterally and/or posteriorly [1].

Pain at the beginning of a movement is typical for pain of joint origin. Therefore, the zygapophysial joints often hurt when moving from a sitting to a standing position or while sleeping when turning from one side to the other. Morning stiffness with difficulty to put on socks in a standing position and pain early in the morning that is relieved during the next hours and with walking will be reported often. The considerations of the normal movements (Sect. 16.3) allow one to form opinions regarding the movements that cause pain. Twisting or rotational movements, extension, and rotation in flexion are more likely to increase pain. Sitting with a round back and a relaxed musculature results in a substantial load for the joints. Also the monotonous seating position in a car might be a strain on the joints.

An acute onset of a sharp, penetrating low back pain with immobilization of the lumbar spine (*acute locked back*) is called *Hexenschuss* (witch's shot) in German. This term illustrates the medieval idea that diseases are inflicted to people through an arrow shot (Fig. 16.7) by supernatural beings (e.g., witches, elves). Even today, the pain sometimes comes so unexpectedly out of a normal movement that an explanation seems to be difficult.

And indeed, the cause remains speculative. However, theories have been advanced involving the concept of meniscal entrapment [2]. Upon flexion, these meniscoid structures (Sect. 16.2.1) are trapped in the subcapsular pockets of the joint [142]. This condition might be amenable to manipulative therapy.

The description of the pathological changes of the zygapophysial joints (Sect. 16.4) makes it clear that zygapophysial joint pain is often only one component of a more complex syndrome. Spinal canal stenosis is often symptomatic with



**Fig. 16.7** Illustration of a *Hexenschuss* on a print by Johann Zainer around the year 1489/1490. Woodcut from the *Tractatus von den bösen Weibern, die man Hexen nennt* by Ulrich Molitor

neurogenic claudication and radiculopathy and, at the same time, pain deriving from the zygapophysial joints. However, back pain from the zygapophysial joints can occur together with radicular pain or even with a radiculopathy if the spinal nerve is irritated or compressed by an additional pathology, like a herniated disk, neuroforaminal stenosis, or a synovial cyst.

# 16.6 Diagnosis

# 16.6.1 Clinical Findings

No historic or physical examination variables exist to identify a zygapophysial joint as the pain source [143, 144]. Target joints might be identified by the pain pattern, local tenderness over the area, and provocation of pain with deep pressure. The neurological examination is usually normal. When performing the straight leg rise test (Lasègue's sign), the patient often experiences back pain. However, there should be no sciatic pain.

Revel et al. [118] identified seven variables associated with a positive response to facet joint anesthesia: age greater than 65 years and pain not exacerbated by coughing, not worsened by hyperextension, not worsened by forward flexion, not worsened when rising from forward flexion, not worsened by extension-rotation, and well relieved by recumbency. However, subsequent investigations have also failed to corroborate the findings of Revel et al. [118].

Of course, the clinical examination serves to delineate zygapophysial joint pain from other pain sources. As shown above (Sect. 16.5), zygapophysial joint pain appears often in combination with other pathologies (e.g., spinal canal stenosis, spondylolisthesis, or a herniated disk).

# 16.6.2 Radiologic Findings

The prevalence of abnormal zygapophysial joint changes on radiologic imaging depends on the age and presence of symptoms in the study population, the imaging modality used, and the threshold use for rendering a diagnosis of abnormal. In studies conducted in patients with low back pain, the incidence of degenerative facet disease on computed tomographic scanning ranges from around 40 % in some studies [73, 145] to upward of 85 % in others [146]. MRI is considered to be somewhat less sensitive than CT imaging in detecting degenerative facet changes [146–148], although several studies conducted in chronic low back pain patients found both the sensitivity and specificity of MRI to be more than 90 % compared with those of CT [88, 147].

On plain radiographs, osteoarthrosis (Fig. 16.8) appears as commonly in asymptomatic individuals as in patients with back pain [63, 149]. In addition, CT scans do not have a diagnostic value for lumbar zygapophysial joint pain [150].

Particularly striking results in MRI (Fig. 16.6) can be helpful to identify the level of the pain source. An increased joint volume indicates spinal instability [112].



Fig. 16.8 Plain radiography with severe osteoarthrosis L4–L5

In summary, the evidence in the literature does not support the routine use of radiologic imaging to diagnose zygapophysial joint pain.

# 16.6.3 Medial Branch Blocks

A detailed description of the technique of medial branch blocks (Fig. 16.9), the evaluation of results, and the validity is given in Chap. 33.

Medial branch blocks are a diagnostic tool. They are used to test if the pain stems from a zygapophysial joint because the medial branch innervates the joint. For this reason medial branch blocks are also referred to as zygapophysial joint blocks or facet joint blocks. The fundamental indication for medial branch blocks is the desire to know if the zygapophysial joints are the pain source. Diagnostic lumbar zygapophysial joint nerve blocks are recommended in patients with suspected zygapophysial joint pain. Of



**Fig. 16.9** AP view of needles in position for an L4 medial branch block and L5 dorsal ramus block after application of a contrast medium (Reproduced from Klessinger [151])

course, the response must affect the management. The only validated treatment for pain mediated by the medial branches is radiofrequency neurotomy [152].

Because the singular reason for performing diagnostic medial branch blocks is to obtain information, the evaluation of the patient's response is essential. A positive response to a block is complete relief of that part of the pain that the blocks are expected to provide relief for the duration commensurate with the expected duration of the local anesthetic's effect. If more than one pain source is known, only a proportion of the pain will be relieved [151].

Lumbar medial branch blocks are the most thoroughly validated of all spinal interventional procedures [153, 154]. Single diagnostic blocks are not valid because they carry an unacceptable high false-positive rate of 25–45 % [8, 155–159]. In order to reduce the likelihood of responses being false positive, controlled blocks are mandatory [152]. Uncontrolled blocks or intraarticular blocks lack validity [8]. In addition, false-negative results after medial branch blocks are reported [160]. For a detailed discussion on false-positive and false-negative blocks, see Chap. 33.

The degree of relief that should occur after medial branch blocks remains contentious [161]. Ideally, diagnostic blocks should produce complete relief of pain or near-complete relief. This would occur only when the patient's sole or principal source of pain lies in the joints innervated by the nerves blocked. Some investigators, however, use a more liberal criterion, such as >50 % relief of pain. This criterion allows medial branch neurotomy to be used to provide substantial, but not necessarily complete, relief of pain, which is nevertheless clinically worthwhile [8]. Adopting lesser diagnostic criteria admits more patients for treatment, but the outcomes are poorer. The implication is that physicians will be treating more patients but not achieving optimal outcomes. Adopting more stringent diagnostic criteria admits fewer patients for treatment, but the outcomes achieved are of greater quality (see discussion in Chap. 33).

# 16.6.4 Arthrography

Arthrography is the demonstration of the internal contours of the joint by injecting a contrast medium [162]. Although various interesting features of a zygapophysial joint can be demonstrated, none of these features has been shown to be diagnostic of any disorder, and none has been shown to determine if the joint is a source of pain. Consequently, lumbar zygapophysial arthrography has no established diagnostic value [162].

# 16.6.5 Intra-articular Blocks

In this procedure, a local anesthetic is injected into the joint. The objective is to test if anesthetizing a particular joint relieves the patient's pain. The validity of intra-articular blocks of the lumbar zygapophysial joints has never been tested and has never been established [8, 163, 164]. For intra-articular blocks, there is no consequent treatment [163].

Moreover, several advantages of medial branch blocks exist [152]: Medial branch blocks are easier to perform. Entering a narrow joint space can be difficult. Sometimes osteophytes or degenerative changes may block the entry. Medial branches are safer because bone prevents overpenetration of the needle and entering the spinal canal. Target nerves can be anesthetized with different agents whose duration of effect is known. If the response to medial blocks is positive, radiofrequency neurotomy is a therapeutic utility with predictive validity [151].

# 16.7 Therapy

# 16.7.1 Conservative Treatment

No specific conservative treatment for zygapophysial joint pain exists. Patients with zygapophysial joint pain are treated in the same way as patients with low back pain emerging from a different pain source. There are no clinical studies specifically assessing pharmacotherapy or noninterventional treatment for lumbar arthropathy [1]. The treatment of low back pain (and also of zygapophysial joint pain) consists of a multimodal approach comprising conservative therapy, medical management, procedural interventions, and, if indicated, psychotherapy. Nonsteroidal antiinflammatory drugs are widely considered first-line drugs for the treatment of low back pain, with little evidence to support one particular drug over another [165–167].

## 16.7.2 Radiofrequency Denervation

Guidelines only exist for radiofrequency denervation of the zygapophysial joints, published by the International Spine Intervention Society [43]. Radiofrequency denervation is the direct consequence after the diagnosis of zygapophysial joint pain was validated by controlled medial branch blocks, and it is the only validated treatment for pain mediated by the medial branches [152].

Percutaneous denervation procedures offer pain relief by denervation of the nerves that innervate painful joints. It is a percutaneous therapeutic procedure in which a radiofrequency electrode is used to coagulate one or more of the medial branches of the lumbar dorsal rami, or the L5 dorsal ramus, in order to relieve back pain mediated by these nerves.

For medial branch neurotomy to be anatomically accurate and effective, the electrodes should be placed parallel to the target nerve. Also, lesions should be placed along the maximal available length of the nerve to optimize duration effect [8]. Therefore, exact anatomic knowledge is essential.

Medial branch neurotomy is performed in patients experiencing pain for at least 3 months and in those who did not respond to conservative treatment. Controlled medial branch blocks are mandatory as a diagnostic test to prove that the target nerve is responsible for the pain. Radiofrequency neurotomy provides good evidence-based results whenever patients have been selected correctly and when anatomically accurate surgical techniques have been used.

Thermal radiofrequency neurotomy is a procedure distinct from pulsed radiofrequency or dorsal root ganglion radiofrequency. Thermal radiofrequency deliberately produces a lesion in the target nerve by denaturing its constituent proteins at the site at which the electrode is applied. The other procedures do not do so [43].

# 16.7.2.1 Patient Selection

The optimal patient for medial branch neurotomy is one who has been experiencing pain for at least 3 months and whose pain did not respond to conservative treatment. The patient should have a realistic expectation. Previous surgery does not preclude neurotomy [168, 169]. Repeat radiofrequency neurotomy after recurrence of pain is possible [170]. It is quite safe to coagulate one or two nerves, but it is not known how many more nerves can be coagulated with safety.

For various reasons, medial branch blocks are the only acceptable and validated diagnostic test as an indication for radiofrequency neurotomy [8]. Medial branch blocks have been validated for validity [171], target specificity [172], and construct validity [173]. Patients with positive responses to controlled blocks can expect to have substantial and lasting responses to medial branch neurotomy [173]. Uncontrolled blocks or intra-articular blocks lack validity [8].

Even after cervical or lumbar spine surgery or in patients with spondylolisthesis, pain emerging from zygapophysial joints can be treated with radiofrequency neurotomy [168–170]. In these patients, the zygapophysial joints are often not the only pain source.

#### 16.7.2.2 Contraindications

Absolute contraindications for radiofrequency exist in patients unwilling or unable to consent to the procedure, patients with systemic infection or bleeding diathesis, or those on anticoagulants with a high risk of bleeding and pregnancy. Relative contraindications exist in patients using pacemaker equipment, after immunosuppression, in patients with unrealistic expectations, and in uncooperative patients [43].

# 16.7.2.3 Technique

For radiofrequency neurotomy, a high-frequency electrical current is alternating between a large

surface area on a ground plate and a small area on the uninsulated tip of the electrode. The electrical field becomes denser at the electrode tip, and therefore, charged molecules around the tip start to oscillate [174]. Where the current is strong enough, this oscillation heats the tissues sufficiently to coagulate them. The volume of the tissue assumes the form of a spheroid. Coagulation occurs principally in a radial direction perpendicular to the long axis of the electrode [175, 176]. The dimensions of the lesions generated are proportional to the length and the width of the electrode. As a rule, in the radial direction, tissues up to 1.6 or 2.3 electrode-widths away from the electrode surface are coagulated [43, 175]. For practical implication, it is important to know that the electrode does not reliably coagulate in the distal direction. Therefore, electrodes that are placed perpendicular to the nerve may miss coagulating the nerve. Consequently, the electrode must be placed parallel to the target nerve [175,176]. Because the lesion size is proportional to the width of the electrode, small-gauge electrodes should be avoided.

The size of the lesion also depends on the temperature and duration of coagulation. Coagulation starts at a temperature of 65 °C [175, 176]. The volume of the lesion expands as the temperature increases to 80 °C. The optimal duration of coagulation lies between 60 and 90 s at 80 °C. During neurotomy, the temperature should be increased slowly [177].

Several ways exist in which radiofrequency neurotomy is currently practiced [178]. In this overview the technique recommended by the International Spine Intervention Society is described [43]. A steep caudocephalad axial tilt of the fluoroscopy beam along with a 20° lateral tilt is used [43]. The cannula can be positioned precisely parallel to the target nerve. However, the appearance of the vertebral structures might be unusual. A spinal needle inserted primarily as a guide might be needed. The distance from skin to the target nerve might be long (Fig. 16.10).

Lumbar medial branch neurotomy is performed as an outpatient procedure. The patient is placed prone on a radiolucent fluoroscopy table. The patient's back is prepared and draped in a sterile manner. An adhesive grounding pad is placed on the upper back and connected to the radiofrequency generator. Generally, no sedation, systemic analgesia, or premedication is required. The procedure can be performed under local anesthesia.

The technique is analogous in all lumbar levels. Only the terminology is different. The L5 dorsal ramus is itself targeted where it crosses the ala of the sacrum instead of its medial branch. One way to find the medial branch is to perform a medial branch block using a standard technique (Sect. 16.6.3 and Chap. 36) and leave the block needle in place. It can be used for the administration of local anesthetics. The tip of the needle will always be pointing to where the target nerve lies irrespective of the type of fluoroscopic views used. The target point is the lateral surface of the superior articular surface just above its junction with the root of the transverse process. The tip of the electrode must be placed proximally from the mamillo-accessory ligament (Fig. 16.5). At the sacrum, the mamillo-accessory ligament is rudimentary.

At all levels, the electrode needs to be as closely parallel to the nerve as possible. Typically, it needs to be  $15-20^{\circ}$  oblique to the sagittal plane and it must be inserted somewhere below the target level (Fig. 16.10). Sometimes multiple parallel placements of the electrode are necessary to coagulate the nerve properly. As a larger gauge



**Fig. 16.10** Illustration of a lateral view of the lumbar spine. The optimal trajectory of the electrode with an insertion point below the target area. (L3-L5) vertebral body, (I) inferior articular process, (S) superior articular process (Reproduced from Klessinger [42])

electrode makes larger lesions, an 18-G electrode with a 10-mm tip is recommended.

The accurate placement of the needle and the electrode must be documented with hard copy films, images on paper, or digital storage (Fig. 16.11). Now the nerve and the surrounding tissue can be anesthetized. The lesion is made by increasing the temperature slowly until it reaches 80 °C. This temperature is maintained for 60–90 s.

The radiofrequency electrode includes the possibility of nerve stimulation, with sensory and motor capabilities, which allows the precise localization of the target nerve. However, in the guidelines of the International Spine Intervention Society, electrical stimulation is considered unnecessary [43].

For lumbar radiofrequency medial branch neurotomy, several theoretical risks apply. These include hematoma, infection, and allergic reactions to local anesthetics. Provided that the electrodes are placed correctly, they penetrate only the skin and posterior back muscle. The spinal nerve and the ventral ramus lie more anterior. Skin burns should not be a risk when a correct ground plate is used. In the literature, no reports of adverse effects can be found [179, 180]. Examples from medicolegal proceedings are known.

#### 16.7.2.4 Results

A comprehensive narrative review of lumbar medial branch neurotomy was presented by Bogduk et al. [8]. Two main problems in the assessment of studies were described: (1) a technique without parallel needle placement and (2) an inconsistent patient selection [8].

Considering the historical development of radiofrequency neurotomy (Sect. 16.1.1), it is obvious that different techniques were used, which cannot be compared with one another. The position of the electrode plays an essential role. The earliest studies with the technique described by Shealy in 1974–1976 [14–17, 181] claimed good success even if it was not possible to coagulate the nerve with the described technique. In the later study of Leclaire et al. [182], the operative technique was not described. The outcome was



**Fig. 16.11** Different views of an electrode placed for an L4 medial branch neurotomy. (a) Anteroposterior view, (b) corresponding oblique view, (c) anteroposterior view of an electrode placed for an L5 medial branch neurotomy

poor. Negative results were also found in the study of van Wijk et al. [183]. Again, an inaccurate surgical technique was used [184].

In other studies, patient selection was questionable. Van Kleef et al. [185] did not select patients on the basis of controlled medial branch blocks but did require 50 % pain relief after single diagnostic blocks. A low success rate with a short duration was the result. Nevertheless, active treatment was superior to placebo treatment. Nath et al. [186] included patients with different pain sources. Controlled blocks and a correct technique were used. Complete and enduring pain relief was not reported because patients still had other sources of persisting pain. However, for the pain for which patients were treated, the study showed significant improvements after radiofrequency neurotomy compared with sham treatment. Another study [187] designed to test pulsed radiofrequency showed that conventional radiofrequency neurotomy was significantly more effective than sham treatment.

The first study with the appropriate selection criteria and correct surgical technique was the descriptive study of Dreyfuss et al. [173]. A total of 60 % of the patients treated with radiofrequency neurotomy achieved 80 % pain relief lasting at least 12 months and 80 % achieved 60 % pain relief. Similar outcomes were found in a study of Gofeld et al. [188]. Approximately 68 %

of the patients maintained at least 50 % pain relief lasting between 6 and 24 months. A third descriptive study of Burnham et al. [189] recorded high patient satisfaction.

The evidence shows that radiofrequency treatment fails when patients are wrongly selected or when an inaccurate technique is used [8]. Yet reviews in the past have acknowledged procedurally flawed studies. The reviews of Boswell et al. [190, 191] and Manchikanti et al. [192] provide strong evidence for short-term and moderate evidence for long-term relief. Manchikanti et al. [193] provide a strong recommendation for lumbar radiofrequency neurotomy. The strength of evidence in the review of Datta et al. [194] is II-2 to II-3 with a recommendation of 1B or 1C. Whenever patients have been selected correctly and when anatomically accurate surgical techniques have been used, the expectations of success have been met [8].

The available data vindicate the use of lumbar medial branch neurotomy provided that the correct surgical technique is used and patients are selected rigorously using controlled blocks [43]. There are no data that vindicate any other technique [43]. If the criterion for a positive response to diagnostic blocks is raised to complete relief, some 56 % of patients achieve complete relief of pain [195]. They restore their normal activities, and the need for other health care is eliminated.

#### 16.7.3 Cryoneurolysis

Different methods are used for peripheral nerve destruction (e.g., alcohol, phenol, or surgical lesions). However, there have not been any significant publications or systematic reviews with pulsed radiofrequency, cryoneurolysis, and laser neurotomy [191]. Since cryoneurolysis is established in some regions, it is discussed.

The cold temperature of the cryoprobe is created by passing a gas at a relatively high pressure through a small central channel in the probe. Once the gas reaches the tip of the probe, it enters an expansion chamber, which is at a much lower pressure. The high-pressure gas entering the chamber rapidly expands, causing the cooling of the gas and of the tip of the cryoprobe. Generally, the tips of these probes may be cooled to as low as -70 °C. This extreme cold then causes the tissues nearby to form an ice ball at the tip of the probe. The size of the ice ball is between 3.5 and 6 mm [196]. When close enough to a nerve, the ice ball causes the entrapment of the nerve. The application of cold to tissues creates a conduction block, similar to the effect of local anesthetics. At 10 °C, larger myelinated fibers stop conducting, but all nerve fibers stop conducting at -20 °C. The extent and duration of the effect is therefore a function of the degree of cold obtained and the duration of exposure [197]. Long-term pain relief from nerve freezing occurs because ice crystals create vascular damage to the vasa nervorum, which produces severe endoneural edema. This disrupts the nerve structure and creates Wallerian degeneration but leaves the myelin sheath and endoneurium intact [198].

Brechner [199] studied the effects of percutaneous cryoneuroablation of the zygapophysial joints in patients with neck and low back pain. There was 70 % pain relief after 1 h, but by 3 months, pain had returned to baseline. Schuster [200] studied 52 patients observed for a 13-month period. A total of 90.4 % had significant relief of low back pain after cryoneuroablation. Ross [201] described 21 out of 23 patients with complete relief for a follow-up of 6 months to 2 years. These studies are limited by a lack of certainty that the zygapophysial joint was the sole pain generator. Prognostic blocks, if performed, did not use currently recommended techniques. A current review about cryoanalgesia was presented by Trescot [198].

# 16.7.4 Therapeutic Zygapophysial Joint Blocks

Medial branch blocks are well known for their diagnostic capability (Sect. 16.7.1 and Chap. 36). However, they have been shown to be effective for long-term therapy. The results in the literature conflicting. are The International Spine Intervention Society does not mention this procedure in its guidelines [43, 152, 162], but the American Society of Interventional Pain Physicians assesses the evidence of therapeutic joint blocks with and without steroid for managing chronic low back pain as fair to good for short- and long-term improvement [202].

Two high-quality randomized trials [203, 204] reported positive results with or without steroids. In one study [204], 85 % of patients receiving local anesthetic and 90 % of patients receiving local anesthetic and steroids were successfully treated with approximately five or six procedures on average over a period of 2 years. The second study compared local anesthetic blocks with radiofrequency neurotomy [203]. At the end of 1 year, 90 % of the patients in the radiofrequency group and 69 % of the patients in the zygapophysial joint block group showed significant improvement. However, they did not use controlled blocks for selection.

The exact mechanism of the therapeutic effect is not known. Lumbar zygapophysial joint blocks may be repeated to reinstate the pain relief.

In other reviews, the indicated level of evidence is level II-1 or II-2 with a strong recommendation (1B or 1C) for the use of therapeutic joint blocks [194] or rather moderate for short-term and longterm pain relief after therapeutic medial branch blocks [190, 191]. In particular a strong recommendation (1B or 1C) is given for the use of therapeutic lumbar zygapophysial joint nerve blocks to provide both short-term and long-term relief in the treatment of chronic zygapophysial joint pain [205].

# 16.7.5 Intra-articular Steroids

Intra-articular injections are always performed with a therapeutic intention, not for diagnostic reasons.

Several reviews exist about lumbar zygapophysial joint injections with results from moderate evidence [190, 191] for short-term and long-term improvement in low back pain over very weak recommendation [194] to no recommendation for therapeutic blocks [8, 205].

Intra-articular steroids have not been tested in patients with a diagnosis of zygapophysial joint pain proven by controlled diagnostic blocks [162]. Therefore, injection of steroids is speculative. The available evidence argues strongly against intra-articular steroids having any effect greater than that of sham therapy [8, 206].

# 16.7.6 Treatment of Cysts

Lumbar zygapophysial joint access can be used in the management of synovial cysts of the lumbar zygapophysial joints [162]. The cyst can be distended and ruptured by forceful injection of contrast medium. Or it can simply be aspirated; or steroids can be injected into the joint [207, 208]. Descriptive studies are the sole level of evidence for such procedures. Success rates differ between 46 % [209] and 72 % [208], between 6 and 12 months after treatment.

# 16.7.7 Radiologic Imaging

#### 16.7.7.1 CT

In the guidelines of the International Spine Intervention Society, fluoroscopy is mandatory [43, 152, 162]. If CT is to be used, multisliced and pulsed CT is necessary to avoid a higher radiation exposure for the patient and the physician. Several images are obtained at one tube rotation. These may be displayed simultaneously or reconstructed. The optimization of new CT protocols has shown a possible effective dose reduction of 85 % compared with standard lumbar spine CT protocol [210]. Usage of CT might be more time-consuming than fluoroscopy. No real-time view of the track of the needle is available. An advantage of CT is that the target and any vulnerable structures, such as soft tissue, are directly visualized with CT rather than inferred from expected relationships to bony landmarks.

Continuous fluoroscopy, during and throughout the injection of a contrast medium, is the best available means of demonstrating intra-arterial flow away from the site of injection [179, 211]. Vascular opacification on CT is a dot or line traceable over multiple axial images. It is present during contrast injection and washes out after cessation of injection [212, 213]. There is no data to demonstrate that CT detects vascular uptake at a rate comparable to fluoroscopy or digital subtraction angiography.

The literature supporting the efficacy of CT-guided spine interventions is sparse and often of poor quality [214, 215].

# 16.7.7.2 MRI

MRI-guided zygapophysial joint infiltration provides an alternative to CT and fluoroscopy. The advantage is the avoidance of radiation for both the patient and physician, a high tissue contrast, and unrestricted multiplanar imaging capabilities. Clinical outcomes are comparable with conventional fluoroscopy [216].

Disadvantages include limited patient access inside the scanner and the enormous technical and financial expenses. MR-compatible equipment including injection needles is necessary.

#### 16.7.7.3 Ultrasound

Ultrasound injections have a place in interventional pain management. They may render some procedures safer and more effective. However, the best indications for ultrasound injections are not at the spine. Fluoroscopy is still a very strong competitor.

The biggest advantage of ultrasound is not having radiation. It provides continuous imaging and visualization of nearby vulnerable structures and real-time visualization of fluid injection. Smaller volumes of local anesthetic are needed. Ultrasound devices are portable.

Disadvantages include poor needle visualization and ultrasound shadows (no visibility behind
bones and air). Patient-related factors (e.g., body weight) affect visibility. Working under sterile conditions is more complicated. The success very much depends on practitioner's expertise.

In patients with zygapophysial joint pain, ultrasound injections may be an alternative to fluoroscopy for diagnostic blocks. However, there is no chance of seeing the lumbar medial branches or the L5 dorsal ramus [217]. Nevertheless, over 90 % successful needle placements seem possible by experienced practitioners [217]. There is also some evidence for feasibility of ultrasound-guided intra-articular injections, although the joint cleft is difficult to visualize and injected fluid cannot be visualized.

#### 16.7.8 Surgery

Surgery is occasionally performed to treat facet arthropathy despite a lack of evidence supporting fusion for degenerative spinal disorders [218, 219]. Not surprisingly, the results of studies evaluating the use of zygapophysial joint blocks to predict lumbar arthrodesis outcomes are discouraging [220–224].

New methods for percutaneous fusion of the zygapophysial joints are being developed but are not yet sufficiently investigated.

#### 16.8 Prevalence

The prevalence of zygapophysial pain is very difficult to specify. In the literature, studies with different prerequisites are found. Different criteria for success were applied: 50 % relief of pain, 80 % relief of pain, or complete relief of pain was used. To be valid, controlled blocks must be used to obviate false-positive results. However, in some studies, only uncontrolled blocks were used for diagnosis. Furthermore, the patient population, which was investigated in different studies, was unequal. Sometimes only certain age groups or only patients with injuries were included, and with patients degenerative changes were excluded. Almost all studies excluded patients with neurologic signs or symptoms secondary to a herniated disk, and many excluded patients

with previous back surgery [1]. Actually, all studies investigate only the presence of isolated zygapophysial joint pain, especially, of course, if 100 % pain relief was used as a diagnostic criterion in medial branch blocks.

Original prevalence studies used 50 % pain relief as a criterion. The patients studied were younger injured workers. The prevalence was 10-20 % [155]. Later studies with 75 % relief as a criterion reported prevalence rates of 27 % [159], 31 % [158], 38 % [156], and 45 % [155]. In elderly patients with no trauma in history and 90 % pain relief as a criterion, the prevalence was 40 % [225]. Using single diagnostic blocks but looking for complete pain relief, the prevalence of zygapophysial joint pain is only 10 % or less if controlled blocks were used [151].

In patients with thigh pain, it was demonstrated that the source of chronic low back pain (internal disk disruption, zygapophysial joint pain, or sacroiliac pain) varies with age. Older age was predictive of zygapophysial joint pain with a predicted probability of more the 50 % in 60-year-old patients and more than 85 % in patients over 80 years old [226].

Especially when considering degenerative diseases, zygapophysial joint pain often comes along with other pain sources (spinal canal stenosis, spondylolisthesis, disk degeneration, etc.). With different pain sources existing simultaneously, it would be great to know which proportion of pain derives from the zygapophysial joints. Unfortunately, no study exists about the prevalence of zygapophysial joint pain in patients with additional painful conditions. Assuming multiple sources of pain are possible at the same time, it is not possible to claim for complete pain relief after medial branch blocks and incomplete pain relief after radiofrequency neurotomy must also be accepted [227].

# 16.9 Zygapophysial Joint Pain in Selected Patients

Particularly well studied is zygapophysial joint pain in patients without comorbidities. In this group of patients, diagnostic standards

(Sect. 16.6.1) can be applied best and success rates after a specific therapy (Sect. 16.8.1) can be measured. In this chapter, the significance of zygapophysial joint pain is elaborated in patient groups in which zygapophysial joint pain is clinically relevant but does not occur as an isolated and independent disease. It is thus expected that diagnostic and therapeutic methods are only partially successful. For the patients, this can nevertheless make a significant difference in their daily lives.

#### 16.9.1 Elderly Patient

The degenerative changes described in Sect. 16.4 are more common in older age. The joints can be affected by osteoarthritis, which is believed to be a possible cause of zygapophysial joint pain [71–74]. Compared with other sources of low back pain (e.g., diskogenic pain or sacroiliac joint pain), zygapophysial joint pain becomes the most important pain source [226].

However, there is often an image of mixed pain of various causes. Especially in combination with diskogenic changes, spinal canal stenosis, and degenerative spondylolisthesis, several pain sources might exist.

In elderly patients, a common situation is that a surgical solution (e.g., decompression of the spinal canal or stabilization) is not taken into consideration. In this respect, interventional pain therapy is significantly prioritized as an additional treatment method to medication and physical therapy. Each, decrease in pain, however small, or a reduction of pain medicine is a success for the patient and means a relief in daily life.

#### 16.9.2 Spinal Canal Stenosis

In degenerative spinal canal stenosis, we are dealing with elderly patients who, on the one hand, have a symptomatology coming from the stenosis and the compression of the nerves in the dural sac. These symptoms are called *claudicatio spinalis* and are manifested in a restricted walking distance with pain, a sensory disturbance in the legs, or even neurologic deficits.

On the other hand, the most important reason for the development of a spinal canal stenosis is the destruction of the zygapophysial joints [105]. Therefore, patients suffer at the same time from pain deriving from the zygapophysial joints. Radiofrequency denervation might be a useful tool to treat the low back pain deriving from the zygapophysial joints. Of course it cannot be expected that the radiating pain or the claudication will improve. No results are known from literature.

#### 16.9.3 Spondylolisthesis

In degenerative spondylolisthesis, the whole upper vertebra slips relative to the lower vertebra [106] due to degenerative changes in the spine. To treat low back pain in patients with degenerative spondylolisthesis with radiofrequency denervation seems obvious because morphological abnormalities of the lumbar zygapophysial joints are a predisposing factor in the development of degenerative spondylolisthesis [109], pathology of the zygapophysial joints is a significant cause of low back pain within the lumbar spine [108], and radiofrequency neurotomy is the "gold standard" for treating zygapophysial joint pain [144].

A sufficient pain reduction could be achieved in 65 % of the treated patients for a reasonable time [116, 227]. It is known that these patients might have sources of pain other than just the zygapophysial joints [115]. In particular the often additionally present spinal canal stenosis causes symptoms not treated by medial branch neurotomy. The second pathology often interlinked with degenerative spondylolisthesis is disk degeneration [106, 107]. Diskogenic pain also is not treated by medial branch neurotomy.

#### 16.9.4 Failed Back Surgery

The treatment of postsurgery syndrome is difficult. Some patients will not improve with conservative measures, and therapies that are more interventional will be required. The evidence base for these interventions has grown in recent times [228, 229]. Zygapophysial joints are an important pain source not only in patients with chronic low back pain but also in patients after disk surgery [168, 169]. Therefore, a specific therapy against zygapophysial joint pain is rational. Continued pain following lumbar spine surgery has been hypothesized to be secondary to multiple causes, including epidural fibrosis, acquired stenosis, sacroiliac joint pain, and zygapophysial joint pain [230–232]. It is difficult in postlumbar surgery syndrome to identify paingenerating structures [233]. The prevalence of zygapophysial joint pain in patients with postlumbar laminectomy syndrome is 32 %. In patients after disk surgery, the prevalence of zygapophysial joint pain is 7 % and 28 % in patients with persistent back pain after surgery [169].

The reasons why the zygapophysial joints are involved even if the joint was untouched during the operation might be inflammatory processes, low-level trauma, changes in disk height, or stretching of the joint capsule [144]. The process of degenerative disk disease, particularly when enhanced by a herniated disk or diskectomy, results in progressive loss of intervertebral disk volume and disk height and increased load to the joints, which might be a reason for pain [234].

Zygapophysial joint pain can be identified and treated with a radiofrequency neurotomy with a success rate of 58.8 % [169] in patients after disk surgery.

After spinal fusion, zygapophysial joint pain can occur due to residual mobility in the index segment or in adjacent segments due to overload. Studies on the effectiveness of a specific joint therapy after spinal fusion do not exist.

#### Conclusions

Zygapophysial joints meet all prerequisites to be a source of pain. They are often involved in back pain and radiating pain and should not be underestimated. The prevalence of isolated zygapophysial joint pain increases with age. In addition, zygapophysial joint pain appears also in combination with other common spine diseases, such as disk degeneration, spinal canal stenosis, and spondylolisthesis. If the diagnosis is made with controlled medial branch blocks, radiofrequency denervation is the only validated treatment for pain mediated by the medial branches.

#### References

- Cohen SP, Raja SN. Pathogenesis, diagnosis, and treatment of lumbar zygapophysial (facet) joint pain. Anesthesiology. 2007;106:591–614.
- Bogduk N. Low back pain. In: Clinical and radiological anatomy of the lumbar spine. 5th revised edn. Edinburgh. Elesevier, Churchill Livingstone; 2012. p. 173–205.
- Goldthwaith JE. The lumbosacral articulation. An explanation of many cases of lumbago, sciatica and paraplegia. Boston Med Surg J. 1911;64:365–72.
- Putti V. New concepts in the pathogenesis of sciatica pain. Lancet. 1927;2:53–60.
- Ghormley RK. Low back pain with special reference to the articular facets, with presentation of an operative procedure. JAMA. 1933;101:1773–7.
- Badgley CE. The articular facets in relation to lowback pain and sciatic radiation. J Bine Joint Surg. 1941;23:481–96.
- Brunori A, De Caro GM, Giuffrè R. Surgery of lumbar disk hernia: historical perspective. Ann Ital Chir. 1998;69:285–93.
- Bogduk N, Dreyfuss P, Govind J. A narrative review of lumbar medial branch neurotomy for the treatment of back pain. Pain Med. 2009;10:1035–45.
- Rees WS. Multiple bilateral subcutaneous rhizolysis of segmental nerves in the treatment of the intervertebral disc syndrome. Ann Gen Pract. 1971;16:126–7.
- Rees WS. Multiple bilateral percutaneous rhizolysis. Med J Aust. 1975;1:536–7.
- Bogduk N, Colman RR, Winer CE. An anatomical assessment of the "percutaneous rhizolysis" procedure. Med J Aust. 1977;1:397–9.
- Taha JM. Percutaneous radiofrequency trigeminal ganglionolysis. In: Burchiel K, editor. Surgical management of pain. New York: Thieme; 2002. p. 841–8.
- White JV, Sweet WH. Pain and the neurosurgeon. Springfield: C.C. Thomas; 1969. p. 193–7.
- 14. Shealy CN. Facets in back and sciatic pain. Minn Med. 1974;57:199–203.
- 15. Shealy CN. The role of the spinal facets in back and sciatic pain. Headache. 1974;14:101–4.
- Shealy CN. Percutaneous radiofrequency denervation of spinal facets. J Neurosurg. 1975;43:448–51.
- Shealy CN. Facet denervation in the management of back sciatic pain. Clin Orthop. 1976;115:157–64.
- Bogduk N, Long DM. The anatomy of the so-called articular nerves and their relationship to facet denervation in the treatment of low back pain. J Neurosurg. 1979;51:172–7.

- Bogduk N, Long DM. Percutaneous lumbar medial branch neurotomy. A modification of facet denervation. Spine. 1980;5:193–200.
- Lord SM, Barnsley L, Wallis BJ, et al. Percutaneous radio-frequency neurotomy for chronic cervical zygapophyseal-joint pain. N Engl J Med. 1996;335:1721–6.
- Birkenmaier C, Terzis A, Wegener B, et al. The gel box—a testing device for the characterization of cryo- and radiofrequency lesions employed in interventional pain therapy. Pain Physician. 2010;13:263–71.
- Baerlocher CB, Krauss JK, Seiler RW. Kryorhizotomy: an alternative technique for lumbar medial branch rhizotomy in lumbar facet syndrome. J Neurosurg. 2003;98:14–20.
- Birkenmaier C, Veihelmann A, Trouillier H, et al. Percutaneous cryodenervation of lumbar facet joints: a prospective clinical trial. Int Orthop. 2007;31:525–30.
- Stander M, Marz U, Steude U, et al. The facet syndrome: frequent cause of chronic backaches. MMW Fortschr Med. 2006;148:33–4.
- Arthur JM, Racz GB. Cryolysis. In: Raj PP, editor. Pain medicine: a comprehensive review. 2nd ed. Philadelphia: Mosby; 2003. p. 297–303.
- Lloyd JW, Barnard JDW, Glynn CJ. Cryoanalgesia, a new approach to pain relief. Lancet. 1976;2:932–4.
- Nehme AE, Warfield CA. Cryoanalgesia: freezing of peripheral nerves. Hosp Pract. 1987;22(1A):71–7.
- Barnard JD, Lloyd JW. Cryoanalgesia. Nurs Times. 1977;73:897–9.
- Barnard D. The effects of extreme cold on sensory nerves. Ann R Coll Surg Engl. 1980;62:180–7.
- Evans PJ, Lloyd JW, Green CJ. Cryoanalgesia: the response to alterations in freeze cycle and temperature. Br J Anaesth. 1981;53:1121–6.
- Bogduk N. The zygapophysial joints detailed structure. In: Clinical and radiological anatomy of the lumbar spine. 5th revised edn. Edinburgh. Elesevier, Churchill Livingstone; 2012. p. 29–38.
- Glover JR. Arthrography of the joints of the lumbar vertebral arches. Orthop Clin North Am. 1977;8:37–42.
- Benini A. Das kleine Gelenk der Lendenwirbelsäule. Fortschr Med. 1979;97:2103–6.
- 34. Horwitz T, Smith RM. An anatomical, pathological and roentgenological study of the intervertebral joints of the lumbar spine and of the sacroiliac joints. Am J Roentgenol. 1940;43:173–86.
- Czervionke LF, Fenton DS. Facet joint injection and medial branch block. In: Fenton DS, Czervionke LF, editors. Image-guided spine intervention. Philadelphia: Saunders; 2003. p. 9–50.
- Bogduk N. Movements of the lumbar spine. In: Clinical and radiological anatomy of the lumbar spine. 5th revised edn. Edinburgh. Elesevier, Churchill Livingstone; 2012. p. 73–92.
- Dorr WM. Über die Anatomie der Wirbelgelenke. Arch Orthop Unfallchir. 1958;50:222–2334.

- Yamashita T, Minaki Y, Ozaktay AC, Cavanaugh JM, King AI. A morphological study of the fibrous capsule of the human lumbar facet joint. Spine. 1996;21:538–43.
- Yong-Hing K, Reilly J, Kirkaldy-Willis WH. The ligamentum flavum. Spine. 1976;1:226–34.
- Twomey LT, Taylor JR. Age changes in the lumbar articular triad. Aust J Physiother. 1985;31:106–12.
- Wolf J. The reversible deformation of the joint cartilage surface and its possible role in joint blockage. Rehabilitacia. 1975;8:30–6.
- Klessinger S. Denervation of the zygapophysial joints of the cervical and lumbar spine. Tech Orthop. 2013;28(1):23–34.
- 43. International Spine Intervention Society. Lumbar medial branch thermal radiofrequency neurotomy. In: Bogduk N. (ed). Practice Guidelines for spinal diagnostic and treatment procedures. 2nd edn. International Spine Intervention Society. San Francisco, 2013. p 601–29.
- Bogduk N. The lumbar mamillo-accessory ligament. Its anatomical and neurosurgical significance. Spine (Phila Pa 1976). 1981;6:162–7.
- Bogduk N. The innervation of the lumbar spine. Spine. 1983;8:286–93.
- Lippitt AB. The facet joint and its role in spine pain. Management with facet joint injections. Spine. 1984;9:746–50.
- 47. Hirsch C, Ingelmark BE, Miller M. The anatomical basis for low back pain. Studies on the presence of sensory nerve endings in ligamentous, capsular and intervertebral disc structures in the human lumbar spine. Acta Orthop Scand. 1963;33:1–17.
- Bogduk N, Tynan W, Wilson AS. The nerve supply to the human lumbar intervertebral discs. J Anat. 1981;132:39–56.
- Bogduk N, Wilson AS, Tynan W. The human lumbar dorsal rami. J Anat. 1982;134:383–97.
- Jackson HAC, Winkelmann RK, Bickel WH. Nerve endings in the human lumbar spinal column and related structures. J Bone Joint Surg. 1966;48:1272–81.
- 51. Ashton IK, Ashton BA, Gibson SJ, Polak JM, Jaffray DC, Eisenstein SM. Morphological basis for back pain: the demonstration of nerve fibers and neuropeptides in the lumbar facet joint capsule but not in ligamentum flavum. J Orthop Res. 1992;10:72–8.
- Giles LG. Human lumbar zygapophyseal joint inferior recess synovial folds: a light microscope examination. Anat Rec. 1988;220:117–24.
- Giles LG, Taylor JR. Innervation of lumbar zygapophyseal joint synovial folds. Acta Orthop Scand. 1987;58:43–6.
- Adams MA, Hutton WC. The mechanical function of the lumbar apophyseal joints. Spine. 1983;8:327–30.
- Lorenz M, Patwardhan A, Vanderby Jr R. Loadbearing characteristics of lumbar facets in normal and surgically altered spinal segments. Spine. 1983;8:122–30.
- 56. El-Bohy AA, Yang KH, King AI. Experimental verification of facet load transmission by direct measure-

ment of facet lamina contact pressure. J Biomech. 1989;22:931–41.

- Adams MA, Hutton WC. The effect of posture on the role of the apophysial joints in resisting intervertebral compressive forces. J Bone Joint Surg. 1980;62:358–62.
- Cyron BM, Hutton WC. The tensile strength of the capsular ligaments of the apophyseal joints. J Anat. 1981;132:145–50.
- Dunlop RB, Adams MA, Hutton WC. Disc space narrowing and the lumbar facet joints. J Bone Joint Surg Br. 1984;66:706–10.
- Adams MA, Dolan P, Hutton WC. The lumbar spine in backward bending. Spine. 1988;13:1019–26.
- Asano S, Kaneda K, Umehara S, Tadano S. The mechanical properties of the human L4-5 functional spinal unit during cyclic loading. The structural effects of the posterior elements. Spine. 1992;17:1343–52.
- Torgerson WR, Dotter WE. Comparative roentgenographic study of the asymptomatic and symptomatic lumbar spine. J Bone Joint Surg Am. 1976;58:850–3.
- Lawrence JS, Bremner JM, Osteo-arthrosis BF. Prevalence in the population and relationship between symptoms and x-ray changes. Ann Rheum Dis. 1966;25:1–24.
- 64. Bogduk N. Age changes in the lubar spine. In: Clinical and radiological anatomy of the lumbar spine. 5th revised edn. Edinburgh. Elesevier, Churchill Livingstone; 2012. p. 157–64.
- Twomey L, Taylor J. Age changes in lumbar intervertebral discs. Acta Orthop Scand. 1985;56:496–9.
- Taylor JR, Twomey LT. Age changes in lumbar zygapophyseal joints. Observations on structure and function. Spine. 1986;11:739–45.
- Hilton RC, Ball J, Benn RT. In-vitro mobility of the lumbar spine. Ann Rheum Dis. 1979;38:378–83.
- Tanz SS. Motion of the lumbar spine; a roentgenologic study. Am J Roentgenol Radium Ther Nucl Med. 1953;69:399–412.
- Vernon-Roberts B, Pirie CJ. Degenerative changes in the intervertebral discs of the lumbar spine and their sequelae. Rheumatol Rehabil. 1977;16:13–21.
- Mooney V, Robertson J. The facet syndrome. Clin Orthop. 1976;115:149–56.
- Eisenstein SM, Parry CR. The lumbar facet arthrosis syndrome. Clinical presentation and articular surface changes. J Bone Joint Surg Br. 1987;69:3–7.
- Abel MS. The radiology of low back pain associated with posterior element lesions of the lumbar spine. Crit Rev Diagn Imaging. 1984;20:311–52.
- Carrera GF, Williams AL. Current concepts in evaluation of the lumbar facet joints. Crit Rev Diagn Imaging. 1984;21:85–104.
- Lynch MC, Taylor JF. Facet joint injection for low back pain. A clinical study. J Bone Joint Surg Br. 1986;68:138–41.
- Willburger RA, Wittenberg RH. Prostaglandin release from lumbar disc and facet joint tissue. Spine. 1994;19:2068–70.

- 76. Tachihara H, Kikuchi S, Konno S, Sekiguchi M. Does facet joint inflammation induce radiculopathy?: an investigation using a rat model of lumbar facet joint inflammation. Spine. 2007;32:406–12.
- 77. Dong L, Guarino BB, Jordan-Sciutto KL, Winkelstein BA. Activating transcription factor 4, a mediator of the integrated stress response, is increased in the dorsal root ganglia following painful facet joint distraction. Neuroscience. 2011;193:377–86.
- 78. Bär K, Natura G, Telleria-Diaz A, Teschner P, Vogel R, Vasquez E, Schaible HG, Ebersberger A. Changes in the effect of spinal prostaglandin E2 during inflammation: prostaglandin E (EP1–EP4) receptors in spinal nociceptive processing of input from the normal or inflamed knee joint. J Neurosci. 2004;24:642–51.
- Lin C, Amaya F, Barrett L, Wang H, Takada J, Samad TA, Woolf CJ. Prostaglandin E2 receptor EP4 contributes to inflammatory pain hypersensitivity. J Pharmacol Exp Ther. 2006;319:1096–103.
- Vasquez E, Bär K, Ebersberger A, Klein B, Vanegas H, Schaible HG. Spinal prostaglandins are involved in the development but not the maintenance of inflammation-induced spinal hyperexcitability. J Neurosci. 2001;21:9001–8.
- 81. Xu D, Sun Y, Bao G, Liu W, Zhu X, Cui S, Fan J, Cui Z. MMP-1 overexpression induced by IL-1β: possible mechanism for inflammation in degenerative lumbar facet joint. J Orthop Sci. 2013;18: 1012–9.
- 82. Moneta GB, Videman T, Kaivanto K, Aprill C, Spivey M, Vanharanta H, Sachs BL, Guyer RD, Hochschuler SH, Raschbaum RF, et al. Reported pain during lumbar discography as a function of anular ruptures and disc degeneration. A re-analysis of 833 discograms. Spine. 1994;19:1968–74.
- Kirkaldy-Willis WH, Wedge JH, Yong-Hing K, Reilly J. Pathology and pathogenesis of lumbar spondylosis and stenosis. Spine. 1978;3:319–28.
- Gottfried Y, Bradford DS, Oegema TR. Facet joint changes after chemo- nucleolysis induced disc space narrowing. Spine. 1986;11:944–54.
- Panjabi MM, Krag MH, Chung TQ. Effects of disc injury on mechanical behavior of the human spine. Spine. 1984;9:707–13.
- Haher TR, O'Brien M, Dryer JW, Nucci R, Zipnick R, Leone DJ. The role of the lumbar facet joints in spinal stability. Spine. 1994;19:2667–71.
- Adams MA, Freeman BJ, Morrison HP, Nelson IW, Dolan P. Mechanical initiation of intervertebral disc degeneration. Spine. 2000;25:1625–36.
- Fujiwara A, Tamai K, Yamato M, An HS, Yoshida H, Saotome K, Kurihashi A. The relationship between facet joint osteoarthritis and disc degeneration of the lumbar spine: an MRI study. Eur Spine J. 1999;8:396–401.
- Mattei TA, Goulart CR, McCall TD. Pathophysiology of regression of synovial cysts of the lumbar spine: the 'anti-inflammatory hypothesis'. Med Hypotheses. 2012;79:813–8.

- 90. Shin KM, Kim MS, Ko KM, Jang JS, Kang SS, Hong SJ. Percutaneous aspiration of lumbar zygapophyseal joint synovial cyst under fluoroscopic guidance -a case report. Korean J Anesthesiol. 2012;62:375–8.
- Sabo RA, Tracy PT, Weinger JM. A series of 60 juxtafacet cysts: clinical presentation, the role of spinal instability, and treatment. J Neurosurg. 1996;85:560–5.
- Hemminghytt S, Daniels DL, Williams AL, Haughton VM. Intraspinal synovial cysts: natural history and diagnosis by CT. Radiology. 1982;145:375–6.
- Howington JU, Connolly ES, Voorhies RM. Intraspinal synovial cysts: 10-year experience at the Ochsner Clinic. J Neurosurg. 1999;91:193–9.
- Rapin PA, Gerster JC. Calcified synovial cysts of zygapophyseal joints. J Rheumatol. 1993;20:767–8.
- 95. Doyle AJ, Merrilees M. Synovial cysts of the lumbar facet joints in a symptomatic population: prevalence on magnetic resonance imaging. Spine. 2004;29:874–8.
- Kalichman L, Suri P, Guermazi A, Li L, Hunter DJ. Facet orientation and tropism: associations with facet joint osteoarthritis and degeneratives. Spine. 2009;34:E579–85.
- Berlemann U, Jeszenszky DJ, Buhler DW, et al. Facet joint remodeling in degenerative spondylolisthesis: an investigation of joint orientation and tropism. Eur Spine J. 1998;7:376–80.
- Boden SD, Riew KD, Yamaguchi K, et al. Orientation of the lumbar facet joints: association with degenerative disc disease. J Bone Joint Surg Am. 1996;78:403–11.
- Fujiwara A, Tamai K, An HS, et al. Orientation and osteoarthritis of the lumbar facet joint. Clin Orthop Relat Res. 2001;385:88–94.
- 100. Grogan J, Nowicki BH, Schmidt TA, et al. Lumbar facet joint tropism does not accelerate degeneration of the facet joints. AJNR Am J Neuroradiol. 1997;18:1325–9.
- 101. Sato K, Wakamatsu E, Yoshizumi A, et al. The configuration of the laminas and facet joints in degenerative spondylolisthesis: a clinicoradiologic study. Spine. 1989;14:1265–71.
- 102. Do DH, Taghavi CE, Fong W, Kong MH, Morishita Y, Wang JC. The relationship between degree of facet tropism and amount of dynamic disc bulge in lumbar spine of patients symptomatic for low back pain. Eur Spine J. 2011;20:71–8.
- 103. Kong MH, He W, Tsai YD, Chen NF, Keorochana G, Do DH, Wang JC. Relationship of facet tropism with degeneration and stability of functional spinal unit. Yonsei Med J. 2009;50:624–9.
- 104. Suri P, Hunter DJ, Rainville J, Guermazi A, Katz JN. Presence and extent of severe facet joint osteoarthritis are associated with back pain in older adults. Osteoarthritis Cartilage. 2013;21:1199–206.
- 105. Aebi M. The adult scoliosis. Eur Spine J. 2005;14:925–48.
- Kalichman L, Hunter DJ. Diagnosis and conservative management of degenerative lumbar spondylolisthesis. Eur Spine J. 2008;17:327–35.

- 107. Sengupta DK, Herkowitz HN. Degenerative spondylolisthesis: review of current trends and controversies. Spine. 2005;30:S71–81.
- 108. Berven S, Tay BB, Colman W, Hu SS. The lumbar zygapophyseal (facet) joints: a role in the pathogenesis of spinal pain syndromes and degenerative spondylolisthesis. Semin Neurol. 2002;22: 187–96.
- 109. Dai LY. Orientation and tropism of lumbar facet joints in degenerative spondylolisthesis. Int Orthop. 2001;25:40–2.
- Fitzgerald JA, Newman PH. Degenerative spondylolisthesis. J Bone Joint Surg Br. 1976;58(2):184–92.
- 111. Toyone T, Ozawa T, Kamikawa K, Watanabe A, Matsuki K, Yamashita T, Wada Y. Facet joint orientation difference between cephalad and caudad portions: a possible cause of degenerative spondylolisthesis. Spine. 2009;34:2259–62.
- 112. Hasegawa K, Kitahara K, Shimoda H, Hara T. Facet joint opening in lumbar degenerative diseases indicating segmental instability. J Neurosurg Spine. 2010;12:687–93.
- 113. Alicioglu B, Sut N. Synovial cysts of the lumbar facet joints: a retrospective magnetic resonance imaging study investigating their relation with degenerative spondylolisthesis. Prague Med Rep. 2010;110:301–9.
- Schinnerer KA, Katz LD, Grauer JN. MR findings of exaggerated fluid in facet joints predicts instability. J Spinal Disord Tech. 2008;21:468–72.
- 115. Jensen TS, Karppinen J, Sorensen JS, Niinimäki J, Leboeuf-Yde C. Vertebral endplate signal changes (Modic change): a systematic literature review of prevalence and association with non-specific low back pain. Eur Spine J. 2008;17:1407–22.
- 116. Klessinger S. Radiofrequency neurotomy for treatment of low back pain in patients with minor degenerative spondylolisthesis. Pain Physician. 2012;15:E71–8.
- 117. Revel ME, Listrat VM, Chevalier XJ, Dougados M, Nguyen MP, Vallee C, Wybier M, Gires F, Amor B. Facet joint block for low back pain: identifying predictors of a good response. Arch Phys Med Rehabil. 1992;73:824–8.
- 118. Revel M, Poiraudeau S, Auleley GR, Payan C, Denke A, Nguyen M, Chevrot A, Fermanian J. Capacity of the clinical picture to characterize low back pain relieved by facet joint anesthesia: proposed criteria to identify patients with painful facet joints. Spine. 1998;23:1972–6.
- 119. Jackson RP, Jacobs RR, Montesano PX. Facet joint injection in low-back pain: a prospective statistical study. Spine. 1988;13:966–71.
- 120. Song KJ, Lee KB. Bilateral facet dislocation on L4-5 without neurologic deficit. J Spinal Disord. 2005;18:462–4.
- De Das S, McCreath SW. Lumbosacral fracture dislocations: a report of four cases. J Bone Joint Surg. 1981;63:58–60.
- 122. Fabris D, Costantini S, Nena U, Lo Scalzo V. Traumatic L5-S1 spondylolisthesis: report of

three cases and a review of the literature. Eur Spine J. 1999;8:290–5.

- 123. Kaplan SS, Wright NM, Yundt KD, Lauryssen C. Adjacent fracture-dislocations of the lumbosacral spine: case report. Neurosurgery. 1999;44:1134–7.
- 124. Verlaan JJ, Oner FC, Dhert WJ, Verbout AJ. Traumatic lumbosacral dislocation: case report. Spine. 2001;26:1942–4. 116. Veras del Monte LM, Bago J. Traumatic lumbosacral dislocation. Spine. 2000;25:756–9.
- 125. Adams MA, Hutton WC. The relevance of torsion to the mechanical derangement of the lumbar spine. Spine. 1981;6:241–8.
- Lamy C, Bazergui A, Kraus H, Farfan HF. The strength of the neural arch and the etiology of spondylolysis. Orthop Clin North Am. 1975;6:215–31.
- 127. Sullivan JD, Farfan HF. The crumpled neural arch. Orthop Clin North Am. 1975;6:199–214.
- 128. Taylor JR, Twomey LT, Corker M. Bone and soft tissue injuries in post-mortem lumbar spines. Paraplegia. 1990;28:119–29.
- Twomey LT, Taylor JR, Taylor MM. Unsuspected damage to lumbar zygapophyseal (facet) joints after motor-vehicle accidents. Med J Aust. 1989;151: 210–7.
- de Vlam K, Mielants H, Verstaete KL, Veys EM. The zygapophyseal joint determines morphology of the enthesophyte. J Rheumatol. 2000;27:1732–9.
- 131. Guillaume MP, Hermanus N, Peretz A. Unusual localisation of chronic arthropathy in lumbar facet joints after parvovirus B19 infection. Clin Rheumatol. 2002;21:306–8.
- 132. Ball J. Enthesopathy of rheumatoid and ankylosing spondylitis. Ann Rheum Dis. 1971;30:213–23.
- Campbell AJ, Wells IP. Pigmented villonodular synovitis of a lumbar vertebral facet joint. J Bone Joint Surg. 1982;64:145–6.
- 134. Smida M, Lejri M, Kandara H, Sayed M, Ben Chehida F, Ben Ghachem M. Septic arthritis of a lumbar facet joint case report and review of the literature. Acta Orthop Belg. 2004;70:290–4.
- Dreyfuss PH, Dreyer SJ, Herring SA. Lumbar zygapophysial (facet) joint injections. Spine. 1995;20:2040–7.
- 136. Fujishiro T, Nabeshima Y, Yasui S, Fujita I, Yoshiya S, Fujii H. Pseudogout attack of the lumbar facet joint: a case report. Spine. 2002;27:E396–8.
- 137. Merskey H, Bogduk N, editors. Classification of chronic pain. Descriptions of chronic pain syndromes and definitions of pain terms. 2nd ed. Seattle: IASP Press; 1994.
- Bogduk N. On the definitions and physiology of back pain, referred pain and radicular pain. Pain. 2009;147:17–9.
- McCall IW, Park WM, O'Brien JP. Induced pain referral from posterior lumbar elements in normal subjects. Spine. 1979;4:441–6.
- 140. Fenton DS, Czervionke LF. Facet denervation. In: Fenton DS, Czervionke LF, editors. Image-guided spine intervention. Philadelphia: Saunders; 2003. p. 51–71.

- 141. Fairbank JC, Park WM, McCall IW, O'Brien JP. Apophyseal injection of local anesthetic as a diagnostic aid in primary low-back pain syndromes. Spine. 1981;6:598–605.
- Bogduk N, Jull G. The theoretical pathology of acute locked back: a basis for manipulative therapy. Man Med. 1985;1:78–82.
- 143. Hancock MJ, Maher CG, Latimer J, et al. Systematic review of tests to identify the disc, SIJ or facet joint as the source of low back pain. Eur Spine J. 2007;16:1539–50.
- 144. van Kleef M, Vanelderen P, Cohen SP, Lataster A, Van Zundert J, Mekhail N. 12. Pain originating from the lumbar facet joints. Pain Pract. 2010;10: 459–69.
- 145. Carrera GF. Lumbar facet joint injection in low back pain and sciatica: preliminary results. Radiology. 1980;137:665–7.
- 146. Murtagh FR. Computed tomography and fluoroscopy guided anesthesia and steroid injection in facet syndrome. Spine. 1988;13:686–9.
- 147. Weishaupt D, Zanetti M, Boos N, Hodler J. MR imaging and CT in osteoarthritis of the lumbar facet joints. Skeletal Radiol. 1999;28:215–9.
- 148. Leone A, Aulisa L, Tamburrelli F, Lupparelli S, Tartaglione T. The role of computed tomography and magnetic resonance in assessing degenerative arthropathy of the lumbar articular facets [in Italian]. Radiol Med. 1994;88:547–52.
- 149. Magora A, Schwartz A. Relation between the low back pain syndrome and X-ray findings. 1. Degenerative osteoarthritis. Scand J Rehabil Med. 1976;8:115–25.
- 150. Schwarzer AC, Wang SC, O'Driscoll D, Harrington T, Bogduk N, Laurent R. The ability of computed tomography to identify a painful zygapophysial joint in patients with chronic low back pain. Spine. 1995;20: 907–12.
- 151. Klessinger S. Medial branch blocks of the cervical and lumbar spine. Tech Orthop. 2013;28:18–22.
- 152. International Spine Intervention Society. Lumbar medial branch blocks. In: Bogduk N. (ed). Practice Guidelines for spinal diagnostic and treatment procedures. 2nd edn. International Spine Intervention Society. San Francisco, 2013. p 559–600.
- 153. Bogduk N. Diagnostic nerve blocks in chronic pain. In: Breivik H, Shipley M, editors. Pain. Best practice & research compendium. Edingurgh: Elsevier; 2007. p. 47–55.
- 154. Curatolo M, Bogduk N. Diagnostic and therapeutic nerve blocks. In: Fishman SM, Balantyne JC, Rathmell JP, editors. Bonica's management of pain. 4th ed. Philadelphia: Wolters Kluwer; 2010. p. 1401–23.
- 155. Manchikanti L, Pampati V, Fellows B, Bakhit CE. Prevalence of lumbar facet joint pain in chronic low back pain. Pain Physician. 1999;2:59–64.
- 156. Manchikanti L, Pampati V, Fellows B, Bakhit CE. The diagnostic validity and therapeutic value of lumbar facet joint nerve blocks with or without adjuvant agents. Curr Rev Pain. 2000;4:337–44.

- 157. Schwarzer AC, Aprill CN, Derby R, Fortin J, Kine G, Bogduk N. The false-positive rate of uncontrolled diagnostic blocks of the lumbar zygapophysial joints. Pain. 1994;58:195–200.
- 158. Manchikanti L, Boswell MV, Singh V, Pampati V, Damron KS, Beyer CD. Prevalence of facet joint pain in chronic spinal pain of cervical, thoracic, and lumbar regions. BMC Musculoskelet Disord. 2004;5:15.
- 159. Manchukonda R, Manchikanti KN, Cash KA, Pampati V, Manchikanti L. Facet joint pain in chronic spinal pain: an evaluation of prevalence and false-positive rate of diagnostic blocks. J Spinal Disord Tech. 2007;20:539–45.
- 160. Derby R, Melnik I, Choi J, Lee JE. Indications for repeat diagnostic medial branch nerve blocks following a failed first medial branch nerve block. Pain Physician. 2013;16:479–88.
- 161. Klessinger S. Cervical medial branch radiofrequency neurotomy. Pain Med. 2012;13:621.
- 162. International Spine Intervention Society. Lumbar zygapophysial joint access. In: Bogduk N. (ed). Practice Guidelines for spinal diagnostic and treatment procedures. 2nd edn. International Spine Intervention Society. San Francisco, 2013. p 373–92.
- 163. Lilius G, Harilainen A, Laasonen EM, Myllynen P. Chronic unilateral low-back pain. Predictors of outcome of facet joint injections. Spine. 1990;15:780–2.
- 164. Carette S, Marcoux S, Truchon R, Grondin C, Gagnon J, Allard Y, Latulippe M. A controlled trial of corticosteroid injections into facet joints for chronic low back pain. N Engl J Med. 1991;325:1002–7.
- 165. Zerbini C, Ozturk ZE, Grifka J, Maini M, Nilganuwong S, Morales R, Hupli M, Shivaprakash M, Giezek H. Efficacy of etoricoxib 60 mg/day and diclofenac 150 mg/day in reduction of pain and disability in patients with chronic low back pain: results of a 4-week, multinational, randomized, doubleblind study. Curr Med Res Opin. 2005;21:2037–49.
- 166. Videman T, Osterman K. Double-blind parallel study of piroxicam versus indomethacin in the treatment of low back pain. Ann Clin Res. 1984;16:156–60.
- 167. Mens JM. The use of medication in low back pain. Best Pract Res Clin Rheumatol. 2005;19:609–21.
- 168. Klessinger S. Radiofrequency neurotomy for the treatment of therapy-resistant neck pain after ventral cervical operations. Pain Med. 2010;11:1504–10.
- 169. Klessinger S. Zygapophysial joint pain in post lumbar surgery syndrome. The efficacy of medial branch blocks and radiofrequency neurotomy. Pain Med. 2013;14:374–7.
- Schofferman J, Kine G. Effectiveness of repeated radiofrequency neurotomy for lumbar facet pain. Spine. 2004;29:2471–3.
- 171. Kaplan M, Dreyfuss P, Halbrook B, Bogduk N. The ability of lumbar medial branch blocks to anesthetize the zygapophysial joint. Spine. 1998;23:1847–52.
- 172. Dreyfuss P, Schwarzer AC, Lau P, Bogduk N. Specificity of lumbar medial branch and L5 dor-

sal ramus blocks: a computed tomographic study. Spine. 1997;22:895–902.

- 173. Dreyfuss P, Halbrook B, Pauza K, et al. Efficacy and validity of radiofrequency neurotomy for chronic lumbar zygapophysial joint pain. Spine. 2000;25:1270–7.
- Organ LW. Electrophysiologic principles of radiofrequency making. Appl Neurophysiol. 1976;39:69–76.
- 175. Lord SM, McDonald GJ, Bogduk N. Percutanous radiofrequency neurotomy of the cervical medial branches: a validated treatment for cervical zygapophysial joint pain. Neurosurg Q. 1998;8:288–308.
- Bogduk N, Macintosh J, Marsland A. Technical limitations to the efficacy of radiofrequency neurotomy for spinal pain. Neurosurgery. 1987;20:529–35.
- 177. Alberts WW, Wright Jr EW, Feinstein B, Von Bonin G. Experimental radiofrequency brain lesion size as a function of physical parameters. J Neurosurg. 1966;25:421–3.
- 178. Gofeld M, Faclier G. Radiofrequency denervation of the lumbar zygapophysial joints – targeting the best practice. Pain Med. 2008;9:204–11.
- 179. Bogduk N, Dreyfuss P, Baker R, Yin W, Landers M, Hammer M, Aprill C. Complications of spinal diagnostic and treatment procedures. Pain Med. 2008;9:11–34.
- Kornick C, Kramarich SS, Lamer TJ, Sitzman TB. Complications of lumbar facet radiofrequency denervation. Spine. 2004;29:1352–4.
- 181. Gallagher J, di Valdo PL P, Wedley JR, et al. Radiofrequency facet joint denervation in the treatment of low back pain: a prospective controlled double-blind study to assess its efficacy. Pain Clin. 1994;7:193–8.
- 182. Leclaire R, Fortin L, Lamber R, et al. Radiofrequency facet joint denervation in the treatment of low back pain: a placebo controlled clinical trial to assess efficacy. Spine. 2001;26:1411–7.
- 183. van Wijk RMA, Geurts JWM, Wynne HJ, et al. Radiofrequency denervation of lumbar facet joints in the treatment of chronic low back pain. A randomized, double-blind sham lesion-controlled trial. Clin J Pain. 2004;21:335–44.
- Bogduk N. Lumbar radiofrequency neurotomy. Clin J Pain. 2006;22:409.
- 185. van Kleef M, Barendse GA, Kessels A, et al. Randomized trial of radiofrequency lumbar facet denervation for chronic low back pain. Spine. 1999;24:1937–42.
- 186. Nath S, Nath CA, Pettersson K. Percutaneous lumbar zygapophysial (facet) joint neurotomy using radiofrequency current, in the management of chronic low back pain. A randomized double-blind trial. Spine. 2008;33:1291–7.
- 187. Tekin I, Mirzai H, Ok G, Erbuyun K, Vatansever D. A comparison of conventional and pulsed radiofrequency denervation in the treatment of chronic facet joint pain. Clin J Pain. 2007;23:524–9.
- 188. Gofeld M, Jitendra J, Faclier G. Radiofrequency denervation of the lumbar zygapophysial joints:

10-year prospective audit. Pain Physician. 2007;10:291–300.

- 189. Burnham RS, Hollistski S, Dimnu I. A prospective outcome study on the effects of facet joint radiofrequency denervation on pain, analgesic intake, disability, satisfaction, cost, and employment. Arch Phys Med Rehabil. 2009;90:201–5.
- 190. Boswell MV, Colson JD, Sehgal N, Dunbar EE, Epter R. A systematic review of therapeutic facet joint interventions in chronic spinal pain. Pain Physician. 2007;10:229–53.
- 191. Boswell MV, Trescot AM, Datta S, et al. Interventional techniques: evidence-based practice guidelines in the management of chronic spinal pain. Pain Physician. 2007;10:7–111.
- 192. Manchikanti L, Singh V, Vilims BD, Hansen HC, Schultz DM, Kloth DS. Medial branch neurotomy in management of chronic spinal pain: systematic review of the evidence. Pain Physician. 2002;5:405–18.
- 193. Manchikanti L, Boswell MV, Singh V, Derby R, Fellows B, Falco FJ, Datta S, Smith HS, Hirsch JA, ASIPP. Comprehensive review of neurophysiologic basis and diagnostic interventions in managing chronic spinal pain. Pain Physician. 2009;12:E71–120.
- 194. Datta S, Lee M, Falco FJ, Bryce DA, Hayek SM. Systematic assessment of diagnostic accuracy and therapeutic utility of lumbar facet joint interventions. Pain Physician. 2009;12:437–60.
- 195. MacVicar J, Borowczyk JM, MacVicar AM, Loughnan BM, Bogduk N. Lumbar medial branch radiofrequency neurotomy in New Zealand. Pain Med. 2013;14:639–45.
- 196. Evans PJ, Lloyd JW, Jack TM. Cryoanalgesia for intractable perineal pain. J R Soc Med. 1981;74:804–9.
- 197. Myers RR, Powell HC, Heckman HM, et al. Biophysical and pathological effects of cryogenic nerve lesion. Ann Neurol. 1981;10:478–85.
- Trescot AM. Cryoanalgesia in interventional pain management. Pain Physician. 2003;6:345–60.
- 199. Brechner T. Percutaneous cryogenic neurolysis of the articular nerve of Luschka. Reg Anesth, Cincinnati, Ohio. 1981;6:18–22.
- Schuster GD. The use of cryoanalgesia in the painful facet syndrome. J Neurol Orthop Surg. 1982;4:271–4.
- 201. Ross EL. Cryoneurolysis of lumbar facet joints for the treatment of chronic lower back pain. Presented at the American Society of Regional Anesthesia, Cincinnati, Ohio, 6 Apr 1991. Reg Anesth 1991;16 Suppl 1:1–91.
- 202. Manchikanti L, Abdi S, Atluri S, Benyamin RM, Boswell MV, Buenaventura RM, Bryce DA, Burks PA, Caraway DL, Calodney AK, Cash KA, Christo PJ, Cohen SP, Colson J, Conn A, Cordner H, Coubarous S, Datta S, Deer TR, Diwan S, Falco FJ, Fellows B, Geffert S, Grider JS, Gupta S, Hameed H, Hameed M, Hansen H, Helm 2nd S, Janata JW, Justiz R, Kaye AD, Lee M, Manchikanti KN,

McManus CD, Onyewu O, Parr AT, Patel VB, Racz GB, Sehgal N, Sharma ML, Simopoulos TT, Singh V, Smith HS, Snook LT, Swicegood JR, Vallejo R, Ward SP, Wargo BW, Zhu J, Hirsch JA. An update of comprehensive evidence-based guidelines for interventional techniques in chronic spinal pain. Part II: guidance and recommendations. Pain Physician. 2013;16:S49–283.

- 203. Civelek E, Cansever T, Kabatas S, Kircelli A, Yilmaz C, Musluman M, Ofluoglu D, Caner H. Comparison of effectiveness of facet joint injection and radiofrequency denervation in chronic low back pain. Turk Neurosurg. 2012;22:200–6.
- 204. Manchikanti L, Singh V, Falco FJ, Cash KA, Pampati V. Lumbar facet joint nerve blocks in managing chronic facet joint pain: one-year follow-up of a randomized, double-blind controlled trial. Pain Physician. 2008;11:121–32.
- 205. Manchikanti L, Boswell MV, Singh V, Benyamin RM, Fellows B, Abdi S, Buenaventura RM, Conn A, Datta S, Derby R, Falco FJE, Erhart S, Diwan S, Hayek SM, Helm S, Parr AT, Schultz DM, Smith HS, Wolfer LR, Hirsch JA. Comprehensive evidence-based guidelines for interventional techniques in the management of chronic spinal pain. Pain Physician. 2009;12:699–802.
- 206. Bogduk N. Evidence-informed management of chronic low back pain with facet injections and radiofrequency neurotomy. Spine J. 2008;8:56–64.
- 207. Lutz GE, Shen TC. Fluoroscopically guided aspiration of a symptomatic lumbar zygapophyseal joint cyst: a case report. Arch Phys Med Rehabil. 2002;83:1789–91.
- Allen TL, Tatli Y, Lutz GE. Fluoroscopic percutaneous lumbar zygapophyseal joint cyst rupture: a clinical outcome study. Spine J. 2009;9:387–95.
- 209. Martha JF, Swaim B, Wang DA, Kim DH, Hill J, Bode R, Schwartz CE. Outcome of percutaneous rupture of lumbar synovial cysts: a case series of 101 patients. Spine J. 2009;9:899–904.
- 210. Schmid G, Schmitz A, Borchardt D, Ewen K, von Rothenburg T, Koester O, Jergas M. Effective dose of CT- and fluoroscopy-guided perineural/epidural injections of the lumbar spine: a comparative study. Cardiovasc Intervent Radiol. 2006;29:84–91.
- 211. Kennedy DJ, Dreyfuss P, Aprill CN, Bogduk N. Paraplegia following image-guided transforaminal lumbar spine epidural steroid injection: two case reports. Pain Med. 2009;10:1389–94.
- 212. Kranz PG, Raduazo P, Gray L, Kilani RK, Hoang JK. CT fluoroscopy- guided cervical interlaminar steroid injections: safety, technique, and radiation dose parameters. AJNR Am J Neuroradiol. 2012;33:1222–4.
- 213. Kranz PG, Raduazo PA. Technique for CT fluoroscopy-guided cervical interlaminar steroid injections. AJR Am J Roentgenol. 2012;198:675–7.
- Wagner AL. Selective lumbar nerve root blocks with CT fluoroscopic guidance: technique, results, procedure time, and radiation dose. Am J Neuroradiol. 2004;25:1592–4.

- 215. Wald JT, Maus TP, Geske JR, Carter RE, Diehn FE, Kaufmann TJ, Morris JM, Murthy NS, Thielen KR. Safety and efficacy of CT-guided transforaminal cervical epidural steroid injections using a posterior approach. Am J Neuroradiol. 2012;33:415–9.
- 216. Freyhardt P, Hartwig T, De Bucourt M, Maurer M, Renz D, Gebauer B, Hamm B, Teichgräber UK, Streitparth F. MR-guided facet joint injection therapy using an open 1.0-T MRI system: an outcome study. Eur Radiol. 2013;23:3296–303. [Epub ahead of print].
- 217. Greher M, Kirchmair L, Enna B, Kovacs P, Gustorff B, Kapral S, Moriggl B. Ultrasound-guided lumbar facet nerve block: accuracy of a new technique confirmed by computed tomography. Anesthesiology. 2004;101:1195–200.
- Deyo RA, Nachemson A, Mirza SK. Spinal-fusion surgery: the case for restraint. N Engl J Med. 2004;350:722–6.
- Gibson JN, Waddell G, Grant IC. Surgery for degenerative lumbar spondylosis. Cochrane Database Syst Rev. 2000;3:CD001352.
- 220. Jackson RP. The facet syndrome: myth or reality? Clin Orthop Relat Res. 1992;279:110–21.
- 221. Esses SI, Botsford DJ, Kostuik JP. The role of external spinal skeletal fixation in the assessment of lowback disorders. Spine. 1989;14:594–601.
- 222. Esses SI, Moro JK. The value of facet joint blocks in patient selection for lumbar fusion. Spine. 1993;18:185–90, 265.
- 223. Lovely TJ, Rastogi P. The value of provocative facet blocking as a predictor of success in lumbar spine fusion. J Spinal Disord. 1997;10:512–7.
- 224. Bough B, Thakore J, Davies M, Dowling F. Degeneration of the lumbar facet joints: arthrography and pathology. J Bone Joint Surg (Br). 1990;72:275–6.
- 225. Schwarzer AC, Wang SC, Bogduk N, McNaught PJ, Laurent R. Prevalence and clinical features of lum-

bar zygapophysial joint pain: a study in an Australian population with chronic low back pain. Ann Rheum Dis. 1995;54:100–6.

- 226. Laplante BL, Ketchum JM, Saullo TR, DePalma MJ. Multivariable analysis of the relationship between pain referral patterns and the source of chronic low back pain. Pain Physician. 2012;15:171–8.
- Klessinger S. In response: does the diagnosis of spondylolisthesis matter? Pain Physician. 2012;15:E158.
- 228. Chan CW, Peng P. Failed back surgery syndrome. Pain Med. 2011;12:577–606.
- 229. Frey ME, Manchikanti L, Benyamin RM, Schultz DM, Smith HS, Cohen SP. Spinal cord stimulation for patients with failed back surgery syndrome: a systematic review. Pain Physician. 2009;12:379–97.
- 230. Fritsch EW, Heisel J, Rupp S. The failed back surgery syndrome: reasons, intraoperative findings, and long-term results: a report of 182 operative treatments. Spine. 1996;21:626–33.
- 231. Manchikanti L, Singh V, Cash KA, Pampati V, Datta S. Management of pain of post lumbar surgery syndrome: one-year results of a randomized, doubleblind, active controlled trial of fluoroscopic caudal epidural injections. Pain Physician. 2010;13:509–21.
- 232. Manchikanti L, Singh V, Cash KA, Pampati V, Datta S. Preliminary results of a randomized, equivalence trial of fluoroscopic caudal epidural injections in managing chronic low back pain: part 3 – post surgery syndrome. Pain Physician. 2008;11:817–31.
- 233. Manchikanti L, Manchukonda R, Pampati V, Damron KS, McManus CD. Prevalence of facet joint pain in chronic low back pain in postsurgical patients by controlled comparative local anesthetic blocks. Arch Phys Med Rehabil. 2007;88:449–55.
- 234. Burton CV, Kirkaldy-Willis WH, Yong-Hing K, Heithoff KB. Causes of failure of surgery on the lumbar spine. Clin Orthop Relat Res. 1981;157:191–9.

Part IV

**Lumbar Disk Herniations** 

# Advanced Scientific Considerations for Surgery in Patients with Lumbar Disk Herniation

Wilco C.H. Jacobs and Wilco C. Peul

# 17.1 Lumbar Disk Herniation

Typically, patients present to their general physician (GP) with complaints of radiating low back pain, which they may or may not have been experiencing for a while. In first instance, in absence of emergency surgical indications, conservative care is prescribed. With persistent complaints, the GP refers to a second-line physician, such as a neurologist, who calls for an MRI. When the MRI shows a concordant disk herniation or nerve root compression, referral to a neurosurgeon often results in disk surgery.

Considerable practice variation exists [1] in the occurrence of spine surgery across geographic locations. The probable cause for this practice variation is inconsistency in the referral patterns as well as differences in treatment decisions. If patient history and diagnostic assessments do not lead to treatment choice in a

Medical Center Haaglanden, The Hague, The Netherlands e-mail: w.c.peul@lumc.nl reliable, reproducible manner, choice for treatment is care provider dependent, while it should be patient dependent, based on disease characteristics and patient preferences. Standardization of these pathways and diagnostic assessments are needed, and once established in guidelines, these should be followed to improve quality of care.

Another aspect of the diagnostic pathway is the utilization of second-line, in-hospital care. The journey through the health-care milieu by patients with disk herniation results in additional diagnostic assessments, leading to extra cost and interventions. Alternatively, a wait and see policy can often perfectly be maintained by the GP and can result in satisfactory, comparable, outcomes in many patients. The spine community should make it a priority that patients are adequately equipped and educated to make a balanced decision. Patients are best informed in the primary care environment, including first-line GPs, physiotherapists, but also second-line neurologists and neurosurgeons. Modern e-health applications can help in this regard.

# 17.2 Diagnosis

In primary care, patient history and physical examination are the key aspects for diagnosis of the cause of sciatica, either defined as disk herniation or nerve root compression. In this process,

© Springer-Verlag Berlin Heidelberg 2016 J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_17 17

<sup>W.C.H. Jacobs, PhD (⊠)
Department of Neurosurgery,
Leiden University Medical Center,
PO Box 9600, 2300, RC, Leiden, The Netherlands
e-mail: wch.jacobs@gmail.com</sup> 

W.C. Peul, MD, PhDDepartment of Neurosurgery,Leiden University Medical Center,PO Box 9600, 2300, RC, Leiden, The Netherlands

these compressive causes of sciatica should be distinguished from other causes such as inflammatory irritation of the nerve root, lumbar stenosis, or malignant causes. Either with primary or secondary care, the following considerations are important before a decision for surgery should be considered.

#### 17.2.1 Relevance of History Taking

In a systematic review, "male gender", "pain worse in the leg than in the back", and "nonsudden onset" have been shown to possess prognostic value for the presence of nerve root compression [2]. For the presence of a disk herniation, relevant factors were "BMI<30", "non-sudden onset", and "sensory loss". This information should be considered before requesting further diagnostic tests such as MRI.

## 17.2.2 Relevance of Clinical Assessments

Physical tests to diagnose disk herniation include usually reflex testing, sensory test, and motor strength tests. These components of the neurological examination, either individually or in combination, are associated with suboptimal clinically relevant accuracy in identifying the level of herniation on an MRI [3]. Physical and neurological examination are especially poor in isolation [4], although this was only studied in surgical populations. Further research is necessary regarding value of combined tests in primary care.

The optimal window for surgical intervention for disk herniation is thought to be between 6 and 8 weeks and 9 months. As a consequence, referral to second-line neurologist, with possibilities for diagnostic imaging such as MRI, is not indicated before 6–8 weeks [5]. Waiting too long might predispose patients for chronic complaints and should also be avoided [6], one of the recommendations from the Choosing Wisely Netherlands campaign (Table 17.1).

**Table 17.1** Recommendations from the Choosing Wisely Netherlands campaign

Recommendation
Do not operate in case of isolated low back pain
Do not operate with short existence of leg pain (<6 weeks), but also do not wait too long (>9 months)
Do not call for an MRI during natural recovery or after disk surgery
Do not operate when the compressed nerve does not match the affected dermatome
Only apply new surgical techniques and implants in context of research

# 17.2.3 Interpretation and Consequences of MRI

In patients that are referred to a neurologist for suspicion of a herniated disk, an MRI is a commonly used tool to identify the probability of a disk herniation. MRI was found to be reliable in diagnosing the affected disk levels, the affected nerve roots, and the probability of nerve root compression [7]. However, most MRI findings have a poor correlation with clinical symptoms or outcome [8]. In a conservative group of patients from a randomized trial, 55 patients received delayed surgery due to persistent complaints [9]. MRI at baseline did not predict which of the conservatively treated patients eventually needed delayed surgery [10].

In case of persisting complaints after surgical or conservative treatment of sciatica, a follow-up MRI did not distinguish between favourable outcome and unfavourable outcome [11]. From an economical point of view, it is thus not advised to perform a repeat MRI when confronted with a patient that has persisting complaints after 1 year. This also was adopted by the Choosing Wisely Netherlands campaign (Table 17.1).

To distinguish recurrent herniated disk from scar tissue, contrast-enhanced MRI with gadolinium is often used. This was found to possess slight to fair agreement between observers to identify enhancement. Further there was also no correlation between enhancement and clinical outcome 1 year after the initial surgery [12]. It should be noted that reliability applies to the interpretation of an existing MRI, not for a repeated MRI, which might even show lower reliability.

In conclusion, MRI appears to be mainly useful in identifying and confirming the affected level. For other applications of MRI findings, we certainly need more research.

## 17.2.4 Prognostic Factors

Little is known about which factors have prognostic value in patients with lumbar disk herniation. In patients indicated for conservative interventions, a recent review [13] only identified leg pain intensity as a prognostic factor for subsequent surgery. Age, body mass index, smoking, and sensory disturbance did not possess prognostic abilities for outcome. Especially when we want to assess the value of published literature, we need to know which factors influence outcome of patients with sciatica at different stages of the disease. More research needs to be done in this area.

## 17.3 Decision for Surgery

# 17.3.1 Evidence for Effectiveness of Surgery Compared with Conservative Interventions

There are five trials [9, 14-17] that compare surgical with conservative interventions, either being conservative management (three trials), prolonged conservative management with optional surgery (one trial), or steroid injections (one trial). Unfortunately, these trials are heterogeneous regarding interventions, some have a high risk of bias, and some suffer from poor reporting which prohibits quantitative analysis [18]. The general conclusion after evaluation of these trials does not support either surgery or nonoperative intervention [18]. However, surgery appeared to result in faster recovery in at least two of the trials. This implies that the choice for surgery over continuation of conservative care comprises balancing fast recovery against surgical risks, cost, and burden.

The three studies [14-16] that compared surgery with conservative management showed conflicting results. One older, high risk of bias study including 126 patients [14] demonstrated that diskectomy was significantly better regarding patient and observer ratings than conservative treatment at 1 year. Twenty-four of the 66 patients (36 %) in the conservative care group versus 39 of the 60 patients (65 %) in the surgery group reported a good outcome. This difference disappeared after 4 and 10 years. One high risk of bias trial including 56 patients [15] found no significant differences for leg pain or back pain and subjective disability throughout the 2 years of follow-up. VAS leg pain scores, however, improved more rapidly in the diskectomy group; 6-week score in the surgery group was 12 (SD 20) versus 25 (SD 27) in the conservative group. The per-protocol analysis demonstrated no statistically significant differences. A large low risk of bias trial including 501 patients showed that both the surgery and the conservative treatment group improved substantially over 2 years for all primary and secondary outcome measures [16]. The intention-to-treat analysis showed no statistically significant differences for any of the primary outcome measures. There was considerable crossover: 50 % of the patients randomized to surgery and 30 % of the patients randomized to conservative treatment. After 2 years this was 45 and 40 %.

The sciatica trial randomized 283 patients with severe sciatica for 6–12 weeks to early surgery or prolonged conservative treatment followed by surgery if needed [9]. In this study, crossover was anticipated and offered surgery to patients that did not improve after 6 months. After 2 weeks, 89 % of patients randomized to early surgery underwent micro-diskectomy, while 39 % of patients randomized to conservative treatment underwent surgery after a mean of 19 weeks. Relief of leg pain was faster for patients assigned to early surgery. Intention-totreat analysis showed statistically significant more leg pain relief was found in favour of early surgery compared with prolonged conservative care at 3 months (MD -17.70, 95 % CI -23.1 to -12.3). There was no significant overall difference between the two groups regarding disability scores during the first year. The median time to recovery was 4.0 weeks (95 % confidence interval (CI), 3.7-4.4) for early surgery and 12.1 weeks (95 % CI, 9.5-14.9) for prolonged conservative treatment. During the first year, early surgery achieved a faster rate of perceived recovery with a hazard ratio of 1.97 (95 % CI 1.72–2.22, P<0.001). At 1 year of follow-up, however, 95 % of patients in both treatment groups had experienced satisfactory recovery, and no subsequent differences were found. This lack of a difference between groups was maintained after 2 years and also after 5 years [19]. The same pattern was found for the subset of 150 patients with a motor deficit [20]. Motor deficit recovered significantly faster in patients randomized to early surgery, but there were no differences found after 1 year.

One high risk of bias trial including 100 patients compared micro-diskectomy with epidural steroid injection [17]. Patients undergoing diskectomy had the most rapid decrease in their symptoms. The decrease in leg pain in the diskectomy group was significantly greater than epidural steroid injection group at 3- and 6-month follow-up intervals, but not beyond 1 year. There were no significant differences between groups for back pain throughout the follow-up. Twentyseven of 50 patients receiving a steroid injection had a subsequent micro-diskectomy. Outcomes in this crossover group were similar to those of the surgery group.

# 17.3.2 How Long to Wait Before Indicating Surgery

The current status of evidence does not support a definite choice for conservative or surgery, at least for the indications that were studied in these trials. The general consensus from guidelines is to wait for at least 6–8 weeks before considering surgery [5]. Also, some of the choosing wisely initiatives provide definitive advice not to proceed with surgery too early but also not to wait too long (more than 9 months) [21]. Between these margins, choice for either surgery or conservative intervention should be based on preferences of wellinformed patients (Fig. 17.1). Only if the patients have information about the advantages and disadvantages of surgery on the short term and the equivalent outlook of both interventions on the longer term, an informed choice can be made [9]. Decision tools can guide these decisions [22].

## 17.4 Surgical Techniques

Once the need and preference for disk surgery are established, preoperative planning can begin and the choice for surgical approach can be made. Several techniques are available and they differ in invasiveness, approach, extent of disk resection, and use of co-interventions such as preventive measures of scar tissue.

## 17.4.1 Evidence for Effectiveness for Different Surgical Techniques

The most common type of surgery is microscopic diskectomy, which is defined as the surgical removal of part of the disk, performed with the use of an operating microscope or other magnifying tools. Most studies refer to Caspar [23], Yasargil [24], and Williams [25] when diskectomy is performed with microscope and to Foley and Smith [26] or Greiner-Perth et al. [27] when diskectomy is performed with tubular, muscle-splitting, retractor systems and endoscope. However, some have returned to using a microscope while retaining the less invasive muscle-splitting approach of Foley and Smith [26]. The result is an array of surgical approaches for which it is difficult to acquire sufficient evidence from randomized trials comparing all techniques.



#### Weeks of leg pain

**Fig. 17.1** The optimal window for surgery for lumbar disk herniation. Before 6–8 weeks and after 9 months (36 weeks) can be regarded inappropriate care. Note that

complaints of over 9 months are not a contraindication for surgery but that one should have treated the majority of these patients earlier

# 17.4.2 Open Versus Microscopic Diskectomy

There are eight trials that have compared open diskectomy with minimal invasive techniques, including microscopic diskectomy, video-assisted microscopic diskectomy, automated percutaneous micro-diskectomy, or micro-endoscopic diskectomy.

Six trials compared the classical open diskectomy, also called standard diskectomy or macrodiskectomy, with microscopic diskectomy [28–33]. There is a consistent finding in these studies that microscopic diskectomy leads to an increased operating time with a pooled effect of 12 min (95 % CI 2.20–22.3; p=0.02; moderate quality of evidence). No differences were found for length of stay, which was only reported in five studies with a total of 452 patients. The mean difference was 0.18 days in favour of open diskectomy (95 % CI -0.09 to +0.45 days; p=0.47; moderate quality of evidence). Blood loss was

reported in two studies; in one study with 119 patients, microscopic diskectomy resulted in less blood loss [31], while in the other study with 60 patients, there was no difference [28]. The quality of evidence for blood loss was "very low". The length of incision was reported in three studies with together 353 patients and found to be shorter for microscopic diskectomy in two studies [30, 33]. The quality of evidence for incision was "low". Leg pain was reported in four studies with together 453 patients and was significantly less for microscopic diskectomy by 2.01 mm (95 % CI 0.57–3.44; *p*=0.006; moderate quality of evidence), while this can hardly be regarded as a clinical relevant difference. Further outcomes (pain, return to work) were found to be comparable, except for a higher return to work at 4 weeks for microscopic diskectomy [33] in one study with 114 patients where two other studies with together 140 patients found no difference at 10.4 weeks [28] and 14.9 months [29]. It should

be noted that all but one of these trials was associated with a high risk of bias.

Two trials compared open diskectomy with micro-endoscopic diskectomy [32, 34]. Huang et al. [34] reported results of a very small, high risk of bias, trial with only 22 patients. The microendoscopic diskectomy group had shorter postoperative hospital stay and less intraoperative blood loss compared with the open diskectomy group, but duration of the operation was longer. There were no differences in pain severity and MacNab criteria between the groups. Teli et al. [32] showed in a larger trial including 220 patients that the micro-endoscopic group compared to open and microscopic diskectomy suffered more dural tears (7 %, 3 %, 3 %, respectively), root injuries (3 %, 0%, 0%, respectively), and a recurrent herniation (7 %, 4 %, 3 %, respectively).

One low risk of bias trial with 60 patients found that patients who had received videoassisted arthroscopic micro-diskectomy had similar satisfactory outcomes compared with open laminotomy and diskectomy, but patients who had had an arthroscopic micro-diskectomy had a shorter duration of postoperative disability and used narcotics for a shorter period [35].

## 17.4.3 Different Minimally Invasive Techniques

There is evidence on the comparative effectiveness of the different minimal invasive techniques for diskectomy such as endoscopic diskectomy, video-assisted diskectomy, percutaneous transforaminal diskectomy, etc.

Eight trials with an accumulative 1047 patients evaluated different approaches for less invasive diskectomy, such as micro-endoscopic diskectomy, tubular microscopic diskectomy, microscopic-assisted percutaneous nucleotomy, minimal access trocar/microsurgical microdiskectomy, percutaneous endoscopic diskectomy, or sequestrectomy. We analysed the comparisons between these techniques, keeping the differences muscle damage and differences in use of microscope or endoscope in mind. The results of these trials are given in Table 17.1.

Seven (six high risk of bias) trials with 923 patients compared tubular diskectomy with conventional microscopic diskectomy [32, 36-41]. Of these, four used an endoscope [32, 36, 37, 39]. There was low to moderate quality of evidence for incision length and this was consistently shorter for tubular diskectomy in all three studies (n=260)that reported this outcome [32, 36, 39]. However, results could not be pooled due to sparse data on variation (SD). The quality of evidence for the remaining outcome parameters was "low' to "very low", so no further meta-analyses could be performed. Inconsistent results were found for operative morbidity. Two studies (n=368) of the six studies (n=718) reporting operative time found a longer duration for tubular diskectomy [32, 36], while one study (n=100) found a shorter duration [38]. No differences were found for blood loss in three studies. Length of stay was longer (2 h) for conventional microscopic diskectomy in only one of four studies [36]. One study found a faster improvement in pain scores for tubular diskectomy before discharge [37], while the only low risk of bias study found a slightly better pain score for conventional diskectomy at 2 years [41]. All other outcomes for pain as measured with VAS, for Oswestry or Roland-Morris score, or for SF36 scores were not significantly different between the two surgical techniques. For Shin et al. [37], baseline values for back pain were not comparable. In one trial, the postoperative analgesic consumption was significantly less in the tubular diskectomy group [40].

One high risk of bias trial [42] with 40 patients compared percutaneous endoscopic diskectomy (cannula inserted into the central disk) with microscopic diskectomy. This trial showed comparable clinical outcomes after the two procedures but contained a small sample size.

## 17.4.4 Techniques to Prevent Scarring

Evidence regarding techniques that are applied for the prevention of scar tissue is relatively sparse. Recent trials of an interposition gel covering the dural sheet, fat, preservation of the ligamentum flavum, and use of a drain show promising effects in reducing epidural scar formation, but no effect on clinical outcomes.

Thirteen studies considered the effect of different techniques to prevent formation of intraspinal scarring following diskectomy, as assessed by magnetic resonance imaging or enhanced computerized tomography. Ten studies evaluated the use of an interposition membrane. The types of membrane used are autologous free fat graft or commercially available gels. The results of these trials are given in Table 17.1.

Four high risk of bias studies compared the use of fat graft versus no fat graft [43–46]. These studies failed to show any improvement in clinical outcomes following use of fat. Three studies evaluated fibrous tissue formation on CT or MRI, two found a decrease for fat graft [44, 46], and one small subsample of MacKay et al. [43] found no difference. The pooled effect with a moderate quality of evidence yielded a significant decrease in scar tissue for fat graft (OR 0.22 (95 % CI 0.08–0.62). One study reported a lesser number of painful episodes 1 year after surgery [46], but this was evaluated by the surgeon.

The synthetic gels in the studies consist of bioresorbable carbohydrate polymer gels (Adcon-L, derived from porcine collagen and dextran sulphate), polytetrafluoroethylene (Prelude), polyethylene oxide with carboxymethylcellulose (Oxiplex/SP), or polyethylene glycol (DuraSeal Xact). There are three trials of ADCON-L [47-49]. Two of these studies [47, 48] show conflicting results. Twelve-month results are reported from a pilot study of Oxiplex/ SP gel [50]. Although there is a trend suggesting that treatment diminishes leg pain severity and lower limb weakness, the study was very small and the results reported are not statistically significant. Fransen et al. [51] compared DuraSeal Xact with no gel in a small, double-blinded low risk of bias study. One small, high risk of bias, study [52] did not find a difference on any of the clinical outcomes or on scar volume between ADCON-L and Preclude Spinal Membrane (PSM). The third arm of MacKay et al. [43] included gelfoam and also found no differences for this additional group.

Three studies evaluated the effect of other approaches, namely, the use of antibiotics, drains, or preservation of ligamentum flavum. One low risk of bias study assessed the effect of locally applied mitomycin C on peridural fibrosis during micro-diskectomy lumbar [53] (N=60).Mitomycin C is isolated from Streptomyces caespitosus and is purported to suppress fibroblast proliferation. At a median follow-up of 18 months, there were no differences between the group with or without mitomycin C on postoperative evaluation of the MR images, pain scores, and neurological function. One small high risk of bias study compared micro-diskectomy with preservation of the ligamentum flavum with usual micro-diskectomy [54]. There were no differences in clinical outcomes (pain, functional status, and straight leg raising), but the group with preserved ligamentum flavum had less epidural fibrosis as assessed with MRI at 6-month followup compared with the usual micro-diskectomy group. However, this study had an inadequate randomization procedure, was not blinded, and only included 20 patients. One high risk of bias study compared micro-diskectomy with a drain aimed at reducing postoperative epidural hematoma with usual micro-diskectomy [55]. The study found no differences in clinical outcomes after 6 months. The group with a drain had statistically significant less postoperative epidural hematoma. After 6 months there was less epidural fibrosis in the drain group, but this difference was not statistically significant. The study had an inadequate method of randomization.

## 17.4.5 Considerations on the Extent of Disk Removal

One low risk of bias trial [56] with 84 patients compared clinical outcomes and recurrence rates after sequestrectomy (removal of only the sequestration while leaving the remaining disk intact) and standard micro-diskectomy (removing the herniated material and resection of disk tissue from the intervertebral space). There were no statistically significant differences in back and leg pain and quality of life up to 2 years of follow-up [57].

## 17.5 Methodological Considerations

The evaluation of surgical interventions for sciatica suffers from a number of challenges. Surgical performance is likely to be surgeon dependent, the timing of surgery differs, there is a big placebo effect from surgery, the surgical evolution of techniques and medical devices is highly technology driven, and relevant follow-up assessments are in the very far future. Especially comparing conservative and surgery is difficult, as conservative intervention would normally imply continuation of an already ineffective treatment from the patient point of view.

# 17.5.1 Randomized Trials

There are apparent methodological problems in randomized trials in comparing surgery with nonsurgical interventions. First, there is considerable crossover in all of these trials. An intentionto-treat analysis is confounded by this crossover, but also by re-interventions. The intention-totreat analysis should be interpreted as a treatment strategy analysis, except that there were no reallife treatment strategies compared as interventions are likely to have been applied different from usual care in the context of the trial. As an alternative, the per-protocol analysis will not provide insight in patient outcomes for a specific choice of treatment strategy, let alone specific interventions. Second, blinding is impossible for patients and care providers and difficult for outcome assessors. Lack of blinding for patients could result in disappointment when no surgical intervention is offered. Third, as many care providers and patients are hesitant to participate in these trials, external validity is compromised. These problems limit the applicability of randomized trials in these comparisons.

Nevertheless, randomized trials still hold value and likely we should make use of all research designs and decide for the best design for each question at hand. Randomized trials are mainly applicable, within spinal surgery, for comparisons between surgical techniques.

## 17.5.2 Alternatives

Alternatives to randomized trials include the use of the opt-out strategy, where patients are included in an experimental study, unless they object. This has however some serious ethical problems. Another approach is the use of clinical equipoise, where a patient is included in an observational study when a team of care providers decides (independently) that there is no consensus about the most appropriate treatment [58].

#### 17.5.3 Spine Registries

The evolving spine registries will provide an opportunity for additional analyses, which will be more representative of usual care, with a higher patient participation and better representativeness. The best current examples of such registries are in the Scandinavian countries (SweSpine in Sweden, NorSpine in Norway, and DaneSpine in Denmark), as well as the Netherlands, the United States, Spain, and Switzerland. New initiatives are to be expected in Canada, Australia, Russia, Turkey, and Singapore.

Spine Tango from the Spine Society of Europe serves individual clinics around the world. This approach could result in a selected inclusion of patients when participating clinics differ from the average clinic in a country. On the other hand, also the national registries are not yet obligatory in all countries.

Most of these registries are initiated for surgical interventions, while others focus on conservative interventions such as the Spanish registry. For a comprehensive view on the patient with lumbar disk herniation, we will need to assess the patients through the whole pathway of care, from first-line general practitioner, to second-line neurologist, and if applicable to neurosurgeon. Only then we can evaluate which patients need which treatments at which point in their disease.

A recent important step has been the global consensus on the outcome parameters that need to be assessed in these registries and assessment timing [59], in order to facilitate global benchmark and comparisons between interventions. A

global group of spine specialists agreed on using the Oswestry Disability Index (version 2.1a), NRS for leg and back pain, and EQ5D as outcome measures along with a series of other measures for assessing complications, reoperation, need for medication, and work status. The existing registries as well as the new registries will hopefully adopt this guidance.

The identification and assessment of relevant prognostic factors are essential for a valid analysis of observational data. This is necessary for proper case mix adjustment of other advanced analysis methods such as propensity score modelling. Most registries now collect age, gender, marital status, duration of complaints, baseline pain and function scores, analgesic use, work absenteeism, and educational level. For example, the patient selection in the Spine Tango registry should be evaluated for these variables. Additional research is still needed to identify the relevant prognostic factors.

### References

- Deyo RA, Mirza SK. Trends and variations in the use of spine surgery. Clin Orthop Relat Res. 2006;443:139–46.
- Verwoerd AJ, Peul WC, Willemsen SP, Koes BW, Vleggeert-Lankamp CL, El Barzouhi A, et al. Diagnostic accuracy of history taking to assess lumbosacral nerve root compression. Spine J. 2014; 14(9):2028–37.
- Hancock MJ, Koes B, Ostelo R, Peul W. Diagnostic accuracy of the clinical examination in identifying the level of herniation in patients with sciatica. Spine (Phila Pa 1976). 2011;36(11):E712–9.
- 4. van der Windt DA, Simons E, Riphagen II, Ammendolia C, Verhagen AP, Laslett M, et al. Physical examination for lumbar radiculopathy due to disc herniation in patients with low-back pain. Cochrane Database Syst Rev. 2010;2:CD007431.
- Koes BW, van Tulder MW, Peul WC. Diagnosis and treatment of sciatica. BMJ. 2007;334(7607):1313–7.
- Orde van Medisch Specialisten, ZonMw. Choosing wisely Netherlands Campaign. 2014.
- El Barzouhi A, Vleggeert-Lankamp CL, Lycklama ANG, Van der Kallen BF, van den Hout WB, Verwoerd AJ, et al. Magnetic resonance imaging interpretation in patients with sciatica who are potential candidates for lumbar disc surgery. PLoS One. 2013;8(7):e68411.
- El Barzouhi A, Vleggeert-Lankamp CL, Nijeholt GJ, Van der Kallen BF, van den Hout WB, Koes BW, et al.

Influence of low back pain and prognostic value of MRI in sciatica patients in relation to back pain. PLoS One. 2014;9(3):e90800.

- Peul WC, Van Houwelingen HC, van den Hout WB, Brand R, Eekhof JA, Tans JT, et al. Surgery versus prolonged conservative treatment for sciatica. N Engl J Med. 2007;356(22):2245–56.
- El Barzouhi A, Vleggeert-Lankamp CL, Nijeholt GJ, Van der Kallen BF, van den Hout WB, Koes BW, et al. Predictive value of MRI in decision making for disc surgery for sciatica. J Neurosurg Spine. 2013;19(6):678–87.
- El Barzouhi A, Vleggeert-Lankamp CL, Nijeholt GJ, Van der Kallen BF, van den Hout WB, Jacobs WC, et al. Magnetic resonance imaging in follow-up assessment of sciatica. N Engl J Med. 2013;368(11): 999–1007.
- 12. El Barzouhi A, Vleggeert-Lankamp CL, Lycklama ANG, Van der Kallen BF, van den Hout WB, Koes BW, et al. Reliability of gadolinium-enhanced magnetic resonance imaging findings and their correlation with clinical outcome in patients with sciatica. Spine J. 2014;14(11):2598–607.
- Verwoerd AJ, Luijsterburg PA, Lin CW, Jacobs WC, Koes BW, Verhagen AP. Systematic review of prognostic factors predicting outcome in nonsurgically treated patients with sciatica. Eur J Pain. 2013;17(8):1126–37.
- Weber H. Lumbar disc herniation. A controlled, prospective study with ten years of observation. Spine (Phila Pa 1976). 1983;8(2):131–40.
- Osterman H, Seitsalo S, Karppinen J, Malmivaara A. Effectiveness of microdiscectomy for lumbar disc herniation: a randomized controlled trial with 2 years of follow-up. Spine (Phila Pa 1976). 2006;31(21):2409–14.
- Weinstein JN, Tosteson TD, Lurie JD, Tosteson AN, Hanscom B, Skinner JS, et al. Surgical vs nonoperative treatment for lumbar disk herniation: the Spine Patient Outcomes Research Trial (SPORT): a randomized trial. JAMA. 2006;296(20):2441–50.
- Buttermann GR. Treatment of lumbar disc herniation: epidural steroid injection compared with discectomy. A prospective, randomized study. J Bone Joint Surg Am. 2004;86-A(4):670–9.
- Jacobs WC, van Tulder M, Arts M, Rubinstein SM, van Middelkoop M, Ostelo R, et al. Surgery versus conservative management of sciatica due to a lumbar herniated disc: a systematic review. Eur Spine J. 2011;20(4):513–22.
- Lequin MB, Verbaan D, Jacobs WC, Brand R, Bouma GJ, Vandertop WP, et al. Surgery versus prolonged conservative treatment for sciatica: 5-year results of a randomised controlled trial. BMJ Open. 2013;3(5).
- Overdevest GM, Vleggeert-Lankamp CL, Jacobs WC, Brand R, Koes BW, Peul WC. Recovery of motor deficit accompanying sciatica-subgroup analysis of a randomized controlled trial. Spine J. 2014; 14(9):1817–24.

- Orde van Medisch Specialisten, Zorgonderzoek Nederland – Medische Wetenschappen. Choosing wisely Netherlands Campaign. 2014.
- LUMC. Audiovisuele keuzehulp voor patiënten met een hernia in de onderrug (LRS). Audiovisual decision aids for patients with a herniated disc in the lower back 2012. www.youtube.com/watch?v=UkFIR42Rzy0.
- Caspar W. A new surgical procedure for lumbar disc herniation causing less damage through a microsurgical approach. In: Wullenweber R, Brocck M, Hamer J, editors. Advances in neurosurgery. Berlin: Springer; 1977. p. 74–7.
- Yasargil M. Microsurgical operation of the herniated lumbar disc. In: Wullenweber R, Brocck M, Hamer J, editors. Advances in neurosurgery. Berlin: Springer; 1977. p. 81–4.
- Williams RW. Microlumbar discectomy: a conservative surgical approach to the virgin herniated lumbar disc. Spine (Phila Pa 1976). 1978;3(2):175–82.
- Foley K, Smith MM. Microendoscopic discectomy. Tech Neurosurg. 1997;3:301–7.
- 27. Greiner-Perth R, Bohm H, ElSaghir H, El Ghait A. The microscopic assisted percutaneous approach to posterior spine – a new minimally invasive procedure for treatment of spinal processes. Zentralbl Neurochir. 2002;63(1):7–11.
- Tullberg T, Isacson J, Weidenhielm L. Does microscopic removal of lumbar disc herniation lead to better results than the standard procedure? Results of a one-year randomized study. Spine (Phila Pa 1976). 1993;18(1):24–7.
- Lagarrigue J, Chaynes P. Comparative study of disk surgery with or without microscopy. A prospective study of 80 cases. Neurochirurgie. 1994;40(2):116–20.
- Henriksen L, Schmidt K, Eskesen V, Jantzen E. A controlled study of microsurgical versus standard lumbar discectomy. Br J Neurosurg. 1996;10(3):289–93.
- 31. Katayama Y, Matsuyama Y, Yoshihara H, Sakai Y, Nakamura H, Nakashima S, et al. Comparison of surgical outcomes between macro discectomy and micro discectomy for lumbar disc herniation: a prospective randomized study with surgery performed by the same spine surgeon. J Spinal Disord Tech. 2006;19(5):344–7.
- 32. Teli M, Lovi A, Brayda-Bruno M, Zagra A, Corriero A, Giudici F, et al. Higher risk of dural tears and recurrent herniation with lumbar micro-endoscopic discectomy. Eur Spine J. 2010;19(3):443–50.
- Tureyen K. One-level one-sided lumbar disc surgery with and without microscopic assistance: 1-year outcome in 114 consecutive patients. J Neurosurg. 2003;99(3 Suppl):247–50.
- Huang TJ, Hsu RW, Li YY, Cheng CC. Less systemic cytokine response in patients following microendoscopic versus open lumbar discectomy. J Orthop Res. 2005;23(2):406–11.
- 35. Hermantin FU, Peters T, Quartararo L, Kambin P. A prospective, randomized study comparing the results of open discectomy with those of video-assisted arthroscopic microdiscectomy. J Bone Joint Surg Am. 1999;81(7):958–65.

- 36. Righesso O, Falavigna A, Avanzi O. Comparison of open discectomy with microendoscopic discectomy in lumbar disc herniations: results of a randomized controlled trial. Neurosurgery. 2007;61(3):545–9.
- Shin DA, Kim KN, Shin HC, Yoon DH. The efficacy of microendoscopic discectomy in reducing iatrogenic muscle injury. J Neurosurg Spine. 2008;8(1): 39–43.
- 38. Franke J, Greiner-Perth R, Boehm H, Mahlfeld K, Grasshoff H, Allam Y, et al. Comparison of a minimally invasive procedure versus standard microscopic discotomy: a prospective randomised controlled clinical trial. Eur Spine J. 2009;18(7):992–1000.
- 39. Ryang YM, Oertel MF, Mayfrank L, Gilsbach JM, Rohde V. Standard open microdiscectomy versus minimal access trocar microdiscectomy: results of a prospective randomized study. Neurosurgery. 2008;62(1):174–81.
- Brock M, Kunkel P, Papavero L. Lumbar microdiscectomy: subperiosteal versus transmuscular approach and influence on the early postoperative analgesic consumption. Eur Spine J. 2008;17(4):518–22.
- Arts MP, Brand R, van den Akker ME, Koes BW, Bartels RH, Peul WC. Tubular diskectomy vs conventional microdiskectomy for sciatica: a randomized controlled trial. JAMA. 2009;302(2):149–58.
- Mayer HM, Brock M. Percutaneous endoscopic discectomy: surgical technique and preliminary results compared to microsurgical discectomy. J Neurosurg. 1993;78(2):216–25.
- 43. MacKay MA, Fischgrund JS, Herkowitz HN, Kurz LT, Hecht B, Schwartz M. The effect of interposition membrane on the outcome of lumbar laminectomy and discectomy. Spine (Phila Pa 1976). 1995;20(16):1793–6.
- 44. Jensen TT, Asmussen K, Berg-Hansen EM, Lauritsen B, Manniche C, Vinterberg H, et al. First-time operation for lumbar disc herniation with or without free fat transplantation. Prospective triple-blind randomized study with reference to clinical factors and enhanced computed tomographic scan 1 year after operation. Spine (Phila Pa 1976). 1996;21(9):1072–6.
- 45. Bernsmann K, Kramer J, Ziozios I, Wehmeier J, Wiese M. Lumbar micro disc surgery with and without autologous fat graft. A prospective randomized trial evaluated with reference to clinical and social factors. Arch Orthop Trauma Surg. 2001;121(8):476–80.
- 46. Gambardella G, Gervasio O, Zaccone C, Puglisi E. Prevention of recurrent radicular pain after lumbar disc surgery: a prospective study. Acta Neurochir Suppl. 2005;92:151–4.
- 47. Richter HP, Kast E, Tomczak R, Besenfelder W, Gaus W. Results of applying ADCON-L gel after lumbar discectomy: the German ADCON-L study. J Neurosurg. 2001;95(2 Suppl):179–89.
- 48. de Tribolet N, Porchet F, Lutz TW, Gratzl O, Brotchi J, van Alphen HA, et al. Clinical assessment of a novel antiadhesion barrier gel: prospective, randomized, multicenter, clinical trial of ADCON-L to inhibit postoperative peridural fibrosis and related symptoms

after lumbar discectomy. Am J Orthop (Belle Mead NJ). 1998;27(2):111–20.

- 49. Ronnberg K, Lind B, Zoega B, Gadeholt-Gothlin G, Halldin K, Gellerstedt M, et al. Peridural scar and its relation to clinical outcome: a randomised study on surgically treated lumbar disc herniation patients. Eur Spine J. 2008;17(12):1714–20.
- 50. Kim KD, Wang JC, Robertson DP, Brodke DS, BenDebba M, Block KM, et al. Reduction of leg pain and lower-extremity weakness for 1 year with Oxiplex/SP gel following laminectomy, laminotomy, and discectomy. Neurosurg Focus. 2004;17(1):ECP1.
- Fransen P. Reduction of postoperative pain after lumbar microdiscectomy with DuraSeal Xact Adhesion Barrier and Sealant System. Spine J: Off J N Am Spine Soc. 2010;10(9):751–61.
- 52. Ivanic GM, Pink PT, Schneider F, Stuecker M, Homann NC, Preidler KW. Prevention of epidural scarring after microdiscectomy: a randomized clinical trial comparing gel and expanded polytetrafluoroethylene membrane. Eur Spine J. 2006;15(9):1360–6.
- Celik SE, Altan T, Celik S, Goksu K, Ince I, Kapran Z. Mitomycin protection of peridural fibrosis in lumbar disc surgery. J Neurosurg Spine. 2008;9(3):243–8.

- 54. Ozer AF, Oktenoglu T, Sasani M, Bozkus H, Canbulat N, Karaarslan E, et al. Preserving the ligamentum flavum in lumbar discectomy: a new technique that prevents scar tissue formation in the first 6 months postsurgery. Neurosurgery. 2006;59(1 Suppl 1):ONS126–33.
- Mirzai H, Eminoglu M, Orguc S. Are drains useful for lumbar disc surgery? A prospective, randomized clinical study. J Spinal Disord Tech. 2006;19(3):171–7.
- Thome C, Barth M, Scharf J, Schmiedek P. Outcome after lumbar sequestrectomy compared with microdiscectomy: a prospective randomized study. J Neurosurg Spine. 2005;2(3):271–8.
- Barth M, Weiss C, Thome C. Two-year outcome after lumbar microdiscectomy versus microscopic sequestrectomy: part 1: evaluation of clinical outcome. Spine (Phila Pa 1976). 2008;33(3):265–72.
- 58. Stadhouder A, Oner FC, Wilson KW, Vaccaro AR, Williamson OD, Verbout AJ, et al. Surgeon equipoise as an inclusion criterion for the evaluation of nonoperative versus operative treatment of thoracolumbar spinal injuries. Spine J. 2008;8(6):975–81.
- 59. Low Back Pain Working Group, ICHOM. The ICHOM standard set for low back pain. 2014.

# Surgery for Lumbar Disk Herniations

Luca Papavero

## 18.1 When to Operate?

As you leave the OR after an apparently successful microdiscectomy, remember that a successful surgical outcome depends 90 % on patient selection and only 10 % on technique. John A. McCulloch [1]

Up to 15-fold variation in regional lumbar diskectomy rates in the United States [2] and lower rates internationally raise questions regarding the appropriateness of some of these surgeries [3, 4]. James N. Weinstein, SPORT [5]

## 18.2 How to Operate?

# 18.2.1 Microsurgical Versus Nonmicrosurgical Surgical Techniques

The spectrum of open surgical treatment of lumbar disk herniations (DHs) ranges from conventional technique without optical magnification to the use of loupes or microscope. Although there is a still ongoing debate about the benefits of the use of microscope for diskectomy for the medium- and long-term outcome, the short-term

L. Papavero

Department of Spinal Surgery, Clinic for Spine Surgery, Schön Klinik Hamburg Eilbek, Hamburg, Germany

e-mail: lpapavero@Schoen-Kliniken.de

advantages include less damage to the paravertebral muscles, decreased blood loss, and reduced postoperative morbidity and by far outweigh the relative disadvantages such as the learning curve [6–9]. In our experience, once this hurdle has been overcome, there is no reason to operate without the aid of a microscope. The surgical techniques described in this chapter are best performed with the aid of the microscope, although they are feasible also with loupes.

# 18.2.2 Fragmentectomy Versus Diskectomy

When symptoms are caused by an extruded disk fragment, the disk space should not be cleared. The removal of disk material from the disk space does not lower the recurrence rate but may increase the postsurgical back pain due to segmental instability [10, 11]. Furthermore, the mean operation time is shorter in the fragmentectomy group and there is no risk of abdominal vascular and visceral injuries [12].

# 18.2.3 Subperiosteal Versus Transmuscular Approach

The subperiosteal approach requires the incision or retraction of the ligamentous insertions of the paravertebral muscles from the spinous processes.

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016 J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_18

The intraoperative injury of the posterior supporting structures of the lumbar spine may lead to increased postoperative back pain [13-15].

The micro-endoscopic diskectomy (MED) was introduced by Foley and Smith [16]. It was the first technique that addressed the shortcomings of the conventional subperiosteal approach. Many investigators have reported that MED is associated with less postoperative pain, a shorter hospital stay, and more rapid return to work [17–19]. However, MED has some limitations related to a small operation field, visualized through a cylindrical tubular retractor [20].

The paraspinal muscle-splitting or "Wiltse" approach along the natural cleavage planes of the paraspinal muscles has shown to cause less damage and retraction of the paraspinal muscles compared to the subperiosteal approach [21]. This leads to decreased back pain and less postoperative analgesic consumption during

the early postoperative period [15]. The choice of the transmuscular approach may be left to the surgeon's preference in patients with fatty degenerated muscles. It is recommended whenever minimizing muscle traumatization becomes an issue.

## 18.2.4 Retractors: Tubular Versus Conventional

The introduction of transmuscular approaches via tubular or miniaturized speculum retractors has prompted the development of miniaturized surgical tools which are sized between the conventional microsurgical instruments and the endoscopic ones. Their design facilitates the intraoperative view of the surgical target area (Fig. 18.1). Surgical times are comparable to open conventional techniques.



**Fig. 18.1** (a) Conventional vs. modified minimally invasive speculum retractor; (b) conventional (*bottom left*) vs. miniaturized rongeurs; (c) miniaturized speculum retractor in situ; (d) the speculum-counter-retractor system can

be docked to a self-holding arm; ( $\mathbf{e}$ ) expandable tubular retractor with holding arm; ( $\mathbf{f}$ ) close-up view of an exiting nerve root through the tubular retractor

# 18.3 Surgical Techniques

## 18.3.1 Primary Disk Herniations

## 18.3.1.1 Interlaminar Approach [22, 23]

## Indications

- All contained disk herniations (DHs) and extruded fragments between the midline and the medial border of the pedicle. In relation to the disk space, the fragments may be caudally or cranially extruded. In the latter case, the translaminar approach is preferred.
- DH combined with central/recess stenosis or with asymptomatic segmental instability.
- Recurrent DH.

## Contraindications

• Foraminal or far lateral DHs which are located lateral to the lateral border of the pedicle

#### **Preoperative Planning**

• Biplanar plain radiographs

Optional in first surgery cases, provided that the MRI investigation encloses a coronal slice (scoliosis!). Mandatory (1) in recurrent DHs cases to evaluate bony defects and (2) whenever MRI leads to suspect a bony abnormality (spina bifida, pars interarticularis defects)

• MRI

Sagittal slices: Contained disk herniation (DH) or extruded fragment? Caudal or cranial (suitable for translaminar approach) fragment dislocation? Mid-vertebral body herniation (halfway between two disk spaces)? Foraminal slice: Black neuroforamen? Extraforaminal slice: Disk fragment still apparent? Axial slices: Axillary disk fragment? How much of the DH is underneath the thecal sac, intraforaminal or extraforaminal (Fig. 18.2)? Pseudomeningocele in recurrent disk surgery? Coronal slices: Which approach for combined intra- and extraforaminal DH? Gadoliniumenhanced slices: Amount of scar tissue on the way to and into the spinal canal? Differentiation between recurrent DH and scar tissue?

# • CT scan

Second choice whenever MRI contraindicated or not available. *Disko-CT* (diskography + CT): helpful in suspected extraforaminal DH. *CM-enhanced CT*: indicated for recurrent disk and differentiation between intraforaminal DH and neurinoma

## Positioning

We recognize that several positioning could provide good clinical results, especially with experienced operating room personnel (ORP). The features of our preferred positioning technique are described below:

- The patient is placed prone on the Wilson frame. Advantages: Hip and knee joints are only moderately flexed, especially important in obese patients. The lordosis of the lumbar spine should be reduced as required by increasing the height of the arches. The distance between the laminae can be adjusted according to the size of the patient in order to allow a free-hanging abdomen to reduce bleeding (Fig. 18.3).
- The head is positioned into a ProneView mask (Manufacturer: Dupaco Inc, Oceanside, California, USA). Eyes, nose, and chin are protected: The anesthesiologist is able to check them intraoperatively by use of a mirror (Fig. 18.3).
- For safety reasons the patient is secured with a belt on the gluteal area: This becomes helpful when the OR table has to be tilted away from the surgeon, e.g., in dealing with extraforaminal or far lateral disk herniations (EFDHs).
- The OR table is tilted until the lumbar spine is parallel to the floor.
- X-ray localization: A 2 cm skin incision does not allow a "seek and find" surgery. Therefore the correct X-ray localization of the surgical target area is of paramount importance. The needle is always inserted contralateral to the intended surgical side in order to avoid subcutaneous or intramuscular hematoma and off the midline in order to prevent inadvertent CSF leakage. The needle is perpendicular to



**Fig. 18.2** Teaching case: A 64-year-old lady presented with mild low back pain and severe left-sided L5 pain requiring opioids since 3 weeks. The examination demonstrated a left-sided foot dorsiflexion weakness. (a) Because the sagittal MRI slices were not performed lat-

the target area (and to the floor): Soft tissue dissection is easier straightforward down. Even small oblique deviations can lead to the wrong level, especially in obese patients. eral enough, the L5/S1 disk was reported as normal. Conservative therapy was advised. (b) MRI was repeated with appropriate lateral slicing. The small intraforaminal disk herniation squeezing the left-sided L5 root (*bottom right*) was removed surgically

The needle should point to the equator of the target disk. With increasing experience, the surgical field may be narrowed further to only the extruded disk fragment.



**Fig. 18.3** Positioning for open lumbar disk surgery: (a) The face is embedded in anatomically tailored foam. (b) The mirror enables a continuous monitoring of the eyes and of the tube. (c) The lumbar spine is parallel to the floor. The belts secure the patient during tilting the table

 $30^{\circ}$  away from the surgeon, as required in extraforaminal disk surgery. (d) The Wilson frame can be adjusted according to the size of the patient and may open up the interlaminar window by decreasing the lumbar lordosis

#### Soft-Tissue Approach

The interlaminar space can be approached via a subperiosteal (SP) or a transmuscular or paramedian (TM) route. Although the use of the microscope "from skin to skin" is optional, its advantages will be appreciated in dealing with a miniaturized surgical corridor. The most relevant steps are described below:

- Prophylactic antibiotic coverage (e.g., cephazoline 2 g) 30 min before skin incision
- *Skin*: 2 cm incision, 5 mm (SP), or 10 mm (TM) off the midline
- Fascia: (SP) Slightly semicircular incision toward the midline. Five holding sutures on the medial lip secured to a clamp with weights. (TM) Straight incision with one holding suture on each side.
- *Muscle*: (SP) Paramedian retraction of the paravertebral muscles from the interspinous ligament. Sharp dissection of the rotators from the lower rim of the superior lamina and from the facet joint capsule. Insertion of a miniaturized speculum-counter-retractor system (Fig. 18.1c; manufacturer: Medicon, Tuttlingen, Germany).

(TM) Blunt splitting with the index finger until the lamino-facet junction can be palpated.

Opening of the muscular corridor with miniaturized muscle retractors or with a dilator. Insertion of an expandable tubular retractor (Fig. 18.1e; manufacturer: Medicon, Tuttlingen, Germany) of 15 mm or 18 mm diameter. Both the speculum and the tube may be secured to the OR table with a self-holding arm nicknamed the "snake" (Fig. 18.1e).

- *Interlaminar space*: From this step onward, the surgical technique is identical. The lower rim of the cranial lamina, the medial border of the facet joint, and the yellow ligament are the area of interest. Radiographic confirmation of the level is performed. Following a lateral flavectomy or flavotomy with suspension sutures, the epidural fat is exposed. The medial border of the inferior articular process is undercut or drilled off until the shoulder of the root is palpated.
- *Epidural dissection*: Up-down dissection of the epidural fat performed with a microdissector and a flat sucker along with careful bipolar coagulation of veins which opens access to the root-DH complex.

#### **Exposure of the Herniated Disk**

• *Management of the DH*: The local anatomy will dictate the necessary steps. Usually, a gentle

dissection between root and disk material is accomplished first. In our experience the root retraction is performed intermittently with a flat sucker rather than with a conventional root retractor. Free disk fragments are removed with miniaturized forceps (Fig. 18.1b, manufacturer: Medicon, Tuttlingen, Germany). If indicated, the annulus is split bluntly with the dissector or with a scalpel and further disk material is removed. In the authors' experience, additional diskectomy is performed in 20–30 % of the cases.

#### Closure

 The disk space, when opened, is rinsed with normal saline. The opening of the annulus is closed with a collagen sponge coated with fibrinogen and thrombin (Tachosil<sup>®</sup>, manufacturer: Behring, Marburg, Germany). The epidural fat is mobilized in order to cover the root. Careful hemostasis goes along with closure by layers.

## 18.3.1.2 Translaminar Approach [24–28]

### Indications

- Cranially extruded disk fragments pushing the exiting root against the lower rim of the pedicle. Usually they are located within the root canal between two lines marking the medial and lateral border of the superior facet.
- Recurrent cranially extruded disk fragments of DH previously removed via an interlaminar approach.

## Contraindication

• Lack of an adequate bony lamina, e.g., severe spinal canal stenosis and spina bifida

#### **Preoperative Planning**

• *MRI* (*sagittal slices*): Measure the distance between the upper border of the disk space and the lower rim of the cephalad pedicle. The translaminar hole will be centered on the halfway of this distance. *Axial slices*: Look at how much of the bulk of the DH is underneath the thecal sac and how much is lateral of it or even intraforaminal. The translaminar hole is centered on the lateral border of the dural sac.

#### Positioning

- Basically the same as for the interlaminar approach.
- Important: The target lamina should be parallel to the floor! This may require the surgeon to tilt the OR table in a reverse-Trendelenburg position. The advantages of a horizontal target lamina are twofold: The placement of the retractor blade and the drilling of the hole become easier (see Figs. 18.6 and 18.7).
- Radiographic localization: The needle should point to the largest portion of the DH which is usually halfway between the upper border of the target disk and the lower rim of the cranial pedicle. At the beginning of the learning curve, these landmarks may be labeled on the skin incision centered in between.

#### Soft-Tissue Approach

- The lamina can be approached via a subperiosteal (SP) or a transmuscular (TM) route. The soft tissue approach mirrors the interlaminar approach. Remember: The width and the overlapping of the lamina in relation to the disk space increase in the caudal-cranial direction, whereas the width of the isthmus decreases. This means that the translaminar hole will be more medially and more ovalshaped in the upper lumbar levels (Figs. 18.4 and 18.5).
- Lamina: Irrespective of the type of retractor used, the lateral border of the lamina should be visible underneath the retractor valve. A dissector is placed onto the lamina where the bulk of the DH is suspected and a fluoroscopic localization is performed. At this point the lamina should have been tilted parallel to the floor, so that the high-speed cutting burr can be held easily perpendicular to the lamina. With slow circular movements, a round (L5) or oval-shaped (L4 and cranially) hole of about 10 mm in diameter is performed (Figs. 18.6 and 18.7). Three layers "white" (outer cortical bone), "red" (spongy bone), and "white" (inner cortical bone) will be



**Fig. 18.4** The up-down length of the lamina (*white figures*) increases, whereas the width of the isthmus (*black figures* in mm) decreases in the caudal-cranial direction. That means that the overlap of the disk by the lamina increases also in the upper lumbar levels. Furthermore, there the translaminar hole becomes more paramedian and oval-shaped

drilled off. For the sake of safety, the inner cortical bone should be drilled with a diamond burr. Remarks: (1) At least 3 mm of the lateral border should be spared in order to avoid a fracture of the pars interarticularis (Fig. 18.6); (2) usually the translaminar hole is located just cephalad to the cranial insertion of the yellow ligament. So, after removal of the thin shell of inner cortical bone with small patches, epidural fat will appear.

• Epidural dissection: Up and down dissection of the fat along the lateral border of the dura.

That should be continued cranial up to the axilla of the exiting root.

#### **Exposure of the Herniated Disk**

Usually an extruded or subligamentous disk fragment(s) can be mobilized. After decompression, the root slips caudally into the visible field (Fig. 18.7). The foramen is then probed with a double-angled hook or blunt probe. If an extensive annular perforation is detected, the disk space should be cleared. In our experience that was required in merely 20 % of the cases. The rate of recurrence was 7 %.

# Closure

 Gelfoam soaked with a long-acting steroid to fill in the hole is optional, but should be avoided if the disk space has been cleared.

## 18.3.1.3 Extraforaminal or Far-Lateral Approach [29–31]

#### Indication

• DH whose bulk is located at least two-thirds lateral to the pedicle

### Contraindication

• Foraminal DH located more than two-thirds inside the root canal

#### **Preoperative Planning**

 MRI – sagittal slices: Usually scans are not lateral enough, i.e., lateral to the root canal, and may miss the extraforaminal disk herniation (EFDH). Axial slices: Compare the amount and distribution of the extraforaminal fat tissue on both sites. Coronal slices: Although rarely performed, they are of invaluable assistance to show the spatial relationship between the exiting root, root canal, and extraforaminal compartment.

#### Positioning

- Basically the same as for the interlaminar approach.
- For safety reasons the patient should be belted on the gluteal region: The OR table has to be tilted 20–30° away from the surgeon in order



**Fig. 18.5** Clinical case: (a) The sagittal MRI shows cranially extruded disk herniations at the level L3/l4 and L4/L5. (b) The DH encroaches the root L3 and (c) L4 on the right side. (d) Because the 28-year-old lady complained about a three-fifth weakness of the m. quadriceps, both

DHs were removed via translaminar holes. Note that the L3 hole is more medial and more oval-shaped due to the narrower pars. Clearing of the disk spaces was not necessary



**Fig. 18.6** (a) The 3-D CT shows a translaminar hole at L3 on the left side. Note: The facet joint L3/L4 is intact and a sufficient lateral rim (5 mm, *arrow*) of the pars is maintained where the bone is strongest. (b) Intraoperative

view: The internal lamina is drilled off inferomedially where the upper rim of the yellow ligament (*star*) appears. The lateral rim of the pars (*arrow*) is the lateral boundary



**Fig. 18.7** (a) A right-sided 10 mm translaminar hole at L4 with an intact inner cortical bone is seen through the expandable tubular retractor (15 mm  $\emptyset$ ); (b) following dissection of the epidural fat, a large extruded disk frag-

to get a better oblique view of the extraforaminal compartment. Morbidly obese patients may risk to "roll over" on to their abdomen.

- Radiographic localization *lateral view*: Insert a spinal needle one finger's breadth lateral to the spinous process, perpendicular to the skin and projecting toward the affected disk space. Draw a horizontal line at this level (Fig. 18.8a, DS).
- *AP view*: Two horizontal lines are drawn, (1) the affected disk (DS) and (2) the lower border of the transverse process above the affected disk (TP). Two vertical lines are also drawn: (1) the midline (row of the spinous processes, M, and (2) a line about 4 cm off to the midline, marking the lateral boundary of the pedicle above and below the affected disk. The distance between the two horizontal lines (SI) is the skin incision and will be 3 cm in length and about 4 cm paramedian (Fig. 18.8a).

#### Soft-Tissue Approach

The paraspinal transmuscular blunt-splitting approach to EFDH at the level L4/L5 or more cranially can be performed with an expandable tubular retractor or with a miniaturized speculum combined with medial and lateral counterretractor blades (Fig. 18.8). At the level L5/S1, the author recommends the use of two counterretractors inserted perpendicular to each other. That allows to choose four blades of different lengths matching with the following structures: facet joint (medial), transverse plane (lateral),

ment appears in the axilla of the exiting L4 nerve root; (c) after the removal of the disk fragments, the L4 nerve root slips back into the visible field

transverse process (cranial), and ala (caudal) (Fig. 18.9). Furthermore, the use of the microscope "from skin to skin" is advised.

- *Skin*: 3 cm in length 4 cm off the midline.
- *Transmuscular route*: After incision of the fascia of m. erector spinae, the muscle is dissected bluntly using the index finger along the cleavage plane between the multifidus and the longissimus muscle (Fig. 18.8b). If this intermuscular plane cannot be palpated, the muscle is split downward to the medial third of the transverse processes. The selected retractor is then introduced so that the tips rest firmly on the lower half of the upper transverse process and on the upper half of the lower one. The lateral surface of the pars interarticularis represents the medial border of the surgical exposure. Fluoroscopic confirmation at this point of the procedure is essential (Fig. 18.8d).
- *Extraforaminal approach*: Tilting the OR table by 20–30° away from the surgeon gives a better view of the area lateral to the pedicle. Drilling off bone is usually not necessary, except in the case of an extremely hypertrophied facet joint or at the L5/S1 level. The medial half of the intertransverse muscle is incised and pushed laterally, thereby exposing the intertransverse membrane, also called the "intertransverse ligament." Use of bipolar cautery is essential to maintain hemostasis and blood-free surgical field. After incision of the membrane, the fat surrounding the nerve appears. Because of



**Fig. 18.8** (a) Preoperative labeling for a left-sided paraspinal approach at L3/L4: midline (M), disk space (DS), transverse process (TP), and skin incision. (b) The blunt muscle-splitting approach uses a cleft between the multifidus muscle (medially) and the thoracic longissimus muscle (laterally). (c) The index finger palpates the junction

between the medial part of the transverse process L3 and the ascending facet joint (*yellow arrow*). The 3-D CT shows the target point. (**d**) The expandable tubular retractor points to the extraforaminal area. (**e**) The L3 nerve is displaced laterally, cranially, and superficially by (**f**) the underlying extruded disk fragment



**Fig. 18.9** (a) Operative site of a left-sided paraspinal approach L5/S1: four slim retractor blades of different lengths are inserted. (1) Ascending facet (medial); (2) thoracic longissimus muscle (lateral); (3) transverse process

the proximity of the nerve, the accompanying vessels, and DH, the sucker should also be used as a nerve retractor. However, beware

L5 (cranial); (4) ala sacri (caudal); the *white arrow* shows the point where the nerve L5 is to be expected; (**b**) intraoperative fluoroscopic check: The dissector points to the target point (*white arrow*)

of an excessive retraction of the dorsal root ganglion in order to minimize the incidence of postoperative burning dysesthesias which should be counseled to the patient preoperatively. Branches of the radicular artery should be dissected carefully and spared whenever possible. The accompanying veins can be cauterized, if they hinder the access to the disk fragment.

#### Exposure of the Herniated Disk

*Management of the DH*: Typically, we find the nerve and the ganglion pushed laterally and cranially by the free disk fragment (Fig. 18.8c). Usually, removal of the fragment alone is sufficient. If an extensive perforation of the annulus is evident, clearing of the disk space should be considered. After probing the root canal with a double-angled blunt hook for residual fragments, the nerve may be covered with a gelfoam soaked with crystalline steroid.

#### Closure

- Placing a drain is optional and in our experience seldom necessary. Muscles do not require suturing.
- Special considerations for the L5/S1 level: Because of the particular anatomical relationship between disk space, transverse process L5, and ala, the microsurgical muscle-splitting approach at the lumbosacral level should be practiced by a surgeon who is already familiar with the technique at the more cranial levels. Repeated intraoperative fluoroscopic checks may also be necessary. If difficulties should arise, switching to the conventional macroapproach should be considered.

## 18.3.2 Recurrent Disk Herniation [32, 33]

A prevalence of 7–10 % recurrent herniations is reported in the literature independently from the surgical technique used. To deal with a recurrent DH usually does not mean to perform a "redo surgery" identical to the first procedure. The peculiarities of surgery for recurrent DH will be addressed.

#### 18.3.2.1 Preoperative Planning

- The use of the microscope in our view is a must as it facilitates the differentiation between scar tissue and the thecal sac.
- Bypassing most of the scar tissue is imperative. This can be achieved either by using a wider approach than the previous one exposing the lower edge of the upper lamina or the medial wall of the pedicle where unscarred dura can be found. The translaminar approach of a recurrent cranially extruded disk fragment can be used via a virgin translaminar route instead of dissecting the interlaminar scar tissue.
- Preoperative radiographs show the amount of previously resected bone. This is of paramount importance if previous surgery has been performed elsewhere. If doubts persist, CT scan shows the bony landmarks which will guide the surgical approach.
- Gadolinium-enhanced MRI shows the relationship between scar tissue and true recurrent disk material and may differentiate the two. However, this holds true if the recurrent disk herniation occurs roughly within 3 years after the first surgery. Furthermore, endplate Modic lesions and CSF collections should be also be closely examined.

#### 18.3.2.2 Positioning

• The same as performing first-time surgery

#### 18.3.2.3 Soft-Tissue Approach

- Fluoroscopic labeling of the target area is mandatory as the scar of the skin may slip depending on the positioning and on the amount of subcutaneous tissue.
- Cautious sharp subperiosteal dissection is recommended to the interlaminar window. The lower border of the cranial lamina, the medial border of the remnant facet joint, and the upper border of the caudal lamina should be clearly visible.
- Drilling off bone with a diamond-coated burr is the most commonly used entry point in the area between the cranial border of the epidural scar and the virgin dura. If this approach should fail, the area between the medial border

of the pedicle and the shoulder of the traversing root is an alternative option.

## 18.3.2.4 Exposure of the Herniated Disk

- A generous decompression of the root in the lateral recess should precede the mobilization of the nerve from the annulus or from the extruded disk fragment. An intraoperative single shot of steroids may be helpful at this stage of the procedure.
- "Peeling" of the fibrous tissue from the nerve carries a high risk of injuring the dura and does not provide a better clinical outcome. Before biting with the Kerrison punch, a light tug may show the scarred dura jump: a sentinel sign of imminent dural tear!
- The disk space can be entered laterally from the border of the dural sac and cleared. This reduces the pressure on the extruded disk material. Repeated flushing with saline within the disk space may bring further disk material to the surface. We do not recommend to curette the endplates. The fibrous pocket containing the extruded disk material is opened and its content removed bluntly with straight/ angled probes of different length. A tiny fibrous shell is left adherent to the dural sac. There is no evidence that forced "neurolysis" provides a better clinical outcome.

## 18.4 Postoperative Care [34]

## 18.4.1 Uncomplicated Surgery

The patient is encouraged to leave the bed 6 h after surgery. Sitting is allowed starting from the first postoperative day. Physiotherapy starts the morning after surgery. Hospital staying is usually 1-3 days.

### 18.4.2 Standard Dural Repair

Forty-eight hours of bed rest with the head in slightly Trendelenburg position (head down). If intraoperative loss of CSF was significant, the patient is treated with intravenous hydration + promethazine + analgesic regimen.

## 18.4.3 Very Difficult Dural Repair Not Watertight at the Time of Wound Closure

Closed subarachnoid drainage obtained by puncture at one level above the dural opening and the catheter placed at the thoracic-lumbar junction. The amount of CSF can be controlled by the level of the collection bag relative to the lumbar spine. CSF drainage can be continued up to 1 week and should cause mild headache.

# 18.5 Complications

The literature lists several "generic" complications such as deep vein thrombosis, pulmonary embolism, and urinary infections which are fortunately rare. Retroperitoneal major vessel injuries and postoperative visual disturbances (risk factors: diabetes mellitus, long operation time) are even more rare.

Microsurgical diskectomies have significantly less severe intraoperative complications as compared to non-microsurgical diskectomies [35]. Experienced surgeons have significantly less complications (2.2 %) than beginners (10.7 %) [36]. Recurrent surgeries are burdened with a higher incidence of complications [37].

The most common complications of even refined microsurgical techniques are: wrong-level surgery, dural opening/CSF leakage (2-7 %), root injury (0.06 %), and spondylodiskitis (0.4-1 %). Some remarks about the first two pitfalls are as follows:

#### 18.5.1 Wrong-Level Surgery

- The surgical target area should be as much parallel to the floor as possible.
- Use a spinal needle (for CSF tap: more expensive, but also more radiopaque, especially in adipose patients) perpendicular to the back, one finger's breadth off the midline, down to

the lamina, contralateral to the intended surgical approach, and pointing to the upper rim of the disk space.

- Label the corresponding horizontal line, the midline, and the skin incision.
- Drape the C-arm (lateral view) and park it conveniently in the surgical suite.
- Time-out procedure: Confirm the correct level and side of surgery
- Repeat the trajectory of the cannula inserting the retractor. Remember: In overweight patients, a minimal oblique trajectory can lead a surgeon to the wrong level.
- C-arm Check of the level before drilling off bone.
- Do not rely on scars in recurrent surgery; mark the skin incision with the aid of radiographic localization.
- The intraoperative threshold for obtaining a fluoroscopic confirmation should be low.

## 18.5.2 Dural Opening

Each dural opening requires a specific treatment depending on location, shape, and size of the lesion, potential concomitant injury of the cauda fibers, and microsurgical skills of the surgeon – just to mention the most important factors.

The following acronym "Bird Mc Dove" may aid in remembering a sequence of steps (Fig. 18.10):

- 1. *B*one removal until you see the whole dural tear.
- 2. Intradural look.
- 3. Repone the fibers, if necessary.
- 4. *D*o an inside patch (e.g.,Tachosil<sup>®</sup> with the yellow surface to the dura).
- 5. Dural closure, at best by suturing.
- 6. Outside patch (the same as step 4).
- 7. Valsalva maneuver (e.g.,  $40 \text{ cm H}_2\text{O} \times 30 \text{ s}$ ).



**Fig. 18.10** (a) Large dural defect not suitable for direct repair. (b) The opening has been closed from the inside with haemostatic (*yellow surface*) collagen fleece (Tachosil<sup>®</sup>). (c) The same has been done from the epidural side with the haemostatic yellow surface inward

(sandwich technique). (d) A tension-free pedicled epidural muscle flap has been swung to fill the epidural space (drawing). (e) Intraoperative site seen from the left. The *brown square* points to the epidural space

- Epidural pedicled muscle flap (from the paraspinal muscles in order to fill the epidural dead space).
- 9. *Multilayer closure* (muscle layer anchored to the spinous process also in the depth).
- 10. CSF drainage, if necessary.

Of course not all of the abovementioned steps are necessary every time. The first goal is to get a watertight closure of the dura (5). Should that fail, then three steps became mandatory: to seal the dural opening (4+6), to achieve a watertight wound closure (9), and to lower the postoperative CSF pressure (10) [34].

## **18.6 Critical Evaluation**

But successful lumbar disk microsurgery is also based on the surgeon's appreciation of the facts that microsurgery is not seek-and-find surgery, and that the microscope does not do the surgery. [23]

The first half of the citation stresses the importance of preoperative planning: the careful evaluation of location, size, and shape of the disk herniation and its relationship to the disk space, to the exiting and to the traversing root, to the lateral recess, and to the root or spinal canal require excellent MRI-imaging and dictate the soft tissue approach and the intraspinal handling of the pathology. A surgical plan tailored for the individual disk herniation becomes necessary. At that point, the short skin incision, the reduced muscle traumatization, the conservative bone drilling, and the plain removal of the offending disk fragment become the practical implementation of the microsurgical philosophy.

The second half of the citation points out that, although "small is beautiful," it should never be an end in itself. Especially at the beginning of the microsurgical learning curve, switching to a larger approach should be considered whenever problems arise. However, with increased experience, all difficult situations will be addressed even more effectively with the aid of the microscope. The gap which has been bridged between macro- and microsurgery is going to be overcome by the latter and endoscopic spinal techniques. Robotic nano-surgery is waiting behind the corner in the future. The goals do not change: to get a good clinical result with the least iatrogenic trauma.

## 18.7 Key Points

- Even though this chapter has dealt with technical aspects of the surgical treatment of lumbar disk herniations, proper indications with the right technique at the right moment is the most important factor influencing the clinical outcome.
- 2. The use of the microscope provides many advantages.
- 3. The most important aspect of the microsurgical philosophy which supports minimally invasive surgery is the "mental planning" of the procedure beforehand.
- 4. The assumption that one approach fits all subgroups of lumbar disk herniations has been substituted by the conviction that tailored approaches such as the interlaminar, translaminar, and paraspinal approaches address the different pathologies in a less traumatizing manner.
- 5. The use of a paramedian muscle-splitting approach and use of tubular retractors when available have even further reduced the approach morbidity, especially in obese patients.

## References

- Mc Culloch JA, Young PH. Chapter 20. In Essentials of spinal microsurgery. Lippincott-Raven; 1998. p. 380.
- Weinstein JN. Musculoskeletal Dartmouth Atlas Working Group. Dartmouth atlas of health care. Chicago: American Hospital Association Press; 2000.
- Deyo RA, Weinstein JN. Low back pain. N Engl J Med. 2001;344:363–70.
- Weinstein JN, Bronner KK, Morgan TS, et al. Trends and geographic variations in major surgery for degenerative diseases of the hip, knee,
and spine. Health Aff (Millwood). 2004;(Suppl Variation):VAR81–9.

- Weinstein JN, Tosteson TD, Lurie JD, et al. Surgical vs nonoperative treatment for lumbar disk herniation. The Spine Patient Outcomes Research Trial (SPORT): a randomized trial. JAMA. 2006;296(20):2441–50.
- Mayer HM. Principles of microsurgical discectomy in lumbar disc herniations. In: Mayer HM, editor. Minimally invasive spine surgery. Heidelberg: Springer; 2005. p. 278–81.
- Caspar W. A new surgical procedure for lumbar disc herniation causing less tissue damage through a microsurgical approach. Adv Neurosurg. 1977;4:74– 80. Springer, Heidelberg.
- Katayama Y, Matsuyama Y, Yoshihara H, et al. Comparison of surgical outcomes between macro discectomy and micro discectomy for lumbar disc herniation: a prospective randomized study with surgery performed by the same spine surgeon. J Spinal Disord Tech. 2006;19:344–7.
- Koebbe CJ, Maroon JC, Abla A, et al. Lumbar microdiscectomy: a historical perspective and current technical considerations. Neurosurg Focus. 2002;13(2), E3.
- Fakouri B, Patel V, Bayley E, et al. Lumbar microdiscectomy versus sequestrectomy/free fragmentectomy: a long-term (>2 y) retrospective study of the clinical outcome. J Spinal Disord Tech. 2011;24(1):6–10.
- Onik GM, Kambin P, Chang MK. Minimally invasive disc surgery. Nucleotomy versus fragmentectomy. Spine. 1997;22(7):827–8; discussion 828–30.
- Baek GS, Kim YS, Lee MC, et al. Fragmentectomy versus conventional microdiscectomy in single-level lumbar disc herniations: comparison of clinical results and recurrence rates. J Korean Neurosurg Soc. 2012;52(3):210–4.
- Anand N, Baron EM, Bray Jr RS. Benefits of the paraspinal muscle sparing approach versus the conventional midline approach for posterior nonfusion stabilization: comparative analysis of clinical and functional outcomes. SAS J. 2007;1:93–9.
- 14. Anand N, Baron EM, Bray Jr RS. Modified musclesparing paraspinal approach for stabilization and interlaminar decompression: a minimally invasive technique for pedicle screw-based posterior nonfusion stabilization. SAS J. 2008;2:40–2.
- Brock M, Kunkel P, Papavero L. Lumbar microdiscectomy: subperiosteal versus transmuscular approach and influence on the early postoperative analgesic consumption. Eur Spine J. 2008;17:518–22.
- Foley KT, Smith MM. Microendoscopic discectomy. Tech Neurosurg. 1997;3:301–7.
- Shin DA, Kim KN, Shin HC, et al. The efficacy of microendoscopic discectomy in reducing iatrogenic muscle injury. J Neurosurg Spine. 2008;8:39–43.
- Riesenburger RI, David CA. Lumbar microdiscectomy and microendoscopic discectomy. Minim Invasive Ther Allied Technol. 2006;15:267–70.
- 19. Sasaoka R, Nakamura H, Konishi S, et al. Objective assessment of reduced invasiveness in

MED. Compared with conventional one-level laminotomy. Eur Spine J. 2006;15:577–82.

- Kim YB, Hyun SJ. Clinical applications of the tubular retractor on spinal disorders. J Korean Neurosurg Soc. 2007;42:245–50.
- Kawaguchi Y, Matsui H, Tsuji H. Changes in serum creatine phosphokinase MM isoenzyme after lumbar spine surgery. Spine. 1997;22:1018–23.
- Mayer HM. Lumbar disc herniations: the microsurgical interlaminar, paramedian approach. In: Mayer HM, editor. Minimally invasive spine surgery. Heidelberg: Springer; 2005. p. 283–96.
- McCulloch JA, Young PH. Microsurgery for lumbar disc herniation. In: Young PH, McCulloch JA, editors. Essentials of spinal microsurgery. Philadelphia: Lippincott-Raven; 1998. p. 329–82.
- 24. Di Lorenzo N, Porta F, Onnis G, et al. Pars interarticularis fenestration in the treatment of foraminal lumbar disc herniation: a further surgical approach. Neurosurgery. 1998;42:87–90.
- Bernucci C, Giovanelli M. Translaminar microsurgical approach for lumbar herniated nucleus pulposus (HNP) in the "hidden zone": clinical and radiologic results in a series of 24 patients. Spine. 2007;32(2):281–4.
- Papavero L. Lumbar disc herniations: the translaminar approach. In: Mayer HM, editor. Minimally invasive spine surgery. Heidelberg: Springer; 2005. p. 279–303.
- Soldner F, Helper BM, Wallenfang T, et al. The translaminar approach to canalicular and craniodorsolateral lumbar disc herniations. Acta Neurochir (Wien). 2002;144:315–20.
- Vogelgesang JP. The translaminar approach in combination with a tubular retractor system for the treatment of far cranio-laterally and foraminally extruded lumbar disc herniations. Zentralbl Neurochir. 2007;68(1):24–8.
- Papavero L. Lumbar disc herniations: the extraforaminal approach. In: Mayer HM, editor. Minimally invasive spine surgery. Heidelberg: Springer; 2005. p. 304–14.
- Tessitore E, de Tribolet N. Far-lateral lumbar disc herniation: the microsurgical transmuscular approach. Neurosurgery. 2004;54(4):939–42.
- McCulloch JA, Young PH. Foraminal and extraforaminal lumbar disc herniation. In: Young PH, McCulloch JA, editors. Essentials of spinal microsurgery. Philadelphia: Lippincott-Raven; 1998. p. 383–428.
- 32. McGirt MJ, Ambrossi GL, Datoo G, et al. Recurrent disc herniation and long-term back pain after primary lumbar discectomy: review of outcomes reported for limited versus aggressive disc removal. Neurosurgery. 2009;64(2):338–44; discussion 344–5.
- 33. Aizawa T, Ozawa H, Kusakabe T, et al. Reoperation for recurrent lumbar disc herniation: a study over a 20-year period in a Japanese population. J Orthop Sci. 2012;17(2):107–13.
- McCulloch JA, Young PH. Postoperative care of the lumbar microsurgical patient. In: Young PH,

McCulloch JA, editors. Essentials of spinal microsurgery. Philadelphia: Lippincott-Raven; 1998. p. 487–92.

- Wildfoerster U. Intraoperative complications in lumbar intervertebral disc operations. Comparative study of the spinal study group of the German Society of Neurosurgery. Neurochirurgia. 1991;34(2): 53–6.
- 36. Wiese M, Kraemer J, Bernsmann K, et al. The related outcome and complication rate in primary lumbar microscopic disc surgery depending on the surgeon's experience: comparative studies. Spine J. 2004;4(5):550–6.
- Schuewer U, Roosen K. Complications in lumbar intervertebral disc operations. Neurochirurgia. 1988;31 Suppl 1:192–5.

# **Lumbar Disk Revision Surgery**

19

# Christoph Mehren and H. Michael Mayer

#### 19.1 Introduction

With the increasing numbers of lumbar disk surgeries, the quantity of revision cases is increasing as well. Especially recurrent disk herniations or diskogenic restenosis of the spinal canal generates challenges in spinal microsurgery.

Primary lumbar diskectomy or fragmentectomy is performed with minimally invasive surgical techniques. Microsurgical or endoscopic removal of lumbar disk herniations have become international standards. However, irrespective of the surgical technique, the incidence of recurrent disk herniation still varies in the literature from 1 up to 38 % [1–8].

The huge range of these figures is explainable by different follow-up times and by the method of detecting the recurrent herniation. If the patient is asymptomatic, a postoperative MRI is usually not performed. However in routinely performed MRIs 2 years after diskectomy, 56 % of the patients had asymptomatic disk reherniations [3].

Recurrences of a disk herniation are multifactorial. In a number of studies, potential risk factors have been identified. Gender (male), smoking,

C. Mehren, MD (⊠) • H.M. Mayer, MD, PhD Department of Spine Surgery, Schön Klinik München Harlaching, München, Germany e-mail: cmehren@schoen-kliniken.de

and heavy work seem to be independent risk factors [9–11]. But it is most probably the genetic predisposition with ongoing or accelerated degeneration which predominantly influences the recurrence rate. Disk height index and sagittal range of motion show a correlation with higher recurrence rates [9]. Additionally, the competence of the disk annulus and the type of herniation can predict the postoperative clinical outcome following lumbar diskectomy. The lowest rates of reherniation and reoperation (1 %) with the best clinical result can be expected for sequestrated herniations with a small annular defect followed by contained, fragmented disk herniations also with a small annular tear (10%). In patients with extruded fragments and massive posterior annular loss recurrent herniation can occur in up to 27 % and the reoperation rate can reach 21 %. In case of contained non-fragmented herniations, 38 % of the patients had recurrent or persistent sciatica [7].

# 19.2 Surgical Techniques at Primary Surgery

The role of different surgical techniques for the risk of developing a recurrent disk herniation is not clear. The epidemiologic data, suggesting that recurrence rates following closed endoscopic diskectomy seem to be somewhat higher when compared to "open" microsurgical techniques,

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_19

have not been substantiated by sound randomized controlled trials.

The same is true for the question whether limited versus more aggressive disk removal diminishes the risk of recurrence. Randomized controlled trials have suggested that the clinical results seem to be comparable; however, the morphology of the annulus defect as well as the amount of preserved disk height might play a more dominant role [5, 12, 13].

This seems to be supported by Carragee's study [4]. The clinical outcome – regarding low back pain – and the recurrence rate are competing with each other with reverse results. It was shown that the incidence of a recurrent disk herniation was lower in the diskectomy group, but in the 2-year follow-up, the clinical outcome regarding low back pain was 2–2.5-fold worse than in the limited disk removal group [2, 4, 5].

Thus, so far general recommendations cannot be established according to scientific data. There is however agreement that patients with higher disk heights and large annular defects seem to have higher risk of recurrencies.

#### 19.3 Diagnostic Considerations

MRI with/without contrast enhancement is the primary diagnostic tool for recurrent disk herniations as it defines precise location of the recurrent disk herniation for accurate planning of the approach. MRI also allows to distinguish disk material from fibrotic tissue and provides information about the amount of intraspinal fibrosis (Figs. 19.1, 19.2, and 19.3).

Should there be contraindications for an MRI (e.g., pacemaker, prosthetic heart valve, clips, etc.), CT scan and myelo-CT are alternative options. CT scan allows the evaluation of the amount of bony defect created by the first surgery.

This can also be assessed with preoperative radiographs in standing position as well as flexion/extension x-rays. The alignment of the segment, the global sagittal balance, and the presence of gross instabilities can be evaluated with radiographs.



Fig. 19.1 Sagittal lumbar spine MR T2-weighted image depicts a cranially sequestered recurrent disk herniation



**Fig. 19.2** The disk material does not show an enhancement of contrast media and is therefore distinguishable from scar tissue in doubtful cases

# 19.4 Indication for Surgery

The chances of a successful conservative therapy in case of recurrent disk herniations are lower as compared to primary disk herniations.



**Fig. 19.3** The axial MR T2-weighted image clearly demonstrates nerve root compression of S1 on the left side due to herniated disk material

This can be explained by the fact that the neural structures are less "mobile" in the spinal canal due to adhesions or scar tissue without the possibility to "give way." On the other hand, it was shown that a high percentage of the recurrent disk herniations include endplate disk material, with harder consistency and less ability to shrink than "pure" nucleus material [14].

Indication for surgery as well as the choice for different types of surgical techniques follows the clinical symptoms of the patient.

If signs of nerve root compression are paramount, the primary surgical option is the decompression/re-diskectomy. Unless functional factors such as gross instability or leading back pain are predominant, there is no need to consider additional stabilization.

The options for surgical treatment range from endoscopic or microsurgical re-sequestrectomy/ diskectomy to a fusion/reconstruction procedure. Besides the natural process of degeneration of the functional spinal unit, the technique of the previous surgery plays an important role regarding a segmental instability with consecutive narrowing of the spinal canal. Laminectomy, extensive resection of the facet joint and iatrogenic discontinuation of the pars interarticularis are predictable factors of a postoperative segmental instability [15, 16]. In other words minimally invasive techniques like endoscopic or microsurgical procedures are able to prevent postoperative instability due to less resection of bony structures [17]. Therefore the surgeon has the option for minimally invasive techniques in revision surgery as well.

In the setting of a clinically dominant sciatica, the solitary revision of the spinal canal is indicated, irrespective of the type of a previous surgery. If the pathology leading to nerve root compression is caused by a segmental instability, additional reconstruction of the functional spinal unit for the purpose of spinal fusion is necessary. Most of the time, the patients report also significant back pain.

Up to now there is a lack of clear algorithms regarding the treatment of recurrent disk herniations. A survey among neurological and orthopedic spine surgeons in the USA concerning the surgical treatment patterns of one- and twotime recurrent lumbar disk herniations revealed significantly different opinions. The surgical treatment options were revision microdiskectomy, revision microdiskectomy with in situ fusion, posterolateral fusion using pedicle screws, and PLIF/TLIF or ALIF procedures in combination with posterior instrumentation. Experienced surgeons with more than 15 years of practice were more likely to select just microdiskectomy in contrast to surgeons with fewer years who were more likely to select the microdiskectomy in combination with PLIF/ TLIF. Overall, the probability that two randomly selected spine surgeons would disagree on the surgical procedure of two-time recurrent disk herniation was 69 % [18].

#### 19.5 Surgical Technique

The previous surgical technique influences the revision surgery strategy. In contrast to the first operation, the surgeon is faced with scar tissue and altered anatomical landmarks in terms of bony defects. In principle all operative techniques (endoscopic, microsurgical, and macrosurgical with and without fusion procedures) are also evaluated in revision surgery. The selection of the appropriate technique should be adapted to the surgeon's experience. The steps of microsurgical re-diskectomy/re-decompression will be described.

The exact localization of the recurrent disk herniation should be known as well as its topographic relation to the nerve root and/or scar tissue. This can be best achieved with an MRI with contrast media. The bony landmarks (e.g., medial facet border, lamina border, isthmus) can be seen on the x-rays. If necessary a CT scan can give detailed information about the extension of the previous laminotomy, hemilaminectomy, facetectomy, etc. It also provides data concerning potential ossification of the herniated disk (Fig. 19.4).

The disk space height is localized under fluoroscopic control and a 2 cm skin incision is placed centered over the disk space or over the maximum extension of the recurrence. Sharp subperiostal dissection is done preferably from the remaining superior lamina down to the transition zone of the superior lamina and the inferior facet. The reliable exposure of the bony "edges" is essential. The scar tissue must be safely detached from the bony rim of the lamina and the medial border of the inferior facet. Extreme care has to be taken if MRI shows a bulging dura dorsal to the lamina border. With a small blunt dissector or a diamond drill, the caudal border of the superior lamina is undercut until untouched ligamentum flavum or healthy dura is exposed under the lamina. Once healthy dura is identified, dissection of the scar tissue is started along the medial border of the inferior facet until the rim of the superior facet is identified.

Blunt dissection is performed between bone and scar tissue until the exposure of the lateral border of the exiting nerve root is achieved. This is followed by the decompression along the shoulder of the nerve root until the caudal pedicle. If significant fibrosis is found, a layer of scar tissue may be left on the nerve root, respectively, on the dural sac to avoid dural tear.

The exposure of the lateral dural margin, respectively, of the lateral nerve root margin is followed by a careful mobilization of the nerve root to the middle. There are often adherences of the nerve with the disk space. In these cases it is advisable to leave the nerve in place and open the scar tissue lateral to the nerve to get a safe access to the recurrent herniation. The herniated disk can then be mobilized carefully with a blunt dissector or with a small nerve hook.

At the end of the operation, the neural structures, especially the dura, are checked again for integrity and sufficient decompression. Careful hemostasis and irrigation of the approach and the epidural space finalize the intervention. A drainage is not needed in most of the cases; the patients are allowed to stand up immediately after they regained circulatory stability and consciousness. If there may be high risk for a recurrent disk



Fig. 19.4 (a) Bony defect in the lamina on the left side. (b) Calcified disk material in the lateral recess on the left side

herniation, the use of a soft brace for 4–6 weeks postoperatively is advised.

# 19.6 When Is Additional Stabilization Necessary?

If the patient is suffering from significant low back pain prior to the planned operation or if there is radiologically proven severe instability of the functional spinal unit, microsurgical decompression of the spinal canal should be combined with a stabilizing procedure.

The pedicle screw-rod system allows the physiologic alignment of the vertebral bodies to be restored. The decompression of the neurological structures takes place in an analogous microsurgical technique. Due to stability reasons, the 360° fusion may be favored, through a PLIF or TLIF fusion, as well as for a very high disk space or for severe instabilities via an anterior retroperitoneal approach with an ALIF procedure. From the aforementioned issues concerning nerve root dissection from fibrotic tissue, a posterior intervertebral fusion procedure is obvious. Especially in the presence of severe epidural fibrosis, the feasibility to mobilize the dura is limited. The technical option to choose an access into the disk space lateral of the dura – like the TLIF procedure - may reduce the risk of dural tears or nerve root irritation postoperatively. The additional anterior fusion (ALIF) ensures a higher primary stability and a higher fusion ability [19] (Figs. 19.5, 19.6, 19.7, and 19.8).

#### 19.7 Prevention

The decision-making concerning disk removal whether to clear the disk space with a diskectomy and its possible adverse effects like accelerated disk degeneration and likely low back pain or to just perform a sequestrectomy with potential higher recurrence rate is an individual challenge each time. It is up to the surgeon's consideration and estimation if the patient benefits more from the one or the other technique. Trails to prevent a



**Fig. 19.5** Lateral x-ray at the time of microsurgical diskectomy L3/4. Status after fusion L4/5 8 years ago

recurrent disk herniation with the aid of special techniques (annulus suture) or implants to close the defect in the dorsal annulus (e.g., Barricaid®) did not yet show a significant improvement in the postoperative outcome in the actual literature especially due to the fact that long-term results are lacking [20, 21]. For interspinous spacer, the database regarding recurrent herniation is negligible. There is rather a general tendency so far that these modalities of implant are not able to prevent a recurrent disk herniation [22, 23].



Fig. 19.6 X-Ray in flexion (a) and extension (b) 3 years after the diskectomy at the level L3/4



Fig. 19.7 The MRI of the lumbar spine is showing a huge recurrent disk herniation (a) and a severe segmental instability (b) at the level L3/4



**Fig. 19.8** Lateral x-ray following a re-diskectomy and dorsoventral fusion at the level L3/4

#### References

- Ruetten S, et al. Use of newly developed instruments and endoscopes: full-endoscopic resection of lumbar disc herniations via the interlaminar and lateral transforaminal approach. J Neurosurg Spine. 2007;6(6):521–30.
- McGirt MJ, et al. Recurrent disc herniation and longterm back pain after primary lumbar discectomy: review of outcomes reported for limited versus aggressive disc removal. Neurosurgery. 2009;64(2):338–44; discussion 344–5.
- Lebow RL, et al. Asymptomatic same-site recurrent disc herniation after lumbar discectomy: results of a prospective longitudinal study with 2-year serial imaging. Spine (Phila Pa 1976). 2011;36:2147–51.

- Carragee EJ, et al. A prospective controlled study of limited versus subtotal posterior discectomy: shortterm outcomes in patients with herniated lumbar intervertebral discs and large posterior annular defect. Spine (Phila Pa 1976). 2006;31(6):653–7.
- Barth M, Weiss C, Thome C. Two-year outcome after lumbar microdiscectomy versus microscopic sequestrectomy: part 1: evaluation of clinical outcome. Spine (Phila Pa 1976). 2008;33(3):265–72.
- Ruetten S, et al. Full-endoscopic interlaminar and transforaminal lumbar discectomy versus conventional microsurgical technique: a prospective, randomized, controlled study. Spine (Phila Pa 1976). 2008;33(9):931–9.
- Carragee EJ, et al. Clinical outcomes after lumbar discectomy for sciatica: the effects of fragment type and annular competence. J Bone Joint Surg Am. 2003;85-A(1):102–8.
- Ambrossi GL, et al. Recurrent lumbar disc herniation after single-level lumbar discectomy: incidence and health care cost analysis. Neurosurgery. 2009;65(3):574–8; discussion 578.
- Kim KT, Park SW, Kim YB. Disc height and segmental motion as risk factors for recurrent lumbar disc herniation. Spine (Phila Pa 1976). 2009;34(24): 2674–8.
- Miwa S, et al. Risk factors of recurrent lumbar disc herniation: a single center study and review of the literature. J Spinal Disord Tech. 2013;28(5):E265–9.
- Shimia M, et al. Risk factors of recurrent lumbar disk herniation. Asian J Neurosurg. 2013;8(2):93–6.
- Barth M, et al. Two-year outcome after lumbar microdiscectomy versus microscopic sequestrectomy: part 2: radiographic evaluation and correlation with clinical outcome. Spine(Phila Pa 1976). 2008;33(3): 273–9.
- Thome C, et al. Outcome after lumbar sequestrectomy compared with microdiscectomy: a prospective randomized study. J Neurosurg Spine. 2005;2(3):271–8.
- Brock M, Patt S, Mayer HM. The form and structure of the extruded disc. Spine (Phila Pa 1976). 1992;17(12):1457–61.
- Bisschop A, et al. Which factors prognosticate rotational instability following lumbar laminectomy? Eur Spine J. 2013;22(12):2897–903.
- Bisschop A, et al. Torsion biomechanics of the spine following lumbar laminectomy: a human cadaver study. Eur Spine J. 2013;22(8):1785–93.
- Hamasaki T, et al. Biomechanical assessment of minimally invasive decompression for lumbar spinal canal stenosis: a cadaver study. J Spinal Disord Tech. 2009;22(7):486–91.
- Mroz TE, et al. Differences in the surgical treatment of recurrent lumbar disc herniation among spine surgeons in the United States. Spine J. 2014;14(10):2334–43.
- Mayer HM. The ALIF concept. Eur Spine J. 2000;9 Suppl 1:S35–43.

- Bailey A, et al. Prospective, multicenter, randomized, controlled study of annular repair in lumbar discectomy: two-year follow-up. Spine (Phila Pa 1976). 2013;38(14):1161–9.
- Parker SL, et al. Effect of an annular closure device (barricaid) on same level recurrent disc herniation and disc height loss after primary lumbar discectomy: two-year results of a multi-center prospective cohort study. J Spinal Disord Tech. 2013. Nov 5.
- 22. Floman Y, et al. Failure of the Wallis interspinous implant to lower the incidence of recurrent lumbar disc herniations in patients undergoing primary disc excision. J Spinal Disord Tech. 2007;20(5):337–41.
- Sur YJ, Kong CG, Park JB. Survivorship analysis of 150 consecutive patients with DIAM implantation for surgery of lumbar spinal stenosis and disc herniation. Eur Spine J. 2011;20(2):280–8.

Part V

Surgical Treatment of Lumbar Degenerative Disk Disease: Doubts, Decisions, and Techniques

# The Decision to Operate: Advanced Concepts

20

James P. Lawrence and Todd J. Albert

## 20.1 Introduction

Low back pain continues to be a vexing and disabling problem affecting the vast majority of the population at some point in their lives [1]. A small subset of this large segment of the back pain population will continue to suffer chronic disabling pain from degenerative disk disease (DDD) requiring treatment, and this care consumes large resources for patients, payors, and providers [2]. The traditional methods for treatment of the patient with symptomatic degenerative disk disease of the lumbar spine involves "conservative treatment," consisting of analgesic and anti-inflammatory medications, limitations of activity or "behavioral modification," and active muscle exercise to recondition and train the truncal musculature. Although there has been progress in the understanding of the pathology underlying DDD and its correlation with pain, the commonality of degenerative changes in the population at large and the still indeterminate

J.P. Lawrence, MD, MBA (🖂)

Division of Orthopedics, Albany Medical College, 1367 Washington Avenue, Albany, NY 12206, USA e-mail: jlawspine@gmail.com

T.J. Albert, MD Department of Orthopedics, Thomas Jefferson University and Hospitals, 925 Chestnut St, Philadelphia, PA 19107, USA e-mail: tjsurg@aol.com

nature of the "pain generator" in DDD obscure the clarity of the disease as well as the pathway toward treatment recommendations. Assessment of patient pathology is of course critical in the recommendation of a surgical option, and the decision to operate should represent the best understanding of the underlying pathology and its consequences, the belief that operative treatment will be efficacious in itself, and the knowledge that the choice of procedure will be preferable to other treatment alternatives for that patient. An appropriately thorough history and physical examination, assessment of imaging modalities, and tailoring of preoperative patient care should provide the most informed body of information on which to base appropriate patient selection for surgery in the setting of symptomatic DDD. However, appropriate patient selection remains a difficult endeavor that reflects both the art and science of medicine. While the surgical options for DDD have expanded to include diskectomy and fusion, arthroplasty, posterior dynamic stabilization, nucleus replacement, and other alternatives, the identification of the appropriate patient (or patients) for these procedures remains a challenging process.

# 20.2 Pathology of DDD

The challenging aspects of DDD that affect the difficulty in recommending surgery stem from the fact that the radiographic changes resulting

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016 J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_20

from the very same findings can occur frequently in asymptomatic patients [3]. Particularly as it concerns an older population, the disk space narrowing, endplate sclerosis, formation of osteophytes, and other radiographic changes characteristic of DDD can occur typically as part of the aging process. It is critical, therefore, that pathologic findings on imaging are distinguished from normal findings and correlated with patient symptoms.

#### 20.3 Patient Evaluation

#### 20.3.1 History

Beyond the identification of patients with symptomatic DDD unresponsive to conservative care, there are numerous aspects of the patient history and course that are predictive of favorable and unfavorable outcomes with surgical intervention. The patient evaluation should provide not only a comprehensive history but should also serve to alert the physician to other items on the differential diagnosis and provide insight into the potential for recovery. Furthermore, the history should serve to eliminate any issues relating to secondary gain.

As many as 3 % of patients with low back pain presenting to the musculoskeletal specialist have a non-spinal cause requiring identification and medical evaluation [4]. Gastric ulceration, abdominal aortic aneurysm, renal disorders, and pancreatitis can all present with back pain and will require attention from appropriate specialists. Red flags for a more extensive generalized patient workup include unexpected weight loss, nocturnal sweating, fevers, or generalized malaise. Any history of previous surgery involving the abdomen or lumbar region should be reviewed.

The patient's previous treatment and treatment response should be carefully delineated. Previous therapeutic modalities including medical treatment, physical therapy, and injections (including facet and medial bundle branch blocks, selective nerve root blocks, and epidurals) should be documented. Notes describing the specific nature of these procedures thus not relying only on patient recollection are extremely helpful. The response to these interventions can reflect patient compliance to prescribed treatments and may serve to estimate the likelihood of improvement with surgery.

#### 20.3.2 Examination

A thorough examination should be performed to identify areas of pain, assess patient ROM, identify areas of neurologic impairment indicative of compressive pathology, and help rule out other disease entities which may result in similar complaints (sacroiliac disease, hip osteoarthritis).

Patients with symptomatic DDD generally have a relatively nonspecific examination devoid of focal posterior tenderness, although muscle spasm may be present. Generally, there is a reduction in flexion-extension, rotation, and lateral bending motion limited by pain. Motor, sensory, and reflex changes are not typically seen. Tension signs are typically negative. Waddell's criteria [5], involving a series of tests designed to indicate nonorganic causes of low back pain, are also helpful to identify.

The patient with neurologic impairment (radicular or myelopathic) involving motor weakness, sensory impairment, or gait instability requires a well-documented examination and comprehensive imaging to localize and define ongoing compressive pathology and differentiate from isolated intradiskal pathology unlikely to result in neurologic impairment.

Based on observational studies, favorable predictors of success with spinal surgery include a high degree of motivation, an absence of active psychopathology, lack of workmen's compensation status, and an absence of tobacco and narcotic use [6, 7]. The patient's psychological profile, although a significant determinant of outcome [8, 9], can be difficult to correlate with surgical outcomes, even using appropriate psychological testing such as the Minnesota Multiphasic Personality Inventory (MMPI) index [10].

#### 20.4 Diagnostic Imaging

Pathologic characteristics identifiable on imaging that predispose to a favorable surgical outcomes include focal disease at one or two levels, significant disk collapse, the presence of Modic changes, and a positive provocative diskogram with negative surrounding control levels.

#### 20.4.1 Radiographs

The initial imaging modality used in the patient with low back pain who fails to demonstrate improvement by approximately 6 weeks are standing plain radiographs. These radiographs serve to define the sagittal and coronal alignment of the lumbar spine, assess the functional disk height, and identify visible pathology such as spondylolysis (Fig. 20.1), spondylolisthesis (Fig. 20.2), or chronic degenerative changes (vertebral body osteophytes, Modic changes). For the patient under consideration for surgery (in particular for interbody arthroplasty), the radiographs should also be examined to assess morphology of the vertebral endplates, as the contour of the endplates may affect the implant bone interface and predispose to or malalignment or subsidence if surgery is contemplated. Flexionextension radiographs have not been shown to readily identify pathology not recognizable on plain films if employed routinely for patients with isolated lumbar complaints [11]. However, they are useful in the evaluation of patients with risk factors for segmental instability (i.e., prior trauma, prior posterior decompression, congenital anomalies, or inflammatory disease such as rheumatoid arthritis) and in providing detailed quantitative information on the motion segment in the preoperative patient (Fig. 20.3).

#### 20.4.2 Computed Tomography

CT scans, although also not employed as part of the routine evaluation of the patient with isolated lumbar complaints, provide useful information in



**Fig. 20.1** Lateral radiograph demonstrating defect of the pars interarticularis (*arrow*) (spondylolysis)



Fig. 20.2 Lateral radiograph demonstrating a degenerative spondylolisthesis

the preoperative assessment of the patient with recalcitrant DDD. The excellent osseous imaging capacity of CT is most useful in illustrating the



**Fig. 20.3** Flexion-extension lateral radiographs in a patient with chronic LBP. Note the disk collapse and loss of height at L5–S1 in the absence of segmental instability

presence of facet joint degeneration (Fig. 20.4) and fractures of the pars interarticularis (Fig. 20.5), which can compromise the use of anterior-based motion-sparing procedures. Furthermore, CT scanning can provide an indirect assessment of bone density as well as provide detailed measurements of the bony anatomy for implant sizing (vertebral body and pedicular anatomy). Quantitative CT may serve as an alternative to bone scintigraphy in the assessment of bone density, although there is a larger dose of radiation imparted to the patient.

#### 20.4.3 MRI

MRI provides great imaging detail on both normal anatomic and abnormal pathologic entities that can affect decision-making (Fig. 20.6). The identification of diskal pathology (herniation, annular tears, dehydration), degenerative changes (vertebral endplate edema, facet joint degenerative



**Fig. 20.4** Axial CT scan demonstrating L4–5 facet joint arthritis on the right side

pathology), or other compressive pathologies (ligamentum flavum infolding, facet hypertrophy) has all been greatly enhanced by the routine use of MRI. As detailed by Boden et al. [3], however, abnormal findings on MRI (such as disk degeneration, annular fissures, small protrusions, and



**Fig. 20.5** Axial CT scan demonstrating bilateral fractures of the pars interarticularis (*arrows*)



**Fig. 20.6** Sagittal T2-weighted MRI demonstrating the characteristic dehydration and loss of disk height seen with degenerative disk disease, seen here at L5–S1

facet arthritis) can be commonly identified in asymptomatic patients. This phenomenon requires that a measured evaluation of MR imaging occurs in light of the patient's history, exam, age, and suspected cause of symptoms. The primary advantages of MRI in the evaluation of the lumbar patient lie in the evaluation of compressive pathology, the staging of disk degeneration, and the evaluation of facet joint degenerative changes. No validated staging system exists, however, for degenerative disk disease using MR imaging.

#### 20.4.4 Bone Scintigraphy

Bone scanning using either dual-energy X-ray absorptiometry (DEXA), dual photon absorptiometry (DPA), or quantitative CT should be performed in all patients suspected of osteopenia or osteoporosis, particularly postmenopausal women, men over age 50, and chronic smokers. DEXA scan has become the standard modality used for this purpose due to its relative low cost and lower dose of imparted radiation versus the alternatives.

## 20.4.5 Diskography

As discussed in an earlier chapter, the usefulness of diskography is primarily as an adjunct to the clinical and diagnostic imaging evaluation. The current limited recommendations for diskography include the definition of the symptomatic level in the patient with multilevel pathology and in distinguishing between spinal and extraspinal causes of pain. Although numerous studies have questioned the positive predictive value of diskography [12-14], it remains the sole modality to permit identification of an intradiskal pain generator, and it can allow for the determination of the symptomatic level in a patient with multilevel pathology. It is important to note that the adequate pressurization of the disk requires annular competency, and an incompetent annulus may result in false-negative results using diskography. More recent data has demonstrated that even using modern administration techniques, diskography has deleterious consequences to the nuclear cells, the annulus, and the disk itself [15]. And although the mechanical insult of the needle introduction and pressurization alone have been shown to contribute to the insult, both radiocontrast and local anesthetics have been shown to have markedly adverse consequences for the diskal environment [16, 17], While the highest yield for diskography may be in the symptomatic back pain patient who has undergone all methods of conservative treatment without relief and has evidence of isolated degenerative disk disease in the absence of other pathology, the risks of the procedure, its limited diagnostic yield, and its known side effects have largely relegated it to the historical realm.

# 20.4.6 Pain Management for Diagnostic and Therapeutic Purposes

Lumbar facet blocks (both with and without steroids), medial branch blocks, and radiofrequency ablation are techniques employed for both the diagnosis and treatment of disorders attributable to the zygapophyseal joints resulting in low back pain. Furthermore, these techniques have been expanded to attempt pain relief of low back pain attributable to the sacroiliac joints as well. Historical data have shown that facet blocks (both intra-articular and medial branch) provide adequate accuracy, reproducibility and safety in the diagnosis, and management of facet-mediated pain [18]. Interestingly, the addition of steroids may not add to the efficacy of this modality [19]. Based on a Cochrane-style review of the literature, diagnostic anesthetic blocks were given a level-1 recommendation, with level-II values associated with therapeutic medial branch blocks and radiofrequency ablation [20].

A successful diagnostic or therapeutic injection does not appear to accurately predict a successful surgical outcome, however [21]. In fact, one historical study appears to indicate that pain relief with temporary external fixation seems to offer better a better prediction of surgical success when compared with other modalities [22]. This is historical data, surely, but the conclusion is that despite their efficacy in diagnosis and treatment, pain management modalities cannot currently be said to predict surgical success.

#### 20.4.7 Spinal Cord Stimulation (SCS)

SCS has emerged as a valuable technique for the management of chronic back and leg pain and unresolved symptoms after thoracolumbar spine surgery, the termed "failed back surgery syndrome" [23, 24]. Patients suffering from complex regional pain syndrome (CRPS) also may benefit from this technology [25]. The data to support its use for the patient with predominantly low back pain as a result of DDD, however, remains lack-ing. Leg pain as the predominant symptom has been the main if not the only predictor of positive outcomes following SCS to this point [26].

#### 20.5 Clinical Data on Surgery

Clinical data that can be considered concerning surgery for DDD consists of randomized controlled trials (RCT) comparing arthrodesis with conservative care and RCTs comparing various procedures with arthrodesis as the control. Although there are many case series of emerging techniques and devices for diskogenic pain, there is not an existing evidence basis to support posterior interspinous devices or nucleus replacement for the treatment of diskogenic back pain.

The landmark paper published by Fritzell et al. in 2001 was the first to compare surgical treatment with nonoperative treatment of chronic low back pain in a well-matched randomized trial. This study was notable for the improved low back pain outcomes seen in the surgical group (33 %) versus the control (9 %) at 2-year follow-up. Criticisms of this trial include the indeterminate nature of the causative entity of the low back pain seen in the control patients. Diagnosis was made on the basis of history, physical examination, and radiographs alone, and MRI and diskography were not employed. Furthermore, the patients randomized to the surgical group were treated with three different fusion options: posterolateral uninstrumented, posterolateral instrumented with pedicle screws, and circumferential (360°) fusion. Also, for the patients treated conservatively, there was considerable variability in the nonoperative care. No

formal physiotherapy protocol was defined, and the use of injections and other pain modalities (acupuncture) was not standardized [27]. This paper was followed by an RCT from Norway by Brox et al. comparing posterolateral arthrodesis with cognitive intervention and exercise for chronic low back pain [28]. The authors used a more defined nonoperative treatment evaluation employing patient education, range of motion exercises, and intensive physiotherapy. At 1-year follow-up, there were no noted differences in improvement in low back pain or in disability as measured using the Oswestry Disability Index (ODI). Fairbank and colleagues performed a similar RCT comparing surgery with conservative treatment and found reductions in disability in both groups 2 years after treatment, but minimal differences using the ODI and a shuttle walking test [29]. A systematic review of these trials was performed in 2007 by Mirza et al. The authors noted that surgery, when compared with cognitive-behavioral therapy and a highly structured rehabilitation protocol for "diskogenic" back pain, resulted in only a modest improvement in back-specific disability [30]. In 2013, Mannion et al. reported long-term follow-up of three randomized trials comparing surgery and nonoperative treatment (multidisciplinary cognitive-behavioral and exercise rehabilitation) for chronic low back pain and found no differences in patient-reported outcomes at an average of 11 years [31].

Numerous RCTs have been performed to assess the efficacy of lumbar arthroplasty for the treatment of diskogenic low back pain, in each case using arthrodesis as a control. In general, the clinical results comparing disk arthroplasty with lumbar fusion have failed to demonstrate significant differences between the interventions in terms of functional outcomes, pain, use of medication, or occupational disability [32–34], with the exception of the investigational device exception (IDE) trial examining the ProDisc-L, which demonstrated marginal superiority of arthroplasty results on the VAS, ODI, and return to employment. However, although the treatment alternatives have been demonstrated to be comparable in this patient population, approximately

half of the clinical trial patients, despite the stringent eligibility criteria, appear to not meet the criteria for clinical success. Furthermore, when following the criteria used for the FDA studies, it becomes apparent that the contraindications to this procedure are common when considering the surgical population in a spine practice, as was demonstrated in a review by Huang et al. [35].

When considering outcomes following any surgical intervention (using health-related quality of life, or HRQOL, instruments), it is important to gauge the magnitude and clinical significance of the score difference and not only the presence of a statistically measurable difference. Using functional scoring scales such as the ODI, for example, the degree of change in the ODI (or ODI delta) may be a more accurate predictor of the functional impact of the intervention. The "minimal clinically important difference" (MCID) has also been described in order to clarify the minimum amount of necessary change in an outcome score to suggest a tangible clinical benefit to the patient [36]. For the ODI, the MCID has been defined variously as ranging from 5.2 to 16.3 historically [37, 38]. Copay, Glassman, et al. performed a prospective study of patients gathered from the Lumbar Spine Study Group and determined the MCID for patients as assessed by the ODI to be 12.8 points for patients undergoing lumbar spinal surgery [39].

It is in this context, then, that the more recent trials for lumbar surgery, including trials examining surgery for DDD, should be viewed. Weinstein et al., in the degenerative spondylolisthesis arm of the SPORT (Spine Patient Outcomes Research Trial) trial, reported a mean improvement in the ODI of -24.2 in the surgery group at 2 years, versus -7.5 in the nonoperative group, for a treatment effect of -16.7 (in light of a MCID of 12.8). In the larger trials of lumbar arthroplasty, the ODI has served as a critical tool in the functional outcome assessment and as a statistical piece of the overall assessment of clinical success. The degree of the clinical effect, however, remains an issue of discussion. In the pivotal trial of the Charite prosthesis, Blumenthal et al. [34] noted changes in the arthroplasty group from a mean preoperative ODI score of 50.6-37.7 (-12.9), 29.9 (-20.7), 27.5

(-23.1), 26(-24.6), and 26.3(-24.3) at 6 weeks and 3, 6, 9, 12, and 24 months, respectively. In the pivotal trial of the ProDisc-L implant, Zigler et al. also used the ODI as part of functional outcome assessment the FDA IDE study [40]. The investigational group had a mean initial ODI of 63.4 (reflecting use of an alternative ODI scale, which can result in a higher disability core), 25.3 % higher than in the Charite trial. At intervals of 6 weeks, 3 months, and 6 months, the authors reported statistically significant differences from control in the ODI and a trend toward significance at 24 months. Versus the baseline values, although precise data were not published, it appears that the range of mean ODI following arthroplasty was between 34.5 and 47, with a value (34.5) provided at 24 months. From baseline ODI, this value reflects a -28.9 point change (46.1 %). The FDA criteria for outcomes were specified to include a decrease of 15 % on the ODI for "ODI success." Although there was a significant decrease following arthroplasty (and fusion, which resulted in a mean ODI of 39.8 at 24 months, for a change of -22.9 points, or 36 %), the patients remained with some degree of disability as reflected by an ODI of 34.5 at 24 months in the investigational group. Furthermore, the differences in the initial disability between the two trials demonstrate that the administration of functional scoring scales has some inherent variability depending on the content and application of the instrument. These data would suggest that the procedure (and the control), although they meet the statistical criteria, the specified FDA criteria, and MCID criteria for success of the intervention (and the control), results in improvement with a remnant of continued functional disability (mean ODI of 26.3 in the Charite trial and 34.5 in the ProDisc trial).

#### 20.6 Summary

Patient selection for the surgical treatment of degenerative disk disease involves an appropriate and thorough evaluation, an understanding of the pathology and pathoanatomy, a recognition of responsiveness to previous treatments and motivation, a discussion of the appropriate operative course, and a clarification of the expectations following surgery both for the surgeon and the patient.

As previously described, degenerative changes of the intervertebral disk or the finding on MR imaging of a so-called black disk can simply be a feature of normal aging in the asymptomatic patient [3] or can represent a pathologic entity capable of causing disabling pain and dysfunction. The keys in the differentiation between these populations involve the detailed history, exam, appropriate imaging, and the judicious usage of targeted pain management procedures. Diskography appears to have been largely invalidated. Ultimately, there is no definable role for invasive procedures in the patient with a "black disk" in the absence of a correlative history, exam, and unsuccessful conservative care. Rather, a combination of the pathoanatomybased approach (seeking the pain generator) and the psychosocial approach (including the psychosocial components that contribute to LBP) should be included in the evaluation of the disabled low back pain patient. In general terms, surgery should not be considered unless there is significant functional disability, a prolonged period of pain (>6 months), and a failure of improvement after a significant period of conservative treatment. The history, physical examination, patient course, and imaging modalities should reflect the suspected disease cause, and other causes of pain should be eliminated. Issues of secondary gain, active psychological or psychiatric disease, low motivation, workmen's compensation status, or a history of tobacco/narcotic use are less likely to result in positive outcomes with surgery. Once the decision has been made to explore a surgical option, diskography, targeted injections, and a careful assessment of the imaging to identify pathoanatomy unsuitable for motion-sparing procedures (facet arthrosis, spondylolysis, endplate disease, or segmental instability) can assist in the decision among surgical alternatives: diskectomy with arthrodesis, intradiskal techniques, or a motion-sparing alternative. An understanding of the evidence basis for surgical treatment and the factors predisposing to favorable and unfavorable outcomes can be critical in improving patient

selection and future outcomes, and over time more information should become available to assist both patient and surgeon in this regard. Currently, the clinical data suggests that the outcomes of all surgical treatment for degenerative disk disease are less than optimal. It appears that even in the mostly rigorously controlled trials of surgery for the best indications, the outcomes result in functional improvement yet a resultant degree of functional impairment. Therefore, caution should be maintained in the consideration of surgery for degenerative disk disease. Lastly, surgeons should be cognizant of their particular skill set and present options to the patient accordingly.

#### References

- Waddell G. Low back pain: a twentieth-century enigma. Spine. 1996;21(24):2820–5.
- Frymoyer JW, Montgomery DM. Diagnosis and treatment of discogenic low back pain. Orthop Clin North Am. 1991;22(2):263–71.
- Boden SD, Davis DO, Dina TS, Patronas NJ, Wiesel SW. Abnormal magnetic-resonance scans of the lumbar spine in asymptomatic subjects. A prospective investigation. J Bone Joint Surg Am. 1990;72(3):403–8.
- Hanley Jr EN, Levy JA. Surgical treatment of isthmic lumbosacral spondylolisthesis. Spine. 1989;22(24):2959–67.
- Waddell G, McCulloch JA, Kummel E, Venner RM. Nonorganic physical signs in low-back pain. Spine. 1980;5:117–25.
- Parker LM, Murrell SE, Boden SD, Horton WC. The outcome of posterolateral fusion in highly selected patients with discogenic back pain. Spine. 1996;21(16):1909–16.
- Moon MS. The outcome of posterolateral fusion in highly selected patients with discogenic low back pain. Spine. 1997;22(12):1419–20.
- Carragee EJ, Alamin TF, Miller JL, Carragee JM. Discographic, MRI and psychosocial determinants of low back pain disability and remission: a prospective study in subjects with benign persistent back pain. Spine J. 2005;5(1):24–35.
- Burton AK, Tillotson KM, Main CJ, Hollis S. Psychosocial predictors of outcome in acute and subchronic low back trouble. Spine. 1995;20(6):722–8.
- Bieliauskas LA, Graziano GP, Kullgren K, Roper BL. Failed back surgery and Minnesota multiphasic personality inventory (MMPI) profiles. J Clin Psychol Med Settings. 1994;1(2):1068–75.

- Hammouri QM, Haims AH, Simpson AK, Alqaqa A, Grauer JN. The utility of dynamic flexion-extension radiographs in the initial evaluation of the degenerative lumbar spine. Spine. 2007;32(21):2361–4.
- Carragee EJ, Barcohana B, Alamin T, van den Haak E. Prospective controlled study of the development of lower back pain in previously asymptomatic subjects undergoing experimental discography. Spine. 2004;29(10):1112–7.
- Carragee EJ, Alamin TF, Carragee JM. Low-pressure positive discography in subjects asymptomatic of significant low back pain illness. Spine. 2006;31(5):505–9.
- Carragee EJ, Lincoln T, Palmar VS, Alamin T. A gold standard evaluation of the "discogenic pain" diagnosis as determined by provocative discography. Spine. 2006;31(18):2115–23.
- Carragee EJ, Don AS, Hurwitz EL, Cuellar JM, Carrino JA, Herzog R. ISSLS prize winner: does discography cause accelerated progression of degeneration changes in the lumbar disc: a ten-year matched cohort study. Spine. 2009;34(21):2338–45.
- 16. Gruber HE, Rhyne 3rd AL, Hansen KJ, Phillips RC, Hoelscher GL, Ingram JA, Norton HJ, Hanley EN, Hanley Jr EN. Deleterious effects of discography radiocontrast solution on human annulus cell in vitro: changes in cell viability, proliferation, and apoptosis in exposed cells. Spine J. 2012;12(4):329–35.
- Eder C, Pinsger A, Schildboeck S, Falkner E, Becker P, Ogon M. The influence of intradiscal medication on nucleus pulposus cells. Spine J. 2013;13:1556–62. S1529-943000316-1.
- Sehgal N, Dunbar EE, Shah RV, Colson J. Systematic review of diagnostic utility of facet (zygapophysial) joint injections in chronic spinal pain: an update. Pain Physician. 2007;10(1):213–28.
- Manchikanti L, Singh V, Falco FJ, Cash KA, Pampati V. Evaluation of lumbar facet joint nerve blocks in managing chronic low back pain: a randomized, double-blind, controlled trial with a 2-year follow-up. Int J Med Sci. 2010;7(3):124–35.
- Datta S, Lee M, Falco FJ, Bryce DA, Hayek SM. Systematic assessment of diagnostic accuracy and therapeutic utility of lumbar facet joint interventions. Pain Physician. 2009;12(2):437–60.
- Cohen SP, Hurley RW. The ability of diagnostic spinal injections to predict surgical outcomes. Anesth Analg. 2007;105(6):1756–75.
- Esses SI, Botsford DJ, Kostuik JP. The role of external spinal skeletal fixation in the assessment of low-back disorders. Spine. 1989;14(6):594–601.
- Kumar K, Taylor RS, Jacques L. Spinal cord stimulation versus conventional medical management for neuropathic pain: a multicentre randomised controlled trial in patients with failed back surgery syndrome. Pain. 2007;132:179–88.
- North RB, Kidd DH, Farrokhi F, Piantadosi SA. Spinal cord stimulation versus repeated lumbosacral spine surgery for chronic pain: a randomized, controlled trial. Neurosurgery. 2005;56:98–106.

- Simpson EL, Duenas A, Holmes MW, Papaioannou D, Chilcott J. Spinal cord stimulation for chronic pain of neuropathic or ischaemic origin: systematic review and economic evaluation. Health Technol Assess. 2009;13(17):1–154.
- 26. Taylor RS, Desai MJ, Rigoard P, Taylor RJ. Predictors of pain relief following spinal cord stimulation in chronic back and leg pain and failed back surgery syndrome: a systematic review and meta-regression analysis. Pain Pract. 2013. doi:10.1111/papr.12095 [Epub].
- 27. Fritzell P, Hagg O, Wessberg P, Nordwall A, Group SLSS. 2001 Volvo award winner in clinical studies: lumbar fusion versus nonsurgical treatment for chronic low back pain: a multicenter randomized controlled trial from the Swedish Lumbar Spine Study Group. Spine. 2001;26:2521–32.
- Brox JI, Sorensen R, Friis A, et al. Randomized clinical trial of lumbar instrumented fusion and cognitive intervention and exercises in patients with chronic low back pain and disc degeneration. Spine. 2003;28:1913–21.
- 29. Fairbank J, Frost H, Wilson-MacDonald J, Yu LM, Barker K, Collins R. Randomised controlled trial to compare surgical stabilisation of the lumbar spine with an intensive rehabilitation programme for patients with chronic low back pain: the MRC spine stabilisation trial. BMJ. 2005;330(7502):1233.
- Mirza SK, Deyo RA. Systematic review of randomized trials comparing lumbar fusion surgery to nonoperative care for treatment of chronic back pain. Spine. 2007;32(7):816–23.
- Mannion AF, Brox JL, Fairbank JC. Comparison of spinal fusion and nonoperative treatment in patients with chronic low back pain: long-term follow-up of three randomized controlled trials. Spine J. 2013;13(11):1438–48.
- 32. Freeman BJ, Davenport J. Total disc replacement in the lumbar spine: a systematic review of the literature. Eur Spine J. 2006;15(S3):S439–47.

- 33. Delamarter RB, Bae HW, Pradhan BB. Clinical results of ProDisc-II lumbar total disc replacement: report from the United States clinical trial. Orthop Clin North Am. 2005;36(3):301–13.
- 34. Blumenthal S, McAfee PC, Guyer RD, et al. A prospective, randomized, multicenter Food and Drug Administration investigational device exemptions study of lumbar total disc replacement with the CHARITE artificial disc versus lumbar fusion: part I: evaluation of clinical outcomes. Spine. 2005;30(14):1565–75.
- Huang RC, Lim MR, Girardi FP, Cammisa Jr FP. The prevalence of contraindications to total disc replacement in a cohort of lumbar surgical patients. Spine. 2004;29(22):2538–41.
- Jaeschke R, Singer J, Guyatt GH. Measurement of health status. Ascertaining the minimal clinically important difference. Control Clin Trials. 1989;10:407–15.
- 37. Suarez-Almazor ME, Kendall C, Johnson JA, Skeith K, Vincent D. Use of health status measures in patients with low back pain in clinical settings: comparison of specific, generic and preference-based instruments. Rheumatology. 2000;39:783–90.
- Taylor SJ, Taylor AE, Foy MA, Fogg AJ. Responsiveness of common outcome measures for patients with low back pain. Spine. 1999;24:1805–12.
- 39. Copay AG, Glassman SD, Subach BR, Berven S, Schuler TC, Carreon LY. The minimum clinically important difference in lumbar spine surgery patients: a choice of methods using the Oswestry disability index, medical outcomes study questionnaire short form 36, and pain scales. Spine J. 2008;8(6):968–74.
- 40. Zigler J, Delamarter R, Spivak JM, Linovitz RJ, et al. Results of the prospective, randomized, multicenter Food and Drug Administration investigational device exemption study of the ProDisc-L total disc replacement versus circumferential fusion for the treatment of 1-level degenerative disc disease. Spine. 2007;32(11):1155–62.

# To Fuse or Not to Fuse: That's the Question

## Ryan P. Ponton, Eric B. Harris, and Alan Hilibrand

#### 21.1 Introduction

Over the last few decades, significant advances have been made with regard to fusion technologies in the lumbar spine. The advent of pedicle screw fixation, lateral interbody grafting techniques, recombinant human bone morphogenetic proteins, and minimally invasive surgical approaches has resulted in a significant increase in the number of fusion procedures annually [1–4]. As an aging, active population continues to expand, more patients seek appropriate therapies to alleviate pain, restore function, and maintain their active lifestyles [5]. While the technologies and techniques available to the spine surgeon have improved and expanded, the debate continues over the appropriate indications for fusion in the setting of lumbar degenerative disk disease [4, 6]. The purpose of this chapter is to review current evidence for lumbar spinal fusion procedures and provide recommendations for eval-

R.P. Ponton, MD • E.B. Harris, MD (⊠) Department of Orthopaedic Surgery, Naval Medical Center San Diego, 34800 Bob Wilson Drive, San Diego, CA 92134-5000, USA e-mail: ryan.ponton@med.navy.mil; ericbharris@gmail.com; eric.harris@med.navy.mil

A. Hilibrand, MD

Professor of Orthopaedic Surgery at Jefferson Medical College, Director of Orthopaedic Medical Education, Professor of Neurosurgery at Jefferson Medical College, 25 Chestnut Street 5th Floor, Philadelphia, PA 19107, USA e-mail: ahilibrand@gmail.com uation, clinical workup, and surgical decision making in this challenging patient population.

#### 21.2 Overview

Lumbar degenerative disease can present with a multitude of symptoms, so it is important to distinguish between various pathologic processes. An appropriate trial of nonoperative therapy should always be attempted before consideration of surgical intervention. If the patient's symptoms persist, imaging studies as well as other diagnostic testing must be carefully reviewed prior to formulating a surgical plan.

Lumbar degenerative disease clinically manifests in five basic forms: axial low back pain, spinal stenosis, radiculopathy, degenerative spondylolisthesis, and degenerative scoliosis. These conditions may occur alone or in combination with one another. Surgical decision making, with regard to the need for fusion, for any of these entities depends upon multiple factors, which will be discussed in relation to each disease process.

#### 21.3 Axial Low Back Pain

Chronic axial low back pain has a high socioeconomic impact, with a lifetime prevalence of 30–50 % for moderate and severe back pain [7]. Pure axial low back pain without deformity is successfully treated nonsurgically in the vast

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_21

majority of cases [8]. Nonoperative modalities include medications, short-term bed rest, physical therapy, and chiropractic therapy. Medications include nonsteroidal anti-inflammatories (NSAIDs), acetaminophen, muscle relaxants, and short durations of narcotics.

After an initial trial with one of the above treatments, epidural and/or facet injections may be useful. In patients with symptoms refractory to nonoperative management for greater than 6 months, fusion may be indicated. However, there is a lack of consensus regarding the efficacy of fusion for discogenic axial back pain [9]. Multiple studies have demonstrated only marginal improvement in back pain following fusion [10–14]. Because of these less than optimal results, many surgeons advocate diagnostic testing to attempt to isolate the pain generator prior to surgical intervention [15]. Provocative discography is the most common test utilized in this scenario, although there is debate in the literature over whether it actually predicts outcomes of spinal fusion [16–19].

In a 2006 comparison of single-level fusion patients with axial low back pain and concordant discogram to a matched control group of patients with unstable spondylolisthesis, Carragee et al. reported a dismal 43 % satisfactory outcome in the discography group at 2 years compared to 92 % in the controls [19].

In a recent systematic review, no subset of patient with chronic back pain could be identified for whom spinal fusion is a predictable and effective treatment. They concluded best evidence does not support the use of provocative discography for patient selection in clinical practice [20]. Clearly, there is lack of agreement upon whether concordant discography can predict the outcome of a fusion for axial low back pain. Due to this fact, it is important to keep in mind that discography is merely one diagnostic component to consider when developing a treatment plan for this challenging problem.

We consider all the factors in the axial low back pain patient's case and, in this subset of patients, would only recommend fusion after failure of a minimum of 6 months nonoperative therapy. We utilize provocative discography, performed by an experienced pain management specialist, in patient who have failed this regimen and have radiographic evidence of single-level degenerative disease. We offer surgery only if a concordant disk correlates with MRI findings and adjacent disks are non-concordant [21].

#### 21.4 Lumbar Spinal Stenosis

Unlike axial low back pain, the question of whether to fuse in cases of lumbar spinal stenosis (LSS) is somewhat more straightforward. If nonoperative measures fail to relieve symptoms in this subset of patients, most obtain meaningful relief of symptoms from surgical decompression [22–24]. Patients with pure stenosis in the absence of degenerative spondylolisthesis or scoliosis may be treated with decompression alone. Numerous studies have shown excellent outcomes in this patient population with regard to relief of leg pain and paresthesias, lower extremity weakness, and walking tolerance [22–24].

LSS patients who demonstrate radiographic evidence of instability should be strongly considered for a fusion procedure [25, 26]. Gross instability may be detected on plain lateral radiographs in the form of spondylolisthesis that translates with active flexion or extension. Additionally, lateral listhesis seen on AP images may also indicate instability. Finally, excessive facet joint fluid seen on T2-weighted trans-axial MRI images is a sign of a more subtle, but clinically significant, instability. Patients with such instability, who undergo decompression alone, are more likely to experience worsening instability and back pain secondary to the loss of posterior stabilizing elements including the supraspinous ligament, intra-spinous ligament, ligamentum flavum, and varying amounts of the facet joints and pars interarticularis. Even in patients without preoperative instability, removal of greater than 50 % of the facet joint or pars interarticularis can lead to iatrogenic instability resulting in poor outcomes and potential need for further surgery [27].

One recent study evaluated nationwide trends in the surgical management of patients with lumbar spinal stenosis (LSS) with and without coexisting spondylolisthesis and scoliosis from 2004 to 2009. They demonstrated simple fusion surgery has increased for treatment of LSS compared with decompression only [4].

As previously mentioned, a clinician must take into account multiple patient factors, including obesity, which has been shown to negatively impact lumbar surgery. One study demonstrated inferior results of surgery for lumbar stenosis, [28] while another demonstrated longer operative times and an increased rate of infection [29].

In today's era of advanced imaging studies, it is important to remember to obtain good anteroposterior, lateral, and flexion/extension lumbar spine radiographs to rule out degenerative scoliosis or spondylolisthesis. Both of these entities can easily be missed if only an MRI is examined during preoperative planning. If scoliosis or spondylolisthesis is identified in a patient with symptomatic spinal stenosis who has failed nonoperative treatment, it is our typical practice to recommend decompression and fusion in addition to at the scoliotic or unstable levels. Patients undergoing this procedure should be extensively counseled that the purpose of the procedure is to relieve lower extremity symptoms and that relief of back pain is less predictable.

#### 21.5 Lumbar Radiculopathy

If nonoperative management fails, radiculopathy secondary to herniated nucleus pulposus with or without coexisting stenosis is reliably treated with decompression through laminoforaminotomy and excision of the disk fragment [30]. As with symptomatic lumbar stenosis, we recommend fusion for patients with lumbar radiculopathy only in the setting of degenerative scoliosis or spondylolisthesis.

Some spine surgeons believe that fusion is necessary for treating disk reherniation. As repeated diskectomy for either ipsilateral or contralateral recurrence requires the removal of more disk material and posterior elements, such as lamina or facet joint, further invasion at the same surgical level can increase the risk of segmental instability [31]. A recent study evaluated the treatment patterns for recurrent lumbar disk herniation among US spine surgeons. The study highlighted the lack of consensus with regard to treatment plan, as some surgeons preferred repeat microdiskectomy, while others preferred microdiskectomy with posterior lumbar interbody fusion/transforaminal lumbar interbody fusion (PLIF/TLIF) [32].

Rarely, patients with lumbar radiculopathy may also have a significant component of low back pain. In these patients, if there is clear evidence via discography or single-level disk degeneration on MRI that the back pain may be referable to the same level as the source of radiculopathy, concomitant fusion may be contemplated. However, it should be noted that studies have shown that such back pain often resolves with diskectomy alone [30, 33].

Among patients with radiculopathy who have a degenerative spondylolisthesis or degenerative scoliosis on imaging studies, decompression with fusion should be strongly considered, regardless of whether or not a significant back pain component is present. Failure to identify these instability patterns and address them at time of surgery may lead to progressive instability and pain.

# 21.6 Degenerative Spondylolisthesis

Surgical treatment, consisting of decompression and fusion with or without instrumentation, has been shown to be an effective treatment of symptomatic degenerative spondylolisthesis [25, 34, 35]. Numerous retrospective and prospective studies have demonstrated satisfactory results with a variety of fusion techniques [35]. Standing flexion and extension radiographs may be the key images necessary to identify this diagnosis as it is often a dynamic rather than a static process and may be missed with MRI alone. Advanced imaging studies should be obtained to identify any areas of stenosis to ensure an adequate decompression and fusion may then be limited to the levels demonstrating instability.

#### 21.7 Degenerative Scoliosis

With an aging population in the United States and an increased attention to quality of life versus cost issues, degenerative scoliosis has become a considerable health concern [36, 37]. Many female patients with mechanical low back pain over the age of 60 will have a degenerative form of lumbar scoliosis. Most of these patients do not have significant radicular or claudication symptoms; therefore surgical treatment is not indicated. However, when these patients present with symptoms of stenosis or radiculopathy unresponsive management, decompression with instrumented fusion is appropriate. In addition to the scoliosis, advanced imaging studies will usually reveal disk degeneration, foraminal stenosis, and nerve root compression. Many of these patients will have a rotatory component to their deformity that may also be addressed with reduction maneuvers utilizing a pedicle screw construct. If there are a large rotatory component and significant lateral listhesis causing a broad curve across several levels, these authors will sometimes recommend a circumferential fusion technique to restore anterior column integrity, obtain better curve correction, and hopefully improve longterm outcomes [38–42].

Despite the risks of scoliotic surgery, there is an increasing demand for operative intervention in this population [43–45]. A recent systemic review analyzed 16 studies including 553 patients with DS and reported that despite an overall 49.0 % complication rate and 15.3 % rate of repeat procedures, surgery is still an effective and reasonable treatment option for DS [46].

#### Conclusions

In the absence of scoliosis or spondylolisthesis, fusion is typically not indicated in the treatment of degenerative disease in the lumbar spine. In the rare case of single-level degeneration with concordant discography, many surgeons would consider fusion. Patients with recurrent disk herniations above L5–S1 may also benefit from fusion at reoperation, although no literature exists to support this theory. Finally, numerous studies have shown that fusion in patients with degenerative scoliosis or spondylolisthesis is not only beneficial from an outcomes standpoint but usually required to prevent further instability and deformity.

#### References

- Burkus JK, Sandhu HS, Gornet MF, Longley MC. Use of rhBMP-2 in combination with structural cortical allografts: clinical and radiographic outcomes in anterior lumbar spinal surgery. J Bone Joint Surg Am. 2005;87:1205–12.
- Gaines Jr RW. The use of pedicle-screw internal fixation for the operative treatment of spinal disorders. J Bone Joint Surg Am. 2000;82-A:1458–76.
- West 3rd JL, Bradford DS, Ogilvie JW. Results of spinal arthrodesis with pedicle screw-plate fixation. J Bone Joint Surg Am. 1991;73:1179–84.
- Bae HW, Rajaee SS, Kanim LE. Nationwide trends in the surgical management of lumbar spinal stenosis. Spine (Phila Pa 1976). 2013;38:916–26.
- Moorin RE, Holman CD. The impact of the evolution of invasive surgical procedures for low back pain: a population based study of patient outcomes and hospital utilization. ANZ J Surg. 2009;79:610–8.
- Asghar FA, Hilibrand AS. The impact of the Spine Patient Outcomes Research Trial (SPORT) results on orthopaedic practice. J Am Acad Orthop Surg. 2012;20:160–6.
- Forster M, Mahn F, Gockel U, et al. Axial low back pain: one painful area-many perceptions and mechanisms. PLoS One. 2013;8:e68273.
- Lu Y, Guzman JZ, Purmessur D, et al. Nonoperative management of discogenic back pain: a systematic review. Spine (Phila Pa 1976). 2014;39:1314–24.
- Mirza SK, Deyo RA, Heagerty PJ, Turner JA, Martin BI, Comstock BA. One-year outcomes of surgical versus nonsurgical treatments for discogenic back pain: a community-based prospective cohort study. Spine J. 2013;13:1421–33.
- Hanley Jr EN, David SM. Lumbar arthrodesis for the treatment of back pain. J Bone Joint Surg Am. 1999;81:716–30.
- Grubb SA, Lipscomb HJ. Results of lumbosacral fusion for degenerative disc disease with and without instrumentation. Two- to five-year follow-up. Spine (Phila Pa 1976). 1992;17:349–55.
- Brox JI, Sorensen R, Friis A, et al. Randomized clinical trial of lumbar instrumented fusion and cognitive intervention and exercises in patients with chronic low back pain and disc degeneration. Spine (Phila Pa 1976). 2003;28:1913–21.
- 13. Fairbank J, Frost H, Wilson-MacDonald J, et al. Randomised controlled trial to compare surgical stabilisation of the lumbar spine with an intensive rehabilitation programme for patients with chronic low

back pain: the MRC spine stabilisation trial. BMJ. 2005;330:1233.

- 14. Fritzell P, Hagg O, Wessberg P, Nordwall A, Swedish Lumbar Spine Study Group. 2001 Volvo Award Winner in Clinical Studies: lumbar fusion versus nonsurgical treatment for chronic low back pain: a multicenter randomized controlled trial from the Swedish Lumbar Spine Study Group. Spine (Phila Pa 1976). 2001;26:2521–32; discussion 2532–4.
- Coppes MH, Marani E, Thomeer RT, Groen GJ. Innervation of "painful" lumbar discs. Spine (Phila Pa 1976). 1997;22:2342–9; discussion 2349–50.
- Colhoun E, McCall IW, Williams L, Cassar Pullicino VN. Provocation discography as a guide to planning operations on the spine. J Bone Joint Surg Br. 1988; 70:267–71.
- Derby R, Howard MW, Grant JM, Lettice JJ, Van Peteghem PK, Ryan DP. The ability of pressurecontrolled discography to predict surgical and nonsurgical outcomes. Spine (Phila Pa 1976). 1999;24: 364–71; discussion 371–2.
- Madan S, Gundanna M, Harley JM, Boeree NR, Sampson M. Does provocative discography screening of discogenic back pain improve surgical outcome? J Spinal Disord Tech. 2002;15:245–51.
- Carragee EJ, Lincoln T, Parmar VS, Alamin T. A gold standard evaluation of the "discogenic pain" diagnosis as determined by provocative discography. Spine (Phila Pa 1976). 2006;31:2115–23.
- Willems PC, Staal JB, Walenkamp GH, de Bie RA. Spinal fusion for chronic low back pain: systematic review on the accuracy of tests for patient selection. Spine J. 2013;13:99–109.
- Parker LM, Murrell SE, Boden SD, Horton WC. The outcome of posterolateral fusion in highly selected patients with discogenic low back pain. Spine (Phila Pa 1976). 1996;21:1909–16; discussion 1916–7.
- 22. Atlas SJ, Keller RB, Wu YA, Deyo RA, Singer DE. Long-term outcomes of surgical and nonsurgical management of lumbar spinal stenosis: 8 to 10 year results from the maine lumbar spine study. Spine (Phila Pa 1976). 2005;30:936–43.
- Weinstein JN, Tosteson TD, Lurie JD, et al. Surgical versus nonsurgical therapy for lumbar spinal stenosis. N Engl J Med. 2008;358:794–810.
- Weinstein JN, Tosteson TD, Lurie JD, et al. Surgical versus nonoperative treatment for lumbar spinal stenosis four-year results of the Spine Patient Outcomes Research Trial. Spine (Phila Pa 1976). 2010;35:1329–38.
- Herkowitz HN, Kurz LT. Degenerative lumbar spondylolisthesis with spinal stenosis. A prospective study comparing decompression with decompression and intertransverse process arthrodesis. J Bone Joint Surg Am. 1991;73:802–8.
- Tosteson AN, Lurie JD, Tosteson TD, et al. Surgical treatment of spinal stenosis with and without degenerative spondylolisthesis: cost-effectiveness after 2 years. Ann Intern Med. 2008;149:845–53.

- Yuan PS, Booth Jr RE, Albert TJ. Nonsurgical and surgical management of lumbar spinal stenosis. Instr Course Lect. 2005;54:303–12.
- Knutsson B, Michaelsson K, Sanden B. Obesity is associated with inferior results after surgery for lumbar spinal stenosis: a study of 2633 patients from the Swedish spine register. Spine (Phila Pa 1976). 2013;38:435–41.
- 29. McGuire KJ, Khaleel MA, Rihn JA, Lurie JD, Zhao W, Weinstein JN. The effect of high obesity on outcomes of treatment for lumbar spinal conditions: sub-group analysis of the spine patient outcomes research trial. Spine (Phila Pa 1976). 2014;39:1975–80.
- Weinstein JN, Lurie JD, Tosteson TD, et al. Surgical vs nonoperative treatment for lumbar disk herniation: the Spine Patient Outcomes Research Trial (SPORT) observational cohort. JAMA. 2006;296:2451–9.
- Chen Z, Zhao J, Liu A, Yuan J, Li Z. Surgical treatment of recurrent lumbar disc herniation by transforaminal lumbar interbody fusion. Int Orthop. 2009;33:197–201.
- 32. Mroz TE, Lubelski D, Williams SK, et al. Differences in the surgical treatment of recurrent lumbar disc herniation among spine surgeons in the United States. Spine J. 2014;14:2334–43.
- Buttermann GR. Treatment of lumbar disc herniation: epidural steroid injection compared with discectomy. A prospective, randomized study. J Bone Joint Surg Am. 2004;86-A:670–9.
- Booth KC, Bridwell KH, Eisenberg BA, Baldus CR, Lenke LG. Minimum 5-year results of degenerative spondylolisthesis treated with decompression and instrumented posterior fusion. Spine (Phila Pa 1976). 1999;24:1721–7.
- 35. Weinstein JN, Lurie JD, Tosteson TD, et al. Surgical compared with nonoperative treatment for lumbar degenerative spondylolisthesis. four-year results in the Spine Patient Outcomes Research Trial (SPORT) randomized and observational cohorts. J Bone Joint Surg Am. 2009;91:1295–304.
- Kotwal S, Pumberger M, Hughes A, Girardi F. Degenerative scoliosis: a review. HSS J. 2011; 7:257–64.
- 37. Fu L, Chang MS, Crandall DG, Revella J. Comparative analysis of clinical outcomes and complications in patients with degenerative scoliosis undergoing primary versus revision surgery. Spine (Phila Pa 1976). 2014;39:805–11.
- Grob D, Scheier HJ, Dvorak J, Siegrist H, Rubeli M, Joller R. Circumferential fusion of the lumbar and lumbosacral spine. Arch Orthop Trauma Surg. 1991;111:20–5.
- Gertzbein SD, Hollopeter M, Hall SD. Analysis of circumferential lumbar fusion outcome in the treatment of degenerative disc disease of the lumbar spine. J Spinal Disord. 1998;11:472–8.
- Hee HT, Castro Jr FP, Majd ME, Holt RT, Myers L. Anterior/posterior lumbar fusion versus transforaminal lumbar interbody fusion: analysis of complications and predictive factors. J Spinal Disord. 2001;14:533–40.

- 41. Soegaard R, Bunger CE, Christiansen T, Hoy K, Eiskjaer SP, Christensen FB. Circumferential fusion is dominant over posterolateral fusion in a long-term perspective: cost-utility evaluation of a randomized controlled trial in severe, chronic low back pain. Spine (Phila Pa 1976). 2007;32: 2405–14.
- 42. Videbaek TS, Christensen FB, Soegaard R, et al. Circumferential fusion improves outcome in comparison with instrumented posterolateral fusion: longterm results of a randomized clinical trial. Spine (Phila Pa 1976). 2006;31:2875–80.
- 43. Li G, Passias P, Kozanek M, et al. Adult scoliosis in patients over sixty-five years of age: outcomes of operative versus nonoperative treatment at a minimum

two-year follow-up. Spine (Phila Pa 1976). 2009;34:2165–70.

- 44. Daubs MD, Lenke LG, Cheh G, Stobbs G, Bridwell KH. Adult spinal deformity surgery: complications and outcomes in patients over age 60. Spine (Phila Pa 1976). 2007;32:2238–44.
- 45. Bridwell KH, Baldus C, Berven S, et al. Changes in radiographic and clinical outcomes with primary treatment adult spinal deformity surgeries from two years to three- to five-years follow-up. Spine (Phila Pa 1976). 2010;35:1849–54.
- 46. Liang CZ, Li FC, Li H, Tao Y, Zhou X, Chen QX. Surgery is an effective and reasonable treatment for degenerative scoliosis: a systematic review. J Int Med Res. 2012;40:399–405.

# Biomechanics of the Lumbar Degenerative Intervertebral Disk

E. Emily Abbott, Bjorn Lobo, and Edward C. Benzel

# 22.1 Introduction

Degenerative lumbar disk disease is one of the most common causes of low back pain and disability in patients above 45 years of age [1, 2]. Diagnosis and treatment of lumbar herniated disks remain perplexing at times. Magnetic resonance imaging (MRI) is the imaging study of choice for the identification of herniated lumbar disks; however, there remains only a moderate correlation between imaging findings and patients symptoms [2]. Disk degeneration is a broad term that includes a variety of changes noted in gross specimens and radiographs from patients with clinical dysfunction. These changes may lead to derangement of the normal biomechanics of the lumbar spine, such as pathological and dysfunctional intervertebral motion - with a commensurate clinical response.

E.C. Benzel, MD  $(\boxtimes)$ 

Department of Neurological Surgery, Center for Spine Health, Neurological Institute, Cleveland Clinic, 9500 Euclid Ave, S-40, Cleveland, OH 44195, USA e-mail: abbotte@ccf.org; lobob@ccf.org; benzele@ccf.org

# 22.2 Structure of the Intervertebral Disk

The human spine contains 23 intervertebral disks (6 cervical, 12 thoracic, and 5 lumbar); there are no disks between the atlas (C1) and axis (C2) or in the sacrum and coccyx. Each intervertebral disk lies between adjacent vertebrae. Cartilaginous endplates surround each disk superiorly and inferiorly; the anterior longitudinal ligament is ventral to each disk and the posterior longitudinal ligament is dorsal. The disk contains fibroblasts, cartilage cells, and few notochord cells and is composed of the nucleus pulposus centrally, annulus fibrosus radially, and annular epiphysis cranially and caudally.

The nucleus pulposus is composed of proteoglycan loosely held together by an unorganized matrix of type II collagen and elastin fibers. The proteoglycan component, mainly aggrecan, is highly anionic and provides a colloid osmotic pressure that pulls water into the nucleus. This is crucial for shock absorption, as the hydrostatic pressure generated from osmotic swelling resists spinal compression [3].

The annulus fibrosus is composed of bands of collagen type I fibers arranged in parallel. The orientations of these fibers are at 30° from the cartilaginous endplates and are reversed in each successive lamella so that in each adjacent lamella, fibers are at 120° angles. The peripheral lamellae are tightly adhered to the ring epiphysis of the

E.E. Abbott, MD • B. Lobo, MD

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016 J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_22



**Fig. 22.1** Annular fibers are oriented in a  $30^{\circ}$  orientation with the endplate (**a**). This permits a significant torsion prevention potential (*arrows*) (**b**). In fact, they are more optimally oriented for torsion prevention than for distraction (or compression) prevention. If the annular fibers are

lax (c), torsion resistance diminishes (*arrows*) (d). Chronic instability and mechanical pain may result. Lax annular ligaments also predispose to the more commonly observed imaging correlate of chronic instability and sub-luxation (e)

vertebral bodies by Sharpey's fibers. This oblique laminated fiber pattern helps resist axial rotation by acting like tendons resisting force along their longitudinal orientation (Fig. 22.1). Transverse and radial annular tears have been shown to disrupt this tendon-like longitudinal force resistance and allow for greater axial rotation [4].

The rostral and caudal endplates are composed largely of hyaline cartilage and are firmly attached to the adjacent vertebral bodies by calcified cartilage. The endplates act as a semipermeable barrier *between* the vascular medullary portion of the vertebral bodies and the disk. Disks in adults and children are avascular and nutrients must enter through the adjacent endplates or capillaries into the annulus fibrosis [5]. In contrast, infant's disk spaces are vascular and therefore nutrients enter the annulus via arteries or arterioles.

# 22.3 Function of the Intervertebral Disk Space

The intervertebral disk space composes 20-25 % of the vertebral column height. In axial loading the disk acts as shock absorber for the relatively incompressible vertebral body. The disk is exposed to variable compression loads throughout the day. The effects of position on the hydrostatic disk have been measured and confirmed by various authors [6–8]. As body weight increases with aging or when one rises from a supine position, the load on the disk space increases, predominantly in lower lumbar levels [5]. In the

supine position with knees bent and supported, the lumbar spine is relaxed in a slightly less lordotic posture. The disk pressure in this position is the least and is primarily produced by surrounding ligaments and musculature. As a standing position is taken, the pressure in the disk increases as it takes on more of the axial load of the body. In a resting standing position, the majority of the pressure is borne in the center of the disk and is fourfold greater than that of being supine. As the nucleus pulposus compresses under axial loading, it bulges out laterally and is contained by the annulus fibrosus. The collagen fibers of the annulus are placed under tension and thus act as a capsule to contain nucleus pulposus. With bending, the nucleus migrates away from the increased force and the annulus bulges in the direction of the concavity of the curve. For instance, bending forward produces more kyphosis in the lumbar spine. In this position, the pressure in the ventral annulus is increased and the nucleus pulposus moves dorsally, where the pressure is less. This dorsal displacement places the dorsal annulus under tension as the nucleus pushes against it (Fig. 22.2).

With asymmetric loading of the spine, the intervertebral disk provides a measure of stiffness that is augmented by extra-discal structures such as spinal ligaments, muscles, and facet joints [7, 9]. To achieve both stability and motion, each of the components of the intervertebral disk plays different yet critical roles.

Under compression, the nucleus pulposus plays an important role in maintaining equal pressure along the endplates and maintaining



**Fig. 22.2** An axial load causes an equally distributed force application to the disk (**a**). An eccentric force application results in annulus fibrosus bulging on the side of the greatest force application (i.e., the concave side of the bend) (**b**). The nucleus pulposus moves in the opposite direction. *Dashed lines* indicate the positions of structures during force application (**c**)

intervertebral disk height. The osmotic properties of the intervertebral disk space are crucial regarding these roles. This osmotic system is centered about the hydration of the nucleus pulposus and abilities of the annulus fibrosus and the cartilaginous endplates to act as semipermeable membranes through which water moves.

The movement of water is governed by several factors. The forces that keep water within the disk are the extra-discal hydrostatic pressure and the oncotic pressure of the nucleus ground substance. The forces that drive water out of the disk are the extra-discal oncotic pressure and the intra-discal hydrostatic pressure. The semipermeable barriers through which water moves are the endplates and the annulus fibrosus. As axial loading increases, the intra-discal hydrostatic pressure increases, driving fluid out of the disk. As fluid leaves, the anionic charge of the mucoid ground substance becomes more concentrated and thus pulls fluid back into the disk so that the following equilibrium is maintained: intra-discal hydrostatic pressure + extra-discal oncotic pressure = extra-discal hydrostatic pressure + intradiscal oncotic pressure. With changes in compressive forces (i.e., position changes during the day), fluid shifts between the intra-discal substance and medullary bone to maintain this equilibrium [6, 7, 10]. These fluid shifts facilitate the transport of nutrients and wastes and may be important for the overall health of the disk.

As previously noted, the vertebral endplates act as firm surfaces that transfer axial loads to the intervertebral disk. They also act as semipermeable membranes that allow not only water but also nutrients to diffuse through from the vascular medullary bone of the vertebral bodies into the relatively avascular disk space. Disruption of the vertebral endplates is linked with disk degeneration and is discussed in further detail later.

# 22.4 Degenerative Changes in the Intervertebral Disk Space

Many events can occur to weaken the main components of the intervertebral disk leading to degenerative changes. The most common level of disk degeneration is at L4/5 and degeneration is more common at multiple levels as compared to a single level [1]. Degeneration results in decreased nuclear energy dissipation, swelling pressure, and compressive modulus relative to a normal nucleus; additionally, the annulus has a higher compressive stiffness and matrix function of the disk undergoes breakdown weakening biomechanical behavior [3]. Imaging findings that are associated with degeneration include: diminished disk height, facet narrowing, endplate spondylophytes and sclerosis, canal stenosis, lateral recess narrowing, desiccation, fibrosis or diffuse bulging of the annulus beyond disk space, fissuring or mucinous degeneration of annulus, defects or sclerosis of endplates, and osteophytes at vertebral apophyses [1, 11]. Mechanical behavior of the disk is related to biphasic-swelling properties as well as individual components: annulus, nucleus pulposus, and cartilaginous endplates [12].

The degenerated disk displays several interrelated changes. Annular tears develop as chondrocytes and fibroblasts are no longer able to produce quality fibers that can resist the expansile pressure of mobile central disk [5]. These tears are seen as early as the second generation of life and are present primarily in three different patterns: circumferential, transverse, and radial tears [13]. Transverse and radial tears are of greater clinical importance in degeneration as they have been shown to increase segmental motion markedly in rotation as well as moderately in flexion and extension but not significantly in lateral bending [4]. It has also been proposed that transverse tears in the posterior annulus act as a conduit for nucleus pulposus to herniate and cause nerve root compression. Though if the external layer of annulus fibrosus remains intact, there exists a possibility the protruded disk matter can return to its original position [5].

Decreased water content has also been noted in disk degeneration and has been linked to proteoglycan degradation. It is greatest in the nucleus pulposus [1]. Up to 70 % reduction in hydration is observed, resulting in desiccation of both the nucleus pulposus and other components of the disk [1]. The decrease of hydrostatic pressure of nucleus in disk degeneration has been noted both in vivo and in vitro studies [10, 14, 15]. This loss of water decreases the ability of the nucleus to weight bear. This progressively leads to increased weight bearing by the annulus and thus degeneration over time.

Weakening of the endplates may result from trabecular micro-damage and are identified on MRI as disk bulging into the vertebral bodies [16, 17]. With complete endplate disruption, herniation of nucleus fragments into vertebral bodies may occur. These are termed Schmörl's nodes when they become calcified. The loss of the endplate's ability to provide a firm surface for translation of compressive forces leads to the loss of the hydrostatic nucleus to weight bear in compression. The resultant compressive forces are then passed to the annulus and the developing fibrocartilaginous scar that forms peripherally. Further calcification of ligaments that span the intervertebral space can result in added fixation of the spinal segment [5]. With increased compressive stress, buckling of the annulus inward into the decompressed nucleus and outward beyond the margins of the vertebral endplates may occur and lead to a further decrease in disk height.

Disk height plays an important role in reducing compressive loads to the articular surface of the facet joints. With decreased height, load bearing in neural arch may increase to up to 50 % of the total compressive forces [18]. With this increased stress on the articular surface, facet joints undergo osteoarthritic changes that lead to facet injury and hypertrophy.

# 22.5 Spinal Instability

Instability of a vertebral motion segment may be defined as pathological or as increased motion of a joint [19]. Kirkaldy-Willis and Farfan argued against using this term alone, as there may be increased motion in asymptomatic individuals. For this reason, they adopted the term clinical instability and defined it as increased abnormal joint motion that produces a clinical response, such as pain or deformity [20]. The intervertebral disk is the main motion segment of the spine [5]. When the disk space is compromised by degeneration,



**Fig. 22.3** A typical stress/strain curve for a biological tissue, such as a ligament. (*AB*) The neutral zone. (*BC*) The elastic zone. When the elastic limit (yield point) (*C*) is reached, permanent deformation can occur (permanent set). (*CD*) The plastic zone where a permanent set occurs. Past (*D*), failure occurs, and the load diminishes. *Hashed* plus *dotted areas* quantitatively represent strength, whereas the *dotted area* represents resilience

injury, or diskectomy, instability can occur and cause further degeneration of the spine [5].

The assessment of mechanical instability in the degenerated lumbar disk obligates the consideration of several additional concepts. Krismer, in 1997, defined instability as being associated with the following mechanical abnormalities: excessive dorsoventral transitional movement, pathological accompanying movements, enlarged neutral zone (NZ), and pathological center of rotation [5, 21]. The instantaneous axis of rotation (IAR) is the point around which a moment arm causes rotation in any given plane. The NZ is the component of motion in which the joint moves in response to an applied force with minimal internal resistance. The limit or "end" of the NZ is the point at which substantial resistance to motion is first exhibited [15]. As joint motion passes the limits of the neutral zone, it enters the elastic zone. This is the zone in which greater resistance to an applied force is observed, and motion is linearly related to the applied force. At the extreme of the elastic zone, a point is reached that results in permanent deformation in response to the applied force or injury to the affected structure (Fig. 22.3) [22]. Therefore, the summation of the NZ and elastic zone gives rise to the full range of motion (ROM) of a joint. The neutral zone ratio (NZR) has been defined as NZ/ROM and is used as a mechanical index of instability [15, 23]. A high NZR is associated with a clinical situation in which a joint was excessively lax (or unstable). Lastly, as degenerative disk disease progresses, ROM, NZ, and NXR all increase [19].

Clinical and radiographic observations of lumbar disk degeneration have led to numerous in vitro and in vivo studies that were designed to determine how disruption of one or more of the functional structures of the disk may result in instability [8, 24]. However, the link of disk degeneration to overt instability is not clear. Kirkaldy-Willis and Farfan postulated that there are three stages in lumbar disk degeneration: (1) temporary dysfunction, (2) unstable, and (3) stabilization [20].

The temporary dysfunction stage may be initiated in younger patients by an inciting damage, such as trauma, or functional changes (slackening of disks) resulting in degeneration. Intradiscal tissue displacement and disk protrusion can produce pain without radiographic correlate [5]. The unstable phase generally occurs in middle-aged patients in whom mechanical pain is accompanied by instability, loss of disk height, and/or arthritic joint changes [5]. The stabilization phase, most commonly observed in the elderly, is defined by re-stabilization of the joints as the vertebral joints stiffen with age.

Using cadaver models, several authors have investigated the role of physiologic compressive forces on the lumbar spine and the resulting changes in the motion segment [14]. In these studies, endplate damage was commonly noted and resulted in a rapid decompression of the nucleus. This, in turn, caused increased load bearing by the annulus. Zhao and colleagues evaluated the role of decreased water content and resulting endplate disruption in the lumbar motion segment [15]. They found increased NZRs in flexion and lateral bending, as well as a decreased bending stiffness, when the water content was decreased. These changes were greater when endplate disruption occurred. Furthermore, horizontal translation was also noted to increase when the disks were subjected to compressive forces. The coupling of horizontal translation with angular rotation may have resulted in the observed increased translocation of the IAR toward the direction of rotation. Of note, in extension, the IAR moved toward the inferior facet. This phenomenon correlates with the observation of increased load bearing by the facet joints in extension. These results suggest that with initial dysfunction, the joint becomes more lax in flexion and lateral bending. With further damage or inadequate healing, the joints may enter an unstable phase that is characterized by increased abnormal segment motion. This, in turn, may produce mechanical low back pain.

Cadaver studies, however, do not reveal the cellular mechanism of healing. Traction vertebral osteophytes, facet hypertrophy, and the healing of annular tears via scarring are reactive changes noted with increasing age and disk degeneration [13, 25]. These changes usually lead to a stiffer, albeit altered, motion segment. With further reactive changes, the motion segment may enter the stabilization phase, in which the previously lax motion segment becomes progressively stiffer.

#### References

- Saleem S, Aslam HM, Rehmani MA, Raees A, Alvi AA, Ashraf J. Lumbar disc degenerative disease: disc degeneration symptoms and magnetic resonance imaging findings. Asian Spine J. 2013;7:322–34.
- Bajpai J, Saini S, Singh R. Clinical correlation of magnetic resonance imaging with symptom complex in prolapsed intervertebral disc disease: a crosssectional double blind analysis. J Craniovertebr Junction Spine. 2013;4:16–20.
- Cheng KK, Berven SH, Hu SS, Lotz JC. Intervertebral disc from spinal nondeformity and deformity patients have different mechanical and matrix properties. Spine. 2013;13:S1529–9430.
- Haughton VM, Schmidt TA, Keele K, et al. Flexibility of lumbar spinal motion segments correlated to type of tears in the annulus fibrosus. J Neurosurg (Spine 1). 2000;92:81–6.
- Kramer J. Intervertebral disc disease: causes, diagnosis, treatment, and prophylaxis. 3rd ed. New York: George Thieme Verlag; 2008. p. 15–32, 43–54.

- Adams MA, Hutton WC. The effect of posture on the fluid content of lumbar intervertebral discs. Spine. 1983;6:665–71.
- Nachemson AL. Disc pressure measurements. Spine. 1981;5:93–7.
- Sato H, Kikuchi S. The natural history of radiographic instability of the lumbar spine. Spine. 1993;18: 2075–9.
- Gillespie KA, Dickey JP. Biomechanical role of lumbar spine ligaments in flexion and extension: determination using parallel linkage robot and a porcine model. Spine. 2004;29:1208–16.
- Sato K, Kikuchi S, Yonezawa T. In vivo intradiscal pressure measurement in healthy individuals and in patients with ongoing back problems. Spine. 1999;24:2468–74.
- Modic MT, Ross JS. Lumbar degenerative disk disease. Radiology. 2007;245:43–61.
- Cortes DH, Jacobs NT, Delucca JF, Elliott DM. Elastic, permeability and swelling properties of human intervertebral disc tissues: a benchmark for tissue engineering. J Biomech. 2013;13:00648–9.
- Vernon-Roberts B, Moore RJ, Fraser RD. The natural history of age-related disc degeneration: the pathology and sequelae of tears. Spine. 2007;32:2797–804.
- Adams MA, Freeman BJC, Morrison HP, et al. Mechanical initiation of intervertebral disc degeneration. Spine. 2000;25:1625–36.
- Zhao F, Pollintine P, Hole BD, et al. Discogenic origins of spinal instability. Spine. 2005;30:2621–30.
- Adams MA, Roughley PJ. What is intervertebral disc degeneration, and what causes it? Spine. 2006; 31:2151–61.
- Vernon-Roberts B, Pirie CJ. Healing trabecular microfractures in the bodies of lumbar vertebrae. Ann Rheum Dis. 1973;32:406–11.
- Pollintine P, Przybyla AS, Dolan P, Adams MA. Neural arch load-bearing in old and degenerated spines. J Biomech. 2004;37:197–204.
- Quint U, Hans-Joachim W. Grading of degenerative disk disease and functional impairment: imaging versus patho-anatomical findings. Eur Spine J. 2008; 17:1705–13.
- Kirkaldy-Willis WH, Farfan HF. Instability of the lumbar spine. Clin Orthop. 1982;165:110–23.
- Krismer M, Haid C, Ogon M, Behensky H, Wimmer C. Biomechanics of lumbar instability. Orthopade. 1997;26:516–20.
- 22. Benzel EC. Biomechanics of spine stabilization. Rolling Meadows: AANS Press; 2001. p. 1–60.
- Mimura M, Panjabi MM, Oxland TR, et al. Disc degeneration affects the multidirectional flexibility of the lumbar spine. Spine. 1994;19:1371–80.
- 24. Weiler PJ, King GJ, Gertzbein SD. Analysis of sagittal plane instability of the lumbar spine in vivo. Spine. 1990;15:1300–6.
- Fujiara A, Lim T, An HS, et al. The effect of disc degeneration and facet joint osteoarthritis on the segmental flexibility of the lumbar spine. Spine. 2000;25:3036–44.

# Adjacent Segment Disease: Natural History of Lumbar Degeneration or Consequence of Fusion?

Mark P. Coseo, Nelson S. Saldua, Eric B. Harris, and Alan Hilibrand

# 23.1 Introduction

Spinal arthrodesis is the surgical gold standard treatment for symptomatic lumbar degenerative disk disease. Over the last decade, the number of lumbar fusion procedures in the United States has risen dramatically [1, 2]. However, there is significant concern over the potential affects lumbar fusion procedures have on the health of adjacent segments. Loss of a lumbar motion segment has been shown to alter lumbar mechanics and increase disk pressures and loading of endplates and facet joints [3–5]. Long-term follow-up studies have shown rates of adjacent segment degeneration after lumbar fusion procedures over 80 % [6-8]. The question remains as to whether adjacent segment degeneration and subsequent disease are a result of lumbar fusion procedures or the result of natural history of progression of lumbar disk disease.

N.S. Saldua, MD • E.B. Harris, MD (⊠) Department of Orthopaedic Surgery, Naval Medical Center San Diego, San Diego, CA, USA e-mail: nelsonsaldua@gmail.com; ericbharris@gmail.com

# 23.2 Adjacent Segment Disease: Definitions

Adjacent segment disease is symptomatic degenerative changes of an adjacent level following surgical intervention, whereas adjacent segment degeneration is the presence of abnormal imaging findings in the adjacent level following a surgical intervention at the index level that is asymptomatic. While there is no classification system for adjacent segment disease, it constitutes a myriad of signs and symptoms including axial low back pain, instability, stenosis, and radiculopathy of the cranial or caudal adjacent segment [9, 10].

# 23.3 Natural History of Lumbar Degeneration

The lumbar spine is susceptible to arthritic changes with age. Under pure compression, healthy lumbar disks demonstrate near uniform stress distribution. Physiologic forces are generally distributed equally along the endplate [11]. As a disk degenerates, stress concentrations shift to the annulus, progressing to disk bulges and protrusions. In addition, the posterior elements encounter greater transmission of forces. The affected segment becomes more susceptible to shear forces and less able to transfer load appropriately [12]. With moderate disk degeneration, there is greater translation under compression as

M.P. Coseo, MD

Department of Orthopaedic Surgery, Naval Medical Center Portsmouth, 34800 Bob Wilson Drive, San Diego, CA 92134-5000, USA e-mail: mark.coseo@med.navy.mil

A. Hilibrand, MD Thomas Jefferson University Medical College, 925 Chestnut Street 5th Floor, Philadelphia, PA 19107, USA e-mail: ahilibrand@gmail.com

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016 J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_23

the instantaneous axis of rotation of the spinal functional unit moves inferiorly and has a wider range toward the periphery [13, 14]. This results in an increase in motion of the spinal functional unit with moderate degeneration. However, as degenerative changes progress, motion decreases [12]. These degenerative changes occur most frequently at the L5-S1 level, followed by L4-L5, and then L3–L4 [15] and occur in asymptomatic patients. Multiple studies have shown a disconnect between imaging findings of degeneration and symptom profile. MRI studies of asymptomatic patients have demonstrated abnormalities in the lumbar spine in greater than 50 % of patients over 60 [15]. Even in asymptomatic patients, lumbar degeneration has been shown to progress over time. In a 2002 study, Elfering et al. examined 41 asymptomatic individuals with MRI studies over a 5-year period. The investigators found 41 % of patients had evidence of progression of lumbar disk degeneration over the 5-year span [16]. However, there was no statistically significant correlation of progression of degenerative change to development of symptoms.

If the progression of lumbar degeneration is a result of natural history as these studies indicate, are certain patient populations at higher risk? Multiple studies have shown there is a likely genetic predisposition to lumbar degenerative disease [17, 18–22]. Much of this evidence has been the result of twin studies. Battie et al. published on the results of MRI studies of 116 pairs of monozygotic Finnish twins. The investigators found familial and genetic influences accounted for variance in disk degeneration for 61 % in the upper lumbar region and 34 % in the lower lumbar region [19]. Another classic twin study, from Sambrook et al., reviewed lumbar MRI studies of 172 monozygotic and 154 dizygotic twins. They showed 64 % heritability for severe disease in the lumbar spine and overall heritability of lumbar degenerative disease of 74 % (95 % CI 64-81 %) [17]. A follow-on twin study of monozygotic and dizygotic Finnish twins by Battie et al. supported the high rate of genetic influence in lumbar degenerative disk disease. They estimated heritability

ranging from 29 to 54 % depending on lumbar level and phenotype [23]. The higher concordance rate seen in the monozygotic twin populations has clearly shown there is a genetic predisposition to the development of lumbar degenerative disk disease.

However, it is also clear there is no signal gene responsible for this increased risk. Progress has been made on delineating possible candidate genes yielding the greatest influence on lumbar degeneration. Polymorphisms of the genes coding proinflammatory cytokines and matrix metalloproteinases leading to overexpression in the intervertebral disk have been hypothesized to be a contributing factor. Of particular concern is overexpression of MMP-2, MMP-3, and IL-1 leading to increased inflammatory response and increased susceptibility to lumbar disk degeneration [24-27]. In addition, healthy disks contain high concentrations of aggrecan and collagen-9. Certain polymorphisms in the genes encoding both these proteins have also been shown to be associated with increased risk of lumbar degenerative change [28–31]. Asporin is a leucine-rich repeat protein found in extracellular matrix. Overrepresentation of the D-14 allele of the asporin gene has been associated with increased risk of osteoarthritis and lumbar degeneration [32–34].

While progress has been made and certain gene alleles have been identified that are associated with lumbar degeneration, there is still much that is unknown. As gene sequencing becomes faster and computing power increases, genomewide studies across multiple populations will become more feasible. This will not only help delineate the genotypes most at risk for developing lumbar degenerative disease but also may identify genetic profiles that are more resistant to disease progression. While the entire genomic influence in lumbar degenerative disease is unclear, it is evident from available studies there are populations that are more susceptible. This lends credence to the belief adjacent segment disease may be progression of natural history, rather than as a result of surgical intervention in the lumbar spine.

# 23.4 Patient Factors in Adjacent Segment Disease

In addition to genetic predisposition, certain patient factors have been hypothesized to increase risk of adjacent segment disease after fusion. There is evidence to suggest facet tropism and laminar inclination could predispose toward disease [35-38]. Okuda et al. reviewed 87 patients who underwent PLIF for L4 degenerative spondylolisthesis and found that facet tropism at L3-4 and horizontal lamina at L3 correlated with increased rate of degenerative change at L3-4 after L4-5 posterior lumbar interbody fusion [36]. The investigators hypothesized that facet joint asymmetry and increased lamina inclination angle resulted in abnormal motion of the spinal unit with increased intradiscal pressures and subsequent degradation [36]. In a follow-on study of this population, investigators found patients undergoing reoperation rates for adjacent segment disease had higher rates of facet tropism and lamina horizontalization in the cranial level to the fused segment [39].

Patient age is also thought to be a risk factor for adjacent segment disease [6, 9]. In a retrospective review of 3188 patients undergoing lumbar fusion procedures, Ahn et al. found a correlation between increasing age at index procedure and need for a reoperation on adjacent segments [6]. Interestingly, the study also found males were more at risk for reoperation [6]. Harrop et al. found similar conclusions in another retrospective review. Older age was associated with development of adjacent segment disease [9].

Finally, it has been hypothesized that obesity and history of smoking play a role in adjacent segment degeneration, while some studies reported no evidence of increased risk with higher BMI or nicotine intake [6, 9, 36, 40, 41]. Cho et al. reviewed 154 patients in a retrospective study and found age, BMI, and preoperative existing degenerative changes at the cranial level had the most significant risk for requiring repeat operative procedures for adjacent segment disease [42].

# 23.5 Prevalence of Adjacent Segment Disease

Many long-term studies have shown the progression of radiographic evidence of degenerative changes on adjacent segments following surgical intervention. While adjacent segment degeneration is common, rates of adjacent segment disease are much lower. A recent meta-analysis reviewed 94 studies with 34,716 patients included. In subgroup analysis, investigators found pooled prevalence rates for adjacent segment degeneration from 21.8 to 37.4 % and rates of 3.2–12.1 % of adjacent segment disease [43]. Harrop et al. in a meta-analysis found similar rates of adjacent segment degeneration and disease. The investigators reported rates of adjacent segment degeneration of 34 % (314/926) and disease rate of 14 % (173/1216) [9]. There is good evidence showing adjacent segment disease is more prevalent in the levels proximal to surgical intervention [9, 43-45]. Celestre et al. found in patients undergoing L4-5 fusion, 75 % developed adjacent segment degeneration at L3-4, while only 25 % had similar progression at L5-S1 [44]. The study found in all cases of adjacent segment disease the cranial levels were affected 90 % of the time [44].

The conclusions from these meta-analyses must be interpreted with caution due to their retrospective nature, differences in symptom evaluation, follow-up time, and imaging studies. However, our best available evidence does show a significant rate of adjacent segment degeneration and disease following lumbar fusion. The literature also shows cranial levels are at highest risk of development.

# 23.6 Is Fusion a Risk Factor?

Lumbar spinal motion is complex, as the functional spinal unit does not have a fixed center of rotation. It is inherently difficult to study given lumbar kinematic behavior is nonlinear. The lumbar spine is viscoelastic and motion coupling
occurs in vivo [46]. This makes in vitro studies of lumbar spinal motion following arthrodesis difficult. However, multiple in vitro models have been developed to study the effect fusion has on adjacent levels [5, 47-49]. In a cadaveric study, Weinhoffer et al. measured intradiscal pressures in uninstrumented, single-level bilateral instrumented L5-S1 and multiple-level bilateral instrumented (L4-S1) models [48]. The investigators found intradiscal pressures were increased in the levels above the instrumentation and more significantly increased as more levels were instrumented [48]. The findings of this study were corroborated by a similar cadaveric study by Chow et al., in 1996. Single-level instrumentation at L4-5 increased intradiscal pressures in the cranial segment. Double-level instrumentation from L4-S1 resulted in marked increase in intradiscal pressure of the cranial segments [5]. This study also measured mobility of the cadaveric spines prior to and following instrumentation. They found one- and two-level instrumentation resulted in increased motion of the unfused segments closer to the extremes of their functional range [5]. Similar results have also been seen in in vitro studies of calf lumbosacral spine specimens. Shono et al. used this model to study the differences in motion following one-, two-, and three-level instrumentation in flexion, extension, lateral bending, and rotation [47]. As levels of instrumentation increased, the motion of the segments proximal to the construct had greater increases in motion [47]. This finding has also been seen in vivo. Axelsson et al. utilized roentgen stereophotogrammetric analysis of six patients undergoing fusion for low-grade spondylolisthesis to study the mobility of pre-fused and fused lumbar segments. Mobility of L4-5 and L5-S1 segments was measured preoperatively and after fusion. They found increased mobility of the proximal segment following fusion procedure [49]. While the results of these studies show concerning effects of loss of a motion segment, they must be interpreted critically. Current in vitro studies do not fully recreate the complexities of in vivo spinal motion, and available in vivo studies are limited by sample size. Improvements are needed in in vitro studies,

and progress has been made to develop models that more accurately replicate spinal motion in vivo for bench studies [50].

It is very plausible biomechanical changes to the lumbar spine secondary to fusion result in abnormal forces on the adjacent segments and subsequent disease. However, does surgical intervention alone without fusion increase this risk? Some hypothesize decompressive laminectomy alone without fusion may alter lumbar mechanics and predispose to disease [51]. Biomechanical studies have shown posterior element excision in laminectomy produces increased motion of the surgical segment in flexion, extension, and rotation [52]. Using comprehensive national data from Sweden, Jansson et al. examined a 10-year follow-up on 9664 patients who underwent decompressive laminectomy for spinal stenosis [53]. They found progression of symptoms and need for reoperation in 11 % of patients on 10-year follow-up [53]. The results of this study and those like it show that fusion alone is not the sole risk for developing adjacent segment disease. The process of degeneration proceeds with non-fusion procedures as well.

While available data does show there is an increase in intradiscal pressures and motion in the levels surrounding a fusion mass, the question still remains if this change increases risk of adjacent segment disease. Models that more accurately represent spinal motion may provide insight if changes following loss of motion segments in the lumbar spine are well compensated in vivo or if they lead to symptomatic degeneration. Adjacent segment disease occurs after spinal fusion; however, there is no definitive evidence in the literature to date proving this phenomenon occurs as a result of the fusion.

#### 23.7 Surgical Intervention: Does Method of Fusion Matter?

It is clear there are multiple methods to achieve fusion of a functional unit in the lumbar spine. However, it is still unclear if approach and fusion method utilized affects risk of adjacent segment disease. Some believe posterior approaches and pedicle screw-based instrumentation may predispose adjacent segment disease over anteriorbased techniques [52, 54, 55]. Chen et al., in 2008, utilized CT scans following pedicle screw fixation for lumbar fusions to assess the rate of superior segment facet joint violation. Of this study cohort, 47 % of patients had evidence of superior facet joint violation after instrumentation [54]. It is possible that violation of the facets at the fusion interface may increase the risk of degeneration. If posterior approaches and instrumentation increase the risk of adjacent segment disease, it would follow then that anterior lumbar interbody fusion procedures would have lower rates of reoperation for adjacent disease. There is some level IV data to support this hypothesis. While not examining adjacent segment disease directly, Strube et al. found a significant difference in patient satisfaction, Oswestry disability index, and visual analog scale in patients undergoing anterior lumbar interbody fusion (ALIF) alone versus anteroposterior lumbar fusion [56]. Wai et al. performed MRI studies on 39 patients after ALIF with a minimum of 20-year followup. Only three patients in this group required additional surgery. The investigators also found 30.7 % of patients had advanced degeneration, but of that group, 17.9 % had preservation of the adjacent segment. They concluded the rates in their study were similar to unfused patients [8]. A study by Min et al. compared 48 patients who had undergone either L4-5 ALIF or PLIF for degenerative spondylolisthesis. They reported rates of adjacent segment degeneration of 44 % in the ALIF group and 82.6 % in the PLIF group. However, there was no significant difference in adjacent segment disease rate [57]. It is difficult to ascertain the clinical significance of the differential finding in adjacent segment degeneration in these two groups. Though likely, it is more consistent with progression of natural history of the degenerative spine rather than fusion related. Finally, while these studies may show a benefit to stand-alone anterior lumbar interbody fusion in association with development of adjacent segment disease, there is also evidence fusion method has no effect. Analyzing outcomes from

the SPORT trial, Abdu et al. found no significant

difference in 4-year outcomes comparing posterolateral in situ fusion, posterolateral instrumented fusion, and 360° fusion [58]. Given the disparity in the literature, there is no definitive answer as to the effect the method of fusion has on the development of adjacent segment disease.

Finally, potentially avoidable surgical results have been associated with increased risk of adjacent segment disease. Over distraction at time of fusion is hypothesized to pathologically load the posterior elements resulting in earlier degeneration [59]. In a study of 85 patients after posterior lumbar interbody fusion at L4-5, Kaito et al. found 13 out of 85 patients developed adjacent segment disease at 2-year follow-up. In the group developing adjacent segment disease, there was an average distraction of 6.1 mm at L4-5, while in the asymptomatic group, there was an average distraction of 3.1 mm [59]. There is also evidence to suggest the importance of maintaining near anatomic sagittal balance following fusion [60-63]. Keorochana et al. reviewed MRI findings of 430 patients with low back pain and found the patients with hyperlordosis (>50°) and hypolordosis (<20°) had increased disk degeneration compared to those with normal lordosis [63]. Djurasovic et al. corroborated these findings, where they found hypolordosis at the fused lumbar segment correlated with adjacent segment degeneration [62]. These results clearly show the importance of good surgical technique when performing lumbar fusion procedures. Particular attention must be paid in avoiding over-distraction and changes in postoperative sagittal alignment.

#### 23.8 Motion-Sparing Surgery: Is Arthroplasty the Answer?

Concern of adjacent segment disease has helped lead to the development of disk and spinal functional unit arthroplasties of the lumbar spine. The goals of motion-preserving surgical interventions are to replace the diseased segment and remove the pain generator while maintaining as close to normal spinal kinematics and motion. It is thought this would help limit the risk fusion has on the development of adjacent segment disease. There is literature to support this claim. In a recent meta-analysis, Harrop et al. included 27 retrospective studies to compare the development of adjacent segment degeneration and disease in patients who had undergone lumbar arthrodesis versus arthroplasty. They found rate of adjacent segment disease in the arthrodesis cohort to be 14 % (173/1216) and a rate in the arthroplasty group to be 1 % (7/595) [9]. Another recent metaanalysis of 1584 patients at 2-year follow-up found a significant improvement in ODI and VAS scores of patients following lumbar disk arthroplasty compared to those after fusion [64]. These results appear promising in support of motionsparing technology; however, they must be interpreted with caution given the nature of studies included in the meta-analysis. In addition, there is relatively little data on the long-term survivability of lumbar disk arthroplasty. The longest follow-up data is available from case series out of Europe [65, 66]. David reported on 106 patients with mean 13.2-year follow-up after one-level lumbar disk arthroplasty. In this case series, 82.1 % reported good to excellent outcomes, 90.6 % were still mobile with mean flexion and extension range of 10.1°, and only three cases (2.8 %) of adjacent segment disease [66]. Huang et al. reported on 42 patients at a mean of 8.7 years following lumbar disk arthroplasty. They found an adjacent segment degeneration rate of 24 % but noted no adjacent segment degeneration developed in patients with motion greater than  $5^{\circ}$  at their arthroplasty site [67]. Siepe et al. performed MRI studies on 93 patients at a mean of 53.4 months following lumbar disk arthroplasty and found an incidence of adjacent level degeneration of 10.2 % but characterized the level of degeneration as mild in all cases [68]. Comparing motion at adjacent segments following fusion versus arthroplasty suggests there is less change in postoperative range of motion following disk replacement [69, 70]. Berg et al. compared 72 patients following arthrodesis and 80 patients after total disk arthroplasty at 2-year follow-up and found significant difference in preservation of preoperative mobility of the adjacent segments in the arthroplasty group [70].

Compared to arthroplasty in other joints, lumbar arthroplasty is in its infancy. The data available is promising in the preservation of adjacent segments following surgery. However, there is still concern of abnormal forces on the posterior elements of the involved segment following disk replacement. This has continued to lead to development of more optimal arthroplasty techniques, including total lumbar functional unit arthroplasty. As this technology develops, continued research is needed to elucidate if lumbar arthroplasty is more effective in preventing adjacent segment disease over fusion.

#### Conclusion

The best available evidence to date suggests that adjacent segment disease is a multifactorial process. There are populations of patients more prone to lumbar degenerative disease from a combination of underlying genetic risk factors and environmental factors. However, there is also good evidence to suggest surgical intervention plays a role in its development. Fusion procedures of the lumbar spine likely hasten the progression of natural degeneration in the lumbar spine. Good surgical technique, limited dissection of the posterior elements, and maintenance of sagittal alignment have been shown to be crucial in helping prevent this condition. As lumbar arthroplasty technology continues to develop, it may prove to be superior to current fusion methods in treating pain generators of the lumbar spine without progression of adjacent segment disease.

#### References

Awe OO, Maltenfort MG, Prasad S, Harrop JS, Ratliff JK. Impact of total disc arthroplasty on the surgical management of lumbar degenerative disc disease: analysis of the Nationwide Inpatient Sample from 2000 to 2008. Surg Neurol Int. 2011;2:139. doi:10.4103/2152-7806.85980.

Rajaee SS, Bae HW, Kanim LE, Delamarter RB. Spinal fusion in the United States: analysis of trends from 1998 to 2008. Spine. 2012;37(1):67–76. doi:10.1097/BRS.0b013e31820cccfb.

- Goto K, Tajima N, Chosa E, Totoribe K, Kubo S, Kuroki H, Arai T. Effects of lumbar spinal fusion on the other lumbar intervertebral levels (threedimensional finite element analysis). J Orthop Sci Off J Japan Orthop Assoc. 2003;8(4):577–84. doi:10.1007/ s00776-003-0675-1.
- Cunningham BW, Kotani Y, McNulty PS, Cappuccino A, McAfee PC. The effect of spinal destabilization and instrumentation on lumbar intradiscal pressure: an in vitro biomechanical analysis. Spine. 1997;22(22):2655–63.
- Chow DH, Luk KD, Evans JH, Leong JC. Effects of short anterior lumbar interbody fusion on biomechanics of neighboring unfused segments. Spine. 1996;21(5):549–55.
- Ahn DK, Park HS, Choi DJ, Kim KS, Yang SJ. Survival and prognostic analysis of adjacent segments after spinal fusion. Clin Orthop Surg. 2010;2(3):140–7. doi:10.4055/cios.2010.2.3.140.
- Remes VM, Lamberg TS, Tervahartiala PO, Helenius IJ, Osterman K, Schlenzka D, Yrjonen T, Seitsalo S, Poussa MS. No correlation between patient outcome and abnormal lumbar MRI findings 21 years after posterior or posterolateral fusion for isthmic spondylolisthesis in children and adolescents. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deformity Soc Eur Sect Cervical Spine Res Soc. 2005;14(9):833–42. doi:10.1007/s00586-005-0950-2.
- Wai EK, Santos ER, Morcom RA, Fraser RD. Magnetic resonance imaging 20 years after anterior lumbar interbody fusion. Spine. 2006;31(17):1952–6. doi:10.1097/01.brs.0000228849.37321.a8.
- Harrop JS, Youssef JA, Maltenfort M, Vorwald P, Jabbour P, Bono CM, Goldfarb N, Vaccaro AR, Hilibrand AS. Lumbar adjacent segment degeneration and disease after arthrodesis and total disc arthroplasty. Spine. 2008;33(15):1701–7. doi:10.1097/ BRS.0b013e31817bb956.
- Hilibrand AS, Robbins M. Adjacent segment degeneration and adjacent segment disease: the consequences of spinal fusion? Spine J Off J N Am Spine Soc. 2004;4(6 Suppl):190S–4. doi:10.1016/j. spinee.2004.07.007.
- McNally DS, Adams MA. Internal intervertebral disc mechanics as revealed by stress profilometry. Spine. 1992;17(1):66–73.
- Niosi CA, Oxland TR. Degenerative mechanics of the lumbar spine. Spine J Off J N Am Spine Soc. 2004;4(6 Suppl):202S–8. doi:10.1016/j.spinee.2004.07.013.
- Fujiwara A, Lim TH, An HS, Tanaka N, Jeon CH, Andersson GB, Haughton VM. The effect of disc degeneration and facet joint osteoarthritis on the segmental flexibility of the lumbar spine. Spine. 2000;25(23):3036–44.
- Fujiwara A, Tamai K, An HS, Kurihashi T, Lim TH, Yoshida H, Saotome K. The relationship between disc degeneration, facet joint osteoarthritis, and stability of the degenerative lumbar spine. J Spinal Disord. 2000;13(5):444–50.

- Boden SD, Davis DO, Dina TS, Patronas NJ, Wiesel SW. Abnormal magnetic-resonance scans of the lumbar spine in asymptomatic subjects. A prospective investigation. J Bone Joint Surg Am. 1990;72(3):403–8.
- Elfering A, Semmer N, Birkhofer D, Zanetti M, Hodler J, Boos N. Risk factors for lumbar disc degeneration: a 5-year prospective MRI study in asymptomatic individuals. Spine. 2002;27(2):125–34.
- Sambrook PN, MacGregor AJ, Spector TD. Genetic influences on cervical and lumbar disc degeneration: a magnetic resonance imaging study in twins. Arthritis Rheum. 1999;42(2):366–72. doi:10.1002/1529-0131(199902)42:2<366::AID-ANR20>3.0.CO;2-6.
- Kalichman L, Hunter DJ. The genetics of intervertebral disc degeneration. Familial predisposition and heritability estimation. Joint Bone Spine Rev Rhum. 2008;75(4):383–7. doi:10.1016/j.jbspin.2007.11.003.
- Battie MC, Videman T, Gibbons LE, Fisher LD, Manninen H, Gill K. 1995 Volvo Award in clinical sciences. Determinants of lumbar disc degeneration. A study relating lifetime exposures and magnetic resonance imaging findings in identical twins. Spine. 1995;20(24):2601–12.
- Battie MC, Videman T, Levalahti E, Gill K, Kaprio J. Heritability of low back pain and the role of disc degeneration. Pain. 2007;131(3):272–80. doi:10.1016/j.pain.2007.01.010.
- Matsui H, Terahata N, Tsuji H, Hirano N, Naruse Y. Familial predisposition and clustering for juvenile lumbar disc herniation. Spine. 1992;17(11):1323–8.
- Richardson JK, Chung T, Schultz JS, Hurvitz E. A familial predisposition toward lumbar disc injury. Spine. 1997;22(13):1487–92; discussion 1493.
- Battie MC, Videman T, Levalahti E, Gill K, Kaprio J. Genetic and environmental effects on disc degeneration by phenotype and spinal level: a multivariate twin study. Spine. 2008;33(25):2801–8. doi:10.1097/BRS.0b013e31818043b7.
- Solovieva S, Kouhia S, Leino-Arjas P, Ala-Kokko L, Luoma K, Raininko R, Saarela J, Riihimaki H. Interleukin 1 polymorphisms and intervertebral disc degeneration. Epidemiology. 2004;15(5):626–33.
- Le Maitre CL, Freemont AJ, Hoyland JA. The role of interleukin-1 in the pathogenesis of human intervertebral disc degeneration. Arthritis Res Ther. 2005;7(4):R732–45. doi:10.1186/ar1732.
- 26. Takahashi M, Haro H, Wakabayashi Y, Kawa-uchi T, Komori H, Shinomiya K. The association of degeneration of the intervertebral disc with 5a/6a polymorphism in the promoter of the human matrix metalloproteinase-3 gene. J Bone Joint Surg Br. 2001;83(4):491–5.
- 27. Dong DM, Yao M, Liu B, Sun CY, Jiang YQ, Wang YS. Association between the -1306C/T polymorphism of matrix metalloproteinase-2 gene and lumbar disc disease in Chinese young adults. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deformity Soc Eur Sect Cervical Spine Res Soc. 2007;16(11):1958–61. doi:10.1007/s00586-007-0454-3.

- Roughley P, Martens D, Rantakokko J, Alini M, Mwale F, Antoniou J. The involvement of aggrecan polymorphism in degeneration of human intervertebral disc and articular cartilage. Eur Cell Mater. 2006;11:1–7; discussion 7.
- 29. Roughley PJ, Melching LI, Heathfield TF, Pearce RH, Mort JS. The structure and degradation of aggrecan in human intervertebral disc. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deformity Soc Eur Sect Cervical Spine Res Soc. 2006;15 Suppl 3:S326–32. doi:10.1007/s00586-006-0127-7.
- Sivan SS, Tsitron E, Wachtel E, Roughley P, Sakkee N, van der Ham F, Degroot J, Maroudas A. Agerelated accumulation of pentosidine in aggrecan and collagen from normal and degenerate human intervertebral discs. Biochem J. 2006;399(1):29–35. doi:10.1042/BJ20060579.
- 31. Annunen S, Paassilta P, Lohiniva J, Perala M, Pihlajamaa T, Karppinen J, Tervonen O, Kroger H, Lahde S, Vanharanta H, Ryhanen L, Goring HH, Ott J, Prockop DJ, Ala-Kokko L. An allele of COL9A2 associated with intervertebral disc disease. Science. 1999;285(5426):409–12.
- 32. Kizawa H, Kou I, Iida A, Sudo A, Miyamoto Y, Fukuda A, Mabuchi A, Kotani A, Kawakami A, Yamamoto S, Uchida A, Nakamura K, Notoya K, Nakamura Y, Ikegawa S. An aspartic acid repeat polymorphism in asporin inhibits chondrogenesis and increases susceptibility to osteoarthritis. Nat Genet. 2005;37(2):138–44. doi:10.1038/ng1496.
- Iida A, Kizawa H, Nakamura Y, Ikegawa S. Highresolution SNP map of ASPN, a susceptibility gene for osteoarthritis. J Hum Genet. 2006;51(2):151–4. doi:10.1007/s10038-005-0337-6.
- 34. Song YQ, Cheung KM, Ho DW, Poon SC, Chiba K, Kawaguchi Y, Hirose Y, Alini M, Grad S, Yee AF, Leong JC, Luk KD, Yip SP, Karppinen J, Cheah KS, Sham P, Ikegawa S, Chan D. Association of the asporin D14 allele with lumbar-disc degeneration in Asians. Am J Hum Genet. 2008;82(3):744–7. doi:10.1016/j.ajhg.2007.12.017.
- Boden SD, Riew KD, Yamaguchi K, Branch TP, Schellinger D, Wiesel SW. Orientation of the lumbar facet joints: association with degenerative disc disease. J Bone Joint Surg Am. 1996;78(3):403–11.
- Okuda S, Iwasaki M, Miyauchi A, Aono H, Morita M, Yamamoto T. Risk factors for adjacent segment degeneration after PLIF. Spine. 2004;29(14):1535–40.
- Noren R, Trafimow J, Andersson GB, Huckman MS. The role of facet joint tropism and facet angle in disc degeneration. Spine. 1991;16(5):530–2.
- Pfirrmann CW, Metzdorf A, Zanetti M, Hodler J, Boos N. Magnetic resonance classification of lumbar intervertebral disc degeneration. Spine. 2001;26(17): 1873–8.
- Okuda S, Oda T, Miyauchi A, Tamura S, Hashimoto Y, Yamasaki S, Haku T, Kanematsu F, Ariga K, Ohwada T, Aono H, Hosono N, Fuji T, Iwasaki

M. Lamina horizontalization and facet tropism as the risk factors for adjacent segment degeneration after PLIF. Spine. 2008;33(25):2754–8. doi:10.1097/BRS.0b013e31817bb9c2.

- Min JH, Jang JS, Jung B, Lee HY, Choi WC, Shim CS, Choi G, Lee SH. The clinical characteristics and risk factors for the adjacent segment degeneration in instrumented lumbar fusion. J Spinal Disord Tech. 2008;21(5):305–9. doi:10.1097/BSD.0b013 e318142b960.
- Aalto TJ, Malmivaara A, Kovacs F, Herno A, Alen M, Salmi L, Kroger H, Andrade J, Jimenez R, Tapaninaho A, Turunen V, Savolainen S, Airaksinen O. Preoperative predictors for postoperative clinical outcome in lumbar spinal stenosis: systematic review. Spine. 2006;31(18):E648–63. doi:10.1097/01. brs.0000231727.88477.da.
- 42. Cho TK, Lim JH, Kim SH, Rhee WT, Kim WJ, Ha SI, Jang IT. Preoperative predictable factors for the occurrence of adjacent segment degeneration requiring second operation after spinal fusion at isolated L4–L5 level. J Neurol Surg A Cent Eur Neurosurg. 2013. doi:10.1055/s-0033-1349331.
- Xia XP, Chen HL, Cheng HB. Prevalence of adjacent segment degeneration after spine surgery: a systematic review and meta-analysis. Spine. 2013;38(7): 597–608. doi:10.1097/BRS.0b013e318273a2ea.
- 44. Celestre PC, Montgomery SR, Kupperman AI, Aghdasi B, Inoue H, Wang JC. Lumbar clinical adjacent segment pathology: predilection for proximal levels. Spine. 2014;39(2):172–6. doi:10.1097/ BRS.000000000000094.
- 45. Lawrence BD, Wang J, Arnold PM, Hermsmeyer J, Norvell DC, Brodke DS. Predicting the risk of adjacent segment pathology after lumbar fusion: a systematic review. Spine. 2012;37(22 Suppl):S123–32. doi:10.1097/BRS.0b013e31826d60d8.
- 46. Schmidt H, Heuer F, Claes L, Wilke HJ. The relation between the instantaneous center of rotation and facet joint forces – a finite element analysis. Clin Biomech (Bristol, Avon). 2008;23(3):270–8. doi:10.1016/j. clinbiomech.2007.10.001.
- 47. Shono Y, Kaneda K, Abumi K, McAfee PC, Cunningham BW. Stability of posterior spinal instrumentation and its effects on adjacent motion segments in the lumbosacral spine. Spine. 1998;23(14):1550–8.
- Weinhoffer SL, Guyer RD, Herbert M, Griffith SL. Intradiscal pressure measurements above an instrumented fusion. A cadaveric study. Spine. 1995;20(5):526–31.
- Axelsson P, Johnsson R, Stromqvist B. The spondylolytic vertebra and its adjacent segment. Mobility measured before and after posterolateral fusion. Spine. 1997;22(4):414–7.
- Panjabi MM. Hybrid multidirectional test method to evaluate spinal adjacent-level effects. Clin Biomech (Bristol, Avon). 2007;22(3):257–65. doi:10.1016/j. clinbiomech.2006.08.006.

- Radcliff KE, Kepler CK, Jakoi A, Sidhu GS, Rihn J, Vaccaro AR, Albert TJ, Hilibrand AS. Adjacent segment disease in the lumbar spine following different treatment interventions. Spine J: Off J N Am Spine Soc. 2013;13(10):1339–49. doi:10.1016/j. spinee.2013.03.020.
- 52. Bresnahan L, Ogden AT, Natarajan RN, Fessler RG. A biomechanical evaluation of graded posterior element removal for treatment of lumbar stenosis: comparison of a minimally invasive approach with two standard laminectomy techniques. Spine. 2009;34(1):17–23. doi:10.1097/BRS.0b013e318191438b.
- 53. Jansson KA, Nemeth G, Granath F, Blomqvist P. Spinal stenosis re-operation rate in Sweden is 11% at 10 years – a national analysis of 9,664 operations. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deformity Soc Eur Sect Cervical Spine Res Soc. 2005;14(7):659–63. doi:10.1007/s00586-004-0851-9.
- 54. Chen Z, Zhao J, Xu H, Liu A, Yuan J, Wang C. Technical factors related to the incidence of adjacent superior segment facet joint violation after transpedicular instrumentation in the lumbar spine. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deformity Soc Eur Sect Cervical Spine Res Soc. 2008;17(11):1476–80. doi:10.1007/s00586-008-0776-9.
- 55. Cardoso MJ, Dmitriev AE, Helgeson M, Lehman RA, Kuklo TR, Rosner MK. Does superior-segment facet violation or laminectomy destabilize the adjacent level in lumbar transpedicular fixation? An in vitro human cadaveric assessment. Spine. 2008;33(26):2868–73. doi:10.1097/BRS.0b013e31818c63d3.
- 56. Strube P, Hoff E, Hartwig T, Perka CF, Gross C, Putzier M. Stand-alone anterior versus anteroposterior lumbar interbody single-level fusion after a mean follow-up of 41 months. J Spinal Disord Tech. 2012;25(7):362–9. doi:10.1097/BSD.0b013e3182263d91.
- Min JH, Jang JS, Lee SH. Comparison of anteriorand posterior-approach instrumented lumbar interbody fusion for spondylolisthesis. J Neurosurg Spine. 2007;7(1):21–6. doi:10.3171/SPI-07/07/021.
- Abdu WA, Lurie JD, Spratt KF, Tosteson AN, Zhao W, Tosteson TD, Herkowitz H, Longely M, Boden SD, Emery S, Weinstein JN. Degenerative spondylolisthesis: does fusion method influence outcome? Four-year results of the spine patient outcomes research trial. Spine. 2009;34(21):2351–60. doi:10.1097/BRS.0b013e3181b8a829.
- Kaito T, Hosono N, Mukai Y, Makino T, Fuji T, Yonenobu K. Induction of early degeneration of the adjacent segment after posterior lumbar interbody fusion by excessive distraction of lumbar disc space. J Neurosurg Spine. 2010;12(6):671–9. doi: 10.3171/2009.12.SPINE08823.
- 60. Kumar MN, Baklanov A, Chopin D. Correlation between sagittal plane changes and adjacent segment degeneration following lumbar spine fusion. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deformity Soc Eur Sect Cervical Spine Res Soc. 2001;10(4):314–9.

- 61. Korovessis P, Repantis T, Papazisis Z, Iliopoulos P. Effect of sagittal spinal balance, levels of posterior instrumentation, and length of follow-up on low back pain in patients undergoing posterior decompression and instrumented fusion for degenerative lumbar spine disease: a multifactorial analysis. Spine. 2010;35(8):898–905. doi:10.1097/BRS.0b013e3181d51e84.
- 62. Djurasovic MO, Carreon LY, Glassman SD, Dimar 2nd JR, Puno RM, Johnson JR. Sagittal alignment as a risk factor for adjacent level degeneration: a casecontrol study. Orthopedics. 2008;31(6):546.
- 63. Keorochana G, Taghavi CE, Lee KB, Yoo JH, Liao JC, Fei Z, Wang JC. Effect of sagittal alignment on kinematic changes and degree of disc degeneration in the lumbar spine: an analysis using positional MRI. Spine. 2011;36(11):893–8. doi:10.1097/BRS.0b013e3181f4d212.
- 64. Rao MJ, Cao SS. Artificial total disc replacement versus fusion for lumbar degenerative disc disease: a meta-analysis of randomized controlled trials. Arch Orthop Trauma Surg. 2013. doi:10.1007/s00402-013-1905-4.
- 65. Lemaire JP, Carrier H, el Sariali H, Skalli W, Lavaste F. Clinical and radiological outcomes with the Charite artificial disc: a 10-year minimum follow-up. J Spinal Disord Tech. 2005;18(4):353–9.
- 66. David T. Long-term results of one-level lumbar arthroplasty: minimum 10-year follow-up of the CHARITE artificial disc in 106 patients. Spine. 2007;32(6):661–6. doi:10.1097/01. brs.0000257554.67505.45.
- 67. Huang RC, Tropiano P, Marnay T, Girardi FP, Lim MR, Cammisa Jr FP. Range of motion and adjacent level degeneration after lumbar total disc replacement. Spine J Off J N Am Spine Soc. 2006;6(3):242–7. doi:10.1016/j.spinee.2005.04.013.
- 68. Siepe CJ, Zelenkov P, Sauri-Barraza JC, Szeimies U, Grubinger T, Tepass A, Stabler A, Mayer MH. The fate of facet joint and adjacent level disc degeneration following total lumbar disc replacement: a prospective clinical, X-ray, and magnetic resonance imaging investigation. Spine. 2010;35(22):1991–2003. doi:10.1097/BRS.0b013e3181d6f878.
- 69. Putzier M, Funk JF, Schneider SV, Gross C, Tohtz SW, Khodadadyan-Klostermann C, Perka C, Kandziora F. Charite total disc replacement – clinical and radiographical results after an average follow-up of 17 years. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deformity Soc Eur Sect Cervical Spine Res Soc. 2006;15(2):183–95. doi:10.1007/ s00586-005-1022-3.
- 70. Berg S, Tropp HT, Leivseth G. Disc height and motion patterns in the lumbar spine in patients operated with total disc replacement or fusion for discogenic back pain. Results from a randomized controlled trial. Spine J Off J N Am Spine Soc. 2011;11(11):991–8. doi:10.1016/j.spinee.2011.08.434.

# How to Obtain the Best Lumbar Lordosis

# Jean-Paul Steib and Yann Philippe Charles

Instrumented lumbar fusion is a common operation worldwide. Nevertheless, lumbar lordosis is not always restored [1]. Lumbar lordosis, however, is essential for standing and ambulation in a balanced position [2]. During growth, for a child to stand, the human spine develops a lumbar lordosis. This lordosis changes during daily activities and with position (sitting, laying, or standing) [3]. It also changes over time, when degenerative changes occur in the lumbar spine [4]. Back pain can affect lordosis [5]. The best way to assess lumbar lordosis is via measurement on a lateral standing radiograph. Other imaging studies, such as CT or MRI, are not able to provide a good assessment of lordosis since this parameter should be assessed in the erect position. However, its measurement remains controversial [6]. It seems that global lordosis is the most realistic measurement. Lumbar lordosis is the key to sagittal balance of the spine. Lordosis is individual but significantly correlates with pelvic incidence [7]. After surgery, a lack of lordosis disturbs normal gait, imbalances the spine and is compensated by hyperlordosis at the adjacent level,

which in turn may lead to a new degenerative disease – adjacent segment degeneration [8, 9]. Therefore, a proper establishment of lordosis is important when considering lumbar fusion. The purposes of a spine fusion may be to restore lordosis, with instrumentation, and to fuse the spine in a proper position. As two thirds of the lumbar lordosis occurs at L4-S1 [10] and two thirds of lumbar fusions address this region, the ability to achieve a good lordosis of the lumbosacral spine represents the basis of surgery. Today, total lumbar disk replacement remains a popular strategy for treating lumbar disk degeneration. The prosthesis is a tool, which allows the patient to find the optimal lordosis during different daily activities. In this chapter, we describe the strategies to achieve optimal lordosis during lumbar spine during surgery.

#### 24.1 Patient Positioning on the Operating Table

The position of the patient on the table is the first step for a successful surgery.

• In prone position

The knee-chest position decreases lumbar lordosis. It is useful for non-instrumented spinal canal decompression or microdiskectomy, since it enlarges the interlaminar space. However, this position should be avoided for

J.-P. Steib, MD (⊠) • Y.P. Charles, PhD Service de Chirurgie du Rachis, Hôpitaux Universitaires de Strasbourg, 1 Place de l'hôpital, BP 426, 67091 Strasbourg, France e-mail: jean-paul.steib@chru-strasbourg.fr; yann.philippe.charles@chru-strasbourg.fr

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_24

instrumented fusion of the lumbar spine. The best position is flat on the table without flexion of the hips. This neutral position allows a natural lordotic posture and makes additional internal correction maneuvers possible that facilitate the acquisition of optimal lordosis through the instrumentation.

• In supine position

A roll placed under the patient's back or bending the operating table enables to increase lordosis in supine position. This position facilitates the correction and setting of segmental lordosis in anterior approaches. On the other hand, it can complicate the surgery by stretching the iliac vessels, thus making the exposition of the disks difficult and dangerous. Therefore, a flat patient positioning on the table is certainly the most reasonable option for the anterior access to the lumbar spine.

# 24.2 Release of the Spine

• Opening of facet joints

The purpose of posterior instrumented fusion is to obtain an arthrodesis of the lumbar spine in a lordotic position. Opening the facet joints during the posterior approach represents a useful method of releasing the lumbar spine, which in turn allows increasing lordosis and further grafting of the facets. This release can be performed using chisels or a saw by resecting the inferior articular process (the cranial part of the facet joint). This bilateral release enhances segmental mobility, especially in case of severe osteoarthritis with major osteophytes. Posterior compression maneuvers lead to shortening of resected posterior elements and open up the anterior intersomatic disk space, which might eventually be addressed with an additional cage. Section of the disks

The opening of the disks can be achieved either by anterior, posterior, or combined approaches. The posterior approach uses posterior cages (PLIF or TLIF) to maintain the intersomatic height with additional fusion between vertebral bodies. In anterior approaches using stand-alone cages (ALIF) with a primary stabilization system, the release of the anterior longitudinal ligament and the major part of the disk allows progressive dilatation of the intervertebral space and segmental lordosis restoration. In combined approaches, it can be very useful to release the disk by anterior approach first, prior to posterior deformity correction (kyphoscoliosis or posttraumatic malunion). The anterior approach can be minimal invasive video assisted. It is important to cut at least two thirds of the annulus and to decorticate the superior and inferior endplates. The rib, harvested during the anterior approach, can be used as a graft (Fig. 24.1). It is inserted as an onlay or a strut graft. The remaining intervertebral space is then filled with cancellous bone from the vertebral body. A postopera-



Fig. 24.1 Rib used as an inlay or a strut graft

tive CT analysis has demonstrated that this rib graft mixed with cancellous bone leads to massive fusion in the anterior column [11]. With this technique, the malleable bone graft allows the mobility during posterior reduction maneuvers and facilitates lordosis correction in combined approaches of the thoracolumbar spine [12].

• Osteotomies

A bony fusion between two vertebrae represents an obstacle to allow a good surgical lordosis restoration.

- The Smith-Petersen osteotomy is the easiest manner to obtain a mobilization of the spine when the anterior column is still mobile. We start the osteotomy by a Farcy's osteotomy (Fig. 24.2), which is achieved by removing the superior articular process (the caudal part of the facet joint). After exposure of the superior articular process, a curette is passed along the lateral part of the facet joint in order to cut all soft tissues adherent to it and to separate the isthmic vessels from



Fig. 24.2 Farcy's osteotomy

the bone. A saw is then used to cut the superior articular process straight ahead: the blade of the saw is set perpendicular to the facet in the transversal plane, adjacent to the cranial edge of the transverse process. The osteotomy is then completed using the chisel to ensure a safe separation of the facet towards the intervertebral foramen. Once the facet is cut, it is removed using a curette inserted in the articular cavity. The last insertions of the flavum ligament attached to the facet are removed. The same procedure is performed bilaterally. It is important to avoid electrocoagulation at the recessus and the foramen.

# 24.3 Instrumentation

 Hooks or pedicle screws are implants that facilitate lordosis reduction and fixation of the spine.

Hooks provide a posterior anchoring of the vertebrae. Compression on hooks at posterior spinal elements induces lordosis. On the other hand, posterior distraction has a kyphosing effect at the level of the disk. In the lumbar spine, hooks can only be used on the laminae. Hooks cannot be placed in a neutral position: a tension between the hook and the bone is required to ensure implant stability. For that reason, lamina hooks are used in claws bridging two adjacent vertebrae. Pedicle screws provide an anchoring in the three columns of the vertebra. As opposed to hooks, screws can be placed in a neutral position. Today, indications for hooks in lumbar spine are very restricted. Special situations and difficulties during pedicle screw placement may however require the use of hooks as a backup option.

Rods

It seems obvious that a rod in lumbar spine should be lordotic according to physiological sagittal curvatures of the spine. Therefore, it is not usually appropriate to use a straight rod in a curved spine. The shape of the rod and the amount of bending represent key factors in order to achieve the best possible lumbar lordosis. A rod that follows the spinal curvature is always better than a rod bridging the spine when low profile instrumentations are used.

#### 24.4 Reduction

- Compression
  - Hooks: As previously mentioned, the use of hooks required tension and posterior compression to achieve lordosis (Fig. 24.3). Compression of posterior vertebral elements induces a posterior narrowing of the intervertebral space and an anterior elevation of the superior vertebra.
  - Screws: The anchorage of the screws is posterior and anterior at each vertebra. Compression on monoaxial screws does not create or induce lordosis if the shape of the rods is straight or curved (Fig. 24.4). Therefore, a lordotic bending of the rods is mandatory to achieve sagittal alignment correction. The center of rotation is posterior with a rod-screw construct if the orientation is 90° between the rod and the monoaxial screw. Polyaxial screws work like hooks: a posterior compression on the screws induces a lordosis on the anterior column by distraction at the level of the disk (Fig. 24.5).



Fig. 24.3 Posterior compression using hooks



Fig. 24.4 Posterior compression using lordotic rod and monoaxial screws







Fig. 24.6 Bending of the rod pushing the screws forward

• Translation

Translation (approximation) is very popular today with instrumentations using persuader systems of the rod. This technique is used on a regular base with percutaneous instrumentation in MIS. With this practice, care should be taken that the vertebrae are not pulled backward to the rod, which might happen if the shape of the rod is too flat. It is mandatory to pull back the end vertebrae of the lordotic construct and to push the apical vertebrae anteriorly. Accurate bending of the rod is mandatory and is achieved by giving a lordotic shape to the rod prior translation and rotation of the apical vertebrae in the sagittal plane. Polyaxial screws facilitate the rod-screw connection with multilevel persuaders. However, the screws must be in line in the sagittal plane to allow lordotic opening of the spine by pushing the curved rod into the screws with the persuaders.

In situ bending

The principle is to mimic the shape of the spine and then to make the spine follow the shape of the rod. Instead making the vertebra following a prebent rod as with the approximation technique, the rod is inserted without any stress onto the uncorrected (flat) spine. The screws are closed but not locked once the rod is in place. Bending irons are set on the two rods, right and left symmetrically, and the rods are then bent inside the patient by pushing the bending irons together. This maneuver applied to the rod induces a movement of the screws, and subsequently of the vertebra, which are pushed anteriorly into a lordotic alignment. This maneuver is not dangerous while the screw is pushed anterior and not pulled back (Fig. 24.6). Bending has to be repeated along the rod, by little forces given in many times. The reduction is easily obtained by a 90° rodscrew connection with monoaxial screws, thus giving the spine the shape of the rod (Fig. 24.7).

#### 24.5 Bone Graft

The bone graft is used to maintain the lordosis at long term. The purpose of the instrumentation is to obtain and to maintain the correction until the spine is fused. The absence of fusion or pseudarthrosis is mechanically associated with micro movement, followed by macro movement, mobilization of the implants with bony lysis around the pedicle screws (radiologic evidence of halo), and finally rod breakage. The consequence is back pain and loss of lordotic correction. The site of grafting



Fig. 24.7 Example of correction obtained by in situ bending

(anterior, posterior, or 360° fusion) can be relevant for the success of fusion. In the case of suspected non-fusion during follow-up, it is important to verify the consolidation of the graft using radiographs (measurement of segmental lordosis, screw, or rod breakage), CT scan (evidence of fusion between the vertebrae), and isotopic CT (hyper- or hypo-fixation). The combination of CT and isotopic CT is useful to determine the exact area of non-fusion.

- Fusion material
  - Autologous bone

Iliac crest remains the gold standard and the best way of obtaining fusion. The incidence of pain caused by graft harvesting (donor-site morbidity) is closely related to the surgical technique. Cancellous bone should be preferably harvested without cracking the cortex and damaging the sacroiliac joint. Furthermore, a good reconstruction of the iliac crest is mandatory to avoid a painful traction on the gluteus medius. The iliac crest can be used as bone chips or as a tricortical bone block.

- Pieces of osteotomies

In some cases, the amount of bony resection from the vertebrae is sufficient to allow bone grafting. Soft tissues need to be entirely stripped from the bone to avoid the risk of pseudarthrosis. The harvested bone should not be mixed with cartilage from the facet joints or necrotic bone from vertebral bodies.

- Bone substitutes

There is no evidence or clear consensus on the possibility of obtaining a bony fusion with bone substitutes. They are used to induce or to facilitate fusion after preparation of bone surfaces on the spine

- Posterior fusion
  - Posterior fusion at the level of the posterior arch is not able to maintain lordosis because of a certain amount of remaining elasticity at the anterior column. This graft works in tension and is in bad position to keep the correction.
  - Posterolateral fusion is located between the transverse processes, at the level of the posterior wall and the vertebral bodies. This type of fusion is closer to the axis of rotation in flexion-extension. Therefore, posterolateral fusion is in neutral position and leads to a better resistance, thus maintaining the lordotic correction.
  - Articular fusion is optimal because it is in line with the two posterior columns of the spine, at the level of movement through the facet joints. A continuous fusion through all facet joints leads to adequate stability.



Fig. 24.8 Cage closing the anterior gap between the two anterior endplates

- Anterior fusion
  - PLIF, TLIF, ALIF, XLIF, and other cages are currently used today. The method of placing the cages followed an evolution with time. The main purpose of using a cage is to add an anterior support and to close the anterior gap between the two endplates of the intervertebral space (Figs. 24.8 and 24.9). Mechanically, an interbody fusion offers the best support to keep segmental lordosis. It is important to avoid the complications related to the specific approaches of each cage: nerve root stretching, destabili-

zation of the spine by large arthrectomy, and vascular or femoral nerve injury.

– An anterior parallel distraction of the disk does not provide lordosis and the endplates should not stay parallel once the cage is inserted. The segmental lordosis of the spine is triggered by the lordosis of the cage. The anterior support helps to keep the anterior opening of the disk. An anterior bone graft seems necessary when the operated disk is narrow preoperatively and elevated after surgery. A gap between two endplates after posterior segmental lordosis



Fig. 24.9 Example of result obtained with an anterior cage

correction will be a weak point in the construct and needs to be filled by an anterior support.

#### 24.6 Total Lumbar Disk Replacement

Obviously, the difference with fusion is the movement, thus allowing changes of segmental lordosis, during positional changes. The surgeon does not set the spine into fixed lordosis, but the implant allows the patient to adapt his own lordosis depending on his position. The patient is operated in supine position: flat on the table. There is no lordosis in supine position. The extension of the lumbar spine stretches the vessels and may complicate the anterior access to the disk. Once the disk is exposed, it is largely opened anteriorly. The intervertebral space is progressively distracted while disk material is removed. Overdistraction is not recommended to avoid a possible painful decompensation at the level of the facet joints. The posterior longitudinal ligament is usually retracted in the degenerative lumbar spine. In order to



**Fig. 24.10** To achieve a parallel distraction of both endplates and adequate prosthesis placement, it is recommended to release the posterior longitudinal ligament

achieve a parallel distraction of both endplates and adequate prosthesis placement, it is recommended to release this ligament (Fig. 24.10). The artificial disk has to be inserted while maintaining the endplates parallel intraoperatively. The posterior ligament release will further allow the automatic lordotic setting of the prosthesis and enhance range of motion during flexion of the spine. Total disk replacement is not recommended when the pelvic incidence is too high since the segmental lordosis is relatively high at L4–L5 and L5–S1, thus preventing proper functioning of the prosthesis. Segmental instability in flexion-extension and associated facet degeneration represent two other main contraindications for total disk replacement.

#### Conclusion

Lumbar lordosis represents a key factor that needs to be considered and restored during lumbar surgery. The role of the surgeon is to position the spine of his patient to obtain and to keep the best possible sagittal angle depending on the amount of pelvic incidence. One of the main drawbacks of lumbar surgery is the lack of lordosis, which will lead to adjacent segment degeneration and possibly a bad clinical result at short or midterm. We described some technical aspects of lumbar surgery including some tricks that may allow successful lumbar lordosis acquisition.

# References

- 1. Schwab F, Patel A, Ungar B, et al. Adult spinal deformity-postoperative standing imbalance: how much can you tolerate? An overview of key parameters in assessing alignment and planning corrective surgery. Spine. 2010;35:2224–31.
- Been E, Kalichman L. Lumbar lordosis. Spine J. 2014;14:87–97.
- Andreasen ML, Langhoff L, Jensen TS, Albert HB. Reproduction of the lumbar lordosis: a comparison of standing radiographs versus supine MRI obtained with straightened lower extremities. J Manipulative Physiol Ther. 2007;30:26–30.
- Papadakis M, Papadokostakis G, Kampanis N, et al. The association of spinal osteoarthritis with lumbar lordosis. BMC Musculoskelet Disord. 2010;11:1.
- Murrie VL, Dixon AK, Hollingworth W, et al. Lumbar lordosis: study of patients with and without low back pain. Clin Anat. 2003;16:144–7.
- Polly DW, Kilkelly FX, McHale KA, et al. Measurement of lumbar lordosis. Evaluation of intraobserver, interobserver and technique validity. Spine. 1996;21:1530–5.
- Roussouly P, Nnadi C. Sagittal plane deformity: an overview of interpretation and management. Eur Spine J. 2010;19:1824–36.
- Umehara MN, Zindrick MR, Patwardhan AG, et al. The biomechanical effect of postoperative hypolordosis in instrumented lumbar fusion on instrumented and adjacent spinal segments. Spine. 2000;25:1617–24.
- Kumar MN, Baklanov A, Chopin D. Correlation between sagittal plane changes and adjacent segment degeneration following lumbar spine fusion. Eur Spine J. 2001;10:314–9.
- Le Huec JC, Charosky S, Barrey C, et al. Sagittal imbalance cascade for simple degenerative spine and consequences: algorithm of decision for appropriate treatment. Eur Spine J. 2011;20:S699–703.
- Antoni M, Charles YP, Walter A, Schuller S, Steib JP. Fusion Rates of Different Anterior Grafts in Thoracolumbar Fractures. J Spinal Disord Tech. 2013. [Epub ahead of print] PMID: 24077416.
- Steib JP, Mezghani S, Charles YP, Mitulescu A. Double approach in thoraco-lumbar malunions. Eur Spine J. 2010;19 Suppl 1:S48–51.

# **Bone Substitutes**

Alem Yacob, Glenn S. Russo, and Jonathan N. Grauer

#### 25.1 Introduction

Spinal fusion is a common surgical procedure and is often considered for degenerative pathologies as well as deformity, trauma, or tumor indications. Approximately 200,000-250,000 patients in the USA annually undergo a spinal fusion procedure, the majority of which are lumbar spinal fusions [1, 2]. Spinal fusion constitutes over half of all bone graft procedures each year [1]. Spinal arthrodesis eliminates motion between segments with the aim of decreasing instability, maintaining alignment, and protecting the neural elements. Fusion is a biologic event in which bridging bone forms between adjacent motion segments.

In order to promote biologic fusion, spinal segments can be immobilized with or without instrumentation. Spinal arthrodesis with internal fixation hardware provides transient stability to the spinal segments that can facilitate fusion.

A. Yacob, MD, MSc • G.S. Russo, MD, MS J.N. Grauer, MD (⊠) Department of Orthopaedics and Rehabilitation, Yale–New Haven Hospital, Yale University School of Medicine, PO Box 208071, New Haven, CT 06520-8071, USA e-mail: glenn.russo@yale.edu; jonathangrauer@yahoo.com; jonathan.grauer@yale.edu However, if biologic fusion is not realized over time, hardware failure will generally occur.

Autologous bone remains the "gold standard" graft material for spinal fusion and is most often obtained from the iliac crest or from bone local to the fusion/decompression site. Autologous bone graft may be limited in its supply and often requires a second surgical site with the associated additional risks. Due to these considerations, the use of iliac crest bone autograft (ICBG) has decreased over the past decade, and the use of bone graft substitutes and bone graft extenders has increased.

As a result of the above considerations, there has been significant work to develop alternative graft materials to supplement or even replace autogenous bone graft. This chapter will discuss the biological processes underlying spinal fusion and the mechanisms by which these graft materials serve to facilitate arthrodesis.

# 25.2 The Biology of Spinal Fusion

Spinal fusion consists of a tightly controlled series of cellular and molecular events that are dependent upon the biologic conditions of the fusion environment. There are three main cellular components that must be present for bone formation to occur. Osteogenic cells are precursors that populate the fusion site and can be induced to differentiate into osteoblasts (bone-forming cells).

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016 J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_25

Osteoinductive factors are proteins that induce the recruitment and differentiation of the osteogenic cells. Finally, osteoconductive matrices serve as a scaffold onto which bone formation and neovascularization can occur.

The process of bone formation at a spinal fusion site starts with inflammation, continues with reparative processes, and then finishes by remodeling [3]. Overall, this is a cascade that mirrors fracture healing. During the inflammatory phase, there is a hematoma that surrounds the fusion site with inflammatory cells and allows for neovascularization and the recruitment of cells with osteogenic potential. Next, in the reparative phase, the soft hematoma begins to solidify into new bone via intramembranous and endochondral bone formation while the graft bone is gradually resorbed [4]. Finally, during the remodeling phase, there is organization of the new bone and continued resorption of the graft bone.

There are a myriad of systemic factors that may compromise a patient's ability to achieve a fusion. Issues can be with the local fusion environment, the mechanical stability at the fusion site, or systemic host factors.

In terms of the local fusion environment, the preparation of the fusion site is believed to be critical. The bony surfaces to be fused need to be decorticated. This allows for delivery of local osteogenic cells and provides for a local blood supply to the fusion site. Maximizing the area of decorticated bone increases the surface area available for fusion. Minimizing trauma to adjacent anatomic structures, and allowing for vascularity of surrounding structures, such as muscle edges, allows for or access of further precursor cells and blood supply.

In terms of mechanical stability at the fusion site, this is needed to allow for bone formation. Often, this is achieved with instrumentation, but bracing or immobilization can help achieve this as well [5]. As the newly formed bone is loaded, it matures according to Wolff's law of bone formation. This explains why bone is found to form more easily under compression than tension. It is for this reason that anterior column interbody constructs are thought to have an advantage [6, 7]. In terms of systemic host factors, examples such as nicotine, nutrition, and others can affect bone formation [6]. For example, individuals who smoke have consistently demonstrated higher rates of pseudarthrosis because nicotine use interferes with angiogenesis, bone metabolism, and regeneration [8–12]. Similarly, drugs that inhibit the systemic inflammatory cascade such as nonsteroidal anti-inflammatory have been shown to inhibit new bone formation [13, 14]. Additionally, decreased thyroid and/or growth hormone function can increase the risk of delayed union. Others have demonstrated that nutritional status can influence bone healing as well [15].

Lumbar fusion is the most commonly performed spinal fusion procedure. Traditionally, posterolateral fusion is the most common approach and seeks to achieve bone formation between adjacent transverse processes. Historic nonunion rates for uninstrumented posterolateral fusions have been reported to be between 40 and 60 % [16, 17]. In order to minimize motion, decrease the reliance on bracing, and facilitate biologic fusion, instrumentation has become the standard.

Interbody fusion has been developed to directly support the anterior column of the lumbar spine. This technique is used to increase the surface areas for fusion and decrease the biomechanical forces on posterior constructs, especially when considering the limited posterior bony surface area when applying posterior minimally invasive techniques [18].

Autologous corticocancellous bone was the first and initially only option that spinal surgeons used for bone graft and it remains the gold standard. Its widespread utility stems from the fact that it is the only single material that contributes all of the elements required for bone regeneration (i.e., osteogenic cells, osteoinductive signaling molecules, and an osteoconductive matrix). However, as discussed earlier, there are reasons to strive to avoid autograft and this has fueled the desire for bone graft supplements and substitutes.

Commonly used bone graft substitutes include allograft products, recombinant osteoinductive factors, and ceramics.

#### 25.3 Mineralized Allograft

Allogeneic bone is one of the most commonly transplanted tissues among humans, and it has been used for numerous orthopedic applications including spinal fusion [19]. Bone allografts are obtained from a deceased donors, are readily available, and thus do not face the same supply limitations as autograft. Mineralized allograft (fresh frozen and freeze-dried) is considered non-osteogenic, osteoconductive, and very mildly osteoinductive [20].

Allograft can consist of cortical bone, cancellous bone, or a corticocancellous mixture. The cortical allograft bone provides better structural support through its ability to resist compressive forces but is slower to incorporate and resorb. Cancellous allograft bone has limited ability to resist compressive forces and resorbs more quickly but provides a more intricate framework for osteoconduction.

Allograft bone is processed either by freezing (fresh frozen) or lyophilization (freeze-drying) after it is harvested from a human cadaver in a manner that avoids infectious agents, decreases its antigenicity, and preserves the grafts for storage. This sterilization process can biomechanically weaken the bone (compared to fresh allograft), significantly decreases any osteoinductive potential, and eliminates osteogenic cells [21].

There is the underlying question of allograft being capable of transferring disease, such as human immunodeficiency virus (HIV). HIV transmission has been documented in non-spinal fresh-frozen allografts, but not documented in freeze-dried allograft cases [22].

When structural allograft is implanted under compression in the anterior column, it is associated with relatively high fusion rates, both in the cervical and the thoracolumbar spine [23, 24]. Fusion rates for single-level ACDF with allograft are similar to those with autograft, but the union rate decreases for allograft more so than autograft when used in multilevel ACDFs. As mentioned earlier, this is in accordance with Wolff's law of bone formation.

Cancellous allograft is preferred to cortical allograft for posterior spinal fusion as it allows

for better vascular ingrowth and has better osteoinductive properties. Cortical bone isn't needed for structural support because the compressive forces are sufficiently low. When nonstructural allograft bone is placed under tension, as in the posterior spine, it incorporates at a slower rate than similarly positioned autograft and leads to lower rates of fusion when used alone [25, 26]. Thus, for posterior spinal applications, allograft is often employed as a bone graft extender in combination with autograft rather than as a substitute for autograft.

An exception to this is in the pediatric scoliosis population, where allograft alone appears to be a reasonable alternative to autograft for posterolateral fusions. In these patients, the use of allograft resulted in similar rates of arthrodesis compared to autograft, with far less morbidity [27–29].

# 25.4 Demineralized Bone Matrix Allograft

Demineralized bone matrix (DBM) was introduced in the early 1990s. DBMs are also derived from cadaveric bone, have lesser osteoconductive properties, and have some low-level osteoinductive properties.

The donor bone is decalcified (removal of calcium and phosphate) by acid extraction, exposing the extracellular matrix and a composite of type I collagen and noncollagenous proteins, including low concentrations of bone morphogenetic proteins (BMPs) and other constitutively expressed growth factors [24]. There are many proprietaryprocessing techniques, but they all have the above-described process in common. Once extracted, these products are combined with carriers to improve product and handling properties (e.g., pastes, putty, chips, and gels) [2].

Research has shown that there is variability in the osteoinductive activities of commercial DBMs both when comparing different company brands and within a single company's brand [30, 31]. In one study comparing the osteoinductive properties of several commercially available DBM products, the fusion rates of athymic rats ranged from 0 to 80 %, in part demonstrating the great deal of variability between the different DBM formulations [32]. This is thought to be due to the varying quantity and composition of osteoinductive factors in various DBM preparations [33].

DBMs have been used successfully as autograft extenders in several animal studies [34, 35]. Animal studies have also shown that posterolateral arthrodesis rates were significantly improved with a composite graft consisting of DBM and allograft bone compared to allograft alone [36, 37].

In a multicenter, prospective equivalency trial with each patient serving as his/her own control, researchers compared Grafton DBM gel/autograft on one side vs. iliac crest autograft on the contralateral side of the patient's instrumented posterolateral lumbar spine [38]. Of 120 patients followed up at 24 months, fusion was found for 42 cases (52 %) on the Grafton DBM (Osteotech Inc., Eatontown, NJ)/autograft side and in 44 cases (54 %) on the autograft side. The authors concluded that the Grafton DBM gel allowed them to use less ICBG.

Another prospective, multicenter randomized clinical trial assigned 46 patients undergoing a single-level instrumented posterior lumbar fusion to receive Grafton DMB/local bone or only iliac crest bone autograft [39]. At 2 years, no statistical difference was noted in the radiographic fusion rate of the Grafton (86 %) vs. the ICBG (92 %) groups (p=0.23). The authors concluded that for this application, Grafton combined with local bone is not different than ICBG.

One study compared one- and two-level posterolateral fusions using lamina autograft (i.e., local bone graft) and demineralized bone matrix (Osteofil/ICM, Sofamor Danek, Memphis, TN) in a 50:50 mix. Based on dynamic radiograph assessment, the study found the one-level fusion rate to be 98 % and the two-level fusion rate to be 96 % with similarly improved SF-36 clinical outcomes 1-year after surgery [40].

In a separate study of posterolateral uninstrumented lumbar fusions, 79 patients underwent an average of 4.9-level lumbar laminectomies with an average 2.0-level non-instrumented posterolateral fusions using lamina autograft and DBM [41]. Six months postoperatively, 17.3 % patients were found to have radiograph (computed tomography and dynamic x-ray) confirmed pseudarthrosis. Only 1 of the 13 patients diagnosed with pseudarthrosis had persistent clinical symptoms that lead to reoperation.

These studies support the use of demineralized bone matrix in the posterior lumbar spine for reducing the amount of autograft needed to achieve similar fusion rates. The studies found similar results in both posterior and posterolateral fusions as well as single- and two-level fusions.

There is minimal clinical data on the use of DBM for anterior lumbar interbody fusion (ALIF). A study by Thalgott et al. retrospectively studied 50 patients who received DBM combined with coralline hydroxyapatite, a synthetic bone graft material, in an anterior cage and an instrumented posterior fusion [42]. Their intention was to assess the utility of coralline hydroxyapatite in the anterior spine. Ninety-six percent of patients achieved radiographic fusion at 3-5 years after surgery. The authors concluded that the combination of titanium mesh cages, coralline hydroxyapatite, and demineralized bone matrix is effective for anterior interbody fusion of the lumbar spine when used as part of a rigidly instrumented circumferential fusion.

The relatively fewer clinical studies utilizing demineralized bone matrix in the anterior lumbar spine may reflect the anterior spinal column's better fusion environment relative to the posterior spinal column.

#### 25.5 Synthetics

Another alternative to autograft for spinal fusion is a synthetic osteoconductive scaffold, such as ceramic. Examples of synthetic ceramic materials that facilitate new bony ingrowth include beta-tricalcium phosphate, hydroxyapatite, and calcium sulfate.

These products do not function well by themselves because they only provide an osteoconductive matrix. It is for this reason that they must be used as a bone graft extender in conjunction with other materials that provide the osteoinductive and osteogenic characteristics.

These synthetic bone graft extenders have been frequently used in part based on their inability to cause an immunogenic response, no risk of human disease transmission, and ample supply. One of the inherent disadvantages of ceramics is that most are relatively weak and brittle. When ceramics are introduced into the anterior spinal column, they must be protected from the significant compressive forces by internal fixation until the graft is incorporated into the new bone growth [24]. Like allograft, ceramics are also inferior to autogenous bone when placed under tension, as in the posterior spine [43, 44].

Other biologically active materials such as DBM, osteoinductive growth factors, or ions (e.g., silicate, magnesium) may be combined with a ceramic osteoconductive matrices to form a composite graft that can then lead to increased amounts of bone formation [45, 46]. Silicate and other bioactive ions are reported to play a role in bone metabolism via stimulation of osteocytes as well as angiogenesis [47]. One animal study on magnesium didn't show a difference in the rate of bone fusion vs. controls but did show an improvement of the bone fusion quality by histological and scanning electron microscopy assessment [48].

Concerns about ceramics include their ability to withstand compressive loads, particularly those seen in the anterior spinal column, their ability to deliver osteogenic and osteoinductive factors, and their resorption rates. Ceramic biomechanics involves a trade-off in porosity and resorption rates [49]. As porosity increases, so does surface area, which allows for increased interaction between the ceramic and the local bone fusion environment. Less porous materials can have slower resorption rates which if long enough can lead to the implant being semipermanent [50]. This is not ideal as it can cause a foreign body reaction and is a potential nidus for infection.

In a clinical study, a non-randomized retrospective review of 1- or 2-level lumbar degenerative disorders assessed the impact of silicate on an instrumented posterolateral bone fusion. Using silicate hydroxyapatite ceramic as a bone graft substitute, without the addition of iliac crest bone graft, researchers showed CT confirmed successful lumbar posterolateral fusion rates of 77 % in patients with 2-year follow-up [49].

As mentioned, Thalgott et al. set out to assess the clinical and arthrodesis efficacy of coralline hydroxyapatite as a synthetic osteoconductive bone graft substitute in the anterior lumbar spine in addition to a posterolateral fusion [42]. The coralline hydroxyapatite's porosity is similar to that of cancellous bone. A prospective case series involving 50 patients (including 14 smokers and 29 revision lumbar surgeries) who received 1- and 2-level lumbar interbody fusions as well as a posterolateral fusion procedure, reported a 96 % success rate after implanting titanium mesh cages packed with a DBM and a coralline hydroxyapatite carrier [42].

#### 25.6 Bone Morphogenetic Proteins and Osteoinductive Growth Factors

The discovery of bone morphogenetic proteins (BMPs) is attributed to Urist and colleagues in the 1960s [51–55]. BMPs are osteoinductive in nature and operate by inducing pluripotent mesenchymal stem cells to become bone-forming cells. With the use of recombinant gene technology, large quantities of osteoinductive growth factors can now be produced in a purified form for clinical use. Multiple strongly positive animal studies supported the continued research on rhBMP-2 in clinical trials [56–58]. The efficacy of these recombinant human bone morphogenetic proteins (rhBMPs), including rhBMP-2 and rhBMP-7 (also known as osteogenic protein-1, or OP-1), has been studied in a number of clinical studies [59–61].

In a multicenter, prospective, randomized trial, 279 patients with lumbar degenerative disk disease underwent ALIF with lumbar-tapered fusion devices (LT-CAGE, Medtronic) filled with either rhBMP-2 (1.5 mg/mL) on an absorbable collagen sponge carrier or autogenous iliac crest bone graft [59]. The radiographic fusion rate of

the patients receiving rhBMP-2 was 94.5 % and that of the ICBG patients was 88.7 % at 24 months. The authors concluded that rhBMP-2 in combination with the LT-CAGE for this surgical indication and approach was an appropriate bone graft substitute. As a result of this and other studies, the FDA approved the combination of rhBMP-2 with an absorbable collagen sponge (INFUSE, Medtronic Sofamor Danek, Memphis, TN) for use with threaded fusion devices in the anterior lumbar spine in the treatment of degenerative disk disease at one level from L4 to S1.

Other uses of INFUSE have also been studied. Some of these have been in off-label cohort studies (such as the cervical spine) and others have been in randomized trials (such as a higher dose in posterolateral, called AMPLIFY).

After encouraging results from a pilot study [62], Dimar et al. carried out a randomized prospective study, which was also part of the Food and Drug Administration (FDA) Investigational Drug Exemption (IDE) for rhBMP-2. The researchers looked at 463 patients treated with single-level instrumented posterolateral fusion [63]. The authors compared iliac crest bone autograft to an rhBMP-2 matrix. The matrix was a 20-cm<sup>3</sup> block of bovine type I collagen carrier containing 15 % hydroxyapatite and 85 %  $\beta$ -tricalcium phosphate particles to create a compression-resistant graft. All patients had symptomatic single-level lumbosacral degenerative disease (with no greater than grade one spondylolisthesis). In addition to showing that the rhBMP-2 group had statistically significant shorter operative times and decreased blood loss, the authors noted that at 24 months, fusion was evident in 96 % of the patients in the recombinant human bone morphogenetic protein-2 matrix group compared with 89 % in the iliac crest bone graft group (p = 0.014).

Regarding cancer risk related to rhBMP-2, Dimar et al. noted that at 24 months the rate of all cancers in the investigational arm of this AMPLIFY study was 3.3 % vs. 0.9 % in the control arm (p=0.107) [63, 64]. Follow-up data submitted to the FDA at 60 months showed the cancer incidence in the same subjects to be 5.0–6.3 % for the rhBMP-2 arm versus 2.2 % for the ICBG arm (no p-value is given). In 2011, based in part on data that lead to the question of associating rhBMP-2 with an increased cancer rate, the FDA rejected Medtronic's application for approval of rhBMP-2 at the higher dose (AMPLIFY) for use in posterolateral lumbar fusion.

In a retrospective study comparing off-label use of rhBMP-2 with local bone graft (LBG) in 70 patients undergoing primary one- or two-level posterior lumbar interbody fusion (PLIF) and transforaminal lumbar interbody fusion (TLIF), the authors found similar rates of radiographic fusion (89.5 % LBG vs. 94.1 % rhBMP-2, p=0.61) but a higher complication rate for the rhBMP-2 (41.2 % vs. 10.5 %, P=0.05) [65]. These complications included radiculopathy, radiculitis, adjacent segment disease, development of ectopic bone, and vertebral osteolysis.

In recent years, with the increasing on- and off-label use of INFUSE as well as reported clinical complications, the efficacy and safety of INFUSE has come under scrutiny. This was somewhat unexpected given that the 13 original industry-associated BMP-2 studies did identify several surgical complications but ultimately reported no (0 %) device-/BMP-2-related or unanticipated adverse events.

Carragee et al. compared the published data from 13 industry-associated trials on rhBMP-2 covering 780 patients and compared this with "available FDA data summaries, follow-up publications, and administrative and organizational database analyses" [66]. The authors concluded that rhBMP-2 was associated with adverse events 10–50 % of the time.

Carragee et al. found that ALIF was associated with higher rates of "implant displacement, subsidence, infection, urogenital events, and retrograde ejaculation" compared to controls.

Further, they noted that PLIF was "associated with radiculitis, ectopic bone formation, osteolysis, and poorer global outcomes" at higher rates than controls. The authors ultimately questioned the methodology and reporting of biases of the original rhBMP-2 trials.

Faced with several questions regarding the complications of INFUSE [67–72], Medtronic

(Minneapolis, Minnesota) voluntarily submitted all the INFUSE clinical trial individual participant level data as well as supplied funding for analysis to the Yale University Open Data Access (YODA) Project in 2011. The YODA Project then commissioned two outside academic groups to independently analyze the individual participant data and report their conclusions [73, 74].

Fu et al. concluded that INFUSE was similarly effective to ICBG for lumbar ALIF and PLIF. They found that INFUSE was associated with increased risk of complications (adverse events, wound complications, dysphagia or dysphonia) when used for anterior cervical spine fusion, an increased risk of cancer (all approaches and levels grouped together), and associated with a reporting bias that failed to give sufficient attention to the risks of its use [73]. They state that an earlier, more complete reporting of the data would have been helpful to clinicians and patients.

Simmonds et al. concluded that INFUSE did improve the rate of bony fusion when compared with ICBG but they found that this was not associated with less post-fusion pain [74]. Further, they found that the data was suggestive of a higher rate of adverse events but stopped short of making firm conclusions regarding the true rate of adverse events associated with rhBMP-2 given the small sample size and numerous nonrandomized studies that were used in the analysis. Consistent with other study findings, Simmonds et al. found that the rate of cancer was "nearly double" that of ICBG. They concluded that despite this, the absolute rate of cancer was low for both, and it is unclear if the risk is clinically significant [74].

Of historical interest but not clinically approved for lumbar fusion by the FDA at this time is BMP-7 also known as osteogenic protein-1 (OP-1). The OP-1 gene was identified in the 1980s, approved for long bone fractures in 2001 under a Humanitarian Device Exemption (HDE), and in 2004, the FDA approved OP-1 Putty (Stryker and subsequently Olympus) under an HDE for performing revision posterolateral lumbar spinal fusion based on several positive animal and early clinical studies [16, 75–80]. Based on a clinical trial [81] that failed to show non-inferiority, in 2009, an advisory panel to the FDA rejected Premarket Approval (PMA) for OP-1 Putty for instrumented posterolateral lumbar spine fusion.

#### 25.7 Gene Therapy: Intracellular Signaling Proteins

The benefits of gene therapy are thought to be twofold: first, using current replication techniques producing complimentary deoxyribonucleic acid (cDNA) is cheaper than producing an osteoinductive protein and second, delivery of the DNA rather than the protein may allow for a more prolonged delivery of the final osteoinductive protein through activation of the cascade [1].

Importantly, gene therapy requires a vector to deliver the intracellular proteins. In one experiment, the cDNA was delivered via an adenovirus. Alternatively, a plasmid was used but the transfection rate was lower than hoped for [82, 83]. As the cellular and molecular mechanisms underlying these novel osteoinductive signaling proteins continue to be elucidated, the clinical indications may expand to include them as alternatives to autogenous bone for spinal fusion and numerous other orthopedic applications.

Several cell signaling proteins, such as LMP-1, NELL-1, and rhGDF-5, have been studied for their use in ex vivo gene therapy. LIM mineralization protein-1 (LMP-1) is an osteoinductive growth factor that potentiates the cellular response to exogenous BMPs [82-88]. NEL-like molecule-1 (NELL-1) is a protein expressed predominantly in neural tissue that encodes epidermal growth factors in cells of the osteochondral lineage; it was first noted to be overexpressed in patients with craniosynostosis [89, 90]. Recombinant human growth and differentiation factor-5 (rhGDF-5) is another example of another type of signaling molecule that has earned some attention from the scientific community [86, 87].

Gene therapy techniques may allow for the sustained local release of these molecules at more physiologic levels. The hope is that these techniques would result in a more potent osteoinductive signal to the surrounding tissues. Presently, however, the safety and efficacy of gene therapy techniques have not yet been sufficiently established to justify its widespread use in a clinical setting.

#### 25.8 Autologous Stem Cells

Autologous bone marrow represents another source of osteogenic cells and osteoinductive material for spinal fusion. The most significant advantage of this technique is that aspiration of bone marrow has much less morbidity than the procurement of iliac crest autograft. When used in combination with an osteoconductive matrix, bone marrow aspirate forms a composite graft that may be an effective alternative to autogenous bone.

In one animal study, aspirates were used as bone graft extenders and were found to significantly increase the rate of arthrodesis for posterolateral fusions in rabbits [91]. The major limitation to this technique is that unfractionated bone marrow has only moderate osteogenic potential. There is a low concentration of mesenchymal stem cells in the graft harvest - 1 in 10,000 cells. However, adipose tissue has been noted to have a higher concentration of these stem cells (1 in 4000 cells). With this as a foundation, Zuk et al. were able to use the BMP signaling cascade on human adipose-derived stem cells to induce cellular differentiation into the osteoblastic lineage [92]. Moreover, Hsu et al. used these stem cells, transfected with AdBMP-2 (adenoviral vector containing BMP-2), to achieve posterolateral spine fusion in an athymic rat model [93].

To further address the problem of limited stem cell availability, attempts have been made to increase the effective concentration of osteoprogenitor cells in these aspirates by selectively retaining these cells within an osteoconductive matrix or by expanding the number of these mesenchymal stem cells in cell culture before transferring them to the matrix [91].

In a prospective randomized study, 24 patients undergoing one-, two-, or three-level lumbar fusions were randomized to receive autologous bone marrow concentrate (from the iliac crest) combined with allograft on one side of their spinal fusion and then iliac crest bone autograft on the other side of their spinal fusion [94]. The authors concluded that there was no difference in the radiographic fusion rate for autogenous bone marrow concentrate combined with allograft as compared to autologous iliac crest bone graft in the lumbar spine.

Bone graft substitutes utilizing stem cell options are continuing to be the focus of significant research. As of yet, there is insufficient evidence to conclude whether bone marrow aspirates or bone graft substitutes will serve only as bone graft extenders or if they will be true alternatives to autogenous bone.

#### Conclusion

The development of bone graft alternatives has progressed rapidly over recent years, with several options now available for various clinical applications. Iliac crest autograft is still recognized as the gold standard graft material for spinal fusion, because it is the only graft option that contains osteogenic cells, osteoinductive growth factors, and an osteoconductive matrix.

Since no other single method to date provides all of these elements necessary for bone formation, the future of bone graft research will likely be focused on creating new composite grafts. These composite grafts will consist of multiple biologically active materials implanted together and acting synergistically to enhance spinal fusion. Further advances in the understanding of the intricate cascade of molecular and cellular events underlying spinal fusion will allow for continued refining and development of new grafting techniques. As novel approaches are developed, such as gene therapy, additional studies will be required to evaluate the efficacy, safety, and cost-effectiveness of both the new and existing techniques.

However, regardless of future advances, the success of these complex surgeries will remain dependent upon the basic principles essential to achieving a solid arthrodesis: proper patient selection, selecting an appropriate bone graft material, optimizing the biological environment, preparation of the fusion bed, and maintaining adequate biomechanical stability during bone formation.

#### References

- Yoon ST, Boden SD. Spine fusion by gene therapy. Gene Ther. 2004;11:360–7. doi:10.1038/ sj.gt.3302203.
- Aghdasi B, Montgomery SR, Daubs MD, Wang JC. A review of demineralized bone matrices for spinal fusion: the evidence for efficacy. Surg J R Coll Surg Edinb Irel. 2013;11:39–48. doi:10.1016/j. surge.2012.08.001.
- Boden SD, Schimandle JH, Hutton WC, Chen MI. Volvo award in basic sciences. The use of an osteoinductive growth factor for lumbar spinal fusion. Part I: biology of spinal fusion. Spine. 1995;20:2626–32.
- Boden SD, Schimandle JH, Hutton WC. An experimental lumbar intertransverse process spinal fusion model. Radiographic, histologic, and biomechanical healing characteristics. Spine. 1995;20:412–20.
- Bible JE, et al. Postoperative bracing after spine surgery for degenerative conditions: a questionnaire study. Spine J Off J N Am Spine Soc. 2009;9:309–16. doi:10.1016/j.spinee.2008.06.453.
- Boden SD, Sumner DR. Biologic factors affecting spinal fusion and bone regeneration. Spine. 1995;20:102S–12.
- Buckwalter JA, Grodzinsky AJ. Loading of healing bone, fibrous tissue, and muscle: implications for orthopaedic practice. J Am Acad Orthop Surg. 1999;7:291–9.
- Brown CW, Orme TJ, Richardson HD. The rate of pseudarthrosis (surgical nonunion) in patients who are smokers and patients who are nonsmokers: a comparison study. Spine. 1986;11:942–3.
- de Vernejoul MC, et al. Evidence for defective osteoblastic function. A role for alcohol and tobacco consumption in osteoporosis in middle-aged men. Clin Orthop Relat Res. 1983(179):107–15.
- Hollo I, Gergely I, Boross M. Smoking results in calcitonin resistance. JAMA J Am Med Assoc. 1977;237:2470.
- Kwiatkowski TC, Hanley Jr EN, Ramp WK. Cigarette smoking and its orthopedic consequences. Am J Orthop (Belle Mead NJ). 1996;25:590–7.
- 12. Andersen T, et al. Smoking as a predictor of negative outcome in lumbar spinal fusion. Spine. 2001;26:2623–8.
- Deguchi M, Rapoff AJ, Zdeblick TA. Posterolateral fusion for isthmic spondylolisthesis in adults: analysis of fusion rate and clinical results. J Spinal Disord. 1998;11:459–64.
- Nilsson OS, Bauer HC, Brosjo O, Tornkvist H. Influence of indomethacin on induced heterotopic bone formation in rats. Importance of length

of treatment and of age. Clin Orthop Relat Res. 1986(207):239-45.

- Einhorn TA, Bonnarens F, Burstein AH. The contributions of dietary protein and mineral to the healing of experimental fractures. A biomechanical study. J Bone Joint Surg Am. 1986;68:1389–95.
- Vaccaro AR, et al. Comparison of OP-1 Putty (rhBMP-7) to iliac crest autograft for posterolateral lumbar arthrodesis: a minimum 2-year follow-up pilot study. Spine. 2005;30:2709–16.
- Vaccaro AR, et al. A pilot study evaluating the safety and efficacy of OP-1 Putty (rhBMP-7) as a replacement for iliac crest autograft in posterolateral lumbar arthrodesis for degenerative spondylolisthesis. Spine. 2004;29:1885–92.
- Sandhu HS, Grewal HS, Parvataneni H. Bone grafting for spinal fusion. Orthop Clin North Am. 1999;30:685–98.
- Prolo DJ, Rodrigo JJ. Contemporary bone graft physiology and surgery. Clin Orthop Relat Res. 1985(200):322–42.
- Cornell CN. Osteoconductive materials and their role as substitutes for autogenous bone grafts. Orthop Clin North Am. 1999;30:591–8.
- Ehrler DM, Vaccaro AR. The use of allograft bone in lumbar spine surgery. Clin Orthop Relat Res. 2000(371);38–45.
- Asselmeier MA, Caspari RB, Bottenfield S. A review of allograft processing and sterilization techniques and their role in transmission of the human immunodeficiency virus. Am J Sports Med. 1993;21:170–5.
- Malloy KM, Hilibrand AS. Autograft versus allograft in degenerative cervical disease. Clin Orthop Relat Res. 2002(394):27–38.
- Vaccaro AR, et al. Bone grafting alternatives in spinal surgery. Spine J Off J N Am Spine Soc. 2002;2:206–15.
- 25. Jorgenson SS, Lowe TG, France J, Sabin J. A prospective analysis of autograft versus allograft in posterolateral lumbar fusion in the same patient. A minimum of 1-year follow-up in 144 patients. Spine. 1994;19:2048–53.
- Nugent PJ, Dawson EG. Intertransverse process lumbar arthrodesis with allogeneic fresh-frozen bone graft. Clin Orthop Relat Res. 1993(287):107–11.
- Blanco JS, Sears CJ. Allograft bone use during instrumentation and fusion in the treatment of adolescent idiopathic scoliosis. Spine. 1997;22:1338–42.
- Dodd CA, Fergusson CM, Freedman L, Houghton GR, Thomas D. Allograft versus autograft bone in scoliosis surgery. J Bone Joint Surg Br. 1988;70:431–4.
- Jones KC, Andrish J, Kuivila T, Gurd A. Radiographic outcomes using freeze-dried cancellous allograft bone for posterior spinal fusion in pediatric idiopathic scoliosis. J Pediatr Orthop. 2002;22:285–9.
- 30. Bae HW, et al. Intervariability and intravariability of bone morphogenetic proteins in commercially available demineralized bone matrix products. Spine. 2006;31:1299–306. doi:10.1097/01. brs.0000218581.92992.b7; discussion 1307–8.

- Bae H, et al. Variability across ten production lots of a single demineralized bone matrix product. J Bone Joint Surg Am. 2010;92:427–35. doi:10.2106/ JBJS.H.01400.
- Lee YP, et al. The efficacy of different commercially available demineralized bone matrix substances in an athymic rat model. J Spinal Disord Tech. 2005;18:439–44.
- 33. Peterson B, Whang PG, Iglesias R, Wang JC, Lieberman JR. Osteoinductivity of commercially available demineralized bone matrix. Preparations in a spine fusion model. J Bone Joint Surg Am. 2004;86-A:2243–50.
- Frenkel SR, Moskovich R, Spivak J, Zhang ZH, Prewett AB. Demineralized bone matrix. Enhancement of spinal fusion. Spine. 1993;18:1634–9.
- 35. Martin Jr GJ, Boden SD, Titus L, Scarborough NL. New formulations of demineralized bone matrix as a more effective graft alternative in experimental posterolateral lumbar spine arthrodesis. Spine. 1999;24:637–45.
- 36. Kiely PD, et al. Evaluation of a new formulation of demineralized bone matrix putty in a rabbit posterolateral spinal fusion model. Spine J Off J N Am Spine Soc. 2014. doi:10.1016/j.spinee.2014.01.053.
- Morone MA, Boden SD. Experimental posterolateral lumbar spinal fusion with a demineralized bone matrix gel. Spine. 1998;23:159–67.
- 38. Cammisa Jr FP, et al. Two-year fusion rate equivalency between Grafton DBM gel and autograft in posterolateral spine fusion: a prospective controlled trial employing a side-by-side comparison in the same patient. Spine. 2004;29:660–6.
- 39. Kang J, et al. Grafton and local bone have comparable outcomes to iliac crest bone in instrumented single-level lumbar fusions. Spine. 2012;37:1083–91. doi:10.1097/BRS.0b013e31823ed817.
- 40. Epstein NE, Epstein JA. SF-36 outcomes and fusion rates after multilevel laminectomies and 1 and 2-level instrumented posterolateral fusions using lamina autograft and demineralized bone matrix. J Spinal Disord Tech. 2007;20:139–45. doi:10.1097/01. bsd.0000211261.36120.3e.
- Epstein NE. Fusion rates and SF-36 outcomes after multilevel laminectomy and noninstrumented lumbar fusions in a predominantly geriatric population. J Spinal Disord Tech. 2008;21:159–64. doi:10.1097/ BSD.0b013e318074ddaa.
- 42. Thalgott JS, Giuffre JM, Klezl Z, Timlin M. Anterior lumbar interbody fusion with titanium mesh cages, coralline hydroxyapatite, and demineralized bone matrix as part of a circumferential fusion. Spine J Off J N Am Spine Soc. 2002;2:63–9.
- 43. Bucholz RW, Carlton A, Holmes RE. Hydroxyapatite and tricalcium phosphate bone graft substitutes. Orthop Clin North Am. 1987;18:323–34.
- 44. Miller CP, et al. The efficacies of 2 ceramic bone graft extenders for promoting spinal fusion in a rabbit bone paucity model. Spine. 2012;37:642–7. doi:10.1097/ BRS.0b013e31822e604e.

- 45. Damien CJ, et al. Effect of demineralized bone matrix on bone growth within a porous HA material: a histologic and histometric study. J Biomater Appl. 1995;9:275–88.
- 46. Kania RE, Meunier A, Hamadouche M, Sedel L, Petite H. Addition of fibrin sealant to ceramic promotes bone repair: long-term study in rabbit femoral defect model. J Biomed Mater Res. 1998;43:38–45.
- 47. Li H, Xue K, Kong N, Liu K, Chang J. Silicate bioceramics enhanced vascularization and osteogenesis through stimulating interactions between endothelia cells and bone marrow stromal cells. Biomaterials. 2014;35:3803–18.
- 48. Kaya RA, et al. The effects of magnesium particles in posterolateral spinal fusion: an experimental in vivo study in a sheep model. J Neurosurg Spine. 2007;6:141–9.
- Jenis LG, Banco RJ. Efficacy of silicate-substituted calcium phosphate ceramic in posterolateral instrumented lumbar fusion. Spine. 2010;35:E1058–63. doi:10.1097/BRS.0b013e3181df196f.
- Lee JH, et al. Negative effect of rapidly resorbing properties of bioactive glass-ceramics as bone graft substitute in a rabbit lumbar fusion model. Clin Orthop Surg. 2014;6:87–95. doi:D – nlm: pmc3942607 oto – notnlm.
- Nogami H, Urist MR. A morphogenetic matrix for differentiation of cartilage in tissue culture. Proc Soc Exp Biol Med. 1970;134:530–5.
- Urist MR. A morphogenetic matrix for differentiation of bone tissue. Calcif Tissue Res. 1970;Suppl:98–101.
- Urist MR, Strates BS. Bone morphogenetic protein. J Dent Res. 1971;50:1392–406.
- Urist MR, Silverman BF, Buring K, Dubuc FL, Rosenberg JM. The bone induction principle. Clin Orthop Relat Res. 1967;53:243–83.
- Brand RA. Marshall R. Urist, 1914–2001. Clin Orthop Relat Res. 2009;467:3049–50. doi:10.1007/ s11999-009-1067-4.
- Boden SD, Martin Jr GJ, Horton WC, Truss TL, Sandhu HS. Laparoscopic anterior spinal arthrodesis with rhBMP-2 in a titanium interbody threaded cage. J Spinal Disord. 1998;11:95–101.
- Sandhu HS, et al. Evaluation of rhBMP-2 with an OPLA carrier in a canine posterolateral (transverse process)spinal fusion model. Spine. 1995;20:2669–82.
- Sandhu HS, et al. Histologic evaluation of the efficacy of rhBMP-2 compared with autograft bone in sheep spinal anterior interbody fusion. Spine. 2002;27:567–75.
- Burkus JK, Gornet MF, Dickman CA, Zdeblick TA. Anterior lumbar interbody fusion using rhBMP-2 with tapered interbody cages. J Spinal Disord Tech. 2002;15:337–49.
- 60. Friedlaender GE, et al. Osteogenic protein-1 (bone morphogenetic protein-7) in the treatment of tibial nonunions. J Bone Joint Surg Am. 2001;83-A Suppl 1:S151–8.
- 61. Govender S, et al. Recombinant human bone morphogenetic protein-2 for treatment of open tibial

fractures: a prospective, controlled, randomized study of four hundred and fifty patients. J Bone Joint Surg Am. 2002;84-A:2123–34.

- 62. Boden SD, Kang J, Sandhu H, Heller JG. Use of recombinant human bone morphogenetic protein-2 to achieve posterolateral lumbar spine fusion in humans: a prospective, randomized clinical pilot trial: 2002 Volvo Award in clinical studies. Spine. 2002;27:2662– 73. doi:10.1097/01.BRS.0000035320.82533.06.
- 63. Dimar 2nd JR, et al. Clinical and radiographic analysis of an optimized rhBMP-2 formulation as an autograft replacement in posterolateral lumbar spine arthrodesis. J Bone Joint Surg Am. 2009;91:1377–86. doi:10.2106/JBJS.H.00200.
- 64. Devine JG, Dettori JR, France JC, Brodt E, McGuire RA. The use of rhBMP in spine surgery: is there a cancer risk? Evid Based Spine Care J. 2012;3(2):35– 41. doi:10.1055/s-0031-1298616.
- Adams CL, Ogden K, Robertson IK, Broadhurst S, Edis D. Effectiveness and safety of recombinant human bone morphogenetic protein-2 versus local bone graft in primary lumbar interbody fusions. Spine. 2014;39:164–71. doi:10.1097/ BRS.0000000000000089.
- 66. Carragee EJ, Hurwitz EL, Weiner BK. A critical review of recombinant human bone morphogenetic protein-2 trials in spinal surgery: emerging safety concerns and lessons learned. Spine J Off J N Am Spine Soc. 2011;11:471–91. doi:10.1016/j. spinee.2011.04.023.
- 67. Anderson PA. Letter to the editor regarding "A critical review of recombinant human morphogenetic protein-2 trials in spine surgery: emerging safety concerns and lessons learned". Spine J. 2012;12:356. http://dx.doi. org/10.1016/j.spinee.2012.03.005.
- Carragee EJ. Carragee responds. Spine J. 2012;12:356–7. http://dx.doi.org/10.1016/j. spinee.2012.03.004.
- Zdeblick TA. Science please. Spine J Off J N Am Spine Soc. 2011;11:686. doi:10.1016/j.spinee.2011.06.005; author reply 687–90.
- Carragee EJ, Hurwitz EL, Weiner BK, Scuderi GJ, Bono CM. Authors and editors combined response to Zdeblick letter (revised 28 June 2011). Spine J Off J N Am Spine Soc. 2011;11:687–90.
- 71. Dimar Ii JR, et al. Reply to "A critical review of recombinant human bone morphogenetic protein-2 trials in spinal surgery: emerging safety concerns and lessons learned". Spine J. 2011;11:1082–3. http:// dx.doi.org/10.1016/j.spinee.2011.09.029.
- Carragee EJ, Hurwitz EL, Weiner BK. Carragee et al respond. Spine J. 2011;11:1083–6. http://dx.doi. org/10.1016/j.spinee.2011.10.003.
- 73. Fu R, et al. Effectiveness and harms of recombinant human bone morphogenetic protein-2 in spine fusion: a systematic review and metaanalysis. Ann Intern Med. 2013;158:890–902. doi:10.7326/0003-4819-158-12-201306180-00006.
- 74. Simmonds MC, et al. Safety and effectiveness of recombinant human bone morphogenetic protein-2

for spinal fusion: a meta-analysis of individualparticipant data. Ann Intern Med. 2013;158:877–89. doi:10.7326/0003-4819-158-12-201306180-00005.

- 75. Cunningham BW, et al. Osseointegration of autograft versus osteogenic protein-1 in posterolateral spinal arthrodesis: emphasis on the comparative mechanisms of bone induction. Spine J Off J N Am Spine Soc. 2002;2:11–24.
- 76. Grauer JN, et al. 2000 young investigator research award winner. Evaluation of OP-1 as a graft substitute for intertransverse process lumbar fusion. Spine. 2001;26:127–33.
- Patel TC, et al. Osteogenic protein-1 overcomes the inhibitory effect of nicotine on posterolateral lumbar fusion. Spine. 2001;26:1656–61.
- 78. Blattert TR, et al. Successful transpedicular lumbar interbody fusion by means of a composite of osteogenic protein-1 (rhBMP-7) and hydroxyapatite carrier: a comparison with autograft and hydroxyapatite in the sheep spine. Spine. 2002;27:2697–705. doi:10.1097/01.BRS.0000035269.55324.C3.
- 79. Johnsson R, Stromqvist B, Aspenberg P. Randomized radiostereometric study comparing osteogenic protein-1 (BMP-7) and autograft bone in human noninstrumented posterolateral lumbar fusion: 2002 Volvo Award in clinical studies. Spine. 2002;27:2654–61. doi:10.1097/01.BRS.0000035339.83704.60.
- White AP, et al. Clinical applications of BMP-7/OP-1 in fractures, nonunions and spinal fusion. Int Orthop. 2007;31:735–41. doi:10.1007/s00264-007-0422-x.
- 81. Vaccaro AR, et al. The safety and efficacy of OP-1 (rhBMP-7) as a replacement for iliac crest autograft in posterolateral lumbar arthrodesis: a long-term (>4 years) pivotal study. Spine. 2008;33:2850–62. doi:10.1097/BRS.0b013e31818a314d.
- Viggeswarapu M, et al. Adenoviral delivery of LIM mineralization protein-1 induces new-bone formation in vitro and in vivo. J Bone Joint Surg Am. 2001;83-A:364–76.
- Boden SD, et al. Lumbar spine fusion by local gene therapy with a cDNA encoding a novel osteoinductive protein (LMP-1). Spine. 1998;23:2486–92.
- 84. Sangadala S, Boden SD, Viggeswarapu M, Liu Y, Titus L. LIM mineralization protein-1 potentiates bone morphogenetic protein responsiveness via a novel interaction with Smurf1 resulting in decreased ubiquitination of Smads. J Biol Chem. 2006;281:17212–9. doi:10.1074/jbc.M511013200.
- Cha CW, Boden SD. Gene therapy applications for spine fusion. Spine (Phila Pa 1976). 2003;28:S74–84.
- 86. Jahng TA, Fu TS, Cunningham BW, Dmitriev AE, Kim DH. Endoscopic instrumented posterolateral lumbar fusion with Healos and recombinant human growth/differentiation factor-5. Neurosurgery. 2004;54:171–80; discussion 180–1.
- Magit DP, et al. Healos/recombinant human growth and differentiation factor-5 induces posterolateral lumbar fusion in a New Zealand white rabbit model. Spine. 2006;31:2180–8. doi:10.1097/01. brs.0000232823.82106.0a.

- Pola E, et al. Efficient bone formation by gene transfer of human LIM mineralization protein-3. Gene Ther. 2004;11:683–93.
- Zhang X, et al. Craniosynostosis in transgenic mice overexpressing Nell-1. J Clin Invest. 2002;110:861– 70. doi:D – NLM: PMC151127 EDAT- 2002/09/18 10:00 MHDA- 2002/10/31 04:00 CRDT- 2002/09/18 10:00 AID – 10.1172/JCI15375 [doi] PST – publish.
- Lu SS, et al. The osteoinductive properties of Nell-1 in a rat spinal fusion model. Spine J. 2007;7:50–60.
- 91. Curylo LJ, Johnstone B, Petersilge CA, Janicki JA, Yoo JU. Augmentation of spinal arthrodesis with

autologous bone marrow in a rabbit posterolateral spine fusion model. Spine. 1999;24:434–8; discussion 438–9.

- 92. Zuk PA. Viral transduction of adipose-derived stem cells. Methods Mol Biol. 2011;702:345–57.
- Hsu WK, et al. Stem cells from human fat as cellular delivery vehicles in an athymic rat posterolateral spine fusion model. J Bone Joint Surg Am Vol. 2008;90:1043–52.
- 94. Johnson RG. Bone marrow concentrate with allograft equivalent to autograft in lumbar fusions. Spine. 2014. doi:10.1097/BRS.00000000000254.

# Techniques for the Osteoporotic Spine That Needs Fusion

26

Paul F. Heini

#### 26.1 Introduction

The increasing number of elderly people is a worldwide phenomenon. Degenerative changes are related to age and so is osteoporosis that is present to a certain extent nearly in every elderly patient [1, 2]. In the majority of cases, standard approaches for treating degenerative spine problems are also applicable in the elderly, as long as the degree of osteoporosis is moderate, which means a T-score of the lumbar spine above -1.5. Problems may arise when the surgeon faces deformities and/or segmental instabilities in combination with significant osteoporosis (T-score lower than -1.5). The bone mineral density correlates with the holding power of pedicle screws, as demonstrated in several in vitro and in vivo studies [3]. The individual situation must be assessed for every single screw, as high loads on the bone-implant interface are present until fusion is achieved.

Until recently, such cases appeared exceptional; however, in the near future, such problems will become routine component of spine surgery as an increasing number of elderly people will be seeking a surgical solution for their spine

Spine Service, Orthopaedic Department, Sonnenhof Hospital, Buchserstrasse 30, Bern CH-3006, Switzerland e-mail: paulheini@sonnenhof.ch troubles. If a surgical stabilization is performed in the osteoporotic spine, the delicate biomechanical equilibrium might be disturbed, leading to implant loosening and increased risk for fractures of adjacent vertebrae or accelerated adjacent segmental degeneration. This should be taken into consideration when dealing with spine problems in osteoporotic patients. There exists a lack of reliable information regarding complications [4]. However, keeping in mind that the early fracture risk in deformity correction in the elderly is reported by 13 % and by 26 % for junctional kyphosis, we can expect significantly higher numbers in the severely osteoporotic patient [5].

The treatment of vertebral fractures in the osteoporotic spine with cement reinforcement is well established [6], especially in acute fractures with a progressive vertebral body collapse [7, 8]. On the other hand, the widespread use of these techniques is questioned by two randomized controlled studies comparing vertebroplasty with a sham procedure [9, 10]. This note is necessary as the solution for critical osteoporotic situations suggests the use of bone cement in order to enhance stability. A recent survey in Germany reveals that nearly 80 % of German spine surgeons use cemented screw fixation in their daily work [11].

Some surgeons advise the use of long fixation procedures with multiple anchoring points, while others use combined screw and hook constructs for increased holding power [12]. The latter type

P.F. Heini, MD, PhD

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_26

of fixation provides optimal stability at the periphery of a construct. The hook prevents the screw pullout efficiently; in vitro measures demonstrate a significant increase of the stiffness of the screw hook construct by nearly 50 % – the same effect as cement reinforcement [13].

#### 26.2 Unique Considerations

In degenerative problems in the elderly with osteoporosis, the following main topics are discussed and presented:

- Spinal stenosis in relation with an osteoporotic fracture or segmental instability
- Treatment option for degenerative scoliosis
- Adjacent segmental degeneration and junctional kyphosis
- Discoplasty an alternative for anterior column support

# 26.3 Spinal Stenosis in the Setting of Osteoporotic Fracture or Segmental Instability

Spinal stenosis is a disorder which incidence is increasing dramatically as the population ages [14, 15]. Oftentimes, the problem remains unnoticed by the osteoporotic patients, as their demands are low. However, a mild compression fracture may turn a silent stenosis into an acute problem with sciatica or even cauda equina-like symptoms. For this situation, surgical treatment is required and consists of decompression and stabilization. The stabilization might be achieved either by cement reinforcement of the fractured vertebra or with instrumentation [16–19]. The decision is based on the general health situation of the patient, as well as the specific local spine situation (deformity, slip, severity of osteoporosis). In these patients a decompression only is usually not appropriate due to its destabilizing effect [18, 20].

The possibility of a decompression in combination with vertebroplasty, kyphoplasty, or stentoplasty might be taken into consideration as long as there is no segmental instability in the presence of competent facet joints in combination with a minor vertebral fracture. Some authors combine such a procedure with an interspinous stabilization procedure [8, 21–24]. Technically, a microsurgical decompression is performed according to the needs in order to clear the spinal canal – then a percutaneous cement augmentation if needed in combination with a kyphoplasty or stentoplasty is performed [8]. A detailed description of the surgical technique for vertebroplasty is available elsewhere [25].

In spinal stenosis with segmental instability, i.e., degenerative slip with or without a fracture, stabilization and fusion are considered.

The application of PMMA to increase the stability of spinal implants has been used for a long time [26–28]. Vertebroplasty offers the option of using the same fixation principles as those for normal bone [29]. To increase the stability of the screws, a standard open or closed vertebroplasty procedure is performed in which the pedicle screws are implanted before the cement has set. This technique allows the use of any standard stabilization system. Alternatively, perforated or fenestrated screws can be placed in a standard fashion and the cement subsequently injected through the screws. The amount of cement per screw varies between 1 and 4 cc. Leakage remains a critical issue and therefore highly viscous cement is a must the newer cement generations (Vertecem® or Confidence®) have very long working times and provide enough time to perform a safe augmentation and controlled screw placement. The question of whether or not an instrumented vertebra requires augmentation remains obscure. A tool for the intraoperative assessment of bone quality did provide useful information in a pilot study - however, it is not yet widely available [3]. If increasing the implant size (length and/or diameter) appears insufficient to achieve good fixation, the use of cement should be considered [30]. So far, the use of cement seems to not have negative effects, except the increased surgical efforts and the risk of cement injection. Furthermore, an increased fracture risk is related to the length of fixation [4]. If the T-score of the spine is very low, the use of cement-reinforced fixation should be considered. However, the bone density assessed by a standard



**Fig. 26.1** A 84-year-old obese female in good health. Spontaneous fracture of the L5 vertebra occurred with simply sitting. The patient noted acute onset of bilateral leg pain when sitting or standing. (a) Standing plain radiographs revealed a mild compression fracture of the L5 (*asterisk*) that was confirmed on MR with evidence of bony edema. Spinal stenosis was present. (b) Treatment

involved decompression and cemented fixation of the pedicle screws in the L4 vertebral body. 3 cc of cement was administered per screw (A). The interbody defect was filled with a discoplasty procedure (B). The screws in L5 and S1 demonstrated sufficient purchase and therefore were not cemented. The maximal screw dimensions were chosen in terms of length and diameter (C)

DEXA measure per se is not a reliable indicator for cement augmentation, and therefore, clinical experience often supports the benefit of cement usage to decrease implant loosening and improve screw insertional torque [3, 31, 32]. In multilevel fixation procedures, the risk of fracture of the terminal instrumented vertebra is significant. Therefore, even in the "osteopenic" spine, cement reinforcement of the most rostral screws should be considered. Furthermore, there is increasing consensus that protective cement reinforcement/ vertebroplasty of the adjacent vertebrae should be considered [4, 33, 34] (Fig. 26.1).

# 26.4 Degenerative Scoliosis and Loss of Spinal Balance

Aging degenerative deformities are complex situations [35]. With them, postural problems are very common. A central or foraminal stenosis often coexists [36]. Traditional open surgery with long posterior stabilization is related with a major complication risk [37, 38].

Lateral interbody fusion through a transpsoas approach has been shown to be extremely effective in deformity correction by a less invasive technique [39, 40]. In the osteoporotic spine, however, the issues are related to cage subsidence and secondary loss of correction [40, 41]. Prophylactic cement reinforcement can help to overcome this limitation and provide sufficient support for the cages [40]. There is a need for aggressive cement filling. The vertebroplasty procedure can be performed in the same session prior to the lateral correction or as a staged procedure. The latter strategy is the one preferred by the author, especially when multiple vertebrae are to be treated. The intervention is performed under local anesthesia/sedation. In multiple-level cases, two sessions are required, as the amount of cement should not exceed 30 cc in one session due to the risk of fat embolism. The main procedure consists of the scoliosis correction using a mini-open lateral approach. The technique aims to be a stand-alone procedure as long as the rotational deformity remains moderate (Fig. 26.2). Otherwise additional posterior percutaneous



**Fig. 26.2** Degenerative scoliosis and spinal stenosis: An 80-year-old patient presented with equal leg and back pain. (a) The standing plain radiographs demonstrated loss of lumbar lordosis and a degenerative scoliosis related to shortening of her left leg. (b) MRI revealed stenosis at L3–L4 and L4–L5 and a severe foraminal narrowing at L3–L4 and L2–L3 (*asterisk*). (c) CT demonstrated a vacuum phenomenon at L2–L3, L3–L4, and L4–L5 (*asterisk*). (d) Treatment consisted of a vertebroplasty procedure from L2 to L5 and then a segmental anterior correction via the lateral approach from the right side. Iliac crest bone graft was harvested and enhanced with demineralized bone matrix.

X-rays at 2-year follow-up demonstrated a well-restored and well-maintained lumbar lordosis when compared to the preoperative standing films. Fusion was not yet complete at all levels. The patient has had a very satisfactory postoperative course without leg symptoms and only minor back pain, not requesting any medication. The reason for a vertebroplasty in this patient was her general health condition: the presence of obesity, diabetes, and renal disease. A DEXA demonstrated a T-score of -2.0 in the lumbar spine. CT did not reveal any relevant reactive bony sclerosis at the concavity of the deformity (c). These factors were taken into consideration in deciding to use cement augmentation



Fig. 26.2 (continued)



Fig. 26.3 (a) Distribution of levels treated for degenerative deformity correction (60 patients, 179 levels). (b) Clinical outcome 1 year postoperative. COMI/ individual parameters (mean, min, max). *COMI* core outcome measure index

screw fixation is performed in combination with cement-reinforced screws.

We studied 60 cases operated through a lateral approach for degenerative deformities (scoliosis/ loss of lordosis) in patients with osteoporosis treated with vertebroplasty and segmental correction. Seven patients underwent an additional posterior percutaneous stabilization within the first 3 months after the lateral intervention due to secondary loss of correction. The clinical outcomes were encouraging. However, 60 % of patients did note approach-related discomfort and side effects in the early phase postoperatively which include pain and numbness in the groin and anterior thigh and weakness of the psoas on the side of approach. The use of iliac crest bone graft is of limited value, especially in the osteoporotic patient. Osteoconductive bone substitutes and allograft material also demonstrate unreliable fusion rates with frequent secondary loss of correction. We have found that BMPII works best in these cases for stand-alone interventions, although it must be clear that evidence for its efficacy in this application has not been shown and a myriad of complications are reported in connection with its use in this setting [42]. The distribution of levels and the 1-year clinical outcome are presented in Fig. 26.3.

# 26.5 Junctional Kyphosis in the Osteoporotic Spine

Posterior deformity correction is associated with a significant potential for complications such as adjacent-level degeneration or vertebral fractures. There is no consensus on how the degree and extent of the instrumentation may affect the risk of junctional failure. Many factors are related to junctional failure. Sagittal balance is not the only critical parameter; sarcopenia plays an important role. Furthermore neurodegenerative disorders, such as Parkinson's disease, have a significant impact [43]. Failure usually occurs during the early postoperative period. It typically involves the adjacent segments, as demonstrated by instability when comparing the standing postoperative X-ray with a CT or an MRI in the supine position. Extension of the construct is often needed, and in the presence of a junctional fracture, the use of cemented screw fixation and protective percutaneous cement augmentation may be required [4, 5]. The use of a "discoplasty procedure" can provide anterior column support and thereby avoid an additional anterior procedure (Fig. 26.4).

# 26.6 Discoplasty: Anterior Column Support with PMMA

In cases of segmental instability with a collapsed disk space, the disk height restoration can occur just by positioning the patient on the operating table. The same is true for certain fractures with a kyphotic deformity. Interbody spacers (TLIF, PLIF cages) that work effectively in a healthy

Fig. 26.4 (a) 78-year-old female patient presented 5 years after a successful decompression and spinal fusion from L2 to L5 with new leg symptoms due to a severe adjacent-level degeneration. The disk space was collapsed at L1-L2 and T12-L1 (asterisk). Furthermore there was a vertebral fracture of T11. (b) A decompression and extension of stabilization was performed to T10. The disk spaces at T12-L1 and L1-L2 were treated with cement injection. Early postoperatively (6 months), an adjacent-level fracture occurred that resulted in paraparesis due to cord compression at the T9–T10 level. A revision surgery was performed with decompression and extension of the stabilization to T7 with a protective vertebroplasty of the adjacent levels up to T4. The patient had a stable postoperative course as demonstrated 2 years after her last intervention (c)





Fig. 26.4 (continued)



**Fig. 26.5** Discoplasty for anterior column support: The segmental collapse of L1–L2 in this patient was related to foraminal root compression. After surgical correction there was an obvious void in the disk space (*asterisk*). A filling cannula was inserted through the canal beside the

thecal sac into the disk space and the void was filled with PMMA. An instrumented fusion was performed in addition. The discoplasty procedure provided a custom-made spacer with optimal anterior column support

bone show a very limited supporting surface and are therefore associated with a significant risk of subsidence in the osteoporotic spine [44]. In addition, the surgical procedure must be expanded considerably for disk removal and cage placement. Therefore, the use of a transdiscal cement injection either in an open fashion through the spinal canal or percutaneously via the foramen through Kambin's triangle can provide optimal anterior column support. Discoplasty provides a custom-made spacer with a maximal supporting surface. The technique is effective in cases in which the disk space is empty and collapsed but the segment remains mobile. The annulus must be intact and there exists no need to evacuate the disk space. A filling cannula for vertebroplasty is placed into the disk space and high viscous cement is injected into the disk – the annulus provides a containment that prevents the cement from leaking. The procedure is performed under fluoroscopic control. Additional stabilization and fusion of the motion segment(s) are mandatory (Fig. 26.5).

#### Conclusion

The increasing number of elderly people represents a huge socioeconomic burden in general and is specifically challenging for the
medical field [45]. For the spine surgeon, the osteoporotic spine represents the most challenging problem [1, 2]. There are certainly well-established treatment modalities for many spine disorders (stenosis, deformity, instability) - but in combination with osteoporosis, common techniques may fail (i.e., a decompression may provoke a secondary vertebral fracture, a vertebral fracture can turn a silent spinal stenosis into an acute sciatica or paresis that need emergent surgery, a segmental stability may promote an adjacent vertebral fracture, and so on) [4, 5]. So far the most efficient means to overcome the mechanical limitations related with the osteoporotic spine is the use of bone cement [29, 46]. PMMA can enhance the anchoring of pedicle screws, and it can protect adjacent vertebrae from a fracture [34]. The use of PMMA is related with specific risks (cement leakage/embolism, fat embolism) that have to be weighed against the possible benefits [25]. Besides the mechanical problems related with the osteoporotic spine, one must consider the general health state of a patient and carefully balance the risks and benefits of a specific treatment individually for each patient [37].

# References

- Incidence of vertebral fracture in Europe: results from the European Prospective Osteoporosis Study (EPOS). J Bone Miner Res. 2002;17(4):716–24.
- Lu C, Chen D, Cai Y, Wei S. Concordane of OSTA and lumbar spine BMD by DXA in identifying risk of osteoporosis. J Orthop Surg Res. 2006;1:14.
- Popp AW, Schwyn R, Schiuma D, Keel MJ, Lippuner K, Benneker LM. DensiProbe spine: an intraoperative measurement of bone quality in spinal instrumentation. A clinical feasibility study. Spine J: Off J N Am Spine Soc. 2013;13(10):1223–9.
- Hart RA, Prendergast MA, Roberts WG, Nesbit GM, Barnwell SL. Proximal junctional acute collapse cranial to multi-level lumbar fusion: a cost analysis of prophylactic vertebral augmentation. Spine J: Off J N Am Spine Soc. 2008;8(6):875–81.
- DeWald CJ, Stanley T. Instrumentation-related complications of multilevel fusions for adult spinal deformity patients over age 65: surgical considerations and treatment options in patients with poor bone quality. Spine (Phila Pa 1976). 2006;31(19 Suppl):S144–51.

- Klazen CA, Lohle PN, de Vries J, Jansen FH, Tielbeek AV, Blonk MC, et al. Vertebroplasty versus conservative treatment in acute osteoporotic vertebral compression fractures (Vertos II): an open-label randomised trial. Lancet. 2010;376:1085–92.
- Lee HM, Park SY, Lee SH, Suh SW, Hong JY. Comparative analysis of clinical outcomes in patients with osteoporotic vertebral compression fractures (OVCFs): conservative treatment versus balloon kyphoplasty. Spine J: Off J N Am Spine Soc. 2012;12(11):998–1005.
- Heini PF, Teuscher R. Vertebral body stenting/stentoplasty. Swiss Med Wkly. 2012;142:w13658.
- Buchbinder R, Osborne RH, Ebeling PR, Wark JD, Mitchell P, Wriedt C, et al. A randomized trial of vertebroplasty for painful osteoporotic vertebral fractures. N Engl J Med. 2009;361(6):557–68.
- Kallmes DF, Comstock BA, Heagerty PJ, Turner JA, Wilson DJ, Diamond TH, et al. A randomized trial of vertebroplasty for osteoporotic spinal fractures. N Engl J Med. 2009;361(6):569–79.
- Goost H, Kabir K, Wirtz DC, Deborre C, Karius T, Pflugmacher R, et al. PMMA augmentation of pedicle screws: results of a survey in Germany. Z Orthop Unfall. 2012;150(3):318–23.
- Hasegawa K, Takahashi HE, Uchiyama S, Hirano T, Hara T, Washio T, et al. An experimental study of a combination method using a pedicle screw and laminar hook for the osteoporotic spine. Spine (Phila Pa 1976). 1997;22(9):958–62; discussion 963.
- Tan JS, Kwon BK, Dvorak MF, Fisher CG, Oxland TR. Pedicle screw motion in the osteoporotic spine after augmentation with laminar hooks, sublaminar wires, or calcium phosphate cement: a comparative analysis. Spine (Phila Pa 1976). 2004;29(16):1723–30.
- Snyder DL, Doggett D, Turkelson C. Treatment of degenerative lumbar spinal stenosis. Am Fam Physician. 2004;70(3):517–20.
- Kauppila LI, Eustace S, Kiel DP, Felson DT, Wright AM. Degenerative displacement of lumbar vertebrae. A 25-year follow-up study in Framingham. Spine (Phila Pa 1976). 1998;23(17):1868–73; discussion 1873–4.
- Nguyen HV, Ludwig S, Gelb D. Osteoporotic vertebral burst fractures with neurologic compromise. J Spinal Disord Tech. 2003;16(1):10–9.
- Korovessis P, Maraziotis T, Piperos G, Spyropoulos P. Spontaneous burst fracture of the thoracolumbar spine in osteoporosis associated with neurological impairment: a report of seven cases and review of the literature. Eur Spine J. 1994;3(5):286–8.
- Kim KT, Suk KS, Kim JM, Lee SH. Delayed vertebral collapse with neurological deficits secondary to osteoporosis. Int Orthop. 2003;27(2):65–9.
- Heggeness MH. Spine fracture with neurological deficit in osteoporosis. Osteoporos Int. 1993;3(4):215–21.
- Natelson SE. The injudicious laminectomy. Spine (Phila Pa 1976). 1986;11(9):966–9.
- 21. Bonaldi G, Cianfoni A. Percutaneous treatment of lumbar compression fracture with canal stenosis

and neurogenic intermittent claudication: combining kyphoplasty and interspinous spacer. J Vasc Interv Radiol. 2012;23(11):1437–41.

- Hiwatashi A, Westesson PL. Vertebroplasty for osteoporotic fractures with spinal canal compromise. AJNR Am J Neuroradiol. 2007;28(4):690–2.
- Miller JD, Nader R. Treatment of combined osteoporotic compression fractures and spinal stenosis: use of vertebral augmentation and interspinous process spacer. Spine (Phila Pa 1976). 2008;33(19):E717–20.
- 24. Singh K, Heller JG, Samartzis D, Price JS, An HS, Yoon ST, et al. Open vertebral cement augmentation combined with lumbar decompression for the operative management of thoracolumbar stenosis secondary to osteoporotic burst fractures. J Spinal Disord Tech. 2005;18(5):413–9.
- Heini P, Kleinschidt M. Vertebroplasty and percutaneous cement reinforcement techniques. In: Roger H, Korge A, editors. Minimally invasive spine surgery – techniques, evidence, and controversies. Stuttgard: Thieme; 2013. p. 51–66.
- Choma TJ, Pfeiffer FM, Swope RW, Hirner JP. Pedicle screw design and cement augmentation in osteoporotic vertebrae: effects of fenestrations and cement viscosity on fixation and extraction. Spine (Phila Pa 1976). 2012;37(26):E1628–32.
- Kuhns CA, Reiter M, Pfeiffer F, Choma TJ. Surgical strategies to improve fixation in the osteoporotic spine: the effects of tapping, cement augmentation, and screw trajectory. Glob Spine J. 2014;4(1):47–54.
- Becker S, Chavanne A, Spitaler R, Kropik K, Aigner N, Ogon M, et al. Assessment of different screw augmentation techniques and screw designs in osteoporotic spines. Eur Spine J. 2008;17(11):1462–9.
- 29. Hoppe S, Loosli Y, Baumgartner D, Heini P, Benneker LM. Influence of screw augmentation in posterior dynamic and rigid stabilization systems in osteoporotic lumbar vertebrae: a biomechanical cadaveric study. Spine (Phila Pa 1976). 2014;39(6):E384–9.
- Helgeson MD, Kang DG, Lehman Jr RA, Dmitriev AE, Luhmann SJ. Tapping insertional torque allows prediction for better pedicle screw fixation and optimal screw size selection. Spine J: Off J N Am Spine Soc. 2013;13(8):957–65.
- 31. Okuyama K, Abe E, Suzuki T, Tamura Y, Chiba M, Sato K. Influence of bone mineral density on pedicle screw fixation: a study of pedicle screw fixation augmenting posterior lumbar interbody fusion in elderly patients. Spine J: Off J N Am Spine Soc. 2001;1(6):402–7.
- 32. Lee JH, Lee JH, Park JW, Shin YH. The insertional torque of a pedicle screw has a positive correlation with bone mineral density in posterior lumbar pedicle screw fixation. J Bone Joint Surg Br. 2012;94(1):93–7.
- Kobayashi N, Numaguchi Y, Fuwa S, Uemura A, Matsusako M, Okajima Y, et al. Prophylactic

vertebroplasty: cement injection into non-fractured vertebral bodies during percutaneous vertebroplasty. Acad Radiol. 2009;16(2):136–43.

- 34. Kebaish KM, Martin CT, O'Brien JR, LaMotta IE, Voros GD, Belkoff SM. Use of vertebroplasty to prevent proximal junctional fractures in adult deformity surgery: a biomechanical cadaveric study. Spine J: Off J N Am Spine Soc. 2013;13(12):1897–903.
- Schwab F, Dubey A, Gamez L, El Fegoun AB, Hwang K, Pagala M, et al. Adult scoliosis: prevalence, SF-36, and nutritional parameters in an elderly volunteer population. Spine (Phila Pa 1976). 2005;30(9):1082–5.
- de Vries AA, Mullender MG, Pluymakers WJ, Castelein RM, van Royen BJ. Spinal decompensation in degenerative lumbar scoliosis. Eur Spine J. 2010;19(9):1540–4.
- 37. Smith JS, Shaffrey CI, Glassman SD, Berven SH, Schwab FJ, Hamill CL, et al. Risk-benefit assessment of surgery for adult scoliosis: an analysis based on patient age. Spine (Phila Pa 1976). 2010; 36(10):817–24.
- Sansur CA, Smith JS, Coe JD, Glassman SD, Berven SH, Polly Jr DW, et al. Scoliosis research society morbidity and mortality of adult scoliosis surgery. Spine (Phila Pa 1976). 2011;36(9):E593–7.
- Ozgur BM, Aryan HE, Pimenta L, Taylor WR. Extreme lateral interbody fusion (XLIF): a novel surgical technique for anterior lumbar interbody fusion. Spine J: Off J N Am Spine Soc. 2006;6(4):435–43.
- 40. Kleinschmidt M, Heini P. Degenerative deformities of the lumbar spine. Treatment approach by minimal invasive anterior correction by extreme lateral interbody fusion. Technical and radiological aspects and review of the literature. Minerva Ortop Traumatol. 2011;62:361–72.
- Dua K, Kepler CK, Huang RC, Marchenko A. Vertebral body fracture after anterolateral instrumentation and interbody fusion in two osteoporotic patients. Spine J: Off J N Am Spine Soc. 2010;10(9):e11–5.
- Tannoury CA, An HS. Complications with the use of bone morphogenetic protein 2 (BMP-2) in spine surgery. Spine J. 2014;14(3):552–9.
- 43. Charosky S, Guigui P, Blamoutier A, Roussouly P, Chopin D. Complications and risk factors of primary adult scoliosis surgery: a multicenter study of 306 patients. Spine (Phila Pa 1976). 2012;37(8):693–700.
- Inaoka M, Tada K, Yonenobu K. Problems of posterior lumbar interbody fusion (PLIF) for the rheumatoid spondylitis of the lumbar spine. Arch Orthop Trauma Surg. 2002;122(2):73–9.
- Barrett-Connor E. The economic and human costs of osteoporotic fracture. Am J Med. 1995;98(2A):3S–8.
- 46. Sven H, Yannick L, Daniel B, Paul H, Lorin B. Influence of screw augmentation in posterior dynamic and rigid stabilization systems in osteoporotic lumbar vertebrae: a biomechanical cadaveric study. Spine (Phila Pa 1976). 2014;39(6):E384–9.

# The Choice of TLIF for Lumbar Interbody Fusion

David Straus, Ricardo B.V. Fontes, and Vincent C. Traynelis

# 27.1 Introduction and Literature Review

The development of posterior approaches for lumbar interbody fusion within the context of degenerative disk disease (DDD) is inherently related to the concept of discogenic pain and its treatment as described by Cloward in 1953 [1]. Lumbar interbody fusion (LIF) had been occasionally utilized since the beginning of the twentieth century for the treatment of spinal tuberculosis and other infectious etiologies -Muller reported on the transperitoneal LIF for tuberculosis as early as 1906 [2]. Its utilization for degenerative conditions, however, was seen as some sort of taboo at that time - most spine surgeons, particularly those with a neurosurgical background, viewed leg pain resulting from direct neural compression and inflammation secondary to a herniated disk fragment as the only degenerative condition worth operating on, one that simple diskectomy could properly address [3]. However, when back pain ensued following a diskectomy procedure, it was still seen as an

Department of Neurosurgery,

Rush University Medical Center, 1725 W Harrison, Suite 855, Chicago, IL 60610, USA e-mail: ricardo\_fontes@rush.edu unfortunate occurrence that should be graciously endured by the patient as an acceptable trade-off for the relief of debilitating leg pain.

In order to address the issue of back pain, noninstrumented, in situ fusions such as described by Hibbs were attempted in conjunction with diskectomy, and the results were not only considered unsatisfactory, but the entire procedure was felt to be unnecessary: Dandy famously stated in 1944 that thorough debridement of the disk space was enough to induce collapse and arthrodesis following diskectomy [4, 5]. Van Wagenen and Dandy may have packed the interspace with autologous bone following diskectomy even before 1953, but Cloward is usually credited as the first to describe posterior LIF in greater detail, along with developing a pathophysiological theory to support it, carefully planning the grafts and designing a series of instruments to assist in nerve root and dural retraction [1, 4]. Cloward reported excellent results, including a perfect fusion rate, supporting this technique so staunchly as to state that it should be performed following every diskectomy [6].

Despite Cloward's praise for the technique he developed, it was slow to gain acceptance. Concerns over nerve root retraction, donor site morbidity, graft protrusion, long hospital stays, and the degree of bed rest and bracing necessary to achieve arthrodesis were important factors [7]. Fusion rate on the other hand was far from the alleged 100 % and collapse due to violation of

D. Straus, MD • R.B.V. Fontes, MD, PhD (🖂)

V.C. Traynelis, MD

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_27

the vertebral body end plate still a frequent occurrence [8]. Most authors advocating noninstrumented PLIF also maintained that integrity of the posterior joints was necessary to avoid instability; therefore, some residual pain could be a result of inadequate decompression [**9**]. Introduction in 1988 of pedicle-based posterior spine instrumentation in conjunction with LIF enabled a more thorough decompression since reconstruction of a posterior tension band was possible; earlier mobilization after surgery and obviation of postoperative bracing were added benefits [7]. PLIF is now almost universally done with the adjunct of pedicle screw-based instrumentation; addition of intertransverse graft material has been variably employed, with conflicting reports of its efficacy [10]. However, even today, reports claiming fusion rates are similar with non-instrumented LIF are found in the literature [11].

The next important development happened in the 1990s, with the description by Harms of posterior interbody fusion through a so-called transforaminal approach, or TLIF [12, 13]. This approach to the disk is performed through the intervertebral foramen, i.e., a more lateral approach than the Cloward's PLIF, therefore necessitating no manipulation of the thecal sac or roots. Though initially described as an open procedure with unilateral diskectomy and backed up by bilateral pedicle screw instrumentation, a number of modifications have been reported in the 15 years since the original report. These variants include more invasive operations, designed to achieve better fusion rates, e.g., with bilateral access to the disk, and also less invasive variants, including unilateral pedicle screw fixation and the so-called "mini-open" (transmuscular approach with pedicle screw insertion under direct vision) and minimally invasive techniques (only the interbody component is performed through a tubular system, backed up by percutaneous screw placement).

The minimally invasive variants of TLIF have sought to reduce damage to surrounding structures; this has been demonstrated in terms of reduced blood loss during surgery, serum markers of muscle injury, in-hospital narcotic use, length of hospital stay, return to work rates, and late postoperative imaging, for example. Late outcome data, however, is essentially equivalent across all techniques when fusion does take place: these have almost universally helped patients with radiated complaints that can be successfully localized to a vertebral level, of which the perfect example is spondylolisthesis. Less satisfactory results have been obtained with back pain and revision surgery but TLIF can still be an option when nonoperative treatment fails. While there is no high-level data, long-term clinical outcomes from these various TLIF techniques have been consistently similar across a number of studies [14–18]. Blood loss [19], postoperative pain [16], hospital length of stay [19], and muscle damage [20, 21], however, have been shown to be decreased in the mini-open technique as compared to the open technique, generating a possible cost-effectiveness advantage of this procedure over the open TLIF technique [16]. The miniopen technique, however, consistently requires increased radiation exposure to both the patient and surgeon [19]. Though reports have varied, a recent meta-analysis identified nonsignificant trends for increased incidence of graft malposition, nonunion, and reoperation rates with the mini-open technique and nonsignificant trends toward increased incidence of durotomy and infection with the open technique [19]. Overall operative time was similar between the miniopen and open TLIF technique [19], though individual reports have demonstrated significant differences in both directions [20, 22-25], indicating operator familiarity plays an important role in the execution of each technique.

Adjunctive instrumentation is routinely used in current practice. Most commonly, bilateral pedicle screw fixation is utilized to provide internal fixation and posterior tension band reconstruction for LIF procedures. Even though unilateral screw fixation may be sufficient for TLIF, clinical reports have reported conflicting and potentially worse outcomes [26, 27], and there is biomechanical evidence that bilateral pedicle screw fixation most closely approximates the properties of the native lumbar spine, whereas non-instrumented and unilaterally instrumented interbody constructs show increased mobility especially in lateral bending [26]. Two recent small randomized, controlled trials have revealed significantly less improvement in radiated complaints and more cage migration when compared to bilateral pedicle screw fixation [28, 29]. While various alternative contralateral instrumentation techniques such as facet screws have also been reported [30], there is insufficient outcomes data in the literature to make any further assessment of their role in TLIF.

There is also variation in the type of interbody spacer to use for LIF procedures. Autologous iliac crest graft has been used extensively in spine fusions and has demonstrated a favorable profile, though concerns related to donor-site morbidity [31, 32] have led to the search for additional options. Machined allograft spacers, bioabsorbable spacers, and cages of various compositions including BAK, titanium, PEEK, and ceramic spacers have been employed as interbody grafts [33]. Cage shape does not appear to alter the biomechanical properties of the interbody graft if posterior pedicle screw fixation is employed [34], and the risk of graft subsidence is most related to patient factors, particularly bone mineral density [35].

There is no consensus in the literature regarding the ideal fusion technique for the lumbar spine. Circumferential fusion has several theoretical advantages over posterolateral fusion: according to Wolff's law, a distraction-type/interbody graft has the highest fusion potential; the bone graft in LIF is placed in the load-bearing surfaces of the anterior and middle spinal columns, and the vertebral interspace has significantly better vascularity than the posterolateral space [36]. From the biomechanical standpoint, LIF has the potential to restore interspace height, lumbar lordosis, and coronal and sagittal balance of the lumbar spine, also resulting in foraminal decompression. Posterior approaches for LIF have the added advantages of allowing direct neural element decompression through the same route. However, clinical trials looking mainly at 1- and 2-level fusions have failed to demonstrate a clear advantage of P/TLIF over posterolateral fusion in terms of validated outcome measures. The frequently quoted randomized trial from the Swedish Lumbar Spine Study Group established the superiority of arthrodesis over nonoperative care in patients with chronic low back pain thought to originate from spondylosis. In a sub-analysis of the surgical arm comparing the three different techniques employed in that study (in situ fusion, instrumented posterolateral, and circumferential, of which approximately 25 % were PLIFs), it was found that there was no statistically significant difference in outcome [37]. A follow-up study from the same group comparing posterolateral fusion and PLIF for patients with isthmic spondylolisthesis also failed to demonstrate a difference in outcome [38]. A sub-analysis of the surgical group from the SPORT study has failed to reveal significant differences in the treatment of degenerative spondylolisthesis between non-instrumented in situ fusion, posterolateral fusion with pedicle instrumentation, and posterior interbody fusion in the long term. These results, however, stem from a nonrandomized, non-blinded study design with considerable selection bias, as the authors themselves pointed out [39].

It is uncertain how the utilization of rhBMP-2 will impact this comparison; its off-label use in TLIF has become very popular in the United States [40, 41]. Recent concerns related to unintended complications from rhBMP-2 have been raised and include reports of postoperative radiculitis, heterotopic ossification, and vertebral body osteolysis [42]. Furthermore, the biomechanical advantages of circumferential fusion may become more evident over longer fusions where sagittal balance is a strong consideration, high-grade spondylolisthesis, and/or revision surgery - given its transforaminal nature, the surgeon may be able to completely avoid a prior laminectomy defect to introduce the interbody graft. Given its safety and reliability proven in the 15-20 years of utilization, TLIF is the author's fusion technique of choice for the lower lumbar spine.

# 27.2 Surgical Indications and Technique

Despite the paucity of clear level 1 evidence to elucidate the precise indications and contraindications for LIF, these techniques have emerged as core procedures in lumbar spinal surgery and represent commonly employed means of achieving lumbar arthrodesis in a variety of pathological conditions. LIF is often used in the management of both mobile and immobile spondylolistheses [43], degenerative lumbar deformities [44], pseudoarthroses [45], and radiculopathies from recurrent herniated nucleus pulposus [22]. LIF may also be applied to select patients with degenerative disk disease and concordant intractable axial back pain [18, 46]. The primary technical contraindication for LIF is incompetent vertebral bodies or end plates (from trauma, tumor, infection, or osteoporosis) [33]. Contraindications specific to the PLIF technique - given the necessary retraction of the thecal sac and nerve roots to achieve interbody graft placement - include epidural fibrosis, lesions at or above L3 (given the risk of retracting the conus medullaris), and conjoined nerve roots [33].

As discussed above, LIF offers a number of important biomechanical advantages over posterolateral lumbar fusion (PLF), including circumferential fusion, foraminal decompression, and improved environment for arthrodesis. It also provides an important technique for lumbar arthrodesis when alternative techniques are suboptimal. Anterior lumbar interbody fusion (ALIF) offers similar biomechanical advantages of LIF, but circumferential fusion with ALIF requires a staged or "front-back" procedure with increased procedure time and blood loss [47], and approachrelated complications including retrograde ejaculation and visceral and vascular injuries detract from this approach [48]. Lateral lumbar interbody fusion (LLIF) provides advantages similar to ALIF but exchanges the risk of sympathetic, visceral, and vascular injury for the risk of lumbar plexus injury and the necessity of disrupting the iliopsoas muscle for exposure [49]. LIF offers an important surgical technique for achieving lumbar arthrodesis offering a distinct risk-benefit profile that has established itself as a mainstay in surgical treatment of degenerative lumbar spinal diseases.

Appropriate preoperative imaging should be available for review in the operating room. In many

cases this will include MRI or CT myelogram, standing radiographs with flexion and extension views and noncontrast CT. Close examination should identify sites of radiographic compression and have relevant clinical correlations. Noncontrast CT may provide further data for planning and sizing of pedicle screws. Attention should be paid to identifying transitional lumbosacral vertebral levels, and in such cases extra attention must be given to intraoperative fluoroscopic localization. After induction of general anesthesia, the patient is flipped prone onto an open Jackson table with all pressure points padded. Physiologic alignment is attempted in order to achieve acceptable alignment of the fusion construct. Fluoroscopy is used to confirm alignment and level and to plan the skin incision. The patient is prepped and draped in the standard manner. In our current practice, we have abandoned the medial PLIF technique in favor of a transforaminal approach in order to minimize nerve root retraction. For the conventional open variant of TLIF, a midline incision is made through the epidermis and dermis and superficial hemostasis is obtained. Following radiological confirmation of the correct vertebral level, exposure of lamina, facets, pars interarticularis, and transverse processes is performed while avoiding removal of soft tissue on the facet joints and ligaments outside of the fusion segment, particularly at the rostral level (Fig. 27.1a).

The osseous exposure of the TLIF often requires a complete, unilateral facetectomy; while some surgeons may prefer to leave a medial bridge of the lamina and articular process, we strive to maximize our decompression of the neural elements. This is begun by performing a hemilaminotomy and connecting this to a transverse osteotomy at the level of the pars interarticularis (Fig. 27.1b). The remainder of the superior articular process of the caudal vertebrae is removed to the level of the pedicle (Fig. 27.1c). The ligamentum flavum is resected and the lateral dural edge is exposed; this may be tailored to the degree of central stenosis. This should provide exposure of the axilla of the exiting nerve root; any retraction is rarely needed to access the disk (Fig. 27.1d).



**Fig. 27.1** Steps for an open right L4–L5 TLIF. (a) Exposure of the lamina and articular and transverse processes. (b) Osteotomy at the level of the right L4 pars interarticularis. (c) Resection of the right superior L5 articular process exposes the L4 nerve root and the L4–L5

disk. (d) Thorough diskectomy is performed without retraction of L4 nerve root. (e) Insertion of the cage is performed at a  $45^{\circ}$  angle and positioned across the midline. (f) Postoperative radiograph demonstrates ideal cage position and pedicle instrumentation

Bleeding from the epidural venous plexus is controlled with bipolar cautery and hemostatic agents. The annulus is incised, posterior osteophytes are removed, and a diskectomy is performed using curettes, rongeurs, and disk shavers. The trajectory of the diskectomy and the interbody graft is at a 45° angle to the disk space (as opposed to the directly ventral trajectory of the PLIF interbody graft) in order to achieve a graft placement that crosses the midline of the vertebral body (Fig. 27.1e). The end plates are prepared with a variety of instruments according to surgeon's preference and an interbody graft is placed with fluoroscopic guidance. Interbody graft placement may vary by case: more anteriorly placed grafts provide increased lordosis in cases where deformity correction is prioritized, and more posteriorly positioned grafts provide increased foraminal decompression. Biomechanically, the TLIF interbody graft should cross the midline to avoid inducing a coronal imbalance, cover more than 20 % of the end plate surface [50], and avoid the central region where the end plate is structurally weakest [51]. Uni- or bilateral pedicle screw instrumentation, posterolateral fusion, hemostasis, and wound closure are performed in the usual manner (Fig. 27.1f). Off-label utilization of rhBMP-2 is also utilized by some surgeons in the interspace (inside or outside the allograft) or the posterolateral space, particularly in patients at high risk of pseudarthrosis.

Mini-open and tubular variations of TLIF are also favored by our group. Utilizing fluoroscopy or intraoperative navigation, a paramedian incision is planned 4 cm off the midline. A series of tubular dilators are used to create a transmuscular (Wiltse-type) working channel. Ideal docking position for a TLIF is over the ipsilateral pars interarticularis with an expandable (mini-open variant) or fixed 26 mm working tube. Neural element decompression and preparation of the interspace are performed in the same manner utilizing adapted instruments with a bayoneted shaft so as not to obstruct the field of view. Expandable dilators allow for a working space of 45-50 mm, so single-level pedicle screw instrumentation can then be placed under direct vision; otherwise percutaneous instrumentation

is utilized. Given the increased radiation exposure inherent to mini-open TLIF procedures, a further alternative is to replace fluoroscopy with O-arm guidance, thus decreasing radiation exposure [52, 53] and potentially improving accuracy of pedicle screw insertion [54].

# 27.3 Complications

Fusion rates are comparable between P/TLIF procedures, with both achieving high rates of arthrodesis, upward of 90 % in most reported studies (average-pooled fusion rate of 93.2 %) [55-60]. A recent review described the average published rates of major complications associated with LIF procedures [61], noting a 4.9 % (0–7 %) incidence of neurological injury, a 5.3 % (0-11 %) rate of radiculitis, a 10.6 % (0-35 %) rate of hardware or graft migration, a 7.3 % (2-14 %) incidence of durotomy, and a 3.7 % (0-9 %) incidence of infection. The higher reported rate of neurological complications in PLIF as compared to TLIF has in part heralded the increased utilization of TLIF over PLIF approaches for posterior lumbar interbody fusions [58].

Insertion of the intervertebral cage is the additional step that sets P/TLIF apart from other forms of posterior instrumented fusion and therefore carries the additional risks. Intervertebral graft displacement into the spinal canal was a much-feared complication before the advent of supplementary pedicle screw fixation but now is exceedingly rare; the cage must be snug against the end plates and ideally distract slightly the interspace. This distraction needs to be counterbalanced with the risk of end plate violation and fracture, particularly in osteoporotic patients and more recent self-expanding cages. Distraction in these patients can also be performed on the pedicle screws firmly anchored in cortical bone, less affected by osteoporosis. Other maneuvers such as alternatively distracting the interspace from both sides may also be employed.

The anterior annulus fibrosus should never be violated during the diskectomy and end plate preparation process: extreme care must be taken during this step, particularly with automatic dis-



**Fig. 27.2** Axial image (**a**) and sagittal reconstruction (**b**) of a CT of the abdomen demonstrating violation of the anterior annulus and intra-abdominal intervertebral

kectomy devices and stackable or self-expanding cages. Expulsion of the graft into the abdomen may be catastrophic at L4–L5 or above due to large vessel injury. Immediate general or vascular surgery assistance and emergent laparotomy may be a lifesaving measure in these cases (Fig. 27.2).

As discussed above, variable fusion rates can be found in the literature for posterior LIF; there is no clear advantage of one technique over another. It is our opinion that thorough debridement of the interspace is the biggest determinant of fusion, rather than utilization of one specific posterior LIF variation. Less nerve root retraction is necessary for TLIF, but extreme care should be taken to avoid cautery around the dorsal root ganglion, which may lead to postoperative dysesthesia that may be particularly difficult to treat.

# Conclusion

Posterior interbody fusion is a safe and timetested technique that may lead to excellent stabilization of the motion segment even after a very aggressive neural element decompression. As with any other surgical techniques applied to degenerative spinal disorders, its

implant. This occurred at the L5–S1 level and therefore below the aortic bifurcation; the patient underwent laparotomy and ALIF

effectiveness is limited by the ability of the clinician to adequately examine the patient, correlate his or her complaints to radiological findings, and select the appropriate surgical technique.

# References

- Cloward RB. The treatment of ruptured lumbar intervertebral discs by vertebral body fusion. I. Indications, operative technique, after care. J Neurosurg. 1953;10(2):154–68. doi:10.3171/jns.1953.10.2.0154.
- Muller W. Transperitoneale Freilegung der Wirbelsaule bei Tuberkuloser Spondylitis. Dtsch Z Chir. 1906;85:128–35.
- Poppen JL. The ruptured intervertebral disc. Bull N Engl Med Cent. 1944;6:403–12.
- 4. Dandy WE. Newer aspects of ruptured intervertebral disks. Ann Surg. 1944;119(4):481–4.
- Hibbs RA. The treatment of deformities of the spine caused by poliomyelitis: a report of eight cases in which fusion operations were performed. J Am Med Assoc. 1917;LXIX(10):787. doi:10.1001/ jama.1917.02590370023010.
- Cloward RB. Changes in the vertebra caused by ruptured intervertebral discs; observations on their formation and treatment. Am J Surg. 1952;84(2):151–60.
- Steffee AD, Sitkowski DJ. Posterior lumbar interbody fusion and plates. Clin Orthop. 1988;227:99–102.

- Lin PM. A technical modification of Cloward's posterior lumbar interbody fusion. Neurosurgery. 1977;1(2):118–24.
- Lin PM, Cautilli RA, Joyce MF. Posterior lumbar interbody fusion. Clin Orthop. 1983;180:154–68.
- Kim K-T, Lee S-H, Lee Y-H, Bae S-C, Suk K-S. Clinical outcomes of 3 fusion methods through the posterior approach in the lumbar spine. Spine. 2006;31(12): 1351–7. doi:10.1097/01.brs.0000218635.14571.55; discussion 1358.
- Kotil K, Ali Akçetin M, Savaş Y. Clinical and radiologic outcomes of TLIF applications with or without pedicle screw: a double center prospective pilot comparative study. J Spinal Disord Tech. 2013;26(7):359– 66. doi:10.1097/BSD.0b013e318249599f.
- Harms J, Rolinger H. A one-stager procedure in operative treatment of spondylolistheses: dorsal tractionreposition and anterior fusion (author's transl). Z Für Orthop Ihre Grenzgeb. 1982;120(3):343–7. doi:10.10 55/s-2008-1051624.
- Harms J, Jeszensky D. Die posteriore, lumbale, interkorporelle Fusion in unilateraler transforaminaler Technik. Oper Orthop Traumatol. 1998;10(2):90–102.
- Adogwa O, Parker SL, Davis BJ, et al. Costeffectiveness of transforaminal lumbar interbody fusion for grade I degenerative spondylolisthesis: clinical article. J Neurosurg Spine. 2011;15:138–43.
- Ntoukas V, Müller A. Minimally invasive approach versus traditional open approach for one level posterior lumbar interbody fusion. Min-Minim Invasive Neurosurg. 2010;53:21–4.
- 16. Parker SL, Adogwa O, Bydon A, Cheng J, McGirt MJ. Cost-effectiveness of minimally invasive versus open transforaminal lumbar interbody fusion for degenerative spondylolisthesis associated low-back and leg pain over two years. World Neurosurg. 2012;78:178–84.
- Peng CWB, Yue WM, Poh SY, Yeo W, Tan SB. Clinical and radiological outcomes of minimally invasive versus open transforaminal lumbar interbody fusion. Spine. 2009;34:1385–9.
- Schizas C, Kulik G, Kosmopoulos V. Disc degeneration: current surgical options. Eur Cell Mater. 2010;20: 306–15.
- Tian N-F, Wu Y-S, Zhang X-L, Xu H-Z, Chi Y-L, Mao F-M. Minimally invasive versus open transforaminal lumbar interbody fusion: a meta-analysis based on the current evidence. Eur Spine J. 2013;22:1–9.
- 20. Fan S, Hu Z, Zhao F, Zhao X, Huang Y, Fang X. Multifidus muscle changes and clinical effects of onelevel posterior lumbar interbody fusion: minimally invasive procedure versus conventional open approach. Eur Spine J. 2010;19(2):316–24.
- Gejo R, Matsui H, Kawaguchi Y, Ishihara H, Tsuji H. Serial changes in trunk muscle performance after posterior lumbar surgery. Spine. 1999;24:1023–8.
- Lee JK, Amorosa L, Cho SK, Weidenbaum M, Kim Y. Recurrent lumbar disk herniation. J Am Acad Orthop Surg. 2010;18:327–37.
- Scheufler K-M, Dohmen H, Vougioukas VI. Percutaneous transforaminal lumbar interbody fusion for the treatment

of degenerative lumbar instability. Neurosurgery. 2007; 60:203–13.

- 24. Wang H-L, Lu F-Z, Jiang J-Y, Ma X, Xia X-L, Wang L-X. Minimally invasive lumbar interbody fusion via MAST quadrant retractor versus open surgery: a prospective randomized clinical trial. Chin Med J Beijing. 2011;124:3868.
- 25. Wang J, Zhou Y, Zhang ZF, Li CQ, Zheng WJ, Liu J. Minimally invasive or open transforaminal lumbar interbody fusion as revision surgery for patients previously treated by open discectomy and decompression of the lumbar spine. Eur Spine J. 2011;20:623–8.
- 26. Harris BM, Hilibrand AS, Savas PE, et al. Transforaminal lumbar interbody fusion: the effect of various instrumentation techniques on the flexibility of the lumbar spine. Spine. 2004;29(4):E65–70.
- Javernick MA, Kuklo TR, Polly Jr DW. Transforaminal lumbar interbody fusion: unilateral versus bilateral disk removal–an in vivo study. Am J Orthop Belle Mead N J. 2003;32(7):344–8; discussion 348.
- Aoki Y, Yamagata M, Ikeda Y, et al. A prospective randomized controlled study comparing transforaminal lumbar interbody fusion techniques for degenerative spondylolisthesis: unilateral pedicle screw and 1 cage versus bilateral pedicle screws and 2 cages. J Neurosurg Spine. 2012;17(2):153–9. doi:10.3171/201 2.5.SPINE111044.
- Duncan JW, Bailey RA. An analysis of fusion cage migration in unilateral and bilateral fixation with transforaminal lumbar interbody fusion. Eur Spine J. 2012;22(2):439–45. doi:10.1007/s00586-012-2458-x.
- Jang J-S, Lee S-H. Minimally invasive transforaminal lumbar interbody fusion with ipsilateral pedicle screw and contralateral facet screw fixation. J Neurosurg Spine. 2005;3:218–23.
- Ahlmann E, Patzakis M, Roidis N, Shepherd L, Holtom P. Comparison of anterior and posterior iliac crest bone grafts in terms of harvest-site morbidity and functional outcomes. J Bone Joint Surg. 2002;84:716–20.
- Banwart JC, Asher MA, Hassanein RS. Iliac crest bone graft harvest donor site morbidity: a statistical evaluation. Spine. 1995;20:1055–60.
- DiPaola CP, Molinari RW. Posterior lumbar interbody fusion. J Am Acad Orthop Surg. 2008;16(3):130–9.
- 34. Chen S-H, Lin S-C, Tsai W-C, Wang C-W, Chao S-H. Biomechanical comparison of unilateral and bilateral pedicle screws fixation for transforaminal lumbar interbody fusion after decompressive surgery– a finite element analysis. BMC Musculoskelet Disord. 2012;13:72.
- 35. Jost B, Cripton P, Lund T, et al. Compressive strength of interbody cages in the lumbar spine: the effect of cage shape, posterior instrumentation and bone density. Eur Spine J. 1998;7:132–41.
- Mummaneni PV, Lenke L, Haid RW. Spinal deformity: a guide to surgical planning and management. St. Louis: Quality Medical Pub; 2008.
- 37. Fritzell P, Hägg O, Wessberg P, Nordwall A, Swedish Lumbar Spine Study Group. 2001 Volvo award winner in clinical studies: lumbar fusion versus nonsurgical treatment for chronic low back pain: a multicenter

randomized controlled trial from the Swedish Lumbar Spine Study Group. Spine. 2001;26(23):2521–32; discussion 2532–4.

- Ekman P, Möller H, Tullberg T, Neumann P, Hedlund R. Posterior lumbar interbody fusion versus posterolateral fusion in adult isthmic spondylolisthesis. Spine. 2007;32(20):2178–83. doi:10.1097/BRS.0b013 e31814b1bd8.
- Abdu WA, Lurie JD, Spratt KF, et al. Degenerative spondylolisthesis: does fusion method influence outcome? Four-year results of the spine patient outcomes research trial. Spine. 2009;34(21):2351–60. doi:10.1097/BRS.0b013e3181b8a829.
- 40. Crandall DG, Revella J, Patterson J, Huish E, Chang M, McLemore R. Transforaminal lumbar interbody fusion with rhBMP-2 in spinal deformity, spondylolisthesis, and degenerative disease–part 1: large series diagnosis related outcomes and complications with 2- to 9-year follow-up. Spine. 2013;38(13):1128–36. doi:10.1097/BRS.0b013e31828864e6.
- 41. Rahman RK, Buchowski JM, Stephens B, Dorward IG, Koester LA, Bridwell KH. A comparison of TLIF with rhBMP-2 vs. No TLIF and higher posterolateral rhBMP-2 dose at L5–S1 for long fusions to the sacrum with sacropelvic fixation in primary adult deformity patients. Spine. 2013. doi:10.1097/ BRS.0000000000000045.
- 42. Adams CL, Ogden K, Robertson IK, Broadhurst S, Edis D. The effectiveness and safety of recombinant human bone morphogenetic protein-2 versus local bone graft in primary lumbar interbody fusions. Spine. 2013;13:1118–25.
- Yan D, Pei F, Li J, Soo C. Comparative study of PILF and TLIF treatment in adult degenerative spondylolisthesis. Eur Spine J. 2008;17:1311–6.
- 44. Hey HWD, Hee HT. Lumbar degenerative spinal deformity: surgical options of PLIF, TLIF and MI-TLIF. Indian J Orthop. 2010;44:159.
- Lee SH, Kim YB, Kim TS, et al. Transforaminal lumbar interbody fusion for the treatment of nonunion after posterolateral lumbar fusion. J Korean Soc Spine Surg. 2004;11:223–30.
- 46. Molinari MRW, Gerlinger MT. Functional outcomes of instrumented posterior lumbar interbody fusion in active-duty US servicemen: a comparison with nonoperative management. Spine J. 2001;1:215–24.
- 47. Andersson GB, Mekhail NA, Block JE. Treatment of intractable discogenic low back pain: a systematic review of spinal fusion and intradiscal electrothermal therapy (IDET). Pain Physician. 2006;9:237.
- 48. Fritzell P, Hägg O, Nordwall A. Complications in lumbar fusion surgery for chronic low back pain: comparison of three surgical techniques used in a prospective randomized study. A report from the Swedish Lumbar Spine Study Group. Eur Spine J. 2003;12(2): 178–89.

- Pumberger M, Hughes AP, Huang RR, Sama AA, Cammisa FP, Girardi FP. Neurologic deficit following lateral lumbar interbody fusion. Eur Spine J. 2012;21: 1192–9.
- Tan J-S, Bailey CS, Dvorak MF, Fisher CG, Oxland TR. Interbody device shape and size are important to strengthen the vertebra–implant interface. Spine. 2005;30:638–44.
- Ferguson SJ, Steffen T. Biomechanics of the aging spine. Eur Spine J. 2003;12:S97–103.
- Kim CW, Lee Y-P, Taylor W, Oygar A, Kim WK. Use of navigation-assisted fluoroscopy to decrease radiation exposure during minimally invasive spine surgery. Spine J. 2008;8:584–90.
- 53. Smith HE, Welsch MD, Sasso RC, Vaccaro AR. Comparison of radiation exposure in lumbar pedicle screw placement with fluoroscopy vs computerassisted image guidance with intraoperative threedimensional imaging. J Spinal Cord Med. 2008; 31:532.
- 54. Silbermann J, Riese F, Allam Y, Reichert T, Koeppert H, Gutberlet M. Computer tomography assessment of pedicle screw placement in lumbar and sacral spine: comparison between free-hand and O-arm based navigation techniques. Eur Spine J. 2011;20:875–81.
- Chen Z, Zhao J, Liu A, Yuan J, Li Z. Surgical treatment of recurrent lumbar disc herniation by transforaminal lumbar interbody fusion. Int Orthop. 2009;33:197–201.
- 56. Faundez AA, Schwender JD, Safriel Y, et al. Clinical and radiological outcome of anterior–posterior fusion versus transforaminal lumbar interbody fusion for symptomatic disc degeneration: a retrospective comparative study of 133 patients. Eur Spine J. 2009;18: 203–11.
- Haid Jr RW, Branch Jr CL, Alexander JT, Burkus JK. Posterior lumbar interbody fusion using recombinant human bone morphogenetic protein type 2 with cylindrical interbody cages. Spine J. 2004;4:527–38.
- Mehta VA, McGirt MJ, Garcés Ambrossi GL, et al. Trans-foraminal versus posterior lumbar interbody fusion: comparison of surgical morbidity. Neurol Res. 2011;33:38–42.
- Vaidya R, Weir R, Sethi A, Meisterling S, Hakeos W, Wybo C. Interbody fusion with allograft and rhBMP-2 leads to consistent fusion but early subsidence. J Bone Joint Surg Br. 2007;89:342–5.
- Xu H, Tang H, Li Z. Surgical treatment of adult degenerative spondylolisthesis by instrumented transforaminal lumbar interbody fusion in the Han nationality: clinical article. J Neurosurg Spine. 2009;10: 496–9.
- Chrastil J, Low JB, Whang PG, Patel AA. Complications associated with the use of the recombinant human bone morphogenetic proteins for posterior interbody fusions of the lumbar spine. Spine. 2013;38(16):E1020–7. doi:10.1097/BRS.0b013e3182982f8e.

# Anterior Lumbar Interbody Fusion (ALIF)

28

# Seth M. Zeidman and Daniel J. Hoh

Anterior lumbar interbody fusion (ALIF) is a valuable surgical technique for complete disk removal, restoration, and maintenance of intervertebral and foraminal height, improving lumbar lordosis, and immobilization of painful or unstable motion segments. ALIF compared to other posterior-based fusion techniques has the distinct advantage of preserving the posterior supporting elements of the spine including the paraspinal muscles and the posterior bony and ligamentous structures as well as avoiding any direct nerve root manipulation. An anterior approach to the lumbar spine also facilitates a more thorough disk excision and consequently allows for placement of a larger interbody fusion device. While ALIF is a popular technique for lumbar arthrodesis, it is often combined with posterior fixation (anterior/ posterior fusion) to provide a more rigid biomechanical construct compared to ALIF alone.

The history and development of ALIF reflect the progression of spinal surgery, as advances in techniques, devices, and fusion adjuncts have driven evolution of ALIF to improve patient outcomes and safety. Over the decades, a variety of operative methods and materials have been developed to facilitate both access to the disk space and achieving arthrodesis. This includes a range of surgical approaches including standard open laparotomy and transperitoneal, retroperitoneal, laparoscopic, and minimally invasive exposures. Various interbody grafts and devices have also been explored to increase the likelihood of fusion including autograft, allograft, synthetics, and osteobiologic agents as well as interbody cage technology designed for restoring intervertebral height, lordosis, and fixation.

To truly understand anterior lumbar interbody fusion is to gain a fundamental understanding of the ongoing quest to improve patient outcomes through a solid bony arthrodesis, with minimized tissue destruction, optimized spinal alignment and biomechanics, and protection of neural elements.

# 28.1 History

The first reported case of an anterior approach to the lumbar spine was in 1933 by Capener and Burns who described their experience with the transperitoneal approach for the treatment of spondylolisthesis. In 1948, Lane and Moore demonstrated good outcomes with ALIF via a transperitoneal approach with allogeneic bone graft to treat lumbar degenerative disorders. Cloward

S.M. Zeidman, MD (⊠) Rochester Brain and Spine Neurosurgery, Rochester, NY, USA e-mail: drseth@frontiernet.net

D.J. Hoh, MD Department of Neurological Surgery, University of Florida, Gainesville, Florida, USA

Center for Spine Health, Cleveland Clinic Hospitals, Cleveland, OH, USA

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_28

described ALIF in the early 1950s for the treatment of low back pain for degenerative spine conditions; however, this procedure did not gain immediate favor due to fairly high nonunion rates (30-40 %). Prior to the 1950s, most anterior lumbar approaches were extensive transperitoneal exposures, with their indications extended by the application of anterior interbody fusion methods. The Hong Kong procedure as described by Hodgson and Stock in 1956 employed ALIF as a successful modality for treating select cases of spinal tuberculosis. In 1957, Southwick and Robinson introduced the retroperitoneal approach, as an alternative exposure with potentially decreased morbidity. The transperitoneal approach involves incision of both the anterior and posterior peritonea. In contrast, the retroperitoneal approach maintains the integrity of the peritoneum by accessing the spinal column laterally behind the peritoneal sac. As a result, the retroperitoneal approach is associated with less intraoperative and postoperative bowel complications.

Since the 1990s, there has been a resurgence of anterior lumbar surgery. Increased popularity for ALIF has coincided with the advent of new interbody fusion and fixation devices such as threaded or tapered titanium cages and the development of less invasive approaches including endoscopic, laparoscopic, and mini-open techniques.

# 28.2 Indications

# 28.2.1 Degenerative Disk Disease

ALIF is most commonly indicated for the treatment of lumbar degenerative disease. The use of ALIF to treat chronic low back pain associated with degenerative disk disease is one of its most frequent indications, yet remains arguably the most controversial. The cause of chronic low back pain is often difficult to determine. Potential etiologies associated with persistent low back pain include degenerative disk disease, spondylolysis, spondylolisthesis, or iatrogenic segmental instability.

The intervertebral disk has been implicated as an etiology of chronic low back pain based on clinical, basic science, and epidemiological research.

However, there remains a lack of consensus regarding the appropriate and accurate diagnosis and treatment of intervertebral disk disorders. Radiographic imaging, often with multiple diagnostic studies used in conjunction, is necessary to make an accurate diagnosis. Functional testing such as diagnostic spinal injections or provocative discography may also play a role in identifying the source of pain. Diagnostic injections, including epidural steroids and selective nerve or facet blocks, are commonly used to better identify the pain generator. However, there is little data to support or refute the use of injections as screening tools for lumbar fusion. Provocative discography, which includes disk stimulation and morphological evaluation, is often used to distinguish a painful disk from other potential sources of pain. Despite the extensive literature, controversy continues about the accuracy and therefore utility of provocative discography.

Lumbar discography should be performed by those well experienced with the procedure and in sterile conditions with a double needle technique and fluoroscopic imaging for proper needle placement. Information assessed should include the volume of contrast injected, pattern of dye distribution, and pain response with particular emphasis on locality and similarity to presenting complaints. Frequently, discography is followed by axial computed tomography to further characterize disk morphology and degree of degeneration.

Most of the current literature supports the use of discography in select situations. Indications for discography include, but are not limited to, the following:

- (1) Further evaluation of abnormal disks to help assess the extent of degeneration and correlate with the clinical symptoms. Such symptoms may include recurrent pain from a previously operated disk or lateral disk herniation.
- (2) Patients with persistent, severe symptoms in whom other diagnostic tests have failed to confirm a suspected disk as the source of pain.
- (3) Assessment of patients who have failed to respond to surgical intervention for either painful pseudarthrosis, a symptomatic disk in a posteriorly fused segment, or a possible recurrent disk herniation.

- (4) Assessment of disks prior to fusion to determine if the disks within the proposed fusion segment are symptomatic and to determine if disks adjacent to the proposed fusion are asymptomatic.
- (5) Assessment of candidates for minimally invasive surgical intervention to confirm a contained disk herniation or to investigate dye distribution pattern before chemonucleolysis or percutaneous procedures.

Despite continued debate regarding the appropriate surgical indications for chronic low back pain, ALIF has become a popular treatment modality for lumbar degenerative diseases, although surgery is generally recommended only when all reasonable conservative measures (pain medications, nerve sheath injections, physical therapies, braces, etc.) have failed. ALIF is particularly well suited for degenerative disk disease. ALIF indirectly decompresses exiting nerve roots by increasing intervertebral height and opening narrowed neural foramina. Further, ALIF immobilizes the motion segment by fusing the vertebra adjacent to the disk space.

#### 28.2.2 Spondylolisthesis

ALIF can also be effective for treating spondylolisthesis or other degenerative or iatrogenicrelated instability; however, in these instances, supplemental posterior fixation to reestablish the posterior tension band and for rigid stabilization is frequently necessary. Some may argue that patients with a low-grade (grade I) spondylolysis or spondylolisthesis may be effectively treated with ALIF as an isolated procedure. Present data regarding the effectiveness of stand-alone ALIF in grade II spondylolisthesis is inconclusive. Biomechanical data related to the degree of vertebral translation concomitant with grade III or greater spondylolisthesis suggest that stand-alone ALIF in these settings may be predisposed to a high pseudarthrosis rate [1]. Therefore, in grade III or greater spondylolisthesis, instrumented posterior fusion in addition to ALIF is strongly recommended, whereas a stand-alone ALIF is only a reasonable option in grade I situations.

#### 28.2.3 Failed Spine Surgery

Attempting posterior revision surgery in patients with failed prior posterior surgery can be hazardous. In this clinical scenario, an anterior approach may obviate many of these potential risks and increase likelihood for a successful patient outcome. An anterior approach avoids working through scar tissue and the significant risk of both neural injury and dural violation. Additionally, some patients continue to have pain after posterolateral spinal fusion despite apparently solid arthrodesis [2]. One potential etiology is pain that arises from an abnormal disk within the fused levels. Low back pain that continues or recurs after apparently solid posterolateral spinal fusion may be caused by painful disk(s) at motion segment(s) within the fusion, as a solid posterolateral spinal fusion may not protect residual disks from injury. Anterior interbody fusion can provide significant improvements in pain and function and a high degree of patient satisfaction in this clinical setting.

#### 28.2.4 Other Indications

ALIF frequently plays an important role in the treatment of spinal deformity. Structural interbody support is an effective method for minimizing longitudinal rod and screw-bone interface strain. Moreover, anterior load-bearing structural grafts and interbody devices have been shown to increase construct stiffness, decrease the incidence of posterior implant failure, permit the use of smaller diameter longitudinal rods, and may enhance the rate of successful spinal arthrodesis [1]. Therefore, in clinical situations such as correction of deformity where the spinal implants are subjected to greater biomechanical stress, incorporating ALIF into the surgical construct facilitates correction, provides load sharing, and improves arthrodesis.

An anterior approach for debridement and fusion with autologous bone graft is also an effective surgical modality for treating pyogenic spondylodiscitis. ALIF allows for thorough removal of infected tissue; establishment of viable, normal bony surfaces; and arthrodesis with generally autologous graft to prevent chronic instability and the development of chronic pain. Recent studies have focused on the safety and use of titanium and synthetic implants as well as osteobiologic agents in the setting of spinal infection.

# 28.3 Contraindications

ALIF has a broad range of applications and indications for pathologies affecting the lumbar spine, with relatively few contraindications. In general, contraindications to ALIF are limited to approach and bone quality. Patients with preexisting medical conditions or prior surgeries involving the abdomen or retroperitoneal space may have an unacceptably high risk associated with performing an anterior surgical approach. In these instances, the assistance of an experienced vascular, urologic, or general surgeon is recommended to obtain exposure. Alternatively, a strictly posterior-based approach may be entertained in order to completely avoid the attendant risk of performing an anterior surgical procedure in these patients.

Because ALIF relies on the integrity of the adjacent vertebra in compressive loading of the interbody graft, ALIF is contraindicated in patients with severely low bone mineral density. Patients with osteoporosis are at high risk for failure of the vertebral end plates with subsidence of the interbody graft or device into the vertebral body, resulting in instability, deformity, pain, and potential neurologic compromise. In patients with low bone mineral density that require ALIF, supplemental posterior pedicle screw instrumentation is recommended to off-load stress at the implant–endplate interface.

# 28.4 Techniques

Anterior lumbar approaches provide excellent visualization of the vertebra and the entire disk space for diskectomy, fusion bed preparation, graft placement, and, in certain cases, instrumentation. During the surgical exposure, numerous vascular, visceral, muscular, urogenital, and nervous structural elements are at risk. The anatomic proximity of these structures to the site of decompression, grafting, and instrumentation further increases the chance of injury. Other more distal structures are also at risk with overly aggressive mobilization and retraction. A detailed understanding of the relevant regional anatomy and the potential for iatrogenic injury is critical to minimize morbidity associated with these procedures.

ALIF may be utilized as an isolated procedure or in conjunction with posterior spinal fusion. The method by which ALIF is accomplished depends largely on the surgeon's preference, training, and experience. There are many factors to consider in determining whether to use anterior-only or combined anterior and posterior techniques in the pediatric and adult population. Anterior-only approaches are more likely to apply to younger, healthier patients with normal bone stock and limited pathology. Patients with large deformities, borderline bone stock, and multi-segmental pathologies are more likely to benefit from a combined approach. Minimally invasive techniques - open or laparoscopic require greater intraoperative attention to detail and preoperative surgical planning.

While overall techniques can be complex and variable, all can be divided into four basic parts:

- A. Preoperative Planning/Templating: Before surgery, the surgeon refers to various imaging studies to determine what size implant(s) the patient will need. The implants are used to help promote fusion of two vertebrae in the spine.
- B. Approach/Exposure: A variety of methods can be used to access the anterior lumbar spine. These include laparoscopic, open, mini-open, and balloon-assisted endoscopic retroperitoneal gasless (BERG) techniques.

# 28.4.1 Laparoscopic/Transperitoneal

Both laparoscopic and transperitoneal techniques are effective ALIF approaches. Both approaches involve surgical exposure that passes through the abdominal cavity and the peritoneal sac. While the laparoscopic approach does involve using smaller port incisions, compared to a larger single abdominal incision for a transperitoneal approach, the laparoscopic approach does not demonstrate a definitive advantage particularly at the L5–S1 level.

With the transperitoneal approach, the surgeon makes a skin incision; dissects through the subcutaneous tissue, fascia, and muscles; and incises the anterior abdominal peritoneum to enter the abdominal cavity. Retraction of the abdominal viscera is performed, and the surgeon incises the posterior peritoneum to expose the great vessels (aortic artery, vena cava vein, and common iliac arteries and veins) and anterior aspect of the spine.

In the 1960s, Harmon reported the advantage of avoiding damage to the nerve roots with an anterior approach. He also described decreased blood transfusion, reduced hospital stay, elimination of pain, and high fusion rates. Zdeblick and David [3] reported their experience with 50 patients undergoing either laparoscopic transperitoneal or open retroperitoneal approaches. They found a significantly higher incidence of complications in the laparoscopic group (4 % vs. 20 %) and a tendency for decreased exposure resulting in limited fusions. Because of the increased incidence of retrograde ejaculation, and the increased operating room time, they ultimately abandoned the transperitoneal insufflation technique and the retroperitoneal gasless video-assisted technique. They found that with either the mini-open laparotomy approach or the traditional flank open approach, there was a lower incidence of retrograde ejaculation, decreased blood loss, shorter operating time, decreased cost, as well as the ability to use standard instruments.

#### 28.4.2 BERG Technique

Balloon-assisted endoscopic retroperitoneal gasless (BERG) techniques for anterior lumbar interbody fusion have also been described. The retroperitoneal space is accessed similarly to the technique of total extraperitoneal laparoscopic hernia repairs, via a balloon spacer and carbon dioxide insufflation. Also termed lumboscopy, this technique has the advantage of being performed minimally invasively and does not require violation of the peritoneum. A recent experience with 46 patients reported complications in 3 patients (7 %), requiring hardware removal in 1 patient [4].

Gazzeri et al. [5] analyzed their series of a simplified endoscopic approach to the anterior spine and made a review of the retroperitoneal endoscopically assisted approach to the anterior lumbar spine in the international literature. They determined that the BERG technique is a safe, effective, simplified, less technically demanding alternative approach when performing ALIF procedures, without the morbidity associated with laparoscopic or traditional approaches.

#### 28.4.3 Mini-Open

The mini-open retroperitoneal approach possesses a number of theoretical advantages; however, the individual surgeon's preference ultimately is likely to be the dictating factor in selecting this operative approach. Surgical exposure is created by bluntly dissecting behind the abdominal cavity. In this approach, the peritoneal sac is mobilized and retracted laterally. The peritoneum is dissected away from the great vessels, and the anterior spine is exposed without entering the peritoneal cavity.

The open retroperitoneal exposure for anterior spine surgery is technically challenging yet ultimately rewarding. The distinct advantage of accessing the entire disk anteriorly as opposed to limited disk access posteriorly, as well as the emergence of lumbar total disk arthroplasty, has resulted in a rise in popularity for the mini-open approach.

With a mini-open retroperitoneal approach, the superior hypogastric plexus is mobilized, with the peritoneum sweeping the small neural plexus from left to right, thus protecting it from surgical trauma. This is a distinct advantage compared to the transperitoneal approach in which the surgical dissection occurs directly over the superior hypogastric plexus, increasing the likelihood of damage.

- C. Diskectomy and Disk Space Preparation: After exposure of the disk space and the adjacent vertebral end plates has been obtained, the surgeon begins the diskectomy. Complete diskectomy is performed while maintaining integrity of the bony end plates. Release of the posterior annulus is not necessary; however, in certain instances, it may be beneficial. Preserving the bony end plates is critical for preventing subsidence of the bone graft and/ or interbody device into the adjacent end plates. Careful but gentle scraping with a rasp exposes bleeding subchondral bone without appreciably compromising endplate integrity. Once this step is complete, the graft or device to be implanted is sized and selected to maximize fit and fill within the interspace.
- D. Instrumentation/Devices: After appropriate preparation of the disk space, a graft or interbody device is inserted to promote fusion of the two adjacent vertebrae. Fusion adjuncts or osteobiologic agents are often supplemented to facilitate bony arthrodesis.

# 28.4.4 Femoral Ring Allograft (FRA)

Femoral ring allograft (FRA) is ideal for ALIF due to its osteoconductivity and large loadbearing surface which provides some immediate stability. Also, FRA eventually is resorbed and replaced by host bone via reverse creeping substitution. In contrast to bone used for posterolateral onlay fusion, FRA in ALIF is placed under compressive forces which according to Wolff's law increases the likelihood of successful arthrodesis. However, the use of stand-alone FRA without spinal fixation demonstrates a particularly high pseudarthrosis rate, suggesting that complete immobilization of the motion segment and graft may be necessary for optimal fusion.

FRA, which is primarily a cortical allograft, undergoes the same incorporation process as cancellous autograft but at a slower rate because of its compact nature. High bone density limits angiogenesis, and incorporation can only occur

following osteoclast invasion. Initially, this process leads to a mechanically weaker construct because bone resorption proceeds more rapidly than new bone formation. This process is known as "reverse creeping substitution" in contrast to what is experienced by cancellous grafts. If the graft is subjected to excessive strain, microcracks may develop, and, if revascularization is not adequate, clinically significant fractures may develop before reverse creeping substitution produces a successful fusion. The structural support can decrease by as much as 40-50 % of the initial strength at 6 months after implantation. Allograft bone also tends to cause resorption of the patient's own bone (osteolysis) at the graft/vertebral endplate interface early in the postoperative course and can actually lead to instability.

# 28.4.5 Threaded Cortical Bone Dowels (TCBD)

Allograft threaded cortical bone dowels (TCBD) were briefly popular in the mid-1990s as an interbody graft option. However, the incidence of technical failures and complications related to bone dowels exceeded 17 % in one series and resulted in a high rate of poor outcomes in patients when used as a stand-alone ALIF device [6]. It should be noted however that the incidence of clinical failures with the use of TCBDs was decreased when supplemented with anterior or posterior spinal fixation [7] or when combined with the osteoinductive growth factor, recombinant human bone morphogenetic protein-2 (rhBMP-2) [8].

Burkus et al. [8] randomized 131 patients with lumbar spondylosis undergoing single-level ALIF with TCBD to receive either rhBMP-2 or autologous bone graft. At 12 months, all patients treated with TCBD and rhBMP-2 demonstrated radiographic evidence of new bone formation and incorporation of the allograft into the adjacent vertebral end plates. Conversely, only 89 % of patients treated with TCBD and autograft had evidence of fusion at 1 year, with a decrease in fusion rate to 81.5 % at 24 months. Given the 100 % fusion rate with the addition of rhBMP-2, they concluded that TCBD can be used for standalone ALIF when combined with this particular osteoinductive agent. Because the bone dowels are threaded, they further suggested that TCBD may function to stabilize the bone–implant interface, resisting motion and risk of expulsion.

# 28.4.6 Threaded Titanium Cylindrical Cages

Threaded titanium cylindrical cages are interbody fusion devices that also provide some degree of fixation without requiring harvest of large structural autograft with a broad loadbearing surface. These devices are implanted in the interspace after reaming a screw path through the subchondral bone of the vertebra adjacent to the disk space. Threaded titanium cages do not truly fixate adjacent vertebral bodies like pedicle screw–rod-based systems; however, they do provide interference fixation. Ultimately, ligamentous annular tension due to interspace distraction combined with loading forces maintains the construct until bony fusion occurs.

There is data to suggest, however, that ligamentous annular tensioning rapidly decreases within 15 min due to stress relaxation of the soft tissues. The underlying biologic property of creep intrinsic to annular tissue allows for gradual relaxation of annular tension and loss of construct stability. It also stands to reason that the magnitude of preload across the disk space and cage due to body weight and muscle activity varies with daily activities in the absence of supplemental posterior spinal fixation. Therefore, until complete bony arthrodesis occurs, stand-alone ALIF with threaded titanium cages does not completely immobilize the motion segment. Because of the suboptimal biomechanics of stand-alone ALIF with threaded cages, this construct design is generally limited to collapsed disk spaces where there is maximal annular tensioning and at L5-S1 where there is minimal motion. Stand-alone ALIF with threaded titanium cages is not recommended for multi-level fusions or at higher lumbar segments.

Another limitation of threaded titanium cages is that bone must be removed from the subchondral end plates in order to create a screw tract for the cage. Particularly in larger disk spaces, more bone must be reamed, thereby weakening the integrity of the adjacent vertebral end plates and increasing the risk of subsidence. Increased subsidence has been observed particularly at the L4–L5 level compared to L5–S1 when using threaded cages for ALIF [9]. Subsidence is also associated with increased reaming depth and with larger cage sizes [9]. Lower risk of subsidence is seen with single-level dual-standard cage constructs.

Sasso et al. [10] prospectively studied fusion rate after stand-alone ALIF at L4–L5 or L5–S1 with either FRA or titanium threaded cages. Both interbody devices were packed with autogenous iliac crest bone graft. At 12 months, 97 % of patients treated with titanium threaded cages demonstrated radiographic fusion compared to only 40 % of patients with FRA. At 24 months, 97 % of the titanium cage group continued to show evidence of fusion, while only 52 % of the FRA group was fused. While back disability index and neurologic scores did not significantly differ between groups, the titanium cage group did have a significantly higher fusion rate and had fewer secondary supplemental fixation procedures compared to the FRA group.

One frequent criticism of titanium cages is that the extent and quality of new bone growth, incorporation, and fusion is difficult to assess on plain x-ray due to the radiopaque metal composition of the cage. Polyether ether ketone (PEEK) is radiolucent, and similar cylindrical threaded cages have been designed with this material. Postoperative assessment of new bone growth after ALIF with PEEK cages is better visualized through the interbody device. PEEK cages however do not adhere to the bony end plates as well as titanium cages, resulting in decreased pull out force, and therefore are rarely used as stand-alone devices [11]. The advent of three-dimensional computed tomography reconstruction, however, has facilitated radiographic evaluation of fusion even with titanium-based interbody cages.

#### 28.4.7 Threaded Lordotic Cages

Tapered or lordotic threaded cages were developed in order to better maintain or restore lumbar lordosis (Fig. 28.1a, b). Maintaining lordosis at



**Fig. 28.1** Posteroanterior (**a**) and lateral (**b**) radiographs demonstrating L5–S1 anterior lumbar interbody fusion with titanium lordotic cage

the fused segment is critical. Loss of lordosis, even at a single level, can affect the mechanics of adjacent segments and overall spinal alignment. Further, excessive disk space distraction with a parallel cage can distract facet joints at the same level resulting in loss of segmental stiffness or hypermobility in extension.

A retrospective study of patients undergoing stand-alone ALIF for degenerative disk disease was performed to determine if different aspects of disk space preparation and cage design affect clinical outcome [12]. The investigators observed that endplate preservation during disk space preparation was associated with improved anterior and posterior disk space distraction. They also found that use of a lordotic or tapered cage led to greater restoration of segmental lordosis than a standard cylindrical cage. These benefits associated with endplate preservation and use of a lordotic cage resulted in improved clinical outcomes as early as 3 months postoperatively and were maintained over a 2-year follow-up period.

Segmental lumbar lordosis can be achieved with parallel cylindrical cages through asymmetric reaming of the vertebral end plates. By removing more of the posterior aspect of the disk space, the adjacent vertebra rotates sagittally about their respective internal axis of rotation to settle on the cylindrical cage, creating segmental lordosis. Over-reaming of the posterior aspect of the disk however inhibits interbody distraction and thereby decreases restoration of foraminal height.

# 28.4.8 Posterior Spinal Fixation: Pedicle Screw and Translaminar Screw Constructs

The use of stand-alone interbody devices for ALIF has been met with mixed clinical success. As a result, many supplement ALIF with posterior spinal fixation with or without posterior intertransverse fusion (360° or circumferential fusion) to increase stabilization, enhance arthrodesis, and ideally improve clinical outcomes (Fig. 28.2a, b). The use of supplementary posterior fixation has been demonstrated to improve stabilization in multiple directions and increase fusion. Holte et al. [13] found that FRA combined with posterior spinal instrumentation resulted in a 98 %



**Fig. 28.2** Posteroanterior (**a**) and lateral (**b**) radiographs demonstrating L4–5, L5–S1 anterior lumbar interbody fusion with femoral ring allograft and supplemental posterior transpedicular fixation

fusion rate compared to a 75 % fusion rate with stand-alone FRA. A recent study using thinsection computed tomography revealed an 89 % fusion rate with pedicle screw–rod fixation compared to a 51 % fusion rate with stand-alone ALIF [14].

Transpedicular fixation with a pedicle screwrod construct remains the biomechanical gold standard for internal stabilization. Pedicle screws when affixed to a connecting rod have the unique capacity for three-dimensional control with restriction of motion in all planes. Conventional open pedicle screw placement, however, adds considerably to the morbidity of the surgical procedure. Besides a separate posterior incision, extensive muscle dissection and retraction is necessary to visualize the appropriate anatomic landmarks for pedicle screw placement. This can result in increased tissue injury, blood loss, operative time, postoperative pain, recovery period, and potential for nerve root or facet injury.

Alternative less invasive options for posterior spinal fixation exist. Translaminar facet screws are placed either via a mini-open technique or percutaneously over a guide wire, thereby reducing operative morbidity. A study investigated fusion rates for patients undergoing either stand-alone ALIF or ALIF with either translaminar, unilateral pedicle, or bilateral pedicle screw constructs. Thin-section computed tomography was used to assess for radiographic evidence of fusion. Patients undergoing unilateral or bilateral pedicle screws had higher fusion rates (89 % and 88 %, respectively) compared to translaminar screw (58 %) fixation or stand-alone ALIF (51 %) [14]. Alternatively, Best and Sasso [15] found that translaminar screws were associated with decreased pain scores, fewer complications, and decreased incidence of reoperation compared to pedicle screws. Recently, percutaneous transpedicular screw fixation has been developed that allows for percutaneous placement of cannulated pedicle screws over a guide wire. Novel instrumentation has been designed to allow for percutaneous insertion of a connecting rod. The long-term benefit of percutaneous pedicle screws compared to conventional open pedicle screws in terms of stabilization, enhancing fusion, and operative morbidity, however, remains to be determined.



**Fig. 28.3** Posteroanterior (**a**) and lateral (**b**) radiographs demonstrating L5–S1 anterior lumbar interbody fusion with a composite cage–screw device

# 28.4.9 Anterior Spinal Fixation: Anterior Lumbar Plate and Composite Device

In 1959, Humphries et al. first reported the use of an anterior lumbar plate to stabilize the motion segment after ALIF. Until recently, anterior lumbar plate fixation, however, did not gain popularity due to several factors. The primary concern was that obtaining access for application of the implant was limited due to surrounding vascular, gastrointestinal, and urologic structures. Implants that provided sufficient stability were often too bulky or cumbersome. Additionally, problems with device migration and screw backout plagued many of the early iterations of such instrumentation. Recently, however, lower profile anterior lumbar screws and plates have been designed which achieve better screw-bone purchase by fixating the cortical bone of the apophyseal ring. More recently composite devices which incorporate anterior screws through an interbody cage allow for an even lower profile design, effectively eliminating the offset of the anterior plate as well as providing direct fixation of the cage to the adjacent vertebral bodies. As a result, the use of anterior lumbar fixation combined with ALIF has increased in popularity, largely due to the benefit of a single anterior approach technique.

Anterior plate fixation with an interbody device improves construct stiffness and reduces range of motion compared to a stand-alone interbody construct. Glazer et al. [16] found that in human lumbar cadaveric spines, anterolateral instrumentation enhances stability of a femoral ring allograft. Similarly, Kuzhupilly et al. [17] reported significant improvement of the stability of FRA in extension when anterior crossed screws were inserted through the FRA into the adjacent vertebral bodies. Gerber et al. [18] found that a triangular anterior plate is equivalent to pedicle screw-rod fixation in limiting flexion, extension, axial rotation, and shear forces. However, pedicle screw-rod fixation remains superior to anterior plate fixation in restricting lateral bending [19]. It should be noted that in this study, specimens were tested after diskectomy and bilateral facetectomy, which may not represent the most common clinical scenario in which these devices may be employed. Recent data suggests that a composite device consisting of anterior lumbar screws that thread through an interbody cage into the adjacent vertebral bodies provides equivalent stabilization to an anterior cage with pedicle screw fixation and equivalent if not greater stabilization than ALIF with translaminar facet screws (Fig. 28.3a, b).

# 28.5 Graft Material

Autogenous bone remains the gold standard for graft material. Autograft combines the essential properties of osteogenicity, osteoinductivity, and osteoconductivity necessary for successful arthrodesis. In addition, autograft can be harvested to include cortical bone for structural support in load sharing. In the past, structural iliac crest bone graft was harvested for ALIF. This technique, however, is associated with a high complication rate including chronic pain, blood loss, infection, and pelvic fracture. Further, iliac crest bone graft has limited load-sharing capacity given the available size of the graft and therefore is susceptible to fracture, particularly without supplemental posterior fixation. Alternatively, autologous cancellous bone marrow can be harvested and packed into FRA or cages to combine the optimal biomechanical support of these interbody devices with the enhanced fusion potential of autologous bone.

Alternative strategies for harvesting autologous bone for ALIF have been explored. One technique involves obtaining a core of local bone from the adjacent vertebral body for autograft and replacing the void with a beta-calcium triphosphate plug [20]. This method has been evaluated both in animal and clinical studies and has been demonstrated to be effective. Harvesting a cylinder of autograft from the adjacent vertebral body is an efficient and less morbid technique than iliac crest bone graft. However, when using this technique with posterior pedicle screw stabilization, careful planning with regard to the site of bone removal and screw trajectory must be made to ensure optimal screw fixation without violating the defect.

Allograft in the form of FRA is a commonly used graft material for ALIF. FRA is particularly attractive as it provides a broad load-sharing surface, is readily available, and avoids the morbidity of harvesting autologous bone. FRA is primarily an osteoconductive graft and lacks any osteoinductive or osteogenic potential. As a result, stand-alone ALIF with FRA results in low fusion rates. Therefore, FRA for ALIF generally requires either rigid immobilization with spinal fixation and/or the addition of fusion adjuncts with osteogenicity or osteoinductivity to promote arthrodesis.

The discovery of bone morphogenetic proteins (BMP) and the subsequent ability to use recombinant human gene technology to produce synthetic BMP have revolutionized ALIF. BMPs are a group of osteoinductive proteins that form part of the TGF- $\beta$  superfamily. Several different types of BMPs have been identified and have been implicated in bone and cartilage formation as well as angiogenesis. BMPs appear to operate by binding to mesenchymal stem cell receptors, initiating a complex cascade of events that leads to cell differentiation and proliferation, promoting in vivo bone formation. Among bone morphogenetic proteins, BMP-2 and BMP-7 have been the subject of considerable attention as particularly powerful osteoinductive agents.

Commercial manufacturing of recombinant human BMP-2 (rhBMP-2) as InFuse (Medtronic, Memphis, TN, USA) has been most widely studied as an osteobiologic adjunct for fusion. A Food and Drug Administration (FDA) study was performed to assess the safety and efficacy of rhBMP-2 in a tapered interbody cage for ALIF [21, 22]. The investigators found that rhBMP-2 is safe and could effectively replace autogenous bone graft for ALIF, thereby avoiding donor site complications. Not unexpectedly, recent clinical trials have been performed to demonstrate equivalency of rhBMP-2 to autologous iliac crest bone graft in both anterior and posterolateral lumbar fusions.

It should be noted, however, that BMP-2 does not only increase bone formation but appears to upregulate bone resorption as well. This is likely due to BMP-2 stimulation of both osteoblast and osteoclast differentiation from progenitor cells. This characteristic is clinically relevant as the use of rhBMP-2 with FRA in stand-alone ALIF has been demonstrated to result in increased graft fracture and nonunion compared to FRA packed with autologous iliac crest bone marrow [23]. Likely, rhBMP-2-induced osteoclast activity resulted in advanced resorption of the femoral ring allograft as well as erosion of the adjacent vertebral end plates leading to graft fracture, subsidence, and nonunion. Therefore, posterior pedicle screw fixation is generally recommended when using rhBMP-2 combined with FRA to maintain stability during the upregulated osteoclast phase until new bone formation occurs.

#### 28.6 Complications

Complications associated with ALIF can be divided into those related to the surgical exposure and those that occur with diskectomy, graft insertion, and hardware placement. Most intraoperative complications of concern during ALIF are associated with the surgical approach, because with appropriate exposure, potential problems with diskectomy, graft insertion and hardware placement are generally minimized. In fact, one of the main advantages of the ALIF approach is that effective anterior exposure facilitates disk removal, endplate preparation, insertion of a biomechanically optimal graft or device, and placement of anterior fixation if necessary, while protecting the neural elements and surrounding dura.

The list of critical structures at risk in the abdomen and retroperitoneum during surgical exposure for ALIF is lengthy. These include the psoas muscle, small intestine, colon, rectum, bladder, kidneys, ureters, diaphragm and crura, medial arcuate ligament, esophageal hiatus, thoracic duct, lumbosacral plexus, greater splanchnic nerves, phrenic nerves, sympathetic chain, superior and inferior hypogastric plexus, aorta, vena cava, segmental and radicular arteries, common iliac vessels, iliolumbar veins, and medial sacral artery.

#### 28.6.1 Vascular Injury

The major risk of ALIF with perhaps the most critical consequence is injury to a great vessel. Depending on the vertebral level, ALIF is associated with risk of injury to the iliac arteries and veins. Vascular injury with anterior exposure is a potentially life-threatening complication as rapid excessive blood loss can occur with a large tear or avulsion of a vessel. Rate of vascular injury after anterior lumbar exposure is reported to be 1-15 %. Venous injury is particularly difficult as these vessels are not easily repaired, even by an experienced vascular surgeon. The incidence of venous injury in primary anterior surgery for lumbar disorders has been reported from 0 to 25 %, depending on case series, approach, and degree of venous injury recorded.

In a review of 345 anterior lumbar procedures performed on 338 patients, the incidence of a major vascular complication was 2.9 % [24]. There were nine injuries involving the common iliac vein and one aortic injury. Current or previous osteomyelitis or discogenic infection, previous anterior spinal surgery, spondylolisthesis, osteophyte formation, transitional lumbosacral vertebra, and anterior migration of the interbody device are associated with increased risk of vascular complication as these lead to scarring and adherence of vessels. Proper identification with gentle blunt dissection of these vessels from the peritoneum and the anterior spine minimizes the risk of vascular injury. Liberal use of topical hemostatic agents can help control minor bleeding from small vessel injuries while preserving vascular patency. Radicular vessels that are sacrificed require proper suture or clip ligation as these vessels tend to retract when cut and may lead to bleeding that is difficult to control otherwise. In some cases where the vessels cannot be safely mobilized or when bleeding is difficult to manage, the spinal procedure may need to be aborted and a posterior approach taken once the patient is medically stable.

A vascular structure that is uniquely troublesome during ALIF and deserves special mention is the iliolumbar vein. The iliolumbar vein is commonly, but inconsistently, seen as a vessel branching directly off the vena cava or the left iliac vein. When originating from the vena cava, it is technically termed the L5 vein, whereas it is designated the iliolumbar vein when it arises from the left iliac. After its origin, the iliolumbar vein courses directly lateral to join with the highly variable ascending lumbar vein complex. This ascending venous system, which drains blood from the extraspinal venous plexi, is located on the lateral aspect of the vertebral body at the level of the neural foramen. The iliolumbar vein is frequently tethered and is at risk for being avulsed during dissection of the neurovascular and soft tissues structures of the anterior lumbar spine. Complete iliolumbar vein disruption can result in copious bleeding that cannot be controlled with direct pressure or topical hemostatic agents. Furthermore, the use of excessive cautery in this region carries an unacceptable risk of hypogastric plexus injury. Therefore, special attention is necessary when identifying the iliolumbar vein during exposure and prior to vena cava or iliac vein mobilization. When the iliolumbar vein is identified, it is recommended to suture ligate and divide it, thereby effectively eliminating risk of avulsion or laceration.

#### 28.6.2 Retrograde Ejaculation

In 1965, the first description of retrograde ejaculation as a consequence of anterior lumbar surgery was reported. Since then, the true incidence of retrograde ejaculation has been debated with the reported incidence ranging from 0.42 to 22 %. Many have argued that postoperative retrograde ejaculation is frequently under reported contributing to a lack of consensus as to its actual incidence. Retrograde ejaculation occurs in males as a potential complication unique to anterior lumbar exposure to the L5-S1 disk space. Retrograde ejaculation is believed to be due to damage to the superior hypogastric plexus of the sympathetic system located ventral to the L5 and S1 vertebral bodies. This plexus is formed by contributions from the paramedian lumbar sympathetic chains with bilateral damage causing ejaculatory disturbances. Small nerves directly over the interspace control a valve mechanism that causes semen to be expelled with sympathetic activity during sexual intercourse. Dissection over the disk space may injure the nerves, thereby disabling coordinated innervation of the valve. As a result, ejaculate travels into the bladder rather than out the urethra. While many patients describe that the sensation of ejaculation is unchanged, in order to reproduce,

special harvesting techniques must be utilized or preoperative banking of sperm is necessary. While the incidence of retrograde ejaculation is still unclear, complete or partial resolution appears to occur in approximately one-third of patients over 1-2 years.

Several factors may increase the risk for postoperative retrograde ejaculation. Surgeon inexperience, use of monopolar cautery, and laparoscopic or transperitoneal approach have been associated with increased risk [25, 26]. In a study of 215 laparoscopic ALIF performed with a threaded titanium cylindrical cage, a 5.1 % incidence of retrograde ejaculation was observed [27]. In a similar study of laparoscopic ALIF with threaded bone dowels or titanium cages, 15.9 % of male patients developed retrograde ejaculation [28]. Regarding transperitoneal versus retroperitoneal approach, in a survey of 20 surgeons with 15-20 years experience and a total of 4,500 anterior lumbar fusions, researchers found that incidence of retrograde ejaculation was not related to approach [29]. The authors suggested that careful surgical technique, proper visualization of the nerves prior to mobilization off the disk space, and avoiding electrocautery in the area are more relevant towards decreasing the risk of retrograde ejaculation. Additional recommendations include that incision of the posterior peritoneal plane along the right side of the aorta and right common iliac artery is helpful as the left-sided sympathetic nerves are commonly dominant. Subsequent mobilization of this flap with the plexus from right to left has been associated with decreased incidence of inadvertent injury to the hypogastric plexus. Injection of saline solution into the dorsal peritoneal tissue to gently develop the plane is also a useful technique for elevating the peritoneum from the nerves.

# 28.6.3 Ilioinguinal/Iliohypogastric Injury

Injury to the ilioinguinal and the lateral branch of the iliohypogastric nerves can occur as they course over the iliac crest, although this complication has rarely been reported after exposure for ALIF. Patients with ilioinguinal or iliohypogastric injury present with numbness or paresthesias affecting the medial upper thigh. Of reported cases of ilioinguinal or iliohypogastric injury after ALIF, most had complete resolution of symptoms within 6 months.

# 28.6.4 Sympathetic Plexus Injury

The lumbar sympathetic chain courses along the lateral aspect of the vertebral body and is at risk for injury with anterior lumbar exposure. Characteristically, patients with sympathetic chain injury complain of a cold foot on the side contralateral to the approach. This paradoxical effect is due to loss of sympathetic tone with increased vasodilation in the foot ipsilateral to the approach, providing the sensation that the contralateral foot is cold. Regardless, patients who report decreased temperature in either limb after anterior lumbar surgery necessitate evaluation of distal pulses for the possibility of arterial compromise. Ultimately, sympathetic nerve injury can be an unavoidable surgery-related complication after anterior-based approaches. Fortunately, for most patients there are no significant long-term sequelae, although a minority may suffer from prolonged dysesthesias.

#### 28.6.5 Lymphocele

Postsurgical lymphoceles can occur at sites of surgical disruption of lymphatic circulation. While postsurgical lymphocele is a common complication of thoracic, pelvic, or groin surgery, they rarely occur after abdominal exposures for spine surgery, despite an extensive lymphatic network surrounding the abdominal aorta. Particular concern for lymphatic disruption, however, may be at the L5–S1 level where there may be greater risk of lymphatic violation around the iliac arteries and veins. The uncommon experience of lymphocele after abdominal approaches may be secondary to reabsorption of extravasated lymphatic fluid by the large surface area of the peritoneal cavity during transperitoneal exposure. By

contrast, during a retroperitoneal approach where the peritoneal sac is maintained, lymphatic fluid may become sequestered in the retroperitoneal space. When postsurgical lymphocele is suspected with a retroperitoneal fluid collection, the differential diagnosis includes abscess, ureteral injury, pancreatic injury with pseudocyst formation, and CSF leak with pseudomeningocele [30]. Of note, in the rare instances of reported postsurgical lymphocele, they frequently were undetected at the time of the primary surgical procedure.

# 28.6.6 Revision Anterior Lumbar Surgery

With the increase in popularity of anterior lumbar surgery, there has consequently been a rise in anterior revision procedures. While complications associated with revision anterior surgery have been described, the incidence of complications for revision procedures is unclear. After prior surgery, scar tissue and adhesions to vascular and visceral structures pose risk for inadvertent injury during repeat surgery. The risks associated with anterior revision surgery and interbody device removal are suspected to be three to five times higher than for the initial surgery [31, 32]. Particularly, complication rates are higher for revision surgery at a previously operated level, compared to revision surgery at an adjacent or higher level.

# 28.6.7 Adjacent Segment Degeneration

A primary complication of any spinal fusion procedure is pseudarthrosis. Fusion rates for ALIF are generally high with many reporting 90–95 % arthrodesis, with better success for single-level procedures and nonsmokers. High fusion rates after ALIF are not surprising given that ALIF facilitates effective endplate preparation, allows for placement of a large interbody graft, and loads the graft in compression which further increases bone healing. Nonunion is more likely to occur in smokers, patients with failed prior surgery, multi-level fusions, and patients who have been previously treated with radiation. It is also important to note that radiographic evidence of fusion is not necessarily required for successful clinical outcome, as a stable, fibrous nonunion may also sufficiently provide enough stiffness to alleviate symptoms.

A potentially more relevant concern after ALIF is the development of adjacent segment degeneration at levels above or below a successful fusion. Biomechanical studies have demonstrated that intradiscal pressures are abnormally elevated at levels adjacent to a simulated fusion in cadaveric specimens. Animal studies have shown that increased facet loading and facet motion occurs at levels adjacent to a motion segment that has been rigidly fixated. Increased pressure or asymmetric and localized stresses within the disk can affect changes in the proteoglycans and ultimately result in increased risk of disk degeneration and prolapse.

Interestingly, adjacent segment changes may be more likely to occur after an anterior lumbar fusion compared to a posterior fusion. Cadaveric spines fixated with screws and wires in an experimental model of anterior lumbar fusion demonstrated increased motion at the superior adjacent segment. Using methyl methacrylate to simulate anterior and posterior fusion, Esses et al. showed that anterior fusion resulted in greater increase in motion at adjacent levels than posterior fusion and was similar to the effect seen with a simulated circumferential fusion.

One issue that remains to be elucidated is the effect of parallel versus lordotic cages on the development of adjacent segment degeneration. Fixation of a lumbar motion segment in increasing degrees of kyphosis is known to increase posterior column loading and laminar strain at the superior adjacent level. Kyphotic malalignment of a fused segment causes compensatory hyperlordosis at the rostral segment which in turn leads to contractures of the posterior ligamentous complex at that level. Alternatively, placement of a tapered or lordotic cage may similarly result in compensatory mechanisms at the adjacent levels. In an animal model of ALIF with paired tapered cages, a significant increase in intervertebral motion and intradiscal pressure was observed at the adjacent segments with flexion loading suggesting an attempt to compensate for the lordosis created at the fused segment [33].

#### Conclusion

Anterior lumbar interbody fusion is an important staple in the armamentarium of spine surgeons. The benefits of an anterior lumbar approach include broad exposure of the disk space allowing for thorough diskectomy, effective endplate preparation, appropriate sizing and placement of an optimal interbody graft, and anterior fixation if necessary. ALIF also has the advantage of protecting the neural elements and surrounding dura and, as a result, is a particularly attractive option in patients with prior posterior surgery. With the placement of broad interbody grafts or devices under compressive loads, ALIF imparts high fusion rates and provides an extremely rigid construct when supplemented with posterior spinal fixation. Therefore, ALIF is a potentially useful complement to fusion constructs that are under large biomechanical stresses such as in deformity surgery or complex reconstructive procedures. While there are potentially significant risks associated with an anterior approach, a variety of techniques are available and may be adopted by appropriately trained spine surgeons or performed with the assistance of experienced vascular, urologic, or general surgeons. Ultimately, decision to perform ALIF as opposed to other spinal fusion techniques is dictated by the patient's underlying pathology, anatomy, and associated medical comorbidities paired with the surgeon's preference, education, experience, and overall expertise.

## References

- Cunningham BW, Polly Jr DW. The use of interbody cage devices for spinal deformity: a biomechanical perspective. Clin Orthop Relat Res. 2002;394:73–83.
- 2. Barrick WT, Schofferman JA, Reynolds JB, Goldthwaite ND, McKeehen M, Keaney D, White

AH. Anterior lumbar fusion improves discogenic pain at levels of prior posterolateral fusion. Spine (Phila Pa 1976). 2000;25:853–7.

- Zdeblick TA, David SM. A prospective comparison of surgical approach for anterior L4-L5 fusion: laparoscopic versus mini anterior lumbar interbody fusion. Spine (Phila Pa 1976). 2000;25:2682–7.
- Vazquez RM, Gireesan GT. Balloon-assisted endoscopic retroperitoneal gasless (BERG) technique for anterior lumbar interbody fusion (ALIF). Surg Endosc. 2003;17:268–72.
- Gazzeri R, Tamorri M, Galarza M, Faiola A, Gazzeri G. Balloon-assisted endoscopic retroperitoneal gasless approach (BERG) for lumbar interbody fusion: is it a valid alternative to the laparoscopic approach? Minim Invasive Neurosurg. 2007;50:150–4.
- Lekovic GP, Han PP, Kenny KJ, Dickman CA. Bone dowels in anterior lumbar interbody fusion. J Spinal Disord Tech. 2007;20:374–9.
- Barnes B, Rodts GE, McLaughlin MR, Haid Jr RW. Threaded cortical bone dowels for lumbar interbody fusion: over 1-year mean follow up in 28 patients. J Neurosurg. 2001;95:1–4.
- Burkus JK, Sandhu HS, Gornet MF. Influence of rhBMP-2 on the healing patterns associated with allograft interbody constructs in comparison with autograft. Spine (Phila Pa 1976). 2006;31:775–81.
- Beutler WJ, Peppelman Jr WC. Anterior lumbar fusion with paired BAK standard and paired BAK Proximity cages: subsidence incidence, subsidence factors, and clinical outcome. Spine J. 2003;3:289–93.
- Sasso RC, Kitchel SH, Dawson EG. A prospective, randomized controlled clinical trial of anterior lumbar interbody fusion using a titanium cylindrical threaded fusion device. Spine (Phila Pa 1976). 2004;29:113– 22; discussion 121–2.
- Spruit M, Falk RG, Beckmann L, Steffen T, Castelein RM. The in vitro stabilising effect of polyetheretherketone cages versus a titanium cage of similar design for anterior lumbar interbody fusion. Eur Spine J. 2005;14:752–8.
- Burkus JK, Schuler TC, Gornet MF, Zdeblick TA. Anterior lumbar interbody fusion for the management of chronic lower back pain: current strategies and concepts. Orthop Clin N Am. 2004;35: 25–32.
- Holte DC, O'Brien JP, Renton P. Anterior lumbar fusion using a hybrid interbody graft. A preliminary radiographic report. Eur Spine J. 1994;3:32–8.
- Anjarwalla NK, Morcom RK, Fraser RD. Supplementary stabilization with anterior lumbar intervertebral fusion–a radiologic review. Spine (Phila Pa 1976). 2006;31:1281–7.
- Best NM, Sasso RC. Efficacy of translaminar facet screw fixation in circumferential interbody fusions as compared to pedicle screw fixation. J Spinal Disord Tech. 2006;19:98–103.

- Glazer PA, Colliou O, Lotz JC, Bradford DS. Biomechanical analysis of lumbosacral fixation. Spine (Phila Pa 1976). 1996;21:1211–22.
- Kuzhupilly RR, Lieberman IH, McLain RF, Valdevit A, Kambic H, Richmond BJ. In vitro stability of FRA spacers with integrated crossed screws for anterior lumbar interbody fusion. Spine (Phila Pa 1976). 2002;27:923–8.
- 18. Gerber M, Crawford NR, Chamberlain RH, Fifield MS, LeHuec JC, Dickman CA. Biomechanical assessment of anterior lumbar interbody fusion with an anterior lumbosacral fixation screw-plate: comparison to stand-alone anterior lumbar interbody fusion and anterior lumbar interbody fusion with pedicle screws in an unstable human cadaver model. Spine (Phila Pa 1976). 2006;31:762–8.
- Beaubien BP, Derincek A, Lew WD, Wood KB. In vitro, biomechanical comparison of an anterior lumbar interbody fusion with an anteriorly placed, lowprofile lumbar plate and posteriorly placed pedicle screws or translaminar screws. Spine (Phila Pa 1976). 2005;30:1846–51.
- 20. Arlet V, Jiang L, Steffen T, Ouellet J, Reindl R, Aebi M. Harvesting local cylinder autograft from adjacent vertebral body for anterior lumbar interbody fusion: surgical technique, operative feasibility and preliminary clinical results. Eur Spine J. 2006;15:1352–9.
- Burkus JK. Bone morphogenetic proteins in anterior lumbar interbody fusion: old techniques and new technologies. Invited submission from the Joint Section Meeting on Disorders of the Spine and Peripheral Nerves, March 2004. J Neurosurg Spine. 2004;1:254–60.
- Burkus JK, Gornet MF, Dickman CA, Zdeblick TA. Anterior lumbar interbody fusion using rhBMP-2 with tapered interbody cages. J Spinal Disord Tech. 2002;15:337–49.
- 23. Pradhan BB, Bae HW, Dawson EG, Patel VV, Delamarter RB. Graft resorption with the use of bone morphogenetic protein: lessons from anterior lumbar interbody fusion using femoral ring allografts and recombinant human bone morphogenetic protein-2. Spine (Phila Pa 1976). 2006;31:E277–84.
- 24. Fantini GA, Pappou IP, Girardi FP, Sandhu HS, Cammisa Jr FP. Major vascular injury during anterior lumbar spinal surgery: incidence, risk factors, and management. Spine (Phila Pa 1976). 2007;32: 2751–8.
- 25. Escobar E, Transfeldt E, Garvey T, Ogilvie J, Graber J, Schultz L. Video-assisted versus open anterior lumbar spine fusion surgery: a comparison of four techniques and complications in 135 patients. Spine (Phila Pa 1976). 2003;28:729–32.
- Sasso RC, Kenneth Burkus J, LeHuec JC. Retrograde ejaculation after anterior lumbar interbody fusion: transperitoneal versus retroperitoneal exposure. Spine (Phila Pa 1976). 2003;28:1023–6.

- Regan JJ, Yuan H, McAfee PC. Laparoscopic fusion of the lumbar spine: minimally invasive spine surgery. A prospective multicenter study evaluating open and laparoscopic lumbar fusion. Spine (Phila Pa 1976). 1999;24:402–11.
- Kleeman TJ, Michael Ahn U, Clutterbuck WB, Campbell CJ, Talbot-Kleeman A. Laparoscopic anterior lumbar interbody fusion at L4-L5: an anatomic evaluation and approach classification. Spine (Phila Pa 1976). 2002;27:1390–5.
- Flynn JC, Price CT. Sexual complications of anterior fusion of the lumbar spine. Spine (Phila Pa 1976). 1984;9:489–92.
- 30. Patel AA, Spiker WR, Daubs MD, Brodke DS, Cheng I, Glasgow RE. Retroperitoneal lymphocele

after anterior spinal surgery. Spine (Phila Pa 1976). 2008;33:E648-52.

- Gumbs AA, Hanan S, Yue JJ, Shah RV, Sumpio B. Revision open anterior approaches for spine procedures. Spine J. 2007;7:280–5.
- 32. Schwender JD, Casnellie MT, Perra JH, Transfeldt EE, Pinto MR, Denis F, Garvey TA, Polly DW, Mehbod AA, Dykes DC, Winter RB, Wroblewski JM. Perioperative complications in revision anterior lumbar spine surgery: incidence and risk factors. Spine (Phila Pa 1976). 2009;34:87–90.
- Rao RD, David KS, Wang M. Biomechanical changes at adjacent segments following anterior lumbar interbody fusion using tapered cages. Spine (Phila Pa 1976). 2005;30:2772–6.

# Interbody Fusion Through the Transpsoas Approach

29

Lee A. Tan, Manish K. Kasliwal, and Richard G. Fessler

# 29.1 Introduction

Spinal fusion has been used extensively in the management of various lumbar spine pathologies such as tumors, spinal instability, deformity, and stenosis. There has been a tremendous advancement in minimally invasive spine surgery (MISS) techniques that can help avoid the morbidity of traditional open anterior or posterior surgery while achieving the same clinical and functional outcome. The advantages of minimally invasive surgery including less tissue trauma during the surgical approach, less postoperative pain, shorter hospital stays, and faster return to activities of daily living have been very appealing for both patients and surgeons alike [1]. The transpsoas approach for lateral lumbar interbody fusion (XLIF (extreme lateral interbody fusion): NuVasive, Inc., San Diego, CA ARIA: Stryker, Inc., Kalamazoo, MI, COUGAR: Depuy Spine, Inc., Raynham, MA Ravine: K2M, Inc., Leesburg, VA DLIF (direct lateral interbody fusion): Medtronic, Inc., Minneapolis, MN Transcontinental: Globus Medical Inc.,

L.A. Tan, MD • M.K. Kasliwal, MD, MCh (🖂)

R.G. Fessler, MD, PhD

Department of Neurosurgery,

RUSH University Medical Center,

1725 W Harrison Street, Suite 855,

Chicago, IL 60612, USA e-mail: Manish\_Kasliwal@rush.edu

Audubon, PA) was developed as an alternative to the traditional anterior approach [2]. This minimally invasive technique can be used to gain access to the lumbar spine via a lateral approach that passes through the retroperitoneal fat and psoas major muscle (Fig. 29.1). It provides a less invasive access to the anterior and lateral aspect of the lumbar spine while minimizing potential complications such as postoperative ileus, bowel and vascular injury, and retrograde ejaculation associated with the traditional anterior approach. Moreover, this approach can be accomplished without the need of access surgeons. In addition, the preservation of the anterior longitudinal ligament and posterior tension band offers additional stability compared to traditional anterior or posterior approaches.

The minimally invasive transpsoas approach to the lumbar spine was first described by Pimenta et al. [3] and Bergey et al. [4] using the endoscope in 2001 and 2004, respectively, and possibly evolved from the initial endoscopic minimally invasive laparoscopic procedures described by McAfee and Fedder in the 1990s [5, 6]. It was subsequently described by Ozgur et al. [2] using the microscope and expandable tubular retractors in 2006 for lumbar interbody fusion. Over the recent years, the transpsoas approach has gained tremendous popularity, and its application has been broadened from diskectomy and interbody fusion to treatment of vertebral fractures, tumors, and spinal deformities [7].

© Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_29



**Fig. 29.1** Illustration demonstrating the trajectory of transpsoas approach (Image reproduced with permission from Medtronic, Inc: Medtronic Sofamor Danek; 2004. This spinal system incorporates technology developed by Gary K. Michelson, MD)

# 29.2 Indications

The transpsoas approach provides a relatively easy and safe surgical corridor to the anterior and lateral aspect of the vertebral column and disk space without the need for retraction of the peritoneum or mobilization of the great vessels. The lateral corridor provided allows the spine surgeon to place a relatively large graft more anteriorly, thereby facilitating the restoration of disk height and lumbar lordosis while proving good axial support. Indirect foraminal decompression and unbuckling of posterior longitudinal ligament can be achieved through restoration of disk height. Additional stabilization can be provided using lateral plate or percutaneous posterior instrumentation (Fig. 29.2). A recent biomechanical study by Fogel et al. [8] found that the laterally inserted cage alone can provide significant reduced range of motion in flexion-extension; additional stabilization with lateral plate, posterior spinous process plate, or bilateral pedicle screws can provide additional stability in lateral bending and axial rotation, but the exact type of fixation used did not make a statistically significant difference in stability. The original indication for lateral interbody fusion delineated by Ozgur et al. was for patients with low back pain associated with degenerative disk disease but without severe central canal stenosis. Subsequently, the transpsoas approach was increasingly performed for degenerative disk disease or adjacent segment disease where interbody fusion is desired, most commonly from L1 to L5 [9–12]. With the increased familiarity with the transpsoas approach and advancement in spinal instrumentation technology such as expandable cages, the application of the lateral approach has been broadened to include treatment of vertebral fractures, tumors, and spinal deformities [7]. Transpsoas approach has also been used for lumbar disk replacement with favorable results [13]. Table 29.1 summarizes the indications suitable for the transpsoas approach. Relative contraindications to this technique may include vascular abnormalities precluding access, significant spondylolisthesis, previous retroperitoneal surgery, and severely collapsed disk spaces.

# 29.3 Anatomy

The anatomical structures relevant to the transpsoas approach include the external and internal oblique muscles, transversus abdominis muscle, transversalis fascia, retroperitoneal fat, quadratus lumborum, psoas muscle, and the lumbar plexus [12]. A thorough understanding of local anatomy is essential for optimal surgical outcome and complication avoidance. The external oblique, internal oblique, and transversalis muscles are three layers of muscle that form the lateral abdominal wall. The transversalis fascia is immediately deep to the transversus abdominis muscle, covering the retroperitoneal fat. The retroperitoneal fat has a characteristic yellow appearance that serves as a helpful indicator of the presence of retroperitoneal space. In the retroperitoneal



**Fig. 29.2** (**a**, **c**) Preoperative radiograph of a patient with L3–L4 degenerative disk disease with asymmetric disk collapse and resultant far lateral foraminal stenosis; (**b**, **d**) postoperative X-rays of the same patient after L3–L4

transpsoas lumbar interbody fusion and posterior transfacet screws demonstrating restoration of disk height and indirect foraminal decompression

space, the quadratus lumborum muscle originates from the last rib and the transverse processes of the upper lumbar vertebrae and inserts to the internal lip of the iliac crest. The psoas muscle is situated anterior to the quadratus lumborum; it originates from the transverse processes and lateral aspect of the lumbar vertebrae and joins the iliacus muscle inferiorly and insert into the lesser trochanter of the femur. The transverse process and quadratus lumborum muscle are good anatomical landmarks for the posterior boarder of the psoas muscle.

The lumbar plexus is formed by the lumbar nerve roots with minor contribution from the T12 root (Fig. 29.3a). It travels between superficial and deep parts of the psoas major muscle. Major branches of the lumbar plexus include the

Tal	ble	29.	1	Ind	icati	ions	for	the	trans	psoas	ap	pro	ac	h
-----	-----	-----	---	-----	-------	------	-----	-----	-------	-------	----	-----	----	---

Transpsoas approach for lumbar interbody fusion
Degenerative disk disease
Low-grade spondylolisthesis (grade I or II)
Adjacent segment disease
Foraminal stenosis from collapsed disk without need for posterior decompression
Degenerative scoliosis
Transpsoas approach for total disk replacement
Degenerative disk disease
Transpsoas approach for corpectomies
Burst fracture
Tumor
Deformity

iliohypogastric, ilioinguinal, genitofemoral, lateral femoral cutaneous, obturator, and femoral nerves in addition to the lumbosacral trunk. The superior part of the plexus is usually located in the posterior one-fourth of the L1 to L2 vertebral body and travels progressively more anteriorly as the lumbar plexus descends. With the exception of the genitofemoral nerve, majority of the lumbar plexus branches are located in the posterior half of the L1 to L4 vertebral body, which makes the anterior half of the L2 to L4 vertebrae the optimal surgical corridor for the transpsoas approach [11]. Uribe et al. [12] and Moro et al. [11] in a cadaveric study nicely demonstrated the lumbar plexus anatomy in relation to the transpsoas approach and divided the area between the anterior and posterior edges of the vertebral body into four zones, Zone I (anterior quarter), Zone II (middle anterior quarter), Zone III (middle posterior quarter), and Zone IV (posterior quarter), as shown in Fig. 29.3b. The safe zone to prevent direct nerve injury from L1-L2 to L3-L4 was located at the middle posterior quarter of the VB (midpoint of Zone III), and the safe anatomical



**Fig. 29.3** (a) Illustration demonstrating lumbar plexus anatomy; and (b) "safe zones" for transpsoas approach (Image reproduced with permission from Medtronic, Inc: Medtronic Sofamor Danek; 2004)

zone at the L4-L5 disk space was at the midpoint of the VB (Zone II-Zone III demarcation). The genitofemoral nerve is formed by the L1 and L2 nerve roots and assumes a more anterior course (Zone II at the L2–L3 space and in Zone I at the lower lumbar levels L3–L4 and L4–L5) than the rest of the lumbar plexus, which predisposes itself for injury during transpsoas approach especially at L3 and below which may result in pain and paresthesias in the medial thigh and scrotal area. Apart from within the psoas muscle, there is potential risk of injury to the ilioinguinal, iliohypogastric, and lateral femoral cutaneous nerves in the retroperitoneal space where these nerves run along the posterior abdominal wall and then course obliquely, inferiorly, and anteriorly across the surgical corridor within the abdominal muscles to reach the iliac crest and the abdominal wall and it should be kept in mind [14]. Transpsoas approach to the L4-L5 disk space may be associated with higher risk of injury to the lumbar plexus and can often be limited by a high-riding iliac crest. Flexing the operating table or resection of the iliac crest can help to obtain lateral access in the setting of high-riding iliac crest. Transpsoas approach to the L1-L2 disk space may be limited by a low-lying 12th rib, which may require rib resection or intercostal approach. Though attempted in cadaveric studies with additional surgical maneuvers, L5-S1 is typically not amenable to transpsoas approach without significant neurological complications due to the traversing lumbar plexus at this level [15].

# 29.4 Operative Technique

#### 29.4.1 Preoperative Planning

Thorough review of preoperative imaging studies and careful surgical planning are imperative to any surgical procedure, but they are especially important in the transpsoas approach. The anatomy of the psoas muscle, spinal curve, adjacent vessels (aorta, vena cava, iliac vessels, etc.), as well as the location of the iliac crest and the 12th rib should be carefully studied to ensure the intended level can be safely reached from the lateral approach. Any prior abdominal or retroperitoneal surgery should be noted since scarring may complicate the dissection from that particular side. Though it is the surgeon's preference, it may be advantageous to approach the spine from concavity of the curve if multiple levels need to be treated as they can be reached through a single skin incision. When a single level is treated, or when there is a significant rotatory scoliosis, approaching from the convexity side may provide shorter working distance to the disk space and a more open disk space. In general, the anterior half of the disk space should be targeted (Fig. 29.4a); in the setting of lowgrade spondylolisthesis, the inferior vertebral body should be used as reference. Patients with high-grade spondylolisthesis and severe deformity have dramatically higher risk for complications and alternative approaches should be considered.

#### 29.4.2 Neuromonitoring Setup

Real-time EMG monitoring of the lumbar plexus and nerve roots is paramount to ensure safe passage of the tubular retractors during psoas dissection [16]. Nerve stimulation probe should also be set up to facilitate safe muscle dissection and tubular retractor placement. Clear communication with the anesthesia team is also important to avoid using any long-acting paralytics during induction and/or other agents that can interfere with EMG monitoring during the procedure. The dilators for this approached are designed in such a way that they can be integrated with EMGstimulating capabilities unidirectionally, with an isolated stimulating surface on the dilator. Using triggered electromyography (tEMG), as the dilator is rotated within the psoas muscle, stimulating areas are localized circumferentially which are very helpful in predicting the position of motor nerves. In general, tEMG thresholds for response below 5 mA indicate direct contact, between 5 and 10 mA indicate close proximity, and 11 mA or more indicate farther proximity from intrapsoas nerves [16].



**Fig. 29.4** (a) Intraoperative radiograph demonstrating using K-wire to target the anterior half of the disk space; (b) illustration demonstrating using K-wires to mark the



skin incision (Image reproduced with permission from Medtronic, Inc: Medtronic Sofamor Danek; 2004)



#### 29.4.3 Patient Positioning

**Fig. 29.5** Intraoperative photo demonstrating patient positioning. Notice the iliac crest is positioned over the break of the operating table

(arrow)

The patient is placed on a radiolucent, articulating table in true lateral decubitus position with the sagittal plane of the patient being 90° to the floor. The patient's iliac crest should be at the break of the articulating table (Fig. 29.5). The bottom leg is kept straight with a slight bend at the knee; the top leg is flexed to relax the ipsilateral psoas muscle to facilitate dilation. All pressure points should be well padded. The patient is then secured to the operating table with a 3 in. tape with or without a bean bag. The table is then flexed to increase the angle between the iliac crest and rib cage to facilitate access to the disk space, which can be particularly important when the intended level is the L4–L5 disk space and the iliac crest is in the way. All EMG leads should be checked to ensure proper connection.

# 29.4.4 Fluoroscopic Localization

The C-arm is brought into the field and an AP radiograph is first taken to ensure the patient is in a true lateral position. Two K-wires can then be

used as radiomarkers to localize the intended disk space with one K-wire in the middle of the intended disk space and the other one bisecting the anterior and posterior halves of the vertebral body (Fig. 29.4b). A 1.5 in. horizontal skin incision is then marked. When multiple levels are treated, a single vertical or oblique incision may be used with the incision centered at the middle of the intervening vertebrae.

#### 29.4.5 Surgical Technique

The patient is prepped and draped in the standard fashion. The previously marked skin incision is opened and dissection is carried through the subcutaneous fat layer to expose the external oblique fascia. Subsequently, a scissor should be used to make a fascial incision along the orientation of the muscle fibers. A Kelly clamp is then used for blunt dissection through the muscle layers of the lateral abdominal wall (external oblique, internal oblique, and transversalis abdominis muscles). The transversalis facia is located just deep to the transversalis abdominis muscle and a blunt opening is made to access the retroperitoneal space. The characteristic yellow color of the retroperitoneal fat confirms entry into the retroperitoneal cavity. A blunt dissection is then performed to palpate and visually expose the psoas muscle just deep to the retroperitoneal fat and the quadratus lumborum muscle. The transverse process marks the posterior boarder of the psoas muscle. Finger palpation is also used to ensure the peritoneal contents are displaced anterior to avoid peritoneal injury during docking of the tubular retractors.

A neural stimulation probe with 8 mA stimulation is carefully guided down and inserted into the psoas muscle, with it centered at the anterior half of the intended disk space (Fig. 29.6a). Lateral X-rays are used to confirm its proper location. If any neural structure is identified during insertion, the probe should be repositioned slightly anterior to avoid the traversing nerve. After the probe is properly placed, AP and lateral X-rays are obtained to confirm probe location. A K-wire is then placed through the probe cannula into the intended disk space. AP and lateral X-rays are again taken to confirm proper location of the guidewire. The initial dilator is then inserted over the K-wire. Special attention should be paid to any EMG changes during dilation at this time. EMG changes may indicate nerve compression and may require reposition of the dilators. After the final tubular dilator is placed, retractor blades with proper depth are inserted and secured. The retractor is then expanded if necessary to allow adequate access to the disk place. Stability shims or pins are then inserted to secure retractors in place. AP and lateral radiographs are again used to confirm proper retractor placement. As the most common neurological postoperative complications are transient motor weakness/palsy and sensory dysesthesia, which can sometimes be permanent, a supra-psoas shallow-docking method has been described which might be a safer alternative and may help minimize morbidities by eliminating or reducing direct psoas injury [17]. The annulus is then incised and a complete diskectomy is carried in the usual fashion similar to the anterior and posterior approaches. Once the diskectomy is completed, a Cobb is inserted into the disk space and a mallet is used to release the contralateral annulus and/or osteophyte on the contralateral side using fluoroscopic guidance (Fig. 29.6b). Gentle tapping should be used with a sudden loss of resistance a confirmation for adequate release. The end plates are then carefully prepared and freed of any residue cartilage or disk material.

The disk space is then distracted with progressively larger trial spacers to obtain the desired disk height and to achieve indirect foraminal decompression. Once desired height is determined and confirmed by radiographs, the interbody spacer is then placed under fluoroscopic guidance. Care should be taken to insert an adequate size implant so as to cover the apophyseal ring on both the sides to reduce the chances of subsidence. An optional lateral plating system may be used at this time if posterior instrumentation is not planned. The retractor system is then removed with attention to hemostasis. The retraction time of the psoas muscle should be recorded and minimized to minimize nerve compression



**Fig. 29.6** (a) Illustration demonstrating using finger to guide the stimulation probe through the psoas muscle. (b) Intraoperative X-rays demonstrating releasing of the

and stretch. The external oblique fascia layer is closed with interrupted sutures carefully so as not to include the nerves traversing the muscle in the stitch and the skin closed in a standard fashion.

# 29.5 Potential Complications

Despite the numerous advantages the transpsoas approach offers, it has its own sets of complications. Postoperative sensory disturbances including transient thigh pain and paresthesias have been reported in up to 63 % of patients, though majority of these symptoms do resolve within 1 year [18, 19]. Postoperative motor deficit such as iliopsoas weakness has been reported ranging from 0.7 to 33.6 %, which also improves over 2–3 weeks and is likely due to dissection through

contralateral annulus using a Cobb (Image reproduced with permission from Medtronic, Inc: Medtronic Sofamor Danek; 2004)

the psoas muscle and placement of the tubular retractors [20]. Many surgeons would not even consider transient iliopsoas weakness as a complication as it is more or less an expected outcome following a transpsoas approach. Other more serious and permanent complications such as lumbar plexus or nerve root injuries are reported to occur in up 3.4 % of patients [20]. Postoperative abdominal paresis (pseudohernia) has also been reported to occur in 4.2 % of patients [21]. Understanding the potential risk of injury to the ilioinguinal, iliohypogastric, and lateral femoral cutaneous nerves not in the retroperitoneal space where they travel obliquely, inferiorly, and anteriorly to reach the iliac crest and the abdominal wall may help avoid this complication. Devastating but rare complications such as bowel, kidney, ureter, or vascular injuries
can also occur, and timely intraoperative consultations with general surgery or vascular surgery colleagues are essential to minimizing morbidity and optimize patient outcome [22]. Understanding the regional anatomy and careful review of all preoperative radiology is key to avoiding these complications. A detailed and thorough discussion regarding these potential complications with the patient should be done preoperatively.

#### Conclusion

In summary, the transpsoas approach offers many advantages and provides a minimally invasive surgical corridor to the anterior and lateral thoracolumbar spine. However, like with any surgical approach, proper patient selection and sound surgical indication in addition to meticulous surgical technique are keys for good surgical outcomes. The spine surgeons should understand the limitations and potential complications with the transpsoas approach and have a frank preoperative discussion with patients. The transpsoas approach is a valuable and important tool for spine surgeons, in addition to the anterior and posterior approaches. It can provide excellent outcome in appropriately selected patients with the right indications.

#### References

- Smith ZA, Fessler RG. Paradigm changes in spine surgery—evolution of minimally invasive techniques. Nat Rev Neurol. 2012. doi:10.1038/nrneurol.2012.110. Published Online First: 19 June.
- Ozgur BM, Aryan HE, Pimenta L, et al. Extreme Lateral Interbody Fusion (XLIF): a novel surgical technique for anterior lumbar interbody fusion. Spine J. 2006;6:435–43. doi:10.1016/j.spinee.2005.08.012.
- Pimenta L. Lateral endoscopic transpsoas retroperitoneal approach for lumbar spine surgery. In: The 8th Brazilian Spine Society Meeting. Belo Horizonte, Minas Gerais, 2001.
- Bergey DL, Villavicencio AT, Goldstein T, et al. Endoscopic lateral transpsoas approach to the lumbar spine. Spine. 2004;29:1681–8.
- Regan JJ, Yuan H, McAfee PC. Laparoscopic fusion of the lumbar spine: minimally invasive spine surgery. A prospective multicenter study evaluating open and laparoscopic lumbar fusion. Spine. 1999;24: 402–11.

- McAfee PC, Regan JJ, Geis WP, et al. Minimally invasive anterior retroperitoneal approach to the lumbar spine. Emphasis on the lateral BAK. Spine. 1998;23:1476–84.
- Arnold P, McGuire R, Anderson K. The lateral transpsoas approach to the lumbar and thoracic spine: a review. Surg Neurol Int. 2012;3:198. doi:10.4103/2152-7806.98583.
- Fogel GR, Parikh RD, Ryu SI, et al. Biomechanics of lateral lumbar interbody fusion constructs with lateral and posterior plate fixation: laboratory investigation. J Neurosurg Spine. 2014;20:291–7. doi:10.3171/2013.11.SPINE13617.
- Benglis DM, Vanni S, Levi AD. An anatomical study of the lumbosacral plexus as related to the minimally invasive transposa approach to the lumbar spine: laboratory investigation. J Neurosurg Spine. 2009;10:139–44. doi:10.3171/2008.10.SPI08479.
- Kepler CK, Bogner EA, Herzog RJ, et al. Anatomy of the psoas muscle and lumbar plexus with respect to the surgical approach for lateral transpoas interbody fusion. Eur Spine J. 2010;20:550–6. doi:10.1007/ s00586-010-1593-5.
- Moro T, Kikuchi S, Konno S, et al. An anatomic study of the lumbar plexus with respect to retroperitoneal endoscopic surgery. Spine. 2003;28:423–8. doi:10.1097/01.BRS.0000049226.87064.3B; discussion 427–8.
- Uribe JS, Arredondo N, Dakwar E, et al. Defining the safe working zones using the minimally invasive lateral retroperitoneal transpsoas approach: an anatomical study: laboratory investigation. J Neurosurg Spine. 2010;13:260–6. doi:10.3171/2010.3.SPINE09766.
- Pimenta L, Oliveira L, Schaffa T, et al. Lumbar total disc replacement from an extreme lateral approach: clinical experience with a minimum of 2 years' follow-up: clinical article. J Neurosurg Spine. 2011;14:38–45. doi:10.3171/2010.9.SPINE09865.
- 14. Dakwar E, Vale FL, Uribe JS. Trajectory of the main sensory and motor branches of the lumbar plexus outside the psoas muscle related to the lateral retroperitoneal transpsoas approach: laboratory investigation. J Neurosurg Spine. 2011;14:290–5. doi:10.3171/2010.10.SPINE10395.
- Fontes RBV, Traynelis VC. Iliac crest osteotomy to enhance exposure of the L4–5 interspace in minimally invasive lateral transpoas interbody fusion: a cadaveric feasibility study. J Neurosurg Spine. 2013;18:13– 7. doi:10.3171/2012.10.SPINE12311.
- Uribe JS, Vale FL, Dakwar E. Electromyographic monitoring and its anatomical implications in minimally invasive spine surgery. Spine. 2010;35:S368– 74. doi:10.1097/BRS.0b013e3182027976.
- Acosta FL, Drazin D, Liu JC. Supra-psoas shallow docking in lateral interbody fusion. Neurosurgery. 2013;73:ons48–52. doi:10.1227/NEU.0b013e3182 88a202.
- Cummock MD, Vanni S, Levi AD, et al. An analysis of postoperative thigh symptoms after minimally invasive transpsoas lumbar interbody fusion: clinical

article. J Neurosurg Spine. 2011;15:11–8. doi:10.317 1/2011.2.SPINE10374.

- Fontes RBV, Traynelis VC. Editorial: transpsoas approach and complications. J Neurosurg Spine. 2011;15:9–10. doi:10.3171/2010.12.SPINE10741.
- 20. Ahmadian A, Deukmedjian AR, Abel N, et al. Analysis of lumbar plexopathies and nerve injury after lateral retroperitoneal transpsoas approach: diagnostic standardization: a review. J Neurosurg Spine. 2013;18:289–97. doi:10.3171/2012.11.SPINE12755.
- Dakwar E, Le TV, Baaj AA, et al. Abdominal wall paresis as a complication of minimally invasive lateral transpoas interbody fusion. Neurosurg Focus. 2011;31:E18. doi:10.3171/2011.7.FOCUS11164.
- 22. Tormenti MJ, Maserati MB, Bonfield CM, et al. Complications and radiographic correction in adult scoliosis following combined transpsoas extreme lateral interbody fusion and posterior pedicle screw instrumentation. Neurosurg FOCUS. 2010;28:E7. doi:10.3171/2010.1.FOCUS09263.

# **Oblique Lumbar Interbody Fusion**

## Clément Silvestre and Pierre Roussouly

## 30.1 Introduction

With the increasing number of spinal fusion performed [1-3], spine surgeons need to be acquainted with a variety of fusion procedures. Lumbar interbody fusion has become a popular technique for treating spinal conditions such as spondylolisthesis, degenerative disk disease, recurrent disk herniation, pseudarthrosis, and spinal deformity. Thus, treatment strategies have evolved from posterior fusion alone or anterior fusion alone to 360° fusion. The latter two require an interbody fusion, either during the posterior approach (posterior lumbar interbody fusion or transforaminal lumbar interbody fusion) or during a second-stage procedure such as anterior lumbar interbody fusion (ALIF), oblique lumbar interbody fusion (OLIF), or direct lumbar interbody fusion (DLIF or XLIF®). These different

Service de chirurgie orthopédique,

Centre des Massues, 92 rue Edmond Locard, Lyon 69005, France e-mail: c.silvestre@cmcr-massues.com

P. Roussouly, MD Department of Orthopedic Surgery, Centre Médico-Chirurgical de Réadaptation des Massues, Lyon, France e-mail: p.roussouly@cmcr-massues.com

approaches to reach the interbody space have been described, as have their advantages and disadvantages. Anterior lumbar interbody fusion provides direct access to the disk with potential improvement of fusion rate but also carries the risk of injury to the iliac vessels, peritoneal content, and ureteral and autonomic nervous system [4]. The traditional anterior retroperitoneal approach can also result in pain, muscular atony, or herniation of the abdominal wall [5]. In an attempt to decrease the complications related to traditional exposures, various minimally invasive techniques have been developed to minimize the incidence of pain and abdominal wall atony or herniation after anterior lumbar interbody fusions [5–7]. Laparoscopic procedures have been proposed but are not widely used due to the steep learning curve, technical complexity, and limited visualization of the spine associated with the technique, as well as the absence of clear benefit over open procedures in terms of complication rate and outcome [8–11]. Conversely, mini-open techniques have gained wider acceptance among surgeons performing anterior lumbar interbody fusion because they allow direct access and visualization of intervertebral disks in order to achieve more complete diskectomy and theoretically a better fusion, while potentially decreasing morbidity [5–7, 11].

Mayer [12] described a minimally invasive anterior approach to the lumbar spine through a retroperitoneal access for L2–3 to L4–5 disks and

C. Silvestre, MD (🖂)

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_30

a transperitoneal access for L5-S1 disk, performed after prior posterior instrumentation and fusion. He presented his technique on 25 patients and observed solid anterior fusion for all patients with minimal blood loss and no evidence of technique-related complication. The technique involves a muscle-splitting approach through a 4-cm oblique skin incision parallel to the fibers of the external oblique abdominal muscle that is extended to 6 cm if exposure of two disks is required. Kaiser et al. [11] reported their experience for single- or two-level anterior interbody fusion using the technique described by Mayer [12] on 51 patients, showing 3.9 % and 17.6 % of intraoperative and immediate postoperative complication rates, respectively. They also suggested that the mini-open technique is associated with decreased incidence of retrograde ejaculation. Saraph et al. [13] compared the technique of Mayer [12] to the traditional anterior retroperitoneal approach for anterior interbody fusion. After a mean follow-up of 5.5 years, fusion rate and complication rate were similar between the two groups, but intraoperative blood loss, operation time, and postoperative back pain were decreased with the mini-open technique. Interestingly, there were three patients with postoperative weakness of abdominal muscles in the group undergoing traditional approach (n=33), as opposed to none in the mini-open group (n=23). Other mini-open anterior approaches to the lumbar spine have also been proposed [10, 14], but these techniques involve opening the rectus sheath and mobilizing the rectus abdominis muscle with theoretically increased potential for abdominal wall morbidity. However, these techniques are useful when a more direct anterior approach is required.

Recently, the authors have used a minimally invasive retroperitoneal anterior approach referred to by the authors as the oblique lumbar interbody fusion (OLIF) [15]. This approach is similar to that of Mayer [12] for anterior lumbar interbody fusion. The authors progressively slide from the Mayer approach, which was during the 1990s and beginning of the 2000 more aggressive than now with many comorbidities such as wall pain or atony. Those were due to a long and posterior incision carrying the risk to injure the troncular nerves for wall muscles. Initially, the results of the study concerning the comorbidities of minimal invasive OLIF were published [15], on a series of 179 patients and the potential complications associated with the technique. Now, our expertise has evolved and increased until more than 733 cases.

### 30.2 Materials and Methods

## 30.2.1 Minimally Invasive Oblique Lumbar Interbody Fusion

The patient is positioned in lateral decubitus and a radiograph is made in order to identify the intervertebral levels to approach. A 4-cm skin incision, centered on the spinal segment to expose, is made in the lateral abdominal region parallel to the fibers of the external oblique muscle (Fig. 30.1a). The incision is made perpendicular to the line joining the anterior superior iliac spine to the umbilicus at one third of the distance from the anterior superior iliac spine, similar to the McBurney incision. We can also define this point as a soft point of the muscular wall; indeed, as for the neck and C6, you can reach and touch the spine, sometimes feel the psoas, with a deep palpation. And this is possible even on fatty patient. The approach is usually carried on from the left side but can also be performed from the right side such as for right lumbar scoliosis. External oblique, internal oblique, and transverse abdominal muscles are then dissected along the direction of their fibers in this muscle-splitting approach (Fig. 30.1b). The retroperitoneal space is accessed by blunt dissection and the peritoneal content is mobilized anteriorly. The psoas muscle is identified. We must emphasize at this time of the procedure that the psoas must be reclined posteriorly while the sympathetic chain and the ureter are mobilized anteriorly. The safer way to manage this stage is to be in the sheath of the psoas. It is important to minimize as much as possible the retraction of the psoas in order to decrease the incidence of postoperative pain - in particular cruralgia - secondary to injury of the lumbar plexus or psoas fibers. Four Steinman



**Fig. 30.1** (a) A 4-cm skin incision (*solid arrow*) is made in the lateral abdominal region along the fibers of the external oblique muscle. The level of the L4–L5 disk (*dotted arrow*) was located using the C-arm. (b) External

oblique, internal oblique, and transverse abdominal muscles are dissected along the direction of their fibers. (c) The intervertebral disk is exposed using handheld retractors and Steinman pins. (d) Skin closure

pins are used to expose the intervertebral disk without having to ligate segmental vessels (Fig. 30.1c). A window of only about 1 cm made in the annulus fibrosis is required anterolaterally to perform the diskectomy and insert the cage. A radiograph is done to confirm the proper level before proceeding to interbody fusion. It provides you indication about your anteroposterior positioning in the disk and may be helpful during the diskectomy and insertion of the implant. Segmental vessels usually do not need to be ligated unless the vertebral body needs to be exposed. At L4–L5, the disk space can be obstructed by the iliolumbar vein, in which case it needs to be ligated.

Up to three disks can be approached using the same 4-cm incision through a "sliding window" technique without the need to extend the incision, by taking advantage of the mobility of the abdominal wall. The described minimally invasive technique is well suited for exposure of L2– L3 to L4–L5 disks, but rarely, L1–L2 and L5–S1 disks can also be exposed. Exposure of L1–L2 disk is limited by the chest cage and can be performed only in the presence of relatively horizontal and mobile floating ribs. As for L5–S1 disk, its access is limited by the iliac wing and by the need to mobilize iliac vessels.

After diskectomy (Fig. 30.2a), vertebral endplates are prepared in order to expose the subchondral bone. The cage is filled with bone graft and/or substitute (Fig. 30.2b) and inserted in a press-fit fashion into the exposed disk spaces that remained open after the posterior procedure (Fig. 30.2c). If required, autogenous iliac graft can also be harvested from the same incision.



Fig. 30.2 (a) Exposure of disk space. (b) Filling of banana-shaped PEEK cage using bone substitute. (c) Cage inserted into exposed disk space after endplate preparation

Abdominal muscle planes are closed sequentially and the skin is closed using subcutaneous and subcuticular sutures (Fig. 30.1d). Although the procedures were performed without magnifying loupes or surgical microscope, it can be used for improved vision. In addition, a headlight or better a small light inserted in the wound by the retractors can be useful especially in deepest or overweight patients.

## 30.3 Results

Here are described main results about the first 179 patients of our series [15].

The main results and main complications are still the same, even if we reach the top of the learning curve. Patients were aged  $54.1 \pm 10.6$  years (range: 14.9–77.4). There were 148 females and 31 males aged  $54.5 \pm 11.0$  years (range: 14.9–77.4) and  $52.2 \pm 8.7$  years (range: 27.2–67.7), respectively. There were 118 primary cases and 61 revision cases. There were few occurrences of revision after previous anterior approach. Diagnosis at time of surgery is shown in Table 30.1. Weight and BMI were, respectively,  $67.1 \pm 14.5$  kg (range: 35–116) and 24.8 ± 4.1 kg/m<sup>2</sup> (range: 15.6–38.6).

Four patients with scoliosis and one patient with L4–L5 degenerative spondylolisthesis had a right-sided approach.

Details of the levels approached with the respective operative blood loss, operative time, and length of hospital stay are provided in Table 30.2. The procedure was performed at

L1–L2 in 4, L2–L3 in 54, L3–L4 in 120, L4– L5 in 134, and L5–S1 in 6 patients.

The procedure was done at a single level for 31 %, two levels for 60 %, and three levels for 9 % of patients. Figure 30.3 shows radiographs of a patient with three-level OLIF at L2–L5, while Fig. 30.4 shows two different patients with L1–L3 and L4–S1 OLIF, demonstrating the potential use of the described technique for approaching L1–L2 and L5–S1 levels, respectively.

In three patients of this series, and very few of all the patients treated, the procedure was aborted for one level mainly due to the too narrow disk space and sometimes due to an iliolumbar vein

Table 30.1	Diagnosis	at time	of surgery

	Number of
Diagnosis	patients
Primary surgery	
Spinal deformity	65
Spondylolisthesis	32
Degenerative disk disease/facet arthrosis	19
Post-traumatic kyphosis	2
Revision surgery	
Pseudarthrosis	18
Adjacent segment disease	18
Spinal deformity or imbalance	13
Spinal stenosis/post-laminectomy syndrome	10
Spondylolisthesis	2

too important in regard to the disk L4–L5 whose mobilization would be too risky. Sometimes, approaching L2–L3 level was not possible due to a prominent rib cage, and only L3–L5 OLIF was performed.

Operative blood loss was  $99.5 \pm 254.0$  ml for all patients, averaging  $56.8 \pm 131.3$  ml per level. It was lowest for single-level approaches  $(53.9 \pm 78.3 \text{ ml})$  and highest for two-level approaches  $(124.1 \pm 319.1)$ . In 98 % of cases, operative blood loss was 400 ml or less.

As for operative time, it was  $53.8 \pm 18.7$  min for all patients with an average of  $32.5 \pm 13.2$  min per level. It was lowest for single-level surgery  $(42.4 \pm 16.8 \text{ min})$ , increasing to  $57.4 \pm 14.8 \text{ min}$ for two-level and  $70.3 \pm 26.4$  min for three-level approaches. The length of hospital stay was  $7.1 \pm 3.5$  days for all patients. It was similar for patients undergoing single-level  $(6.5 \pm 2.3 \text{ days})$ , two-level  $(7.5 \pm 4.0 \text{ days})$ , and three-level  $(6.7 \pm 3.4 \text{ days})$  procedures. However, some patients had to stay longer at the hospital while waiting for transfer in a rehabilitation center. Now, this length of hospital stay has decreased to 4–5 days. Maybe the length of stay could be less, but due to our health system policy, patient must stay four nights in the hospital.

The main complications are exposed in Table 30.3. The most common complication was incisional pain (2.2 %), followed by lower extremity symptoms from sympathetic chain

**Table 30.2** Levels approached with respective operative blood loss, operative time, and length of hospital stay

Approach	Number of patients	Operative blood loss (ml)	Operative time (min)	Length of hospital stay (days)
Single-level	55	53.9±78.3	42.4±16.8	6.5±2.3
L1-L2	1	150	50	4
L2–L3	5	$60.0 \pm 82.2$	44.0±17.1	8.2±1.9
L3-L4	7	41.4±35.2	37.9±16.0	6.1±2.6
L4-L5	43	53.0±83.5	42.7±17.4	6.4±2.3
Two-level	108	124.1±319.1	$57.4 \pm 14.8$	7.5±4.0
L1–L3	2	200.0±212.1	67.5±10.6	12.5±2.1
L2-L4	29	$104.5 \pm 104.5$	$58.3 \pm 14.9$	7.9±4.0
L2-L3, L4-L5	2	$500.0 \pm 707.1$	72.5±17.7	4.0±1.4
L3-L5	68	123.6±378.6	$55.7 \pm 15.0$	7.2±4.0
L4-S1	6	75.0±61.2	63.2±10.1	8.0±3.5
Three-level	16	93.8±106.3	$70.3 \pm 26.4$	6.7±3.4
L1-L4	1	200	75	16
L2–L5	15	86.7±106.0	70.0±27.3	6.1±2.3



**Fig. 30.3** Preoperative (a, b) and postoperative (c, d) radiographs of a 45-year-old female with degenerative scoliosis undergoing three-level OLIF, showing the pres-

ence of radiopaque markers of the interbody cages from L2 to L5 (*full arrows*)



Fig. 30.4 Postoperative radiographs of two different patients undergoing OLIF at L1–L3 (a, b) and L4–S1 (c, d)

Complication	Number of patients
Incisional pain	4
Lower extremity symptoms related to sympathetic chain injury	3
Neurological deficit	2
Iliac vein laceration and bilateral femoral deep venous thrombosis	1
Iliac vein laceration	1
Iliolumbar vein laceration	1
Pseudomembranous colitis	1
Ileus	1
Peritoneal laceration	1
Cerebrovascular accident	1
Postoperative peripheral ischemia in lower extremities	1
Ipsilateral transient psoas paresis	1
Ipsilateral transient groin numbness	1
Symptomatic pseudarthrosis requiring revision ALIF	1

Table 30.3Complications

injury (1.7 %). There was no occurrence of abdominal muscle weakness, nor herniation, nor retrograde ejaculation.

In the original series, there were two patients with neurological deficit after left-sided L3-L5 OLIF. There was one more patient if we consider the whole series (733 patients). The first patient had left L4 paresthesia and L3-L4 motor weakness (grade 4 strength) presumably due to nerve stretching from restoration of disk height. For this case, surgery was uneventful and postoperative imaging did not show any misplacement of the interbody cages. The neurological deficit remained stable, but she was diagnosed with pancreatic adenocarcinoma and died 4 months after the OLIF procedure. A second patient had right L4-L5 paresthesia and weakness (grade 0 strength), as well as grade 3 strength in right S1 postoperatively. Preoperatively, she already had weakness of her right lower extremity as a sequelae of poliomyelitis at a young age. CT scan showed prominent cage of 36 mm length at L3-L4 and L4-L5 compressing the dural sac contralaterally on the right side. She then underwent revision through the same incision with placement of shorter cages of 30 mm length at L3–L4 and L4–L5 but did not recover from her neurological injury. Since this incident, we made a new implant with a new shape, and we also changed the inserter in order to have a better central positioning of the implant instead of a too latero-oblique traject.

For the last patient who had a neurological injury, during the procedure, the approach performed was too posterior and the surgeon reclined the psoas anteriorly instead of posteriorly. Consequently, during the diskectomy, the L4 nerve root was injured. The patient suffered for a partial deficit that recovered after few months of rehabilitation.

One patient presented with ipsilateral weakness (grade 4 strength) of hip flexion after L3–L5 OLIF but recovered full strength after 15 days. Due to the transient nature of the weakness, it was attributed to local pain from the surgical approach (manipulation of abdominal and/or psoas muscles). Another patient undergoing L3–L5 OLIF had hypoesthesia at the upper medial aspect of the left thigh after surgery, which returned to normal as noted at the 9-month follow-up visit. It was presumed to be caused by stretching of the ilioinguinal nerve located between internal oblique and transverse abdominal muscles at L4– L5 level near the anterior part of the iliac crest.

Two patients sustained intraoperative iliac vein laceration that was repaired primarily with nonabsorbable sutures. One of these patients lost 100 cc intraoperatively and presented bilateral edema in lower extremities postoperatively due to deep femoral venous thrombosis requiring anticoagulation treatment. Another patient had an iliolumbar vein laceration leading to 1,000 ml blood loss that ceased after ligation. One patient decompensated from preexisting peripheral arterial disease and presented rest pain in both lower extremities postoperatively due to peripheral ischemia. He improved with nonsurgical treatment consisting of fluid repletion and aspirin. One patient sustained a left-sided cerebrovascular accident secondary to a patent foramen ovale associated with an aneurysm of the interatrial septum. He was treated with thrombolysis and had no residual deficit from his cerebrovascular accident.

#### 30.4 Discussion

We present the largest cohort to date in the literature pertaining to this approach. As opposed to other studies referring to this approach [11-13], the current study shows that the original technique can be modified in order to address three levels through a "sliding window" using the same 4-cm incision. It also shows that L1-L2 disk can be approached in selected cases for which floating ribs are relatively horizontal and mobile. As for L5–S1 disk, OLIF through a retroperitoneal approach was performed successfully in six patients, but it had to be aborted in one patient. In addition, one patient required revision of L5-S1 interbody fusion due to symptomatic pseudarthrosis after L4-S1 OLIF. Due to the technical complexity of approaching L5-S1 using the retroperitoneal OLIF technique secondary to the need to mobilize iliac vessels and to the presence of the iliac wing, the authors suggest that another approach such as the transperitoneal approach described by Mayer [12] be strongly considered when anterior fusion of L5-S1 is required.

Surprisingly, operative time  $(53.8 \pm 18.7 \text{ min})$ was markedly decreased in the current series when compared to previous reports [11-13]. The authors hypothesize that three factors could have contributed to that finding. First, no microscope was used, thereby decreasing the number of manipulations during surgery, especially when radiographs are needed. Second, all surgeries were performed through the same retroperitoneal approach while previous reports used a transperitoneal approach for L5-S1 disk. Lastly, fusion was performed using bone substitute only, without harvesting autogenous iliac crest bone graft, which can increase operative time. Although autogenous iliac crest bone graft can be harvested from the same incision, it was not performed in the current study in order to avoid donor site morbidity and because the authors believe that using bone substitute was sufficient to achieve adequate rate of fusion clinically, in the context that all patients had also been stabilized posteriorly using segmental instrumentation.

Overall, minimally invasive OLIF carries about the same risks (rate and type of complications) as in traditional anterior approaches [4]. In this series, the most common complications were incisional pain (2.2 % of patients) and lower extremity symptoms due to sympathetic chain injury (1.7 % of patients). Vascular injury (iliac or iliolumbar vein) occurred in three patients (1.7 %) and could be repaired successfully despite the small incision. There are many potential advantages related to the OLIF technique. First, because it is a muscle-splitting approach, the incidence of abdominal wall pain is decreased and it becomes easier to develop a "sliding window" to access multiple levels through a small incision. Second, the incision for the OLIF technique is more anterior to the traditional anterior approach and therefore spares the proximal nervous trunks innervating the abdominal muscles. Accordingly, there was no occurrence of abdominal wall atony or herniation in the current series. In addition, the OLIF technique requires only minimal posterior retraction of the psoas to insert the banana-shaped cage, thereby reducing the incidence of postoperative crural- or psoasrelated pain. Finally, the OLIF technique could decrease the length of hospital stay although it remains to be verified. In the present study, the mean length of hospital stay was only  $7.1 \pm 3.5$  days, but it has to be mentioned that some patients had to stay longer at the hospital while waiting for transfer in a rehabilitation center.

Based on the results, age, weight, BMI, and the number of levels approached were not associated with the occurrence of complications. Although the procedure has been performed on patients with weight and BMI of up to 116 kg and 38.6 kg/m<sup>2</sup>, it could be use safely in these cases. Moreover, the positioning of the patient on the table makes the approach easier due to the fact that bowels and the abdominal fat go anteriorly.

A banana-shaped cage was used in association with the OLIF procedure in an attempt to facilitate the insertion of the cage and to minimize the rate of neurological injury. With such cage, less posterior reclination of the psoas is needed to insert a cage, and the concavity of the cage decreases the risk of injury to the dural sac centrally. However, as shown in one case with a neurological deficit, the risk of injuring contralateral traversing nerve roots is still present and therefore underlines the need to adequately assess the position of the cage either visually or radiographically. In order to decrease the incidence of this complication, the authors recommend using cages of 30 mm length or shorter in the lumbosacral spine. In the first 179 results, the authors used a banana-shaped cage normally designed for a TLIF approach so the ancillary was not adapted for an OLIF approach. Now, with a new design of the cage and a review of the inserter, the cage is better positioned at the center of the disk space and the trajectory in the disk is strictly laterolateral and less oblique. Thus, the risk of contralateral lesion of the root has decreased, and using long cages as 36 mm or more is no more damageable.

All the techniques used to achieve an anterior fusion with a posterior fusion and instrumentation have some benefits and inconvenient. The surgeons have to choose the most appropriate technique, adapted to their skills and to the better strategy for the patient. Thus, Table 30.4 shows the advantage and inconvenient for all the techniques in order to choose the better strategy.

#### 30.5 Evolution

We have seen that OLIF procedure is a safe and mini-invasive procedure. With the consequences of the development of the technique and evolution of this application, the surgeons' skills improve too. Actually, with the same OLIF approach, we can perform a partial corporectomy as we tend to after posterior transpedicular osteotomy, in order to provide more stability of the spine and avoid

**Table 30.4** Main advantage and inconvenient of each technique

	OLIF	Direct LIF	TLIF
Operative time	Decreases the posterior procedure	Decreases the posterior procedure	Increases the posterior procedure
	Requires a second and mini-invasive stage	Requires a second stage	One stage circumferential fusion
Disk access	2 or 3 levels	1 or 2 levels	1 or 2 levels. Three levels are rare
Nerve roots injury	Almost zero	Needs neuromonitoring due to a high risk of damage	Radicular pain postoperatively is very common
Disk shape	Can be done in all shape of disk. Even in disks closed posteriorly	Can be done in all shape of disk. Even in disks closed posteriorly	Easier in disk posteriorly opened
Height disk	Can be done in all height of disk. Even in very height disk (>14 mm)	Can be done in all height of disk	Difficult when more than a 10- or 12-mm cage is required
L5–S1 access	Possible in selected cases	Impossible	Possible, but more difficult
L2–L3 access	Possible	Impossible	Possible, but more difficult
L1–L2 access	Possible in selected cases	Impossible	Possible in selected cases



**Fig. 30.5** Preoperative radiographs of one patient after lumbar PSO. (a) Postoperative radiographs of the patient after partial corporectomy on L4 by OLIF approach ( $\mathbf{b}$ ,  $\mathbf{c}$ )

rod breakage and pseudarthrosis. The procedure is strictly the same; the skin incision is not so long and could be stretched by the retractors. The four Steinman pins are inserted in the vertebra above and below the osteotomized vertebra. Next, diskectomies of the two disks above and below are done. Managing the vertebra's pedicular vessels could be required, because most of the time, a lot of fibrosis recover this vessels. Thus, their identification, dissection, and ligature may be sometimes difficult. Then, the vertebra is osteotomized with bone scissor. Calibration of the corporectomy cage is done and finally, the cage is inserted (Fig. 30.5).

Of course, this OLIF approach can be done for a stand-alone anterior fusion with a stabilized cage provided on the market or with an anterior instrumentation (Fig. 30.6).



**Fig. 30.6** Postoperative radiograph of one patient after anterior fusion alone by OLIF approach

#### Conclusion

The technique was used effectively and safely for up to three levels from L2 to L5 using a "sliding window" approach. The described technique is associated with a risk of complications that is similar to that reported for traditional anterior approaches, with the advantage of decreasing the risk of abdominal wall weakness or herniation. For selected cases, it can also be performed at L1–L2 and L5–S1, although another approach might be preferred at L5–S1 due to the risks associated with mobilization of the iliac vessels and to the presence of the iliac wing. Because of the new design of implants and development of this technique, the risk of contralateral compression of dural sac and nerve roots during insertion of interbody devices becomes quite rare and the procedure is now quite safe. Thus, OLIF can be applied to other procedures besides diskectomy and interbody fusion.

#### References

- Bederman SS, Kreder HJ, Weller I, et al. The who, what and when of surgery for the degenerative lumbar spine: a population-based study of surgeon factors, surgical procedures, recent trends and reoperation rates. Can J Surg. 2009;52:283–90.
- Deyo RA, Nachemson A, Mirza SK. Spinal-fusion surgery – the case for restraint. N Engl J Med. 2004;350:722–6.
- Weinstein JN, Lurie JD, Olson PR, et al. United States' trends and regional variations in lumbar spine surgery: 1992–2003. Spine. 2006;31:2707–14.
- Sasso RC, Best NM, Mummaneni PV, et al. Analysis of operative complications in a series of 471 anterior lumbar interbody fusion procedures. Spine. 2005;30:670–4.
- Mummaneni PV, Haid RW, Rodts GE. Lumbar interbody fusion: state-of-the-art technical advances. J Neurosurg Spine. 2004;1:24–30.
- Eck JC, Hodges S, Humphreys SC. Minimally invasive lumbar spinal fusion. J Am Acad Orthop Surg. 2007;15:321–9.
- Shen FH, Samartzis D, Khanna AJ, et al. Minimally invasive techniques for lumbar interbody fusions. Orthop Clin North Am. 2007;38:373–86.
- Regan JJ, Hansen Y, McAfee PC. Laparoscopic fusion of the spine: minimally invasive spine surgery: a prospective multicenter study evaluating open and laparoscopic lumbar fusion. Spine. 1999;24:402–11.
- Chung SK, Lee SH, Lim SR, et al. Comparative study of laparoscopic L5–S1 fusion versus open mini-ALIF, with a minimum 2-year follow-up. Eur Spine J. 2003;12:613–7.
- Zdeblick TA, David SM. A prospective comparison of surgical approach for anterior L4–5 fusion. Laparoscopic versus mini anterior lumbar interbody fusion. Spine. 2000;25:2682–7.
- Kaiser MG, Haid Jr RW, Subach BR, et al. Comparison of the mini-open versus laparoscopic approach for anterior lumbar interbody fusion: a retrospective review. Neurosurgery. 2002;51:97–105.
- Mayer HM. A new microsurgical technique for minimally invasive anterior lumbar interbody fusion. Spine. 1997;22:691–9.

- Saraph V, Lerch C, Walochnik N, et al. Comparison of conventional versus minimally invasive extraperitoneal approach for anterior lumbar interbody fusion. Eur Spine J. 2004;13:425–31.
- Brau SA. Mini-open approach to the spine for anterior lumbar interbody fusion: description of the procédure, results and complications. Spine J. 2002;2:216–23.
- Silvestre C, Mac Thiong JM, Hilmi R, Roussouly P. Minimally invasive anterior oblique lumbar interbody fusion (OLIF). Description of the technique and complications on 179 patients. Asian Spine J. 2012 Jun;6(2):89–97. doi: 10.4184/asj.2012.6.2.89. Epub 2012 May 31

# Instrumented PLIF in Lumbar Degenerative Spine: Principles, Indications, Technical Aspects, Results, Complications and Pitfalls

### Olivier Launay, Gilles Perrin, and Cédric Barrey

## 31.1 Introduction

Historically, lumbar fusion has been described as a treatment of symptomatic spondylolisthesis, degenerative scoliosis and spinal stenosis associated with instability [1–3]. Lumbar fusion is also performed after posterior decompressive procedure when evidence of preoperative lumbar spinal deformity or instability that could worsen after laminectomy alone exists [4].

Burns [5] reported the first case of lumbar interbody fusion in 1933. From an anterior approach (anterior lumbar interbody fusion, i.e. ALIF), he used an autogenous tibial peg to treat an adolescent with spondylolisthesis. The posterior lumbar interbody fusion (PLIF) procedure was first described in 1944 by Briggs and Milligan [6], who used laminectomy bone chips in the disk space as interbody

O. Launay, MD

G. Perrin, MD

Hôpital Neurologique Groupement Hospitalier Est Hospices Civils de Lyon, Neurochirurgie C Chirurgie du Rachis, Lyon, France e-mail: giperrin@numericable.fr

C. Barrey, MD, PhD (⊠) Department of Neurosurgery C and Spine Surgery, Neurological Hospital, University Claude Bernard Lyon 1, Lyon, France e-mail: c.barrey@wanadoo.fr graft. In 1946, Jaslow [7] modified the technique by positioning an excised portion of the spinous process within the intervertebral space. It was not until 1953 when Cloward [1] described his technique, which used impacted blocks of iliac crest autograft that the popularity of PLIF technique increased. The PLIF procedure was found to have substantially increased fusion rates, often in excess of 85 %. Despite controversy about the efficacy of lumbar interbody fusion, because of the introduction of pedicle screw fixation [8], some clinicians have continued to use this procedure as Lin [9], Branch [10] and Takeda [11]. Then, advances in bone physiology, biomechanics, and fusion techniques with synthetic interbody implants have renewed interest in posterior interbody fusion. The BAK cage, which is a perforated stainless steel cylinder and filled with local autologous bone graft, was developed by Bagby and Kulisch [12, 13]. The concept was to use two parallel implants interposed between the vertebral bodies, with distraction, that restored the disk space, and the compression of the implants against the subchondral bone produces immediate stability [14]. More recently, interbody cages have become popular and are now composed of a wide range of materials, such as titanium mesh, carbon fibre and polyether ether ketone (PEEK) [15]. Finally, the addition of pedicle screws increases the stability of the construct and has been reported to significantly increase the fusion rate of this procedure compared with stand-alone grafts [16, 17].

Service de Chirurgie du Rachis, Clinique de Fontaine Fontaine-les-Dijon, France e-mail: o.launay@hotmail.fr

### 31.2 Rationale

Damage and degeneration of the lumbar disk can be the result of ageing, activity and trauma. Therefore, the degradation of the disk matrix leads to loss of disk height with or without bulging of the intervertebral disk and distension of the ligaments that create segmental instability. Thus, constraints on facet joints are increased and that may cause deformation, hypertrophy or subluxation of the facet joints like in spondylolisthesis. Moreover, mechanical stress causes hypertrophy and fibrosis of the ligamentum flavum. All these processes, including decreased disk height, facet joint and ligamentum flavum hypertrophy and vertebral endplate osteophytosis, may result in central canal stenosis, and/or lateral recess stenosis and/or foraminal stenosis [18]. Moreover, spinal stenosis may be emphasised by congenital abnormalities, or disorder of postnatal development [19].

Nerve root and cauda equina compression may arise from the combination of prolapsed disk, vertebral bone lesions anteriorly, but also from degenerated facet joint and hypertrophied ligamentum flavum posteriorly. Most often, degenerative spondylolisthesis occurs at the fourth lumbar vertebrae in middle-age women. As a result of a slipping forward of the vertebrae, cauda equina and spinal nerve roots may be tightened between the edge behind the top of lower vertebrae and frontal edge of the lower part of upper lamina, also linked to the subluxation of the facet joints.

Imaging studies are indispensable for diagnostic evaluation and treatment planning in symptomatic patients. There are many morphometric methods for the description of the spinal canal. Such terms as absolute and relative spinal stenosis are defined by purely radiological criteria and lack any clinical correlation. Lumbar MRI is the standard procedure for the demonstration of stenosis and cauda equina compression. As reported in the literature, its sensitivity is 87–96 % and its specificity is 68–75 % [20]. Lumbar CT may be useful for the assessment of bone condition and potential osteoporosis with a view towards the planning of surgery. On the other hand, lumbar myelography with postmyelographic CT should now only be performed in exceptional cases. The main indications for this invasive study are the presence of metal implants in the lumbar spine that would make MRI uninterpretable because of artefacts [21]. In our practice, we perform routinely full spine radiograph in order to analyse sagittal balance and lumbar dynamic radiographs to explore segmental instability. Electrophysiological studies are mainly useful in that they can reveal potential differential diagnoses.

One of the major objectives of spinal fusion is to relieve pain arising from spinal structures by removing potentially pain-generating disk tissue and stabilising one or more motion segments. Various methods of posterior lumbar fusion (PLF) have long been used for this purpose. Interbody fusion procedures became more widely used for their stabilising effect on the spine segment and as the role of the lumbar disk as a pain generator became better appreciated. The primary concept behind lumbar interbody fusion is that by removing all or most of the disk and stabilising the operated segment with bone graft, the primary pain generator is removed. Stabilising the segment should then eliminate mechanical stimulation that may provoke symptoms and may avoid future problems associated with collapse of an unsupported space. In a biomechanical study, interbody fusion was found to be stiffer than posterior lumbar fusion [22]. In addition, the surface area between the host bone and the graft is much greater with interbody fusion than with intertransverse process fusion.

The two primary purposes of interbody fusion are to relieve pain and stabilise the symptomatic spine segment. In cases of disk-related pain, the symptom-related tissue is removed. However, the removal of this tissue may cause the disk space to collapse with a concomitant narrowing of the foramen and related changes of the facet joints, causing nerve root compression. By filling the disk space with bone graft, the disk space height is re-established. This may also increase the height of the foramen. The bone graft grows into the bone of the adjacent vertebra, fusing them into a single unit. This stabilising effect is particularly important in cases of pseudarthrosis, spondylolisthesis, spinal instability and postlaminectomy syndrome.

Evidence supports interbody fusion over posterior fusion alone in the treatment of lumbar disk-related pain. Weatherley [23] reported using discography to identify symptomatic disks at the level of a solid posterior fusion. More recently, successful outcome was reported for such patients with persistent symptoms despite a solid posterior fusion when symptomatic disks within the previously fused segment were treated with ALIF [24]. Results of a biomechanical study found that following simulated posterior fusion with pedicle screw fixation, the intradiscal pressure during spinal flexion was as great as that measured in the intact, nonoperated segment [25]. These studies provide biomechanical and clinical support for the need to use an interbody fusion technique to adequately address pain arising from the disk. ALIF and PLIF have been found to be effective in the treatment of diskrelated pain [26–31], particularly that associated with a chemically sensitised disk identified by discography [32]. Fusion not involving an interbody technique has yielded poor results for diskrelated pain [32-34]. The potential benefits of using cages in interbody fusion procedures are that they may increase the chances of achieving a successful fusion and they provide some immediate stability to the operated segment whilst the bone graft incorporates [35].

Several cages are designed to be implanted into the disk space using either the anterior or posterior approach. Based on the review of the literature, there is no general preference for the approach to be used. The decision regarding the type of approach should be made based on several factors, such as the sagittal balance, pathology present, spinal anatomy, patient's history of prior surgery (either approach may be more difficult if there is significant scarring from prior surgeries), vascular anatomy (and conditions that may make an anterior procedure more difficult, such as calcification of vessels) and the surgeon's individual training and experience.

The main challenge in the surgical treatment of lumbar degenerative spinal stenosis is to achieve adequate decompression of the neural structures without inducing iatrogenic instability, keeping or restoring a good lordosis and correcting or preventing spinal deformity. Sometimes nerve root decompression could be achieved only by restoration of the height of the intervertebral space and by a large opening of the lateral recesses and the foramen. Large bone resection may be indeed required. Decompression surgery for spinal stenosis due to degenerative changes producing claudication is successful in most patients. According to the literature the rate of further spinal instability is from 5 to 10 % and the risk of postoperative additional forward slip in degenerative spondylolisthesis is assessed between 10 and 18 % of the patients treated without fusion [36, 37]. Even if further horizontal dislocation did not lead to worse clinical results, it is logical for the surgical treatment not only to aim the most efficient decompression of the neurological structures by using adequate bone resection and restoration of the intervertebral height by the distractive interbody fusion, but also the second aim for surgery is to prevent postoperative destabilisation by using the same intervertebral fusion.

# 31.3 Indications

The principal indication for lumbar interbody fusion surgery is the stabilisation and fusion of adult spinal instability and/or deformity. Therefore, lumbar fusion has been described as a treatment of symptomatic spondylolisthesis, degenerative scoliosis and spinal stenosis associated with instability [1, 2, 9, 38]. For those with lumbar stenosis but without spondylolisthesis (deformity), the surgical management has traditionally involved posterior decompressive procedures, including laminectomy or laminotomy, and judicious use of partial medial facetectomies and foraminotomies, with or without diskectomy [39, 40]. In patients with evidence of spinal instability, however, in situ posterior lumbar fusion is recommended as a treatment option in addition to decompression in the setting of

lumbar stenosis [39]. Due to early surgical failures (the mean rate of poor outcome is 20 % in large series of laminectomies [36, 37]) and late deteriorations due to iatrogenic instability (5–18 %), restenosis (7 %) or disk herniation at adjacent spinal levels (10 %), careful selection of patients for fusion must be carried out by assessing radiological parameters that are associated with the greatest risk of postoperative destabilisation.

Secondary indications include recurrent lumbar disk herniation, where extensive bony removal is necessary for exposure of the disk fragments, lateral or massive disk herniations, failed previous lumbar fusions by other techniques and discogenic low back pain [38]. Because the cause of spinal pain is not completely understood and remains controversial, surgical efforts to treat such conditions also remain controversial [41]. The description of spinal pain is often referred to as "lumbar segmental instability" [42, 43] caused by degenerative disk disease [34] or facet joint syndrome [42, 44] when no signs of increased motion or spondylolisthesis exist [45].

As a consequence, the main parameters for indication of fusion after surgical decompression are (Fig. 31.1) as follows:

- Sagittal orientation of the facet joints
- Total facetectomy
- Lumbar stenosis associated with lumbar previous idiopathic scoliotic deformity
- Degenerative scoliosis
- Intracanal synovial cysts alone or associated with listhesis
- Flat back with loss of lordosis
- Degenerative spondylolisthesis
- Recurrent lumbar disk herniation
- Secondary displacement after failed previous decompressive surgery



Antero-posterior fusion by combined approach is an alternative for all these indications but requiring two surgical procedures

Fig. 31.1 Surgical indications of fusion in lumbar stenosis

### 31.4 Technical Aspects (Fig. 31.2)

To perform PLIF, patients are positioned in the prone on chest and iliac crests rolls in order to lower intra-abdominal pressure and improve venous drainage. Arms are placed on arm boards with abduction limited to 80° as to prevent brachial plexus injury.

A dorsal midline incision is made and subcutaneous tissues are dissected with monopolar until the deep fascia. This fascia is incised adjacent to the spinous processes bilaterally preserving supra-spinous ligament. Then the para-spinous muscles are released from the laminae in a subperiosteal fashion, and the dissection is taken out to the facets bilaterally until the transverse processes are visualised. Lateral radiographs should be obtained to confirm the operative levels prior to arthrodesis. Then, soft tissues should be removed on and around the lamina, pars, facet joint and dorsal transverse pro-



**Fig. 31.2** Ten main surgical steps for PLIF procedure: (1) Complete exposure of the posterior arches of the two adjacent vertebras, (2) bilateral facetectomy (inferior facets of upper vertebra and superior facets of lower vertebra), (3) insertion of pedicle screws, (4) laminotomy with control of the adjacent nerve roots (i.e. the two upper and the two lower roots), (5) complete diskectomy via bilateral approach, (6) intervertebral distraction through the disk space, (7) cleaning of the end plates using curettes and/or dedicated rasps, (8) insertion of lordotic peek cages filled with autologous bone graft, (9) contouring of the rod, (10) rod placement with compression performed along the rod between the screw heads

cesses including the facet capsule and intrafacet synovium. After all, the fixation with pedicle screws is realised prior to the decompression, therefore limiting the risk of neural and dura mater injury during screws insertion and reducing the timing with the canal opened (associated with potential epidural bleeding). In addition slight and gentle distraction between the screw head using appropriate distractor could facilitate the insertion of the interbody implants. For this procedure, a facetectomy is done, keeping the bone that will be morselised for future graft, and then pedicle screws are inserted with lateral radiograph control bilaterally for all interbody fusions. Laminotomy and foraminotomy can be performed as needed for neural decompression of the thecal sac and nerve roots.

In most cases, complete laminectomy is not necessary and only partial laminotomy of the

upper vertebra is sufficient to perform the decompression and to permit the insertion of the interbody cages. Epidural veins must be coagulated to avoid bleeding and cut to move apart neural elements without tether and discover disk space. Care should be taken to protect neural structures with nerve root retractor. Another cause of epidural bleeding is the emissary vein of the vertebral body, which can be plugged by haemostatic gauze. Complete diskectomy and endplate preparation are performed, also removing the cartilaginous end plates using rasps. Then, spacers are inserted in order to progressively distract the disk space and determine the adequate gauge implant size (Fig. 31.3). Morselised autogenous bone, obtained from the laminectomy, is packed anteriorly before the implants are placed. According to our experience, the dimensions of the cages have to be high



**Fig. 31.3** Main surgical steps of PLIF procedure with perioperative views. Control of the four adjacent nerve roots, i.e. right and left L4 and L5 roots for L4–L5 level, is crucial to avoid any damage to neurologic structures.

Intervertebral distraction on one side can be helpful to complete the decompression on the other side. *uf* upper facet, *ds* dural sac

enough (at least 10 mm) and large (25 mm) to obtain a good primary stabilisation and thus a good fusion. Also, wedge-shaped cages (8° lordotic at minimum) are superior to rectangular cages in restoring segmental lordosis and sagittal alignment and avoiding flat back deformity [46]. The cages, filled with autogenous bone (perfectly cleaned with removal of all soft tissues), are inserted into the disk space with the medial aspect on the pedicles bilaterally. Then pedicle screws and rods are compressed to restore segmental lordosis and promote fusion by graft compression. After haemostasis is ensured, the wound is irrigated and closed in layers. A subfascial drain may be left.

### 31.5 Advantages/Limitations

Unlike posterolateral intertransverse fusion, PLIF is a biomechanically optimal fusion because the graft and/or the interbody implant maintains the disk height (i.e. the lateral foraminal opening), protects the nerve roots, restores weight bearing to anterior structures and controls both horizontal and vertical instabilities. The cagelike implants (titanium or polyether ether ketone (PEEK) cages) meet the mechanical requirements for PLIF by serving both a mechanical function and a biologic bone growth function. The cages stretch the intervertebral space to its normal anatomic height and prevent the postoperative collapse of the graft. The implant is packed with cancellous bone graft obtained from the laminectomy [47]. PLIF and anterior lumbar interbody fusion (ALIF) with cages, without a complementary posterior fixation for 360° stabilisation, are associated with pseudo-arthrosis, secondary displacement and subsequent complications. The role of the pedicle screwbased posterior fixation is first to carry out temporary control of AP, lateral or rotational translation before the achievement of the definitive bone fusion, second to enhance osteogenesis and third to allow early mobilisation without the need of a postoperative corset to avoid external contention (except in case of osteoporosis), loss of lordosis and further destabilisation at the adjacent level to the arthrodesis.

PLIF is neither useful nor safer when reoperations are performed and in which the spinal canal was already opened. There exists an increased risk of dural breach and neural injury due to fibrosis and nerve root distortion. ALIF or TLIF may be a good alternative for these patients, thus avoiding the dissection in the region of the epidural fibrosis. Another drawback of this technique is the blood loss that can be excessive, particularly in older patients. Also, in patients with a high pelvic incidence, ALIF may be a better alternative. ALIF facilitates a good fusion and restores an optimal sagittal balance. This parameter is crucial to respect, because the L4-S1 segment represents two-third of the total lumbar lordosis. As a consequence, arthrodesis should be performed with these parameters in mind.

#### 31.6 Complications (Table 31.1)

Posterior lumbar interbody fusion provides circumferential release of the dural sac and/or nerve roots as well as a biomechanically stable construct with anterior and middle-column load sharing combined with pedicle screw devices. However, PLIF has some risks for surgical complications [48]. Along with risks related to the surgical approach, the use of implants increases

Table 31.1 Complications due to PLIF procedures

Perioperative complications	Late complications
Dural laceration, cerebrospinal	Subsidence rare
fluid (CSF) leakage: 4–17 %	Pseudarthrosis: 2–15 %
Neurological complications:	Cage migration: rare
Transient (radicular pain, weakness) 3–17 %	Adjacent segment disease (no specific to PLIF): 3–11 %
Permanent (radicular pain, weakness) 0–7.5 %	
Deep wound infection: 0.5–5 $\%$	
Hematoma: 1.2 %	
Pedicle screw misplacement: 4 %	
Injury to major abdominal	
vessels	
Pulmonary embolism: 0.4 %	

the risk for additional complications [49]. Complications are divided here into perioperative complications that occurred during and within 1 month of surgery and late complications after 1 month of surgery.

#### 31.6.1 Perioperative Complications

The incidence of perioperative complications following single-level PLIF has been reported to be 18-37.5 % [48, 50], and the incidence after two-level PLIF has been reported as 46 % [51]. Moreover, Deyo et al. found that patients who underwent lumbar surgery with fusion had a complication rate twice as high as those who underwent surgery without fusion [49]. Amongst several kinds of fusion techniques, PLIF is considered one of the most technically demanding procedures and a definite learning curve exists. One of the most dangerous manipulations in PLIF is excessive retraction of the dural sac with the cauda equine and nerve roots whilst removing disk material and inserting cages and bones. Nerves are often taut and immobile because of severe adhesion due to canal stenosis. Surgeons may unknowingly retract the dural sac beyond a critical pressure and/or period whilst concentrating on the disk space. Neurological deficits have been reported in only 2 % of patients after posterolateral lumbar fusion, in which access to the disk is not required [52]. Hosono et al. found that the surgery duration was the only significant risk factor for neurological complications and therefore suggested that the dural sac or roots should have been retracted for unusually long periods in patients presented with neurological deficits [49]. Also, the rate of neurological complications in procedures with total facetectomy is much lower than procedures with partial preservation of facet joints. It may reduce the intensity and period of retraction of the dural sac and nerve roots and the risks of neurological complications by taking advantages of the large working space provided by total excision of bilateral facet joints.

As a consequence, perioperative complications of PLIF procedures are as follows:

- Dural laceration, cerebrospinal fluid (CSF) leakage: 4–17 % [53, 54]
- Neurological complications Transient (radicular pain, weakness) 3–17 % [54, 55]
  - Permanent (radicular pain, weakness) 0–7.5 % [54, 55]
- Deep wound infection: 0.5–5 % [49, 53]
- Hematoma: 1.2 %
- Pedicle screw misplacement: 4 %
- Injury to major abdominal vessels [56]
- Pulmonary embolism: 0.4 %

#### 31.6.2 Late Complications

The intracorporeal penetration on the cages or subsidence, and thus the loss of the restored intervertebral height, is perhaps the most significant late complication. It mainly occurs in osteo-porotic patients, but remains rare – one patient in the authors' series [55].

Pseudarthrosis is an uncommon complication of PLIF – less than 2 % [54, 55].

Cage retropulsion after PLIF is another complication that has been described. The risk factors are insufficient cage size, multilevel fusion, inadequate seating of the cage anteriorly and surgery at segment L5/S1. Fundamental techniques in performing PLIF must be mastered as follows:

- The degenerated disk materials must be removed and the end plates cleaned from cartilaginous layers thoroughly.
- The cage must be inserted without damaging the bony end plates.
- Undersized cages should not be selected.
- Adequate compressive force must be applied to the disk space by the pedicle screws.
- Use of lordotic cages [57].

A prospective randomised study reported that fusion accelerates degenerative changes at the adjacent segment of the fused spine, compared with naturally occurring changes [58]. Spinal fusion alters the biomechanics of spinal motion and increases intradiscal pressure or the load on facet joints of the adjacent motion segment of the fused spine [59]. Within 5 years of lumbar fusion surgery, the clinical incidence of symptomatic adjacent segment disease (ASD) is reportedly 5.2-18.5 % [59] and the incidence of additional surgery for symptomatic ASD is reportedly 3–11 % [60, 61]. Moreover, the deterioration rate for repeat PLIF (44%) [62] is higher than that for initial PLIF (5.2-18.5 %) [59]. Biomechanical studies have demonstrated greater intradiscal pressure at the adjacent segment in double-level fusion than in single-level fusion [59]. This is one reason why repeat PLIF leads to higher incidence of ASD than the initial PLIF. Devo et al. [63] reported in their study of 31,543 patients with surgery for lumbar stenosis that prior spinal surgery was the strongest risk factor for repeat surgery and that the hazard ratio for this was 1.58. These results suggest that patients undergoing repeat PLIF for ASD would incur more risk factors for additional surgery than those undergoing single- or double-level PLIF at the initial surgery. Furthermore, age was reported to be a major risk factor for ASD [59, 60].

# 31.7 A Comparison of PLIF and TLIF

Interbody fusion techniques have been developed to preserve the load-bearing capacity of the spine, restore local lordosis and facilitate compressive loading onto interbody graft – all of which enhance the potential for fusion acquisition [64]. Lumbar interbody fusion with supplemental posterior pedicle screw fixation ("circumferential" fusion), based on biomechanical evaluation, stabilises all three columns of the spine and has been used routinely for the operative treatment of painful spinal disorders. PLIF, TLIF and ALIF approaches are the most frequently performed options and, when accompanied by posterior pedicle screw fixation, result in circumferential fusion. Each of the former procedures has advantages and drawbacks.

Posterolateral graft and fixation is easily added to the PLIF, further enhancing spinal stability and the induction of fusion.

Unfortunately, the PLIF is usually limited to use at levels below L3, because of the risk of damage to the conus medullaris and to the cauda equina that may result from bilateral root retraction here. The suggested modification of PLIF presented by Harms and Jeszenszky [65], the TLIF, is equivalent to the PLIF and is simpler and safe, and some believe superior in result. The technical advantages of the TLIF include avoidance of thecal sac and/or nerve root retraction injury, safe performance below L3 and a decrease in epidural bleeding and scarring [65–67]. Harms and Jeszenszky, in their presentation of the original TLIF procedure, as well as many other authors of biomechanical reports, have recommended additional posterior pedicle screw fixation to enhance stability.

The PLIF and TLIF are familiar to most spine surgeons and both require only a single approach. These two procedures have therefore recently become the most popularly used techniques to treat spinal disorders. They are associated with a few differences with regard to the actual surgical technique, however. The TLIF requires a complete unilateral facetectomy and spares the contralateral lamina, facets, and pars interarticularis. The PLIF procedure requires a bilateral laminotomy as well as partial, and at times complete, facetectomy to place an adequate interbody spacer device. The TLIF implants are usually semilunar and only one is implanted, whereas those used for PLIF are cubic or cylindrical in shape and are placed in pairs resulting in a greater surface of bone graft and better distribution of loads. With the PLIF procedure, a portion of the posterior longitudinal ligament (PLL) is cut to position the interbody space devices, whereas the TLIF procedure preserves most of the PLL [68].

On the other hand, the diskectomy and clearing of end plates performed during PLIF procedure

via bilateral approach are probably more complete and have better quality compared to TLIF.

From a biomechanical consideration perspective, Sim et al. showed that the PLIF provides a higher immediate stability than the TLIF, especially for the lateral bending motion. The implant position in the disk space, however, is not an important factor for the immediate stability of a single-level TLIF. If the TLIF implant is placed further anteriorly, although there were no statistically significant differences in this study, there is a tendency for this position to be more stable [68].

Another difference between PLIF and TLIF is that performing a TLIF at the L5–S1 segment is quite difficult due to the pelvic position that prevents the good positioning of the cage in the disk space.

## 31.8 Tips and Tricks

- During exposure, a goal should be to avoid dural tear and nerve roots injury that can result from the manipulation of instruments and also to reduce the amount of bleeding coming from the canal (epidural veins).
- During the procedure when the nerve roots are retracted to prepare the disk space, there is most often significant bleeding that arises from the emissary vein of the vertebral body. This may be difficult to stop. The most effective strategy is to clog the vein with

haemostatic gaze and thus to perform an embolisation of the vein.

- When preparing the disk space, the vertebral bodies must be maximally distracted in order to put the higher cage (10–12 mm of height in most cases). The distractor used should be 1 or 2 mm higher compared to the implant to facilitate the insertion of the cage on the contralateral side. It also facilitates the decompression of the nerve roots in the foramen and confers maximum stability to the spine thus avoiding the retropulsion of the cage.
- Morselised and perfectly cleaned bone is compressed into the cage, but the area of fusion is quite reduced. To enhance the chance of fusion, we also put morselised bone into the anterior disk space before inserting the cages.
- The five key points to restore lumbar lordosis during PLIF procedure are as follows:
  - 1. Patient positioning
  - 2. Use of lordotic cages
  - 3. Optimal size for the cages
  - 4. Optimal AP placement of the cages (placed at the anterior part of the disk space)
  - 5. Posterior inter-pedicular compression after cage insertion

#### 31.9 Clinical Cases

Clinical cases are illustrated in Figs. 31.4, 31.5, 31.6 and 31.7.



**Fig. 31.4** CT scan in sagittal and coronal views that demonstrate a solid fusion with remodelling of the bone graft between L4 and L5 after a PLIF procedure



**Fig. 31.5** A 54-year-old woman operated with an L4–L5 PLIF procedure because of a degenerative L4–L5 spondylolisthesis with an intra-canalicular synovial cyst



**Fig. 31.6** A 56-year-old man who underwent an operation for lumbar stenosis via an L3–L4 PLIF procedure that suffered postoperatively from left cruralgia. An emergency CT scan was performed, which demonstrated morselised bone graft located in the left recess (*red arrows*). The patient underwent immediate reoperation to decompress the nerve root and remove the bone graft that had migrated into the canal



**Fig. 31.7** CT scan with sagittal and axial views that demonstrate bilateral cage retropulsion into the recess (*red arrows*). The patient suffered from bilateral sciatica, which necessitated reoperation for cage repositioning

### References

- Cloward RB. The treatment of ruptured lumbar intervertebral discs by vertebral body fusion. Indications, operative technique, after care. J Neurosurg. 1953;10:154–68.
- Collis JS. Total disc replacement: a modified posterior lumbar interbody fusion. Report of 750 cases. Clin Orthop Relat Res. 1985;193:64–7.
- Cole CD, Mc Call TD, Schmidt MH, et al. Comparison of low back fusion techniques: transforaminal lumbar interbody fusion (TLIF) or posterior lumbar interbody fusion (PLIF) approaches. Curr Rev Musculoskelet Med. 2009;2:118–26.
- Resnick DK, Choudhri TF, Dailey AT, et al. Guidelines for the performance of fusion procedures for degenerative disease of the lumbar spine. Part 1: introduction and methodology. J Neurosurg Spine. 2055;2:637–8.
- Burns BH. An operation for spondylolisthesis. Lancet. 1933;1:1233.
- Briggs H, Milligan P. Chip fusion of the low back following exploration of the spinal canal. J Bone Joint Surg. 1944;26:125–30.
- Jaslow I. Intracorporeal bone graft in spinal fusion after disc removal. Surg Gynecol Obstet. 1946;82: 215–22.
- Roy-Camille R, Saillant G, Mazel C. Internal fixation of the lumbar spine with pedicle screw plating. Clin Orthop. 1986;203:7–17.
- Lin PM. Posterior lumbar interbody fusion technique. Complications and pitfalls. Clin Orthop. 1985;193: 90–102.
- Branch Jr CL. The case of posterior lumbar interbody fusion. Clin Neurosurg. 1996;43:252–67.

- Takeda M. Experience in posterior lumbar interbody fusion. Unicortical versus bicortical autologous grafts. Clin Orthop. 1985;193:120–6.
- Butts M, Kuslick S, Bechtold J. Biomechanical analysis of a new method for spinal interbody fusion. Boston: American Society of Mechanical Engineers; 1987.
- 13. Bagby G. The Bagby and Kuslich (BAK) method of lumbar interbody fusion. Spine. 1999;24:1857.
- Bagby GW. Arthrodesis by the distractioncompression method using a stainless steel implant. Orthopedics. 1988;11:931–4.
- Brantigan JW, Steffee AD, Geiger JM. A carbon fiber implant to aid interbody lumbar fusion. Mecha Test Spine. 1991;16:S277–82.
- Steffee AD, Sitkowski DJ. Posterior lumbar interbody fusion and plates. Clin Orthop Relat Res. 1988;227:99–102.
- Brodke DS, Dick JC, Kunz DN, et al. Posterior lumbar interbody fusion. A biomechanical comparison, including a new threaded cage. Spine. 1997;22: 26–31.
- Genevay S, Atlas SJ. Lumbar spinal stenosis. Best Pract Res Rheumatol. 2010;24:253–65.
- Ciricillo SF, Weinstein PR. Lumbar spinal stenosis. West J Med. 1993;158(2):171–7.
- 20. Wassenaar M, Van Rijn RM, Van Tulder MW, et al. Magnetic resonance imaging for diagnosis lumbar spinal pathology in adult patients with low back pain or sciatica: a diagnostic systematic review. Eur Spine J. 2012;21:220–7.
- Kalff R, Ewald C, Waschke A, et al. Degenerative lumbar spinal stenosis in older people: current treatment options. Dtsch Arztebl Int. 2013;110(37): 613–24.
- Lee CK, Langrana NA. Lumbosacral spinal fusion: a biomechanical study. Spine. 1984;9:574–81.

- Weatherley CR, Prickett CF, O'Brein JP. Discogenic pain persisting despite solid posterior fusion. J Bone Joint Surg Br. 1986;68:142–3.
- Barrick WT, Schofferman JA, Reynolds JB, et al. Anterior lumbar fusion improves discogenic pain at levels of prior posterolateral fusion. Spine. 2000;25:853–7.
- Abe E, Nickel T, Buttermann GR, et al. Lumbar intradiscal pressure after posterolateral fusion and pedicle screw fixation. Tohoku J Exp Med. 1998;186: 243–53.
- Blumenthal SL, Baker J, Dossett A, et al. The role of anterior lumbar fusion for internal disc disruption. Spine. 1988;13:566–9.
- Gill K, Blumenthal SL. Functional results after anterior lumbar fusion at L5–S1 in patients with normal fusion at L5–S1 in patients with normal and abnormal MRI scans. Spine. 1992;17:940–2.
- Lee CK, Vessa P, Lee JK. Chronic disabling low back pain syndrome caused by internal disc derangements. The results of disc excision and posterior lumbar interbody fusion. Spine. 1995;20:356–61.
- Linson MA, Williams H. Anterior and combined anteroposterior fusion for lumbar disc pain. A preliminary study. Spine. 1991;16:143–5.
- Newman MH, Grinstead GL. Anterior lumbar interbody fusion for internal disc disruption. Spine. 1992; 17:831–3.
- Schechter NA, France MP, Lee CK. Painful internal disc derangements of the lumbosacral spine: discographic diagnosis and treatment by posterior lumbar interbody fusion. Orthopedics. 1991;14:447–51.
- Derby R, Howard MW, Grant JM, et al. The ability of pressure-controlled discography to predict surgical and nonsurgical outcomes. Spine. 1999;24:364–71.
- Parker LM, Murrell SE, Boden SD, et al. The outcome of posterolateral fusion in highly selected patients with discogenic low back pain. Spine. 1996;21:1909–16.
- Wetzel FT, LaRocca SH, Lowery GL, et al. The treatment of lumbar spinal pain syndromes diagnosed by discography. Lumbar arthrodesis. Spine. 1994;19:792–800.
- Blumenthal SL, Ohnmeiss DD. Intervertebral cages for degenerative spinal diseases. Spine J. 2003;3: 301–9.
- Johnsson KE, Redlund-Johnell I, Udén A, et al. Preoperative and postoperative instability in lumbar spinal stenosis. Spine. 1989;14:591–3.
- Johnsson KE, Willner S, Johnsson K. Postoperative instability after decompression for lumbar spinal stenosis. Spine. 1986;11:107–10.
- Mummaneni PV, Haid RW, Rodts GE. Lumbar interbody fusion: state-of-the-art technical advances. Invited submission from the Joint Section Meeting on Disorders of the Spine and Peripheral Nerves, March 2004. J Neurosurg Spine. 2004;1:24–30.
- Resnick DK, Choudhri TF, Dailey AT, et al. Guidelines for the performance of fusion procedures for degenerative disease of the lumbar spine. Part 9: fusion in patients with stenosis and spondylolisthesis. J Neurosurg Spine. 2005;2:679–85.

- 40. Resnick DK, Choudhri TF, Dailey AT, et al. Guidelines for the performance of fusion procedures for degenerative disease of the lumbar spine. Part 10: fusion following decompression in patients with stenosis without spondylolisthesis. J Neurosurg Spine. 2005;2:686–91.
- Zdeblick TA. The treatment of degenerative lumbar disorders. A critical review of the literature. Spine. 1995;20:126S–37.
- Esses SI, Botsford DJ, Kostuik JP. The role of external spinal skeletal fixation in the assessment of low-back disorders. Spine. 1989;14:594–601.
- Stokes IA, Frymoyer JW. Segmental motion and instability. Spine. 1987;12:688–91.
- Mooney V, Robertson J. The facet syndrome. Clin Orthop Relat Res. 1976;115:149–56.
- 45. Kuslich SD, Ulstrom CL, Michael CJ. The tissue origin of low back pain and sciatica: a report of pain response to tissue stimulation during operations on the lumbar spine using local anesthesia. Orthop Clin North Am. 1991;22:181–7.
- Gödde S, Fritsch E, Dienst M, et al. Influence of cage geometry on sagittal alignment in instrumented posterior lumbar interbody fusion. Spine. 2003;28(15):1693–9.
- 47. Perrin G. Surgical treatment of severe lateral and foraminal spine degenerative stenosis. In: Robert G, Marek S, editors. Lumbar spinal stenosis. Philadelphia: Lippincott Williams and Wilkins; 2000. p. 313–20.
- Hosono N, Nakameta M, Makino T, et al. Perioperative complications of posterior lumbar interbody fusion for nonisthmic spondylolisthesis: analysis of risk factors. J Neurosurg Spine. 2008;9:403–7. Clinical article.
- Deyo RA, Ciol MA, Cherkin DC, et al. Lumbar spinal fusion. A cohort study of complications, reoperations and resources use in the medicare population. Spine. 1993;18:1463–70.
- Okuda S, Miyauchi A, Oda T, et al. Surgical complications of posterior lumbar interbody fusion with total facetectomy in 251 patients. J Neurosurg Spine. 2006;4:304–9.
- Makino T, Hosono N, Mukai Y, et al. Perioperative complications of patients undergoing two-level posterior lumbar interbody fusion for degenerative lumbar diseases. Rinsho Shinkei Geka. 2008;43: 459–63.
- Carreon LY, Puno RM, Dimar II JR, et al. Perioperative complications of posterior lumbar decompression and arthrodesis in older adults. J Bone Joint Surg Am. 2000;85:2089–92.
- 53. Sakaura H, Yamashita T, Miwa T, et al. Outcomes of 2-level posterior lumbar interbody fusion for 2-level degenerative lumbar spondylolisthesis. J Neurosurg Spine. 2013;19:90–4. Clinical article.
- Mehta VA, McGirt MJ, Garcés Ambrossi GL, et al. Trans-foraminal versus posterior lumbar interbody fusion: comparison of surgical morbidity. Neurol Res. 2011;33(1):38–42.
- 55. Perrin G, Barrey C. Chapter 22. When should PLIF (intervertebral fusion) be done in lumbar canal

decompression for degenerative stenosis? In: Advanced concepts for lumbar degenerative diseases. Rio de Janeiro: Dilivros; 2010. p. 259–72.

- Postacchini R, Cinotti G, Postacchini F. Injury to major abdominal vessels during posterior lumbar interbody fusion. A case report and review of the literature. Spine J. 2013;13:7–13.
- Kimura H, Shikata J, Odate S, et al. Risk factors for cage retropulsion after posterior lumbar interbody fusion. Analysis of 1070 cases. Spine. 2012;37(13):1164–9.
- Ekman P, Moller H, Shalabi A, et al. A prospective randomized study on the long-term effect of lumbar fusion on adjacent disc degeneration. Eur Spine. 2009;18:1175–86.
- Park P, Garton HJ, Gala VC, et al. Adjacent segment disease after lumbar or lumbosacral fusion: review of the literature. Spine. 2004;29:1938–44.
- Cho KS, Kang SG, Yoo DS, et al. Risk factors and surgical treatment for symptomatic adjacent segment degeneration after lumbar spinal fusion. J Kor Neurosurg Soc. 2009;46:425–30.
- Sears WR, Sergides IG, Kazemi N, et al. Incidence and prevalence of surgery at segments adjacent to a previous posterior lumbar arthrodesis. Spine J. 2011; 11:11–20.

- 62. Miwa T, Sakura H, Yamashita T, et al. Surgical outcomes of additional posterior lumbar interbody fusion for adjacent segment disease after single-level lumbar interbody fusion. Eur Spine J. 2013;22(12):2864–8.
- Deyo RA, Martin BI, Kreuter W, et al. Revision surgery following operation for lumbar stenosis. J Bone Joint Surg Am. 2011;93:1979–86.
- McLaughlin MR, Haid Jr RW, Rodts GE, et al. Posterior lumbar interbody fusion: indications, techniques and results. Clin Neurosurg. 2000;47:514–27.
- Harms JG, Jeszenszky D. The unilateral, transforaminal approach for posterior lumbar interbody fusion. Orthop Traumatol. 1998;6:88–99.
- 66. Hee HT, Castro Jr FP, Madj ME, et al. Anterior/posterior lumbar fusion versus transforaminal lumbar interbody fusion: analysis of complications and predictive factors. J Spinal Disord. 2001;14:533–40.
- Humphreys SC, Hodges SD, Patxardhan AG, et al. Comparison of posterior and transforaminal approaches to lumbar interbody fusion. Spine. 2001;26:567–71.
- 68. Sim HB, Murovic JA, Cho BY, et al. Biomechanical comparison of single level posterior versus transforaminal lumbar interbody fusion with bilateral pedicle screw fixation: segmental stability and the effect on adjacent motion segments. J Neurosurg Spine. 2010;12:700–8.

Part VI

Minimally Invasive Techniques

32

# Degenerative Disk Disease: Stages of Degeneration, Low Back Pain, and Insights on Intradiskal Therapies

Jean-Louis Husson, Jean Lombard, and Florian Cueff

# 32.1 Stages of Degeneration and Low Back Pain (and Why We Need a New Classification System)

# 32.1.1 Spinal Degeneration Cascade

Although the spinal degeneration cascade proposed by Kirkaldy-Willis and Farfan [1] is still very relevant, it is incomplete. It does not capture the early lesions in some of the stabilizing disk and ligament structures from a purely histological point of view, despite being the subject of a significant body of research (cell culture, genomics, etc.), nor does it capture lesions that do not cause excessive joint motion.

J. Lombard, MD Department of Orthopedic Surgery, Centre Hospitalier de Niort, Niort 79000, France

Department of Orthopedic Surgery, Centre Hospitalo–Universitaire Pontchaillou, Rennes 35033 Cedex 9, France

Department of Orthopedic Surgery, Centre Hospitalier Privé des Côtes d'Armor, Centre Orthopedique Briochin, Plerin 22190, France e-mail: florian.qf@gmail.com

# 32.1.2 Acquired Degenerative Intervertebral Dysfunction (ADIVD)

To eliminate confusion surrounding the use of the term "instability," which only corresponds to stage II of the Kirkaldy-Willis cascade, we would like to propose a new term: *acquired degenerative intervertebral dysfunction (ADIVD)* [2]. This condition refers to a benign mechanical disorder of segmental vertebral function due to degeneration and is hence acquired. It takes into account all of the lesions in the stabilizing structures, with movement disorders characterized as being qualitative and/or quantitative in nature, whether to a greater or lesser extent.

# 32.1.3 New Four-Stage Classification for Progressive Spinal Degeneration

This new term allows us to represent every stage of the progressive loss of stability in the degenerative lumbar spine due to ADIVD (Fig. 32.1) [2]. This four-stage classification encompasses the same I, II, and III stages proposed by Kirkaldy-Willis and Farfan [1], but adds stage 0, where minimal dysfunction is present.

#### Stage 0: Minimal Dysfunction

This is the initial phase of elastic deformation due to the disk losing its viscoelastic properties. Lesions are only visible on histologic

© Springer-Verlag Berlin Heidelberg 2016 J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_32

J.-L. Husson, MD (🖂)

Department of Orthopedic Surgery, Centre Hospitalo–Universitaire Pontchaillou, Rennes 35033 Cedex 9, France e-mail: jeanlouis.husson@wanadoo.fr

F. Cueff, MD

examination and only rarely manifest themselves as acute lumbar immobilization.

Stage I: Minor Dysfunction

This is the intermediate phase of elastic deformation with pure loss of stability, marked by lower back pain and brief episodes of posterior facet joint locking, which can occur at one or multiple levels and cause referred pain. This corresponds to Maigne's notion of painful minor intervertebral dysfunction (PMID).

Stage II: Major Dysfunction

This corresponds to Kirkaldy-Willis' stage of instability. This is the advanced stage of elastic deformation, with a dynamic, progressive loss of stability. Radiological and clinical signs of dynamic stenosis appear, leading to lumbar and sciatic symptoms due to changes in the spinal canal volume without anatomical modifications. Any disk protrusion, posterior facet osteoarthritis, or retrolisthesis can then alter the volume of the nerve root canal during the stage of static-dynamic stenosis and trigger neurogenic claudication symptoms. Next, the disks undergo plastic deformation because of water loss. This results in permanent lateral stenosis, made worse by the consequences of the loss of stability (osteophytes, abnormal movements).

- In terms of functional signs, the only aspect that all authors agree on is the presence of mundane mechanical low back pain in combination with pseudoradicular pain, or Maigne's referred pain; true radicular pain is rare.
- There is no correlation between various clinical examination techniques and objective measurements. A comparative study of patients with suspected instability and patients in a control group was performed using a twist CT scan [3]. The findings disproved Graf's hypothesis [4] that posterior facet separation



**Fig. 32.1** Diagram capturing the progressive loss of stability due to acquired degenerative intervertebral dysfunction (*ADIVD*)

was a pathognomonic sign of instability, since separation was also observed in patients who were asymptomatic. Pain in the posterior joint structures during trunk rotation or when striking the heel to the ground is only one of the many clinical signs of instability. The duration of pain relief after peri- or intra-articular lidocaine or corticosteroid injection is shorter in cases of facet pain due to instability than in cases of pain due to facet joint arthritis.

Stage III: Maximal Dysfunction

This corresponds to the final degenerative phase, characterized by structures that become wedged together and restabilize the spine, or the appearance of Junghanns degenerative spondylolisthesis or rotational dislocation, which causes degenerative lumbar scoliosis in adults.

The clinical symptoms including pain can either stem from the bone, adjacent spinous processes rubbing together (Baastrup's disease), the posterior facet joints, or even be referred.

## 32.1.4 Need for Additional Examinations

The lack of specific symptoms, their multifactorial nature, and the lack of relationship between movement quantification and pain intensity make it challenging to understand the clinical picture of ADIVD. It also explains why the physician must turn to various types of additional tests [2]. Standard A/P and lateral and three fourth X-rays form the basis of the evaluation, along with dynamic views, which were first proposed by Nachemson in 1944. These were followed by many studies attempting to define segmental vertebral motion, quantify it, define standards, and as a consequence, get closer to the pathology using quantitative and hopefully reproducible data. Although these were all high-quality studies, there was no general agreement. However, White and Panjabi [5] were credited for showing that the vertebral unit had 6° of freedom. The twist CT scan is the only dynamic test with some evaluation potential, not necessarily by using true measurements, but by subjectively evaluating posterior facet joint separation in extreme positions of active rotation [3]. Magnetic resonance imaging (MRI) can reveal early signs of nucleus pulposus dehydration, which shows up as reduced T2 hypersignal. Modic underlines the lack of correlation between three levels of spinal cord signal intensity and degenerative disk disease and between clinical symptoms and anatomical disruptions [6, 7]. Pfirmann [8] subsequently proposed a treatment algorithm based on MRI classification of lumbar intervertebral disk degeneration.

## 32.2 Overview of Intradiskal Therapies

Conservative treatment encompasses several well-known nonsurgical and rehabilitation methods. Among the various intradiskal therapies currently being used to treat herniated lumbar disks and diskogenic lower back pain, we discuss the two we are most familiar with: radiofrequency (RF) ablation and nucleus pulposus implant.

# 32.2.1 Radiofrequency Intradiskal Techniques for Treating Herniated Lumbar Disks and Diskogenic Lower Back Pain

### 32.2.1.1 Introduction

Percutaneous intradiskal RF techniques are an integral part of the fairly complex treatment of low back pain or radicular diskogenic lumbar pain. These techniques came to the forefront when the chymopapain enzyme used for chemo-nucleolysis was discontinued. They were developed in parallel with nucleolysis techniques, which use chemical agents instead.

All RF techniques are performed through a minimally invasive transforaminal approach, the same approach used for a diskogram in Kambin's triangle [9, 10] under the outgoing nerve root. A thin catheter is used to deliver a variable dose of thermal energy to a specific part of the lumbar disk; this catheter is connected to a preprogrammed external RF generator that can deter-

mine the disk impedance, among other variables. The catheter is removed at the end of the procedure. This procedure, which is carried out in an ambulatory care setting under local anesthesia or with neurosedative agents, must be performed in the appropriate aseptic environment with fluoroscopy. Prophylactic antibiotics during the procedure are recommended.

The exact mechanism of action is not well understood. It is based on the principles of decompressing the disk and/or destroying the disk's peripheral neoinnervation and neovascularization, along with altering the disk's collagen.

The clinical outcomes for these procedures are a function of patient selection and the technical requirements of the procedure, which are very specific. These techniques are aimed at relatively young patients who have failed a full course of conservative treatment and physical therapy, prior to discussing surgical disk stabilization. Even if only temporarily effective, these radiofrequency techniques can be used as patient selection criteria for disk arthroplasty or interbody fusion, as the success of intradiskal treatment confirms that the pain originates in the disk. If not effective, another surgical procedure can be performed without any additional problems.

#### 32.2.1.2 History and Classification

In the mid-1970s, chemical nucleolysis was only performed with chymopapain. Since chymopapain was first introduced in 1963, its efficacy (70–80 % good and very good results) has been demonstrated in several randomized studies and compared with surgical treatment [11–13]. When the manufacture and sale of chymopapain were discontinued in 2001 for financial reasons, other percutaneous intradiskal therapies such as RF developed rapidly.

There are two main types of RF therapies:

- Those that target the nucleus pulposus, such as disk decompression (e.g., Nucleoplasty® by ArthroCare)
- Those that target the annulus fibrosis, such as annuloplasty with intradiskal electrothermal therapy (IDET) (e.g., SPINECATH and ACUTHERM from NeuroTherm)

Among these techniques, some are best suited to nerve root compression (Nucleoplasty, ACUTHERM) and others to isolated disk disease (SPINECATH).

#### 32.2.1.3 Advantages

Generally, the risks associated with this procedure are considerably reduced because intradiskal RF ablation is performed through a percutaneous approach and the procedure is short (less than 20 min).

- Local anesthesia or neurosedative agents are sufficient for these minimally painful procedures. This allows for continued oral communication with the patient about pain and neurological signs.
- The spinal canal is not breached and there are no epidural scars. The posterior musculature is not touched.
- The risk of sepsis is negligible.
- Only a very small part of the disk structure is removed.

## 32.2.1.4 Common Mechanisms of Action

Percutaneous techniques are indicated in early degenerative disk disease, at a stage where the disk is still hydrated.

Diskogenic pain is provoked by excessive pressure on the disk, along with inflammation and hyperinnervation of annulus fibers within the fissures. Contained disk herniation within the annulus is only one of the degenerative stages.

Several mechanisms of actions have been proposed to explain the effectiveness of these procedures:

- Denaturation of type I collagen (greater stiffness and dehydration)
- Nuclear cavitation leading to reduction in nucleus volume and reintegration of annular fragments (disk nucleoplasty)
- Destruction of annular fissures and destruction of peripheral inflammatory neovascularization and newly formed pain receptors (annuloplasty)
- Disintegration of the herniated mass (annuloplasty)

#### 32.2.1.5 Patient Selection and Results

RF ablation techniques are aimed at patients with diskogenic lower back pain, with or without small contained disk herniation, with or without compressive or referred radiculalgia. As with every treatment, the outcomes are a function of patient selection. The outcomes are also affected by the procedure quality, catheter placement, navigation within the disk, and the surgeon's practices.

The minimum imaging assessment consists of weight-bearing and dynamic X-rays, MRI without contrast, and CT scan.

The indications are:

- Diskogenic lower back pain and lumbar radicular pain due to contained disk herniation that has not responded to at least 6 months of wellconducted conservative treatment and physical therapy. Individual background variables such as the presence of secondary gains or social and professional conflicts must also be evaluated. Components of the diskogenic lower back pain diagnosis may include the outcome of the corset test and a negative result after facet joint block. Other diagnostic tools include the diskography results associated with provoked pain and evidence of posterior fissure. This evaluation is recommended before performing RF ablation to evaluate disk pressure (if the goal is to reduce it) or to eliminate the presence of non-contained disk herniation.
- Involvement of one or two levels. The procedure is harder to perform at L5–S1 in men because the disk is embedded into the iliac crests.
- MRI: type 0 Modic changes.
- Disk height >70 % on weight-bearing X-rays.

The contraindications are suspicion of facet-related low back pain, extruded disk, spondylolisthesis or retrolisthesis, symptomatic lumbar stenosis, disk asymmetry, scoliosis, type I Modic changes (inflammatory changes in vertebral end plates), collapsed disk, and epidural leakage on diskogram (noncontained hernia).

## 32.2.1.6 Intradiskal RF Ablation Techniques

There are two types of techniques, aimed either at isolated disk degeneration or contained disk herniation.

#### Treatment of Isolated Disk Degeneration

The SPINECATH IDET [14] is an annuloplasty procedure where a bipolar RF catheter is inserted into the posterior or lateral section of the annulus (until the fissure is reached). Fluoroscopy is used to verify the catheter position (Fig. 32.2) and then thermal energy (40–60 °C) delivered over a 5-cm area for 16 in. Optimal placement of the catheter can be challenging, as the entire posterior annulus must be covered without breaching it (Fig. 32.3). Moderate resurgence of the diskogenic pain during the procedure is a sign of efficacy.

Several nonrandomized studies have reported moderate improvements with 50 % pain reduction after 1 year [15–18]. Two randomized studies found good results versus placebo in terms of reduction in the Oswestry Disability Index (ODI) [19, 20].

 Percutaneous intradiskal RF thermocoagulation (PIRFT) techniques are used to perform nucleoplasty with monopolar radiofrequency; ALAR, RADIONIC, and DISKIT have been gradually abandoned because of disappointing results and lack of reliable basic research [21, 22].

#### **Treatment of Contained Disk Herniation**

Laser diskectomy was first introduced in 1986 [23]. This is not actually an RF technique, but one where the nucleus is vaporized using a laser diode. This procedure is aimed at treating lumbar radicular pain due to contained disk herniation. The needle or catheter is introduced in the center of the disk and then moved to the posterolateral side toward the suspected hernia. Intermittent bursts of energy are given up to a total dose of 1,200–1,600 J to



**Fig. 32.2** Unrolled configuration of the SPINECATH IDET (NeuroTherm) on A/P (*left*) and lateral (*right*) fluoroscopy views



**Fig. 32.3** Drawing of the optimal position of the SPINECATH catheter

vaporize part of the nucleus [23]. This technique is demanding and requires optimal placement of the powerful catheter. Secondary end plate inflammatory lesions have been described by Cvitanic in 30 % of cases [24], as well as a few cases of thermal discitis with flare-up of the low back pain.

Several studies have reported good results and pain reduction in about 70 % of cases [25-27]. This pain reduction occurs 6–8 weeks after the procedure. The best indication seems to be contained herniation or failed disk decompression using radiofrequency energy



**Fig. 32.4** Drawing of catheter positioning used during disk decompression (Nucleoplasty® by ArthroCare)

(nucleoplasty) for the same indications. However, the efficacy of this technique has yet to be truly demonstrated [28].

 DISK nucleoplasty<sup>™</sup> is a newer technique introduced by ArthroCare. This technique consists of RF catheter coblation (cold ablation) to induce nucleus cavitation (Fig. 32.4) and reduce intradiskal pressure. Less heat (40–70 °C) is produced than with a laser, thus the risk of end plate damage is lower. The indications are midline- or lateral-contained


**Fig. 32.5** Drawing of ACUTHERM<sup>TM</sup> decompression catheter (NeuroTherm) being used to treat a lateral disk herniation

herniation in minimally degenerated disks. In an animal model, this treatment was shown to alter cytokines such as IL-1 and IL-8 [29]. Cohort studies have shown significant decrease in VAS pain and with good results in 70–80 % of cases [30–32].

• The ACUTHERM IDET is an annuloplasty technique derived from the SPINECATH system (same RF generator) that uses a catheter to provide targeted disk decompression. A catheter heats the area of the disk-nerve root impingement, which alters the collagen and reduces the hernia by desiccation. This technique can be used with foraminal hernias (Fig. 32.5).

### **Other Percutaneous Techniques**

These other techniques revolve around chemonucleolysis (injection of a chemical agent) and percutaneous diskectomy (mechanical action), for example:

- Nucleolysis with Hexatrione<sup>®</sup>, which has been abandoned because of the risk of disk calcification
- Ozone chemonucleolysis
- Absolute alcohol nucleolysis
- Discogel® nucleolysis

Discogel<sup>®</sup> (jellified ethanol, manufactured by Gelscom in France) is an implantable device that

is a promising treatment for contained disk herniation. It aims at improving water diffusion from the periphery to the center of the disk. The gel's viscosity keeps the alcohol from leaking outside the disk. Results of a pilot study were encouraging [33]. A study funded by the PHRC in France is currently under way. We will soon be publishing the results of 35 patients that were treated with Discogel<sup>®</sup>.

#### 32.2.1.7 Conclusion

Percutaneous intradiskal RF techniques are easy to perform in the hands of a trained surgeon. They are minimally invasive for disk and perivertebral structures and have a low complication rate, but require specific instrumentation and consumable products (RF generator and catheter).

Their main role resides in treating diskogenic lower back pain at an early, nonsurgical stage of the disease progression and as an alternative to traditional diskectomy surgery for treating disk herniation. However, the uptake of these techniques has been limited because of persistent questions about their efficacy, despite numerous publications.

# 32.2.2 Nucleus Pulposus Implant: Preliminary Results of a Memory-Coiling Spiral Implant

### 32.2.2.1 Introduction

The aftereffects of disk nucleus ablation have been evaluated clinically and biomechanically. During an in vitro study, Brinckmann [34] found progressive loss of disk height and increased radial disk bulging proportional to the mass of the excised nucleus tissue. This loss of disk height can lead to overload of the posterior facet joints and biological changes in the joint cartilage [35, 36]. These changes can cause painful spondylarthrosis and eventually require surgical treatment. After nucleus removal, kinematic studies have shown increased range of motion and clear displacement of the center of rotation in flexion/extension and lateral bending [37–39].

# 32.2.2.2 Design Specifications for a Memory-Coiling Spiral Nucleus Pulposus Implant

Our primary goal was to come up with an original method to replace only the nucleus pulposus through a minimally invasive procedure that would be an extension of the surgical diskectomy procedure used to treat disk herniation and to maintain the physiology of the mobile spinal segment and adjacent disks [40-44]. For this "memory-coiling spiral" implant, the memory effect is achieved through a specific manufacturing process where the base polycarbonate urethane elastomer (Sulene<sup>TM</sup> PCU, previously produced by Centerpulse, now Zimmer) is modified at the molecular level. When implanted in its unrolled state into the empty intradiskal cavity, the implant spontaneously regains its pre-formed spiral shape and completely fills the cavity without mechanical fixation (Fig. 32.6).

By restoring or maintaining maximum disk height, normal lumbar lordosis is preserved and the congruence between the facet joints is preserved as best as possible. Together, these limit the occurrence of secondary facet-related pain. This design provides a certain degree of pain-free interbody stability, preserves the mobility of the intervertebral segment, and avoids the need for another surgical procedure.

To meet these objectives, the intervertebral disk implant needs to have the following properties: non-compressible, elastic, and able to withstand continuous cyclic loading in various directions and no mechanical axis or fixed center of rotation so as to not introduce abnormal loads that could lead to painful joint dysfunction or localized overloading of the adjacent vertebral end plates. The centers of rotation of this new implant do not need to be positioned precisely during surgery as do other implants [45]. The implant's center continuously changes during movements within an area that is specific to each level and each movement.

The implant must fill the intradiskal void created by percutaneous diskectomy as completely as possible. This will restore internal disk preloads and, as a consequence, the biomechanical conditions required for healing of the type II collagen fibers in the outer annulus (which have a half-life of about 7 months, provided they are not in an advanced stage of degenerative disk disease). In this scenario, the outer annulus has some functional recovery capacity and is able to maintain its "new" nucleus without causing disk prolapse or predisposing it to expulsion into the medullary canal.



Fig. 32.6 Drawing of the nucleus pulposus memory-coiling spiral implant (*left*) and photo of the actual product (*right*)

This implant truly differs from other nucleus replacement implants because it preserves the adjacent vertebral end plates [45] and avoids or delays continued degeneration of the outer annulus by keeping it in a functional state.

This implant must be biocompatible, reliable, and not undergo any material degradation over time. In addition, it must be easy to remove through the same surgical approach.

### 32.2.2.3 Materials and Methods

#### **Properties of the Implant Material**

Some of the implant's most important properties-non-compressible, highly compressive, tensile and shear strength, and excellent fatigue and tear resistance after being cut-are made possible by the raw material used. The spiral implant is made of polycarbonate urethane elastomer that has excellent biocompatibility (ISO 10993, FDA Tripartite Guidance) and biological stability. Although it has been used in other implantable medical devices, it had never been inside implanted an intervertebral disk. Biocompatibility testing was performed by placing the implant in the cervical disks of two sheep who where then sacrificed 3 and 6 months later. Histology studies showed excellent biocompatibility of this material inside a disk [40, 41].

#### Implant Geometry

The implant has a spiral shape. A specific manufacturing process is used where the base polycarbonate urethane elastomer is modified at the molecular level to give it memory-shape properties (Fig. 32.6). Because of the device's properties, it can be inserted through a small opening in the annulus fibrosis. This opening is the same one used to treat the disk herniation. There seems to be no theoretical or practical reasons why a helixshaped disk implant cannot be used, as suggested by Edeland [46, 47].

This "new" nucleus restores internal preloads and seems to have the ability to directly selfanchor. It is protected because the outer annulus has regained its basic functionality due to the disk height being restored and the fibers being correctly reoriented and reloaded. Together, this ensures that the surgical opening made for the diskectomy and implantation closes itself without the need for surgical glue.

Based on disk height measurements obtained at L3–L4, L4–L5, and L5–S1 from a large number of patients, the ideal implant thickness was determined to be 6 and 8 mm. These are the two sizes currently available.

#### Mechanical Properties

The implant must be able to withstand several million compressive cycles without tearing or wearing down. Dynamic and static compression tests confirmed that this objective was met, as there were no implant failures. Fatigue testing was performed under physiological conditions (37 °C Ringer solution) with 1200 N compressive loads placed in multiple directions (axial compression,  $\pm 5^{\circ}$  off-axis compression) for up to 50 million cycles.

In situ biomechanical studies with three human cadaver specimens were performed to evaluate the effect of the proposed implant on spinal kinematics from full flexion to full extension of L4 relative to L5. Three conditions were tested in sequence: intact disk, after nucleotomy (disk removed), and then after spiral implant insertion [38, 39]. While nucleotomy reduces disk height and increases facet joint compression, the spiral implant restores the lost height and resets the facet joints in their proper position. Although the sample size was too small to perform statistical testing (n=3), the curves measured from full flexion to full extension led us to conclude that the spiral implant does not change the overall kinematics of the segment (Fig. 32.7).

To determine the implant's effect on deformation of the adjacent vertebral end plates, seven functional spinal units were instrumented with strain gauges [48]. Up to 500 N was applied in axial compression, off-center flexion, extension, lateral bending (left and right), anterior-posterior shear, and medial-lateral shear. Three conditions were tested in sequence: intact disk, after nucleotomy (disk removed), and then after spiral implant insertion. There were no differences in the deformation of the central part of the end plate relative to the intact condition (except for



**Fig. 32.7** Kinematic *curves* measured in situ in the sagittal plane (full flexion to full extension) of L4 relative to L5. Measurements taken with an intact disk, after nucle-

otomy (disk removed), and then after spiral implant insertion in cadaver specimen No. 2 (*left*); FasTrak simulator (*right*) [38]

posterior shear). The spiral implant increased the deformation of the central part of the end plate relative to the nucleotomy condition, thereby reducing the loads on the facet joints. There was no implant migration, no matter the direction of load application.

### **Patient Selection**

For the pilot clinical study and then the multicenter European study, very narrow inclusion and exclusion criteria were applied during patient selection [49].

Inclusion criteria: age 18–65 years, suffering from single-level radiculalgia or lumbar radicular pain with more leg pain than lower back pain due either to lateral disk herniation or single-segment acquired intervertebral disorder [2] secondary to stage II degenerative disk disease (instability in the Kirkaldy-Willis classification), and has not responded to well-conducted conservative treatment. These two etiological presentations had to be localized to a single level between L2 and S1. MRI was used to confirm the annulus was not significantly altered and still had some functional recovery capacity, and the posterior disk height in the involved level was at least 5 mm.

*Exclusion criteria*: previous spine surgery; multilevel spinal disorder (disk herniation,

acquired degenerative intervertebral dysfunction); medial disk herniation; factors that could increase the risk of faulty positioning at the involved level (highly biconvex-shaped disk, lesion in adjacent vertebral end plates due to intracancellous hernia); significant central, lateral, or foraminal spinal stenosis; symptomatic degenerated facet joints; degenerative spondylolisthesis beyond grade I; isthmic spondylolisthesis; severe osteoporosis, active infection; bone tumor or congenital bone abnormality; associated severe disease; and more than 40 % overweight.

*Study variables:* According to the study protocol, the patients received clinical and imaging examinations before the procedure, immediately postoperative, then at 3, 6 months, and 1, 2, and 5 years after the surgery. The clinical outcomes were visual analog scale (VAS) for low back pain and radiculalgia according to Huskisson, type and amount of analgesics used, neurological evaluation (sensory, motor, reflexes), angulation while looking for Lasègue's sign (positive straight leg raise test), activity level according to the Oswestry Disability Index (ODI), and patient satisfaction.

The imaging assessment combined static A/P and lateral X-rays with an MRI exam. The latter was intended to evaluate the position of the radiolucent implant, the disk height, and the vitality



**Fig. 32.8** Pain (VAS) and functional outcome (Oswestry Disability Index) for the entire patient series with a memorycoiling spiral implant, where FUP is the time after implantation in months

and function of the outer annulus. Segmental motion was evaluated using dynamic A/P and lateral X-rays performed 2 and 5 years after implantation. The function of the facet joints was evaluated 2 years postoperative using a dynamic twist CT-scan test [3].

#### 32.2.2.4 Results

The clinical and radiological results of the first nine cases included in the prospective, pilot, noncomparative, multicenter European study initiated in September 2001 will be reviewed here [49]. There were eight men and one woman. The average age at the time of surgery was 37.4 years (range 23–51). The follow-up ranged between 6 and 64 months. The surgical indication in all patients was disk herniation with radicular symptoms, with six also having low back pain. L5–S1 was involved in six cases and L4–L5 in three cases. In six cases, the surgical approach was on the right side and in three cases, on the left. The average diameter of the spiral implant was 21 mm (range 17–24).

Clinical results for the entire series are provided in Fig. 32.8. The VAS for radiculalgia and low back pain was clearly lower in all patients after the surgery. There were no intraoperative complications. No neurological deficits were observed during the follow-up examinations. Function based on the ODI was improved in all patients. Three intradiskal implant modifications were visible on MRI. Although there were no clinical symptoms, all three implants were explanted as a preventative measure.

The clinical, X-ray, and MRI assessments were used to evaluate these outcomes and look for direct and indirect signs of changes in implant positioning and its potential effects. The patient with the longest follow-up (24-yearold female) underwent implantation on May 27, 1997. She was a licensed nursing assistant who had ongoing L5 right radiculalgia with motor deficit in the foot associated with ongoing low back pain due to lateral L4-L5 disk herniation on the right side (Fig. 32.9a). Since conservative treatment was not effective, she underwent surgical diskectomy and then spiral implant insertion. She was evaluated 3, 6, 12, 24, 48, and 64 months after the procedure. After 64 months, the clinical assessment found no radicular pain or low back pain, no functional limitations with a finger-to-floor distance of 0 mm, normal extension (Fig. 32.9b), and no social, professional, or recreational limitations. The patient was very satisfied with the implantation procedure because she was able to bring two pregnancies to term and was able to return to work in the same job.

MRI was used to evaluate any changes in position, shape, coiling, translation, rotation, or displacement within the implant's cut plane. No changes were observed in six of the nine cases (Fig. 32.10a), including patient 1 with 64 months of follow-up (Fig. 32.10b). In the remaining three



**Fig. 32.9** (a) Preoperative MRI for patient no. 1; *left* lateral disk herniation at L4–L5. (b) Dynamic flexion-extension X-rays at 64 months postoperative for patient no. 1; normal spinal mobility and preserved disk height. *arrows* operating level



**Fig. 32.10** (a) Results at 12 months postoperative for patient no. 4; no change in the position, shape, coiling, translation, rotation, or cut plane of the implant. (b) MRI at 64 months postoperative for patient no. 1; no changes in the implant

cases, the clinical assessment was satisfactory at 12 months postoperative, but the MRI revealed two cases of intradiskal shape change (uncoiling), posterior translation, rotation of the central part, and displacement within the cut plane, and one case of general posterior intradiskal movement without uncoiling. Based on the MRI findings, the implants were removed in all three patients as a preventative measure.

The MRI also revealed direct signs of efficacy of the nucleus pulposus implant. We found continued vitality and function of the outer annulus after 64 months (Fig. 32.10b) and maintained disk height in all patients both on X-rays (Fig. 32.9b) and MRI (Fig. 32.10b).

The linear mobility in flexion/extension was found to be normal in dynamic X-rays 1, 2, and 5 years postsurgery in the first three patients, and proper lumbar lordosis had been restored and/or maintained (Fig. 32.10b). Rotational mobility was evaluated in the first two patients at 2 and 5 years postsurgery using the twist CT scan [3]. This test revealed that the facet joint function had been fully preserved (Fig. 32.11).



Fig. 32.11 Twist CT scan at 24 months postoperative for patient no. 1; facet joints appear to function normally

### 32.2.2.5 Conclusion

The memory-coiling lumbar nucleus pulposus spiral implant restores spine biomechanics by reestablishing normal spine kinematics and by restoring and/or maintaining intervertebral disk height and proper facet joint positioning through better load distribution. Biomechanical testing has demonstrated that the spinal implant stays in place and that it distributes loads with proper support from the anterior spine.

Patients who have received this implant up to now have achieved significant pain reduction and were very satisfied with the procedure. Under these conditions, the memory-coiling spiral implant seems to be sound, tangible, and effective, but it can only be used if the outer annulus is intact. Since this excludes post-diskectomy indications, the implant design will be further developed to integrate new ideas. Its effectiveness and easy application-due to a standardized approach—its shape memory properties, and the potential for minimally invasive insertion are all steps forward within the realm of non-fusion techniques, especially relative to other nucleus implants such as the prosthetic disk nucleus (PDN) device [45].

# 32.3 Proposed Conservative and Surgical Treatment Indications for Various Stages of Lumbar Spine Degeneration

Treatment indications for cases of acquired degenerative intervertebral dysfunction (ADIVD) are first and foremost a function of the four stages of the new classification for loss of stability in the degenerative lumbar spine described earlier. They also take into consideration several other criteria: presence of spinal stenosis (dynamic or static), presence of translational deformity, presence of posterior facet osteoarthritis, disk condition, recovery potential of the annulus fibrosis evaluated on MRI using the Modic stages [6, 7] and the Pfirrmann algorithm [8], and posterior disk height (more or less than 5 mm) (Fig. 32.12).

#### Stage 0: Minimal Dysfunction

The goal is to prevent the changes observed in the basic structural components from getting worse by using good spine health habits, such as daily exercise to maintain the capacity of the stabilizing muscles.

ADIVD*	Type of stenosis	Posterior disk height >5mm	Reversible annulus changes	Reducible	Facet joint osteoarthritis	Translational deformation	Therapeutic indications	
Minimal		+	+	+	-	-	Prevent aggravation Conservative treatment	
Minor		+	+	+	-	-		
Major	Dynamic	+	+	+	-	-	mpression	Nucleus pulposus implant
		+	-	+	-	-		Nucleus pulposus implant & Dynamic stabilization
		-	-	+	-	-		
	Static	-	-	+	-	_		Total disk replacement
		-	-		_		ll deco	Fusion
Maximal	Permanent lateral	-	-		+		Surgica	
		-	-	-	+	+		
					+			

Fig. 32.12 Indications for conservative and surgical treatment of acquired degenerative intervertebral dysfunction (*ADIVD*)

## Stage I: Minor Dysfunction

After 48-72 h of conservative treatment with analgesics, muscle relaxants, and NSAIDs to reduce the acute phase, manipulations can be performed within the limits of pain, along with relaxation massage and gentle physical therapy. Intra-articular injections may be needed in cases of strong periarticular or interspinous ligament reactions. Rehabilitation that combines lumbar and pelvic stabilization, by reinforcing the dynamic control over the lumbar, abdominal, and pelvic muscles, with relaxation of the anterior or posterior lower pelvic girdle muscles through specific postural stretching, is often indicated, along with overall rebalancing of the spine-pelvis-femur complex.

## Stage II: Major Dysfunction

- The first step is to apply *conservative treatment*, which consists of standard medical treatment (analgesics, muscle relaxants, and NSAIDs), immobilization of the thorax and pelvis using braces with lumbar-abdominal muscle reinforcement, and rebalancing of the spinepelvis-femur complex.
- *Surgical treatment* is only indicated after wellconducted conservative treatment has failed and would be performed in two progressive conditions:
  - In cases of *dynamic stenosis*, when there is no translational deformity or posterior facet arthritis, reduction is possible, and the posterior disk height is more than 5 mm, a nucleus pulposus implant can be proposed

if the annulus fibrosis is minimally altered on MRI according to the Modic stages [6, 7] and the Pfirrmann algorithm [8], and the fibrous annulus has changes that can be reversed [2, 49].

- If the spinal canal volume is altered but no anatomical changes are present, posterior or posterolateral dynamic stabilization can be proposed if the loss of stability is reducible and there is no posterior facet joint osteoarthritis.
- In cases of *static stenosis* where the anatomical volume of the nerve root canal is altered, either due to disk protrusion or retrolisthesis during extension, or due to posterior facet joint osteoarthritis, rehabilitation can be proposed in combination with gentle vertebral traction in a pool; if this fails, total disk replacement can be proposed as long as the deformity has not translated and there is no posterior osteoarthritis; otherwise, only fusion is indicated, especially if two separate levels are affected, in combination with either a disk-specific procedure or nerve root decompression. Any surgical procedure at this stage that modifies the statics and dynamics of the spine-pelvis-femur complex must be followed by overall rebalancing treatment.

#### Stage III: Maximal Dysfunction

Conservative treatment should be used initially while making sure not to interfere with spontaneous restabilization by needlessly trying to increase vertebral mobility. Instead, the goal should be to correct static and dynamic imbalances by stimulating spinal antigravity muscles and readjusting muscles mainly in the anterior and posterior lower pelvis area. Use of external bracing is indicated for more at-risk patients. Surgical treatment is only indicated after well-conducted conservative treatment has failed. The goal will be to decompress the nerve root, possibly in combination with fusion or fusion alternatives.

# Conclusion

The clinician, who must attempt to quantify purely subjective functional signs such as pain, while relying on a challenging physical examination, can only deliver a subjective interpretation because of the lack of paraclinical, morphological, static, dynamic, electrophysiological, histological, and pathognomonic criteria on which a decision can be made.

These uncertainties, along with the risk of being misled due to semantics, led us to propose the term "acquired degenerative intervertebral dysfunction (ADIVD)" to more accurately take into account the pathological, biomechanical, and clinical reality of progressive lumbar spine degeneration and to help frame the indications. We hope this will make it easier to choose the appropriate treatment among the many available.

Acknowledgments The authors wish to thank Joanne Archambault and Réto Braunschweiler for the help provided during the preparation of this book chapter.

### References

- 1. Kirkaldy-Willis WH, Farfan HF. Instability of the lumbar spine. Clin Orthop. 1982;165:110–23.
- Husson JL. Instabilité vertébrale à l'étage lombaire. Conférence d'Enseignement 1995. Cahier d'Enseignement de la S.O.F.O.C.T., vol 52. Paris: Expansion Scientifique Française ; 1995, p. 63–78.
- Husson JL, Poncer R, De Korvin B, Meadeb J. Apport du Scanner en Twist-Test dans la mesure de l'instabilité du Rachis Lombaire. Rev Chir Orthop. 1993;79(1):11.
- Graf H. Instabilité vertébrale. traitement à l'aide d'un système souple. RACHIS. 1992;4:123–37.
- White A, Panjabi M. Clinical biomechanics of the spine. Philadelphia: J.B. Lippincott Company; 1978.
- Modic MT, Massaryk TJ, Ross JS, et al. Imaging of degenerative disk disease. Radiology. 1988;168:177–86.
- 7. Modic MT, Steinberg PM, Ross JS, et al. Degenerative disk disease: assessment of changes in vertebral body marrow with MR imaging. Radiology. 1988;166:193–9.
- 8. Pfirmann CW, Metzdorf A, Zanetti M, Hodler J, Boos N. Magnetic resonance classification of

lumbar intervertebral disc degeneration. Spine. 2001;26(N°17):1873–8.

- Kambin P, Sampson S. Posterolateral percutaneous suction-excision of herniated lumbar intervertebral discs. Report of interim results. Clin Orthop Relat Res. 1986;207:37–43.
- Kambin P, Gellman H. Percutaneous nucleotomy: a new treatment method for lumbar disk herniation. J Toden Hosp. 1975;5:39–44.
- Dabezies EJ, Langford K, Morris J, Shields CB, Wilkinson HA. Safety and efficacy of chymopapain (Discase) in the treatment of sciatica due to a herniated nucleus pulposus. Results of a randomized, double-blind study. Spine. 1988;13:561–5.
- Fraser RD. Chymopapain for the treatment of IV disk herniation. A preliminary report of a double blind study. Spine. 1984;9(8):815–8.
- Crawshaw C, Frazer AM, Merriam WF, Mulholand RC, Web JK. A comparison of surgery and chemonucleolysis in the treatment of sciatica. A prospective randomized trial. Spine. 1984;9(2):195–8.
- Saal JA, Saal JS. Intradical electrothermal treatment for chronic discogenic low back pain: prospective outcome study with a minimal 1-years follow up. Spine. 2000;25(20):2622–7.
- Saal JA, Saal JS. Intradiscal electrothermal treatment for chronic discogenic low back pain: prospective outcome study with a minimum 2-year follow-up. Spine. 2002;27(9):966–73; discussion 973–4.
- Bogduk N, Karasek M. Two-year follow-up of a controlled trial of intradiscal electrothermal annuloplasty for chronic low back pain resulting from internal disc disruption. Spine J. 2002;2:343–50.
- Karasek M, Bodduk N. Twelve-month follow-up of a controlled trial of intradiscal thermal annuloplasty for back pain due to internal disc disruption. Spine. 2000;25(20):2601–7.
- Lee MS, Cooper G, Lutz GE, Lutz C, Hong HM. Intradiscal electrothermal therapy (IDET) for treatment of chronic lumbar discogenic pain: a minimum 2-year clinical outcome study. Pain Phys. 2003;6:443–8.
- Pauza KJ, Howell S, Dreyfuss P, Peloza JH, Dawson K, Bogduk N, et al. A randomized, placebo-controlled trial of intradiscal electrothermal therapy for the treatment of discogenic low back pain. Spine J. 2004;4:27–35.
- 20. Freeman BJC, Fraser RD, Cain CMJ, Hall DJ, Chapple DCL, et al. A randomized, double-blind, controlled trial: intradiscal electrothermal therapy versus placebo for the treatment of chronic discogenic low back pain. Spine. 2005;30:2369–77.
- Barendse GAM, Van den Berg SGM, Kessels AHF, et al. Randomized controlled trial of percutaneous intradiscal radiofrequency thermo-coagulation for chronic discogenic back pain. Lack of effect from a 90 second 70°C lesion. Spine. 2001;26:287–92.
- 22. Ercelen O, Bolutcu E, Oktenoglu T, et al. Radiofrequency lesioning using two different time modalities for the treatment of lumbar discogenic low back pain: a randomized trial. Spine. 2003;28:1922–7.

- Choy DSJ, Case R, Fielding W, Hugues J, Liebler W, Asher P. Percutaneous laser nucleolysis of lumbar disc. N Engl J Med (Lett). 1987;12:771–2.
- Cvitanic OA, Schimandle J, Casper GD, Tirman PF. Subchondral marrow changes after laser diskectomy in the lumbar spine: MR imaging findings and clinical correlation. AJR. 2000;174:1363–9.
- Choy DSJ, Asher PW, Ranu HS, Saddekni S, Ackaiti D, Liebler W, Hugues J, Altman P. Percutaneous laser decompression: a new therapeutic modality. Spine. 1992;17:949–56.
- Nerubay J, Caspi J, Levinkopf M. Percutaneous carbon dioxide laser nucleolysis with 2 to 5 year follow up. Clin Orthop. 1997;337:45–8.
- Liebler WA. Percutaneous laser nucleotomy. Clin Orthop. 1995;310:58–66.
- HAS (Haute Autorité de Santé). Destruction d'un disque intervertébral par laser (Nucléotomie) par voie transcutanée avec guidage radiologique, S.e.e.d. à professionnels, editor. Paris; 2005, p 1–35.
- O'Neill CW, Liu JJ, Leibenberg E, Hu SS, Deviren V, Tay BK, et al. Percutaneous plasma decompression alters cytokine expression in injured porcine intervertebral discs. Spine J. 2004;4:88–98.
- Mirzai H, Tekin I, Yaman O, Bursali A. The results of nucleoplasty in patients with lumbar herniated disc: a prospective clinical study of 52 consecutive patients. Spine J. 2007;7:88–92. discussion 92-3.
- Sharps LS, Isaac Z. Percutaneous disc decompression using nucleoplasty. Pain Phys. 2002;5:121–6.
- Yakovlev A, Tamimi MA, Liang H, Eristavi M. Outcomes of percutaneous disc decompression utilizing nucleoplasty for the treatment of chronic discogenic pain. Pain Phys. 2007;10:319–28.
- 33. Theron J, Guimaraens L, Casasco A, Sola T, Cuellar H, Courtheoux P. Percutaneous treatment of lumbar intervertebral disk hernias with radiopaque gelified ethanol: a preliminary study. J Spinal Disord Tech. 2007;20:526–32.
- Brinckmann P, Grootenboer H. Change of disc height, radial disc bulge, and intradiscal pressure from discectomy. An in vitro investigation on human lumbar discs. Spine. 1991;16:641–6.
- Dunlop RB, Adams MA, Hutton WC. Disc space narrowing and the lumbar facet joints. J Bone Joint Surg (Br). 1984;66:706–10.
- Gotfried Y, Bradford DS, Oegema TR. Facet joint changes after chemonucleolysis- induced disc space narrowing. Spine. 1986;11:944–50.
- 37. Steffen R, Wittenberg RH, Nolte LP, Hedtmann A, Kolditz D, Herchenbach T. Experimentelle Untersuchungen zur Drehpunktveränderung des Bewegungssegmentes nach Bandscheibenausräumung. Z Orthop Ihre Grenzgeb. 1991;129:248–54.
- Freudiger S, Husson JL. Nucléoplastie intersomatique par voie postérieure perdiscectomie: étude biomécanique sur simulateur. In: Husson JL, Le Huec JC, editors. Restabilisation inter-somatique du rachis lombaire, vol. 8. Montpellier: Sauramps Médical; 1996. p. 321–31.

- Schärer N, Husson JL, Froehlich M. Translation of the endplate during flexion and extension on intact human spines: an in-vitro study. Eur Spine J. 1999;8:S52–3.
- 40. Husson JL, Baumgartner W, Le Nihouannen JC. Nucléoplastie inter- somatique par voie postérieure perdiscectomie: concept et étude expérimentale. In: Husson JL, Le Huec JC, editors. Restabilisation inter-somatique du rachis lombaire. Montpellier: Sauramps; 1996. p. 311–20.
- Husson JL, Schärer N, Le Nihouannen JC, Freudiger S, Baumgartner W, Polard JL. Nucleoplasty during discectomy: concept and experimental study. Rachis. 1997;9:145–52.
- 42. Husson JL, Schärer N. Key aspects and biomechanical approach for the development of a nucleoplasty. In: 7th international meeting on advanced spine techniques (IMAST). 6–8 Jul 2000, Barcelona.
- Husson JL, Korge A, Nydegger T et al. Artificial nucleus replacement with a spiral implant: early clinical results. Spine arthroplasty meeting. 6–8 May 2002, Montpellier.

- 44. Korge A, Nydegger T, Polard JL, Mayer HM, Husson JL. A spiral implant as nucleus prosthesis in the lumbar spine. Eur Spine J. 2002;11 Suppl 2:S149–53.
- 45. Ray CD. The PDN prosthetic disc-nucleus device. Eur Spine J. 2002;11 Suppl 2:S137–42. Epubg 2002 Jun 4.review.
- Edeland HG. Suggestions for a total elasto-dynamic intervertebral disc prosthesis. Biomater Med Dev Artif Organs. 1981;9:65–72.
- Edeland HG. Some additional suggestions for an intervertebral disc prosthesis. J Biomed Mater Resour Appl Biomater. 1989;23:189.
- Frei HP, Rathonyi G, Orr TE, et al. Effect of a nucleus prosthetic device on the biomechanical behavior of the functional spinal unit. Eur Spine J. 1999;8:S36–7.
- 49. Husson JL, Korge A, Polard JL, Nydegger T, Kneubühler S, Mayer HM. A memory coiling spiral as nucleus pulposus prosthesis. Concept, specifications, bench testing and first clinical results. J Spinal Disord Tech. 2003;16(4):405–11.

# Lumbar Spine Injections: An Evidence-Based Review

33

Byron J. Schneider, Neal Varghis, and David J. Kennedy

# 33.1 Introduction

Lumbar spine injections are frequently used for both diagnostic and therapeutic purposes in patients suffering from painful disorders of the spine. There are a wide variety of potential pain generators in the lumbar spine including the intervertebral disk, zygapophyseal joint, and nerve roots among others. Similar to surgery, injection therapy is by definition target specific. Given this variety of pain generators and the invasive and focal nature of an injection, it is crucial to accurately review the efficacy literature on these techniques. It is imperative to not combine nonspecific diagnoses such as low back pain when evaluating the efficacy of a procedure. It is equally imperative to consider that not all techniques are the same. For instance, the literature has repeatedly demonstrated that nonimageguided (aka "blind") injections have a very low

Division of Physical Medicine and Rehabilitation, Department of Orthopaedic Surgery, Stanford University, 450 Broadway St., Pavilion C, MC 6342, Redwood City, CA 94063, USA e-mail: bjschn2@stanford.edu;

nvarghis@stanford.edu; djkenned@stanford.edu

accuracy, thus negating the ability to be target specific. Nonimage-guided injections, such as muscle trigger point injections, have not only been shown to have poor accuracy, but they also have been found to be no more effective than sham treatments. These procedures will therefore not be discussed in this chapter. There are also other injections that have no credible published efficacy data, and these too will not be covered. This chapter will thus focus on the commonly preformed image-guided injections. In doing so it will focus on the literature behind the most common injections. Each section will cover the basic principles, evidence-based outcomes, and technical considerations. The chapter will cover procedures targeting the lumbar intervertebral disks, the zygapophyseal joint, and the epidural space.

# 33.2 Lumbar Zygapophyseal Joints

Joel E. Goldthwait first suggested that lumbar zygapophyseal (aka Z-joints or facet joints) could cause low back pain back in 1911 [1]. Later, the discovery of facet joint nociceptors gave credence to this [2]. Much more recently, Kaplan then demonstrated that lumbar z-joint pain could be provoked by capsular distention [3]. Lumbar z-joint pain is the most common source of low back pain, and depending on how the diagnosis is

B.J. Schneider, MD • N. Varghis, MD

D.J. Kennedy, MD (🖂)

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_33

made, it is the predominate source of low back pain up to 45 % of the time [4–6]. The most common cause of lumbar z-joint pain is osteoarthritis; however, other causes such as trauma, infection, synovial impingement, inflammatory arthritides, villonodular synovitis, and pseudogout are also painful processes that can affect the lumbar z-joints [7–11]. Risk factors for lumbar z-joint degeneration include age, genetics, joint tropism, disk degeneration, and abnormal spine alignment [12, 13]. Cadaveric studies have found that nearly 60 % of people demonstrate signs of z-joint osteoarthritis by the age of 30 and nearly everyone has signs of z-joint osteoarthritis by the age of 60 [14], with the most common level affected being L4/L5 [14, 15].

#### 33.2.1 Diagnosis

It is important to note that the degree of osteoarthritic degeneration is not directly correlated with degree of pain, and unfortunately, there are no specific physical exam maneuvers or radiographic findings that can always correctly identify those suffering from z-joint pain [12]. There are however clinical findings that are more suggestive of low back pain arising from the z-joints. These include age over 65, pain relieved by lying down, and the absence of pain being aggravated by coughing and absence of pain being aggravated by flexion [16]. Another study suggested that paraspinal tenderness correlates with z-joint arthropathy [17]. While the presence of these has been suggested to increase the likelihood of low back pain being due to z-joints, they are not specific. One study found that there was no diagnostic value in any clinical exam maneuvers for identifying z-joint-mediated low back pain [5, 16, 18–20]. In a review of the literature in 2008, Bogduk concluded that "the failure of multiple studies to identify any clinical features that are indicative of lumbar z-joint pain leaves diagnostic blocks as the only means of diagnosing this entity" [21].

Intra-articular z-joint injections were first described in 1976 by Mooney and Robertson [22]. Intra-articular anesthetic blocks involve

injection of anesthetic into the z-joint to evaluate if the patient's pain is alleviated by anesthetizing the z-joint itself. However, in the initial study by Kaplan that demonstrated lumbar z-joint pain could be provoked by capsular distention, he also demonstrated that anesthetizing the medial branches prior to repeating the provocative capsular distention negated the painful response in eight of nine subjects [3]. This led to the concept of medial branch blocks, which can be defined as a "diagnostic procedure designed to test if a patient's pain is mediated by one or more of the medial branches of the lumbar dorsal rami" [23]. In other words, pain relief from lumbar medial branch blocks implies that the anesthetized nerve mediates the patient's pain. Using a triple placebo-controlled block as the gold standard, Lord et al. found that intra-articular injections had less sensitivity and specificity than diagnostic medical branch blocks [24]. Utilizing appropriate image-guided technique, lumbar medial branch blocks have also been proven to be target specific [25]. Therefore medial branch blocks are the preferred method for diagnosing z-joint pain and have superiority over intra-articular blocks because they are easier to perform, safer to perform, more easily subjected to controls, have better sensitivity and specificity, and have proven therapeutic utility in that they predict response to radiofrequency neurotomy [23].

#### 33.2.2 Relevant Anatomy

The lumbar z-joint is a diarthrodial synovial joint that contains a joint space within a fibrous capsule. Anteriorly the capsule blends with the ligamentum flavum [20]. They are crescent shaped with the concave side facing the spinous process, resulting in the posterior aspect of the joint with a more sagittal orientation compared to the more coronal orientation of the anterior aspect. This is important to note as during an intra-articular injection, the posterior aspect of the joint is targeted. Moving caudally the general orientation of the z-joints becomes more coronal. The more coronal orientation allows the lower lumbar spine to better resist shear forces [13]. There are two recesses which are located at the superior and inferior aspects of the posterior z-joint; the inferior recess is larger of the two, while the superior recess is closer to the spinal nerves, and thus the inferior recess is the target of intra-articular injections [26].

The innervation to the z-joints has been well described. Dorsal rami arise from the spinal nerve in the intervertebral foramen and branch into the medial, intermediate, and lateral branches. An exception to this is the L5 dorsal ramus, which does not have an intermediate branch. Z-joints are innervated by the medial branches of the dorsal rami, once again with the exception of the L5 dorsal ramus, which innervates the respective z-joint prior to branching into a medial and lateral branch. The medial branch of the dorsal rami courses inferiorly between the superior articular process (SAP) and transverse process, hooking medially around the SAP and passing under the mamillo-accessory ligament. As such, any given lumbar medial branch actually runs along the vertebral body below that of the spinal nerve from which it arises. For example, the L4 medial branch courses along the L5 vertebral body. The resulting nomenclature is that each z-joint is innervated by branches from the level above the level of the z-joint itself. For example, the L4-L5 z-joint is innervated by medial branches of the L3 and L4 dorsal rami.

# 33.2.3 Contraindications

Absolute contraindications include systemic or localized infection, bleeding diathesis, and possible pregnancy. Relative contraindications include hypersensitivity or allergies to the medications to be used.

# 33.2.4 Intra-articular Joint Injection

The procedure is performed with the patient in the prone position. Correct identification of the target level is identified using fluoroscopy with an AP view. The fluoroscope is then positioned to the



Fig. 33.1 Needle enter right L5/S1 zygapophyseal joint

minimal obliquity that is required to identify the posterior joint space of the target z-joint. Local anesthetic is injected superficially. A spinal needle is then inserted parallel to the trajectory view toward the posterior inferior joint space. The needle is advanced until the joint capsule is penetrated (Fig. 33.1). Once the joint capsule is penetrated, confirmation of correct placement is performed with use of AP and lateral views. At this point, a small amount of nonionic contrast is administered under live fluoroscopy (Fig. 33.2). Care should be taken to notice any vascular uptake. If flow into the joint is confirmed in the absence of vascular uptake, the injectate is administered. The injectate is most often a combination of an anesthetic and steroid; however, the small volume of the joint only facilitates a 1-1.5 cc injectate. Full technical description of the procedure is available in the second edition of the International Spine Intervention Society (ISIS) guidelines [23].

### 33.2.5 Medial Branch Blocks

Fluoroscopic imaging, starting with an AP view and transitioning to more oblique views as



Fig. 33.2 L5/S1 zygapophyseal joint injection with intra-articular contrast

needed, is used to visualize the target (Fig. 33.3). The target is the space between the superior border of the transverse process and the mamilloaccessory notch at the junction of the superior articular process and the transverse process. The superficial skin can be anesthetized with a small amount of local anesthetic, followed by the use of a spinal needle, which is then introduced and directed toward the target until a bony stop is reached. Care must be taken to avoid needle placement that is too far posterior along the bulk of the superior articular process (SAP) itself or placement too far lateral along the transverse process itself. AP and lateral views must be used to confirm correct positioning (Fig. 33.4). A small amount of contrast is injected while visualizing under live fluoroscopy to assure proper placement and the absence of venous flow (Fig. 33.5). If vascular flow is encountered, the needle should be repositioned and contrast reinjected to assure that the flow has abated. Next a small amount of concentrated anesthetic (0.3-0.5 mL) is injected. Larger volumes should be avoided so as to minimize unintended spread of anesthetic. Targeting of the L5–S1 facet joint is slightly different as the



**Fig. 33.3** Oblique view with needles over L4 medial branch and L5 dorsal ramus to anesthetize the L5/S1 facet joint



Fig. 33.4 Depth view for diagnostic medial branch blocks



**Fig. 33.5** AP view for diagnostic medial branch block with contrast placed to assure no aberrant vascular flow

L5 dorsal ramus crosses the sacra ala as opposed to the transverse process. The target for this injection is 5 mm below the superior junction between the sacral ala and the S1 SAP. When MBBs are performed according to the guidelines, no significant complications have been reported [23]. Technical complications such as thecal puncture only occur if the procedural guidelines are not followed and the needle is grossly misplaced. Full technical description of the procedure is available from ISIS [23].

The purpose of an MBB is to precisely deliver local anesthetic to anesthetize a medial branch and subsequently evaluate the patient's response. Therefore, careful documentation of the patient's pre- and post-procedure pain level is required. In general the patient must be experiencing his or her typical pain prior to the procedure. After the injection pain level should be recorded in a pain diary in a systematic fashion, at intervals of at least 30–60 min for the remainder of the day. Pain levels should also be recorded at less frequent time intervals over the course of the next few days. The utility in the test truly lies in the information obtained from a properly conducted block and not in the execution of the block itself [23]. Without an accurately maintained pain diary, a perfectly performed MBB loses all potential benefit. Maintaining an accurate pain diary in real time limits the potential for recall bias. The 2004 ISIS guidelines recommend a formal post-MBB assessment by an assessor other than the performing physician; however, this is not practical in most practices. Ideally relief based on a numeric or visual analog scale, patients should also track pain medication requirements and ability or inability to perform tasks that are typically pain limited.

# 33.2.6 Intra-articular Z-Joint Injections

In terms of diagnostic utility, intra-articular z-joint injections have fallen out of favor. Birkenmaier found that using pericapsular injection of anesthetic to predict response to denervation procedures had poorer outcomes than when MBBs were used [27]. Three randomized controlled trials that evaluated the use of intraarticular injections of local anesthetic as a prognostic tool have been published, two of which were equivocal [28, 29] and one that was definitively negative [30]. In addition to improved diagnostic utility, medial branch blocks are also used in place of intra-articular anesthetic blocks because they are easier to perform, safer to perform, more easily subjected to controls, and have proven therapeutic utility in that they predict response to radiofrequency neurotomy [23].

There is limited data on the therapeutic utility of intra-articular z-joint injections as well. In uncontrolled studies the success rate for lumbar intra-articular steroid injections has varied widely, with reported success rates between 18 and 63 % [31, 32]. Only two RCTs evaluating the efficacy of lumbar z-joint intra-articular steroid injections have been published. In one of the studies, 8 mL of injectate was used and compared intra-articular anesthetic, extra-articular anesthetic, or intra-articular saline and found no difference at 3 months between the groups [33]. However, since the z-joint can only hold 1–2 mL, the validity of this study is in question. The other study evaluated patients who had immediate pain relief from intra-articular anesthetic injection and randomized them to receive either intra-articular corticosteroid or intra-articular saline [34]. There was no difference between the groups at 1 and 3 months; however, at 6 months the intra-articular steroid group was statistically more likely to have improved. The authors attributed this to concurrent therapies received by certain patients in the study, even though the results at 6 months remained statistically significant even after assuming that all patients with concurrent treatment did not improve.

In 1996, Dolan reported that positive findings on single-photon emission computed tomography (SPECT) scans may correlate with greater levels of pain relief for up to 3 months following intra-articular z-joint steroid injections [35]. Similarly, Ahmad and Ackerman published a study of patients with low back pain and isolated z-joint inflammation seen on SPECT imaging and compared outcomes at 12 weeks between those that received intra-articular steroid plus anesthetic and those that received medial branch perineural steroid and anesthetic. The number of patients with greater than 50 % pain relief at 12 weeks was 61 % in the intra-articular group, which was statistically significantly greater than in the perineural group (26 %) [36]. A study by Pneumaticos also found that the patients with SPECT positive imaging had significantly better outcomes at 1 month and 3 months with intraarticular injections compared to patients that did not have facet joint abnormalities on SPECT [37]. However there was no difference between the groups at 6 months. Another recent study that compared the effect of intra-articular z-joint steroid injection with intramuscular steroid injection for presumed z-joint-mediated low back pain found slight increase in outcomes with respect to physical function and reduction in NSAID use in the intra-articular group [38]. However the effect size of these results was small, there was no control arm to the study, and all subjects in the intraarticular group received bilateral injections at L3–L4, L4–L5, and L5–S1 as opposed to selecting the joints thought to be most likely involved.

Others have reported that intra-articular steroids are no better than sham injections [21]. Given the mixed literature, it is likely that there is a subset of the population that does significantly respond to intra-articular steroids, that being those with joint inflammation seen on SPECT.

As opposed to steroids, hyaluronic acid is another injectate that has been studied for the treatment of z-joint-mediated low back pain. Hyaluronic acid has been used in other joints with osteoarthritis to theoretically improve the viscoelastic properties of the defective synovial fluid and thus decrease pain [39]. Fuchs et al. compared the efficacy of intra-articular hyaluronic acid versus intra-articular steroid with 6-month follow-up and found that both groups demonstrated significant improvement in pain scores and function over the 6-month period [40]. There were no significant differences between the two groups in any of the outcome measures except for faster onset of symptom relief within the steroid group. Unfortunately the study did not include a control arm, which given the limited data on the efficacy of intra-articular steroids limits the utility of such a non-inferiority design study. However, at the very least there is theoretical benefit of hyaluronic acid over steroids given the reduced side effect profile of hyaluronic acid compared to steroids.

Given the presence of a better diagnostic test and limited evidence on the therapeutic utility of therapeutic intra-articular facet injections, defined indications for facet joint injections have not been published.

### 33.2.7 Medial Branch Blocks

Medial branch blocks have a well-established diagnostic utility and are thus possibly indicated in a patient that has chronic or subacute low back pain thought to be mediated by structures innervated by the medial branches, in most cases the z-joints, and confirmation of this will alter management. Not all patients with low back pain require MBBs. Examples of patients not needing an MBB would be those with low-level pain that does not result in functional limitations, when the pain is thought to be due to other structures, when pain is still in the acute stages, or when the first step in treatment is conservative therapy regardless MBBs are not indicated [23]. It is also important to have ruled out more serious possible causes of low back pain such as infection and tumors, at least through history and examination if not through other diagnostic tests, prior to proceeding with MBBs as a diagnostic test.

To fully address the use of MBBs requires a more in-depth discussion about the false positives of MBBs, false negatives of MBBs, what constitutes a positive finding with a single MBB, the number of medial branch blocks required, and ultimately establishing cutoffs for what constitutes a true positive result from MBBs.

While a full discussion of radiofrequency ablation (RFA) is covered elsewhere in this textbook, the practical use of what level of response to MBB constitutes a positive MBB is directly tied to the ability of MBB to predict response to the efficacy of radiofrequency neurotomy. As the criteria for what constitutes a positive MBB become more stringent, the likelihood of successful RFA also increases. In general, when nearly complete relief of symptoms is required from MBB in order to proceed with RFA, outcome data on the efficacy of RFA is very strong [41, 42], whereas relaxed criteria for percent relief of symptoms from MBB have led to less profound outcomes in the subsequent RFA that were performed [28, 30, 43]. When looking at pain relief as a general topic however, studies have found that as little as 30 % pain relief is clinically meaningful in chronic pain conditions [44]. Other studies indicate that 50 % in pain reduction improves a patient's quality of life [45]. Elsewhere in the pain literature, 50 % pain relief is the most commonly used dichotomous outcome measure [46]. So while the academic benefit of using more stringent criteria for what constitutes a positive MBB and how this then lends itself to having more robust positive findings when investigating RFA, at least some consideration must also be made to the proponents of less stringent criteria for what constitutes a positive MBB even if this results in less robust pain relief in the less ideal patients for RFA in light of the fact of what other literatures suggest that is clinically meaningful to the patient. Regardless of these theoretical differences, on the most basic level, pain relief from lumbar medial branch blocks implies that the patient's pain is mediated by the anesthetized nerve, and if there is no pain relief, the target nerve is deemed not to be contributing to the patient's pain. Nonetheless, additional safeguards must be in place to minimize the chance of false positives. Consideration must also be made about possibilities of false negatives.

### 33.2.7.1 False Positives

Depending on how a positive response is defined, rates of false-positive results from single MBBs in the lumbar spine have been reported to be between 17 and 41 % [5, 47, 48]. Many argued that such high false-positive rate when only single MBB is used has rendered them invalid [48– 50]. To mitigate false-positive responses, both dual and triple blocks have been proposed. Dual anesthetic blocks include performing the procedure twice with two different anesthetic of differing duration and evaluating if the patient's pain relief is concordant with the duration of each anesthetic. In order to prevent false positives due to the patients' potential bias toward desiring a positive response, the patient should not be instructed as to the expected duration of relief if pain relief may be achieved. A triple block includes a dual block in addition to performing the procedure with saline as a placebo. Triple blocks are not commonly performed for a variety of reasons, including but not limited to cost, efficiency, and the ethics of performing invasive promedications. cedures with placebo ISIS recommends dual comparative blocks as a viable alternative to placebo-controlled blocks [23]. True positive finding with comparative local anesthetic blocks is when a patient reports duration of pain relief that corresponds to the expected duration of action of the anesthetic used, in essence that the patient experiences pain relief with both blocks but longer relief when the longer-acting anesthetic is used [51].

A study that looked at cervical facet pain found that while the specificity of dual comparative MBBs was 88 % specific, it was only 54 % sensitive [52]. Accordingly, a 12 % false-positive rate is much lower than when single blocks are used and is well within acceptable levels when compared to other diagnostic tests. However, the low sensitivity also suggests that false negatives must also be considered when performing MBBs. And while the risk of a false-positive test is that a patient may undergo an un-needed radiofrequency neurotomy, false negatives are potentially worse than false positives as they withhold potentially beneficial treatment from a patient.

#### 33.2.7.2 False Negatives

False negatives will theoretically increase as more stringent criteria are applied for what constitutes a positive MBB. There are many reasons a false-negative response may occur, including but not limited to concurrent pain generators, inaccurate technique, excessive or inadequate use of superficial local anesthetic, and vascular uptake of injectate [53]. Procedural-related pain may preclude the patient from properly identifying that their typical pain has been alleviated. Fortunately, there is evidence that a single-needle approach to block multiple medial branches reduces procedural-related pain compared to conventional multiple needle site entry techniques [54]. There may also only be partial pain relief in the instance that there are multiple pain generators. Pain can concurrently be originating from the contralateral side, additional segmental levels, or an alternate painful structure. The patient may have difficulty in recognizing relief of their z-joint-mediated pain if these other possible pain generators continue to cause pain, resulting in a false-negative response. Some argue that it is unreasonable to expect complete pain relief from MBB as z-joint degeneration and pain rarely occurs in isolation of other potential sources of LBP. Radiologic studies demonstrate that significant z-joint degeneration never occurs in the absent of disk degeneration [55]. Even more, loss of disk height can accelerate or precipitate z-joint degeneration [56]. Conversely other studies have suggested that multiple pain generators simultaneously contributing to pain occur in less than 5 % of low back pain patients [57]. Regardless, the argument can be made that if only partial pain relief is achieved by a MBB, simultaneous anesthetization of the other painful structure should still enable total pain relief and confirmation of pain generators [21]. Unfortunately, the literature and techniques with respect to anesthetic injections for diagnosis of other sources of low back pain are not established enough to put this theoretical argument into practice.

If the procedure is performed without the patient in their usual state of pain, eliciting enough pain relief to be significantly noticeable and thus be considered a positive test may be difficult or even impossible. Vascular uptake of the anesthetic during MBB can also contribute to false negatives, which has been documented in 3.5 % of lumbar MBBs [58]. The risk of vascular uptake can be mitigated by use of real-time contrast injection [23]. It has also been hypothesized that there are potentially anomalous innervations of the facet joint other than the medial branch blocks. In the original Kaplan study that found that anesthetizing the medial branches blocked painful responses from facet joint capsular distention in eight of the nine patients, the other patient has been postulated to have anomalous innervations [3]. If the target nerve is missed and not bathed in anesthetic, a false negative may also occur. One study that compared cervical (not lumbar) low-volume MBBs (0.25 cc vs 0.5 cc) found that the target nerve was missed 7 % of the time in both cases, but that unintentional aberrant spread occurred twice as often (38 % compared to 16 %) in the high-volume group [59]. Alternatively, another study that performed CT scans after lumbar MBB to evaluate location of contrast found that the target nerve was bathed in injectate in all 120 injections [25]. Regardless all MBBs should be done with low volumes (0.25-0.3 cc) with live fluoroscopy to assure the lack of venous uptake.

### 33.2.8 Facet Cyst Injections

Facet joints can develop cysts that can be readily seen on MRI imaging and if in the anterior aspect of the joint may result in radicular pain. If symptomatic, symptoms are often that of radicular pain or stenosis, likely because of proximity of the cyst to the spinal nerve. Rupture of the cysts can be attempted. One study achieved 72 % success defined by avoidance of surgery and improvement in symptoms. In the same study there was a 37.5 % recurrence rate, but 45 % of recurrences responded to repeat cyst rupture [60]. Another study found that 46 % of the time surgery could be avoided [61]. Technique of cyst rupture involves anesthetizing the spinal nerve using a transforaminal approach, followed by high-volume facet intra-articular injection with anesthetic and contrast.

# 33.3 Epidural Steroid Injections

By definition epidural steroid injections target the spinal nerves. They are thus indicated for radicular pain. The most common causes of radicular pain are intervertebral disk herniations and spinal stenosis. It has been shown that the degree of nerve root compression does not correlate to the level of pain [62-64]. It has also been shown that pure mechanical compression of the spinal nerves produces paresthesia and motor weakness but not pain [65]. As such, it reasons that radicular pain must be caused by additional factors in addition to, if not exclusive of, root compression. Multiple studies have shown that inflammation is an essential component to the painful symptoms experienced in patients with radiculopathy [13, 66–68]. Inflammatory mediators such as phospholipase A2, prostaglandin E2, leukotrienes, nitric oxide, immunoglobulins, and cytokines such as interleukin-6 and tumor necrosis factor alpha are all involved in the inflammatory component of radiculopathy [69–71]. Various biochemical inflammatory markers must be present in order for the dorsal root ganglion to generate painful discharges [72]. Moreover, many of these inflammatory mediators such as phospholipase 2 have been found within the nucleus pulposus itself and are found in high concentrations along with inflammatory cells such as macrophages at sites of disk herniation [68, 73, 74]. Histopathological findings consistent with inflammation have also been found in nerve root

specimens taken from decompression surgery [68, 75]. Corticosteroids inhibit phospholipase 2 and leukocyte aggregation at sites of inflammation; prevent degranulation of granulocytes, mast cells, and macrophages and transmission of nociceptive C-fibers; and stabilize ectopic discharge of neuronal membranes [70, 76, 77]. As such local administration of corticosteroid can theoretically result in symptom relief [78].

### 33.3.1 Contraindications

Contraindications include bleeding diathesis, local infection at the injection site, systemic infection, cardiovascular instability, uncontrolled diabetes, uncontrolled glaucoma, cauda equina syndrome, pregnancy, and allergy to local anesthetic or steroid medication.

## 33.3.1.1 Types of Epidural Steroid Injections

# Caudal and Interlaminar Epidural Steroid Injection

Caudal epidural steroid injections (CESI) involves administration of steroid into the epidural space through the sacral hiatus. In 1957 Cyriax reported the first use of caudal epidural steroid injection for pain relief [79]. The first interlaminar epidural injection was described by Pages in 1921 [80]. Interlaminar injections involve delivery of medication into the posterior epidural space between the dura anteriorly and ligamentum flavum posteriorly. Both of these techniques have limited literature on their efficacy and have thus somewhat fallen out of favor when compared to transforaminal epidural steroid injection. In light of this a full description of their techniques is not warranted in this chapter.

#### Transforaminal Epidural Steroid Injection

The first reported TFESI was described in 1952 by Robecchi and Capra [81]. TFESI delivers steroid to the epidural space in close proximity to the affected nerve root. Compared to other approaches, it is more targeted in bathing the affected spinal nerve root close to its dorsal root ganglion and theoretically maximizes the therapeutic effect. Derby first postulated that a transforaminal approach theoretically is superior because it can provide a high concentration of injectate directly to the posterior annulus and the ventral epidural space [82].

#### **General Technique**

The patient is positioned prone and sterilely draped. The skin is anesthetized with local anesthetic. Using the AP view the transforaminal space that is the target is identified by first squaring off the inferior end plate of the target level. Then using an oblique approach the spinal needle is advanced under the pedicle, making sure not to pass medial to the 6 o'clock position of the pedicle when viewed from a true anterior-posterior (AP) view (Figs. 33.6 and 33.7). The needle should be deep to the lamina on the lateral view and in the safe triangle on the AP view. The safe triangle is composed of the base of the corresponding pedicle, lateral border of the vertebral body, and lateral border of the exiting nerve root. Needle placement must be confirmed by both AP and lateral views. The AP view is used to confirm that the needle has not been placed too medially which increases risk of dural puncture (Fig. 33.7). The lateral view is used to confirm depth (Fig. 33.8). Contrast medium is then injected using live fluoroscopy and evaluated for transforaminal epidurogram spread and assuring that no intravascular or intrathecal spread has occurred (Fig. 33.9). After confirming appropriate contrast, spread small-volume injectate, usually consisting of 1–2 cc of anesthetic followed by 1–2 cc of steroid [23, 78].

#### **Complications and Side Effects**

Overall incidence of complications for TFESIs ranges between 5.5 and 9.6 % [83, 84]. Rates are higher for multilevel injections versus single-level injections [83]. The most common side effects include injection site pain, vasovagal reaction (3.5 %), increased radicular pain, light-headedness, increased pain caused by direct trauma to the spinal nerve, nausea, non-positional headache, vomiting, facial flushing, and elevated blood pressure [78, 84–86]. Anesthetic



Fig. 33.6 (a) Transforaminal oblique view (without needle). (b) Transforaminal oblique view (with needle)



Fig. 33.7 TF AP view



Fig. 33.9 TF AP post-contrast injection



Fig. 33.8 TF lateral view

medications, contrast, and steroids can all cause allergic reaction. Steroids can also cause myopathy, fluid retention, hypertension, mood abnormalities, menstrual irregularities, hyperglycemia, and iatrogenic Cushing syndrome [87–90].

Bleeding is another possible complication. Patients with coagulopathy or on anticoagulation medications are at increased risk of bleeding complications [91–93]. Very rarely does bleeding result in epidural hematoma and compression of the spinal cord and spinal nerves, reported to occur 1 in 150,000 injections [94, 95]. Surgical evacuation is warranted in the rare event it does occur.

Infection is another known risk of all spine injections. Infection risk includes epidural abscess, diskitis, osteomyelitis, and meningitis [96–100]. Given proximity to the pelvic and abdominal cavity, gram-negative infections are more likely. If the needle is advanced too far ventral or lateral, there is also a risk of abdominal cavity puncture leading to infection [101]. Serious infection is extremely rare, occurring only 0.001–0.1 % of the time. If present, serious infections require surgical intervention 70 % of the time and often do not fully recover [102]. Fifty-three percent of the time, an infection presents as worsening pain, most often around 7 days postinjection [102].

Dural puncture is another potential complication of TFESI [103]. Dural puncture can result in positional headache. If not identified, as evidenced by flashback of CSF or by recognition of poor positioning on fluoroscopy, intrathecal administration of anesthetic can cause cauda equina, arachnoiditis, or meningitis [101]. Intradiscal injection can occur during TFESI [104–106]. The primary concern with intradiscal injection is diskitis, and prophylactic antibiotics are usually given if this complication is encountered.

Rate of intravascular injection has been reported as high as 11.2 % for all lumbar TFESIs and as high as 21.3 % for S1 TFESIs [107] (Fig. 33.10). The risk of intravascular injection is double in patients over 50 years old [108]. Most of these injections are venous in nature. The real concern is intra-arterial injection into the spinal radiculomedullary arteries. The artery of Adamkiewicz, which supplies the anterior third of the spinal cord, is often implicated in intraarterial injections due to its location in the neural foramen. There is variability in the anatomy of the artery of Adamkiewicz as it is located on the left 63 % of the time and is between the T9 and L2 level only 85 % of the time [109–111]. Intraarterial injection with a particulate corticosteroid



Fig. 33.10 Venous flow on S1 TF ESI

during TFESI has been reported to cause spinal infarction and subsequent paraplegia [109–111].

Smuck reported that intermittent fluoroscopy only identifies 57 % of vascular injections as opposed to continuous fluoroscopy [112]. Even more concerning is that confirmation of epidural spread does not rule out concomitant vascular uptake [107, 112]. Digital subtraction angiography can also be used in adjunct with continuous fluoroscopy to further enhance the ability to detect intravascular flow [101, 113].

An anesthetic test dose can also be used to reduce the risk of intra-arterial injection. And anesthetic challenge dose involves administration of anesthetic such as lidocaine after needle position has been confirmed with contrast injection and evaluating for patient response. Reported symptoms such as tinnitus, metallic taste in mouth, headache, dizziness, and sensorimotor changes in either all four or bilateral lower extremities are suggested of intra-arterial infiltration. If positive the procedure should be terminated. Despite these safeguards, irreversible paraplegia has been reported even when continuous fluoroscopy, digital subtraction angiography, and anesthetic test dose have all been implemented [114]. In addition to potentially catastrophic events, intravascular uptake may also reduce the efficacy of TFESIs [108, 115]. Additionally the use of a non-particulate, preservative-free corticosteroid such as dexamethasone could also reduce the risks of inadvertent intra-arterial injection.

## 33.3.2 Dosing and Number of Injections

No standard dose of steroid exists for TFESI though in a recent comprehensive review of the literature MacVicar reported in most studies that investigated TFESI used either low (40 mg)-dose methylprednisolone or high (80 mg)-dose methylprednisolone, equivalent dosing of triamcinolone or betamethasone, and that less extensive use of dexamethasone has been found in the literature [116].

There is no current literature that specifically investigates the ideal number of injections to achieve maximal benefit. However, MacVicar pooled the number of injection data from all studies that reported categorical data on patients that achieved at least 50 % pain relief (totaling 9 studies with a total of 727 injection included) and revealed that  $94 \pm 2$  % of patients with successful outcomes from TFESI did so with only 1 injection [116]. Of the 15 patients that had relief in a study by Ghahreman, only 5 required a second injection, and none required more than 2 [117]

# 33.3.3 Evidence-Based ESI for Radicular Pain Due To Disk Herniation

### 33.3.3.1 CESI Efficacy

The data on CESI for disk herniation is quite limited. Even more problematic is that the majority of available studies utilize blind CESI. Blind interlaminar and blind caudal approaches demonstrate a 30–40 % rate of missing the epidural space [2, 118]. Current standard of care dictates that fluoroscopy be used for such injections. This further minimized the usefulness of available literature. Also worth considering is that spread of injectate via CESI is at best up to L3–L4 and more likely only up to the L4–L5 level and that L4–L5 is the most cephalic level of pain generation that has been reported to be amenable to treatment with caudal injection [119–121].

The first evidence that CESI may be beneficial for radicular pain was published in 1971 [122]. In 1987 Matthews published results of a series of three blind CESI in patients and showed benefit in pain reduction at 3 months but not 1, 6, or 12 months [123]. It was not until Bush and Hiller published a randomized placebo-controlled study in 1991 that more significant evidence became available. They reported significant gains in mobility and quality of life at 4 weeks in the group that received CESI compared to placebo [124]. Unfortunately, the study did not differentiate between radicular pain due to stenosis and disk herniation and was limited by a very small sample size (n=23). Moreover, the procedures were performed without fluoroscopy. At 1 year differences between the groups were no longer present, as the anesthetic-only control group demonstrated similar gains by that time [124].

Dincer et al. studied the efficacy of blind CESI compared to 1 month of NSAID therapy for radicular pain due to disk herniation in 64 patients and found that the CESI group had statistically greater improvement in VAS at 2 weeks, 4 weeks, and 12 weeks and in Oswestry scores at 2 and 4 weeks [125].

Another study was designed to evaluate if targeted placement of steroid using endoscopically placed steroid around the affected nerve root had greater effect compared to less targeted steroid placement via fluoroscopically guided CESI. It evaluated patients with radicular pain but excluded those with "chronic stenosis" as defined by symptoms of 18 months or longer. Both groups showed significant improvement in pain at 6 weeks, 3 months, and 6 months but no difference between the two groups [126]. While there was no control group, the results of the CESI group can be evaluated as a cohort independent of the endoscopic group, and the study can be used as evidence that CESI can produce favorable improvements in pain for up to 6 months in patients with radicular pain of less than 18 months.

There are no studies that directly evaluate the efficacy of fluoroscopically guided CESI for radicular pain due to disk herniation. At best, reviewing the available literature including the blind CESI and group subset analysis of other studies shows that there is evidence, while minimal, that fluoroscopically guided CESI may provide short-term pain relief for acute and subacute radicular pain.

### 33.3.3.2 Interlaminar Epidural Steroid Injection (ILESI) Efficacy

There is limited literature that supports the use of ILESI. In perhaps the best designed study aimed at evaluating ILESI, Carette published in the NEJM a randomized double-blind trial in 158 patients with radicular pain due to herniated nucleus pulposus that compared blind ILESI of methylprednisolone to interlaminar injections with normal saline and found that there was significant improvement in leg pain in the steroid group at 3 weeks and 6 weeks, but by 3 months

there was no difference between the groups. All other parameters evaluated including ODI did not reveal any differences between the two groups at 3, 6, or 12 weeks. Only group mean data was evaluated; no categorical data was included in the study [127]. In 2003 Valat et al. published a randomized double-blind control trial comparing blind interlaminar saline to blind interlaminar steroid for radicular pain due to "presumed" disk herniation with 85 patients total. Primary outcome was whether or not patient deemed their symptoms "resolved" or "markedly improved" at day 20, whereas "worse" and "slight improvement" were deemed as failure. Use of NSAIDs or surgery was also considered failure. At day 20 there was no significant difference between the groups with an intention to treat analysis; however, once patients that were lost to follow-up or excluded (11 of the 85 subjects) there was a strong trend (p=0.054) toward treatment "success" in the steroid group compared to the saline group. By day 35 the groups were found to have equal outcome [128]. In 2009 Parr reviewed the best available literature for interlaminar epidural injections for low back pain, including the two studies mentioned above, and concluded that the evidence is "limited for blind interlaminar epidurals in managing all types of pain except for short-term relief of pain secondary to disk herniation and radiculitis" [129]. Perhaps more importantly though, Parr noted that the evidence identified does not represent contemporary interventional pain management practices given that none of the studies identified utilized fluoroscopy. Strictly speaking, the evidence may not be extrapolated to fluoroscopically directed lumbar interlaminar epidural injections. Currently there continues to be a paucity of literature that is investigating, much less in support of, fluoroscopically guided interlaminar steroid injection. This is most likely in large part due to the major shift in clinical practice toward transforaminal epidural steroid injections.

### 33.3.3 TFESI Efficacy

Certainly the literature for caudal and interlaminar epidural steroid injections is limited and in many instances has not found these interventions to be

more effective than sham treatments [127, 128, 130]. However, careful review of the available literature that specifically investigates transforaminal injections for radicular pain reveals significant and positive findings, most dramatically for herniated disk pathology. Unfortunately, some systematic reviews consider all types of epidural steroid injections as equivalent and have thus inappropriately dismissed the reported effectiveness of a TFESI [131]. Promising research evaluating the use and efficacy of TFESIs has been more abundant over the last 20 years. Some of the earlier studies demonstrated a significant surgical sparing effect of TFESI for herniated disks causing radicular pain. In 1997 Weiner and Fraser reported that 27/30 patients with severe lumbar radiculopathy had immediate pain relief after TFESI, and 22 of the 28 (79 %) patients available for longer-term follow-up had considerable and sustained relief [132]. In a randomized controlled double-blind study by Riew in 2000 that evaluated the surgical sparing effect of transforaminal anesthetic compared to transforaminal anesthetic plus steroid in 55 patients with lumbar radicular pain due to either canal stenosis or disk herniation that were scheduled for surgery, the group that received anesthetic plus steroid deferred surgical intervention 71 % of the time compared to only 33 % of the patients that received anesthetic alone [133]. This effect was maintained for 5 years [133]. The publication reported stratifying the data based on patients with stenosis and those with lumbar disk herniation but did not present the raw data nor did they comment on the efficacy of TFESI in preventing surgery based on pathology (stenosis vs disk herniation). However, they did report that in the group that avoided surgery, which was predominantly composed of patients that received both steroid and anesthetic, that the patients with stenosis had "significant relief of low back pain" (p < 0.008) and those with disk herniation trended toward "significant relief of low back pain" (p < 0.07) [133]. The surgical sparing effect of TFESI was also demonstrated by Wang in 2002 [134]. In a retrospective review of 69 patients with symptomatic herniated lumbar disks whom had failed conservative management with anti-inflammatories and physical

therapy who were now requesting diskectomy, only 16 (23 %) eventually went on to having surgery after receiving between 1 and 6 TFESI with a mean follow-up of 1.5 years [134]. All three studies combined provide strong evidence that for patients with radicular pain due to herniated nucleus pulposus, TFESI is an effective means of preventing surgical intervention in a significant amount of patients.

Beyond a surgical sparing effect, the literature has also supported the use of TFESI as a means of providing symptomatic relief. 50 % pain relief has been established as what patients considered a "much improved" for pain in general as well as the minimal clinically important change for radicular pain [44, 135]. Subsequently, much of the literature on ESI has appropriately used this to define what is categorically considered to be a "positive response." Another important consideration to make when evaluating studies with ordinal data such as VAS is that if mean data is used for statistical analysis of outcomes, the outcomes must be in a normal distribution. Otherwise the data should have the mode and interquartile range evaluated, not the mean. The lumping of responders and nonresponders can result in group mean scores that are worthless. Rarely are pain scores distributed in a normal bell-shaped distribution. This emphasizes that the need or importance of using categorical outcome, predefining what success is and who achieves success, is a vital step in evaluating these types of studies.

There are multiple studies that evaluated patient cohorts of various sizes that demonstrated significant pain relief in varying percentages of patients ranging from 38 to 75 % [136–140]. In 1998, Lutz reported a prospective case series of 69 patients in which "52 of 69 (75 %) patients had a successful long-term outcome, reporting at least a >50 % reduction between pre-injection and post-injection pain scores, as well as an ability to return to or near their previous levels of functioning after only 1.8 injections per patient" with a mean follow-up of 80 weeks [136]. More recently, a large retrospective study of 2,024 patients undergoing single lumbar TFESI for radicular pain either due to disk herniation or foraminal stenosis reported that 45.6 % had at least 50 % pain

relief at 2 months. For patients with less than 3 months of pain, the success rate increased to 68.3 % at 2 months [137]. Unfortunately, data specifically on disk herniations or lumbar stenosis was not presented separately. In a study by Cyteval, when looking only at the subgroup of patients with radiculopathy solely due to disk herniation who had failed conservative therapy, 65 of 172 patients (38 %) had at least 50 % pain relief at 2 weeks, and 88 % of these had continued relief through 1 year [138]. Yet another study, by Narozny, found that 12 of 20 (60 %) patients with radicular pain due to disk herniation had at least 60 % pain relief at 6 months after TFESI [139]. Jeong reported on two different transforaminal approaches for lumbar radicular pain due to either central canal stenosis or disk herniation. Overall, at 4 weeks 99 of 122 (89 %) in the preganglionic approach group had good or excellent results (defined by at least 50 % reduction in pain) compared to 90 out of 127 (70 %) in the ganglionic group [140]. The difference between the two groups was no longer significant at 6 months. Overall analysis of the entire group as a single cohort of patients with radicular pain due to disk herniation that received TFESI is valuable in this case though. Overall there was still good or excellent response in 142 of the 222 subjects (64 %) for 6 months [140]. In the subset of patients with radicular pain due to disk herniation, 118 of 191 patients (62 %) had at least 50 % pain relief at 6-month follow-up [140].

As mentioned, dangerous pitfalls arise in randomized controlled trials when comparing mean pain scores from studies with non-normally distributed data as opposed to reporting predefined categorical data. A randomized controlled trial that evaluated the efficacy of transforaminal steroid versus transforaminal saline, first published in 2001, did not find benefit for TFESI at 3 or 6 months when analyzing mean group data [92]. However with subgroup analysis of the same data evaluating radicular pain due to contained herniations, the steroid group was found to have shortterm benefit for radicular pain and disability [141]. At 1 year, TFESI was found to prevent progression to surgery within the same subgroup of disk herniations and when compared to the control group found a saving of \$12,666 per responder on average [141]. Additional well-designed studies that consider predefined categorical outcomes have repeatedly demonstrated convincing data in support of the use of TFESI for disk herniation causing radicular pain. In 2002 Vad published a randomized (patient selected) controlled trial that compared the efficacy of TFESIs to trigger point injections for lumbosacral radiculopathy secondary to herniated nucleus pulposus [142]. Successful outcome was categorized as predefined improvement in all three categories of patient satisfaction, Roland Morris score, and VAS pain reduction. "After an average follow-up period of 1.4 years, the group receiving transforaminal epidural steroid injections had a success rate of 84 %, as compared with 48 % for the group receiving trigger point injections" [142]. The best designed study to date that investigated the efficacy of TFESI was by Ghahreman et al. in 2010. The study was a prospective, randomized study with five arms that compared TFESI to TF local anesthetic, to TF saline, to intramuscular steroid, and to intramuscular saline and used categorical outcomes to evaluate success. They found that a significantly greater proportion of patients in the TFESI group (54 %) achieved at least 50 % pain relief at 1 month compared to the other four arms (ranging between 7 and 21 %) [117]. Pain relief was "corroborated by significant improvements in function and disability and reduction in the use of other health care" in the TFESI group. Long-term relief at 1 year was also greater in the TFESI group (25 %) than the other four groups; however, the results did not reach statistical significance. The authors of the study argued that 25 % success rate at 1 year is "patently cost-effective" considering that the cost of the alternative (surgery) is much greater than the cost of a single injection. The study also found that using transforaminal saline as a control/placebo compared to TFESI, the number needed to treat to obtain at least 50 % pain relief at 4 weeks is only three [117]! Lastly, the design of the study also makes evident that both the medication (steroid) and route of delivery (transforaminal) combine to form a compound intervention that is unique and different from systemic steroids and from transforaminal administration of other compounds [117].

In 2013 MacVicar et al. published a thorough and comprehensive review of the literature with accompanying systematic analysis of all published data regarding TFESI [116]. After plotting the success rates of outcome studies, pragmatic studies, and explanatory studies, MacVicar et al. summarized it best in saying that with regard to TFESI for radicular pain due to disk herniation, the literature is "abundant" and of "higher quality" and reveals that "about 60 % of patients seem to achieve at least 50 % relief of pain at between 1 and 2 months but that only 40 % maintain this outcome for 12 months" [116]. Specifically, their conclusions included that TFESIs are effective (more so in patients with contained disk herniations, low-grade compression, and acute symptom duration) [140, 143–145], are statistically more than placebo effects [117, 133], reduce the burden of disease by improving function [117, 142, 146] and reducing need for surgery [117, 133, 134], and ultimately are cost-effective [141].

### 33.3.4 Predictors of Response to TFESI

In the prospective case series published by Lutz, in patients who had pain for less than 36 weeks, the success rate was nearly 80 % versus only 65 % in patients with symptoms that present longer for longer durations [136]. Similarly, in the study by Jeong, univariant analysis revealed that pain of less than 6 months had better therapeutic effect than those with greater than 6 months of symptom duration regardless of symptoms being due to stenosis or disk herniation [140]. In a large retrospective study of 2,024 patients undergoing single lumbar TFESI for radicular pain either due to disk herniation or foraminal stenosis, the proportion of responders was significantly higher when there was less than 3 months of pain that was present (odds ratio 2.42) [137]. In the study by Cyteval on TFESI for lumbar radicular pain due to either foraminal stenosis or disk herniation, patients that had at least 75 % pain relief had a mean duration of symptoms of 3 months compared to the group that had less than 25 % pain relief who had a mean duration of symptoms of

8 months at 2-week follow-up [138]. The review article by MacVicar also pooled data from three studies that provided data on duration of radicular symptoms, in addition to other inclusion criteria, and its effect on success rates of TFESI. They concluded that while there is a statistical difference that exists of patients with pain present less than 6 month being more likely to have a positive response, the clinical effect is negligible, concluding that 70 % of patients with acute pain can expect to benefit, but up to 60 % of patients with chronic pain can still benefit as well [116]. Also of note though is that the 95 % confidence intervals between the two groups in the combined data overlapped [116].

In a review of 71 patients with lumbar radicular pain due to disk herniation treated with TFESI, clinical and radiological features were assessed for predictors of positive response, defined by at least 50 % reduction in pain 1 month postinjection [147]. The only feature that was found to be significant was grade of nerve root compression. Low-grade nerve root compression responded favorably 75 % of the time as compared to only a 26 % response rate with high-grade compression. For paracentral disk herniation, high-grade compression that is associated with poor response was defined as obliteration of periradicular CSF or fat or morphologically distorted nerve root. For foraminal/far lateral herniation, high-grade compression was defined as perineural fat obliteration in all four directions or morphologic distortion of the nerve root. Duration of symptoms, presence of neurologic symptoms, abnormal neurologic findings on exam, level of herniation, location of herniation, and morphology of disk herniation were all evaluated and not found to be a predictive response to TFESI [147]. Alternatively, in the original Ghahreman study, there was no association between the need to progress to surgery and the size of the disk herniation [117].

### 33.3.5 Comparative Studies

It is clear that the evidence in support of TFESI for relief of radicular pain is robust and definitive compared to the level of evidence available in support of CESI and ILESI. This has driven clinical practice largely toward predominately using this approach. Additionally, there is evidence that has attempted to directly compare the efficacy of TFESI, CESI, and ILESI.

In a retrospective review of 40 patients with radicular pain due to herniated lumbar disk, Schaufele found that 14 of 20 (70 %) of patients that received TFESI had improvement of at least 2 on a 0–10 pain score compared to only 9 of 20 (45 %) in those that received fluoroscopically guided ILESI. Follow-up was limited to 19 days only [148]. A prospective randomized trial by Thomas that compared fluoroscopically guided TFESI to blind ILESI for radicular pain secondary to lumbar disk herniation in 31 patients found that mean VAS pain score was statistically significantly lower (p < 0.04) in the TFESI group (VAS improvement 56.8 mm) than in the ILESI group (VAS improvement 40.3 mm) at 6 months [119]. The study is limited by small sample size, ILESI being performed blind, and lack of categorical data. Nonetheless, this is valuable evidence in support of the superiority of TFESI over ILESI given with a prospective and randomized design. A similar prospective randomized trial with 64 patients that compared fluoroscopically guided TFESI to fluoroscopically guided ILESI in patients with radicular pain due to lumbar disk herniation was published by Rados in 2011. Comparing the mean pain scores and mean Oswestry scores between the two groups found that while both groups improved at 6 months, there was no statistical difference between the groups [149]. At 6 months, when comparing the percentage of patients that improved by at least two points on VAS (TFESI 84 % vs ILESI 75 %), by at least 50 % on VAS (63 % TFESI vs ILESI 53 %) or ten-point improvement on Oswestry (TFESI 66 % vs ILESI 50 %), there was again no statistically significant difference, but there was a noticeable trend toward more favorable outcomes with TFESI [149]. Collectively, the three studies support the use of TFESI over ILESI for radicular pain.

Lee reported a retrospective nonrandomized study of 233 patients who had failed conservative therapy that compared the efficacy of TFESI versus CESI versus ILESI for radicular pain caused by either disk herniation or spinal stenosis [150]. When evaluating patients with at least 50 % pain relief at 2 months, Lee found that both TFESI and ILESI were superior to CESI for patients with either disk herniation or spinal stenosis [150]. Further evaluation of the subgroup data only in patients with radicular pain caused by disk herniation, 25 out of 38 (65.8 %) had at least 50 % pain relief at 2 months when treated with TFESI compared to only 3 out of 14 (21.4%) in those treated with CESI and 16 out of 31 in those treated with ILESI (51.6 %) [150]. Overall, the study provides relatively strong evidence that CESI is inferior to ILESI and TFESI, and while not statistically significant, also weakly corroborates other available evidence that TFESI is likely superior to ILESI.

Ackerman evaluated 90 patients with L5/S1 disk herniations with S1 radicular pain who were randomly assigned to CESI, ILESI, and TFESI and followed for 24 weeks [145]. The number of injections in that time period ranged between 1 and 3. At 12-week follow-up, there was a trend toward TFESI being more likely to provide partial or complete relief than CESI or ILESI, but the results were not statistically significant. At 24 weeks both CESI and TFESI were statistically more likely to provide "complete relief" compared to ILESI, but there was no difference between CESI and TFESI [145].

Of the available studies, three suggest that TFESI is superior to ILESI, and two demonstrate more robust and statistically significant evidence of this. In the two studies that also included CESI, TFESI was found superior to CESI in one and equivalent in the other. Looking at the direct comparative data as whole, it becomes readily evident that TFESI provides the best outcomes of the three approaches. Ackerman also evaluated contrast spread pattern and found a statistically significant relationship between "complete pain relief" and ventral spread of contrast. A transforaminal approach was the approach most likely to achieve ventral spread [145]. This provides at least some theoretical support that explains why TFESI is more efficacious.

### 33.4 Lumbar Intervertebral Disks

Intervertebral disk pathology can play a role in low back pain [151]. Disks are innervated with nociceptive pain receptors by branches of the sinuvertebral nerves, gray rami communicantes, and lumbar ventral rami [152–155]. The greatest concentration of nociceptors is found within the posterior annulus fibrosus. In general the outer third of the annulus fibrosus has the greatest innervations, with there being only some innervations of the middle third and little to no innervation of the inner third [152–155]. Discography is defined by the injection of contrast into the nucleus pulposus to evaluate disk morphology and was first described in the 1940s [156–158]. This led to the development of provocative discography, which is the injection of contrast medium into the nucleus pulposus of a disk in an attempt to recreate a patient's pain and thus diagnose the ultimate source of the pain. In provocative discography, fluoroscopic imaging is used to confirm placement of contrast as opposed for morphologic evaluation of the disk. If further anatomic confirmation of positive provocative test is desired, pursual of subsequent MRI or CT scans of the injected disk can then be pursued. In large part this evolution from using discography for morphologic evaluation to provocative discography to assess for patient response to the procedure itself occurred because of a need to identify which one of the multiple morphologically abnormal disks was the primary pain generator, as often many of the morphologically abnormal disks were not painful with provocation [159].

# 33.4.1 General Principles for Provocative Discography

The general principle of provocative discography is that pressurizing a disk, which is the patient's pain generator, will reproduce that pain as opposed to stressing an asymptomatic disk, which should either be pain-free or generate unfamiliar pain. Delineation between reproducing the patient's pain, termed concordant pain, and generating other pain, termed discordant pain, is important. Because the test is provocative, there is an inherent increased risk of false positives as provocative maneuvers. This necessitates internal controls for provocative discography, which have historically been a provocation of intervertebral disks adjacent to the target level. Thus according to both the International Association for the Study of Pain and International Spine Interventional Society, the diagnosis of discogenic pain requires that provocation of the target disk reproduces the patient's concordant pain and that provocation of adjacent disk does not reproduce concordant pain [23].

### 33.4.1.1 Procedural Considerations

### Indications

The main purpose of this injection is to accurately diagnose the pain generator for potential treatment purposes. There are currently no studies available that demonstrate a better response to a given therapy in subjects with discography-proven disks as opposed to those selected by clinicians. Until appropriate medications or interventional treatments have been found, the ultimate role for provocative discography may be in surgical planning. This is especially true if the surgeon is trying to avoid fusion, or if surgery is already being planned as discography has been used in assessing if additional levels should be fused [160–163].

#### Contraindications

Absolute contraindications: patient unwilling or unable, inability to assess patient response during procedure, untreated localized infection, and pregnancy

Relative contraindications: allergy to materials involved, bleeding diathesis or anticoagulation therapy, systemic infection, and anatomical derangements that may make the procedure unsafe

### Antibiotic Prophylaxis

Antibiotic prophylaxis such as cefazolin 1 g, clindamycin 900 mg, or ciprofloxacin 400 mg IV should be used. Reported rates of infectious

diskitis are from 0.05 to 1.3 % per disk injection without the use of prophylactic antibiotics [23]. Meta-analysis reviewing the risk of diskitis with discography reveals a 0.24 % risk without antibiotic use and no reported cases when antibiotics are used [164]

#### Immediate Procedural Complications

Short-term complications include allergic reactions, vasovagal reactions, increased pain, bleeding, superficial infection, and penetration of the ventral ramus. More significant of a concern is infectious diskitis.

### Long-Term Complications

A 10-year matched cohort evaluated asymptomatic patients that underwent three-level provocative discography and MRI evaluation and were compared with matched cohort that underwent MRI evaluation but no discography [165]. MRI evaluation looked at degenerative changes, Modic changes, and herniations again 7-10 years later. All graded MRI parameters demonstrated that disks that had been exposed to discography had significantly higher progression toward disk pathology [165]. Even more, new herniations were found to be significantly associated with the ipsilateral side of disk injection [165]. Using raw data to calculate 95 % confidence intervals, as opposed to the published chi-squared analysis, rendered the association with Modic changes and degenerative changes less statistically significant [166].

#### Technique

An AP view of the lumbar spine is used to properly identify the target disk. Often an adjacent disk, which is not to be the primary pain generator, is selected first. From the AP view, the fluoroscope is transitioned to an oblique view in order to visualize the ipsilateral superior articular process of the inferior portion of the target segment, with one-third of the disk being posterior to this point. Using a two-needle technique to decrease the likelihood of infection, the needle is then advanced over the inferior articular process and into the center of the intervertebral disk, while being careful to avoid the ventral ramus. In general, the needle is introduced from the side opposite to that of the predominant side of pain in order to avoid false positives via iatrogenic procedural pain. It is essential to confirm accurate placement with both anterior-posterior (AP) and lateral views (Fig. 33.11). Once proper position inside the center of the disk is confirmed, the stylette is removed, and a pressure transducer manometer in line with a closed system containing antibiotics and contract medium is connected. Contrast medium is then slowly injected under live fluoroscopy (Figs. 33.12 and 33.13). Once contrast media is visible, this corresponds to the opening pressure that is then recorded. Opening pressure should be noted, as low opening pressure can be suggestive of annular tears, and increased opening pressure can be suggestive of incorrect needle placement. The pressurecontrolled intradiscal injection continues until one of the four situations in encountered: the patient's low back pain is reproduced, contrast escapes the disk, pressures exceed 100 psi, or >4 mL of contrast is injected. If pain is reproduced, the pressure at which this occurred should be recorded. During the procedure sham provocation should be performed as well, in which the patient is told that pressurization is occurring when it is in fact not to in order to see if pain is reproduced without provocation. Full procedural guidelines are available through the ISIS guidelines [23].

#### Interpretation of Results

The volume of fluid injected is proportional to the pressure applied. It reasons to suggest that any disk can be painful if enough pressure is applied, which would lead to false positives. Therefore it is important to define the pressures used during provocative discography. It is suggested that thresholds for pressure that cause concordant pain be considered when it is experienced with pressure less than 50 psi or less than 15 psi above opening pressure [23]. Pain intensity must also be considered, and if only minor pain is elicited, then the increased likelihood of false positives must be considered. This has led some authors to suggest that only a pain intensity of 7/10 should be considered positive [23].



Fig. 33.11 Intradiscal AP pre-contrast



Fig. 33.12 Intradiscal lateral post-contrast

Obviously, the more strict the criteria that are used to consider a response as positive, the lower the false-positive rate becomes.



Fig. 33.13 Intradiscal AP post-contrast

#### **False Positives**

Holt published a study of asymptomatic volunteer prisoners that underwent provocative discography and found a 37 % false-positive rate; however, this study has since been refuted [167, 168], and false positives have been reported to be as low as 1 % with correct technique [169]. One systematic review with meta-analysis reported that the false-positive rate did not exceed 10 % and was possibly as low as 6 % [170]. There has been much research and debate about the falsepositive rate of provocative discography. Because of the provocative nature of this technique, there is an inherent theoretical risk of higher rate of false positives. It has also been theorized that asymptomatic individuals cannot have false positives identified as concordant pain because they do not have baseline pain [171].

Carragee reported false-positive rates of 10 % (95 % CI 0–29 %) in asymptomatic patients, 40 % (95 % CI 10–70 %) in chronic pain patients, and 75 % (95 % CI 33–100 %) in somatization patients [172]. Another study looking at only asymptomatic patients found that no one experienced pain at any level with pressures of 20 psi or less suggesting that in asymptomatic patients using a threshold of 20 psi virtually eliminates

false-positive response [166, 173]. Proponents of discography often point to the asymptomatic groups in such studies as proof that the falsepositive rate of provocative discography is acceptably low. However, asymptomatic patients should not be undergoing provocative discography in the first place. In light of this, the falsepositive rate for provocative discography must include the rate found in those with chronic pain. Bogduk et al. applied the manometric criteria from the aforementioned study by Derby and applied the aforementioned study by Carragee and determined that the prevalence of false positives decreases from 40 to 30 % in those with chronic pain and from 75 to 33 % in those with somatization disorders [27, 166, 172]. A larger study of 50 patients later reported no difference in false positives among patients with somatization disorder [174].

At best one can conclude that the false-positive rate for provocative discography including in asymptomatic individuals does not exceed 10 % and that in patients with chronic pain, falsepositive rates are likely higher, making interpretation of positive responses even more difficult to interpret in this population.

#### Therapeutic Validity

The validity of provocative discography as a diagnostic test still remains unproven/controversial. A study by Carragee attempted to compare discography to a constructed "gold standard" of clinical outcomes [175]. Patients without any medical comorbidities with positive single-level low-pressure discography underwent 360° single-level fusion including diskectomy. The presumption was that removal of the primary pain generator should completely alleviate the patients' pain. Results were compared to strictly matched cohort undergoing the same procedure for single-level unstable spondylolisthesis. The hypothesis was that definitive surgical ablation of both of these lesions should be highly and equally effective in relieving pain and restoring function. Predetermined requirements for high-level success by measure of pain score, ODI, medication use, and return to usual work revealed that only 27 % patients in the positive discography group had highly effective clinical outcome compared to 72 % in the spondylolisthesis group [175]. Similar results were found when evaluating for less stringent minimal acceptable outcome measures. Positive predictive value when compared to a theoretical gold standard of clinical outcome was only 42–43 %. In other words, in patients found to have discogenic pain with positive provocative discography, less than half had resolution of pain after surgical fusion.

This appears to be in contrast with data reported by Colhoun in 1988 that reported that 89 % of patients with positive provocative discography responded well to fusion compared to 52 % of patients responding well to fusion with negative provocative discography [176]. Full interpretation of the study is limited to lack of available information on details of what qualified as "positive" discography and if fusion was limited to only the level found positive on discography. Regardless, a 52 % success rate even with negative discography suggests that etiologies of pain other than discogenic pain were also addressed in this study, and nothing can be said about the specificity of discography from this study or its utility in determining who should get surgery. One would argue that, in light of the aforementioned Carragee study, performing discography in order to be able to offer patients at least a 40 % chance at having pain relieved by surgery would then also have to accept the statement that, in consideration of the Colhoun study, even with negative discography a 52 % chance at success would also prompt surgical intervention. Only if the Carragee study is completely disregarded and Calhoun's study be taken at face value can the argument that provocative discography has validated be a therapeutic utility to predict success with surgical fusion.

## 33.4.1.2 CT or MRI Discography After Provocative Discography

It has been theorized that there is value in postdisk stimulation CT or MRI scan in that it may reveal annular tears that would not otherwise be seen. This may provide additional evidence of what would have otherwise been missed as nucleus pulposus extravasation causing chemical radiculitis [177]. The Dallas discogram scale describes the degree of annular tear as grade I tear of the inner third, grade II tear of the middle third, grade III tear of the outer third, and grade IV tear being circumferential spread of contrast of at least 30°. There is evidence that greater pain is associated with greater degree of annular disruption; grade 0 and grade 1 disruptions were rarely painful compared to 75 % of grade III or IV disruptions being associated with concordant pain [178, 179].

#### 33.4.1.3 Analgesic Discography

Analgesic discography is an appealing theoretical alternative to provocative discography to evaluate the etiology of a patient's pain. Instead of pressurizing the disk and looking for concordant pain, an analgesic discography relies on relief of pain after the injection of a local anesthetic. Therefore it has the significant advantage of not requiring a normal healthy disk to be injected as a control. It also would eliminate the element of manometry currently required for provocative discography. Derby et al. compared the proportions of patients that reported relief of pain following local anesthetic injection into disks that were found to be concordantly painful with provocative testing using four different protocols [180]. One group underwent pressure-controlled discography with a mixture of anesthetic and contrast and found that 7 % of patients had greater than 50 % pain relief and 3 % had greater than 80 % pain relief [181]. The second group underwent injection with anesthetic alone while being evaluated for concordant pain during injection of the anesthetic. This group had 40 % of patients that reported 50 % pain relief and 20 % reported at least 80 % pain relief [180]. The third group underwent injection of intradiscal anesthetic only into disks that were positive with provocative discography using contrast. This group demonstrated that 46.2 % of patients reported at least 50 % pain relief and 30 % reported at least 80 % pain relief [180]. Lastly, the fourth group had a catheter left in after provocative discography, and anesthetic was injected 45 min post-procedure and showed that 80 % of patients had at least 50 % pain relief and 25 % had at least 80 % relief [182].

Notably all four groups demonstrated similar positive tests with of rates provocative discography (28–41 %) [180]. However, when looking at pain relief in disks found to be painful on provocative discography, the rates differed significantly between groups. With using 50 % pain relief as a cutoff, only two groups had similar rates (40 and 46 %) with the two other groups lying on either side of this (7 and 80 %) [180]. Using an 80 % pain relief demonstrated similarity between three of the groups (20 %, 35 %, 30 %) with the other showing a lower percent response (3 %) [180]. Theoretically when compared to provocative discography, there may be discordant rates of positive tests that can be attributable to either false positives during provocative testing or false negatives during analgesic testing. If considering 80 % pain relief as a positive result, which occurred in 20-30 % of the disks found to be painful with provocative discography, one can postulate that provocative discography has false-positive rates of 70-80 % or that anesthetic discography has false negatives of 70–80 %. Without a clear reference standard, the discrepancy between the two cannot be fully defined.

Clearly analgesic discography still lacks a standardized protocol to perform the procedure, standardized testing parameters during the procedure, and a standardized way to interpret the test result before it can be a viable diagnostic tool for discogenic pain. Whether it is potentially a test that can be used in lieu of provocative discography or one that can be used in conjunction with to formulate a compound criteria including positive response to both provocative discography and analgesic discography is another question that will need to be answered in the future.

# 33.4.1.4 Potential Intradiscal Therapeutic Agents

Aside from the diagnostic value of either provocative or analgesic discography, the ultimate goal of medicine is to not only diagnose but also offer safe and effective therapeutic agent. There have been several treatments postulated for discogenic low back pain, including intradiscal corticosteroids, methylene blue, and fibrin sealant. Methylene blue has been postulated as a possible effective intradiscal therapy given its theoretical neurotropic effects and theoretical action as a direct inhibitor of NO synthase. Peng et al. reported significant improvements in pain, satisfaction, and disability score in a randomized placebo-controlled trial with the intradiscal injection of methylene blue versus placebo [183]. However attempts to replicate these findings by Kim et al. in a 1-year prospective study have failed, and to this point the Peng study stands alone in demonstrating utility in intradiscal methylene blue treatment [184]. Perhaps the greatest area of potential for therapeutic intervention lays in nucleus pulposus replacement, annulus fibrosus replacement, and/or sealants/adhesives. Both synthetic polymeric materials and natural biopolymers have been studied as injectable nucleus pulposus tissue replacement [185–191]. Many of these have been able to match the compression and shear moduli of native nucleus pulposus tissue [191]. However progress is still needed before these are viable therapeutic options. Also, an injectable nucleus pulposus replacement would still have the theoretical problems of leaving a puncture site through the annulus fibrosus, and as previously discussed there is evidence that this alone predisposes disks to herniation and degeneration. Schek et al. have published a possible fibrin-genipin hydrogel adhesive with a tunable mechanical property that may be able to act as a plug mechanism without causing mechanical stress on the surrounding tissue [192]. There have also been attempts to produce biomaterials for annulus fibrosus replacement; however, no materials have been found thus far that are able to match the mechanical properties of the annulus fibrosus [193–196]. More promising perhaps are potential scaffolds that would support the growth of annulus fibrosus cells. Many attempts are underway to identify materials for nucleus pulposus and annulus fibrosus repair; however, currently significant progress remains to be made before these are viable therapeutic options for patients.

In light of there being no conclusive evidence that there is any effective therapeutic intervention for discography-proven discogenic pain, coupled with the long-term detrimental sequelae of discography, its utility is questionable at best, and use of the test should be extremely limited. In patients with chronic low back pain that are potentially undergoing fusion, there may remain a small role in discography in aiding surgical planning if the patient is willing to consider that positive discography leading to fusion of an additional level may increase chances at successful surgical outcome by up to 50 %. Otherwise, until new techniques that mitigate the proven longterm risks of discography are developed in addition to the development of successful outcomes for discography-proven discogenic pain, the clinical utility of discography is marginal.

#### Conclusions

The judicious use of appropriately performed injections can aid significantly in the ability to diagnose and treat patients with spine pathologies. Injections by definition are target specific and, to be useful from a diagnostic standpoint, must be done with imaging guidance. To be appropriately therapeutic they must also be judiciously used in appropriately selected patients while following appropriate technical guidelines. When done appropriately these image-guided injections have a large body of literature demonstrating their efficacy for certain disease conditions.

### References

- Goldthwait JE. The lumbo-sacral articulation; an explanation of many cases of "Lumbago", "Sciatica" and Paraplegia. Boston Med Surg J. 1911;164(11):365–72.
- Yamashita T, Cavanaugh JM, el-Bohy AA, Getchell TV, King AI. Mechanosensitive afferent units in the lumbar facet joint. J Bone Joint Surg Am. 1990;72(6):865–70.
- Kaplan M, Dreyfuss P, Halbrook B, Bogduk N. The ability of lumbar medial branch blocks to anesthetize the zygapophysial joint. A physiologic challenge. Spine. 1998;23(17):1847–52.
- Manchikanti L, Singh V, Pampati V, Damron KS, Barnhill RC, Beyer C, et al. Evaluation of the relative contributions of various structures in chronic low back pain. Pain Physician. 2001;4(4):308–16.
- Schwarzer AC, Wang SC, Bogduk N, McNaught PJ, Laurent R. Prevalence and clinical features of lumbar zygapophysial joint pain: a study in an Australian

population with chronic low back pain. Ann Rheum Dis. 1995;54(2):100–6.

- Manchikanti L, Manchikanti KN, Manchukonda R, Cash KA, Damron KS, Pampati V, et al. Evaluation of lumbar facet joint nerve blocks in the management of chronic low back pain: preliminary report of a randomized, double-blind controlled trial: clinical trial NCT00355914. Pain Physician. 2007;10(3):425–40.
- De Vlam K, Mielants H, Verstaete KL, Veys EM. The zygapophyseal joint determines morphology of the enthesophyte. J Rheumatol. 2000;27(7):1732–9.
- Fujishiro T, Nabeshima Y, Yasui S, Fujita I, Yoshiya S, Fujii H. Pseudogout attack of the lumbar facet joint: a case report. Spine. 2002;27(17):E396–8.
- Smida M, Lejri M, Kandara H, Sayed M, Ben Chehida F, Ben Ghachem M. Septic arthritis of a lumbar facet joint case report and review of the literature. Acta Orthop Belg. 2004;70(3):290–4.
- Campbell AJ, Wells IP. Pigmented villonodular synovitis of a lumbar vertebral facet joint. J Bone Joint Surg Am. 1982;64(1):145–6.
- 11. Loeser RF, Shakoor N. Aging or osteoarthritis: which is the problem? Rheum Dis Clin North Am. 2003;29(4):653–73.
- Kalichman L, Hunter DJ. Lumbar facet joint osteoarthritis: a review. Semin Arthritis Rheum. 2007;37(2):69–80.
- Bogduk N, Towmey LT. Clinical anatomy of the lumbar spine and sacrum. New York: Churchill Livingstone; 1997.
- Eubanks JD, Lee MJ, Cassinelli E, Ahn NU. Prevalence of lumbar facet arthrosis and its relationship to age, sex, and race: an anatomic study of cadaveric specimens. Spine. 2007;32(19):2058–62.
- Kalichman L, Li L, Kim DH, Guermazi A, Berkin V, O'Donnell CJ, et al. Facet joint osteoarthritis and low back pain in the community-based population. Spine. 2008;33(23):2560–5.
- Revel M, Poiraudeau S, Auleley GR, Payan C, Denke A, Nguyen M, et al. Capacity of the clinical picture to characterize low back pain relieved by facet joint anesthesia. Proposed criteria to identify patients with painful facet joints. Spine. 1998;23(18):1972–6. discussion 1977.
- Cohen SP, Hurley RW, Christo PJ, Winkley J, Mohiuddin MM, Stojanovic MP. Clinical predictors of success and failure for lumbar facet radiofrequency denervation. Clin J Pain. 2007;23(1):45–52.
- 18. Hancock MJ, Maher CG, Latimer J, Spindler MF, McAuley JH, Laslett M, et al. Systematic review of tests to identify the disc, SIJ or facet joint as the source of low back pain. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2007;16(10):1539–50.
- Schwarzer AC, Aprill CN, Derby R, Fortin J, Kine G, Bogduk N. Clinical features of patients with pain stemming from the lumbar zygapophysial joints. Is the lumbar facet syndrome a clinical entity? Spine. 1994;19(10):1132–7.
- Bogduk N. Evidence-informed management of chronic low back pain with facet injections and
radiofrequency neurotomy. Spine J Off J N Am Spine Soc. 2008;8(1):56–64.

- Mooney V, Robertson J. The facet syndrome. Clin Orthop. 1976;115:149–56.
- International Spine Intervention Society. Practice guidelines for spinal diagnostic and treatment procedures. 1st ed. San Francisco: International Spine Intervention Society; 2004. 347 p.
- Lord SM, Barnsley L, Wallis BJ, McDonald GJ, Bogduk N. Percutaneous radio-frequency neurotomy for chronic cervical zygapophyseal-joint pain. N Engl J Med. 1996;335(23):1721–6.
- Dreyfuss P, Schwarzer AC, Lau P, Bogduk N. Specificity of lumbar medial branch and L5 dorsal ramus blocks. A computed tomography study. Spine. 1997;22(8):895–902.
- Taylor JR, Twomey LT. Age changes in lumbar zygapophyseal joints. Observations on structure and function. Spine. 1986;11(7):739–45.
- Jefferies B. Facet steroid injections. Spine. 1988;2:409–17.
- Birkenmaier C, Veihelmann A, Trouillier H-H, Hausdorf J, von Schulze Pellengahr C. Medial branch blocks versus pericapsular blocks in selecting patients for percutaneous cryodenervation of lumbar facet joints. Reg Anesth Pain Med. 2007;32(1): 27–33.
- 28. Van Wijk RMAW, Geurts JWM, Wynne HJ, Hammink E, Buskens E, Lousberg R, et al. Radiofrequency denervation of lumbar facet joints in the treatment of chronic low back pain: a randomized, double-blind, sham lesion-controlled trial. Clin J Pain. 2005;21(4):335–44.
- 29. Gallagher J, Perticcione Di Vadi P, Wedley J, Hamann W, Ryan P, Chikanza I, et al. Radiofrequency facet joint denervation in the treatment of low back pain: a prospective controlled double-blind study to assess its efficacy. Pain Clin. 1994;7(3):193.
- 30. Leclaire R, Fortin L, Lambert R, Bergeron YM, Rossignol M. Radiofrequency facet joint denervation in the treatment of low back pain: a placebocontrolled clinical trial to assess efficacy. Spine. 2001;26(13):1411–6. discussion 1417.
- Cohen SP, Raja SN. Pathogenesis, diagnosis, and treatment of lumbar zygapophysial (facet) joint pain. Anesthesiology. 2007;106(3):591–614.
- 32. Cohen SP, Williams KA, Kurihara C, Nguyen C, Shields C, Kim P, et al. Multicenter, randomized, comparative cost-effectiveness study comparing 0, 1, and 2 diagnostic medial branch (facet joint nerve) block treatment paradigms before lumbar facet radiofrequency denervation. Anesthesiology. 2010;113(2):395–405.
- 33. Lilius G, Laasonen EM, Myllynen P, Harilainen A, Grönlund G. Lumbar facet joint syndrome. A randomised clinical trial. J Bone Joint Surg Br. 1989;71(4):681–4.
- 34. Carette S, Marcoux S, Truchon R, Grondin C, Gagnon J, Allard Y, et al. A controlled trial of corticosteroid injections into facet joints for chronic low back pain. N Engl J Med. 1991;325(14):1002–7.

- 35. Dolan AL, Ryan PJ, Arden NK, Stratton R, Wedley JR, Hamann W, et al. The value of SPECT scans in identifying back pain likely to benefit from facet joint injection. Br J Rheumatol. 1996;35(12):1269–73.
- 36. Ackerman 3rd WE, Ahmad M. Pain relief with intraarticular or medial branch nerve blocks in patients with positive lumbar facet joint SPECT imaging: a 12-week outcome study. South Med J. 2008;101(9):931–4.
- Pneumaticos SG, Chatziioannou SN, Hipp JA, Moore WH, Esses SI. Low back pain: prediction of short-term outcome of facet joint injection with bone scintigraphy. Radiology. 2006;238(2):693–8.
- Ribeiro LH, Vilar Furtado RN, Konai M, Andreo AB, Rosenfeld A, Natour J. The effect of facet joint injection versus systemic steroids in low back pain: a randomized controlled trial. Spine. 2013;38(23):1995–2002.
- Balazs EA. Viscosupplementation for treatment of osteoarthritis: from initial discovery to current status and results. Surg Technol Int. 2004;12:278–89.
- Fuchs S, Erbe T, Fischer H-L, Tibesku CO. Intraarticular hyaluronic acid versus glucocorticoid injections for nonradicular pain in the lumbar spine. J Vasc Interv Radiol JVIR. 2005;16(11):1493–8.
- Dreyfuss P, Halbrook B, Pauza K, Joshi A, McLarty J, Bogduk N. Efficacy and validity of radiofrequency neurotomy for chronic lumbar zygapophysial joint pain. Spine. 2000;25(10):1270–7.
- 42. Nath S, Nath CA, Pettersson K. Percutaneous lumbar zygapophysial (Facet) joint neurotomy using radiofrequency current, in the management of chronic low back pain: a randomized double-blind trial. Spine. 2008;33(12):1291–7. discussion 1298.
- 43. Van Kleef M, Barendse GA, Kessels A, Voets HM, Weber WE, de Lange S. Randomized trial of radiofrequency lumbar facet denervation for chronic low back pain. Spine. 1999;24(18):1937–42.
- 44. Farrar JT, Young Jr JP, LaMoreaux L, Werth JL, Poole RM. Clinical importance of changes in chronic pain intensity measured on an 11-point numerical pain rating scale. Pain. 2001;94(2):149–58.
- Fairbank JC, Park WM, McCall IW, O'Brien JP. Apophyseal injection of local anesthetic as a diagnostic aid in primary low-back pain syndromes. Spine. 1981;6(6):598–605.
- Moore A, McQuay H, Gavaghan D. Deriving dichotomous outcome measures from continuous data in randomised controlled trials of analgesics. Pain. 1996;66(2–3):229–37.
- Manchikanti L, Manchikanti KN, Cash KA, Singh V, Giordano J. Age-related prevalence of facet-joint involvement in chronic neck and low back pain. Pain Physician. 2008;11(1):67–75.
- Schwarzer AC, Aprill CN, Derby R, Fortin J, Kine G, Bogduk N. The false-positive rate of uncontrolled diagnostic blocks of the lumbar zygapophysial joints. Pain. 1994;58(2):195–200.
- Bogduk N. Diagnostic nerve blocks in chronic pain. Best Pract Res Clin Anaesthesiol. 2002;16(4):565–78.

- Manchikanti L, Pampati V, Fellows B, Bakhit CE. The diagnostic validity and therapeutic value of lumbar facet joint nerve blocks with or without adjuvant agents. Curr Rev Pain. 2000;4(5):337–44.
- Barnsley L, Lord S, Bogduk N. Comparative local anaesthetic blocks in the diagnosis of cervical zygapophysial joint pain. Pain. 1993;55(1):99–106.
- Lord SM, Barnsley L, Bogduk N. The utility of comparative local anesthetic blocks versus placebocontrolled blocks for the diagnosis of cervical zygapophysial joint pain. Clin J Pain. 1995;11(3):208–13.
- Cohen SP, Huang JHY, Brummett C. Facet joint pain-advances in patient selection and treatment. Nat Rev Rheumatol. 2013;9(2):101–16.
- 54. Stojanovic MP, Dey D, Hord ED, Zhou Y, Cohen SP. A prospective crossover comparison study of the single-needle and multiple-needle techniques for facet-joint medial branch block. Reg Anesth Pain Med. 2005;30(5):484–90.
- 55. Fujiwara A, Tamai K, Yamato M, An HS, Yoshida H, Saotome K, et al. The relationship between facet joint osteoarthritis and disc degeneration of the lumbar spine: an MRI study. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 1999;8(5):396–401.
- Gottfried Y, Bradford D, Oegema D. Facet joint changes after chemonucleolysis induced disc space narrowing. Spine. 1986;11:944–54.
- Schwarzer AC, Aprill CN, Derby R, Fortin J, Kine G, Bogduk N. The relative contributions of the disc and zygapophyseal joint in chronic low back pain. Spine. 1994;19(7):801–6.
- Verrills P, Mitchell B, Vivian D, Nowesenitz G, Lovell B, Sinclair C. The incidence of intravascular penetration in medial branch blocks: cervical, thoracic, and lumbar spines. Spine. 2008;33(6):E174–7.
- 59. Cohen SP, Strassels SA, Kurihara C, Forsythe A, Buckenmaier 3rd CC, McLean B, et al. Randomized study assessing the accuracy of cervical facet joint nerve (medial branch) blocks using different injectate volumes. Anesthesiology. 2010;112(1):144–52.
- Allen TL, Tatli Y, Lutz GE. Fluoroscopic percutaneous lumbar zygapophyseal joint cyst rupture: a clinical outcome study. Spine J Off J N Am Spine Soc. 2009;9(5):387–95.
- Martha JF, Swaim B, Wang DA, Kim DH, Hill J, Bode R, et al. Outcome of percutaneous rupture of lumbar synovial cysts: a case series of 101 patients. Spine J Off J N Am Spine Soc. 2009;9(11):899–904.
- Halperin N, Agasi M, Hendel D. Painless root compression following disc extrusion. A report of three cases. Arch Orthop Trauma Surg Arch Für Orthop Unf-Chir. 1982;101(1):63–6.
- Wiesel SW, Tsourmas N, Feffer HL, Citrin CM, Patronas N. A study of computer-assisted tomography. I. The incidence of positive CAT scans in an asymptomatic group of patients. Spine. 1984;9(6):549–51.
- 64. Boden SD, Davis DO, Dina TS, Patronas NJ, Wiesel SW. Abnormal magnetic-resonance scans of the lumbar spine in asymptomatic subjects. A

prospective investigation. J Bone Joint Surg Am. 1990;72(3):403-8.

- Macnab I. The mechanism of spondylogenic pain. In: Hirsch C, Zotterman Y, eds. Cervical Pain. Oxford: Pergamon, 1972:89–95
- 66. Lindahl O, Rexed B. Histologic changes in spinal nerve roots of operated cases of sciatica. Acta Orthop Scand. 1951;20(3):215–25.
- Bobechko WP, Hirsch C. AUTO-immune response to nucleus pulposus in the rabbit. J Bone Joint Surg Br. 1965;47:574–80.
- McCarron RF, Wimpee MW, Hudkins PG, Laros GS. The inflammatory effect of nucleus pulposus. A possible element in the pathogenesis of low-back pain. Spine. 1987;12(8):760–4.
- 69. Svennerholm L, Boström K, Fredman P, Månsson JE, Rosengren B, Rynmark BM. Human brain gangliosides: developmental changes from early fetal stage to advanced age. Biochim Biophys Acta. 1989;1005(2):109–17.
- Takahashi H, Suguro T, Okazima Y, Motegi M, Okada Y, Kakiuchi T. Inflammatory cytokines in the herniated disc of the lumbar spine. Spine. 1996;21(2):218–24.
- Goupille P, Jayson MI, Valat JP, Freemont AJ. The role of inflammation in disk herniation-associated radiculopathy. Semin Arthritis Rheum. 1998;28(1):60–71.
- Murphy RW. Nerve roots and spinal nerves in degenerative disk disease. Clin Orthop. 1977;129:46–60.
- Saal JS, Franson RC, Dobrow R, Saal JA, White AH, Goldthwaite N. High levels of inflammatory phospholipase A2 activity in lumbar disc herniations. Spine. 1990;15(7):674–8.
- 74. Virri J, Sikk S, Grönblad M, Tolonen J, Seitsalo S, Kankare J, et al. Concomitant immunocytochemical study of macrophage cells and blood vessels in disc herniation tissue. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 1994;3(6):336–41.
- Hirsch C. Studies on the pathology of low back pain. J Bone Joint Surg Br. 1959;41-B(2):237–43.
- 76. Cohen SP, Bogduk N, Dragovich A, Buckenmaier 3rd CC, Griffith S, Kurihara C, et al. Randomized, double-blind, placebo-controlled, dose-response, and preclinical safety study of transforaminal epidural etanercept for the treatment of sciatica. Anesthesiology. 2009;110(5):1116–26.
- Johansson A, Hao J, Sjölund B. Local corticosteroid application blocks transmission in normal nociceptive C-fibres. Acta Anaesthesiol Scand. 1990;34(5):335–8.
- DePalma MJ. ISpine evidence-based interventional spine care [Internet]. New York: Demos Medical; 2011. [cited 2014 Jan 31], Available from: http://site. ebrary.com/id/10482351.
- Cyriax J. Textbook of orthopedic medicine. 3rd ed. London: Cassell; 1957. p. 460–9.
- Pages E. Anestesia metamerica. Rev Sanid Mil Madr. 1921;11:351–80.
- Robecchi A, Capra R. Hydrocortisone (compound F); first clinical experiments in the field of rheumatology. Minerva Med. 1952;43(98):1259–63.

- 82. Derby R, Kine G, Saal JA, Reynolds J, Goldthwaite N, White AH, et al. Response to steroid and duration of radicular pain as predictors of surgical outcome. Spine. 1992;17(6 Suppl):S176–83.
- Stalcup ST, Crall TS, Gilula L, Riew KD. Influence of needle-tip position on the incidence of immediate complications in 2,217 selective lumbar nerve root blocks. Spine J Off J N Am Spine Soc. 2006;6(2):170–6.
- Botwin KP, Gruber RD, Bouchlas CG, Torres-Ramos FM, Freeman TL, Slaten WK. Complications of fluoroscopically guided transforaminal lumbar epidural injections. Arch Phys Med Rehabil. 2000;81(8):1045–50.
- Huston CW, Slipman CW, Garvin C. Complications and side effects of cervical and lumbosacral selective nerve root injections. Arch Phys Med Rehabil. 2005;86(2):277–83.
- 86. Kennedy DJ, Schneider B, Casey E, Rittenberg J, Conrad B, Smuck M, et al. Vasovagal rates in fluoroscopically guided interventional procedures: a study of over 8,000 injections. Pain Med Malden Mass. 2013;14(12):1854–9.
- Manchikanti L. Role of neuraxial steroids in interventional pain management. Pain Physician. 2002;5(2):182–99.
- Boonen S, Van Distel G, Westhovens R, Dequeker J. Steroid myopathy induced by epidural triamcinolone injection. Br J Rheumatol. 1995;34(4):385–6.
- Ward A, Watson J, Wood P, Dunne C, Kerr D. Glucocorticoid epidural for sciatica: metabolic and endocrine sequelae. Rheumatol Oxf Engl. 2002;41(1):68–71.
- Silbergleit R, Mehta BA, Sanders WP, Talati SJ. Imaging-guided injection techniques with fluoroscopy and CT for spinal pain management. Radiogr Rev Publ Radiol Soc N Am Inc. 2001;21(4):927–39. discussion 940–942.
- Lippert JA, McGraw JK. Spine interventions. Semin Roentgenol. 2002;37(4):266–81.
- Karppinen J, Malmivaara A, Kurunlahti M, Kyllönen E, Pienimäki T, Nieminen P, et al. Periradicular infiltration for sciatica: a randomized controlled trial. Spine. 2001;26(9):1059–67.
- Layton KF, Kallmes DF, Horlocker TT. Recommendations for anticoagulated patients undergoing image-guided spinal procedures. AJNR Am J Neuroradiol. 2006;27(3):468–70.
- 94. Horlocker TT, Bajwa ZH, Ashraf Z, Khan S, Wilson JL, Sami N, et al. Risk assessment of hemorrhagic complications associated with nonsteroidal antiin-flammatory medications in ambulatory pain clinic patients undergoing epidural steroid injection. Anesth Analg. 2002;95(6):1691–7. table of contents.
- Stoll A, Sanchez M. Epidural hematoma after epidural block: implications for its use in pain management. Surg Neurol. 2002;57(4):235–40.
- Hooten WM, Kinney MO, Huntoon MA. Epidural abscess and meningitis after epidural corticosteroid injection. Mayo Clin Proc. 2004;79(5):682–6.
- Abram SE, O'Connor TC. Complications associated with epidural steroid injections. Reg Anesth. 1996;21(2):149–62.

- Tham EJ, Stoodley MA, Macintyre PE, Jones NR. Back pain following postoperative epidural analgesia: an indicator of possible spinal infection. Anaesth Intensive Care. 1997;25(3):297–301.
- 99. Yue W-M, Tan S-B. Distant skip level discitis and vertebral osteomyelitis after caudal epidural injection: a case report of a rare complication of epidural injections. Spine. 2003;28(11):E209–11.
- Gutknecht DR. Chemical meningitis following epidural injections of corticosteroids. Am J Med. 1987;82(3):570.
- 101. Goodman BS, Posecion LWF, Mallempati S, Bayazitoglu M. Complications and pitfalls of lumbar interlaminar and transforaminal epidural injections. Curr Rev Musculoskelet Med. 2008;1(3–4):212–22.
- 102. Hooten WM, Mizerak A, Carns PE, Huntoon MA. Discitis after lumbar epidural corticosteroid injection: a case report and analysis of the case report literature. Pain Med Malden Mass. 2006;7(1):46–51.
- 103. Goodman BS, Bayazitoglu M, Mallempati S, Noble BR, Geffen JF. Dural puncture and subdural injection: a complication of lumbar transforaminal epidural injections. Pain Physician. 2007;10(5):697–705.
- 104. Haspeslagh S, Van Zundert J, Puylaert M, Heylen R, van Kleef M, Vissers K. Unilateral diagnostic infiltration of lumbar L3 nerve root resulting in an inadvertent discogram: the importance of fluoroscopic guidance in interventional pain therapy. Anesthesiology. 2004;100(4):1019–21.
- 105. Cohen SP, Maine DN, Shockey SM, Kudchadkar S, Griffith S. Inadvertent disk injection during transforaminal epidural steroid injection: steps for prevention and management. Pain Med Malden Mass. 2008;9(6):688–94.
- 106. Finn KP, Case JL. Disk entry: a complication of transforaminal epidural injection–a case report. Arch Phys Med Rehabil. 2005;86(7):1489–91.
- 107. Furman MB, O'Brien EM, Zgleszewski TM. Incidence of intravascular penetration in transforaminal lumbosacral epidural steroid injections. Spine. 2000;25(20):2628–32.
- Sullivan WJ, Willick SE, Chira-Adisai W, Zuhosky J, Tyburski M, Dreyfuss P, et al. Incidence of intravascular uptake in lumbar spinal injection procedures. Spine. 2000;25(4):481–6.
- 109. Boll DT, Bulow H, Blackham KA, Aschoff AJ, Schmitz BL. MDCT angiography of the spinal vasculature and the artery of Adamkiewicz. AJR Am J Roentgenol. 2006;187(4):1054–60.
- Houten JK, Errico TJ. Paraplegia after lumbosacral nerve root block: report of three cases. Spine J Off J N Am Spine Soc. 2002;2(1):70–5.
- 111. Kennedy DJ, Dreyfuss P, Aprill CN, Bogduk N. Paraplegia following image-guided transforaminal lumbar spine epidural steroid injection: two case reports. Pain Med Malden Mass. 2009;10(8):1389–94.
- 112. Smuck M, Fuller BJ, Chiodo A, Benny B, Singaracharlu B, Tong H, et al. Accuracy of intermittent fluoroscopy to detect intravascular injection during transforaminal epidural injections. Spine. 2008;33(7):E205–10.

- 113. Ergin A, Yanarates O, Sizlan A, Orhan ME, Kurt E, Guzeldemir ME. Accuracy of caudal epidural injection: the importance of real-time imaging. Pain Pract Off J World Inst Pain. 2005;5(3):251–4.
- 114. Chang Chien GC, Candido KD, Knezevic NN. Digital subtraction angiography does not reliably prevent paraplegia associated with lumbar transforaminal epidural steroid injection. Pain Physician. 2012;15(6):515–23.
- 115. Furman MB, Giovanniello MT, O'Brien EM. Incidence of intravascular penetration in transforaminal cervical epidural steroid injections. Spine. 2003;28(1):21–5.
- 116. MacVicar J, King W, Landers MH, Bogduk N. The effectiveness of lumbar transforaminal injection of steroids: a comprehensive review with systematic analysis of the published data. Pain Med Malden Mass. 2013;14(1):14–28.
- 117. Ghahreman A, Ferch R, Bogduk N. The efficacy of transforaminal injection of steroids for the treatment of lumbar radicular pain. Pain Med Malden Mass. 2010;11(8):1149–68.
- 118. Renfrew DL, Moore TE, Kathol MH, el-Khoury GY, Lemke JH, Walker CW. Correct placement of epidural steroid injections: fluoroscopic guidance and contrast administration. AJNR Am J Neuroradiol. 1991;12(5):1003–7.
- 119. Kim KM, Kim HS, Choi KH, Ahn WS. Cephalic spreading levels after volumetric caudal epidural injections in chronic low back pain. J Korean Med Sci. 2001;16(2):193–7.
- 120. Mohamed MMM, Ahmed M, Chaudary M. Caudal epidural injection for L4-5 versus L5-S1 disc prolapse: is there any difference in the outcome? J Spinal Disord Tech. 2007;20(1):49–52.
- 121. Conn A, Buenaventura RM, Datta S, Abdi S, Diwan S. Systematic review of caudal epidural injections in the management of chronic low back pain. Pain Physician. 2009;12(1):109–35.
- 122. Béliveau P. A comparison between epidural anaesthesia with and without corticosteroid in the treatment of sciatica. Rheumatol Phys Med. 1971;11(1):40–3.
- 123. Mathews JA, Mills SB, Jenkins VM, Grimes SM, Morkel MJ, Mathews W, et al. Back pain and sciatica: controlled trials of manipulation, traction, sclerosant and epidural injections. Br J Rheumatol. 1987;26(6):416–23.
- 124. Bush K, Hillier S. A controlled study of caudal epidural injections of triamcinolone plus procaine for the management of intractable sciatica. Spine. 1991;16(5):572–5.
- 125. Dincer U, Kiralp MZ, Cakar E, Yasar E, Dursan H. Caudal epidural injection versus non-steroidal anti-inflammatory drugs in the treatment of low back pain accompanied with radicular pain. Joint Bone Spine Rev Rhum. 2007;74(5):467–71.
- 126. Dashfield AK, Taylor MB, Cleaver JS, Farrow D. Comparison of caudal steroid epidural with targeted steroid placement during spinal endoscopy for chronic sciatica: a prospective, randomized, double-blind trial. Br J Anaesth. 2005;94(4):514–9.

- 127. Carette S, Leclaire R, Marcoux S, Morin F, Blaise GA, St-Pierre A, et al. Epidural corticosteroid injections for sciatica due to herniated nucleus pulposus. N Engl J Med. 1997;336(23):1634–40.
- 128. Valat J-P, Giraudeau B, Rozenberg S, Goupille P, Bourgeois P, Micheau-Beaugendre V, et al. Epidural corticosteroid injections for sciatica: a randomised, double blind, controlled clinical trial. Ann Rheum Dis. 2003;62(7):639–43.
- 129. Parr AT, Diwan S, Abdi S. Lumbar interlaminar epidural injections in managing chronic low back and lower extremity pain: a systematic review. Pain Physician. 2009;12(1):163–88.
- Dilke TF, Burry HC, Grahame R. Extradural corticosteroid injection in management of lumbar nerve root compression. Br Med J. 1973;2(5867):635–7.
- 131. Armon C, Argoff CE, Samuels J, Backonja M-M, Therapeutics and Technology Assessment Subcommittee of the American Academy of Neurology. Assessment: use of epidural steroid injections to treat radicular lumbosacral pain: report of the Therapeutics and Technology Assessment Subcommittee of the American Academy of Neurology. Neurology. 2007;68(10):723–9.
- Weiner BK, Fraser RD. Foraminal injection for lateral lumbar disc herniation. J Bone Joint Surg Br. 1997;79(5):804–7.
- 133. Riew KD, Yin Y, Gilula L, Bridwell KH, Lenke LG, Lauryssen C, et al. The effect of nerve-root injections on the need for operative treatment of lumbar radicular pain. A prospective, randomized, controlled, double-blind study. J Bone Joint Surg Am. 2000;82-A(11):1589–93.
- 134. Wang JC, Lin E, Brodke DS, Youssef JA. Epidural injections for the treatment of symptomatic lumbar herniated discs. J Spinal Disord Tech. 2002;15(4):269–72.
- 135. Giraudeau B, Rozenberg S, Valat J-P. Assessment of the clinically relevant change in pain for patients with sciatica. Ann Rheum Dis. 2004;63(9):1180–1.
- Lutz GE, Vad VB, Wisneski RJ. Fluoroscopic transforaminal lumbar epidural steroids: an outcome study. Arch Phys Med Rehabil. 1998;79(11):1362–6.
- 137. Kaufmann TJ, Geske JR, Murthy NS, Thielen KR, Morris JM, Wald JT, et al. Clinical effectiveness of single lumbar transforaminal epidural steroid injections. Pain Med Malden Mass. 2013;14(8):1126–33.
- 138. Cyteval C, Fescquet N, Thomas E, Decoux E, Blotman F, Taourel P. Predictive factors of efficacy of periradicular corticosteroid injections for lumbar radiculopathy. AJNR Am J Neuroradiol. 2006;27(5):978–82.
- Narozny M, Zanetti M, Boos N. Therapeutic efficacy of selective nerve root blocks in the treatment of lumbar radicular leg pain. Swiss Med Wkly. 2001;131(5–6):75–80.
- 140. Jeong HS, Lee JW, Kim SH, Myung JS, Kim JH, Kang HS. Effectiveness of transforaminal epidural steroid injection by using a preganglionic approach: a prospective randomized controlled study. Radiology. 2007;245(2):584–90.

- 141. Karppinen J, Ohinmaa A, Malmivaara A, Kurunlahti M, Kyllönen E, Pienimäki T, et al. Cost effectiveness of periradicular infiltration for sciatica: subgroup analysis of a randomized controlled trial. Spine. 2001;26(23):2587–95.
- 142. Vad VB, Bhat AL, Lutz GE, Cammisa F. Transforaminal epidural steroid injections in lumbosacral radiculopathy: a prospective randomized study. Spine. 2002;27(1):11–6.
- 143. Tafazal S, Ng L, Chaudhary N, Sell P. Corticosteroids in peri-radicular infiltration for radicular pain: a randomised double blind controlled trial. One year results and subgroup analysis. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2009;18(8):1220–5.
- 144. Lee JW, Kim SH, Choi J-Y, Yeom J-S, Kim K-J, Chung S-K, et al. Transforaminal epidural steroid injection for lumbosacral radiculopathy: preganglionic versus conventional approach. Korean J Radiol Off J Korean Radiol Soc. 2006;7(2):139–44.
- 145. Ackerman 3rd WE, Ahmad M. The efficacy of lumbar epidural steroid injections in patients with lumbar disc herniations. Anesth Analg. 2007;104(5):1217– 22. tables of contents.
- 146. Thomas E, Cyteval C, Abiad L, Picot MC, Taourel P, Blotman F. Efficacy of transforaminal versus interspinous corticosteroid injection in discal radiculalgia – a prospective, randomised, double-blind study. Clin Rheumatol. 2003;22(4–5):299–304.
- 147. Ghahreman A, Bogduk N. Predictors of a favorable response to transforaminal injection of steroids in patients with lumbar radicular pain due to disc herniation. Pain Med Malden Mass. 2011;12(6): 871–9.
- 148. Schaufele MK, Hatch L, Jones W. Interlaminar versus transforaminal epidural injections for the treatment of symptomatic lumbar intervertebral disc herniations. Pain Physician. 2006;9(4):361–6.
- 149. Rados I, Sakic K, Fingler M, Kapural L. Efficacy of interlaminar vs transforaminal epidural steroid injection for the treatment of chronic unilateral radicular pain: prospective, randomized study. Pain Med Malden Mass. 2011;12(9):1316–21.
- 150. Lee JH, Moon J, Lee S-H. Comparison of effectiveness according to different approaches of epidural steroid injection in lumbosacral herniated disk and spinal stenosis. J Back Musculoskelet Rehabil. 2009;22(2):83–9.
- Crock HV. Internal disc disruption. A challenge to disc prolapse fifty years on. Spine. 1986;11(6):650–3.
- 152. Bogduk N, Tynan W, Wilson AS. The nerve supply to the human lumbar intervertebral discs. J Anat. 1981;132(Pt 1):39–56.
- 153. Yoshizawa H, O'Brien JP, Smith WT, Trumper M. The neuropathology of intervertebral discs removed for low-back pain. J Pathol. 1980;132(2):95–104.
- 154. Bogduk N. The innervation of the lumbar spine. Spine. 1983;8(3):286–93.
- 155. Groen GJ, Baljet B, Drukker J. Nerves and nerve plexuses of the human vertebral column. Am J Anat. 1990;188(3):282–96.

- Lindblom K. Diagnostic puncture of intervertebral disks in sciatica. Acta Orthop Scand. 1948;17(3–4):231–9.
- Lindblom K. Technique and results in myelography and disc puncture. Acta Radiol. 1950;34(4–5):321–30.
- Lindblom K. Technique and results of diagnostic disc puncture and injection (discography) in the lumbar region. Acta Orthop Scand. 1951;20(4):315–26.
- Massie W, Stevens D. A critical evaluation of discography. J Bone Joint Surg. 1967;49A:1243–4.
- Brodsky AE, Binder WF. Lumbar discography. Its value in diagnosis and treatment of lumbar disc lesions. Spine. 1979;4(2):110–20.
- 161. Doyle T, Tress B, Gillot R. Combined discography and metrizamide myelography in evaluation of confusing low back pain. Australas Radiol. 1985;29(3):217–22.
- 162. Gresham JL, Miller R. Evaluation of the lumbar spine by diskography and its use in selection of proper treatment of the herniated disk syndrome. Clin Orthop. 1969;67:29–41.
- 163. Hartman JT, Kendrick JI, Lorman P. Discography as an aid in evaluation for lumbar and lumbosacral fusion. Clin Orthop. 1971;81:77–81.
- 164. Willems PC, Jacobs W, Duinkerke ES, De Kleuver M. Lumbar discography: should we use prophylactic antibiotics? A study of 435 consecutive discograms and a systematic review of the literature. J Spinal Disord Tech. 2004;17(3):243–7.
- 165. Carragee EJ, Don AS, Hurwitz EL, Cuellar JM, Carrino JA, Carrino J, et al. 2009 ISSLS prize winner: does discography cause accelerated progression of degeneration changes in the lumbar disc: a ten-year matched cohort study. Spine. 2009;34(21):2338–45.
- 166. Bogduk N, Aprill C, Derby R. Lumbar discogenic pain: state-of-the-art review. Pain Med Malden Mass. 2013;14(6):813–36.
- 167. Holt Jr EP. The question of lumbar discography. J Bone Joint Surg Am. 1968;50(4):720–6.
- Simmons JW, Aprill CN, Dwyer AP, Brodsky AE. A reassessment of Holt's data on: "The question of lumbar discography.". Clin Orthop. 1988;237:120–4.
- 169. Pauza KJ, Howell S, Dreyfuss P, Peloza JH, Dawson K, Bogduk N. A randomized, placebo-controlled trial of intradiscal electrothermal therapy for the treatment of discogenic low back pain. Spine J Off J N Am Spine Soc. 2004;4(1):27–35.
- 170. Wolfer LR, Derby R, Lee J-E, Lee S-H. Systematic review of lumbar provocation discography in asymptomatic subjects with a meta-analysis of falsepositive rates. Pain Physician. 2008;11(4):513–38.
- 171. Derby R, Kim B-J, Lee S-H, Chen Y, Seo K-S, Aprill C. Comparison of discographic findings in asymptomatic subject discs and the negative discs of chronic LBP patients: can discography distinguish asymptomatic discs among morphologically abnormal discs? Spine J Off J N Am Spine Soc. 2005;5(4):389–94.
- 172. Carragee EJ, Tanner CM, Khurana S, Hayward C, Welsh J, Date E, et al. The rates of false-positive lumbar discography in select patients without low

back symptoms. Spine. 2000;25(11):1373-80. discussion 1381.

- 173. Derby R, Lee S-H, Kim B-J, Chen Y, Aprill C, Bogduk N. Pressure-controlled lumbar discography in volunteers without low back symptoms. Pain Med Malden Mass. 2005;6(3):213–21. discussion 222–224.
- 174. Manchikanti L, Singh V, Pampati V, Fellows B, Beyer C, Damron K, et al. Provocative discography in low back pain patients with or without somatization disorder: a randomized prospective evaluation. Pain Physician. 2001;4(3):227–39.
- 175. Carragee EJ, Lincoln T, Parmar VS, Alamin T. A gold standard evaluation of the "discogenic pain" diagnosis as determined by provocative discography. Spine. 2006;31(18):2115–23.
- 176. Colhoun E, McCall IW, Williams L, Cassar Pullicino VN. Provocation discography as a guide to planning operations on the spine. J Bone Joint Surg Br. 1988;70(2):267–71.
- 177. Marshall LL, Trethewie ER, Curtain CC. Chemical radiculitis. A clinical, physiological and immunological study. Clin Orthop. 1977;129:61–7.
- 178. Vanharanta H, Sachs BL, Spivey MA, Guyer RD, Hochschuler SH, Rashbaum RF, et al. The relationship of pain provocation to lumbar disc deterioration as seen by CT/discography. Spine. 1987;12(3):295–8.
- 179. Moneta GB, Videman T, Kaivanto K, Aprill C, Spivey M, Vanharanta H, et al. Reported pain during lumbar discography as a function of anular ruptures and disc degeneration. A re-analysis of 833 discograms. Spine. 1994;19(17):1968–74.
- 180. Derby R, Aprill CN, Lee J-E, DePalma MJ, Baker RM. Comparison of four different analgesic discogram protocols comparing the incidence of reported pain relief following local anesthetic injection into concordantly painful lumbar intervertebral discs. Pain Med Malden Mass. 2012;13(12):1547–53.
- 181. Derby R, Lee J-E, Lee S-H. Analgesic discography: effect of adding a local anesthetic to routine lumbar provocation discography. Pain Med Malden Mass. 2010;11(9):1335–42.
- 182. DePalma MJ, Lee J-E, Peterson L, Wolfer L, Ketchum JM, Ketcum J, et al. Are outer annular fissures stimulated during diskography the source of diskogenic low-back pain? An analysis of analgesic diskography data. Pain Med Malden Mass. 2009;10(3):488–94.
- 183. Peng B, Pang X, Wu Y, Zhao C, Song X. A randomized placebo-controlled trial of intradiscal methylene blue injection for the treatment of chronic discogenic low back pain. Pain. 2010;149(1): 124–9.
- 184. Kim S-H, Ahn S-H, Cho Y-W, Lee D-G. Effect of intradiscal methylene blue injection for the chronic discogenic low back pain: one year prospective follow-up study. Ann Rehabil Med. 2012;36(5):657–64.

- 185. Carl A, Ledet E, Yuan H, Sharan A. New developments in nucleus pulposus replacement technology. Spine J Off J N Am Spine Soc. 2004; 4(6 Suppl):325S–9.
- Coric D, Mummaneni PV. Nucleus replacement technologies. J Neurosurg Spine. 2008;8(2):115–20.
- 187. Goins ML, Wimberley DW, Yuan PS, Fitzhenry LN, Vaccaro AR. Nucleus pulposus replacement: an emerging technology. Spine J Off J N Am Spine Soc. 2005;5(6 Suppl):317S–24.
- 188. Risbud MV, Albert TJ, Guttapalli A, Vresilovic EJ, Hillibrand AS, Vaccaro AR, et al. Differentiation of mesenchymal stem cells towards a nucleus pulposus-like phenotype in vitro: implications for cell-based transplantation therapy. Spine. 2004;29(23):2627–32.
- 189. Mwale F, Roughley P, Antoniou J. Distinction between the extracellular matrix of the nucleus pulposus and hyaline cartilage: a requisite for tissue engineering of intervertebral disc. Eur Cell Mater. 2004;8:58–63. discussion 63–64.
- 190. Cloyd JM, Malhotra NR, Weng L, Chen W, Mauck RL, Elliott DM. Material properties in unconfined compression of human nucleus pulposus, injectable hyaluronic acid-based hydrogels and tissue engineering scaffolds. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2007;16(11):1892–8.
- 191. Iatridis JC, Nicoll SB, Michalek AJ, Walter BA, Gupta MS. Role of biomechanics in intervertebral disc degeneration and regenerative therapies: what needs repairing in the disc and what are promising biomaterials for its repair? Spine J Off J N Am Spine Soc. 2013;13(3):243–62.
- 192. Schek RM, Michalek AJ, Iatridis JC. Genipincrosslinked fibrin hydrogels as a potential adhesive to augment intervertebral disc annulus repair. Eur Cell Mater. 2011;21:373–83.
- 193. Allen MJ, Schoonmaker JE, Bauer TW, Williams PF, Higham PA, Yuan HA. Preclinical evaluation of a poly (vinyl alcohol) hydrogel implant as a replacement for the nucleus pulposus. Spine. 2004;29(5):515–23.
- 194. Richardson SM, Curran JM, Chen R, Vaughan-Thomas A, Hunt JA, Freemont AJ, et al. The differentiation of bone marrow mesenchymal stem cells into chondrocyte-like cells on poly-L-lactic acid (PLLA) scaffolds. Biomaterials. 2006;27(22):4069–78.
- 195. Wang BH, Campbell G. Formulations of polyvinyl alcohol cryogel that mimic the biomechanical properties of soft tissues in the natural lumbar intervertebral disc. Spine. 2009;34(25):2745–53.
- 196. Vernengo J, Fussell GW, Smith NG, Lowman AM. Synthesis and characterization of injectable bioadhesive hydrogels for nucleus pulposus replacement and repair of the damaged intervertebral disc. J Biomed Mater Res B Appl Biomater. 2010;93(2):309–17.

# Endoscopic Procedures for the Lumbar Spine: A Comprehensive View

34

Sebastian Ruetten, Martin Komp, Semih Oezdemir, and Patrick Hahn

## 34.1 Terminology and Definitions

Full-endoscopic lumbar surgery describes the surgical technique associated with lumbar spinal canal decompression, carried out under continuous visual control and irrigation using an approach associated with minimal trauma. It is not an endoscopic-assisted procedure through a tubular retractor. Rather, it is a uniportal technique using endoscopes with intraendoscopic working channels. Apart from reduced surgical trauma, it yields the benefits of arthroscopic procedures, such as improvement in visualization and light conditions. Two differing approaches are applied – the full-endoscopic interlaminar and the full-endoscopic transforaminal/extraforaminal approaches.

Center for Spine Surgery and Pain Therapy, Center for Orthopaedics and Traumatology of the St. Elisabeth Group – Catholic Hospitals Rhein-Ruhr, St. Anna Hospital Herne/Marien Hospital Herne University Hospital/Marien Hospital Witten, Hospitalstrasse 19, 44649 Herne, Germany e-mail: spine-pain@elisabethgruppe.de, info@s-ruetten.com; martinkomp@web.de; semih23@gmx.de; patrick-hahn@gmx.de

# 34.2 Surgical Principle

Over the past 90 years, many different operations on the lumbar spine have been described [1-4]. In the meantime, these operations have been modified [5-12]. The focus was frequently on reducing invasiveness and improving visualization during the course of surgical intervention. The microscopically assisted technique, introduced nearly 40 years ago, remains the standard for spinal decompression operations [6, 9, 13].

Gaining access to the spinal canal using an interlaminar approach by means of complete or partial laminectomy was described at the beginning of the twentieth century [1-4]. Alternative methods for carrying out operations relating to pathology of the intervertebral disks were developed 30 years later [11]. The posterolateral approach for taking biopsies from vertebrae was described at the end of the 1940s [14]. Percutaneous operations have been applied since the beginning of the 1970s [10, 12, 15–17]. The microsurgical procedure using the microscope was also developed in the 1970s and has achieved the status of gold standard for interlaminar decompression in the spinal canal [6, 9, 13]. Endoscopes have been used since the beginning of the 1980s. Initially, they were used in order to inspect the intervertebral space after open decompressions were considered completed [8]. This developed into the endoscopic transforaminal technique with posterolateral access [18-22]. This technique was

S. Ruetten, MD, PhD (⊠) • M. Komp, MD S. Oezdemir, MD • P. Hahn, MD

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_34



**Fig. 34.1** (a) The lateral transforaminal approach for endoscopic surgery. (b) The interlaminar approach for endoscopic surgery

the most common procedure used for endoscopic operations carried out on patients with diseases of lumbar disks. Manuscripts regarding endoscopically assisted procedures have been published since the early 1990s [7, 23–26]. This relates to visualization of the open operation site using an endoscope on a monitor.

The endoscopic posterolateral transforaminal approach allows the intervertebral space to be accessed within the intervertebral foramen between the exiting and traversing spinal nerves. Direct removal of intraforaminal and extraforaminal sequestrated disk extravasations is possible through this approach [19, 27]. The removal of displaced disk material within the spinal canal under a retrograde resection, i.e., intradiskally through the annulus defect, was described and termed as the "in-out technique" [18, 28, 29]. However, there are frequently technical constraints due to specific anatomical and pathological circumstances that do not allow this technique. Direct access to extradiskal ventral epidural space with continuous visualization is hence necessary for adequate decompression. This is prevented if the posterolateral approach is being used, particularly in the caudal segments when a small intervertebral foramen deflects the endoscope from the epidural space into the disk [28, 30, 31].

As a result of these problems, the fullendoscopic lateral trans-/extraforaminal approach was developed. It provides access to the spinal canal, with a continuous view of vital anatomy (Fig. 34.1a). Working with irrigation fluid provides excellent visualization. No measurement is carried out in centimeters to define an entry point in the skin, but an individual anatomical determination is performed under radiographic control [32–34]. Despite the lateral approach and the possibility of bone resection, there are clearly defined indications – and hence also constraints. These relate to mobility and obstruction of the approach by the pelvis or organ systems [32–34].

The constraints associated with the transforaminal approach motivated development of the full-endoscopic interlaminar approach which also permits surgery on pathologies that are outside the indication spectrum of the transforaminal procedure [33–36] (Fig. 34.1b).

Today, the combination of the new surgical approaches and advanced technical developments permits the first full-endoscopic procedure under visualization, which is equivalent to conventional operations. The transforaminal procedure is subject to more constraints compared with the interlaminar procedure, but at the same time provides optimal preservation of tissue. The anatomical and pathological conditions mean that the percentage ratio of transforaminal to interlaminar procedures is about 30–70.

## 34.3 Patient Selection

#### 34.3.1 General Indication

The indication for surgery must be determined according to today's standards on the basis of radicular symptoms, neurogenic claudication, and existing neurological deficits [37, 38]. Isolated back pain cannot usually be improved by decompressing operations. Existing secondary pathologies, such as instabilities, may have to be treated at the same time using other procedures. The following anatomical indications are currently unequivocal:

- Sequestered or non-sequestered lumbar disk herniations inside the spinal canal
- Sequestered or non-sequestered lumbar disk herniations intra- or extraforaminal
- Sequestered or non-sequestered recurrent disk herniations independent of localization
- Lateral spinal canal stenosis (recess stenosis) (bony, ligamentous, diskal)
- Central spinal stenosis (bony, ligamentous, diskal)
- Spinal stenosis due to cysts of the zygapophyseal joint
- In special cases, positioning of implants in the intervertebral space
- In special cases, intervertebral debridement and draining in spondylodiskitis or epidural abscess

# 34.3.2 Indications for Trans-/ Extraforaminal Approach

• All intraforaminal and extraforaminal disk herniations or cysts of the zygapophyseal joints are indications for the transforaminal approach.

In disk herniations within the spinal canal, the following inclusion criteria need to be taken into account due to the limited mobility:

- Location of the sequestered disk reaching cranially and caudally to maximally at the start of the pedicle above and below the level in question (Fig. 34.2a)
- In lateral X-ray pelvic overlay of the level in question to maximally at the middle of the pedicle above (Fig. 34.2b)

In the case of lateral spinal canal stenosis, only foraminal stenosis caused by intraforaminal/ extraforaminal cysts of the zygapophyseal joints is regarded as an indication for the transforaminal/ extraforaminal approach.



**Fig. 34.2** (a) Maximal sequestration for the transforaminal approach (*lines*, start of the pedicle above and below the level in question). (b) Maximal overlaying of the

pelvic for the transforaminal approach (*arrows*, iliac crest; *line*, middle of the pedicle above the level in question)

# 34.3.3 Indications for Interlaminar Approach

- All disk herniations or cysts of the zygapophyseal joint located within the spinal canal which cannot be operated technically in the transforaminal approach because of the criteria cited are taken as indications for the interlaminar approach.
- The interlaminar technique can be used as an alternative for surgery in all spinal disk herniations in the spinal canal comprised within the inclusion criteria for transforaminal approach.
- Recess stenosis due to bony/ligamentous/diskal pathologies.
- Recess stenosis due to cysts of the zygapophyseal joints.
- Central spinal stenosis due to bony/ligamentous/diskal pathologies.
- Central spinal canal stenosis based on cysts of the zygapophyseal joints.

# 34.3.4 Contraindications

- All criteria which generally apply as contraindications to decompressing operations, taking into consideration the specific technical possibilities and the inclusion criteria of each surgical procedure are considered contraindications.
- Isolated back pain caused by associated pathologies, e.g., instabilities, deformities.
- Cauda equina syndrome: a conventional procedure should be considered here, particularly for legal reasons.

# 34.4 Advantages and Disadvantages

# 34.4.1 Advantages

Conventional open surgical procedures are indispensable today and will remain so in the future. The possible complications and consequential damage entailed by such procedures are familiar. New techniques must guarantee sufficient possibilities of attaining the surgical goal which are equal to those of established procedures [39].

Full-endoscopic operation, such as a truly minimally invasive procedure, offer advantages. These correspond largely to the advantages of microscope-assisted surgery, cited in each case, over the conventional open procedure. Fullendoscopic operations may thus be classified as the next step for technical advances in surgical techniques.

- Facilitation for the surgeon, due to excellent visualization, good illumination, and expanded field of vision with 25° endoscopes
- Cost-effective procedure due to short operating time, rapid rehabilitation, high rate of return to earlier activity levels, and low postoperative costs of care
- Reduced trauma and the resultant consequences for the surrounding tissue, the stabilizing structures of the spinal canal, and the epidural space
- Facilitated revision operations
- Reduced complication rate, such as dural injury, bleeding, infections, etc.
- Monitor image as training basis for assistants
- High level of patient acceptance

## 34.4.2 Disadvantages

The following are cited as specific disadvantages:

- Inclusion criteria for the different approaches must be complied with:
- Limited mobility in the transforaminal approach
- Limited possibility to expand the operation in the event of unforeseen hindrances
- Full-endoscopic suturing of dura technically not possible
- Challenging learning curve
- Lumbar transforaminal risk of injury to the emerging nerves as a result of the approach

## 34.5 Preoperative Planning

The preoperative preparation is the same as the preparation for conventional, microsurgical operations.

#### 34.5.1 Examinations

As with all microsurgical techniques, the intraoperative procedure must be planned preoperatively based on imaging and clinical findings. The goal is to perform the resection of spinal canal structures as sparingly as possible depending on the pathology and provide adequate neurological decompression. Conventional X-rays of the lumbar spine and MR imaging are obligatory. In applying the lateral transforaminal approach, the access pathway may not be shifted by abdominal structures. Particular attention must be paid to this in the levels cranial to L4–L5. If the findings are not entirely clear, a single abdominal CT scan should be made through the disk level for evaluation and preoperative planning.

#### 34.5.2 Patients' Informed Consent

Patients must be informed about their disease, its possible long-term course, and consequences and, despite the minimal invasiveness and attendant advantages of the surgical procedure, all known side effects, complications, and therapeutic possibilities must be explained, as for conventional procedures. With reference to the full-endoscopic procedure, it is important to highlight that even with minimally invasive interventions, scarring may not be completely avoided. It is also important to emphasize that a switch to an open procedure may be required during the operation or subsequently in an additional procedure should unforeseen complications arise.

#### 34.5.3 Preparation

The preoperative preparation of the patients is the same as in microsurgical techniques.

A single-shot antibiotic is applied for infection prophylaxis.

#### 34.5.4 Anesthesia

Full-endoscopic operations can usually be performed under local or general anesthesia. General anesthesia has advantages because it is more convenient for both the patient and the surgeon, permits positioning as required, and also facilitates complex work within the spinal canal. In cases of local anesthesia in the interlaminar approach, anesthesia for the route of access and also of the neural structures is necessary. Due to inflammatory processes, epidural anesthesia alone is frequently not sufficient, and therefore, intrathecal administration of local anesthetic must be carried out. In addition, systemic sedation is necessary for immobilization. Positioning entails costly control of vital parameters and correction of anesthesiological problems can be difficult.

In transforaminal approach, there is a risk of damaging the exiting nerve route passing the foramen. Theoretically, the risk can be reduced with the possibility of communicating with the patient. Thus, the operation under local anesthesia is also prevalent.

#### 34.5.5 Positioning

The operation is performed with the patient in prone position on a radiolucent table, under twoplane radiological control. The patient lies on a hip and thoracic rolls to relieve the abdominal and thoracic organs. The operating table is lordotically or kyphotically adjustable intraoperatively at lumbar level depending on the anatomy and pathology.

### 34.5.6 Equipment

A radiolucent, electrically adjustable operation table and a C-arc are necessary. In addition to the surgical instruments and endoscopes, general equipments for endoscopic operations under fluid flow are needed, such as monitor, camera unit, light source, documentation system, fluid pump, shaver system, or radiofrequency generator. Equipment available for arthroscopy or endoscopy can be used. Depending on indication, the rod-lens endoscope has an outer diameter of 6.9 or 9.9 mm. The endoscope contains an intraendoscopic, eccentric working channel with diameter of 4.1 or 6.5 mm. The angle of vision is 25°. The working sheaths used have a beveled opening which enables creation of visual and working fields in an area without clear anatomically preformed cavity (Fig. 34.3a, b).

### 34.6 Surgical Technique

# 34.6.1 Lateral Transforaminal Approach

First, the skin incision is localized. The goal is to reach the spinal canal as tangentially as possible. At levels L4–L5 and L3–L4, in lateral X-ray path, the posterior line of the descending facet usually serves as the boundary which should not be crossed toward the ventral direction (Fig. 34.4a). To avoid injury to abdominal organs, a single abdominal CT scan through the individual disk should be made for evaluation and preoperative planning, especially at the cranial levels when findings are not unequivocal. Depending on the scan, an individual, less lateral approach should be selected.

An atraumatic spinal needle is inserted through the skin incision parallel running to the disk space in the target area. A practicable end point is the contact of the dorsal annulus in the medial pedicle line. After a target wire is inserted and the cannula removed, the cannulated dilator is inserted. It is absolutely essential to ensure that the dilator is located for all work steps at the level of the intervertebral disks and not displaced cranially as this can lead to damage of the emerging spinal nerves. The target wire is removed and the operation sheath with beveled opening is pushed through the dilator. When an appropriate position is attained, the site of the sheath opening is located at the medial pedicle line and the opening itself in the lateral ray path is positioned half in the ventral epidural space and in the dorsal annulus (Fig. 34.4b, c). From this point on, decompression is performed under visualization and continuous irrigation with isotonic saline without any special additives. The entire system is left open as standard so that the irrigation fluid can flow out. Further entry into the epidural space which may be required is made under visual control.

Annulus fragments are resected for dissection medially, until the disk herniation is localized and exposed. A rongeur is used to remove the disk herniation entirely or in parts. After complete resection, an unobstructed view is provided of the decompressed area. Depending on previous dissection, the dorsal longitudinal ligament can still be seen, which can be opened as necessary. The intervertebral space can then be cleared until the free intradiskal fragments are resected (Fig. 34.5a-c). The operation is implemented in the same way even if previous operations have been carried out in the operating area. After the operation has been completed, the instrument set is removed and the stab incision is closed. Drainage is not necessary.

# 34.6.2 Extraforaminal Approach

The extraforaminal approach can be used for intraforaminal/extraforaminal pathologies or for anatomical/pathological conditions which preclude a harmless, direct passage of the foramen due to the restricted diameter or the position of the emerging spinal nerves.

The spinal cannula is pushed forward to the caudal pedicle under X-ray control. This is a safe zone in which the emerging spinal nerve is not damaged. Dilator and operating sheath are then inserted (Fig. 34.6a). From this point on, the operation is carried out under visualization and continuous irrigation with isotonic saline. The entire system is left open as standard so that the irrigation fluid can drain away.

Pedicle, ascending facet, and disk are dissected and the foramen is exposed. The operating sheath is used as an instrument to hold the



Fig. 34.3 (a) Various endoscopes with intraendoscopic working channel. (b) Different instruments for the endoscopic surgery



**Fig. 34.4** (a) The posterior line (*green line*) of the descending facet (*red line*) should not be crossed toward the ventral direction for marking the skin incision. (b) Site of the sheath opening is located at the medial pedicle line

emerging spinal nerves cranially and ventrally. The extraforaminal port should also be selected maximally lateral so that it passes under the nerve cranially without significant manipulation.

The operation then continues from this position, such as direct decompression in the foramen, entry into the spinal canal through the foramen, or prior bone resection.

In the case of intraforaminal/extraforaminal disk herniation, the approach is determined by the location of the herniation, which is generally sequestrated rostrally. The exiting spinal nerve is moved further rostrally with the operating sheath and identified. The herniation is localized, dissected, and resected. In order to gain access further cranially under the spinal nerve, it can be lifted with the movable shaft of the rongeur. In the

(*red line*). (c) Opening of the sheath is positioned half in the ventral epidural space and in the dorsal annulus (*red line*, posterior wall of vertebral bodies)

same way, additional parts sequestrated cranially in the spinal canal can be resected. The intervertebral space can then be cleared (Fig. 34.6b, c).

In the case of intraforaminal/extraforaminal cysts of the zygapophyseal joints, dissection and precise identification of spinal nerve and cysts are important. The cysts are then opened, the material inside is removed, and the cyst walls are resected as far as possible.

Bone resection can be carried out in order to generally enlarge the foramen if it is constricted and to create a passage, but most frequently in order to achieve more mobility dorsally or caudally (Fig. 34.6d). Depending on the pathology, bone is resected in the ventral area of the ascending facet or in the cranial area of the caudal pedicle.



**Fig. 34.5** (a) Transforaminal operation inside the spinal canal. (b) Intraoperative view after decompression (*1*, traversing spinal nerve; *thin arrows*, posterior longitudinal

ligament; 2, intradiskal; *bold arrow*, radiofrequency electrode). (c) Intraoperative radiographic view

The extraforaminal approach at L5–S1 or in the last level presents a special situation, since the pelvis and the transverse process exert a particular influence on the approach. The caudal pedicle (S1) is the target for the spinal cannula. On account of the pelvis, the spinal cannula generally has a steep to a virtually posterior pathway after reaching the end position. After inserting the dilator, operating sheath, and endoscope, subsequent dissection is equivalent to the process for the standard extraforaminal approach but differs in implementation as the passage selected becomes steeper. This can result in the emerging spinal nerve being dissected and exposed directly from a dorsal position after the bony structures have been exposed, similar to the interlaminar technique. The operating sheath must also be used in a similar way. The precise performance of decompression depends on the findings of each case. Drainage is not necessary.

#### 34.6.3 Interlaminar Approach

The skin incision is made as medially as possible through the interlaminar window. The craniocaudal localization depends on the findings of the pathology.

The dilator is inserted bluntly on the lateral edge of the ligamentum flavum or on the descending facet of the zygapophyseal joint under radiographic posterior-anterior control. From this point onward, the operation is performed under radiographic lateral control. The operation sheath with beveled opening is inserted via the dilator in the direction of the ligament. The subsequent procedure is then performed under visualization and continuous irrigation with isotonic saline solution. The entire system is left open as standard so that the irrigation fluid can drain away.

In order to reach the spinal canal, the ligamentum flavum is incised laterally to approx.



**Fig. 34.6** (a) Target point (*green circle*) for the cannula in the extraforaminal approach. (b) The extraforaminal operation starts at the caudal pedicle. (c) Intraoperative

view after intraforaminal decompression (*thin arrows*, exiting nerve; *bold arrows*, radiofrequency electrode). (**d**) Bone resection to enlarge the foramen

3-5 mm. The subsequent procedure is enabled by the elasticity of the ligament. By rotating, the operation sheath with beveled opening can be used as a second instrument and serves, for example, as a nerve hook in shifting the neural structures in the medial direction.

The neural structures are identified prior to operating on the primary disk herniation, in particular the lateral boundary. If the recess cannot be visualized laterally to the traversing spinal nerve, medial portions of the ascending facet can be resected with the punch. If adequate space is available in the recess, the operating sheath can be introduced with the opening aligned medially on the floor of the spinal canal in order to shift the neural structures in a medial direction. The sheath is rotated with continuous contact to the base of the spinal canal. If the operating sheath cannot be introduced directly in the recess, the maneuver can be carried out with the aid of the dissector. A partial decompression through the axilla must be carried out prior to the maneuver involving sheath rotation. At the same time, this prevents parts of the disk herniation from being displaced medially together with the neural structures. The protruding disk herniation material is dissected and



**Fig. 34.7** (a) Interlaminar operation inside the spinal canal. (b) Intraoperative after decompression (*thin arrows*, flavum ligament; *bold arrow*, traversing spinal nerve; *1*,

resected. The intervertebral space can be cleared (Fig. 34.7a–c).

If the bony diameter of the interlaminar window does not permit passage or for large sequestered herniations, the window is enlarged using a burr and instruments. The descending facet joint is dissected and the medial edge and caudal pole are exposed. An incision is made on the surface leaf of the ligamentum flavum along its process at the medial edge of the descending joint facet. Bone resection begins at the caudal pole of the descending facet and continues along the medial part of the descending facet and toward the cranial lamina. Since the ligamentum flavum is

dura of cauda equina). (c) Intraoperative radiographic view. (d) Bone resection on the contralateral side in overthe-top technique

inserted caudally, directly at the bony edge of the lamina, burrs are used to thin the lamina here and the intervention is then continued with resection using a punch.

When revision operations are performed, no assessment on the implication of the ligamentum flavum can be made in advance of the operation – so that the dilator and operating sheath are introduced directly at the descending facet joint. The ongoing approach varies with each individual and depends on the degree of scarring and the type of operation carried out previously. If there is significant scarring, dissection directly along the edge of the bone and the descending facet in a ventral direction has proved effective. If direct entry to the recess is not possible, the medial edge of the ascending facet joint is dissected and the approach is carried out strictly at the bone in the direction of the spinal canal. Once the recess has been adequately dissected, the operating sheath is introduced. Depending on the degree of scarring, the maneuver involving rotation of the operating sheath displaces all the tissue en bloc medially. The neural structures may be fixed depending on the scarring. There may be an increased risk of damage as a result of manipulation. The force applied when displacing the neural structures therefore needs to be carefully moderated. If it is not possible to enter the recess, bone resection described above has to be carried out in advance.

As described above, bone resection has to be performed for operating on a recess stenosis. The medial portion of the descending facet joint or parts of the cranial lamina have to be resected as part of the standard procedure until the cranial tip of the ascending facet is reached. Experience indicates that bone resection caudally is adequate if it extends to the middle of the caudal pedicle. The ligamentum flavum is frequently involved in the pathology and then has to be resected in the lateral area over the entire craniocaudal extension. Depending on the characteristics of the pathology, the medial bony edge of the ascending facet is resected using a punch or burr until the recess has been exposed. The maneuver involving rotation of the sheath is used to shift the neural structures medially. If compression is caused by a protruding annulus or ventral osteophytes, they must be resected.

If a single-sided port is used for the central spinal canal stenosis, the approach used in the lateral stenosis of the ligamentum flavum has to be expanded by resecting medially up to the midline. In the case of contralateral decompression in the over-the-top technique with a single-sided approach, bone is already removed medially up to the spinous process during dissection of the caudal lamina depending on the characteristics of the pathology. After ipsilateral decompression has been completed, the operating sheath is inserted contralaterally and the contralateral liga-

mentum flavum, the contralateral bone of the lamina, and the descending facet and the medial edge of the ascending facet are resected until the spinal canal and the recess are exposed (Fig. 34.7d). As in microsurgical interventions, the central spinal canal stenosis can also be sufficiently decompressed contralaterally in the over-the-top technique. The detailed decompression of the recess in the cranial and caudal area is frequently subject to ipsilateral decompression. A bilateral approach with independent ports on both sides should therefore be considered in cases of bilateral recess stenosis with radicular symptoms. This enables the complete median area of the spinal canal and its structures to be retained, which is not involved with the pathology in recess stenosis.

The intraspinal cyst of the zygapophyseal joints generally occurs on one side and leads to symptoms of lateral or central spinal canal stenosis depending on the extent and localization. In addition to cyst resection, it may be necessary to carry out spinal canal expansion as described above. After opening the ligamentum flavum, this is resected until the cyst can be dissected and completely visualized. If possible, the cyst wall is separated from the dura. The cyst is opened, the material inside is removed, and the cyst wall is resected as far as possible. If necessary, a burr can be used on the joint side in order to resect all the parts of the cyst in contact with the joint.

### 34.7 Postoperative Care

The length of stay in hospital depends on the surgical measures carried out. Pure diskectomies or simple decompressions are treated with brief hospitalization or, if patient care at home is adequate, on an outpatient basis. Mobilization is immediate, as soon as this is possible following recovery from anesthesia. No medication is required for pain following the operation. Apart from patients with neurological deficits, no rehabilitative measures are necessary. Isometric and coordinative exercises can be performed without supervision once they have been learned. A passive lumbar brace is prescribed during the day for about 6 weeks. The level of exercise can be increased depending on the pathology and the patient's subjective sense of well-being. Return to work and sports are possible under the same conditions after the wound has healed. Limitations are imposed only to the extent that there should be no increase in pain during any activity. After more complex operations, the postsurgical treatment regimen is usually more restrictive and depends on the individual and the interventions carried out.

# 34.8 Complications and Avoidance

Possible complications during microsurgical procedures are known and there is a wide body of literature on this subject. A minimally invasive procedure is able to reduce the complication rate, although statistically this cannot be completely avoided. In principle, all the complications from conventional operating procedures may occur. These include the following:

- *Intraoperative complications:* surgery on the wrong segment, epidural bleeding, insufficient decompression, injuries to the dura, injuries to neural structures, injuries to vessels, injuries to organs
- Direct postoperative complications: persistent or preexisting radicular symptoms, cauda equina syndrome, urinary retention, consequences of injury to vessels or organs
- Delayed postoperative complications: soft tissue infection, spondylodiskitis, CSF fistula, delayed consequences of injury to vessels or organs, further radicular symptoms, surgically induced symptoms (failed back surgery syndrome)

As far as full-endoscopic procedures are concerned, it is important to emphasize that a onesided or two-sided switch to an open procedure may be necessary to carry out therapy in the event of a complication. In particular, endoscopic suture of a dural injury is not technically possible. Theoretically, if operating times are extended and blockage for outflow of irrigation fluid is overlooked, the consequences of increased pressure within the spinal canal and the attached and neighboring structures cannot be completely ruled out. Operations should therefore be performed leaving the system open such that the irrigation fluid can overflow.

With the interlaminar approach, a long-lasting and uninterrupted excessive retraction of the neural structures with the working sheath in the medial direction must be avoided or made only intermittently in order to avoid the risk of neurological damage.

In the transforaminal approach, the risk of injury to the exiting nerves cannot be completely eliminated. The highest risk occurs while performing the approach itself. If the risk is to be avoided, it is necessary to remain strictly within the caudal aspect of the foramen (Fig. 34.8). Alternatively, if the foramen is narrowed, an extraforaminal approach should be performed if necessary. When the lateral access is used, it is important to ensure that abdominal organs do not block the access path. It is particularly important to take account of this at the levels cranial to L4–L5. If the findings are not entirely clear, a single abdominal CT scan should be made through the



**Fig. 34.8** Remain strictly within the caudal aspect (*green plane*) of the foramen during the approach to avoid damaging the exiting nerve (*red arrow*)

disk for purposes of evaluation, preoperative planning, and measuring of the approach. An important factor for the outcome is the correct indication for the procedure itself and the proper approach.

Especially during the learning curve, there is an increased risk of complications occurring in the initial surgeries, as with any new technique. Prior observation of and assisting in procedures and workshops involving practice on cadavers might be instructive. Strict adherence to the indication criteria for the appropriate full-endoscopic approach is necessary. In the first instance, operations should be carried out on "simple" cases where no difficulties are to be expected in view of the anatomical situation. The possibility of an intraoperative switch to a standard procedure is helpful if problems are encountered. Nonetheless, it is important to remember that difficulties can never be ruled out during the learning curve.

### 34.9 Critical Evaluations

The objective in the development of surgical therapy for radicular compression syndromes caused by disk herniations or spinal canal stenoses is to provide adequate decompression under optimum visualization conditions with minimal trauma induced by surgery and the resulting negative consequences of such trauma. When new techniques are introduced, the clinical results of conventional standard procedures must be attained as a minimum criterion. At the same time, advantages in surgical technique and/or clinical variables must be the objective.

The development of new rod-lens endoscopes with a large intraendoscopic working channel and appropriate instrument sets has provided the technical platform for full-endoscopic operation on all lumbar disk herniations, inside and outside the spinal canal, and on spinal stenosis [32–36, 40–42]. In order to ensure complete decompression with certainty, operations on disk herniations and spinal canal stenosis must also be carried out using a full-endoscopic technique and under continuous visualization. The development of the lateral transforaminal approach optimizes

and facilitates access to the spinal canal and working under continuous visualization [32-34]. This eliminates problems associated with the posterolateral approach. However, the lateral approach also entails clear inclusion and exclusion criteria and hence also constraints [32–34]. Today, the interlaminar approach can be used in cases that are inoperable using the transforaminal approach for technical reasons [33, 34, 36, 40-42]. The full-endoscopic techniques that have been developed can now produce results where the clinical outcome coincides with the conventional microsurgical procedure. Simultaneously, significant advantages are evidenced that remain consistent over subsequent follow-up periods of examination [32–36, 40–42].

Despite the developments over the past 10 years, there are clear limits to full-endoscopic techniques. Open and maximally invasive procedures are necessary today and will remain so in the future. Surgeons must be able to perform such operations, not simply so that they are in a position to offer patients the most appropriate procedure for their particular circumstances but also to enable them to deal safely with any problems and complications that may emerge during fullendoscopic interventions as in any other invasive procedure. The development of full-endoscopic techniques should not be evaluated as a replacement for existing standard operations, but as a complementary procedure and alternative within the overall concept of spine surgery.

## 34.9.1 Indication for Disk Herniation

Possible negative consequences of conventional operations on the lumbar spine are well known and documented in the literature [43–53]. A comparison of the literature and the underlying studies reveals that the full-endoscopic procedure reduces operating times, tissue trauma, and complications [34, 54–58]. This corresponds to the published benefits of a minimally invasive intervertebral and epidural approach. The current state of knowledge indicates that it is possible to avoid instabilities as a result of the ability to reduce or eliminate bone and ligament resection

combined with atraumatic curettage of the intervertebral space. The full-endoscopic technique minimizes the annulus defect and this appears to exert a protective influence [43, 59–68]. Rehabilitation measures following an operation are not necessary and comparatively high return to the performance level required for job and sporting activities is achieved [69]. There is no evidence of increased morbidity resulting from accompanying factors [55, 57]. The rate of recurrence demonstrates no significant differences with the conventional approach in a comparison of the literature and within the studies [70-74]. Revisions can be carried out using the same technique. The form of the disk herniation and the annulus defect appear to exert a greater influence on the rate of recurrence than the extent of curettage of the intervertebral space [5, 71, 75]. No relevant disadvantages for the application of the full-endoscopic technique when operating on disk herniations have been identified overall [32-34, 36]. At the same time, there is evidence of advantages in operating technique and reduced trauma in the area of the access path to the structures of the spinal canal. The transforaminal approach is evaluated as inducing less trauma on account of reduced bone and ligament resection. It is therefore assessed as the approach of first choice. However, the anatomical and pathological prerequisites entail significant restrictions so that the interlaminar approach has the greater spectrum of application.

# 34.9.2 Indication for Recurrent Disk Herniation

Recurrent disk herniations following diskectomies can never be completely excluded. The rate of relapse is described in the literature as being between 5 % and more than 20 % depending on the fragment type and annulus defect [5, 70–74]. When operating on recurring disk herniations, the risk of dura and nerve injuries is increased if there is already epidural scarring [57, 76–78]. Extensive dissection generally needs to be carried out in the operating area in order to reduce such injury and the resulting increase in trauma has to be taken into account. As a result, conditions such as segmental instability following surprogressive degeneration, increasing gery, epidural scarring, or arachnoiditis may occur [43, 46-50, 53, 79-82]. This may induce clinical symptoms and create difficulties for further revisions. The scarring connection between dura and paravertebral musculature may cause so-called tethering of the cauda equina [83-86]. The increasing resection of stabilizing structures is conducive to instability following surgery [43, 46–50, 53]. The trauma caused by the access pathway in the innervation area of the dorsal branch of the spinal nerve may exert a negative effect on the stabilizing and coordinative system [44, 87]. The aim is, therefore, to use techniques designed to preserve tissue when carrying out revisions, as in the case of primary operations [88, 89]. When using the full-endoscopic procedure, the parameters determining the results and the benefits are comparable with those evinced by the indication of primary spinal disk herniation in respect of reduced operating times, tissue trauma, and complications [34, 42, 54–58, 90, 91]. No relevant disadvantages have been identified by comparison with the conventional microscopically assisted technique [34, 42]. The same inclusion and exclusion criteria are applicable. The transforaminal access is particularly effective because it completely circumvents the existing epidural scarring caused by the previous operation.

## 34.9.3 Indication for Spinal Canal Stenosis

The same problems are discussed in relation to operations on spinal canal stenosis as in diskectomy [43–49, 52, 53, 76, 92, 93]. Resection of joints and soft tissue structures in the lateral and ventral area is generally more extensive on account of the pathology. Hence, any instability induced as a result of surgery always needs to be taken into account [43, 46–50, 53, 94]. Extensive decompressions, or additional instabilities and deformities, may require additive fusion. Attempts to reduce trauma are made through the

use of various tissue-conserving techniques [86, 95-101]. A key prerequisite for using fullendoscopic techniques was the development of appropriate abraders that permit bone resection under visualization. This provides the technical capability for adequate decompression of spinal canal stenosis [33, 34, 42]. When using the fullendoscopic approach, the parameters determining results and the benefits are comparable with those evinced by the indication of primary or recurring disk herniation in respect of complications, tissue trauma, and reduced operating times [34, 40–42, 54–58, 90, 91]. Also in this case, no relevant disadvantages have been identified by comparison with the conventional microscopically assisted technique [40, 41]. Only a small number of these stenoses meet the inclusion criteria for a transforaminal approach on account of anatomical and pathological constraints. This approach is therefore restricted to a small number of individual cases.

## 34.10 Other Areas of Application

The transforaminal and interlaminar approaches are possible in the area of the thoracic spine depending on pathology and anatomy. The main indication is constituted by thoracic spinal disk herniations without significant spinal cord compression that continue to induce symptoms despite conservative therapy. Generally, only pathologies in a lateral position are operable, since manipulations of the spinal cord have to be avoided due to the risk of lesion and a lateral transforaminal approach is precluded by the organs located in the thorax. Technical implementation of both ports is equivalent to the lumbar procedure and is possible from the cervical-thoracic to the thoracic-lumbar junction. Other indications may be posterior pathologies like facet cyst, epidural abscess, or spinal stenosis. Contrary to the lumbar spine, in the case of thoracic spine, there is a higher overall risk of injury to neural and surrounding structures and, on account of these constraints, in the implementation of the ports and during the surgical procedure. In borderline cases relating to anatomy,

pathology, and symptoms, an operation using the conventional procedure may be the only appropriate option.

The anterior transdiskal and the posterior foraminotomy techniques are available in the area of the cervical spine [88, 93, 102-106]. Only relatively small endoscopes with more restricted visualization relations and small intraendoscopic working channel can be used for the anterior transdiskal procedure. Some specific work stages have to be carried out in the absence of direct visualization under radiographic control, mobility in the spinal canal may be restricted, and adequate bone resection is limited. Greater mobility is provided dorsally as well as the possibility of carrying out all work stages in vision. The spinal disk is retained. Surgery focuses primarily on lateral pathologies due to the risk of damage to the spinal cord during manipulation, as known from standard posterior foraminotomy. Other indications may be posterior pathologies like facet cyst, epidural abscess, or spinal stenosis.

#### References

- Mixter WJ, Barr JS. Rupture of the intervertebral disc with involvement of the spinal canal. N Engl J Med. 1934;211:205–10.
- 2. Putti V. Pathogenesis of sciatic pain. Lancet. 1927;2:53.
- Steinke CR. Spinal tumors: statistics on a series of 330 collected cases. J Nerv Ment Dis. 1918;47:418–26.
- Stookey B. Compression of spinal cord due to ventral extradural chondromas: diagnosis and surgical treatment. Arch Neurol Psychiatry. 1928;20:275–91.
- Carragee EJ, Han MY, Suen PW, Kim D. Clinical outcomes after lumbar discectomy for sciatica: the effects of fragment type and anular competence. J Bone Joint Surg Am. 2003;85:102–8.
- Caspar W. A new surgical procedure for lumbar disc herniation causing less tissue damaging through a microsurgical approach. Adv Neurosurg. 1977;7:74– 7. In: Wüllenweber R, Brock M, editors.
- Destandau J. A special device for endoscopic surgery of lumbar disc herniation. Neurol Res. 1999;21:39–42.
- Forst R, Hausmann G. Nucleoscopy: a new examination technique. Arch Orthop Trauma Surg. 1983;101:219–21.
- Goald HJ. Microlumbar discectomy follow-up of 147 patients. Spine. 1978;3:183–5.
- Hijikata S. Percutaneous discectomy: a new treatment method for lumbar disc herniation. J Toden Hosp. 1975;5:5–13.

- Hult L. Retroperitoneal disc fenestration in low back pain and sciata. Acta Orthop Scand. 1956;20:342–8.
- Maroon JC, Onik G, Sternau L. Percutaneous automated discectomy: a new approach to lumbar surgery. Clin Orthop. 1989;238:64–70.
- Wilson DH, Kenning J. Microsurgical lumbar discectomy: preliminary report of 83 consecutive cases. Neurosurgery. 1979;42:137–40.
- Valls J, Ottolenghi CE, Schajowicz F. Aspiration biopsy in diagnosis of lesions of vertebral bodies. JAMA. 1948;136:376.
- Gottlob C, Kopchok G, Peng SH, Tabbara M, Cavaye D, White RA. Holmium:YAG laser ablation of human intervertebral disc: preliminary evaluation. Lasers Surg Med. 1992;12:86–91.
- Kambin P, Gellman H. Percutaneous lateral discectomy of the lumbar spine: a preliminary report. Clin Orthop. 1983;174:127–32.
- Smith L, Garvin PJ, Gesler RM, Jennings RB. Enzyme dissolution of the annulus pulposus. Nature. 1963;198:1311–2.
- Kambin P, Casey K, O'Brien E, Zhou L. Transforaminal arthroscopic decompression of the lateral recess stenosis. J Neurosurg. 1996;84: 462–7.
- Lew SM, Mehalic TF, Fagone KL. Transforaminal percutaneous endoscopic discectomy in the treatment of far-lateral and foraminal lumbar disc herniations. J Neurosurg. 2001;94:216–20.
- Mathews HH. Transforaminal endoscopic microdiscectomy. Neurosurg Clin N Am. 1996;7:59–63.
- Mayer HM, Brock M. Percutaneous endoscopic discectomy: surgical technique and preliminary results compared to microsurgical discectomy. J Neurosurg. 1993;78:261.
- Savitz MH. Same-day microsurgical arthroscopic lateral-approach laser-assisted (SMALL) fluoroscopic discectomy. J Neurosurg. 1994;80:1039–45.
- Brayda-Bruno M, Cinnella P. Posterior endoscopic discectomy (and other procedures). Eur Spine J. 2000;9:24–9.
- Nakagawa H, Kamimura M, Uchiyama S, Takahara K, Itsubo T, Miyasaka T. Microendoscopic discectomy (MED) for lumbar disc prolapse. J Clin Neurosci. 2003;10:231–5.
- Perez-Cruet MJ, Foley KT, Isaacs RE, Rice-Wyllie L, Wellington R, Smith MM. Microendoscopic lumbar discectomy: technical note. Neurosurgery. 2002;51:129–36.
- Schick U, Doehnert J, Richter A, Konig A, Vitzkun HE. Microendoscopic lumbar discectomy versus open surgery: an intraoperative EMG study. Eur Spine. 2002;11:20–6.
- Jang JS, An SH, Lee SH. Transforaminal percutaneous endoscopic discectomy in the treatment of foraminal and extraforaminal lumbar disc herniations. J Spinal Disord Tech. 2006;19:338–43.
- Kambin P, O'Brien E, Zhou L, Schaffer JL. Arthroscopic microdiscectomy and selective fragmentectomy. Clin Orthop. 1998;347:150–67.

- Stuecker R. The transforaminal endoscopic approach. In: Mayer HM, editor. Minimally invasive spine surgery. Berlin: Springer; 2000. p. 201–6.
- Lee SH, Kang BU, Ahn Y, Choi G, Choi YG, Ahn KU, et al. Operative failure of percutaneous endoscopic lumbar discectomy: a radiologic analysis of 55 cases. Spine. 2006;31:285–90.
- Schubert M, Hoogland T. Endoscopic transforaminal nucleotomy with foraminoplasty for lumbar disc herniation. Oper Orthop Traumatol. 2005;17:641–61.
- 32. Ruetten S, Komp M, Godolias G. An extreme lateral access for the surgery of lumbar disc herniations inside the spinal canal using the full-endoscopic uniportal transforaminal approach. – Technique and prospective results of 463 patients. Spine. 2005;30:2570–8.
- 33. Ruetten S, Komp M, Merk H, Godolias G. Use of newly developed instruments and endoscopes: fullendoscopic resection of lumbar disc herniations via the interlaminar and lateral transforaminal approach. J Neurosurg Spine. 2007;6:521–30.
- Ruetten S, Komp M, Merk H, Godolias G. Fullendoscopic interlaminar and transforaminal lumbar discectomy versus conventional microsurgical technique: a prospective, randomized, controlled study. Spine. 2008;33:931–9.
- Ruetten S. The full-endoscopic interlaminar approach for lumbar disc herniations. In: Mayer HM, editor. Minimally invasive spine surgery. Berlin: Springer; 2005. p. 346–55.
- 36. Ruetten S, Komp M, Godolias G. A new fullendoscopic technique for the interlaminar operation of lumbar disc herniations using 6 mm endoscopes: prospective 2-year results of 331 patients. Minim Invasive Neurosurg. 2006;49:80–7.
- Andersson GBJ, Brown MD, Dvorak J, Herzog RJ, Kambin P, Malter A, et al. Consensus summary on the diagnosis and treatment of lumbar disc herniation. Spine. 1996;21:75–8.
- McCulloch JA. Focus issue on lumbar disc herniation: macro- and microdiscectomy. Spine. 1996;21:45–56.
- Maroon JC. Current concepts in minimally invasive discectomy. Neurosurgery. 2002;51:137–45.
- 40. Komp M, Hahn P, Merk H, Godolias G, Ruetten S. Bilateral operation of lumbar degenerative central spinal stenosis in full-endoscopic interlaminar technique with unilateral approach: prospective 2-year results of 74 patients. J Spinal Disord Tech. 2011;24:281–7.
- 41. Ruetten S, Komp M, Merk H, Godolias G. Surgical treatment for lumbar lateral recess stenosis with the full-endoscopic interlaminar and transforaminal approach versus conventional microsurgical technique: a prospective, randomized, controlled study. J Neurosurg Spine. 2009;10:476–85.
- 42. Ruetten S, Komp M, Merk H, Godolias G. Recurrent lumbar disc herniation following conventional discectomy: a prospective, randomized study comparing full-endoscopic interlaminar and transforaminal versus microsurgical revision. J Spinal Disord Tech. 2009;22:122–9.

- Abumi K, Panjabi MM, Kramer KM, Duranceau J, Oxland T, Criso JJ. Biomechanical evaluation of lumbar spinal stability after graded facetectomies. Spine. 1990;15:1142–7.
- 44. Cooper R, Mitchell W, Illingworth K, Forbes WS, Gillespie JE, Jayson MI. The role of epidural fibrosis and defective fibrinolysis in the persistence of postlaminectomy back pain. Spine. 1991;16:1044–8.
- 45. Fritsch EW, Heisel J, Rupp S. The failed back surgery syndrome: reasons, intraoperative findings and long term results: a report of 182 operative treatments. Spine. 1996;21:626–33.
- 46. Haher TR, O'Brien M, Dryer JW, Nucci R, Zipnick R, Leone DJ. The role of the lumbar facet joints in spinal stability. Identification of alternative paths of loading. Spine. 1994;19:2667–70.
- Hopp E, Tsou PM. Postdecompression lumbar instability. Clin Orthop. 1988;227:143–51.
- Kaigle AM, Holm SH, Hansson TH. Experimental instability in the lumbar spine. Spine. 1995;20:421–30.
- Kato Y, Panjabi MM, Nibu K. Biomechanical study of lumbar spinal stability after osteoplastic laminectomy. J Spinal Disord. 1998;11:146–50.
- Kotilainen E, Valtonen S. Clinical instability of the lumbar spine after microdiscectomy. Acta Neurochir. 1993;125:120–6.
- Ruetten S, Meyer O, Godolias G. Endoscopic surgery of the lumbar epidural space (epiduroscopy): results of therapeutic intervention from 93 patients. Minim Invasive Neurosurg. 2003;46:1–4.
- Schoeggl A, Maier H, Saringer W, Reddy M, Matula C. Outcome after chronic sciatica as the only reason for lumbar microdiscectomy. J Spinal Disord Tech. 2002;15:415–9.
- Sharma M, Langrana NA, Rodrigues J. Role of ligaments and facets in lumbar spinal stability. Spine. 1995;20:887–900.
- Mayer HM. The microsurgical interlaminar, paramedian approach. In: Mayer HM, editor. Minimally invasive spine surgery. Berlin: Springer; 2000. p. 79–91.
- Ramirez LF, Thisted R. Complications and demographic characteristics of patients undergoing lumbar discectomy in community hospitals. Neurosurgery. 1989;25:226–31.
- Rantanen J, Hurme M, Falck B, Alaranta H, Nykvist F, Lekto M. The lumbar multifidus muscle five year after surgery for a lumbar intervertebral disc herniation. Spine. 1993;18:568–74.
- Stolke D, Sollmann WP, Seifert V. Intra- and postoperative complications in lumbar disc surgery. Spine. 1989;14:56–9.
- Wildfoerster U. Intraoperative complications in lumbar intervertebral disc operations. Cooperative study of the spinal study group of the German Society of Neurosurgery. Neurochirurgica. 1991;34:53–6.
- 59. Aydin Y, Ziyal IM, Dumam H, Turkmen CS, Basak M, Sakin Y. Clinical and radiological results of lumbar microdiscectomy technique with preserving of ligamentum flavum comparing to the standard micro-discectomy technique. Surg Neurol. 2002;57:5–13.

- De Devitiis E, Cappabianca P. Lumbar discectomy with preservation of the ligamentum flavum. Surg Neurol. 2002;58:68–9.
- Ebara S, Harada T, Hosono N, Yonenobu K, Hiroshima K, Ono K. Intraoperative measurement of lumbar spinal instability. Spine. 1992;17:44–50.
- Faulhauer K, Manicke C. Fragment excision versus conventional disc removal in the microsurgical treatment of herniated lumbar disc. Acta Neurochir. 1995;133:107–11.
- Goel VK, Nishiyama K, Weinstein JN, Lin YK. Mechanical properties of lumbar spinal motion segments as affected by partial disc removal. Spine. 1986;11:1008–12.
- 64. Iida Y, Kataoka O, Sho T, Sumi M, Hirose T, Bessko Y, et al. Postoperative lumbar spinal instability occurring or progressing secondary to laminectomy. Spine. 1990;15:1186–9.
- Johnsson KE, Redlund-Johnell I, Uden A, Willner S. Preoperative and postoperative instability in lumbar spinal stenosis. Spine. 1989;14:591–3.
- 66. Mochida J, Toh E, Nomura T. The risks and benefits of percutaneous nucleotomy for lumbar disc herniation. A 10-year longitudinal study. J Bone Joint Surg Br. 2001;83:501–5.
- Zander T, Rohlmann A, Kloeckner C, Bergmann G. Influence of graded facetectomy and laminectomy on spinal biomechanics. Eur Spine J. 2003;12: 427–34.
- Zollner J, Rosendahl T, Herbsthofer B, Humke T, Eysel P. The effect of various nucleotomy techniques on biomechanical properties of the intervertebral disc. Z Orthop. 1999;137:206–10.
- Donceel P, Du Bois M. Fitness for work after lumbar disc herniation: a retrospective study. Eur Spine. 1998;7:29–35.
- Boyer P, Srour R, Buchheit F, Krause D, Albuquerque M. Lumbar disc hernia. Excision of hernia with or without complementary discectomy? Neurochirurgie. 1994;40:259–62.
- Carragee EJ, Spinnikie AO, Alamin TF, Paragioudakis S. A prospective controlled study of limited versus subtotal posterior discectomy: short-term outcomes in patients with herniated lumbar intervertebral discs and large posterior anular defect. Spine. 2006;31:653–7.
- Hirabayashi S, Kumano K, Ogawa Y, Aota Y, Machiro S. Microdiscectomy and second operation for lumbar disc herniation. Spine. 1993;18:2206–11.
- Stambough JL. Lumbar disk herniation: an analysis of 175 surgically treated cases. J Spinal Disord. 1997;10:488–92.
- Wenger M, Mariani L, Kalbarczyk A, Groger U. Long-term outcome of 104 patients after lumbar sequestrectomy according to Williams. Neurosurgery. 2001;49:329–34.
- Yorimitsu E, Chiba K, Toyama Y, Hirabayashi K. Long-term outcomes of standard discectomy for lumbar disc herniation. Spine. 2001;26:652–7.
- Kim SS, Michelsen CB. Revision surgery for failed back surgery syndrome. Spine. 1992;17:957–60.

- Law JD, Lehman RAW, Kirsch WM. Reoperation after lumbar intervertebral disc surgery. J Neurosurg. 1978;94:259–63.
- Waddell G, Reilly S, Torsney B, Allan DB, Morris EW, Di Paola MP. Assessment of the outcome of low back surgery. J Bone Jt Surg Br. 1988;70:723–7.
- Connolly ES. Surgery for recurrent lumbar disc herniation. Clin Neurosurg. 1992;39:211–6.
- Ebeling U, Kalbarcyk H, Reulen HJ. Microsurgical reoperation following lumbar disc surgery. Timing, surgical findings, and outcome in 92 patients. J Neurosurg. 1989;70:397–404.
- Fandino J, Botana C, Viladrich A, et al. Reoperation after lumbar disc surgery: results in 130 cases. Acta Neurochir Wien. 1993;122:102–4.
- Jonsson B, Stromqvist B. Repeat decompression of lumbar nerve roots: a prospective two-year evaluation. J Bone J Surg Br. 1993;75:894–7.
- Hall S, Bartleson JD, Onofrio BM, et al. Lumbar spinal stenosis: clinical features, diagnostic procedures, and results of surgical treatment in 68 patients. Ann Intern Med. 1985;103:271–5.
- LaRocca H, MacNab I. The laminectomy membrane. Studies in its evaluation, characteristics, effects and prophylaxis in dogs. J Bone J Surg Br. 1974;56:545–50.
- Markwalder TM. Surgical management of neurogenic claudication in 100 patients with lumbar spinal stenosis due to degenerative spondylolisthesis. Acta Neurochir. 1993;120:136–42.
- Ragab AA, Fye MA, Bohlmann HH. Surgery of the lumbar spine for spinal stenosis in 118 patients 70 years of age or older. Spine. 2003;28:348–53.
- Lewis PJ, Weir BKA, Broad RW, Grace MG. Longterm prospective study of lumbosacral discectomy. J Neurosurg. 1987;67:49–54.
- Ahn Y, Lee SH, Shin SW. Percutaneous endoscopic cervical discectomy: clinical outcome and radiographic changes. Photomed Laser Surg. 2005;23:362–8.
- Isaacs RE, Podichetty V, Fessler RG. Microendoscopic discectomy for recurrent disc herniations. Neurosurg Focus. 2003;15:E11.
- Caspar W, Campbell B, Barbier DD, Kretschmmer R, Gotfried Y. The Caspar microsurgical discectomy and comparison with a conventional standard lumbar disc procedure. Neurosurgery. 1991;28:78–87.
- Wilson DH, Harbaugh R. Lumbar discectomy: a comparative study of microsurgical and standard technique. In: Hardy RW, editor. Lumbar disc disease. New York: Racen Press; 1982. p. 147–56.
- 92. Annertz M, Jonsson B, Stromqvist B, Holtas S. No relationship between epidural fibrosis and sciatica in the lumbar postdiscectomy syndrome. A study with contrast-enhanced magnetic resonance imaging in symptomatic and asymptomatic patients. Spine. 1995;20:449–53.

- Kotilainen E. Percutaneous nucleotomy in the treatment of cervical disc herniation: report of three cases and review. Minim Invasive Neurosurg. 1999;42:152–5.
- Kotilainen E. Clinical instability of the lumbar spine after microdiscectomy. In: Gerber BE, Knight M, Siebert WE, editors. Lasers in the musculoskeletal system. Berlin: Springer; 2001. p. 241–3.
- Frank EH, Hsu FP. An endoscopic dural retractor for spinal stenosis surgery. Minim Invasive Neurosurg. 2002;45:136–8.
- Getty CJM, Johnson JR, Kirwan E, O'Sullivan MF. Partial undercutting facetectomy for bony entrapment of the lumbar nerve root. J Bone J Surg Br. 1981;63:330–5.
- Guiot BH, Khoo LT, Fessler RG. A minimally invasive technique for decompression of the lumbar spine. Spine. 2002;27:432–8.
- Khoo LT, Fessler RG. Microendoscopic decompressive laminotomy for the treatment of lumbar stenosis. Neurosurgery. 2002;51:146–54.
- 99. Mayer HM, List J, Korge A, Wiechert K. Microsurgery of acquired degenerative lumbar spinal stenosis. Bilateral over-the-top decompression through unilateral approach. Orthopaede. 2003;32:889–95.
- Sanderson PL, Getty CJM. Long-term results of partial undercutting facetectomy for lumbar lateral recess stenosis. Spine. 1996;21:1352–6.
- 101. Young S, Veerapen R, O'Laoire SA. Relief of lumbar canal stenosis using multilevel subarticular fenestration as an alternative to wide laminectomy: a preliminary report. Neurosurgery. 1988;23:628–33.
- 102. Ruetten S, Komp M, Merk H, Godolias G. A new full-endoscopic technique for cervical posterior foraminotomy in the treatment of lateral disc herniations using 6.9-mm endoscopes: prospective 2-year results of 87 patients. Minim Invas Neurosurg. 2007;50:219–26.
- 103. Ruetten S, Komp M, Merk H, Godolias G. Fullendoscopic cervical posterior foraminotomy for the operation of lateral disc herniations using 5.9-mm endoscopes: a prospective, randomized, controlled study. Spine. 2008;33:940–8.
- 104. Ruetten S, Komp M, Merk H, Godolias G. Fullendoscopic anterior decompression versus conventional anterior decompression and fusion in cervical disc herniations. Int Orthop Int Orthop. 2008. doi:10.1007/s00264-008-0684-y.
- 105. Tajima T, Sakamoto H, Yamakawa H. Diskectomy cervicale percutanee. Rev Med Orthop. 1989;17: 7–10.
- 106. Zhou YC, Zhou YQ, Wang CY. Percutaneous cervical discectomy for treating cervical disc herniation – a report of 12 cases. J Tongji Med Univ. 1994;14:110–3.

# Minimally Invasive Lumbar Disk Herniation Surgery with Tubular Retractors: Indications and Technical Aspects

35

# Carter S. Gerard, Ricardo B.V. Fontes, Laura A. Snyder, Lee A. Tan, and Richard G. Fessler

# 35.1 Introduction

Removal of herniated lumbar disk material is one of the most common procedures performed by spine surgeons, with the ultimate goal of alleviating radiculopathy through decompression of the nerve root [1]. Surgical treatment of patients with neural element compression due to lumbar disk herniation was first described in the American literature in 1829 by A. G. Smith [2, 3]. While the goals of surgery have not changed, surgical technique has seen considerable evolution. In the early twentieth century, full laminectomy with transdural approach was abandoned in favor of a hemilaminectomyextradural approach [4]. In the 1970s, Yasargil, Caspar, and Williams introduced microsurgical technique to the treatment of lumbar disk disease, applying principles of minimal and careful dura and root manipulation to spine surgery, initially envisioned for application in cranial surgery [5]. The principles of manipulation of neural elements were mostly laid out at that time, and accordingly, neurological morbidity

L.A. Snyder, MD • L.A. Tan, MD •

R.G. Fessler, MD, PhD

Rush University Medical Center, 1725 W Harrison, Suite 855,

has remained stable since then. Further technical advances would basically try to minimize soft tissue and osseous trauma: Faubert and Caspar introduced tubular retractors in 1991, which allowed the development of the microendoscopic technique in 1997 by Foley and Smith [6, 7]. These technical advances have allowed minimally invasive surgery (MIS) to become increasingly popular with spine surgeons.

Lumbar diskectomy through a tubular retractor allows the surgeon to address the vast majority of disk herniations (Fig. 35.1) while minimizing collateral damage to superficial structures. Multiple randomized prospective trials comparing MIS diskectomy to the established open diskectomy have produced similar results in regard to outcomes [8–12]. Other studies have found muscle-splitting tubular retractors to reduce postoperative pain, limit blood loss, decrease rate of infection, and minimize damage to the paraspinal musculature and ligaments [12–14]. A reduction in damage to surrounding anatomy is not only vital in determining satisfactory outcomes in the immediate postoperative period but may also avoid longterm complications such as iatrogenic spondylolisthesis. In this chapter, we describe a minimally invasive lumbar microdiskectomy/ foraminotomy technique via tubular retractors that provides access to the neural foramen, preserves the midline structures, and results in minimal muscular injury.

J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_35

C.S. Gerard, MD (🖂) • R.B.V. Fontes, MD, PhD

Department of Neurosurgery,

Chicago, IL 60610, USA

e-mail: Carter\_Gerard@rush.edu

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016



**Fig. 35.1** Right-sided paramedian L5–S1 disk herniation with nerve compression amenable to tubular diskectomy (**a**, **b**). (**c**, **d**) Demonstrate a foraminal herniation that can

be addressed through the far lateral technique described in the chapter

# 35.2 Surgical Technique

Once diagnosis is established and a decision to undergo surgical treatment is made, patients should be examined to assess the access site and the body habitus. Obese patients may particularly benefit from the use of the tubular retraction systems [15, 16]. With standard open techniques, obese patients often require an incision two to four times the standard length to provide adequate exposure, whereas with tubular retractors the incision is the same for all patients. Patients should have a complete medical work-up prior to surgery and meet the criteria for general anesthesia. The majority of patients can be safely treated as outpatients. Age in isolation should not be a contraindication to surgery; the senior author has performed the procedure successfully on multiple patients in their 90s. Depending on their level of independence in the community and their home situation, elderly patients or patients with other comorbidities may benefit from an overnight hospital stay for observation.

#### 35.2.1 Equipment

A tubular retractor system such as the MetRx system (Medtronic Sofamor Danek, Memphis, TN) is required. Systems should include all of the necessary instruments for retractor placement such as the dilators, retractors, and retractor arm, but also provide extra-long, anti-glare-coated instruments including a variety and sizes of curettes, disk punches, and rongeurs. If utilizing the conventional microscope, bayonet-shaped instruments and an angled drill attachment are also helpful.

While the tubular retractor can be used with an operating microscope or a 30° angled endoscope, it is our belief that endoscopic diskectomy provides better visualization of the surgical anatomy over that provided by conventional microscopy. The coupler provided with the MetRx kit allows for the use of a variety of different endoscopes with either the 16 or 18 mm working channel. Lateral fluoroscopic guidance is required for localization and during the dilation phase and electrophysiological monitoring is not routinely utilized.

#### 35.2.2 Anesthesia and Positioning

This procedure can be safely performed under general, local, or spinal anesthesia [17, 18]. If being performed under general anesthesia, nondepolarizing neuromuscular blockers are avoided to allow for improved feedback from nerve root manipulation during surgery. Any position/frame/ bed combination that allows for a free-hanging abdomen is acceptable so pressure is not transmitted to the epidural veins and causes bothersome bleeding during surgery (Fig. 35.2). The surgeon should stand on the same side as the patient's pathology and a rail attachment placed on the opposite level for the table-mounted arm; typically, this is aligned with the patient's hip. Video monitor for the endoscope is also placed across the table so the surgeon can face it directly. If a microscope is to be used in lieu of endoscopy, the microscope base can be brought in from the side opposite to the C-arm base. In either case, it is usually easiest to position the C-arm monitor at the foot of the operative table.

## 35.2.3 Surgical Technique: Paramedian Disk Herniation

The whole lumbar area is prepped and draped. The fluoroscopy C-arm is also draped in sterile fashion in an under-table position to enable quick radiological confirmation throughout the case. Localization of the level of interest is performed under lateral fluoroscopy counting cranially from the sacrum, noting vertebrae with transitional morphology. Specific adjustments may be made for migrated fragments. When the radiopaque marker is in the correct location, incision is planned 1.5 cm off midline on the side of the intended approach.

The operative site is then infiltrated with local anesthetic, while the assistant secures the clamp and retractor arm to the operative table (Fig. 35.3). An 11-blade scalpel is used to make a small 20-mm incision exposing the subcutaneous fat. A stainless steel guide pin or Kirschner wire is passed through the stab incision and soft tissues on to the underlying bone. Fluoroscopy should be used to confirm location of the guide pin during such passage to avoid penetrating the interlaminar space and causing a dural leak, especially if positioned on a Wilson frame. The safest approach is to direct the guide pin perpendicular to the entry point, so that the pin will contact the zygapophysial joint or even the transverse process rather than the lamina. Once the pin is docked on the zygapophysial joint, the first dilator is passed over the pin and removed; with the dilator, it is safe to angle medial and dock at the spinous process-lamina junction, directly over the disk space (Fig. 35.3). The remaining dilators are then passed sequentially and lastly, the



**Fig. 35.2** Patient positioned on a Wilson frame (**a**) which distracts the lamina at the level of interest. The articulating arm is attached contralaterally at hip level (**b**) and the 2-cm incision is marked 1.5 cm off-midline (**c**)

**Fig. 35.3** The incision is carried through the lumbar fascia (**a**). The guide pin is inserted and contacts the lamina (**b**, **c**). Subsequent dilators are placed one over another,

followed by the tubular retractors  $(\mathbf{d}, \mathbf{e})$ . The articulating arm is attached and the microscope or endoscope is brought into the field  $(\mathbf{f})$ 



working channel. These devices are introduced with a twisting motion so the dorsal lumbar fascia and underlying musculature are split along their fibers rather than torn. It is important to utilize fluoroscopy during this stage to confirm the trajectory and that the dilators are resting against the bone; this minimizes the amount of tissue that needs to be removed later. A working channel of 16 or 18 mm may be utilized for endoscopic applications and up to 22 mm for use with the microscope. In either case, it is connected to the table-mounted arm and secured in position, and the dilators are removed from its interior. Following a final fluoroscopic position check, the endoscope is attached to the working channel. Should the working tube be repositioned during the procedure, it is also recommended that fluoroscopy be used: even a seemly small adjustment may prove enough to aim at the wrong disk space.

Monopolar cautery and pituitary rongeurs are used to clear the remaining soft tissue off the lamina and inferior articular process, taking care to start the dissection over solid bone laterally. All muscle tissue should be detached completely along the circumference of the tube prior to attempting removal of the muscle (Fig. 35.4). Failure to do so can result in excessive bleeding and pulling more muscle into the field beneath the edges of the working channel. The "cut" setting on the cautery appears to be more effective than "coagulation" during initial exposure. Once the muscle has been cleared, a straight curette is used to identify and clear any residual soft tissue from the caudal edge of the lamina. The surgeon's view should be centered on the spinous process-lamina junction and the tubular retractor may need to be readjusted at this point. Straight and up-angled curettes are used to detach the ligamentum flavum medially from the overlying lamina and then work laterally under the lamina. Fluoroscopy should be used to confirm the location under the lamina dorsal to the desired disk space. Once an adequate plane has been established, 3 and 4 mm Kerrison rongeurs should be used in a rostral direction to perform a laminotomy. The up-angled curette is routinely utilized to confirm separation between the ligamentum flavum and lamina, in order to prevent injury to

the underlying dura and nerve roots. In general, the laminotomy should extend the length of the neural foramen, from pedicle to pedicle; the rostral insertion of the ligamentum flavum normally marks the cranial extent and exposed dura is seen adjacent to it. At the junction of the lamina and facet complex, it is usually necessary to use a high-speed drill to perform a medial resection of the inferior articular process until an up-angled curette can be passed easily into the neural foramen. It can be useful at this point to aim the tube slightly lateral and to utilize fluoroscopy with a probe or up-angled curette in the foramen to confirm the extent of the laminotomy and foraminotomy. With these maneuvers, a medial foraminotomy is performed, which can be a stand-alone procedure or a routine, initial part of every paramedian diskectomy. Bony bleeding is easily addressed by the use of bone wax. In most cases, the bony opening exposes the ligament superficial to the shoulder and lateral margin of the traversing nerve root that exits at the foramen one level caudally.

Attention is then directed toward removal of the ligamentum flavum. A blunt, delicate dissector is used to elevate the rostral end of the flavum from the dura. If another opening has been made elsewhere, it can also be used. A curved curette is then used to establish a plane between the dura and ligament and can be used to detach the ligament from the bone along the edges of the laminotomy. Right-angle rather than 60-degree-angled punches are useful for removal of the ligamentum. The ligament can be detached laterally from the superior articular process with an up-angled curette. In cases of disk herniation, the nerve root may be distorted by the disk herniation and it may be necessary to carefully resect more of the medial zygapophysial joint in order to adequately visualize the shoulder of the nerve root. Once the shoulder of the root is exposed, gentle medial retraction usually reveals the disk fragment in extruded cases or the protruding annulus with a contained fragment. This step may be difficult if a tethering fragment is present at the axilla; careful dissection at the axilla and initial partial removal of a fragment there may be required. A combination nerve root retractor and sucker is



**Fig. 35.4** Residual muscle is removed with cautery and rongeurs  $(\mathbf{a}, \mathbf{b})$ . A laminotomy is performed with bone punches and/or a drill  $(\mathbf{c})$ . Removing the ligamentum fla-

used to retract the thecal sac medially, or a gentle, blunt-tip suction may be utilized. Extruded fragments are removed with a variety of disk punches. In case an extruded disk fragment is expected but not initially visualized, a spatula or blunt hook can be used to explore the ventral aspect of the dura and the level should be checked with fluoroscopy.

vum exposes the nerve root (d) which is then retracted medially (e). Diskectomy and foraminotomy fully decompress the root (f)

If the nucleus fragment is still retained under the posterior longitudinal ligament and annulus, once the root is retracted, epidural veins may be coagulated with the bipolar forceps in a low setting. Coagulation should be minimized around the root, as well as manipulation through the axilla. The annulus is incised and nucleus fragment removed with a disk forceps. It is our routine to not aggressively explore the interspace and remove contained fragment but just the disk material that is loose and removed easily. If the 30° endoscope is used for the procedure, the endoscope can be directed medially to help with decompression of more medially located disks, a maneuver not possible with the operating microscope. When dealing with calcified annulus or large osteophytes adjacent to the disk, a downangled curette can be utilized to fracture the osteophyte into the annular opening and decompress the root. At the end of the decompression, a ball tip probe should be passed into the foramen to confirm adequacy.

Hemostasis is achieved while minimizing coagulation; simple tamponade with a number of different agents may be performed. We do not advocate closing the annular defect. The retractor arm is unlocked and the tube is removed under direct vision: bipolar cautery is used to coagulate more significant muscle arteries. Closure is performed in the usual fashion with fascial and subcuticular suture layers.

## 35.2.4 Surgical Technique: Foraminal Disk Herniation

A tubular MIS approach is particularly useful in the management of foraminal or "far lateral" disk herniations as it obviates a very large muscular dissection. In this situation, the incision is made 5-6 cm lateral to midline rather than the 1.5 cm used for paramedian herniations. Dilation is performed as described above. Desired site for docking of the initial dilator is at the caudal edge of the medial transverse process or, alternatively, the lateral margin of the pars interarticularis. The monopolar cautery cannot be utilized in this setting as the root is exposed: a straight curette is utilized to detach the muscle fibers from the transverse process and sweep it in a lateral-caudal direction. Once the lateral limits of the neural foramen are defined, a lateral foraminotomy can be performed with the high-speed drill and Kerrison punches. The exiting nerve root can be identified at the cranial part of the foramen just caudal to the foramen. It is then retracted

cranially, thus exposing the disk and diskectomy proceeds in the usual fashion. A greater or smaller degree of foraminotomy may be performed as desired based on the amount of disk expected to be removed (e.g., soft disk herniation versus calcified disk bulge with adjacent osteophytes).

## 35.2.5 Tips and Tricks

Working through the tubular retractor is not inherently different than standard microsurgical technique. If the endoscope is being utilized, it is important to attempt to maintain the working instrument and the suction tip in parallel and avoid "crossing" instruments.

· Frequent repositioning of the tubular retractor

The small working area of the tubular retractor can be multiplied by sequentially unlocking the retractor arm or adjusting or "wanding" the tubular retractor as necessary to visualize another area. This can be more effectively done with the last dilator inserted within the tube, to create a longer moment arm, while the assistant unlocks and locks the retractor arm. In this manner, the surgeon can better visualize relevant anatomy and address neural compression. It is important to repeat fluoroscopy imaging after each adjustment in order to avoid migration to an adjacent level.

• Use of the bipolar cautery for soft tissue and ligamentum flavum retraction

Once the tubular retractor is positioned, soft tissue may "creep under" the retractor and reduce visualization. We utilize the bipolar cautery with a special angled tip to coagulate and retract this soft tissue under the tube. This may also be utilized for remnants of the ligamentum flavum or annulus fibrosus that may be protruding into the spinal canal.

Wide exposure for very large disk herniations

For cases with large, extruded disk fragments, a wide exposure with complete hemilaminectomy is recommended. We strongly advise to obtain visualization of the normal anatomy cranial and/ or caudal to the fragment before attempting to retract the nerve root medially. Partial removal of whatever fragment is evident and loose may also be performed before root retraction.

Additional partial facetectomy

While instability is a concern in open cases, the surgeon must always be aware the visualized field is minimal when working through a tube (16 or 18 mm in diameter). Therefore, lateral angulation of the tube and additional removal of 2–3 mm of the facet is unlikely to result in instability, as discussed below. In 14 years performing this procedure, the senior author has only one case of delayed, clinically significant instability following a microdiskectomy. We thus advocate for additional removal of facet when the root cannot be clearly identified as opposed to blindly retracting the root and/or dura.

Use of a suction-retractor

In open microdiskectomy, a root retractor is normally utilized and the assistant may also utilize suction. In the restrained environment of the tubular retractor, the assistant is unable to assist with both these tasks. A combined root retractorsuction tip can be found in many commercially available sets of tubular systems and is particularly useful, as the assistant is then able to perform both tasks. The surgeon may then work two dissecting instruments, which may be particularly useful when trying to separate the root and dura from the herniated disk.

## 35.3 Postoperative Care

Patients are allowed to ambulate immediately after the procedure: 80–90 % of all our diskectomies are performed in the outpatient setting. Patients are discharged on mild narcotics and analgesics and seen in clinic two weeks postoperatively. NSAIDs and steroids may be utilized at the surgeon's discretion.

# 35.4 Potential Complications and Avoidance

Acute radiculopathy may occur in cases with large disk herniations requiring aggressive mobilization of the nerve root. In these patients a rapid methylprednisolone taper has proven quite useful. If suspected during surgery, deposit steroids may be utilized in situ. Although painful, the symptoms usually resolve within 7-10 days. Persistence of radiculopathy at the 6-week follow-up in well-indicated cases should raise the suspicion of incomplete decompression or retained fragment. Similarly, initial improvement and recurrence of symptoms may be an indicator of acute re-herniation. Both conditions are assessed with repeat MRI, and re-exploration is a consideration versus continued pain management. We have found that patients with acute disk re-herniations are likely to experience the same improvement with re-exploration they had initially.

Clinical instability requiring fusion is a recognized complication of lumbar diskectomy that occurs in 6-9 % of open cases [19, 20]. Preservation of the ligaments, intervertebral disk, and facet joints that compose the posterior tension band is vital to prevent destabilization. Facetectomy may be necessary to access lateral disks; however, this should be done sparingly and be less than 40 % to avoid significant mobility [21]. Although prospective data is lacking, case series demonstrate that MIS techniques allow for similar outcomes with lower iatrogenic instability rates, while biomechanical data support this affirmation due to preservation of the posterior tension band and smaller degree of facet removal [22, 23]. Ulterior arthrodesis of the operated segment may also be required due to persistent back pain. While aggressive removal of the disk significantly decreases the likelihood of recurrence, loss of disk height is directly related to subsequent instability and poor outcomes, but in this case, this complication is inherent to both open and MIS techniques [24, 25]. Therefore, diskectomy should focus on fragment removal and should stop once the nerve root is adequately decompressed.

Accidental durotomy is a known complication of diskectomy and happens in 4-9 % of cases. The incidence of such complication may actually be higher for MIS techniques due to the constraints of operating within a narrow channel and the learning curve [26, 27]. Paradoxically, the incidence of symptomatic pseudomeningocele requiring reoperation is far smaller with MIS techniques due to the absence of dead space and perfect apposition of soft tissue layers at the end of the procedure. Although intradural surgery is commonly performed through tubular channels including direct dural repair, it is a very challenging task through the 16-18-mmdiameter tubular retractors and ultimately unnecessary [28]. In durotomies of small to moderate sizes, we have been lately simply covering the defect with muscle, fat, blood-soaked Gelfoam (Pharmacia and Upjohn, Kalamazoo MI), or a dural substitute along with a layer of fibrin glue or dural sealant (Fig. 35.5). The patient is maintained on overnight bed rest in the hospital and ambulation is resumed the following morning. For larger dural tears, this strategy is complemented by 2-3 days of lumbar cerebrospinal fluid drainage to allow for healing of the surgical site. In addition, a larger tear may allow placement of one to two sutures to oppose the dural edges [26, 27].

Vascular and intra-abdominal injury is a potential complication of the MIS diskectomies during the localization and diskectomy phases. The incidence of arterial injury has been estimated at 5 per 10,000 diskectomy cases with a mortality rate reaching 65 % [29]. Fluoroscopy is essential during the dilation step to confirm the depth of the guide pin and ensure it never reaches deeper than the transverse process. During the diskectomy portion of the procedure, fluoroscopy can be used to confirm the depth of the instruments within the disk so that the anterior annulus is not violated, but the general rule of "minimal necessary diskectomy" outlined above should be enough to prevent anterior penetration. Arterial bleeding from the wound is frequently not seen because of a "self-sealing" effect of the anterior annulus [30]. Suspicion of an anterior breach along with hemodynamic instability should thus immediate and aggressive volume resuscitation and abdominal exploration. With prompt action, these injuries are survivable [30].

Fig. 35.5 Following an unintended durotomy, a piece of Gelfoam is placed over the defect (a) followed by dural sealant (b) and closure of the wound

#### Conclusion

Minimally invasive spinal surgery has rapidly been adopted by surgeons throughout the world as a philosophy to minimize unintended trauma to surrounding tissues. Tubular retractors are valuable tools within this context to adequately perform the end-procedure; lumbar diskectomy is just the simplest of them. There are multiple factors driving the acceptance of minimally invasive diskectomy including ease of use for the surgeon, patient demand, and the improved outcomes from minimizing exposure-related damage. Most surgeons in training today receive more or less extensive training in minimally invasive techniques. Those that have mastered open techniques are able, and encouraged, to apply that valuable knowledge to mastering lumbar diskectomy through tubular retractors and, in so doing, offering their patients the best care possible.

### References

- Rutkow IM. Orthopaedic operations in the United States, 1979 through 1983. J Bone Joint Surg Am. 1986;68(5):716–9.
- Robinson JS. Sciatica and the lumbar disk syndrome: a historic perspective. South Med J. 1983;76(2):232–8.
- Oppenheimer JH, DeCastro I, McDonnell DE. Minimally invasive spine technology and minimally invasive spine surgery: a historical review. Neurosurg Focus. 2009;27(3):E9. doi:10.3171/2009.7.FOCUS09121.
- Parisien RC, Ball PA. William Jason Mixter (1880– 1958). Ushering in the "dynasty of the disc". Spine. 1998;23(21):2363–6.
- Maroon JC. Current concepts in minimally invasive discectomy. Neurosurgery. 2002;51(5 Suppl):S137–45.
- Faubert C, Caspar W. Lumbar percutaneous discectomy. Initial experience in 28 cases. Neuroradiology. 1991;33(5):407–10.
- Foley KT, Smith MM. Microendoscopic discectomy. Tech Neurosurg. 1997;3:301–7.
- Franke J, Greiner-Perth R, Boehm H, et al. Comparison of a minimally invasive procedure versus standard microscopic discotomy: a prospective randomised controlled clinical trial. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2009;18(7):992–1000. doi:10.1007/ s00586-009-0964-2.

- Arts MP, Brand R, van den Akker ME, et al. Tubular diskectomy vs conventional microdiskectomy for the treatment of lumbar disk herniation: 2-year results of a double-blind randomized controlled trial. Neurosurgery. 2011;69(1):135–44. doi:10.1227/ NEU.0b013e318214a98c; discussion 144.
- Arts MP, Brand R, van den Akker ME, et al. Tubular diskectomy vs conventional microdiskectomy for sciatica: a randomized controlled trial. JAMA J Am Med Assoc. 2009;302(2):149–58. doi:10.1001/jama.2009.972.
- Dasenbrock HH, Juraschek SP, Schultz LR, et al. The efficacy of minimally invasive discectomy compared with open discectomy: a meta-analysis of prospective randomized controlled trials. J Neurosurg Spine. 2012;16(5):452–62. doi:10.3171/2012.1.SPINE11404.
- German JW, Adamo MA, Hoppenot RG, Blossom JH, Nagle HA. Perioperative results following lumbar discectomy: comparison of minimally invasive discectomy and standard microdiscectomy. Neurosurg Focus. 2008;25(2):E20. doi:10.3171/FOC/2008/25/8/E20.
- Shin DA, Kim KN, Shin HC, Yoon DH. The efficacy of microendoscopic discectomy in reducing iatrogenic muscle injury. J Neurosurg Spine. 2008;8(1):39–43. doi:10.3171/SPI-08/01/039.
- Bresnahan L, Fessler RG, Natarajan RN. Evaluation of change in muscle activity as a result of posterior lumbar spine surgery using a dynamic modeling system. Spine. 2010;35(16):E761–7. doi:10.1097/ BRS.0b013e3181e45a6e.
- Cole 4th JS, Jackson TR. Minimally invasive lumbar discectomy in obese patients. Neurosurgery. 2007;61(3):539–44. doi:10.1227/01. NEU.0000290900.23190.C9; discussion 544.
- Park P, Upadhyaya C, Garton HJL, Foley KT. The impact of minimally invasive spine surgery on perioperative complications in overweight or obese patients. Neurosurgery. 2008;62(3):693–9. doi:10.1227/01. neu.0000317318.33365.f1; discussion 693–9.
- Chen H-T, Tsai C-H, Chao S-C, et al. Endoscopic discectomy of L5–S1 disc herniation via an interlaminar approach: prospective controlled study under local and general anesthesia. Surg Neurol Int. 2011;2:93. doi:10.4103/2152-7806.82570.
- Yilmaz C, Buyrukcu SO, Cansever T, Gulsen S, Altinors N, Caner H. Lumbar microdiscectomy with spinal anesthesia: comparison of prone and knee-chest positions in means of hemodynamic and respiratory function. Spine. 2010;35(11):1176–84. doi:10.1097/ BRS.0b013e3181be5866.
- Schaller B. Failed back surgery syndrome: the role of symptomatic segmental single-level instability after lumbar microdiscectomy. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2004;13(3):193–8. doi:10.1007/s00586-003-0632-x.
- Parker SL, Xu R, McGirt MJ, Witham TF, Long DM, Bydon A. Long-term back pain after a single-level discectomy for radiculopathy: incidence and health
care cost analysis: clinical article. J Neurosurg Spine. 2010;12(2):178–82. doi:10.3171/2009.9.SPINE09410.

- Smith ZA, Vastardis GA, Carandang G, et al. Biomechanical effects of a unilateral approach to minimally invasive lumbar decompression. PLoS One. 2014;9(3):e92611. doi:10.1371/journal.pone.0092611.
- Lee MJ, Bransford RJ, Bellabarba C, et al. The effect of bilateral laminotomy versus laminectomy on the motion and stiffness of the human lumbar spine: a biomechanical comparison. Spine. 2010;35(19):1789– 93. doi:10.1097/BRS.0b013e3181c9b8d6.
- Smith JS, Ogden AT, Shafizadeh S, Fessler RG. Clinical outcomes after microendoscopic discectomy for recurrent lumbar disc herniation. J Spinal Disord Tech. 2010;23(1):30–4. doi:10.1097/BSD.0b013e318193c16c.
- Mochida J, Nishimura K, Nomura T, Toh E, Chiba M. The importance of preserving disc structure in surgical approaches to lumbar disc herniation. Spine. 1996;21(13):1556–63; discussion 1563–4.
- 25. McGirt MJ, Ambrossi GLG, Datoo G, et al. Recurrent disc herniation and long-term back pain after primary lumbar discectomy: review of outcomes reported for limited versus aggressive disc removal.

Neurosurgery. 2009;64(2):338–44. doi:10.1227/01. NEU.0000337574.58662.E2; discussion 344–5.

- Ruban D, O'Toole JE. Management of incidental durotomy in minimally invasive spine surgery. Neurosurg Focus. 2011;31(4):E15. doi:10.3171/2011.7.FOCUS11122.
- 27. Wong AP, Shih P, Smith TR, et al. Comparison of symptomatic cerebral spinal fluid leak between patients undergoing minimally invasive versus open lumbar foraminotomy, discectomy, or laminectomy. World Neurosurg. 2013. doi:10.1016/j.wneu.2013.11.012.
- Fontes RB, Tan LA, O'Toole JE. Minimally invasive treatment of spinal dural arteriovenous fistula with the use of intraoperative indocyanine green angiography. Neurosurg Focus. 2013;35 Suppl: Video 5. doi:10.3171/2013.V2.FOCUS13191.
- Papadoulas S, Konstantinou D, Kourea HP, Kritikos N, Haftouras N, Tsolakis JA. Vascular injury complicating lumbar disc surgery. A systematic review. Eur J Vasc Endovasc Surg Off J Eur Soc Vasc Surg. 2002;24(3):189–95.
- Yip S-L, Woo S-B, Kwok T-K, Mak K-H. Nightmare of lumbardiskectomy:aortalaceration.Spine.2011;36(26): E1758–60. doi:10.1097/BRS.0b013e3182194e1c.

# Minimally Invasive Transforaminal Lumbar Interbody Fusion (TLIF): Indications and Techniques

Richard Assaker, Fahed Zairi, and Mohamed Allaoui

### 36.1 Introduction

In the past two decades, the development of spinal instrumentation devices and of minimally invasive approaches and the understanding of spinal biomechanics have lead to dramatically increase the use of spinal fusion procedures [1]. It is estimated that almost 300,000 spinal fusion procedures are performed annually in the United States alone. In the present time, approximately 75 % of cases are performed for spinal degenerative changes especially in the lumbar spine [2]. Of all available techniques, transforaminal lumbar interbody fusion (TLIF) is very popular as it allows the surgeon to achieve a 360° fusion using a single posterior approach. TLIF has gained popularity because of some advantages such as minimal epidural dissection or its efficacy in revision cases [3]. The main disadvantage of this technique is directly related to the morbidity of the approach, which requires a midline incision and a wide soft tissue retraction. Detachment and retraction of the paraspinal muscles are associ-

R. Assaker, MD, PhD (⊠) • F. Zairi, MD M. Allaoui, MD Department of Neurosurgery - Hôpital Roger Salengro, Lille University, Lille, France e-mail: Richard.assaker@chru-lille.fr, r-assaker@chru-lille.fr; mohamed.allaoui@gmail.com; Fahed.ZAIRI@CHRU-LILLE.FR ated with increased blood loss, postoperative pain and muscle necrosis, which may alter the kinematics and stability of the motion segment [4–6]. In order to avoid these drawbacks and especially to preserve the muscle anatomy, minimally invasive TLIF has been introduced [7] and has become an increasingly popular technique in recent years. We describe the indications, the technical aspects and complications of the minimally invasive TLIF.

## 36.2 Indications and Contraindications

Minimally invasive TLIF is generally performed for the same indications than open procedure [8]: low-grade isthmic or degenerative spondylolisthesis, degenerative disk disease and failed back surgery syndrome requiring revision surgery with fusion and instrumentation [9]. However, there are some limitations to the minimally invasive approach, which must be screened by an exhaustive radiological evaluation. Preoperative radiological evaluation, as with an open procedure, includes full-spine standing X-rays, CT scan and MRI. The pelvic parameters should be evaluated on standing X-rays. One should ensure that the pedicles are well identified on lateral and AP views. The foramen should be carefully evaluated on MRI, including its size and the exiting nerve root. The facet complex is better evaluated on the

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_36

CT scan. Limitations of the minimally invasive approach can be classified as follows:

- Number of levels: Most often, minimally invasive TLIF is achieved to treat one level [10]. However, many reports have demonstrated the feasibility of performing TLIF on two adjacent levels using the minimally invasive approach [11]. Indeed, expandable retractors allow exposing two adjacent foramina without altering the paraspinal musculature. Contemporary, available instrumentations do not allow performing more than two levels of fusion.
- High-grade spondylolisthesis: Many tools allow the surgeon to perform reduction manoeuvres, making minimally invasive TLIF an effective treatment option for the management of lowgrade (1 or 2) spondylolisthesis [12]. Currently, a high-grade spondylolisthesis that require significant reduction manoeuvres remains a contraindication to the minimally invasive approach.
- *High pelvic incidence*: It is mandatory to detect preoperatively, on full-spine standing X-rays, patients with high pelvic incidence, which require the restoration of a significant lumbar lordosis. Such a situation is not a real contraindication, but the risk of causing a flat back must be well known. If this approach is still preferred, we recommend to insert the cage as anteriorly as possible and to perform a bilateral arthrectomy in order to enhance the posterior compression and then to increase the lordosis.
- Anatomical considerations: We should ensure preoperatively that the pedicles are well visible on plane X-rays especially on the AP view. In such cases, an open procedure is more suit-

able. However, when available, the new navigation technologies guided by CT scan make it possible to overcome this limitation [13]. Nerve root abnormalities, such as a conjoined root, are also a contraindication for TLIF in general. In our experience, in one case we decided not to insert the interbody fusion device because of a conjoined root that obstructed the route for disk access. As it was not detected preoperatively, we converted to minimally invasive posterior fusion and instrumentation. If needed, the cage should be inserted through an anterior approach.

## 36.3 Surgical Technique

#### 36.3.1 Patient Positioning

The patient should be positioned prone on a radiolucent table, allowing AP and lateral fluoroscopy (Fig. 36.1). Radiolucent rolls are placed on the chest and the iliac crest to ensure the physiological placement of the lumbar spine. Note that excessive lordosis can narrow the foramina and makes the transforaminal disk approach more difficult. The abdomen and the chest must hang free to avoid elevation of the thoracic pressure or vena cava compression. Before draping the patient, AP and lateral fluoroscopy are obtained to ensure proper visualisation of the pedicles. Needles can be placed percutaneously to locate more precisely the pedicles and the disk space. Landmarks are drawn to locate the projection of these structures of interest (midline, pedicles).



**Fig. 36.1** Minimally invasive approach. The skin incisions are marked lateral to the projections of the pedicle entry points (**a**). Opening of the thoracolumbar fascia (**b**) and muscle-splitting approach (**c**)

#### 36.3.2 Screw Placement

Pedicle screws can be inserted in a true percutaneous manner as described by Foley et al. [14] or through an expandable tubular retractor. When a purely percutaneous technique is chosen, it is essential to have a high-quality fluoroscopic guidance, with properly aligned images (Fig. 36.2). The surgeon should place the C-arm in order to obtain a true AP view with no shadow on the superior end plates. The pedicles should be well seen and located at the cranial part of the vertebral body, and the spinous process observed must be strictly in the midline. The lateral view should be obtained with strictly superimposed pedicles and single posterior vertebral body wall. Once the surgeon ensured the reliability of AP and lateral views, a Jamshidi needle can be introduced in the pedicle. Under fluoroscopic guidance, the needle is introduced through a 2 cm skin incision and docked gently on the lateral margin of the pedicle. On AP view, the tip of the needle is at the lateral cortex of the pedicle while it appears at the posterior edge of the pedicle on the lateral view. The Jamshidi needle is then hammered deeper in the pedicle, targeting its centre on the AP view and the posterior wall of the vertebral body on the lateral view. Once the Jamshidi needle is correctly placed, a K-wire is then introduced in the vertebral body through the needle. The needle can be removed and a cannulated tap is used over the wire to prepare the placement of a cannulated screw. Once the screws are inserted on one side, the rod can be placed but not locked, before the placement of the interbody cage.

#### 36.3.3 Placement of the Expendable Retractor

The major principle of the minimally invasive approach is to minimise muscle trauma by



**Fig. 36.2** Optimal placement of the Jamshidi needle on AP (**a**) and lateral (**b**) fluoroscopy. K-wires are introduces through the needles into the vertebral body (**c**, **d**). After

taping (e), the polyaxial screws are introduced over the guide wire  $(\mathbf{f})$ 



**Fig. 36.3** Introduction of the expandable retractor. Introduction of the dilatators  $(\mathbf{a}, \mathbf{b})$ . A mediolateral retractor  $(\mathbf{c})$  allows a better lateral exposure. Photographs  $(\mathbf{d}-\mathbf{f})$ showing the anatomical exposure after progressive

opening of the retractor. The different anatomical structures were identified (\*, facet joint; *1*, disk space; 2, spinous process; *3*, transverse process; *arrows*, pedicle entry points)

lessening the soft tissue dissection and retraction (Fig. 36.3). It is crucial to properly place the retractor to ensure adequate exposure, creating a "working corridor" for safe disk access with minimum muscle injury. On the other hand, this "working corridor" limits the anatomical exposure. There are two technical points to become

familiar with: first, handling and placing properly the retractor and, second, identifying the global anatomy with minimum bone exposure. The side of the transforaminal approach is based on the side of the radiculopathy, in order to achieve a large decompression of the affected nerve root. The site of the skin incision must be identified on



Fig. 36.3 (continued)

AP fluoroscopy. A 3 cm skin incision is done 1-2 cm to the lateral margin of the pedicles. Notice that the incision may be more lateral in overweight patients, to maintain an adequate obliquity. The lumbosacral fascia is opened sagittally, and a blunt dissection is carried on in the plane between the multifidus and the longissimus, as described by Wiltse et al. [15]. The paraspinal muscles are split parallel to their fibres, creating a route to the junction between the transverse processes and the facet joint. The entry points of the cranial and caudal pedicles, overlying the disk of interest, are then identified. Using this plane, serial dilatators are introduced in the direction of the disk and centred on the facet joint. An appropriate length retractor is introduced over the dilatators and secured against the facet joint. Many retractors dedicated to this procedure are currently available. We recommend expandable retractors that can accommodate to the patient's anatomy and expand the exposure when needed. The retractor is secured to the operating table via a flexible arm. A mediolateral retractor can be used in combination to widen the exposure of the segment of interest. This mediolateral retractor maintains the muscle split during the procedure. It

is recommended to loosen this mediolateral retractor every 30 min to release the pressure and to limit the risk of muscle trauma and necrosis. One of the main advantages of the expandable retractor is the direct visual access to the pedicle's entry point. In this setting and according to the surgeon's preference, an operating microscope or operating loupes could be used. We prefer the microscope for the improved visualisation, the quality of the light, the possibility of video transmission and teaching purposes.

#### 36.3.4 The Transforaminal Approach

The dorsal and lateral surfaces of the facet complex are exposed using electrocautery and forceps (Fig. 36.4). Typical exposure includes both pedicle entry points and the transverse processes achieving posterior fusion. A total facetectomy is then performed using high-speed drill or osteotomes. During this step, the resected bone must be saved as bone graft. That is why osteotomes should be preferred rather than the drill. The inferior facet of the above vertebra is completely removed. The lamina can be partially drilled or opened using



**Fig. 36.4** Operative settings. The microscope is used to perform the muscle splitting and to expose the right facet complex using electrocautery (a-c). We performed a wide diskectomy between the dura and the exiting root (d)

Kerrison rongeurs to expose the canal. In case of spinal canal narrowing, it is also possible to perform a wide laminectomy using this approach. The superior facet of the below vertebra is also cut or drilled without breaching the caudal pedicle to not alter the screw anchorage. For better identification, the pedicles should be located and tapped first. Once the foramen is opened, ligaments are carefully removed to expose both exiting and passing nerve roots. The surgeon should be aware of the location of both nerve roots and especially the exiting root. Epidural dissection can be carried on using bipolar cautery and spatulas. At this step, bayoneted instruments are required for optimum visualisation with microscope, in a small working channel. Step by step, the surgeon ensures enough space between the lateral margin of the dural sac and the passing root and the medial margin of the

exiting root. This is mandatory to have a safe and wide exposure of the disk and mobile roots to be manipulated safely. The disk is then incised to perform an ipsilateral diskectomy using bayoneted forceps and curettes.

#### 36.3.5 Cage Placement and Grafting

The interbody space is gradually distracted using sequential distractors until reaching the optimal reduction and suited height (Fig. 36.5). Sometimes, an osteophyte must be resected first using osteotomes to access the disk space. Once the desired distraction is obtained, it is maintained by locking the rod on the other side. The distractor is removed and the diskectomy can be completed. The end plates must be carefully



**Fig. 36.5** Different steps for disk distraction and interbody grafting. A bone chisel can be used to cut an osteophyte for disk access (a). Progressive distraction of the disk space ( $\mathbf{b}$ - $\mathbf{e}$ ). The distraction is maintained by lock-

ing the contralateral construct ( $\mathbf{f}$ ). Preparation of the end plates by using forceps and scrapers ( $\mathbf{g}$ ,  $\mathbf{h}$ ). Insertion of the graft and of a wide PEEK cage as anteriorly as possible ( $\mathbf{i}$ )

prepared using dedicated straight and angled curettes and scrapers. The disk space is filled with the graft material. In our experience, the graft is performed using resected bone expanded with BCP (biphasic calcium phosphate). The fusion rate reported using such graft material is up to 90 %. Currently some teams use rh-BMP-2 (bone morphogenetic protein) and report a fusion rate of 100 % [16]. However, many complications have been reported such as postoperative radiculitis, bone osteolysis and symptomatic ectopic bone formation. As its use is not without risk, the reader should be aware of the reported complications related to the offlabel use of rh-BMP-2 in minimally invasive TLIF [17]. Once the graft is inserted in the disk space, a straight or shaped TLIF cage is introduced after proper sizing. During the cage insertion, care should be taken on the roots especially the exiting root. A nerve root retractor might be needed to protect the root during this step. Finally, the ipsilateral screws and rod are placed under direct vision. Before locking, a posterior contraction is needed to secure the cage and to increase the lordosis. The posterior graft is placed and a drain is suitable on the side of the transforaminal approach.

## 36.4 Outcomes/Advantages of the Minimally Invasive Approach

### 36.4.1 Clinical Results

Numerous studies have demonstrated that the two techniques can achieve similar clinical outcomes. In these studies, the back and leg pain and functional scores were not significantly improved by the minimally invasive approach in comparison with the standard approach [10, 18, 19]. This means that there is no tangible evidence that muscle preservation may lead to improve the functional status of patients. Additional randomised trials with a longer follow-up are needed. However, recent works have demonstrated that the minimally invasive approach reduces postoperative pain and analgesic consumption. The reduced surgical invasiveness is also associated with a shorter hospital stay [18].

#### 36.4.2 Complications

The main advantage of the minimally invasive approach is the significant reduction of the overall complication rate [20]. The minimally invasive approach reduces dramatically the blood loss, making use of blood transfusion very rare. Moreover, the risk of infection is also reduced, which is an important concern especially for high-risk patients [10, 18–20]. Many other complications can be reported during and after such a procedure. Misplacement of pedicle screws is not uncommon at the beginning of the surgeon's experience [21]. Usually this is due to inadequate fluoroscopy or lack of experience. The quality of the C-arm and its manipulation are crucial for the safety of pedicle screw placement in a minimally invasive fashion. Many studies demonstrated that in trained hands the accuracy of screw placement is at least as safe as an open technique. Dural tears may be encountered. It is difficult to close such tears through a tube. The use of collagen, biological glue and fat-tissue flap is recommended for small tears. However, this complication is rarely reported in the published series and also in our experience [10]. The oblique and lateral access, the little needed exposure of the dura and the use of the microscope may be sufficient explanations. However, to limit the risks related to the transforaminal approach, especially with the nerve exposures, it is mandatory to respect a learning curve. We recommend starting the minimally invasive experience by transmuscular tubular approaches for lumbar diskectomies and decompressions. It may be also suitable to have training in anatomical labs.

#### 36.5 Perspective

The main drawback of this technique is the radiation exposure. It is an important concern not only for patients but also for surgeons. The insertion of the pedicle screws and the cage and the manoeuvres of reduction are performed under strict AP and lateral fluoroscopic control. The radiation exposure for patient and surgeon is much more important in minimally invasive approach than in open surgery [22]. Surgeon exposure is significant and requires strict protection and careful monitoring. Annual dose limits for the eyes and the thyroid can be exceeded if a large number of these minimally invasive procedures are performed. Many new protocols and navigation-assisted fluoroscopy systems have been developed in order to overcome this drawback [23, 24]. The CT navigation system is the most recent one [25]. It provides an excellent accuracy and a very low radiation exposure, especially for the surgical team (Fig. 36.6). The recent development of these tech-



**Fig. 36.6** CT navigation system. The frame is fixed to the upper spinous process (**a**). Placement of the K-wires under navigation system without fluoroscopic control (**b**)

nologies is likely to improve the safety of the procedure while decreasing operative time and radiation exposure.

#### Conclusion

Minimally invasive TLIF is a safe and effective technique to achieve circumferential fusion for the treatment of various degenerative diseases at the lumbar spine. The limits of this technique must be well known by surgeons who should comply with a learning curve for patient's best care.

#### References

- Deyo RA, Nachemson A, Mirza SK. Spinal-fusion surgery-the case for restraint. N Engl J Med. 2004;350: 722–6.
- Katz JN. Lumbar spinal fusion: surgical rates, costs, and complications. Spine. 1995;20:78–83.
- Khan IS, Sonig A, Thakur JD, Bollam P, Nanda A. Perioperative complications in patients undergoing open transforaminal lumbar interbody fusion as a revision surgery. J Neurosurg Spine. 2013;18:260–4.
- Gejo R, Matsui H, Kawaguchi Y, et al. Serial changes in trunk muscle performance after posterior lumbar surgery. Spine. 1999;24:1023–8.
- Styf JR, Willen J. The effects of external compression by three different retractors on pressure in the erector spine muscles during and after posterior lumbar spine surgery in humans. Spine. 1998;23:354–8.

- Tsutsumimoto T, Shimogata M, Ohta H, Misawa H. Mini-open versus conventional open posterior lumbar interbody fusion for the treatment of lumbar degenerative spondylolisthesis: comparison of paraspinal muscle damage and slip reduction. Spine. 2009;34:1923–8.
- Foley KT, Holly LT, Schwender JD. Minimally invasive lumbar fusion. Spine. 2003;28:S26–35.
- Salehi SA, Tawk R, Ganju A. Transforaminal lumbar interbody fusion: surgical technique and results in 24 patients. Neurosurgery. 2004;54:368–74.
- Mummaneni PV, Rodts GE. The mini-open transforaminal lumbar interbody fusion. Neurosurgery. 2005;57:256–61.
- Zairi F, Arikat A, Allaoui M, Assaker R. Transforaminal lumbar interbody fusion: comparison between open and mini-open approaches with twoyears follow-up. J Neurol Surg A Cent Eur Neurosurg. 2013;74:131–5.
- Scarone P, Lepeintre JF, Bennis S, Aldea S, Dupuy M, Gaillard S. Two-levels mini-open transforaminal lumbar interbody fusion: technical note. Minim Invasive Neurosurg. 2009;52:275–80.
- Park P, Foley KT. Minimally invasive transforaminal lumbar interbody fusion with reduction of spondylolisthesis: technique and outcomes after a minimum of 2 years' follow-up. Neurosurg Focus. 2008;25:E16.
- Silbermann J, Riese F, Allam Y, Reichert T, Koeppert H, Gutberlet M. Computer tomography assessment of pedicle screw placement in lumbar and sacral spine: comparison between free-hand and O-arm based navigation techniques. Eur Spine J. 2011;20:875–81.
- Foley KT, Gupta SK. Percutaneous pedicle screw fixation of the lumbar spine: preliminary clinical results. J Neurosurg. 2002;97:7–12.

- Wiltse LL, Bateman JG, Hutchinson RH, Nelson WE. The paraspinal sacrospinalis-splitting approach to the spine. J Bone Joint Surg Am. 1968;50:919–26.
- Villavicencio AT, Burneikiene S, Nelson EL, Bulsara KR, Favors M, Tharmann J. Safety of transforaminal lumbar interbody fusion and intervertebral recombinant human bone morphogenetic protein-2. J Neurosurg Spine. 2005;3:436–43.
- Singh K, Nandyala SV, Marquez-Lara A, Cha TD, Khan SN, Fineberg SJ, et al. Clinical sequelae after rhBMP-2 use in minimally invasive transforaminal lumbar interbody fusion. Spine J. 2013;13:1118–25.
- Gu G, Zhang H, Fan G, He S, Cai X, Shen X, Guan X, Zhou X. Comparison of minimally invasive versus open transforaminal lumbar interbody fusion in two-level degenerative lumbar disease. Int Orthop. 2014;38(4): 817–24.
- Seng C, Siddiqui MA, Wong KP, Zhang K, Yeo W, Tan SB. Five-year outcomes of minimally invasive versus open transforaminal lumbar interbody fusion: a matchedpair comparison study. Spine. 2013;38:2049–55.
- 20. Zairi F, Allaoui M, Thines L, Arikat A, Assaker R. Transforaminal lumbar interbody fusion: goals of the

minimal invasive approach. Neurochirurgie. 2013;59: 171–7.

- Nandyala SV, Fineberg SJ, Pelton M, Singh K. Minimally invasive transforaminal lumbar interbody fusion: one surgeon's learning curve. Spine J. 2013. doi:10.1016/J.spinee.2013.08.045.
- 22. Bindal RK, Glaze S, Ognoskie M, Tunner V, Malone R, Ghosh S. Surgeon and patient radiation exposure in minimally invasive transforaminal lumbar interbody fusion. J Neurosurg Spine. 2008;9: 570–3.
- Clark JC, Jasmer G, Marciano FF, Tumialan LM. Minimally invasive transforaminal interbody fusions and fluoroscopy: a low-dose protocol to minimize ionizing radiation. Neurosurg Focus. 2013;35:E8.
- Kim CW, Lee YP, Taylor W, Oygar A, Kim WK. Use of navigation-assisted fluoroscopy to decrease radiation exposure during minimally invasive spine surgery. Spine J. 2008;8:584–90.
- 25. Nottmeier EW. A review of image-guided spinal surgery. J Neurosurg Sci. 2012;56:35–47.

# Minimally Invasive Operation for Lumbar Fusion, Canal Stenosis, Degenerative Scoliosis, and Spondylolisthesis. Is It Possible?

37

## Christopher C. Gillis and Richard G. Fessler

## 37.1 Introduction

Minimally invasive spinal surgery (MISS) is one of the most recent revolutions to occur in the treatment of the spine. Minimally invasive techniques compared to "open" techniques are considered those which use smaller skin incisions, spare the muscle and ligamentous complexes, and often utilize a tubular muscle retractor [1-3]. Using these principles as guidelines, it is definitely possible to perform minimally invasive techniques to achieve lumbar fusion, to decompress lumbar canal stenosis, and to correct or stabilize degenerative scoliosis and spondylolisthesis. The efficacy in comparison to open techniques will be discussed, as well as a brief description of techniques that can be implemented.

The idea of using smaller incisions and less tissue disruption has been present in all branches of surgery and has expanded rapidly in spinal surgery over the past 50 years [4, 5]. The majority of evolution in technique occurred concurrently with rapid expansion in technology allowing for intraoperative spinal imaging, improved microscopic and endoscopic video imaging techniques, and the development of the tubular dilation and retraction system. The majority of MISS techniques use a progressive muscle dilation and tubular retraction technique. From the basic indication of the microendoscopic diskectomy, the indications for MISS have rapidly expanded to a wide variety of procedures and pathologies. MISS has been developed for the cervical, thoracic, and lumbar spine in both anterior and posterior approaches. Over the past decade, the field has been rapidly advancing as more surgeons train and become experts in these techniques to the degree that MISS approachable pathology now includes intradural spinal tumors and even spinal deformity. In this chapter, we focus on the literature and techniques as they pertain to the lumbar spine.

This topic will be subdivided into three general sections discussed in the order of their evolution as minimally invasive techniques: (1) decompression of lumbar canal stenosis, (2) treatment of spondylolisthesis, and (3) lumbar scoliosis correction.

## 37.1.1 Decompression of Lumbar Canal Stenosis

Lumbar stenosis is a recognized and welldescribed etiology of neurologic disease, usually resulting from a combination of hypertrophied facet joints, ligaments, disk herniation, osteophyte overgrowth, and underlying

C.C. Gillis, MD (⊠) • R.G. Fessler, MD, PhD Department of Neurosurgery, Rush University Medical Center, Chicago, IL, USA e-mail: chriscgillis@gmail.com; rfessler@rush.edu

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016 J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_37

spondylolisthesis. This combination of degenerative arthritic changes is termed spondylosis and is prevalent in the relatively mobile cervical and lumbar spines. Decompression for lumbar stenosis is the most common surgery for patients over age 65 [3, 6]. The resultant narrowing of the spinal canal can cause compression on nerve roots or the cauda equina leading to neurogenic claudication with intermittent symptoms, or intense radiating pain into the buttocks or legs. Often patients are trialed on nonsurgical options which can include narcotic, anti-inflammatory and neuropathic pain medications, epidural steroid injections, and physiotherapy. Patients who experience progression of symptoms, severe symptoms, and no resolution of their symptoms with conservative measures or are greatly limited in their activities leading to a decreased quality of life then are considered for surgical management.

The Spine Patient Outcomes Research Trial (SPORT) [7] demonstrated statistically significant improvement in patient outcomes with surgical management and continued long-term improvement of the beneficial effect. The traditional approach for lumbar stenosis has been through an open, decompressive laminectomy with or without facetectomies. The difficulty with open approaches is the inadvertent creation of iatrogenic spinal instability, subsequently leading to additional procedures to stabilize the spine. Radiographic and cadaver studies have illustrated both the decompressive effectiveness of open procedures and the disruption of the native anatomic support structures, including the paraspinal musculature, the supraspinous ligament, the intraspinous ligament, and the facet joints. It is believed that the muscle atrophy from dissection and loss of posterior supporting ligaments lead to long-term instability [3, 4]. Knowledge of these issues led to the development of minimally invasive spine procedures. Through the use of muscle splitting serial tube dilators and retractors, the musculature is left intact and subsequent atrophy minimized. Through a more lateral approach, the midline posterior ligaments are also left intact. Anatomic maintenance and tissue sparing is the main goal of the minimally invasive approach, leading to decreased long-term instability. Traditionally, MISS approaches were associated with a higher rate of cerebrospinal fluid (CSF) leak than the open approaches, and controversy has persisted about the ability to decompress the contralateral side when a unilateral approach has been taken. Both of these points have been disproven as the techniques have improved and long-term literature has been reported. MISS techniques are increasingly used, and lumbar decompression is one of the most frequent indications.

The historical efficacy of open laminectomies is quoted as 64–83 % of patients achieving clinical improvement with durotomy rates as high as 18 % [3, 7]. We can compare this number to the review of Polikandriotis et al. [6] looking at 320 patients treated with an endoscopic lumbar laminotomy and foraminotomy and evaluated out to an average of 18 months postoperatively. The surgical technique used was very similar to that preferred by the senior author, and the surgeries all done on an outpatient basis. The rate of surgical complications was 2.2 %, with all the complications being CSF leaks (c.f. 18 % open laminectomy).

#### 37.1.1.1 Technique Description

The favored minimally invasive technique of the senior author is described as the microendoscopic decompression for stenosis (MEDS). Fluoroscopy is used to identify the surgical level and an 18 mm incision made paramedian to the midline. A stab incision is made with a scalpel, and a Kirschner (k) wire or guide wire is used to further localize the level at the facet region of the vertebral body that has the stenosis. Once the level is correctly identified, serial dilators are placed over the k wire until enough space is created for insertion of the tubular retractor to an 18 mm diameter. The retractor is fixed to the surgical table by a fixed articulating arm (Fig. 37.1). Once the retractor is inserted, some surgeons prefer to use a microscope or loop vision to continue the dissection; however, the senior author prefers to use a specially designed endoscope which fits in the medial side of the retractor (Fig. 37.2). The trajectory to the lamina is lateral, and once a hemilaminectomy is made on the side of entry, the

**Fig. 37.1** Photograph of equipment used for MIS procedures. Shown are a series of dilator tubes, used for the muscular dissection over a K wire, and the 18 mm retractor tube. The retractor tube has a forked attachment arm to secure to the articulating arm affixed to the operating room table





**Fig. 37.2** Photograph of the special endoscope used for MEDs procedures. This endoscope fits within the retractor tubes (As illustrated in Fig. 37.1)

tube can be angled medially, and through a specially designed drill with a retractable singlesided guide, the contralateral lamina can be drilled from underneath the spinous process. The visual angle to the contralateral lateral recess is excellent and can be well decompressed from the unilateral approach (Fig. 37.3).

Rosen et al. [8] described the effective treatment of lumbar spondylosis and stenosis in patients greater than age 75 in a retrospective database review. The results were documented on VAS, ODI, and SF-36 showing great improvement over baseline. Due to disparate measure-



Fig. 37.3 Axial computed tomography scan (CT) demonstrating a left-sided MEDS procedure and the bony decompression achieved

ments, age groupings, and clumping of decompression and fusion patients together, it is difficult to compare this particular patient group outcome to open techniques. The result of this paper is critical to the concept of MISS techniques for lumbar disease due to the correlation of spinal stenosis and degenerative disease with increasing age and with the increasingly aging patient population we are all faced with. It is interesting to note that there were better results for the older patients compared to the younger patients, the reoperation rate was only 2 %, and the hospital stay was significantly shorter than that in other "open" technique series (29 h vs 3.9–11.6 days). Overall the follow-up for this paper was short at mean 7 months, but the prevailing theory is that MISS technique benefit will only increase over time with preservation of anatomic structures leading to decreased instability, adjacent segment disease, and thus delayed reoperation.

Smith et al. [4] evaluated the biomechanical effects of the unilateral MISS lumbar decompression using cadaver specimens in comparison to bilateral decompression and a wide facetectomy. Evaluated was the range of motion in flexion and extension, axial rotation, and lateral bending. The findings of statistically significant increased range of motion in flexion and extension, ipsilateral lateral bending and axial rotation in those treated with a traditional unilateral decompression compared to MISS, point toward the relative destabilization of the open approach. This was further apparent with the wide facetectomy procedure, which caused significant hypermobility in all tested movements, especially in axial rotation.

#### 37.1.1.2 Interspinous Devices

A brief discussion of interspinous devices and decompression of lumbar stenosis is warranted given the popularity of these devices. Gazzeri et al. [9] discussed the controversies related to these devices and their ability to provide relief of degenerative lumbar disease. Although not an entirely minimally invasive procedure, the insertion of an interspinous device follows the principle of minimal anatomical disruption. These devices are used to treat a range of lumbar pathologies including canal stenosis, segmental instability, facet pain syndrome, and discogenic low back pain. Reported benefits to use of an

interspinous device include local anesthetic, preservation of bone and soft tissue, reduced risk of epidural scarring, reduced risk of CSF leak, and even reversibility of the procedure [9]. Looking more closely at the use of the interspinous devices simply for lumbar decompression, the selection criteria are specific to those patients who have relief of their stenotic pain with flexion of the lumbar spine. The mechanism of flexion of the lumbar spine is believed to stretch the redundant ligamentum flavum, which causes posterior compression and stenosis. Once the ligamentum is stretched by the continuous flexion of the interspinous device, there is increased canal diameter and enlargement of the neural foramina, relieving lower extremity symptoms. As reviewed by Gazzeri et al., mean expansion of the spinal canal with insertion of a device is reported between 18 % and 22 % and neural foramina area increased by approximately 25 %. The interspinous devices can be inserted through small, tissue-sparing midline incisions, or even percutaneously. There are a multitude of different devices on the market. The only device with level 1 data is the X-Stop (Medtronic, Tolochenaz, Switzerland; Formerly St. Francis Medical Technologies, Alameda, CA). Looking at published literature the short-term results with the use of the X-Stop device had a significant benefit over nonoperative therapy. Compared to open decompression, the X-Stop showed no significant difference in symptoms or function at 24 months; however, there was a significant increase in reoperation rates for those in the X-Stop group (26 % vs 6 %) [9]. Overall the current evidence does not support a benefit to an interspinous device over surgical decompression, either open or minimally invasive.

A major consideration in the use of MISS is the learning curve encountered when beginning the use of the technique [3, 10-12]. All surgical techniques have a learning curve, and in advanced neurosurgical fields such as complex spine and skull base surgery, there are often a higher incidence of complications encountered during the initial phase of a surgeon's application of a new technology. Depending on the similarity of a technique to one already in use or familiar to the surgeon, it can be understood that some techniques have steeper curves than others. Generally it is believed that familiarity of a surgical technique occurs after approximately 30 cases, while mastery of a technique may continue for some number of cases after. Complications encountered early in the learning process include unintentional durotomy, nerve root injury, inferior facet fracture, wrong level surgery, infections, and new neurologic deficits. Looking at Ikuta et al.'s [12] retrospective review of 114 patients, they found that the incidence of neurologic complications in the first 34 patients was 18 %, whereas it was down to 6.3 % in the last 80 patients.

Minimally invasive spine surgery can be done for lumbar spinal stenosis. The literature confirms that MISS is equivalent or better than open procedures in clinical patient outcome measures [3, 7, 10, 12, 13]. It allows for less blood loss, shorter overall operative time, shorter hospital stay, decreased postoperative narcotic use, decreased rate of infection, decreased incidence of symptomatic CSF leak, and a decrease in the time required for the patient to return to work [3]. As well, as was illustrated in Smith et al.'s biomechanical study [4], MISS has the advantage of less tissue disruption allowing for decreased rates of spinal instability and subsequent requirement for future procedures. One may thus conclude that MISS decompression is a superior technique to a traditional open approach, once the technique has been mastered.

#### 37.1.2 Treatment of Spondylolisthesis

Minimally invasive TLIF is covered extensively in another chapter, and it is a TLIF that has emerged as the current gold standard treatment for spondylolisthesis. We provide a brief summary of the literature outlining the efficacy and feasibility of MISS TLIF (Fig. 37.4).

MISS TLIF is being increasingly performed and has shown reduction in perioperative complications with comparable clinical and radiological outcomes. Compared to an open technique, the benefits of MISS TLIF are decreased postopera-



**Fig. 37.4** Intraoperative photograph of osteotomy being performed through retractor tube for MISS TLIF

tive pain, decreased intraoperative blood loss, shorter postoperative hospital stay, faster return to normal activity, and reduced reoperation rates for adjacent segment disease [6, 14]. In example, Parker et al. [15] demonstrated a significant benefit to MISS TLIF with only 2.6 weeks of postoperative narcotic use compared to 6.5 weeks in open cases and a return to work at 8.3 weeks compared to 16.3 weeks. Lee et al. [16] showed a mean operating time of 166 min for MISS TLIF compared to 181 min with open. The mean blood loss was 50.6 mL compared to 447.4 mL and mean hospital stay 3.2 days compared to 6.8 days.

Quality of life in MISS TLIF was studied prospectively by Perez-Cruet et al. [14] with mean 47 months of follow-up in 304 patients. It was found that MISS TLIF provided statistically significant improvement in clinical outcome for patients with symptomatic spondylolisthesis and degenerative disk disease with or without stenosis. There was a high rate of fusion (>95 % majority achieved by 6.8 months) with the technique and very low rate of interbody device failure (1%) and very low rate of adjacent segment disease requiring reoperation (2%). Looking at quality-of-life scores: visual analog scale (VAS) improved by 35.7% in the immediate short-term and this benefit was maintained long term; the Oswestry Disability Index (ODI) also demonstrated immediate and long-term significant improvement as did the SF-36 physical and mental component scores.

Concern about obtaining bilateral decompression with a unilateral approach was discussed by Kim et al. [17] via magnetic resonance imaging (MRI) analysis. Sagittal T1 MRI was used to calculate the foraminal stenosis at the level of the mid pedicle both quantitatively and qualitatively and central canal stenosis examined both qualitatively and quantitatively on axial T2 MRI. They found significant improvement in all measurements following unilateral TLIF. Contralateral decompression is achieved during the procedure by angling of the retractor medially providing access to the contralateral lateral recess and indirect decompression of the neural foramen by placement of a distracting intervertebral cage. By restoring disk height with an interbody cage, it is believed the disk bulge is reduced and unfolding of the ligamentum flavum occurs which provides central stenosis decompression.

In refinement of technique, the idea of using unilateral versus bilateral percutaneous instrumentation was examined by Dahdaleh et al. [18]. This study was done as a single-center randomized controlled trial with follow-up out to approximately 1 year. Results demonstrated no significant difference between the groups in terms of clinical outcome (measured on validated outcome scores) lordosis correction or in fusion rate. Biomechanical studies have demonstrated superiority of bilateral pedicle screws in both open and MISS constructs, but unilateral fusion may be sufficient for measurement of clinical and radiographic improvement and outcome. Unilateral fusion after TLIF is an option, however, only for single level disease.

The cost benefit of MISS TLIF compared to open was examined by Perez-Cruet et al. and Parker et al. [14, 15]. Parker et al. calculated a 2 year treatment mean cost difference of \$8,731 in favor of MISS TLIF compared to open. Wang et al. [19] showed that hospital stay cost was \$2,106 less per patient with two-level MISS versus an open approach, but no significant difference with one level approaches. Surgical site infections were examined by McGirt et al. [20] and with 6.1 % for open compared to 4.5 % in MISS procedures there was a direct cost difference for investigation and treatment of infections at \$3,593,862 for open and \$1,024,950 for MISS.

## 37.1.2.1 Reduction of Spondylolisthesis

In treatment of spondylolisthesis, reduction is a consideration in those patients with Meyerding grade 1 or greater. The decision of whether to reduce or fuse in situ remains an often discussed issue with benefit of reduction believed to be restoration of spinal anatomy leading to improved sagittal alignment resulting in improved neurologic decompression and fusion rate. Disadvantages of reduction are believed to include neurologic deficits and prolonged operative time. A recent paper by the senior author [21] looked at MISS TLIF in 282 patients, 162 with reduction, and 120 without reduction with at least 1 year follow-up. The authors found found no statistically significant difference in operative time, length of stay, or complication rate. The estimated blood loss and the rate of fusion (84.5 % vs 70.8 % p < 0.05) was significantly higher in those patients with reduction. Reduction is achievable through an MIS approach, led to a higher fusion rate in the study, and was not associated with a higher rate of neurologic complications. If reduction cannot be achieved, fusion in situ is an acceptable alternative.

Thus, MISS TLIF is an accepted and useful technique and can be employed for reduction of spondylolisthesis. Outcome measures are least equivalent for open and MISS procedures. There are lower rates of intraoperative complications, a more rapid return to baseline patient functioning and recent studies show a possible benefit in terms of cost benefit analysis.

#### 37.1.3 Lumbar Scoliosis Correction

Adult degenerative scoliosis is an increasingly identified condition presenting to spine surgeons for consideration, usually along with lumbar spondylosis and varying degrees of radiculopathy or spinal stenosis. Given the aging population in developed nations, the incidence is only expected to increase in the near future. The correction of lumbar scoliosis is traditionally performed with open surgical approaches involving significant rates of complication and blood loss.

The latest in MISS techniques covers those surgeries that are often the maximally invasive of open procedures: the scoliosis deformity correction. MISS deformity correction has arisen over the past decade and is a field that is continuously evolving. With these developing procedures, it is possible to correct a variety of coronal and sagittal lumbar deformity with percutaneous screws and hybrid MISS techniques for osteotomies and TLIF at appropriate levels. Early versions of the technique, despite being internally tissue sparing of the fascia and muscle, would often leave multiple small incisions which could be cosmetically displeasing. This has evolved to a midline incision for cosmetic reasons with suprafascial dissection and a far lateral muscle splitting (similar to a Wiltse approach) approach to the facet joints which allows for multi-level osteotomies and interbody cages. This is augmented with percutaneous screws placed in a muscle splitting fashion and through lateral trans-psoas approaches (DLIF, XLIF, LLIF etc.) which allow for direct coronal plane correction through manipulation of the anterior column. Some surgeons have also advocated for release of the anterior ligamentous structures which allow for more extensive manipulation of the anterior column. We will briefly discuss these techniques and the evidence to support their use.

The main goal of lumbar scoliosis correction is pain relief with the secondary goals of neural decompression, creating or maintaining coronal and sagittal balance, and stabilization of the construct. Deformity correction progresses from the discussion of MISS TLIF. The basic building block of lumbar deformity correction and fusion is the complete removal and fusion reconstruction of a single motion segment; this is achieved with MISS TLIF or as we will discuss, lateral lumbar interbody fusion (LLIF). This involves a single level central and foraminal decompression of the local neural elements, fusion, and placement of a cage after disk removal and depending on instrumentation used can stabilize or improve deformity present at that level. It stands to reason, then, that expanding upon the basis of the minimally invasive (MISS) TLIF, we can achieve multi-level decompression, fusion, and correction of a variety of lumbar spine disease.

Anand et al. [22, 23] described their extensive long-term experience with MISS correction of scoliosis in 187 patients. This paper was in both thoracic and lumbar deformities but has the longest published follow-up experience and serves as our starting point. They found that the technique was possible and useful but limited in the ability to achieve normal sagittal vertebral angle (SVA; SVA less than 47 mm optimal) in patients who had a preoperative value less than 100 mm, and limited in the ability to correct pelvic incidence to lumbar lordosis mismatch (PI-LL  $\pm 10^{\circ}$ ). The maximum SVA correction achieved was 89 mm and the PI-LL could only be normalized in those patients with starting values of 38° or less. The average curve reduction achieved was 61 %. They also noted five of their patients developed L5-S1 pseudoarthrosis. The conclusion by Anand et al. was that their circumferential technique was effective on patients who had preoperative deformity measurements within the confines of the limitations of the technique, based on the values given above. In moving forward from this, it is important to note the techniques included used direct lateral trans-psoas approaches for interbody fusions and MISS percutaneous pedicle screw and rod placement. In those patients requiring L5-S1 fixation, transsacral axial lumbar interbody fusion was used. The limitations in the lateral trans-psoas approaches are likely the limitation in sagittal correction found in this study, similar to Acosta et al. [24], who found that lateral trans-psoas approaches did not improve lumbar lordosis.

As mentioned, some surgeons advocate for release anterior longitudinal ligament (ALL) when performing lateral trans-psoas interbody techniques. In deformity correction, it is often necessary to release anteriorly to achieve greater deformity correction, and it can be conferred that without this release, a limitation in the ability to achieve maximum correction is reached. In MISS deformity correction, a direct anterior approach is replaced by the lateral trans-psoas approach, and some surgeons believe that depending on the degree of correction required, the release of the anterior longitudinal ligament may be needed to achieve this. However, the release of the ALL is associated with significant intraoperative complications, specifically injury to the great vessels and/or anterior dislodgement of the cage after placement, as a result this can be considered a higher-risk maneuver. Due to the risks associated with ALL release, it is likely best avoided for those unfamiliar with its use. Another newer technique allowing for greater deformity correction from the lateral approach is the approval of increasingly hyperlordotic interbody cages. Using these devices, which have far greater angulation than the usual TLIF or PLIF interbody device, increasing severity of sagittal deformity is able to be corrected through lateral approaches alone. In some cases, the lateral approach alone

may be all that is required to achieve the correction. The senior author prefers to perform the lateral interbody procedure and then obtain upright postoperative imaging before progressing to a posterior approach. Overall, lateral interbody techniques continue to evolve and add to the toolbox of the MISS deformity surgeon. As technology continues to progress, so will the level of achievable deformity correction.

In Dahdaleh et al.'s review of the literature [11], the lateral trans-psoas approaches and their role in deformity correction are examined in detail. The technique involves a small incision with a lateral window created through the psoas muscle (Fig. 37.5). A retraction device is mounted to the operating room table, and cages of sizes up to the anterior-posterior diameter of the vertebral body can be placed for lumbar interbody fusion. Coronal deformity can be corrected from 3.0 to 5.9° and sagittal correction from 2.2 to 3.3° per level [11]. Lateral interbody fusion also allows for indirect decompression of the foramina and central canal [11], and the lateral approach allows for a superior result compared to TLIF for lordosis. Complications of the lateral trans-psoas approach include injury to the lumbar plexus in its path through the psoas muscle. This can present as groin and anterolateral thigh sensory symptoms and muscle weakness in ipsilateral



**Fig. 37.5** Patient positioning for LLIF procedure. Patient is positioned lateral, on a bean bag support with the break in the operating room table just below the iliac crest, where it is used to maximize the space between the iliac crest and ribs to provide maximum lateral exposure



**Fig. 37.6** Intraoperative fluoroscopic image illustrating the insertion of the lateral interbody cage through the LLIF approach. There are pins in the vertebral body above and below to stabilize the tube retractor over the lateral disk space. The subjacent level has already had an interbody cage placed

psoas or quadriceps. The published rates of numbress vary from 18 to 40 % postoperatively and the rates of weakness range from 25 to 55 %; the majority of these symptoms are resolved by 6 month to 1 year follow-up (Fig. 37.6).

In patients with appropriate sagittal balance and pure coronal scoliosis symptoms, a lateral interbody fusion technique with percutaneous screws alone can be used. When more extensive sagittal correction is also required, depending on the degree of deformity, as discussed previously, more extensive posterior work is required than percutaneous screws alone. This more extensive posterior work can be performed through essentially a MISS TLIF approach, allowing for access to the lamina, the disk and posterior ligamentous elements, and the facet and foraminal zone. This allows for facet osteotomies as well as Smith-Peterson facet osteotomies for achieving greater degrees of sagittal deformity correction.

To illustrate what we have surmised from the literature, Dangelmajer et al. [25] performed a meta-analysis comparing differences in patient selection and outcomes between MISS lumbar scoliosis correction and open surgery. They included 12 studies for analysis, of which 8 used lateral interbody fusion or XLIF technique and 4 studies did decompression without fusion. The analysis showed that the patients selected for MISS were older with more medical comorbidities than those offered open surgery. Those patients with more severe deformity as measured by greater Cobb angles and greater sagittal imbalance had open surgery, with a higher rate of reoperation. Overall there were no significant differences between complications between the two groups, but given the differences in patient selection, this cannot be taken as a direct comparison between MISS and open deformity surgery. Given what we have discussed, these results are not unexpected; MISS deformity correction has a limitation and open surgery is favored in the case of severe deformity, as well as those patients who are likely unable to tolerate a large open surgery (elderly, medical comorbidities) will better tolerate MISS procedures and with acceptable complication rates.

Currently, MISS techniques are not as versatile as open deformity techniques, and some reports detail suboptimal correction or pseudoarthrosis with sagittal plane deformity, despite the ability to achieve excellent coronal correction. To examine the current consensus of what can be corrected by MISS, Mummaneni et al. [26] created a decision-making algorithm for MISS deformity correction. They tested the reliability of their algorithm by presenting 20 representative cases to 11 fellowship trained spine surgeons at two separate time points. There was very good intraobserver reliability and moderate good interobserver (intraobserver kappa 0.86, interobserver 0.58 and 0.69) reliability with their survey. Overall the agreement was echoed almost identical to Anand et al. [23] in that open approaches were required for the most severe deformities, with the values given as SVA > 7 cm, with a rigid curve; LL-PI mismatch greater than 30° with a pelvic tilt greater than 25°; and/or thoracic hyperkyphosis greater than 60°. Deformity with values less than this can be considered for MISS procedures ranging from one- to two-level TLIF with minimal deformity and for those with moderate deformity more extensive MISS including lateral interbody fusion and muscle sparing minimally invasive or mini open posterior procedures combined with percutaneous pedicle screws.

MISS scoliosis correction is a newly developed technique, which currently has technical limitations in the degree of deformity that can be corrected. As techniques, especially the lateral interbody fusion, continue to expand and develop, the limit of deformity that can be corrected slowly diminishes. Within these limitations, MISS deformity correction is an accepted and possible surgical option. Both MISS techniques alone or in a hybrid procedure with osteotomies are an emerging tool in the field of spinal deformity correction.

#### References

- Fessler RG, O'Toole JE, Eichholz KM, Perez- Cruet MJ. The development of minimally invasive spine surgery. Neurosurg Clin N Am. 2006;17(4):401–9.
- Singh K, Nandyala SV, Marquez-Lara A, Fineberg SJ, Oglesby M, Pelton MA, et al. A perioperative cost analysis comparing single-level minimally invasive and open transforaminal lumbar interbody fusion. Spine J. 2013;14(8):1694–701.
- Wong AP, Smith ZA, Lall RR, Bresnahan LE, Fessler RG. The microendoscopic decompression of lumbar stenosis: a review of the current literature and clinical results. Minim Invasive Surg. 2012;2012:325095.
- Smith ZA, Vastardis GA, Carandang G, Havey RM, Hannon S, Dahdaleh N, et al. Biomechanical effects of a unilateral approach to minimally invasive lumbar decompression. PLoS One. 2014;9(3):e92611.
- Snyder LA, O'Toole J, Eichholz KM, Perez-Cruet MJ, Fessler R. The technological development of minimally invasive spine surgery. Biomed Res Int. 2014;2014:293582.
- Polikandriotis JA, Hudak EM, Perry MW. Minimally invasive surgery through endoscopic laminotomy and foraminotomy for the treatment of lumbar spinal stenosis. J Orthop. 2013;10(1):13–6.
- Weinstein JN, Tosteson TD, Lurie JD, et al. Surgical versus nonoperative treatment for lumbar spinal stenosis four year results of the spine patient outcomes research trial. Spine. 2010;35(14):1329–38.
- Rosen DS, O'Toole JE, Eichholz KM, Hrubes M, Huo D, Sandhu FA, et al. Minimally invasive lumbar spinal decompression in the elderly: outcomes of 50 patients aged 75 years and older. Neurosurgery. 2007; 60(3):503–9; discussion 509–10.
- Gazzeri R, Galarza M, Alfieri A. Controversies about interspinous process devices in the treatment of degenerative lumbar spine diseases: past, present, and future. Biomed Res Int. 2014;2014:975052.

- Ang C-L, Phak-Boon Tow B, Fook S, Guo C-M, Chen JL-T, Yue W-M, et al. Minimally invasive compared with open lumbar laminotomy: no functional benefits at 6 or 24 months after surgery. Spine J. 2013. doi: http://dx.doi.org/10.1016/j.spinee.2013.07.461.
- Dahdaleh NS, Smith ZA, Snyder LA, Graham RB, Fessler RG, Koski TR. Lateral transpoas lumbar interbody fusion: outcomes and deformity correction. Neurosurg Clin N Am. 2014;25(2):353–60.
- Ikuta K, Arima J, Tanaka T, et al. Short-term results of microendoscopic posterior decompression for lumbar spinal stenosis. Technical note. J Neurosurg Spine. 2005;2(5):624–33.
- Mobbs RJ, Li J, Sivabalan P, Raley D, Rao PJ. Outcomes after decompressive laminectomy for lumbar spinal stenosis: comparison between minimally invasive unilateral laminectomy for bilateral decompression and open laminectomy. J Neurosurg Spine. 2014;30:1–8.
- Perez-Cruet MJ, Hussain NS, White GZ, Begun EM, Collins RA, Fahim DK, et al. Quality-of-life outcomes with minimally invasive transforaminal lumbar interbody fusion based on long-term analysis of 304 consecutive patients. Spine (Phila Pa 1976). 2014; 39(3):E191–8.
- Parker SL, Mendenhall SK, Shau DN, Zuckerman SL, Godil SS, Cheng JS, et al. Minimally invasive versus open transforaminal lumbar interbody fusion for degenerative spondylolisthesis: comparative effectiveness and cost-utility analysis. World Neurosurg. 2014;82(1–2):230–8.
- Lee K, Yue W, Yeo W. Clinical and radiological outcomes of open versus minimally invasive transforaminal lumbar interbody fusion. Eur Spine J. 2012;21: 2265–70.
- 17. Kim MC, Park JU, Kim WC, Lee HS, Chung HT, Kim MW, et al. Can unilateral-approach minimally invasive transforaminal lumbar interbody fusion attain indirect contralateral decompression? A preliminary report of 66 MRI analysis. Eur Spine J. 2014;23(5):1–6.
- Dahdaleh NS, Nixon AT, Lawton CD, Wong AP, Smith ZA, Fessler RG. Outcome following unilateral versus bilateral instrumentation in patients undergoing minimally invasive transforaminal lumbar interbody fusion: a single-center randomized prospective study. Neurosurg Focus. 2013;35(2):E13.
- Wang M, Lerner J, Lesko J, et al. Acute hospital costs after minimally invasive versus open lumbar interbody fusion: data from a us national database with 6106 patients. J Spinal Disord Tech. 2012;17:324–8.
- McGirt M, Parker S, Lerner J, et al. Comparative analysis of perioperative surgical site infection after minimally invasive versus open posterior/transforaminal lumbar interbody fusion: analysis of hospital billing and discharge data from 5170 patients. J Neurosurg Spine. 2011;14:771–8.

- 21. Scheer JK, Auffinger B, Wong RH, Lam SK, Lawton CD, Nixon AT, Dahdaleh NS, Smith ZA, Fessler RG. Minimally invasive transforaminal lumbar interbody fusion (TLIF) for spondylolisthesis in 282 patients: in situ arthrodesis versus reduction. World Neurosurg. 2015;84(1):108–13.
- Anand N, Baron EM, Kahwaty S. Evidence basis/outcomes in minimally invasive spinal scoliosis surgery. Neurosurg Clin N Am. 2014;25(2):361–75.
- 23. Anand N, Baron EM, Khandehroo B. Limitations and ceiling effects with circumferential minimally invasive correction techniques for adult scoliosis: analysis of radiological outcomes over a 7-year experience. Neurosurg Focus. 2014;36(5):E14.
- 24. Acosta FL, Liu J, Slimack N, Moller D, Fessler R, Koski T. Changes in coronal and sagittal plane align-

ment following minimally invasive direct lateral interbody fusion for the treatment of degenerative lumbar disease in adults: a radiographic study. J Neurosurg Spine. 2011;15(1):92–6.

- Dangelmajer S, Zadnik PL, Rodriguez ST, Gokaslan ZL, Sciubba DM. Minimally invasive spine surgery for adult degenerative lumbar scoliosis. Neurosurg Focus. 2014;36(5):E7.
- 26. Mummaneni PV, Shaffrey CI, Lenke LG, Park P, Minimally Invasive Surgery Section of the International Spine Study Group, et al. The minimally invasive spinal deformity surgery algorithm: a reproducible rational framework for decision making in minimally invasive spinal deformity surgery. Neurosurg Focus. 2014;36(5):E6.

## Spine Injections for Persistent Lumbar and Radicular Pain After Lumbar Spine Surgery

38

## Lee R. Wolfer, Richard Derby, and Jeong-Eun Lee

Patients with persistent lumbar pain and/or radicular leg pain after lumbar surgery pose significant diagnostic and therapeutic challenges. Patients who do not improve after lumbar surgery are commonly labeled with a diagnosis of "failed back surgery syndrome (FBSS)." The term "FBSS" was coined before by spine surgeons prior to 1970 when reporting surgical outcomes [1, 2]. Some nonsurgical specialists have served only to obfuscate the etiology of persistent lumbar and/or leg pain after lumbar surgery by stating that FBSS is "easy to recognize but hard to define" [3]. FBSS is "a euphemistic term used for a heterogeneous group of disorders in patients complaining of back and leg pain" [4]. Burton et al. [5], lacking MRIs and diagnostic/therapeutic spine injections, incorrectly attributed persistent pain after spine

R. Derby, MD, FIPP, ABIPP

Spinal Diagnostics and Treatment Center, 901 Campus Drive, Suite 310, Daly City, CA 94015, USA e-mail: leewolfermd@gmail.com; rderby@spinaldiagnostics.com

J.-E. Lee, PT

Spinal Diagnostics and Treatment Center, Graduate School of Medicine, Korea University, Anam-Dong 5-Ka, Seongbuk-Gu, Seoul 136-701, South Korea

Department of Anesthesiology, Spinal Diagnostics and Treatment Center, 901 Campus Drive, Suite 310, Daly City, CA 94015, USA e-mail: grin2649@gmail.com

surgery to prima facie psychological comorbidities and secondary gain. Based on his collective experience of treating over 800 FBSS cases per year, Burton et al. [5] stated: "Patients with this condition can rarely be 'cured' because FBSS is actually a spectrum of organic disease processes complicated by secondary financial gain and learned chronic pain behavior." Unfortunately the label implies (is often equated with) both psychological pain exaggeration and a centralized self-sustaining pain process in which further diagnostic tests are inappropriate and treatment options directed at a specific structural cause are useless. Patients with persistent pain after surgery do not have a primary psychological problem, responsible for their pain; however, psychological comorbidities are present in this population, some premorbid, and others likely reactive to the life circumstances of persistent pain. With increasing number of physicians and societies devoted to pain management and musculoskeletal spine and sport medicine, the diagnosis and nonoperative/noninterventional treatment algorithms used in these practices have evolved to include conservative care modalities such as aggressive pharmacologic pain management, functional restoration, behavioral modification, and psychological counseling; however, in a subset of patients, these treatments may at best be adjunct treatments or may in fact miss the opportunity to diagnose and treat pain generators definitively.

L.R. Wolfer, MD, MS (🖂)

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_38

The FBSS of 30 years ago was nonspecific and non-diagnostic (Slipman et al. [6]); however, we can now state that "failed back surgery syndrome" (FBSS) is an outmoded, inaccurate, and pejorative term. A current, more accurate definition of FBSS is: "persistent or recurrent pain, mainly in the region of the lower back and legs, even after technically, anatomically successful lumbosacral spine surgeries" [7]. The primary label used by the authors of this chapter is "postlumbar surgery syndrome" (PLSS) [8]. PLSS is defined as a disease entity representing "a cluster of nomenclature and syndromes following spine surgery wherein the expectations of the patient and spine surgeon are not met, with persistent pain following lumbar surgery" [9]. With modern imaging innovations, as well as refined diagnostic and therapeutic spine interventions including spine injections and re-operation, more than 90 % of patients can be given a specific diagnosis and treatment for their persistent low back and/or leg pain after lumbar surgery.

The postsurgical patient with persistent lumbar pain and/or leg pain poses diagnostic and therapeutic challenges unlike those of the patient with chronic low back pain. The most common causes of persistent pain after lumbar spine surgery currently are: recurrent/residual or new disk herniation, painful internal disk disruption, sacroiliac pain, facet joint pain, instability, fibrosis (epidural, perineural, or intraneural), neuropathic pain, acquired stenosis, and adjacent segment disease. Rare patients may have a surgical complication or technical failure.

The goal of this chapter is to inform the reader about the most common spine injections currently used to diagnose and treat PLSS. Due to the influence of payors and policymakers on surgical decision-making, PLSS will also be briefly presented within an epidemiologic, historic, and socioeconomic context. Payors, policymakers, researchers, and other stakeholders have sounded an "alarm" over the increase in spine surgeries. They have sounded the same alarm over the proliferation of interventional procedures in nonoperative spine practices. Detractors argue that some of these procedures are dubious, risky, and scientifically unproven and do not lead to long functional improvement. Some interventions have stood the test of time, e.g., epidural injections, diagnostic facet and sacroiliac joint blocks, and medial branch neurotomy, but other treatments such as pulsed radiofrequency treatments of spinal nerves or various intradiscal therapies are recent interventions and may or may not be in common use in the future. With the meteoric rise in spine injections over the last decade, often without appropriate indications, spine interventionalists have come under intense scrutiny to justify the medical necessity of their procedures. Ultimately, both patient and the physician are the losers with performance of medically unnecessary spine injections. Because spine interventionalists did not utilize rigorous indications for successful diagnostic and therapeutic interventions in the last decade, payors and policymakers intervened.

Interventional procedures are directed at specific peripheral sources of pain originating within the posterior, middle, or anterior columns, and for the most part, their use requires the identification of an ongoing nociceptive/neuropathic source via diagnostic lumbar spine injections. The most important task of the surgeon evaluating a patient with persistent lumbar pain and/or radicular pain after surgery is to provide a diagnosis based on history, physical exam, imaging, and diagnostic/therapeutic spine injections. The role of diagnostic and therapeutic spine injections is to assist the surgeon in precision identification of pain generators to optimize surgical outcomes. Once a diagnosis is obtained, the spine surgeon can consider the best treatment options, as presented in this chapter, based on the most up-to-date, evidence-based review of the spine literature on PLSS currently available [10, 11].

As with any surgery, spine surgery has risks and may not completely eradicate pain and functional impairment. Ideally, patients undergoing primary or revision surgery should have exhausted noninterventional, conservative care modalities, as well as interventional spine care treatments. For example, in patients with primary chronic axial low back pain, spine injections performed per the International Spine Intervention Society criteria for facet pain, sacroiliac pain, and provocative discography should be considered before lumbar fusion if the patient meets appropriate criteria [10, 11]. If spine interventionalists drop the bar by adopting less stringent diagnostic criteria for diagnostic blocks, outcomes suffer and payors and policymakers will continue to decrease reimbursements. If the diagnostic bar is set high for facet blocks, SIJ blocks, epidural steroid injections, and so forth, both patients and physicians will benefit. If the bar is set high with more stringent definitions of a successful lumbar surgery, as well as more rigid surgical indications, the incidence of PLSS should also decrease.

The esteemed spine surgeons Burton et al. [5] presciently stated the following – which still rings true today to guide spine specialists to the best possible outcomes:

The answer to the problem is not better FBSS salvage, but avoiding the causes of failure of lumbar spine surgery. It relates to the prevention of induced iatrogenic disease; it also relates to adequate diagnosis and treatment based on that diagnosis. Moreover, it relates to preventive care and early, aggressive, comprehensive, conservative management of low back pain. Conservative treatment is often inadequate and employed for insufficient periods of time. Finneson [12, 13] pointed out in a survey of surgical patients that '80 % should not, in our opinion, have been brought to surgery' ... When surgery is indicated, adequate diagnostic tests and the execution of appropriate procedures based upon this information should largely prevent the failed back surgery syndrome.

### 38.1 Epidemiology of Chronic Low Back Pain

To understand the etiology of persistent lumbar and/or radicular pain after lumbar surgery, it is critical to briefly review the epidemiology of low back pain. According to the Global Burden of Disease study from 2010: "*Low back pain causes more global disability than any other condition.*" Disability weights were applied to overall prevalence to obtain overall disability expressed in years lived with a disability (YLD). The years lived with a disability (YLD) increased from 58 million in 1990 to 81 million in 2010 [14]. All agree that these patients represent a tremendous social and financial burden. In the United States, it was estimated that direct healthcare expenditure for back pain was \$90.7 billion in 1998 [15]. A significant percentage of chronic low back pain patients will fail conservative care and seek out diagnostic and therapeutic spine injections. A small percentage of chronic low back patients will fail both conservative care and interventional spine treatments and undergo surgery.

Among the US adult population, the prevalence of chronic pain ranges from 2 % to 40 %, with a median point prevalence of 15 %. Chronic spinal pain is the most common source of chronic pain [16]. The lifetime prevalence of spinal pain has been reported as 54–80 % [16]. Some studies have reported an increased prevalence of "chronic, impairing low back pain" [17]. A telephone survey of greater than 4,000 North Carolina households showed a 3.9 % prevalence in 1992 and a 10.2 % prevalence in 2006.

Other researchers have looked at healthcare utilization as a surrogate marker for patients with chronic spine problems. Martin et al. [18] studied trends in healthcare expenditures, utilization, and health status among US adults with spine problems from 1997 to 2006. There was an increase in the estimated number (and treated prevalence) of people reporting care for spine problems in the United States from 14.8 million (10.8 % of the population) in 1997 to 21.9 million (13.5 %) in 2006. Per-user expenditures for spine-related inpatient, outpatient, pharmacy and emergency services were followed. From 1997 to 2006, a 49 % increase in the number of patients seeking spine-related care was the largest contributor to outpatient expenses. Among people with spine problems, a 660 % increase in expenditures for opioid medications was reported, from \$246 million in 1997 to \$1.9 billion in 2006 [18].

## 38.2 "Nonspecific Low Back Pain": A Closer Look

How a physician defines low back pain determines how they will treat a patient with low back pain. A physician, payor, or policymaker who has learned that 80 % of low back pain is "nonspecific" or "idiopathic" will be a minimalist and be unlikely to endorse spine injections or surgery. A majority of healthcare professionals, insurance industry, and policymakers have been taught that the majority of low back pain has no clear cause and no compelling reason. Patients suffering from chronic low back pain also continue to be told their pain is psychogenic or due to secondary gain motives. In contradistinction, a spine surgeon or spine interventionalist, who has identified greater than 90 % of the causes of low back pain, will approach the treatment of chronic, disabling low back pain much differently.

A brief look origin of the statement that "80 % of low back pain is nonspecific" is not simply an academic epidemiologic tangent. This statement of "fact" has been used to frame and guide spine care for almost 50 years. This statistic continues to be used as "evidence" for limiting spine surgery, limiting spine injections, and perhaps causing unnecessary pain and suffering in patients with low back pain. Primary care physicians and many neurologists have learned this statistic as fact and patients are principally given only conservative care options. This statistic from one isolated study has also contributed to the creation of a climate of fear of back surgery, quite unlike attitudes toward other orthopedic surgeries.

In 1982, prominent spine surgeon, Augustus A. White quoted a study in which the authors found that in 79 % of first attacks of low back pain in men (and 89 % in women), the "specific cause was unknown" [19, 20]. From this time on, it became generally accepted that 80 % of low back pain was idiopathic or nonspecific. This statistic came to be broadly misapplied to all patients with low back pain, acute and chronic. Taking a closer look, the original study published in the 1960s included patients with less than 2 weeks of low back pain; diagnostic tools included only history and physical exam [19]. The conclusions of this 60-year-old observational study continue to be inappropriately applied to the diagnosis and treatment of patients with prima facie chronic low back pain and post-lumbar surgery persistent pain. The evidence for treating chronic low back pain can almost always be divided into two polarized camps, each camp continuing to provide evidence for its position. Much of our current division has its historic roots in the acceptance of the view that the etiology of low back pain cannot be determined in 80 % of patients.

Beginning in the late 1970s, the use of fluoroscopic imaging and contrast during injection procedures allowed for the development of more precise localized injection techniques [21] in which pain reproduction and local anesthetic pain relief could be used to help confirm that the target structure was a source of pain. These techniques have evolved and culminated in the International Spine Intervention Society (ISIS) Practice Guidelines for injection standards for diagnosing and treatment of chronic spinal pain [10, 11]. Based on Schwarzer's studies [22–27] using precision diagnostic injection as the criterion standard for diagnosis, Bogduk [28] postulated that a definite diagnosis could be made in 70-80 % of patients with chronic lumbar pain. Schwarzer et al. [23, 27] identified the z-joint (9%), SI joint (15%), or disk (39%) as the primary sources of pain. Although arriving at a different percentage prevalence of specific pain sources, Manchikanti et al. [29] using the same techniques in 120 consecutive patients with spinal pain was able to identify the z-joint (40 %), SI joint (2 %), or disk (26 %) as the primary source of pain in 68 % of patients. High-quality, randomized controlled trial studies, particularly in last 20 years, using diagnostic spine injections (facet, sacroiliac blocks, and provocative discography) have definitively disproven the general statement that 80 % of low back pain is nonspecific or idiopathic. These same diagnostic techniques have been refined and utilized to determine specific etiologies of persistent lumbar and/or leg pain after surgery. Advances in imaging have also contributed significantly to our understanding of low back pain. In the 1950s, the only imaging tool was x-ray. MRI, especially in the post-lumbar surgery spine, has proven to be an indispensable tool for diagnosis.

The "literature legends" of "nonspecific" and "idiopathic" low back pain have been passed from publication to publication for over 50 years without critical review of the original study. The study from which these terms are derived applies narrowly to patients presenting with acute (<2 weeks) low back pain in the 1950s, not to patients with chronic, disabling low back pain presenting in the twenty-first century. In 1966 Dillane et al. published a study entitled, "Acute low syndrome: A study from general practice" [19]. The incidence of *acute low back pain epi*sodes lasting less than 2 weeks was collected in a general practice setting from patients seen between 1957 and 1960. X-rays were not performed because of their reported unreliability for diagnosing disk herniations [30]. According to Troup [30], x-ray evidence of a "disk" problem is known to be unreliable. At this time discogenic pain and painful internal disk disruption were not well understood as a cause of low back pain; only herniated disks causing radiculopathy were recognized. Dillane et al. [19] reported that the cause of low back pain was not evident in 79 % of males and 89 % of females (see Table 38.1). The report of 79 % of males with a "cause not evident" was rounded up to 80 %, thus giving rise to the claim that 80 % of low back pain is idiopathic or nonspecific. Female patients were even more mysterious, with 90 % of women having a "cause not evident." The reader can see how this data could be misused with many physicians attributing low back pain to an organic psychogenic problem or the attitude that the patient will recover if given enough time. Dillane et al. [19] reported that a majority of patients (62 %) recovered in 2 weeks. (Of note, no follow-up was undertaken on the 38 % of patients who did not recover in 2 weeks.) Forty-five percent (45 %) of patients experienced a reoccurrence within 4 years.

 Table 38.1
 Cause of index attack [19]

Cause of index attack	Males (%)	Females (%)
Cause not evident	79.3	88.9
Strain	10.9	4.3
Proved disk <sup>a</sup>	7.6	5.6
Ankylosing spondylitis and neoplasms	0.5	0.6
Direct trauma	0.5	-
Others	1.1	0.6

<sup>a</sup>A diagnosis of prolapsed disk was accepted only if there was objective evidence of nerve root pressure – paralysis or wasting of muscle of an area of anesthesia

The population reported on by Dillane et al. [19] is not the population routinely seen by spine surgeons; however, the notion of nonspecific low back pain has persisted and remains a common retort for those against spine interventions for both chronic low back pain patient and patient with persistent low back or leg pain patient after surgery. Many physicians persist in offering only pharmacologic management, cognitive behavioral therapy, functional restoration, and/or physical therapy under the belief that the spinal pain is primarily psychogenic in origin or that the pain has become centralized, and therefore the search for a pain generator is futile and unsupported by evidence. Interesting, as briefly noted above, an important historic study by Rowe [31] was published in 1969 shortly after the Dillane study, but has been ignored from an epidemiologic perspective. Rowe (1969) studied 500 employees with chronic low back pain for 7 years at the Kodak factory [31]. Rowe states the reasons for his 7-year observational study best: "Studies based upon short-term observation of patients with low back disability have yielded information nearly as tunnel-sighted as that gleaned from the single etiology approach. Experience indicates that backache is characteristically intermittent, episodic, and recurrent and can meaningfully be studied only as a continuum which stretches through the active years in a man's life." Rowe [31] reported a diagnosis in approximately 80 % of his patients, including disk pain (54 % both intrinsic discogenic pain and pain due to disk herniation), inflammatory backache (14 %), and miscellaneous causes (10 %). Rowe did not have use of CT scans, MRI, or diagnostic spine injections to identify facet or sacroiliac joint pain. The use of provocative discography is not reported. Furthermore, in the late 1960s, discography had fallen out of favor based on the negative discography study by Holt [32]. Holt's study was later discredited as he utilized a nonstandard technique on prisoners receiving incentives for study participation.

In summary, the statements on low back pain being almost entirely "nonspecific" and "idiopathic" are based on a single, general practice study on *acute low back pain* (less than 2 weeks) in the late 1960s [19]. The terms "idiopathic low back pain" and "nonspecific low back pain" then reentered the literature with the 1982 Symposium on Idiopathic Low Back Pain publication by eminent spine surgeons Augustus White and Alf Nachemson [20]. Research published at the same time as the Dillane study was ignored [31]. While Dillane et al. [19] accurately reported his experience as a general practice doctor (in 1950s Boston, MA USA) caring for patients with less than 2 weeks of low back pain, his results should not be applied beyond that time or narrow clinical setting. However, the statistic continues to be taught in medical school and utilized by many authors, payors, and policymakers to apply to all causes of low back pain and most importantly to patients with chronic low back pain. The statistic is an example of a "literature legend" wherein the original study, clinical setting, and relevance have been lost over time. "Nonspecific low back pain" and "idiopathic low back pain" are terms used primarily by detractors of spine surgery and spine injections. Just as the majority of etiologies of failed back surgery syndrome have been elucidated, so have the etiologies of "nonspecific low back pain." The terms "idiopathic" and "nonspecific" low back pain are not supported by the current published evidence.

## 38.3 History of "Failed Back Surgery Syndrome (FBSS)"

It is useful to examine the history of failed back surgery syndrome (FBSS) or persistent pain after lumbar surgery (post-lumbar surgery syndrome) before looking to the present state-of-the-art in diagnosis and treatment of the condition. Patients with persistent pain after lumbar spine surgery represent a heterogeneous group which has changed over the years due to advances in surgical techniques and diagnostic/therapeutic spinal injections. Until the early 1980s, largely singlesurgeon or single-institution case series results were reported, but no evidence-based, critical quantitative studies. A brief historical review demonstrates the significant advances achieved in both diagnosis and treatment of low back and leg pain before and after lumbar surgery. Reviews of spine surgery outcomes published in the 1980s [5, 33] lacked MRIs and diagnostic/therapeutic spine injections and were unable to diagnose a significant percentage of PLSS patients. The authors attributed PLSS in many patients to psychosocial comorbidities and secondary gain as primary pain drivers. Long et al. [33] were unable to find a cause for the patients' pain in up to 44%of patients: 21 % of patients were classified as "normal," and 21 % were diagnosed as having expected postoperative changes/mild spondylosis – not consistent with their persistent pain. A majority of patients were instead diagnosed with primary psychiatric and addiction problems as well as prominent motivations for secondary gain. Pain medicine specialists, also lacking diagnostic tools at that time, diagnosed a majority of patients with persistent pain after lumbar surgery as having a "centralized pain syndrome," as opposed to ongoing, correctable peripheral nociceptive lesions. Patients with centralized pain syndromes and "learned chronic pain behavior" were relegated to functional restoration, medication management, and cognitive behavioral therapy interventions.

Seven seminal quantitative studies are reviewed here (see Table 38.2). In 1981, Burton et al. [5] published the first quantitative study on the etiologies of "failure of surgery in the lumbar spine." Collectively, this paper reflects the experience of surgeons who treated an average of "800 FBSS cases per year." The authors summarize the status of patients with persistent pain after lumbar surgery, circa 1970s: "The FBSS entity represents a highly complex challenge to the physician. Patients with this condition can rarely be "cured" because FBSS is actually a spectrum organic of disease processes complicated by secondary financial gain and learned chronic pain behavior. While many of these patients can be "salvaged" to varying degrees by comprehensive rehabilitation programs, it is uncommon to achieve complete pain relief by any combination of therapeutic measures. This is due in part to the great difficulty in quantitating pain and associated psychologic, occupational, social, monetary, intellectual,

1			0 5 5		1	·		
	Burton et al. [5]	Burton et al. [5]	Long et al. [33] <sup>c</sup>	Bernard [34] <sup>d</sup>	Fritsch et al. [2]	Waguespack et al. [35]	Slipman et al. [6]	DePalma et al. [36] <sup>i</sup>
Surgical								
Spinal stenosis, total	64	72	5	29	-	-	21.5	_
Stenosis, foraminal	-	-	-	-	-	-	12.4	_
Stenosis, central <sup>a</sup>	7	14	-	29	-	-	5.9	-
Stenosis, lateral	57	58	-	-	0	29	3.2	0
Internal disk disruption/ disk degeneration	-	-	-	29	12	25 <sup>f</sup>	21.5 <sup>h</sup>	11 <sup>j</sup>
Severe spondylosis	-	-	5	-			-	
Instability	-	-	-	2	12	5	0.5	0
Spondylolisthesis	-	-	-	4			1.6	
Recurrent or retained HNP	12	16	1	33	62°	11 <sup>g</sup>	12.4	0
HNP at a new level	-	-	-	7	22	-	-	-
Scoliosis	-	-	1	-	-	-	-	-
Pseudoarthrosis	<5	-	-	29	-	14	-	-
Foreign body	<5	-	-	-	-	_	-	-
Surgery performed at the wrong level	<5	-	-	-	-	-	-	-
Traumatic meningocele	-	-	1	_	-	-	_	-
Tarsal tunnel syndrome	-	_	1	_	_	_	_	_
Fractured hip	-	-	1	_	-	-	_	_
Compression fracture	-	_	1	_	_	_	_	_
Synovial cyst	-	-	-	_	-	-	1.1	-
Vascular claudication	-	-	-	_	-	_	1.1	_
Pseudomeningocele	-	-	-	_	-	-	0.5	_
Nonsurgical					1			
Arachnoiditis	6 <sup>b</sup>	16 <sup>b</sup>	13	11	-	1	0.5	0
Epidural or intraneural fibrosis	6 <sup>b</sup>	8 <sup>b</sup>	14	-	4	-	14.6	0
Nerve injury during surgery/battered root	<5	-	6	-	-	-	1.6	-
Chronic mechanical pain	<5	-	-	-	-	-	-	-
Transitional syndrome	<5	-	-	-	-	_	-	-
Unknown	<5	-	-	-	-	6	5.6	0
Normal	-	-	21	-	-	-	-	-
Expected postop/minor spondylotic	-	-	21	_	-	-	-	-
Cancer	-		4					
Musculoskeletal abnormality	-	-	3	_	-	-	-	-
Degenerative spondylosis	-	-	-	9	-	-	-	-
Mechanical low back pain	-	-	-	-	-	-	9.1	_
Radiculopathy	-	-	-	_	-	_	5.4	_
Radicular pain	-	-	-	_	_	_	4.8	_
Deconditioning	-	-	-	_	-	_	3.8	-
Facet syndrome	-	-	-	-	-	_	2.7	18

 Table 38.2
 Comparison of studies on failed back surgery syndrome (reported as %)

531

(continued)

	Burton	Burton	Long	Bernard	Fritsch	Waguespack	Slipman	DePalma
	et al. [5]	et al. [5]	et al. [33] <sup>c</sup>	[ <mark>34</mark> ] <sup>d</sup>	et al. [2]	et al. [35]	et al. [ <mark>6</mark> ]	et al. [36] <sup>i</sup>
Sacroiliac joint syndrome	_	-	-	-	-	-	1.6	43
Complex regional pain syndrome	-	-	-	-	-	-	0.5	-
Fibromyalgia	_	_	-	-	_	-	0.5	
Diskitis/infection	_	-	-	-	-	1	0.5	-
Hardware/soft tissue irritation	_	_	-	-	-	-	-	14
Neuropathic pain	_	_	-	-	-	9	-	0
Psychological	_	_	-	-	_	3	-	-

#### Table 38.2 (continued)

<sup>a</sup>Stenosis, central: includes fusion overgrowth

<sup>b</sup>Burton et al. [5]: Arachnoiditis or epidural fibrosis not defined

<sup>c</sup>Long et al. [33]: 78 patients; 64 underwent surgery

<sup>d</sup>Bernard [34]: 19 patients had more than one diagnosis

<sup>e</sup>Fritsch et al. [2]: 40 % recurrent HNP; 22 % recurrent HNP and new HNP

Waguespack et al. [35]: 17 % IDD; 3 % within fusion; 5 % IDD with stenosis

<sup>g</sup>Waguespack et al. [35]: 6 % recurrent HNP; 5 % recurrent HNP and stenosis

<sup>h</sup>Slipman et al. [6]: Internal disk disruption syndrome is included in surgical diagnoses although nonsurgical intradiscal thermal and orthobiologic treatments are being researched

<sup>i</sup>DePalma et al. [36]: Patients presented with primarily axial low back pain

<sup>j</sup>DePalma et al. [36]: 3 cases, 2 cases adjacent to surgical level and 1 case removed by one level from surgical level

motivational and educational factors" [5]. During the 1970s, diagnostic imaging was limited to x-ray, CT scan, CT myelography, and selective use of discography. MRI (particularly with and without gadolinium postoperatively) has contributed significantly to understanding the etiologies of PLSS.

In the first substantive review on FBSS, Burton et al. [5] reported (see Table 38.2) almost 75 % of the patients with persistent pain had correctable lesions caused either by lateral stenosis (58 %) or retained/recurrent disk herniation (12-16 %). Nonsurgical etiologies diagnosed (without MRI) included: arachnoiditis (6-16 %) and epidural fibrosis (6-8 %). With the introduction of MRI (with and without gadolinium) and diagnostic spine injections, diagnostic sensitivity and specificity improved markedly over the next three decades. Surgeons also responded to this early data with improved surgical techniques or revision strategies to eliminate or reduce certain persistent postsurgical pain problems, such as residual lateral/foraminal stenosis, re-herniation/ residual disk herniation, "battered root syndrome"/traumatic neuritis, epidural fibrosis, and arachnoiditis.

Long et al. [33] studied 78 patients admitted to the inpatient Johns Hopkins multidisciplinary pain clinic program for an average of 18 days. Of 1,541 admitted to the pain clinic, two-thirds were noted to suffer from disabling low back and leg pain. The "typical patient" had undergone three spine surgeries and six myelograms and was "misusing narcotics and psychotropic agents." Approximately one-third of patients were diagnosed with primarily iatrogenic complications from spine surgery. A prospective study of patients admitted from 1979 to 1981 was performed. The average age was 43 years old (range of 19-67 years). The average duration of pain was 7.2 years. The first surgery occurred in the early to mid-1970s. Fifty-three patients (68 %) were involved with disability litigation or receiving disability compensation. This study is the only PLSS study obtained from a pain clinic. Patients were independently evaluated by both a neurosurgeon and orthopedic surgeon. Demographically, this patient cohort represents the "worst of the worst" in terms of persistent pain after lumbar surgery. The authors summarized their findings, noting that 67% of patients did not meet generally accepted surgical criteria

per neurosurgery. Twenty-six patients (33 %) met clinical criteria for surgery. Second surgery criteria were met in 18 patients (40 %). The authors found that 52 patients (67 %) had "persistent pain" of with normal preoperative imaging or showed "nonspecific DDD." These patients also had significant "underlying psychiatric abnormalities." Sixteen patients (21 %) had no physical abnormalities to explain their complaint and were classified as normal. Another 16 patients (21 %) had "minor postoperative changes insufficient to cause disabling pain." Twenty-seven patients (35 %) had a complication from the previous surgery. (Sixty-four patients underwent 171 surgeries; patients underwent an average of 2.6 surgeries.) Thirteen patients (17 %) had spondylotic disease and six patients (8 %) received a new diagnosis. The authors found significant psychiatric pathology to which persistent pain after lumbar surgery was attributed: 10 patients (13 %) had a definitive psychiatric diagnosis, 34 patients (44 %) had a maladaptive personality disorder, and 34 patients were normal. Thus 57 % were classified with significant psychiatric disorders. Sixty-seven patients (86 %) experienced reactive depression. In terms of pain medication used, the authors reported extensive aberrant pain medication usage: 58 patients (74 %) "misused narcotics," 9 patients had (12 %) "drug addiction," and 54 (69 %) patients had withdrawal symptoms. In summary, the most common diagnoses given were "normal" in 16 patients (21 %) and "expected postop changes/minor spondylosis" in 16 patients (21 %) and various primary psychiatric disorders. Expected postop changes were defined as: "insufficient to warrant any interventional procedure and did not necessarily explain intractable pain." Therefore in 42 % of patients, no primary diagnosis and/or no "objective physical cause" was identified to explain pain. The remaining most common diagnoses (total 33 %) were attributed to surgical complications: localized epidural scar, 11 patients (14 %), arachnoiditis, 10 patients (13 %), and traumatic neuritis 5 patients (6 %). (The surgical complication rate among this 78 patient cohort was 12 %.) Six patients had an unexpected source of back/ leg pain which was incorrectly diagnosed.

Viewed historically, Long et al.'s [33] ability to diagnose the cause of persistent pain suffered greatly from lack of advanced imaging (MRI) and diagnostic spine injections; moreover, many of these patients were given an incorrect primary psychiatric diagnosis for pain and labeled as addicts or substance "misusers."

Bernard [34] reported on etiologies of PLSS in 45 patients in a single-author practice audit. His diagnostic workup of the etiology of persistent pain was extremely robust - with extensive diagnostic imaging and an average of 2-year follow-up after surgery for correctable causes of pain. The author reported that intrathecally enhanced CT, MRI, and provocative discography with CT scan were critical to obtain a diagnosis; a single imaging study revealed surgical abnormalities in only 61 % of cases. The following are variables found to be statistically significant in predicting a successful repeat surgery: noncompensable injury (p < 0.04), return to work (p < 0.002), and achieving a solid fusion (p < 0.0012). Variables not statistically significant (p>0.05) are as follows: age, absence of litigation, number of previous operations, psychological diagnosis, and postoperative diagnosis. Bernard's study made significant contribution in terms of the role of diagnostic imaging to obtain a specific diagnosis. CT myelography was conclusive in 40 % of cases; however, it missed contained disk herniation at 21 adjacent levels. Diagnosis of contained disks can only be shown by provocative discography with CT scan. Intrathecally enhanced CT scanning picked up lateral recess stenosis and small disk herniations overlooked with CT myelography. MRI use in this study was limited due to the recent introduction in the technology. MRI with gadolinium is highly sensitive and specific for distinguishing scar tissue from disk herniations. MRI did show decrease in T2-weighted signal in patient disks; however, provocative discography was necessary to determine if the disk was a true pain generator. Provocative discography disclosed 25 symptomatic contained disks. Lumbar discography was conclusive in 32/34 patients. The author did not find that a "poor psychological profile" predicted an unsuccessful surgical outcome. The author stated: "patients with poor psychological profiles who are experiencing pain may improve when the pain is alleviated" [34]. Thirty-six out of 45 patients underwent fusion for their persistent pain. Fusion was undertaken when the workup revealed recurrent disk herniation, contained painful discogenic disease (with positive discogram), or instability. A solid fusion was achieved in 34/36 patients. Achieving a solid fusion predicted a successful outcome (P < 0.0012). Diagnostic facet blocks or SI joint blocks were not done. It is likely that some degree of symptomatic facet arthropathy was present in this cohort. Fusion addresses more than one pain generator: disk, facet, and instability of the index segment.

Fritsch et al. [2] retrospectively reviewed 182 revisions on FBSS from 1965 to 1990. The authors previously reported a reintervention rate of 10.8 % in 1,500 lumbar diskectomies. A total of 182 revisions were performed on 136 patients; 44 patients (34 %) were revised multiple times. Imaging was used to assist in diagnosis as follows: CT myelography in 54 patients (40 %), CT scan 41 patients (33 %), and MRI 4 patients (4 %). Clearly, MRI now provides much greater diagnostic information than CT scan; moreover, with the dyes previously used in CT myelography, the risk of arachnoiditis was not trivial. Re-intervention was primarily for recurrent or un-influenced sciatic pain and neurologic deficiency. A total of 84 % patients were diagnosed with: true re-occurrent disk herniation (44 %); true reoccurrence and new herniation (22%); and new herniation (22 %). Four percent (4 %) of patients had epidural fibrosis as a primary diagnosis. The other reason for reoperation was "instability," reported in 12 % of patients based on history, x-ray, CT, and physical exam findings. Laminectomy performed in the primary surgery was reported as the only factor leading to a higher rate of revisions. Epidural fibrosis and instability increased to greater than 60 % in multiple revision patients.

The next quantitative study on the etiology of long-term failures from lumbar spine surgery was published by Waguespack et al. [35]. This study was the first to employ diagnostic spine injections to precisely identify pain generators. A total of 181 patients with persistent pain after lumbar surgery between 1995 and 1997 underwent extensive diagnostic testing including: highresolution CT scan, MRI, standing flexionextension x-rays, discography, additional diagnostic spinal injections, and a psychiatric evaluation. With the advent of MRI and improvements in surgical techniques, the incidence of residual stenosis dropped to 29 % (from 58 % in Burton et al. [5]). Four other etiologies were identified as causes of persistent pain: retained/ recurrent disk (7 %), painful disk (17 % at surgical level, 3 % at nonoperated level), neuropathic pain (9%), and instability (5%). Recognition of the adverse effects of dyes (i.e., iophendylate (Pantopaque®)) used in CT myelography and the introduction of MRI reduced the incidence of arachnoiditis to very rare. Painful internal disk disruption was increasingly recognized as a significant contributor to persistent axial low back pain through the use of discography (albeit controversial).

Two additional quantitative studies, performed by interventional spine physicians, have since been published. Slipman et al. [6] studied 197 patients and identified 23 various diagnoses. He found that 55 % of patients with persistent pain after lumbar spine surgery had a surgical etiology and that 95 % of patients could be provided a diagnosis. The most common diagnoses were: residual stenosis (21.5 %), retained/recurrent disk (12 %), painful internal disk disruption (22 %), and neural fibrosis (14.5 %). DePalma et al. [36] used modern zygapophyseal joint blocks, sacroiliac joint blocks, provocative discography, and hardware blocks to evaluate 28 patients presenting with primarily persistent axial pain after lumbar fusion. Given this referral population bias, certain types of pain patients were likely referred to other centers for treatment of persistent nonsurgical leg pain with spinal cord stimulators. Evidencing improved diagnosis and treatment of PLSS, DePalma et al. [36] did not diagnose any patient with PLSS from residual stenosis, retained/residual herniation, or epidural fibrosis. He reported the following etiologies of PLSS: sacroiliac pain (43 %), painful internal disk disruption (IDD) (25 %), facet pain (18 %), and hardware pain/soft tissue irritation (14 %). All patients (100 %) received a diagnosis. The early studies by Burton (1981) [37] and Waguespack et al. [35] did not report the sacroiliac joint (SIJ) as a potential pain generator; Slipman et al. [6] reported a 1.6 % incidence of SIJ pain vs. 43 % by DePalma et al. [36]. Painful internal disk disruption incidence stayed relatively stable from 17 % in Waguespack et al. [35], 22 % per Slipman et al. [6], to 25 % with DePalma et al. [36]. All these investigators used provocative discography to diagnose painful internal disk disruption. Facet joint pain also took greater prominence, from no report of this pain generator in the earlier surgical studies to 2.7 % by van Wijk et al. [38] and then 18 % reported by North et al. [39]. Fortin et al. [40] also reported more hardware-related pain than prior investigators (14 %).

## 38.4 Spinal Fusion and Spine Injections: The Socioeconomic Controversy

To estimate the number of patients per year with persistent pain after lumbar surgery, accurate numbers on the how many spine surgeries are performed per year as well as surgical success rates

are needed. In 1997, there were more than 293,000 spine surgeries performed [41] with costs at ~\$5 billion dollars. In 2002, not long after the introduction of pedicle screws and the "cage rage" (based on data derived from proprietary sources on the spine industry), more than one million spinal surgeries were being performed per year: 600,000 uninstrumented and 400,000 instrumented cases, respectively [37, 42, 43]. In 2002, orthopedic industry data sources reported a yearly growth rate 3-5 % for uninstrumented cases and 6-8 % for instrumented cases [44-46]. In 2002, the spine market was forecast to compound at 22 % annually due to increased use of spinal instrumentation and aging of the baby boomer generation [37, 46]. By 2004, spinal fusions costs alone generated \$16 billion dollars. Researchers critical of the rise in spine surgery noted that the United States had the highest rates of lumbar fusions in the world (see Fig. 38.1), two to five times higher than Sweden, the Netherlands, Australia, and the United Kingdom [47].

Deyo [48] reported a 220 % increase in lumbar fusions between 1990 and 2000 without an apparent improvement in clinical outcomes [49, 50] (see Fig. 38.2). This 2007, "anti-fusion" publication was widely read and disseminated by the US national media, thus placing spinal fusion under intense scrutiny and apparently single-



**Fig. 38.1** Ratios of back surgery rates in selected countries to back surgery rate in the United States (1988– 1989) (Adapted from Cherkin et al. [47])



**Fig. 38.2** Lumbar spine fusion rates, US National Inpatient Sample (Adapted from Deyo and Mirza [50])

handedly leading to decrease spinal surgery rates in the United States.

Responding to the increase in spinal fusion rates and costs, policymakers and payors began scrutinizing the medical necessity of lumbar fusion, particularly for degenerative disk disease. In 2010, the insurer CIGNA set a new policy for lumbar fusions: to obtain a lumbar fusion, patients had to first participate in a 6-month, physician-directed functional restoration program with behavioral modification and physical therapy [51]. Policymakers and payors are also looking more closely at the risks of spinal fusion. One researcher reported increased mortality after spinal fusion. Juratli et al. [52] examined mortality after spinal fusion among Washington state injured workers from 1994 to 2001. A total of 2,378 patients underwent lumbar fusion; 103 (4.3 %) were deceased by 2004. Analgesic-related deaths were the number one cause of death among this patient group, responsible for 21 % of all deaths. The risk of analgesicrelated death was highest in men ages 45-54 years with degenerative disk disease (rate 7.45) who received instrumentation or intervertebral cage devices. Further investigation is needed into this report of increased mortality after lumbar fusion and to determine if patients receiving instrumentation indeed were at increased risk of analgesic-related death versus patient receiving bone grafts alone.

In 2008, beginning with the Great Recession and a confluence of other factors, the number of spine surgeries began to decline. Factors affecting the decline in number of spine surgeries include multiple publications critical of lumbar fusion [48, 50], increased scrutiny from payors and policymakers on spine surgery rates, costs and outcomes, and increased scrutiny on physician-owned distributorships. By 2011, a total of ~465,000 fusion operations were reported according to the Agency for Healthcare Research and Quality [51]. The research firm GlobalData reported that about 87 % of spinal procedures in 2013 were fusions [51].

According to spine industry analysts, a continued reduction in the number of spinal fusions per year is projected. The Sunshine Act (2013) is expected to inhibit collaboration and innovation between spine surgeons and industry [53]. GlobalData lowered its predicted compound annual growth rate for spinal fusions from 10 % annually to 5 % through 2020 due to multiple factors [54]: stricter reimbursement policies with surgeons reporting more denials for fusion, healthcare reform with mounting economic pressures, loss of physician autonomy over patient care, and a shifting focus to motion-preserving spinal technologies and intradiscal biologics. In January 2014, the Millennium Research Group reported that the global market for spinal nonfusion technologies is expected to triple in size through 2022, surpassing \$1.6 billion, driven by emerging markets in Asia Pacific, Brazil, India, and China [54].

Spine interventionists have also come under increasing scrutiny and control due to dramatic increase in spine injections over the last decade. Interventional pain management (IPM) techniques for chronic spinal pain increased markedly from 2000 to 2011 based on data from the CMS (Centers for Medicare & Medicaid Services) [55]. IPM techniques overall increased by 228 % over this interval or 177 % per 100,000 Medicare beneficiaries (see Fig. 38.3). The population growth of this demographic during this interval was 18 %. The highest increases were noted for facet injections and sacroiliac blocks with a total increase of 386 % and 310 % per 100,000 beneficiaries, respectively. Other types of injections reported the following percent increases (total and per 100,000 Medicare beneficiaries): 168 % and 127 % for epidural and



**Fig. 38.3** Illustration of distribution of procedural characteristics by type of procedures from 2000 to 2011 (Adapted from Manchikanti et al. [55])

adhesiolysis procedures, 150 % and 111 % for other types of nerve blocks, and 28 % and 8 % for percutaneous disk procedures. The geometric annual average was 11.4 % for all IPM procedures with facet injections showing the highest increase at 13.7 % per year.

In 2008, the Office of the Investigator General (OIG) of the US Department of Health and Human Services focused its attention on this explosive rate of growth. Sixty-three percent (63 %) of injections did not meet criteria for medical necessity. Subsequently, facet joint injections have declined by 6 % since 2008. A recent audit by Noridian reported 60-95 % of facet joint injections to be medically unnecessary [56]. The United States Office of Inspector General (OIG) in 2008 [57] reported that Medicare paid over \$2 billion in 2006 for IPM procedures. These investigators reported that 63 % of facet joint injection services and 34 % of transforaminal epidural injections did not meet the medical necessity criteria, leading to a total of \$175 million of improper payments between 2003 and 2007.

The spinal interventionalist faces many of the same challenges as spine surgeons from payors and policymakers. IPM physicians have to take into account the risks and technical difficulties of with each procedure while keeping in mind the external payor pressure from CMS and the current healthcare cost reform environment. One of the goals of this chapter is to provide the highquality evidence from the literature for the appropriate and efficacious use of diagnostic and therapeutic spine interventions in the PLSS patient.

## 38.5 Lumbar Spinal Surgery Outcomes: A Brief Review

Spine surgeons and spine interventionalists are under increased scrutiny for spine surgery and spine injection indications and outcomes. Increasingly, payors and policymakers are dictating how surgeons should diagnose and treat chronic low back pain patients. Spine specialists and their patients will be best served by an evidence-based diagnostic and therapeutic approach to spinal pain to rebut critics. Spine surgeons and interventionalists must independently establish transparent, rigorous indications for spine surgery, continue technical refinements, and publish high-quality literature – or spine care will be increasingly withheld from patients. The lower the reported success rate for primary and revision surgeries, the less likely payors and policymakers will pay for spine surgery. A selfcritical look at the definition of success rates is also needed among spine specialists. It is no longer sufficient for an author to personally ask if a patient is satisfied with their surgery. A surgeon's or spine interventionalist's results are considered more robust and therefore unbiased, if independent assessment of success is undertaken, along with inclusion of a range of primary outcomes, including: functional status (ODI), reduction in medication use, and return to ADLs/work. Evidence now is also weighted by type of study: retrospective reviews are ranked much lower than randomized controlled trials. In plain terms, to a payor or policymaker, a patient who is not a success is a failure; thus, that term is used in this chapter. However, among spine specialists it is well known that even partial pain relief, when a patient suffers from intractable pain, is meaningful.

Based on literature from the 1960s to 1980s, FBSS is quoted to affect up to 40 % of patients undergoing lumbar spinal surgery [39]. Based on recent data using the most stringent criteria for success, reports of persistent lumbar and/or leg pain after lumbar surgery can range from 20 % to 67 % for microdiskectomy and lumbar fusion, respectively [58, 59]. Patients undergoing microdiskectomy for primary leg pain have the best outcomes. Fusion outcomes vary considerably. Reports of success also vary considerably according to the "view" of the publishing authors, i.e., whether the authors are considered to be "prospine surgery" or not. Some surgeons also are very conservative in their use of fusion for DDD. Each study must be scrutinized carefully, particularly in terms of how the authors define "success." The true prevalence of PLSS is difficult to assess because of the heterogeneous nature of the syndrome, variable definitions of success, and access to public and proprietary data sources for the number, types, and outcomes of surgeries performed each year. Both chronic spinal pain and persistent pain after lumbar surgery present huge individual and societal costs. In an era of increasing healthcare restructuring and "reform," spine specialist are "awash" in "guidelines" and increasing restrictions on spine surgery and spine injections.

Regarding surgery outcomes, all interested parties question spine surgery studies which report greater than 90-95 % success rates. When Asch et al. [58], a neurosurgeon, used more rigorous criteria for assessing success after microdiskectomy among his patients, he reported leg pain relief (per VAS) at 80 % and low back pain relief at 77 % (success per VAS score defined as pain 0-4; VAS failure pain 5-10). These figures mean 20-23 % of microdiskectomies could be classified as unsuccessful. Santiago-Dieppa et al. (2014) reported on outcomes in older patients with spinal stenosis after non-instrumented arthrodesis, finding a ~60 % incidence of persistent back pain and leg pain with 31 % of patients requiring reoperation over the 7-year follow-up [60]. Regarding lumbar fusion, Fritzell et al. [61] reported a 63 % success rate with 37 % of patients who were unchanged or worse. Using very strict success criteria, from surgeons who perform fusion for DDD as a last resort, success rates are reported as low as 33 % and 43 % [59, **62**].

The definition of success after spine surgery remains controversial, with marked variation in criteria applied often lacking independent review. Currently "success" in spine surgery is primarily being defined by authors well known for taking a conservative approach to fusion for DDD. Carragee and Cheng [63] performed a study on 165 patients undergoing spinal fusion to determine a preoperatively determined, patient-acceptable definitions of minimally acceptable outcomes for success. In summary, the patient-acceptable, minimal clinical important difference (MCID) was: at least a 3/10point decrease in pain, an improvement in Oswestry Disability Index (ODI) of 20 or more\*, discontinuation of opioid medication, and return to some occupational activity. Of note, increasing the definition of a successful functional outcome
from an 11-point ODI increase to a 20-point increase is appropriate. Historically, spine surgery studies have only set the bar at an 11-point change, which is not a meaningful functional improvement. A 20-point change in ODI is also the most likely to improve a patient by at least one disability grade. The ODI scoring system includes a description of degrees of disability relating to scores. Scores from 0 % to 20 % indicate minimal disability; 20–40 %, moderate disability; 40–60 %, severe disability; 60–80 %, crippled; and 80–100 %, bedbound or exaggerating [64]. Both the Asch and Caragee's studies are commendable for increasing the bar for a successful functional improvement by raising the ODI to a 20-point change.

Using the most rigorous composite definition of success to date in the surgical literature, Mirza et al. [59] reported success rate of 33 % for spinal fusion for painful internal disk disruption/discogenic pain, thus obtaining a failure rate of 67 % [59]. Published lumbar fusion failure rates range from 30 % to 67 % [60-62, 65-68]. Reports of surgical success rates have come under increasing scrutiny for setting the bar to low [58]. As the bar for success is raised, often by opponents of spinal fusion, published success rates fall. It is generally accepted that success rates for spinal fusion are best for spondylolisthesis (78–91 %) as opposed to degenerative disk disease [62]. Mirza et al. [59] are well known for taking a more "conservative" approach to spinal fusion for DDD. A statistical "composite" definition of success was defined as 30 % improvement in Roland Score, 30 % improvement in pain, and no opioid medication use and working (if relevant). This definition for success appears to be the most stringent presented to date in the spine literature. However, despite the high-bar, surgical intervention was superior to nonsurgical intervention at 1-year. Nonsurgical treatment obtained a 15 % success rate. A more detailed survey of additional lumbar surgery studies is discussed later in the chapter.

The authors of this chapter report the data from the literature with an attempt to provide a "balanced" view of current evidence, from the spectrum of both "positive" and "negative" spine surgery publications, as well as publications between these extremes. This is not a critical review or systematic review of the evidence. Every attempt has been made to identify the group publishing the report as knowing the authors' surgical approach typically predicts the outcomes they will report. Across the board, outcome reporting obtained by the operating surgeon from the patient underreports pain; therefore, careful attention has been paid to the demographics of the study and the definition of success. Throughout the orthopedic community, surgical outcome reporting has become more stringent out of medical necessity and increased scrutiny and external pressure from nonphysician stakeholders in the age of healthcare reform. It is now standard to include functional outcomes (Oswestry Disability Index), medication intake, and return to work. Ultimately, this expansion of the definition of success will benefit both patient and surgeon. To estimate the number of patients with persistent low back and/or leg pain, current outcome data on the most commonly performed surgeries in the lumbar spine are reviewed for: microdiskectomy, spinal stenosis decompression, lumbar fusion, and lumbar total disk arthroplasty. To arrive at best estimates of success after lumbar surgery, this brief focused review of the literature includes recent systematic reviews, randomized controlled trials, and large cohort studies.

### 38.6 Lumbar Microdiskectomy: Success Rates

For lumbar microdiskectomy, patient satisfaction rates in the literature have varied from 40 % to 98 % [69] with microdiskectomy having an 88–98 % satisfaction rate over open diskectomy. However, in this same study, patient outcomes are not as robust when reported by grade. Patients reported the following outcomes after microdiskectomy: excellent (39 %), good (34 %), satisfactory (19 %), and failed (9 %). Moreover, the longer patients are followed up after spine surgery, the greater decline in satisfaction. Fritsch et al. [2] reported on results of 182 patients undergoing revision diskectomy between 1965 and 1990. In 80 % of the patients, the results were satisfactory in short-term evaluation, decreasing to 22 % in long-term follow-up (2–27 years).

Asch et al. [58] were critical of publications reporting greater 90-95 % success rates after microdiskectomy. The authors concluded that outcome success rates of 75-80 % are more "realistic" than studies claiming greater than or equal to 90-95 % improvement. A prospective study of surgical outcomes in 212 consecutive patients was performed. Independent examiners collected and analyzed outcome data. The author points out that potential bias may be introduced into the literature when outcomes are reported by the operating surgeons and not an independent examiner. Results obtained from independent observers are typically less favorable. More recently, important functional outcome measures have been added to strengthen postsurgical assessments. Asch et al. [58] reported the following: leg pain relief, 80 %; back pain relief, 77 %; improvement in Oswestry Low Back Disability Index, 78 % (ODI <40); satisfaction with surgery results, 76 %; return to normal daily activities, 65 %, and, return to work, 61 %. Worker's compensation status and increasing age had negative effects on outcome. Based on this data, 20-25 % of patients would have persistent pain after microdiskectomy. Asch et al. [58] estimates are consistent with some of the most recent figures on outcomes after microdiskectomy as reported by other authors. Klessinger [70] reported a 25 % (120/479) prevalence of persistent axial low back pain after lumbar microdiskectomy. A 9 % reoperation rate was reported for recurrent lumbar disk herniation.

McGirt et al. [71] performed a meta-analysis of 54 studies with 13,359 patients comparing incidence of short-term and long-term (2-year) low back or leg pain after aggressive (AD) versus limited diskectomy (LD) between 1980 and 2007. After 2 years the reported incidence of recurrent back or leg pain was 11.6 % for LD versus 27.8 % for AD or a 2.5-fold decrease in persistent pain among patients undergoing limited diskectomy. Aggressive diskectomy involves aggressive removal of the herniated disk material with curettage of the remaining disk. The limited diskectomy technique involves removal of the disk fragment only with minimal incursion into the disk space. The AD technique has been criticized as leading to accelerated disk degeneration with loss of disk height and increased incidence of persistent low back and/or leg pain. Conversely, the LD technique has been criticized for an increased incidence of recurrent disk herniation (RDH). Overall, this paper suggests that ~10-30 % patients may have persistent axial low back pain or leg pain based on the type of microdiskectomy. Repeat surgery is less successful, and according to several studies, only 60-82 % of patients with recurrent disk herniation improve after surgery. In patients who have only epidural scar tissue, the success rate of reintervention is as low as 17–38 % [72, 73]. Recently, Lurie et al. (2014), known to belong to a conservative care spine research group, reported 8-year follow-up results from the Spine Outcomes Research Trial (SPORT) for surgical versus nonsurgical treatment for lumbar disk herniation. Researchers found superior outcomes for pain, physical functions, and ODI in the surgical group [74].

# 38.7 Non-instrumented Lumbar Fusion for Spinal Stenosis: Success Rates

Santiago-Dieppa et al. [60] reported long-term (median follow-up 7.7 years) outcomes of 376 patients undergoing non-instrumented lumbar arthrodesis at a single institution over a 20-year period. The primary outcomes were: symptom resolution, development of adjacent segment disease (ASD), and need for reoperation. The patient's mean age was 61 years at presentation with neurogenic claudication due to multi-level spinal stenosis. At follow-up, the presence of back pain decreased from 91.5 % to 61 %; radiculopathy decreased 81-58 %. The cumulative rate of ASD was 18 % (69 patients). The reoperation rate due to non-improvement or worsening of symptoms was 31 % (115 patients), necessitating additional stabilization techniques. Overall, 60 % of patients continued to have back pain,

42 % of patients continued to have leg pain, and 31 % needed reoperation.

## 38.8 Lumbar Fusion: Success Rates

Studies vary widely regarding lumbar fusion success rates. Rates of success also vary widely based on diagnosis and on the definition of success. Complications after lumbar fusion include: instrument failure (7 %), iliac crest bone-donor site pain (11 %), neural injuries (3 %), and pseudoarthrosis (15 %) [75]. Secondary surgeries for persistent pain or surgery-related complications are reported in 20 % of patients after lumbar surgery [76]. Lumbar spinal surgery success rates drop to 30 % after a second spine surgery, 15 % after a third surgery, and approximately 5 % after the fourth surgery [77].

Philips et al. [65] published a systematic review of 3,060 patients undergoing lumbar spine fusion for CLBP due to degenerative disk disease (DDD). The weighted average improvement in back pain was 37/100, Oswestry Disability Index (ODI) 22, and 12 points on the SF-36 physical component scale. Patient satisfaction averaged 71 %. These results indicate that ~30 % of patients were not satisfied. The reoperation rate was 12.5 %, with 9 % at the index level.

Caragee et al. [62] compared outcomes of fusion for lumbar DDD versus unstable spondylolisthesis. In the spondylolisthesis group, 23 of 32 patients (72 %) met the highly effective success criteria compared to 8 of 30 (27 %) of the presumed discogenic pain cohort. The minimal acceptable outcome was met in 29 of 31 patients (91 %) in the spondylolisthesis group versus 13 of 30 (43 %) of the discogenic pain group. From this data approximately 10 % of patients would be considered a failure after fusion for spondylolisthesis versus 57 % for presumed discogenic pain.

In a randomized controlled trial with 2-year follow-up, Fritzell et al. [61] reported fusion to be superior to usual physical therapy care directed by a family practice doctor in patients with severe axial low back pain of an average of 8 years. All primary outcomes measures favored surgical intervention with statistically significant improvements. Back pain was reduced by 33 % (64-43) or 19-point/~decrease of 2-point VAS in the surgical group vs. 7 % (63–58) in the nonsurgical group. ODI score was reduced 25 % (47–36) or 11 points. In the surgical group, a total of 63 % of patients rated themselves as "much better" (29 %) or "better" (34 %) vs. 29 % in the nonsurgical group. Among fusion patients 29 % rated themselves as "much better" vs. 14 % in the nonsurgical group. The net back to work rate was in favor of surgical treatment versus usual care group, 36 % vs. 13 %, respectively. In total, 37 % of surgical patients failed to respond to the intervention and rated themselves as unchanged (24 %) or worse (14 %). On every metric, surgical patients did better than nonsurgical patients. In patients with an average of 8 years pain, cure is not likely; the objective instead is to meaningfully reduce the amount of the patients' pain. In summary, substantial improvement was seen in 25-33 % of surgical patients vs. 4–8 % of nonsurgical patients. In addition, depression was reduced by 20 % in the surgical group vs. 7 % in the nonsurgical group. Of note, 28 % of patients randomized to usual care ultimately decided to undergo surgery at 2 years. This study reported surgical treatment of DDD as superior to usual care, 63 % of patients reported being "much better" or "better," leaving 37 % of patients with no improvement or worse pain. Success in this study was a VAS decrease in 2 points (i.e., from six to four) and an ODI decrease of 11 points. The criteria for success were minimally based compared to more rigorous independent outcome measures. Overall, 29 % of patients rated themselves as much better, 34 % better, 24 % unchanged, and 14 % worse. This data suggests a 38 % failure rate. Three other RCTs, comparing fusion to intensive rehabilitation [78] and cognitive behavioral therapy with exercise [66, 68], did not show fusion to be a superior intervention.

Brox et al. [66] compared lumbar fusion with posterior transpedicular screws (not anterior/posterior fusion) to cognitive behavioral therapy (CBT) and exercise at 1 year, using the ODI as the primary outcome. Patients were ages 25-60 years with low back pain lasting longer than 1 year, imaging evidence of disk degeneration at L4-5 and L5-S1, and ODI score of at least 30 points. The average age of subjects was ~43 years, and duration of low back pain ~10 years; working was 24 % fusion group and 22 % CBT/exercise group; and 70-80 % of patients were either on sick leave, in rehabilitation, or on a disability pension. In the Norwegian healthcare systems, individuals on sick leave more than a year are entitled to a rehabilitation benefit or a disability pension. Approximately 50 % were taking analgesics and ~40 % were smokers. In terms of medical comorbidities, approximately ~70-80 % of subjects had comorbidities not specifically described. We know from the SPORT study that subjects with comorbidities such as diabetes and depression have inferior surgical outcomes [66]. Subjects were randomized to fusion or cognitive behavioral therapy. No difference was observed in ODI between groups at 1-year follow-up with only a 3 % lost to follow-up rate. ODI was reduced from 41 to 26 (15 points) after surgery which was not statistically significant from cognitive behavioral therapy (CBT) and exercise 42-30 (8 points). An independent observer rated success, reporting 70 % after surgery and 76 % after CBT/exercise. The early complication rate in the surgical group was 18 %. Seventy percent (70 %) success would give a 30 % failure rate. Brox et al. [68] also studied the 4-year follow-up results. Fifteen (23 %) patients assigned to surgery underwent reoperation. ODI decreased from 14 points, from 44.1 to 29.7 which was not statistically significantly different between groups. Both randomized groups reported less pain and better function at 9 years vs. baseline; more operated patients were on pain medication and not working. In 2006, Brox et al. [67] published another study comparing instrumented fusion with cognitive behavioral therapy plus exercise in patients with CLBP greater than 1 year. Patients had already undergone a diskectomy and had persistent pain. The primary outcome measure was a reduction in ODI. The fusion group averaged a 9-point ODI decrease

from 47 to 38 versus the CBT/exercise group with a 13-point decrease from 45 to 32. At 1-year follow-up, such small ODI decreases represent an unimpressive outcome measure for defining success. The authors reported  $\sim 50$  % success rate in both groups. This would give a 50 % rate of patients with persistent pain after instrumented lumbar fusion for persistent pain after diskectomy. At 11 years, Mannion et al. [79] were to perform long-term follow-up at 11 years and no difference was found in ODI (-0.7 vs. -0.8) or patient self-reported outcomes between fusion and CBT/exercises groups, respectively. The study had a very high lost to follow-up rate of 45 %. Of note, by 4 years, 25 % of surgical patients underwent reoperation. In summary, in the various RCTs by Brox et al., their early study reported 1-year failure rates or unchanged/worsened pain rate of 30 %; the midterm study reported a "failure rate" of 50 %.

Recently, a community-based study comparing surgical versus nonsurgical treatments for discogenic back pain was published by Mirza et al. [59]. This research group is well known for opposition to the increased rates of lumbar fusion and the use of fusion to treat discogenic pain [50]. Mirza et al. [59] enrolled 495 patients; 86 patients (17 %) underwent surgery. Grading success with a never-before utilized composite criteria (30 % improvement in Roland score, 30 % improvement in pain, no opioid pain medication use and working - if relevant), the 1-year success rate was 33 % for surgery and 15 % for nonstructured conservative care. Longer-term follow-up is critical to assess outcomes in the surgical group. Based on 1-year follow-up, 33 % of patients had a successful outcome and 67 % of patients did not meet success criteria.

## 38.9 Lumbar Disk Arthroplasty: Success Rates

Our data on total disk arthroplasty is limited. Summarized here are two RCTs comparing TDR to circumferential fusion and two different TDR implants. Zigler et al. (2012) reported 5-year results of a prospective, randomized trial comparing circumferential arthrodesis to the ProDisc L for single-level DDD [80]; TDR was noninferior to fusion by a 12.5 % margin. Seventy-seven (77 %) patients in both groups were satisfied with surgery. Both groups had a 48 % decrease in VAS scores and similar ODI decreases [80]. These outcomes suggest similar outcomes between TDR and fusion patients. Patients' satisfaction was reported as 77 % among both groups. Eighty-three percent (83 %) of TDR patients vs. 68 % of fusion patients stated they would have surgery again.

Guyer et al. (2014) compared two different disk arthroplasty implants with a 2-year followup [81]. Success was defined by decrease in VAS, 15-point decrease in ODI, no complications, and lack of reoperation. Success rates were similar: 68 % for the Kineflex L-disc versus 67 % for the Charité Disk. At 2-year follow-up, greater than 90 % of patients in each group were satisfied with outcome. Reoperation was performed in 10.3 % of the investigational group versus 8.4 % of the Charité group. A total of 32-33 % of patients did not meet criteria for success. Likely, the disparity between the high self-reported satisfaction rate versus traditional outcome metrics is due to the novelty of the artificial disk. This phenomenon is not uncommon with the introduction of new drugs or medical devices.

# 38.10 An Algorithmic Approach to the PLSS Patient

An algorithmic approach to working up residual pain after lumbar spine surgery is useful. The spine surgeon or interventionalist utilizes history, physical exam, imaging, and diagnostic/therapeutic spine injections to identify and treat persistent pain after lumbar surgery. Increasingly, spine and pain specialists are also carefully scrutinizing patient's comorbidities as treatment of these conditions may help optimize patients for better surgical outcomes.

There are various approaches to diagnosing the cause of PLSS: time from surgery, predominance of back versus leg symptoms, and the three-column approach. Briefly, Crock [82] described two postsurgical failure patterns: "outright failure" and "failure following initial temporary relief." He emphasized that outright failure is usually related to wrong diagnosis. Immediate failure can also be due to technical error, instrumentation failure, or residual disk. There is temporary relief followed by pain within a few weeks: recurrent disk herniation or infection. There is onset of pain more than 6 months after surgery: re-herniation/ new herniation, posterior column pain (facet/SI joint), or epidural fibrosis. Longer-term failures are as follows: loss of stability or stenosis at the surgical site or adjacent site, adjacent facet pain, SIJ pain, painful internal disk disruption (IDD),

Table 38.3       Causes of failure after spinal fusion	Time	Back pain predominates	Leg symptoms
	Early (weeks)	Infection	Nerve impingement by fixation device or cement
		Wrong level fusion	
		Insufficient levels fused	
		Psychosocial distress	
	Midterm (months)	Pseudoarthrosis	Fixation loose
		Disk disruption	Early adjacent disk degeneration
		Early adjacent disk degeneration	Graft donor site
		Inadequate reconditioning	
		Graft donor site	
	Long term (years)	Late pseudoarthrosis	Disk with pseudoarthrosis
		Adjacent level instability	Adjacent level stenosis
		Acquired spondylolysis	Adjacent level disk
		Abutment symptoms	Stenosis above fusion
		Compression fracture above fusion	

Adapted from Kostuik's failures after spinal fusion 1997b [83]

Column	Posterior	Middle	Anterior
	Facet joint	Radicular pain/	Discogenic pain: painful IDD at surgery site or adjacent level
	Sacroiliac joint	Radiculopathy	Recurrent/residual disk herniation
	Soft tissue: "fusion disease"	Epidural fibrosis	Pseudoarthrosis
	Myofascial pain	Transition syndrome/vertical disease	Adjacent segment disease
		Arachnoiditis	

**Table 38.4** Algorithmic approach to the most common causes of residual lumbar and/or leg pain after lumbar surgery by column

and pseudoarthrosis. Kostuik [83] (see Table 38.3) summarizes the causes of failure after fusion by time and predominance of back versus leg symptoms. Another straightforward approach is to divide the spine into three columns: posterior, middle, and anterior (see Table 38.4).

### 38.11 History

Accurate history can help identify the cause of pain in many cases. The patient should describe their pre- and postoperative pain in great detail (best to worst pain rating, intermittent versus constant pain, nocturnal pain, pain aggravating or relieved by certain positions, etc.), particularly if there is a change in quality or location. The qualities of nociceptive nerve pain and neuropathic pain are well known to the surgeon. With the 2013 publication of Hannah Albert's work on Modic type I changes [84], greater attention is being paid to patient's description of constant, compelling, deep, aching axial pain that does not respond to or is aggravated by usual conservative care such a physical therapy. Dr. Albert asserts that Modic type I changes on MRI are due to disk and endplate infection with Propionibacterium acnes which may be treated with 100 days of oral amoxicillin-clavulanate three times a day.

The examiner should determine whether the patient has primarily axial lumbosacral pain or leg pain or a combination. The workup for primary persistent lumbar pain is different for primary leg pain. The emerging standard for persistent leg pain (with no correctable surgical cause) is referral for spinal cord stimulator. Some consider spinal cord stimulation a first-choice treatment in PLSS due to lumbosacral fibrosis [85]. The reader is referred to outside sources for a more detailed description of this technology. In the United States FBSS is the most common reason for the use of spinal cord stimulation therapy [86]. The analgesic efficacy ranges from 52% to 72% and a recent analysis of the national Italian register of implantable systems found that 81 % of the patients reported a positive assessment for pain control with a lowering of drug needs in 71 % of the positive responders [87].

If axial back pain is the principal complaint, the disk, facets, and SI joint must be considered; painful hardware may also be an issue. Revel's criteria has not been validated or reproduced for predicting facet pain based on correlation with diagnostic facet blocks. Revel's criteria for facet pain (5/7) are: age greater than 65 years and pain not exacerbated with coughing, not worsened with hyperflexion, not worsened by forward flexion, not worsened when rising from forward flexion, not worsened with extensionrotation, and well relieved by recumbency. Revel's results have never been duplicated; moreover, to validate his history questions specific to facet pain, he only performed a single diagnostic anesthetic block [88]. The false-positive rate for a single facet block in this setting is 49 % [89]. The gold standard for the diagnosis of facet pain is dual, comparative diagnostic blockade. However, the finding of axial lumbosacral pain below L5 is a sensitive finding on history and is specific for sacroiliac pain and validity by dual-controlled SI joint diagnostic blocks [90]. The examiner should therefore endeavor to identify lumbar versus sacral pain when evaluating axial pain.

Medical comorbidities should be reviewed. Anecdotally, the first author of the study evaluates all chronic low back pain patients, potential primary lumbar surgery patients, and PLSS patients for presence of obesity, prediabetes/type 2 diabetes mellitus, visceral adiposity, metabolic syndrome, subclinical hypothyroidism, hypovitaminosis D, hypogonadism, andropause, menopause, and osteopenia/osteoporosis. If hardware is being considered for a fusion or if the patient has pseudoarthrosis, a recent DEXA scan for bone density is useful to assess bone health. The lower the bone density, from osteopenia to osteoporosis, the less the bone will be able to protect the disk and the greater risk for increased axial loading on the disk and posterior column as well as soft tissues (muscles, fascia, and ligament dysfunction). Hardware is also more likely to fail in osteopenic/osteoporotic patients, both women and men. Many women undergoing back surgery have entered menopause and no longer have the bone protection afforded by estrogen. Surgeons should take a careful history to see if their female patients have undergone surgery and entered surgical menopause or normal menopause. The average age for women to enter menopause is 51 years old. Estrogen is a key hormone for bone metabolism for men and women. Menopause is associated with a loss of bone mineral density (BMD): 10-year cumulative loss was 11 % in the lumbar spine. Serum estradiol (E2) concentrations predict fractures. Older women with total E2 levels <5 pg/ml have a 2.5 increased risk of hip and spine fractures independent of age and body weight; similar associations are found in men [91].

Similarly, we now appreciate that men also experience an age-related decrease in testosterone. Surgeons may consider asking their male patients about loss of vigor, depression, reduced sexual function, and muscle weakness to assess if their male patients have entered andropause. From a lab standpoint, hypogonadism is defined by some authors as a total testosterone less than 400 ng/dL; total testosterone less than 275 ng/dL is frank hypogonadism. Men with suboptimal testosterone also appear to be more prone to osteopenia/osteoporosis. Reduced muscle mass may also impair spine stability and efficacy of postoperative rehabilitation. We are awaiting the 2015 publication results of the Testosterone Trial which is testing the hypothesis that testosterone treatment of men in unequivocally low total serum testosterone (<275 ng/dL) will increase bone density of the spine [91–93].

Surgeons already make sure that their patients stop smoking for at least 6 months or quit altogether; NSAID use is also discouraged. New research shows that there is much more surgeons and bioidentical hormone specialists can do to help achieve optimal outcomes in surgery. This is clearly an emerging field of research. Koerner et al. [94] identified patient factors associated with the best outcome after diskectomy from the Spine Outcomes Research Trial (SPORT). Patient factors associated with the largest improvement in ODI at 4 years with either surgical or nonoperative treatment included: higher baseline ODI, body mass index (BMI) < 30, not being depressed, being insured, having no litigation or worker's compensation claim pending, and having symptoms for <6 weeks. Patients with diabetes did not benefit with surgery versus nonoperative care. Knowledge of these factors is useful to both patient and surgeon whether further surgical interventions are undertaken or not. Psychosocial issues, depression, anxiety, and so forth should be identified and treated as they can make the primary organic problem more difficult to diagnose and treat. When treating persistent lumbar and/or radicular pain after lumbar surgery, it is prudent to identify and modify as many risk factors as possible: abnormal lab values, obesity, prediabetes, type 2 diabetes, alcohol intake, smoking, depression and hormonal imbalances, and depression. Anecdotally, the first author's preference is the use of bioidentical hormones instead of synthetic hormones. Some surgeons may consider checking labs to help stratify and optimize their patients for surgical success or sending their patients to a specialist in bioidentical hormone replacement (a full discussion of the pros and cons of bioidentical hormone replacement versus synthetic hormones or traditional agents used to increase bone density is beyond the scope of this chapter). Briefly, some of the most interesting

findings in terms of vitamin and hormone function will be presented. The face validity of optimizing hormonal health and thus tissue health, bone strength, and healing capacity (especially under surgical stress such as fusion) would seem to make sense in terms of decreasing the incidence of persistent pain after lumbar surgery. Anecdotally, when potential primary surgery as well as postsurgical candidates present to the first authors' practice, they undergo a complete hormone and lab panel, to attempt to remove any barriers to healing. The first author has found the abovementioned medical comorbidities to be much more common in spine patients, especially patients with persistent pain after lumbar surgery. This is an area of research in which little research has been published.

Chronic pain patients, including low back pain patients, are known to have insufficient vitamin D 25-hydroxy (vitamin D3) [95, 96]. Severe to moderate deficiency of vitamin D25-OH, defined as <10 ng/mL, was statistically significantly associated with Modic type I changes, with an odds ratio of 0.30 (95 % CI 0.12; 0.75). The other two-thirds of patients had vitamin D25-OH >20 ng/mL which these authors considered "normal." Many authors disagree with this as the cutoff for normal, let alone "optimal" for bone and muscle healing. Dissenting authors have defined optimal as >50 ng/mL [97].

Briefly, we will discuss vitamin D3 as it is linked to bone health, falls prevention, immune function, cancer protection, and type 2 diabetes. However, controversy remains about the definition of insufficient versus optimal; thus the normal range is defined from 20 to 100 ng/ mL. Deficient is based on the author ranging from <20 to <50 ng/mL. Great interest is emerging in having an "optimal" versus "normal" level of vitamin D. Anecdotally, some physicians work to optimize modifiable factors such as vitamin D3 level and prevention of type 2 diabetes, so patients can achieve the best outcomes if undergoing spine surgery. The optimal level for bone health is >50 ng/mL. According to the Third National Health and Nutrition Examination Survey, 61 % of white and 91 % of black Americans suffer from vitamin D insufficiency

(25[OH]D <32 ng/mL). Recent studies have demonstrated that a minimum 25(OH)D level of 32 ng/mL is necessary for optimal protection from fracture and intestinal absorption of calcium. Calcium supplements for bone health will be ineffective if a patient is vitamin D deficient [98]. If a patient has cancer or heart disease, some studies recommended obtaining a blood level of 70-100 ng/mL. An intake of 2,000 IU of vitamin D3 will raise the blood level by 20 ng/ mL over several months. Some patients may need to take 5,000-8,000 IU of vitamin D3 to obtain these blood levels. The current US Recommended Daily Allowance (USRDA) of 600 IU of vitamin D3 per day is considered inadequate by many; the RDA in Canada is 800-2,000 IU/day [99]. It may be prudent for surgeons considering fusion to optimize vitamin D 25-hydroxy levels to at least 50 at least 3 months before surgery. Lastly, the combination of any low sex steroid hormone (estrogen/testosterone) and 25-hydroxy vitamin D is associated with an increased fracture risk [91].

The majority of patients with PLSS are also on opiate medication which depresses the hypothalamic-pituitary-adrenal (HPA) axis. Valverde-Filho et al. (2014) studied chronic spinal and oral-morphine induced neuroendocrine and metabolic changes in non-cancer patients referred to a pain center [100]. The authors compared three groups, patients with intrathecal morphine pumps (0.2-10 mg/day), patients on oral opiates (60-120 mg/day), and patients receiving non-opioid analgesics. All patient groups experienced improvement in pain scores. Libido, reduced potency, hot flashes, and menstrual cycle dysfunction occurred more often in opiate groups vs. the non-opioid group. Low total testosterone (hypogonadotrophic hypogonadism) was significantly more prevalent in the morphine groups (58 % and 70 %, respectively) vs. non-opioid group (17 %). This group reported clinical symptoms concordant with lab findings. Total cholesterol >200 mg/dL and high-sensitivity CRP (C reactive protein) was significantly more frequent in opiate groups. Lastly, total bone mineral density was below normal in men receiving spinal morphine (p=0.014). Growth hormone, thyroidstimulating hormone, and adrenocortical hormones were also suboptimal, but did not reach the same degree of statistical significance.

This author does not recommend statins due to the negative effects on nerve and muscle. At a minimum, statin doses should be the lowest possible "effective dose." In meta-analyses of RCTs, muscle adverse effects (AEs) are greater with statins versus placebo. Factors such as metabolic syndrome, thyroid disease, and genetic mutations linked to mitochondrial dysfunction can amplify AEs of statins. There is emerging evidence of additional statin-induced AEs such as neuropathy and cognitive impairment [101]. The reader is encouraged to read emerging literature in this area to obtain optimal surgical outcomes in patients, especially patients with persistent neuropathic or radicular pain after lumbar surgery.

### 38.11.1 Physical Exam

For the PLSS patient, the surgeon repeats a detailed preoperative spine exam with a few additions. Comparison of pre- and postoperative exam findings is essential. New pain generators and possible overlooked pain generators are sought. In some cases the patient may have an entrapment of the superior cluneal nerve (amenable to local surgical release), which travels through an osteofibrous tunnel over the medial iliac crest, approximately 7-8 cm lateral to midline at L4-5. Infrequently, osteoarthritis of the hip can refer to the low back, shin, and calf areas [102]. The presence of low back pain was also statistically more common in patients with longer duration of symptomatic end-stage hip osteoarthritis. Rarely, hip labral tears can also refer to the ischial tuberosity or anterior thigh regions [103]. Hip exam maneuvers for osteoarthritis (i.e., decreased internal rotation) and labral tears/ femoral acetabular impingement tests (i.e., flexion-adduction-internal rotation – FAIR test) are critical both preoperatively and postoperatively to screen for less typical hip pain referral patterns. A neurologic exam must be performed to assess for focal deficits, particular residual impairment after disk herniation. If patients have symmetric hypoactive or absent ankle jerks, they likely have a comorbid peripheral neuropathy (most likely due to type 2 diabetes, alcohol, or subclinical hypothyroidism). Abnormalities in each of these areas adversely affect generally healing, especially nerve recovery. If the patient is diabetic, further surgical intervention can be expected to have a statistically significantly inferior outcome compared to nondiabetics. This finding also applies to efficacy of various pharmacologic and interventional spine treatments and is helpful in counseling patients regarding expectations for treatment [94].



Fig. 38.4 The distribution of pain required for inclusion in sacroiliac pain study (Adapted from Maigne et al. [151])



**Fig. 38.5** The distraction test (testing right and left SIJ simultaneously). *Note*: Vertically oriented pressure is applied to the anterior superior iliac spinous processes directed posteriorly, distracting the sacroiliac joint



**Fig. 38.6** The thigh thrust test (testing the right SIJ). *Note*: The thigh and sacrum are fixed against the table with the left hand, and a vertically oriented force is applied through the line of the femur directed posteriorly, producing a posterior shearing force at the SIJ

Physical exam findings for spinal pain such as for facet pain are not reliable; diagnostic facet blocks are the gold standard. As the most recent quantitative study of PLSS reports the sacroiliac joint to be responsible for 43 % of persistent axial low back pain in PLSS [36], more attention is being directed to its precise diagnosis and treatment. The typical location of pain from the SI joint is 10 cm caudal and 3 cm lateral to the posterior superior iliac spine (Fig. 38.4). This referral pattern was obtained from asymptomatic



**Fig. 38.7** Gaenslen test (testing the right SIJ in posterior rotation and the left SIJ in anterior rotation). *Note*: The pelvis is stressed with a torsion force by a superior/posterior force applied to the right knee and a posteriorly directed force applied to the left knee



**Fig. 38.8** The compression test (testing right and left SIJ). *Note*: A vertically directed force is applied to the iliac crest directed toward the floor, i.e., transversely across the pelvis, compressing the SIJs

subjects undergoing provocative arthrography [90]. Researchers have shown that a battery of  $\geq$ 3/5 provocative SI joint tests are highly predictive of SIJ pain based on studies correlated with results from fluoroscopically guided, comparative SIJ blocks [104–106]. The five SIJ tests include: distraction, compression, thigh thrust, Gaenslen, and sacral thrust. The five SIJ tests are reproduced for the reader (Figs. 38.5, 38.6, 38.7, 38.8, and 38.9) [36].



**Fig. 38.9** The sacral thrust test (testing right and left SIJ simultaneously). *Note*: A vertically directed force is applied to the midline of the sacrum at the apex of the curve of the sacrum, directed anteriorly, producing a posterior shearing force at the SIJs with the sacrum nutated

### 38.11.2 Imaging

To help determine the cause of persistent pain after lumbar surgery, imaging is a useful first step to help guide evidence-based spine interventions. First, sagittal plane and standing, weight-bearing radiographs (AP/lateral/flexion-extension) are obtained to assess for stability of the fusion construct. Ideally they can be compared to preoperative x-rays. Most consider listhesis greater or equal to 4 mm to be indicative of clinically important instability. Thin-section CT scans (2–3 mm) with sagittal and coronal reformations are used to assess disk levels with hardware. High-resolution CT scan can detect pseudoarthrosis (absence of bridging bone), foraminal stenosis, misplaced bone graft or cement, and migrated/misaligned hardware. Hardware loosening is seen with stress fatigue and typically associated with a 2 mm halo around the hardware [107]. Complications associated with instrumentation are assessed by CT scan and need to be ruled out in the patient with leg pain. After transpedicular screw placement, the rate of nerve root irritation is 1 %, typically caused by inappropriately low and medial screw placement [108].

Postoperative MRIs are typically obtained if surgical outcomes are not achieved in a timely fashion. The optimal MRI is performed with angled T2-weighted sections from T12-L1 to L5–S1, with 1 mm stacked images through the fusion level. Metal artifact reduction sequences (MARS) can be used to better image fusion levels. Post-contrast (gadolinium) T1-weighted images are particularly useful for residual/recurrent disk herniation, new herniation, enlargement/ enhancement of spinal nerves, perineural/epidural fibrosis, and/or arachnoiditis; however, postoperative MRIs with GAD must be interpreted with caution and guided by relevant clinically concordant symptoms. Caution against overinterpreting fibrosis is particularly critical in the early (< than 6 months) postoperative period. The extent and degree of scar tissue enhancement decrease within the first 6 months. In a study of MRI in 34 successful diskectomy patients, intrathecal nerve root enhancement was present in 6 (18 %) patients 6 weeks after surgery, facet joint enhancement was seen in the majority of subjects at 6 months (63 %), and eight patients (20 %) had residual mass effect on neural elements with an enhancement pattern suggesting a disk fragment [73, 109]. Grane et al. [110] reported a 19 % prevalence of residual/recurrent disk herniation in asymptomatic, post-diskectomy patients. Epidural scarring is typically more pronounced if patients receive an MRI before 12 months [110].

The label "segmental or mechanical instability" has been used much too loosely to diagnose the source of pain in patients with degenerative spondylolisthesis (DS) and rotational translation (RT). Lattig et al. [111] reported that facet joint effusion on conventional MRI (mean effusion size 2.15 mm) indicated abnormal motion in DS and RT. The authors identified 160 post-surgical patients status post either decompression alone or decompression with fusion. Mean age of patients was 69 years old. Twenty-five percent (40/160) of patients had no facet joint effusion with the % slip on upright x-ray and supine MRI  $\leq 3$  %. In 77 % (108/140) of patients, % slip between x-ray and MRI was >3 % (mean 10.6 %, range 4–29 %) and was associated with a mean facet effusion size of  $2.15 \pm 0.85$  mm). Extent of facet effusion was significantly associated with increased slippage on standing x-ray versus supine MRI (p < 0.001); moreover, the extent of right/left difference in effusion was associated with the presence of rotational translation (RT  $1.31\pm0.8$  mm vs. no RT  $0.23\pm0.17$  mm, p < 0.0001). If patients have evidence of % slippage >3 %, facet effusion >2.15 mm, or asymmetric facet effusion, diagnostic facet medial branch blocks would be useful in determining if these are pain generators. More study is needed to determine the role of facet effusion on MRI in surgical decisionmaking for optimal surgical treatment (decompression vs. decompression with fusion).

Modic type I changes are receiving great scrutiny since the publication of Hannah Albert's controversial work in 2013 [84]. Dr. Albert asserted that these Modic Type I changes are associated with an infection by Propionibacterium acnes infection of the endplates and/or disk. This is not a typical "diskitis" or osteomyelitis; thus patients will have normal white blood counts and sedimentation rates. Blood cultures are negative. Dr. Albert's work follows in the footsteps of Barry Marshall who discovered that ulcers were in fact caused by H. pylori which required treatment with antibiotics. At 1-year follow up, after completion of a 100-day course of antibiotics, Albert et al. [84] found that patients with Modic type I changes had a statistically significant reduction in back pain and disability. Recently, another group showed Modic type I changes to be correlated with low back pain outcome at 1 year. Patients with Modic type I changes on MRI had more pain, more disability, and greater incidence of an unsuccessful return to work at 1 year [112]. The Modic type I change was the only degenerative manifestation found on MRI to correlate negatively with clinical outcome.

Gates et al. [113] evaluated patients with SPECT scan after lumbar surgery. Sixty-three patients with persistent back pain after spine surgery were evaluated. Fifty percent of patients underwent scanning after at least 2 years and 25 % between 1 and 2 years. Bone scan can detect biomechanical stresses at the surgical level or above/below the operative level. Bone SPECT scan was used to detect pseudoarthrosis, abnormal facets uptake, disk pathology, and sacroiliitis. In the 63 patients, 132 bone scan lesions were detected. Overall, 75 of 132 lesions (57 %) were identified in the operative field, with 57 abnormalities (43 %) outside the field. Sixty percent of the abnormalities in the facets, disk spaces, and vertebral bodies were at the operative level with 29 % above the operative level (primarily facet) and 11 % below (facet and sacroiliac joint). The most common abnormalities were as follows: 37 % facet (n=51), 22 % disk spacecentered conditions (degenerative disk disease, post-surgical reactive findings, no diskitis) (n=29), 15 % pseudoarthrosis (n=20), 14 % sacroiliac joint (n=18), and 7 % vertebral body lesions (n=9) and miscellaneous lesions. Sacroiliac joint uptake was seen in 15 patients for a total of 18 joints or 24 % of patients. Four fractures were uncovered and three in the vertebral bodies above the operative site. Many patients had more than one lesion.

Pseudoarthrosis causes continued motion at the operative site. Typically 6–9 months must elapse to see a solid fusion on x-ray. Two years is needed for a solid fusion to completely remodel. Based on studies with SPECT scans, increased tracer uptake at the fusion level at 1 year suggests raises possibility of pseudoarthrosis. All bone SPECT scans were performed at or beyond the usual window for surgical recovery. The SPECT scan can be a useful tool to add to the typical persistent lumbar pain patients after surgery. Additional diagnostic blocks can then be used to confirm or refute bone scan findings to distinguish between painless radiologic findings and actual pain generators.

## 38.12 Etiologies of PLSS: The Three-Column Approach

A three-column approach can be taken to diagnosis and treatment of the most common etiologies for persistent lumbar and/or radicular pain after lumbar surgery. The spine is divided into posterior, middle, and anterior columns. This approach assists the clinician with a straightforward algorithm for PLSS. In the remainder of the chapter, the primary etiologies of persistent pain are discussed, along with the current best evidencebased approaches to treatment. This review will not cover intraoperative and perioperative complications as these areas are well reviewed in other chapters. Intrathecal pumps and spinal cord stimulators are not covered in this chapter. The authors present stringent diagnostic and therapeutic criteria for spine injections. If spine interventionalists adopt less stringent diagnostic criteria or use suboptimal techniques, outcomes suffer. Poor outcomes affect the patient, spine interventionalist, and surgeon adversely and make payors reluctant to pay for treatment.

### 38.12.1 Posterior Column: Facet Pain

Lumbar surgery changes the stresses on the posterior column and could result in new pain generators in the z-joints, sacroiliac joints, or muscles/ligamentous tissues. Such effects might include increased segmental motion above or below a spinal fusion or increased or abnormal segmental motion due to partial removal of the intervertebral disk. The most common pain generators are the facet and sacroiliac joints and soft tissues. According to DePalma et al. [36], posterior column structures comprised 75 % of pain generators in his series of 28 patients with persistent axial lumbar pain after surgery: facet 18 %, sacroiliac joint 43 %, and hardware/soft tissue 14 %. The remaining 25 % of patients were diagnosed with painful internal disk disruption by provocative discography.

In the patient with persistent axial lumbosacral pain, zygapophyseal joint arthropathy (ZJA) pain must be considered as a possible treatable source of persistent pain. Pathophysiologically, we know that after lumbar anterolateral interbody fusion, increased capsular strain is placed on the adjacent segment and index segment facet joint capsules. The facet joint has both nociceptors and mechanoreceptors. The mechanoreceptors are also thought to function in a proprioceptive role [114]. Little et al. [115] examined the effects of a single-level, L4-5 anterolateral interbody fusion on adjacent facets. Fixation increased the moment for all levels for all motions. Intervertebral angle and plane strains were increased at L3-4 and L5-S1 levels after fixation. The L4–5 facet capsules demonstrated decreased and increased strains ipsilateral and contralateral, respectively, in response to anterolateral interbody fixation.

Attempts to identify a clinical "facet syndrome" have largely been fruitless. Revel et al. [88] reported that patients with five of seven clinical variables distinguished 92 % of patients responding to a single lidocaine block with  $\geq$ 75 % pain relief as positive response. Revel's criteria are as follows: age greater than 65 years and pain not exacerbated with coughing, not worsened with hyperflexion, not worsened by forward flexion, not worsened when rising from forward flexion, not worsened with extensionrotation, and well relieved by recumbency. The author's stated that these criteria could be used to select patients for facet injections; however, "these characteristics should not be considered as definite diagnostic criteria of lower back pain originating from the facet joint." Revel's study has never been duplicated; moreover, we know the single block false-positive rate to be unacceptably high (up to 49 %), thus making the claim of distinguishing 92 % of ZJA patients difficult to reproduce. Spine interventionalists fortunately solved the diagnostic dilemma with comparative blocks.

In patients with nonoperated, chronic low back pain, the prevalence of specific posterior column pain generators, i.e., painful zygoapophyseal joint arthropathy (ZJA), has been reported to range from 15 % to 40 % among persons with chronic low back pain [24, 26]. The first population studied had a prevalence of 15 % painful facet joint arthropathy. The study subjects were primarily male with a median age of 38 years and also receiving worker's compensation. The first study utilized a superior dual-block technique. The second study, which reported a 40 % prevalence of painful facet joint arthropathy, included primarily women with a median age of 59 years, using only a single intraarticular injection with a positive response defined as  $\geq 50$  % pain relief. Studies requiring at least 75 % pain relief reported prevalence rates of 27-45 % using controlled comparative local anesthetic medial branch blocks [116–119]. For a criterion of 90 % relief, the prevalence was 40 % in elderly female patients with no history of trauma [26].

Specifically to the question of prevalence of facet joint pain in patients with persistent axial low back pain after spine surgery, several studies have been performed. Prevalence ranges from 2.7 % to 33 % using varying criteria. Slipman et al. [6] reported a 2.7 % prevalence of facet joint pain (positive response criteria not reported). Using the standard technique of dual comparative medial branch blocks (with  $\geq$ 75 % relief) in postlumbar surgery patients, facet-mediated pain was reported as 16 % by Manchikanti et al. [89] with a 49 % false-positive rate for single blocks. The surgical group was composed of patients status post lumbar laminectomy (61 %), fusion with bone (20 %), fusion with hardware (18 %), microdiskectomy (8 %), and others (33 %, including disk arthroplasty) [89]. Greater than 80 % pain relief threshold was defined as a positive response; a screening block yielded a falsepositive rate of 49 % [89].

DePalma et al. [36] reported an 18 % prevalence of ZJA. Klessinger [70] using a dual-block  $\geq$ 80 % relief criterion reported an 8 % prevalence of facet syndrome after lumbar microdiskectomy. Siepe et al. [120] report a higher prevalence, likely due to using  $\geq 50$  % cutoff for positive response to screening blocks. These authors studied post-TDR pain etiologies with single, fluoroscopically guided facet intraarticular injections with local anesthetic and corticosteroids (>50 %relief) finding a 12.6 % incidence of z-joint pain in patients with chronic pain after TDR (ProDisc II), predominantly at the index level (86.4 %). (Intraarticular facet injections, however, lack specificity and sensitivity and are not considered the standard for diagnosing symptomatic facet joints [11].) Worse outcomes and a greater incidence of posterior joint pain were seen for TDR at L5/S1 (21.6 %) and two-level TDR (33.3 %). TDR may compromise the index segment while endeavoring to reduce adjacent segment degeneration [120].

Several studies of etiology of persistent pain after lumbar surgery show a lower prevalence of facet syndrome than in the nonsurgical patient literature; however, further study is needed. It is known that more stringent positive response criteria reduce prevalence.

To provide their patient with the best in spine care, today's spine surgeon is well served by making certain their patient's facets are evaluated using the evidence-based criterion standard for diagnosing ZJA, using at least a  $\geq$ 75–80 % relief with dual comparative anesthetic medial branch blocks to be considered a positive response. Practice guidelines for facet evaluation have recently been updated from the International Spine Intervention Society [10, 11]. Falsepositive rates are unacceptably high with single blocks reported at 37 % [121], 41 % [117], and 47 % [29] and 49 % [89]. When appropriate technique is utilized, medial branch blocks are target specific with a low false-negative rate of 8 %, due to unrecognized intravascular injection of local anesthetic [122]. Many surgeons will also have potential surgical patients undergo facet blocks prior to surgery.

### 38.12.1.1 Radiofrequency Ablation (RFA) of Medial Branches

Multiple clinical guidelines and systematic reviews have been published with unfavorable conclusions regarding the efficacy and utility of radiofrequency ablation (RFA) [123-132]. However, these reviews included studies with technically incorrect electrode placement and inadequately inclusion criteria to diagnosis zygoapophyseal pain [10, 11]. Correct technique for RFA of the lumbar medial branch is paramount. Bogduk et al. determined that the radiofrequency techniques used by Shealy in the 1970s [133, 134] and additional researchers through the 2000s [135–137] were inadequate due to lack of knowledge of medial branch nerve anatomy [138, 139]. Moreover, researchers demonstrated that thermal RF electrodes coagulate radially around their tip, not distal to the tip [140]. This discovery led to improved success rates for researchers who coagulated medial branch nerves by placement of RF electrodes "parallel" versus perpendicular to the target nerve. The RCT of RFA of lumbar facet joints by Van Wijk et al. [137] showed no difference from sham; however, the RF electrodes were placed perpendicular to instead of parallel to medial branch nerves, thus creating a small, ineffective ablative lesion no better than sham treatment [10, 11].

Many of the recent RFA clinical guidelines and systematic reviews also suffer from the inclusion of patients without facet-mediated pain. Recall that false-positive rates for a single block technique (with  $\geq 50$  % pain relief as positive response) can be as high as 50 % [89]. In many studies the bar for a positive response was set too low. The degree of diagnostic confidence is directly proportional to treatment outcome. If the dual comparative block approach is utilized, with  $\geq$ 70 % pain relief or complete pain relief and appropriate electrode placement, the patient is much more likely to benefit from RFA [117, 141]. Operational and technical guidelines for optimal performance of RFA are set forth in the recent edition of the International Spine Intervention Society Practice Guidelines [10, 11].

In a narrative review of papers utilizing correct indications and technique, medial branch neurotomy was determined to be clinically effective [142]. In summary, the studies using a positive response criterion of 50 % pain relief achieved an approximate 40 % success rate (defined as 50 % relief of pain) [143]. However, if the criterion for a positive response is elevated to complete relief after diagnostic blocks, 56 % of patients achieve complete pain relief for a median duration of 13 months per treatment [144]. Dreyfuss et al. [145] reported outcomes using strict operational criteria with 60 % of the patients obtaining 90 % relief of pain at 12 months and 87 % of patients obtaining at least 60 % relief. Derby et al. [141] reported favorable RF neurotomy outcomes with a 70 % cutoff value for subjective relief after dual blocks. Klessinger [70] reported on the efficacy of RF neurotomy for facet pain in post-diskectomy patients. Using a  $\geq$ 80 % diagnostic dual-block criterion to receive RFA, 59 % of patients reported at least 50 % pain relief for 6 months post neurotomy.

Recently increasing interest has emerged in treatment with pulsed radiofrequency (PRF). The mechanism of action of PRF is undergoing intensive investigation. Classic RF achieves temperature of 70–90 °C to achieve thermal coagulation of neural structures. PRF appears to work by both

nonthermal electromagnetic field effects and very brief heat spikes, from 45 to 50 °C. Very little tissue destruction occurs beyond 0.5 mm from the tip. In general PRF produces markedly stronger electromagnetic fields than RF. A recent review included six RCTs using PRF: PRF versus epidural steroid injections, PRF versus sham, and the remaining studies PRF versus conventional RF [146]. The best evidence of PRF is in the cervical spine, prior studies with RF to the DRG produced transient neuritis and motor dysfunction. Van Zundert et al. performed PRF to the cervical DRG in patients with chronic radicular pain [147]. At 3 months patients reported >50 % reduction in pain vs. sham and a 20-point reduction in VAS. Patients had PRF to the cervical DRG for complaints of radicular pain. A similar study in the lumbar spine was poor quality and could not be assessed for efficacy.

#### 38.12.1.2 Sacroiliac Joint Pain

In evaluating 368 patients presenting to a single spine surgeon's office for pain, including 25 % with prior diskectomy [148], 14.5 % of patients were found to have sacroiliac pain (SIJP) based on evaluation with diagnostic blocks. As sacroiliac pain can present with axial, lumbosacral, and pseudoradicular patterns, it is a critical diagnosis to consider both prior and after lumbar surgery (see Fig. 38.4: SIJ pain referral map). Early studies of PLSS reported no sacroiliac pain (see Table 38.2); Slipman et al. [6] reported a 2.7 % prevalence of SIJP in a heterogeneous PLSS population. By 2011, studying a more homogeneous population of patients presenting with persistent axial low back pain only after surgery, DePalma et al. [36] reported a 43 % prevalence of SI joint pain.

Based on current research using diagnostic blocks, sacroiliac joint (SIJ) pain currently appears to be an underappreciated source of chronic axial low back pain both pre- and postoperatively. Schwarzer et al. [27] were the first to use diagnostic blocks for SIJ pain. Using a validated historical factor, patient's subjective report of pain below L5 [149], Schwarzer et al. [27] reported a 30 % prevalence of sacroiliac pain in patients with CLBP using a single block technique. Using a dual comparative technique, reported prevalence of SIJP in patients with chronic LBP varies: 2 % Pang et al. [150], 6 % Manchikanti et al. [29], and 18.5 % Maigne et al. [151]. Maigne reported 30–47 % false-positive rates with single screening blocks [151].

Studies in postsurgical patients using a single, screening block protocol reported a 32-35 % prevalence of SIJ pain after lumbar fusion. Katz et al. [152] reported the SI joint as the cause of persistent pain after lumbar fusion in 32 % of patients; however, the authors utilized a single injection with local anesthetic and corticosteroid which would have an unacceptably high falsepositive rate. A positive response was defined at >75 % pain relief from local anesthetic and at least 10 days of continued relief. Maigne and Planchon [153] reported a 35 % prevalence, using a single block technique with >75 % relief. Of note, however, they also found that the only historical factor predictive of SIJ pain was pain of different character than preoperative pain, typically arising greater than 3 months postoperatively. Sacroiliac joint pain was not related to iliac crest donor site or presence of fusion to the sacrum or not. The authors did not find that iliac crest bone harvesting close to the joint was a risk factor, as was suggested by Ebraheim et al. [154]. Studies using a dual-block protocol, by Liliang et al. [155] and DePalma et al. [36], reported a 40 % and 43 % prevalence of SI joint pain, respectively. Both investigators studied patients with persistent pain below L5 after lumbar fusion.

An in-depth look at Depalma's recent work evaluated [36] 28 patients with CLBP after lumbar fusion; SIJs were symptomatic in 43 % (12/28) of the fusion cases. In DePalma's workup of patients with chronic axial lumbosacral pain after fusion, SIJD was the most common etiology for pain, followed by painful IDD 25 % (7/28), painful ZJA 18 % (5/28), and soft tissue irritation by fusion hardware 14 % (4/28 cases). More than 80 % (10/12) of positive SIJD cases had lumbar fusion to the sacrum versus L5 (p=0.0032). DePalma concluded that inclusion of the sacrum into the fusion construct appears to be a risk factor for subsequent development of SIJ pain. Biomechanical studies of the sacrum after fusion show increased angular motion and stress across the SIJ increase after lumbar fusion [156].

Liliang et al. [155] evaluated 130 patients with persistent lumbar and/or radicular leg pain after lumbar fusion. Fifty-two patients [52] met inclusion criteria: pain below L5 and at least three positive provocative SI joint maneuvers. Triple block criteria, requiring >75 % relief, were utilized. A false-positive rate of 26 % was reported if patients did not undergo a third diagnostic block. Twenty-one patients [21] or 40 % of patients met criteria for SIJ pain. Of the initial 130 patients referred, this represented 16 % (21/130) of patients with persistent pain after lumbar fusion. Liliang et al. [155] reported the following predictive factors: unilateral pain (76 %) versus bilateral (24 %), positive responses to more than three provocative physical exam maneuvers, and pain with characteristics different from preoperative pain. Number of levels fused or fusion across L5-S1 did not predict SIJ pain.

After lumbosacral fusion, Emani et al. [157] reported that 15 % of 60 patients with long fusions to the sacrum required removal of painful implants, primarily iliac as opposed to sacral screws. In some patients, pain after fusion to the sacrum is mechanical; in other cases it appears to be a soft tissue hyperinflammatory reaction to the wear particles from implanted materials [158, 159]. In a study of causes for persistent axial low back pain after total disk replacement (TDR) with fluoroscopically guided injections, the SI joint was identified as a cause of pain in 12 % of cases [120].

There are varying hypotheses for the etiology of SIJ pain after fusion. Frymoyer et al. [160] reported that the SI joint was essentially noncontributory to persistent pain after posterior fusion based on an imaging-only comparison of lumbar fusion patients (with and without donor site pain) versus diskectomy patients. The authors could find no difference in terms of flexionextension mobility or degenerative changes on x-rays of the SI joint. Subsequent researchers have found increased degeneration on CT scan in post-fusion patients. Ebrahim et al. (2000) performed CT scans of 22 patients with persistent pain in 24 sacroiliac joints, after posterior iliac crest bone graft harvesting, and found that disruption of the synovial part of the joint led to severe degeneration [154]. Of 16 joints with disruption of the inner table at the ligamentous part of the joint, 10 had mild degenerative findings and 6 showed moderate degenerative finding. Three joints with disruption of the synovial portion of the inner table showed severe degeneration. Five joints had disruption of the inner table without degenerative changes. In summary, the authors concluded that there is a high prevalence of inner table disruption after posterior iliac crest harvesting, with more severe degeneration seen with disruption of the synovial portion of the inner table. Graft harvesting was recommended in only zone 1, versus 2 or 3, to avoid any inner table disruption. The shortcoming of this study was the lack of correlation with diagnostic SI joint blocks. Ha et al. [161] followed the CT scans in patients 5 years post-lumbar fusion, with and without sacral fixation. Seventy-five percent (75 %) of patients developed degeneration of the joint versus the control group. Greater SIJ degeneration was seen in the group with fusion to the sacrum.

Investigators have reported increased SI pain post-fusion (SIJ pain diagnosed by ≥75 % pain relief after SIJ block) when there are a decrease of the sacral slope (resulting in a more vertical sacrum) and an increase of pelvic retroversion [162]. This sacral position replicates a sitting position for which the individual compensates with a loss of lumbar lordosis and increased hip stress. Post-fusion pain is significantly more likely in patients with a more vertical sacrum (less sacral slope) and more pelvic tilt [163]. Mechanically, the lumbar paraspinal muscles have to exert greater force to maintain upright sagittal alignment, theoretically leading to increased posterior column pain in muscles "fusion disease" and in the facet and sacroiliac joints. Over time, this constellation of findings can lead to increased axial loading on the disk [163].

In these studies on the sacroiliac joint as a cause of persistent pain after lumbar fusion, the Duval-Beaupère criteria are used to assist sagittal spine alignment (Fig. 38.10) [164]. The lumbopelvic



**Fig. 38.10** Lumbopelvic spine, lateral x-ray parameters. *LL* lumbar lordosis, *PI* pelvic incidence, *PT* pelvic tilt, *SS* sacral slope [162]

parameters include lumbar lordosis (LL), pelvic incidence (PI), pelvic tilt (PT), and sacral slope (SS) on the standing radiographs (Fig. 38.10).

Anatomically, the sacroiliac joint has both synovial and ligamentous components. The anterior inferior one-half to two-thirds of the SI joint is a classic synovial joint with adjoining hyaline cartilage surfaces, whereas, in the upper part of the joint, the sacrum and ilium are not in direct contact and are bridged by anterior and posterior interosseous ligaments. One other intriguing theory about SI joint pain addresses the ligamentous structures of the joint as pain generators.

SIJ pain may be more complex than first appreciated. Controversy remains over primarily dorsal versus mixed dorsal and ventral innervations of the joint. Fortin et al. [40] dissected SIJs and reported that the joint was wholly supplied by sacral loops from the dorsal rami from S1 to S3 (S4). These loops also innervate the interosseous ligaments and short and long dorsal SI ligaments; therefore potential extra-articular sources of SIJ pain cannot be ignored as noted by Murakami et al. [165]. Horwitz [166] reported that the long and short posterior SI ligaments are to contain dorsal rami branches from S1 to S2. Studies have shown that both the synovial portion and ligamentous portion of the SI joint have nociceptors [167].

Murakami et al. [165] screened for patients using a pain provocation test; then patients received either a fluoroscopically guided intraarticular or periarticular SI joint injection with lidocaine. The periarticular injection was effective in all patients, while the intraarticular injection was effective in only 9/25 patients. All of the intraarticular injection patients then responded to a periarticular injection. (For a more detailed description of the technique, the reader is referred to the article.) Vertical loading on the SI joint is primarily through the posterior and interosseous ligaments, where the posterior capsule is rudimentary or absent [168, 169]. The authors also found that the fluoroscopic injection into the middle one-third section of the SI joint had the highest yield, relieving more than 50 % of the pain. In a comprehensive review of SI joint pain [170], the authors reported evidence supports both intra- and extra-articular causes for SIJ pain, with clinical studies demonstrating intermediate-term benefit for both intra- and extra-articular steroid injections.

If the sacroiliac joint is determined with dual diagnostic blocks to be the pain generator, SI joint neurotomy can be considered for longer lasting relief. Cooled RF lateral branch neurotomy has also been shown to provide significant relief both short term [171] and long term for up to 2 years [172]. Hansen et al. [173] performed a systematic review of therapeutic interventions for sacroiliac pain; 11/56 studies met inclusion criteria. The evidence was fair in favor of cooled radiofrequency neurotomy and poor for short-term and long-term relief from intraarticular steroid injections, periarticular injections with steroids or botulinum toxin,

pulsed radiofrequency, and conventional radiofrequency neurotomy.

# 38.12.1.3 Myofascial Pain/"Fusion" Disease

Myofascial pain may predate surgery as part of the patient presentation. Myofascial pain or "fusion disease" may also result from extensive muscular/fascial dissection and prolonged retraction with increased intramuscular pressures. "Fusion disease" may also result from post-fusion compensatory hyperextension and a more vertical sacrum leading to increased firing of the erector spinae muscles and increased loading to the posterior column [163]. Straightforward myofascial pain with trigger points may be amenable to trigger point injections, botulinum toxin injection, or dry needling; however, "fusion disease" is unlike to respond to usual myofascial treatments.

# 38.12.1.4 Miscellaneous Causes of Posterior Column Pain: Superior Cluneal Neuropathy

Superior cluneal neuropathy was first described by Strong and Davila in 1957 [174]. In the rheumatology literature, an "iliac crest pain syndrome" (ICPS) was reported in 30-50 % of patients with chronic low back pain in general practice [175]. Iliac crest syndrome is described by rheumatologists as a legitimate, distinctive cause of chronic low back pain. It is diagnosed by physical exam with tenderness over the medial iliac crest, 6-8 cm from the midline at L4-5. Detection of this sign is reliable with a kappa score of 0.57 for inter-rater reliability. In 1999, Bogduk in regard to ICPS stated that "there is no evidence of its mechanism or cause" and that it constituted no more than a single clinical sign. Bogduk [176] posited possible causes of ICPS (but not substantiated at the time) to include: sprains of the lumbar intermuscular aponeurosis or iliolumbar ligament; multifidus muscle or gluteus maximus muscle strain; trigger point activity in the quadratus lumborum muscle; or, entrapment of the lateral branches of dorsal rami in the fascia attached to the iliac crest. In 2005, Akbas

et al. [177], reported a case of a man admitted to the hospital after 6 months of severe right-sided low back pain post ipsilateral decubitus ulcer surgery. On exam he had two exquisite tender points at 6.5 and 7.5 cm lateral to midline over the medial iliac crest. Local injections with anesthetic and corticosteroid were performed and the pain dissipated in minutes. Recent researchers have since shown that the iliac crest pain syndrome is in fact entrapment of the superior cluneal nerve in a rigid osteofibrous tunnel penetrating the thoracolumbar fascia at the medial iliac crest. Prevalence was reported as 10 % [178]. The SCN provides sensory innervation to the posterior iliac crest and upper buttocks. The SCN originates from the L1 to L3 nerves and also penetrates the psoas major. Ermis et al. [179] diagnosed 25 patients with medial superior cluneal nerve entrapment via ultrasound and used local anesthetic with steroid for treatment. Recently, Morimoto et al. [180] reported on the surgical treatment of SCN neuropathy. A microinvasive release was performed of the osteofibrous tunnel containing the SCN in 34 patients. The patients presented with pain 7 cm lateral to midline over the iliac crest. Average duration of pain was ~5 years. Pain was both unilateral and bilateral. After a diagnostic block with lidocaine with >75 % relief, and failure of blocks with corticosteroid, surgery was performed. All patients reported immediate and total relief of pain once the SCN was released from its osteofibrous tunnel. At 10 months, all patients reported complete pain relief; none experienced recurrence of pain.

# 38.12.1.5 Technical Note: Diagnostic Blocks, Performed with or Without Sedation

A recent study has brought into sharp focus the effect of sedation on the validity, accuracy, and treatment outcomes after diagnostic injections. Based on the results of this randomized controlled study [181], it is recommended that conscious sedation (including agents such as propofol, midazolam, and fentanyl) should not be used because of the statistically significant prob-

ability of obtaining a false-positive diagnostic block. To properly diagnose facet-mediated or sacroiliac joint pain, dual comparative anesthestic blocks are needed. Patients with false-positive results are likely to be subject to misdiagnoses and unnecessary procedures. Patients receiving blocks with sedation reported statistically significant greater mean reductions in pain diary scores and less procedure-related pain. Critically, a higher proportion of patients receiving sedation obtained >50 % relief (70 % vs. 54 %, p=0.039).

#### 38.12.2 Middle Column

## 38.12.2.1 Etiologies of Persistent Leg Pain After Surgery

The most common causes of middle column pain are: recurrent/residual disk herniation, new herniation at a new level, epidural fibrosis, and, over time, adjacent segment disease. Historically, residual lateral stenosis, far lateral herniations, sequestered fragments, and arachnoiditis were more prominent causes of PLSS in early surgical outcome studies. In the first quantitative study of PLSS by Burton et al. (1981), CT myelography with pantopaque was still common, and the authors reported an incidence of 6-16 % of arachnoiditis [5]. Once dye use was discontinued and MRI became available, the incidence of arachnoiditis became rare. Chronic leg pain is one of the most disabling sources of pain after lumbar surgery. Radicular leg pain/radiculopathy pain needs a critical analysis with a search for a correctable cause due to the significant morbidity and mortality of FBSS with primary neuropathic leg pain.

Post-diskectomy re-herniation, either ipsilateral or contralateral, ranges from 5 % to 15 % of cases at the surgical level [182–186]. MRI with gadolinium is the study of choice postoperatively; however, it must be carefully correlated with the clinic presentation. Many surgeons also recommend an epidural steroid injection to attempt to manage the pain without need for revision surgery. In a study of MRI in recurrent sciatic patients 1 year after surgery, 42 % of symptomatic patients had a disk herniation and 19 % of asymptomatic patients had a disk herniation [110]. Many patients with persistent or residual leg pain after surgery, without an obvious correctable lesion, simply need adequate time for the nerve to recover from prolonged compression and inflammation. Battered root syndrome can occur with prolonged and aggressive retraction and presence of a conjoined nerve root and/or excessive bleeding [187]. Persistent postsurgical leg pain can in rare cases result from arachnoiditis (3 %), not including lesions caused by prior myelography [188]. Three patterns are described for arachnoiditis, which albeit rare should be recognized (on T2-weighted MRI): nerve root clumping (type I, mildest type), "empty sac" with peripheral adhesions of nerve roots to thecal sac (type II, moderate severity), and an intermediate signal mass filling the subarachnoid space below the conus (type III, severe). The onset of the leg symptoms after an approximately 6-month painfree period can also suggest possible epidural fibrosis. Complications may also arise from hardware migration or misalignment. X-ray and thinsection CT scan can assess hardware position. Rare reports have also found pars interarticularis fracture after instrumentation. Iliac crest graft donor site has been reported as a cause of diffuse leg pain; however, iliac crest bone harvest techniques have improved markedly to spare the superior cluneal nerve. Hardware placement and iliac crest graft donor site have also improved to avoid disrupting the sacroiliac joint and accelerated degeneration which could present with lumbosacral pain as well as "pseudoradicular" leg pain. Leg pain occurring over the long term may represent adjacent segment disease (ASD) post-fusion.

When a correctable cause of persistent leg cannot be found, the etiology is typically attributed to epidural fibrosis (including intraneural and perineural). There is controversy about the true incidence of epidural fibrosis. Advocates of an anti-adhesion barrier gel and the lysis of adhesions procedure tend to report literature with a higher percentage of clinically significant epidural fibrosis, whereas the quantitative reviews noted in Table 38.1 report a range of epidural fibrosis from 4 % to 15 %, which is currently a

more reliable epidemiologic estimate. Thomson and Jacques [189] reported on the demographics of a subset of PLSS patients with severe neuropathic leg greater than back pain who were undergoing treatment with spinal cord stimulators. These patients have higher pain levels, increased disability/inability to work, poorer quality of life than patients with complex regional pain syndrome, rheumatoid arthritis, and fibromyalgia. This group likely constitutes the patients with the worst outcomes after fusion and is not representative of all patients with persistent pain after fusion. As compared to other chronic pain patients with complex regional pain syndrome (CRPS, previously called reflex sympathetic dystrophy or RSD) or rheumatoid arthritis, FBSS patients are severely disabled with an ODI of 56 (which is considered near "crippled" by pain) vs. 27 (low moderate disability) [189]. Demographically, these patients report an average back pain of 5/10 with leg pain 7.5/10 versus RA patients with pain ranging from 3.4 to 6.0/10. Quality of life by the EQ-5D is 0.16 or very poor. Seventy-eight percent (78 %) of FBSS are unable to work versus 31 % of CRPS patients.

In this study, patients were recruited from Canada, Western Europe, and Australia with severe neuropathic pain after PLSS. The prevalence of this PLSS subgroup was reported as 0.61 %, similar to rheumatoid arthritits (0.5 %)[189]. To estimate prevalence in the United States, we have to look at international spine surgery rates. Cherkin et al. (1994) presented an international comparison of back surgery rates from the mid to late 1980s [47]; see Fig. 38.1. The US rate of spine surgery is double the rate of surgery in Canada, Western Europe, and Australia and five times the rate in the United Kingdom [47, 51]. The estimated figure PLSS with severe neuropathic leg greater than back pain would be 1.2 %. Based on census data from 2012, with 314 million people in the United States, the number of Americans living with this subtype of PLSS would be ~3.8 million.

Burton et al. [5] reported a 6–8 % incidence of epidural fibrosis as a cause of persistent pain. Waguespack et al. [35] reported 4 % incidence of epidural fibrosis the primary cause of persistent leg pain; however, the authors noted that 44 % of PLSS patients had some degree of epidural fibrosis noted at reoperation. Slipman et al. [6] reported 8.1 % epidural and 6.5 % intraneural fibrosis. Depalma et al. [36] did not attribute any cases of persistent axial low back pain to epidural fibrosis. Early evaluation of PLSS from the 1970s, when there was a known high incidence of residual lateral stenosis, arachnoiditis, and lack of MR imaging, published reports of up to 24 % of PLSS cases caused by epidural fibrosis [190, 191]. All spine specialists, however, agree that the more surgeries undertaken, the greater likelihood of significant scar tissue formation. Fritsch et al. [2] reviewed 182 surgical cases and reported a primary rate of 4 % of epidural fibrosis after primary diskectomy progressing to a 60 % rate of epidural fibrosis and instability after multiple revisions. Repeat neurolysis or fibrinolysis attempts to remove the scar tissue were largely unsuccessful. The authors concluded that in severe diskotomy syndrome with epidural fibrosis, local arachnoiditis, and progressive instability, fusion provided a better option than multiple fibrinolyses. Revision surgeries performed for peridural fibrosis have unfavorable long-term outcomes [192, 193].

While researching an anti-adhesion barrier gel, Ross et al. [194] investigated the presence of a correlation between recurrent radicular pain 6 months post-diskectomy and the amount of lumbar peridural fibrosis on gadoliniumenhanced MR imaging. The authors reported that a patient with extensive peridural scar is 3.2 times more likely to experience radicular pain (extensive peridural scar was defined as scar >75 % of the affected quadrant). A later study by same group, Petrie and Ross (1996), then reported reduced peridural scaring and improved patient clinical outcomes with use of anti-adhesion barrier gel [195]. As a follow-up to the possibly biased barrier gel studies, Vogelsang et al. (1999) reviewed post-diskectomy MRIs and found only rare cases of fibrosis and no correlation with radicular pain [196].

For the relatively infrequent cases of epidural, perineural, or intraneural fibrosis, we have pathophysiologic data to suggest mechanisms for persistent pain; however, the etiology is likely multifactorial including effects from the primary

disk herniation (both mechanical and chemical effects) and intraoperative surgical factors (tissue manipulation, bleeding, etc). The application of nucleus pulposus (NP) to pig spinal nerves, without compression, reduces nerve conduction velocity [197]. The NP may also damage axons and myelin sheaths, increasing vascular permeability and intravascular coagulation and reducing intraneural blood flow [198]. Once perineural fibrosis exists, it interferes with normal cerebrospinal fluid mediated nutrition, making the nerve roots hyperesthetic and hypersensitive to compression [199]. We also know that the nerve root obtains as much as 50 % of its nutrition through cerebral spinal fluid within the dural cuff [200]. Postsurgical fibrosis of the dural cuff could potentially cause nerve root ischemia and neuropathic pain. Peridural scarring can also tether the nerve roots and dura compromising axoplasmic transport, increasing neural tension, and disrupting arterial and venous blood supply. In a subgroup of PLSS patients with arachnoiditis, defects in fibrinolytic activity with fibrin deposits are described [201]. Scar tissue reduces nerve vascularization, resulting in deafferentation pain [202].

As noted above, the nerve itself does not have to be compressed to become fibrotic. If the nerve becomes ischemic, fibrosis can result. In a cadaveric study of 160 lumbar foramens (range 35–91 years), distention of the venous plexus was more common than direct nerve root compression; furthermore, subjects with direct nerve compression from a disk herniation did not develop nerve fibrosis. In the absence of direct nerve compression, the most severe neural pathology was observed with compression, congestion, and dilatation of foraminal veins. The nerves demonstrated peri- and intraneural fibrosis, edema, and focal demyelination. Inflammatory cells were absent. The authors proposed a mechanism of venous obstruction with ischemia, thereby leading to perineural and intraneural fibrosis.

## 38.12.2.2 Epidural Steroid Injections (ESIs)

Epidural steroid injections (ESIs) are the most commonly used type of spine injection postoperatively in patients with persistent leg pain for both diagnostic and therapeutic purposes. Both negative and positive responses to ESIs have prognostic value. Low-volume, local anestheticonly, selective nerve root blocks can be utilized; however, most spine specialists add corticosteroid for potential temporary partial pain relief. An ESI has critical diagnostic and prognostic value for determining whether a patient has reversible cause of persistent leg pain versus neuropathic pain. A patient with ongoing leg pain without any correctable surgical lesion should have at a minimum one ESI although this is not an acceptable long-term management strategy. Patients diagnosed with persistent neuropathic leg pain are typically referred for a spinal cord stimulator trial if an ESI provides no relief.

In the nonoperated spine, epidural injections are routinely used to manage primarily leg pain associated with lumbar disk herniation or spinal stenosis. Epidural steroids can be delivered via the transforaminal, interlaminar, or caudal routes. General systematic reviews on the subject report moderate evidence for short-term relief and fair evidence for long-term relief. MacVicar et al. [203] performed a recent systematic review of the data on use of transforaminal epidural steroid injections (TFESIs), reporting that 70 % of patients achieve 50 % pain relief at 1 or 2 months and 30 % achieve complete relief. Over the last decade, the prevailing wisdom has been that the transforaminal route is superior to the interlaminar route for treating radicular pain, ostensibly to bring the steroid closer to the herniated disk/ nerve root interface. As such there has been a tremendous rise in utilization with an annual 20.4 % growth rate of TFESIs [204-206] vs. 2 % for caudal and ILESI procedures [205] (Fig. 38.3). However, recently, a systematic review reported equal efficacy for both transforaminal epidural steroid injections (TFESIs) and interlaminar epidural steroid injections (ILESIs) in reducing pain and improving function for unilateral leg pain. There is evidence that a parasagittal ILESI provides comparable pain relief and functional improvement versus TFESIs [207]. Parasagittal ILESIs can achieve 100 % ventral flow versus midline ILESIs with 36 % ventral flow on epidurography. Further research is warranted, given the known serious complications of TFESIs in the lumbar spine which have included paraplegia and permanent neurologic injury [208]. Only a minor 15 % superiority was seen for TFESIs at 2 weeks, but in 1 months and 6 months, outcomes were not statistically significant difference in terms of pain relief, 43 % vs. 54 %. Functional improvement was >50 % in both groups. It is not known if this effect would also be seen in post-lumbar surgery patients wherein interlaminar spine access may be distorted status post laminotomy or laminectomy.

Few studies have looked specifically at postsurgical patients and epidural steroid injections. Klessinger [209] studied TFESIs in patients with persistent or recurrent radicular pain after microdiskectomy. Of 1,009 diskectomy patients, 156 (15.5 %) had persistent radicular pain. Nine percent (91/1009) of patients required reoperation. Patients received a TFI if they have pain persisting >3 months. A positive response was defined as >50 % relief. TFIs were performed with triamcinolone 10 mg and 0.25 % bupivacaine. TFIS achieved at least 50 % reduction in 31 % of these patients. The odds ratio is 17.5. No patient with positive result after TFIS had to undergo a reoperation. Klessinger et al. [70] reported both diagnostic and therapeutic value in performing a TFI in patients with persistent radicular pain >3 months after radiculopathy. Diagnostically, a negative TFI was highly sensitive for recurrent disk herniation (true positive rate 94 %) with a 17.5 odds ratio and the need for reoperation. Therapeutically, 31 % of patients can achieve at least 50 % reduction in pain which makes patient and surgeon less likely to consider a repeat surgery or further invasive procedures. The results also suggest performing the TFI during the first 3 month after the operation for better results (44.4 % good results early after operation versus 22.8 % later, P=0.02).

Derby et al. [210] also reported inferior surgical outcomes in patients undergoing an ESI after versus before 1 year. Derby et al. [210] findings suggest possible fibrotic remodeling over time with persistent pain, thereby making a nerve more difficult to treat the longer a patient's pain is unrelieved. Derby et al. [210] retrospectively studied a group of patients undergoing both primary and repeat lumbar spinal surgery for extremity pain and determined the correlation between>=50 % relief of extremity pain for >1 week following selective epidural block with local anesthetic and corticosteroids to 50 % or greater relief of leg pain at 1 year post-spinal surgery. All patients had temporary relief of their leg pain for the duration of local anesthetic. The study found that when duration of leg pain was less than a year, both steroid responders and nonresponders had good outcomes; however when the duration of leg pain was greater than 1 year, nonresponders had a 5 % favorable outcome compared to 85 % who responded to corticosteroids. Derby et al. concluded that when leg pain was present more than a year in duration, especially in patients with previous surgery and underwhelming structural abnormalities, the chances of leg pain resolution following surgery were unlikely [210].

Manchikanti et al. (2010) evaluated caudal injections in post-lumbar surgery syndrome patients [9]. The RCT compared fluoroscopically guided caudal injections with local anesthetic (LA) versus local anesthetic plus Celestone. Postsurgical patients averaged ~50 years old with a mean of ~13 years of persistent pain. At 1 year, pain relief ( $\geq$ 50 %) and disability reduction were noted in 53 % LA group and 59 % LA plus steroid group, with no statistically significant difference noted. Subjects averaged three to four procedures per year with an average of 38 weeks of pain relief. Functionally subjects improved from moderate to mild disability per ODI. Patients also reported a statistically significant drop in opioid use. Average relief was 4-6 weeks per injection. In terms of pain relief, 70-75 % of patients reported significant pain relief, albeit short-lasting. For a patient with an average of 13 years of persistent postsurgical pain, a ~50-60 % reduction in pain and disability obtained by an average of 38 weeks relief with three to four injections with either LA or LA plus steroids is significant. This study in fact serves to highlight how difficult neuropathic pain is to treat. Given the choice, LA seems superior due to potential negative steroid side effects including adrenal and immune suppression. Yet reimbursement favors steroid injections currently. Local anesthetic, saline, and dextrose may have roles in treating radicular pain which are equal to corticosteroids. Smigel and Reeves [211] presented 1-year outcome data using 10 cc of D5W via fluoroscopically guided caudal injections for chronic low and/or leg pain. In a study of 25 patients, outcomes using D5W at 1 year were noninferior to results obtained using steroids.

Devulder et al. [212] studied patients with persistent leg pain status post diskectomy. Sixty post-lumbar surgery patients were diagnosed with "pronounced epidural fibrosis" around one to two nerve roots, confirmed by MRI, epidurogram, and positive EMGs. Three protocols were compared with transforaminal injections along nerves identified as fibrotic: Group A with 1 ml bupivacaine 0.50 % and 1,500 units of hyaluronidase and 1 ml saline, Group B with 1 ml bupivacaine 0.5 % plus 40 mg methylprednisolone, and Group C with bupivacaine 0.5 %, 1,500 units of hyaluronidase, and 40 mg of methylprednisolone. Patients received a total of two injections. A score of 3 indicated 50-80 % relief; and 4, complete relief. No statistically significant difference was found between solutions at 1 month, 3 months, and 6 months. Injections without corticosteroids were as effective as those without steroids. This study does not mean that steroids and/ or hyaluronidase is ineffective; indeed it may more accurately characterize a group of postoperative patients with "pronounced epidural fibrosis surrounding the nerve root" unfortunately resistant to our best treatments. In this particularly intractable group with persistent radicular chronic pain group, a result of >50 % pain relief at 6 months is clinically and statistically significant.

Clearly, local anesthetic, dextrose 5 % in sterile water (D5W), and normal saline 0.9 % are not placebos; in fact they are all biologically active agents. In regional anesthesia blocks, both local anesthetic and normal saline affect the muscle twitch response; however, the current required for electrostimulation of the femoral or sciatic nerve is greater after injection of normal saline (NS) than D5W [213]. Many older studies assessing efficacy of steroid injections compared steroid to normal saline or local anesthetic (LA) as placebos, which are now understood to not be true placebos. Alternatively, these results suggest that the "active" agents (corticosteroid and hyaluronidase) offered no extra benefit versus local anesthetic and normal saline, perhaps due to a mechanical volume effect of lessening pain in affected fibrotic nerves by disrupting scar tissue such that local blood flow and axoplasmic flow could be improved. A systematic review examined the connection between volume of solution injection and relief of radicular pain during ESI. The authors found that the greater the volume injected, the greater relief achieved [214]. The proposed mechanisms of efficacy for additional volume include: washout of cytokines, lysis of adhesions, suppression of ectopic neural discharges, and enhanced blood flow to ischemic nerve roots.

In regard to radicular pain due to adjacent segment spinal stenosis, no primary studies have been done in this post-lumbar surgery syndrome subgroup. Recently, the efficacy of TFESIs for moderate to severe radicular pain in 400 patients with lumbar spinal stenosis has also been thrown into controversy. Friedly et al. (2014) performed a multicenter RCT [215]. The study was funded by the Agency for Healthcare Research and Quality (AHRQ). Epidural injection of glucocorticoid with lidocaine offered "minimal or no short-term" benefit vs. lidocaine alone. The editorial after the article recommended no ESIs for radicular leg pain and proceeding either with surgery or conservative care. This paper, published in the highly prestigious New England Journal of Medicine and widely publicized, has been criticized by Manchikanti et al. (2014), who stated: "The interventional pain management community believes that there are severe limitations to this study, manuscript, and accompanying editorial. The design, inclusion criteria, outcomes assessment, analysis of data and interpretation, and conclusions of this trial point to the fact that this highly sophisticated and much publicized randomized trial may not be appropriate and lead to misinformation" [216]. Reanalysis of the data shows that epidural local anesthetic with steroids was clearly superior at 3 weeks and potentially superior at 6 weeks.

Although this is a lumbar spine injection chapter, this is a straightforward, potential worthwhile "pearl" with a very low "downside" for physicians treating FBSS with primarily leg pain of all etiologies: strategic use of gabapentin as an adjunct to ESI and for postoperative pain control. For patients undergoing on ESI for FBSS due to epidural fibrosis, a recent study found that the addition of 1,200 mg of gabapentin TID resulted in a statistically significant reduction in pain compared to an alternate regimen and usual ESI care. The surgeon may consider a 1-month trial of gabapentin 1,200 mg TID for 1 month post ESI to potentially obtain significantly greater follow-up at 6 months. The study compared the addition of gabapentin for 1 month to the Group K versus Group G. Group K received 1 month of naproxen sodium 1,100 mg/day BID and 12 mg of tizanidine and vitamin B and C complex for 1 month versus 1 month of gabapentin. The gabapentin group did report increased mild sedation for the 1 month post ESI; otherwise no adverse effects were reported. An interlaminar, fluoroscopically guided ESI at L3-4 was performed in the midline with 80 mg of methylprednisolone. The average age of the group ranged from 42 to 47. The baseline pain in Group K and Group G was equal to VAS 7.70 and 8.05, respectively (p=0.142). At 1-month, 3-month, and 6-month follow-up, Group K showed statistically significant and clinical significant reduction in pain: at 1-month follow-up, Group K and Group G reported VAS scores 1.90 and 0.70 (p = 0.003); at 3-month follow-up, Group K and Group K reported VAS scores 2.50 and 1.54 (p=0.004); and at 6-month follow-up, Group K and Group G reported VAS scores of 5.60 and 2.60, respectively (p < 0.001). Gabapentin is a structural analog of  $\gamma$ -aminobutyric acid and acts on the  $\alpha 2\delta 1$  subunits of voltage-dependent calcium channels. Gabapentin works via primary afferent neurons, dorsal root ganglia, dorsal horn neuron, and supraspinal sites [217]. Gabapentin also decreased movement associated pain in animal models of neuropathic pain and peripheral inflammation [218]. Surgeons may also seriously consider preoperative gabapentin in a single dose of 900 or 1,200 mg administered either pre- or post-incision. In a RCT on patients undergoing laminectomy receiving gabapentin, these required statistically significant less morphine for postop pain control postlaminectomy. Note: the 600 mg dose of gabapentin did not achieve significant pain relief [219]. Theoretically, patients should obtain the same results undergoing microdiskectomy or fusion; however, this has not been studied yet.

#### 38.12.2.3 Lysis of Adhesions

Whether or not fibrosis is a source of pain remains controversial because fibrosis is often seen in both patients with and without continued spinal pain following surgery. Fritsch et al. [2] report epidural fibrosis as the primary cause of FBSS in 4 % of patients; however, its presence was noted in 44 % of FBSS cases. Surgical neurolysis of adhesions has generally shown poor long-term results [2] which prompted the investigation and use of various percutaneous neurolysis procedures via the caudal route, with and without catheter use. Lysis of adhesions has also been used for patients with lumbar spinal stenosis. Various agents have been injected including hyaluronidase, hypertonic saline, local anesthetics, and corticosteroids. Hyaluronidase was observed to markedly reduce post-laminectomy peridural fibrosis in animal studies [220]. However, as noted early, the data are mixed on the subject. Devulder et al. [212] compared transforaminal epidural injections of hyaluronidase, local anesthetic, and corticosteroid. All groups in this RCT showed approximately a 30 % success rate (>50 % pain relief) at 1-, 3-, and 6-month follow-up.

Another neurolysis method is the forceful injection of a volume of fluid through a needle placed into the caudal epidural space. Using this approach, several randomized comparative studies have investigated the efficacy of providing pain relief in postoperative patients with back and leg pain without an obvious structural source other than epidural fibrosis. Using fluoroscopic and contrast verified injections, both Revel et al. [221] and Meadeb et al. [222] study compared the efficacy of the forceful injection of saline with or without added corticosteroids (40 and 20 ml, respectively) versus low-volume corticosteroid alone. At 6 months, Revel et al. [221] reported a success rate of 45 % vs. 19 % for relief of leg pain

(p=0.03) with and 29 % vs. 6 % relief of back pain comparing forceful injection of saline with and without steroids. At 18 months post-injection protocol, forceful injection remained superior showing statistically significant pain reduction of 39 % of patients with sciatica and 31 % of patients with back pain compared to the steroid alone group. The Meadeb study [222] used a smaller volume of 20 ml and did not have enough cases to reach statistical significance, but a modest decrease of 15 % or greater was seen in 47 % of the forceful injection group at 4 months following three injections performed at 1-month interval. No complications were reported.

A more aggressive approach involves fluoroscopically guiding an epidural catheter to the site of fibrosis and using both injected fluid pressure and mechanical disruption by the catheter to lyse adhesions. The caudal approach has been advocated and pioneered by first by Gabor Racz and Holubec [223] and advocated more recently by Manchikanti et al. [224]. Originally Racz followed a 3-day in-hospital protocol using repeat treatments and included in addition to normal saline and corticosteroids the injection of a hypertonic (10 %) saline solution which presumably decreased edema and attenuated small pain fiber activity. In a randomized comparative study to compare the efficacy of hyaluronidase versus hypertonic saline in relieving spinal pain, the Racz group found a 25 % or more reduction in VAS scores in 83 % of the patients at 1 month and a 50 % reduction in VAS scores at 3, 6, 9, and 12 months. No difference in the outcomes was found between groups, although the groups receiving hypertonic saline with or without hyaluronidase required slightly fewer treatments.

Manchikanti et al. [224] performed a RCT in 120 post-lumbar surgery syndrome patients with presumed symptomatic epidural fibrosis. Patients were followed for 2 years. The control group received a fluoroscopically guided caudal injection with catheterization up to S3 with local anesthetic (lidocaine 2 %, 5 mL), nonparticulate betamethasone (6 mg, 1 mL), and 6 mL of 0.9 % sodium chloride solution. The intervention group received percutaneous adhesiolysis of the targeted area, with targeted delivery of lidocaine 2 % (5 mL), 10 % hypertonic sodium chloride solution (6 mL), and nonparticulate betamethasone (6 mg). The intervention group was treated with the Racz catheter, guided to the regions of maximum fibrosis. At 2-year follow-up, 82 % of intervention patients versus 5 % of control patients experienced at least 50 % relief of pain. The intervention group received an average of 6 1-day injections with an overall total relief of approximately 78 out of 114 weeks.

### 38.12.3 Anterior Column

The primary structure in the anterior column responsible for continued or recurrent pain postlumbar surgery is the intervertebral disk. In the unoperated spine the prevalence of pain due to an internally disrupted disk, as evaluated by discography, has been reported as 26 % [29] and 39 % [22]. In the quantitative studies summarized in Table 38.2, the reports of painful internal disk disruption (IDD) and pseudoarthrosis are reported as a group by some authors, and other authors separate pseudoarthrosis from painful IDD at a nonoperated level. Burton et al. [5] reported <5 % pseudoarthrosis; Fritsch reported no cases; Bernard [34] reported 29 % painful IDD; Wageuspack et al. [35] reported 17 % psuedoarthrosis; Slipman reported 22 % painful IDD; and DePalma et al. [36] reported 25 % painful IDD (four cases pseudoarthrosis; three cases painful IDD).

If a patient has significant adjacent segment disease with discogenic pain or pseudoarthrosis, the standard is to extend the fusion and revise the index surgical level, respectively. However, potential new treatments are emerging for adjacent segment discogenic pain, including more effective thermal treatments and the injection of "regenerative" agents such as platelet-rich plasma, and growth factors, and "regenerative solutions" which include dextrose. Much more research is needed in these areas.

Because of nerve ingrowth into the pathologically painful degenerated disks or the postoperative disk, continued pain from the remaining disk tissue may be responsible for ongoing symptoms [225, 226]. While the existence of discogenic pain is now well accepted, the use of discography

as the standard for identifying a painful disk remains controversial. The reader is also referred to a chapter devoted entirely to discography on this topic. The authors of this chapter recommend the current International Spine Intervention Society guidelines [10, 11]. Based on a metaanalysis of all of discography studies, if the International Spine Intervention Society guidelines and operational criteria are adhered to during discography [68], a false-positive rate of less than 10 % is attainable [227]. Regarding the question of whether discography causes disk degeneration [228], a critical review of this study has determined that the study is inconclusive, with significant statistical shortcomings, including the use of an atypical control group [229]. The control group had a much lower prevalence of Modic changes (11 % vs. 39 %) [230] and disk herniations [230, 231] than seen in the general population, which would have created an incorrect conclusion about discography accelerating degeneration or causing disk herniations. Moreover, there is a weak association between the mild degenerative findings seen in the study and the presence of back pain. If discography were indeed as damaging as the authors claim, we would be facing an epidemic of disk herniations after discography, which has not materialized [229].

## 38.12.3.1 Intradiscal Thermal Treatments

If pain following disk injury is in part due to the ingrowth of unmyelinated nerve fibers into the annulus, annular disk and nucleus pulposus may be prone to subsequent sensitization to mechanical and chemical stimuli. Pain reduction may occur by reducing or eliminating nociceptive input by destroying pain-sensitive fibers using heat. However, an early RCT of percutaneous intradiscal heating for 90 s to 70 °C was ineffective versus sham [232]. Helm et al. (2012) performed a systematic review of thermal annular procedures for the disk [233]. Forty-three studies were identified; however, only three RCTs and one observational study met inclusion criteria. The evidence for IDET was fair [234]. The evidence for disctrode was poor. Results of the most recent biacuplasty study are reviewed below, suggesting good evidence thus far.

The first randomized, placebo-controlled study of transdiscal radiofrequency (biacuplasty) was recently completed by Kapural et al. [235]. A total of 1,894 patients were screened; 64 patients met inclusion criteria. Note: patients with prior lumbar surgery were excluded; this treatment may therefore be considered first for adjacent segment IDD. There are two cooled radiofrequency probes in a bipolar array to lesion nociceptors in the posterolateral annulus. The sham treatment was exactly the same except that the RF probes were not placed intradiscal nor was a RF current delivered. At 6 months, the biacuplasty group showed statistically significant reduction in pain (p=0.006), physical disability (p=0.029), and (p=0.037).disability Biacuplasty patients reported a trend toward decreased opioid use with a 16 mg reduction in opioid use. Other thermal treatments are discussed elsewhere in the book.

#### 38.12.3.2 Intradiscal Injections

Intradiscal steroids have been shown to be ineffective for discogenic pain in a prospective, double-blind trial and have been abandoned [236]. A pilot study performed intradiscal injections with a solution containing agents thought to induce proteoglycan synthesis including: glucosamine, chondroitin, hypertonic dextrose and dimethylsulfoxide (DMSO) [237]. This pilot study was compromised of 30 patients with an average age of 45 years with intractable low back pain of an average 8.5 years duration [237]. On follow-up statistically significant improvement was found in pain, RMDQ. Fifty-seven percent of patients 17/30 improved markedly with 72 % reduction in disability scores and 76 % improvement in VAS scores. The remaining 43 % ([13] patients) had little or no improvement. Patients who did poorly had FBSS, spinal stenosis, and long-term disability. Further research is ongoing on this intradiscal therapy. This could be considered as a therapy in an adjacent pain disk to the surgical construct.

As noted above, the current standard of care for painful IDD or pseudoarthrosis is surgical intervention; however, there is emerging interest in intradiscal injections. On review of the latest business sites in treatment of spinal pain, there are 12 start-up companies working on intradiscal injections: including platelet-rich plasma/ growth factors, stem cells, growth factors, etc. [238]. Terry et al. [239] presented the first human study on PRP in the disk for discogenic pain as diagnosed by a positive discogram. PRP is currently being used with success for various tendinopathies, hamstring/rotator cuff tears, as well as osteoarthritis. PRP is made by concentrating the patients' platelets. Typically, the ideal concentration is 4-6× baseline so as to exceed a cell count of one million. Platelets house multiple anabolic growth factors that include: PDGF, FGF, IL-1, IL-2, IL-8, and VEGF. Inclusion criteria included pain present greater than 3 months, greater than 50 % disk height preserved on MRI, no grade IV tears, and a positive discogram. A total of 41 patients were initially randomized and 11 were excluded. A 2:1 patient recruitment strategy was utilized; thus a total of 19 patients received PRP and 11 patients received contrast as a placebo. At 1 year, the PRP group has a statistically significant decrease in pain, including current pain, best pain, and worst pain. Patients receiving PRP also reported superior satisfaction per the NASS survey. Other researchers believe that the growth factors released with PRP alone or as single growth factors alone cannot be effective because of the short half-life. Some researchers are injecting transfected intervertebral disk cells with genes encoding the active proteins into the disk. Researchers are also implanting stem cells with PRP or in biologic scaffolds to restore disk structure and function [240–242].

### Conclusion

- Persistent lumbar and radicular pain after lumbar surgery can best be termed post-lumbar surgery syndrome (PLSS) which is defined as a cluster of nomenclature and syndromes following spine surgery wherein the expectations of the patient and the spine surgeon are not met with persistent pain following lumbar surgery.
- Failed back surgery syndrome (FBSS) 30 years ago was largely non-diagnostic of major etiologies of persistent pain after lumbar surgery.

With more rigorous surgical indications, advanced imaging, and use of diagnostic spine injections, 70–100 % of causes of persistent pain have been elucidated. A more acceptable current definition of FBSS is: persistent or recurrent pain, mainly in the region of the lower back and legs, even after technically, anatomically successful lumbosacral spine surgeries.

- Chronic low back pain is neither "nonspecific" nor "idiopathic;" the causes of chronic low back pain can be determined in the majority of patients with history, physical exam, imaging, and diagnostic/therapeutic spine injections.
- Spine surgery outcomes and rates are closely monitored by payors and policymakers. More stringent definitions for success across all orthopedic surgery disciplines are expected and should include pain relief, functional assessment, medication usage, return to activities, and return to work.
- Surgical outcomes are negatively affected by obesity (BMI >30), type 2 diabetes mellitus, and depression. Studies have not borne out psychiatric pathology as the primary reason for persistent pain after lumbar surgery. However, if patients have significant pre- or postoperative co-morbid psychopathology, their psychiatric wellbeing should still be addressed as part of a multidisciplinary treatment approach.
- Emerging research shows that chronic pain patients and patients who take opiates may suffer from low vitamin D, low testosterone, low estrogen, osteopenia/osteoporosis, and other medical comorbidities. In addition to addressing smoking and NSAID use prior to surgery, optimizing treatment for medical comorbidities with bioidentical hormones may further improve surgical outcomes.
- The most common causes of persistent axial low back pain after fusion (as diagnosed by controlled injections) are: facet arthropathy, painful internal disk disruption, sacroiliac joint pain, hardware irritation/soft tissue pain, and adjacent segment disease.
- The most common causes of persistent leg pain after lumbar surgery are: recurrent/residual disk herniation or new herniation, neuropathic pain, and epidural fibrosis.
- Conscious sedation (including agents such as propofol, midazolam, and fentanyl) should

not be used because of the statistically significant probability of obtaining a false-positive diagnostic block. To properly diagnose facetmediated or sacroiliac joint pain, dual comparative anesthetic blocks are needed. Patients with false-positive results are likely to be subject to misdiagnoses and unnecessary procedures. Patients receiving blocks with sedation reported statistically significant greater mean reductions in pain diary scores and less procedure-related pain.

 If spine interventionalists adopt less stringent diagnostic criteria or surgeons use less rigorous surgical indications, outcomes suffer. Poor outcomes negatively affect the patient, spine interventionalist, and surgeon and make payors and policymakers reluctant to pay for treatment.

### References

- 1. Long DM. Failed back surgery syndrome. Neurosurg Clin N Am. 1991;2(4):899–919.
- Fritsch EW, Heisel J, Rupp S. The failed back surgery syndrome: reasons, intraoperative findings, and longterm results: a report of 182 operative treatments. Spine Affiliated Soc Meet Abstr. 1996;21(5):626–33.
- 3. Onesti ST. Failed back syndrome. Neurologist. 2004;10(5):259–64.
- Bogduk N. The rationale for patterns of neck and back pain. Patient Manag. 1984;13:17–28.
- Burton CV, Kirkaldy-Willis WH, Yong-Hing K, Heithoff KB. Causes of failure of surgery on the lumbar spine. Clin Orthop Relat Res. 1981;157:191–9.
- Slipman CW, Shin CH, Patel RK, Isaac Z, Huston CW, Lipetz JS, et al. Etiologies of failed back surgery syndrome. Pain Med. 2002;3(3):200–14.
- Leveque JC, Villavicencio AT, Bulsara KR, Rubin L, Gorecki JP. Spinal cord stimulation for failed back surgery syndrome. Neuromodulation Technol Neural Interface. 2001;4(1):1–9.
- Manchikanti L, Singh V, Cash KA, Pampati V, Datta S. Fluoroscopic caudal epidural injections in managing post lumbar surgery syndrome: two-year results of a randomized, double-blind, active-control trial. Int J Med Sci. 2012;9(7):582–91.
- Manchikanti L, Singh V, Cash KA, Pampati V, Datta S. Management of pain of post lumbar surgery syndrome: one-year results of a randomized, doubleblind, active controlled trial of fluoroscopic caudal epidural injections. Pain Phys Off J Assoc Pain Manag Anesthesiol. 2010;13(6):509–21.
- Bogduk N. Practice guidelines for spinal diagnostic and treatment prodedures. San Francisco: International Spine Intervention Society; 2004.

- Bogduk N. Practice guidelines for spinal diagnostic and treatment procedures. 2nd ed. San Francisco: International Spine Intervention Society (ISIS); 2013.
- Finneson BE, Cooper VR. A lumbar disc surgery predictive score card. A retrospective evaluation. Spine (Phila Pa 1976). 1979;4(2):141–4.
- Finneson BE. A lumbar disc surgery predictive score card. Spine (Phila Pa 1976). 1978;3(2):186–8.
- Hoy D, March L, Brooks P, Woolf A, Blyth F, Vos T, et al. Measuring the global burden of low back pain. Best Pract Res Clin Rheumatol. 2010;24(2):155–65.
- Luo X, Pietrobon R, Sun SX, Liu GG, Hey L. Estimates and patterns of direct health care expenditures among individuals with back pain in the United States. Spine (Phila Pa 1976). 2004;29(1): 79–86.
- Manchikanti L, Singh V, Datta S, Cohen SP, Hirsch JA, American Society of Interventional Pain P. Comprehensive review of epidemiology, scope, and impact of spinal pain. Pain Phys Off J Assoc Pain Manag Anesthesiol. 2009;12(4):E35–70.
- Freburger JK, Holmes GM, Agans RP, Jackman AM, Darter JD, Wallace AS, et al. The rising prevalence of chronic low back pain. Arch Intern Med. 2009;169(3):251–8.
- Martin BI, Turner JA, Mirza SK, Lee MJ, Comstock BA, Deyo RA. Trends in health care expenditures, utilization, and health status among US adults with spine problems, 1997–2006. Spine Affiliated Soc Meet Abstr. 2009;34(19):2077–84.
- Dillane JB, Fry J, Kalton G. Acute back syndrome-a study from general practice. Br Med J. 1966;2(5505): 82–4.
- White AA. Introduction to symposium of idiopathic low back pain. In: Augustus A, White I, Stephen G, editors. American Academy of Orthopedic Surgeons. St. Louis: CV Mosby Company; 1982.
- Derby R. Diagnostic block procedures: use in pain localization. In: White AH, editor. Spine: state of the art reviews, failed back surgery syndrome. Philadelphia: Hanley & Belfus, Inc.; 1986. p. 47–64.
- Schwarzer AC, Aprill CN, Derby R, Fortin J, Kine G, Bogduk N. The prevalence and clinical features of internal disc disruption in patients with chronic low back pain. Spine. 1995;20(17):1878–83.
- Schwarzer AC, Aprill CN, Derby R, Fortin J, Kine G, Bogduk N. The relative contributions of the disc and zygapophyseal joint in chronic low back pain. Spine. 1994;19(7):801–6.
- 24. Schwarzer AC, Aprill CN, Derby R, Fortin J, Kine G, Bogduk N. Clinical features of patients with pain stemming from the lumbar zygapophysial joints. Is the lumbar facet syndrome a clinical entity? Spine. 1994;19(10):1132–7.
- Schwarzer AC, Derby R, Aprill CN, Fortin J, Kine G, Bogduk N. The value of the provocation response in lumbar zygapophyseal joint injections. Clin J Pain. 1994;10(4):309–13.
- Schwarzer AC, Wang SC, Bogduk N, McNaught PJ, Laurent R. Prevalence and clinical features of lumbar

zygapophysial joint pain: a study in an Australian population with chronic low back pain. Ann Rheum Dis. 1995;54(2):100–6.

- Schwarzer AC, Aprill CN, Bogduk N. The sacroiliac joint in chronic low back pain. Spine (Phila Pa 1976). 1995;20(1):31–7.
- Bogduk N. Musculoskeletal pain: toward precision diagnosis. Progress in pain research and management. In: Jensen TS, Turner JA, Wiesenfeld-Hallin Z, editors. Proceedings of the 8th World Congress on Pain. Seattle: IASP Press; 1997. p. 507–25.
- Manchikanti L, Singh V, Pampati V, Damron KS, Barnhill RC, Beyer C, et al. Evaluation of the relative contributions of various structures in chronic low back pain. Pain Physician. 2001;4(4):308–16.
- Troup JD. Relation of lumbar spine disorders to heavy manual work and lifting. Lancet. 1965;1(7390):857–61.
- Rowe ML. Low back pain in industry. A position paper. JOM J Occup Med. 1969;11(4):161–9.
- Holt EP. The question of lumbar discography. J Bone Joint Surg Am Vol. 1968;50(4):720–6.
- Long DM, Filtzer DL, BenDebba M, Hendler NH. Clinical features of the failed-back syndrome. J Neurosurg. 1988;69(1):61–71.
- Bernard Jr TN. Repeat lumbar spine surgery. Factors influencing outcome. Spine Affiliated Soc Meet Abstr. 1993;18(15):2196–200.
- Waguespack A, Schofferman J, Slosar P, Reynolds J. Etiology of long-term failures of lumbar spine surgery. Pain Med. 2002;3(1):18–22.
- DePalma MJ, Ketchum JM, Saullo TR. Etiology of chronic low back pain in patients having undergone lumbar fusion. Pain Med. 2011;12(5):732–9.
- Future of orthopaedics, strategic forecast for a service line under siege. The Health Care Advisory Board, Washington, DC. Available at: www.advisory.com. 2003.
- van Wijk RM, Geurts JW, Wynne HJ. Long-lasting analgesic effect of radiofrequency treatment of the lumbosacral dorsal root ganglion. J Neurosurg. 2001;94(2 Suppl):227–31.
- North RB, Campbell JN, James CS, Conover-Walker MK, Wang H, Piantadosi S, et al. Failed back surgery syndrome: 5-year follow-up in 102 patients undergoing repeated operation. Neurosurgery. 1991;28(5):685–90.
- Fortin JD, Kissling RO, O'Connor BL, Vilensky JA. Sacroiliac joint innervation and pain. Am J Orthop. 1999;28(12):687–90.
- 41. National Hospital Discharge Survey: Annual Summary, 1997. In: U.S. Department of Health and Human Services CfDCaPNCfHS, editor. Vital and health statistics from the Centers for Disease Control and Prevention/National Center for Health Statistics. Hyattsville: DHHS Publication No. (PHS) 2000-1715; 1999 Dec.
- The Worldwide Orthopaedic Market 2001–2002, Orthopaedic Knowledge Enterprises, Inc. Dorland's biomedical information. Available at: www.orthoworld. com. 2002.
- The Edge; Orthopaedics Briefing 2003. SG- 2 LLC, Evanston. Available at: www.sg-2.com. 2003.

- Lieberman IH. Disc bulge bubble: spine economics 101. Spine J. 2004;4(6):609–13.
- 45. 2000 Spinal Surgery Update. Orthopaedic Network News October. 2000;11(4): Available at: http://www. orthopedicnetworknews.com.
- 46. NASS preview—now this is spinal tap. Merrill Lynch Global Securities Research and Economics Group. 2003 Oct.
- Cherkin DC, Deyo RA, Loeser JD, Bush T, Waddell G. An international comparison of back surgery rates. Spine. 1994;19(11):1201–6.
- Deyo RA. Back surgery–who needs it? N Engl J Med. 2007;356(22):2239–43.
- Deyo RA, Gray DT, Kreuter W, Mirza S, Martin BI. United States trends in lumbar fusion surgery for degenerative conditions. Spine. 2005;30(12):1441–5.
- Deyo RA, Mirza SK. The case for restraint in spinal surgery: does quality management have a role to play? Eur Spine J. 2009;18:331–7.
- 51. Lee J. Rethinking health care: some healthcare systems are moving beyond surgery to help patients. Available at: http://www.modernhealthcare.com/ article/20140322/MAGAZINE/303229985 Accessed 7 Sept 2014. Modern Healthcare; 22 Mar 2014.
- Juratli SM, Mirza SK, Fulton-Kehoe D, Wickizer TM, Franklin GM. Mortality after lumbar fusion surgery. Spine Affiliated Soc Meet Abstr. 2009;34(7):740–7.
- 53. Dyrda L. Young spine surgeons entering the field: 5 key trends, Available at: http://www.beckersspine. com/spine/item/21135-young-spine-surgeonsentering-the-field-5-key-trends. Accessed 3 Sept 2014. Becker's Spine Review; 11 Jun 2014.
- 54. Dyrda L. 5 key trends in the spinal fusion market, Available at: http://www.beckersspine.com/spine/ item/21138-5-key-trends-in-the-spinal-fusionmarket. Accessed 27 Aug 2014. Becker's Spine Review; 12 Jun 2014.
- 55. Manchikanti L, Falco FJ, Singh V, Pampati V, Parr AT, Benyamin RM, et al. Utilization of interventional techniques in managing chronic pain in the Medicare population: analysis of growth patterns from 2000 to 2011. Pain Phys Off J Assoc Pain Manag Anesthesiol. 2012;15(6):E969–82.
- 56. Manchikanti L, Pampati V, Singh V, Falco FJ. Assessment of the escalating growth of facet joint interventions in the medicare population in the United States from 2000 to 2011. Pain Phys Off J Assoc Pain Manag Anesthesiol. 2013;16(4):E365–78.
- 57. Levinson DR. US Department of Health and Human Services, Office of Inspector General: Medicare Payments for Facet Joint Injection Services, OEI-05-07-00200. Available at: http://oig.hhs.gov/oei/ reports/oei-05-07-00200.pdf. Sept 2008.
- Asch HL, Lewis PJ, Moreland DB, Egnatchik JG, Yu YJ, Clabeaux DE, et al. Prospective multiple outcomes study of outpatient lumbar microdiscectomy: should 75 to 80% success rates be the norm? J Neurosurg. 2002;96(1 Suppl):34–44.
- Mirza SK, Deyo RA, Heagerty PJ, Turner JA, Martin BI, Comstock BA. One-year outcomes of surgical versus nonsurgical treatments for discogenic back

pain: a community-based prospective cohort study. Spine J. 2013;13(11):1421–33.

- 60. Santiago-Dieppa D, Bydon M, Xu R, De la Garza-Ramos R, Henry R, Sciubba DM, et al. Long-term outcomes after non-instrumented lumbar arthrodesis. J Clin Neurosci. 2014;21(8):1393–7.
- 61. Fritzell P, Hägg O, Wessberg P, Nordwall A, Swedish Lumbar Spine Study G. 2001 Volvo Award Winner in clinical studies: lumbar fusion versus nonsurgical treatment for chronic low back pain: a multicenter randomized controlled trial from the Swedish Lumbar Spine Study Group. Spine. 2001;26(23):2521–32.
- Carragee EJ, Lincoln T, Parmar VS, Alamin T. A gold standard evaluation of the "discogenic pain" diagnosis as determined by provocative discography. Spine. 2006;31(18):2115–23.
- Carragee EJ, Cheng I. Minimum acceptable outcomes after lumbar spinal fusion. Spine J. 2010;10(4):313–20.
- Davidson M, Keating J. Oswestry disability questionnaire (ODQ). Aust J Physiother. 2005;51(4):270.
- 65. Phillips FM, Slosar PJ, Youssef JA, Andersson G, Papatheofanis F. Lumbar spine fusion for chronic low back pain due to degenerative disc disease: a systematic review. Spine. 2013;38(7):E409–22.
- 66. Brox JI, Sørensen R, Friis A, Nygaard Ø, Indahl A, Keller A, et al. Randomized clinical trial of lumbar instrumented fusion and cognitive intervention and exercises in patients with chronic low back pain and disc degeneration. Spine. 2003;28(17):1913–21.
- 67. Brox JI, Reikerås O, Nygaard Ø, Sørensen R, Indahl A, Holm I, et al. Lumbar instrumented fusion compared with cognitive intervention and exercises in patients with chronic back pain after previous surgery for disc herniation: a prospective randomized controlled study. Pain. 2006;122(1–2):145–55.
- Brox JI, Nygaard ØP, Holm I, Keller A, Ingebrigtsen T, Reikerås O. Four-year follow-up of surgical versus non-surgical therapy for chronic low back pain. Ann Rheum Dis. 2010;69(9):1643–8.
- Ebeling U, Reichenberg W, Reulen HJ. Results of microsurgical lumbar discectomy. Review on 485 patients. Acta Neurochir. 1986;81(1–2):45–52.
- Klessinger S. Zygapophysial joint pain in post lumbar surgery syndrome. The efficacy of medial branch blocks and radiofrequency neurotomy. Pain Med. 2013;14(3):374–7.
- McGirt MJ, Ambrossi GL, Datoo G, Sciubba DM, Witham TF, Wolinsky JP, et al. Recurrent disc herniation and long-term back pain after primary lumbar discectomy: review of outcomes reported for limited versus aggressive disc removal. Neurosurgery. 2009;64(2):338–44.
- Shafaie FF, Bundschul C, Jinkins JR. The posttherapeutic lumbosacral spine. In: Jenkins JR, editor. Postherapeutic neurodiagnostic imaging. Philadelphia: Lippincott-Raven; 1997. p. 223–43.
- Van Goethem JW, Van de Kelft E, Biltjes IG, van Hasselt BA, van den Hauwe L, Parizel PM, et al. MRI after successful lumbar discectomy. Neuroradiology. 1996;38:S90–6.

- 74. Lurie JD, Tosteson TD, Tosteson AN, Zhao W, Morgan TS, Abdu WA, et al. Surgical versus nonoperative treatment for lumbar disc herniation: eightyear results for the spine patient outcomes research trial. Spine. 2014;39(1):3–16.
- Turner JA, Ersek M, Herron L, Haselkorn J, Kent D, Ciol MA, et al. Patient outcomes after lumbar spinal fusions. JAMA J Am Med Assoc. 1992;268(7): 907–11.
- Martin BI, Mirza SK, Comstock BA, Gray DT, Kreuter W, Deyo RA. Reoperation rates following lumbar spine surgery and the influence of spinal fusion procedures. Spine Affiliated Soc Meet Abstr. 2007;32(3):382–7.
- Hazard RG. Failed back surgery syndrome: surgical and nonsurgical approaches. Clin Orthop Relat Res. 2006;443:228–32.
- 78. Fairbank J, Frost H, Wilson-MacDonald J, Yu LM, Barker K, Collins R, et al. Randomised controlled trial to compare surgical stabilisation of the lumbar spine with an intensive rehabilitation programme for patients with chronic low back pain: the MRC spine stabilisation trial. BMJ (BMJcom). 2005;330(7502): 1233.
- Mannion AF, Brox JI, Fairbank JC. Comparison of spinal fusion and nonoperative treatment in patients with chronic low back pain: long-term follow-up of three randomized controlled trials. Spine J. 2013;13(11):1438–48.
- 80. Zigler JE, Delamarter RB. Five-year results of the prospective, randomized, multicenter, Food and Drug Administration investigational device exemption study of the ProDisc-L total disc replacement versus circumferential arthrodesis for the treatment of single-level degenerative disc disease. J Neurosurg Spine. 2012;17(6):493–501.
- Guyer RD, Pettine K, Roh JS, Dimmig TA, Coric D, McAfee PC, et al. Comparison of 2 lumbar total disc replacements: results of a prospective, randomized, controlled, multicenter Food and Drug Administration trial with 24-month follow-up. Spine. 2014;39(12):925–31.
- Crock HV. Observations on the management of failed spinal operations. J Bone Joint Surg Br Vol (JBJS BR). 1976;58(2):193–9.
- Kostuik JP. Chapter 18: The Surgical Treatment of Failures of Laminectomy and Spinal Fusion. Spinal Instability. Holtzman RNN, et al. (eds.). Springer-Verlag New York, Inc. 1993. p. 407–88.
- 84. Albert HB, Sorensen JS, Christensen BS, Manniche C. Antibiotic treatment in patients with chronic low back pain and vertebral bone edema (Modic type 1 changes): a double-blind randomized clinical controlled trial of efficacy. Eur Spine J. 2013;22(4):697–707.
- 85. Fiume D, Sherkat S, Callovini GM, Parziale G, Gazzeri G. Treatment of the failed back surgery syndrome due to lumbo-sacral epidural fibrosis. Acta Neurochir Suppl. 1995;64:116–8.
- North RB, Ewend MG, Lawton MT, Kidd DH, Piantadosi S. Failed back surgery syndrome: 5-year follow-up after spinal cord stimulator implantation. Neurosurgery. 1991;28(5):692–9.

- Soldati E. National Italian Register of implantable systems for spinal cord stimulation (SCS): analysis of preliminary data. Neuromodulation Technol Neural Interface. 2002;5(1):7–15.
- Revel M, Poiraudeau S, Auleley GR, Payan C, Denke A, Nguyen M, et al. Capacity of the clinical picture to characterize low back pain relieved by facet joint anesthesia. Proposed criteria to identify patients with painful facet joints. Spine. 1998;23(18):1972–6.
- 89. Manchikanti L, Manchukonda R, Pampati V, Damron KS, McManus CD. Prevalence of facet joint pain in chronic low back pain in postsurgical patients by controlled comparative local anesthetic blocks. Arch Phys Med Rehabil. 2007;88(4):449–55.
- Fortin JD, Dwyer AP, West S, Pier J. Sacroiliac joint: pain referral maps upon applying a new injection/ arthrography technique. Part I: asymptomatic volunteers. Spine (Phila Pa 1976). 1994;19(13):1475–82.
- 91. Cauley JA. Estrogen and bone health in men and women. Steroids. 2015;99:11–5.
- 92. Moran JM, Lavado-Garcia JM, Roncero-Martin R, Pedrera-Canal M, Vera V, Fernandez P, et al. Testosterone levels and bone mineral density in healthy elderly men. J Am Geriatr Soc. 2015;63(1): 206–7.
- Huhtaniemi IT. Andropause–lessons from the European Male Ageing Study. Ann Endocrinol (Paris). 2014;75(2):128–31.
- Koerner JD, Glaser J, Radcliff K. Which variables are associated with patient-reported outcomes after discectomy? Review of SPORT Disc Herniation Studies. Clin Orthop Relat Res. 2015;473:2000–6.
- Plotnikoff GA, Quigley JM. Prevalence of severe hypovitaminosis D in patients with persistent, nonspecific musculoskeletal pain. Mayo Clin Proc. 2003;78(12):1463–70.
- Al Faraj S, Al Mutairi K. Vitamin D deficiency and chronic low back pain in Saudi Arabia. Spine (Phila Pa 1976). 2003;28(2):177–9.
- 97. Johansen JV, Manniche C, Kjaer P. Vitamin D levels appear to be normal in Danish patients attending secondary care for low back pain and a weak positive correlation between serum level Vitamin D and Modic changes was demonstrated: a cross-sectional cohort study of consecutive patients with nonspecific low back pain. BMC Musculoskelet Disord. 2013;14:78.
- Khazai N, Judd SE, Tangpricha V. Calcium and vitamin D: skeletal and extraskeletal health. Curr Rheumatol Rep. 2008;10(2):110–7.
- Bischoff-Ferrari HA. Optimal serum 25-hydroxyvitamin D levels for multiple health outcomes. Adv Exp Med Biol. 2014;810:500–25.
- 100. Valverde-Filho J, Cunha Neto MB, Fonoff ET, Meirelles ES, Teixeira MJ. Chronic spinal and oral morphine-induced neuroendocrine and metabolic changes in noncancer pain patients. Pain Med. 2015;16:715–25.
- 101. Golomb BA, Evans MA. Statin adverse effects: a review of the literature and evidence for a mitochon-

drial mechanism. Am J Cardiovasc Drugs. 2008;8(6):373–418.

- 102. Hsieh PH, Chang Y, Chen DW, Lee MS, Shih HN, Ueng SW. Pain distribution and response to total hip arthroplasty: a prospective observational study in 113 patients with end-stage hip disease. J Orthop Sci. 2012;17(3):213–8.
- 103. Arnold DR, Keene JS, Blankenbaker DG, Desmet AA. Hip pain referral patterns in patients with labral tears: analysis based on intra-articular anesthetic injections, hip arthroscopy, and a new pain "circle" diagram. Phys Sportsmed. 2011;39(1):29–35.
- Laslett M, Aprill CN, McDonald B, Young SB. Diagnosis of sacroiliac joint pain: validity of individual provocation tests and composites of tests. Man Ther. 2005;10(3):207–18.
- 105. van der Wurff P, Buijs EJ, Groen GJ. A multitest regimen of pain provocation tests as an aid to reduce unnecessary minimally invasive sacroiliac joint procedures. Arch Phys Med Rehabil. 2006;87(1):10–4.
- Laslett M. Evidence-based diagnosis and treatment of the painful sacroiliac joint. J Man Manip Ther. 2008;16(3):142–52.
- 107. Herrera Herrera I, Moreno de la Presa R, González Gutiérrez R, Bárcena Ruiz E, García Benassi JM. Evaluation of the postoperative lumbar spine. Radiologia. 2013;55(1):12–23.
- Berquist TH. Imaging of the postoperative spine. Radiol Clin North Am. 2006;44(3):407–18.
- Glickstein MF, Sussman SK. Time-dependent scar enhancement in magnetic resonance imaging of the postoperative lumbar spine. Skeletal Radiol. 1991;20(5):333–7.
- 110. Grane P, Tullberg T, Rydberg J, Lindgren L. Postoperative lumbar MR imaging with contrast enhancement. Comparison between symptomatic and asymptomatic patients. Acta Radiol. 1996;37(3 Pt 1):366–72.
- 111. Lattig F, Fekete TF, Grob D, Kleinstück FS, Jeszenszky D, Mannion AF. Lumbar facet joint effusion in MRI: a sign of instability in degenerative spondylolisthesis? Eur Spine J. 2012;21(2):276–81.
- 112. Jensen OK, Nielsen CV, Sørensen JS, Stengaard-Pedersen K. Type 1 Modic changes was a significant risk factor for 1-year outcome in sick-listed low back pain patients: a nested cohort study using magnetic resonance imaging of the lumbar spine. Spine J. 2014;14(11):2568–81.
- Gates GF, McDonald RJ. Bone SPECT of the back after lumbar surgery. Clin Nucl Med. 1999;24(6): 395–403.
- 114. Ianuzzi A, Little JS, Chiu JB, Baitner A, Kawchuk G, Khalsa PS. Human lumbar facet joint capsule strains: I. During physiological motions. Spine J. 2004;4(2):141–52.
- 115. Little JS, Ianuzzi A, Chiu JB, Baitner A, Khalsa PS. Human lumbar facet joint capsule strains: II. Alteration of strains subsequent to anterior interbody fixation. Spine J. 2004;4(2):153–62.
- 116. Manchukonda R, Manchikanti KN, Cash KA, Pampati V, Manchikanti L. Facet joint pain in

chronic spinal pain: an evaluation of prevalence and false-positive rate of diagnostic blocks. J Spinal Disord Tech. 2007;20(7):539–45.

- 117. Manchikanti L, Pampati V, Fellows B, Bakhit CE. Prevalence of lumbar facet joint pain in chronic low back pain. Pain Physician. 1999;2(3):59–64.
- 118. Manchikanti L, Pampati V, Fellows B, Bakhit CE. The diagnostic validity and therapeutic value of lumbar facet joint nerve blocks with or without adjuvant agents. Curr Rev Pain. 2000;4(5):337–44.
- 119. Manchikanti L, Boswell MV, Singh V, Pampati V, Damron KS, Beyer CD. Prevalence of facet joint pain in chronic spinal pain of cervical, thoracic, and lumbar regions. BMC Musculoskelet Disord. 2004;5:15.
- 120. Siepe CJ, Korge A, Grochulla F, Mehren C, Mayer HM. Analysis of post-operative pain patterns following total lumbar disc replacement: results from fluoroscopically guided spine infiltrations. Eur Spine J. 2008;17(1):44–56.
- 121. Manchikanti L, Pampati V, Fellows B, Baha AG. The inability of the clinical picture to characterize pain from facet joints. Pain Phys Off J Assoc Pain Manag Anesthesiol. 2000;3(2):158–66.
- 122. Dreyfuss P, Schwarzer AC, Lau P, Bogduk N. Specificity of lumbar medial branch and L5 dorsal ramus blocks. A computed tomography study. Spine (Phila Pa 1976). 1997;22(8):895–902.
- 123. Neilens H, Van Zundert J, Mairiaux P, Gailly J, Van Den Hecke N, Mazina D, et al. Chronic low back pain. Good Clinical practice (GCP). Brussels: Belgian Health Care Knowledge Centre (KCE); 2006. KCE reports 48 C (D/2006/10.273/71).
- 124. Airaksinen O, Brox JI, Cedraschi C, Hildebrandt J, Klaber-Moffett J, Kovacs F, et al. Chapter 4. European guidelines for the management of chronic nonspecific low back pain. Eur Spine J. 2006;15 Suppl 2:S192–300.
- 125. (NICE) NIfHaCE. Low back pain: early management of persistent non-specific low back pain. NICE clinical guideline 88. May 2009.
- 126. Chou R, Loeser JD, Owens DK, Rosenquist RW, Atlas SJ, Baisden J, et al. Interventional therapies, surgery, and interdisciplinary rehabilitation for low back pain: an evidence-based clinical practice guideline from the American Pain Society. Spine. 2009;34(10):1066–77.
- 127. Niemistö L, Kalso E, Malmivaara A, Seitsalo S, Hurri H, Cochrane Collaboration Back Review G. Radiofrequency denervation for neck and back pain: a systematic review within the framework of the cochrane collaboration back review group. Spine. 2003;28(16):1877–88.
- 128. Niemisto L, Kalso E, Malmivaara A, Seitsalo S, Hurri H. Radiofrequency denervation for neck and back pain. A systematic review of randomized controlled trials. Cochrane Database Syst Rev (CDSR). 2003;1, CD004058.
- 129. Chou R, Atlas SJ, Stanos SP, Rosenquist RW. Nonsurgical interventional therapies for low back pain: a review of the evidence for an American

Pain Society clinical practice guideline. Spine (Phila Pa 1976). 2009;34(10):1078–93.

- 130. Geurts JW, van Wijk RM, Stolker RJ, Groen GJ. Efficacy of radiofrequency procedures for the treatment of spinal pain: a systematic review of randomized clinical trials. Reg Anesth Pain Med. 2001;26(5):394–400.
- 131. Slipman CW, Bhat AL, Gilchrist RV, Issac Z, Chou L, Lenrow DA. A critical review of the evidence for the use of zygapophysial injections and radiofrequency denervation in the treatment of low back pain. Spine J. 2003;3(4):310–6.
- 132. Resnick DK, Choudhri TF, Dailey AT, Groff MW, Khoo L, Matz PG, et al. Guidelines for the performance of fusion procedures for degenerative disease of the lumbar spine. Part 13: injection therapies, lowback pain, and lumbar fusion. J Neurosurg Spine. 2005;2(6):707–15.
- 133. Shealy CN. Technique for percutaneous spinal facet rhizotomy. Burlington: Radionics; 1974.
- 134. Bogduk N. Practice guidelines for spinal diagnostic and treatment procedures, 2nd edition. In: Bogduk N, editor. Efficacy of lumbar medial branch thermal radiofrequency neurotomy. San Francisco: International Spine Intervention Society; 2013. p. 631–41.
- 135. Mehta M, Sluijter ME. The treatment of chronic back pain. A preliminary survey of the effect of radiofrequency denervation of the posterior vertebral joints. Anaesthesia. 1979;34(8):768–75.
- 136. Sluijter ME, Mehta M. Treatment of chronic back and neck pain by percutaneous thermal lesions. In: Lipton S, Miles J, editors. Persistent pain. Modern methods of treatment, vol. 3. London: Academic; 1981. p. 141–79.
- 137. van Wijk RM, Geurts JW, Wynne HJ, Hammink E, Buskens E, Lousberg R, et al. Radiofrequency denervation of lumbar facet joints in the treatment of chronic low back pain: a randomized, double-blind, sham lesion-controlled trial. Clin J Pain. 2005;21(4):335–44.
- 138. Bogduk N, Long DM. The anatomy of the so-called "articular nerves" and their relationship to facet denervation in the treatment of low-back pain. J Neurosurg. 1979;51(2):172–7.
- Bogduk N, Long DM. Percutaneous lumbar medial branch neurotomy: a modification of facet denervation. Spine (Phila Pa 1976). 1980;5(2):193–200.
- Bogduk N, Macintosh J, Marsland A. Technical limitations to the efficacy of radiofrequency neurotomy for spinal pain. Neurosurgery. 1987;20(4):529–35.
- 141. Derby R, Melnik I, Lee JE, Lee SH. Correlation of lumbar medial branch neurotomy results with diagnostic medial branch block cutoff values to optimize therapeutic outcome. Pain Med. 2012;13(12):1533–46.
- 142. Bogduk N, Dreyfuss P, Govind J. A narrative review of lumbar medial branch neurotomy for the treatment of back pain. Pain Med. 2009;10(6):1035–45.
- 143. Burnham RS, Holitski S, Dinu I. A prospective outcome study on the effects of facet joint radiofrequency denervation on pain, analgesic intake,

disability, satisfaction, cost, and employment. Arch Phys Med Rehabil. 2009;90(2):201–5.

- 144. Macvicar J, Borowczyk JM, Macvicar AM, Loughnan BM, Bogduk N. Lumbar medial branch radiofrequency neurotomy in New Zealand. Pain Med. 2013;14:639–45.
- 145. Dreyfuss P, Halbrook B, Pauza K, Joshi A, McLarty J, Bogduk N. Efficacy and validity of radiofrequency neurotomy for chronic lumbar zygapophysial joint pain. Spine (Phila Pa 1976). 2000;25(10):1270–7.
- 146. Chua NH, Vissers KC, Sluijter ME. Pulsed radiofrequency treatment in interventional pain management: mechanisms and potential indications-a review. Acta Neurochir (Wien). 2011;153(4):763–71.
- 147. Van Zundert J, Patijn J, Kessels A, Lamé I, van Suijlekom H, van Kleef M. Pulsed radiofrequency adjacent to the cervical dorsal root ganglion in chronic cervical radicular pain: a double blind sham controlled randomized clinical trial. PAIN®. 2007;127(1–2):173–82.
- 148. Sembrano JN, Polly Jr DW. How often is low back pain not coming from the back? Spine. 2009;34(1):E27–32.
- 149. Fortin JD, Falco FJ. The Fortin finger test: an indicator of sacroiliac pain. Am J Orthop. 1997;26(7):477–80.
- 150. Pang WW, Mok MS, Lin ML, Chang DP, Hwang MH. Application of spinal pain mapping in the diagnosis of low back pain–analysis of 104 cases. Acta Anaesthesiol Sin. 1998;36(2):71–4.
- 151. Maigne JY, Aivaliklis A, Pfefer F. Results of sacroiliac joint double block and value of sacroiliac pain provocation tests in 54 patients with low back pain. Spine (Phila Pa 1976). 1996;21(16):1889–92.
- 152. Katz V, Schofferman J, Reynolds J. The sacroiliac joint: a potential cause of pain after lumbar fusion to the sacrum. J Spinal Disord Tech. 2003;16(1): 96–9.
- 153. Maigne JY, Planchon CA. Sacroiliac joint pain after lumbar fusion. A study with anesthetic blocks. Eur Spine J. 2005;14(7):654–8.
- 154. Ebraheim NA, Elgafy H, Semaan HB. Computed tomographic findings in patients with persistent sacroiliac pain after posterior iliac graft harvesting. Spine (Phila Pa 1976). 2000;25(16):2047–51.
- 155. Liliang PC, Lu K, Liang CL, Tsai YD, Wang KW, Chen HJ. Sacroiliac joint pain after lumbar and lumbosacral fusion: findings using dual sacroiliac joint blocks. Pain Med. 2011;12(4):565–70.
- 156. Ivanov AA, Kiapour A, Ebraheim NA, Goel V. Lumbar fusion leads to increases in angular motion and stress across sacroiliac joint: a finite element study. Spine. 2009;34(5):E162–9.
- 157. Emami A, Deviren V, Berven S, Smith JA, Hu SS, Bradford DS. Outcome and complications of long fusions to the sacrum in adult spine deformity: luque-galveston, combined iliac and sacral screws, and sacral fixation. Spine. 2002;27(7):776–86.
- 158. Hallab NJ, Cunningham BW, Jacobs JJ. Spinal implant debris-induced osteolysis. Spine. 2003;28(20):S125–38.

- 159. Mody DR, Esses SI, Heggeness MH. A histologic study of soft-tissue reactions to spinal implants. Spine (Phila Pa 1976). 1994;19(10):1153–6.
- 160. Frymoyer JW, Howe J, Kuhlmann D. The long-term effects of spinal fusion on the sacroiliac joints and ilium. Clin Orthop Relat Res. 1978;134:196–201.
- 161. Ha KY, Lee JS, Kim KW. Degeneration of sacroiliac joint after instrumented lumbar or lumbosacral fusion: a prospective cohort study over five-year follow-up. Spine. 2008;33(11):1192–8.
- 162. Cho DY, Shin MH, Hur JW, Ryu KS, Park CK. Sagittal sacropelvic morphology and balance in patients with sacroiliac joint pain following lumbar fusion surgery. J Korean Neurosurg Soc = 대한신경 외과학회지. 2013;54(3):201-6.
- 163. Lazennec JY, Ramaré S, Arafati N, Laudet CG, Gorin M, Roger B, et al. Sagittal alignment in lumbosacral fusion: relations between radiological parameters and pain. Eur Spine J. 2000;9(1):47–55.
- 164. Duval-Beaupère G, Schmidt C, Cosson P. A Barycentremetric study of the sagittal shape of spine and pelvis: the conditions required for an economic standing position. Ann Biomed Eng. 1992;20(4):451–62.
- 165. Murakami E, Tanaka Y, Aizawa T, Ishizuka M, Kokubun S. Effect of periarticular and intraarticular lidocaine injections for sacroiliac joint pain: prospective comparative study. J Orthop Sci. 2007;12(3):274–80.
- 166. Horwitz MT. The anatomy of (A) the lumbosacral nerve plexus—its relation to variations of vertebral segmentation, and (B), the posterior sacral nerve plexus. Anat Rec. 1939;74:91–107.
- 167. Sakamoto N, Yamashita T, Takebayashi T, Sekine M, Ishii S. An electrophysiologic study of mechanoreceptors in the sacroiliac joint and adjacent tissues. Spine (Phila Pa 1976). 2001;26(20):E468–71.
- Gaenslen FJ. Sacro-iliac arthrodesis: indications, author's technic and end-results. JAMA. 1927;89:2031–5.
- 169. Bowen V, Cassidy JD. Macroscopic and microscopic anatomy of the sacroiliac joint from embryonic life until the eighth decade. Spine (Phila Pa 1976). 1981;6(6):620–8.
- 170. Cohen SP, Chen Y, Neufeld NJ. Sacroiliac joint pain: a comprehensive review of epidemiology, diagnosis and treatment. Expert Rev Neurother. 2013;13(1):99–116.
- 171. Patel N, Gross A, Brown L, Gekht G. A randomized, placebo-controlled study to assess the efficacy of lateral branch neurotomy for chronic sacroiliac joint pain. Pain Med. 2012;13(3):383–98.
- 172. Ho KY, Hadi MA, Pasutharnchat K, Tan KH. Cooled radiofrequency denervation for treatment of sacroiliac joint pain: two-year results from 20 cases. J Pain Res. 2013;6:505–11.
- 173. Hansen H, Manchikanti L, Simopoulos TT, Christo PJ, Gupta S, Smith HS, et al. A systematic evaluation of the therapeutic effectiveness of sacroiliac joint interventions. Pain Physician. 2012;15(3):E247–78.

- 174. Strong EK, Davila JC. The cluneal nerve syndrome; a distinct type of low back pain. Ind Med Surg. 1957;26(9):417–29.
- 175. Njoo KH, van der Does E, Stam HJ. Interobserver agreement on iliac crest pain syndrome in general practice. J Rheumatol. 1995;22(8):1532–5.
- 176. Bogduk N (on behalf of The Australasian Faculty of Musculoskeletal Medicine) (1999) Evidence based clinical guidelines for the management of acute low back pain. (Draft clinical practice guidelines for the management of acute low back pain.) The National Musculoskeletal Medicine Initiative. National Health & Medical Research Counctil. Available at: http://www.emia.com.au/MedicalProviders/ EvidenceBasedMedicine/afmm/. Accessed July 14, 2015.
- 177. Akbas M, Yegin A, Karsli B. Superior cluneal nerve entrapment eight years after decubitus surgery. Pain Pract. 2005;5(4):364–6.
- 178. Kuniya H, Aota Y, Saito T, Kamiya Y, Funakoshi K, Terayama H, et al. Anatomical study of superior cluneal nerve entrapment. J Neurosurg Spine. 2013;19(1):76–80.
- 179. Ermis MN, Yildirim D, Durakbasa MO, Tamam C, Ermis OE. Medial superior cluneal nerve entrapment neuropathy in military personnel; diagnosis and etiologic factors. J Back Musculoskelet Rehabil. 2011;24(3):137–44.
- 180. Morimoto D, Isu T, Kim K, Imai T, Yamazaki K, Matsumoto R, et al. Surgical treatment of superior cluneal nerve entrapment neuropathy. J Neurosurg Spine. 2013;19(1):71–5.
- 181. Cohen SP, Hameed H, Kurihara C, Pasquina PF, Patel AM, Babade M, et al. The effect of sedation on the accuracy and treatment outcomes for diagnostic injections: a randomized, controlled, crossover study. Pain Med. 2014;15(4):588–602.
- Babar S, Saifuddin A. MRI of the post-discectomy lumbar spine. Clin Radiol. 2002;57(11):969–81.
- 183. Mobbs RJ, Newcombe RL, Chandran KN. Lumbar discectomy and the diabetic patient: incidence and outcome. J Clin Neurosci. 2001;8(1):10–3.
- 184. Carragee EJ, Han MY, Suen PW, Kim D. Clinical outcomes after lumbar discectomy for sciatica: the effects of fragment type and anular competence. J Bone Joint Surg Am. 2003;85-A(1):102–8.
- 185. Suk KS, Lee HM, Moon SH, Kim NH. Recurrent lumbar disc herniation: results of operative management. Spine Affiliated Soc Meet Abstr. 2001;26(6):672–6.
- 186. Ross JS. Magnetic resonance assessment of the postoperative spine. Degenerative disc disease. Radiol Clin North Am. 1991;29(4):793–808.
- 187. Meyer RS, Garfin SR. Recurrent lumbar disc herniation. In: Albert TJ, Vaccaro AR, editors. Mastercases: spine surgery. New York: Thieme; 2001. p. 143–7.
- Fitt GJ, Stevens JM. Postoperative arachnoiditis diagnosed by high resolution fast spin-echo MRI of the lumbar spine. Neuroradiology. 1995;37(2):139–45.
- Thomson S, Jacques L. Demographic characteristics of patients with severe neuropathic pain secondary

to failed back surgery syndrome. Pain Pract. 2009;9(3):206–15.

- 190. Burke FD. Lumbar disc surgery. A review of a series of patients. Br J Clin Pract. 1976;30(2):29–31.
- 191. Spangfort EV. The lumbar disc herniation. A computer-aided analysis of 2,504 operations. Acta Orthop Scand Suppl. 1972;142:1–95.
- 192. Ebeling U, Kalbarcyk H, Reulen HJ. Microsurgical reoperation following lumbar disc surgery. Timing, surgical findings, and outcome in 92 patients. J Neurosurg. 1989;70(3):397–404.
- 193. Fandiño J, Botana C, Viladrich A, Gomez-Bueno J. Reoperation after lumbar disc surgery: results in 130 cases. Acta Neurochir. 1993;122(1–2):102–4.
- 194. Ross JS, Robertson JT, Frederickson RC, Petrie JL, Obuchowski N, Modic MT, et al. Association between peridural scar and recurrent radicular pain after lumbar discectomy: magnetic resonance evaluation. ADCON-L European Study Group. Neurosurgery. 1996;38(4):855–61.
- 195. Petrie JL, Ross JS. Use of ADCON-L to inhibit postoperative peridural fibrosis and related symptoms following lumbar disc surgery: a preliminary report. Eur Spine J. 1996;5:S10–7.
- 196. Vogelsang JP, Finkenstaedt M, Vogelsang M, Markakis E. Recurrent pain after lumbar discectomy: the diagnostic value of peridural scar on MRI. Eur Spine J. 1999;8(6):475–9.
- 197. Olmarker K, Rydevik B, Nordborg C. Autologous nucleus pulposus induces neurophysiologic and histologic changes in porcine cauda equina nerve roots. Spine. 1993;18(11):1425–32.
- 198. Mulleman D, Mammou S, Griffoul I, Watier H, Goupille P. Pathophysiology of disk-related sciatica. I.–Evidence supporting a chemical component. Joint Bone Spine. 2006;73(2):151–8.
- 199. Hoyland JA, Freemont AJ, Jayson MI. Intervertebral foramen venous obstruction. A cause of periradicular fibrosis? Spine. 1989;14(6):558–68.
- 200. Rydevik B, Holm S, Brown MD, Lundborg G. Diffusion from the cerebrospinal fluid as a nutritional pathway for spinal nerve roots. Acta Physiol Scand. 1990;138(2):247–8.
- Pountain GD, Keegan AL, Jayson MI. Impaired fibrinolytic activity in defined chronic back pain syndromes. Spine. 1987;12(2):83–6.
- 202. Cooper RG, Mitchell WS, Illingworth KJ, Forbes WS, Gillespie JE, Jayson MI. The role of epidural fibrosis and defective fibrinolysis in the persistence of postlaminectomy back pain. Spine. 1991;16(9):1044–8.
- 203. MacVicar J, King W, Landers MH, Bogduk N. The effectiveness of lumbar transforaminal injection of steroids: a comprehensive review with systematic analysis of the published data. Pain Med. 2013;14(1):14–28.
- 204. Vad VB, Bhat AL, Lutz GE, Cammisa F. Transforaminal epidural steroid injections in lumbosacral radiculopathy: a prospective randomized study. Spine (Phila Pa 1976). 2002;27(1):11–6.
- Manchikanti L, Pampati V, Falco FJ, Hirsch JA. Assessment of the growth of epidural injections

in the medicare population from 2000 to 2011. Pain Phys Off J Assoc Pain Manag Anesthesiol. 2013;16(4):E349–64.

- 206. Manchikanti L, Pampati V, Falco FJ, Hirsch JA. Growth of spinal interventional pain management techniques: analysis of utilization trends and Medicare expenditures 2000 to 2008. Spine. 2013;38(2):157–68.
- 207. Candido KD, Raghavendra MS, Chinthagada M, Badiee S, Trepashko DW. A prospective evaluation of iodinated contrast flow patterns with fluoroscopically guided lumbar epidural steroid injections: the lateral parasagittal interlaminar epidural approach versus the transforaminal epidural approach. Anesth Analg. 2008;106(2):638.
- 208. Chang-Chien GC, Knezevic NN, McCormick Z, Chu SK, Trescot AM, Candido KD. Transforaminal versus interlaminar approaches to epidural steroid injections: a systematic review of comparative studies for lumbosacral radicular pain. Pain Phys Off J Assoc Pain Manag Anesthesiol. 2014;17(4):E509–24.
- 209. Klessinger S. Diagnostic value of transforaminal injections of steroids in recurrent disc herniations. J Spine Neurosurg. 2013;2(2):1–4.
- 210. Derby R, Kine G, Saal JA, Reynolds J, Goldthwaite N, White AH, et al. Response to steroid and duration of radicular pain as predictors of surgical outcome. Spine. 1992;17(6 Suppl):S176–83.
- 211. Smigel L, Reeves D. Blinded analgesic effect and one year outcome of caudal D5W injection in allcomers with chronic low back pain and either buttock or leg pain. American Osteopathic Association Prolotherapy Regenerative Medicine, Kansas City, 6th May 2014.
- 212. Devulder J, Deene P, De Laat M, Van Bastelaere M, Brusselmans G, Rolly G. Nerve root sleeve injections in patients with failed back surgery syndrome: a comparison of three solutions. Clin J Pain. 1999;15(2):132–5.
- 213. Pham Dang C, Lelong A, Guilley J, Nguyen JM, Volteau C, Venet G, et al. Effect on neurostimulation of injectates used for perineural space expansion before placement of a stimulating catheter: normal saline versus dextrose 5% in water. Reg Anesth Pain Med (Sci Direct). 2009;34(5):398–403.
- Rabinovitch DL, Peliowski A, Furlan AD. Influence of lumbar epidural injection volume on pain relief for radicular leg pain and/or low back pain. Spine J. 2009;9(6):509–17.
- 215. Friedly JL, Comstock BA, Turner JA, Heagerty PJ, Deyo RA, Sullivan SD, et al. A randomized trial of epidural glucocorticoid injections for spinal stenosis. N Engl J Med. 2014;371(1):11–21.
- 216. Manchikanti L, Candido KD, Kaye AD, Boswell MV, Benyamin RM, Falco FJ, et al. Randomized trial of epidural injections for spinal stenosis published in the New England Journal of Medicine: further confusion without clarification. Pain Phys Off J Assoc Pain Manag Anesthesiol. 2014;17(4): E475–87.

- 217. Sutton KG, Martin DJ, Pinnock RD, Lee K, Scott RH. Gabapentin inhibits high-threshold calcium channel currents in cultured rat dorsal root ganglion neurones. Br J Pharmacol. 2002;135(1):257–65.
- 218. Patel S, Naeem S, Kesingland A, Froestl W, Capogna M, Urban L, et al. The effects of GABA(B) agonists and gabapentin on mechanical hyperalgesia in models of neuropathic and inflammatory pain in the rat. PAIN®. 2001;90(3):217–26.
- 219. Khan ZH, Rahimi M, Makarem J, Khan RH. Optimal dose of pre-incision/post-incision gabapentin for pain relief following lumbar laminectomy: a randomized study. Acta Anaesthesiol Scand. 2011;55(3):306–12.
- 220. Songer MN, Rauschning W, Carson EW, Pandit SM. Analysis of peridural scar formation and its prevention after lumbar laminotomy and discectomy in dogs. Spine. 1995;20(5):571–80.
- 221. Revel M, Auleley GR, Alaoui S, Nguyen M, Duruoz T, Eck-Michaud S, et al. Forceful epidural injections for the treatment of lumbosciatic pain with postoperative lumbar spinal fibrosis. Rev Rhum Engl Ed. 1996;63(4):270–7.
- 222. Meadeb J, Rozenberg S, Duquesnoy B, Kuntz JL, Le Loët X, Sebert JL, et al. Forceful sacrococcygeal injections in the treatment of postdiscectomy sciatica. A controlled study versus glucocorticoid injections. Joint Bone Spine. 2001;68(1):43–9.
- 223. Racz GB, Holubec JT. Lysis of adhesions in the epidural space. In: Racz GB, editor. Techniques of neurolysis. Boston: Kluwer Academic Publishers; 1989. p. 57–72.
- 224. Manchikanti L, Singh V, Cash KA, Pampati V. Assessment of effectiveness of percutaneous adhesiolysis and caudal epidural injections in managing post lumbar surgery syndrome: 2-year follow-up of a randomized, controlled trial. J Pain Res. 2012;5:597–608.
- 225. Freemont AJ, Peacock TE, Goupille P, Hoyland JA, O'Brien J, Jayson MI. Nerve ingrowth into diseased intervertebral disc in chronic back pain. Lancet. 1997;350(9072):178–81.
- 226. Freemont AJ, Watkins A, Le Maitre C, Baird P, Jeziorska M, Knight MT, et al. Nerve growth factor expression and innervation of the painful intervertebral disc. J Pathol. 2002;197(3):286–92.
- 227. Wolfer LR, Derby R, Lee J-E, Lee S-H. Systematic review of lumbar provocation discography in asymptomatic subjects with a meta-analysis of falsepositive rates. Pain Physician. 2008;11(4):513–38.
- 228. Carragee EJ, Don AS, Hurwitz EL, Cuellar JM, Carrino J, Herzog R. Does discography cause accelerated progression of degeneration changes in the lumbar disc: a ten-year matched cohort study. Spine (Phila Pa 1976). 2009;13(21):2338–45.
- 229. Bogduk N, Aprill C, Derby R. Lumbar discogenic pain: state-of-the-art review. Pain Med. 2013;14(6):813–36.
- Jensen MC, Brant-Zawadzki MN, Obuchowski N, Modic MT, Malkasian D, Ross JS. Magnetic reso-

nance imaging of the lumbar spine in people without back pain. N Engl J Med. 1994;331(2):69–73.

- 231. Boden SD, McCowin PR, Davis DO, Dina TS, Mark AS, Wiesel S. Abnormal magnetic-resonance scans of the cervical spine in asymptomatic subjects. A prospective investigation. J Bone Joint Surg Am. 1990;72(8):1178–84.
- 232. Barendse GA, van Den Berg SG, Kessels AH, Weber WE, van Kleef M. Randomized controlled trial of percutaneous intradiscal radiofrequency thermocoagulation for chronic discogenic back pain: lack of effect from a 90-second 70 C lesion. Spine. 2001;26(3):287–92.
- 233. Helm Ii S, Deer TR, Manchikanti L, Datta S, Chopra P, Singh V, et al. Effectiveness of thermal annular procedures in treating discogenic low back pain. Pain Phys Off J Assoc Pain Manag Anesthesiol. 2012;15(3):E279–304.
- 234. Pauza KJ, Howell S, Dreyfuss P, Peloza JH, Dawson K, Bogduk N. A randomized, placebo-controlled trial of intradiscal electrothermal therapy for the treatment of discogenic low back pain. Spine J. 2004;4(1):27–35.
- 235. Kapural L, Vrooman B, Sarwar S, Krizanac-Bengez L, Rauck R, Gilmore C, et al. A randomized, placebo-controlled trial of transdiscal radiofrequency, biacuplasty for treatment of discogenic lower back pain. Pain Med. 2013;14(3):362–73.
- Simmons JW, McMillin JN, Emery SF, Kimmich SJ. Intradiscal steroids. A prospective double-blind clinical trial. Spine (Phila Pa 1976). 1992;17(6 Suppl):S172–5.
- 237. Klein RG, Eek BC, O'Neill CW, Elin C, Mooney V, Derby RR. Biochemical injection treatment for discogenic low back pain: a pilot study. Spine J. 2003;3(3):220–6.
- List of 12 biologic start ups. 2014. Available at: http://orthostreams.com/list-of-biologics-startups/. Accessed 4 Sept 2014.
- 239. Terry A, LaSalle E, Shah B, Campos J, Nguyen J, Solomon J, et al. Lumbar intradiscal platelet rich plasma injections: a prospective, double-blind, randomized controlled trial. International Spine Intervention Society- 21st Annual Scientific Meeting Research Abstracts. 2013;14:1269–976.
- 240. Crevensten G, Walsh AJ, Ananthakrishnan D, Page P, Wahba GM, Lotz JC, et al. Intervertebral disc cell therapy for regeneration: mesenchymal stem cell implantation in rat intervertebral discs. Ann Biomed Eng. 2004;32(3):430–4.
- 241. Hiyama A, Mochida J, Iwashina T, Omi H, Watanabe T, Serigano K, et al. Transplantation of mesenchymal stem cells in a canine disc degeneration model. J Orthop Res. 2008;26(5):589–600.
- 242. Zhang YG, Guo X, Xu P, Kang LL, Li J. Bone mesenchymal stem cells transplanted into rabbit intervertebral discs can increase proteoglycans. Clin Orthop Relat Res. 2005;430:219–26.
# Cell Transplantation for Lumbar Spine Degenerative Disk Disease

39

### Christian Hohaus and Hans Jörg Meisel

#### 39.1 Introduction

Low back pain is a very common symptom, affecting nearly three-quarters of the population some point in their life. While 90 % of the population recovers within 3 months, in some patients, chronic back or leg pain leads to long-term physical disability and a reduced quality of life. Disk anatomy likely plays a pivotal role in the underlying pain, yet abnormal spine and disk morphology, including disk herniation, have been described as normal findings in the asymptomatic population [1]. Why is it that some patients remain asymptomatic, while some symptomatic patients with degenerative changes may be treated?

Given that disk herniation is thought to be an extension of progressive disk degeneration that attends the normal aging process, seeking an effective therapy that staves disk degeneration has been considered a logical attempt to reduce back pain. Previous studies have validated genetic factors [2–5] and implicated nutrition [6] as relevant to the degenerative process. However, the high prevalence across diverse populations suggests that a myriad of unidentified factors likely contribute to the symptom complex.

As no effective therapies to retard or reverse disk degeneration have yet been devised, a variety of surgical procedures have been developed to treat disk degeneration and back pain. Unfortunately, the procedures currently available fail to provide an outcome that is structurally sound and at the same time physiologic. Surgery tends to limit motion. Fusion in particular seems to shift excessive stresses to adjacent spinal segments. Equally concerning in selecting fusion as an option is the fact that non-unions have been reported in 5–35 % of patients [7, 8] and that patients undergoing a repeat fusion for lumbar spine failed surgery have a clinical failure rate as high as 40 % [9–11]. The advent of tissue engineering has broadened the options for considering treatments that tailor repair to distinct anatomy. In particular, the use of cell and gene therapy to provide specific properties or repair specific tissues is widely considered an emerging modality for effecting treatment.

C. Hohaus, MD • H.J. Meisel, MD, PhD (⊠) Department of Neurosurgery, Professional Association Hospital Bergmannstrost, Halle, Germany

Translational Center of Regenerative Medicine, University of Leipzig, Leipzig, Germany e-mail: Christian.Hohaus@Bergmannstrost.com; Meisel@Bergmannstrost.com

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016 J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_39

Numerous scientific studies have provided observations concerning the biochemistry and biomechanics of the disk, offering insights and theories into structure-function-failure relationships [3, 12, 13]. The most apparent cellular and biochemical changes attributable to degeneration include a decrease in cell density in the disk that is accompanied by a reduction in synthesis of cartilage-specific extracellular matrix components, such as Type II collagen and aggrecan. As the proteoglycan content of the disk decreases, the resulting loss of water-binding capacity by the disk matrix, coupled with a subsequent reduced capacity for dissipating spinal forces, is thought to lead to degenerative disk disease [7, 14, 15].

Collagen plays a key load-bearing role in the disk, and changes in its extracellular matrix content have been attributed to aging as well as to the pathology of degeneration [16]. In normal intervertebral disks, at least seven different types of collagen are present (i.e., Types I, II, III, V, VI, IX, and XI), although Types I and II are the most abundant [17–22]. The anulus fibrosus contains more Type I collagen than Type II, whereas the nucleus pulposus is composed mainly of Type II collagen.

Calcification of the vertebral endplates is another factor thought to be relevant to disk degeneration. The passage of nutrients and waste products across the endplate depends on fluid flowing into the disk (during the night at bed rest) and flowing out during the day when we walk about [23]. Thus, shortcomings of permeability would be expected to adversely effect chondrocyte metabolism [24–26].

While cells constitute only 1 % of the adult disk tissue by volume, their role in matrix synthesis and metabolic turnover is vital. Most assessments of intervertebral disk failure have focused on degenerative, morphologic changes in disk tissue morphology that affect the biomechanical performance of the motion segment [13, 27]. In this vein, mechanical failure is little more than a corollary of matrix structure, which in turn depends on balanced cell metabolism for efficient maintenance of the disk matrix. Given the value of cells to the metabolic health of the disk, one therapeutic strategy could be to replace, regenerate, or augment the intervertebral disk cell population, with the goal of correcting matrix insufficiencies and restoring normal segment biomechanics.

Recent work has shown that disk aging and degeneration are accompanied by a decline in the number of cells in the disk, a change attributable to both necrosis and apoptosis [28]. Perhaps a more important outcome of this work and that of others has been to demonstrate that disk cells retain an ability to respond to both genetic endowment and in vivo stimulation and that when returned to the disk under controlled conditions, integrate with the surrounding tissue [6, 28–30].

With this in mind, we designed a study using the dog as our model to investigate the hypothesis that repair of the damaged disk is technically feasible, autologous cells can be reproducibly cultured under defined and controlled conditions, percutaneous delivery is possible, and disk cells will integrate with the surrounding tissue, produce the appropriate intervertebral disk extracellular matrix, and potentially provide a functional solution to disk repair.

## 39.2 Canine Trial of Chondrocyte Transplantation

The goal of this study was to test the hypothesis that restoration of intervertebral disk morphology could be achieved by transplantation of cultured autologous chondrocytes into the nucleus pulposus. As a natural model of degeneration has not been described in a large mammal, this study was fashioned after established work demonstrating that degeneration can be stimulated by damaging the outer anulus [31]. Under institutional guidelines of the Institutional Animal Care and Use Committee (IACUC), 18 purpose-bred, 2-yearold female dogs, weighing between 20 and 25 Kg, were studied to see whether the introduction of cultured autologous disk-derived cells would repair a damaged disk and inhibit degenerative changes. Prior to surgery, 125 ml of blood was obtained from each of the dogs to serve as a serum supplement for autologous cell culture. As blood loss was insignificant during the surgical procedure, this approximate 6–8 % loss of total blood volume was not considered an additional risk to the animals.

The dogs were divided into two basic groups: 4 animals receiving autologous cells containing bromodeoxyuridine (BrdU) as a nuclear marker and the other 14 receiving autologous cells without a nuclear marker. Animals were radiographed to establish a baseline for preexisting spine pathology. Under general anesthesia, a minimal invasive approach was made to the posterolateral aspect of the canine lumbar spine. Lumbar intervertebral disks at levels L1/L2, L2/ L3, and L3/L4 were identified as study levels for the procedure and disk tissue was collected. Approximately 200 mg of tissue was collected from the lateral aspect of the anulus, 100 mg of anulus material, and 100 mg of nucleus pulposus material.

The sampled disk cells were expanded in culture through several passages, with a goal of establishing a population of disk cells capable of producing matrix and sustaining an expanded volume within the damaged disk. The average number of cells expanded and transplanted in each L3–L4 disk was approximately six million cells. This procedure was done by the Co.don AG Teltow/Germany.

In this study, tissue was removed from the L1– L2 intervertebral disk, but chondrocyte transplantation was not performed. The L2–L3 disk was approached but not violated, thus serving as a surgical control, while the L3–L4 level underwent disk material removal and underwent chondrocyte transplantation 12 weeks later. The wound sites were closed with resorbable suture and the animals returned to their holding area. None of the animals developed problems related to the surgery and all regained full function.

An important criterion for evaluating the success of cell transplantation for the disk repair procedure was identifying that matrix regeneration was attributable to transplanted cultured expanded disk cells, rather than a result of inherent disk capacity for self-repair. BrdU, an analog nucleotide of thymidine, was incorporated into the nucleus during DNA synthesis and could later be identified by immunohistochemical techniques. As such, it was possible to analyze morphology in situ after repair and delineate cells that were transplanted from those already present in the host tissue. To verify the source of disk repair and matrix regeneration, BrdU was used as a cell marker in four animals.

During the last 4 days in monolayer culture, the cells in passage 2 were tagged by adding a small concentration of BrdU (1:1000) to the culture medium. To perform growth curves, monolayer cells in passage 1 were cultivated in six-well plates and the cell number in each well was determined daily. Viability of the cells was assessed by staining with trypan blue.

Twelve weeks after the disk tissue had been harvested, the autologous disk cell cultures were transplanted at L3–L4 on each of the dogs. The intervertebral disk between L1 and L2 served as the control for untreated degeneration. Cells were shipped from Teltow, Germany, overnight at 4–8 °C for transplantation. Animals were anesthetized, placed in right lateral recumbence, and the L3–L4 level was located by fluoroscopic imaging. As the previous surgeries had been performed from the right lateral side, the cultured cells were introduced through the left side of the anulus.

The animals were humanely euthanized 3 months (3 dogs), 6 months (7 dogs), 9 months (4 dogs), and 12 months (4 dogs) following the cell transplantation. Immediately after the dogs were killed, their lumbar spines were removed and the tissue analyzed (Fig. 39.1), MRI and X-ray analysis and coronal slices of the spinal column were performed to interpret disk height.

Tissue analyses included light microscopy and immunohistochemistry for assessing BrdU content (Fig. 39.2) and collagen expression.

The canine study evaluated whether autologous disk cell transplantation might be an appropriate therapeutic treatment to repair disk



**Fig. 39.1** Gross pathology 12 month follow-up after autologous chondrocyte transplantation in the Canine model. Level L3–L4 was transplanted, level L1–L2 received no treatment and displayed more scar tissue, L2–L3 was the control level with a normal intervertebral disk

damage and inhibit degeneration. In this context, several important observations emerged:

- 1. Autologous disk cells were expanded in culture and returned to the disk by a minimally invasive procedure after 12 weeks. Under defined conditions, it was possible to assure phenotype and assess metabolic capacity of the cells prior to transplantation.
- Disk cells remained viable after transplantation as shown by BrdU incorporation and maintained a capacity for proliferation after transplantation as depicted by histology.
- Transplanted disk cells produced an extracellular matrix that contained components similar to normal intervertebral disk tissue. Positive evidence of proteoglycan content was supported by accepted histochemical staining techniques such as safranin O-fast green.

- Both Type II and Type I collagens were demonstrated in the regenerated intervertebral disk matrix by immunohistochemistry following chondrocyte transplantation.
- There was a statistically significant correlation between transplanting cells and retention of disk height that was demonstrated at longer intervals following transplantation.

Although a morphotypic nucleus pulposus was not generated, cells that could appropriately be considered disk cells were identified in the intervertebral disks that had received disk cell transplantation. The observed matrix to cell ratio suggested strongly that these cells were elaborating a cartilage-specific matrix that was appropriate with respect to both collagen and proteoglycan components. No evidence of necrotic change was present, nor were there any active signs of tissue vascularization. Absence of bone in the intervertebral space and the productive matrix synthesis suggested that active remodeling and expression were guided by the demands of the anatomy and that cell response after transplantation was dependent on both the phenotypic identity of the cells and the biomechanical cues of the anatomy.

Cell viability and their capacity for matrix synthesis were particularly encouraging outcomes of this study. In the light of a 12-week interval between disk tissue sampling and cell transplantation, cells were placed into an environment that had fundamentally changed in both composition and function. Under the provision of central delivery and pressurized containment, the transplanted cells were prepared for the environment of the nucleus pulposus. The high cell to volume ratio of the transplanted cells, the deformable nature of the regional anatomy, and the inherent capacity of the cells to respond to new loading regimens all supported the vitality of the transplant conditions.

Extracellular matrix change, biomechanical variation, altered morphology, and cell viability are acknowledged steps leading to intervertebral disk degeneration. In the process of invigorating the population of vital disk cells and achieving matrix transformation, a positive observation regarding the morphology of the disk was made. The ability to control cell conditions, potentially to imbue the cells with additional genetic capac-



**Fig. 39.2** Staining of paraffin sections of the regenerated intervertebral disk 6 months following cell transplantation. BrdU-containing chondrocytes were detected and stained by immunohistochemical procedures using DAB as the chromogen. Sections were counterstained by eosin.

BrdU-positive cells are colored *black*. (a) Nucleus regenerates overview  $(25\times)$ ; (b) BrdU-stained transplanted cells  $(200\times)$ ; (c, d) single BrdU-stained transplanted chondrocytes, pericellular de novo synthesis of nucleus matrix  $(1,000\times)$ 

ity, and the availability of autologous tissue from diskectomy procedures make this technology both feasible and attractive.

#### 39.3 EuroDisc Randomized Trial

After these positive and promising results, the EuroDisc Randomized Trial was initiated to embrace a representative patient group, examining not only the traumatic, less degenerative disk, but also to include patients with persistent symptoms that had not responded to conservative treatment and in whom a surgical treatment is considered.

Interventional surgery for disk herniation is one of the most widely used and effective treatments for back pain that emerges within the broad scope of disk degeneration. Successful removal of impinging tissue offers the individual patient substantial relief for associated pain. However, the reduction of tissue involved with the surgical procedure anatomically compromises the function of the affected disk and effects a load transfer to adjacent disks. Biological restoration with interventional cell therapy offers a potential for accentuating disk metabolism with an underlying intent to restore spine mechanics.

Patients who were to undergo surgical intervention at one level were eligible for participation in the trial; patients requiring treatment at more than one level were excluded from the study. Prior to their participation, all patients were advised of the potential risks and signed a letter of consent. No placebo group was commit-



**Fig. 39.3** Intraoperative setting. (a) Fluoroscopic-guided minimal invasive puncture of the intervertebral disk form the opposite side, (b) pressure-volume-test, (c) cell transplantation

ted to this study; each patient participating in the clinical trial was to undergo surgical treatment for their disk prolapse, and the prospective basis of cell transplantation separated the active treatment group from the control group. Patients were not blinded to their treatment. Randomization was done after the open microdiskectomy. Eligibility was limited to patients between 18 and 60 years of age, with a body mass index (BMI) below 28. Exclusion criteria for participating in the study included sclerotic changes, edema, Modic changes of Types II or III, and spondylolisthesis among other accepted criteria such as pregnancy, etc. Operative procedures were performed as minimally invasive open sequestrectomies performed by an experienced neurosurgeon under general anesthesia. The harvested cells from the sequestered disk material were cultured by the Co.don AG Teltow/Germany under GMP conditions. More than five million living disk cells were included in the solution for transplantation.

A single puncture with a minimal caliber cannula was used to achieve a precise delivery with minimal trauma to the patient and to the anulus (Fig. 39.3). The technique was developed with respect to literature that has demonstrated a sizespecific correlation between anular injury and disk degeneration. A simple, minimal invasive technique was necessary to reduce the wound site trauma and effectively support cell injection without further injury to the anulus. Cells were transplanted approximately 12 weeks following sequestrectomy to assure that the anulus has healed and will contain the cells. Using a pressure-volume test prior to the delivery of any chondrocytes, cells could be placed with the confidence that they would be retained at the site of delivery.

One hundred and twelve patients have been enrolled in the EuroDisc Study; the primary criteria follow-up was intended to occur at 1 year, an interim analysis scheduled at 2 years, and the final analysis will be completed at 4 years. The primary clinical evaluation criterion was the Oswestry Low Back Pain Disability Questionnaire. Secondary criteria include the SF-36, Prolo Score [32], Quebec Back Pain Disability Scale, MRI, and X-ray evaluation. The use of the Oswestry disability questionnaire in clinical trials is recommended by the German Orthopedic Society (DGOT), demonstrating acceptable test quality and satisfactory test-retest reliability. The Quebec Back Pain Disability Scale, another self-rating scale, was professionally developed using factor analysis comprising high internal consistency, high item discriminability, and high test-retest reliability. Finally, the SF-36, an often used scale to assess patients' general condition and quality of life, and a VAS will be used to standardize measureable pain.

The interim analysis, made by a cut in January of 2006 to assess whether intervention correlated with positive clinical outcomes, forms the basis for this report. Within the analysis, successive 3-month, 6-month, 12-month, and 24-month assessments are stratified within the continuum of the study. The information within this study allows a broad interpretation of the general progress made over 2 years following a clinical intercession with autologous disk cells. Interim analysis was performed on the first 28 patients who reached 24-months follow-up with regard to autologous disk cell transplantation (ADCT). These first 28 patients were randomized in three different centers.

For descriptive analysis of efficacy, the total sum score as well as the disability index of the Oswestry Low Back Pain Disability Questionnaire (OPDQ) and the total sum score of the Quebec Back-Pain Disability Scale (QBPD) were taken into account from the initial presurgical presentation through the 2-year follow-up. The outcomes are depicted in Table 39.1. Based on the mean total sum score as well as the disability index of the OPDQ, differences in initial presentations between the control group and those receiving autologous cells were not minimal. Surgery as an intervention was a positive experience and, as expected, substantially reduced the patient's disability and pain. The trend in reduction of the total sum score continued to decrease in the patients whose treatment was supplemented by cell transplantation, while the control group did not sustain continual improvement. Two years following the therapeutic intervention with cells, both the total sum score and the disability index of the OPDQ were lower in the ADCT group compared with the control.

Descriptive analyses of the mean total sum score of the QBPD prior to sequestrectomy, prior to ADCT/control, and 3 months after ADCT/control demonstrated a decrease in mean and median sum scores in both groups. Although the mean and median values for both the ADCT and the control group decreased between 1 and 2 years, the assessments for the ADCT group were clearly lower (Table 39.2). Patient global assessment of pain demonstrated some fluctuation although both groups received substantial relief from the surgical intervention. However, as patients were tracked over the course of the 2-year follow-up, changes emerged that suggest that the ADCTtreated patients have a lower pain scores (Table 39.3).

MRI was used to assess the respective disk height along the course of the analyses from the date of the sequestrectomy until the 2-year follow-up (Fig. 39.4). In addition to the disk height, the content of the liquid component was evaluated as a means of assessing matrix content. Results of the analysis of the intervertebral disk height compared affected (treated with surgery, or with surgery and cells) with non-affected adja-

Total sums	score								
		Ν	Mean	SD	Min	Lower quartile	Median	Upper quartile	Max
Visit -1	ADCT	12	28.42	9.30	13.00	20.00	29.50	36.00	45.00
	Control	16	26.88	9.99	14.00	18.00	25.50	34.00	46.00
Visit 0.5	ADCT	12	8.00	6.89	0.00	2.50	7.50	12.50	24.00
	Control	15	8.40	4.69	1.00	4.00	9.00	13.00	15.00
Visit 1	ADCT	11	6.73	8.56	0.00	0.00	5.00	12.00	28.00
	Control	14	7.14	6.36	0.00	1.00	5.50	13.00	19.00
Visit 2	ADCT	10	9.10	10.72	0.00	1.00	6.50	12.00	35.00
	Control	14	7.79	7.42	0.00	2.00	6.50	12.00	26.00
Visit 3	ADCT	11	7.82	8.46	0.00	2.00	4.00	15.00	25.00
	Control	14	7.07	5.94	0.00	1.00	7.00	12.00	19.00
Visit 4	ADCT	12	6.00	8.89	0.00	0.00	2.00	8.50	29.00
	Control	16	7.56	6.52	0.00	2.50	6.00	13.00	19.00
Disability	index (%)								
Visit -1	ADCT	12	56.83	18.60	26.00	40.00	59.00	72.00	90.00
	Control	16	53.75	19.97	28.00	36.00	51.00	68.00	92.00
Visit 0.5	ADCT	12	16.06	13.73	0.00	5.33	15.00	25.00	48.00
	Control	15	16.80	9.37	2.00	8.00	18.00	26.00	30.00
Visit 1	ADCT	11	13.45	17.11	0.00	0.00	10.00	24.00	56.00
	Control	14	14.29	12.72	0.00	2.00	11.00	26.00	38.00
Visit 2	ADCT	10	18.64	21.53	0.00	2.00	13.89	26.67	70.00
	Control	14	15.62	14.80	0.00	4.44	13.00	24.00	52.00
Visit 3	ADCT	11	15.64	16.92	0.00	4.00	8.00	30.00	50.00
	Control	14	14.14	11.88	0.00	2.00	14.00	24.00	38.00
Visit 4	ADCT	12	12.00	17.79	0.00	0.00	4.00	17.00	58.00
	Control	16	15.19	12.99	0.00	5.50	12.00	26.00	38.00

**Table 39.1** Total sumscore and disability index of the OPDQ based on patients who had been followed for 2 years after autologous disk chondrocyte transplantation

Visit -1: Sequestrectomy. Visit 0.5: ADCT/control. Visit 1: 3 months after ADCT/control visit 0.5 Visit 2: 6 months after ADCT/control visit 0.5, Visit 3: 12 months after ADCT/control visit 0.5 Visit 4: 24 months after ADCT/control visit 0.5

Table 39.2	Total sumscore of the QBPD based on patients with at least 2 years follow-up after autologous	disk chon-
drocyte trans	splantation	

Total sum	score								
		Ν	Mean	SD	Min	Lower quartile	Median	Upper quartile	Max
Visit -1	ADCT	12	45.08	17.60	23.00	31.50	42.00	55.00	82.00
	Control	16	49.69	18.69	21.00	34.00	45.00	65.00	81.00
Visit 0.5	ADCT	12	14.75	16.07	0.00	4.50	8.50	17.50	50.00
	Control	15	18.27	11.04	1.00	6.00	19.00	25.00	38.00
Visit 1	ADCT	11	10.64	16.05	0.00	1.00	4.00	15.00	55.00
	Control	14	13.29	9.72	3.00	6.00	8.50	24.00	30.00
Visit 2	ADCT	10	15.00	20.77	0.00	1.00	10.00	19.00	70.00
	Control	14	13.93	11.76	1.00	4.00	12.50	18.00	41.00
Visit 3	ADCT	11	11.09	16.71	0.00	2.00	4.00	19.00	57.00
	Control	14	12.71	12.55	2.00	4.00	9.50	17.00	48.00
Visit 4	ADCT	12	9.33	15.33	0.00	0.50	3.50	12.50	55.00
	Control	16	13.94	12.61	0.00	5.00	8.00	22.50	41.00

Visit -1: Sequestrectomy, Visit 0.5: ADCT/control, Visit 1: 3 months after ADCT/control visit 0.5 Visit 2: 6 months after ADCT/control visit 0.5, Visit 3: 12 months after ADCT/control visit 0.5 Visit 4: 24 months after ADCT/control visit 0.5

Global as	sessment of	f pain (	(100 mm V	AS)					
		N	Mean	SD	Min	Lower quartile	Median	Upper quartile	Max
Visit -1	ADCT	11	59.45	22.76	15.00	48.00	60.00	76.00	96.99
	Control	16	57.31	28.51	0.00	27.00	70.00	79.50	88.98
Visit 0.5	ADCT	12	19.17	19.37	0.00	2.50	13.00	31.50	65.00
	Control	15	17.20	14.70	0.00	3.00	14.00	31.00	46.00
Visit 1	ADCT	11	12.82	19.37	0.00	0.00	3.00	24.00	61.99
	Control	14	14.36	10.59	1.00	4.00	15.00	22.00	33.00
Visit 2	ADCT	10	21.00	22.85	0.00	8.00	16.50	23.00	78.99
	Control	14	14.00	16.51	1.00	2.00	5.50	19.00	51.00
Visit 3	ADCT	11	18.00	18.73	2.00	3.00	9.00	25.00	56.00
	Control	14	15.07	12.16	0.00	3.00	12.00	29.00	37.00
Visit 4	ADCT	12	11.17	13.48	0.00	1.00	5.00	17.00	39.00
	Control	16	15.62	15.16	1.00	3.00	12.50	26.50	53.99

**Table 39.3** Global assessment of pain based on patients with at least 2-years follow-up after autologous disk chondrocyte transplantation

Visit -1: Sequestrectomy. Visit 0.5: ADCT/control, Visit 1: 3 months after ADCT/control visit 0.5 Visit 2: 6 months after ADCT/control visit 0.5, Visit 3: 12 months after ADCT/control visit 0.5 Visit 4: 24 months after ADCT/control visit 0.5



**Fig. 39.4** A 28-year-old woman undergoing diskectomy in level L5–S1. The 60-month follow-up MRI displays increased signal intensity in the transplanted L5–S1 level. (a) Pre-transplantation. (b) 1 day post transplantation. (c)

3 month post transplantation. (d) 12 month post transplantation. (e) 24 month post transplantation. (f) 60 month post transplantation

cent segments in the same patients, and also measured the relative vertebral heights as a means of assessing patient demographics and morphologic variation. Comparison of the mean intervertebral disk heights and the vertebral heights revealed no differences between the groups.

An analysis of fluid content of the intervertebral disk at each visit demonstrated that more than 80 % of the affected segments showed decreased hydration 3 months following surgery (Table 39.4). In general, the proportion of affected segments with a diminished water content decreased over the course of the trial. Of particular interest was the outcome at 2 years, where the ADCT-treated group showed a substantially higher normalization as a group; 41 % normal fluid content compared with only 25 % normal content in the control group. Perhaps the most interesting of all the data to emerge from this study comes from inspecting disks either one or two segments from the treated intervertebral disk. Fluid levels at both of these segments

Content o	of liquid									
		N	Affected s	segment	1. Non-aff	fected segment	2. Non-af	2. Non-affected segment		
			Normal (%)	Decreased (%)	Normal (%)	Decreased (%)	Normal (%)	Decreased (%)		
Visit -1	ADCT	12	16.67	83.33	83.33	16.67	83.33	16.67		
	Control	15	13.33	86.67	86.67	13.33	46.67	53.33		
Visit 0.5	ADCT	12	25.00	75.00	81.82ª	18.18 <sup>a</sup>	50.00	50.00		
	Control	14	0.00	100.0	78.57	21.43	28.57	71.43		
Visit 3	ADCT	11	27.27	72.73	90.91	9.09	63.64	36.36		
	Control	13	23.08	76.92	76.92	23.08	53.85	46.15		
Visit 4	ADCT	12	41.67	58.33	91.67	8.33	66.67	33.33		
	Control	16	25.00	75.00	86.67 <sup>b</sup>	13.33 <sup>b</sup>	56.25	43.75 %		

Table 39.4 Analysis of fluid content of the intervertebral disk with at least 2-years follow-up after autologous disk chondrocyte transplantation

Visit -1: Sequestrectomy, Visit 0.5: ADCT/control, Visit 3: 12 months after ADCT/control visit 0.5

Visit 2: 6 months after ADCT/control visit 0.5, Visit 3: 12 months after ADCT/control visit 0.5

Visit 4: 24 months after ADCT/control visit 0.5

<sup>a</sup> : only 11 values available

<sup>b</sup>: only 15 values available

showed a substantially higher percentage of normal fluid content despite the fact that they were distant from the surgical intervention site.

The interim analysis of the EuroDisc Study revealed the following:

- 1. Disk cells that had been removed as a normal part of sequestrectomy could be expanded in culture under GMP conditions and returned to the patient after the anulus had been allowed to heal for 12 weeks.
- 2. Disk cell transplantation could be delivered by percutaneous technique.
- 3. Patients who received autologous disk cell transplantation had greater pain reduction at 2 years compared with patients who did not receive cells following their diskectomy surgery.
- 4. Disks in patients that received cells demonstrated a significant difference, as a group, in the fluid content of their treated disk when compared to control.
- 5. Adjacent intervertebral disks, both at one level and two levels from the intervertebral disk that received the cell therapy also demonstrated a difference in fluid content.

The results of this study are encouraging from several perspectives. First morphologic outcomes mirrored that seen in our preclinical animal study

[33]. Second, the pain relief seen in the pilot study which served as a basis for this clinical trial was sustained for the course of this 2-year interim analysis. This gives cause to the success of the cell-based intervention. Chondrocytes for transplantation into degenerated disks are limited in daily practice. Different cell types with regenerative capacity are required to treat patients with clinical symptoms but earlier levels of disk degeneration.

Adipose tissue provides such an alternative source of regenerative cells with little donor site morbidity. These regenerative cells are able to differentiate into a nucleus pulposus-like phenotype when exposed to environmental factors similar to disk and offer the inherent advantage of availability without the need for transporting, culturing, and expanding the cells [34, 35].

#### 39.4 **Canine Trial of Adipose-Derived Regenerative Cell** Transplantation

In an effort to develop a clinical option for cell placement and assess the response of the cells to the postsurgical milieu, adipose-derived cells were collected, concentrated, and transplanted under fluoroscopic guidance directly into a surgically damaged disk.

For this study, 12 dogs 2 years of age, were obtained. Adipose cells were harvested from the super-scapular region of the neck (scruff) and adherent cells separated, collected, and labeled with DAPI. Adipose tissue has been known for some time to contain regenerative cells in addition to fat cells [36]. Three lumbar intervertebral disk levels in each dog underwent a partial nucleotomy; other levels served as nonoperated controls. Levels of intervention, as well as the regimen of treatment, were dually randomized. Three interventions were used in this study; adipose-derived cells in hyaluronic acid (HA) carrier, HA alone, or no intervention - all deliveries were guided by fluoroscopy. Assessments were made by MRI, radiography, microscopy, RT-PCR, and ELISA.

Six dogs were radiographed, received MRI scans (Fig. 39.5), and then were euthanized by 6 months. The disk tissue was harvested from the lumbar spine in each dog (Fig. 39.6). Cells were seen to be viable in the tissue (Fig. 39.7). Matrix composition was assessed; assays were made of aggrecan, Types I and II collagen by both RT-PCR and ELISA to assess and compare matrix regeneration. mRNA and protein from each level were presented with respect to normal values defined as the 100 % expression (Table 39.5).

Table 39.6 depicts the relative protein levels as measured by ELISA.

The data were calculated with two samples t-test, comparing control with interventions at P < 0.05 and P < 0.01. Statistical differences were found between the control and each intervention at P < 0.01, whereas the difference between control and HA plus cells was only significant at P < 0.05. No significant difference could be shown between HA alone and no intervention. These evaluations and other morphometric assessments support:

- 1. Cell viability follows implantation.
- 2. Supplementing adipose cells following injury supports regeneration.
  - Morphology was maintained.
  - Intervertebral disk height was not lost.
  - MRI signal remained similar to native control.
- 3. Hyaluronic acid was insufficient to prevent disk degeneration or desiccation.



**Fig. 39.5** Six-month follow-up MRI sagittal section (same dog as in Fig. 39.2)

- 4. Lack of intervention resulted in progressive degeneration.
- A limited nucleotomy procedure, similar to that which would be experienced following clinical microdiskectomy, resulted in prolapsed anulus tissue into the central space of the nucleus pulposus.
- 6. No significant regeneration of cells or matrix occurred without treatment.

The results of the study provide evidence that cells harvested from adipose tissue might offer a reliable source of regenerative potential capable of bio-restitution. Such makes the case for using adipose-derived cells; first, cells can be transplanted percutaneously; and second cells survive and functionally adapt and produce an appropriate matrix. The span of this study was sufficient to show that freshly isolated cells will survive the trauma associated with postsurgical inflammation. The time to treat, the cell carrier, and the ability of the cells to integrate into the disk matrix were all certainly convincing.



**Fig. 39.6** Six-month follow-up gross section pathology. (a) L3–L4 no treatment. (b) L4–L5 hyaluronic acid alone. (c) L5–L6 adipose-derived stem cells in hyaluronic acid. (d) L6–L7 normal disk



**Fig. 39.7** Cell viability and cell sustenance were judged positive by DAPI-stained nuclei that were present 6 months following transplantation of the adipose-derived stem cells

### 39.5 Discussion

Cell transplantation for degenerative disk disease is possible, and the good results of the aforementioned clinical trial provide a promising outlook for the future treatment for patients with intervertebral degenerative disk disease.

Autologous disk cell transplantation after sequestrectomy is a safe and technical feasible procedure. Transplanted chondrocytes are viable in situ and create a functional matrix.

The first results of the EuroDisc Study give strong evidence for the safety and efficiency of the disk-derived cell transplantation applied following sequestrectomy to delay or inhibit ongoing processes of disk degeneration. After transplantation a statistically significant decrease in OPDQ, QBPD, and VAS in ADCT-treated patients for disability and pain is measured. In the MRI analysis, less decreased liquid content in affected intervertebral disks in the ADCT group could be displayed. A statement for the ideal group of patients who profits most from the autologous disk cell transplantation is at this time not possible.

The technique of autologous disk cell transplantation is only possible for patients who underwent a sequestrectomy. It might be better to transplant disk cells in the earlier stage of degeneration without any loss of matrix from the intervertebral disk, but this unfortunately requires an operation. Surgery to harvest cell material in the early stages of disk degeneration introduces the risk of infection to the disk and bone, the risk of nerve root injury, and other potential complications. Therefore, there is no ethical or medical foundation for this consideration.

In patients with symptoms of disk degeneration and MRI findings according to the Pfirrmann classification [37] of one to three along with symptomatic nerve root compression by a sequestered nucleus pulposus prolapse who plan to undergo a minimal invasive operative procedure, the possibility of autologous disk cell transplantation may now be entertained.

For patients with intervertebral degenerative disk disease who do not need to undergo operative treatment with sequestrectomy

	Control	Hyaluronic acid alone	Adipose-derived stem cells with hyaluronic acid	No intervention
Aggrecan	100	43.6	85.6	37.9
Type I col	100	73.2	87.1	67.2
Type II col	97.5	41.5	82.8	41.35

Table 39.5 Relative mRNA for specific matrix proteins – comparison between treatments

 Table 39.6
 Relative protein for specific matrix proteins – comparison between treatments

			Adipose-derived stem cells with	
	Control	Hyaluronic acid alone	hyaluronic acid	No intervention
Aggrecan	100	62.3	83.0	58.4
Type I col	97.7	74.9	88.4	71.2
Type II col	99	55	85	53

concerning by root nerve compression with untreatable pain or neurological deficits, a therapeutic option could be the transplantation of adipose-derived stem and regenerative cells.

We have presented in the animal model the evidence that these cells can be transplanted at surgery using fluoroscopic guidance and that these cells can be injected directly into the intervertebral disk with the expectation that they will remain viable and produce appropriate, tissue-specific matrix [38, 39].

If the future results of the adipose-derived and regenerative cell study present demonstrate promising outcomes, this may represent a safe and easy technique of producing cells for minimal invasive transplantation into a symptomatic degenerative intervertebral disk without open operative intervention.

In a first clinical study using stem cells for treatment of intervertebral degenerative disk disease, Orozco et al. illustrate the technical feasibility and safety [40]. Ten patients with chronic back pain diagnosed with lumbar disk degeneration with intact anulus fibrosus were treated with autologous expanded bone marrow MSC injected into the nucleus pulposus area. The follow-up time was 1 year and there was no control group. Treated patients exhibited rapid improvement of pain and disability. The authors compare this favorably with the results of other procedures such as spinal fusion or total disk replacement. These positive effects should be studied further more and should transfer into a controlled randomized study.

Another option for these patients could be the transplantation of juvenile allogenic chondrocyte cells. Coric et al. could demonstrate in a small group of patients the technical feasibility and safety of this procedure by using juvenile allogenic chondrocyte cells harvested from the articular surface of cadaveric donor tissue [41]. These cells were transplanted using fibrin glue like carrier.

#### Conclusion

A total regeneration of the degenerated intervertebral disk is today not possible. The goals of regenerative medicine at this time are to prevent further progression of disk degeneration and its associated symptoms.

The now available, safe, and well-studied strategy for arresting and reversing degenerative disk disease is the autologous disk-derived cell transplantation after sequestrectomy [39].

Our own experience currently embraces more than 120 patients treated with autologous disk cell transplantation over the last 10 years. This is the largest number of patients with degenerated intervertebral disk disease treated with cell transplantation under strong study conditions. All patients profit from transplantation via reduced back pain and increased quality of life. All of the patients with an employment contract could return to work after transplantation. Over time, MR images demonstrate a stable disk height in the transplanted segment. The reduction of the reherniation rate was by 52 % compared with the control group. We didn't see any inflammation in all these patients.

However, the use of autologous chondrocytes requires the ex vivo expansion of cells, which is costly, time-consuming, and highly regulated, making it an intricate procedure.

An alternative that might circumvent these disadvantages is the use of a one-step procedure, using stem cells obtained from autologous adipose tissue. Ongoing studies should provide more information about the possibility of using stem cells for regenerative therapies in intervertebral degenerative disk disease.

Pitfalls and factors for associated drawback of these regenerative therapies should include the alteration of regulatory frameworks as the European Union – Advanced Therapy Medical Products (EU-ATMP) regulation, as well as refined reimbursement strategies.

#### References

- Boden SD, McCowin P, Davis DO, Dina TS, Mark AS, Wiesel S. Abnormal magnetic-resonance scans of the cervical spine in asymptomatic subjects. A prospective investigation. J Bone Joint Surg Am. 1990;72:1178–84.
- Annunen S, et al. An allele of COL9A2 associated with intervertebral disc disease. Science. 1999; 285(5426):409–12.
- Kawaguchi Y, et al. Association between an aggrecan gene polymorphism and lumbar disc degeneration. Spine. 1999;24(23):2456–60.
- Paassilta P, et al. Identification of a common risk factor for lumbar disk disease. JAMA. 2000;285: 1843–9.
- Videman T, et al. Intragenic polymorphisms of the vitamin D receptor gene associated with intervertebral disc degeneration. Spine. 1998;23(23):2477–85.
- Sun Y, et al. Characterization of nucleus pulposus-like tissue formed in vitro. J Orthop Res. 2001;19(6): 1078–84.
- Bibby SR, et al. The pathophysiology of the intervertebral disc. Joint Bone Spine. 2001;68(6):537–42.
- Steinmann JC, Herkowitz HN. Pseudarthrosis of the spine. Clin Orthop Relat Res. 1992;284:80–90.
- Flynn JC, Hoque MA. Anterior fusion of the lumbar spine. End-result study with long-term follow-up. J Bone Joint Surg Am. 1979;61(8):1143–50.

- Waddell G, et al. Failed lumbar disc surgery and repeat surgery following industrial injuries. J Bone Joint Surg Am. 1979;61(2):201–7.
- West 3rd JL, Bradford DS, Ogilvie JW. Results of spinal arthrodesis with pedicle screw-plate fixation. J Bone Joint Surg Am. 1991;73(8):1179–84.
- Doers TM, Kang JD. The biomechanics and biochemistry of disc degeneration. Curr Opin Orthop. 1999;10: 117–21.
- Gruber HE, Hanley Jr EN. Analysis of aging and degeneration of the human intervertebral disc. Comparison of surgical specimens with normal controls. Spine (Phila Pa 1976). 1998;23(7):751–7.
- Lauerman WC, et al. Results of lumbar pseudarthrosis repair. J Spinal Disord. 1992;5(2):149–57.
- Norwig J, et al. Integrated isolator technology-based sterile production of cell-based drugs. Pharm Ind. 2000;63:780–4.
- Antoniou J, et al. The human lumbar intervertebral disc: evidence for changes in the biosynthesis and denaturation of the extracellular matrix with growth, maturation, ageing, and degeneration. J Clin Invest. 1996;98(4):996–1003.
- Beard HK, Roberts S, O'Brien JP. Immunofluorescent staining for collagen and proteoglycan in normal and scoliotic intervertebral discs. J Bone Joint Surg Br. 1981;63B(4):529–34.
- Beard HK, et al. Immunochemical localization of collagen types and proteoglycan in pig intervertebral discs. Immunology. 1980;41(2):491–501.
- Eyre DR. Collagens of the disc. In: Ghosh P, editor. The biology of the intervertebral disc. Boca Raton: CRC Press; 1988. p. 171–88.
- Roberts S, et al. Does the thickness of the vertebral subchondral bone reflect the composition of the intervertebral disc? Eur Spine J. 1997;6(6): 385–9.
- 21. Roberts S, et al. Type III collagen in the intervertebral disc. Histochem J. 1991;23(11–12):503–8.
- Wu JJ, Eyre DR, Slayter HS. Type VI collagen of the intervertebral disc. Biochemical and electronmicroscopic characterization of the native protein. Biochem J. 1987;248(2):373–81.
- Malko JA, Hutton WC, Fajman WA. An in vivo MRI study of the changes in volume (and fluid content) of the lumbar intervertebral disc after overnight bed rest and during an 8-hour walking protocol. J Spinal Disord Tech. 2002;15(2):157–63.
- 24. Buckwalter JA. Aging and degeneration of the human intervertebral disc. Spine. 1995;20(11):1307–14.
- Maroudas A, et al. Factors involved in the nutrition of the human lumbar intervertebral disc: cellularity and diffusion of glucose in vitro. J Anat. 1975;120(Pt 1):113–30.
- Nerlich AG, Schleicher ED, Boos N. 1997 Volvo Award winner in basic science studies. Immunohistologic markers for age-related changes of human lumbar intervertebral discs. Spine. 1997;22(24): 2781–95.

- Thompson JP, et al. Preliminary evaluation of a scheme for grading the gross morphology of the human intervertebral disc. Spine. 1990;15(5):411–5.
- Gruber HE, et al. Autologous intervertebral disc cell implantation: a model using Psammomys obesus, the sand rat. Spine. 2002;27(15):1626–33.
- Ganey T, et al. Disc chondrocyte transplantation in a canine model: a treatment for degenerated or damaged intervertebral disc. Spine. 2003;28(23): 2609–20.
- Okuma M, et al. Reinsertion of stimulated nucleus pulposus cells retards intervertebral disc degeneration: an in vitro and in vivo experimental study. J Orthop Res. 2000;18(6):988–97.
- Osti OL, Vernon-Roberts B, Fraser RD. 1990 Volvo Award in experimental studies. Anulus tears and intervertebral disc degeneration. An experimental study using an animal model. Spine. 1990;15(8): 762–7.
- 32. Prolo DJ, Oklund SA, Butcher M. Toward uniformity in evaluating results of lumbar spine operations. A paradigm applied to posterior lumbar interbody fusions. Spine. 1986;11(6):601–6.
- Ganey TM, Meisel HJ. A potential role for cell-based therapeutics in the treatment of intervertebral disc herniation. Eur Spine J. 2002;11 Suppl 2:S206–14.

- Strem BM, et al. Multipotential differentiation of adipose tissue-derived stem cells. Keio J Med. 2005; 54(3):132–41.
- Zuk PA, et al. Human adipose tissue is a source of multipotent stem cells. Mol Biol Cell. 2002;13(12): 4279–95.
- Katz AJ, et al. Emerging approaches to the tissue engineering of fat. Clin Plast Surg. 1999;26(4): 587–603. viii.
- Pfirrmann CW, et al. Magnetic resonance classification of lumbar intervertebral disc degeneration. Spine. 2001;26(17):1873–8.
- Ganey T, et al. Intervertebral disc repair using adipose tissue-derived stem and regenerative cells: experiments in a canine model. Spine (Phila Pa 1976). 2009;34(21):2297–304.
- Hohaus C, et al. Cell transplantation in lumbar spine disc degeneration disease. Eur Spine J. 2008;17 Suppl 4:492–503.
- Orozco L, et al. Intervertebral disc repair by autologous mesenchymal bone marrow cells: a pilot study. Transplantation. 2011;92(7):822–8.
- Coric D, et al. Prospective study of disc repair with allogeneic chondrocytes presented at the 2012 Joint Spine Section Meeting. J Neurosurg Spine. 2013; 18(1):85–95.

## Spinal Robotics: Present Indications and Trends

40

Konstantin Hockel, Gottlieb Maier, Marcos Tatagiba, and Florian Roser

### 40.1 Introduction: Image-Guided Surgery

The success of spinal surgery is highly dependent on accuracy. Surgeons are aware of the close local relations when manipulating the anatomical structures of the spine, i.e., vascular and neuronal structures, bone, ligaments, and joint space, which eventually determines the clinical outcome.

With conventional open-spine surgery, frequently consisting of decompression and stabilization, the target structures and trajectories are identified by exposure of the surface anatomy in combination with 2D fluoroscopy. Not only high-grade degenerative disease, spinal deformities, and revision surgery but also the increasing

K. Hockel, MD • M. Tatagiba, MD

Neurosurgical Department,

University Hospital Tuebingen, Tuebingen, Germany e-mail: Konstantin.Hockel@med.uni-tuebingen.de; marcos.tatagiba@med.uni-tuebingen.de

G. Maier, MD Department of Neurosurgery, Klinikum Stuttgart, Stuttgart, Germany e-mail: gottlieb.maier@googlemail.com

F. Roser, MD (⊠) Department of Neurosurgery, Cleveland Clinic Abu Dhabi, Neurological Institute, Abu Dhabi, United Arab Emirates e-mail: f.roser@gmx.de, roserf@clevelandclinicabudhabi.ae numbers of minimal invasive techniques create a challenge for the anatomical orientation, even for the experienced surgeon.

There exist several computer-assisted image guidance systems that help to visualize the anatomical relations in a 3D environment and facilitate the accurate placement of spinal instrumentation [1-3]. Such systems typically improve the accuracy of screw placement while reducing the need for extensive fluoroscopic imaging [4]. Several manufacturers have introduced computer-assisted navigation systems designed to increase implant placement accuracy in spine surgery.

A meta-analysis published by Kosmopoulos and Schizas, covering 37,337 pedicle screw implants in total, determined in a subgroup of in vivo implants a median accuracy of 95.1 % and 90.3 % with and without the assistance of navigation, respectively (15,358 screws) [1]. Verma et al. were determined to assess the actual functional benefit for the patient by applying computer-assisted navigation [3]. Their evaluation of 23 studies confirmed the superior accuracy of navigation over conventional pedicle screw placement. On the other hand, no statistical significant difference could be found for complication rates and the study was unable to make conclusion regarding functional outcome.

In recent years, the implementation of intraoperative imaging has further refined the possibilities and helped to overcome some drawbacks of

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_40

image-guided navigation for spinal instrumentation. Intraoperative 3D fluoroscopy and computed tomography (CT) increases accuracy of navigation system and facilitates the registration process [5]. While the benefits of intraoperative image-guided surgery are evident, there is further demand for alternative active assistance in the surgical workflow of spinal procedures.

### 40.2 Surgical Robots for Spinal Procedures

In many ways, spine surgery is ideally suited for the integration of robot-assisted surgical procedures. Although navigation assists the surgeon in display of trajectories in a complex 3D environment, the surgical task, e.g., cannulating the pedicle, has traditionally been carried out "free hand." This manipulation has to be performed in close proximity to critical structures that are more frequently accessed through minimally invasive approaches. A robotic interface actually represents the so far missing link between preoperative imaging, trajectory planning, and surgical execution. Thereby, it can significantly improve microsurgical dexterity and perform repetitive tasks with precision and reproducible results [6, 7].

A variety of surgical robots for different applications have been introduced, e.g., general surgery, cardiac surgery, urology, and spine surgery. Surgical robots can be classified into three broad categories, varying in their degree of automation [8]: (1) supervisory-controlled systems in which the surgeon plans the operation offline, specifying the motions that the robot must follow to perform the operation, and the robot then performs the procedure autonomously with the surgeon closely supervising; (2) telesurgical systems that allow the surgeon to directly control the surgical instruments held by the robot via a joystick or hand controls in which task execution can be either passive or active; and (3) shared-control systems that allow both the surgeon and the robot to directly control the surgical instrument at the same time.

Only a handful of robotic systems are specifically designed for spinal applications [9–15]. The Miro system (German Aerospace Center (DLR), Cologne, Germany), comprised of a robotic arm and an optical tracking system, positions a drill holder, through which the surgeon performs the procedure, has been evaluated for the placement of pedicle screws [10]. The Georgetown Robot (Johns Hopkins University, Baltimore, MD, USA), introduced in 2002, was designed as a percutaneous needle driver for minimally invasive spine procedures under biplane fluoroscopic guidance [15].

In addition, there are two readily commercially available robotic systems, i.e., the da Vinci (Intuitive Surgical, Sunnyvale, CA, USA) and the Renaissance/SpineAssist platform (Mazor Robotics Ltd., Israel). The teleoperated da Vinci enables the surgeon, who operates the control handles, to translate his hand movements via a robotic arm to the surgical field, facilitating complex manipulations, motion scaling, and tremor reduction. In spine surgery, it has been implemented in animal studies performing anterior lumbar interbody fusion, paraspinal schwannoma resection, laminotomy, disk incision, and dural suturing procedures [12–14, 16]. Thus far, this system was evaluated in humans for transoral odontoidectomy and for decompression of the craniocervical junction [11, 17]. The da Vinci system offers improved visualization and dissection of soft tissue structures [18] in deep cavities (e.g., transthoracic approaches) but lacks bonehandling instruments.

Today, the Renaissance/SpineAssist Surgical Guidance Robot available for routine clinical use in spinal procedures resembles a semiautonomous, shared-control system. The robot is a cylindrical  $(50 \times 90 \text{ mm})$  device equipped with an end effector with  $6^{\circ}$  of freedom. The bonemounting platform interfaces the robotic device with the patient's skeleton, and the robotic arm is positioned according to preplanned, imageguided trajectories, and the surgeon directs the instruments (e.g., drill guide, cannulas) along the predefined path. Applying this concept, the device is not only applicable for spinal instrumentation but also for collecting biopsies, tumor excisions, cement augmentations, and extraforaminal disk prolapses in distorted anatomic spaces [19–24].

## 40.3 Present Data for the Use of Robotic Systems in the Lumbar Spine

#### 40.3.1 Instrumentation Accuracy and Radiation Exposure

So far, a clinical evaluation of surgical robots in treatment concepts of the lumbar spine has only been performed for the SpineAssist platform. Since 2005, numerous cadaveric and clinical studies on the SpineAssist system have been evaluating the technical feasibility, radiation exposure, and the accuracy of pedicle screw instrumentation at the thoracic, lumbar, and sacral spine [20, 23, 25–31].

A recent meta-analysis by Marcus et al. evaluated all available studies, including two randomized, two cohort studies, and one cadaver study [20, 21, 25–27], comparing pedicle screw placement with robot assistance vs. fluoroscopy guidance [32]. Among the total of 1,308 pedicle screws, 729 were instrumented robot assisted and 579 under fluoroscopy guidance with a satisfactory position of 94.1 % and 92.7 %, respectively.

A retrospective, multicenter evaluation of 3,271 SpineAssist-supported spinal implantations between 2005 and 2009, involving a large percentage of pediatric scoliosis patients, reported a clinically acceptable placement rate of 98 % assessed by intraoperative fluoroscopy [19]. Postoperative CT-based evaluation of a subset of patients included 646 pedicle screws and demonstrated an accuracy of 98.3 % according to Gertzbein and Robbins A and B criteria [33], with a mean axial and sagittal plane deviation of  $1.2 \pm 1.49$  mm and  $1.1 \pm 1.15$  mm, respectively. Only two screws deviated >4 mm from the pedicle wall, but without irreversible neurological deficits reported.

Pechlivanis et al. in their prospective study report successful integration of the device in 31 cases, with 98.5 % of screws demonstrating axial and longitudinal accuracy, i.e., deviations less than 2 mm [34]. Another case series evaluated a total of 960 pedicle screws in 102 patients, primarily presenting spinal deformities and or revision surgeries. Accurate positioning was determined for 98.9 % of the screws [22]. Eleven screws were considered misplaced, of which ten screws were manually corrected during surgery. One patient required implant removal within 3 days of surgery as a result of radiculopathy. Screw misplacement was presumably caused by tool "skiving," i.e., the tip of the drill holder or guiding tool skids off the intended entry point leading to an aberrant trajectory.

Similar aspects for inaccurate screw placement have been pointed out recently by Ringel et al. [25]. In contrast to all other comparative data, this randomized, prospective study assessed screw accuracy in favor of freehand fluoroscopyguided over robot-assisted instrumentation with a satisfactory placement rate (Gertzbein and Robbins A and B on postoperative CT) of 93 and 85 %, respectively. They concluded that besides displacement of the entry point by soft tissue pressure, an unstable attachment of the robot to the spine may have contributed to the assessed inaccuracies.

During conventional fluoroscopy-guided pedicle instrumentation, the patient and the surgical team are exposed to a significant amount of radiation. Several clinical studies have shown that robot-assisted instrumentation decreases the occupational risk for surgeons and operating room staff members by significantly reducing intraoperative radiation doses, particularly in minimally invasive surgery (MIS) procedures [21, 27, 35].

A retrospective cohort analysis comparing conventional open to robotic-guided open/percutaneous pedicle screw insertions measured up to 70 % less radiation exposure in the robot-guided procedures, 64 % of which were performed using percutaneous approaches, when compared to 57 freehand surgeries, all executed using an open approach [21].

Similarly, in a prospective, randomized comparative trial of freehand, navigation-guided, and robotic-guided spinal procedures reports a twofold reduction in radiation time and dose when integrating the robotic system to insert pedicle screws in a minimally invasive approach, as compared to freehand surgeries in a conventional open approach [20]. In another single-center study of conventional versus robotic-assisted percutaneous spine fusion surgeries, a median 40 % lower radiation exposure was measured for the robot cohort [24].

Upon implementation of SpineAssist in a controlled, cadaveric implantation trial, overall radiation exposure among 87 % of the surgeons was below 1 mrem, versus a mean 136 mrem exposure when using traditional surgical approaches [27]. When calculating the average radiation per screw, a 98.2 % reduction in ionization exposure was observed in robot-guided procedures, with an average 0.2 mrem per screw, versus a mean 10.1 mrem per screw in the control group. Fluoroscopy time was reduced by a similar degree, from a mean 33.0 s per screw in the control group and 0.7 s per screw.

#### 40.3.2 Surgical Robots for Minimally Invasive Surgery

The present experience confirms a high level of accuracy for robot-assisted pedicle instrumentation for both the conventional open and percutaneous approach. In "regular" open cases in which the anatomy is clearly exposed, the advantage is less significant and the extra input might not be justified. Whereas in high-grade degenerative disease, spondylolysis, revision surgery, and in deformity cases, visualization of surface anatomy alone may not be sufficient, and robotic guidance can be helpful. Therefore, the system has clear advantages in percutaneous or MIS procedures [21]. The multicenter evaluation of robot-guided cases by Devito, in which a significant subgroup had complex deformities, already showed that 49 % of the screws in their study were placed in a percutaneous approach [19].

In general, MIS procedures in spine surgery have gained more acceptance and are applied more frequently as they are associated with less postoperative pain, lower infection rates, less blood loss and less paraspinal muscle trauma, and reduced recovery period and tissue scarring [36– 41]. On the other hand, MIS procedures are usually associated with increased operating time and have been reported to expose patients and surgeons to high radiation doses [21, 27, 35, 42, 43].

For instance, interbody fusion techniques represent a reasonable strategy for the management of lumbar degeneration, in isthmic spondylolisthesis, postlaminectomy situations, and pseudoarthrosis [44, 45]. Particularly the TLIF (transforaminal lumbar interbody fusion) is particularly suited for minimal invasive or percutaneous approaches due to its posterior, unilateral access to the spine [45]. Numerous comparative studies, MIS TLIF versus open TLIF, have been performed in recent years and results are fairly homogeneous [42, 43, 46-50]. The clinical outcome reported was generally favorable with special highlight on reduced postoperative pain, reduced intraoperative blood loss, and shortened hospital stay. A meta-analysis by Karikari et al. quantified the blood loss from 150 to 456 mL for MIS TLIF and 366.8-1,147 mL for open TLIF. The duration of postoperative hospital stay ranged in the MIS TLIF from 3.0 to 10.6 days and from 4.2 to 14.6 days for open TLIF [41]. The clinical outcomes are ranging from favorable to excellent (assessed by VAS and ODI), especially during the immediate postoperative course [48], but long-term outcome is not significantly better for MIS than for open TLIF in most studies [42, 47, 49]. Whereas studies from Peng et al. and Wang et al. found both operating time and radiation exposure to be increased in MIS TLIF procedures [42, 43]. Concerns whether equally solid bony fusion is feasible in MIS TLIF, in which preparation of the disk space through narrow corridors is more challenging, were invalidated by numerous studies [46, 47]. Recently, Wu et al. reported in a meta-analysis a fusion rate of 94.8 % and 90.9 % for MIS versus open TLIF [51].

Therefore, the integration of a robotic platform in a minimal invasive TLIF procedure is reasonable. We evaluated a heretofore unpublished case series of 28 patients with monosegmental lumbar disk disease, composed of either isthmic or degenerative spondylolisthesis (Table 40.1). The study was initiated after the authors had gathered experience with the

Mean age	$55 \pm 10$ yea	ar	
Sex	Female $n=16$	Male $n=12$	
Spondylolisthesis	Isthmic $n=15$	Degenerative $n=13$	
Meyerding	Grade 0 n=3	Grade I n=16	Grade II $n=9$
TLIF segments	L3/4 n=3	L4/5 n=12	L5/S1 n=13
Postop Meyerding	Grade 0 n=15	Grade I n=12	Grade II $n=1$
Clinical improvement (VAS 3–6 months)	Radicular pain 95 %	Back pain 60 %	

**Table 40.1** Clinical characteristics in robot-assisted MIS TLIF (n=28)

SpineAssist system in pedicle screw placement in about 50 cases. The primary objective was to integrate the robotic platform in a completely minimally invasive and more complex procedure with attention on surgical technique and workflow, operating time, radiation exposure, and clinical patient outcome.

#### 40.3.3 Surgical Technique

The surgical workflow involves acquiring a CT of the patient's spine for the preoperative planning stage (Figs. 40.1 and 40.2). In the operating room, the robotic platform is attached to the patient's spine involving pins to the spinous process and to the iliac crest or to the OR table. Two fluoroscopic images (anteroposterior and 60° oblique to the lateral plane) of the spine and the robotic platform (marked by a 3D fiducial array) are used for registration and matching to the preoperative CT. Following 3D synchronization, the robot is attached to the platform and the robotic arm is dispatched to the calculated trajectories. After stab skin incision (1-1.5 cm) and robotassisted k-wire insertion, the four pedicle screws are placed percutaneously followed by removal of the robotic system. In the majority of cases, the rod construct is inserted and temporarily tightened. A small midline incision (about 3 cm) is followed by subperiostal placement of the retractor system to gain access to the ipsilateral lamina and facet joint. Microsurgical facetectomy is performed using high-speed burr and rongeurs. Bone may be harvested for graft material. If indicated, ipsilateral and/or bilateral decompression of the dural sac and the exiting nerve root is feasible. Eventually, lateral exposure of the annulus allows for meticulous preparation of the disk space, and implantation of a TLIF cage was performed under fluoroscopic control. This is followed by compression of the screw-rod system and final tightening of the screw caps.

Postoperative CT scans were assessed for screw placement accuracy. Ninety-six percent of the overall 120 screws met Gertzbein and Robbins criteria A and B. Misplaced screws (grades C and D, three screws) were located at the lateral pedicle wall but were not associated with neurological deficits as deviance was to the lateral superior aspect of the pedicle. The overall mean operation time was  $165 \pm 30$  min and mean blood loss was assessed to be  $162 \pm 99$  ml, which are on the lower range of the previously reported MIS procedures. Under robot guidance, a mean of 17±7 min was needed for k-wire placement and further  $15\pm8$  min for final screw insertion. Radiation time per screw was assessed  $4.1 \pm 1.9$  s and radiation dose was  $3.4 \pm 2.1$  mGy per screw. Observed complications included one patient with postoperative epidural hematoma that needed evacuation and one patient with a L5 syndrome postoperatively that was not related to screw misplacement. No infections were observed.

### 40.4 Discussion and Future Outlook

The literature is replete with reports of spinal surgery outcomes, yet varies widely in the diagnostic criteria and methods employed to judge clinical efficacy. Image-guided navigation in spinal surgery has demonstrated its feasibility and qualities for spinal instrumentation in numerous studies, but lacks sufficiently powered data to prove the benefit on clinical outcome.



**Fig. 40.1** Surgical workflow for SpineAssist. (a) Referencing of the spine-fixated platform prior to robot assembly with marker array and fluoroscopy. (b) Robot-guided trajectory of the percutaneous drill cannula. (c)

Intraoperative lateral fluoroscopy with k-wires in L4 and drill cannula directed to L5. (d) Inserted k-wires before screw placement

Correspondingly, the available, mainly observational, data established a superior accuracy for robot-assisted pedicle instrumentation in the thoracic, lumbar, and sacral spine to conventional surgery along with a reduced radiation exposure. The only two RCTs, however, show inconsistent results [20, 25]. Here, further prospectively and randomized controlled data on accuracy and especially patient outcome is still needed for the evaluation of robotic guidance systems. Besides, clinical studies have to identify treatment strategies and surgical procedures where robotic systems can realize their potentials best. Minimally invasive and percutaneous procedures represent a reasonable field of application. A growing number of technological advancements have been implemented in this field to enhance the possible benefits of MIS spine surgery, i.e., reduced soft tissue trauma and recovery period, reduced complication rates, and improved



**Fig. 40.2** Example case of a 54-year-old isthmic L5/S1 spondylolisthesis grade I for MIS TLIF using SpineAssist. (a) Preoperative sagittal CT scans. (b) Pedicle screws, placed via SpineAssist on coronal CT scan. (c) Postoperative

axial CT scans showing the TLIF cage after right lateral decompression and facetectomy. (d) Postoperative (3 months) skin scarring of 3 cm midline incision and lateral stab incision for percutaneous pedicle screws

outcome. The integration of robotic systems could in addition reduce radiation exposure and accelerate the workflow.

Surgery of the cervicothoracic junction and cervical spine, in which the need for accuracy is crucial, is likewise dependent on fixation implants. The target bone volume is much more delicate and the critical anatomical structures, i.e., the vertebral arteries, the spinal cord, and nerve roots, are in close proximity and their violation is related with significant morbidity. The preliminary application of SpineAssist devices at the cervical spine clearly demonstrated the limitations as the cervical spine displays a high intersegmental mobility and even more vulnerable structures. However, robotic assistance may be helpful in revision cases, which will confront surgeons more frequently due to the increasing number of spinal instrumentation. The altered anatomy and scar tissue in a previously operated situs make orientation more difficult and bears the risk for further complications.

Despite strong evidence for the accuracy of robotic systems, continuous refinement is necessary. For the SpineAssist system, for example, technical hardware and software solutions are meant to enhance the stability of the mounted platform, to minimize the influence of tool deviation due to skiving or soft tissue pressure and to reduce the rate of conversion to conventional manual instrumentation, which in some reports lies between 10 % and 16 % [19, 22].

Future advancements, e.g., including a range of different surgical tools, are conceivable and should be evaluated.

The available surgical robot technology does not only facilitate spinal instrumentation and guided bone work (SpineAssist) but has also demonstrated a high degree of dexterity for soft tissue dissection and preparation in non-openly exposed deep cavities (da Vinci). To combine these robotic philosophies would, at least in theory, potentiate their individual potential.

Integration in other upcoming technologies, e.g., intraoperative imaging modalities such as 3D fluoroscopy and intraoperative CT and MRI, would promote direct intraoperative planning and omit the preoperative preparation steps (imaging and planning), thus further condensing the workflow. Moreover, a gain in accuracy and direct onsite control over the robot-assisted procedures would be feasible.

To connect imaging studies, 3D anatomy, presurgical planning, and in vivo execution through a computerized platform create a new interface that gives essential support but also requires special attention. So far, in a shared-control and hands-on robotic system, the surgeon is the eventual executing factor of the process, which allows but also demands for a continuous reinsurance and validity check of each step. This dichotomy is further continued in the issue to what extent an extension of robotic automation is possible and justifiable. For instance, the tasks of drilling the pedicle and driving-in the screw implants into the sometimes fragile bony structures require tight control of the trajectory axis, which could certainly be performed with high accuracy and precision by a robotic arm. Whether surgeons should leave these critical steps, where the bony anchorage of the implant is prepared and perceived, to the robot computer is debatable.

The question whether the surgeon enables a robot to completely perform delicate positioning of screws remains unanswered. Even markedly reduced, there is an immanent risk of nerve lesion due to the robot, a responsibility bearded solely by the surgeon. The more robots are autonomous, the more likely it is that the responsibility seems to drift away from the surgeon. However, the decision-making process of the surgeon is completely different than that of the robot, as the surgeon additionally incorporates patient's history, his personal experiences, and intuition to the case.

The financial investment in a surgical robot or other computerized surgical systems is substantial. While navigation systems have been introduced over two decades ago and have proved to produce superior results, only 11 % of spine surgeons responding to a survey reported using such technology in their surgical protocols [52]. Costefficiency is a critical factor for the appraisal and adoption of new technologies in the clinical setting, and expectations will increase further. If the clinical value of computer-assisted spinal surgery, for example, in MIS TLIF surgery, which is promising but has to be further validated, consists in significantly reduced hospital stay, reduced reconvalescence and postoperative pain, and thus improved patient outcome, general acceptance and financial compensation will escalate.

Moreover, the growing acceptance of computer-assisted systems, to the degree of newly defined standards of care, will have medicolegal implications for spinal surgery as well. Although experienced spine surgeons achieve high accuracy and excellent results in conventional freehand techniques, the application of these technologies facilitates a detailed documentation of every single step of a surgical procedure, starting from preoperative imaging studies and planning, referencing of the virtual anatomy to the surgical situs, execution according to predefined pathways, and eventually confirmation by intraoperative imaging modalities.

It has become obvious that computer-assisted guidance technology, navigation, and robotics have a significant learning curve for surgeons and OR staff [53]. The preoperative planning, the robot assembly, and referencing do consume extra time that can only be minimized by sufficient training and experience. To reserve the application of these systems for the few really demanding cases seems not advisable. Only use on a daily basis, i.e., for standard instrumentation as well, will provide the necessary experience to enable rapid workflows, short operating times, low complication rates, and eventually satisfactory outcome.

#### References

- Kosmopoulos V, Schizas C. Pedicle screw placement accuracy: a meta-analysis. Spine (Phila Pa 1976). 2007;32:E111–20.
- Tian N-F, Huang Q-S, Zhou P, Zhou Y, Wu R-K, Lou Y, et al. Pedicle screw insertion accuracy with different assisted methods: a systematic review and metaanalysis of comparative studies. Eur Spine J. 2011;20:846–59.
- Verma R, Krishan S, Haendlmayer K, Mohsen A. Functional outcome of computer-assisted spinal pedicle screw placement: a systematic review and meta-analysis of 23 studies including 5,992 pedicle screws. Eur Spine J. 2010;19:370–5.
- Gebhard FT, Kraus MD, Schneider E, Liener UC, Kinzl L, Arand M. Does computer-assisted spine surgery reduce intraoperative radiation doses? Spine (Phila Pa 1976). 2006;31:2024–7; discussion 2028.
- Scheufler K-M, Franke J, Eckardt A, Dohmen H. Accuracy of image-guided pedicle screw placement using intraoperative computed tomography-based navigation with automated referencing. Part II: thoracolumbar spine. Neurosurgery. 2011;69:1307–16.
- Kelly PJ. Neurosurgical robotics. Clin Neurosurg. 2002;49:136–58.
- Louw DF, Fielding T, McBeth PB, Gregoris D, Newhook P, Sutherland GR. Surgical robotics: a review and neurosurgical prototype development. Neurosurgery. 2004;54:525–36; discussion 536–7.
- Nathoo N, Cavuşoğlu MC, Vogelbaum MA, Barnett GH. In touch with robotics: neurosurgery for the future. Neurosurgery. 2005;56:421–33; discussion 421–33.
- Bertelsen A, Melo J, Sánchez E, Borro D. A review of surgical robots for spinal interventions. Int J Med Robot. 2013;9:407–22.
- Ortmaier T, Weiss H, Döbele S, Schreiber U. Experiments on robot-assisted navigated drilling and

milling of bones for pedicle screw placement. Int J Med Robot. 2006;2:350–63.

- Lee JYK, Lega B, Bhowmick D, Newman JG, O'Malley BW, Weinstein GS, et al. Da Vinci robotassisted transoral odontoidectomy for basilar invagination. ORL J Otorhinolaryngol Relat Spec. 2010; 72:91–5.
- Kim MJ, Ha Y, Yang MS, Yoon DH, Kim KN, Kim H, et al. Robot-assisted anterior lumbar interbody fusion (ALIF) using retroperitoneal approach. Acta Neurochir (Wien). 2010;152:675–9.
- Yang MS, Yoon DH, Kim KN, Kim H, Yang JW, Yi S, et al. Robot-assisted anterior lumbar interbody fusion in a swine model in vivo test of the Da Vinci surgicalassisted spinal surgery system. Spine (Phila Pa 1976). 2011;36:E139–43.
- Ponnusamy K, Chewning S, Mohr C. Robotic approaches to the posterior spine. Spine (Phila Pa 1976). 2009;34:2104–9.
- Cleary K, Watson V, Lindisch D, Taylor RH, Fichtinger G, Xu S, et al. Precision placement of instruments for minimally invasive procedures using a "needle driver" robot. Int J Med Robot. 2005;1:40–7.
- Perez-Cruet MJ, Welsh RJ, Hussain NS, Begun EM, Lin J, Park P. Use of the da Vinci minimally invasive robotic system for resection of a complicated paraspinal schwannoma with thoracic extension: case report. Neurosurgery. 2012;71:209–14.
- Lee JYK, O'Malley BW, Newman JG, Weinstein GS, Lega B, Diaz J, et al. Transoral robotic surgery of craniocervical junction and atlantoaxial spine: a cadaveric study. J Neurosurg Spine. 2010;12:13–8.
- Yang MS, Kim KN, Yoon DH, Pennant W, Ha Y. Robot-assisted resection of paraspinal Schwannoma. J Korean Med Sci. 2011;26:150–3.
- Devito DP, Kaplan L, Dietl R, Pfeiffer M, Horne D, Silberstein B, et al. Clinical acceptance and accuracy assessment of spinal implants guided with SpineAssist surgical robot: retrospective study. Spine (Phila Pa 1976). 2010;35:2109–15.
- Roser F, Tatagiba M, Maier G. Spinal robotics: current applications and future perspectives. Neurosurgery. 2013;72 Suppl 1:12–8.
- 21. Kantelhardt SR, Martinez R, Baerwinkel S, Burger R, Giese A, Rohde V. Perioperative course and accuracy of screw positioning in conventional, open roboticguided and percutaneous robotic-guided, pedicle screw placement. Eur Spine J. 2011;20:860–8.
- Hu X, Ohnmeiss DD, Lieberman IH. Robotic-assisted pedicle screw placement: lessons learned from the first 102 patients. Eur Spine J. 2013;22:661–6.
- Togawa D, Kayanja MM, Reinhardt MK, Shoham M, Balter A, Friedlander A, et al. Bone-mounted miniature robotic guidance for pedicle screw and translaminar facet screw placement: part 2 – evaluation of system accuracy. Neurosurgery. 2007;60:ONS129– 39; discussion ONS139.
- Schoenmayr R, Kim I-S. Why do I use and recommend the use of navigation? Argo Spine News J. 2011;22:132–5.

- 25. Ringel F, Stüer C, Reinke A, Preuss A, Behr M, Auer F, et al. Accuracy of robot-assisted placement of lumbar and sacral pedicle screws: a prospective randomized comparison to conventional freehand screw implantation. Spine (Phila Pa 1976). 2012;37:E496–501.
- Schizas C, Thein E, Kwiatkowski B, Kulik G. Pedicle screw insertion: robotic assistance versus conventional C-arm fluoroscopy. Acta Orthop Belg. 2012;78:240–5.
- Lieberman IH, Hardenbrook MA, Wang JC, Guyer RD. Assessment of pedicle screw placement accuracy, procedure time, and radiation exposure using a miniature robotic guidance system. J Spinal Disord Tech. 2012;25:241–8.
- Sukovich W, Brink-Danan S, Hardenbrook M. Miniature robotic guidance for pedicle screw placement in posterior spinal fusion: early clinical experience with the SpineAssist. Int J Med Robot. 2006;2:114–22.
- Barzilay Y, Liebergall M. Robotic guidance for spine surgery—introduction of a novel system and analysis of challenges encountered during the clinical development phase at two spine centres. Med Robot. 2006;2:146–53.
- 30. Lieberman IH, Togawa D, Kayanja MM, Reinhardt MK, Friedlander A, Knoller N, et al. Bone-mounted miniature robotic guidance for pedicle screw and translaminar facet screw placement: part I-technical development and a test case result. Neurosurgery. 2006;59:641–50; discussion 641–50.
- Shoham M, Lieberman IH, Benzel EC, Togawa D, Zehavi E, Zilberstein B, et al. Robotic assisted spinal surgery–from concept to clinical practice. Comput Aided Surg. 2007;12:105–15.
- Marcus HJ, Cundy TP, Nandi D, Yang G-Z, Darzi A. Robot-assisted and fluoroscopy-guided pedicle screw placement: a systematic review. Eur Spine J. 2014;23:291–7.
- Gertzbein SD, Robbins SE. Accuracy of pedicular screw placement in vivo. Spine (Phila Pa 1976). 1990;15:11–4.
- 34. Pechlivanis I, Kiriyanthan G, Engelhardt M, Scholz M, Lücke S, Harders A, et al. Percutaneous placement of pedicle screws in the lumbar spine using a bone mounted miniature robotic system: first experiences and accuracy of screw placement. Spine (Phila Pa 1976). 2009;34:392–8.
- Rampersaud YR, Foley KT, Shen AC, Williams S, Solomito M. Radiation exposure to the spine surgeon during fluoroscopically assisted pedicle screw insertion. Spine (Phila Pa 1976). 2000;25:2637–45.
- O'Toole JE, Eichholz KM, Fessler RG. Surgical site infection rates after minimally invasive spinal surgery. J Neurosurg Spine. 2009;11:471–6.
- Wang MY, Lerner J, Lesko J, McGirt MJ. Acute hospital costs after minimally invasive versus open lumbar interbody fusion: data from a US national database with 6106 patients. J Spinal Disord Tech. 2012;25:324–8.
- German JW, Foley KT. Minimal access surgical techniques in the management of the painful lumbar

motion segment. Spine (Phila Pa 1976). 2005;30:S52–9.

- Foley KT, Holly LT, Schwender JD. Minimally invasive lumbar fusion. Spine (Phila Pa 1976). 2003;28:S26–35.
- Park P, Foley KT. Minimally invasive transforaminal lumbar interbody fusion with reduction of spondylolisthesis: technique and outcomes after a minimum of 2 years' follow-up. Neurosurg Focus. 2008;25:E16.
- Karikari IO, Isaacs RE. Minimally invasive transforaminal lumbar interbody fusion: a review of techniques and outcomes. Spine (Phila Pa 1976). 2010;35:S294–301.
- 42. Wang J, Zhou Y, Zhang ZF, Li CQ, Zheng WJ, Liu J. Comparison of one-level minimally invasive and open transforaminal lumbar interbody fusion in degenerative and isthmic spondylolisthesis grades 1 and 2. Eur Spine J. 2010;19:1780–4.
- 43. Peng CWB, Yue WM, Poh SY, Yeo W, Tan SB. Clinical and radiological outcomes of minimally invasive versus open transforaminal lumbar interbody fusion. Spine (Phila Pa 1976). 2009;34:1385–9.
- Weinstein JN, Lurie JD, Tosteson TD, Hanscom B, Tosteson ANA, Blood EA, et al. Surgical versus nonsurgical treatment for lumbar degenerative spondylolisthesis. N Engl J Med. 2007;356:2257–70.
- 45. Holly LT, Schwender JD, Rouben DP, Foley KT. Minimally invasive transforaminal lumbar interbody fusion: indications, technique, and complications. Neurosurg Focus. 2006;20:E6.
- 46. Schwender JD, Holly LT, Rouben DP, Foley KT. Minimally invasive transforaminal lumbar interbody fusion (TLIF): technical feasibility and initial results. J Spinal Disord Tech. 2005;18(Suppl):S1–6.
- Scheufler K-M, Dohmen H, Vougioukas VI. Percutaneous transforaminal lumbar interbody fusion for the treatment of degenerative lumbar instability. Neurosurgery. 2007;60:203–12; discussion 212–3.
- 48. Isaacs RE, Podichetty VK, Santiago P, Sandhu FA, Spears J, Kelly K, et al. Minimally invasive microendoscopy-assisted transforaminal lumbar interbody fusion with instrumentation. J Neurosurg Spine. 2005;3:98–105.
- Schizas C, Tzinieris N, Tsiridis E, Kosmopoulos V. Minimally invasive versus open transforaminal lumbar interbody fusion: evaluating initial experience. Int Orthop. 2009;33:1683–8.
- Shunwu F, Xing Z, Fengdong Z, Xiangqian F. Minimally invasive transforaminal lumbar interbody fusion for the treatment of degenerative lumbar diseases. Spine (Phila Pa 1976). 2010;35:1615–20.
- Wu RH, Fraser JF, Härtl R. Minimal access versus open transforaminal lumbar interbody fusion: metaanalysis of fusion rates. Spine (Phila Pa 1976). 2010;35:2273–81.
- Härtl R, Lam KS, Wang J, Korge A, Kandziora F, Audigé L. Worldwide survey on the use of navigation in spine surgery. World Neurosurg. 2013;79:162–72.
- Hu X, Lieberman IH. What is the learning curve for robotic-assisted pedicle screw placement in spine surgery? Clin Orthop Relat Res. 2013;22:661–6.

Part VII

**Nonfusion Technologies** 

# Disk Arthroplasty: A 30-Year History

41

## Klaus John Schnake and Frank Kandziora

#### 41.1 Introduction

While the first artificial disk was implanted just 30 years ago, the history of disk arthroplasty is already about 55 years old.

Arthroplasty has been quite successful for joints such as the hip and knee. In contrast, disk arthroplasty is far away from being the standard treatment in spine surgery. Degenerative lesions of the disk consist of a decrease in the hydrophilic properties of the nucleus as well as the appearance of annulus tears. Secondary, osteoarthritis of the facet joints and subchondral bone alterations occur. Thus, the entire functional spinal segment degenerates. While severe hip osteoarthritis can be treated with total hip replacement, successful replacement of a total spinal segment is still not possible.

In the past, surgeons attempted to develop arthroplasty-like implants in order to mimic physiologic motion. The primary purpose was to

F. Kandziora, MD, PhD

restore symptom-free biomechanical function. While there is almost no more indication to perform a hip or knee arthrodesis, it is still unclear if spinal arthrodesis will follow this pattern. Today, the increasing demands and expectations from the patients' side, and the medical desire to avoid fusion with its adverse side effects, have led to the development of multiple implants for disk arthroplasty.

Over the average human lifespan, the spine is exposed to more than 100 million cycles of motion. Per year, the lumbar spine is exposed to 2 million nonsignificant motions and 125.000 significant bends per year. The optimal lifetime for an arthroplasty spinal implant was therefore considered to be approximately 30 million cycles, or 10 million cycles in a fully loaded spine situation [1, 2].

The intervertebral disk is not a simple cartilaginous joint, but rather a complex anatomic structure allowing small movements along and around three main axes. In contrast to peripheral joints, whose stability is mainly achieved by ligaments, the disk on its own provides the major part of its stability. The configuration of nucleus and annulus allows high resistance to external forces.

The biomechanical requirements present quite a challenge for any implanted mechanical device, since the most elastomeric polymers will degrade, metals will wear, each will be exposed to the immune system, and open spaces will lead to tissue ingrowth. Continuous motion and stress

K.J. Schnake, MD (🖂)

Center for Spinal Therapy, Schön Klinik Nürnberg Fürth, Europa-Allee 1, 90762 Fürth, Germany e-mail: kschnake@schoen-kliniken.de

Center for Spinal Surgery and Neurotraumatology, BG-Unfallklinik Frankfurt am Main, Friedberger Landstraße 430, 60389 Frankfurt am Main, Germany e-mail: frank.kandziora@bgu-frankfurt.de

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_41

The development of disk prostheses was heavily influenced by design and engineering principles previously established for total joint arthroplasty. Strategies for immediate and delayed implant fixation have been taken from lessons learned with hip prostheses. The same is true for the use of cobalt chrome alloys and ultrahigh-molecular-weight polyethylene in a majority of devices. Titanium provides excellent biocompatibility and the advantage of MRI compatibility. Knowledge transfer from joint arthroplasty to the spine certainly offers advantages. However, the disk space is a different biomechanical and biologic environment compared to other joints. To complicate matters, the origin of low back pain is not fully understood yet and appears to be more complex than in peripheral joints.

Two key principles can be differentiated for disk arthroplasty [3]:

- Reproduction of the viscoelastic properties of the disk. Those implants are mostly manufactured from various silicones or polymers. Some rely on springs and/or piston systems and some are injected in monomer form and polymerised in situ. The main field of application is nucleus replacement.
- Reproduction of the motion characteristics of the disk. Those implants are usually mechanical devices made from metal and/or polyethylene couples. The main field of application is total disk replacement.

Innumerable different designs for both principles have been developed and patented so far. However, only a few of them reached the level of clinical application, and even fewer were implanted in a bigger number of patients. The story began in Europe in the 1950s with preclinical studies and patents. While today the negative aspects of spinal fusion serve as a rationale for disk arthroplasty, the rationale in the last century was restoration of joint biomechanics.

In 1955, van Steenbrugghe patented a joint replacement for the disk consisting of two

cushions [4]. In the late 1950s, Nachemson injected self-hardening liquid silicone rubber into cadaver disks [5].

However, the virtual clinical pioneer of disk arthroplasty was Fernström. He was the first to implant an "artificial disk" in the late 1950s. In fact, he was using a metal ball in an attempt to reproduce the "ball joint" mechanism of the disk [6].

The Fernström ball seems to have been used in about 250 patients. Because of poor results, the implant was withdrawn [7].

Probably the first attempt of a total disk replacement restoring the motion function was patented by Weber in 1978. The devices consisted of two polyethylene box-like structures anchored in the adjacent vertebrae. In between, a ceramic ovoid core was placed allowing motion. However, the device was never manufactured [8].

The modern history of disk arthroplasty began in the 1980s. Nucleus replacement strategies and total disk replacement were separated and both could reach clinical application in the subsequent course.

### 41.2 Nucleus Replacement

Nucleus replacement strategies reflect the earliest attempts to relieve pain and restore function of the degenerated spinal motion segment. The abovementioned Fernström ball was one of the first replacements. All of those early techniques were neither mechanical nor biological promising.

The history of nucleus replacement suffered from many setbacks. Surgeons and engineers ingenuity lead to numerous devices. The majority of them were either tried and failed or were never tried clinically at all. The early designs were made of virtually any material (metal or polymer) having flexibility, especially elastomeric (rubbery) materials or metal springs and hinges. Literally all of the early designs were mechanical devices. Typical problems of most of the implants were: surgical implanting technique, attachment to the bone, resistance to expulsion, mechanical failure loads, longevity, and tissue compatibility. Despite those problems, nucleus replacements have been used since the late 1990s.





This is probably due to the tremendous efforts of Charles D. Ray, the inventor of the prosthetic disk nucleus (PDN) (Fig. 41.1). He followed the idea to compare the degenerated disk with a flat car tyre. The nucleus loses water and shrinks and consecutively the disk height decreases like a tube of a tyre which deflates. As long as the annulus (respectively, the tyre) maintains its integrity, a restoration of function is possible if the pressure in the disk (respectively, the tube) can be increased again. Coming from this point, he developed a hydrogel which could be hydrated. The first prosthetic disk was patented in 1988 [9].

Ray gave his prosthetic disk nucleus a scientific background. To date, the most often used nucleus replacement is the PDN and PDN-SOLO (Raymedica, Bloomington, USA). The PDN consists of a hydrogel core and an ultra-highmolecular-weight polyethylene jacket. The hydrogel core is polyacrylonitrileа polyacrylamide multi-block copolymer with memory capability. It can absorb 50-90 % of its dry weight in water to become fully hydrated. The woven jackets provide dimensional control of the swollen pellets. Two PDN devices are usually implanted next to each other after removal of the nucleus [10].

Indications that have been proposed are low back pain due to degenerative disk disease, with or without leg pain [11].

While the theory of nucleus replacement with a swelling implant sounds auspicious, early clinical reports revealed high complication rates. Typical complications were implant dislocations and subsidence into the adjacent vertebral body [12, 13].

Obvious problems led to a new implant design, the PDN-SOLO®. Thus, complication rate dropped to less than 10 %, which is, in our opinion, still too high [14, 15]. Only few data concerning the long-term outcome are available showing persistent favourable clinical results. However, quality and quantity of data are too poor [16].

In the past, many other nucleus replacement strategies have been developed and are currently under investigation. Today, nucleus replacement devices can be functionally categorised as elastomeric and mechanical. The latter are in clinical use but are confronted with many of the same problems encountered with the PDN implant [17]. The surgical approach to the nucleus harms the integrity of the annulus. Thus, implanted nucleus devices can be extruded through the primary approach. The sewing of the annulus as well as lateral approaches to the disk did not eliminate this problem [18].

Elastomeric devices are under clinical investigation with encouraging preliminary results. Most of the actual strategies follow the path of injectable materials that undergo in situ polymerisation.

However, up to now, there is no evidence that nucleus replacement offers any clinical benefit. Furthermore, the indications are not clear yet.

In conclusion, further clinical investigation with prospective, randomised pivotal trials is needed to determine the efficacy of nucleus replacement in the treatment of lumbar degenerative disk disease [19, 20].

### 41.3 Total Disk Arthroplasty

The Charité artificial disk was the first available total disk replacement (TDR) system. Three different types have been developed so far. While type I and II Charité artificial disks were manufactured in the former German Democratic Republic (GDR) and, therefore, was never commercially available, the type III Charité disk was distributed by DePuy Spine (Raynham, Massachusetts, USA) until spring 2010.

The Charité type I was developed in East Berlin by Kurt Schellnack and Karin Büttner-Janz in the Charité Hospital in 1982. The idea was based on the biomechanically proven "low friction" principle, which had already been successful in total joint replacement. It consisted of two highly polished metal end plates with teeth for bony anchorage and an ultra-high-molecularweight polyethylene (UHMWPE) sliding core. The device was intended to imitate the movement of the nucleus pulposus of the intervertebral disk within its annular containment. It is therefore considered to be an unconstrained type of total disk replacement. An unconstrained design incorporates a mobile-bearing core and provides independent rotation and translation about three axes. The so-called SB Charité Artificial Disk Mark I was implanted at the Charité Hospital in the years 1984–1985 [21, 22].

Due to axial plane migrations, the device was modified to the SB Charité II.

It was used between 1985 and 1987. In contrast to type I, end plates were enlarged and exhibited bilateral wings to avoid subsidence. Finally, problems of migration and metal fatigue fractures led to the abandonment of type II also [22].

From 1987 on, the Link SB Charité III was manufactured by Waldemar Link GmbH in Hamburg, West Germany. The end plates were changed to cobalt-chromium-molybdenum alloy (CoCrMo) and received a porous coating of plasma-sprayed titanium and calcium phosphate to enhance osteointegration. The free-floating biconvex sliding core is still made of ultra-highmolecular-weight polyethylene (UHMWPE), encased by a metal wire for radiological marking. Primary stability is achieved by press-fit implantation through the teeth on the end plate which anchor into the subchondral bone. Different sizes and angulations were developed later [21].

After being taken over by DePuy Spine (Johnson & Johnson) in 2003, the SB Charité III was called Charité artificial disk until its abandonment (Fig. 41.2).

One of the earliest and most comprehensive clinical review regarding the Charité III was



Fig. 41.2 Charité artificial disk

published by LeMarie in 1997. After 51 months follow-up of 105 patients, he found 79 % excellent results and a return to work rate of 87 % [23].

The American Food and Drug Administration (FDA) approved the Charité artificial disk in October 2004, the first of its kind, for use in treating pain associated with degenerative disk disease (DDD). The device is intended to replace a diseased or damaged intervertebral disk at either the L4–L5 or L5–S1 level.

In a multicenter study, the artificial disk was performed in 205 patients who had been diagnosed with DDD and had failed to have their pain relieved after 6 months of nonsurgical therapy and compared them to 99 patients who received the control device (stand-alone BAK spinal fusion cage using bone graft).

The study showed that 2 years after surgery, patients treated with the artificial disk did no worse than patients treated with intervertebral body fusion. However, patient satisfaction was higher in the artificial disk group. The rates of adverse events from use of the artificial disk were similar to those from treatment with fusion. In addition, the study showed that there was no statistically significant relationship between motion at the level where the disk was implanted and the patient's relief from pain [24, 25]. To conclude, the patients treated with the Charité artificial disk were as good as the patients treated with a standalone cage. However, the latter is considered to be an obsolete treatment in Europe due to unsatisfactory clinical results.

The SB Charité III has the longest clinical follow-up of any TDR. It has been implanted more than 17.000 times worldwide. Following the implantation, most of the authors stated 60–90 % satisfactory to excellent results [26–28].

Despite the good results published otherwise, Michael Putzier from the Charité Berlin published long-term results after an average followup of 17 years. The deflating results revealed spontaneous ankylosis in 60 % and a reoperation rate of 11 % [29].

Interestingly, the last Charité artificial disk was already implanted in the Charité hospital in Berlin in 1989. Despite the fact that the implant carried the name of the biggest university hospital in Germany, it has not been used there since then. In the USA, criticism increased over time, and numerous patients started to sue Johnson & Johnson alleging the pharmaceutical company knew or should have known about the serious complications of the artificial disk surgery. As mentioned above, the Charité artificial disk is not available anymore.

Another early type of artificial disk was the AcroFlex (DePuy Spine), designed by Arthur Steffee. The first implant type consisted of a hexen-based polyolefin rubber cushion attached to two titanium end plates. A pilot study with six patients was published in 1993. Due to concerns regarding potential carcinogenicity of parts of the implant, the clinical trials were suspended. The next generation was proposed by Steffee, Fraser and co-workers. Debris and implant failure were the typical problems of the AcroFlex series [30, 31]. A prospective non-randomised study with almost 10 years follow-up revealed a cumulative survival of only 61 %. The authors concluded that further use of this implant is not justified [32].

Two other types of TDR were developed in the 1990s and showed promising results: the ProDisc and the Maverick artificial disk.

The first ProDisc (Aesculap, Tuttlingen, Germany) was developed by Thierry Marnay in Montpellier in the late 1980s [33].

The ProDisc has two metal (titanium) end plates which are plasma sprayed with titanium



Fig. 41.3 ProDisc-L

and have two vertical fins for fixation to the end plates. The core is made of high-density polyethylene and fits firmly to the inferior end plate. The superior surface of the core is formed convex. The resulting centre of rotation is fixed and located inferior to the disk space. This semiconstrained design includes a fixed axis of rotation that limits translation. This leads to increased stress within the device and at the device-bone interface, resulting in a potentially increased risk of implant loosening.

The first series of 64 patients were operated by Marnay between 1990 and 1993 with promising results. Follow-up ranged between 7 and 11 years with over 75 % excellent or good results. There was no outcome difference between 1 and 2 level implantations. All implants were still intact and functioning without signs of subsidence or migration [34].

The second generation of the ProDisc called ProDisc-L (Fig. 41.3) is distributed by Synthes (Paoli, Pennsylvania, USA). It was introduced to the market in 2000. Different end plate sizes and lordosis angles, different core heights and only one keel per end plate were the main design changes. It has been approved by the American FDA in August 2006 for one- or two-level implantation. The FDA IDE trial compared the ProDisc II with an anteroposterior fusion using femoral ring allografts anteriorly and pedicle screw fixation with autograft posteriorly. The clinical results in the ProDisc group were slightly better than in the fusion group [35].



Fig. 41.4 Maverick

In uncontrolled clinical series, success rates up to 90 % are reported [27, 36]. Long-term data are sparse. In a prospective, single-centre investigation, 181 patients (90 % follow-up rate) could be examined after a mean follow-up of 7.4 years. Clinical results improved significantly after surgery and 87 % of patients were (highly) satisfied at final follow-up. Complication rate was 14 %, about half of which were device related. Reoperation rate was 16 %. The authors concluded that for a carefully selected cohort of patients, results compare favourably to results achieved with fusion [37]. Similar mid- to longterm results have been published by other authors [38–40].

Mathews, Le Huec and co-workers conceived the semiconstrained Maverick artificial disk (Fig. 41.4) (Medtronic Sofamor Danek, Memphis, Tennessee, USA), a metal-on-metal (chrome cobalt) interface implant with a posterior rotation axis [41].

Metal-on-metal disk prostheses have been developed to eliminate polyethylene wear and its potential risks. Although polyethylene debris and osteolysis have not been proven to be significant clinical issues with TDR, the long-term effects remain unclear. As known from hip surgery, metal-on-metal surfaces produce far lesser debris than polyethylene-onmetal surfaces [27]. The Maverick artificial disk is fixated to the end plates with a midline sagittal fin like the ProDisc. The end plates are hydroxyapatite coated. The overall biomechanical profile is similar to the ProDisc. However, the convex caudal component has a slightly smaller radius of curvature compared with the concave superior component.

Like in the other TDR, different sizes, heights and angles are available.

The first implantation was performed in 2002 and a randomised FDA clinical trial was begun in 2003, comparing the Maverick artificial disk to a fusion using cage combined with BMP. LeHuec and co-workers published a prospective study reporting the outcome of 64 Maverick devices implanted between January 2002 and November 2003. The Oswestry score improved for 75 % of patients [42]. Published midterm data 4 years after surgery showed 85 % of patients were again working and 79 % took up their normal sports activities [43].

In the last 6 years, a variety of new implants and techniques have been developed so that nowadays nearly all companies offer implants for disk arthroplasty. Despite extensive biomechanical studies, it currently remains unclear which type of constraint offers the most advantages in total disk replacement. Constrained implants offer stability but lead to high stress on fixation, while unconstrained implants are sensitive to surgical positioning. The normal axis of rotation of the lumbar spine is not fixed but varies. Although a constrained device in good position may provide more controlled motion and consecutive preservation of the facet joints, an unconstrained device may provide greater range of motion and be more forgiving in terms of surgical positioning.

In the recent past, results from multicenter and prospective randomised studies without any potential conflict of interest have been published. Data from 240 patients from a Swiss registry including different types of TDR showed significant, clinically relevant and lasting reduction of pain until 5 years after surgery. Adjacent segment degeneration occurred in 10 % of patients. Revision rate was 4.4 %. Almost 87 % of operated segments remained mobile despite the fact that heterotopic ossifications could be detected in 44 % of patients [44].

In a Swedish prospective randomised controlled trial, 152 patients were either fused posteriorly or received TDR of different types. After 5 years, both groups still showed significant clinical benefit from surgery. However, all parameters were significantly better in the TDR group. No differences were found concerning complications and reoperations [45].

In a recent meta-analysis looking at results after 2-year follow-up, seven relevant RCTs with a total of 1,584 patients could be included. TDR was significantly more effective in ODI, VAS score, shorter duration of hospitalisation and a greater proportion of willing to choose the same operation again. All other parameters showed no significant difference [46].

Nevertheless, the authors of a Cochrane review judged the results slightly different: "Although statistically significant, the differences in clinical improvement were not beyond generally accepted boundaries for clinical relevance. Prevention of adjacent level disease and/or facet joint degeneration was not properly assessed. Therefore, because we think that harm and complications may occur after some years, the spine surgery community should be prudent to adopt this technology on a large scale, despite the fact that total disk replacement seems to be effective in treating low back pain in selected patients, and in the short term is at least equivalent to fusion surgery" [47].

Complications have been reported with all types of TDR and can be divided into two groups: those related to the surgical approach and those related to the prosthesis.

Access to the anterior lumbar spine includes potential injury to major vascular and visceral structures. Total complication rates range from 10 % to 23 %. Complications related directly to device implantation occur in 2.9–6.5 % of patients [26, 37, 44, 46].

Implant-related complications are design specific and appear as subsidence, dislocation and breakage of the prosthesis. Overall complication rates range from 2 % to 26 % of patients [26, 27, 37, 46, 48].

#### Conclusion

Thirty years of clinical application has not revealed a clear advantage of disk arthroplasty over fusion techniques. Even the prospective randomised FDA studies are arguable since the methods used in the control groups are considered to be arguably substandard treatment in some parts of the world. Conflicting results, increasing number of lawsuits and inconsistent demeanour of companies have lead to an increasing mistrust of both surgeons and patients in TDR. During the last decade, in the USA, surgical treatment for lumbar DDD has increased 2.4-fold. Although all fusion procedures significantly increased, TDR did not increase [49].

Therefore, some authors still consider total disk replacement as an experimental procedure [50].

On the other hand, much knowledge has been gained on TDR, and extensive database exist from numerous randomised and nonrandomised studies. The results are as good as with fusion. Complications and reoperations are similar with both techniques. Disk arthroplasty has opened a new era in spinal surgery and has gained a firm place in the operative portfolio of many surgeons. Many patients beyond the clinical trials were treated successfully with disk arthroplasty. As often in spinal surgery, proper patient selection is more important than selection of implant.

Looking to the future, surgeons must be aware of the interests of the manufacturers, which spent billions of dollars for disk arthroplasty technologies. Nevertheless, the lessons we have learned from the past 30 years should lead us to the development of better implants for our patients.

#### References

- White AA, Panjabi MM. The basic kinematic of the lumbar spine. Spine. 1978;3:12–20.
- Hedman TP, Kostuik JP, Fernie GR, Hellier WG. Design of an intervertebral disk prosthesis. Spine. 1991;16(6 Suppl):S256–60.

- Szpalski M, Gunzburg M, Mayer M. Spine arthroplasty: a historical review. Eur Spine J. 2002;11 Suppl 2:S65–84.
- Van Steenbrugghe MH. Perfectionnements aux prothèses articulaires. French Patent 1.122.634, 28 May 1956
- Nachemson A. Some mechanical properties of the lumbar intervertebral disc. Bull Hosp Joint Dis. 1962;23:130–2.
- Fernström U. Arthroplasty with intercorporal endoprosthesis in herniated disc and in painful disc. Acta Orthop Scand. 1966;10(Suppl):287–9.
- Fernström U. Der Bandscheibenersatz mit Erhaltung der Beweglichkeit. In: Herdman H, editor. Zukunftaufgaben für die Erforschung und Behandlung von Wirbelsäulenleiden. Die Wirbelsäule in Forschung und Praxis. Stuttgart: Hippokrates; 1972.
- Weber G. Zwischenwirbel Prothese. Swiss Patent 624573, February 1, 1978
- Ray CD, CorbinT. Prosthetic disc and method of implanting. US Patent 4,772,287; 20 Sept 1988
- Ray CD. The Raymedica prosthetic disc nucleus. an update. In: Kaech DL, Jinkins JR, editors. Spinal restabilization procedures. New York: Elsevier Science B.V; 2002. p. 273–82.
- 11. Klara PM, Ray CD. Artificial nucleus replacement. Clinical experience. Spine. 2002;12:1374–7.
- Schönmayr R, Busch C, Lotz C, Lotz-Metz G. Prosthetic disc nucleus implants: the Wiesbaden feasibility study. 2 years follow-up in ten patients. Riv Neuroradiol. 1999;12:S163–70.
- Shim CS, Lee SH, Park CW, Choi WC, Choi G, Choi WG, Lim SR, Lee HY. Partial disc replacement with the PDN prosthetic disc nucleus device. J Spinal Disord Tech. 2003;16:324–30.
- Bertagnoli R, Karg A, Voigt S. Lumbar partial disc replacement. Orthop Clin N Am. 2005;36:341–7.
- Schnake KJ, Weigert F, Kandziora F, Haas NP. Local vertebral body destruction after migration of a nucleus replacement. Z Orthop Unfall. 2007;145:649–51.
- Selviaridis P, Foroglou N, Tsitlakidis A, Hatzisotiriou A, Magras I, Patsalas I. Long-term outcome after implantation of prosthetic disc nucleus device (PDN) in lumbar disc disease. Hippokratia. 2010;14(3):176–84.
- Balsano M, Zachos A, Ruggiu A, Barca F, Tranquilli-Leali P, Doria C. Nucleus disc arthroplasty with the NUBAC<sup>TM</sup> device: 2-year clinical experience. Eur Spine J. 2011;20 Suppl 1:S36–40.
- Bertagnoli R, Vazquez RJ. The Anterolateral TransPsoaticApproach (ALPA): a new technique for implanting prosthetic disc-nucleus devices. J Spinal Disord Tech. 2003;36:398–404.
- Di Martino A, Vaccaro AJ, Lee JY, Denaro V, Lim MR. Nucleus pulposus replacement: basic science and indications for clinical use. Spine. 2005;30:S16–22.
- Coric D, Mummaneni PV. Nucleus replacement technologies. J Neurosurg Spine. 2008;8:115–20.
- Link HD. History, design and biomechanics of the LINK SB Charité artificial disc. Eur Spine J. 2002;11 Suppl 2:S98–105.

- Büttner-Janz K, Schnellnack K, Zippel H. Eine alternative Behandlungsstrategie beim lumbalen BandscheibenschadenmitderBandscheibenendoprothese Modulartyp SB Charité. Z Orthop. 1987;125:1–6.
- LeMaire JP, Skalli W, Lavaste F, Templier A, Mendes F, Diop A, Sauty V, Laloux E. Intervertebral disc prosthesis. Results and prospects for the year 2000. Clin Orthop Res. 1997;337:64–76.
- 24. Blumenthal S, McAfee PC, Guyer RD, Hochschuler SH, Geisler FH, Holt RT, Garcia Jr R, Regan JJ, Ohnmeiss DD. A prospective, randomized, multicenter Food and Drug Administration investigational device exemptions study of lumbar total disc replacement with the CHARITE artificial disc versus lumbar fusion: part I: evaluation of clinical outcomes. Spine. 2005;30:1565–75.
- 25. McAfee PC, Cunningham B, Holsapple G, Adams K, Blumenthal S, Guyer RD, Dmietriev A, Maxwell JH, Regan JJ, Isaza J. Prospective, randomized, multicenter Food and Drug Administration investigational device exemption study of lumbar total disc replacement with the CHARITE artificial disc versus lumbar fusion: part II: evaluation of radiographic outcomes and correlation of surgical technique accuracy with clinical outcomes. Spine. 2005;30:1576–83.
- 26. Petersilge CA. Lumbar disc replacement. Semin Musculoskelet Radiol. 2006;10:22–9.
- Mayer HM. Total lumbar disc replacement. J Bone Joint Surg Brit. 2005;87:1029–37.
- 28. Guyer RD, McAfee PC, Banco RJ, Bitan FD, Cappuccino A, Geisler FH, Hochschuler SH, Holt RT, Jenis LG, Majd ME, Regan JJ, Tromanhauser SG, Wong DC, Blumenthal SL. Prospective, randomized, multicenter Food and Drug Administration investigational device exemption study of lumbar total disc replacement with the CHARITE artificial disc versus lumbar fusion: five-year follow-up. Spine J. 2009;9(5):374–86. doi: 10.1016/j.spinee.2008.08.007. Epub 2008 Sep 19.
- Putzier M, Funk JF, Schneider SV, Gross C, Tothz SW, Khodadadyan-Klostermann C, Perka C, Kandziora F. Charité total disc replacement – clinical and radiographical results after an average follow-up of 17 years. Eur Spine J. 2006;15:183–95.
- Steffee AD. The Steffee artificial disc. In: Weinstein JN, editor. Clinical efficacy and outcome in the diagnosis and treatment of low back pain. New York: Raven; 1992.
- Fraser RD, Ross ER, Lowery GL, Steffee AD. Spinal disc, United States Patent 6139579, 31 Oct 2000
- 32. Meir AR, Freeman BJ, Fraser RD, Fowler SM. Tenyear survival and clinical outcome of the AcroFlex lumbar disc replacement for the treatment of symptomatic disc degeneration. Spine J. 2013;13(1):13–21.
- Marnay T. Prosthesis for intervertebral discs and instruments for implanting it. United States Patent 5314477, 24 May 1994
- 34. Marnay T. The ProDisc<sup>™</sup>: clinical analysis of an intervertebral disc implant. In: Kaech DL, Jinkins JR, editors. Spinal restabilization procedures. New York: Elsevier Science B.V; 2002. p. 317–31.

- 35. Zigler J, Delamarter R, Spivak JM, Linovitz RJ, Danielson 3rd GO, Haider TT, Cammisa F, Zuchermann J, Balderston R, Kitchel S, Foley K, Watkins R, Bradford D, Yue J, Yuan H, Herkowitz H, Geiger D, Bendo J, Peppers T, Sachs B, Girardi F, Kropf M, Goldstein J. Results of the prospective, randomized, multicenter Food and Drug Administration investigational device exemption study of the ProDisc-L total disc replacement versus circumferential fusion for the treatment of 1-level degenerative disc disease. Spine. 2007;32:1155–62.
- 36. Bertagnoli R, Yue JJ, Shah RV, Nanieva R, Pfeiffer F, Fenk-Mayer A, Kershaw T, Husted DS. The treatment of disabling multilevel lumbar discogenic low back pain with total disc arthroplasty utilizing the ProDisc prosthesis: a prospective study with 2-year minimum follow-up. Spine. 2005;30:2192–9.
- Siepe CJ, Heider F, Wiechert K, Hitzl W, Ishak B, Mayer MM. Mid- to long-term results of total lumbar disc replacement: a prospective analysis with 5- to 10-year follow-up. Spine J. 2014;14(8):1417–31.
- 38. Zigler JE, Delamarter RB. Five-year results of the prospective, randomized, multicenter, Food and Drug Administration investigational device exemption study of the ProDisc-L total disc replacement versus circumferential arthrodesis for the treatment of singlelevel degenerative disc disease. J Neurosurg Spine. 2012;17(6):493–501.
- Park CK, Ryu KS, Lee KY, Lee HJ. Clinical outcome of lumbar total disc replacement using ProDisc-L in degenerative disc disease: minimum 5-year follow-up results at a single institute. Spine. 2012;37:672–7.
- Tropiano P, Huang RC, Girardi FP, Cammisa Jr FP, Marnay T. Lumbar total disc replacement. Seven to eleven-year follow-up. J Bone Joint Surg Am. 2005;87-A:490–6.
- 41. Mathews H, Le Huec JC, Bertagnoli R, Friesem T, Eisermann L. Design rationale and early multicenter evaluation of Maverick total disk arthroplasty. International meeting on Advanced Spine Technologies, 2002, Montreux

- 42. Le Huec JC, Mathews H, Basso Y, Aunoble S, Hoste D, Bley B, Friesem T. Clinical results of Maverick lumbar total disc replacement: two-year prospective follow-up. Orthop Clin North Am. 2005;36:315–22.
- 43. Van de Kelft E, Verguts L. Clinical outcome of monosegmental total disc replacement for lumbar disc disease with ball-and-socket prosthesis (Maverick): prospective study with four-year follow-up. World Neurosurg. 2012;78(3–4):355–63.
- 44. Aghayev E, Bärlocher C, Sgier F, Hasdemir M, Steinsiepe KF, Wernli F, Porchet F, Hausmann O, Ramadan A, Maestretti G, Ebeling U, Neukamp M, Röder C. Five-year results of lumbar disc prostheses in the SWISSspine registry. Eur Spine J. 2014;23: 2114–26.
- Sköld C, Tropp H, Berg S. Five-year follow-up of total disc replacement compared to fusion: a randomized controlled trial. Eur Spine J. 2013;22(10): 2288–95.
- 46. Rao MJ, Cao SS. Artificial total disc replacement versus fusion for lumbar degenerative disc disease: a meta-analysis of randomized controlled trials. Arch Orthop Trauma Surg. 2014;134(2):149–58.
- 47. Jacobs WC, van der Gaag NA, Kruyt MC, Tuschel A, de Kleuver M, Peul WC, Verbout AJ, Oner FC. Total disc replacement for chronic discogenic low back pain: a Cochrane review. Spine (Phila Pa 1976). 2013;38(1):24–36.
- Cinotti G, David T, Postacchini F. Results of disc prosthesis after a minimum follow-up period of 2 years. Spine. 1996;21:995–1000.
- Yoshihara H, Yoneoka D. National trends in the surgical treatment for lumbar degenerative disc disease: US, 2000–2009. Spine J. 2014. pii: S1529-9430(14)01544-7. doi: 10.1016/j.spinee.2014.09.026. [Epub ahead of print].
- De Kleuver M, Oner FC, Jacobs WCH. Total disc replacement for chronic low back pain: background and a systematic review of the literature. Eur Spine J. 2003;12:108–16.

# Biomechanics of Lumbar Disk Arthroplasty

42

Fanor Saavedra, Christopher A. lannotti, Dani Bidros, and Edward C. Benzel

#### 42.1 Introduction

Mechanical low back pain (LBP) from lumbar degenerative disk disease (DDD) is a leading cause of pain and disability for adults in this country with over \$34 billion in annual health costs. Mechanical LBP has been problematic to both diagnose and treat. Nonsurgical treatment is successful in the majority of patients with LBP [1, 2]. Unfortunately, a significant minority of patients at various stages of the degenerative cascade remain debilitated with mechanical LBP. Surgical intervention is often recommended when conservative treatment fails to alleviate the symptoms. Surgical treatment for mechanical LBP has traditionally focused on segmental spinal fusion with concomitant loss of function of the motion segment(s). Traditional lumbar fusion

F. Saavedra, MD Department of Neurosurgery, University of Puerto Rico, San Juan, Puerto Rico

C.A. Iannotti, MD, PhD Arizona Neurosurgery and Spine Specialists, Phoenix, AZ, USA

D. Bidros, MD Lafayette Brain and Spine, Neurological Associates of Louisiana, Lafayette, LA, USA

E.C. Benzel, MD (⊠) Department of Neurosurgery, Neurological Institute, Cleveland Clinic, 9500 Euclid Avenue, S40, Cleveland, OH 44195, USA e-mail: benzele@ccf.org techniques have evolved over the years, and with increasingly sophisticated spinal instrumentation and implants, increased fusion rates have been achieved. However, fusion rates do not reliably predict clinical success rates. While fusion rates have approached 100 %, clinical success rates that range from 60 % to 90 % have been reported [3–7].

Arthrodesis of the three-joint functional spinal unit (anterior disk and two posterior facet joints) results in loss of motion [8–11] with a known incidence of adjacent-level symptomatic degeneration, ranging from 10 % to 30 % [12–15]. Spine arthroplasty techniques allow for pain relief by eliminating presumed pain generators (e.g., the annulus and/or nucleus) while preserving motion and theoretically protecting adjacent levels from additional strains and stresses.

Partial disk replacement was performed as early as the 1960s, when Fernstrom implanted stainless steel balls into the cervical and lumbar spine [16]. The modern era of lumbar arthroplasty originated at Charité Hospital in Berlin when Butner-Janz and Schellnack developed the original Charité artificial disk in the early 1980s. In the late 1980s, Dr. Thierry Marnay, a French orthopedic surgeon, developed the ProDisc-L [17]. Subsequently, total disk replacement has seen a steady evolution with a proliferation of devices and concomitant refinements in implant design, surgical technique, and instrumentation.

© Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_42
Lumbar arthroplasty can be divided into total disk replacement (TDR) and partial disk replacement (PDR) or nucleus replacement. TDR devices may be categorized according to their composite biomaterials (metal on metal or metal on polymer), biomechanics (unconstrained, semi-constrained, constrained), components (one-, two-, or three-piece designs), or fixation (spike or keel). Nucleus replacement devices represent an even more heterogeneous group of devices. Functionally, nucleus replacement devices can be divided into two broad classifications: elastomeric and mechanical.

The primary indication for both mechanical nucleus replacements and TDRs is mechanical low back pain due to DDD. Elastomeric nucleus replacement can also be utilized in the postdiskectomy setting. Generally, nucleus replacement represents a therapeutic intervention aimed at an early stage in the degenerative cascade, consisting of mild to moderate DDD. Advantages of PDR include minimally invasive and multiple approach options, including anterior retroperitoneal, lateral, and posterior approaches. The limited exposure required for insertion via annulotomy allows for multiple revision options, including TDR and fusion. Challenges of PDR include migration or expulsion risk since the devices generally are not fixed to the end plates and subsidence [18-27]. TDR is applicable for more advanced degenerative changes, as seen in moderate to severe DDD. Advantages of TDR include complete disk removal (which addresses pathology involving both the annulus and the nucleus), as well as reliable fixation to bony end plates. The challenges of TDR include avoidance of complications such as implant failure and expulsion, as well as longevity and wear debris.

# 42.2 Lumbar Fusion: Advantages and Disadvantages

The surgical procedure offered to patients with mechanical LBP due to DDD typically consists of arthrodesis of the painful motion segment, which may be combined with decompression for patients with radicular or neurogenic claudication symptoms. The rationale for spinal fusion is based on the assumption that it will eliminate painful nonphysiological motion across the destabilized or degenerated segment, preserve sagittal balance, and restore normal disk space height if combined with interbody spacers. Good to excellent results have been reported in 50-100 % of patients following anterior interbody or posterior spine fusion. Spinal fusion, however, is not a benign treatment. In a long-term study (>20 years), results of lumbar fusion demonstrated approximately 50 % of patients suffered recurrent symptoms requiring medication years after the original procedure, and approximately 15 % of patients reported undergoing additional surgery during the study period [13].

Fusion alters the normal biomechanics of the spine, transmitting forces to adjacent vertebral levels. The increased strain in the neighboring motion segments may accelerate disk degeneration, facet arthropathy, and promote osteophyte formation, all of which lead to recurrent back pain and symptomatic spinal stenosis at levels adjacent to the fusion site [13, 28–30]. Spine fusion also is associated with other complications, including alteration of muscular synergy, loss of spinal mobility, graft collapse resulting in suboptimal sagittal alignment, complications associated with the hardware implanted to achieve immediate stability, and the harvesting of iliac bone that is necessary for bone fusion [31].

## 42.3 Lumbar Disk Replacement

Due to concerns regarding detrimental long-term effects of spinal fusion procedures, there has been a search to develop a more physiological solution to mechanical LBP. It is theorized that the development of the lumbar artificial disk may revolutionize the treatment of lumbar spondylosis, similar to the manner in which hip and knee arthroplasty techniques have revolutionized the treatment of degenerative disease in those joints. Although it remains unproven, many believe that by reconstructing the normal biomechanics of the lumbar spine, complications associated with spinal fusion can be reduced.

Lumbar disk replacements have evolved from early designs, such as the Fernstrom ball, over a period of decades to the current designs present in the Charité SB III and the ProDisc-L. These implants have been offered as an alternative to spine fusion, with the overall goal of treatment of painful degenerated lumbar disks. The Charité was approved for single-level implantation for degenerative lumbar disk disease in the United States after FDA approval in October 2004. A second lumbar disk replacement, ProDisc-L (Synthes), was later approved in August 2006 for use in the United States. Since then, dozens of companies now have either plans or aspirations to market their own disk replacement in order to tap into the lucrative and growing market of disk.

The theoretical advantages of lumbar disk arthroplasty are prevention of adjacent-segment disease by preserving segmental motion, protection of neural elements by restoring disk space height, and shorter recuperation times since patients do not require a prolonged postoperative period to allow for fusion maturation. The current indications for lumbar disk arthroplasty include young, non-osteoporotic patients with one- or two-level symptomatic disk degeneration without severe facet arthropathy, segmental instability, or neural element compression requiring a posterior decompression.

#### 42.4 TDR Implants

#### 42.4.1 Operative Procedure

The general operative technique for various TDR devices utilizes a standard anterior retroperitoneal approach or standard mini-open approach for exposure of anterior disk space at the operative level, similar to the approach used for anterior lumbar interbody fusion (ALIF). In the United States, exposure is often accomplished with an access surgeon, typically either a general or vascular surgeon familiar with this approach. The patient is positioned in a supine, neutral position on a radiolucent operating table. Use of intraoperative fluoroscopy is mandatory. The midline should be preliminarily identified and marked using anatomic landmarks prior to extensive dissection. It is imperative to mobilize the iliac vessels in order to visualize the lateral margins of the anterior disk. Figure 42.1 shows a standard anterior retroperitoneal approach to the lumbar spine.

A total diskectomy is facilitated by mobilizing the disk with interbody distractors and releasing or resecting the posterior annulus and posterior longitudinal ligament (PLL), with possible removal of any posterior osteophytes. The lateral annulus is preserved on both sides. Disk removal is accomplished with standard technique utilizing curettes



Fig. 42.1 Anterior retroperitoneal approach to the lumbar spine

and rongeurs. It is essential to establish a minimum width of 30–35 mm on the anterior part of the disk to perform the implantation safely. Care is taken to remove the cartilaginous end plates while preserving the bony end plates to minimize the risk of implant subsidence. Attention is paid to complete disk removal retaining only the lateral annulus bilaterally. Special care must be taken with lateral disk resection where any retained disk material can be pushed into the foramen during artificial disk placement. The artificial disk is then placed under AP and lateral fluoroscopic guidance using an individualized combination of sizing, trialing, midline verification, and disk placement.

#### 42.4.2 TDR Devices

Concomitant with the Food and Drug Administration's (FDA) approval of the Charité artificial disk in October 2004, the era of spinal arthroplasty in the United States was born. Subsequently, the ProDisc-L (2006) received FDA approval. Several more devices have followed (Maverick, Kineflex, FlexiCore, and Activ-L). Cumulatively, these studies have produced a substantial proportion of class one evidence from prospective, randomized studies confirming TDR's efficacy in the treatment of symptomatic lumbar DDD (Table 42.1).

#### 42.4.3 Charité (DePuy Spine)

The SB Charité III (DePuy Spine) is one of the most extensively studied intervertebral prostheses to date. The first two designs, SB Charité I and SB Charité II, were implanted in a small number of patients in East Berlin, Germany, and were never made commercially available [32]. These early designs included stainless steel end plates, which

Table 42.1 FDA-approved artificial disk devices

Artificial disk device	Company	FDA status
Charité III	DePuy Spine, Inc.	Approval date: October 2004
ProDisc-L	Synthes Spine	Approval date: August 2006

were prone to breaking and subsidence. The thirdgeneration and current design (SB Charité III) was first marketed by Waldemar Link outside the United States in 1987, and DePuy Spine acquired the product rights to the Charité artificial disk in 2004. The Charité artificial disk consists of a three-piece design: two cobalt-chrome alloy end plates with spike fixation, with an intervening unconstrained, ultrahigh molecular weight polyethylene sliding core. This offers the theoretical advantage of allowing the spacer to shift dynamically within the disk space during spinal motion, moving posteriorly with flexion and anteriorly in lumbar extension. The primary attachment of the end plates is made possible by six "teeth" on the inferior and superior end plates, and a titanium/calcium hydroxyapatite coating promotes secondary fixation to the vertebral body by allowing bony ingrowth. Although this device was theorized to decrease facet joint arthropathy [32], the actual biomechanical effect of the device in vitro revealed a maximum local loading of facet joints approximately 2.5 times that of intact normal motion segments [26, 32, 33].

In 2009, Guyer et al. published a randomized, multicenter study comparing Charité artificial disk vs. anterior lumbar interbody fusion (ALIF) with BAK cages with a 5-year follow-up. From 160 patients, they did not found statistical differences regarding clinical outcomes between groups and concluded with a noninferiority report compared to ALIF [34].

## 42.4.4 ProDisc-L Artificial Disk (Synthes)

The ProDisc was designed in the late 1980s by Thierry Marnay, a French orthopedic spine surgeon, and was approved by the US FDA in August 2006 for use at a single lumbar vertebral level. The polyethylene core is secured to the caudal end plate by a modular locking system. Each end plate has a central anchoring keel and two spikes to provide immediate stability. The end plates are coated with a titanium plasmapore surface to enable bony ingrowth for secondary stability. Specially designed instrumentation is used to create a midline groove in the vertebral end plates for implant insertion and to lock the polyethylene core into the caudal end plate. There are two end plate sizes (medium and large), three heights (10, 12, and 14 mm), and two lordosis angles ( $6^{\circ}$  and  $11^{\circ}$ ).

Zigler et al. recently published a prospective, randomized, multicenter study with a 5-year follow-up comparing the ProDisc-L versus circumferential arthrodesis for the treatment of single-level lumbar degenerative disk disease. One hundred sixty-one patients underwent TDRs with the ProDisc-L and 75 patients underwent circumferential fusion. They found that both groups had sustained significant improvement from their preoperative status. No significant differences were found in clinical outcome scales (ODI, SF-36). They concluded that both fusion and TDR are reasonable surgical options [35].

## 42.4.5 Maverick Artificial Disk (Medtronic: Fig. 42.2)

The Maverick<sup>TM</sup> artificial disk consists of a cobaltchrome, two-piece metal-on-metal design with keel fixation and a constrained ball-and-socket design. The center of rotation is fixed and located in the posterior third of the disk space. It is also a semiconstrained device that does not allow for pure translation. These features allow for unloading of the facet joints and reproduction of near-normal force transmission at the operated motion segment. Metal fixation into bone provides immediate stability, and there is no polyethylene wear issue.

Gornet et al. published in 2011 a 2-year followup randomized controlled study with 577 patients. Four hundred five patients received the Maverick disk device and 172 anterior interbody fusion (LT-cage). At the last follow-up, the overall success rate for the TDR group was 73.5 %, and for the control group (ALIF), the rate was 55.3 % with a noninferiority conclusion (P < 0.001). Postoperative outcome scales (SF-36 and ODI) also favored the TDR group. There are no differences on adverse events between groups. There were two implant removals in the Maverick group; one was considered to be related to an allergic reaction. Longer follow-up with this two-piece metal-on-metal implant is needed, particularly in light of emerging complications (e.g., pseudotumor formation) with metal-on-metal hip implants. Despite completion of the investigational device exemption study in 2010, the Maverick<sup>TM</sup> is not currently marketed in the United States due to a patent dispute with another company [36].



**Fig. 42.2** (a, b) The Maverick artificial disk (With permission and image provided by Medtronic, Inc.) (The Maverick<sup>TM</sup> disk is not available or approved in the United States)

# 42.4.6 Kineflex Artificial Disk (SpinalMotion: Fig. 42.3)

The Kineflex artificial disk consists of cobaltchrome, metal-on-metal design with a semiconstrained core. The mobile core is seated within a retention ring, and the superior and inferior end plates have multiple serrations, in addition to a central fin allowing for keel fixation and initial stability. On 2011 Kenneth Pettine published a randomized noninferiority trial comparing the Kineflex artificial disk versus the Charité. Sixty-four patients were randomized and followed for 24 months. Eightythree percent of patients in the Kineflex disk group and 85 % of patients in the Charité group met FDAdefined criteria for clinical success, with no difference between groups (P=0.802) [37].

# 42.4.7 FlexiCore Artificial Disk (Stryker Spine: Fig. 42.4)

FlexiCore is a cobalt-chrome metal-on-metal device that is inserted as a single unit. The superior and inferior portions are linked by a captured ball-and-socket joint (constrained domed core). The FlexiCore can be implanted from a straight anterior or an anterolateral direction and offers the ability to manipulate the position of the implant within the intervertebral space. The FlexiCore features unique dome-shaped baseplate surfaces to approximate the concavities of the vertebral body end plates, with several small spikes or "teeth" for immediate bone fixation. These implants are also coated with titanium plasma spray to assist in delayed bony ingrowth for fixation. FDA IDE studies of the FlexiCore disk commenced in 2003 and enrolled and followed over 500 patients. It was also sold outside the United States beginning in 2005. The FlexiCore disk is not currently commercially available [38].

# 42.4.8 Activ-L Artificial Disk (Aesculap)

The Activ-L Artificial Disk is a next-generation intervertebral disk prostheses consisting of cobaltchrome end plates with a semi-constrained, polyethylene core (ball-and-socket design). There are modular end plates available, in both spike and keel fixation. Low implant height (with a total height of 8.5 mm) may reduce disk space overdistraction. The convex-shaped end plate allows an ideal contact surface with the concave vertebral end plates. Importantly, the device allows for



Fig. 42.3 (a-c) The Kineflex artificial disk (With permission and image provided by SpinalMotion, Inc.)



Fig. 42.4 (a, b) FlexiCore artificial disk (With permission and image provided by Styker)



**Fig. 42.5** (a-d) XL TDR (With permission and image provided by NuVasive, Inc.) (The NuVasive XL TDR® device is not approved in the United States. Caution – investigational device. Limited by US law to investigational use)

rotation and controlled anteroposterior translation of the polyethylene core, similar to the native IAR. The combination of rotation and translation may serve to unload the facets joints, thereby preventing implant-related facet arthrosis. An FDA trial started in 2007, enrollment was completed and currently it is in the final stages of data collection and submission [38].

There are also other devices that are currently on trials, such as the NuVasive XL TDR® (NuVasive – Fig. 42.5). It has metallic design that is implanted from the patient's side, with a similar approach to

extreme lateral interbody fusion (XLIF) cages. An FDA trial started in 2009. The NuVasive XL TDR® device is not approved in the United States and limited by US law to investigational use.

The Freedom Lumbar Disk (AxioMed – Fig. 42.6) is made of a silicone polycarbonateurethane polymer core that is bonded between the two metal end plates to allow controlled motion and enable shock absorption capability. The device has been used in Europe, and FDA trials began in 2008. The Triumph (Fig. 42.7) consists of two metallic end plates with a geom-



**Fig. 42.6** Freedom TDR (With permission and image provided by AxioMed)



**Fig. 42.7** Triumph TDR is an investigation device in the United States (With permission and image provided by Globus Medical, Inc.)

etry allowing it to be inserted through a posterolateral approach. An IDE study started in 2007; it is still ongoing but no longer recruiting.

# 42.5 Biomechanics of TDR Prostheses

# 42.5.1 Intervertebral Disk and Disk Replacement Center of Rotation

The center of rotation of the lumbar spinal motion segment does not move about a single fixed axis of rotation, but rather an elliptical locus of instantaneous axes of rotation, termed the centrode [39–42]. The centrode is located within the posterior half of the disk space in the normal intact spine. In early degenerative disk disease, the centrode lengthens, and in moderate disk degeneration, the centrode migrates caudally. Axial loading does not appear to influence the centrode dimensions or position.

Using an in vitro human cadaveric model, the multidirectional flexibility properties and COR of TDR (SB Charité) were determined and compared to conventional threaded fusion cages (e.g., BAK cages) and BAK cages augmented with transpedicular screw/rod fixation for single-level spinal instrumentation [43, 44]. The SB Charité prosthesis was associated with an increase in axial rotation by 44 % compared to the intact lumbar spine, whereas the BAK and anteroposterior reconstructions decreased range of motion by 29 % and 80 %, respectively. Flexionextension range of motion was slightly increased for the SB Charité (3 %) versus the intact disk, whereas this was decreased in the BAK and anteroposterior stabilization groups (BAK = 57%, anteroposterior = 93 %) when compared to the intact and SB Charité conditions. Based on flexion-extension radiographs, the intervertebral centers of rotation were in the posterior one-third of the operative intervertebral disk only for the SB Charité device and intact spine condition, with definitive evidence of physiologic intervertebral translation. Therefore, the TDR device preserved the kinematic properties and normal mapping of segmental motion at the operative and adjacent intervertebral disk levels, as compared to interbody instrumentation, with or without transpedicular screw/rod fixation.

# 42.5.2 Load Sharing After Lumbar Disk Arthroplasty

Using a finite element model, Dooris et al. [45] implanted a ball-and-cup-type artificial disk via an anterior approach and compared the data with in vitro data. Both small and large annulotomies were performed, and the implant was placed either anteriorly or posteriorly within the disk space. The restoration of an intact ALL was also assessed. Models were subjected to either an axial compression force alone or a combination of flexion-extension moment with axial preload. In this model, the facet loads were found to be more sensitive to the anteroposterior location of the artificial disk than to the degree of annulotomy. Under pure axial compression, implanted models with an anteriorly placed artificial disk exhibited facet loads 2.5 times greater than loads observed with the intact model, whereas posteriorly implanted models demonstrated no facet loads in compression. Implanted models with a posteriorly placed disk also exhibited greater flexibility than the intact and implanted models with anteriorly placed disks. The restoration of the anterior longitudinal ligament resulted in reduced pedicle stresses, facet loads, and extensionrotation to near normal. This study suggested that by altering the placement of the artificial disk in the anteroposterior direction, the motion segment flexural stiffness and posterior load sharing may be modulated [45].

Biomechanical comparisons have also been performed between unconstrained and constrained lumbar artificial disk designs. These studies have revealed differences in facet loading and implant stresses (e.g., polyethylene core) [46]. An unconstrained artificial disk design is less sensitive to placement and unloads the facet joints, compared with a constrained design. The decreased core stress may result in a reduced potential for wear in an unconstrained prosthesis, which may potentially increase the functional longevity of the device.

# 42.5.3 Load-Displacement Curve and Motion Patterns of TDR Devices

The pattern of intervertebral load-displacement curves for Charité TDR-implanted motion segments has been studied by O'Leary et al. [47]. The prosthesis component motion patterns (PCMP) were divided into four types: PCMP1, angular motion predominantly between the upper end plate and core, with little or no visible core translation; PCMP2, lift-off of upper prosthesis end plate from core or of core from lower end plate; PCMP3, core entrapment, resulting in a locked core over a portion of the range of motion; and PCMP4, angular motion between both the upper and lower end plates and core, with visible core translation [47].

A gradually changing motion pattern was observed in normal lumbar segments, while the TDR-implanted segments displayed regions of both relatively small and large angular changes with gradual moment application. The disruption of the ALL and the anterior annulus during the insertion of the Charité TDR removes the biomechanical constraints imposed by these structures. As a result, a larger angular travel was observed in the absence of physiologic compressive preload for the same applied moment in TDR motion segments, compared with the intact motion segments. Under a compressive preload, entrapment and locking of the polyethylene core occurred over a portion of the sagittal plane motion – as reflected by the relatively flat portion of the load-displacement curves in the presence of a preload (Fig. 42.8). Once the core was released, a large angular change occurred, reflected by a sharp rise of the load-displacement curves (Fig. 42.8). The predominant angular motion within the prosthesis occurred between the upper end plate and the polyethylene core.

The effects that these TDR motion patterns may have on the long-term outcome, as well as on load sharing at the implanted level and polyethylene core wear, currently remain unknown. However, the nonuniform motion could potentially influence the wear of the implant. Other movement patterns, such as angular motion at one articulation with only a small amount of core translation, may also potentially influence load sharing within the implanted segment and at adjacent levels. Based on biomechanical studies, it may be concluded that several factors are likely to affect the function of the non- or semiconstrained TDR implants, including implant placement and orientation, intraoperative changes in lordosis, and the magnitude of physiologic compressive preload [47].

Fig. 42.8 Load angular displacement curves of intact lumbar spine and Charité TDR device implanted at the (a) L5-S1 and (b) L4-L5 levels in human lumbar spine. (a) Prosthesis motion pattern demonstrating core entrapment. A locked core is observed over a portion of the range of motion, with large angular change occurring with core release. (b) Prosthesis motion pattern with angular motion between both the *upper* and *lower* end plates and core, with visible core translation (Figure kindly provided by Dr. Avinash Patwardhan)



# 42.5.4 Adjacent-Segment Degeneration

Accelerated adjacent-segment degeneration is one of the most important morbidities associated with

solid spinal fusion. It is for this reason that there is such a significant development of non-fusion motion preservation devices. Short-term results for the lumbar disk arthroplasty devices are encouraging. Long-term results, however, are not yet available. Appropriately designed biomechanical studies of adjacent-level degeneration have provided insight into the potential success or failure of an implant with respect to motion preservation and adjacent-level degeneration. Although many biomechanical studies are available, the results have large variation and are conflicting, mostly due to the use of inappropriate and illdefined methodologies. A relatively new testing method designed to study spinal adjacent-level effects, the hybrid test method, uses unconstrained pure moments to provide rotation input for multidirectional testing. The hybrid test method has four steps: (1) intact spine specimen with entire mobile region is used to measure various biomechanical parameters, e.g., disk pressures, ligament strains, and facet loads; (2) appropriate unconstrained pure moment is applied to the intact specimen and total range of motion is determined; (3) unconstrained pure moment is applied to the spinal construct (specimen with an implant) until the total range of motion of the construct equals that of the intact; and (4) statistical comparison of the biomechanical parameters between the construct and intact then allows for comparison of the adjacentlevel effects [48–51]. Using a hybrid test method with a whole lumbar spine specimen (T12–S1), multidirectional adjacent-level effects due to implantation of one- and two-level ProDisc-L device were evaluated and compared to simulated one- and two-level arthrodesis. The single-level lumbar ProDisc-L preserved rotations at the nonoperated levels but showed increased rotations at the operated level in both lateral bending and torsion. Conversely, two-level simulated fusion decreased the rotations at the fusion site and produced increased adjacent-level rotations in all directions, compared with lumbar arthroplasty devices. Therefore, it may be concluded that rotation lost/gained at the operated level is redistributed over the remainder levels. ProDisc-L implantation produced only small adjacent-level effects in comparison with significant effects produced by spinal fusion. Importantly, the study revealed that the adjacent-level effects were not confined to the adjacent levels alone but were seen throughout the entire specimen (i.e., whole lumbar spine specimen, T12–S1).

Adjacent-level biomechanics after multilevel disk arthroplasty in cadaveric lumbar spines have been evaluated by comparing operativeand adjacent-segment range of motion and intradiscal pressures [52]. The study compared two-level disk arthroplasty versus circumferential arthrodesis using anterior interbody cages and pedicle screws. A kinematics assessment revealed that segmental motion distributed over L2–S1 was preserved in the arthroplasty group but was significantly altered after circumferential fusion. After arthrodesis, adjacent-level range of motion and intradiscal pressures were increased proximally and distally under loading modalities of flexion-extension, axial rotation, and lateral bending. In contrast, there was no significant difference in either the range of motion or intradiscal pressures at adjacent levels between intact control and disk arthroplasty groups.

Harrop et al. [53] have performed a systematic review of the published incidence of radiographic adjacent-segment degeneration and symptomatic adjacent-segment disease after arthrodesis or total disk replacement. Their data supported the use of arthroplasty to reduce adjacent-level disk degeneration and disease, compared to arthrodesis. In the study, 34 % of patients undergoing arthrodesis and 9 % of patients in the total disk replacement group developed adjacent-level degeneration. Increased risk of adjacent-level degeneration was associated with older patients, arthrodesis, and longer follow-up [30, 53]. The risk of adjacent-segment failure appears to be higher for patients in whom lumbar fusion with rigid instrumentation is performed to treat degenerative instability, and the risk seem to be particularly high in postmenopausal women [54]. Adjacent-segment disease developed in 14 % arthrodesis patients compared to 1 % of arthroplasty patients. These results suggested a correlation between fusion and the development of adjacent-level degeneration, compared to arthroplasty. However, the association was dampened by the influence of patient age. Nevertheless, a strong correlation existed between spinal fusion and adjacent-level disease, compared to arthroplasty.

Another systematic review done by Wang et al. compared the incidence of adjacent-segment disease of two randomized controlled trials (Berg et al. and Guyer et al.), after a 2-year follow-up. The risk of clinical adjacent-segment disease (treated surgically) was 1.2 % (2/170) on the TDR group versus 7.0 % (8/115) on the fusion group. Another more recent study done by Zigler et al. published a randomized controlled trial with a 5-year follow-up, comparing the ProDisc-L versus circumferential fusion. Three (1.9 %) of 161 of the TDR group and 3 (4 %) of 75 fusion group required surgery at the adjacent level [38].

The aforementioned studies are challenged, however, by existing data that suggests that incidence of adjacent-segment disease is no greater in fused spines than in spines that did not undergo fusion. Furthermore, the maintenance of a lordotic posture appears to be critical for the prevention of adjacent-segment degeneration and disease [43, 55–68].

#### 42.5.5 Motion Preservation

The preservation of segmental mobility represents a potential advantage of lumbar disk arthroplasty. One of the goals of lumbar arthroplasty is to restore and maintain mobility and to protect adjacent levels from abnormal motion, which may be a factor in transition syndrome following arthrodesis. The restoration of disk motion may preserve lumbar facet structure and function and prevent early degeneration. Results from clinical trials of disk replacement, however, do not yet have sufficient follow-up to determine whether the cascade of facet joint arthropathy is, in fact, halted or retarded. The mobility observed with TDR appears to be approximately 6° in flexionextension at the L5-S1 level and 8° at L4-L5 [69]. Mobility in lateral inclination has also been analyzed and was found to be approximately 3-4°. Clinical and radiologic outcomes with a minimum follow-up of 10 years have been reported for the Charité artificial disk [69]. A total of 107 patients implanted with the Charité prosthesis through a standard anterior retroperitoneal approach (147 prostheses implanted, 54 one-level procedures, 45 two-level procedures, and 1 three-level procedure) were evaluated. Clinically, 62 % had an excellent outcome, 28 % had a good outcome, and 10 % had a poor outcome. Mean flexion-extension motion was 10.3° for all levels (12.0° at L3–L4, 9.6° at L4–L5, 9.2° at L5–S1). The mean lateral motion was 5.4°. In the sagittal plane, 6.1 % of the devices were anterior of geometric center, 34.0 % were centered, and 59.9 % were posterior of center. In the frontal plane, 75 % were centered, and 25 % were lateral of center. No subluxation of the prostheses and no cases of spontaneous arthrodesis were identified. Thus, with a minimum follow-up of 10 years, the Charité artificial disk demonstrated excellent flexion-extension and lateral range of motion with no significant complications [69].

The distribution of in vivo and in vitro range of motion (ROM) following single-level arthroplasty with the SB Charité III artificial disk has been compared with posterolateral fusion [70]. In this study, in vitro ROM in flexion-extension at the implanted and adjacent levels was measured, and the results were compared with in vivo, 2-year postoperative radiographic ROM evaluations. The results showed that single-level arthroplasty appeared to replicate the normal distribution of motion of the intact spine [17, 70].

A human cadaveric biomechanical study was performed to measure the facet forces and the IAR for different spinal positions under simulated weight-bearing conditions before and after total disk replacement at L5-S1 using semi-constrained (3° of freedom; ProDisc) and unconstrained (5° of freedom; Charité) articulated implants. With the ProDisc, the facets were partially unloaded, though the IAR did not match the fixed geometric center of the UHMWPE core, suggesting that joint surface incongruence is developed during movement. With the Charité, the IAR was less variable, yet the facet forces tended to increase particularly during lateral bending. These results highlight the important role that the facets play in guiding movement and that implant constraint influences facet-implant synergy. Rotational hypermobility and excessive torsional forces are potentially undesirable effects. Such hypermobility may induce excessive loads on the posterior facets, promoting facet arthrosis and nerve root impingement. Moreover, the posterior elements in many patients with mechanical LBP are not fully intact, leading to further events of the degenerative cascade (e.g., disk degeneration, facet arthrosis) and subsequently producing pain.

The biomechanics of the lumbar spine treated either by fusion or total disk replacement (TDR) have been compared under severe loading conditions [71]. A three-dimensional model of a twolevel ligamentous lumbar segment was created and simulated through static analyses with the finite element method. The analysis predicted that mobility after arthrodesis on the upper level was reduced in all rotational degrees of freedom by an average of approximately 44 %, relative to healthy normal disks. Conversely, the mobility after TDR on the upper level was increased in all rotational degrees of freedom by 52 %. The level implanted with the artificial disk showed excessive ligament tensions, high facet pressures, and a high risk of instability. The mobility and the stresses in the level adjacent to the arthroplasty were also increased. This model for an implanted artificial disk showed a greater risk of instability and further degeneration than predicted for the arthrodesis model [72].

## 42.5.6 Stability of the Implant-Bone Interface

The stability of an implanted lumbar arthroplasty device may be conceptualized into three categories: (1) short-term or immediate stability, (2) intermediate stability, and (3) long-term stability. The anchoring to the vertebral end plates via spikes or midline keel is considered important for immediate implant stability. The original SB Charité prosthesis was not solidly anchored, and this may have been associated with early expulsion of the device [32]. Several device end plates have been modified with a porous titanium and electrochemically bonded calcium hydroxyapatite coating, which allow for bony ingrowth into the end plate, providing intermediate or secondary stability with a potentially reduced risk of device expulsion [73]. Long-term stability may be defined as the cellular and microscopic changes of bony ingrowth and integration at the bone interfaces.

Following a 6-month survival period, the range of motion of the SB Charité and intact nonoperative controls were examined under axial compression, flexion-extension, and lateral bending. No statistical differences were observed between the groups [70, 71]. Plain film radiographic analysis showed no lucencies or loosening of any metallic prosthetic vertebral end plate, and gross histopathologic analysis of the SB Charité prosthesis demonstrated excellent bony ingrowth at the level of the implant-bone interface, without evidence of fibrous tissue or synovium. Furthermore, histochemical assays showed no local or systemic accumulation of particulate wear debris (titanium, ultrahigh molecular weight polyethylene, or cobalt-chrome) or cytokines, including TNF- $\alpha$ , PG<sub>E</sub>2, IL-1, IL-2, or IL-6. The improved degree of porous ingrowth in total disk replacement prostheses, compared to other joint replacement devices, is likely due to ligamentotaxis, causing sustained compression forces across the metal-bone interface.

#### 42.5.7 Wear Testing

An understanding of the wear potential of total disk replacements (TDRs) is critical as these new devices are increasingly introduced into clinical practice. In contrast to hip and knee wear, little is known about in vivo degradation or the contribution of wear debris to biologically mediated failure mechanisms of lumbar artificial disk devices. Ongoing retrieval studies have provided some evidence that clinically relevant wear and polyethylene degradation may occur in vivo with artificial disks. Implant subsidence, malpositioning, or migration may result in rim damage, plastic deformation, and component fracture. Chronic inflammatory reactions and wear debris in tissues have also been observed surrounding failed artificial disks. However, the clinical significance of wear in the spine remains poorly understood. Although it appears that wear particles may result

in a local inflammatory reaction, few cases of osteolysis around artificial disks have been reported. The wear potential of a ProDisc-L implant was determined using an adaptive finite element technique [1]. The testing scheme was validated using a model of a total hip replacement (THR) and was then used to model the ProDisc-L. The degree of flexion-extension, lateral bending, axial twist, and axial load were applied through ten million simulation cycles.

The polyethylene wear rate of the ProDisc-L TDR was 9.8 mg/million cycles, whereas the comparable wear rate for the THR was 16.1 mg/ million cycles. Thus, the wear potential of the TDR was better than the THR using joint-specific loading standards [1].

Metal-on-metal wear for the Maverick TDR (a semi-constrained, ball-and-socket implant) has been tested extensively, and there is essentially no metal wear at 31.5 years of simulated loading. The potential for toxicity of metallic wear debris to the surrounding neurologic structures has also been examined. The epidural toxicity of wear debris of the Maverick disk replacement was conducted in rabbits and there was no significant difference in epidural debris between control and Maverick-implanted animals [74]. Overall, the production of metal debris by the metal-metal TDR devices appears to be extremely low, which may enable metal-metal TDR devices to obtain an optimal longevity. In cases of THR, the polyethylene wear debris is responsible for prosthesis loosening over time, due to the macrophage reaction to wear debris particles. Similar studies have yet to be reported with TDR devices.

The extent of surface damage of the polyethylene core, including rim fracture and wear, after long-term implantation also remains poorly understood. Studies have been performed on Charité TDR components retrieved from patients undergoing revision TDR surgery and conversion to fusion [75]. All implants were removed due to pain, and the implants were associated with either subsidence, anterior migration, core dislocation, lateral subluxation, wear with wire marker fracture, end plate loosening, or osteolysis. Dimensional measurements and assessment of the extent and severity of polyethylene surface damage were assessed by micro-CT scanning, light microscopy, and white light interferometry. The dominant wear mechanism was adhesive/ abrasive wear at both the dome and rim. End plate penetration (dome wear) was correlated with implantation time. There was also evidence of macroscopic rim damage, including radial and transverse cracking, fracture, plastic deformation, and third-body damage. Radiographic wire marker fracture was always associated with deformation, cracking, or fracture of the polyethylene rim. The TDRs displayed surface damage observed previously in both hip and knee replacements. Because of the evidence of increasing wear with implantation time, along with the demonstrated potential for osteolysis in the spine, regular long-term follow-up for patients undergoing TDRs is warranted.

#### 42.5.8 Shock Absorption Capacity

The ProDisc-L (metal-polyethylene prosthesis) and the Maverick (metal-metal device) have been compared in terms of their shock absorption capacity and transmission of vibrations [74]. In this study, no significant differences were found between the implants regarding shock absorption or vibration transmission, with neither having any significant shock-absorbing effects. In pure axial loading, the axial stiffness of TDR devices is nearly infinite, demonstrating an absence of shock absorption capacity. This is strikingly different from the shock-absorbing properties seen in normal, and even severely degenerated, lumbar disks. However, the long-term consequences of these differences in shock absorption capacity are relatively uncharacterized. It is possible that an increase in posterior load transfer may occur after TDR, shifting a larger proportion of axial compressive load to the facet joints and potentially hastening the occurrence of facet joint arthropathy. Polymeric core disks address this problem. For example, the Freedom Lumbar Disk is made of a silicone polycarbonate-urethane polymer core that is bonded between the two metal end plates to allow controlled motion and enable shock absorption capability.

# 42.6 Complications of TDR Implants

Numerous types of complications have been documented with lumbar disk replacements. These include facet joint degeneration (facet compression or distraction), unexplained radiculopathy and/or back pain, extrusion of the implant, facet fracture, and acquired spondylolysis [31, 76]. These implant failures have been ascribed to surgical errors in either placement or sizing, surgical approach errors, inexperience, inappropriate indications, and patient noncompliance [77, 78]. Of note, the presence of anterior and posterior longitudinal ligaments, and the absence of collateral ligaments, facilitates side-to-side motion at the involved segment(s). Under normal conditions, the ALL may serve to limit excessive extension in the disk space. However, during disk replacement the ALL is incised, which may allow for a significant increase in lumbar extension. Since the ALL is focally absent, extrusion of the implant is a potential problem. It is also thought that abnormally high load sharing occurring posteriorly at the facet joints, along with a concurrent decrease in load sharing anteriorly, may further promote loosening and expulsion of the implant.

Potential complications of TDR include vertebral body fracture and implant subsidence, facet fracture, facet arthrosis, acquired spondylolysis, vascular injury, implant migration or dislodgement, and thecal sac/nerve root injury. A study of patients who presented with persistent or recurrent backache or leg pain after implantation of an artificial disk prosthesis (SB Charité III) revealed that migration of the prosthesis, subsidence into the vertebral body, adjacent-level disk degeneration at one or more neighboring levels, and facet arthrosis were the predominant complications [76]. In several cases, rupture of the metal wire around the polyethylene core occurred, as well as signs of radiological polyethylene wear. Subsidence of a metal end plate of the disk replacement into the vertebral body appears to be associated with fracture of the vertebral end plate. The artificial disk does not possess significant shock absorption capacity, and this may promote a more posterior transfer of load [74]. The excision of the ALL allows for an increased degree of lumbar extension, while the PLL remains intact. The above factors, along with an increase in disk space distraction by the TDR device, may potentially promote fracture of the adjacent end plates and, subsequently, subsidence of the implant.

Siepe et al. [79] published a prospective review on the ProDisc implant with a 5-10-year follow-up. The overall complication rate was 14.4 % (26/181): 11.9 % (18/151) for monosegmental TDRs and 27.6 % (8/29) for bisegmental interventions. The overall reoperation rate was 16.0 % (29/181 cases). Immediate technique-related complications device or occurred in 5 % (9/181), while general surgeryrelated complications occurred in 2.2 % (4/181). Reoperations for the treatment of persisting symptoms of LBP occurred in 5.5 % (10/181). A smaller subgroup in which reoperation was indicated for the treatment of adjacent-segment pathologies occurred in 2.2 % (4/181). Other published studies have reported widely diverging complication rates following TDR, the majority of which ranged between 10 % and 20 % [79].

## 42.7 Nucleus Replacement

There are no nucleus replacements or augmentation devices that are currently approved for use in the United States, but there are a number that are under clinical investigation. Hence, the following discussion revolves around devices in evolution.

#### 42.7.1 Nucleus Replacement Devices

The treatment of chronic low back pain due to advanced disk degeneration requires restabilization of the lumbar column, mainly by restoring disk function. Disk restabilization involves restoration and stabilization of the ligaments and annulus. Such restabilization represents a new concept in treating lumbar disk pathology.

Clinically, indications for nucleus replacement may be separated into two general categories. First, nucleus replacement may be utilized in select post-diskectomy patients as a largely prophylactic procedure to prevent recurrent disk herniations or progressive DDD. It is well documented that a small, but significant, proportion of patients (~3-15 %) undergoing standard microdiskectomy will develop recurrent herniations or progressive degenerative changes with symptomatic low back pain [11, 80, 81]. Iatrogenic changes seen post-diskectomy include loss of disk space height and desiccation [80, 82]. Partial disk replacement in the post-diskectomy setting offers the theoretical benefit of slowing future degenerative changes by maintaining disk space height and normal motion. Similar to TDR, nucleus replacement can also be utilized in patients with mechanical low back pain due to mild to moderate DDD. PDR allows for direct removal of the presumed pain generator (the diseased nucleus) and replacement with a device that maintains functional motion [19–24, 55]. Functionally, nucleus replacement devices can be divided into two broad classifications: elastomeric and mechanical [19-21, 44]. Elastomeric devices can be further subdivided into hydrogel and non-hydrogel replacements. These devices are either preformed or injectable. Injectable nucleus replacements are delivered in liquid form into the nucleotomy void and cure in situ, while preformed nucleus replacements are pre-cured polymers [19-24, 55, 83]. Mechanical devices can be subdivided into one- and two-piece designs [19–21].

#### 42.7.2 Operative Technique

The surgical approach for nucleus replacement is variable depending on the specific device. Generally, devices can be placed via minimally invasive techniques, with multiple approach options available (e.g., anterior, lateral, posterior). The prosthetic disk nucleus (PDN; Raymedica) is implanted via a posterior hemilaminotomy or a transposatic approach [2].

## 42.7.2.1 Elastomeric Nucleus Replacement

The advantages of elastomeric nucleus replacement include the ability to recreate the natural function of the normal nucleus pulposus with uniform stress distribution and shock absorption capability. A central challenge involved in the utilization of preformed devices is implant extrusion due to their inherently deformable nature. For injectable devices, biocompatibility, longterm durability, as well as avoidance of leakage are of crucial importance. There are several elastomeric nucleus devices in active clinical trials [20, 21, 24, 25].

## 42.7.3 Prosthetic Disk Nucleus (PDN) (Centinel Spine)

PDN is composed of a preformed hydrogel core consisting of polyacrylonitrile and polyacrylamide with a polyethylene woven jacket. PDN was first implanted in humans in 1996. PDN remains the most widely studied nucleoplasty device worldwide [22, 48]. PDN is implanted dehydrated and subsequently hydrates and expands. It is designed to absorb up to 80 % of its own weight in water. Extrusion risk with the original design led to design and approach modification [17]. Subsequent designs include the PDN-Solo and, most recently, PDN HydraFlex [55, 77].

# 42.7.4 NeuDisc<sup>™</sup> (Replication Medical)

NeuDisc is a compressible, preformed hydrogel consisting of an aquacryl polymer reinforced with Dacron mesh. It is implanted dehydrated and expands anisotropic to conform to the nucleotomy defect [20, 21, 25].

## 42.7.5 NuCore™ Injectable Disk Nucleus (IDN) (Spine Wave)

NuCore is an injectable non-hydrogel nucleus. It is an rDNA-based synthetic protein copolymer composed of silk and elastin. NuCore has been utilized in both the post-diskectomy and early DDD indication.

# 42.7.6 Dascor™ (Disk Dynamics): Prosthetic Intervertebral Nucleus

Dascor is a constrained, injectable, non-hydrogel nucleus. It is applied as a two-part in situ curable polyurethane and an expandable polyurethane balloon which is inserted into the disk space after the nucleus has been removed. The balloon is then injected under pressure with the flowable polymer that conforms to the shape and size of the disk space. The flowable polymer cures creating a firm but pliable implant.

## 42.7.6.1 Mechanical Nucleus Replacement

The advantages of mechanical nucleus replacement include strength and durability. Challenges include maintaining an even stress distribution and lack of shock absorption. The major weakness revolves around a lack of anchor to the end plates which predisposes to subsidence and expulsion. There are two mechanical nucleus devices in pilot feasibility study [19–21].

# 42.7.7 NuBac<sup>™</sup> Disk Arthroplasty Device (Pioneer Surgical)

NuBac is a two-piece, mechanical nucleus. It is composed of poly-ether-ether-ketone (PEEK) and is the first PEEK-on-PEEK articulated intradiscal arthroplasty device. NuBac has a ball-andsocket design with a large surface contact area to distribute stress and theoretically lower the risk of subsidence. NuBac has been implanted through a lateral (XLIF) approach at L3–L4, L4– L5, as well as anterior retroperitoneal and posterior approaches at L5–S1 [19, 23, 49]. Currently the product is not available for distribution or use in the United States.

## 42.7.8 Regain<sup>™</sup> (Biomet)

Regain is a solid, one-piece mechanical nucleus composed of a graphite substrate with a coating of pyrolytic carbon. It is applied at one-level L4–

L5 or L5–S1 via an anterior retroperitoneal approach. Regain has a convex outer surface intended to conform to the natural convexity of the vertebral body end plates.

## 42.7.8.1 Percutaneous Nucleus Replacement (PNR: TranS1)

The percutaneous nucleus replacement (PNR) released by TranS1 consists of a titanium screw system anchoring itself onto the superior and inferior vertebrae, with a central membrane that is filled with curable material and acts as the nucleus. It is inserted in a caudal-rostral manner anterior to the sacrum axially through the vertebra to replace the damaged disk and restore the natural motion while also preserving the integrity of the annulus fibrosus and ligaments so as to reduce the risk of implant migration. Because of the approach, the system is limited to L5–S1. It has been in limited use since 2008, and its clinical performance has yet to be evaluated [38].

## 42.7.8.2 Perspectives on Lumbar Disk Arthroplasty

With the advent of new technology and techniques in lumbar disk arthroplasty, interest in preserving spinal motion at degenerated motion segments has increased. The goals of lumbar disk arthroplasty are to provide long-term pain relief at the degenerated disk level, to restore disk height to protect neural elements, and to preserve motion to prevent posterior facet arthropathy and adjacentsegment disease. Despite the relatively positive early clinical results of these devices, questions remain about the long-term efficacy in pain relief and maintenance of motion, the results of randomized comparative trials with fusion, and the life span of the devices. In addition, late sequelae and revision options are unknown. Further biomechanical clinical data involving TDR are needed before its large-scale adoption as a surgical procedure. Areas of potential future investigation may include the kinematics of TDR and hybrid constructs at different lumbar levels, effects of TDR on sagittal balance parameters, detailed in vivo retrieval and wear studies, anatomic press-fit designs for end plates, and long-term effects of TDR on adjacent segments and facet joints.

Conceptually, nucleus replacement offers the promise of a novel technology to add to the spectrum of surgical techniques to treat a variety of degenerative spine pathologies. Specifically, nucleus replacement has the potential to address degenerative pathologies more complex than simple disk herniation, but less advanced than severe DDD. Therefore, nucleus replacement may fill a surgical niche between simple diskectomy and TDR or spinal fusion. However, the potential success of nucleus replacement must also be tempered by the lack of long-term clinical results. Today, new technologies for spine surgery are judged on the basis of their safety and efficacy as well as cost-effectiveness. Several nucleus replacement devices are entering or completing pilot feasibility studies. Further clinical investigation with well-designed prospective, randomized pivotal trials is needed to ultimately determine the ideal indications and efficacy of nucleus replacement in the treatment of lumbar DDD.

## References

- Rawlinson JJ, Punga KP, Gunsallus KL, Bartel DL, Wright TM. Wear simulation of the ProDisc-L disc replacement using adaptive finite element analysis. J Neurosurg Spine. 2007;7(2):165–73.
- Ray CD. Lumbar interbody threaded prosthesis. In: Brock M, Mayer HM, Weigel K, editors. The artificial disc. Berlin: Springer; 1991. p. 53–67.
- Bono CM, Lee CK. Critical analysis of trends in fusion for degenerative disc disease over the past 20 years: influence of technique on fusion rate and clinical outcome. Spine. 2004;29(4):455–63.
- Jackson RK, Boston DA, Edge AJ. Lateral mass fusion. A prospective study of a consecutive series with long-term follow-up. Spine. 1985;10:828–32.
- Kiviluoto O, Santavirta S, Salenius P, Morri P, Pylkkanen P. Posterolateral spine fusion. A 1–4-year follow-up of 80 consecutive patients. Acta Orthop Scand. 1985;56:152–4.
- O'Beirne J, O'Neill D, Gallagher J, Williams DH. Spinal fusion for back pain: a clinical and radiological review. J Spinal Disord. 1992;5:32–8.
- Stauffer RN, Coventry MB. Posterolateral lumbarspine fusion. Analysis of Mayo Clinic series. J Bone Joint Surg Am. 1972;54:1195–204.
- Carragee EJ. The surgical treatment of disc degeneration: is the race not to the swift? Spine J. 2005;5:587–8.
- Deyo RA, Nachemson A, Mirza SK. Spinalfusion surgery-the case for restraint. N Engl J Med. 2004;350:722–6.

- Errico TJ, Gatchel RJ, Schofferman K, Benzel EC, Faciszewski T, Eskay-Auerbach M, Wang JC. A fair and balanced view of spine fusion surgery. Spine J. 2004;4:129S–42.
- Sheehan JM, Shaffrey CI, Jane JA. Degenerative lumbar stenosis: the neurosurgical perspective. Clin Orthop Relat Res. 2001;384:61–74.
- Eck JC, Humphreys SC, Hodges SD. Adjacentsegment degeneration after lumbar fusion: a review of clinical, biomechanical, and radiology studies. Am J Orthop. 1999;28:336–40.
- Lee CK, Langrana NA. A review of spinal fusion for degenerative disc disease: need for alternative treatment approach of disc arthroplasty? Spine J. 2004;4:173S–6.
- Lehmann TR, Spratt KF, Tozzi JE, Weinstein JN, Reinarz SJ, el-Khoury GY, Colby H. Long-term follow-up of lower lumbar fusion patients. Spine. 1987;12:97–104.
- Ozgur BM, Aryan HE, Pimenta L, Taylor WR. Extreme lateral interbody fusion (XLIF): a novel surgical technique for anterior lumbar interbody fusion. Spine J. 2006;6:435–43.
- Fernstrom U. Arthroplasty with intercorporal endoprosthesis in herniated disc and in painful disc. Acta Chir Scand Suppl. 1966;357:154–9.
- Hitchon PW, Eichholz K, Barry C, et al. Biomechanical studies of an artificial disc implant in the human cadaveric spine. J Neurosurg Spine. 2005;2:339–43.
- Anderson PA, Rouleau JP, Bryan VE, et al. Wear analysis of the Bryan cervical disc prosthesis. Spine. 2003;28:S186–94.
- Bao QB, Yuan HA. Pioneer surgical technology NUBAC artificial nucleus. In: Kim DH, Cammisa FP, Fessler RG, editors. Dynamic reconstruction of the spine. New York: Thieme; 2006. p. 128–36.
- 20. Bao QB, Yuan HA. New technologies in spine: nucleus pulposus replacement. Spine. 2002;27:1245–7.
- 21. Bao QB, Yuan HA. Prosthetic disc replacement: the future? Clin Orthop. 2002;394:139–45.
- Bertagnoli R, Schonmayr R. Surgical and clinical results with the PDN prosthetic disc nucleus. Eur Spine J. 2002;11:S143–8.
- Bertagnoli R, Vazquez RJ. The anterolateral transpoatic approach (ALPHA): a new technique for implanting prosthetic disc-nucleus devices. J Spine Dis. 2003;16:398–404.
- Bertagnoli R, Karg A, Voigt S. Lumbar partial disc replacement. Orthop Clin N Am. 2005;36:341–7.
- Goins ML, Wimberley DW, Yuan PS, Fitzhenry LN, Vaccaro AR. Nucleus pulposus replacement: an emerging technology. Spine J. 2005;5:317S–24.
- Lemaire JP, Skalli W, Lavaste F, et al. Intervertebral disc prosthesis: results and prospects for the year 2000. Clin Orthop. 1997;337:64–76.
- Martino AD, Vaccaro AR, Lee JY, Denaro V, Lim MR. Nucleus pulposus replacement: basic science and indications for clinical use. Spine. 2005;30:S16–22.
- Frymoyer JW, Durrett CL. The economics of spinal disorders. In: Frymoyer JW, editor. The adult spine:

principles and practice. Philadelphia: Lippincott-Raven; 1997. p. 143–50.

- 29. Lee CK, Goel VK. Artificial disc prosthesis: design concepts and criteria. Spine J. 2004;4:209S–18.
- Rahm MD, Hall BB. Adjacent-segment degeneration after lumbar fusion with instrumentation: a retrospective study. J Spinal Disord. 1996;9:392–400.
- Putzier M, Funk JF, Schneider SV, Gross C, Tohtz SW, Khodadadyan-Klostermann C, Perka C, Kandziora F. Charite total disc replacement–clinical and radiographical results after an average follow-up of 17 years. Eur Spine J. 2006;15:183–95.
- Link HD. History, design and biomechanics of the LINK SB Charité artificial disc. Eur Spine J. 2002;11 Suppl 2:S98–105.
- Wang J, Mummaneni PV, Haid RW. Current treatment strategies for the painful lumbar motion segment: posterolateral fusion versus interbody fusion. Spine. 2005;30:S33–43.
- 34. Guyer RD, McAfee PC, Banco RJ, Bitan FD, Cappuccino A, Geisler FH, Hochschuler SH, Holt RT, Jenis LG, Majd ME, Regan JJ, Tromanhauser SG, Wong DC, Blumenthal SL. Prospective, randomized, multicenter Food and Drug Administration investigational device exemption study of lumbar total disc replacement with the CHARITE artificial disc versus lumbar fusion: five-year follow-up. Spine J. 2009;9(5):374–86.
- 35. Zigler JE, Glenn J, Delamarter RB. Five-year adjacent-level degenerative changes in patients with single-level disease treated using lumbar total disc replacement with ProDisc-L versus circumferential fusion. J Neurosurg Spine. 2012;17(6):504–11.
- 36. Gornet MF, Burkus JK, Dryer RF, Peloza JH. Lumbar disc arthroplasty with Maverick disc versus stand-alone interbody fusion: a prospective, randomized, controlled, multicenter investigational device exemption trial. Spine (Phila Pa 1976). 2011;36(25):E1600–11.
- 37. Pettine K, Hersh A. Kineflex lumbar artificial disc versus Charité lumbar total disc replacement for the treatment of degenerative disc disease: a randomized non-inferiority trial with minimum of 2 years' followup. SAS J. 2001;5(4):108–13.
- 38. Wang JC, Arnold PM, Hermsmeyer JT, Norvell DC. Do lumbar motion preserving devices reduce the risk of adjacent segment pathology compared with fusion surgery? A systematic review. Spine (Phila Pa 1976). 2012;37(22 Suppl):S133–43.
- Gertzbein SD, Seligman J, Holtby R, Chan KH, Kapasouri A, Tile M, Cruickshank B. Centrode patterns and segmental instability in degenerative disc disease. Spine. 1985;10(3):257–61.
- Panjabi MM. Centers and angles of rotation of body joints: a study of errors and optimization. J Biomech. 1979;12:911–20.
- Panjabi MM, Goel VK, Walter SD. Errors in kinematic parameters of a planar joint: guidelines for optimal experimental design. J Biomech. 1982;15:537–44.
- Seligman JV, Gertzbein SD, Tile M, Kapasouri A. Computer analysis of spinal segment motion in

degenerative disc disease with and without axial loading. Spine. 1984;9(6):566–73.

- Cho SK, Riew KD. Adjacent segment disease following cervical spine surgery. J Am Acad Orthop Surg. 2013;21(1):3–11.
- Coric D, Mummaneni P. Nucleus replacement technologies. J Neurosurg Spine. 2008;8:115–20.
- 45. Dooris AP, Goel VK, Grosland NM, Gilbertson LG, Wilder DG. Load-sharing between anterior and posterior elements in a lumbar motion segment implanted with an artificial disc. Spine. 2001;26:E122–9.
- 46. Moumene M, Geisler FH. Comparison of biomechanical function at ideal and varied surgical placement for two lumbar artificial disc implant designs: mobilecore versus fixed core. Spine. 2007;32:1840–51.
- O'Leary P, Nicolakis M, Lorenz MA, et al. Response of Charite total disc replacement under physiologic loads: prosthesis component motion patterns. Spine J. 2005;5:590–9.
- Goel VK, Grauer JN, Patel T, et al. Effects of charite artificial disc on the implanted and adjacent spinal segments mechanics using a hybrid testing protocol. Spine. 2005;30:2755–64.
- Panjabi M, Henderson G, Abjornson C, Yue J. Multidirectional testing of one- and two-level ProDisc-L versus simulated fusions. Spine. 2007;32(12):1311–9.
- Panjabi M, Malcolmson G, Teng E, et al. Hybrid testing of lumbar CHARITE discs versus fusions. Spine. 2007;32:959–66.
- Panjabi MM. Hybrid multidirectional test method to evaluate spinal adjacent-level effects. Clin Biomech (Bristol, Avon). 2007;22(3):257–65.
- 52. Cunningham BW, et al. Distribution of in vivo and in vitro ROM following 1-level arthroplasty with the CHARITÉ artificial disc compared with fusion. J Neurosurg Spine. 2008;8:7–12.
- 53. Harrop JS, Youssef JA, Maltenfort M, Vorwald P, Jabbour P, Bono CM, Goldfarb N, Vaccaro AR, Hilibrand AS. Lumbar adjacent segment degeneration and disease after arthrodesis and total disc arthroplasty. Spine. 2008;33(15):1701–7.
- Etebar S, Cahill DW. Risk factors for adjacentsegment failure following lumbar fixation with rigid instrumentation for degenerative instability. J Neurosurg. 1999;90:163–9.
- 55. Barrey C, Jund J, Noseda O, Roussouly P. Sagittal balance of the pelvis-spine complex and lumbar degenerative diseases. A comparative study about 85 cases. Eur Spine J. 2007;16:1459–67.
- 56. Chen WJ, Lai PL, Tai CL, Chen LH, Niu CC. The effect of sagittal alignment on adjacent joint mobility after lumbar instrumentation – a biomechanical study of lumbar vertebrae in a porcine model. Clin Biomech. 2004;19:763–8.
- 57. Hioki A, Miyamoto K, Kodama H, Hosoe H, Nishimoto H, Sakaeda H, Shimizu K. Two-level posterior lumbar interbody fusion for degenerative disc disease: improved clinical outcome with restoration of lumbar lordosis. Spine J. 2005;5:600–7.

- Hresko MT, Labelle H, Roussouly P, Berthonnaud E. Classification of high-grade spondylolistheses based on pelvic version and spine balance. Spine. 2007;32(20):2208–13.
- Hwang SH, Kayanja M, Milks RA, Benzel EC. Biomechanical comparison of adjacent segmental motion after ventral cervical fixation with varying angles of lordosis. Spine J. 2007;7:216–21.
- 60. Keorochana G, Taghavi CE, Lee KB, Yoo JH, Liao JC, Fei Z, Wang JC. Effect of sagittal alignment on kinematic changes and degree of disc degeneration in the lumbar spine. Spine. 2011;36(11):893–8.
- Kolstad F, Nygaard OP, Leivseth G. Segmental motion adjacent to anterior cervical arthrodesis. Spine. 2007;32(5):512–7.
- 62. Kretzer RM, Hu N, Umekoji H, Sciubba DM, Jallo GI, McAfee PC, Tortolani PJ, Cunningham BW. The effect of spinal instrumentation on kinematics at the cervicothoracic junction: emphasis on soft-tissue response in an in vitro human cadaveric model. J Neurosurg Spine. 2010;13:435–42.
- Levin DA, Hale JJ, Bendo JA. Adjacent segment degeneration following spinal fusion for degenerative disc disease. Bull NYU Hosp Joint Dis. 2007;65(1):29–36.
- 64. Nunley PD, Jawahar A, Cavanaugh DA, Gordon CR, Kerr EJ, Utter PA. Symptomatic adjacent segment disease after cervical total disc replacement: re-examining the clinical and radiological evidence with established criteria. Spine J. 2013; 13:5–12.
- Ozer E, Yucesoy K, Yurtsever C, Secil M. Kyphosis one level above the cervical disc disease. J Spinal Disord Tech. 2007;20(1):14–9.
- 66. Park JY, Cho YE, Kuh SU, Cho JH, Chin DK, Jin BH, Kim KS. New prognostic factors for adjacent-segment degeneration after one-stage 360° fixation for spondylolytic spondylolisthesis: special reference to the usefulness of pelvic incidence angle. J Neurosurg Spine. 2007;7:139–44.
- 67. Sudo H, Oda I, Abumi K, Ito M, Kotani Y, Minami A. Biomechanical study on the effect of five different lumbar reconstruction techniques on adjacent-level intradiscal pressure and lamina strain. J Neurosurg Spine. 2006;5:150–5.
- 68. Vialle R, Ilharreborde B, Dauzac C, Lenoir T, Rillardon L, Guigui P. Is there a sagittal imbalance of the spine in isthmic spondylolisthesis? A correlation study? Eur Spine J. 2007;16:1641–9.
- Lemaire JP, Carrier H, el-H S, Skalli W, Lavaste F. Clinical and radiological outcomes with the Charité artificial disc: a 10-year minimum follow-up. J Spinal Disord Tech. 2005;18(4):353–9.
- Cunningham BW, Dmitriev AE, Hu N, McAfee PC. General principles of total disc replacement arthroplasty: seventeen cases in a nonhuman primate model. Spine. 2003;28(20):S118–24.

- Cunningham BW, Gordon JD, Dmitriev AE, Hu N, McAfee PC. Biomechanical evaluation of total disc replacement arthroplasty: an in vitro human cadaveric model. Spine. 2003;28(20):S110–7.
- Denozière G, Ku DN. Biomechanical comparison between fusion of two vertebrae and implantation of an artificial intervertebral disc. J Biomech. 2006;39(4):766–75.
- Cunningham BW. Basic scientific considerations in total disc arthroplasty. Spine J. 2004;4:S219–30.
- LeHuec JC, Kiaer T, Friesem T, Mathews H, Liu M, Eisermann L. Shock absorption in lumbar disc prosthesis: a preliminary mechanical study. J Spinal Disord Tech. 2003;16:346–51.
- Kurtz SM, van Ooij A, Ross R, de Waal MJ, Peloza J, Ciccarelli L, Villarraga ML. Polyethylene wear and rim fracture in total disc arthroplasty. Spine J. 2007;7(1):12–21.
- Van Ooij A, Oner FC, Verbout AJ. Complications of artificial disc replacement: a report of 27 patients with the SB Charite disc. J Spinal Disord Tech. 2003;16:369–83.
- 77. Geisler FH, Blumenthal SL, Guyer RD, McAfee PC, Regan JJ, Johnson JP, Mullin B. Neurological complications of lumbar artificial disc replacement and comparison of clinical results with those related to lumbar arthrodesis in the literature: results of a multicenter, prospective, randomized investigational device exemption study of Charite intervertebral disc. Invited submission from the Joint Section Meeting on Disorders of the Spine and Peripheral Nerves, March 2004. J Neurosurg Spine. 2004;1:143–54.
- Leary SP, Regan JJ, Lanman TH, Wagner WH. Revision and explantation strategies involving the CHARITE lumbar artificial disc replacement. Spine. 2007;32:1001–11.
- 79. Siepe CJ, Heider F, Wiechert K, Hitzl W, Ishak B, Mayer MH. Mid- to long-term results of total lumbar disc replacement: a prospective analysis with 5to 10-year follow-up. Spine J. 2014;14:1417–31. pii: S1529-9430(13)01475-7.
- Lowell TD, Errico TJ, Fehlings MG, DiBartolo TJ, Ladosi L. Microdiskectomy for lumbar disk herniation: a review of 100 cases. Orthopaedics. 1995;18:985–90.
- Yorimitsu I, Chiba K, Toyama Y, Hirabayashi K. Long-term outcomes of standard discectomy for lumbar disc herniation: a follow-up of more than 10 years. Spine. 2001;26:652–7.
- Brinckmann P, Grootenboer H. Change of disc height, radial disc bulge, and intradiscal pressure from discectomy: an in vitro investigation on human lumbar discs. Spine. 1991;16:641–6.
- Bertagnoli R, Sabatino CT, Edwards JT, Gontarz GA, Prewett A, Parsons JR. Mechanical testing of a novel hydrogel nucleus replacement implant. Spine J. 2005;5:672–81.

Pedicle Screw-Based Dynamic Stabilization Devices in the Lumbar Spine: Biomechanical Concepts, Technologies, Classification, and Clinical Results 43

Cédric Barrey, Eurico Freitas, and Gilles Perrin

## 43.1 Introduction

Spinal motion-sparing technologies were introduced to address the adverse effects of traditional spinal fusion: stiffness, pseudarthrosis, mechanical failure, and/or adjacent degenerative disease [1–8]. There are three different basic concepts involved with these technologies:

- Partial/total disk replacement
- Total facet replacement
- Posterior dynamic stabilization (PDS) devices

Compared to disk and facet replacement technologies, PDS systems are designed to stabilize the spinal segments without removing any part of the native disk and facet joints. They are

C. Barrey, MD, PhD (🖂)

Department of Neurosurgery C and Spine Surgery, Neurological Hospital, University Claude Bernard Lyon 1, Lyon, France e-mail: c.barrey@wanadoo.fr

E. Freitas, MD • G. Perrin, MD Neurochirurgie C Chirurgie du Rachis, Hôpital Neurologique Groupement Hospitalier Est Hospices Civils de Lyon, Lyon, France e-mail: gilles.perrin@chu-lyon.fr classically divided into two categories [9–11]: interspinous process spacers and pedicle screw-based systems (Fig. 43.1).

Pedicle screw-based PDS systems have an advantage over total disk replacement in that they are based on techniques familiar to surgeons who have experience with posterior approach and traditional pedicular-based instrumented spinal fusions. The basic concept is to reduce the stiffness of the instrumentation to allow for load sharing between the instrumentation and the functional spine unit (FSU) at the instrumented levels [12]. Various implant designs have thus been developed to achieve this goal: reduction of diameter metallic rods, hinged pedicle screwheads that allow motion, damper components in the longitudinal elements, and more flexible rods made of nonmetallic biomaterials.

Most PDS devices are approved for use as adjuncts to spinal fusion. However, inherent to their design, PDS systems are also utilized to address chronic low back pain by degenerative disk disease (DDD) and spinal stenosis. These devices may relieve symptoms of stenosis as well as discogenic pain by controlling motion, providing a certain degree of disk space distraction and indirect neurologic elements decompression, decreasing intervertebral disk stress, and unloading the facet joints. PDS devices may

© Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_43



Fig. 43.1 Classification of spinal motion preservation technologies

also reduce transmission of stress to the adjacent levels and potentially decrease the incidence of adjacent segment disease [11, 13–16].

The authors recently performed an extensive review of biomechanical and clinical investigations involving pedicle screw-based PDS devices currently available for use clinically [17–22]. The current chapter presents the most significant results from this review in both the biomechanical and clinical fields.

# 43.2 Biomechanical Concepts

To analyze the biomechanical effects of pedicle screw-based PDS devices, we have to distinguish two different indications: non-fusion (i.e., posterior dynamic stabilization) and fusion (i.e., dynamic fusion).

#### 43.2.1 Dynamic Stabilization

The three basic biomechanical requirements for an ideal posterior dynamic stabilization device are [23]:

- Unload the intervertebral disk evenly with a predictable load distribution
- Control physiologic 3D motion (range of motion, ROM, and location of mean axis of rotation, AR)
- Maintain/restore sagittal balance and anatomic alignment





#### 43.2.1.1 Load Distribution

The intervertebral disk is composed of a homogeneous gel of collagen and proteoglycans and forms the main load-bearing structure in the FSU [24, 25]. Through experimental studies, the lumbar intervertebral disk demonstrated very high resistance to axial compression (ranging from 3,000 to 5,000 N) [24] with compressive loads on the nucleus redistributed evenly in a radial direction (Fig. 43.2a).

Degenerative disk disease affects the ability of nucleus to bear compressive loads, and consequently more load is transferred to the peripheral regions of the disk and annulus (Fig. 43.2b) [4, 5, 26]. Profilometry in vivo studies by McNally [13] confirmed this abnormal load distribution in degenerative disks. In addition, disk degeneration results in the loss of disk height and increase of axial compressive loads to the posterior elements, potentially accelerating facet arthrosis.

Sengupta recently proposed that asymmetrical load distribution rather than abnormal motion was the main cause of mechanical low back pain [16]. This concept is supported by correlations between abnormal stress distribution across the disk space and painful disks on discography. The presence of high load zones has been described as the "stone in the shoe" theory by Mulholland [12]. Although current surgical treatment has focused on restricting intervertebral motion, abnormal transmission of load across the disk space is likely a significant source of low back pain. Similar process has been observed for hip and knee joints with asymmetrical load distribution resulting in accelerating arthrosis. Correcting osteotomy in the lower limbs has been therefore proposed to obtain more even load distribution.

Pedicle screw-based PDS devices have been designed as load-bearing devices to unload the degenerated disk and minimize load transfer to the posterior facets (Fig. 43.2c). By reducing the pressure on the degenerated disk and facets, pedicle screw-based PDS devices may reduce mechanical pain associated with these structures [15, 16, 27].

Load transmission through the functional spinal unit (FSU) following implantation of PDS systems has been investigated by finite element analysis and by measuring intradiscal pressure in cadaveric studies using flexible pressure transducers [28] (Table 43.1). Through these biomechanical studies, significant unloading of the intervertebral disk was observed, especially in extension and after distraction, whereas in flexion, intradiscal loads were similar to the loads of the intact spine.

	Schmoelz et al. [29] experimental study		Zander et al. [30] FEA	Rohlmann et al. [31] FEA	
Loading condition	Dynamic	Rigid	Dynamic	Dynamic	Rigid
Standing	NT	NT	ns	Ļ	11
Flexion	ns	ns	ns	ns	ns
Extension	11	↓↓	ns	Ļ	↓↓
Lat bend	Ļ	Ļ	NT	NT	NT
Axial rot	1	Ļ	ns	ns	ns

**Table 43.1** Load transmission at the instrumented level following implantation of PDS device and compared to the intact condition

NT not tested, NS not significant, FEA finite element analysis, Dynamic: posterior dynamic stabilization, Rigid: rigid instrumentation

#### 43.2.1.2 Kinematics

The intervertebral kinematics basically described the relative displacement of the upper vertebra with respect to the lower vertebra. The mean segmental flexion-extension range of motion (ROM) in the lumbar spine ranges from 10 to 18° depending on the vertebral level. In flexion-extension, the mean axis of rotation (AR) is located just below the surface of the upper endplate of the lower vertebra in the posterior one-third of the disk space (Fig. 43.3) [32]. The normal location of COR results from the natural tendency of the vertebra to move spontaneously in the path of least mechanical resistance.

In 1992 Panjabi defined "spinal instability" as an increase of the neutral zone on a loaddisplacement curve [33]. The neutral zone corresponds to the initial segment of the curve where there is low resistance offered by the FSU and subsequently great displacement occurs from minimal load application.

Although the relationship between low back pain and instability is not clearly defined yet, most authors accept that excessive nonphysiologic motion may manifest as mechanical low back pain. According to the staged grading system of Kirkaldy-Willis, instability occurs in the early phase of disk degeneration, whereas the restabilization stage with reduction of motion and ankylosis of the FSU occurs in the late phase [34].



**Fig. 43.3** Location of the mean AR in sagittal rotation (flexion-extension) in the lumbar spine [32]

Results from experimental studies involving pedicle screw-based PDS devices [11, 35–39] support the biomechanical concept of "controlled motion without instability." Scifert et al. has demonstrated in a calf model that PDS devices have the ability to provide significant stability with reduction of ROM compared to intact spines in flexion-extension and lateral bending [40].

By reducing excessive intervertebral motion, we can expect that implantation of such devices will result in relief of mechanical pain related to instability.



**Fig. 43.4** ROM following implantation of the DYNESYS® implant (ROM Instrum) compared to ROM of the intact spines (ROM Intact) (averaged results from four different human cadaveric in vitro studies [35, 41–44])

However, the biomechanical studies have demonstrated that pedicle screw-based PDS systems were more efficient to control ROM in flexion-extension and lateral bending than in axial rotation. This is important clinically as these devices may not be beneficial in cases of rotational instabilities, laterolisthesis, or scoliosis deformities. As examples, ROM following implantation of the DYNESYS® device ranged from 20 to 45 % of intact ROM in flexion, from 33 to 94 % in extension, from 26 to 40 % in lateral bending, and from 76 to 181 % in axial rotation [35, 41–44]. The mean ratio that was measured from experimental studies between DYNESYS<sup>®</sup> ROM and intact ROM is presented in Fig. 43.4.

Compared to soft stabilization devices, especially DYNESYS® implant which was the most investigated pedicle screw-based PDS device through experimental studies, few data are available concerning the effects on ROM following insertion of semirigid devices. Compared to soft PDS systems like DYNESYS®, these devices may probably result in a greater control in 3D motion, especially in axial rotation. That could justify their use in cases of deformity or instability. In case of degenerative spondylolisthesis, although Freudiger et al. found that DYNESYS<sup>®</sup> implant could control partially horizontal translation [41], semirigid devices may also be more suitable than soft PDS.

Restoration of normal segmental kinematics requires the return of ROM in physiologic range as well as physiologic quality of motion with respect to the location of the mean AR. The posterior placement of PDS systems is relatively far away from the physiologic axis of rotation. Subsequently, the posterior shift of the mean AR induced by most of the pedicle screw-based PDS results in a compression effect of the whole disk in flexion (anterior and posterior annulus) and a distraction effect, i.e., the unloading of the whole disk, in extension (Fig. 43.5b). This is in contrast to the normal kinematic behavior of the FSU (Fig. 43.5a).

The non-physiologic kinematics following pedicle screw-based devices may result in overloading the intervertebral disk in some positions, especially when the spine is flexed (sitting position). Otherwise, some authors proposed the use of combination of PDS devices with total disk prosthesis anteriorly to deal with disk and facets degenerative changes [45]. According to us, there is a potential risk of conflict between the kinematic of the disk prosthesis and that of the PDS device. Although combination of the two types of motion preservation technology may be an attractive option, kinematic data suggest that such a combined construct may potentially result in overloading the disk prosthesis in flexion and complete unloading of the prosthesis in extension with the risk of dislocation.

More recently, using a nonlinear 3D finite element model of L4–L5, Jahng et al. [46] found that the center of rotation and stress distribution differed according to the design and materials used and also confirmed that the biomechanical effects induced by the implantation of dynamic stabilization systems produced non-physiological stress on the functional spine unit.

In 2014, using a 3D finite element model of the L4–L5 segment, Alapan et al. [47] confirmed that alterations in the location of the COR results in significant changes of load-sharing characteristics within the spine segment.



**Fig. 43.5** Consequences of posterior shift of the axis of rotation on intervertebral kinematics [17]. (a) Normal kinematics of the FSU in flexion-extension. (b) Kinematics of the FSU after implantation of pedicle screw-based PDS system

#### 43.2.1.3 Alignment/Posture

It is now well accepted that restoration of sagittal balance with respect to anatomic spino-pelvic parameters is paramount in optimizing outcomes after lumbar fusion surgery [48–50]. Referring to classes of pelvis incidence may be particularly helpful to evaluate the amount of lordosis required in a spinal fusion procedure [51]. In general, restoration of adequate lordosis during fusion surgery is achieved by contouring the rod and then realigning the spine along the bended rod.

The use of flexible rods may therefore pose difficulties when trying to manipulate, correct, and maintain spinal alignment in the setting of spinal deformity, making this technology less optimal in these settings. Restoration of a large amount of lordosis may also be challenging.

In addition, most PDS systems provide a certain degree of posterior distraction to unload facet joints and posterior annulus and widen neural foramina potentially resulting in radicular and discogenic pain relief. However, posterior distraction may also lead to focal kyphosis and subsequently increased adjacent level stresses (Fig. 43.6).

Legaye et al. reported a study on the unfavorable influence of pedicle screw-based PDS systems on the sagittal balance [52]. This author, who analyzed the impact of PDS implantation on sagittal alignment, noted that PDS devices were associated with loss of lumbar lordosis and pelvis backtilt, i.e., pelvis retroversion.

More recently, in 2011, Chen et al. [53] analyzed and compared the restoration of lordosis after implantation of hybrid versus purely dynamic instrumentation in the lumbar spine. Twenty-nine patients were included in the study. Local and global lordosis was measured using specific software. The authors concluded that hybrid construct could better preserve lordosis at



the instrumented levels, therefore reducing the compensatory hyperextension of the adjacent spine above. Further study is needed to determine the efficacy of PDS devices in maintaining postoperative physiologic sagittal balance.

Otherwise, further clinical follow-up and correlation between pelvic morphometry (flat versus lordotic alignment) is necessary to better understand the consequences of such occurrences. The forces witnessed in the intervertebral disk depend partly on patient-specific spino-pelvic organization [50, 51] with predominant axial compressive forces noted in flat spines and predominantly shear forces in lordotic spines. Inherent to their design, pedicle screw-based devices are probably more suitable to control axial compression than shear forces, especially those with the presence of a spring/damper.

## 43.2.1.4 Implant Longevity and Adjacent Level

In contrast to traditional spinal fusion, dynamic devices must stay anchored to the bone in the setting of continued intervertebral motion. One of the complications that may therefore arise from

instrumentation without fusion may be mechanical failure of the implant like loosening at the screw-bone interface or screw breakage [54]. Stoll et al. reported that rates of screw loosening, probably underestimated, were approximately 10 % in a series of 73 patients implanted with the DYNESYS<sup>®</sup> system (mean follow-up of 38 months) [55]. Benezech and Mitulescu recently reported the clinical and radiographical results of a series of 33 patients instrumented with the ISOLOCK<sup>®</sup> device without fusion [56]. After a mean follow-up of 45 months, they noted the presence of 5 mechanical complications (3 cases of screw breakage out of 148 screws implanted (2 %), 1 case of unscrewing of the screw nut, and 1 case of material loosening). However, there was no correlation between the presence of a mechanical complication and clinical results (the functional results were good or excellent in 76 % of patients with a return to previous work rate of 87 %).

Although mechanical failure of pedicle screwbased PDS devices may probably be unavoidable with time, such devices should have the advantage to delay the need of spinal fusion.

Otherwise, PDS offers a potential advantage over traditional rigid instrumentation in terms of adjacent segment load transmission [57]. However, to date, no clinical studies have reported their efficacy with regard to this objective. Additionally, some biomechanical studies did not find clear benefits of posterior dynamic stabilization in terms of minimizing adjacent level disease [29, 31, 44, 58]. As examples Schmoelz et al. [29] reported no difference in intradiscal pressure for both dynamic and rigid stabilization compared to intact spines, and Castellvi et al. [59] found only a 5.5 % reduction in maximal stresses provided by dynamic instrumentation versus rigid fixation at the adjacent level. These results suggest that the difference in the biomechanical effect between dynamic and rigid stabilization may not be as high as reported [58]. However, while the stress reduction may be small, it could be clinically significant because this effect is repeated over many loading cycles.

In 2012, Mageswaran et al. [60] reported results from a biomechanical experimental study involving seven human spine specimens and comparing three configurations: intact, one-level rigid instrumentation (L4-L5), and two-level hybrid construct (L3-L5) with dynamic stabilization at the level above (L3-L4). The idea was to evaluate the protective effect of the disk above (L3-L4) from excessive ROM and the creation of a transitional zone. The authors demonstrated that the biomechanical behavior of the dynamic instrumentation was very close to the rigid instrumentation. It reduced ROM at the above disk (L3–L4) but transformed one-level lumbar fusion into two-level lumbar fusion with augmentation of stresses and mobility at the supraadjacent levels (L1-L2 and L2-L3).

Early clinical data suggests that results are as good as those reported using rigid instrumentation; however, further long-term followup studies comparing dynamic versus rigid stabilization are required to determine the efficacy of PDS on adjacent segment disease (Table 43.2). **Table 43.2**Biomechanics and pedicle screw-based PDSdevices: key points

Capacity to unload the intervertebral disk in extension, lateral bending, and axial compression
Kinematics:
Control ROM in flexion-extension and lateral bending
Posterior shift of the mean axis of rotation
Only few data available concerning metallic semirigid devices
DYNESYS <sup>®</sup> implant has been the most investigated device
Risk of hypolordotic construct (inherent to their design)
Reduction of stresses at adjacent segments
Implant longevity as a limitation (high risk of screw loosening)

## 43.2.2 Dynamic Fusion

#### 43.2.2.1 Introduction

Surgery for low back pain due to degenerative disk disease and/or facet arthritis can be divided into three main options (Fig. 43.7):

- Limitation of motion (i.e., dynamic stabilization using motion-preserving devices)
- Restoration of motion (i.e. involving disk/facets replacement implants)
- Intervertebral fusion (considered as the gold standard)

Each option requires specific and adequate instrumentation to achieve the aim of the surgery.

In almost all cases, intervertebral fusion is currently performed using traditional rigid instrumentation. However, some authors advocated using a less rigid instrumentation to enhance intervertebral fusion success, thus introducing the concept of dynamic fusion [61–63].

Historically, most pedicle screw-based posterior dynamic systems (PDS) were initially designed to improve interbody fusion success in combination with an interbody bone graft [17–21]. In fact, most PDS devices are currently approved in the USA and Europe for use as adjuncts to spinal fusion and not as dynamic



Fig. 43.7 Main strategies for low back pain surgery including the concept of dynamic fusion

stabilization system. What is confusing is that these technologies are most often used as nonfusion devices (dynamic stabilization). Most papers and clinical reports on pedicle screwbased PDS concerned the concept of dynamic stabilization without fusion. Using the following keywords in a PubMed search, "lumbar dynamic instrumentation and fusion" and "Dynamic fusion and lumbar spine," we found 47 papers on dynamic instrumentation, including 22 clinical reports, 14 biomechanical studies, and 5 reviews. All the papers involving pedicle screw-based PDS focused on the concept of dynamic stabilization but none on dynamic fusion.

## 43.2.2.2 Concept of Dynamic Instrumentation for Fusion

Dynamic instrumentation for fusion has been introduced in the 1990s to address the adverse effects of traditional spinal fusion observed with rigid instrumentation: pseudarthrosis, bone rarefaction, and mechanical failure.

Some authors suggested that eliminating mechanical loads on an interbody bone graft may result in negative bone remodeling, pseudarthroses, and osteoporosis [23, 64, 65]. This "stressshielding" phenomenon at the disk space level may result from the excessive stiffness of traditional rigid instrumentation. Reducing the stiffness of the instrumentation, pedicle screw-based PDS allows for load sharing between the instrumentation and the functional spine unit (FSU) at the instrumented level(s). Using a finite element model of the lumbar spine, several authors demonstrated that posterior dynamic instrumentation, compared to rigid instrumentation, increases the amount of load transmission through the anterior column and the interbody bone graft, thus avoiding stress-shielding phenomenon. This may favor osteogenesis and enhance interbody fusion in



**Fig. 43.8** Finite element analysis illustrating loadsharing phenomenon using posterior dynamic instrumentation (*right*) versus traditional rigid system (*left*) (From F Lavaste and G Perrin with permission, 1993, Laboratory of Biomechanics, ENSAM, Arts et Metiers Paristech, Paris, unpublished data)

accordance with Wolff's law according to which the bone will adapt to the loads it is placed under, i.e., the structure and shape of bone permanently adapt to the loading conditions [66, 67]. Overload exposes to the risk of graft osteonecrosis, whereas underload may result in bone graft resorption. Thus, the basic concept of dynamic fusion is fewer loads through the instrumentation and more loads through the interbody bone graft without comprising stability, i.e., load sharing versus stress shielding.

In 1993, Lavaste and Perrin (unpublished data), using a finite element model of the lumbar spine, confirmed that dynamic posterior stabilization with *ISOBAR TTL*<sup>M</sup>, compared to rigid instrumentation, increases the amount of load transmission through the anterior column (Fig. 43.8).

Through a finite element analysis (FEA), Duffield et al. [64] compared the effects of three different longitudinal devices (4.8 and 6.3 mm rods and plate). They found that the axial load passing through the FSU was greater with 4.8 mm rod compared to 6.3 mm rod and/or plates (90 % versus 77 %, respectively). By using a canine model, Lim et al. [68] demonstrated that a less rigid stabilization device could reduce device-



**Fig. 43.9** In flexion, predominant load transfers through the system depend on instrumentation stiffness: a dynamic system results in anterior compression and posterior traction, while a rigid system results in axial pullout forces at the ends of the construct (Adapted from Templier et al. [65])

**Table 43.3** Theoretical advantages of dynamic instrumentation for fusion: key points

Load sharing between the instrumentation and interbody bone graft
Avoid stress-shielding phenomenon in the anterior column
Stresses reduction at bone-to-screw interface
Less rigid fused segment

related osteopenia in the stabilized spinal segments and around the pedicle screws. In 1998, Templier et al. [65] using a 3D geometric FE model of the lumbar spine postulated that the TWINFLEX<sup>®</sup> semirigid device could offer a more favorable biomechanical environment for enhanced interbody fusion healing (Fig. 43.9). They evaluated the role of the longitudinal component in load transfer between the FSU and implant and noted that by reducing the stiffness of lumbar fixation, there was more homogeneous load transmission throughout the FSU without significantly reducing the rigidity of the instrumented spinal segment.

Finally, Goel et al. [61] developed a 3D finite element model to compare the load distribution of a hinged-dynamic posterior device versus a rigid construct and confirmed that the dynamic system enabled more load to be transferred through the anterior column as compared with traditional rigid instrumentation without comprising stability.

Theoretical biomechanical advantages of dynamic instrumentation for fusion are summarized in Table 43.3. These advantages may result in increase fusion rates, limitation of bone rarefaction, and reduction of mechanical complications with the ultimate objective to reduce reoperations rates.

Presently, most pedicle screw-based PDS devices are FDA approved as an adjunct for spinal fusion. Although controversies remain regarding the support of such a classification, pedicle screw-based PDS systems are classically divided into semirigid rod systems and tension-band posterior-based systems used as a non-fusion technology [4-6, 10-12, 14, 16]. Only semirigid PDS systems could logically serve for dynamic fusion since excessive flexibility provided by soft stabilization PDS devices may allow for excessive motion and excessive anterior loading of the interbody graft, resulting in endplate failure, subsidence, decreased fusion rates, and sagittal plane deformity (flat back). Consequently, a classification of pedicle screwbased PDS devices based on motion restriction is needed to separate soft from semirigid rod systems with which a fusion is generally intended.

# 43.2.2.3 Biomechanical and Clinical Experience with ISOBAR TTL<sup>™</sup> Technology for Fusion

#### Implant Design

The basic concept of posterior dynamic systems is to reduce the stiffness of the instrumentation to allow for more physiologic load transmission at the instrumented levels. Various technologies have been introduced to allow for partially controlled three-dimensional motion or micromotion: reduction of diameter metallic rods, hinged pedicle screwheads that allow motion, damper components in the longitudinal elements, and more flexible rods made of non-metallic biomaterials [19–21].

In the literature, pedicle screw-based PDS devices are divided as follows: metallic rod systems (considered as semirigid stabilization),



**Fig. 43.10** ISOBAR  $TTL^{M}$ , 1997, evolution of ISOLOCK<sup>®</sup> device 1993, Scient'x – Alphatec, France (Reprinted with permission)

tension-band posterior-based systems (considered as soft stabilization), and hybrid devices.

ISOBAR TTL<sup>™</sup> consists of a metallic semirigid pedicle screw-based PDS made of titanium (minimum artifacts on MRI and CT) (Fig. 43.10) [21].

It contains a damper element in its longitudinal element, a 5.5 mm titanium alloy rod. The damper, i.e., the dynamic component, allows reduced stiffness and limited amount of angular and axial micromotion (Figs. 43.11 and 43.12).

The damper provides  $\pm 2.25^{\circ}$  angular ROM in flexion-extension and lateral bending, no limitation in axial rotation (unconstrained), and  $\pm 0.4$  mm axial ROM (Fig. 43.12).

Concerning the surgical technique, this implant requires the same procedure as fusion performed with standard instrumentation using pedicle screws and rigid rod. Due to the familiarity of spine surgeons with pedicle screws placement, the learning curve for the implantation of the device is practically nonexistent.

#### In Vitro Testing

To support a classification of pedicle screw-based PDS systems, based on motion restriction, in vitro experimental investigations are needed. Concerning *ISOBAR TTL*<sup>imestarrow</sup> device, the experimental evaluation was performed by N'dri in the laboratory of biomechanics, Arts et Metiers Paristech, Paris, France (unpublished data).

Six human L2–S1 spinal specimens were tested intact, injured (laminectomy at L4–L5), and instrumented at L4–L5 using *ISOBAR*  $TTL^{TM}$  implant. Biomechanical tests were carried out using an optoelectronic system. Loads were applied to the upper vertebra (L2) with the lower vertebra (S1) fixed in a container. Pure moments were applied in flexion-exten-



**Fig. 43.11** Stacked washers within the dynamic component (From Scient'x – Alphatec, France, reprinted with permission)



**Fig.43.12** Range of motion provided by ISOBAR TTL<sup>™</sup> implant (Reprinted with permission)

sion, torsion, and lateral bending. Linear and angular displacements were measured using reflective markers rigidly fixed on L4 and L5 vertebrae (Fig. 43.13).

Results for instrumented spines in terms of range of motion, compared to intact spines, are presented in Fig. 43.14.

Through this experimental investigation, the authors found that ROM following implantation of the posterior dynamic implant *ranged from 20 to 50%*, depending on the loading condition. These results suggested that semirigid devices provide a greater control in 3D motion, especially

in axial rotation, in comparison with results reported for soft stabilization devices.

#### **Clinical Experience**

To the best of our knowledge, one of the first introduced semirigid rod is the ISOBAR TTL<sup>™</sup> implant (1997, evolution of ISOLOCK<sup>®</sup> device 1993) which has been now used in Europe for over 15 years and was approved FDA clearance for use as an adjunct to spinal fusion in 1999.



Fig. 43.13 Testing device

of ISOBAR TTL™

The first clinical implantation of the ISOLOCK<sup>™</sup> device was performed by G Perrin in June 1993. In 1996 he published a report on the usefulness of intervertebral titanium cages for PLIF and dynamic posterior fixation. Patients were treated with semirigid ISOLOCK® plates for lumbar degenerative disk disease and spondylolisthesis. The fusion rate was more than 95 % without any mechanical failure of the instrumentation (Figs. 43.15 and 43.16).

Unfortunately, there is no prospective study available comparing rigid versus dynamic instrumentation for fusion in the lumbar spine. Only case series have been reported in the literature [21, 63,69]. The largest series has been reported by G Perrin (800 patients implanted with *ISOBAR TTL<sup>™</sup>*); however, this author unfortunately mixed in his series patients with dynamic stabilization (no fusion), dynamic fusion, and hybrid constructs (rigid + dynamic) making the results difficult to assess. He retrospectively reported an overall fusion rate of 98 % with no mechanical complications.

There is no data available in the literature concerning the fusion period and/or the bone graft volume comparing dynamic versus rigid instrumentation.

In fact we consider dynamic instrumentation as an option to treat degenerative disk disease for given indications and given lumbar levels. The use of metallic rods with dampers may pose difficulties when trying to maintain spinal alignment or to restore a large amount of lordosis. Because of these sagittal balance considerations [48, 49, 51, 52], we estimate that dynamic instrumentation should be avoided at L5-S1 level. The authors feel that the best indications correspond to one or two levels to





**Fig. 43.16** Case 2: L2–L3 severe stenosis treated by L2–L3 interbody fusion using semirigid posterior dynamic stabilization in combination with interbody graft (PEEK

cages). Simple interlaminar decompression without diskectomy was performed at L3–L4  $\,$ 

be fused between L2–L3 and L4–L5 (Fig. 43.17). To restore sufficient segmental lordosis with ISOBAR TTL<sup>TM</sup>, it is essential to apply compression between the screwheads along the rod.

Otherwise, when using dynamic instrumentation for fusion, the authors' preference is to place systematically the bone graft trough the intervertebral space (PEEK cages) rather than to realize an interlaminar and/or inter-transverse graft. Although some authors advocate the use of dynamic instrumentation in combination with posterolateral bone graft, we consider that dynamic instrumentation associated with interbody graft is more pertinent from a biomechanical point of view.

#### 43.2.2.4 Conclusion

In comparison with the cervical spine, dynamic anterior cervical plates have been progressively



**Fig. 43.17** A 55-year-old man presented with severe back and leg pain, not responding to conservative treatment. Imaging revealed a degenerative stenosis in L1–L2 and L2–L3 in relation with an L2–L3 disk hernia and inflammatory DDD at L1–L2. He was treated by decom-

introduced to provide a better graft loading with the ultimate objectives to accelerate spinal fusion and lead to a lower incidence of postoperative mechanical complications [70]. The use of dynamic instrumentation for fusion in the lumbar spine applies the same biomechanical concept, i.e., favoring load sharing versus stress shielding.

Further prospective studies are now needed to confirm the efficacy of PDS devices in enhancing spinal fusion and especially determine the advantages of dynamic instrumentation in terms of fusion period, fusion rates, and fusion quality.

# 43.3 Technologies and Classification of Pedicle Screw-Based Dynamic Stabilization Devices

The basic concept of PDS systems is to reduce the stiffness of the instrumentation to allow for more physiologic load transmission at the instrumented levels. However, the design of these

pression, followed by PLIF on two levels with cages and dynamic instrumentation. Surgical treatment provided a significant extent of immediate symptoms relief. One year after, a CT scan exam showed a solid fusion. Pain and disability levels decreased to 3/10 and 28/100, respectively

devices varies greatly. In the literature, pedicle screw-based PDS devices are divided as follows: metallic rod systems (considered as semirigid stabilization), tension-band posterior-based systems (considered as soft stabilization), and hybrid devices [71].

For these three categories, different implant designs have been introduced into the implant market:

- Implants that have introduced a small diameter rod to allow for greater motion: TWINFLEX<sup>®</sup> and BIOFLEX<sup>®</sup>
- An implant that utilizes a hinged pedicle screwhead to allow motion: COSMIC<sup>®</sup>
- Implants that contain a damper/coupler element in their longitudinal element: ISOBAR TTL<sup>®</sup>, ALADYN<sup>®</sup>, DYNAMO<sup>®</sup>, PERFX-2<sup>®</sup>, DSS<sup>®</sup>, and N-FLEX<sup>®</sup>
- Implants that replace the metallic longitudinal element by an elastic or solid polymer longitudinal element: GRAF ligament, CD HORIZON<sup>®</sup> PEEK rod, EXPEDIUM<sup>®</sup> PEEK rod, FLEXPLUS<sup>®</sup>, and DYNESYS<sup>®</sup>

Pedicle screw-based PDS systems are an attractive type of dynamic stabilization system due to the surgeon's familiarity with pedicle screw placement. Tables 43.4, 43.5, and 43.6 show examples of currently available pedicle screw-based PDS systems.

Device	Technology	Picture
Graf ligament Showa Ika Kohgyo, Japan (reprinted with permission)	8 mm braided polyester nonelastic tension bands	
DYNESYS® (DYnamic NEutralization SYStem) Zimmer Spine, USA (reprinted with permission)	Cylindrical polycarbonate- urethane (PCU) spacer Tensioned polyethylene terephthalate (PET) cord tunneled through the PCU spacer	
DYNESYS DTO® (DYnamic NEutralization SYStem) Zimmer Spine, USA (reprinted with permission)	Combined cord-rod construct and rigid rod offering the ability to transition from a rigid to dynamic system	
CD HORIZON <sup>®</sup> LEGACY <sup>™</sup> PEEK rod Medtronic Sofamor Danek, USA (reprinted with permission)	PEEK rod (polyetheretherketone)	

Table 43.4 Soft PDS pedicular-based devices

Device	Technology	Picture
CD HORIZON BalanC <sup>®</sup> Medtronic Sofamor Danek, USA (reprinted with permission)	Rod with a combination of silicone and PEEK materials	Dynamic Portion Fusion Portion
TRANSITION <sup>®</sup> Globus Medical, USA (reprinted with permission)	Preassembled or intraoperatively assembled PET central cord, PCU flexible spacers, and compressible bumpers Titanium alloy spools	
EXPEDIUM <sup>®</sup> PEEK rod DePuy Synthes, USA (reprinted with permission)	Polyetheretherketone (PEEK) rod	Store Store
FLEXPLUS® Spine Vision, Paris, France	Multiple titanium fibers with a polycarbonate-urethane sheet	

#### Table 43.4 (continued)
Device	Technology	Picture
TWINFLEX <sup>®</sup> (1992) Scient'X – Spine Network, France (reprinted with permission)	Twin rods 2×2.5 mm	
ISOLOCK <sup>®</sup> (1993) Scient'x – Alphatec, France (reprinted with permission)	Intrapedicular connection allowing micro-movements with the limits of 0.8 mm in axial compression and 4° in flexion-extension	
ISOBAR® TTL (1997, evolution of ISOLOCK® device 1993) Scient'x – Alphatec, France (reprinted with permission)	5.5 mm titanium alloy rod Damper allowing reduced stiffness and limited amount of angular and axial micromotion	
ALADYN <sup>®</sup> (2003) Scient'x – Alphatec, France (reprinted with permission)	S-shaped damper inside a rectangular box	

Table 43.5 Metallic rod PDS pedicular-based devices

Device	Technology	Picture
ACCUFLEX <sup>®-</sup> Globus Medical, USA (reprinted with permission)	6.5 titanium rod Helical cuts	
BIOFLEX <sup>®</sup> BioSpine, Korea (reprinted with permission)	4 mm Nitinol spring rod (one or two loops) "Memory metal"	
DSS <sup>®</sup> Paradigm Spine, Germany (reprinted with permission)	Polyaxial screws with slotted and rigid couplers	THEFT
HPS <sup>®</sup> Paradigm Spine, Germany (reprinted with permission)	Rigid spinal fixation with a topping-off option (DSS <sup>®</sup> coupler)	

### Table 43.5 (continued)

(continued)

Device	Technology	Picture
COSMIC <sup>®</sup> Ulrich Medical, Ulm, Germany (reprinted with permission)	6.25 mm rod Integrated hinge in the pedicle screwheads	
DYNAMO <sup>®</sup> Scient'x – Alphatec, USA (reprinted with permission)	5.5 mm titanium alloy rod Damper allowing limited amount of angular and axial micromotion based on Scient'x ISOBAR®	
PERFX-2 <sup>®</sup> Eden Spine, FL, USA	The dynamic rod damper provides ±5° in flexion- extension and ±2 mm of axial ROM	
WAVEFLEX <sup>®</sup> Medyssey, Seoul, Korea	Titanium alloy rod	

### Table 43.5 (continued)

Device	Technology	Picture
STABILIMAX <sup>®</sup> Applied Spine Technologies, New Haven, CT, USA (reprinted with permission)	Dual spring combined with ball-and-socket joint mechanism Cobalt-chromium and titanium alloy	
N-FLEX <sup>®</sup> N Spine Inc., USA (reprinted with permission)	6 mm titanium rod with one end containing a composite titanium and PCU sleeve	
		Neutral Flexion Extension

Table 43.6 Hybrid dynamic PDS pedicular-based devices

### 43.4 Clinical Investigations

### **43.4.1 Graf Ligament** (Table **43.4**)

Introduced in 1992, the Graf ligament was probably the first pedicle screw-based PDS device available for use clinically. It was initially proposed by his conceptor, Henri Graf, as an alternative treatment for spinal instability without fusion [72].

# Summary of Biomechanical Investigations [73, 74]

- In vitro study: 2; FEA: 0
- Significant reduction of ROM in flexionextension and lateral bending

- Failure to reduce intervertebral motion in axial rotation (107–132 % of intact ROM) and translation of the vertebral body
- Application of Graf ligament increases annulus and nucleus stresses
- Risk of exacerbating facet disease and neuroforamen stenosis due to increase of segmental lordosis

Henri Graf theorized that abnormal rotary motion might be the primary source of mechanical low back pain. The Graf ligament was therefore proposed to stabilize rotary motion and realign the segment in physiological lordosis. The "ligament" was also intended to compress the posterior annulus and allow healing of annular tears. Initial outcomes following Graf ligament placement showed only modest improvement in functional ability with a high revision surgery rate.

Grevitt et al. [73] reported outcomes of 50 chronic lower back pain patients that were implanted with the Graf ligament device. They noted improvement in Oswestry Disability Index (ODI) scores from 59 preoperatively to 31. However, de novo postoperative radiculopathy was reported in 12 of 50 (24 %) patients, and 20 revisions or further procedures were performed in 13 patients. Only few mechanical failure of the construct were observed with one screw displacement and two screw breakages (1.2 % of screws used).

Hadlow et al. [75] published a retrospective case-control study comparing clinical and radiological outcomes at 1 and 2 years between soft stabilization with the Graf ligament and instrumented posterolateral fusion. They found significantly better outcomes in patients following a traditional spinal fusion compared to those treated with the Graf ligament, especially due to a higher revision rate in the Graf treatment group (55 and 73 % for Graf ligament group versus 37 and 43 % for the traditional fusion group at 1 and 2 years, respectively).

Rigby et al. [76] presented mid- and long-term results of a series of 51 patients treated with the Graf ligament. The mean follow-up of the study was approximately 4 years. Patients had only a six-point improvement in their ODI scores, and seven patients (14 %) required additional traditional fusion procedure. Finally, 41 % of patients reported that they wouldn't undergo a similar procedure again.

More recent prospective randomized evaluations reported better clinical outcomes in patients undergoing the Graf ligament placement compared to fusion.

In 2003 good to excellent results were reported by Madan et al. [77] in 93 % for patients operated with the Graf ligament group compared to 78 % for those treated by anterior lumbar interbody fusion. Their prospective randomized study compared the efficacy of Graf ligamentoplasty (28 patients) in comparison with rigid instrumented ALIF and insertion of the Hartshill horseshoe cage (27 patients) for similar severity of disk degeneration.

Kanayama et al. [78] reported a retrospective study aiming to assess midterm clinical and radiographic results of Graf artificial ligament stabilization in the treatment of degenerative spondylolisthesis. Sixty-four patients with degenerative spondylolisthesis were treated with the Graf ligament and the mean follow-up was approximately 5 years. Visual analogical scales of low back pain and radicular symptoms were significantly improved from 72 and 76 preoperatively to 14 and 15 postoperatively, respectively. The radiographic results demonstrated that Graf artificial ligament stabilization did not affect vertebral slip and disk height but maintained segmental lordosis (which was  $9.8 \pm 5.9^{\circ}$ before surgery,  $12.8 \pm 6.7^{\circ}$  immediately after surgery, and  $12.2 \pm 8.9^{\circ}$  at the final follow-up) and preserved segmental motion (ROM at the operative level was  $11.2 \pm 5.6^{\circ}$  before surgery and  $4.7 \pm 4.6^{\circ}$  at the final follow-up). Finally, the authors reported additional surgeries in only four cases (6.25 %) for adjacent degenerative disease.

The same authors recently published a retrospective long-term follow-up study with a mean 10-year follow-up [79]. Fifty-six patients who were treated with the Graf system were included in the study. Results showed that the construct maintained segmental lordosis and disability improved in patients with low-grade spondylolisthesis or flexion instability. A patient with scoliosis or lateral listhesis, however, had poor clinical improvement requiring reoperation.

### **43.4.2 TWINFLEX**<sup>®</sup> (Table 43.5)

To the best of our knowledge, the TWINFLEX<sup>®</sup> implant was probably the first metallic semirigid PDS device available for use clinically.

### Summary of Biomechanical Investigations [65]

- In vitro study: 0; FEA: 1
- Load sharing between the anterior column and the instrumentation
- Response to a flexion moment results rather in anterior compression and posterior traction than in axial pullout forces

Korovessis et al. [80] conducted a prospective comparative randomized study to compare the postoperative effects of a rigid versus TWINFLEX® instrumentation for degenerative spine disease and stenosis with a special focus on sagittal lumbar spine alignment. Thirty patients were included in the study with equal number of patients for each type of instrumentation. Evaluation of patients was mainly radiological analysis preoperatively and at 3 months postoperatively. The authors found that both rigid and dynamic instrumentations restored lumbar lordosis, sacral tilt, and distal lordosis and increased the foraminal diameter at the level L4-L5 resulting in an indirect decompression of the nerve roots at this level. They concluded that the dynamic system could be used with the same indications with the rigid in degenerative lumbar spine because it can offer equally good short-term results regarding sagittal spine alignment.

Champain et al. [69] performed a retrospective study to evaluate the interest of quantified radiographic analysis of lumbar spine in global outcome assessment of 49 patients treated by posterolateral lumbosacral fusion with iliac crest bone graft and TWINFLEX<sup>®</sup> dynamic instrumentation. The authors used Beaujon-Lassalle score to assess clinical outcome, and the mean follow-up of the study was 5 years. Sixty-one and 29 % of patients presented very satisfactory and satisfactory clinical outcomes, respectively. Complications were adjacent level degeneration in four cases (8 %), pseudarthrosis in two patients (4 %), and hardware failure in three cases (6 %), comparing favorably with literature data for rigid fusion.

## 43.4.3 ISOBAR TTL® (Table 43.5)

One of the first introduced semirigid rods is also the ISOBAR TTL<sup>®</sup> implant (evolution of ISOLOCK<sup>®</sup> device) which has been used in Europe for over 10 years and was approved FDA clearance for use as an adjunct to spinal fusion in 1999. Composed of a titanium alloy rod with a damper element made of stacked titanium alloy, the device allows a small amount of both axial and angular motion to the rigid rod.

Summary of Biomechanical Investigations [20, 21, 59, 81]

- In vitro study: 2; FEA: 1
- Axial (0.4 mm) and angular (2.25°) micromotion capacity
- Stabilization comparable to a rigid rod in terms of ROM
- · Increase of anterior column loading
- Reduction of maximal stresses in the next adjacent level by approximately 5.5 % in flexion

The first clinical implantation of the ISOLOCK<sup>®</sup> device was performed by G Perrin in June 1993. In 1996 he published a report on the usefulness of intervertebral titanium cages for PLIF and dynamic posterior fixation. Patients were treated with semirigid ISOLOCK<sup>®</sup> plates for lumbar degenerative disk disease and spondylolisthesis [63]. The fusion rates was more than 95 % without any mechanical failure of the instrumentation.

Benezech and Mitulescu recently reported a retrospective study [56] to analyze clinical and radiographical long-term outcomes of 33 patients treated with the ISOLOCK<sup>®</sup> device. The surgical procedure was posterior dynamic stabilization without fusion, and the mean follow-up was 45 months. Seventy-five percent of patients had one-level stabilization and 25 % two levels. Good or excellent outcome was noted in 76 % with a return to previous work rate of 88 %. No spontaneous fusion was observed at the instrumented levels. The authors noted the preservation of preoperative disk height in more than 90 % of cases. Finally, they observed five mechanical complications: three screw breakages (out of 148 screws used), one unscrewing of the screw nut, and one material loosening.

In 2012, Li et al. [82] reported a 2-year follow-up study after insertion of the ISOBAR TTL<sup>®</sup> device. Thirty-seven patients were included consecutively with a minimal follow-up of 12 months. Both VAS and ODI decreased significantly postoperatively. New signs of degeneration at adjacent levels were observed on MRI in 14 patients (39 %), and revision surgery at adjacent segment was necessary for 3 patients (8 %). Finally, the authors observed screw loosening in four patients (11 %). They concluded that clinical improvement after implantation of ISOBAR TTL<sup>®</sup> was good but not really superior to traditional spinal fusion.

## 43.4.4 DYNESYS® (DYnamic NEutralization SYStem for the Spine) (Table 43.4)

The DYNESYS<sup>®</sup> device consists of two titanium pedicle screws connected by a cylindrical spacer (made of polycarbonate-urethane (PCU)) with a tensioned cord (made of polyethylene terephthalate (PET)) tunneled through the PCU spacer [83]. The spacer resists compression during extension and thereby maintains foraminal height while unloading the posterior annulus, and the tensioned PET cord resists tensile forces and provides resistance to spine flexion similar in concept to the Graf system. The device has been approved by the FDA as an adjunct for spinal fusion. Using DYNESIS DTO<sup>®</sup> system allows surgeons to treat different stages of degeneration at contiguous levels.

Summary of Biomechanical Investigations [27, 31, 35, 38, 42–44, 58, 84–86]

- In vitro study: 8; FEA: 4
- DYNESYS<sup>®</sup> was probably the most pedicle screw-based PDS system investigated
- Limits ROM from 20 to 45 % in flexion compared to intact spines, from 33 to 94 % in extension, from 26 to 40 % in lateral bending, and from 76 to 181 % in axial rotation
- Impact of spacer length and posterior intervertebral distraction on kinematics and load transmission
- Unloads the intervertebral disk mainly in extension

Clinical studies on the DYNESYS<sup>®</sup> system used as a non-fusion device have shown heterogenous clinical outcomes compared to traditional rigid fusion [54, 55, 87–92].

The first clinical report was published in 2002 by Stoll et al. [55] who performed a prospective and multicenter study evaluating clinical and radiological outcomes of DYNESYS® in a series of 83 patients. Indications for the surgical procedure were segmental instability combined with spinal stenosis in 60 %, degenerative disk disease in 24 %, disk herniation in 8 %, and previous diskectomy in 6 %. The population of patients included 39 cases of degenerative spondylolisthesis as a secondary diagnosis. In most cases (56/83), dynamic instrumentation was combined with direct decompression and involved one level in 55 cases, two levels in 17 cases, three levels in 8 cases, and four levels in 3 cases. The mean follow-up was approximately 3 years. The mean low back pain score improved significantly from 7.4 to 3.1, mean leg pain scale from 6.9 to 2.4, and ODI from 55.4 to 22.9 %. There were ten mechanical complications: two screw malplacements, one screw loosening requiring reoperation, and seven radiological signs of screw loosening. The authors concluded that the dynamic neutralization proved to be a safe and effective alternative in the treatment of unstable lumbar conditions.

In 2005, Grob et al. [88] published a retrospective study on 50 consecutive patients who were treated with the DYNESYS® system for symptomatic degenerative disk disease or stenosis with associated instability (listhesis). Results were analyzed from clinical and radiological data of 31 patients with a minimum follow-up of 2 years. The surgery involved one level in 32 % of cases, two levels in 52 %, three levels in 13 %, and four levels in 4 %. The authors found that back and leg pain improved in 67 % and 64 % of patients, respectively; however, functional capacity only improved in 40 %, and within the 2-year follow-up period, 19 % of patients required an additional surgical procedure. They conclude that the results of the study provided no support for the notion that dynamic stabilization of the lumbar spine results in better patient-oriented outcomes than those of traditional fusion.

A prospective study published by Schnake et al. in 2006 especially focused on the use of DYNESYS<sup>®</sup> implant in cases of instability [91]. A total of 26 patients with lumbar spinal stenosis and degenerative spondylolisthesis were included in the study. The dynamic stabilization was combined with interlaminar decompression. Twenty-four patients were evaluated after 2 years with a mean follow-up of 26 months. The authors observed that mean leg pain, evaluated with VAS scale, decreased significantly from 80 preoperatively to 23 postoperatively and that mean walking distance improved significantly from 250 m to more than 1,000 m. 87.5 % of patients would undergo the same procedure again. Radiographically, overall progression of spondylolisthesis was not significant and evaluated at only 2.1 %. The authors noted asymptomatic implant failure in four patients: three screw loosening and one screw breakage. They concluded that in patients with spinal stenosis and degenerative spondylolisthesis, dynamic stabilization with the DYNESYS<sup>®</sup> device combined with decompression leads to similar clinical results than those reported in established protocols using decompression and fusion with pedicle screws.

Welch et al. [92] recently reported the results of a multicenter, randomized study (FDA and IDE clinical trial) examining the non-fusion application of the DYNESYS® device in a population of 101 patients. Indications for the surgical procedure were grade I degenerative retrolisthesis (n=3), grade I degenerative spondylolisthesis (n=20), and spinal stenosis (n=66). At 12 months follow-up, the mean VAS scale for back and leg pain and ODI improved significantly from 54 to 29, from 80 to 25, and from 56 % to 26 %, respectively. Fifteen of 101 patients (15 %) required 18 reoperations by the time of the 1-year follow-up evaluation. Ten of these 18 additional surgeries were performed at the same level due to low back pain, radiculopathy, or increased instability. The authors concluded that the early clinical outcomes of treatment with DYNESYS® were promising; however, long-term follow-up care was still needed.

Finally, in 2010, Kocak et al. [90] reported a review of the literature to determine the occurrence of screw loosening after DYNESYS<sup>®</sup> implantation. These authors found that revision surgery due to screw loosening was not necessary in the majority of cases and concluded that it is not clear if screw loosening influences the clinical results.

In 2010, the FDA recommended against approval for DYNESYS<sup>®</sup> as a spinal non-fusion technology because of the risk of unfavorable outcomes (such as screw loosening) and an unclearly defined patient population.

Fay et al. [86] reported in 2013 a retrospective comparative study involving patients with (n=24) and without degenerative spondylolisthesis (n=14) and operated with DYNESYS<sup>®</sup> at one or two levels. Mean follow-up was 41 months. Postoperative ROM at the index level in FE decreased from 10 to 2.7°. The authors observed screw loosening in 4.6 % of screws and 21.1 % of patients. There was no difference between the Another clinical study was reported in 2013 by Haddad et al. [93] with 4-year follow-up. The authors retrospectively compared outcomes of patients operated with DYNESYS<sup>®</sup> system (n=32) versus traditional rigid fusion (n=32). Improvement of VAS (back and leg pain) and ODI was greater for the fusion group, and more patients were satisfied after fusion compared to after DYNESYS.

### **43.4.5 ACCUFLEX**<sup>®</sup> (Table 43.5)

The ACCUFLEX<sup>®</sup> rod has been developed with a double helical cut within a standard 6.5 mm rod to reduce rod stiffness and is presently FDA approved as an adjunct for one-level fusion when used in conjunction with an interbody graft.

### **Summary of Biomechanical Investigations**

- In vitro study: 0; FEA: 0
- · Only fatigue tests were performed
- Intended to limit axial rotation and lateral bending while permitting flexion-extension
- Amount of flexibility depends on the number of circumferential cuts

Mandigo et al. [94] reported a prospective and randomized study of 170 patients in which 54 received the ACCUFLEX<sup>®</sup> rod system. Results at 1-year follow-up were comparable between the dynamic (n=54) and the rigid (n=116) group with similar fusion rates (92 % and 95 %, respectively) and clinical outcomes (evaluated by VAS and short form-16 scales). No instrumentation failure was reported in this series.

## **43.4.6 CD HORIZON<sup>®</sup> LEGACY<sup>™</sup> PEEK Rod** (Table 43.4)

The implant consists of non-metallic rods made of polyetheretherketone (PEEK) with a modulus of elasticity close to that of natural bone. It has been FDA approved as an adjunct for fusion when used in conjunction with an interbody graft (one-level interbody fusion). In addition, the rod is radiolucent allowing the surgeon to assess the quality of spinal fusion using X-rays.

### Summary of Biomechanical Investigations [95]

- In vitro study: 2; FEA: 1
- Intended to allow some motion
- More flexible than titanium but comparable flexion-extension rigidity when PEEK rod is combined with an interbody spacer
- Unloads the bone-screw interface and increases anterior column load (approximately 75 % anterior and 25 % posterior)
- Control 3D motion better in FE and LB compared to AR (80 % of ROM reduction in FE, 70 % in LB, and 54 % in AR)

In 2007, Highsmith et al. reported their preliminary clinical experience with the LEGACY<sup>™</sup> PEEK rod implant [96]. They described three cases in which the PEEK rod was implanted. Indications for surgery were adjacent segment disease, degenerative disk disease, and grade I degenerative spondylolisthesis. In all three cases, the dynamic implant was used as an adjunct for fusion. No data are available concerning clinical and radiological methods used for evaluation.

In 2012, Medtronic introduced the CD HORIZON BalanC<sup>™</sup> spinal system which uses a unique combination of PEEK and silicone

materials and geometry. The rod is made of silicone and PEEK in its dynamic portion, while the fusion portion is entirely made of PEEK, thus creating a transitional zone between the fused and the mobile segments. The device is indicated to treat multilevel spinal surgeries requiring fusion at one or more levels and neutral stabilization (non-fusion) at adjacent level (only for use outside the USA).

In 2013, Athanasakopoulos et al. [97] reported a series of 52 patients who underwent posterior spinal fusion with CD HORIZON PEEK rod with a mean follow-up of 3 years. Mean Oswestry index improved from 76 % preoperatively to 30 % at 1 year. Bone fusion evaluated on standard and dynamic radiographs was observed in 96 % of patients (n=50/52). Two complications were recorded in the series: one infection and one hardware breakage necessitating revision surgery.

### **43.4.7 STABILIMAX**<sup>®</sup> (Table 43.6)

The device is FDA approved since 2007. During normal spine kinematics, the neutral zone (NZ) is a region of intervertebral motion around the neutral posture where little resistance is offered by the passive spinal column. This NZ is increased at the early phase of the disk degeneration process resulting in instability, excessive motion, and mechanical pain. The STABILIMAX<sup>®</sup> uses a rod with concentric springs to maintain the spinal segment in a neutral position during spinal motion.

#### Summary of Biomechanical Investigations [98]

- In vitro study: 1; FEA: 0
- Reduction of the NZ by increasing resistance of the passive spinal system around the NZ while preserving maximal ROM
- No precise data available concerning effects on NZ and ROM

Clinical trials are planned under an investigational device exemption from the US Food and Drug Administration.

### 43.4.8 BIOFLEX® (Table 43.5)

The BIOFLEX<sup>®</sup> system is a pedicle screwbased system COMBINING a Nitinol rod shaped with one or two loops confering stability in flexion, extension, and lateral bending. Nitinol is an alloy of nickel and titanium, commonly referred to as "memory metal" for its ability to return back to its original shape after deformation.

### Summary of Biomechanical Investigations

- In vitro study: 0; FEA: 0
- No data available

Kim et al. reported outcomes on 103 patients who underwent implantation of the BIOFLEX® device [99]. Patients were divided into two groups: dynamic stabilization without fusion (n=46) and dynamic fusion in which the dynamic device was combined with an interbody cage and graft (n=57). In the dynamic stabilization group, mean VAS score decreased from 7.3 preoperatively to 1.4 postoperatively, ODI from 35 % to 12 %, and ROM from 4 to 10°, respectively. In the dynamic fusion group, mean VAS score decreased from 7.4 preoperatively to 2.1 postoperatively and ODI from 38 to 14 %, and the fusion rates of segments treated by BIOFLEX® and PLIF were approximately 90 %. Complications observed in the two groups were one cage retropulsion, presence of a halo in three cases, one screw breakage, unstable levels in eight cases, and one screw loosening. The authors concluded that the BIOFLEX® device was safe and efficient and could serve as a dynamic stabilization device and as an adjunct for dynamic fusion.

### **43.4.9 DSS<sup>®</sup> and DSS<sup>®</sup> HPS** (Table 43.5)

The DSS<sup>®</sup> hybrid performance system (HPS) is designed to combine rigid spinal fixation with a topping-off option including the DSS<sup>®</sup> coupler. The system was developed in collaboration with the Biomechanical Laboratory of Ulm University (Germany). The system allows for mono- and multilevel constructs. It has been implanted in several thousand of cases all over the world with good outcomes.

### **43.4.10 EXPEDIUM® PEEK Rod** (Table 43.4)

The EXPEDIUM<sup>®</sup> PEEK rod more closely mimics the material characteristics of cancellous or cortical bone with a lower modulus of elasticity compared to titanium. It therefore allows for load sharing between the posterior elements and the interbody bone graft. In addition, PEEK rod is associated with imaging compatibility with reduction of artifacts and better visualization of fusion mass.

### **Summary of Biomechanical Investigations**

- In vitro study: 0; FEA: 2
- Reduction of ROM: flexion motion the most limited and axial rotation the least limited

#### Conclusion

To conclude this chapter, we would like to underline the point that pedicle screw-based posterior dynamic systems could serve as *fusion technology* and also as *non-fusion technology*.

Historically, pedicle screw-based posterior dynamic systems (PDS) were initially designed to improve interbody fusion success in combination with an interbody bone graft; however, these technologies are most often used as non-fusion devices (dynamic stabilization) resulting in confusion through the literature.

*Dynamic fusion* using semirigid devices to improve rates of interbody fusion success is now well documented through fundamental investigations; however, further clinical prospective studies are now needed to confirm the efficacy of PDS devices in enhancing spinal fusion and especially determine the advantages of dynamic instrumentation in terms of fusion period, fusion rates, and fusion quality.

Pedicle screw-based PDS devices have also been progressively introduced to serve as *dynamic stabilization* devices. In this last setting, they intend to address the adverse effects of traditional spinal fusion, especially to minimize the incidence of adjacent segment degeneration by restoring some native motion at the instrumented level. Biomechanical investigations have widely demonstrated that such devices could control intervertebral motion without instability while unloading the intervertebral disk. These systems can therefore be considered as an alternative to fusion to treat degenerative disk disease.

Early clinical results suggest that results are as good as those reported using rigid instrumentation. In fact, no clear significant differences were identified between traditional spinal fusion and dynamic stabilization regarding VAS, functional scales, and postoperative complications. Therefore, no clinical data from comparative studies really support the use of dynamic stabilization over standard spinal fusion.

Otherwise, further long-term follow-up studies comparing dynamic versus rigid stabilization are required to determine the efficacy of PDS on adjacent segment disease.

Finally, the question of implant longevity in the setting of continued intervertebral motion is still controversial, and it might be possible that posterior dynamic stabilization systems could serve to delay the need of spinal fusion but not avoid its requirement at long term. Until now, no dynamic stabilization devices received FDA approval for use other than as an adjunct to spinal fusion. The FDA specified that separate approval would be required for "off-label" marketing of these devices, including but not limited to use as stand-alone device (spinal stabilization in the absence of fusion).

### References

- Chou D, Lau D, Skelly A, Ecker E. Dynamic stabilization versus fusion for treatment of degenerative spine conditions. Evid Based Spine Care J. 2011; 2(3):33–42.
- Ghiselli G, Wang JC, Bhatia NN, Hsu WK, Dawson EG. Adjacent segment degeneration in the lumbar spine. J Bone Joint Surg Am. 2004;86:1497–503.
- Goel VK, Lim TH, Gwon J, Chen JY, Winterbottom JM, Park JB, Weinstein JN, Ahn JY. Effects of rigidity of an internal fixation device. A comprehensive biomechanical investigation. Spine. 1991;16:S155–61.
- Huang RC, Girardi FP, Lim MR, Cammisa FP. Advantages and disadvantages of nonfusion technology in spine surgery. Orthop Clin N Am. 2005;36: 263–9.
- Huang RC, Wright TM, Panjabi MM, Lipman JD. Biomechanics of non fusion implants. Orthop Clin N Am. 2005;36:271–80.
- Inceoglu S. Posterior dynamic stabilization of the lumbar spine. World Spine J. 2006;1:62–7.
- Ishihara H, Osada R, Kanamori M, Kawaguchi Y, Ohmori K, Kimura T, Matsui H, Tsuji H. Minimum 10 year follow-up of anterior lumbar interbody fusion for isthmic spondylolisthesis. J Spinal Disord. 2001;14:91–9.
- Kaner T, Ozer AF. Dynamic stabilization for challenging lumbar degenerative diseases of the spine: a review of the literature. Adv Orthop. 2013;753470:13.
- Gomleksiz C, Sasani M, Oktenoglu T, Ozer AF. A short history of posterior dynamic stabilization. Adv Orhop. 2012(2012);629698:12.
- Khoueir P, Kim A, Wang MY. Classification of posterior dynamic stabilization devices. Neurosurg Focus. 2007;22:E3.
- Wright T, Tauber M, Meyers K, Sudin Y, Arnin U, Girardi F. The biomechanics of posterior motion preservation systems. Spine J. 2005;5:143S–4.
- Mulholland RC, Sengupta DK. Rationale, principles and experimental evaluation of soft stabilization. Eur Spine J. 2002;11:S198–205.
- McNally DS, Shackleford IM, Goodship AE, Mulholland RC. In vivo stress measurement can predict pain on discography. Spine. 1996;21:2580–7.
- McNally DS. Rationale for dynamic stabilization. In: Dynamic reconstruction of the spine. New York: Thieme eds; 2006. p. 237–43.

- Nockels RP. Dynamic stabilization in the surgical management of painful lumbar spinal disorders. Spine. 2005;30:S68–72.
- Sengupta DK. Dynamic stabilization devices in the treatment of low back pain. Orthop Clin N Am. 2004;35:43–56.
- Barrey CY, Ponnappan RK, Song J, Vaccaro AR. Biomechanical evaluation of pedicle screw-based dynamic stabilization devices for the lumbar spine. SAS J. 2008;2:159–70.
- Barrey CY, Vaccaro AR, Song J, Ponnappan R, Perrin G. Stabilisation dynamique du rachis lombaire à l'aide de systèmes reposant sur vis pédiculaires : évaluation biomécanique, technologies, et classification [in french]. Le Rachis 2009, vol 6, N°4.
- Barrey CY. Pedicle screw-based dynamic stabilization devices for the lumbar spine. Argospine News J. 2009;21(2):78–84.
- 20. Barrey CY. Dynamic instrumentation for fusion with Isobar TTL<sup>™</sup>: biomechanical and clinical aspects. Argospine News J. 2010;22(2):62–6.
- Barrey CY, Perrin G, Champain S. Pedicle-screw based dynamic systems and degenerative lumbar diseases: biomechanical and clinical experiences of dynamic fusion with Isobar TTL<sup>™</sup>. ISRN Orthopedics. 2013;183702:10.
- 22. Song JJ, Barrey CY, Ponnappan RK, Bessey JT, Shimer AL, Vaccaro AR. Pedicle screw-based dynamic stabilization of the lumbar spine. Pan Arab Journal of Neurosurgery. 2010;14(1):1–8
- Goel VK, Gilbertson LG. Basic science of spinal instrumentation. Clin Orthop Relat Res. 1997;335: 10–31.
- Skalli W, Champain S, Mosnier T. Spine biomechanics. In: Lumbar and lumbo-sacral fusion alternatives. Paris: Elsevier Masson eds; 2007. p. 8–18 [in french].
- 25. Vital JM. Lumbar degenerative process. In: Alternative to lumbar and lumbo-sacral fusion. Paris: Elsevier Masson eds; 2007. p. 1–7 [in french].
- Natarajan RN, Williams JR, Andrsson GB. Modeling changes in intervertebral disc mechanics with degeneration. JBJS. 2006;88(Suppl2):36–40.
- Cabello J, Cavanilles-Walker JM, Iborra M, Ubierna MT, Covaro A, Roca J. The protective role of dynamic stabilization on the adjacent disc to a rigid instrumented level. An in vitro biomechanical analysis. Arch Orthop Trauma Surg. 2013;133(4):443–8.
- Cunningham BW, Kotani Y, McNulty PS, Cappuccino A, McAfee PC. The effect of spinal destabilization and instrumentation on lumbar intradiscal pressure: an in vitro biomechanical analysis. Spine. 1997;22: 2655–63.
- Schmoelz W, Huber JF, Nydegger T, Claes L, Wilke HJ. Influence of a dynamic stabilization system on load bearing of a bridged disc: an in vitro study of intradiscal pressure. Eur Spine J. 2006;15:1276–85.
- Zander T, Rohlmann A, Burra NK, Bergmann G. Effect of a posterior dynamic implant adjacent to a rigid spinal fixator. Clin Biomech. 2006;21: 767–74.

- Rohlmann A, Burra NK, Zander T, Bergmann G. Comparison of the effects of bilateral posterior dynamic and rigid fixation devices on the loads in the lumbar spine: a finite element analysis. Eur Spine J. 2007;16:1223–31.
- Pearcy MJ, Bogduk N. Instantaneous axes of rotation of the lumbar intervertebral joints. Spine. 1988;13: 1033–41.
- Panjabi MM. The stabilizing system of the spine. J Spinal Disord. 1992;5(4):383–96.
- Kirkaldy-Willis WH, Farfan HF. Instability of the lumbar spine. Clin Orthop Relat Res. 1982; 165:110–23.
- Cheng BC, Gordon J, Cheng J, Welch WC. Immediate biomechanical effects of lumbar posterior dynamic stabilization above a circumferential fusion. Spine. 2007;32:2551–7.
- Clausen JD, Goel VK, Sairyo K, Pfeiffer M. A protocol to evaluate semi-rigid pedicle screw systems. J Biomech Eng. 1997;119:364–6.
- Lee JK, Gomez J, Michelsen C, Kim Y, Moldavsky M, Chinthakunta SR, Khalil S. In vitro biomechanical study to quantify range of motion, intradiscal pressure and facet force of 3-level dynamic stabilization constructs with decreased stiffness. Spine. 2013;38(22): 1913–9.
- Xu HZ, Wang XY, Chi YL, Zhu QA, Lin Y, Huang QS, Dai LY. Biomechanical evaluation of a dynamic pedicle screw fixation device. Clin Biomech. 2006;21:330–6.
- 39. Yu AK, Siegfried CM, Chew B, Hobbs J, Sabersky A, Jho DJ, Cook DJ, Bellotte JB, Whiting DM, Cheng BC. Biomechanics of posterior dynamic fusion systems in the lumbar spine: implications for stabilization with improved arthrodesis. J Spinal Disord Tech. 2012.
- 40. Scifert JL, Sairyo K, Goel VK, Grobler LJ, Grosland NM, Spratt KF, Chesmel KD. Stability analysis of an enhanced load sharing posterior fixation device and its equivalent conventional device in a calf spine model. Spine. 1999;24:2206–13.
- Freudiger S, Dubois G, Lorrain M. Dynamic neutralization of the lumbar spine confirmed on a new lumbar spine simulator in vitro. Arch Orthop Trauma Surg. 1999;119:127–32.
- 42. Niosi CA, Zhu Q, Wilson DC, Keynan O, Wilson DR, Oxland TR. Biomechanical characterization of the three-dimensional kinematic behaviour of the Dynesys dynamic stabilization system: an in vitro study. Eur Spine J. 2006;15:913–22.
- Niosi CA, Wilson DC, Zhu Q, Keynan O, Wilson DR, Oxland TR. The effect of dynamic posterior stabilization on facet joint contact forces. Spine. 2008;33: 19–26.
- 44. Schmoelz W, Huber JF, Nydegger T, Claes L, Wilke HJ. Dynamic stabilization of the lumbar spine and its effects on adjacent segments. An in vitro experiment. J Spinal Disord Tech. 2003;16:418–23.
- Bertagnoli R, Tropiano P, Zigler J, Karg A, Voigt S. Hybrid constructs. Orthop Clin North Am. 2005;36:379–88.

- 46. Jahng TA, Kim YE, Moon KY. Comparison of the biomechanical effect of pedicle-based dynamic stabilization: a study using finite element analysis. Spine J. 2013;13(1):85–94.
- 47. Alapan Y, Sezer S, Demir C, Kaner T, Inceoglu S. Load sharing in lumbar spinal segment as a function of location of center of rotation. J Neurosurg Spine. 2014;20(5):542–9.
- Lazennec JY, Ramare S, Arafati N, Laudet CG, Gorin M, Roger B, Hansen S, Saillant G, Maurs L, Trabelsi R. Sagittal alignment in lumbosacral fusion : relations between radiological parameters and pain. Eur Spine J. 2000;9:47–55.
- Legaye J, Duval-Beaupère G, Hecquet J, Marty C. Pelvic incidence: a fundamental pelvic parameter for three-dimensional regulation of spinal sagittal curves. Eur Spine J. 1988;7:99–103.
- Roussouly P, Gollogly S, Berthonnaud E, Dimnet J. Classification of the normal variation in the sagittal alignment of the human lumbar spine and pelvis in the standing position. Spine. 2005;30:346–53.
- Barrey CY, Jund J, Noseda O, Roussouly P. Sagittal balance of the pelvis-spine complex and lumbar degenerative diseases. A comparative study about 85 cas. Eur Spine J. 2007;16:1459–67.
- Legaye J. Unfavorable influence of the dynamic neutralisation system on the sagittal balance. Rev Chir Orthop Reparatrice Appar Mot. 2005;91(6):542–50.
- Chen H, Charles YP, Bogorin I, Steib JP. Influence of 2 different dynamic stabilization systems on sagittal spinopelvic alignment. J Spinal Disord Tech. 2011; 24(1):37–43.
- 54. Wu JC, Huang WC, Tsai HW, Ko CC, Wu CL, Tu TH, Cheng H. Pedicle screw loosening in dynamic stabilization: incidence, risk, and outcome in 126 patients. Neurosurg Focus. 2011;31(4):E9.
- 55. Stoll TM, Dubois G, Schwarzenbach O. The dynamic neutralization system for the spine: a multicenter study of a novel nonfusion system. Eur Spine J. 2002;11:S170–8.
- Benezech J, Mitulescu A. Retrospective patient outcome evaluation after semi-rigid stabilization without fusion for degenerative lumbar instability. Eur J Orthop Surg Traumatol. 2007;17:227–34.
- Weinhoffer SL, Guyer RD, Herbert M, Griffith SL. Intradiscal pressure measurements above an instrumented fusion. A cadaveric study. Spine. 1995;20: 526–31.
- Rohlmann A, Neller S, Bergmann G, Graichen F, Claes L, Wilke HJ. Effect of an internal fixator and a bone graft on intersegmental spinal motion and intradiscal pressure in the adjacent regions. Eur Spine J. 2001;10:301–8.
- 59. Castellvi AE, Huang H, Vestgaarden T, Saigal S, Clabeaux DH, Pienkowski D. Finite element analysis of dynamic instrumentation demonstrates stress reduction in adjacent level discs. In submission.
- Mageswaran P, Techy F, Colbrunn RW, Bonner TF, McLain RF. Hybrid dynamic stabilization: a biomechanical assessment of adjacent and supra-adjacent

levels of the lumbar spine. J Neurosurg Spine. 2012;17(3):232–42.

- 61. Goel VK, Konz RJ, Chang HT, Grosland NM, Grobler LJ, Chesmel KD. Hinged-dynamic posterior device permits greater loads on the graft and similar stability as compared with its equivalent rigid device: a threedimensional finite element assessment. J Prosthetics Orthot. 2001;13:17–20.
- 62. Mazel C, Mitulescu A, Balabaud L, Antonietti P. Spinal arthrodesis with dynamic instrumentation. In: Alternative to lumbar and lumbo-sacral fusion. Paris: Elsevier Masson eds; 2007. p. 100–9 [in french].
- 63. Perrin G. Usefulness of intervertebral titanium cages for PLIF and posterior fixation with semi-rigid Isolock plates. In: Szpalski M, Gunsburg R, Spengler DM, Nachemson A, editors. Instrumented fusion of the degenerative lumbar spine: state of the art, questions and controversies. Philadelphia: Lippincott-Raven Publishers; 1996.
- 64. Duffield R, Carson W, Chen L, Voth B. Longitudinal element size effect on load sharing, internal loads, and fatigue life of tri-level spinal implant constructs. Spine. 1993;18:1695–703.
- Templier A, Denninger L, Mazel C, Lavaste F, Skalli W. Comparison between two different concepts of lumbar posterior osteosynthesis implants. A finite element analysis. Eur J Orthop Surg Traumatol. 1998;8:27–36.
- 66. Frost HM. A 2003 update of bone physiology and Wolff's law for clinicians. Angle Orthod. 2004;74: 3–15.
- 67. Wolff J. The law of bone remodeling. Berlin: Springer; 1986 (translation of the German 1892 edition).
- Lim TH, Goel VK, Winterbottom JM, Kessler B, Ahn JY, Gwon JK, Park JB, Weinstein JN. A comparison of stress-induced porosity due to conventional and a modified spinal fixation device. J Spinal Disord. 1994;7:1–11.
- Champain S, Mazel C, Mitulescu A, Skalli W. Quantitative analysis in outcome assessment of instrumented lumbosacral arthrodesis. Eur Spine J. 2007;16:1241–9.
- 70. Stulik J, Rainer Pitzen T, Chrobok J, Ruffing S, Drumm J, Sova L, Kucera R, Vyskocil T, Ingo Steudel W. Fusion and failure following anterior cervical plating with dynamic or rigid plates: 6-months results of a multi-centric, prospective, randomized, controlled study. Eur Spine J. 2007;16:1689–94.
- Murtagh R, Castellvi AE. Motion preservation surgery in the spine. Neuroimaging Clin N Am. 2014;24(2):287–94.
- Graf H. Lumbar instability: surgical treatment without fusion. Rachis. 1992;412:123–37.
- Grevitt MP, Gardner AD, Spilsbury J, Shackleford IM, Baskerville R, Pursell LM, Hassaan A, Mulholland RC. The Graf stabilization system: early results in 50 patients. Eur Spine J. 1995;4:169–75.
- Strauss PJ, Novotny JE, Wilder DG, Pope MH. Multidirectional stability of the graf system. Spine. 1994;19:965–72.

- Hadlow SV, Fagan AB, Hillier TM, Fraser RD. The Graf ligamentoplasty procedure. Comparison with posterolateral fusion in the management of low back pain. Spine. 1998;23:1172–9.
- Rigby MC, Selmon GP, Foy MA, Fogg AJ. Graf ligament stabilisation: mid- to long-term follow-up. Eur Spine J. 2001;10:234–6.
- Madan S, Boeree NR. Outcome of the Graf ligamentoplasty procedure compared with anterior lumbar interbody fusion with the hartshill horseshoe cage. Eur Spine J. 2003;12:361–8.
- Kanayama M, Hashimoto T, Shigenobu K, Oha F, Ishida T, Yamane S. Non-fusion surgery for degenerative spondylolisthesis using artificial ligament stabilization: surgical indication and clinical results. Spine. 2005;30:588–92.
- Kanayama M, Hashimoto T, Shigenobu K, Togawa D, Oha F. A minimum 10-year follow-up of posterior dynamic stabilization using Graf artificial ligament. Spine. 2007;32:1992–6.
- Korovessis P, Papazisis Z, Lambiris E. The role of rigid versus dynamic instrumentation for stabilization of the degenerative lumbosacral spine. Stud Health Technol Inform. 2002;91:457–61.
- 81. Yu SW, Yen CY, Wu CH, Kao FC, Tu YK. Radiographic and clinical results of posterior dynamic stabilization for the treatment of multisegment degenerative disc disease with a minimum follow-up of 3 years. Arch Orthop Trauma Surg. 2012;132(5):583–9.
- 82. Li Z, Li F, Yu S, Ma H, Chen Z, Zhang H, Fu Q. Twoyear follow-up results of the Isobar TTL semi-rigid rod system for the treatment of lumbar degenerative disease. J Clin Neurosci. 2013;20(3):394–9.
- Dubois GG, De Germay B, Prere J, Schwarzenbach O, Stoll TM. Dynamic neutralization: treatment of mobile vertebral instability. In: Dubois G, Sauramps (eds), Spinal restabilization procedures. Montpellier, France; 2006. p. 17–39.
- Kiapour A, Ambati D, Hoy RW, Goel VK. Effect of graded facetectomy on biomechanics of Dynesys dynamic stabilization system. Spine. 2012;37(10): E581–9.
- Meyers K, Tauber M, Sudin Y, Fleischer S, Arnin U, Girardi F, Wright T. The use of instrumented pedicle screws to evaluate load sharing in posterior dynamic stabilization systems. Spine J. 2008;8(6):926–32.
- 86. Shih SL, Chen CS, Lin HM, Huang LY, Liu CL, Huang CH, Cheng CK. Effect of spacer diameter of the Dynesys dynamic stabilization system on the biomechanics of the lumbar spine: a finite element analysis. J Spinal Disord Tech. 2012;25(5):E140–9.
- Fay LY, Wu JC, Tsai TY, Wu CL, Huang WC, Cheng H. Dynamic stabilization for degenerative spondylolisthesis: evaluation of radiographic and clinical outcomes. Clin Neurol Neurosurg. 2013;115(5):535–41.
- 88. Grob D, Benini A, Junge A, Mannion AF. Clinical experience with the Dynesys semirigid fixation system for the lumbar spine: surgical and patient-oriented outcome in 50 cases after an average of 2 years. Spine. 2005;30:324–31.

- Hu Y, Gu YJ, Xu RM, Zhou LJ, Ma WH. Short-term clinical observation of the Dynesys neutralization system for the treatment of degenerative disease of the lumbar vertebrae. Orthop Surg. 2011;3(3): 167–75.
- Kocak T, Cakir B, Reichel H, Mattes T. Screw loosening after posterior dynamic stabilization – review of the literature. Acta Chir Orthop Cech. 2010; 77(2):134–9.
- Schnake KJ, Schaeren S, Jeanneret B. Dynamic stabilization in addition to decompression for lumbar spinal stenosis with degenerative spondylolisthesis. Spine. 2006;31:442–9.
- 92. Welch WC, Cheng BC, Awad TE, Davis R, Maxwell JH, Delamarter R, Wingate JK, Sherman J, Macenski MM. Clinical outcomes of the Dynesys dynamic neutralization system: 1-year preliminary results. Neurosurg Focus. 2007;22:E8.
- Haddad B, Makki D, Konan S, Park D, Khan W, Okafor B. Dynesys dynamic stabilization: less good outcome than lumbar fusion at 4-year follow-up. Acta Orthop Belg. 2013;79(1):97–103.
- Mandigo CE, Sampath P, Kaiser MG. Posterior dynamic stabilization of the lumbar spine: pedicle

based stabilization with the Accuflex rod system. Neurosurg Focus. 2007;22:E9.

- 95. Gornet MF, Chan FW, Coleman JC, Murrell B, Nockels RP, Taylor BA, Lanman TH, Ochoa JA. Biomechanical assessment of a PEEK rod system for semi-rigid fixation of lumbar fusion constructs. J Biomech Eng. 2011;133(8):p. 12.
- Highsmith JM, Tumialan LM, Rodts Jr GE. Flexible rods and the case for dynamic stabilization. Neurosurg Focus. 2007;22:E11.
- Athanasakopoulos M, Mavrogenis AF, Triantafyllopoulos G, Koufos S, Pneumaticos SG. Posterior spinal fusion using pedicle screws. Orthopedics. 2013;36(7):e951–7.
- Yue JJ, Timm JP, Panjabi MM, Jaramillo de la Torre J. Clinical application of the Panjabi neutral zone hypothesis: the Stabilimax NZ posterior lumbar dynamic stabilization system. Neurosurg Focus. 2007;22:E12.
- 99. Kim YS, Zhang HY, Moon BJ, Park KW, Ji KY, Lee WC, Oh KS, Ryu GU, Kim DH. Nitinol spring rod dynamic stabilization system and Nitinol memory loops in surgical treatment for lumbar disc disorders: short-term follow up. Neurosurg Focus. 2007;22:E10.

## A Word from the Inventor of Intervertebral Dynamic Fixation: On Interspinous Devices

44

## Jacques Sénégas

## 44.1 Introduction

The idea of developing a system of dynamic lumbar fixation came to me in the early 1980s. At that time, the only thing everyone talked about was fusion and pedicle screw systems represented the ultimate technique. I began working in the cadaver laboratory testing various constructs in 1982 with R. Bréard, whose company SEM (Science et Médecine) primarily developed hip prostheses. His engineer, J. Frismand, suggested that I should begin with a system adapted to pedicle screws, which they were starting to manufacture. After several in vitro biomechanical tests, it became apparent to me that the constraints of a dynamic system would go beyond the long-term capacities of pedicle screw fixation, at least at that period of its development, and that the risks of screw loosening were too high. Consequently, I proposed the alternative of a "floating" interspinous device without intraos-

J. Sénégas, MD Clinique Saint Martin, Centre Aquitain du Dos, Pessac, France e-mail: Jacques.senegas@neuf.fr seous fixation. The first patient was operated in 1986, and the patent was obtained in 1987 by Bréard. The intent was to find an alternative to lumbar arthrodesis, which appeared to accelerate degenerative changes at adjacent levels. This initial device consisted of one metal interspinous spacer to which was attached a long polyester cord. The cord was passed around one of the spinous processes limiting that interspinous space, back through a tunnel in the spacer, around the other spinous process, and back through a second tunnel in the spacer where it was prevented from backing out by a Morse taper. This interspinous spacer was made of metal to avoid being split by the Morse taper. The metal spacer could be completed at up to four other interspinous levels with plastic spacers, each containing two tunnels for passage of the cord (the same attached to the primary spacer) in opposing directions. After passage around the spinous processes and through the various spacers, the extremity of the cord was blocked in the steel spacer by the Morse taper [1, 2]. The device was intended to restore, in degenerate intervertebral segments, the high-flexibility zone flexion-extension stiffness, which is diminished in symptomatic degenerative disk disease and worsened by posterior decompressive surgery [3, 4].

My spine team colleagues agreed with me to test the concept. Previously, in many patients with unstable degenerative lumbar disorders that

In this chapter, Professor Sénégas provides a unique perspective regarding a field of spine surgery that he pioneered. His account in conversation form, is unique from both a historical and clinical perspective.

called for decompressive surgery, we performed an arthrodesis when we believed their preoperative instability had contributed to low back pain resistant to conservative treatment for several months. These degenerative disorders requiring decompression included lumbar canal stenosis at one or, more often, more than one level and certain types of disk herniation (recurrent herniated disks, herniated disk of L4-L5 when there was sacralization of L5, and unusually massive herniated disks). These were the criteria of selection of indications for this first study. In patients fulfilling these conditions who gave enlightened consent, we would complete decompressive procedures with interspinous stabilization instead of arthrodesis. The patients were informed that, if their low back pain remained unresolved, they could have the arthrodesis that they initially chose not to undergo. We cautiously proceeded in this manner without commercialization of the device, waiting to judge the results over time.

The trial was very successful for these indications. Actuarial survivorship analysis of the patients operated between 1987 and 1995 demonstrated that only 17 % needed either replacement of the device by arthrodesis or any other lumbar surgical procedure within the first 10 years of the operation with a 95 % confidence interval of plus or minus 6 % [5]. We believed that interspinous dynamic stabilization does indeed delay adjacent segment syndrome given that, even when lumbar osteosynthesis is correctly performed and achieves fusion, additional surgery at adjacent levels has been reported in 36 % of patients within the first 10 years with a confidence interval of plus or minus 10 % [6]. Furthermore, our patients who were not reoperated had qualityof-life scores that were almost the same as those of the age- and sex-matched general population, while the patients in whom arthrodesis was later performed had poorer quality-of-life scores [7]. These differences with the general age-matched quality-of-life scores were similar to those reported elsewhere in a similar group of patients 10 years after osteosynthesis [8].

Positive clinical feedback from this cohort led to the development and widespread diffusion in 2002 (20 years after the first version) of the Wallis system, an improved version of the initial device [9] (Figs. 44.1 and 44.2). We later devel-





Fig. 44.2 Wallis implant





Fig. 44.3 UniWallis implant

oped a third generation of the Wallis system (UniWallis) in 2008 to permit unilateral insertion and simplify placement, tightening, and locking of the tension band (Fig. 44.3).

Our indications for our system have evolved over the years. The current principal indications and contraindications of the Wallis and UniWallis devices are summarized in Tables 44.1 and 44.2.

At the same time as our second-generation interspinous dynamic stabilization implant was being launched, the interspinous implant X-Stop was developed for an entirely different reason, to distract the interspinous space in order to relieve neurogenic claudication without using laminectomy or laminotomy in patients with lumbar canal stenosis [10, 11].

The interspinous dynamic stabilization devices we developed were followed by other interspinous devices that were also intended for dynamic stabilization, not distraction. Among the first of these were the Interspinous U (which later became the Coflex) [12] and the DIAM [13], but **Table 44.1** Interspinous dynamic stabilization is indicated for degenerative lesions with potential or reducible instability (less than 2 mm on dynamic films)

- 1. After disk surgery, especially massive disk herniation, transitional L4–L5 disk if sacralization of L5 is present, and recurrent herniated disk
- 2. Chronic low back pain with degenerated disk and/ or facet joint arthritis, or Modic I changes refractory to conservative treatment
- 3. After decompressive laminotomy for central stenosis
- 4. For dynamic foraminal stenosis with retrolisthesis on dynamic X-rays due to posterior disk collapse
- 5. Stabilization of one or two symptomatic apical levels in lumbar degenerative scoliosis in elderly patients (alternative to deformity corrections with extensive fusion constructs)
- 6. Instability adjacent to a prior lumbar fusion
- 7. "Topping off" for a degenerated adjacent segment above lumbar fusion or total disk replacement

**Table 44.2** Contraindications for interspinous dynamic stabilization include

- 1. Spondylolisthesis of any grade and nonreducible retrolisthesis
- 2. Spinal deformities of children and young adults
- 3. Psychological, social, and professional issues

many others have appeared over the years: LimiFlex [3], In-Space [14], Ligament Vertebral de Renfort [15, 16], BacJak and Viking [17], Dallos [18], InSWing [19, 20], and Locker [21, 22]. Likewise, there was a proliferation of devices intended for interspinous distraction rather than dynamic stabilization that were marketed after the appearance of X-Stop: Aperius [23, 24], an device Kinoshita unnamed from Giken Corporation, Japan [25], In-Space [26, 27], Superion [28, 29], SMID [30, 31], and ExtenSure [32]. This fundamental difference in indications (dynamic stabilization vs. distraction) is not always perceived by authors reporting on interspinous devices in the literature [32–39]. Moreover, devices like In-Space are used for dynamic stabilization by some authors [14], for distraction by some [26], and for both by others [40]. One study even compared the clinical efficacy of distraction alone for canal stenosis with Wallis, Coflex, and X-Stop, with no associated surgical decompression: as would be expected, VAS pain score improvements were poor for all

three devices used in this indication [41]. Although the distraction principle improves symptoms of lumbar canal stenosis more than conservative treatment [26, 42], distraction is well documented to be less effective than surgical decompression [23, 27, 37, 43–48], except in one meta-analysis, which was based upon indirect comparison of these techniques [49]. Furthermore, other disadvantages have been reported in patients treated for canal stenosis with interspinous distraction instead of surgical decompression [50–53]. Reviewers who believe that all interspinous devices are used for distraction have concluded that "interspinous technology" is unproven and unreliable or that the risks outweigh the benefits [54, 55]. Tamburrelli et al. put it well, concluding that "as generally occurs with any new technique, the early contagious enthusiasm-resulting in an excessive and sometimes incorrect use of the device-has resulted in a rising number of failures and in a critical consideration about the indications and the true advantages of the technique [17]." This situation is further complicated by reviewers who consider that the only posterior dynamic stabilization systems are pedicle screw-based devices, appearing to ignore entirely the existence of interspinous dynamic stabilization [56, 57].

Keeping in mind this confusion of interspinous dynamic stabilization with interspinous distraction, a review of available evidence on interspinous implants intended solely for dynamic stabilization indications is very enlightening, albeit still a complex undertaking. First, the biomechanical experimentation varies with the different conceptions of how dynamic stabilization should function. As for clinical results, there are also a variety of hypotheses and explanations for the therapeutic action of the devices and generally good patient outcomes. Lastly, among the available interspinous dynamic stabilization devices, there are fundamental differences that further complicate the picture, primarily whether or not there is a flexionlimiting tension band system coupled with an extension-limiting spacer, which move loads away from painful areas of the lumbar motion segment in flexion and in extension, respectively

[58]. Over the past few years, there has been an explosion of evidence in the literature on interspinous dynamic stabilization, particularly from Asia. This chapter represents, to the best of my knowledge, an exhaustive review of all peerreviewed articles pertaining to interspinous dynamic stabilization that have appeared over the last 5 years.

## 44.2 Results of In Vitro Biomechanical Investigations

- Stabilization in terms of reduced range of motion and increased stiffness,
- Unloading of the disk

The available sizes of interspinous spacers correspond to measurements of the interspinous spaces published in two studies [59, 60]. These studies suggest that the spacers should be placed anteriorly in the interspinous space, where the cortex is thicker and the interspinous process distance is greatest [60]. An in vivo radiographic study found smaller interspinous spaces, suggesting that bone trimming is typically necessary to avoid creating undesired kyphotic changes in segments treated by interspinous dynamic stabilization [61], confirming our own experience.

Maintaining segmental lordosis is capital and merits a short digression regarding several important operative and postoperative aspects: first, in order to select a spacer size that will not induce kyphosis, the patient must be operated in the prone position, not in the genu pectoral position. To insert the Wallis or the UniWallis when the patient is in the prone position, one may use a distractor. If the device fits tightly when the patient is in the genu pectoral position, the bands will loosen whenever the patient is upright. This will certainly not contribute to restoring segmental stiffness! The spacer should be pushed in against friction, but not be large enough to induce kyphosis. It is also important to use a spacer that is not too small. If it is too small, there is a risk of excessive movement that might scratch the spinous processes. Another operative point concerns the supraspinous ligament, the distal end of which is L4 in the majority of subjects. I always thought it important to disinsert the fascia of the paraspinal muscles from the spinous processes after the initial incision, then, at the end of the Wallis implant procedure, to reattach them with a suture through a small hole at the tip of the spinous process. It is also important to insist that the patient must wear a lumbar corset for 1 month following the operation to optimize development of scar tissue around the implant. Finally, just as one should do in selecting patients for fusion, surgeons should take into account psychological, social, or workers compensation-related issues when deciding which patients would best benefit from dynamic stabilization (end of digression).

In interspinous dynamic stabilization implants that do not have flexion-limiting bands, there is conflicting biomechanical evidence. In one study with axial preloading applied to cadaver specimens of the lumbar spine, Coflex, In-Space, and Aperius reduced range of motion (ROM) of the implanted segment in both extension and flexion even though the supraspinous ligament was removed for placement of the Coflex [62]. In a similar study with preloading, In-Space reduced ROM in extension, but not in flexion, lateral bending, or axial rotation; reduced disk pressure of the operated segment, with little effects on ROM or disk pressure in the adjacent segments [63]; and unloaded the facet joints when loaded in extension [64]. In contrast, cadaver studies of spacers alone (without tension bands) without axial preloading show significant limitation of ROM in extension, but not in flexion, as well as no influence on axial rotation or lateral bending [65–67]. This would apply less if the stabilization devices could be implanted without injury to the thoracolumbar fascia, fascia of the longissimus thoracis muscles, or fascia of the multifidus muscles, although avoiding these injuries is difficult when decompressing the lumbar canal [68]. Regarding the segments adjacent to implantation by a spacer alone, Hartmann et al. reported that flexion-extension ROM above and below was increased in most of their cadaver specimens with and without a follower load of 400 N [69]. Mao et al. reported increased ROM only in extension at L2-L3 with a Coflex at L3-L4, which

limited flexion-extension ROM there with little effect on axial rotation or lateral bending, adjacent to L4–L5 with rigid pedicle screw fixation [70]. An in vivo radiographic study has reported that limitation of flexion by the Coflex device is possible in case of bone overgrowth into the device, but generally not expected [71]. Coflex can be used to strongly limit flexion, but the solution calls for rivets through the spinous processes, with high risk of spinous process fracture especially in L5 and in patients with poor bone quality [72].

In an ovine model, Gunzburg et al. showed that an interspinous spacer reduced total segmental ROM in flexion-extension by only 17 % and that the combination of a spacer plus a tension band around the spinous processes reduced the ROM by 46 % [19]. Contrary to some interspinous dynamic stabilization devices, all three generations of our interspinous stabilization devices have a strong cord or band that limits flexion. A recent cadaver study with application of compressive preloads mimicking the stabilizing action of axial musculature demonstrated that anatomical alterations corresponding to degenerative and iatrogenic lesions result in decreased segmental stiffness. This loss of stiffness was less amenable to compensation by axial muscular activity in flexion than in extension, suggesting to the authors the potential usefulness of surgical implants that specifically increase flexion stiffness and limit flexion ROM to counteract the iatrogenic instability resulting from surgical decompression [4]. Without preloading, other cadaver studies have shown that the Wallis system restored ROM to that of the intact specimen in both flexion and extension, while Coflex and DIAM restored ROM only in extension, but not in flexion; all three systems reduced lateral bending by 10 % and axial rotation by 20 % [66, 67]. In a randomized trial of patients operated at L4-L5 for herniated disks or canal stenosis by either PLIF or Wallis dynamic stabilization, Li et al. measured the stiffness of degenerated L4-L5 segment. L4-L5 stiffness before decompression was 37 Nm; after decompression, it fell to 26 Nm, and after Wallis stabilization, it was restored to 46 Nm. The stiffness of the intact overlying

J. Sénégas

adjacent segment (L3-L4) was significantly higher above the Wallis than above the pedicle screw-augmented PLIF (46 Nm vs. 35 Nm; p < 0.05), confirming their own experimental cadaver studies [73, 74]. Other cadaver studies of the second-generation Wallis have shown a 14 % reduction in flexion-extension ROM of the stabilized L3-L4 segment, with small increases in ROM at the uninstrumented L2-L3 and L4-L5 segments of 7 % and 3.5 %, respectively, and little influence on lateral bending and axial rotation [75]. Similar biomechanical results in cadavers have been found with another device that has a spacer and a tension band, which device also was shown to reduce pressures in the posterior annulus and central nucleus [22]. Comparing Wallis to a pedicle screw-based dynamic stabilization system, Schulte et al. reported that the Wallis implant reduced extension by 69 % and flexion by 62 %, with almost no action on lateral bending or axial rotation [76].

A study of distance between spinous processes showed that the variations in interspinous process distance (ISPD) are greater in patients with degenerative disk disease than in healthy subjects, demonstrating the risk of implant dislocation for interspinous devices that do not limit ISPD during flexion [77]. Indeed, among dynamic stabilization devices that do not strongly limit flexion, dislocation has been reported for Coflex in several studies [12, 78-80]. In this respect, the DIAM device appears to be an exception. Even though cadaver studies show that it fails to limit flexion [66, 67], the tethering laces probably explain the lack of reported postoperative dislocations [81]. Next to the Wallis system, other interspinous stabilization devices with strong flexion-limiting bands have been developed to restore segmental stiffness more consistently, including the Ligament Vertebral de Renfort [16], Dallos [18], InSWing [20], and Locker [21]. In a porcine study, the supraspinous ligament (SSL) and the laces of DIAM were both observed to have a mechanical role, leading the authors to recommend preservation of the SSL and use of the laces, which also were thought to prevent postoperative dislocation [82]. The LimiFlex device has flexion-limiting bands, but not an interspinous spacer [3].

## 44.3 Review of Clinical and Radiological Findings

In contrast to the poor efficacy associated with interspinous distraction (see above), over the last 5 years, 29 clinical studies of interspinous devices used for dynamic stabilization have reported improved clinical status and persistence of the improvement regardless of the device: Wallis [83–92], Coflex [78, 80, 86, 89, 90, 93–101], DIAM [102–107], In-Space [14, 108], Ligament Vertebral de Renfort [15, 109], or Dallos [18].

Using interspinous dynamic stabilization devices to decompress nerve roots and to off-load disks and facet joints, many authors report radiological data that shows increased foraminal dimensions, disk height, or both [14, 91, 92, 99, 101, 110, 111], with the exception of DIAM, which has a spacer made of silicone [103]. Even though these mechanisms of action undoubtedly contribute to the efficacy of dynamic stabilization devices [112], we developed the Wallis line of implants primarily to relieve chronic low back pain associated with loss of intersegmental stiffness. Other authors agree that the clinical action of interspinous dynamic stabilization devices is theoretically due to unloading of the facet joints, restoration of foraminal height, and/or increased intervertebral stability [108, 113–116]. As stated above, experimental in vivo proof in patients demonstrates that the Wallis device does indeed restore physiological stiffness of the treated segment without adversely affecting the stiffness of the adjacent segments [73]. However, because direct measurement of stiffness in patients is impractical, to measure stabilization authors report instead radiological flexion-extension ROM restrictions achieved and maintained by dynamic stabilization devices compared to the preoperative ROM. The overall flexion-extension ROM, which is increased by intervertebral degenerative disease and further increased by decompressive procedures, is consistently improved by the placement of an interspinous spacer, even if the spacer has no flexion-limiting attachments. When the implant has nothing that limits flexion, postoperative adherences between the spinous processes and the medial fascia of the paraspinal muscles may limit flexion of the

implanted segment after several weeks. In any case, because spacers do reduce extension, this automatically reduces overall flexion-extension ROM. Most authors who have compared preoperative to follow-up flexion-extension ROM in their patients have reported improved (reduced) ROM of the treated segments at follow-up [18, 84, 95, 97, 99, 104, 107, 113]. Others have reported almost no change between preoperative ROM and ROM at final follow-up [95, 96, 110]. Sun et al. reported more restriction of flexionextension ROM achieved with Wallis (10°) with its flexion-limiting band than with Coflex  $(13^{\circ})$ (p=0.019) [90], and Chao et al. showed that the ROM in extension decreased, but that the ROM in flexion increased in lumbar segments implanted with Coflex [95].

The ROM of the intervertebral segments adjacent to the treated segments is equally important. Above a fused lumbar segment, ROM of the adjacent segment increases, which is thought to accelerate adjacent segment disease [113]. We developed dynamic stabilization devices that would stiffen the treated segment without completely eliminating flexion and extension there in order to preserve physiological functioning in the adjacent segments. Ideally, the flexion-extension ROM in adjacent, healthy segments should not be affected by placement of an interspinous dynamic stabilizer. All authors who have measured adjacent segment ROM report no undesired increase in that ROM during follow-up of the interspinous dynamic stabilization devices that they use [84, 95, 99, 111, 113]. In a study of 60 patients who underwent decompression of L4/L5 for degenerative canal stenosis, Liu et al. reported that the follow-up ROM of L3/L4 was increased and the disk height of L3/L4 was decreased significantly more in the 31 patients who had 360° fusion of L4/L5 compared to the 29 patients stabilized at L4/L5 by an interspinous dynamic stabilization device (p < 0.05), leading those authors to conclude that dynamic stabilization would delay degeneration of L3/L4 [97].

In one study, Kaplan-Meier analysis of survival from failure showed that decompression by laminotomy and flavectomy stabilized by fusion was 76 % at 5 years, with all failures caused by

additional surgery for adjacent level syndrome. In that study, among the patients who had the same operation without fusion, 5-year survival from failure was 92 %, with both failures at the index level [43]. In patients who had lumbar decompressive surgery, Hong et al. compared 18 patients who had no stabilization to 23 who had dynamic interspinous stabilization, resorting to revision by fusion for symptomatic instability in 1 of the 23 patients (4 %) in the stabilized group and in 5 of the 18 patients (28 %) in the unstabilized group [109]. In a matched retrospective comparative study, Liu et al. reported that, compared with isolated PLIF of L5-S1, PLIF at L5-S1 combined with Wallis or Coflex dynamic stabilization at L4-L5 restricts the ROM of L4-L5 in extension and prevents excessive olisthesis of L4 in both extension and flexion. Based upon these findings and convincing MRI evidence of differences in L4-L5 disk degeneration and Modic changes, they concluded that follow-up of their patients longer than 24 months would potentially show that interspinous dynamic stabilization reduces degenerative changes adjacent to fusion [113, 117]. Even the DIAM device, which limits flexion and extension less than the Wallis system [66, 67], has been reported to slow the development of radiological adjacent segment degeneration above a PLIF (p=0.03), although no significant difference in additional surgery at the segment rostral to PLIF was observed in that cohort [81]. In a randomized controlled study of Wallis dynamic stabilization above lumbar osteosynthesis procedures, Korovessis et al. have provided the best evidence that these devices can delay symptomatic adjacent segment disease: the ROM in flexion and extension of the adjacent segments protected by a Wallis implant remained stable after the operation, while there was progressive significant increase (p < 0.02) in the adjacent segment ROM of the control patients who had no protection above the fusion; this was associated with better ODI scores in the Wallis group (p < 0.05) and more adjacent segment revision operations in the control group (14 % vs. none) [85].

In a study of patients who underwent revision surgery for degenerative disease of the segment adjacent to prior fusion, Cho et al. recommended treatment by decompression and an interspinous dynamic stabilization device instead of extending fusion, because clinical results were equally good and dynamic stabilization preserves posterior complex integrity [118].

As we have recommended for our system, authors using other systems also preserve the supraspinous ligament [14, 78, 79], which contributes to segmental stability [119, 120], sends proprioceptive information to the paraspinal musculature [121, 122], and prevents increased ROM in flexion and extension in the adjacent segments [123].

Some authors have reported less favorable results for interspinous dynamic stabilization devices or results not superior to control groups. In each of these reports, the less favorable results can be attributed to either use of the devices in controversial indications or insufficient length of follow-up. Because dynamic stabilization devices are intended to relieve instability-related pain, many months may be necessary before differences in low back pain appear between decompressed patients with and without stabilization. In a study with 24 months of follow-up, no difference was found in clinical outcome, which was good, between patients with or without Coflex stabilization after decompression of canal stenosis (with spondylolisthesis in half of the patients) [115]. In an as yet unpublished randomized controlled trial presented by Mahir and Marsh at the British Orthopaedic Association 2012 Annual Congress in 2012, both groups (30 patients treated by decompression alone compared to 30 patients treated by decompression and Wallis dynamic stabilization), postoperative clinical results were good, practically identical and stable in the two groups at 1 year and 2 years, but the unstabilized group worsened after 3 years while the same good results persisted in the group stabilized by Wallis [124]. Clinical results regarding symptomatic adjacent segment disease are also time dependent, more than 2 years of follow-up being necessary to demonstrate superiority of interspinous dynamic stabilization over arthrodesis in terms of revision surgery for adjacentlevel syndrome [5, 6].

In two examples of less favorable results involving controversial indications, Mayer et al. reported revision surgery within 34 months in 8 of 32 patients in whom they used In-Space for arthrogenic low back pain [40], and, using the Coflex device for distraction in 20 patients who also had isolated facet joint pain, Cabraja et al. reported reduction of 50 % in VAS pain score in only 7 patients (35 %) after 2 years [125]. This suggests that unloading the facet joints with an interspinous spacer for isolated facet joint pain may be a poor indication. These poor results might be attributable, however, to the use of too much distraction of the treated segments (L4-L5), with a radiographically demonstrated loss of lordosis there, (p < 0.001) and increased lordosis at L3–L4 (p < 0.032). These changes induced in the sagittal profile may have contributed to further facet joint degeneration possibly explaining why the clinical outcome of these patients was better at 1-year follow-up than at final follow-up [125].

In patients with grade I degenerative spondylolisthesis, which, in my opinion, is certainly a contraindication, interspinous dynamic stabilization devices have failed to prevent further slippage [16, 21, 126] and, in one study, good clinical results were achieved in only two thirds of the patients [16]. A 6-year study of 23 patients stabilized with an interspinous dynamic device for grade I degenerative spondylolisthesis compared to 22 patients treated for the same indication with pedicle screw-augmented PLIF provides even more convincing confirmation that interspinous dynamic stabilization should not be used for spondylolisthesis [127].

## 44.4 Complications of Interspinous Dynamic Stabilization Devices

The reported complication rates are generally lower in interspinous dynamic stabilization studies than in reports on patients with degenerative disease treated by fusion. In 131 patients treated with Coflex, Xu et al. reported only three implantrelated complications (loosening, wing breakage, and spinous process fracture), along with five other complications requiring additional surgery (recurrent disk herniation at the treated level in two, a residual herniated disk in one, spinal canal hematoma in a patient taking anticoagulants, and incomplete decompression in one) [78]. Zang et al. reported a total of only 13 complications among 133 patients [79]. Nachakian et al. reported one revision procedure (for recurrence of neurologic symptoms) among 134 patients [98]. In a single-unit study of complications in 168 patients who had either Wallis or Coflex dynamic stabilization, the overall complication rate was 10.7 % (18/168), 6.2 % (8/130) in the Wallis group and 26.3 % (10/38) in the Coflex group (p < 0.01) [128]. Xu et al. reported that none of their 96 patients had complications related to Wallis dynamic stabilization [91]. In 48 Wallis patients, Liu et al. observed no intraoperative complications [87]. Other studies have also recorded no implant-related complications with Wallis (n=0/20) [84], (n=0/15) [88], (n=0/25)[85], Coflex (n=0/20) [94], (n=0/29) [97], (n=0/21) [96], DIAM (n=0/8) [107],(n=0/16)[106] (*n*=0/68) [105], and Locker (*n*=0/23) [21].

Among complications of interspinous stabilization devices, unresolved low back pain is not a serious issue, because these systems spare vertebral anatomy; they do not preclude or significantly complicate later treatment with a more definitive procedure (i.e., fusion). Some reports of interspinous dynamic stabilization include a few cases of straightforward removal and replacement by fusion [78, 79, 81, 98, 100]. The complication that naturally occurs at the index level more often after any kind of dynamic stabilization than after fusion is disk herniation, because dynamic stabilization preserves disk function, posterolateral fusion reduces disk function, and lumbar interbody fusion eliminates the disk. As shown by Floman et al. the frequency of recurrent disk depends upon the diskectomy procedure more than upon the dynamic stabilization technique [83]. After a Wallis procedure, Liu et al. reported a recurrent disk in 6 of 48 patients, 3 of whom were treated conservatively and 3 simply by removal and fusion [87]. In a study comparing Wallis in 25 patients to Coflex in 27 patients, Sun

et al. reported 4 recurrent disks in the Coflex patients and none with Wallis [90]. In another series of 68 patients treated by interspinous dynamic stabilization, Li et al. reported 2 cases of recurrent disk [129]. Hrabálek et al. reported no recurrent disks among 68 patients [105].

The rate of intraoperative and postoperative spinous process fractures complicating interspinous dynamic stabilization devices is quite low in my experience with the technique, but I always used small spacer sizes to avoid distraction and preserve segmental lordosis. This contrasts with the high incidence of spinous process fractures when interspinous spacers are used for distraction to treat canal stenosis without undercutting [24, 44, 55], because larger spacers are used to obtain segmental kyphosis instead of preserving segmental lordosis as we recommend in dynamic stabilization to avoid facet joint pain. A report by Fabrizi et al. illustrates the indication-dependent aspect of this complication of interspinous devices. Among 1315 patients in whom they used an interspinous device for dynamic stabilization after decompression, they observed 7 spinous process fractures (0.5 %), whereas among 260 patients in whom the same surgeons used an interspinous device for distraction in elderly patients to avoid surgical decompression and general anesthesia, they reported 3 spinous process fractures (1.2 %) [102]. The interspinous spacers and soft polyester bands of our system may be less aggressive to the spinous process than metallic interspinous dynamic stabilization systems. Spinous process fractures have been reported with the use of Coflex [78, 79, 100, 115]. In a series of 133 patients who had Coflex dynamic stabilization, 3 had an intraoperative spinous process fracture and 2 had postoperative spinous process fracture [80]. However, in a study by Sun et al. of interspinous dynamic stabilization complications in 168 patients with either Wallis or Coflex, no spinous process fracture was reported [128]. Using a soft interspinous spacer in 65 patients, Lee et al. reported that none of the patients had a spinous process fracture [16]. In vivo radiographic analysis in 176 patients shows that the average loads exerted by an interspinous dynamic stabilization spacer on the spinous process and lamina are estimated to be only 11 % and 7 % of their respective static failure load, which would help explain the observed low rates of postoperative fractures [71].

There have been reports of heterotopic ossification leading to fusion between the spinous processes in patients with the Coflex system [130]. After a fortuitous discovery of interspinous fusion around a Coflex device in one patient [131], Tian et al. found this complication in 81 % of their Coflex patients [132]. To the best of my knowledge, this complication has not been reported in other interspinous devices used for dynamic stabilization. Compared to other interspinous stabilization devices, one difference Coflex has is the presence of retaining teeth on the metal wings that are squeezed against the spinous processes. This may result in freshening of the lateral aspects of the spinous processes with bleeding that might mediate the observed heterotopic ossification. Tian et al. hypothesized that the resulting consistent stress at the interface between the bone and implant might stimulate growth of heterotopic bone [132].

There have been reports of osteolysis with interspinous devices [133]. Park et al. reported radiolucent gaps between the Coflex implant and the spinous processes in 17 of 30 patients (57 %), the gaps being most prominent around the spikes on the spacer [126]. In our experience with the firstgeneration interspinous stabilization device, some osteolysis was observed at long-term, systematic follow-up in some patients. Further investigation with CT scans revealed remodeling of the spinous process in contact with spacers or polyester cords. The cortical bone was preserved, but took on the shape of the cord or the spacers, suggesting a slow process of bone remodeling in response to surface stress. Because the distance between interspinous processes diminishes with age [134], the observed remodeling might be due to spinous processes growing around the spacer and cords rather than spacers and cords cutting into the spinous processes. In any event, our discovery of this remodeling process was rare and fortuitous in asymptomatic patients. When the process occurs, it appears to be self-limiting, because no patients have presented with spinous process fractures associated with these macroscopic features.

In the largest series of interspinous dynamic stabilization (1315 patients in whom the 1832 devices were made with polyester), deep infection was observed in only 10 patients (0.8 % of patients; 0.5 % of devices) [102]. Another group using the same implant in 22 patients reported deep infection in 1 (4.5 %) [104]. Using another polyester interspinous stabilization device, Hong et al. reported 1 infection among 23 patients [109]. In 68 patients treated with the latter polyester device, Hrabálek et al. reported no wound seroma or deep infection [105].

Among rare complications of interspinous dynamic stabilization devices are blockage of drainage of an epidural hematoma by the implant with resulting cauda equina syndrome [135], an inflammatory reaction to a polyester device [16], a foreign-body reaction to a polyester and silicone device [17, 136], irritation and fluid around a spacer made of PEEK and titanium [40], and bilateral stress fracture of the inferior spinous process [137].

### Conclusions

In vivo proof has been published showing that an interspinous spacer *combined with tension bands around the spinous processes* actually does restore physiological stiffness of the treated degenerative segment, even after decompressive procedures that further reduce segmental stiffness more than the degenerative process itself. In segments adjacent to interspinous dynamic stabilization, stiffness is not lost and ROM is not increased, contrary to what is observed in segments adjacent to fused lumbar levels.

The rate of implant-related complications is much lower with interspinous dynamic stabilization than with intervertebral fusion procedures. Fewer implant-related complications are reported regarding implants that have an interspinous spacer combined with tension bands than in those that have an interspinous spacer without tension bands. Interspinous dynamic stabilization preserves patient anatomy, facilitating more invasive surgical alternatives if chronic low back pain is not resolved.

In patients with chronic low back pain related to degenerative instability who undergo diskectomy or decompression for canal stenosis, there is very strong published evidence that outcome is better in patients who have interspinous dynamic stabilization than in those who are not stabilized. There is strong evidence that, in these two indications, patients who have interspinous dynamic stabilization have lower rates of revision surgery in general and lower rates of revision for adjacent level disease in particular than patients stabilized by arthrodesis. In patients who undergo lumbar or lumbosacral fusion for degenerative conditions such as spondylolisthesis, there is also strong evidence showing that fewer revision surgeries are necessary for symptomatic adjacent-level disease if an overlying degenerate adjacent segment is stabilized with an interspinous implant than if no stabilization is used. Among patients who have symptomatic adjacent segment disease, those treated by interspinous dynamic stabilization have better outcome than those treated by extension of fusion.

Provided that interspinous dynamic stabilization devices are not used for distraction, i.e., provided that lordosis of the treated segments is preserved, that these devices are not used for spondylolisthesis, and that they are not used to prevent recurrence of disk herniation, no one has observed poor results with this technique. On the contrary, all 29 of the clinical studies regarding interspinous dynamic stabilization for appropriate indications published over the last 5 years have demonstrated good clinical results.

### References

- Senegas J. Surgery of the intervertebral ligaments, alternative to arthrodesis in the treatment of degenerative instabilities. Acta Orthop Belg. 1991;57 Suppl 1:221–6.
- Senegas J, Etchevers JP, Vital JM, Baulny D, Grenier F. Recalibration of the lumbar canal, an alternative to laminectomy in the treatment of lumbar canal stenosis. Rev Chir Orthop Reparatrice Appar Mot. 1988;74:15–22.
- Fielding LC, Alamin TF, Voronov LI, Carandang G, Havey RM, Patwardhan AG. Parametric and cadaveric models of lumbar flexion instability and flexion restricting dynamic stabilization system. Eur Spine J. 2013;22:2710–8.
- Fry RW, Alamin TF, Voronov LI, Fielding LC, Ghanayem AJ, Parikh A, Carandang G, McIntosh BW, Havey RM, Patwardhan AG. Compressive preload reduces segmental flexion instability after pro-

gressive destabilization of the lumbar spine. Spine (Phila Pa 1976). 2014;39:E74–81.

- Senegas J, Vital JM, Pointillart V, Mangione P. Longterm actuarial survivorship analysis of an interspinous stabilization system. Eur Spine J. 2007;16: 1279–87.
- Ghiselli G, Wang JC, Bhatia NN, Hsu WK, Dawson EG. Adjacent segment degeneration in the lumbar spine. J Bone Joint Surg Am. 2004;86-A:1497–503.
- Senegas J, Vital JM, Pointillart V, Mangione P. Clinical evaluation of a lumbar interspinous dynamic stabilization device (the Wallis system) with a 13-year mean follow-up. Neurosurg Rev. 2009;32:335–41; discussion 341–2.
- Glaser J, Stanley M, Sayre H, Woody J, Found E, Spratt K. A 10-year follow-up evaluation of lumbar spine fusion with pedicle screw fixation. Spine (Phila Pa 1976). 2003;28:1390–5.
- Senegas J. Mechanical supplementation by non-rigid fixation in degenerative intervertebral lumbar segments: the Wallis system. Eur Spine J. 2002;11 Suppl 2:S164–9.
- Lindsey DP, Swanson KE, Fuchs P, Hsu KY, Zucherman JF, Yerby SA. The effects of an interspinous implant on the kinematics of the instrumented and adjacent levels in the lumbar spine. Spine (Phila Pa 1976). 2003;28:2192–7.
- Swanson KE, Lindsey DP, Hsu KY, Zucherman JF, Yerby SA. The effects of an interspinous implant on intervertebral disc pressures. Spine (Phila Pa 1976). 2003;28:26–32.
- Kaech DL, Fernandez C, Haninec P. Preliminary experience with the interspinous "U" [Article in French]. Rachis. 2001;13:403–4.
- Caserta S, La Maida GA, Misaggi B, Peroni D, Pietrabissa R, Raimondi MT, Redaelli A. Elastic stabilization alone or combined with rigid fusion in spinal surgery: a biomechanical study and clinical experience based on 82 cases. Eur Spine J. 2002;11 Suppl 2:S192–7.
- Zhou D, Nong LM, Du R, Gao GM, Jiang YQ, Xu NW. Effects of interspinous spacers on lumbar degenerative disease. Exp Ther Med. 2013;5:952–6.
- Lee SH, Enes M, Hoogland T. Soft stabilization with interspinous artificial ligament for mildly unstable lumbar spinal stenosis: a multicenter comparison. Arch Orthop Trauma Surg. 2010;130:1335–41.
- 16. Lee SH, Lee JH, Hong SW, Shim CS, Chung SE, Yoo SH, Lee HY. Factors affecting clinical outcomes in treating patients with grade 1 degenerative spondylolisthesis using interspinous soft stabilization with a tension band system: a minimum 5-year follow-up. Spine (Phila Pa 1976). 2012;37:563–72.
- Tamburrelli FC, Proietti L, Logroscino CA. Critical analysis of lumbar interspinous devices failures: a retrospective study. Eur Spine J. 2011;20 Suppl 1: S27–35.
- Rosales-Olivares LM, Alpizar-Aguirre A, Miramontes-Martinez V, Zarate-Kalfopulus B, Reyes-Sanchez A. Dynamic interspinous stabilization in lumbar discectomy: 4-year follow-up. Cir Cir. 2010;78:492–6.

- Gunzburg R, Szpalski M, Callary SA, Colloca CJ, Kosmopoulos V, Harrison D, Moore RJ. Effect of a novel interspinous implant on lumbar spinal range of motion. Eur Spine J. 2009;18:696–703.
- Pfeiffer M. Interspinous implant "InSWing(R)" for the lumbar spine. Oper Orthop Traumatol. 2010;22:512–23.
- Lee DY, Lee SH, Shim CS, Lee HY. Decompression and interspinous dynamic stabilization using the locker for lumbar canal stenosis associated with lowgrade degenerative spondylolisthesis. Minim Invasive Neurosurg. 2010;53:117–21.
- Shim CS, Park SW, Lee SH, Lim TJ, Chun K, Kim DH. Biomechanical evaluation of an interspinous stabilizing device, Locker. Spine (Phila Pa 1976). 2008;33:E820–7.
- Beyer F, Yagdiran A, Neu P, Kaulhausen T, Eysel P, Sobottke R. Percutaneous interspinous spacer versus open decompression: a 2-year follow-up of clinical outcome and quality of life. Eur Spine J. 2013;22:2015–21.
- Bonaldi G, Bertolini G, Marrocu A, Cianfoni A. Posterior vertebral arch cement augmentation (spinoplasty) to prevent fracture of spinous processes after interspinous spacer implant. AJNR Am J Neuroradiol. 2012;33:522–8.
- 25. Nishida K, Doita M, Kakutani K, Maeno K, Yurube T, Kurosaka M. Development of percutaneously insertable/removable interspinous process spacer for treatment of posture-dependent lumbar spinal-canal stenosis: preclinical feasibility study using porcine model. Eur Spine J. 2012;21:1178–85.
- 26. Hrabalek L, Wanek T, Machac J, Vaverka M, Langova K, Kalita O, Krahulik D, Novak V, Houdek M. Percutaneous interspinous dynamic stabilization (in-space) in patients with degenerative disease of the lumbosacral spine – a prospective study. Rozhl Chir. 2012;91:311–6.
- 27. Kantelhardt SR, Torok E, Gempt J, Stoffel M, Ringel F, Stuer C, Meyer B. Safety and efficacy of a new percutaneously implantable interspinous process device. Acta Neurochir (Wien). 2010;152:1961–7.
- Goyal A, Goel VK, Mehta A, Dick D, Chinthakunta SR, Ferrara L. Cyclic loads do not compromise functionality of the interspinous spacer or cause damage to the spinal segment: an in vitro analysis. J Long Term Eff Med Implants. 2008;18:289–302.
- 29. Loguidice V, Bini W, Shabat S, Miller LE, Block JE. Rationale, design and clinical performance of the Superion(R) Interspinous Spacer: a minimally invasive implant for treatment of lumbar spinal stenosis. Expert Rev Med Devices. 2011;8:419–26.
- 30. Yao QQ, Zheng SN, Cheng L, Yuan P, Zhang DS, Liao XW, Xu Y, Wang LM. Effects of a new shapememory alloy interspinous process device on pressure distribution of the intervertebral disc and zygapophyseal joints in vitro. Orthop Surg. 2010;2:38–45.
- 31. Zheng S, Yao Q, Cheng L, Xu Y, Yuan P, Zhang D, Liao X, Wang L. The effects of a new shape-memory alloy interspinous process device on the distribution

of intervertebral disc pressures in vitro. J Biomed Res. 2010;24:115–23.

- Bono CM, Vaccaro AR. Interspinous process devices in the lumbar spine. J Spinal Disord Tech. 2007;20:255–61.
- 33. Chen H, Ding W. Research advancement of lumbar inter-spinous process non-fusion techniques. Zhongguo Xiu Fu Chong Jian Wai Ke Za Zhi. 2010;24:368–73.
- 34. Crawford RJ, Price RI, Singer KP. The effect of interspinous implant surgery on back surface shape and radiographic lumbar curvature. Clin Biomech (Bristol, Avon). 2009;24:467–72.
- Gomleksiz C, Sasani M, Oktenoglu T, Ozer AF. A short history of posterior dynamic stabilization. Adv Orthop. 2012;2012:629–98.
- Kaner T, Sasani M, Oktenoglu T, Ozer AF. Dynamic stabilization of the spine: a new classification system. Turk Neurosurg. 2010;20:205–15.
- 37. Moojen WA, Arts MP, Jacobs WC, van Zwet EW, van den Akker-van Marle ME, Koes BW, Vleggeert-Lankamp CL, Peul WC. Interspinous process device versus standard conventional surgical decompression for lumbar spinal stenosis: randomized controlled trial. BMJ. 2013;347:f6415.
- Murtagh RD, Quencer RM, Castellvi AE, Yue JJ. New techniques in lumbar spinal instrumentation: what the radiologist needs to know. Radiology. 2011;260: 317–30.
- Sangiorgio SN, Sheikh H, Borkowski SL, Khoo L, Warren CR, Ebramzadeh E. Comparison of three posterior dynamic stabilization devices. Spine (Phila Pa 1976). 2011;36:E1251–8.
- Mayer HM, Zentz F, Siepe C, Korge A. Percutaneous interspinous distraction for the treatment of dynamic lumbar spinal stenosis and low back pain. Oper Orthop Traumatol. 2010;22:495–511.
- 41. Sobottke R, Schluter-Brust K, Kaulhausen T, Rollinghoff M, Joswig B, Stutzer H, Eysel P, Simons P, Kuchta J. Interspinous implants (X Stop, Wallis, Diam) for the treatment of LSS: is there a correlation between radiological parameters and clinical outcome? Eur Spine J. 2009;18:1494–503.
- 42. Zucherman JF, Hsu KY, Hartjen CA, Mehalic TF, Implicito DA, Martin MJ, Johnson 2nd DR, Skidmore GA, Vessa PP, Dwyer JW, et al. A prospective randomized multi-center study for the treatment of lumbar spinal stenosis with the X STOP interspinous implant: 1-year results. Eur Spine J. 2004;13: 22–31.
- 43. Brodke DS, Annis P, Lawrence BD, Woodbury AM, Daubs MD. Reoperation and revision rates of 3 surgical treatment methods for lumbar stenosis associated with degenerative scoliosis and spondylolisthesis. Spine (Phila Pa 1976). 2013;38:2287–94.
- 44. Kim DH, Shanti N, Tantorski ME, Shaw JD, Li L, Martha JF, Thomas AJ, Parazin SJ, Rencus TC, Kwon B. Association between degenerative spondylolisthesis and spinous process fracture after interspinous process spacer surgery. Spine J. 2012;12:466–72.

- Krauss WE. Interspinous distraction devices: too good to be true? Yes. World Neurosurg. 2013;80: 78–9.
- 46. Schizas C, Pralong E, Tzioupis C, Kulik G. Interspinous distraction in lumbar spinal stenosis: a neurophysiological perspective. Spine (Phila Pa 1976). 2013;38:2113–7.
- 47. Sobottke R, Rollinghoff M, Siewe J, Schlegel U, Yagdiran A, Spangenberg M, Lesch R, Eysel P, Koy T. Clinical outcomes and quality of life 1 year after open microsurgical decompression or implantation of an interspinous stand-alone spacer. Minim Invasive Neurosurg. 2010;53:179–83.
- 48. Verhoof OJ, Bron JL, Wapstra FH, van Royen BJ. High failure rate of the interspinous distraction device (X-Stop) for the treatment of lumbar spinal stenosis caused by degenerative spondylolisthesis. Eur Spine J. 2008;17:188–92.
- 49. Chou D, Lau D, Hermsmeyer J, Norvell D. Efficacy of interspinous device versus surgical decompression in the treatment of lumbar spinal stenosis: a modified network analysis. Evid Based Spine Care J. 2011; 2:45–56.
- 50. Barbagallo GM, Corbino LA, Olindo G, Foti P, Albanese V, Signorelli F. The "sandwich phenomenon": a rare complication in adjacent, double-level X-stop surgery: report of three cases and review of the literature. Spine (Phila Pa 1976). 2010;35:E96–100.
- 51. Barbagallo GM, Olindo G, Corbino L, Albanese V. Analysis of complications in patients treated with the X-Stop Interspinous Process Decompression System: proposal for a novel anatomic scoring system for patient selection and review of the literature. Neurosurgery. 2009;65:111–9; discussion 119–20.
- 52. Idler C, Zucherman JF, Yerby S, Hsu KY, Hannibal M, Kondrashov D. A novel technique of intra-spinous process injection of PMMA to augment the strength of an inter-spinous process device such as the X STOP. Spine (Phila Pa 1976). 2008;33:452–6.
- Miller JD, Miller MC, Lucas MG. Erosion of the spinous process: a potential cause of interspinous process spacer failure. J Neurosurg Spine. 2010;12:210–3.
- Kabir SM, Gupta SR, Casey AT. Lumbar interspinous spacers: a systematic review of clinical and biomechanical evidence. Spine (Phila Pa 1976). 2010;35:E1499–506.
- 55. Kim DH, Tantorski M, Shaw J, Martha J, Li L, Shanti N, Rencu T, Parazin S, Kwon B. Occult spinous process fractures associated with interspinous process spacers. Spine (Phila Pa 1976). 2011;36:E1080–5.
- Chou D, Lau D, Skelly A, Ecker E. Dynamic stabilization versus fusion for treatment of degenerative spine conditions. Evid Based Spine Care J. 2011;2:33–42.
- Costi JJ, Freeman BJ, Elliott DM. Intervertebral disc properties: challenges for biodevices. Expert Rev Med Devices. 2011;8:357–76.
- Bonaldi G. Minimally invasive dynamic stabilization of the degenerated lumbar spine. Neuroimaging Clin N Am. 2010;20:229–41.

- Albietz JS, Rosasarellano P, Fleming JC, Gurr KR, Bailey SI, Bailey CS. An anatomic study of the interspinous space of the lumbosacral spine. Eur Spine J. 2012;21:145–8.
- 60. Sobottke R, Koy T, Rollinghoff M, Siewe J, Kreitz T, Muller D, Bangard C, Eysel P. Computed tomography measurements of the lumbar spinous processes and interspinous space. Surg Radiol Anat. 2010;32:731–8.
- Xia Q, Wang S, Passias PG, Kozanek M, Li G, Grottkau BE, Wood KB, Li G. In vivo range of motion of the lumbar spinous processes. Eur Spine J. 2009;18:1355–62.
- Hartmann F, Dietz SO, Hely H, Rommens PM, Gercek E. Biomechanical effect of different interspinous devices on lumbar spinal range of motion under preload conditions. Arch Orthop Trauma Surg. 2011;131:917–26.
- Park SW, Lim TJ, Park J. A biomechanical study of the instrumented and adjacent lumbar levels after In-Space interspinous spacer insertion. J Neurosurg Spine. 2010;12:560–9.
- 64. Lazaro BC, Brasiliense LB, Sawa AG, Reyes PM, Theodore N, Sonntag VK, Crawford NR. Biomechanics of a novel minimally invasive lumbar interspinous spacer: effects on kinematics, facet loads, and foramen height. Neurosurgery. 2010;66:126–32; discussion 132–3.
- 65. Heuer F, Schmidt H, Kafer W, Graf N, Wilke HJ. Posterior motion preserving implants evaluated by means of intervertebral disc bulging and annular fiber strains. Clin Biomech (Bristol, Avon). 2012;27:218–25.
- Wilke HJ, Drumm J, Haussler K, Mack C, Kettler A. Biomechanics of interspinous spacers. Orthopade. 2010;39:565–72.
- Wilke HJ, Drumm J, Haussler K, Mack C, Steudel WI, Kettler A. Biomechanical effect of different lumbar interspinous implants on flexibility and intradiscal pressure. Eur Spine J. 2008;17: 1049–56.
- 68. Kaulhausen T, Siewe J, Eysel P, Knifka J, Notermans HP, Koebke J, Sobottke R. The role of the intersupraspinous ligament complex in stand-alone interspinous process devices: a biomechanical and anatomic study. J Neurol Surg A Cent Eur Neurosurg. 2012;73:65–72.
- 69. Hartmann F, Dietz SO, Kuhn S, Hely H, Rommens PM, Gercek E. Biomechanical comparison of an interspinous device and a rigid stabilization on lumbar adjacent segment range of motion. Acta Chir Orthop Traumatol Cech. 2011;78:404–9.
- Mao ZX, Jiang JM, Yan HB, Zhao WD, Wang FL, Wu Y, Li Y. Effect of Coflex interspinous stabilization and vertebral arch pedicle screw implantation on the stability of three-dimensional motions of the lumbar spine. Nan Fang Yi Ke Da Xue Xue Bao. 2010;30: 863–6.
- Trautwein FT, Lowery GL, Wharton ND, Hipp JA, Chomiak RJ. Determination of the in vivo posterior

loading environment of the Coflex interlaminarinterspinous implant. Spine J. 2010;10:244–51.

- 72. Kettler A, Drumm J, Heuer F, Haeussler K, Mack C, Claes L, Wilke HJ. Can a modified interspinous spacer prevent instability in axial rotation and lateral bending? A biomechanical in vitro study resulting in a new idea. Clin Biomech (Bristol, Avon). 2008;23:242–7.
- Li CD, Sun HL, Lu HZ. Comparison of the effect of posterior lumbar interbody fusion with pedicle screw fixation and interspinous fixation on the stiffness of adjacent segments. Chin Med J (Engl). 2013;126: 1732–7.
- Li CD, Sun HL, Yu ZR. Biomechanical study of interspinous fixational effect on the stiffness of adjacent segments. Beijing Da Xue Xue Bao. 2011;43:657–60.
- Ilharreborde B, Shaw MN, Berglund LJ, Zhao KD, Gay RE, An KN. Biomechanical evaluation of posterior lumbar dynamic stabilization: an in vitro comparison between Universal Clamp and Wallis systems. Eur Spine J. 2011;20:289–96.
- 76. Schulte TL, Hurschler C, Haversath M, Liljenqvist U, Bullmann V, Filler TJ, Osada N, Fallenberg EM, Hackenberg L. The effect of dynamic, semi-rigid implants on the range of motion of lumbar motion segments after decompression. Eur Spine J. 2008;17:1057–65.
- 77. Yao Q, Wang S, Shin JH, Li G, Wood K. Motion characteristics of the lumbar spinous processes with degenerative disc disease and degenerative spondylolisthesis. Eur Spine J. 2013;22:2702–9.
- Xu C, Ni WF, Tian NF, Hu XQ, Li F, Xu HZ. Complications in degenerative lumbar disease treated with a dynamic interspinous spacer (Coflex). Int Orthop. 2013;37:2199–204.
- 79. Zang L, Du P, Hai Y, Su QJ, Lu SB, Liu T. Device related complications of the Coflex interspinous process implant for the lumbar spine. Chin Med J (Engl). 2013;126:2517–22.
- 80. Zang L, Hai Y, Su QJ, Lu SB, Zhang CS, Yang JC, Guan L, Kang N, Meng XL, Liu T, et al. Device implanted complications of Coflex interspinous dynamic stabilization. Zhonghua Wai Ke Za Zhi. 2012;50:782–7.
- 81. Lee CH, Hyun SJ, Kim KJ, Jahng TA, Yoon SH, Kim HJ. The efficacy of lumbar hybrid stabilization using the DIAM to delay adjacent segment degeneration: an intervention comparison study with a minimum twoyear follow-up. Neurosurgery. 2013;73:ons224–32.
- 82. Anasetti F, Galbusera F, Aziz HN, Bellini CM, Addis A, Villa T, Teli M, Lovi A, Brayda-Bruno M. Spine stability after implantation of an interspinous device: an in vitro and finite element biomechanical study. J Neurosurg Spine. 2010;13(5):568–75.
- 83. Floman Y, Millgram MA, Smorgick Y, Rand N, Ashkenazi E. Failure of the Wallis interspinous implant to lower the incidence of recurrent lumbar disc herniations in patients undergoing primary disc excision. J Spinal Disord Tech. 2007;20:337–41.

- Jia YH, Sun PF. Preliminary evaluation of posterior dynamic lumbar stabilization in lumbar degenerative disease in Chinese patients. Chin Med J (Engl). 2012;125:253–6.
- Korovessis P, Repantis T, Zacharatos S, Zafiropoulos A. Does Wallis implant reduce adjacent segment degeneration above lumbosacral instrumented fusion? Eur Spine J. 2009;18:830–40.
- 86. Liu B, Yin D, Wang QM, Chang YB, Zhan SQ, Zeng SX, Ke YH, Wang YS, Xiao D. Lumbar interspinous non-fusion techniques: comparison between Coflex and Wallis. Nan Fang Yi Ke Da Xue Xue Bao. 2010;30:2455–8.
- Liu HY, Gu AQ, Zhu ZQ, Zhou J. The efficacy and complication analysis of interspinous dynamic device (Wallis) in patients of degenerative lumbar disease. Zhonghua Wai Ke Za Zhi. 2012;50:788–91.
- Sandu N, Schaller B, Arasho B, Orabi M. Wallis interspinous implantation to treat degenerative spinal disease: description of the method and case series. Expert Rev Neurother. 2011;11:799–807.
- 89. Sun HL, Li CD, Liu XY, Li H, Yu ZR, Lin JR, Yi XD, Liu H, Lu HL. Retrospective study of combined application of interspinous process fixation system and rigid fixation system for degenerative lumbar diseases. Zhonghua Wai Ke Za Zhi. 2010;48:363–7.
- 90. Sun HL, Li CD, Liu XY, Lin JR, Yi XD, Liu H, Lu HL. Mid-term follow-up and analysis of the failure cases of interspinous implants for degenerative lumbar diseases. Beijing Da Xue Xue Bao. 2011;43: 690–5.
- 91. Xu L, Yu X, Bi LY, Liu GZ, Li PY, Qu Y, Jiao Y. Intermediate and long-term follow-up evaluation of posterior dynamic lumbar stabilization in lumbar degenerative disease. Zhonghua Wai Ke Za Zhi. 2012;50:792–6.
- 92. Zhang ZJ, Pan B, Lu YS, Xu WG, Fu CD. Clinical analysis of an interspinous stabilization system (wallis) in treating lumbar degenerative disease. Zhongguo Gu Shang. 2012;25:463–7.
- Arrotegui I. Coflex interspinous spacer. Use in degenerative lumbar disc herniation. Acta Ortop Mex. 2010;24:187–90.
- 94. Celik H, Derincek A, Koksal I. Surgical treatment of the spinal stenosis with an interspinous distraction device: do we really restore the foraminal height? Turk Neurosurg. 2012;22:50–4.
- Chao L, He Q, Ruan DK. The clinical observation about Coflex of dynamic interspinous implant on the treatment of lumbar spinal stenosis. Zhongguo Gu Shang. 2011;24:282–5.
- Du FT. Clinical analysis of interspinous dynamic internal fixation with the Coflex system in treating lumbar degenerative disease. Zhongguo Gu Shang. 2011;24:291–4.
- 97. Liu J, Liu H, Li T, Zeng J, Song Y, Liu L, Gong Q. A comparative study between Coflex interspinous dynamic reconstruction and lumbar 360 degrees fusion in treating single-level degenerative lumbar

spinal disorders. Zhongguo Xiu Fu Chong Jian Wai Ke Za Zhi. 2011;25:693–8.

- Nachanakian A, El Helou A, Alaywan M. The interspinous spacer: a new posterior dynamic stabilization concept for prevention of adjacent segment disease. Adv Orthop. 2013;2013:637362.
- 99. Ni WF, Xu HZ, Zhou Y, Chi YL, Huang QS, Wang XY, Lin Y, Mao FM, Wu LJ. Clinical evaluation of interspinous process device Coflex for degenerative disk diseases. Zhonghua Wai Ke Za Zhi. 2012;50:776–81.
- 100. Richter A, Schutz C, Hauck M, Halm H. Does an interspinous device (Coflex) improve the outcome of decompressive surgery in lumbar spinal stenosis? One-year follow up of a prospective case control study of 60 patients. Eur Spine J. 2010;19: 283–9.
- 101. Villarejo F, Carceller F, de la Riva AG, Budke M. Experience with coflex interspinous implant. Acta Neurochir Suppl. 2011;108:171–5.
- 102. Fabrizi AP, Maina R, Schiabello L. Interspinous spacers in the treatment of degenerative lumbar spinal disease: our experience with DIAM and Aperius devices. Eur Spine J. 2011;20 Suppl 1:S20–6.
- 103. Ha KY, Seo JY, Kwon SE, Son IN, Kim KW, Kim YH. Posterior dynamic stabilization in the treatment of degenerative lumbar stenosis: validity of its rationale. J Neurosurg Spine. 2013;18:24–31.
- 104. Holinka J, Krepler P, Matzner M, Grohs JG. Stabilising effect of dynamic interspinous spacers in degenerative low-grade lumbar instability. Int Orthop. 2011;35:395–400.
- 105. Hrabalek L, Machac J, Vaverka M. The DIAM spinal stabilisation system to treat degenerative disease of the lumbosacral spine. Acta Chir Orthop Traumatol Cech. 2009;76:417–23.
- 106. Ryu SJ, Kim IS. Interspinous implant with unilateral laminotomy for bilateral decompression of degenerative lumbar spinal stenosis in elderly patients. J Korean Neurosurg Soc. 2010;47:338–44.
- 107. Zhao Y, Wang YP, Qiu GX, Zhao H, Zhang JG, Zhou X. Efficacy of the dynamic interspinous assisted motion system in clinical treatment of degenerative lumbar disease. Chin Med J (Engl). 2010;123: 2974–7.
- Hrabalek L, Wanek T, Adamus M. Percutaneous dynamic interspinous stabilisation for the treatment of juxtafacet cysts of the lumbar spine: prospective study. Acta Chir Orthop Traumatol Cech. 2012;79:144–9.
- 109. Hong SW, Lee HY, Kim KH, Lee SH. Interspinous ligamentoplasty in the treatment of degenerative spondylolisthesis: midterm clinical results. J Neurosurg Spine. 2010;13:27–35.
- 110. Li ZH, Wang SY, Tang H, Ma H, Zhang QL, Ho UT. Spinal fusion combined with dynamic interspinous fixation with Coflex system for lumbar degenerative disease. Zhongguo Gu Shang. 2011;24: 277–81.

- 111. Zhou SY, Chen XS, Jia LS, Zhu W, Fang L, Cai TY. Short-term clinical results of interspinous dynamic fixation of Coflex for the prevention of adjacent segment degeneration after lumbar fusion. Zhonghua Wai Ke Za Zhi. 2012;50:772–5.
- 112. DePalma MJ, Ketchum JM, Saullo T. What is the source of chronic low back pain and does age play a role? Pain Med. 2011;12:224–33.
- 113. Liu HY, Zhou J, Wang B, Wang HM, Jin ZH, Zhu ZQ, Miao KN. Comparison of topping-off and posterior lumbar interbody fusion surgery in lumbar degenerative disease: a retrospective study. Chin Med J (Engl). 2012;125:3942–6.
- 114. Richolt JA, Rauschmann MA, Schmidt S. Interspinous spacers–technique of Coflex implantation. Oper Orthop Traumatol. 2010;22:536–44.
- 115. Richter A, Halm HF, Hauck M, Quante M. 2-year follow-up after decompressive surgery with and without implantation of an interspinous device for lumbar spinal stenosis: a prospective controlled study. J Spinal Disord Tech. 2014;27(6):336–41.
- Schizas C, Duff JM, Tessitore E, Faundez A. Non fusion techniques in spinal surgery. Rev Med Suisse. 2009;5:2574–7.
- 117. Liu HY, Zhou J, Wang B, Wang HM, Jin ZH, Zhu ZQ, Miao KN. The effect of topping-off surgery on preventing adjacent segment degeneration, a retrospective study. Zhonghua Wai Ke Za Zhi. 2012;50:115–9.
- 118. Cho KS, Kang SG, Yoo DS, Huh PW, Kim DS, Lee SB. Risk factors and surgical treatment for symptomatic adjacent segment degeneration after lumbar spine fusion. J Korean Neurosurg Soc. 2009;46:425–30.
- Cusick JF, Yoganandan N, Pintar FA, Reinartz JM. Biomechanics of sequential posterior lumbar surgical alterations. J Neurosurg. 1992;76:805–11.
- 120. Tai CL, Hsieh PH, Chen WP, Chen LH, Chen WJ, Lai PL. Biomechanical comparison of lumbar spine instability between laminectomy and bilateral laminotomy for spinal stenosis syndrome – an experimental study in porcine model. BMC Musculoskelet Disord. 2008;9:84.
- 121. Cavanaugh JM, el-Bohy A, Hardy WN, Getchell TV, Getchell ML, King AI. Sensory innervation of soft tissues of the lumbar spine in the rat. J Orthop Res. 1989;7:378–88.
- 122. Yahia H, Newman N. A light and electron microscopic study of spinal ligament innervation. Z Mikrosk Anat Forsch. 1989;103:664–74.
- 123. Hartmann F, Janssen C, Bohm S, Hely H, Rommens PM, Gercek E. Biomechanical effect of graded minimal-invasive decompression procedures on lumbar spinal stability. Arch Orthop Trauma Surg. 2012;132:1233–9.
- 124. Marsh GD, Mahir S, Leyte A. A prospective randomised controlled trial to assess the efficacy of dynamic stabilisation of the lumbar spine with the Wallis ligament. Eur Spine J. 2014;23(10):2156–60.

- 125. Cabraja M, Abbushi A, Woiciechowsky C, Kroppenstedt S. The short- and mid-term effect of dynamic interspinous distraction in the treatment of recurrent lumbar facet joint pain. Eur Spine J. 2009;18:1686–94.
- 126. Park SC, Yoon SH, Hong YP, Kim KJ, Chung SK, Kim HJ. Minimum 2-year follow-up result of degenerative spinal stenosis treated with interspinous u (coflex). J Korean Neurosurg Soc. 2009;46:292–9.
- 127. Lee SH, Lee JH, Hong SW, Chung SE, Yoo SH, Lee HY. Spinopelvic alignment after interspinous soft stabilization with a tension band system in grade 1 degenerative lumbar spondylolisthesis. Spine (Phila Pa 1976). 2010;35:E691–701.
- 128. Sun HL, Li CD, Liu XY, Yi XD, Lin JR, Liu H, Lu HL, Li H, Yu ZR. Retrospective study of complication of interspinous implants for degenerative lumbar disease. Zhonghua Wai Ke Za Zhi. 2013;51: 35–9.
- 129. Li CD, Sun HL, Liu XY, Lin JR, Yi XD, Liu H, Lu HL. Retrospective study of application of interspinous implants for degenerative lumbar diseases. Zhonghua Yi Xue Za Zhi. 2009;89:3196–200.
- Maida G, Marcati E, Sarubbo S. Heterotopic ossification in vertebral interlaminar/interspinous instrumentation: report of a case. Case Rep Surg. 2012;2012: 970642.

- 131. Tian NF, Zhang XL, Wu YS, Jiang LB, Xu HZ, Chi YL. Fusion after interspinous device placement. Orthopedics. 2012;35:e1822–5.
- 132. Tian NF, Wu AM, Wu LJ, Wu XL, Wu YS, Zhang XL, Xu HZ, Chi YL. Incidence of heterotopic ossification after implantation of interspinous process devices. Neurosurg Focus. 2013;35:E3.
- 133. Barz T, Lange J, Melloh M, Staub LP, Merk HR, Kloting I, Follak N. Histomorphometric and radiographical changes after lumbar implantation of the PEEK nonfusion interspinous device in the BB.4S rat model. Spine (Phila Pa 1976). 2012;38: E263–9.
- 134. Ihm EH, Han IB, Shin DA, Kim TG, Huh R, Chung SS. Spinous process morphometry for interspinous device implantation in Korean patients. World Neurosurg. 2013;79:172–6.
- Limthongkul W, Yingsakmongkol W. Case report: cauda equina syndrome associated with an interspinous device. Clin Orthop Relat Res. 2012;470:1668–72.
- Jerosch J, Moursi MG. Foreign body reaction due to polyethylene's wear after implantation of an interspinal segment. Arch Orthop Trauma Surg. 2008;128:1–4.
- 137. Chung KJ, Hwang YS, Koh SH. Stress fracture of bilateral posterior facet after insertion of interspinous implant. Spine (Phila Pa 1976). 2009;34: E380–3.

## Interspinous Process Fixation for Motion Preservation

45

David M. Benglis Jr., Rishi Wadhwa, Praveen V. Mummaneni, and Regis W. Haid Jr.

## 45.1 Introduction

Neurogenic claudication from lumbar stenosis was first reported by Verbiest in 1954 [1]. It is a debilitating condition that affects 1.7–8 % of the US population and is particularly common in the elderly. Patients often find relief in flexion and exacerbation of symptoms in extension. Causes range from hypertrophy of the facet complexes or ligamentum flavum, disk herniation, and spondylolisthesis [2–4].

When 3–6 months of conservative therapy fails, operative intervention is typically indicated. The "gold standard" treatment for lumbar stenosis includes laminectomy, removal of hypertrophied ligament, and partial facetectomy, with or without diskectomy [5]. Fusions are commonly indicated in those patients with concomitant translational instability, coronal plane deformity, or need to remove a significant amount of the joint in order to achieve adequate

Department of Neurosurgery/Neuroscience, Atlanta Brain and Spine Care, Piedmont Hospital, 2001 Peachtree Road NE, Suite #575, Atlanta, GA 30309, USA e-mail: dbenglis@atlantabrainandspine.com

R. Wadhwa, MD • P.V. Mummaneni, MD Department of Neurosurgery, University of California, San Francisco, CA, USA e-mail: wadhwar@neurosurg.ucsf.edu; vmum@aol.com neurologic decompression. Minimally invasive options for lumbar decompression are also available with the use of tubular intramuscular dilators [6].

Interspinous devices (ID) are an alternative minimally invasive option in the lumbar spine for treatment of neurogenic claudication and back pain from lumbar stenosis. First described by Minns and Walsh, there are now multiple devices on the market. They are applied through a percutaneous or mini-open approach, with or without laminotomy, laminectomy, or diskectomy [7]. The basic premise behind their design is to create a state that prevents extension, thereby offloading the facet complexes and straightening out a "buckled" ligamentum flavum (e.g., for IDs without laminotomy), theoretically preventing a state of persistent nerve compression thereby decreasing the chances of restenosis. Some also provide an alternative to rigid arthrodesis for mild instability.

R.W. Haid Jr., MD Department of Neurosurgery/Neuroscience, Atlanta Brain and Spine Care, Piedmont Hospital, 2001 Peachtree Road NE, Suite #575, Atlanta, GA 30309, USA

Piedmont Spine Center and Neuroscience Service Line, Atlanta Brain and Spine Care, Piedmont Atlanta Hospital, Atlanta, GA, USA e-mail: rhaid@atlantabrainandspine.com

D.M. Benglis Jr., MD (🖂)

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_45

### 45.2 Current Devices

Examples of current IDs available in either the United States or outside US markets are ones that do not require laminectomy and are positioned in between the spinous processes: (1) X-Stop (Medtronic, Minneapolis, MN)-rigid titanium, second-generation PEEK body, through midline mini-open incision (Fig. 45.1); (2) Aperius (Medtronic, Tolochenaz, Switzerland)-rigid titanium through para-midline percutaneous incision and those IDs that are deployed following laminectomy/laminotomy/or diskectomy ; (3) DIAM (Medtronic, Tolochenaz, Switzerland)nonrigid polyester coated silicone with cord to anchor around spinous processes; (4) Wallis (Abbott Spine, Bordeaux, France)-secondgeneration nonrigid polyether ether ketone (PEEK) with ribbon to anchor around spinous processes; and (5) Coflex (Paradigm Spine, New York, NY)-titanium "U"-shaped device that clamps on superior and inferior spinous process (Fig. 45.2). X-Stop and Coflex are FDA approved for use in the United States. In this chapter we provide current evidence supporting or refuting the use of ID for motion preservation in the setting of stenosis without or without spondylolisthesis. Table 45.1 provides descriptions of various IDs.

### 45.3 General Indications for ID Placement

Patients should have relief of buttock, leg, and even back pain in a partially flexed forward position in order to be candidates for some of the devices. Following placement, in the early postoperative period, radiographic and clinical parameters are often significantly improved including foraminal height/width and intervertebral disk height [8, 9]. General contraindications include severe osteoporosis (due to the increased risk of spinous process fracture post placement), radiologic ankylosis/fusion at proposed treatment level, grade 2 or greater spondylolisthesis, severe scoliosis, and greater than two lumbar levels of stenosis (some articles



Fig. 45.1 X-Stop device



Fig. 45.2 Coflex device

Device	Manufacturer	Design	Indications	Potential complications
X-Stop <sup>a</sup>	Medtronic (Minneapolis, MN)	Interspinous spacer providing indirect decompression	Lumbar spinal stenosis	Implant dislodgement
				Improper placement
				Spinous process fracture
				Implant mechanical failure
Aperius	Medtronic (Tolochenaz, Switzerland)	Percutaneous lumbar interspinous decompression system	Lumbar spinal stenosis	Same as above
DIAM	Medtronic (Tolochenaz, Switzerland)	Silicon device secured to spinous processes to provide flexible support	Lumbar spinal stenosis	Same as above
Wallis	Abbott Spine (Bordeaux, France)	PEEK spacer providing indirect decompression and providing resistance against flexion	Lumbar spinal stenosis	Same as above
Coflex <sup>a</sup>	Paradigm (New York, NY)	Interlaminar stabilizer and motion preservation after decompression	Lumbar spinal stenosis	Same as above

Table 45.1 Descriptions of IDs

<sup>a</sup>FDA approved for use in the United States

implant IDs over three levels however), the presence of motor deficits, and the presence of flat back with a lumbar lordosis-pelvic incidence (LL-PI) mismatch over 10°.

## 45.4 Gold Standard Treatment for Neurogenic Claudication Is Laminectomy

Recent studies have supported laminectomy over continued conservative management for neurogenic claudication from lumbar stenosis. The Spine Patient Outcomes Research Trial (SPORT) demonstrated the superiority of laminectomy versus conservative management for lumbar spinal stenosis at both 2- and 4-year time points [10, 11]. In a European prospective study of 159 patients who underwent laminectomy and partial facetectomy for neurogenic claudication, at 5 years 79 % of those patient's reported a "good" global outcome [12].

A certain percentage of patients, however, fail laminectomy due to recurrent stenosis and ultimately require a fusion operation. For example, in the SPORT trial at 2 years, reoperations occurred in 8 % of patients (n=289), half of those reoperations were due to recurrent stenosis (i.e., a failure of the initial operation), and most of those patients received a fusion operation [10]. One explanation for why patients fail laminectomy is due to the presence of worsening deformity from iatrogenic instability [13]. Additionally, the utility of laminectomy alone in the setting of significant back pain or spondylolisthesis is questionable [14, 15]. Fusion may be indicated in these settings, and those patients tend to maintain greater pain relief and improvement in function over time [16, 17].

Proponents of IDs state that lumbar fusions carry nascent risks in both the perioperative and late postoperative periods including pseudoarthrosis and painful hardware. Intraoperative risks are also encountered with the placement of pedicle screws and interbody devices [18]. Adjacent segment degeneration (ASD) is a controversial topic and likely has contributory factors from both the patients' own progressive arthritic disease state and the shear forces on non-fused segments from prior fused vertebrae [19]. Consequently this may lead to further surgery for extensions of fusion.

## 45.5 Do IDs Provide the "Intended" Bridge Between Laminectomy and Laminectomy/Fusion?

The implementation of alternative stabilizing devices (i.e., ID) that preserve motion with or without laminectomy provides a minimally invasive alternative to either laminectomy alone or laminectomy with fusion in certain situations [20]. In a large retrospective cohort analysis (n=99,084) of patients who received an ID, ID + laminectomy, decompression alone, or lumbar fusion (one to two level), the authors found that patients receiving an ID were generally older, yet did not have a higher degree of comorbid conditions preoperatively compared to the other groups.

Patients undergoing ID placement encountered a lower percentage of procedural morbidity (i.e., faster operative times, less blood loss, less CSF leaks, ability to utilize non-general anesthesia) but had higher rates of further inpatient surgery in the future for revision operations. This is the major drawback of the implantation of IDs [21]. Is a small gain in upfront risk associated with the initial procedure worth a higher longterm risk of reoperation compared to other therapies?

Common complications reported in various studies with IDs include spinous process fractures (most are asymptomatic), device malposition over time, failure of the device to provide continued stabilization of the motion segment and decompression of the neural elements, and creation of flat back. Kim and colleagues reported that a higher incidence of spinous process fractures occurred when a particular ID was used in patients with spondylolisthesis [22].

### 45.6 Biomechanical Data

## 45.6.1 Cadaveric

Wilke et al. analyzed 4 IDs (Coflex, DIAM, WILLIS, and X-STOP) in a set of 24 cadavers. In summary all devices similarly limited motion by 50 % in extension compared to the state of the

intact spine without the device. Intradiscal pressure was also reduced in extension compared to the intact spine. Flexion, lateral bending, and axial rotation were affected very little by either of the devices [23]. Lafage et al. noted similar results in a cadaveric study examining only the Wallis implant [24].

An alternate biomechanical study by Tsai et al. of the Coflex ID reported different results, however. They found that Coflex did not allow significantly more or less motion than an intact cadaveric specimen following partial destabilization (removal of supra-/interspinous ligaments, ligamentum flavum, medial facetectomy) in flexion/extension or axial rotation. In this study Coflex restored a spine segment to preoperative levels of stability following decompression [25]. Phillips in a similar study using the DIAM device noted that the ID restored pre-diskectomy stability in flexion/extension and reduced but did not restore in lateral bending. There was little to no effect on axial rotation [26].

Multiple cadaveric biomechanical studies examining the X-Stop device have been performed. Wiseman et al. observed that specimens with an ID had significantly lower forces on the facets, while Swanson noted decreased intradiscal pressures at the instrumented level [27, 28]. No effect of the device was observed at the adjacent-level segments in either of the studies. Richards et al. noted that following placement of an X-Stop ID in a cadaveric specimen, during extension the ID significantly increased the canal diameter and foraminal size [9].

### 45.6.2 In Vivo

Zucherman et al. noted no differences in radiographic parameters (spinous process distance, A/P disk height, foraminal height) when the surgical group was compared to the control nonoperative arm at both 12 and 24 months with the X-Stop device [29]. These findings are in contrast to a study by Lee et al. who noted a 23 % increase in canal diameter in preoperative versus postoperative MRI following X-Stop placement for primarily single-level stenosis [30]. In a retrospective study by Sobottke et al. of 129 patients who received either the DIAM, Wallis, or X-Stop ID, they observed significant changes in foraminal measurements and disk height compared to preop [8]. There was no strong correlation between the magnitude of correction and symptom relief. During the follow-up period, the radiographic findings tended to revert back to pre-op without a return of symptoms.

Holinka et al. noted decreased instability in flexion and extension postoperatively following interlaminar decompression and DIAM implantation in 22 patients with low-grade <5 mm spondylolisthesis [31]. Kong et al. in a case series of 42 patients comparing Coflex to PLIF noted increased motion at the adjacent segment in the PLIF cohort and no increased motion in the Coflex at 1 year [32]. Following lumbar decompression and DIAM placement, Kim noted no change in disk height or sagittal alignment at the mean 12-month follow-up [33]. Korovessis conducted a prospective case-control study examining adjacent segment degeneration (ASD) in patients with fusion alone versus fusion plus the Wallis implant at the adjacent superior segment in an average follow-up of 60 months [34]. Disk height significantly increased and range of motion decreased in all planes in the Wallis group. Over time ASD was noted in a greater number of non-Wallis-treated spines.

## 45.7 Devices: Clinical Literature and Surgical Technique

45.7.1 Interspinous Devices Placed Without Removal of Ligament/Laminectomy or Laminotomy

## 45.7.1.1 X-Stop (Titanium or PEEK Mini-Open Midline Approach): Clinical Literature

The X-Stop device exists as a minimally invasive alternative to open laminectomy in patients with lumbar spinal stenosis and neurogenic claudication. This device is also intended to indirectly open the foramen. It has been on the market for more than 10 years and involves placement of the device in the interspinous space and maintains the segment in some degree of flexion and limits extension, thereby enlarging the spinal canal and neuroforamina and off-loads the facets and disk space at the instrumented level.

It is best utilized in patients with stenosis from ligamentum flavum buckling and not favored in patients with boney stenosis or rigid spinal segments. Biomechanically the X-Stop device has no effect on axial rotation or lateral bending at the instrumented level and has no effect on any parameter at adjacent levels [27, 28, 35]. Although not reported extensively in the literature, there is a second-generation X-Stop titanium outer ring and PEEK-bone interface device available and approved by the FDA that provides greater contact area with bone when compared to the first-generation all titanium [36].

A multicenter randomized controlled trial (RCT) showed that ID (X-STOP) had improved outcomes compared to conservative therapies and epidural injections in both short- and long-term follow-up (4 years) [29, 37]. Anderson et al. also conducted an RCT in patients with spondylolisthesis and showed that patients who received ID had improved measures in all parameters compared to conservative therapy [38]. Verhoof noted that patients with spondylolisthesis and ID placement had a high return rate for revision decompression and fusion within 2 years (7/12, 58.3 %) [39]. A cost-effective analysis of the treatment of the lumbar stenosis population found laminectomy was the lowest cost followed by X-Stop and then conservative management [40].

Patil et al. examined reoperation rates, complications, and costs in patients who had received an ID (X-STOP) in a retrospective study of 498 patients over a mean of 1.2 years. They found that reoperation rates were significantly high (22 %) with the most common reoperation being laminectomy (n=60) versus new ID placement (n=52), mean hospital stay was approximately 1.5 days, and mean hospital cost was close to 20,000 dollars. When compared to a match cohort for laminectomy (n=348), the group found that persons who received an ID had incurred an increased cost at the index hospitalization that
had leveled out by the 12- and 18-month mark and the laminectomy patients had more complications at the 30- and 90-day mark [41].

Other groups have shown cost-effectiveness when it can be performed as an outpatient procedure [42]. Stromqvist et al. conducted a prospective randomized controlled trial of 100 patients comparing this particular ID done under local anesthesia to decompressive laminectomy for one- to two-level stenosis without spondylolisthesis [43]. Mean surgery time and EBL were significantly less in the ID group. Zurich Claudication Questionnaire (ZCQ), visual analog scale (VAS), and SF-36 improved similarly between the groups at 6, 12, and 24 months. Three (6 %) patients in the laminectomy group received a reoperation, while 13 (26 %) received one in the ID group. Eleven of those 13 patients who received an ID had no symptom alleviation throughout the course following the procedure. Although the reoperation rates in that study were particularly high, and the reason for reoperation was ineffective treatment of symptoms by the device, in an alternate prospective series of 175 patients, Kutchta et al. found a much lower reoperation rate at 24 months (n=8, 4.6 %) [44]. Barbagallo reported in a retrospective review of 69 patients who received the X-Stop device seven postoperative complications (10.1 %) (three spinous process fractures and three device dislodgements) all requiring revision surgery [45].

Only one trial has compared the "gold standard" treatment for neurogenic claudication (i.e., laminectomy) versus an ID without laminectomy. Moojen et al. conducted a randomized controlled trial of 159 participants [46]. ZCQ analysis revealed no significant differences in long-term follow-up at 2 years, yet reoperations were significantly higher in the ID (n=21, 29 %) group versus the laminectomy (n=6, 8 %).

# 45.7.1.2 Aperius (Titanium Percutaneous Posterolateral Minimally Invasive): Clinical Literature

Multiple studies without control groups have shown clinical benefit compared to preoperative symptoms following placement of the Aperius device. It is not approved by the FDA for general use in the United States as of 2014. Two retrospective studies using Aperius ID (1 in 40 patients and the other in 152 patients) with a follow-up over a period of 9–12 months reported improvement in measured outcomes (VAS, ZCQ, and Macnab criteria) [47, 48]. In a safety study, Van Meirhaeghe et al. reported on 156 patients who received the device at one to three levels all with 12-month follow-up [49]. The mean duration for one-level procedures was 15.5 min. They noted improvements in walking distances, VAS and ZCQ scores at both 6-week and 12-month time points. Mean lordosis (L1-S1) decreased from 54.1 to 52.4° at 12 months. During the 12-month follow-up, 12 patients (7.7 %) had their ID removed due to persistent symptoms (most common), spinous process fracture, or malposition.

Two more powerful studies have compared the Aperius ID with open laminectomy with less than satisfactory results supporting ID placement. Postacchini compared n=36 receiving the ID and n=35 with laminectomy. The long-term follow-up was over a period of 2 years [50]. They separated the patients into subgroups of moderate versus severe stenosis. Approximately a onethird of patients in each group had grade 1 spondylolisthesis in addition to stenosis. Mean postoperative ODI and ZCQ scores for the laminectomy group were significantly better than the ID at all time points. For the ID, the rate of "good" results was significantly better in patients who had moderate (<40 % constriction of the canal diameter on MRI) and not severe stenosis. Additionally, 6/36 (16.7 %) had their device removed between 2 and 16 months because of no improvement in symptoms or initial improvement with return of symptoms, while only one patient in the open decompression group had revision surgery.

Beyer et al. reported on 2-year outcomes of 45 patients with neurogenic intermittent claudication comparing standard lumbar laminectomy to ID placement in a prospective nonrandomized observational study. Group 1 patients (n=12) received the Aperius ID. 5/12 patients required implant removal due to return of preoperative symptoms. No device dislodgments were noted.

All five patients had devices removed with a standard decompression at a mean of 13 months. The remaining patients in Group 1 showed no significant decrease in back and leg pain or outcome measures ODI/SF-36 over 2 years, but did have significant improvements in walking assessment. Group 2 patients on the contrary who received laminectomy had significant decreases in their back and leg pain at both 12 and 24 months as well as significant improvement in outcome measures and walking assessment [51].

## 45.7.1.3 Surgical Technique: X-STOP and Aperius

The patient is placed in the prone position over a Wilson frame. Monitored anesthesia care (MAC) with local anesthesia or local alone is used. Some groups have advocated the use of spinal epidural anesthesia as well.

A midline (X-STOP—Medtronic, Minneapolis, MN) or posterolateral (1.5 cm from midline— Aperius—(Medtronic, Switzerland) is performed over the respective level in the lumbar spine. The fascia is opened and trocars or spacers are sized through the interspinous ligament for the successive preparation of the interspinous space. Preservation of the supraspinous ligament is achieved.

An initial trial of 8 mm is used, and dilation through the interspinous ligament can continue in increments of 2 mm. The most common size of implant in one study was 12 mm. Maximum trocar size and implant is 14 mm [49]. X-Stop PEEK is sized up to 16 mm. Intraoperative fluoroscopy is utilized to check midline position of the implant before the wings are deployed.

# 45.7.2 Interspinous Device Mini-Open with Microsurgical Decompression (Laminotomy/ Laminectomy With/Without Diskectomy)

#### 45.7.2.1 Coflex

Coflex (Paradigm Spine, New York, NY) is a novel metallic "U"-shaped interspinous device utilized in one- or two-level stenosis of the lumbar region L1–L5. It is designed to impart a stabilizing effect on the operative levels. In preoperative assessment similar to other IDs listed above, the patient should experience relief of symptoms in flexion. Unlike the X-Stop or Aperius, the Coflex is placed after decompression (laminectomy/bilateral laminotomy) of the affected segments. The wings have serrated bone gripping surfaces and are attached to the superior and inferior spinous processes. The device is MRI compatible and is contraindicated in spondylolisthesis equal to or greater than grade 2, prior fusion, total removal of hypertrophic facets that would destabilize the spine, and degenerative scoliosis >25°.

## 45.7.2.2 Coflex: Clinical Literature

Davis et al. conducted a prospective, randomized, multicenter trial evaluating the safety and efficacy of Coflex compared with posterior spinal fusion in treating one- to two-level stenosis (with neurogenic claudication and back pain) with or without spondylolisthesis grade 1 (n=322patients (215 Coflex, 107 fusions) [20]. Patients were randomized to receive laminectomy with Coflex interlaminar stabilization or laminectomy with posterolateral fusion and pedicle screws (*without BMP or interbody*). This was an investigational device exemption (IDE) study. Very few of these types of studies exist in the spine literature.

Coflex was designed to maintain some motion, yet provides stability following laminectomy. The Coflex cohort exhibited increased ZCQ and SF-12 scores versus control, a trend toward better ODI at 24 months, no difference in VAS scores (except early post-op period trend for Coflex with back and leg pain), shorter hospital stays, lower blood loss, and shorter operative times.

At 2 years the Coflex group maintained normal operative and adjacent-level motion versus fusion. The study did not address whether this resulted in lower reoperation rates at the adjacent levels. Adverse events were comparable in both groups. Surgical adverse events were 23.7 % in Coflex group versus 30.8 % in the fusion.

Spinous process fracture incidence (most were asymptomatic) in the Coflex group was 14 %; however, approximately half of those had

healed at 2 years. Reoperation rate was higher in the Coflex group. 10.7 % (23/215) and 13/23 were converted to a fusion versus 7.5 % (8/107)reoperation rate in the fusion group. The authors stated those that were converted to fusion (some > than 12 months) from receiving Coflex that this device may serve as a "bridge" between conservative therapy and fusion. This study did not compare Coflex to the "gold standard" treatment for lumbar stenosis (lumbar laminectomy). In summary Coflex provided equivalent in some and superior in other outcome measures when compared to fusion for lumbar stenosis with neurogenic claudication, with or without low back pain and spondylolisthesis; however, reoperation rates for the device were higher.

Davis et al. also performed a subgroup analysis in the study above of only those patients treated with either Coflex (n=99) or fusion (n=51) for spondylolisthesis [18]. ODIs, SF-12, VAS, and ZCQ were similar among the two groups at 2 years. Preoperative spondylolisthesis in both groups averaged 1.12 mm in the Coflex group versus 0.98 mm in the fusion controls which could be argued as nominal or physiologic in some cases.

As expected, at 2 years in the operative level, the Coflex group had more angulation versus the fusion (4.32° vs. 1.64°), but less at the adjacent above level (3.49° versus 5.42°), and similar at the adjacent below level. No worsened spondylolisthesis was noted in the Coflex group at the operative level at 2 years. At the superior adjacent level, Coflex group experienced a reduction in translation (0.87  $\rightarrow$  0.74 mm) versus an average increase in the fusion group (0.64  $\rightarrow$  0.97 mm).

Spinous process fractures were noted in 18 % of Coflex patients (18/99), while 6.4 % of patients had movement of their Coflex device. The rate of radiographic fusion was 71 % (36/51) (*no BMP or interbody was utilized in this study*), yet no difference in outcome was noted in patients with a solid fusion versus pseudoarthrosis. Fusion rates for local autograft versus iliac crest were similar as well.

In summary both groups met the study's criteria for success; however, this subgroup analysis Coflex patients returned to the operating room a higher number of times versus the fusion group (14.1 % 14/99 vs. 5.9 % 3/51) and 8/14 patients were converted to a fusion.

Richter et al. conducted a 2-year prospective controlled study comparing laminectomy with Coflex device (n=31) versus laminectomy alone (n=31) [52]. No significant differences were noted in all outcome parameters. One implant dislocation occurred, and three patients in the ID group were converted to fusion. Two patients in the laminectomy group required a later fusion.

#### 45.7.2.3 Surgical Technique: Coflex

A midline incision is made exposing and removing the ligamentum flavum, and a bilateral hemilaminotomy is carried out with a high-speed burr along with a partial facetectomy. The supraspinous and interspinous ligament is taken last. Shaping of the two spinous processes with the burr may be necessary for an appropriate fit. Care must be taken to keep at least 14 mm of spinous process present as less increases the risk of spinous process fracture [18]. Trials (smaller first) are then deployed to choose a proper end sized implant. They come in five different sizes and range from 8 to 16 mm in 2 mm increments. The goal is to provide 1-2 mm of facet distraction. Following trial deployment, an appropriately sized Coflex device is inserted into the interspinous space. The wings of the Coflex device can be opened with a bender to facilitate positioning. The device is seated 1-2 mm above the dura or the apex of the "U" at the midline of the facet joint. The wings are then crimped. If necessary removal is achieved with bending of the wings open with chisel and forceps. Postoperative x-rays are demonstrated for a multilevel procedure (Fig. 45.3).

# 45.7.2.4 "Soft Devices": Wallis, Second-Generation (PEEK), and DIAM, Second-Generation (Silicone)

## **Clinical Literature and Surgical Technique**

The Wallis and DIAM devices are referred to as "soft" IDs due to their non-titanium composition. These IDs are not FDA approved for use in the United States as of 2014. They have cords



**Fig. 45.3** Postoperative lateral x-ray demonstrating two-level decompression and placement of Coflex L3/L4, L4/L5 (Complements of Gary Gropper, MD)

that are fixated around the respective spinous processes of the level treated. In some of the reported literature, patients have had a prior lumbar operation (i.e., diskectomy or laminotomy) and placement of the device is intended to stabilize the segment following a revision, or in some cases, a diskectomy is performed along with the implant placement. A portion of the superior and inferior spinous processes must be preserved similar to the Coflex device. These IDs may be placed like the X-Stop without a decompression where the intention is intended to unbuckle the ligamentum flavum or indirectly open the foramen.

They are placed through a midline mini-open incision similar to the X-Stop and Aperius. The Wallis implant is 30 times less rigid than its firstgeneration titanium counterpart [34]. It consists of an interspinous PEEK block that is appropriately sized (10–16 mm) to the interspinous space. The DIAM on the other hand is an "X"-shaped silicone wedge covered in polyester knit. The Wallis requires removal of the interspinous ligament, while the DIAM preserves the supraspinous ligament and fits through the interspinous ligament in procedures that do not require a decompression. Trials are utilized to determine the appropriate final size of the implant. The implant is then inserted and seated with a tamp and mallet. The cords are then wrapped around the superior and inferior spinous process, tightened, and then crimped.

Mariottini et al. performed a study on the DIAM device on 43 patients. They reported 97 % satisfying results over the study period where this device was placed following laminotomy and microsurgical nerve root decompression [53]. Taylor and colleagues also reported satisfactory results with very few device-related revisions in a retrospective study of 104 patients who received the DIAM ID [54].

Senegas et al. examined 107 patients with the Wallis implant retrospectively over a period of 13 years and noted a good outcome (ODI, SF-36, VAS) in 80 % while 20 % had to be reoperated on where the implant was removed and a fusion performed [55]. Kim et al. compared 31 patients (n=8 with recurrent disk herniation with equal back and leg pain, n=15 with disk herniation with equal back and leg pain, n=8 with lumbar stenosis with mild instability) who received a DIAM following lumbar surgery with 31 patients receiving lumbar surgery without ID in a casecontrol safety study with mean of 12-month follow-up. They noted no differences in outcomes between the two groups and a higher complication rate of the DIAM group mainly due to resultant spinous process fractures or recurrent disk herniations (authors did not elaborate on a cause) [33]. Floman et al. did not find a reduced incidence in recurrent disk herniations in Wallis ID versus non-ID diskectomies over a 16-month period, and Korovessis et al. noted increased ASD in patients who had instrumented lumbar fusions without Wallis at the superior adjacent segment [34].

#### Conclusions

The premise behind IDs is to provide stabilization at the operative level and create an environment of restricted extension. A distinction should be made between those devices that provide indirect decompression following deployment (X-Stop/Aperius/ DIAM/Wallis) versus those that can be placed in conjunction with a laminotomy (Coflex, DIAM) or diskectomy (Wallis). The ideal candidate has yet to be conclusively determined for ID placement. While some patients have long-term benefit following ID, a high percentage of patients with IDs need revision operations (this is the biggest disadvantage of this particular therapy). They may be indicated for a limited number of patients (i.e., those who would encounter morbidity with general anesthesia), but the surgeon may be trading off early risk with the initial operation for a higher rate of return to the operating room for revision.

#### **Key Points**

- 1. The "gold standard" treatment for neurogenic claudication from lumbar stenosis without translational or coronal instability is LAMINECTOMY.
- 2. The X-Stop and Aperius ID are placed without a bony decompression, while the Wallis, DIAM, and Coflex are placed following a bony decompression.
- Patients who undergo ID placement generally have a lower procedural morbidity but have higher percentages of return to the operating room for revisions compared to the gold standard laminectomy.
- 4. ID complications include spinous process fractures, device malposition, and failure of the device to provide continued stabilization of the motion segment and decompression of the neural elements.
- 5. IDs tend to limit the spine in extension, reduce intradiscal pressure, and offload facets at the instrumented level, but have little effect on the adjacent levels.
- Avoid ID in patients with flat backs who have LL-PI mismatch over 10° as the devices increase kyphosis.

#### References

- Verbiest H. A radicular syndrome from developmental narrowing of the lumbar vertebral canal. J Bone Joint Surg Br Vol. 1954;36-B(2):230–7.
- De Villiers PD, Booysen EL. Fibrous spinal stenosis. A report on 850 myelograms with a water-soluble contrast medium. Clin Orthop Relat Res. 1976; 115:140–4.
- Kovacs FM, Urrutia G, Alarcon JD. Surgery versus conservative treatment for symptomatic lumbar spinal stenosis: a systematic review of randomized controlled trials. Spine. 2011;36(20):E1335–51.
- Roberson GH, Llewellyn HJ, Taveras JM. The narrow lumbar spinal canal syndrome. Radiology. 1973; 107(1):89–97.
- Siebert E, Pruss H, Klingebiel R, Failli V, Einhaupl KM, Schwab JM. Lumbar spinal stenosis: syndrome, diagnostics and treatment. Nat Rev Neurol. 2009;5(7):392–403.
- Asgarzadie F, Khoo LT. Minimally invasive operative management for lumbar spinal stenosis: overview of early and long-term outcomes. Orthop Clin N Am. 2007;38(3):387–99; abstract vi–vii.
- Minns RJ, Walsh WK. Preliminary design and experimental studies of a novel soft implant for correcting sagittal plane instability in the lumbar spine. Spine. 1997;22(16):1819–25; discussion 26–7.
- Sobottke R, Schluter-Brust K, Kaulhausen T, et al. Interspinous implants (X Stop, Wallis, Diam) for the treatment of LSS: is there a correlation between radiological parameters and clinical outcome? Eur Spine J. 2009;18(10):1494–503.
- Richards JC, Majumdar S, Lindsey DP, Beaupre GS, Yerby SA. The treatment mechanism of an interspinous process implant for lumbar neurogenic intermittent claudication. Spine. 2005;30(7):744–9.
- Weinstein JN, Tosteson TD, Lurie JD, et al. Surgical versus nonsurgical therapy for lumbar spinal stenosis. N Engl J Med. 2008;358(8):794–810.
- Weinstein JN, Tosteson TD, Lurie JD, et al. Surgical versus nonoperative treatment for lumbar spinal stenosis four-year results of the spine patient outcomes research trial. Spine. 2010;35(14):1329–38.
- Mannion AF, Denzler R, Dvorak J, Grob D. Five-year outcome of surgical decompression of the lumbar spine without fusion. Eur Spine J. 2010;19(11): 1883–91.
- Johnsson KE, Willner S, Johnsson K. Postoperative instability after decompression for lumbar spinal stenosis. Spine. 1986;11(2):107–10.
- Kleinstuck FS, Grob D, Lattig F, et al. The influence of preoperative back pain on the outcome of lumbar decompression surgery. Spine. 2009;34(11): 1198–203.
- Herkowitz HN, Kurz LT. Degenerative lumbar spondylolisthesis with spinal stenosis. A prospective study comparing decompression with decompression and intertransverse process arthrodesis. J Bone Joint Surg (Am Vol). 1991;73(6):802–8.

- Weinstein JN, Lurie JD, Tosteson TD, et al. Surgical versus nonsurgical treatment for lumbar degenerative spondylolisthesis. N Engl J Med. 2007;356(22): 2257–70.
- Weinstein JN, Lurie JD, Tosteson TD, et al. Surgical compared with nonoperative treatment for lumbar degenerative spondylolisthesis. Four-year results in the Spine Patient Outcomes Research Trial (SPORT) randomized and observational cohorts. J Bone Joint Surg (Am Vol). 2009;91(6):1295–304.
- 18. Davis R, Auerbach JD, Bae H, Errico TJ. Can lowgrade spondylolisthesis be effectively treated by either coflex interlaminar stabilization or laminectomy and posterior spinal fusion? Two-year clinical and radiographic results from the randomized, prospective, multicenter US investigational device exemption trial: clinical article. J Neurosurg Spine. 2013;19(2):174–84.
- Park P, Garton HJ, Gala VC, Hoff JT, McGillicuddy JE. Adjacent segment disease after lumbar or lumbosacral fusion: review of the literature. Spine. 2004;29(17):1938–44.
- 20. Davis RJ, Errico TJ, Bae H, Auerbach JD. Decompression and coflex interlaminar stabilization compared with decompression and instrumented spinal fusion for spinal stenosis and low-grade degenerative spondylolisthesis: two-year results from the prospective, randomized, multicenter, food and drug administration investigational device exemption trial. Spine. 2013;38(18):1529–39.
- Deyo RA, Martin BI, Ching A, et al. Interspinous spacers compared with decompression or fusion for lumbar stenosis: complications and repeat operations in the medicare population. Spine. 2013;38(10): 865–72.
- Kim DH, Shanti N, Tantorski ME, et al. Association between degenerative spondylolisthesis and spinous process fracture after interspinous process spacer surgery. Spine J. 2012;12(6):466–72.
- Wilke HJ, Drumm J, Haussler K, Mack C, Steudel WI, Kettler A. Biomechanical effect of different lumbar interspinous implants on flexibility and intradiscal pressure. Eur Spine J. 2008;17(8):1049–56.
- 24. Lafage V, Gangnet N, Senegas J, Lavaste F, Skalli W. New interspinous implant evaluation using an in vitro biomechanical study combined with a finiteelement analysis. Spine. 2007;32(16):1706–13.
- 25. Tsai KJ, Murakami H, Lowery GL, Hutton WC. A biomechanical evaluation of an interspinous device (Coflex) used to stabilize the lumbar spine. J Surg Orthop Adv. 2006;15(3):167–72.
- Phillips FM, Voronov LI, Gaitanis IN, Carandang G, Havey RM, Patwardhan AG. Biomechanics of posterior dynamic stabilizing device (DIAM) after facetectomy and discectomy. Spine J. 2006;6(6): 714–22.
- Swanson KE, Lindsey DP, Hsu KY, Zucherman JF, Yerby SA. The effects of an interspinous implant on intervertebral disc pressures. Spine. 2003;28(1): 26–32.

- Wiseman CM, Lindsey DP, Fredrick AD, Yerby SA. The effect of an interspinous process implant on facet loading during extension. Spine. 2005;30(8): 903–7.
- Zucherman JF, Hsu KY, Hartjen CA, et al. A multicenter, prospective, randomized trial evaluating the X STOP interspinous process decompression system for the treatment of neurogenic intermittent claudication: two-year follow-up results. Spine. 2005;30(12): 1351–8.
- Lee J, Hida K, Seki T, Iwasaki Y, Minoru A. An interspinous process distractor (X STOP) for lumbar spinal stenosis in elderly patients: preliminary experiences in 10 consecutive cases. J Spinal Disord Tech. 2004;17(1):72–7; discussion 8.
- Holinka J, Krepler P, Matzner M, Grohs JG. Stabilising effect of dynamic interspinous spacers in degenerative low-grade lumbar instability. Int Orthop. 2011;35(3):395–400.
- Kong DS, Kim ES, Eoh W. One-year outcome evaluation after interspinous implantation for degenerative spinal stenosis with segmental instability. J Korean Med Sci. 2007;22(2):330–5.
- 33. Kim KA, McDonald M, Pik JH, Khoueir P, Wang MY. Dynamic intraspinous spacer technology for posterior stabilization: case-control study on the safety, sagittal angulation, and pain outcome at 1-year follow-up evaluation. Neurosurg Focus. 2007;2 2(1):E7.
- Korovessis P, Repantis T, Zacharatos S, Zafiropoulos A. Does Wallis implant reduce adjacent segment degeneration above lumbosacral instrumented fusion? Eur Spine J. 2009;18(6):830–40.
- 35. Lindsey DP, Swanson KE, Fuchs P, Hsu KY, Zucherman JF, Yerby SA. The effects of an interspinous implant on the kinematics of the instrumented and adjacent levels in the lumbar spine. Spine. 2003;28(19):2192–7.
- Kim DH, Tantorski M, Shaw J, et al. Occult spinous process fractures associated with interspinous process spacers. Spine. 2011;36(16):E1080–5.
- 37. Kondrashov DG, Hannibal M, Hsu KY, Zucherman JF. Interspinous process decompression with the X-STOP device for lumbar spinal stenosis: a 4-year follow-up study. J Spinal Disord Tech. 2006;19(5): 323–7.
- Anderson PA, Tribus CB, Kitchel SH. Treatment of neurogenic claudication by interspinous decompression: application of the X STOP device in patients with lumbar degenerative spondylolisthesis. J Neurosurg Spine. 2006;4(6):463–71.
- 39. Verhoof OJ, Bron JL, Wapstra FH, van Royen BJ. High failure rate of the interspinous distraction device (X-Stop) for the treatment of lumbar spinal stenosis caused by degenerative spondylolisthesis. Eur Spine J. 2008;17(2):188–92.
- Burnett MG, Stein SC, Bartels RH. Cost-effectiveness of current treatment strategies for lumbar spinal stenosis: nonsurgical care, laminectomy, and X-STOP. J Neurosurg Spine. 2010;13(1):39–46.

- Patil CG, Sarmiento JM, Ugiliweneza B, et al. Interspinous device versus laminectomy for lumbar spinal stenosis: a comparative effectiveness study. Spine J. 2014;14:1484–92.
- Skidmore G, Ackerman SJ, Bergin C, et al. Costeffectiveness of the X-STOP(R) interspinous spacer for lumbar spinal stenosis. Spine. 2011;36(5): E345–56.
- 43. Stromqvist BH, Berg S, Gerdhem P, et al. X-stop versus decompressive surgery for lumbar neurogenic intermittent claudication: randomized controlled trial with 2-year follow-up. Spine. 2013;38(17):1436–42.
- 44. Kuchta J, Sobottke R, Eysel P, Simons P. Two-year results of interspinous spacer (X-Stop) implantation in 175 patients with neurologic intermittent claudication due to lumbar spinal stenosis. Eur Spine J. 2009;18(6):823–9.
- 45. Barbagallo GM, Olindo G, Corbino L, Albanese V. Analysis of complications in patients treated with the X-stop interspinous process decompression system: proposal for a novel anatomic scoring system for patient selection and review of the literature. Neurosurgery. 2009;65(1):111–9; discussion 9–20.
- 46. Moojen WA, Arts MP, Jacobs WC, et al. Interspinous process device versus standard conventional surgical decompression for lumbar spinal stenosis: randomized controlled trial. BMJ. 2013;347:f6415.
- 47. Galarza M, Fabrizi AP, Maina R, Gazzeri R, Martinez-Lage JF. Degenerative lumbar spinal stenosis with neurogenic intermittent claudication and treatment with the Aperius PercLID system: a preliminary report. Neurosurg Focus. 2010;28(6):E3.
- 48. Nardi P, Cabezas D, Rea G, Pettorini BL. Aperius PercLID stand alone interspinous system for the

treatment of degenerative lumbar stenosis: experience on 152 cases. J Spinal Disord Tech. 2010;23(3):203-7.

- 49. Van Meirhaeghe J, Fransen P, Morelli D, et al. Clinical evaluation of the preliminary safety and effectiveness of a minimally invasive interspinous process device APERIUS(®) in degenerative lumbar spinal stenosis with symptomatic neurogenic intermittent claudication. Eur Spine J. 2012;21(12):2565–72.
- Postacchini R, Ferrari E, Cinotti G, Menchetti PP, Postacchini F. Aperius interspinous implant versus open surgical decompression in lumbar spinal stenosis. Spine J. 2011;11(10):933–9.
- Beyer F, Yagdiran A, Neu P, Kaulhausen T, Eysel P, Sobottke R. Percutaneous interspinous spacer versus open decompression: a 2-year follow-up of clinical outcome and quality of life. Eur Spine J. 2013; 22(9):2015–21.
- 52. Richter A, Halm HF, Hauck M, Quante M. 2-year follow-up after decompressive surgery with and without implantation of an interspinous device for lumbar spinal stenosis: a prospective controlled study. J Spinal Disord Tech. 2014;27:336–41.
- 53. Mariottini A, Pieri S, Giachi S, et al. Preliminary results of a soft novel lumbar intervertebral prosthesis (DIAM) in the degenerative spinal pathology. Acta Neurochir Suppl. 2005;92:129–31.
- Taylor J, Pupin P, Delajoux S, Palmer S. Device for intervertebral assisted motion: technique and initial results. Neurosurg Focus. 2007;22(1):E6.
- 55. Senegas J, Vital JM, Pointillart V, Mangione P. Clinical evaluation of a lumbar interspinous dynamic stabilization device (the Wallis system) with a 13-year mean follow-up. Neurosurg Rev. 2009;32(3):335–41; discussion 41–2.

**Part VIII** 

DDD and Spine Deformity and Sagittal Balance

# Degenerative Scoliosis: Surgical Treatment

Jeffrey L. Gum and Jacob M. Buchowski

# 46.1 Demographics and Natural History

Adult scoliosis can typically be divided into two types: idiopathic and degenerative (de novo) scoliosis. The former is usually a patient with a history of adolescent idiopathic scoliosis (AIS) that progresses into adulthood [1], whereas adult degenerative scoliosis (ADS) patients have no history of scoliosis and most commonly present in the sixth decade with spinal stenosis symptoms [2, 3]. Other symptoms often include worsening mechanical pain or radiculopathy. Structurally, the lumbar curves typically have an L2 or L3 apex and are associated with a distal fractional curve (L4-sacrum) but not a structural thoracic curve, although a compensatory thoracic curve can be present. ADS curves are thought to develop as the result of asymmetric disk degeneration, osteoporosis, and vertebral compression fractures [4]. Similar to AIS, curve prevalence in

ADS is inversely proportional to curve magnitude with an overall prevalence ranging from 1 to 10 % [5-8].

## 46.2 Preoperative Evaluation

A thorough clinical evaluation should start with a detailed history and physical examination. Inquiries regarding a previous history of scoliosis are helpful to exclude the possibility of a degenerative idiopathic deformity. Although pain is the most common presenting symptom (>90 %), other aspects of the chief complaint can be very helpful when formulating a workup and operative plan [9, 10]. Pain is evaluated in regard to onset, location, duration, characteristics, and aggravating/relieving factors. Mechanical or axial pain is more likely associated with radiographic parameters such as lateral or rotatory subluxation often in the setting of severe spondylitic changes (Fig. 46.1) or sagittal imbalance and therefore should be addressed in the operative plan. Delineation of radicular pain or leg pain is important because lateral recess stenosis requires a more generous decompression compared to that of central stenosis. Although the literature suggests that radicular symptoms typically originate from the apex of the major curve, our experience suggests that the concavity of the fractional curve is the more common location for symptomatic foraminal stenosis [11]. Neurogenic claudication

J.L. Gum, MD (⊠) Department of Orthopaedics, Norton Leatherman Spine Center, Louisville, KY, USA e-mail: jlgum001@gmail.com

J.M. Buchowski, MD, MS Department of Orthopaedic Surgery, Washington University in St. Louis, St. Louis, MO, USA e-mail: buchowskij@wudosis.wustl.edu

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_46

**Fig. 46.1** PA view of patient with degenerative scoliosis showing severe spondylitic changes from L1 to the sacrum with L2–L3 and L3–L4 being the worst. There is significant lateral listhesis at L4–L5 and obliquity at the L5–S1 disk space typical of the fractional curve. The patient complained of left lower extremity pain extending to the dorsum of her foot. Radicular symptoms from the concavity of either the major curve or fractional curve is typical with this type of deformity

pain from central spinal stenosis in ASD (vs purely degenerative lumbar spinal stenosis) is typically not relieved by forward posture, unless the patient sits with his/her trunk supported by the arms [2]. Again, this validation would alter the construct to make sure areas requiring significant decompression were included in the instrumented fusion levels, whereas it may be possible to avoid instrumented fusion on levels with minimal decompression.

It is important to inquire about increasing clinical deformity. Rapidly progressing deformity can be a sign of neurogenic scoliosis, although rare in this adult population. This type of scoliosis results from a central nervous system condition with altered signaling to the spinal musculature and merits complete neural axis imaging [11]. Besides timing of deformity, patients often relate increasing deformity as an increasing rib or paraspinal hump, decreased height, or even the feeling of falling forward as the day progresses. The rotational deformity should be evaluated and documented with a scoliometer at each visit along with height recording. Any shoulder or pelvic asymmetry should be noted. Several subtleties on physical exam can

help illuminate increasing sagittal or coronal imbalance. As the day progresses, these patients have to utilize more and more compensatory mechanisms to keep them upright with minimal energy expenditure and may complain of spinal extensor fatigue. The pelvis retroverts and hips extend while the knees flex. Hyperextension of the neck along with shoulder extension can all be seen in patients with a sagittal imbalance, as they are trying to center their head over their pelvis. These exam findings are reflected on longcassette 36" standing radiographs as well. Overall, it is very important to pay attention to the sagittal profile when deciding which surgery and instrumentation construct as extending the upper instrumented vertebra (UIV) more proximally is likely necessary for significant sagittal imbalance.

Adult deformity patients pose a more challenging medical scenario compared to adolescent patients. And the role of comorbidities has been well documented as an important determinant of postoperative clinical outcome improvement [12]. Surgical interventions, especially long instrumented fusions, tend to be lengthy and maximally invasive and require diligent presurgical screening. Cardiopulmonary status and general medical condition should be evaluated and approved by the patients' general medical physician and/or cardiologist. Nutritional status assessment and bone quality analysis are important for the treatment algorithm as well [13]. A preoperative dual-energy x-ray absorptiometry (DEXA) scan is helpful. If patients are osteoporotic and have yet to be treated, one should consider provisional treatment.

Full-length (36-in.) standing anterior-posterior and lateral radiographs are required for preoperative planning. The addition of "spot" films (lumbosacral or thoracolumbar junction) in the correct profile, as rotation can be significant, can be useful for pedicle assessment. Supine long-cassette radiographs (gravity removed) are utilized to assess flexibility. Push-prone or bending radiographs can aid in the assessment of flexibility although these are typically rigid deformities. Cobb angle measurements, coronal and sagittal balance (sagittal- and coronal-vertical axis), and spinopelvic parameters (lumbar lordosis, pelvic incidence, pelvic tilt) are all determined. Advanced imaging with computed tomography (CT) scan (± myelogram) and/or MRI can be useful to help evaluate the degree of central, lateral recess, or foraminal stenosis. CT myelograms in this population are particularly useful as a significant percentage of these patients will have a contraindication (pacemaker, stent, coil, implanted stimulator) to an MRI. Additionally, a CT myelogram allows for the best appraisal of the bony architecture.

# 46.3 Surgical Options and Navigating a Surgical Algorithm

Patients that fail nonoperative modalities and have symptoms that correlate with radiographic findings should be considered for operative treatment.Specific radiographic parameters that tend to correlate with postoperative clinical improvement include lumbar curves >30-40,  $\geq 6$  mm of lateral listhesis, >3 mm increase in listhesis with flexion/extension, L3 and L4 endplate angulations, thoracolumbar kyphosis, and progressive curves (>10°) [1, 2, 8]. Lenke and Silva et al. describe six levels of operative treatment: I, decompression alone; II, decompression and limited instrumented posterior spinal fusion; III, decompression and lumbar curve instrumented fusion; IV, decompression with anterior and posterior spinal instrumented fusion; V, thoracic instrumentation and fusion extension; and VI, utilization of osteotomies for deformity correction [2].

Level I treatment includes decompression alone. This treatment is typically utilized best in the setting of neurogenic claudication secondary to central stenosis requiring only a limited decompression. The potential for deformity progression has been well documented and must be considered. To minimize this risk, patients with radiographic evidence of deformity stability should be selected. The presence of anterior osteophytes, collapsed disk space, and no more than 2 mm of subluxation can be helpful signs of inherent stability. Additionally, the curve should be  $<30^{\circ}$  without hyperkyphosis and/or sagittal or coronal imbalance. Lastly, these patients should have minimal to no mechanical back pain or deformity complaints as these symptoms will not improve and likely worsen [2].

Level II treatment is level I + limited instrumentation involving only the area of decompression. Patients requiring a more extensive decompression (lateral recess) or evidence of instability prior to decompression are good candidates for this option. Radiographic clues are similar to above, and patients without anterior osteophytes, a well-preserved disk space, and/or more than 2 mm of subluxation may be better served with the addition of instrumented fusion. The curve should still be  $<30^{\circ}$  without hyperkyphosis and/or sagittal or coronal imbalance as only a limited fusion outside of these parameters could promote deformity progression or accelerated adjacent segment breakdown [2, 11]. Daubs et al. present a series of 55 consecutive patients with ADS treated either with decompression alone (level I, 16 patients) or decompression with limited instrumented fusion (level II, 39 patients). Although the level II patients were younger and had larger curves, at a minimum 2-year follow-up, 62 % of the level I versus 82 % of the level II patients reported a good-excellent result (p < 0.05). At 5-year follow-up, 75 % of the decompressiononly patients had recurrent stenosis, while 36 % of the decompression/limited fusion patients had adjacent level stenosis (p=0.008) [14].

Level III treatment encompasses fusion of the entire lumbar curve and any necessary decompressions. As to not stop the upper instrumented vertebra (UIV) at a physiologic apex, this level of treatment typically involves T10 or T11 instrumentation down to the sacrum/pelvis. Clinically, these patients commonly complain of axial or mechanical pain associated with their deformity. The curves are typically >45°, having >2 mm of subluxation, lack anterior osteophytes, but still have reasonable balance in both coronal and sagittal planes (Fig. 46.1) [2]. Although there is no clear literature that ascertains a critical construct length that pelvis fixation should definitely be included (versus just sacrum), it is common



**Fig. 46.2** Level III treatment example. PA (**a**) and lateral (**b**) of patient with lumbar degenerative scoliosis with acceptable coronal and sagittal balance. The patient underwent a posterior spinal fusion from T10 to the

sacrum/pelvis with S2 alar-iliac screws placed (c, d). Decompression was performed from L4 to L5 and L5 to S1 for central stenosis

practice to consider pelvic fixation at our center when the UIV is L2 or proximal [15–19]. Additionally, anterior column support with interbody fixation (via TLIF) should be considered with this surgical option. The combination of anterior column support and sacropelvic fixation may reduce pseudarthrosis rates, screw pullout, and instrumentation failure [20].

Level IV treatment includes both anterior and posterior fusion of the lumbar spine. Traditionally, anterior spinal fusion has played a significant role in the correction of lumbar hyperkyphosis and sagittal imbalance. The load sharing and increased fusion surface area are obvious biomechanical advantages that will help decrease pseudarthrosis rates and instrumentation failure [2]. Additionally, this adds indirect decompression via foraminal distraction. At our center, the refinement of posterior-only techniques has significantly decreased the need for formal anterior exposures and fusions. The current trend and increased utilization of lateral-based procedures will likely further decrease the use of formal anterior lumbar fusion procedures.

Level V treatment includes the extension of the instrumented fusion into the upper thoracic region. This is needed in patients with thoracic hyperkyphosis, thoracic decompensation, thoracolumbar junctional kyphosis, and/or sagittal or coronal imbalance (Fig. 46.2) [2]. O'Shaughnessy et al. evaluated a series of 58 patients from a single center with an average 3-year follow-up and compared outcomes of patients with lower thoracic (LT) versus upper thoracic (UT) primary instrumented fusions in patients with adult scoliosis. The UT group had a greater preoperative thoracic kyphosis and coronal Cobb and increased blood loss. The UT group experienced more perioperative complications (30 % vs 16 %), a higher pseudarthrosis rate (20 % vs 5 %), and a higher prevalence of revision surgery (20 % vs 11 %). The LT cohort developed more proximal junctional kyphosis (18 % vs 10 %) but rarely requiring revision surgery [21].



**Fig. 46.3** Level V treatment example. A 62-year-old female with complaints of increasing deformity and left lower extremity pain. PA (**a**) shows a large,  $102^{\circ}$ , lumbar curve with a 58° fractional curve. Preoperative coronal and sagittal balance (**c**) are acceptable, which is different than most patients requiring level VI treatment. The

patient underwent T2–sacrum/ilium with conventional iliac fixation and three-level TLIFs and five posterior column osteotomies (PCOs) with significant coronal and sagittal plane correction as seen in both the postoperative coronal (**b**) and sagittal (**d**) radiographs

Level VI treatment involves the use of an osteotomy. Patients whose deformity corrects >30 % with supine, push-prone, or bending films do not require osteotomies as these are considered flexible deformities [2]. Not all rigid deformities require an osteotomy as well, especially if well balanced. Again, the preoperative evaluation is critical in this group of patients as long instrumented fusions with 3-column osteotomies are a great physiologic stress on the patients. The majority of patients that require level VI treatment do so secondary to sagittal imbalance (Fig. 46.3) [2]. Again, attention to the sagittal profile is critical. Global balance, segmental or regional balance, and spinopelvic parameters such as pelvic incidence (PI), pelvic tilt (PT), and lumbar lordosis (LL) are all components that contribute to the overall sagittal profile. Smith-Peterson or posterior column osteotomies (PCO) can be utilized at sequential levels for segmental imbalance and potentially avoid a larger, more complex osteotomy. To maximize the corrective affect of a PCO, it is important for the disk space to still have some mobility allowing extension. The pedicle subtraction osteotomy is the next most corrective osteotomy option. Typically, it provides ~30° of lordotic correction without the need for anterior releases or structural grafting making it desirable for global imbalance. It is useful in osteoporotic patients and patients with suboptimal bone healing potential such as diabetics or smokers [2]. The bone-to-bone contact created during osteotomy closure provides relatively high fusion rates. For concurrent coronal imbalance, an asymmetric PSO can be utilized to achieve biplanar correction. The most corrective potential is achieved via a vertebral column resection (VCR) but is rarely needed for this patient population.

### 46.4 Fusion Level Selection

Upper instrumented vertebra (UIV) or proximal fusion level should begin at a neutral and stable vertebra, established to be the center sacral vertical line (CSVL) [22, 23]. The thoracic physiologic apex should be avoided as well [24]. Additionally, the fusion should not stop at a level with signs of radiographic instability such as rotatory subluxation or listhesis. One should consider all the above for determining the lower instrumented vertebra (LIV) as well. There is debate about the sparing of only one motion segment by stopping at L5. This is possible but the integrity of the L5–S1 disk should be considered, and if there is any obliquity, which is typical of the fractional curves >15°, the sacrum and/or pelvis should be included [2].

As mentioned several times previously in the chapter, there are numerous factors, from spondylosis to medical comorbidities, to consider when evaluating health-related quality-of-life outcomes of patients who have surgery for ADS. The largest study comparing level I, level II, and level III treatment options was reported by Transfeldt et al. and included 85 patients retrospectively studied with a minimum 2-year follow-up [25]. They found that the complication rate was highest in level III treatment (56 %) and lowest in level I treatment (10 %) with Oswestry Disability Index only reaching a significant improvement in these two cohorts as well. Overall, the SF-36 scores had significant improvement for all cohorts and the satisfaction questionnaire showed the highest success to be in the level III treatment group with an average radiographic improvement in Cobb from 39 to 19°. Their regression analysis echoed other concepts in deformity surgery, revealing that sacrum to curve apex fusions and positive postoperative sagittal imbalance were associated with poor outcomes. Their conclusion was that both good and poor results were seen with each of the three procedures confirming the variability in this patient population and the difficulty in stratifying treatment [25]. A systematic review by Liang et al. included 16 studies and 553 patients that had operative treatment for degenerative scoliosis. Obviously, a very wide spectrum regarding level of treatment but overall a mean improvment in ODI from 36.0 to 23.3 with a mean reduction in curve magnitude of 48.5 %. The overall incidence of complications was 49.0 % with a rate of revision surgery of 15.3 %. They concluded that despite a relatively high rate of complications, surgery is an effective and reasonable treatment providing significant functional improvement and deformity correction [26].

#### References

- Pritchett JW, Bortel DT. Degenerative symptomatic lumbar scoliosis. Spine (Phila Pa 1976). 1993; 18(6):700–3.
- Silva FE, Lenke LG. Adult degenerative scoliosis: evaluation and management. Neurosurg Focus. 2010;28(3):E1.
- Benner B, Ehni G. Degenerative lumbar scoliosis. Spine (Phila Pa 1976). 1979;4(6):548–52.
- Herkowitz HN, Kurz LT. Degenerative lumbar spondylolisthesis with spinal stenosis. A prospective study comparing decompression with decompression and intertransverse process arthrodesis. J Bone Joint Surg Am. 1991;73(6):802–8.
- Grubb SA, Lipscomb HJ, Coonrad RW. Degenerative adult onset scoliosis. Spine (Phila Pa 1976). 1988; 13(3):241–5.
- Kobayashi T, et al. A prospective study of de novo scoliosis in a community based cohort. Spine (Phila Pa 1976). 2006;31(2):178–82.
- Ploumis A, Transfledt EE, Denis F. Degenerative lumbar scoliosis associated with spinal stenosis. Spine J. 2007;7(4):428–36.
- Schwab FJ, et al. Adult scoliosis: a quantitative radiographic and clinical analysis. Spine (Phila Pa 1976). 2002;27(4):387–92.
- 9. Kostuik JP, Israel J, Hall JE. Scoliosis surgery in adults. Clin Orthop Relat Res. 1973;93:225–34.
- Winter RB, Lonstein JE, Denis F. Pain patterns in adult scoliosis. Orthop Clin N Am. 1988;19(2): 339–45.
- Birknes JK, et al. Adult degenerative scoliosis: a review. Neurosurgery. 2008;63(3 Suppl):94–103.
- Slover J, et al. The impact of comorbidities on the change in short-form 36 and oswestry scores following lumbar spine surgery. Spine (Phila Pa 1976). 2006;31(17):1974–80.
- Schoenfeld AJ, et al. Patient factors, comorbidities, and surgical characteristics that increase mortality and complication risk after spinal arthrodesis: a prognostic study based on 5,887 patients. Spine J. 2013; 13(10):1171–9.
- Daubs MD, et al. Decompression alone versus decompression with limited fusion for treatment of degenerative lumbar scoliosis in the elderly patient. Evid Based Spine Care J. 2012;3(4):27–32.
- Bridwell KH, Lenke LG, Lewis SJ. Treatment of spinal stenosis and fixed sagittal imbalance. Clin Orthop Relat Res. 2001;384:35–44.
- Chen F, Shen JX, Qiu GX. Features of pelvic parameters in adolescent idiopathic scoliosis and their relationships with spinal sagittal parameters. Zhonghua Yi Xue Za Zhi. 2013;93(7):487–90.
- Kostuik JP. Treatment of scoliosis in the adult thoracolumbar spine with special reference to fusion to the sacrum. Orthop Clin N Am. 1988;19(2):371–81.
- Kuklo TR, et al. Minimum 2-year analysis of sacropelvic fixation and L5-S1 fusion using S1 and iliac screws. Spine (Phila Pa 1976). 2001;26(18):1976–83.

- Perra JH. Techniques of instrumentation in long fusions to the sacrum. Orthop Clin N Am. 1994; 25(2):287–99.
- Shen FH, et al. Pelvic fixation for adult scoliosis. Eur Spine J. 2013;22 Suppl 2:S265–75.
- O'Shaughnessy BA, et al. Does a long-fusion "T3-sacrum" portend a worse outcome than a shortfusion "T10-sacrum" in primary surgery for adult scoliosis? Spine (Phila Pa 1976). 2012;37(10):884–90.
- Bridwell KH. Normalization of the coronal and sagittal profile in idiopathic scoliosis: options of treatment. J Orthop Sci. 1998;3(2):125–34.
- 23. Bridwell KH. Selection of instrumentation and fusion levels for scoliosis: where to start and where to stop.

Invited submission from the joint section meeting on disorders of the spine and peripheral nerves. J Neurosurg Spine. 2004;1(1):1–8.

- Bradford DS, Tribus CB. Vertebral column resection for the treatment of rigid coronal decompensation. Spine (Phila Pa 1976). 1997;22(14):1590–9.
- Transfeldt EE, et al. Surgical outcomes of decompression, decompression with limited fusion, and decompression with full curve fusion for degenerative scoliosis with radiculopathy. Spine (Phila Pa 1976). 2010;35(20):1872–5.
- Liang CZ, et al. Surgery is an effective and reasonable treatment for degenerative scoliosis: a systematic review. J Int Med Res. 2012;40(2):399–405.

# The Importance of Sagittal Balance for the Treatment of Lumbar Degenerative Disk Disease

João Luiz Pinheiro-Franco and Pierre Roussouly

# 47.1 Introduction

Low back pain (LBP) has a lifetime prevalence of approximately 80 % [1]. A primary source of LBP, lumbar degenerative disk disease (DDD), is related to aging and may not be symptomatic. Disk degeneration is a natural phenomenon of the aging spine and the distinction between physiological and pathological DDD is not obvious. Lumbar pain in the setting of DDD may be the result of complex and intricate underlying processes with multiple variables involved. DDD is deemed to be induced mechanically and mediated by biochemical responses, often concurrent with aging and probably influenced by genetic particularities [2].

Beyond the disk, important sources of LBP are the vertebral posterior elements (facet joints) and the musculoligamentary system. According to Roussouly et al., global lumbar hyperlordosis

J.L. Pinheiro-Franco, MD (⊠) Neurological and Spine Surgery, Samaritano Hospital, São Paulo, Brazil e-mail: jlpfranco@uol.com.br

P. Roussouly, MD
Chirurgie de la colonne vertébrale,
Croix Rouge Française, Centre Médico-Chirurgical de Réadaptation des Massues,
92 Rue Edmond Locard,
69322 Cedex 05 Lyon, France
e-mail: p.roussouly@cmcr-massues.com and focal lumbar hyperextension (two adjacent functional segments in hyperextension) may result in increased stress on facet joints and may cause lumbar facet pain [3]. Moreover, failure of both discal and facet joint structure may determine segmental instability and this may play a role in LBP.

More recently, iatrogenic sagittal unbalance of the spine has been credited as one of the principal reasons for chronic LBP after spinal fusion surgery. The balance of spine and pelvis in the sagittal plane may be involved in lumbar pain when lordosis requirements are not respected.

Sagittal balance and bipedalism are mutually linked. The verticality of human bipedalism is obtained by a specific combination of shape and orientation of both spine and pelvis. Bipedalism specificities, as pelvis verticality, lumbar lordosis, and C7 plumb line alignment, are encountered in human population only. Spinopelvic sagittal alignment may be analyzed by a combination of spinopelvic and lumbar lordosis parameters and the global spinal balance may be evaluated by C7 plumb line.

Studies on asymptomatic volunteers have drawn the conclusion that there are conditions for an ideal sagittal balance. The first one is the maintenance of the C7 plumb line over the sacral plateau [4]. This is obtained by adaptation of the spinal curvatures, mainly lumbar lordosis, to the different shapes of pelvis. There is an interaction between shape and position of the pelvis with

47

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_47

shape and position of the spine curvatures in order to maintain global balance (C7 plumb line over the sacral plateau).

Four types of spinal curvature settings were described according to the pelvic shape parameter named pelvic incidence [5, 6]. The four different shapes of lordosis may comprehend and represent the overall distribution of mechanical stresses of loading on the spine according to the pelvic spatial orientation. With aging, mechanical stresses may induce specific degenerative patterns regarding each spinopelvic organization. The degeneration of the spine in one or several functional segment units may change the global spinal alignment and even induce sagittal plane compensatory mechanisms in spine or pelvis in order to maintain C7 plumb line in a balanced position. When these compensations are overpassed, C7 plumb line is displaced forward, first forward the sacral plateau and then forward the femoral heads. Even if these compensatory mechanisms allow the patient to maintain a standing position, the situation may be uncomfortable and difficult to maintain for a long time. The principal aim of a surgical treatment for that condition is the restoration of sagittal balance. This implies the repositioning of C7 over the sacral plateau, by adjustment of the spinal curvatures, with decreasing of the pelvis compensation. In surgery, it is mandatory to respect the spinal alignment, according to the pelvic shape given by the pelvic incidence.

The sagittal spinopelvic alignment in spinal disorders has been studied in developmental spondylolisthesis [7, 8], degenerative spondylolisthesis [9], adolescent idiopathic scoliosis [10], and adult spinal deformity [11] and in the asymptomatic population [5, 12]. Some authors also analyzed spinopelvic balance in LBP and lumbar DDD [13–16]. However, the relationship between sagittal alignment and pathology is not well understood.

There is no ideal sagittal balance but a unique physiological sagittal balance for a given individual. In asymptomatic individuals, there is a great variability in sagittal spinopelvic alignment with large standard deviations. Therefore, although statistically significant differences may be found, differences in sagittal spinopelvic alignment in chronic LBP are only small, and clinically, multiple factors other than spinopelvic alignment contribute to LBP and/or lumbar disk degeneration. However, there seems to be trends to specific patterns of degeneration according to specific spinopelvic morphotypes.

For decades, the surgical treatment of spinal pathologies has focused on focal problems (e.g., the neurodecompression in a lumbar disk herniation) with no concern to regional lumbar or global spine sagittal balance. In the last 20 years, however, there has been a marked interest in the spinal sagittal balance. The maintenance of the balance should be one of the pillars of the surgical treatment for spinal diseases.

## 47.2 Bipedalism and Lumbar Lordosis

Over hundreds of thousands of years, man has been transformed. Not just we have been, we are in transformation. The combination of multiple factors made the primitive ancestor to evolve from a quadruped position to a standing one, the bipedalism. For this to elapse over millennia, there were obviously important anatomical modifications in the structure. From the skull base to the lumbar spine, striking changes occurred. The pelvis, previously a linking organ between trunk and lower members, became the pedestal over which the spine must equilibrate. Anatomical and physiological adjustments in the musculoligamentary system occurred to adapt to a new posture.

Human pelvis changed: it is the only one that is retroverted, the sacral plateau being backward the bifemoral heads' axis (balanced spine). In mammals, the sacral plateau is always anterior to the femoral heads' axis. This human disposition is the only mechanical possibility to allow the erect position, with the trunk over the pelvis. Therefore, evolution made possible the *extension* of the lumbar spine, named lumbar lordosis, unique among mammals and all the vertebrates. Bipedalism allows static and dynamic stability. The sacrum, defined by Doubousset "pelvic vertebra," is the keystone in the pelvic-spine structural joint, joining the mobile spine and the hip. Its position in the space is essential to the local (lumbar) and global balance of the spine. If all primates can move in a bipedal way, man is the only primate capable of doing it for a long time and long distances [17]. This is possible due to the sagittal balance of the spine over the hips. Apes have a column marked by a global kyphosis that prevents them from maintaining a stable, steady upright stance. Their hip is high and narrow in the anterior-posterior plane. Using the upper limbs supported in pronation, they can somehow stand for a certain period and even walk. However, their march without support is difficult and limited. Studies have shown that the orangutan is very often bipedal, with his knee in extension (a feature associated with human bipedalism). The kangaroo, like the extinct Tyrannosaurus, walks (or walked) on two legs but have (or had) a stabilizing tail. The bipedalism of birds, as well, in no way resembles the human bipedalism. Hence, human verticalization, lumbar lordosis acquisition, and the obtention of spinal sagittal balance are intrinsically linked.

# 47.3 Sagittal Balance and the Pelvic Parameters of the Spine

The sagittal balance of the spine can be defined as the harmonious balance of the trunk over the pelvis with a minimum expenditure of muscular energy in order to place the weight-bearing axis in a physiological position [18]. The spinopelvic alignment may be assessed by three groups of parameters: pelvic parameters, spinal curvatures, and C7 plumb line positioning.

### 47.3.1 Pelvic Parameters

The most commonly used pelvic parameters were described by Duval-Beaupère [19]. *Pelvic incidence* (PI) is a shape parameter that relates the sacral plateau with the bifemoral axis. Pelvis tilt (PT) is a positional parameter that gives the



**Fig. 47.1** Pelvic parameters: pelvic incidence (*PI*), sacral slope (*SS*), and pelvic tilt (*PT*) (From Barrey [18]; with permission)

rotational position of the pelvis around the femoral heads. Sacral slope (SS) is a positional parameter that gives the angle between the sacral plateau direction and the horizontal.

*PI* angle consists of the angle between two lines, namely, the line drawn between the midpoint of bicoxofemoral axis (midpoint of the line joining the center of both femoral heads) and the middle of the sacral plateau and the line perpendicular to the middle of the sacral plateau (Fig. 47.1). PI is a morphological immutable angle in adulthood, being specific to each individual [18]. PI is anatomically determined by the shape of the sacrum: pelves with high PI have generally a short sacrum tending to horizontalization (sacral plate is far below and behind the top of iliac wings in lateral view) while those pelves with low PI have a long vertical sacrum in lateral view (sacral plate close to the top of iliac wings) (Fig.47.9). Definitive value of PI is acquired when final skeletal growth is achieved. There is some controversy concerning a possible variation of PI (increase) with aging by sacroiliac joint degeneration.

The second pelvic parameter is the sacral slope angle (SS), which is the angle formed by two lines: a horizontal line and the line passing through the sacral endplate (Fig. 47.1). The steeper the sacral plateau, the greater the SS. Thus, the more inclined anteriorly the sacrum (anteverted pelvis), the greater the SS; the more verticalized the sacrum (retroverted pelvis), the lower the SS. The longitudinal analysis of the flattening or the slope of the sacral plateau in a same patient may reflect a compensatory mechanism for an imbalanced spine. When there is a compensatory retroversion, the sacral plateau flattens and SS decreases. The horizontalization of SS may affect the forces acting on disks and facets.

The third pelvic parameter is the *pelvic tilt* angle (PT) (Fig. 47.1), the angle between two lines, namely, a vertical straight line and the line drawn from the midpoint between the two femoral heads' centers to the midpoint of the sacral endplate. The PT sets the oscillating motion of the hips around the bicoxofemoral axis, the femoral heads. This is possible due to the spherical shape of coxofemorals. When the pelvis swings forward, anteversion, the PT is lower; when it swings backward, pelvis retroversion, the PT is higher. The retroversion of the pelvis, the first compensatory mechanism of sagittal imbalance, aims to bring back the spine to the balanced situation, with C7 plumb line behind the posterior part of the sacrum. Thus, PT and SS are positional parameters, variables [12].

Legaye et al. [20, 21] and Duval-Beaupère et al. [19] noted important correlations between pelvic parameters, linked by the geometrical relation PI=PT+SS [19, 20, 21]. The algebraic equation is: A=x+y, where A is constant and x and y are variable. Even if many studies have described a strong correlation between PI and SS (R: 0.8, P<0.001) and PI and PT (R: 0.65 p<0.001), it is mathematically impossible to write a linear relation between PI and SS or PI and PT.

If A=x+y, then x=kA+b or y=k'A+c; but this is not totally true to PI=PT+SS. A direct extraction of PT or SS from PI is a statistical approximation but cannot be written under a mathematical equation. Interpretation of PI=PT+SS could be done in two means:

- For a same person, PI is constant: when PT increases, SS decreases (retroverted pelvis=smaller SS); when PT decreases, SS increases (anteverted pelvis=higher SS).
- For different people, a marked pelvis retroversion (high PT) is possible with higher PI only. Even with a small PI, a high SS is possible when the pelvis is very anteverted (small PT or negative PT).

An increase of the PT is the earliest compensatory mechanism resulting from the loss of lordosis of natural aging or may be the result of lack of lordosis obtention after fusion surgery (flat back syndrome).

## 47.4 Spinal Parameters

# 47.4.1 Spinal Segmentation and Length of Lumbar Lordosis (LL)

To assess the sagittal balance, the authors took into account the part of the spine between C7 and T1 and the sacral plateau, excluding the cervical spine. While in animals the thoracic and lumbar spine have only one kyphotic curvature, in humans, two successive curves are described: thoracic kyphosis and lumbar lordosis. Anatomically, the first, corresponding to the thoracic vertebrae, is limited by C7-T1 disk above and T12-L1 disk below, and the second by T12-L1 proximally and S1 plateau distally (Fig. 47.2a) [12, 19, 22]. However, if we consider that lordosis is composed by vertebrae in the part of the spine where the curve is in extension, the length of extension in lumbar area may be variable (Fig. 47.2b). This concept has brought Berthonnaud et al. to describe a segmentation model of the spine with two curves between vertebrae T1 and S1: thoracic kyphosis and lumbar lordosis, limited by an inflection point where kyphosis turns into lordosis [5]. Thus, lordosis may be defined as the segment of the spine



**Fig. 47.2** (a) Subdivision of the sagittal spinal curvatures. Cervical spine goes from C1 to C7. Thoracolumbar spine is limited by C7–T1 and the sacral plateau. Inflection point is the place where the lordosis curve turns into kyphosis. This is a variable limit between lordosis and

between the sacral endplate and the inflection point, without any reference to a specific anatomical landmark. The inflection point may be at the level of T12–L1 disk, a little bit higher or a little bit lower. In Roussouly type 1 lordosis, a thoracic vertebra, T12, may be at least in part in the lordotic spine, as it has been demonstrated in previous studies [12, 23]. Thus, this concept may affect the planning of an arthrodesis. The same remark may be done when using Cobb measurement. As for frontal scoliosis measurements, sagittal Cobb method has to address the most tilted vertebrae to measure a curve angle.

Stagnara pointed out the relation between LL and SS [24], with a description of static and dynamic spinal curvatures. Statistically, a strong correlation between LL and SS was well established [20] and Duval-Beaupère has extracted a statistical equation: LL = -5, 4 - 1,06SS. More

kyphosis (From Roussouly and Pinheiro-Franco [17]; with permission). (b) There is a great variability of the length of lordosis. Note shorter and marked lordosis on the *left* (lower pelvic incidence) and long and more harmonious lordosis on the *right* (higher pelvic incidence)

recently, Roussouly et al. [6] proposed a classification of LL according to SS (Fig. 47.3). This classification was fundamented by the lordosis concept from the aforementioned segmentation model [5], being lordosis defined from the inflection point to the sacral endplate. LL curvature is divided into two arcs of circle tangent on the horizontal line passing on the apex of LL.

# 47.5 Lumbar Lordosis Geometrical Analysis [17]: Basis for Roussouly Classification

Lumbar lordosis is not homogeneous throughout its extension. The angle of lordosis is not evenly distributed from the sacrum endplate to the proximal limit in the inflection point where the spine



bends into kyphosis. The contribution of each functional segment unit to the lordosis increases progressively from L1 to L5. Two-thirds of the total lordosis is depicted in the lower lumbar spine L4-S1 [18]. A meta-analysis demonstrated that 66 % of LL concentrates in L4-L5 and L5-S1 (n=552) [25]. Geometrical constructions of the spine have been proposed to describe LL: arc of circle [5] and quadrant of an ellipse [26]. The ellipsoid design is very realistic but difficult to use in everyday clinical practice [18]. Berthonnaud et al. [5] described a mathematical construction of lumbar lordosis. These authors proposed that lumbar lordosis could be measured using the Cobb method between the upper plateau of the sacrum and the inflection point where lordosis turns into kyphosis. The point of tangency of a vertical line with the anteriormost part of the convex side of lordosis with the vertical is the *apex of lordosis*. A horizontal line is traced from the apex of lordosis and defines two arches (Fig. 47.4): a lower arch (from the apex horizontal line to the sacral plate line) and an upper arch (from the apex horizontal line to the inflection point into kyphosis). The angle of the lower arch and the SS are equal and vary together by definition [6] (Fig. 47.4). Roussouly et al. demonstrated that the mean value

of the upper arch angle was  $20^{\circ}$  and was stable whatever the value of SS was [6]. This explains why the lordosis is linked to the value of the lower arch (and therefore to SS). This has important implications in lumbar spine fusions.

## 47.6 The Roussouly Classification

Based on the geometrical model aforementioned, Roussouly et al. developed a classification of spinopelvic morphotypes translated by four types of lumbar lordosis according to the angle of SS and PI. The variation of lordosis was defined by the extent of the lower arch, and thus the SS. The separation of the values of the SS, according to Gauss curve, allowed the SS values to be classified into three groups depending on their values: SS <35°, 35° < SS <45°, SS >45° (Fig. 47.3). The values 35° and 45° were previously established [6, 18].

When the SS is small (<35°), two spinopelvic morphotypes of lordosis could be demonstrated:

• Type 1: short acute lordosis (concentrated in L4–L5 and L5–S1). The thoracic kyphosis extends slightly into the lumbar spine and



**Fig. 47.4** (a) Schematic representation of arc of a circle model for type 1 lordosis. Note that the lower arch of the lumbar lordosis (=SS) is too small and the apex of lordosis is very low. There is a backward displacement of the top of lordosis and a very positive weighbridge angle. (b) Schematic representation of arc of a circle model for type 2 lordosis. The lower arch (SS) is slightly larger than in type 1, and the lordosis is still very small. The angle of

weighbridge is from positive to zero. (c) Schematic representation of arc of a circle model for type 3 lordosis. Lordosis is divided between the two arches and the apex is at the center of L4. (d) Schematic representation of arc of a circle model for type 4 lordosis. The inflection point may be at the lower thoracic region. The lordosis angle increases as the number of vertebrae increases in the lordosis. The apex rises above L4 (From Barrey [18]; with permission)

therefore, the thoracolumbar (TL) junction is under the classic T12–L1 disk. The lower arch of the lumbar lordosis (=SS) is too small, and the apex of lordosis is very low. There is a backward displacement of the top of lordosis and a very positive weighbridge angle. As lordosis is short, the kyphosis is long and extends a little beyond the thoracolumbar spine (Fig. 47.4a);

• Type 2: corresponds to "flat back." The lumbar spine is usually quite flat. The lower arch (SS) is slightly larger than in type 1, and the lordosis is still very small. The angle of weighbridge is from positive to zero (Fig. 47.4b).

When SS has an average value (between 35 and  $45^{\circ}$ ):

• Type 3: the best balanced spine. The inflection point is at the thoracolumbar junction. Lordosis is divided between the two arches and the apex is at the center of L4. There are usually four to five vertebrae in the curvature. The angle of weighbridge is from positive to zero (Fig. 47.4c).

When the SS is high  $(>45^{\circ})$ :

• Type 4: The inflection point may be at the lower thoracic region. The lordosis angle increases as the number of vertebrae increases in the lordosis. The apex rises above L4. The toggle angle is generally from zero to negative. As lordosis is extended, the thoracic kyphosis is shortened (Fig. 47.4d).

### 47.6.1 LL and PI Relations

In asymptomatic volunteers, there is a strong correlation between LL and PI but less than between LL and SS. The authors have seen that due to the relation PI=PT+SS, there is a strong correlation between PI and SS but without concluding to a linear relation. Generally for a small PI, SS is smaller and this situation is represented mainly by types 1 or 2 lordosis. On contrary, for high-grade PI, SS is generally higher, with more type 3 or 4 lordosis. Occasionally, the authors have seen that PT may be small (<10°) or even negative in very anteverted pelvis. This situation may allow SS >40° even with a small PI and conducts to type 3 or 4 lordosis with a small PI. The value of PI may bring to a tendency of LL morphotypes, but to extract LL value from PI is an inexact extrapolation.

## 47.6.2 LL and Thoracic Kyphosis Angle (TK)

LL and TK are linked by their length and angles. Spinal parameters LL and TK are interdependent (Fig. 47.2). Jackson and McManus [13] found a significant correlation between LL and TK [27]. One change in one segment induces a change in the reciprocal segment according to the flexibility of the spine. If the TK increases, LL increases in order to maintain C7 in the balanced position. Reversely, when LL decreases, TK decreases, flattening the back.

- Following the inflection point, the distribution of both LL and TK is variable. In case of short LL, TK may reach the thoracolumbar area. With long LL, TK may be shortened in the more proximal thoracic area. Sometimes, both LL and TK curves are separated by a straight segment of a variable number of vertebrae (Fig. 47.2). The importance of this disposition is not well validated but requires further study.
- If we consider the tangent arcs of circle segmentation, there is a direct relation between the upper arc of LL and the lower arc of TK. An increased TK may induce an increased LL. This is more relevant in type 1 (too small lower arch of LL) where the total LL depends mostly on the upper arch of LL that has to compensate the higher lower arc of TK in the thoracolumbar area.

## 47.7 Global Balance of the Spine

## 47.7.1 The Global Balance [17]

To the analysis of the global balance of the spine, several points have been proposed. One of them, the external ear conducts, is useful to indicate the head positioning. The T9 tilt was described by Duval-Beaupère et al. as an indicator of the spine balance at the body mass center level [19, 27]. Historically, a vertical [4] plumb line originating in the center of the C7 vertebral body was deemed to be in approximately the same place in the sagittal plane as a vertical line passing through the patient's center of gravity (Fig. 47.5). Easy to be read on sagittal X-rays, C7 plumb line is the most commonly used index of the global balance. Kuntz et al. [28] noted in a review of the literature that global parameter as C7 plumb line was a stable, reliable index of the sagittal balance,



**Fig. 47.5** Mechanism of compensation of a progressive kyphosis. (a) Balanced spine -C7 PL over the sacral endplate. (b) With progressive loss of lordosis, pelvis retroversion permits maintaining C7 PL behind the femoral

heads. (c) Severe unbalance, retroversion reaches its limits, knees flexed in attempt to "put backwards C7 PL"; note the Pelvifemoral angle (PFA) (From Roussouly and Pinheiro-Franco [3]; with permission)

being maintained in narrow ranges for alignment of the spine over the pelvis and femoral heads. This parameter has been historically quantified by measuring the position of a vertical line originating in the center of the C7 vertebral body with respect to the posterior superior corner of S1.

The center of the vertebral body of C7 may be considered as the upper limit of the thoracic spine and, indeed, of the whole spine below the cervical area. Before recent radiograph quality advances, C7 was considered easier to be identified rather than T1 that was hidden by the shoulder superposition.

The *positioning of C7* has been widely studied and its implication in sagittal balance well established. *Three ways of evaluation* are possible:

• *Distance measurement*: sagittal vertical axis (SVA) is the distance between the posterior edge of the sacral plateau and the C7 plumb line projection. Schwab et al. [29] considered that a value superior to 5 cm is an evidence of

imbalance. Radiograph distance measurement is controversial. It is necessary to have a precise calibration of the radiographies as comparison between the radiographies may lead to potential errors. On another way, if SVA >5 cm is certainly an imbalanced situation, SVA <5 cm does not depict obvious balance.

- The authors prefer to use angles or distance ratio, as these are not subject to calibration variations typical of radiographs.
- Angles measurement: using the line from the center of C7 to the center of the sacral plateau, two angles are designed:
  - C7 *tilt* is a positional parameter with the vertical direction. This angle is very stable in a normal population around 3–5° backward.
  - Spinosacral angle, SSA, is a shape parameter drawn with the sacral plateau line (angle between the line from the center of C7 to the center of the sacral plate and the sacral plate line itself). It is representative of the total kyphosis of the whole thoracolumbar spine (cervical spine excluded). Its very high correlation (R > 0.9) with SS in an asymptomatic population demonstrates the stability of positioning C7 plumb line over the sacrum in normal conditions.
- *Distance ratio*: described by Barrey [18], the ratio between the distance of the C7 plumb line and the posterior edge of the sacrum, and the distance of the C7 plumb line and center of femoral heads, allows the positioning of C7 plumb line relative to these two anatomical landmarks.

C7 plumb line must be related to an anatomical pelvic landmark as the center of the femoral heads (CFH) or the posterior point of the sacral plate (PP-S1). Usually, the horizontal distance between C7 PL and an anatomical landmark is calculated. It is not recommended to use distance to characterize this position. Barrey proposed a ratio between the horizontal distances from C7 PL to CFH and between CFH and PP-S1 (Fig. 47.6). This ratio provides an adimensional value concerning the position of C7 PL in relation to the vertical lines passing through CFH and through PP-S1 (Fig. 47.6):



**Fig. 47.6** In order to avoid errors with distance magnification C. Barrey proposed a ratio between horizontal distances from the center of the femoral heads (FH) and C7PL (R1) and the distance from FH and the posterior edge of the sacral plate (R2). R1/R2 is negative when C7 PL is forward FH, 0 < R1/R2 < 1 when C7PL is between FH and the sacrum, and R1/R2>1 when C7PL is behind the sacrum (From Roussouly and Pinheiro-Franco [17]; with permission)

- When C7 PL is behind the PP-S1 (ideal, balanced situation), the ratio value is superior to 1.
- When C7 PL is between CFH and PP-S1, the ratio is between 0 and 1 (balance is compromised).
- When C7 PL is forward CFH, the ratio value is negative (noncompensated severe imbalance).

#### 47.7.2 Method of Measurement [27]

The analysis of the sagittal balance of the spine requires radiographs in a standardized fashion. The Cobb method may be used. A lateral radiograph of the spine must be made with vertical 30- to 90-cm film with a constant distance from the radiographic source. The knees must be fully extended and the arms flexed forward to  $45^{\circ}$  and resting on supports. According to Vedantam et al. [30], positioning the arms at



**Fig. 47.7** Analysis of the sagittal balance using digitized imaging and software KEOPS (SMAIO, Lyon, France)

 $90^{\circ}$  rather than  $30^{\circ}$  resulted in a negative shift of the sagittal vertical axis. Marks et al. [31] affirmed that shoulder flexion of  $45^{\circ}$  is the best position to use when a lateral radiograph is made, in order to repeatedly measure the sagittal vertical axis.

The radiograph must show the femoral heads caudally and C7 cranially. Radiographs should be digitized, and all measurements may be performed by specialized softwares. The present authors use KEOPS (SMAIO, Lyon, France) (Fig. 47.7). The software permits rapid and precise measurement of all angular parameters on digitized radiographs. Other authors [32] demonstrated in a similar software that the intraobserver and interobserver reliability is very high and that the results obtained by the numerical process are similar to those obtained by manual measurement. Recently, the EOS system was developed allowing full-spine standing imaging with minimal irradiation.

• The KEOPS sagittal balance analyzer is a Web-based software aimed to measure key spinopelvic parameters on long-standing sagittal X-rays from femoral heads to center of ears; compare measured parameters to those of asymptomatic populations using specific cursors and simulate surgical correction on sagittal spinopelvic reconstruction. Accuracy and repeatability were validated with multiple users performing two times 30 measurements [33]. It was observed that the measurements were more reliable than manual measurements performed by the same users. The software proposes a global acquisition mode in which femoral heads, S1 plate, center of C7, and a b-spleen line passing through the middle of each vertebral body are acquired in less than 2 min. A synthesis allows users to compare patient spinopelvic morphology and positioning parameters with those of an asymptomatic population (709 asymptomatic volunteers were assessed). The simulation modules allows to measure effect of pelvic tilt decompensation on C7 balance and simulate PSO as well as Smith-Petersen osteotomies depending on the level and the amount of correction.

# 47.8 Distribution Values of Lumbar Lordosis and Pelvic Parameters

LL, PI, and SS vary considerably in the asymptomatic population. Lordosis is characterized by a relatively large variability in the population with physiological values ranging from 20° to 85° [18]. The variability of SS according to the great range of PI may explain the different spinopelvic morphologies of lordosis that Roussouly et al. classified in four main types.

There are studies in the literature providing physiological standards for pelvic and spinal parameters that describe spinal balance [22, 27]. A study in a cohort of 300 asymptomatic adult volunteers [27] had mean 60° for LL (30-89), mean 41° for SS (17-63), mean 13° for PT, and mean 55° for PI (33–82). A strong correlation was found between SS and PI, LL and SS, LL and PI, and PI and PT. Similar results were found by Boulay et al. [22] in a population of 149 asymptomatic individuals from their medical staff: mean PI = 53 (33-77), mean SS=41 (0,5–19), and mean PT=11,9(-2-30) [22]. The more SS is pronounced and the more important is lordosis, plus the weight-bearing axis is transferred to the posterior structures of the lumbar spine, the facet joints [3].

There is no ideal mean lordosis, as it may vary [13, 24, 25, 34, 35]. Length of LL is an important issue: for a given angle of LL, a short LL on three segments has not the same stress on the posterior structure than a long one. Another parameter to take into account is *the proper flexibility of the spine in extension*. Some spines are less extensible and the maximum of LL angle they may provide may be insufficient according to a high PI value. This limited maximal extension is reached when there is a contact between spinous processes on standing x-ray. There exists optimal lordosis for each individual. It is not possible to define values of the ideal sagittal balance, but indeed a physiological sagittal balance for a given person.

## 47.9 Sagittal Imbalance and Clinical Symptoms

There is a wide range in spinopelvic parameters and spinopelvic alignment in an asymptomatic population. Some authors demonstrated trends to patterns of degenerative changes for each spinopelvic morphotype [3]. LBP is strongly associated with lumbar DDD, a multifaceted problem, and for what will probably never be possible to describe one single factor that causes it.

Glassman [36] et al. reported a correlation between poor clinical outcome and positive sagittal balance. The same finding was verified by other studies. It was suggested that pelvis retroversion correlates with decreased health status scores [7, 24]. Schwab et al. affirmed that pelvic tilt is highly correlated with patient self-reported function (ODI, SF-12, and SRS) [37]. Patients with suboptimal sagittal balance have significantly lower total and self-image subscale SRS-24 outcome scores compared to patients with optimal sagittal balance. Appropriate sagittal plane alignment is an important factor (and sometimes the most important) in clinical outcomes and patient satisfaction [36, 38, 39]. Sagittal balance has been pointed as a quality-of-life indicator [40]. Patients with fixed sagittal imbalance tend to expend more energy in gait and standing leading to chronic pain after surgery. Grobler et al. [41] described the sagittal imbalance with forward thrusting of the trunk as a source of pain and fatigue. Gautier et al. found no correlation between the type of LL (assessed by proximal vs. distal LL) and LBP [42]. These authors did not find any difference in segmental and total LL nor PI when comparing 74 subjects with a history of LBP to 152 asymptomatic subjects.

Chaléat-Valayer et al. reported the largest database in the literature on the evaluation of sagittal spinopelvic alignment in chronic LBP in comparison with the asymptomatic adult population [43]. Prospective adult cohorts of 198 subjects with chronic LBP (LBP cohort) and 709 controls without spinal disorder (control cohort) were compared. Significant but small differences are found for various parameters of pelvic, lumbar, and thoracic segments in subjects with LBP. The type of LL is also distributed differently among subjects with LBP. A significantly increased proportion of subjects with LBP stand with abnormally small SS (<35) and PI associated with a long but small LL, when compared to controls. There was a greater proportion of chronic LBP in patients with low SS, low LL, and small PI, suggesting the relationship between this specific pattern and the presence of chronic LBP.

Jackson and McManus [13] observed decreased total LL associated with decreased distal and increased proximal LL as well as a more vertical sacrum in 100 patients with LBP compared to 100 matched controls. Similarly, Barrey et al. [14] showed similar PI, but decreased SS, LL, and thoracic kyphosis (TK), as well as increased PT in 57 patients with DDD or herniation prior to lumbosacral arthrodesis, compared to 154 controls. Rajnics et al. [15] also observed significant differences for SS, PT, and LL—but not PI or TK—in 50 patients presenting with LBP and disk herniation compared to 30 healthy subjects. Other studies also reported conflicting results concerning [16] LL in patients with LBP.

Type 1 LL is the least common type of LL found in asymptomatic adults and in patients with chronic low back pain [18, 43]. Roussouly stated that type 2 LL has a trend to develop LBP greater than other types of LL. Other authors obtained a significantly greater proportion of subjects with LBP presenting type 2 LL (37.4 %), as compared to controls (23.3%). At contrary, the proportion with type 3 LL was significantly decreased in the LBP cohort when compared to controls (38.9 % vs. 47.7 %) [43]. The proportion of subjects with either type 1 or type 4 LL was similar between the two cohorts. Therefore, the distribution of the types of LL was shifted from type 3 LL (and to a lesser extent type 4 LL) toward type 2 LL in subjects with LBP [43]. This finding confirmed that a greater proportion of subjects with chronic LBP tended to present a small SS (<35) associated with a long but small LL (flat back). Consequently in the LBP cohort, there was also a shift toward a greater proportion of subjects with abnormally small PI (usually associated with type 1 or type 2 LL). Because PI is a morphological parameter and is linked to the type of LL, it is possible that individuals with an abnormally small PI are at increased risk of LBP because of increased disk pressure/degeneration secondary to decreased LL and/or of suboptimal muscular/postural biomechanics needed to maintain adequate balance.

# 47.10 Sagittal Balance and Pathology

# 47.10.1 Mechanical Mechanisms of Compensation in Sagittal Balance Disturbance

In degenerative spine, with aging, the main pathological effect is a decrease of SSA that may be an expression of a true kyphosis or a loss of lordosis and sometimes both. Two main compensatory mechanisms are automatically involved in the restoration of the global balance (C7 plumb line over the sacrum):

- *Hyperextension of the adjacent vertebral functional segment unit* above or below the area of balance defect. This mechanism occurs when the spine is flexible. This local hyperextension may be painful by facet joints hyperpressure.
- ٠ Posterior tilt of the pelvis: increased PT and small PI or retroverted pelvis. This mechanism is possible when the spine is rigid either by degeneration or by iatrogenic unbalanced operated spine. Considering that PI and PT are linked by the relation PI=PT+SS, the theoretical maximum of PT compensatory increase for a very small SS (close to zero) has the same value of PI. It is easy to understand that the bigger is PI, the higher are the possibilities of pelvis retroversion. PT is limited by hip extension. The more the pelvis is retroverted, the more the hips are in extension. When hips extension possibilities are overpassed, the patient has to tilt the femoral shafts by flexing the knees. Even if this mechanism of compensation is efficient, it remains very uncomfortable for the patient, limiting ability of walking and even standing. On another hand, limitation of PT compensation in small PI pelvis (type 1 or 2 lordosis) may bring to an important global imbalance, with C7 plumb line forward the femoral heads even for a small loss of SSA. With aging and likely loss of lordosis, the hip may undergo retroversion, with increase of PT. Mac-Thiong et al. suggested that the nonpathological upper limit of PT would ideally be less than 50 % of PI [44]. A PT approaching the value of SS denotes imbalance. Likewise, the ideal values for SS should exceed 50 % of PI. In pathology, SS never reaches a negative position, less than 0. The minimal value of SS is 0, which is the horizontal sacral plate. This situation corresponds to the maximal retroversion possible.

If the PT is high (active compensation of imbalance), it is the PI that allows differentiating

a degenerated type 2 from a degenerated type 3 or 4. If PI and PT are high, the lordosis is a type 3 or 4 that has lost original lordosis. If PT is high but PI is low, this is a degenerated type 2 lordosis. A lumbosacral arthrodesis for type 3 or 4 should seek to restore lordosis, sometimes via an osteotomy, with the risk of perpetuating the important hip retroversion if the amount of lordosis is not sufficiently assured. If the PI is small, the need of restoring great lordosis has not the same importance. A lumbar fusion should restore the original type of lordosis, be it type 1, 2, 3, or 4.

Concerning normal or optimal value of PT, as PI=PT+SS, it would be reasonable to think that PT increases with PI and that the more the PI is high, the more the PT is high. This would be right without considering hips extension limits. It seems that the maximum of 25° for PT is the acceptable limit of hips extension. Over this limit, knees have to play a role, displacing the center of rotation of the set "pelvis-femur" on the femorotibial joint. This was first described by Mangione et al. [45] who described the pelvifemoral angle (PFA) (Fig. 47.5). PT is becoming an addition of PFA plus hips extension. More recently, Le Huec et al. [46] described the FBI angle considering that the true PT is the visible PT plus PFA and that the LL correction has to follow this hidden PT. In our opinion, the global positioning of the spine is following the true PT and when the LL reduction is obtained, the true PT is decreasing. If the PFA value is reached, hips extension limit is reached and the femur vertical position is now possible.

### 47.10.2 Gravity Stress on the Spine

To maintain the human body in erect position, the system has to "fight" against gravity. Of course, this stress increases in weight-bearing situation. To simplify, the authors take into account only the standing position. Different studies have shown that the gravity line in standing position is passing a little bit behind the center of the femoral heads and just forward the sacral plateau. The normal tendency of gravity is to flex the spine forward. To control the gravity force, the spinal system is built as a crane with the spine as a pylon and the spinal lumbar muscles acting as a counterforce against gravity [3]. As for a crane, the contact force (CF) applied on the basis of the pylon is the addition of both forces of gravity and lumbar muscles. To balance the crane, the moment of the gravity has to be equal to the lumbar muscles moment. Therefore, the more the gravity displaces forward, the more the muscle force increases and the more the CF increases. This is the situation in case of weight bearing where the charge's weight is coming in addition to the body gravity. In pathology, when the spine flexes forward with degenerative changes, the gravity moves forward increasing CF tremendously. The counteraction of the lumbar muscles may be quickly overpassed explaining the painful difficulties of standing position and the progressive flexion of the body during the day.

# 47.11 Spinopelvic Morphology and MR Imaging

Roussouly et al. noted a relationship between the four lumbar spinopelvic morphotypes and the degenerative findings depicted in CT scans and MRIs [47]. In an unbalanced spine, the forward momentum of gravitational force moves forward and the compensatory forces in the paravertebral musculature increase, which increases the contact force. The contact force distribution into the joints of the functional spine unit (into the disk or into facet joints) is made according to the global sagittal orientation of the column. If the spine is in flexion, the contact forces are forward, over the vertebral bodies and disks. If the spine is in extension, contact forces pass backward over the articular facets (Fig. 47.3). Roussouly hypothesized that each of the four spinopelvic morphotypes would have a different tendency to degenerate according to how the forces would interact in the disk or in the facet joints.

Moreover, depending on the orientation of vertebral plates and disk, the contact force may be divided into two as a result, parallel and perpendicular to the lower plateau. A strong slope of the plate increases resultant parallel, shear/slipping forces. When the plateau is slightly inclined, it is the force of pressure that is increased (Fig.).

In "flat back" or in lumbar kyphosis, the resultant contact forces are ahead and the forces of disk pressure are increased. In a lumbar spine with much lordosis or in lumbar hyperextension, the contact forces are exerted on the posterior parts and over the facet joints. The sloping plateau favors slipping and may constitute the olisthesis. The overpressure on posterior articular facet may determine facet arthrosis and degenerative spondylolisthesis.

Beginning with four spinopelvic morphotypes of lordosis and their specific organizations, Roussouly et al. sought to deduct the hypothesis of types of lordosis' degenerative evolution [6].

Type 1 lordosis (Fig. 47.8a): defined by a small SS ( $<35^\circ$ ). The PI is often small ( $<45^\circ$ ), with a very short but marked distal lordosis and a thoracolumbar kyphosis. This geometry may promote a short distal (L4–L5–S1) lumbar hyperextension and a zone of disk overpressure at the level of the thoracolumbar kyphosis. The likely degenerative evolution can combine discopathies by overpressure in the thoracolumbar kyphosis and retrolisthesis in the junctional zone. In the region of short distal lordosis (L4-L5 and mainly L5–S1), disks often have normal appearance on magnetic resonance imaging (MRI). However, mechanical stress is felt posteriorly in the very distal acute lordosis, being observed by interspinous contact when the individual is upright and also with the densification of the joints in the CT scan. This local hyperextension may explain root pain in standing position by narrowing the foramens when extension is at its maximum. The distal lumbar hyperextension is proportional to the amplitude of thoracolumbar kyphosis. Lumbar distal hyperextension (marked distal short acute lordosis) can lead to lysis of L5 isthmus by fatigue fracture due to impaction of the inferior articular facet joints. This explains the L5-S1 "nutcracker" mechanism associated with isthimic spondylolisthesis in patients with low PI [6] (Fig. 47.5). Pain may occur. As PI is small, the possibility of compensation to a sagittal imbalance is small.

Type 2 lordosis (Fig. 47.8b): defined by a small SS (<35°) and PI generally small (<45°). The division between lordosis and kyphosis is smooth, but the overall lumbar lordosis angle is small, featuring a flat back. The small lordosis displaces the contact forces forward. The intervertebral plates are horizontalized. The most important forces affect the distal lumbar plates L4–L5 and L5–S1. The disk overpressure may be responsible for early disk degeneration manifested by a central disk herniation or early disk degeneration at multiple levels. This may be the case of many lumbar disk herniations in individuals in the third or fourth decades of life. Barrey stated that disk herniations in subjects under 45 years old have a trend to develop in type 2 (flat back) lordosis [18]. However, after 45 years old, there is a homogeneous distribution of disk herniations between all four types of lumbar lordosis. A type 2 lordosis is identified through the small PI and there are small retroversion possibilities. When conducting an arthrodesis, there is no need to obtain significant lordosis. However, care must be taken not to further reduce the small amount of lordosis. The patient must be aware of this, understanding that his/her spine has lower possibilities of compensation for loss of lordosis due to aging and that more spinal troubles are a real possibility due to his/her spinopelvic anatomy.

Type 3 lordosis (Fig. 47.8c): harmonious and theoretically is not prone to a particular type of degeneration.

Type 4 lordosis (Fig. 47.8d): has a large SS (>35°) and large PI (>55°). It features smooth curvature with large lordosis. The lordosis angle and the number of vertebrae in lordosis are high. The contact forces are moved backward, and in front, the disks are protected from early degeneration. The slope of the distal disks L4–L5 and L5–S1 may predispose to spondylolisthesis by isthmic lysis by shearing forces. 80 % of the L5 isthmic lyses have a large PI (>60°). The type 4 without isthmic lysis can degenerate by two modes, early or late:

Early: the mechanical loads on the posterior facet joints may cause a later interapophysary osteoarthritis (Fig.), and eventually a degenerative



the thoracolumbar kyphosis. The likely degenerative evolution can combine discopathies by overpressure in the thoracolumbar kyphosis and retrolisthesis in the junctional zone. (c) Type 3 lordosis. (d) Type 4 lordosis, Harmonious but long and intense curvature





spondylolisthesis especially in L4–L5, by functional derangement of the joint and disk.

Late: it is frequent in older individuals, where disk degeneration is linked to aging. The loss of disk height causes a lumbar hypolordosis and, as a result, retroversion of the pelvis. This situation of flat back with retroverted pelvis must be distinguished from a true type 2 where it is a native flat back without pelvis retroversion

The type of lordosis, be it type 2 or type 4, may affect the morphology of posterior vertebral elements (facet joints/interspinous processes). According to the arc of a circle mathematical model, in type 2 lordosis, as the lordosis is smooth and flat, the anterior body line exhibits a huge radius for the circle of the inferior arc and almost the same for the facet joint line. This is not the case in type 4 lordosis where the radiuses of the arcs are smaller, with inconsequence of a smaller length of the posterior facet line (Fig. 47.9a). Consequently, we may expect larger spaces for facets and spinous processes in type 2 than in type 4 lordosis and maybe more contact between facet joints in type 4 lordosis, what would be a predisposing factor for facet joint arthritis and arthrosis (Fig. 47.9b). Facet joints and spinous processes are probably bigger in type 2 than in type 4. Anatomical or radiological studies relating to the size of the facet joints to the spinopelvic morphotype of lordosis are required to confirm this hypothesis.

In an asymptomatic population, types 1 and 2 represent only 30–35 % of subjects [18]. They are significantly more frequent in the group of herniated disks and disk disease. In contrast, there are few types 1 and 2 in the group of degenerative spondylolisthesis. Types 3 and 4 were found in the group of discopathies and herniated disks while they constitute nearly 85 % of lordosis found in the group of degenerative spondylolisthesis. These results show the influence of the very probable spinal statics in the determination of spinopelvic type.

 In the asymptomatic population (n=160), Barrey found 21,2 % as type 1, 11,2 % as type 2, 37,5 % as type 3, and 30 % as type 4. In total, it seems that there are disk herniation, DDD, spondylolisthesis, and lumbar



**Fig. 47.9** (a) In a type 2 lordosis (*left*), the area occupied by posterior elements is longer for a same number of vertebrae than in a type 4 lordosis (*right*). By this way, posterior elements (facets and spinous processes) are smaller in type 4 than in type 2 (From Roussouly and Pinheiro-Franco [17]; with permission). (b) Vertical forces and different spinopelvic morphotypes. Note how these forces may affect posterior vertebral elements, depending on types of lumbar lordosis. Type 4 lordosis (*bottom*) should be more prone to facet joint disease. Less concentrated lordosis (*top left*) should be less prone to facet problems (more facet "impaction" when marked lordosis exists) (From Barrey [18]; with permission)



**Fig. 47.10** When performing a spine fusion, it is mandatory to give the proper lordosis to the fused part of the spine according to the global shape of the spine and pelvis

narrow canal for each type of lordosis but there are fairly clear trends. The less lordotic curvatures are rather prone to disk problems while lordosis with strong curvatures is more susceptible to degenerative slippages. In the group of disk pathology, the spinal profile varies significantly with age. In young subjects with disk problems (lumbar disk herniation or discopathies), the morphotype is formed in nearly two-thirds of cases of lower PI. On the contrary, with aging, the prevalence of disk problems tends to join the normal population. In other words, the subjects presenting discopathies or herniated disk disease early in life have a tendency to present a particular spinal profile (type 1 or 2). With aging, the findings of disk problems including lumbar disk herniations have a much more homogeneous distribution in the range of the four spinopelvic morphotypes.

# 47.12 How to Organize a Spinal Arthrodesis

When performing a spine fusion, it is mandatory to give the proper lordosis to the fused part of the spine according to the global shape of the spine and pelvis (Fig. 47.10).

## 47.12.1 Strategy of Fusion in Case of Small PI

Lumbar lordosis may have an aspect of type 1 or type 2.

*Type 1 evolution*: Degenerative effect may induce TL kyphosis and promote a short hyperlordosis by compensation. In young patients, the surgical strategy has to focus on TL kyphosis reduction in order to obtain a compensative reduction of lumbar hyperlordosis. In older patients, the TL kyphosis reduction may be more demanding and may pose risk for pseudarthrosis. The authors propose to maintain the TL kyphosis with a partial reduction and to maintain the type 1 shape by fusing lower lumbar spine with a short hyperlordosis until the sacrum.

*Type 2 lordosis*: the lumbar spine has a global aspect of a flat back, and in the worse degenerative evolution, a lumbar kyphosis may occur. Restoration of a small angle of lordosis is sufficient. Sometimes, in case of lumbar kyphosis, a PSO may be necessary to restore an adequate lordosis. A small-angle PSO is sufficient enough to rebalance the spine. As there is a poor impact of lumbar reduction on the thoracic kyphosis, the risk of adjacent kyphosis is small, and it is not necessary to address the thoracic spine in long lumbar fusions.

*Type 4 evolution*: Two situations are possible—the spine has still a big amount of lordosis or the spine has lost the lumbar lordosis and the pelvis is in retroversion (high PT).

- When the lumbar spine remains hyperlordotic, it is mandatory to maintain this long and curved lordosis. As we have seen that the lower arch of LL is equal to SS, the more SS is tilted, the more the lower arch is curved. This is the reason why it is mandatory to restore the maximum of curvature between L4, L5, and S1. A short L4–L5 fusion for a degenerative L4–L5 spondylolisthesis without enough extension will bring painful compensatory hyperextension to the adjacent level L3–L4.
- When the lumbar lordosis is lost, the authors have observed that the balance compensation is obtained by pelvis retroversion and, sometimes, by a flat thoracic spine due to the extension of the thoracic area reducing the normal kyphosis. This situation is very challenging. Restoration of a big amount of lordosis necessitates osteotomies either by Smith-Petersen or PSO. Problems of junctional kyphosis may occur because of the spontaneous changes of thoracic spine in relation with the new lumbar shape. This relation between thoracic and lumbar spine lordosis has to be analyzed fur-

ther to better understand the problem of junctional kyphosis.

## Conclusion

To obtain the sagittal balance of the human body in upright posture is not only to put C7 over the sacrum. The various spinopelvic shapes, determined by PI ranges of values, induce different spinal morphologies. The mechanical arrangements of each morphotype permit to explain the physical properties of one spine and the degenerative changes in the elderly. Treatment strategies for spinal pathologies have to take into account this strong relation between spine and pelvis, mainly when planning surgical arthrodesis where mistakes in balance obtention may bring to unsuccessful situations.

## References

- 1. Robertson JT. The rape of the spine. Surg Neurol. 1993;39:5–12.
- Podichetty VK. The aging spine: the role of inflammatory mediators in intervertebral disc degeneration. Cell Mol Biol (Noisy-le-grand). 2007;53(5):4–18.
- Roussouly P, Pinheiro-Franco JL. Biomechanical analysis of the spino-pelvic organization and adaptation in pathology. Eur Spine J. 2011;20:609–18.
- Roussouly P, Gollogly S, Noseda O, Berthonnaud E, Dimnet J. The vertical projection of the sum of the ground reactive forces of a standing patient is not the same as the C7 plumb line: a radiographic study of the sagittal alignment of 153 asymptomatic volunteers. Spine. 2006;31:E320–5.
- Berthonnaud E, Dimnet J, Roussouly P, Labelle H. Analysis of the sagittal balance of the spine and pelvis using shape and orientation parameters. J Spinal Disord Tech. 2005;18:40–7.
- Roussouly P, Berthonnaud E, Dimnet J. Geometrical and mechanical analysis of lumbar lordosis in an asymptomatic population: proposed classification. Rev Chir Orthopédique Réparatrice Appar Mot. 2003;89:632–9.
- Rajnics P, Templier A, Skalli W, Lavaste F, Illés T. The association of sagittal spinal and pelvic parameters in asymptomatic persons and patients with isthmic spondylolisthesis. J Spinal Disord Tech. 2002;15:24–30.
- Roussouly P, Gollogly S, Berthonnaud E, Labelle H, Weidenbaum M. Sagittal alignment of the spine and pelvis in the presence of L5-s1 isthmic lysis and lowgrade spondylolisthesis. Spine. 2006;31:2484–90.

- Barrey C, Jund J, Perrin G, Roussouly P. Spinopelvic alignment of patients with degenerative spondylolisthesis. Neurosurgery. 2007;61:981–6. discussion 986.
- Mac-Thiong J-M, Labelle H, Charlebois M, Huot M-P, de Guise JA. Sagittal plane analysis of the spine and pelvis in adolescent idiopathic scoliosis according to the coronal curve type. Spine. 2003;28:1404–9.
- 11. Kim YJ, Bridwell KH, Lenke LG, Rhim S, Cheh G. An analysis of sagittal spinal alignment following long adult lumbar instrumentation and fusion to L5 or S1: can we predict ideal lumbar lordosis? Spine. 2006;31:2343–52.
- Vaz G, Roussouly P, Berthonnaud E, Dimnet J. Sagittal morphology and equilibrium of pelvis and spine. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 2002;11:80–7.
- Jackson RP, McManus AC. Radiographic analysis of sagittal plane alignment and balance in standing volunteers and patients with low back pain matched for age, sex, and size. A prospective controlled clinical study. Spine. 1994;19:1611–8.
- Barrey C, Jund J, Noseda O, Roussouly P. Sagittal balance of the pelvis-spine complex and lumbar degenerative diseases. A comparative study about 85 cases. Eur Spine J. 2007;16:1459–67.
- Rajnics P, Templier A, Skalli W, Lavaste F, Illes T. The importance of spinopelvic parameters in patients with lumbar disc lesions. Int Orthop. 2002;26:104–8.
- George SZ, Hicks GE, Nevitt MA, Cauley JA, Vogt MT. The relationship between lumbar lordosis and radiologic variables and lumbar lordosis and clinical variables in elderly, African-American women. J Spinal Disord Tech. 2003;16:200–6.
- Roussouly P, Pinheiro-Franco JL. Sagittal parameters of the spine: biomechanical approach. Eur Spine J. 2011;20:578–85.
- Barrey C. Equilibre sagittal pelvi-rachidien et pathologies lombaires degeneratives. 2004. These Doctorat – Université Claude-Bernard, Lyon 1.
- Duval-Beaupère G, Schmidt C, Cosson P. A Barycentremetric study of the sagittal shape of spine and pelvis: the conditions required for an economic standing position. Ann Biomed Eng. 1992;20:451–62.
- 20. Legaye J, Duval-Beaupère G, Hecquet J, Marty C. Pelvic incidence: a fundamental pelvic parameter for three-dimensional regulation of spinal sagittal curves. Eur Spine J Off Publ Eur Spine Soc Eur Spinal Deform Soc Eur Sect Cerv Spine Res Soc. 1998;7:99–103.
- Legaye J, Hecquet J, Marty C, Duval-Beaupère G. Equilibre sagittal du rachis. Relations entre bassin et courbures rachidiennes sagittales en position debout. Rachis. 1993;5:215–26.
- Boulay C, et al. Sagittal alignment of spine and pelvis regulated by pelvic incidence: standard values and prediction of lordosis. Eur Spine J. 2005;15:415–22.
- Roussouly P, Gollogly S, Berthonnaud E, Dimnet J. Classification of the normal variation in the sagittal alignment of the human lumbar spine and pelvis in the standing position. Spine. 2005;30:346–53.

- 24. Stagnara P, et al. Reciprocal angulation of vertebral bodies in a sagittal plane: approach to references for the evaluation of kyphosis and lordosis. Spine. 1982;7:335–42.
- Troyanovich SJ, Cailliet R, Janik TJ, Harrison DD, Harrison DE. Radiographic mensuration characteristics of the sagittal lumbar spine from a normal population with a method to synthesize prior studies of lordosis. J Spinal Disord. 1997;10:380–6.
- Janik TJ, Harrison DD, Cailliet R, Troyanovich SJ, Harrison DE. Can the sagittal lumbar curvature be closely approximated by an ellipse? J Orthop Res. 1998;16:766–70.
- Vialle R, et al. Radiographic analysis of the sagittal alignment and balance of the spine in asymptomatic subjects. J Bone Joint Surg Am. 2005;87:260–7.
- Kuntz 4th C, Levin LS, Ondra SL, Shaffrey CI, Morgan CJ. Neutral upright sagittal spinal alignment from the occiput to the pelvis in asymptomatic adults: a review and resynthesis of the literature. J Neurosurg Spine. 2007;6:104–12.
- International Spine Study Group, et al. Risk factors for major peri-operative complications in adult spinal deformity surgery: a multi-center review of 953 consecutive patients. Eur Spine J. 2012;21: 2603–10.
- Vedantam R, Lenke LG, Bridwell KH, Linville DL, Blanke K. The effect of variation in arm position on sagittal spinal alignment. Spine. 2000;25:2204–9.
- Marks MC, Stanford CF, Mahar AT, Newton PO. Standing lateral radiographic positioning does not represent customary standing balance. Spine. 2003;28:1176–82.
- 32. Rillardon L, et al. Validation of a tool to measure pelvic and spinal parameters of sagittal balance. Rev Chir Orthopédique Réparatrice Appar Mot. 2003;89:218–27.
- 33. www.sagittal-balance.com.
- 34. Gelb DE, Lenke LG, Bridwell KH, Blanke K, McEnery KW. An analysis of sagittal spinal alignment in 100 asymptomatic middle and older aged volunteers. Spine. 1995;20:1351–8.
- Bernhardt M, Bridwell KH. Segmental analysis of the sagittal plane alignment of the normal thoracic and lumbar spines and thoracolumbar junction. Spine. 1989;14:717–21.
- Glassman SD, et al. The impact of positive sagittal balance in adult spinal deformity. Spine. 2005;30:2024–9.
- Schwab F, Lafage V, Patel A, Farcy J-P. Sagittal plane considerations and the pelvis in the adult patient. Spine. 2009;34:1828–33.
- Glassman SD, Berven S, Bridwell K, Horton W, Dimar JR. Correlation of radiographic parameters and clinical symptoms in adult scoliosis. Spine. 2005;30:682–8.
- Schwab F, et al. A clinical impact classification of scoliosis in the adult. Spine. 2006;31:2109–14.
- Lafage V, Schwab F, Patel A, Hawkinson N, Farcy J-P. Pelvic tilt and truncal inclination: two key
radiographic parameters in the setting of adults with spinal deformity. Spine. 2009;34:E599–606.

- Grobler LJ, Moe JH, Winter RB. Loss of lumbar lordosis following surgical correction of thoracolumbar deformities. Orthop Trans. 1978;2:239.
- 42. Gautier J, Morillon P, Marcelli C. Does spinal morphology influence the occurrence of low back pain? A retrospective clinical, anthropometric, and radiological study. Rev Rhum Engl Ed. 1999;66:29–34.
- Chaléat-Valayer E, et al. Sagittal spino-pelvic alignment in chronic low back pain. Eur Spine J. 2011;20:634–40.
- 44. Mac-Thiong J-M, Roussouly P, Berthonnaud E, Guigui P. Sagittal parameters of global spinal balance: normative values from a prospective cohort of seven hundred nine Caucasian asymptomatic adults. Spine. 2010;35:E1193–8.

- 45. Mangione P, Gomez D, Senegas J. Study of the course of the incidence angle during growth. Eur Spine J. 1997;6:163–7.
- 46. Le Huec JC, Leijssen P, Duarte M, Aunoble S. Thoracolumbar imbalance analysis for osteotomy planification using a new method: FBI technique. Eur Spine J. 2011;20:669–80.
- 47. Pinheiro-Franco JL, Roussouly P, Vaccaro AR. Importância do Equilíbrio Sagital no Tratamento Cirúrgico da Doença Degenerativa Discal Lombar. In: Pinheiro-Franco JL, Vaccaro AR, Benzel EC, Mayer H-M, editors. Conceitos Avançados em Doença Degenerativa Discal Lombar. Rio de Janeiro: DiLivros Publisher; 2010. p. 277–86 (in Portuguese).

ions48.1 IntroductionLumbar lordosisThe aging spine is characterized<br/>disk disease, hypertrophic face<br/>bone remodeling, and atrophy of<br/>cles. This may result in a progre

# Abbreviations

LL

PI

PT

SS

SSA

SVA

TK

SVA/SFD

C. Barrey, MD, PhD (🖂)
Laboratory of Biomechanics,
ENSAM, Arts et Metiers ParisTech,
51 boulevard de l'hopital,
75013 Paris, France

Department of Neurosurgery C and Spine Surgery, Neurological Hospital, University Claude Bernard Lyon 1, 59 boulevard Pinel, 69394 Lyon, France e-mail: c.barrey@wanadoo.fr; cedric.barrey@chu-lyon.fr

Spinosacral angle

Thoracic kyphosis

Sagittal vertical axis

Ratio SVA/sacro-femoral distance

J.L. Pinheiro-Franco, MD Neurological Surgery and Spine Surgery, Samaritano Hospital, São Paulo, Brazil J.-C. Le-Huec, MD, PhD Department of Spine Surgery, Pellegrin University Hospital, Bordeaux, France

G. Perrin, MD Department of Neurosurgery C and Spine Surgery, Neurological Hospital, University Claude Bernard Lyon 1, 59 boulevard Pinel, 69394 Lyon, France

P. Roussouly, MD Department of Orthopaedic Surgery, Centre Médico-Chirurgical de Réadaptation des Massues, 92 rue Edmond Locard, 69005 Lyon, France e-mail: p.roussouly@cmcr-massues.com

Compensatory Mechanisms Contributing to the Maintenance of Sagittal Balance in Degenerative Diseases of the Lumbar Spine

Cédric Barrey, João Luiz Pinheiro-Franco, Jean-Charles Le-Huec, Gilles Perrin, and Pierre Roussouly

> The aging spine is characterized by degenerative disk disease, hypertrophic facet joints, arthritis, bone remodeling, and atrophy of extensor muscles. This may result in a progressive kyphosis of the lumbar spine with the risk of developing progressive sagittal imbalance [1, 2]. It has been extensively reported in the literature that patients with multi-segmental lumbar degenerative disk

diseases often present with significant modifications of the sagittal balance and may, in fact, present with anterior sagittal imbalance, loss of lumbar lordosis, and increased pelvic tilt [3–6]. In these patients the anterior imbalance results mainly from the loss of lumbar lordosis due to the degeneration of lumbar disks, but in some cases, the respective part of structural and postural loss of lordosis is difficult to determine. These changes in spinopelvic alignment result in deterioration of clinical status and functional scores [7].

Except for the loss of lordosis, which is related to the degenerative process, the other spinopelvic parameter changes (i.e., decrease of sacral slope, reduction of thoracic kyphosis, increase of upper lumbar lordosis) generally correspond to compensatory mechanisms. These compensations represent a compromise for the patient in order to maintain an erect posture in a physiological position with minimal muscular effort. To optimize the management of lumbar degenerative disorders and to avoid underestimating the severity of degenerative disease, it is important to recognize and take into consideration these compensatory mechanisms.

In the asymptomatic population, correlations between the shape of the pelvis (reflected by the pelvic incidence), the sacral slope, and the sagittal curves of the spine (especially lumbar lordosis) have been well documented in the literature [1, 6, 8-13]. The pelvic incidence represents the only parameter not modified by degenerative changes, thus providing useful information on the original morphology of the spinopelvic complex. Consequently, using these correlations from the normal population, it is relatively easy to understand the changes of spinopelvic parameters for patients with degenerative disorders in the lumbar spine. The compensatory mechanisms contribute to maintaining normal overall sagittal balance of the spine above the pelvis in an economic position, limiting the consequences of lumbar kyphosis in terms of sagittal anterior imbalance, and minimizing muscles efforts. They may involve the spine adjacent to the kyphotic segment, the pelvis, and/or the lower limbs. These mechanisms may present isolated or

combined (in case of severe deformity), depending on the intensity of the sagittal imbalance [14].

The objective of this chapter is to describe in details all of these different compensatory mechanisms in patients with severe degenerative lumbar spine disorders.

# 48.2 Assessment of Global Balance

It is essential to have an optimal congruence between pelvic and spinal parameters in order to achieve an economic posture placing the axis of gravity in a physiological position with minimal muscle efforts [8, 10, 15, 16]. When analyzing the spinopelvic alignment of patients with spinal deformities, the first step should be to evaluate the global balance of the patient. This can be done optimally by using strength plate and measuring positioning of the gravity axis in the sagittal plane [13]. However, in clinical practice, global balance is appreciated more simply by describing the relative positioning of the spine in reference with the pelvis on standing full spine radiographs. Global sagittal alignment is typically determined by calculating the position of the vertebra C7 related to the sacral plate. The offset (distance in mm) is measured between the posterior corner of the sacrum and the vertical line passing through the vertebral body of C7 (i.e., sagittal vertical axis, SVA). Instead of measuring a linear distance, we recommend using angular measurements and/or ratios to characterize the positioning of C7 in relation to the sacrum. Angular parameters are assessed via the spinosacral angle (SSA), and the ratio corresponds to the SVA/sacro-femoral distance ratio (SVA/SFD). These two parameters have already been reported and validated [3, 17].

The SSA was defined as the angle between the sacral plate and the line connecting the centroid of C7 vertebral body and the midpoint of the sacral plate (Fig. 48.1a). In the normal population, the mean value of this angle is  $135^{\circ} \pm 8$  [17].

The method to measure SVA/SFD ratio is presented in Fig. 48.1b. This ratio is equal to zero, when C7 plumb line projects exactly on



**Fig. 48.1** Evaluation of global sagittal alignment using the spinosacral angle (**a**) and the SVA/SFD ratio (**b**). The SSA is defined as the angle between the sacral plate and the line connecting the centroid of C7 vertebral body and the midpoint of the sacral plate [17]. Sacro-femoral distance (*SFD*) is the horizontal distance between the *vertical* bicoxofemoral axis and the vertical line passing through the posterior corner of the sacrum. The horizontal distance between C7 PL and the posterior corner of the

the posterior corner of the sacrum, and to one, when C7 plumb line projects exactly on the bicoxofemoral axis. It is negative when C7 plumb line projects posteriorly to the sacrum and more than one when C7 plumb line projects from anterior to the femoral heads. In the normal population, the value of this ratio is  $-0.9 \pm 1$  [3].

The spinosacral angle and the C7/SFD ratio facilitate the evaluation of the global sagittal alignment of the spine above the pelvis. According to the severity of the imbalance, we propose the identification of four different stages (from I to IV): balanced, compensated, partially compensated, and imbalanced (Fig. 48.2). In the last stage, the compensatory mechanisms are not efficient enough to maintain the sagittal balance and C7 plumb-line falls

sacrum (i.e., SVA) was also measured. Then we calculated the SVA/SFD ratio corresponding to the ratio between SVA distance and SF distance [3]. This ratio is equal to zero, when C7 plumb line projects exactly on the posterior corner of the sacrum, and to one, when C7 plumb line projects exactly on the bicoxofemoral axis. It is negative when C7 plumb line projects posteriorly to the sacrum and more than one when C7 plumb line projects from anterior to the femoral heads [3] (From Barrey et al. [3])

in front of the femoral heads (SVA/SFD ratio >1). An illustration of each situation is presented in Fig. 48.3.

### 48.3 Compensatory Mechanisms

Compensatory mechanisms can be observed in the spine, the pelvis, and/or the lower limb areas and are summarized in Fig. 48.4. Although these mechanisms are rarely observed all together in the same patient, they are usually associated at different degrees, depending mainly on the stiffness of the spine, the musculature status, painful phenomenon, and the severity of the imbalance.

Their basic concept is to extend adjacent segments of the kyphotic spine, facilitating the acquisition of compensated spine. Most of these



**Fig. 48.2** Classification of global sagittal alignment in 4 stages with respect to the severity of the imbalance. *Balanced*: C7 plumb line falls close to the posterior corner of the sacrum, SVA/SFD ratio is close to 0, there is no compensation. *Compensated*: C7 plumb line is closer to the posterior corner of the sacrum than to the femoral heads, SVA/SFD ratio is <0.5, compensation is present.

mechanisms result from muscle action, thus exposing the subject to chronic pain and muscle fatigue.

To understand the variations of positional parameters such as sacral slope (SS), pelvic tilt (PT), LL, and TK in the patients' population, we previously published values of six different classes of pelvic incidence in a normal control group of 154 subjects [4]. Values of positional parameters for each class of PI (from I to VI corresponding to a progressively increase of the PI value) are summarized in Table 48.1. Theoretical normal values for spinopelvic parameters (i.e., theoretical PT and theoretical LL) may also be estimated from mathematical relations (Table 48.2). Otherwise, to analyze

*Partially compensated*: C7 plumb line is closer to the femoral heads than to the posterior corner of the sacrum, SVA/SFD ratio is >0.5, compensation is present. *Imbalanced*: C7 plumb line is placed in front of the femoral heads, SVA/SFD ratio is >1, compensation is present but not sufficient to keep the balance

segmental changes, we have to keep in mind that the L4–S1 segment provides the two third of the total lumbar lordosis [6, 11, 19].

#### 48.3.1 Spine

### 48.3.1.1 Cervical Hyperlordosis

Although in most cases the cervical spine is not well evaluated on full spine radiographs, it should be included in the sagittal balance assessment since compensatory curvature can be observed at this level. Hyperextension of the cervical spine is a typical compensatory mechanism above a thoracic hyperkyphosis in order to maintain the horizontality of the gaze. Inconvenience related to



Fig. 48.3 Illustrations showing the four situations of sagittal balance

this hyperlordosis is not negligible – resulting in acceleration of degenerative changes in the cervical spine (i.e., hypertrophic facet joints arthritis and kissing spinous processes), presence of axial neck pain, foraminal stenosis, and the risk to develop spondylotic myelopathy.

### 48.3.1.2 Reduction of Thoracic Kyphosis

Reduction of thoracic kyphosis permits the limitation of anterior translation of the axis of gravity and is typically observed in young patients with a flexible spine (Fig. 48.5). It is the consequence of active muscle actions and therefore implies a good quality of erectors muscles of the spine. Takemitsu et al. described this mechanism for patients with lumbar kyphosis [20]. In a previous work, we also found that patients with degenerative disk disease and disk herniation were characterized by flat spine with significant reduction of both lumbar lordosis and thoracic kyphosis. This profile was more marked for patients with disk diseases below 45 years old [4]. Our findings were concordant with those reported by Rajnics et al. through a similar study [21]. When the spine is rigid (aging of the spine is kyphosis and ankylosis) or in case of atrophy of spinal erectors, it is not possible for the patient to reduce the magnitude of the thoracic curve.

In severe cases of sagittal imbalance, reduction of thoracic curve is not sufficient (Fig. 48.6). Considering that the compensatory thoracic hypokyphosis is potentially reversible after restoration of sagittal balance, it is thus possible to limit the extent of the fusion and preserve the thoracic spine mobility. Reappearance of physiological thoracic kyphosis may be observed postoperatively after correction of the lumbar spinal deformity (Fig. 48.5).



**Table 48.1** Classes of pelvic incidence and corresponding values of spinopelvic positional parameters from a group control of 154 subjects [4]

	n	PI	РТ	SS	LL	TK
I 28° <pi<37.9°< td=""><td>12</td><td>35.4±1.3 [33.7–37.9]</td><td>3.9±4.5 [-1.5-13.3]</td><td>31.5±5.2 [21.2–38.5]</td><td>53.3±6.6 [41.2–62]</td><td>43.8±9.1 [22.5–51.5]</td></pi<37.9°<>	12	35.4±1.3 [33.7–37.9]	3.9±4.5 [-1.5-13.3]	31.5±5.2 [21.2–38.5]	53.3±6.6 [41.2–62]	43.8±9.1 [22.5–51.5]
II 38° <pi<47.9°< td=""><td>44</td><td>42.7±2.8 [37.9–47.6]</td><td>8.9±4.8 [-5.1-18.2]</td><td>33.8±4.8 [23.1–48.4]</td><td>55.5±8 [41.5–76.5]</td><td>48±8.8 [24-64.7]</td></pi<47.9°<>	44	42.7±2.8 [37.9–47.6]	8.9±4.8 [-5.1-18.2]	33.8±4.8 [23.1–48.4]	55.5±8 [41.5–76.5]	48±8.8 [24-64.7]
III 48° <pi<57.9°< td=""><td>59</td><td>52.6±2.8 [48.2–57.4]</td><td>12.5±5.6 [-1.2-23.2]</td><td>40.1±5.5 [28.2–52.9]</td><td>61.5±8.4 [43.1–81.9]</td><td>47.4±10.7 [24–70.3]</td></pi<57.9°<>	59	52.6±2.8 [48.2–57.4]	12.5±5.6 [-1.2-23.2]	40.1±5.5 [28.2–52.9]	61.5±8.4 [43.1–81.9]	47.4±10.7 [24–70.3]
IV 58° <pi<67.9°< td=""><td>26</td><td>62.6±2.8 [58.2–67.6]</td><td>15.8±4.3 [7.1–26.8]</td><td>46.8±4.2 [37.9–58.5]</td><td>68.3±5.1 [60.9–76.3]</td><td>47.6±7.8 [34.7–64.7]</td></pi<67.9°<>	26	62.6±2.8 [58.2–67.6]	15.8±4.3 [7.1–26.8]	46.8±4.2 [37.9–58.5]	68.3±5.1 [60.9–76.3]	47.6±7.8 [34.7–64.7]
V 68° <pi<77.9°< td=""><td>11</td><td>72.6±2.8 [69.6–77.4]</td><td>19.7±5.5 [12.6–27.9]</td><td>52.9±5.2 [46.2–59.6]</td><td>74.9±6.8 [62.2–81.6]</td><td>46±10.2 [29.7–62]</td></pi<77.9°<>	11	72.6±2.8 [69.6–77.4]	19.7±5.5 [12.6–27.9]	52.9±5.2 [46.2–59.6]	74.9±6.8 [62.2–81.6]	46±10.2 [29.7–62]
VI 78° <pi<87.9°< td=""><td>2</td><td>81.4±3.3 [79.1–81.4]</td><td>21.9±12.3 [13.2–30.6]</td><td>59.5±9 [53.1–65.9]</td><td>76±8.3 [70.1–81.9]</td><td>44.6±12.2 [36–53.3]</td></pi<87.9°<>	2	81.4±3.3 [79.1–81.4]	21.9±12.3 [13.2–30.6]	59.5±9 [53.1–65.9]	76±8.3 [70.1–81.9]	44.6±12.2 [36–53.3]

	- 1		
PI class	PI (°)	Theoretical PT (°)	Theoretical LL (°)
Ι	<38	4	PI+18
Π	38–47	8	PI+13
III	48–57	12	PI+9
IV	58–67	16	PI+6
V	68–77	20	PI+2
VI	>78	24	PI – 5

**Table 48.2** Theoretical pelvic tilt and lumbar lordosis according to the pelvic incidence [4]

As examples, for PI measured to  $40^\circ$ , expected PT should be  $8^\circ$  and LL should be  $53^\circ$ ; for PI measured to  $52^\circ$ , expected PT should be  $12^\circ$  and LL should be  $61^\circ$ ; and for PI measured to  $64^\circ$ , expected PT should be  $16^\circ$  and LL should be  $70^\circ$ 

### 48.3.1.3 Hyperextension of Adjacent Segments

Hyperextension of adjacent segments is a very common local compensatory mechanism to limit the consequences of lumbar kyphosis on the shift of axis gravity (Fig. 48.7). Previous studies demonstrated that low back pain subjects were characterized by less caudal lordosis, a more vertical sacrum, and greater rostral lumbar lordosis [5, 23]. Higher values of proximal lumbar lordosis signify more extension in the upper lumbar spine. Recently, Schuller et al. found that upper lumbar spine (L1–L2



**Fig. 48.5** Patient with lumbar kyphosis due to severe multilevel degenerative disease stenosis from L2 to S1. The patient is unbalanced (C7PL/SFD>1). Thoracic spine is clearly flattened (thoracic curve measured to only 21°). PI was measured to  $26^{\circ}$ , PT was  $12^{\circ}$ , and SS was  $16^{\circ}$ . Compared to group control from normal and asymptom-

atic population, we should expect value of PT around 4°. After corrective surgery in the lumbar spine and restoration of good sagittal balance (C7PL/SFD <0.5), we noted the reappearance of the thoracic kyphosis and reduction of PT (from 12 to 6°)



Fig 48.6 Illustration of patient presenting with thoracolumbar lordosis above a hypolordotic construct from L4 to S1

and L2–L3 segments) was more extended for patients with L4–L5 degenerative spondylolisthesis [24]. On the other hand, in cases of thoracolumbar kyphosis, hyperlordosis in the lower lumbar spine (L4–S1) is very common (Fig. 48.7) [25].

Hyperextension can be regional (multisegmental) or local (mono/bi-segmental). Regional hyperextension may affect the thoracic spine, the thoracolumbar junction, and a part of the lumbar spine (the upper or the lower part). For example, lower lumbar hyperlordosis is typically observed in patients with thoracolumbar kyphosis, regardless of the kyphosis (congenital, degenerative, posttraumatic, etc.).

Local hyperextension in this situation is sufficient to position the upper spine posteriorly; however, this generates increase of stresses on posterior structures (Fig. 48.7), exposes to the risk of retrolisthesis, and may result in accelerated facet joint arthritis, interspinous hyperpressure (Baastrup disease), and sometimes isthmic lysis.

From a biomechanical point of view, we consider that compensatory discopathy must be differentiated from classical aging discopathy (Fig. 48.8). Compensatory discopathy is characterized by a discal hyperextension (more than 15°) compensating for a loss of lordosis. It is associated with increased stresses on the posterior elements with facet arthritis and hyperpressure contact between the spinous processes. On the contrary, aging discopathy (the most frequent type) is characterized by a disk narrowing with parallel end plates resulting in loss of lordosis. The spinal segment is kyphotic in aging DDD but hyperextended in compensatory DDD, with increase of stresses on the posterior elements. The sagittal orientation of disk space (extended, neutral, kyphotic) is therefore very important to consider when analyzing consequences of degenerative discopathy on global balance.



**Fig. 48.7** Patient with lumbar stenosis from L2–L3 to L4–L5 and thoracolumbar kyphosis: full spine radiographs (**a**), X-rays focused on lumbo-pelvic zone (**b**) and sagittal T2-weighted MRI sequence (**c**). The patient is well balanced (C7PL/SFD is -0.3); however, some compensatory mechanisms are present in the lumbar area.

Hyperextension is observed at L5–S1 (*curved black arrow*) (local lordosis was measured to  $24^{\circ}$ ), and there is multilevel retrolisthesis at L2–L3 (*red circle*) and L4–L5 (*large arrow*). The pelvic tilt was quite normal as it was calculated to  $22^{\circ}$  and the PI to  $46^{\circ}$  (From Barrey et al. [22])



**Fig. 48.8** Classification of degenerative disk diseases into aging discopathy and compensatory discopathy. Compressive forces (*arrow*), spine working in extension (*curved arrow*) and stresses on posterior elements (*star*)

#### 48.3.1.4 Retrolisthesis

Retrolisthesis, defined as the posterior slippage of the upper vertebra in reference to the lower vertebra, is typically limited to 2–3 mm slippage in the lumbar spine. It often results in severe foraminal stenosis and more rarely in central stenosis (Fig. 48.7). Retrolisthesis is usually observed at the immediate adjacent lower or upper segments of the lumbar spine that had some kyphotic changes: L5–S1 and upper lumbar spine L1–L2 and L2–L3) are common sites. Retrolisthesis is typically underestimated on supine radiological imaging techniques (CT scan and MRI); however, they can be suspected on MRI imaging in the presence of subluxation of facet joints with fluid collection.

### 48.3.2 Pelvis

The only compensatory mechanism in the pelvic area is pelvic back tilt (also called pelvic

retroversion), defined by the increase of the pelvic tilt and corresponding to the posterior rotation of the pelvis around the femoral heads, similar to that occurring during hip extension (Fig. 48.9). This motion is permitted by contraction of the hip extensor muscles and results in posterior positioning of the sacrum related to the coxofemoral heads. Bringing backwards the sacral plate related to the coxofemoral heads (and increasing the sacro-femoral distance), this mechanism permits the subject to compensate for the anterior translation of the axis of gravity. The pelvic incidence determines the global capacity of pelvic retroversion which is easily achieved for patients with a great pelvic incidence. In fact, considering that PI = SS + PT and that SS cannot be a negative number, one can tilt more with a high PI than with a low PI, since there is a much wider range through which adaptation can occur. Numerous studies reported that patients with chronic low back pain and lumbar degenerative disease were characterized



**Fig. 48.9** Pelvic backtilt mechanism. Increase of pelvic tilt results in posterior placement of sacrum related to the coxofemoral heads, thus increasing the sacro-femoral dis-

tance (*red line*) and bringing back the axis of gravity (From Barrey et al. [26]; with permission)

by decrease of sacral slope and increase of pelvic tilt [4, 5, 23] as demonstrated in our illustrated cases in Fig. 48.3.

### 48.3.3 Lower Limbs

#### 48.3.3.1 Knee Flexion

Knee flexion can be evaluated by the pelvifemoral angle described by Mangione et al. [27]. This is a well-known compensatory mechanism for patients with severe degenerative spine and has already been widely reported [2], Fig. 48.10. It permits to translate posteriorly the axis of gravity with respect to the feet. More recently, Obeid and Vital demonstrated the strong correlation between knee flexum angle and the lack of lordosis (which was estimated from the theoretical lordosis) [28]. Knee flexum is usually observed in severe sagittal imbalance at late stage of the degenerative process with global kyphosis of the lumbar and thoracic spine. The posture with knee flexed is very uncomfortable for the patient with limited walking distance.

### 48.3.3.2 Ankle Extension

Through a prospective study, Lafage et al. recently underlined that the pelvic translation was a parameter as important as the pelvic rotation (measured by the pelvis tilt) and probably induced by extension in ankle joint [15]. Therefore, they suggested that the patients should be analyzed from head to feet.



**Fig. 48.10** Illustrative case with lumbar kyphosis and compensatory knee flexum. After surgical correction of the kyphosis (TPO procedure at L4), the knee flexum sig-

nificantly reduced postoperatively suggesting the improvement of the global sagittal balance

### 48.4 Algorithm

Finally we propose a three-step algorithm to achieve the analysis of sagittal balance and determine the presence or not of compensatory mechanisms:

- *First step* What is the value of the pelvic incidence? The knowledge of the pelvic incidence permits to determine the expected theoretical values of the spinopelvic positional parameters (Tables 48.1 and 48.2).
- Second step Is the patient globally balanced? Global sagittal alignment is evaluated by analyzing the positioning of C7, related to the sacrum, using measurement of SSA and/or SVA/SFD ratio.
- *Third step* Are there some compensatory mechanisms?
  - In spinal area: analysis of this zone consists of measurement of lumbar lordosis and thoracic kyphosis and looking for the presence of compensatory discopathy(ies) and retrolisthesis and/or abnormal regional curves (i.e., thoracolumbar lordosis). Cervical curvature analysis should also be included.
  - In pelvic area: is the pelvic tilt (PT) adequate with respect to the pelvis incidence? The presence of horizontal sacral plate is highly suspected of pelvic backtilt mechanism. The use of PI classes is useful to determine the theoretical PT (Table 48.2).
  - In lower limb area: are the knee flexed? One must care to this point considering that knee flexum minimizes the importance of sagittal imbalance on full spine radiographs. Measurement of knee flexum angle is mandatory but implies full body radiograph (as provided by EOS<sup>™</sup> system).

#### Conclusion

Meticulous and exhaustive analysis of spinopelvic parameters allows for identification of the main compensatory mechanisms observed in patients with sagittal balance disorders. These mechanisms have to be considered prior to therapeutic options. This may probably optimize the management of patients with severe degenerative spine especially when surgical treatment with instrumentation of the spine is planned.

### References

- Gelb DE, Lenke LG, Bridwell KH, Blanke K, MacEnery KW. An analysis of sagittal spinal alignment in 100 asymptomatic middle and older aged volunteers. Spine. 1995;20:1351–8.
- Kobayashi T, Atsuta Y, Matsuno T, Takeda N. A longitudinal study of congruent sagittal spinal alignment in an adult cohort. Spine. 2004;29:671–6.
- Barrey C. Equilibre sagittal pelvi-rachidien et pathologies lombaires dégénératives. Etude comparative à propos de 100 cas (in French). Thèse de Médecine. Lyon: Université Claude Bernard; 2004.
- Barrey C, Jund J, Noseda O, Roussouly P. Sagittal balance of the pelvis-spine complex and lumbar degenerative diseases. A comparative study about 85 cases. Eur Spine J. 2007;16:1459–67.
- Jackson RP, MacManus AC. Radiographic analysis of sagittal plane alignment and balance in standing volunteers and patients with low back pain matched for age, sex and size. Spine. 1994;19:1611–8.
- Jackson RP, Kanemura T, Kawakami N, Hales C. Lumbopelvic lordosis and pelvic balance on repeated standing lateral radiographs of adult volunteers and untreated patients with constant low back pain. Spine. 2000;25:575–86.
- Smith JS, Klineberg E, Schwab F, Shaffrey CI, Moal B, Ames CP, Hostin R, Fu KM, Burton D, Akbarnia B, Gupta M, Hart R, Bess S, Lafage V, International Spine Study Group. Change in classification grade by the SRS-Schwab adult spinal deformity classification predicts impact on health-related quality of life measures: prospective analysis of operative and nonoperative treatment. Spine. 2013;38:1663–71.
- Berthonnaud E, Dimnet J, Roussouly P, Labelle H. Analysis of the sagittal spine and pelvis using shape and orientation parameters. J Spinal Disord Tech. 2005;18:40–7.
- During J, Goudfrooij H, Keessen W, Beeker TW, Crowe A. Towards standards for posture. Postural characteristics of the lower back system in normal and pathologic conditions. Spine. 1985;10:83–7.
- Duval-Beaupère G, Legaye J. Composante sagittale de la statique rachidienne (in French). Rev Rhum. 2004;71:105–19.
- Roussouly P, Gollogly S, Berthonnaud E, Dimnet J. Classification of the normal variation in the sagittal alignment of the human lumbar spine and pelvis in the standing position. Spine. 2005;30:346–53.

- Stagnara P, De Mauroy JC, Dran G, Gonon G, Costanzo G, Dimnet J, Pasquet A. Reciprocal angulation of vertebral bodies in a sagittal plane: approach to references for the evaluation of kyphosis and lordosis. Spine. 1982;7:335–42.
- Vaz G, Roussouly P, Berthonnaud E, Dimnet J. Sagittal morphology and equilibrium of pelvis and spine. Eur Spine J. 2002;11:80–7.
- Lamartina C, Berjano P, Petruzzi M, Sinigaglia A, Casero G, Cecchinato R, Damilano M, Bassani R. Criteria to restore the sagittal balance in deformity and degenerative spondylolisthesis. Eur Spine J. 2012;21(Suppl):27–31.
- Lafage V, Schwab F, Skalli W, Hawkinson N, Gagey PM, Ondra S, Farcy JP. Standing balance and sagittal plane spinal deformity: analysis of spinopelvic and gravity line parameters. Spine. 2008;33:1572–8.
- Legaye J, Duval-Beaupère G, Hecquet J, Marty C. Pelvic incidence: a fundamental pelvic parameter for three dimensional regulation of spinal sagittal curves. Eur Spine J. 1998;7:99–103.
- Roussouly P, Gollogly S, Noseda O, Berthonnaud E, Dimnet J. The vertical projection of the sum of the ground reactive forces of a standing patient is not the same as the C7 plumbline: a radiographic study of the sagittal alignment of 153 asymptomatic volunteers. Spine. 2006;31:E320–5.
- Barrey C, Roussouly P, Le Huec JC, D'Acunzi G, Perrin G. Compensatory mechanisms contributing to keep the sagittal balance of the spine. Eur Spine J. 2013;22:S834–41.
- Korovessis PG, Stamatakis MV, Baikousis AG. Reciprocal angulation of vertebral bodies in the sagittal plane in a asymptomatic greek population. Spine. 1998;23:700–4.

- Takemitsu Y, Harada Y, Iwahara T, Miyamoto M, Miyatake Y. Lumbar degenerative kyphosis. Clinical, radiological and epidemiological studies. Spine. 1988;13:1317–26.
- Rajnics P, Templier A, Skalli W, Lavaste F, Illes T. The importance of spinopelvic parameters in patients with lumbar disc lesions. Int Orthop. 2002;26:104–8.
- Barrey C, Roussouly P, Perrin G, Le Huec JC. Sagittal balance disorders in severe degenerative spine. Can we identify the compensatory mechanisms? Eur Spine J. 2011;20 Suppl 5:626–33.
- 23. Korovessis PG, Dimas A, Iliopoulos P, Lambiris E. Correlative analysis of lateral vertebral radio-graphic variables and medical outcomes study short-form health survey: a comparative study in asymptomatic volunteers versus patients with low back pain. J Spinal Disord Tech. 2002;15:384–90.
- Schuller S, Charles YP, Steib JP. Sagittal spinopelvic alignment and body mass index in patients with degenerative spondylolisthesis. Eur Spine J. 2011;20:713–9.
- Lamartina C, Berjano P. Classification of sagittal imbalance based on spinal alignment and compensatory mechanisms. Eur Spine J. 2014;23(6):1177–89.
- Barrey C, Jund J, Perrin G, Roussouly P. Spino-pelvic alignment of patients with a degenerative spondylolisthesis. Neurosurgery. 2007;61:981–6.
- Mangione P, Sénégas J. Sagittal balance of the spine. Rev Chir Orthop Reparatrice Appar Mot. 1997;83(1): 22–32. French.
- Obeid I, Hauger O, Aunoble S, Bourghli A, Pellet N, Vital JM. Global analysis of sagittal spinal alignment in major deformities: correlation between lack of lumbar lordosis and flexion of the knee. Eur Spine J. 2011;20 Suppl 5:681–5.

# Posterior Impaction Osteotomy for Correction of Sagittal Imbalance in Iatrogenic Flat Back: Surgical Technique

49

Christian Mazel, Laurent Balabaud, and Alexandre Cogan

### 49.1 Introduction

Sagittal imbalance can be defined as the inability to stand erect without compensatory flexion of hips and knees. Loss of lumbar lordosis, often associated to increased thoracic kyphosis, is the most characteristic presentation of this biomechanical disorder, which is often named "flat back." Muscle degeneration is usually present [1]. Several degenerative, inflammatory, and infectious pathologies have been recognized to lead to this clinical picture. Flatback posture is also the common result of distractive instrumentation used in the lumbar spine for a myriad of spinal disorders. Lumbar fusion for scoliosis or other degenerative disorders is a well-established iatrogenic cause of flatback deformity [2] and failed spine surgery. The first line of management is nonoperative treatment that, unfortunately, is often not effective [3]. Most patients with this disabling condition often

Department of Orthopedic Surgery and Spine, Institut Mutualiste Montsouris, 42 Boulevard Jourdan, Paris 75014, France e-mail: cristian.mazel@imm.fr

L. Balabaud, MD • A. Cogan Department of Orthopedic Surgery, Institut Mutualiste Montsouris, Paris, France require an osteotomy to restore a satisfying lumbar lordosis. Two osteotomy techniques are currently performed: Smith-Petersen osteotomy [4] and the transpedicular closing wedge osteotomy frequently attributed to Thomassen [5, 6].

# 49.2 Clinical and Radiological Assessment of Sagittal Imbalance

The most common symptoms reported by patients with flatback syndrome are buttock pain, calmed down by a sitting position, back pain, radicular pain or neurogenic claudication, and neck pain.

Neck pain is largely due to cervical spine extension to maintain horizontal gaze. Radiculopathy and neurogenic claudication could be explained by disk and facet joint complex degeneration causing hypermobility at involved levels. The appearance of a forward pitching position, with insufficient knees flexion to compensate for sagittal imbalance, is a typical finding on clinical assessment (Fig. 49.1). A thorough neurological assessment is required, along with various functional outcome measures such as ODI, SRS, and possibly JOA scores. Hips and knees joint range of motion should be carefully measured.

Radiological assessment of sagittal balance (Fig. 49.2a, b) is performed using full-length lateral radiography, measuring spinal parameters

C. Mazel, MD, PhD (🖂)

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_49



Fig. 49.1 Clinical presentation of a sagittal and coronal imbalance



**Fig. 49.2** (a) Full-length radiography demonstrating a flatback posture post a Harrington procedure. (b) Full-length radiography demonstrating a flatback posture following a multi-segmental scoliosis procedure Pr and post-op

such as lordosis, kyphosis (Cobb's angle), sagittal tilting, and pelvic parameters such as incidence, sacral slope, sacrofemoral tilting, and overhang [7]. Two additional parameters to those previously defined also include regional kyphosis (angle between the superior end plate of the vertebra above the planned osteotomy and inferior end plate of the vertebra below) and the sagittal vertical axis also referred to as the plumb line (distance between a projected vertical line drawn from the center of C7 and the anterosuperior end plate of the sacrum). If the projection passes in front of the anterior aspect of the sacrum, the measure is positive. Van Royen et al. used the posterosuperior end plate of the sacrum as a second anatomical landmark to evaluate sagittal vertical axis [8].

The use of various methods to assess lumbar lordosis makes comparing radiological outcomes between series difficult. The large variability of sacral inclination, for example, makes the superior end plate of S1 an unreliable landmark [9, 10]. In our series, we measured lumbar lordosis from the superior end plate of L1 to inferior end plate of L5.

# 49.3 The Different Surgical Techniques

# 49.3.1 Smith-Petersen Osteotomy (SPO)

The first cases of osteotomy reported by Smith-Petersen et al. in 1945 were for post-ankylosing spondylitis deformity. This technique (Fig. 49.3) consists of posterior element subtraction, taking away the spinous process, lamina, and facet joints on one or more levels. Correction is obtained through external manipulations resulting in compression of the posterior elements, opening of the disk space, and tearing of anterior longitudinal ligament. The major drawback of this technique is that it lengthens the anterior column, potentially injuring the spinal cord and tearing the great vessels. Lengthening of the anterior column has been measured by Scudese and Calabro who noted 1.7 cm of lengthening for a 40° correction [11]. Moreover, closure of the posterior elements may compress the foramen, causing radiculopathy. The average correction that can be expected is  $5-10^{\circ}$  per osteotomy level [12].



Fig. 49.3 Smith-Petersen osteotomy (SPO) is lengthening the anterior column of the spine

#### 49.3.2 Ponte Osteotomy

The *Ponte-type osteotomy* was first described by Ponte et al. in 1984 for Scheuermann kyphosis although the terminology only entered the US literature in 2007 [13]. The Ponte osteotomy was described as wide segmental osteotomies followed by posterior compression along unfused regions of the deformity in Scheuermann patients.

The authors are mostly speaking of multilevel facet and lamina local osteotomy that by their multilevel situation corrects the sagittal deformity. The surgical technique is a multilevel reshaping of the canal associate to an inferior and medial facetectomy, with ligamentum flavum resection (Fig. 49.4).

Although today the terms Smith-Petersen and Ponte are often used interchangeably, the technique currently used is Alberto Ponte's. In addition, SPOs have become a mainstay in the correction of coronal deformities, such as adolescent idiopathic scoliosis; however, they were not originally described for this indication.

### 49.3.3 Posterior Subtraction Osteotomy (PSO)

Corporeal subtraction osteotomy consists of a wedge osteotomy of the vertebral body performed through a posterior approach (Fig. 49.5). It allows shortening of the posterior column without lengthening of its anterior aspect, thus sparing the neural elements. Moreover, it gives a larger surface of compressive bone contact than the Smith-Petersen technique, thereby enhancing stability and fusion.

Comparison between Smith-Petersen osteotomy (SPO) and pedicle subtraction osteotomy (PSO) underscores major differences. The SPO group tends to produce more coronal decompensation and requires multiple levels of osteotomy for substantial correction [13]. There is a higher risk of vascular injury as well as a higher rate of nonunion because the opening is through the disk space. The bulk of axial loads in the standing position passes through the anterior column, so opening of the disk space predisposes to



**Fig. 49.4** Ponte osteotomy. Multilevel resection of the medial part of the *upper* and *lower* facets. Correction is spread on the different adjacent levels



Fig. 49.5 Posterior subtraction osteotomy (PSO) shortens the spinal column

nonunion [14]. A PSO on the other hand is associated with a significantly higher blood loss and a higher union rate [15, 16].

### 49.3.4 Posterior Impaction Osteotomy (PIO)

These results and complications explain our surgical evolution and desire to minimize bleeding and to simplify the osteotomy procedure. The development of the posterior wedge intracorporeal impaction osteotomy is to us a good way to obtain such results.

The main principles of this technique are to accomplish the posterior closing osteotomy by intracorporeally impacting the cancellous bone of the posterior vertebral body wedge. The apex of the osteotomy must be as close as possible to the anterior cortex while preserving a hinge.

The patient is placed in the prone position on an Allen<sup>®</sup> radiolucent carbon surgical table (Fig. 49.6a, b). The chest is supported by two thoracic mobile stands (up and down) while the iliac crest and pelvis are positioned on a fixed one; the fixed lower limbs are positioned along the body's axis. This setup leaves the abdomen free from compression, thereby decreasing bleeding, and favors the natural restoration of the lumbar lordosis by hyperextending the lumbar spine.

A posterior incision is made. Muscle must be handled with care because of its importance in spinal stability and injuries caused by previous surgery. Dissection is gently done with a knife. Knife dissection is preferred because it decreases the risk of dural tears, especially if a laminectomy was performed on previous procedures, and of muscle necrosis in multi-operated cases. Monopolar cautery may be used to dissect soft tissue off any spinal instrumentation. The surface anatomy is thoroughly cleaned of all soft tissue in order to identify essential anatomical landmarks before starting the osteotomy. Intraoperative radiographic control is used to verify the level of the selected vertebra. Anatomical limits of bone resection, depending on the amount of correction, are represented by two lines drawn with monopolar cautery on either side, perpendicular to the spinal axis of pedicles screw holes easily visible after instrumentation removal. Instrumentation length is tailored to each individual case, taking various factors into account (quality of bone, adjacent pathologies, topping of syndrome, nonunion, etc.). A minimum of two levels over and under the osteotomy level is recommended. Bone resection consists of a laminectomy at the selected level and cephalad vertebral level including the target level pedicles, facet joints, and transverse processes. The neural elements, i.e., the nerve roots above and below the pedicle of interest, are protected by cotton pads. The pedicle subtraction is performed using a high-speed diamond drill. This results in a large foramen whose contents include the two exiting nerve roots, maintained in distraction using a distraction clamp (Fig. 49.6a, b). Bleeding from epidural veins can be controlled by haemostatic agents (Surgicel<sup>®</sup>, Surgiflo<sup>®</sup>, or others) or bipolar cautery if necessary. Corporeal osteotomy starts on the posterior wall, using two osteotomes. The



**Fig. 49.6** (a) Radiolucent table: Belly is entirely free of all anterior compression. (b) Radiolucent table giving a lower limb hyperextension capacity



**Fig. 49.7** Peroperative view of the bone scissors before impaction of the wedge-shaped part of the vertebral body (By courtesy of JM Vital)

extent of posterior body resection is determined by the obliquity required to affix the osteotomy margins just behind the anterior vertebral cortex, leaving an anterior bony hinge that prevents forward translation. Osteotome progression inside the vertebral body is controlled by fluoroscopy (Fig. 49.7).

There are two major differences between the PIO and PSO techniques at this step of the procedure. With the PSO technique, it is recommended to progressively remove the cancellous bone from the vertebral body, creating a wedge-shaped void giving place for future closure and correction. With the PIO technique, the cancellous bone, as well as the posterior wall of the vertebral body, is impacted inside the vertebral body. Impacting the bone instead of removing it decreases intraoperative bleeding. It also simplifies the procedure. Bone impactors have been designed specifically for this purpose. They are long enough to deal with obese patients and narrow enough to be easily handled up to the anterior extent of the osteotomy (Fig. 49.8). The cancellous vertebral body bone is progressively impacted out of the osteotomy site, alternating on both sides. This is the first step of the osteotomy procedure.

The second step of wedge creation is the cutting of the lateral vertebral body wall. In the PSO technique, you elevate the lateral soft tissues and then cut the lateral wall with an osteotome. Control of the lumbar segmental vessels is difficult and local bleeding can be voluminous. In



Fig. 49.8 Bone impactors of different sizes enable to compact the bone inside the vertebral body, impaction posterior osteotomy (IPO)

PIO you weaken the lateral edges of the vertebral body with the bone impactor. The tip of the impactor is manipulated from inside the vertebral body, weakening the lateral wall without injuring the lumbar vessels.

The last step of the decancellation is the resection of the vertebral median posterior wall just in front of the dural sac. It is the last hurdle to closing the wedge. The resistance of this median posterior cortical wall depends on the patient's bone quality. In most cases the same weakening technique used to deal with the lateral wall is employed. A smaller bone impactor is used. The instrument is inserted obliquely from both sides in front of the dural sac and a progressive impaction maneuver will break it down.

The wedge is now complete, bleeding is controlled by cancellous bone impaction, and posterior closure is obtained by progressively releasing the distraction clamp (Fig. 49.9a, b). Satisfying lumbar lordosis is obtained in most cases at this point. In stiffer patients, in order to complete the posterior wedge closure, it is necessary to perform an upward translation of the thoracic platform and a hyperextension at the site of the osteotomy by raising the lower limbs until the upper and lower osteotomy limits contact bone on bone. A thorough evaluation of the posterior decompression is assessed to make sure there is no dural impingement or foraminal compression. Careful inspection of each foramina and nerve root is mandatory all along the closure. Root entrapment or impingement is the most frequent possible complication. The procedure is completed by local decortication and posterolateral graft application using locally harvested morsellized bone chips. Mechanical



**Fig. 49.9** (a) Spreading Meary-type clamp. Mostly applied directly on the bone margins, they avoid pedicle screw mobilization of other types of distracting clamp. (b) Pedicle-type distracting clamp

stability is obtained by strict patient fixation to the operative table, thus avoiding instrumentation at this time of the procedure. Spinal fixation is then performed spanning at least two levels above and two levels below the osteotomy. This is possible due to the stability of the patient on the operative table. We use rigid instrumentation, favoring highly rigid chromium cobalt alloy rods to titanium ones. Finally, lumbar lordosis is obtained using the operative table. The desired posture is maintained by placing screws in previous pedicle trajectories and using precontoured adapted rods (Fig. 49.9a, b). There is no need to perform in situ bending. Two suction drains are placed into the epidural space and the wound is closed in layers. The patient is kept supine for the first day postoperatively and then mobilized on the second post-op day with a rigid Boehler-type brace which is used for walking and standing for the first 3 months. Sitting is not allowed for the first 45 days for osteotomies lower than L3.

# 49.3.5 PIO with Partial Pedicle Resection

The complete resection of the pedicle is the standpoint of PSO or IPO techniques; in some cases where the needed correction is lower than  $20^{\circ}$ , it is possible to only remove the lower two-



Fig. 49.10 Partial IPO with preservation of the upper part of the pedicle (*Circle*). Only one foramina is open preserving the adjacent one and decreasing morbidity related to root impingement

third of the pedicle. It avoids violating the upper foramina, thus decreasing morbidity, and shortens the surgical time (Fig. 49.10).

# 49.3.6 Choice of Technique

The choice Ponte osteotomy and PSO/PIO mostly depends on deformity reducibility. Correction of the sagittal imbalance is assessed on lateral full spine X-rays (EOS® system if possible) and flex-ion-extension films (Fig. 49.11a, b).

Multilevel deformities are more often addressed by a Ponte osteotomy. Stiff deformities will need PSO or PIO; reducible ones will be addressed with Ponte osteotomy.

The axial extension of the deformity is also considered. A localized deformity is more efficiently dealt with by a PSO or PIO (Fig. 49.12).



Fig. 49.11 (a) Reducible kyphotic deformity shown on the lateral film in supine position more than on flexionextension lateral films. (b) Post-op films after Ponte multilevel osteotomy

### 49.3.7 Choice of Level in PSO and PIO

Multiple factors (type and level of pathology, local modifications) weigh on the choice of the osteotomy level. The indication is tailored to each pathology and deformity.

Lower levels carry less neurological risk (cauda equina versus cord), and the lower the level from L1 to L5, the higher the correction angle, but the more challenging a stable fixation. With an L5 osteotomy, both transiliac and sacral screws are needed for stability.

The best compromise is L4: low enough to give a good correction, high enough to spare enough levels under the osteotomy to maintain mechanical stability.

In previously operated cases, local conditions can modify this strategy. Priority can be given to a level free of dural scar with no previous vertebral canal exploration or to a nonunion.

# 49.4 Case Series: Results and Discussion

This retrospective series includes 49 patients who underwent corrective surgery for functional sagittal imbalance using a closing wedge osteotomy by intracorporeal impaction over an 8-year period from July 1999 to December 2011.

The mean age was 54 years (range 21–84 years). There were 40 women and 9 men.

All but one patient had previously undergone spinal fusion. The average number of prior spine interventions was 2, with a maximum of 7. The initial spine surgery was a fusion for thoracic or thoracolumbar scoliosis (with sacral fusion) for 23 patients, an instrumented fusion for degenerative indications for 25 patients, and a laminectomy for 1 patient. The average period separating the last spinal procedure and the osteotomy was 47 months (ranging from 12 to 132 months).



Fig. 49.12 Pre-op kyphotic deformity of L3 corrected by a PSO. Radiological and clinical outcome

Preoperative X-ray finding is presented in Table 49.1.

In our series of osteotomy for iatrogenic flatback deformity (Fig. 49.10), follow-up shows clinical improvement in 90 % of patients. Intraoperative data revealed a mean operative time of 208 min (range 120–360) and a mean blood loss of 1856 ml (range 650–4000). The mean hospital stay was 12 days (range 6–34). We used Cell Saver® autologous transfusions to compensate for the vast majority of the peroperative blood loss (Table 49.2). Seventeen patients required additional postoperative heterologous transfusion.

Intraoperative incidents included a dural tear in 11 patients. All were stitched and hermetically closed. Patients with a dural tear were kept in bed the first postoperative day, and non-aspirate drainage was left in place 24 h. There was no cerebrospinal fluid leakage or meningocele. There was no nerve root injury.

Postoperative complications (Table 49.3) included six transient motor nerve root deficits (12.2 %), five painful transient nerve root irritations (10.2 %), and one superficial wound infections (2 %). There was one tragical case of higher level postoperative paraplegia, without any direct relation with the osteotomy site. It is not included in these data. Instrumentation extended up to T3 level in this patient, and early revision of a malpositioned screw in the upper part of the osteosynthesis did not bring neurological improvement.

There was one case of perioperative collapse of the anterior hinge, resulting in a small translation without any neurological consequences.

	Normal values	Average	Minimum	Maximum
Thoracic kyphosis	45°	27.1°	0°	62°
Lumbar lordosis	61°	12.2°	-13°	24°
Incidence angle	53°	62.9°	39°	90°
Pelvic tilt	12°	37.8°	13°	60°
Sacral slope	41°	25.1°	13°	40°
Regional kyphosis	NA	12.6°	-8°	28°
Sagittal alignment	11°	7.1°	-4°	15°
Overhang	23 mm	64.5 mm	25 mm	100 mm
Sagittal vertical axis	0 mm	96.5 mm	20 mm	180 mm

Table 49.1 Preoperative radiological data

 Table 49.2
 Operative time and blood loss

Items	Mazel et al.	Bridwell et al. [19]	Chiffolot et al. [20]	Boachie-Adjej et al. [21]	Ikenaga et al. [17]
Operative	208	750	260	304	277
time(min)					
Blood loss(ml)	1856 <sup>a</sup>	2400	1400	2700	1988

<sup>a</sup>Cell saver autotransfusion is not included

 Table 49.3
 Operative complications of lumbar osteotomy

Complications	Mazel et al.	Bridwell et al. [19]	Chiffolot et al. [20]	Boachie-Adjej et al. [21]	Ikenaga et al. [17]	Kim et al. [22]
Nbre of patients	49	33	34	24	67	45
Dural tears	11	2	5	4	1	-
Infection	1	-	2	2	2	-
Neurol. deficit	6	6	-	0	2	5
Radicular pain	5	-	4	1	1	4
Pseudoarthrosis	2	8	8	3	7	-
Addit. ant. surg	2	14	-	7	7	-

Instrumentation corrected the displacement and resulted in fusion.

An anterior approach was deemed necessary in one case. This patient had an instrumented T5– L3 fusion using Harrington rods for thoracolumbar scoliosis associated with an L5 isthmic spondylolisthesis. Due to decompensation of the adjacent levels, the fusion was elected to be extended to the sacrum. An L4 osteotomy was necessary in order to compensate for the associated previous thoracolumbar flatback deformity. Resection of the L5 posterior arch as part of the procedure (Gill procedure), resulting in local instability, indicated a complementary anterior approach in order to fuse the L4–L5 and L5–S1 interspaces.

A complementary anterior approach should be associated with a transpedicular osteotomy every time posterior instrumentation alone appears insufficient for optimal stability. This complementary approach may be indicated in cases of severe osteoporosis, correction through the disk space, or insufficient correction. In two patients, the correction was mostly at the level of the intervertebral disk and not in the vertebral body itself, but the osteosynthesis was stable enough. In these cases we neither have to harvest additional bone graft nor use an anterior cage.

Two cases of nonunion were observed.

The first patient had Forestier disease with a kyphotic lumbar deformity and lumbar stenosis. He had previously undergone a laminectomy and instrumented fusion from T11 to S1. His postoperative flat back was corrected by an L4 posterior

transpedicular wedge impaction osteotomy. Rod failure due to nonunion occurred 6 months after surgery. The revision consisted of posterior decortication, bone grafting, and new fixation with an anterior L3–L4 fusion during the same procedure.

The second patient had previously undergone a laminectomy and instrumented fusion from L3 to L5 for a lumbar stenosis. Her postoperative flat back was corrected by an L3 posterior transpedicular wedge impaction osteotomy with a T9 to sacrum fusion. She was revised 33 months later for a union at the lumbosacral junction, with an anterior L5–S1 fusion.

Impaction bone grafting preserves cancellous bone and in our opinion reduces blood loss and nonunion. Preservation of the anterior cortical bony hinge decreases the risk of anterior translation with nerve entrapment and probably enhances local stability and fusion rates. The Allen<sup>®</sup> operative table with its thoracic and pubic platform configuration allows minimal abdominal compression and decreases intraoperative bleeding. Moreover, it assists in osteotomy closure following the osteotomy [17, 18].

Postoperative radiological outcomes are presented in Table 49.4.

Many heterogeneous case series of flatback deformity have been reported with PSO. Globally, the average correction for a one-level PSO is approximately 30° [17, 19–22]. In our series, the average improvement in regional kyphosis was 19.6°. The average improvement in lumbar lordosis was 23.5°, less than quoted in

Items/mean correction	Mazel et al.	Bridwell et al. [19]	Chiffolot et al. [20]	Boachie-Adjej et al. [21]	Ikenaga et al. [17]	Kim et al. [22]
Number of patients	49	33	34	24	67	45
Thoracic kyphosis	8.2°	12.2°	-	10°	6.3°	4
Lumbar lordosis	23.5°	33°	31.5°	41°	35.2°	34
Regional lordosis	19.6°	28°	28.9°	-	34.2°	-
Sacrofemor. tilting	-7.8°	-	-16°	-	-	-
Sacral slope	7.5°	-	8	-	-	16°
Sagittal alignment	3.5°	-	-	-	-	-
Overhang (mm)	12.2	-	-	-	-	-
Vertical sagittal Axis (mm)	-70.9	-14.9	-50	-108	-17	-86

**Table 49.4** Summary of mean correction obtained in different series

the literature. This difference may be due to the cause of the deformity. As far as we know, our series is the largest one reported regarding iatrogenic flat back. Furthermore, we did not try to maximize correction but rather to recreate a harmonious balance of the spine [17, 23]. The wide range of normal values of lumbar lordosis implies that each person has its own means of self-balance. We therefore believe that the most reliable criteria to assess correction of sagittal alignment are the vertical sagittal axis [24]. Choosing the osteotomy level is still a matter of debate among authors. Some [24-26] recommend that it should be performed in the lower lumbar spine in order to maximize spinal sagittal correction and avoid spinal cord injury. Others [17] do not hesitate to make osteotomies at the thoracic level.

In addition to the potential for direct trauma during an osteotomy procedure, the procedure itself may lead to loss of blood flow to the spinal cord if performed at the spinal cord level [27]. Bridwell et al. [19] prefer to avoid performing a PSO in close proximity to conus medullaris [19, 21], while Booth et al. [28] perform their osteotomies above L1 at the apex of the deformity. Lagrone et al. [29] prefer to perform the osteotomy at a nonunion level. As a preexisting nonunion may predispose to recurrent postoperative deformity at the same site, it may be the optimal level to improve bone fusion by compression and instrumentation.

In our series, the osteotomies were always performed in the lower lumbar spine, mostly at L4 (Fig. 49.11). We also believe that there is less risk of neurological injury at that level, especially in multi-operated cases. Moreover, an osteotomy at this level allows for the greatest correction of alignment; with the same angular correction, the translational effect on the upper body and head is greater if the rotational arc is longer and thus if the vertebral osteotomy is at a lower level. Such a distal level raises the issue of stable fixation: we routinely use four points of fixation in the sacrum, two screws in S1 and two laterally directed screws in S2. The choice of osteotomy level is also influenced by the presence of an underlying nonunion.

When using a Smith-Petersen osteotomy, it is taught that for every 1 mm of bone resection, the deformity may be corrected by 1° [30]. Wang and Berven [31] recommend preoperative planning by taking a long cassette lateral X-ray and cutting a wedge out on tracing paper, aiming for a final lumbar lordosis of 30° along with moving the plumb line to the posterior sacrum. Pigge et al. [32] reported the results of an eight patient series where they used preoperative surgical planning assessing all pelvic parameters aiming for an optimal sagittal vertical axis as defined by Van Royen et al. [8]. To accomplish this they used the assistance of a computer program called ASKyphoplan to choose the level of osteotomy and degree of correction, correlated to the parameters mentioned above.

We suggest a preoperative planning method using a standing full spine sagittal X-ray. We draw the anterior hinge on the chosen vertebra and then a first osteotomy cut line. Then we cut the X-ray along this line but keep the part of the anterior hinge intact. By rotating the upper part of the X-ray around this hinge, we reposition the spine in the desired position through the use of the vertical sagittal axis. Once the sagittal imbalance is corrected, we just measure the required angle of osteotomy necessary for obtaining this correction. We also measure the angle between the superior and inferior end plates of the osteotomized vertebra which we then compare with the peroperative X-rays. Once the osteotomy closure is finished, the radiographic measurement of the angle between the upper and lower end plates of the osteotomized vertebra indicates if there is enough correction.

#### Conclusion

Taking into account the various parameters of sagittal spinal imbalance is paramount to treat a symptomatic deformity and restore balance.

Pedicle posterior wedge osteotomy is a technically demanding procedure. There are significant potential complications. Pedicle impaction osteotomy may alleviate those complications. Late complications such as nonunion often lead to further challenging surgery.

### References

- Lee JC, Cha JG, Kim Y, Kim Y-II. Quantitative analysis of back muscle degeneration in patients with the degenerative lumbar flat back using a digital image analysis. Spine. 2008;33:318–25.
- Doherty JH. Complications of fusion in lumbar scoliosis. J Bone Joint Surg (Am). 1973;55:438.
- Farcy JP, Schwab FJ. Management of flatback and related kyphotic decompensation syndromes. Spine. 1997;20:2452–7.
- Smith-Petersen MN, Larson CB, Aufranc OE. Osteotomy of the spine for correction of flexion deformity in rheumatoid arthritis. J Bone Joint Surg (Am). 1945;27:1–11.
- Thomassen E. Vertebral osteotomy for correction of kyphosis in ankylosing spondylitis. Clin Orthop. 1985;194:142–52.
- Thiramont N, Netrawichen P. Transpedicular decancellation closed wedge osteotomy for treatment of fixed flexion deformity of spine in ankylosing spondylitis. Spine. 1993;18:2517–22.
- Legaye J, Hecquet J, Marty C, Duval-Beaupere G. Equilibre sagittal du rachis. Relations entre basin et courbures rachidiennes sagittales en position. Rachis. 1993;5:215–26.
- Van Royen BJ, De Gast A, Smith T. Deformity planning for sagittal plane corrective osteotomies of the spine in ankylosing spondylitis. Eur Spine J. 2000;9:492–8.
- Stagnara P, De Mauroy JC, Dran G, Gonon GP, Costanzo G, Dimnet J, Pasquet A. Reciprocal angulation of vertebral bodies in the sagittal plane: approach to references for the evaluation of kyphosis and lordosis. Spine. 1982;7:335–42.
- Propst-Proctor SL, Bleck EE. Radiographic determination of lordosis and kyphosis in normal and scoliotic children. J Pediatr Orthop. 1983;3:344–6.
- Scudese VA, Calabro JJ. Vertebral wedge osteotomy. Correction of rheumatoid (ankylosing) spondylitis. JAMA. 1963;186:627–31.
- Yang BP, Ondra SL, Chen LA, Jung HS, Koski TR, Salehi SA. Clinical and radiographic outcomes of thoracic and lumbar pedicle subtraction osteotomy for fixed sagittal imbalance. J Neurosurg Spine. 2006;5:9–17.
- Geck MJ, Macagno A, Ponte A, Shufflebarger HL. The Ponte procedure: posterior only treatment of Scheuermann's kyphosis using segmental posterior shortening and pedicle screw instrumentation. J Spinal Disord Tech. 2007;20:586–93.
- Bergmark A. Stability of the lumbar spine. A study in mechanical engineering. Acta Orthop Scand. 1989;230(Suppl):1–54.
- Cho KJ, Bridwell KH, Lenke LG, Berra A, Baldus C. Comparison of Smith-Petersen osteotomy versus pedicle subtraction osteotomy for correction of fixed sagittal imbalance. Spine. 2005;30:2030–7.
- Potter BJ, Lenke LG, Kuklo TR. Prevention and management of iatrogenic flatback deformity. J Bone Joint Surg (Am). 2004;86:1793–808.

- Ikenaga M, Shikata J, Takemoto M, Tanaka C. Clinical outcomes and complications after pedicle subtraction osteotomy for correction of thoracolumbar kyphosis. J Neurosurg Spine. 2007;6:330–6.
- Willem KF, Slot GH, Anderson PG, Pavlov PW, De Kleuver M. Spinal osteotomy in patients with ankylosing spondylitis: complications during first postoperative year. Spine. 2004;30:101–7.
- Bridwell KH, Lewis SJ, Edwards C, Lenke LG, Iffrig TM, Berra A. Complications and outcomes of pedicle subtraction osteotomy for fixed sagittal imbalance. Spine. 2003;28:2093–101.
- Chiffolot X, Lemaire JP, Bogorin I, Steib JP. Pedicle closing-wedge osteotomy for the treatment of fixed sagittal imbalance. Revue de chirurgie Orthop. 2006;92:257–65.
- Boachie-Adjej O, Fergusson JA, Pigeon RG, Peskin MR. Transpedicular lumbar wedge osteotomy for fixed sagittal imbalance: surgical technique and early results. Spine. 2006;31:485–92.
- Kim KT, Suk KS, Cho YJ, Hong GP, Park BJ. Clinical outcome results os pedicle subtraction osteotomy in ankylosing spondyitis with kyphotic deformity. Spine. 2002;27:612–8.
- Wiggins GC, Ondra SL, Shaffrey CI. Management of iatrogenic flat-back syndrome. Neurosurg Focus. 2003;15:article 8.
- Mazel C, Zrig M, Antonietti P, De Thomasson E. Impaction posterior wedge osteotomy for the treatment of post surgical flatback. Revue de chirurgie Orthop. 2005;91:530–41.
- Von Royen BJ, Slot GM. Closing-wedge posterior osteotomy for ankylosing spondylitis. J Bone Joint Surg (Am). 1995;77:117–21.
- Min K, Hahn F, Leonardi M. Lumbar spinal osteotomy for kyphosis in ankylosing spondylitis. The significance of the whole body kyphosis angle. J Spinal Disord Tech. 2007;20(2):149–53.
- Kahawara N, Tomita K, Kobayashi T, Abdel-Wanis ME, Murakami H, Akamaru T. Influence of acute shortening on spinal cord: an experimental study. Spine. 2005;30:613–20.
- Booth KC, Bridwell KH, Lenke LG, Baldus CR, Blanke KM. Complications and predictive factors for the successful treatment for flatback deformity (fixed sagittal imbalance). Spine. 1999;24:1712–20.
- Lagrone MO, Bradford DS, Moe JH, Lonstein JE, Winter RB, Ogigview JW. Treatment of symptomatic flatback after spinal fusion. J Bone Joint Surg (Am). 1988;70:569–80.
- Bridwell KH, Lenke LG, Lewis SJ. Treatment of spinal stenosis and fixed sagittal imbalance. Clin Orthop. 2001;384:35–44.
- Wang MY, Berven SH. Lumbar pedicle subtraction osteotomy. Oper Neurosurg. 2007;60:140–6.
- Pigge RR, Scheerder FJ, Smit TH, Mullender M, Van Royen J. Effectiveness of preoperative planning and view in ankylosing spondylitis. Neurosurg Focus. 2008;24(1):1–5.

Part IX

Lessons from a Life

# Rational Evaluation and Management of the Patient with Spinal Pain

50

Donlin M. Long

## 50.1 Introduction

My views on the management of patients with spinal pain have been formed by a group of diverse influences. These began by training in the traditional neurosurgical approaches to the spine concentrating upon the truly herniated lumbar or cervical disk. My mentor and long-time friend Dr. Shelley Chou began collaborating with orthopedic surgery for spinal decompression in complex repairs of the spine in scoliosis. I learned a team approach with neurosurgical skills with the nervous system and orthopedic emphasis upon bone and cartilage combined. I chaired the committee which oversaw the introduction of spinal stimulation for pain into medical practice [1, 2]. Thus, I began a lifelong interest in salvage for the so-called failed back syndrome patients. My current experience with these patients is more than 8000, and nearly 2500 patients have come to surgery or other treatment. With my colleagues Warren Torgerson and Mohammed BenDebba I organized and directed the National Low Back Pain Study [3, 4]. This examination of nearly 4000 patients with first-time or first recurrent

Department of Neurosurgery, The Johns Hopkins University School of Medicine, Baltimore, MD, USA e-mail: office.neuroscienceconsults@comcast.net lumbar disk disease examined the opposite end of the spinal surgery spectrum from the surgical failures. These data on nearly 12,000 patients either personally seen or studied have formulated my general approach to the patient with spinal pain [5].

There is one other influence that has been important. In 1974, I founded a Chronic Pain and Evaluation Treatment Center at Johns Hopkins and was responsible for the clinical operations until 1982. The center functioned in close collaboration with psychiatry. The behavioral and psychological issues identified and treated remain an important part of my patient evaluations [6].

My current evaluation and management of new patients are based upon the 5-year examination of new-onset low back pain with or without sciatica from the National Low Back Pain Study. From that study, I was able to learn natural history, the lack of effect of any of the currently available nonoperative therapies, and the value of surgery for the small number of patients who actually required an operation [7, 8]. From the experience with thousands of patients with failed back syndrome, I was able to understand the causes of the ongoing complaints and thus design better treatments. From the Pain Treatment Center experience, I appreciate the psychosocial factors that can play such an important role in the generation and maintenance of pain complaints as well as the value of cognitive therapies for these patients [5, 9].

D.M. Long, MD, PhD

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), Advanced Concepts in Lumbar Degenerative Disk Disease, DOI 10.1007/978-3-662-47756-4\_50

There is one other important influence in my practice. Dr. Nikolai Bogduk came to my laboratory as a medical student and convinced me of the lack of accurate anatomical understanding of the innervation of the spine. He began his studies in my anatomical laboratory and has continued the anatomical studies and dissections which allow us to understand the innervation of all of the spinal structures and thus have a much better understanding of the origins of spinal pain. I apply these anatomical lessons to practice regularly [10, 11].

### 50.2 Causes of Spinal Pain

One of the major difficulties with dealing with the problem of spinal pain is that there is little definitive evidence for what causes pain in the majority of patients [12]. Low back pain is ubiquitous and virtually the normal human condition. Over a lifetime, three fourths of adults have at least one episode of low back pain. The overall incidence of consistent complaints of low back pain is above 30 %, and at least 10 % of adults are complaining at any one time. Low back pain is one of the most common reasons why anyone sees a physician and one of the most common reasons for referral from one physician to another [13].

Many different groups of therapists treat patients. Generalists usually tell patients the problem is musculoskeletal in origin. This is repeated by all those involved with the utilization of passive physical measures and manipulation of the spine. These therapists usually believe in the musculoskeletal therapy and add correctable malalignment as a cause. Neurosurgeons have focused upon the intervertebral disk as the cause of pain and only in the recent past have begun to emphasize the biomechanical causes of spinal pain. Orthopedic surgeons have emphasized instability and disrupted biomechanics which they try to correct. Spinal surgeons today are more likely to have both skills and therefore treat both the disk and biomechanical issues together [14].

Since the majority of the acute spinal pain syndromes relent spontaneously without treatment, it is virtually impossible to ever determine the cause. Once we go beyond a point that spontaneous improvement is likely to occur, then it becomes much more feasible to actually determine a cause of pain and thus develop the most rational solution for that pain. Unfortunately, many physicians still maintain a stereotyped approach based upon a simplistic assessment of symptoms and without any of the diagnostic adjuncts which are currently available. My goal in this presentation is outline my own highly personal approach to spinal problems causing pain in hopes that this will lead others to develop similar protocols which can be expanded as new information becomes available. The emphasis will be upon pain. This is because the majority of spinal surgery is done for pain, not neurological deficit. When neurological deficits are important, they can be emphasized, but my fundamental premise is that most surgery is carried out for pain and in the absence of pain would not be required no matter what the imaging abnormalities. The few exceptions can be emphasized [3, 13, 15].

# 50.3 The Evaluation of the Patient with Spinal Pain

The first issue is always history. The importance of an accurate history cannot be overemphasized. In the National Low Back Pain Study, the 16 experts who were the investigators in the eight nationally recognized centers of the study made a correct diagnosis and predicted therapy in virtually every patient after the history was complete and before a physical examination was done or imaging studies were reviewed. The correct diagnosis and eventual treatment were chosen in over 90 % of patients, and the changes made after physical examination and imaging studies usually were related to levels rather than fundamental diagnosis. Key issues are the severity of pain, the presence of neurological complaints, the spatial and temporal characteristics of the pain as related to known anatomical radicular patterns, a history of intercurrent disease, and an assessment of the impact of the problem on the patient's life.

The physical examination is textbook and needs no reiteration here. The only important point is the lack of specificity in the physical examination for the majority of patients. We have all learned in training the triads of reflex, sensory, and motor change which specify the root involved or spinal level. In the National Low Back Study, the textbook triads occurred in less than 1 % of patients subsequently shown to have one or more lumbar roots compressed requiring surgery. The nonspecific findings of back tenderness, lack of range of motion, and focal pain to palpation had no value in diagnosis or decision making. Thus, the physical examination is rarely an important issue unless it demonstrates a significant neurological deficient which requires urgent care [7].

A history of bladder or bowel difficulty or findings of significant perineal sensory loss, sphincter disturbance, or a lower extremity neurological loss one would not want to be permanent may all be indications for urgent care.

Remember the history of intercurrent disease should focus upon the possibility of infection, trauma, or cancer as particular causes of pain and deficits.

# 50.4 Management of Acute Back Pain With or Without Sciatica

Some years ago, the Health Policy Institute formulated back pain guidelines through the consensus process with a large number of experts in the management of spinal pain contributing. The conclusions were that in the presence of acute low back pain with or without sciatica, and the absence of a history of trauma or a significant neurological deficit, no imaging of the back pain problem was required. Expectant management with symptomatic relief could be instituted for at least 1 month and if symptoms persisted, then imaging was recommended. With a history of trauma plane spine films immediately could be obtained but were not required. This is more a legal issue in the United States than a medical consideration. Early mobilization was recommended [3, 15].

Intractable pain not easily relieved, a history of trauma, a concern for intercurrent disease, or a significant neurological deficit may all be reasons for proceeding with immediate evaluation depending upon severity.

From the National Back Pain Study, we learned that even patients with classic herniated disk syndromes virtually all improved spontaneously without any therapy [3]. Thus, my approach to the patient with the herniated disk is to treat the pain and mobilize the patient quickly, and the majority will recover without intervention. If the patient has severe radicular pain which is limiting function, a local steroid injection around the root will nearly always give relief. If the patient has severe pain without such a specific radicular component, then oral steroids for 4 or 5 days will usually provide relief. From the National Low Back Pain data, it appears that nearly all patients will be improving at 1 month and most will be fully functional at that time. Nearly all will recover over 3 months. Indications for surgery become intractable pain that cannot be relieved, a significant or progressive neurological deficient, or a social situation that does not tolerate 1-3 months of incapacitation. Some patients are simply unwilling to wait and want a solution.

Another important issue is mobilization following an acute back syndrome. There is excellent data indicating that patients are benefited by early mobilization. When true disk herniation has occurred, I limit vigorous activity such as strenuous sports and heavy lifting for a minimum of 3 months and typically for 6 months until I have MRI evidence of disk resolution.

Over the years, I have operated upon 7-10 % of those patients referred to me with known acute disk herniations. The remainder recovered spontaneously [7].

# 50.5 The Value of the So-Called Conservative Therapy Measures

It is common for patients with acute and chronic low back pain to be offered physical therapy, analgesics, manipulation therapy, or employ a wide variety of nonoperative treatments which they choose for themselves. In the National Low Back Pain Study, we were able to examine the outcome of these therapies for over 2000 patients. We could not determine that any therapy including physical therapy and exercise, manipulation therapy, and acupuncture had any statistically verifiable influence upon the rate of recovery or the eventual outcome for the patient. Therefore, I use none of them in the management of patients [3, 16].

A major issue for American neurosurgeons is the virtual requirement by the majority of insurance carriers that surgery be preceded by a prolonged course of physical therapy. All the evidence we have suggest that this is a waste of time and money. On the other hand, simply waiting or providing the best symptomatic control possible will allow a substantial number of patients to recover spontaneously. Lack of understanding of this natural history has led many practitioners in the field of spinal pain to believe their specific therapies were responsible for recovery when they are simply observing the natural history of the disease [13].

# 50.6 What Do We Do with the Patient Who Fails to Recover?

Even though the majority of patients with acute spinal pain recover, there are a significant number who do not. Because of the enormous numbers of patients with spinal pain worldwide, the small percentage who do not recover remain a very large public health problem. These are the patients typically referred to spinal surgeons for evaluation. When confronted with a patient who has not made the expected spontaneous recovery, it is important to have a rational understanding of spinal pain as a problem and an equally rational evaluation system to try to determine the specific pain generators in an individual patient. It is equally important to identify present or impending neurological issues as well and to understand psychosocial factors.

# 50.7 Clinical Features of Spinal Pain

It is not my purpose in this chapter to try to define every spinal syndrome [12]. Rather it is to emphasize that the history is key in determining whether the patient requires treatment or not and usually will lead to diagnosis and guide that treatment. However, diagnoses made are frequently not specific and need to be verified and supplemented from other sources.

There are some generalities that are helpful. Local spinal pain without radiation suggests an axial problem. This may be muscle/ligament disease, it may be degenerative disk disease, and it may be from vertebrogenic sources. Radicular obviously implies root compression. pain Pseudoradicular pain cannot be differentiated clinically, and referred radicular pain is slowly being recognized as a real clinical phenomenon. It is poorly understood and cannot be completely defined at present. Nevertheless, there is good evidence that irritation of joint capsules, annulus, and posterior longitudinal ligament may produce an apparent radicular pain syndrome without obvious nerve root compression.

The history will usually localize the region of the spine where the pain originates. The severity is a key issue because treatment typically is dependent upon the influence of the pain upon the life of the patient. There is no reason to contemplate a major interventional procedure for a patient whose pain is relatively minor and easily tolerated. The history suggests the possibility of an underlying serious disease. Significant neurological loss emphasizes the need for prompt action. Once the problem is severe enough for referral to a spinal surgeon, then imaging is appropriate if it has not already been accomplished.

# 50.8 Imaging Correlations

The standard examinations are well known. Plane spinal films with oblique and dynamic views are still important. They help appreciate motion and anomalies of the spine which may be important in surgery.

The CT scan is now available with two- and three-dimensional reconstructions. These studies are important for assessment of bony anatomy.

The MRI with and without contrast is important because it gives the best views of soft tissues. The nerve roots and spinal cord with all surrounding structures can be well evaluated [8]. It is important to remember that a rare patient has low lumbar and sacral pain associated with sciatica on the basis of sacral and/or pelvic pathology. So when imaging studies are not definitive for a typical clinical syndrome, then examinations of these areas may be required with the same modalities.

The problem with imaging studies is that nonspecific degenerative changes are present in the majority of the adult population. They do not correlate well with clinical complaints. Simply finding severe degenerative changes in the lumbar spine does not suggest the patient with those changes will even have any symptoms. Minor changes can rarely be associated with specific clinical syndromes.

There are a few obvious diagnoses which correlate very well with the clinical syndrome. Such things as spinal tumor, infection, and spinal stenosis are adequately defined by imaging. The truly herniated disk and/or spinal stenosis is typically well defined also. Severe scoliosis and spondylolisthesis are satisfactorily diagnosed as well. Corrective procedures can be planned for all of such specific syndromes. However, the patient with nonspecific back pain with or without leg pain and degenerative disease without obvious canal or foraminal stenosis may need more definitive evaluation. There has been a great tendency in the past to dismiss these patients without complete exploration of the possible causes of spinal pain and the things which might be done to alleviate it.

Myelography with associated postinjection CT is still occasionally required for specialized situations. For the usual patient with back pain, the myelogram is rarely needed.

# 50.9 Diagnostic Blocks as an Adjunct in Diagnosis

The majority of patients who present with spinal pain will not have a definitive diagnosis made on the basis of history, physical examination, and imaging [17]. These measures will be adequate to diagnose almost all patients needing urgent care since progressive neurological deficit is typically related to a specific imaging finding. The large number of patients without these specific abnormalities are usually dismissed by the surgeon and relegated to ineffective modalities of treatment. I personally think that it is important they be investigated for potential interventions by going further to try to identify the causes of spinal pain. These additional steps require the use of diagnostic blockade [10, 11, 13].

The theory and utility of these blocks require some explanation. They are poorly understood by the majority of physicians even those expert in spinal problems. The theoretical basis for their application is straightforward. The first concept indicates that irritating the painful part may reproduce the pain which the patient experiences. Thus, placing a needle onto a painful joint may cause the same pain the patient suggests. Placing the needle close to an irritated nerve root will have the same effect. Thus the first phase of the procedure is to determine if placing the needle may reproduce the patient's pain. The second concept is that the anesthetization of the structure or its innervation with a local anesthetic will provide temporary respite from the pain. The relief of pain should be related to the duration of action of the anesthetic used or controlled by placebo.

It has been demonstrated that placebo blocks interposed with real blocks will provide the best selectivity and specificity. An alternative proposed by Bogduk is single blinding of the block so the patient does not know the actual structure being blocked and utilization of anesthetic agents of differing durations to assess the veracity of patient responses [13].

Thus, in a typical block situation, the patient does not know what structure is being blocked and does not know the duration of relief expected from the anesthetic. The individual performing the block should be skilled in questioning patients concerning outcome. The patient is queried concerning the production of concordant pain. That is, does the procedure reproduce the patient's usual pain? Then ideally, a third individual not directly related to the procedure or the patient's care should query the patient concerning the outcome for pain relief over the relevant time period. Bogduk and several collaborators have studied selectivity and specificity of these blocks and have demonstrated acceptable values which make them useful adjuncts for the determination of the origins of spinal pain. Placebo control blocks approach 90 % accuracy. Those without placebo control fall more in the 70 % range. A positive block (one following which pain is relieved) has greater value than a negative block (one in which pain is not relieved). Accepting the limitations, these blocks can be helpful in determining origins of pain in patients with indeterminate imaging studies and form an important part of the diagnostic capabilities of the spinal surgeon. There is another important point, however. The decision for surgery is not based upon the outcome of these blocks. The purpose of the block is to determine origins of pain to guide a reparative surgical procedure to be the most specific possible. The decision for surgery is based upon the full patient evaluation and the totality of the examinations.

### 50.9.1 Root Blocks

Individual roots may be blocked using fluoroscopic or CT control. The needle is simply placed in immediate proximity to the root, usually at a foramen, and a small amount of the local anesthetic injected. The block should be characterized by appropriate motor and sensory loss to be verifiable. The blocks are indicated when the specific roots involved in the pathological process cannot be defined or when some uncertain process such as Tarlov cyst involves the root, and the question is whether the involvement is symptomatic.

### 50.9.2 Facet Blocks

The zygapophyseal joints may be blocked by placing the local anesthetic in the periarticular region or by blocking the innervation of the joint by anesthetizing the medial branch of the posterior primary ramus.

Pain relief following blockade can lead to periarticular steroid injection, radiofrequency destruction of the medial branch, and help select levels for fusion.

## 50.9.3 Diskography and Disk Blockade

Diskography has been in use for a half century and has major value in determining painful disk levels particularly with regard to interbody fusion. The technique was originally used to identify degenerated disks, and the decision for surgery was based upon the presence of degeneration alone. That approach was discredited years ago. Now the so-called provocative diskography is much more specific. Needles are carefully placed into the nucleus with image guidance. Injections into any other structures around the spine will be painful and negate the value of the test. Once the needles are in place and position verified the disk is distended by the injection of saline. The most important part of the test is provocation of the patient's usual pain. The patient must be awake enough to respond, and the questioner must be skilled in not leading the patient to an answer. Some diskographers then determine the degree of disruption of disk anatomy by the injection of a contrast agent. When the annulus is intact, a local anesthetic can be injected, and the relief of pain is an added proof of the importance of that particular disk in the painful process. When the annulus is disrupted, injection of local anesthesia has no localizing value. Again it is important to emphasize that the decision to operate is not based upon the provocative diskogram. The purpose of the diskogram is to specify levels for a proposed procedure and sometimes to emphasize the necessity for interbody fusion.

### 50.9.4 Sinuvertebral Block

The sinuvertebral nerve enters the posterior vertebral body at the exit point of the large draining vein which empties posteriorly. This nerve innervates the posterior third of the vertebral body, the posterior longitudinal ligament, some of the dura locally, and periosteum out into the foramina. The possibility that some back pain is purely vertebrogenic in origin is being explored now. The clinical value of these blocks is uncertain, but they are done with a technique similar to vertebroplasty, and this is a new area for exploration.

### 50.10 The Decision for Surgery

When these diagnostic techniques are employed together in totality, they will eventually provide the information required to make a decision for or against a surgical procedure [18]. Surgery is proven to benefit nerve root compression and instability. Most spine surgeons believe that pain can be generated from abnormal facets in the same way that pain is generated from any other abnormal joint. Loss of resiliency, annular degeneration and tears, and abnormal stress distribution from degenerated disks are generally less widely accepted as the causes of spinal pain. Procedures specific to the disk such as interbody fusion have been based upon diskography. It has also been used to help identify the number of levels to be fused. Facet blocks have the same value and may lead to simple percutaneous treatments but may also lead to incorporation of specific joints into posterior fusion. However, the choice for surgery is never based simply upon imaging studies, except in very unusual circumstances, and is never based upon diagnostic blocks alone. The patient's complaint incapacity must first be judged seriously enough to warrant treatment, and then adequate localization of the probable pain generators by imaging and/or diagnostic blockade is required. When all these factors are concordant, then surgery is a reasonable choice [16, 19, 20].

### 50.11 Surgery on the Spine

The decision for surgery with herniated lumbar disk is usually straightforward. Patients with intractable pain unrelieved by the usual measures can go directly to surgery without the interposition of any physical therapy in my opinion. I know of no evidence of that physical therapy presents the need for surgery in these patients. For those who tolerate the pain and can afford the period of disability, spontaneous recovery is the rule and surgery is not required.

Spinal stenosis nearly always requires surgery for correction, so the only question is, are the symptoms serious enough to warrant treatment? Simple decompression is adequate in the majority of patients without demonstrated instability. In my own series, approximately 1 in 20 requires fusion from that category. Of course evidence of instability means fusion is likely indicated [21].

Patients with canal and foraminal stenosis fair just as well with surgery as those with disk herniation though the operation is somewhat more complex [21].

Fusion is required for demonstrated instability or the correction of spinal deformity when that deformity is symptomatic or threatens neurological function. Pain is not an indication for fusion unless the origins of that pain can be defined and shown to be related instability. A patient's complaint without supporting evidence is never an indication for surgery [21].

### 50.12 The Failed Back Syndrome

This term is imprecise and has no diagnostic value but is widely used. The diagnosis vaguely indicates that the patient has had previous procedures upon the spine and has failed to be improved by surgery. There are few specific diagnoses and the term is usually based upon a nebulous spectrum of therapies. The use of such a meaningless diagnosis also affects the behavior of those involved with these patients. It leads to the inaccurate perception that nothing can be done and implies psychiatric disease [16, 19, 20].

The goal in dealing with the patient who has failed previous treatments should be the same as the goal for those to be diagnosed and treated. That is, the surgeon must define the causes of the ongoing complaints and recommend specific therapies related to those specific causes. My experience with the failed back syndrome is over 8000 patients. Nearly 2500 of those have come to reparative surgery. This means that in the majority of patients, no indication for reoperation was found [22]. However, in a very large number, there was an abnormality which could be corrected with the probability of pain relief and/or improvement in neurological function. The majority of those in whom no specific generator could be identified were satisfied and continued to function with their ongoing complaints. Approximately 1000 with more serious complaints were referred for spinal stimulation, and approximately 500 were maintained in chronic pain management programs. Of those in whom specific abnormalities amenable to surgery could be identified, approximately one third were symptomatic because the original surgery had failed to correct the underlying abnormality. Another third had developed a significant direct complication of previous surgery, and the final third had developed a new problem such as transition syndrome. These possibilities should be kept in mind when designing a protocol for the study of the patient who has failed previous treatments.

That actual protocol is very similar to what will be used with a patient with first-time complaints. A history, particularly of previous treatments, is important. The physical examination provides a baseline but rarely as anything of diagnostic value. Imaging studies should include plane dynamic films, CT for bony anatomy, and MRI. CT myelography is required more frequently in these patients because of the artifact induced by the presence of fixators. Two- and three-dimensional reconstructions can overcome the fixator artifact. In evaluating these studies, the three important questions are: (1) did the original surgeon accomplish the goals of the indicated operation? (2) is there a definable complication which could explain this patient's complaint? and (3) is there a new disease which can explain the patient's complaint? [18]

Diagnostic blocks can be helpful. Individual root blocks can define painful roots to correlate with imaging abnormalities. Field blocks of hardware can identify when the fixator is the cause of the pain. Facet blocks and diskography above and below fusion areas or blocks of apparent pseudoarthroses may be helpful in making a decision about the causative abnormality. Only when a clearly reparable problem can be demonstrated should surgery be selected. In my experience, these repeat procedures are as effective as original procedures to the fourth reoperation. That is, three procedures do not impact upon the eventual outcome but beyond three outcomes will be less salutary [18].

## 50.13 Use of Pain-Relieving Procedures

There are two procedures to be used for pain control when reoperation is not feasible and symptoms are severe enough to warrant.

Spinal cord stimulation is the most popular and has the longest history [23]. The technique has been employed for more than 40 years now and in my experience is very successful given specific parameters of selection. The patient should have a definable cause of the problem for the complaint and no serious unresolved psychosocial issues. Drug misuse should not be an issue. Stimulation is better for radicular pain than for back pain, and pelvic pain is the most difficult to relieve. Patients are selected by trials of temporary stimulation and those who achieve good relief are candidates for implanted devices. The overall long-term outcome is approximately equivalent to reoperation, and spinal stimulation is nearly always an alternative if neurological loss is not an important issue [24–32].

By contrast, the implanted drug delivery systems have been much less satisfactory in my experience. I have no patient who has achieved lasting control with such a problem. Therefore, I reserve them for the last resort when all else is failed and the patient is desperate for another trial of pain control [5].

# 50.14 Psychosocial Features of Chronic Pain

The effects of chronic pain are well known [33, 34]. Demoralization, depression, and anxiety are common consequences of unrelieved pain. They
do not necessarily imply that these were primary problems or cause the pain complaint.

On the other hand, pain is a common complaint with depression and other psychosocial issues [6]. The importance of these so-called comorbidities is that they may interfere with the appropriate evaluation of the patient and that patient's outcome of therapy. Pain is a subjective complaint, and, therefore, the complaints must be valid if treatments are to be prescribed and outcomes assessed. The major point is that these comorbidities must be assessed in advance and appropriately treated lest they influence both decision making and posttreatment evaluation.

The issue of litigation and disability is important in many countries. In the National Low Back Pain Study, we identified that every patient working at the time of surgery for herniated lumbar disk returned to the same job without restriction postoperatively. For those patients involved in litigation, not one returned to work during the first 2 years after surgery. However, the anonymous outcomes assessed by every measure were identical for the two groups. The effect of litigation upon complaints and the resolution of those complaints must always be considered [19].

Malingering or exaggerations are also known among these patients. The actual incidence of true malingering is probably small but must be remembered whenever a patient appears to be exaggerating the pain complaints through pain behavior. The signs suggestive of exaggeration were described by Waddeil and those observations remain valid. Pain behavior is a red flag which needs to be considered. Florid pain behavior may be nothing more than the patient's desire to impress upon physician how important the problem is, but it may have much more serious connotations which need to be evaluated. Whenever depression, anxiety, or demoralization appear to be important as factors in the patient's complaints, it is valuable to have the patient seen by a mental health professional to deal with those issues. Florid pain behavior strongly suggests the need for psychiatric consultation and potentially for the cognitive and behavioral therapies which will be of value. In my experience with failed back syndrome, the need for these complex pain

therapies is relatively small, but the important issue is that any patient in whom important comorbidities are suspected, psychiatric and/or psychological evaluation may be helpful [35].

#### 50.15 The Tarlov Cyst

In the recent past, I have recognized that the number of patients with symptomatic Tarlov cysts may be substantially higher than is generally been suspected. These cysts were described in the 1930s, and by midcentury, it was well accepted that many were symptomatic. MRI subsequently demonstrated a number of cysts which were not symptomatic, and the appreciation that some could be was gradually lost. Now it is common for radiologists and neurosurgeons to state that these cysts are never symptomatic. We have identified groups of patients in whom the cysts are obviously symptomatic. These fall in three categories.

Dural ectasia in patients with known connective tissue disorders such as Marfan's syndrome is frequently symptomatic and produces loss of sacral root neurological function.

Dural diverticula, which may be variations of internal meningoceles, are often seen. These are large midline structures, usually confined to the sacrum, unlike the ectasias which often penetrate the sacrum and appear as presacral masses. The dural ectasias are typically found in women; the dural diverticula are typically found in men. The diverticula are often associated with progressive neurological deficit mimicking tethered cord syndrome.

Patients with large perineural cysts on individual nerve roots can also be symptomatic. The typical clinical picture is of pain in the distribution of the affected roots and appropriate accompanying neurological loss.

I have successfully operated upon these patients for many years though the numbers remain quite small (approximately 75). Recently, we have begun a study for aspiration of apparently symptomatic cysts and subsequent filling with a fibrin sealant. The outcomes are promising, but longer evaluations required to determine what the eventual outcome will be. The important issue is that large Tarlov cysts should not be dismissed as incidental findings. Some will be symptomatic and adequate treatment will relieve the patients.

## 50.16 Minimally Invasive Procedures

There are a number of procedures introduced over the many years of my practice for the treatment of low back pain, degenerative disk disease, herniated disk, and related conditions. The injection of chymopapain, an enzyme to dissolve the internal structure of the nucleus and possibly herniated disk fragments, failed because of the small incidence of fatal anaphylactic reactions and neural toxicity. A new form of this enzyme therapy without either drawback is currently being studied and may be an important new adjunct in therapy of back pain related to disk degeneration and herniation.

There have been a number of attempts to remove the nucleus of the disk percutaneously. These have included laser destruction, mechanical excision, and heat. There are no adequately controlled studies to demonstrate significant value of any of these techniques. Heat and the injection of hypertonic solutions have also been used to try to solidify existing nuclear structures and thus improve the biomechanics of the spine. No adequately controlled studies exist to verify the value of these techniques.

Perineural and periarticular steroid injections are commonly employed. It is the experience of many spinal experts that these provide immediate relief of pain in many patients with compressed nerve roots or arthritic facets. The studies are mixed and mainly flawed by lack of selectivity in the patient inclusion criteria. Nevertheless, it is the experience of many that these injections will be of immediate value in many patients with acute and some patients with chronic problems. There is virtually no evidence that the eventual clinical course of these patients will be modified, but pain relief can often be immediate and provide for more rapid mobilization of the patient. Prolotherapy which is the injection of hypertonic solutions designed to strengthen ligaments in the back has limited theoretical value in my opinion. I know of no definitive studies which demonstrate long-term value of the technique.

Radiofrequency destruction of the medial branch of the posterior primary ramus for the treatment of facet-based spine pain was introduced in the early 1970s and has proven to be very successful in a relatively small number of patients. The patient who is a candidate should have demonstrated facet arthritic changes, excellent relief of pain by temporary blockade of the innervation of those joints, and failure of periarticular steroids. About 60 % of such patients will achieve lasting relief with simple percutaneous denervation carried out on an outpatient basis [36]. Unfortunately, a relatively small number of patients appear to have pain generated almost exclusively from lumbar facets. Most have a more complicated set of pain generators.

## 50.17 Newer Concepts of Diskogenic Pain and Its Treatment

Motion and nerve root compression have been the established causes of back pain amenable to surgery. Several newer concepts are suggested to be important in the generation of lumbar pain.

Transient pain from muscle and ligament injury or reaction to injury is well known and typically relents with conservative care. Recently, abnormalities of spinal contour have been shown to be causes of chronic severe pain. Scoliosis with and without rotation has long been known to be associated with pain. Abnormalities of sagittal contour with displacements of the normal spinal center of gravity either anteriorly or posteriorly are increasingly understood to be a cause of chronic pain. These are usually related to degenerative disk disease and may occur with and without spondylolisthesis. Correction of disorders of sagittal contour and scoliosis can reduce ligamentous tension, inflammation, and muscle distortions all of which produce pain [37].

There is increasing evidence that the inflammatory products of the herniated or degenerating disk also induce pain [38]. Products of the inflammatory cascade are found in nerves, the dorsal root ganglion neurons, and even in the spinal cord neurons [39]. Nociceptors are sensitized. Normally quiescent cells are sensitized and respond to all kinds of stimulations by signaling pain. This local inflammatory response may explain why annular tears and small disk protrusions without obvious significant nerve root compression can cause apparent back and radicular pain of severe degree. It is increasingly likely that one of the reasons why locally injected steroids can be so effective in eliminating symptoms is their antiinflammatory influence upon the inflammatory cascade. This concept is supported by laboratory research but remains to be verified in humans [40]. The inflammatory products of the injured disk have been demonstrated. At present, the only practical treatment would be oral administration of antiinflammatory drugs, local measures for the relief of the inflammation, and local steroid injections, which are so often successful in relieving symptoms with acute disk injuries.

The newest field of research in diskogenic pain relief involves reconstitution of annulus and/ or nucleus with stem cells or related cellular techniques [41]. There are promising animal models. Protein growth factors, gene transfers, and cellular therapies utilizing stem and mature cells are all strategies being investigated. Tissue engineering is another possible reconstructive technique for sealing tears and strengthening the fibroblastic response. Sealants may exclude the inflammatory process from the surround. None of these have been applied to humans sufficiently to judge potential as yet [42–45].

A future scenario is likely to be injection of a nucleolytic to dissolve a disk herniation followed by injections to reconstitute the nucleus and annulus. First steps are now being taken to make this possibility a reality.

There are no data which allow the evaluation of surgical biological therapies for these degenerated but not herniated disks as yet. It seems likely that diskectomy with stabilization or replacement of the injured disk with an artificial substitute would be useful but as yet even that has not been systematically investigated for the simple degenerated disk. The biological alternatives would be minimally invasive and thus even more appealing.

#### 50.18 The Guidelines of My Spinal Practice

Low back pain is a common adult complaint throughout the developed world. The majority of acute low back pain episodes will relent spontaneously. Evaluation is not needed immediately if there are no red flags. A small number of patients with acute disk herniation will require surgery based upon severity of pain and real or impending neurological deficits. Many can be relieved of radicular pain by perineural steroid injection and/ or a short course of oral steroids. Rapid mobilization and return to function are helpful. A small number of patients will not improve after 1-3 months, and surgery for demonstrated pain generators is highly successful. Major indications for surgery currently are instability and neural compression. The role of disk degeneration, facet arthritis, and skeletal deformity continues to be explored, but corrective procedures for all three are often of value. The majority of patients with degenerative spinal disease do not need fusion. Simple posterior fusion is often satisfactory, and fixators should be employed for specifically defined purposes. The presence of back pain is not an indication for fusion alone. Patients who fail procedures should be reevaluated to be certain that the original procedure was successful, no complication has occurred, and no new disease has developed. Those with serious psychosocial and behavioral issues should be referred for comprehensive evaluation and treatment. Pain management with narcotics will be useful for some who might otherwise be disabled. The majority can be managed with nonnarcotic analgesics, restriction of activities, and education concerning the realities of a chronic pain back problem. There are many unproven therapies available for these people, and it is important that they all understand what is accepted therapy,

what is proven to be a value, what is investigational, and what is of no proven value. The socalled conservative measures commonly employed in back pain may provide immediate improvement in symptoms but have no lasting effect upon the natural history of the problem. Many acute patients in whom surgery might be contemplated will escape operation if surgery is delayed for 1-3 months following onset of symptoms. A requirement for physical therapy prior to surgery is without merit based upon my experience and literature reviewed. The key issue is that surgery corrects specific abnormalities and should be reserved for those individuals in whom those specific abnormalities can be demonstrated. Surgery for symptoms is not indicated and will rarely be successful. Surgery for the correction of abnormalities should demonstrated relieve related symptoms.

#### References

- Long DM. Electrical stimulation for relief of pain from chronic nerve injury. J Neurosurg. 1973;39(6): 718–22.
- Long DM, Hagfors N. Electrical stimulation in the nervous system: the current status of electrical stimulation of the nervous system for relief of pain. Pain. 1975;1:109–23.
- Long DM, BenDebba M. Diagnosis and management of low back pain, chapter 51. In: Asbury AK, McKhann GM, McDonald WI, Goadsby PJ, McArthur JC, editors. Diseases of the nervous system. Clinical neuroscience and therapeutic principles. 3rd ed. Cambridge, UK: Cambridge University Press; 2002. p. 760–70.
- BenDebba M, Dizerega MD, Long DM. The lumbar spine outcomes questionnaire: its development and psychometric properties. Spine J. 2007;7:118–32.
- Long DM. Chronic back pain. In: Wall PD, Melzack R, editors. Textbook of pain. 4th ed. Edinburgh/New York: Churchill Livingstone; 1999. p. 539–58.
- Long DM. The development of the comprehensive pain treatment program at Johns Hopkins. In: Cohen MJM, Campbell JN, editors. Pain treatment centers at a crossroads: a practical and conceptual reappraisal. Progress in pain research and management, vol. 7. Seattle: IASP Press; 1996. p. 3–24.
- Long DM, BenDebba M, Torgerson WS, Boyd RJ, Dawson EG, Hardy RW, Robertson JT, Syport GW, Watts C. Persistent back pain and sciatica in the United States: patient characteristics. J Spinal Disord. 1996;9(1):40–58. Philadelphia: Lippincott-Raven Publishers.

- Ackerman SJ, Steinberg EP, Bryan RN, Ben Debba M, Long DM. Patient characteristics associated with diagnostic imaging evaluation of persistent low back problems. Spine. 1997;22(14):1634–40.
- Hendler NH, Mollett A, Viernstein M, Schroeder D, Rybock J, Campbell J, Levin S, Long DM. A comparison between the MMPI and the "Hendler Back Pain Test" for validating the complaint of chronic back pain in men. J Neurol Orthop Med Surg. 1985;6(4):333–7.
- Lora J, Long DM. So-called facet denervation in the management of intractable back pain. Spine. 1976;1(2):121–6.
- Bogduk N, Long DM. Percutaneous lumbar medial branch neurotomy. A modification of facet denervation. Spine. 1980;5(2):193–200.
- Long DM. Pain of spinal origin, chapter 126. In: Youmans JR, editor. Neurological surgery, vol. 6. 6th ed. Philadelphia: W. B. Saunders Publishing Co; 1982. p. 3613–26.
- Long DM. Surgical treatment for back and neck pain. In: Wall PD, McMahon SB, Koltzenbarg M, editors. Wall and Melzack's textbook of pain. Philadelphia: Elsevier/Churchill Livingstone; 2006.
- 14. Long DM. Effectiveness of therapies currently employed for persistent low back and leg pain. Commentary Effectiveness of therapies currently employed for persistent low back and leg pain. Commentary. Pain Forum. 1995;4(2):122–5.
- Long DM. Chronic back pain, chapter 5. In: Wall, Melzack, editors. Handbook of pain management: a clinical companion to wall and Melzack textbook of pain. Churchill Livingstone: Elsevier; 2003. p. 67–76.
- Long DM, Zeidman SM. Outcome of low back pain therapy. In: Hadley MN, editor. Perspectives in neurological surgery, vol. 5, no. 1. St. Louis: Qualit, Medical Publishing; 1994. p. 41–51.
- Long DM. Evaluation and treatment of the "multiple surgery" low-back cripple. Contemporary. Neurosurgery. 1979;17:1–6.
- Long DM. Failed back syndrome: etiology, assessment and treatment—chapter 27. In: Burchiel K, editor. Surgical management of pain. New York: Thieme Medical Publishers, Inc; 2002. p. 354–64.
- Zeldman SM, Long DM. Failed back surgery syndrome, chapter 43. In: Menezes AH, Sonntag VKH, editors. Principles of spinal surgery, vol. 1. New York: McGraw-Hill Publishers; 1996. p. 657–79.
- Long DM. Failed back syndrome. In: Kostulk JP, editor. SPINE failed spinal surgery: state of the art reviews-vol. 11, no. 3. Philadelphia: Hanley & Bellus; 1997. p. 439–52.
- Long DM, McAfee PC, editors. Atlas of spinal surgery. Baltimore: Williams and Wilkins; 1992.
- BenDebba M, van Alphen HA, Long DM. Association between peridural scar and activity-related pain after lumbar discectomy. Neurol Res. 1999;21(Supp. 1):537–42.
- Long DM. External electrical stimulation as a treatment of chronic pain. Minn Med. 1974;57:195–8.

- 24. Campbell JN, Long DM. Peripheral nerve stimulation in the treatment of intractable pain. J Neurosurg. 1976;45:692–9.
- 25. Long DM. Use of peripheral and spinal cord stimulation in the relief of chronic pain. In: Bonica JJ, Albe-Fessard, editors. Advances in pain research and therapy. New York: Raven Press; 1976. p. 395–403.
- Long DM. Electrical stimulation for the control of pain. Arch Surg. 1977;112:884–8.
- Long DM. Uses of percutaneous electrical stimulation of the nervous system. Med Prog Technol. 1977; 5:47–50.
- North RB, Fischell TA, Long DM. Chronic dorsal column stimulation via percutaneously inserted epidural electrodes: preliminary results in 31 patients. Appl Neurophysiol. 1977/1978;40:184–91.
- Long DM. Electrical stimulation of the nervous system for pain control. Contemporary Clin Neurophysiol. 1978; (EEG Suppl. No. 34):343–48.
- Long DM, Erickson D, Campbell J, North R. Electrical stimulation of the spinal cord and peripheral nerves for pain control. A 10-year experience. Appl Neurophysiol. 1981;44:207–17.
- Erickson DL, Long DM. Ten-year follow-up or dorsal column stimulation. In: Bonica JJ, editor. Advances in pain research and therapy, vol. 5. New York: Raven; 1982. p. 583–9.
- North RB, Kidd DH, Zahurak M, James CS, Long DM. Spinal cord stimulation for chronic, intractable pain: experience over two decades. Neurosurgery. 1993;32(3):384–95.
- Long DM. Chronic pain syndrome. In: Long DM, editor. Current therapy in neurological surgery 1985– 1986. Toronto: B C. Decker Publisher; 1985. p. 208–9. St. Louis/Toronto/London: C. V. Mosby Company.
- Long DM. Contemporary diagnosis and management of pain. 3rd ed. Newtown: Handbooks in Health Care Co., a Division of Associates in Medical Marketing Co; 2005.
- 35. Long DM. Rehabilitation of the patient with persistent pain. In: Illis LS, editor. Neurological

rehabilitation. 2nd ed. Oxford: Blackwell Scientific Publishers; 1994. p. 394–408.

- 36. Bogduk N, Long DM. The anatomy of the so-called "articular nerves" and their relationship to facet denervation in the treatment of low-back pain. J Neurosurg. 1979;51:172–7.
- Vrtovec T, Janssen MMA, Likar B, Castelein RM, Viergever MA, Pernus F. A review of methods for evaluating the quantitative parameters of sagittal pelvic alignment. Spine J. 2012;12:433–46.
- Schroeder M, Viezens L, Schaefer C, Friedrichs B, Algenstaedt P, Ruther W, Wiesner L, Hansen-Algenstaedt N. Chemokine profile of disc degeneration with acute or chronic pain. Laboratory investigation. J Neurosurg Spine. 2013;18:496–503.
- Aoki Y, Rydevik B, Kikuchi S, Olmarker K. Local application of disc-related cytokines on spinal nerve roots. Spine (Phila PA 1976). 2002;27:1614–7.
- Burke JG, Watson RW, McCormack D, Dowling FE, Walsh MG, Fitzpatrick JM. Intervertebral discs which cause low back pain secrete high levels of proinflammatory mediators. J Bone Joint Surg Br. 2002; 84:196–201.
- DePalma M. Biologic treatments for discogenic low back pain. SpineLine. 2012;3:19–26.
- 42. Nagae M, Ikeda T, Mikami Y, Hase H, Ozawa H, Matsuda K, Sakamoto H, Tabata Y, Kawata M, Kubo T. Intervertebral disc regeneration using a plateletrich plasma and biodegradable gelatine hydrogel microspheres. Tissue Eng. 2007;13(1):147–58.
- Meisel HJ, Siodla V, Ganey T, et al. Clinical experience in cell-based therapeutics: disc chondrocyte transplantation: a treatment for degenerated or damaged intervertebral disc. Biomol Eng. 2007;24:5–21.
- 44. Steck E, Bertram H, Abel R, et al. Induction of intervertebral disc-like cells from adult mesenchymal stem cells. Stem Cells. 2005;23:403–11.
- 45. Yoshikawa T, Ueda Y, Miyazaki K, et al. Disc regeneration therapy using marrow mesenchymal cell transplantation: a report of two case studies. Spine. 2010;35:E475–80.

## **Lessons Learned from a Life**

H. Michael Mayer

## 51.1 The World of Spinal Surgery: Then

In 1982 when I started my career as a young resident at Mario Brock's Department of Neurosurgery at the Free University in (at that time "West"-) Berlin, the "world of spinal surgery" seemed to be easy.

Neurosurgeons performed lumbar (micro-) diskectomies and cervical keyhole (sometimes rather "mousehole") foraminotomies and anterior diskectomies with fusion using autologous bone grafts like Cloward or Smith and Robinson had taught them.

For spinal tumors and metastases, total laminectomies very often were the treatment of choice.

Orthopedic surgeons concentrated on deformity correction. The Harrington rods and the Luque technique were the standard procedures for the treatment of deformities. For trauma surgeons, the treatment of spinal fractures and

Spine Center, Orthopädische Klinik München-Harlaching, Harlachinger Straße 51, 81243 München, Germany e-mail: MMayer@schoen-kliniken.de injuries was at that time a difficult endeavor. Conservative treatment in casts, wires, and rods without the possibility of reduction, and uninstrumented fusion were the most popular treatment modalities. For short-level fusions, a more logical but "dangerous" technique of putting screws into the pedicle and connecting them with rods or plates had just been invented by two European spine surgeons [1, 2] who were about to use this techniques in clinical trials.

In summary the treatment of disk herniations, spinal cord compression of any kind, fractures, and deformities were the main fields which were covered by spinal surgery. Moreover, diagnostic measures were limited. X-rays, myelography, and scintigraphy were standard procedures. The spinal computerized tomography (CT) scan was something new and not used as a standard diagnostic tool at that time.

Patient selection was easy, diagnosis was difficult and invasive, surgical technology was poor, and looking back very critically to those years, the treatment outcomes were poor as well. Surgical treatment of the spine was characterized by high complication rates, loss of correction in deformity cases, the necessity of recurrent operations, and pain. These were the years were a new syndrome was established: the "failed back surgery syndrome" (FBSS). Although we are aware that also today we see patients with persistent pain following spinal surgery, we now know that the term FBS "syndrome" is a semantic error

© Springer-Verlag Berlin Heidelberg 2016

H.M. Mayer, MD, PhD Department of Spine Surgery, Schön Klinik München Harlaching, Munich, Germany

because "syndrome" implies an underlying clinical and morphological entity which is not the case in patients suffering from persistent symptoms following back surgery. Today we are able to find out the specific pain source in virtually every patient.

For most of neurosurgical and orthopedic residents, spine surgery was a potentially dangerous and unknown territory, poorly understood, and not at all an attractive choice for subspecialization. Neurosurgical residents were fascinated by brain surgery or aneurysma clipping, and orthopedic residents focused on the trendy field of arthroscopic knee surgery or total hip replacement – fields with a great future.

So what kind of idiot could I have been to get interested in spinal surgery?

And here we come to the *first lesson* I have learned:

#### Lesson 1: Don't Always Follow the Mainstream

One of the first operations I was allowed to perform at the beginning of my neurosurgical training was a cranial burr hole to insert an intraventricular catheter for intracranial pressure measurements in a brain-injured patient. My second operation was a lumbar diskectomy. These two procedures were at the beginning of the 1980s the typical "resident's cases." Besides the marginal technical differences the main difference I noticed between these two "procedure categories" was that, outcome- and complicationwise, there was not really much to lose in a nearly brain-dead patient, but you could well "destroy" the professional and social life of a patient with a simple disk herniation if the operation was not done properly.

It wasn't really clear to me at that time, why we youngsters were allowed to perform lumbar diskectomies which outcomes could have such a big impact on a patient's life, whereas the excision of a glioblastoma (with a survival rate for the patient which was at that time less than 9 months with or without treatment) was considered to be high-level neurosurgery.

I believe the answer to this policy which was practiced in the majority of neurosurgical departments throughout Europe lies in the fact that the contact time of a disk patient with neurosurgery was short. Pre- and postoperative treatment were completely in the hand of either physiatrists or orthopedic surgeons, neurologists, and pain specialists. Unless it was a recurrent disk herniation or a postoperative pseudo-meningocele, we very rarely saw a patient with a "failed back surgery."

Anyway, I got interested in disk surgery and in the question how the common surgical technique of diskectomy could be improved. It was in the early days of 1983 when John A. McCulloch visited the Neurosurgical Department at the Free University of Berlin and showed us a new technique which was called "chemonucleolysis" [3]. I was fascinated by the idea that a posterolateral percutaneous intradiscal injection of an enzyme which was obtained from the papaya fruit and which was called "chymopapain" could have the same clinical outcome as "open" diskectomy. Until that time, my whole scientific activities had been focused on intracranial pressure, brain metabolism measurement, and intracerebral blood flow under various experimental conditions such as ischemia, brain abscess or brain injury. Now I got interested in a "spinal" topic.

I learned the technique and was allowed to perform it. My scientific interests focused on perioperative intradiscal pressure-volume tests, anesthesiologic aspects, clinical results, and complications such as allergic reactions [4–7].

The results were good, the complication rates were low, and chemonucleolysis was about to become a new minimally invasive standard technique in spinal surgery.

Then I had to learn my second lesson:

#### Lesson 2: Successful Treatments May Fail Without Medical Reasons

Why is this important? The neurosurgeon and orthopedic surgeon's thinking in the middle of the 1980s of the last century was more or less mechanical. A slipped disk causes pressure on the nerve root. If you remove the disk, the nerve root is relieved. The biochemical aspects of disk herniations and root pain were unknown that time. Now if chymopapain is injected into the disk, it hydrolyses the bonds of the matrix mucopolysaccharides and "dissolves" the nucleus pulposus. This leads to a reduction of the intradiscal pressure and thus to a reduction of the pressure on the nerve root. However, it can take some time until this clinical effect is evident for the patient as well as for the surgeon. Now, being patient is not really typical for surgeons. Even though chemonucleolysis achieved good clinical results with low complication rates, it never had the popularity which it should have deserved. Moreover, the reimbursement was much lower as compared to surgical diskectomy.

Three things happened in the middle of the 1980s which initiated the failure of chemonucleolysis. (1) There was a report of a patient who had developed a paraplegia after erroneously intrathecal injection of chymopapain [8, 9]. (2) There were some case reports on anaphylactic reactions to chymopapain [10]. Chymopapain is also an ingredient of "meat tenderizers" or cleaning fluids for contact lenses, so there is a certain sensitization among the normal population. (3) In 1985 a new percutaneous technique was described to mechanically aspirate nucleus pulposus from the center of the lumbar disk [11].

So, I lost chemonucleolysis temporarily out of sight because I was about to learn my *third lesson*:

#### Lesson 3: The Shortest Way Is Not Always the Best Way

I had realized that the spectrum of spinal surgery in neurosurgery would be too small to build a career on. After 3 years, I interrupted my neurosurgical training to start my orthopedic training at the Department for Orthopedic Surgery at the Free University of Berlin to learn spinal fusion techniques as well as anterior approaches to the thoracic and lumbar spine. The advantage of having being skilled already in microsurgical techniques was priceless. Although it was accidentally that I did part of my neurosurgical training first, I learned the lesson that if you are able to practice microsurgery, it will definitely be easier to learn macro-surgical techniques. I enjoyed doing hip and knee surgery through small skin incisions (which I had to justify everyday because nobody was practicing or was even thinking about minimally invasive approaches

for total hip or knee replacement at that time ...). I could learn all types of anterior trans- and retroperitoneal approaches to the lumbar spine because the philosophy of biomechanically stable circumferential fusion had become popular by the work of René Louis from the mid-1970s [12]. I could experience the first steps with a new posterior correction and stabilization system called "CD" according to its inventors Cotrel and Dubousset, two surgeons from France [13, 14]. I learned anterior transthoracic approaches for anterior scoliosis correction with the Zielke system [15]. Orthopedic surgery made fun, but after 3 years I was in the situation that I had completed two half resident's programs and I had to make a decision.

Between 1985 and 1987, Onik's automated percutaneous lumbar diskectomy (APLD) became popular on both sides of the Atlantic Ocean. However, it turned out that obviously the indication spectrum for a central decompression of the lumbar disk was narrow and that the majority of disk herniations could not be treated with this procedure.

The percutaneous approach however was fascinating. I was wondering whether an endoscopic procedure which would follow the principle of arthroscopic surgery could be an innovative approach to lumbar disk herniation. It turned out that I was not the first to have such an idea; however, since the orthopedic patient population at that time did not include a lot of disk patients and since my chief was a traditional hip surgeon who did not support my ideas, I decided to find somebody to support me. It was my former chief Prof. Mario Brock who, due to his experience with chemonucleolysis, was also fascinated by the idea of an endoscopic approach to the disk.

As I said, it was not really my own idea. Parviz Kambin in Philadelphia, USA, was about to organize the first international workshop on "Arthroscopic Microdiskectomy" in 1987 in Philadelphia. I attended this workshop and met with Hijikata from Japan who had performed percutaneous diskectomies since early 1973 [16] and I met Profs. Schreiber and Suezawa from Switzerland who had performed the first endoscopic diskectomies worldwide [17]. In summer 1987, I performed the first endoscopic diskectomy in Germany and spent the following years until the early 1990s with this scientific topic. I learned that a selective diskectomy is possible under endoscopic control with various types of instruments. In experimental and clinical trials, I learned about the use of different

laser technologies for endoscopic diskectomy. I learned that using a laser does not really influence the clinical result of endoscopic diskectomy (but it helped me to write my PhD thesis) [18, 19]. I completed my neurosurgical training and

could show, in a prospective, randomized controlled trial, that percutaneous endoscopic diskectomy achieves similar results as compared to microsurgical diskectomy (with the same indication criteria) [20].

I was happy but not yet a spine surgeon. I realized that I had to leave neurosurgery again to learn the rest of the orthopedic part of spinal surgery.

And there I learned another (the *fourth*) lesson:

# Lesson 4: Don't Sacrifice the Surgical Goal for a New Technique

I was fascinated by small surgical approaches to the human spine. In 1991 Obenchain, an American gynecologist, described a laparoscopic approach for lumbosacral diskectomy [21]. This was more or less the trigger for laparoscopic and thoracoscopic fusions and approaches which were refined in the following years. The long learning curves, demanding technology, complication rates, and high non-fusion rates were the reasons, why laparoscopic fusion did not stand the proof of time. One of the major problems with laparoscopic surgery was that this fascinating approach was combined with the application of threaded fusion cages. These cages never proofed to be sufficient as a stand-alone implant for spinal fusion even if they were applied through "open" approaches. So the fact that the surgical goal to achieve a solid spinal fusion was sacrificed for the new minimally invasive approach was the main reason for the failure of laparoscopic techniques although the idea was fascinating. However, thoracoscopic surgery survived and is now a routine for anterior treatment of fractures and other pathologies of the thoracolumbar spine.

I was frustrated with my first laparoscopic experiences and focused my attention to the development of mini-open anterior approaches to the thoracic and lumbar spine [22, 23]. Again I could apply "microsurgical thinking" to develop a new approach. Mini-open anterior lumbar interbody fusion turned out to be a universal approach for different pathologies without the necessity to use one single type of implant.

It even turned out to be the access technique of choice for modern artificial disk implants.

The era of modern total disk replacement began in 1999, and again I learned another important *lesson* (#5):

#### Lesson 5: Progress in Spinal Surgery Needs the Right Implants, the Right Surgeons with the Right Technology at the Right Time

I had the honor to attend one of the first Charité total disk implantations at the Charité Hospital of the Humboldt University in "East Berlin" in 1984 which was performed by Karin Büttner-Janz and Professor Schellnack, the two inventors of the "Charité disk." At that time, nobody realized that it would take nearly 15 years until this technology would "make it around the world." There have been several reasons why it took so long for a good idea to become accepted. It was the implant material and design which had to be further improved. It was the anterior transabdominal surgical access which was way too aggressive at that time to justify the implantation of a new implant for degenerative disk disease. And last but not least, it was the diagnostic uncertainty which was a very high threshold for surgeons to recommend aggressive types of surgical procedures. I remember that all these arguments were important for us not to start with total lumbar disk replacement at that time. However, the topic was so fascinating that we decided to organize the first international symposium on "The Artificial Disk" which we performed in Berlin in on November 9, 1989 [24]. It was an historic event, not necessarily because of the scientific topic but because of the date. During our faculty dinner on the evening of November 9, 1989, the Berlin Wall opened and we had the opportunity to be part of a historical moment when we went to the Brandenburg Gate that night. Actually one of our overseas faculty members thanked us later on for "the best social program he ever had...."

Although politically it was the right time, total disk replacement had to wait.

However, 10 years later, I was fascinated by the idea to apply our minimally invasive anterior approaches with the new generation of total disk replacement. It was in the mid-1990s when Thierry Marnay visited me in Berlin to discuss minimally invasive anterior approaches for the implantation of anterior fusion devices and total disk replacement.

When Prodisc L second generation was developed, I learned *lesson # 6*:

#### Lesson 6: Total Disk Replacement Works

In 1997 I was chief staff surgeon and vice director of the Department of Orthopedic Surgery at the Free University of Berlin. I had become a spine surgeon and was ready to take over full responsibility. Fortunately I got the chance to build up a spine center at one of the most traditional and well-known orthopedic hospitals in Germany: the Orthopädische Klinik München-Harlaching in Munich, Germany.

I took over the new job in March 1998 and began to structure a spine center. Besides the various managing tasks, I, for the first time in my professional life, had the freedom to practice new techniques. In 1999 we implanted the first artificial lumbar disk through a mini-open retroperitoneal approach. It was a great moment to realize that my access technique actually met the needs for this new implant concept. Motion/function preservation in combination with a minimally invasive surgical access technique turned out to be the key for the success story of total lumbar disk replacement. Although the overall midterm outcomes of total lumbar disk replacement in the last 10 years are more or less comparable to spinal fusion, there is a subgroup of patients who show and represent the cutting edge of this new technology combination. Looking back at spinal fusion results, I had never seen high-level athletes who returned to their full performance strength after lumbar of cervical fusion. Now I could see a professional soccer player who played in 60 matches

in the year after total lumbar disk replacement. I could see a patient becoming Olympic champion 7 months after artificial cervical disk replacement, and I had saw patients who could run marathons with an artificial lumbar disk [25–28].

Our enthusiasm in "motion preservation" urged us to organize the first international scientific meeting on "Spine Arthroplasty" in 2001 in Munich. Munich was the birthplace, and this meeting was the birth place of the Spine Arthroplasty Society (SAS) which was founded during this meeting.

In the following years, the idea of spine arthroplasty grew, the annual meetings of the SAS have become the yearly highlights of the scientific scenery for motion-preserving technology and have also become the forum for the rapidly growing industry behind it.

In 2001 we all did not realize that total disk replacement was just the beginning of a new era in spinal surgery. The hype of replacing fusion surgery with reconstructive technology has triggered other developments such as dynamic pedicle-screw fixation, interspinous "extension stoppers," interspinous spacers with interlaminar fixation, etc.

And here is lesson #7:

#### Lesson 7: We Cannot Beat Nature

If we look at outcome studies for various types of treatment in degenerative low back pain, we will soon realize that the successful so-called long-term results all range between 60 and 80 %. This is true for passive waiting as well as for all types of invasive treatment. And - it is logical. Degenerative changes of the spine are self-limiting pathologies which symptomatology is temporary in the majority of the patients. If you add on whatever kind of more or less aggressive treatment to this "natural course," how can you expect the "end result" to be better than the "uninfluenced" course?

This led me to *lesson* 8 which says:

#### Lesson 8: Surgical Therapy in Degenerative Diseases of the Spine Can Only Modulate the natural Course

The "classic" in this sense are the outcomes of lumbar diskectomies. It is well known from the

literature and from randomized controlled trials (RCT) that the midterm and long-term results of lumbar diskectomy (open, micro, endo) are virtually the same as in non-operated patients. The results converge after 4 years. However, would this justify to tell a patient who is in severe radicular pain due to a lumbar disk herniation or who is suffering from neurological deficits to wait 4 years for the "end result?"

Definitely not! Even if the result of diskectomy is the same as for conservative therapy and even if the risk of surgery is higher than the risk of just waiting, this result can be achieved much quicker through the decompression of the affected nerve root.

The concept of indirect decompression of the central and lateral spinal canal by the implantation of an interspinous spacer and extension stoppers arose in 2003 with the first clinical application of X-Stop [29]. In the past years, this concept has been further elaborated and now a great number of interspinous implants are currently in clinical studies or already in routine use for the treatment of spinal stenosis and/or low back pain. They neither have a significant stabilizing effect nor do they promote spinal fusion. This means that these implants most probably will have a temporary clinical effect.

And this is *lesson* # 9 I learned:

## Lesson 9: Spinal Implants with a Temporary Clinical Effect May Be Acceptable for the Surgeons and the Patients

It is a daily experience I make since a couple of years: patients want to get rid of their pain quickly and reliably. The less invasive a type of treatment is, the higher is the acceptability. How would we or the patient classify, for example, the percutaneous implantation of an interspinous spacer for the treatment of dynamic lumbar spinal stenosis? Most of the patients consider this 12 min procedure more as an aggressive or invasive type of conservative therapy than as a minimally invasive type of surgery. Even though the perspective is only a temporary relief of symptoms, the patient seems to accept this kind of treatment provided that the need for another surgery in case of recurrent symptoms is not impaired by the fact that bridges have been burnt down.

## 51.2 The World of Spinal Surgery: Now

The last lesson is *lesson # 10*:

#### Lesson 10: The Spine World in 2014 Is More Colorful

When I learned the first lesson in my "spine life," the world of spinal surgery was black and white. For a given pathology, there were one or two treatment options. Today we are facing a wide variation of conservative and surgical treatment modalities. This makes the life of a spine surgeon easier and harder. It is easier to choose among different options, but it is more difficult to determine which treatment option is the best in each individual case. But this is what makes spinal surgery so fascinating and what will keep us busy scientifically and clinically for the coming years.

Innovative ideas are not getting less, but the obstacles and difficulties to realize them get more. Failures of most of the global players in the spinal Medtech industry to realize that innovation and progress has always been patient/surgeon – and not engineer – driven have led to the development and marketing of a number of needless and senseless implant systems and concepts in the last 15 years. This on the other hand has expectedly increased the alertness and criticism of payers, insurance companies, and healthcare providers toward new products. The adherence of major companies to the currently dominant and overregulated markets has initiated a downward spiral for innovative spine surgery in the western world.

However, this opens new opportunities for countries and companies which do not comply with this trend. We should always remember that all major steps forward in orthopedic surgery started very nontraditional, off-label, off-budget, uncontrolled, remarkably innovative, and unhibited by failures and relentless. I would like to thank my great teachers and fellow colleagues who have lit the fire for spine surgery and who have helped me to learn my lessons.

#### References

- Roy-Camille R, Demeuleneare C, Barcat E, Saillant G. Dorsal and lumbar spine osteosynthesis by posterior approach. Nouv Presse Med. 1973;2:1309–12.
- Dick W. Osteosynthesis of severe injuries of the thoracic and lumbar spine with internal fixation. Langenbecks Arch Chir. 1984;364:343–6.
- McCulloch JA. Chemonucleolysis. Can Med Assoc J. 1982;126:119–20.
- Artigas J, Brock M, Mayer HM. Complications following chemonucleolysis with collagenase. J Neurosurg. 1984;61:679–85.
- Mayer HM, Lutze M, Wehr M, Kaden B, Brock M. Disc-compliance studies in chemonucleolysis. Altern Spinal Surg. 1985;2:9–11.
- Görge HH, Brock M, Curio G, Mayer HM. Surgical findings in 50 cases of failed chemonucleolysis with chymopapain. Surg Neurol. 1986;25:563–7.
- Mayer HM, Wehr M, Brock M, Kaden B. Skin testing for chymopapain allergy in chemonucleolysis. Surg Neurol. 1986;25:283–9.
- Dyk P. Paraplegia following chemonucleolysis. A case report and discussion of neurotoxicity. Spine. 1985;10:359–62.
- Eguro H. Transverse myelitis following chemonucleolysis. JK Bone Joint Surg. 1983;65:1328–30.
- Bouillet R. Treatment of sciatica. A comparative survey of complications of surgical treatment and nucleolysis with chymopapain. Clin Orthop Rel Res. 1990;251:144–51.
- Onik G. Percutaneous lumbar diskectomy using a new aspiration probe: porcine and cadaver model. Radiology. 1985;155:251–2.
- Louis R, Maresca C. Stable arthrodesis of the lumbosacral region. Rev Chir Orthop Reparatrice Appar Mot. 1976;62 Suppl 2:70–9.
- Cotrel Y, Morel G. The elongation-derotation-flexion technique in the correction of scoliosis. Rev Chir Orthop Reparatrice Appar Mot. 1964;50:59–75.
- Cotrel Y, Dubousset J. A new technique for segmental spinal osteosynthesis using the posterior approach. Rev Chir Orthop Reparatrice Appar Mot. 1984;70:489–94.

- Zielke K, Stunkat R, Beaujean F. Ventral derotation spondylodesis. Arch Orthop Unfallchir. 1976;85: 257–77.
- Hijikata S. Percutaneous nucleotomy. A new concept technique and 12 years experience. Clin Orthop Rel Res. 1989;238:9–23.
- Suezawa Y, Rüttimann B. Indications, methods and results in percutaneous nucleotomy in lumbar disk hernia. Z Orthop Ihre Grenzgeb. 1983;121:25–9.
- Mayer HM, Brock M, Sedlmaier B, Berlien H-P, Müller G, Dörschel K. Ultrastructure of human nucleus pulposus following application of Erbium: YAG 2940 nm – laser. Laser Med Surg. 1990;6/4:190–7.
- Mayer HM, Brock M, Berlien HP, Weber B. Percutaneous endoscopic LASER discectomy (PELD) – a new surgical technique for nonsequestrated lumbar discs. Acta Neurochir (Suppl). 1992;54:53–8.
- Mayer HM, Brock M. Percutaneous endoscopic discectomy – surgical technique and preliminary results compared to microsurgery. J Neurosurg. 1993;78: 216–25.
- Obenchain TG. Laparoscopic lumbar discectomy: case report. J Laparoendosc Surg. 1991;1:145–9.
- Mayer HM. A new, microsurgical technique for minimal invasive anterior lumbar interbody fusion (MINIALIF). Spine. 1997;22:691–700.
- 23. Mayer HM. The ALIF concept. Eur Spine J. 2000;9:S35–43.
- Brock M, Mayer HM, Weigel K, editors. The artificial disc. Berlin: Springer; 1991.
- Siepe C, Mayer HM, Wiechert K, Korge A. Clinical results of total lumbar disc replacement with prodisc II: – 3-year-results for different indications. Spine. 2006;31:1923–32.
- Siepe CJ, Wiechert K, Khattab MF, Korge A, Mayer HM. Total lumbar disc replacement in athletes: clinical results, return to sport and athletic performance. Eur Spine J. 2007;16:1001–13.
- Siepe CJ, Mayer HM, Heinz-Leisenheimer M, Korge A. Total lumbar disc replacement. Different results for different levels. Spine. 2007;32:782–90.
- Mayer HM, Siepe C. Total lumbar disc arthroplasty. Curr Orthop. 2007;21:17–24.
- 29. Zucherman JF, Hus KY, Hartjen CA, Mehalic TF, Implicito DA, Martin MJ, Johnson DR, Skidmore GA, Vessa PP, Dwyer JW, Puccio ST, Cauthen JC, Rm O. A multicenter, prospective randomized controlled trial evaluating the X-Stop interspinous process decompression system for the treatment of neurogenic intermittent claudication. Spine. 2005;30:1351–8.

## Lessons from 35 Years in the Trenches

52

Edward C. Benzel, Kene Ugokwe, and Nina Z. Moore

## 52.1 Introduction

Mastery of the art and science of the spine surgery is a career-long endeavor. Spine surgeons are exposed to experiences throughout their career that are not necessarily a part of their formal training, yet mold their practice and their academic life. Though a solid understanding of the biomechanical and surgical principles is essential, the lessons learned from "experience" are equally valuable and when coupled with prior knowledge result in "wisdom." These "lessons learned" are not limited to the act of surgery but also include the perioperative period. Surgeons can get caught up in the "act of surgery," while ignoring essential nonoperative factors. The patient selection process and the "social aspects"

K. Ugokwe, MD Department of Surgery, Mercy Health Youngstown St. Elizabeth's Hospital, Youngstown, OH, USA e-mail: kenechukwu11@gmail.com

N.Z. Moore, MD, MSE Department of Neurosurgery, Center for Neurological Restoration, Cleveland Clinic, Neurological Institute, Cleveland, OH, USA e-mail: mooren4@ccf.org of surgery and the decision-making process are critical components of the "art of surgery." The "social aspect" of care takes into account the patient's desires, expectations, and feelings. It involves an obligatory ongoing dialog with the patient and the patient's family. This chapter focuses on the senior author's reflections regarding the art and science of surgery, as interpreted by the junior authors.

## 52.2 The Preoperative Period

The game is won or lost in the outpatient clinic.

Patient selection for surgery can be challenging. This is clearly an understatement. To simplify this process for lumbar degenerative disk disease, one might divide patients into several categories: nonoperative, potentially operative, and definitively operative. Nonoperative patients often have an undiagnosed chronic pain syndrome. These individuals often have numerous pain complaints (multiple unrelated somatic complaints) that are not attributable to a specific dermatome or myotome. Their pain is often characteristically atypical, e.g., burning pain. One of the most significant mistakes in the lumbar degenerative disease decision-making process is to treat chronic pain as if it were acute pain. These patients often have a low energy level and complain of nonrestorative sleep. Unfortunately,

E.C. Benzel, MD (⊠) Department of Neurosurgery, Cleveland Clinic, Neurological Institute, Cleveland, OH, USA e-mail: benzele@ccf.org

778

they may also have "pathology" on imaging studies. This "pathology" often tempts the surgeon to recommend surgery. The outcome is often predictably suboptimal. The bottom line here is to maintain a low threshold for establishing the diagnosis of a chronic pain syndrome and to treat the patient accordingly – usually without surgery.

Operative patients can be divided into two groups based on a simple question. Would the surgeon have this surgery if he/she were the patient? If the surgeon honestly answers this question, the rest of the process is greatly simplified. During a national spine meeting approximately 8 years ago, audience members during a plenary session were asked if they would recommend surgery for a case presented to the group. Eighty percent of the surgeons responded in the affirmative. When asked if they would have the same surgery themselves, only 20 % of the audience responded in the affirmative. This is damning to the profession of spine surgery. Performing an operation that one would not undergo him/ herself, at the very least, raises questions regarding the surgeon's specific operative indication process and the motives for such. There are, of course, patients that harbor a clear-cut indication for surgery and are, therefore, much easier to select. Regardless of the strategic plan derived, the spine surgeon should be fiscally responsible when crafting the plan of attack on the patient's malady. The surgeon should act (i.e., manage and advise the patient) as if he/she were paying the bill.

#### 52.2.1 Nonsurgical Management

Most patients should undergo aggressive nonsurgical management prior to any consideration for surgery. Surgery should be most often employed as a treatment of last resort. It is imperative that the surgeon be familiar with, and be willing to prescribe, relevant nonsurgical clinical modalities. Offering such should not be construed as a sign of failure, but rather as an indication of good judgment. Physical therapy by a familiar and competent physical therapist is an often underutilized tool. Other nonoperative techniques that should be considered are epidural steroid injections, traction, TENS units, and nonnarcotic meds including muscle relaxants and NSAIDs. An appropriate exercise program should consist of a core-strengthening and flexibility training that also incorporates weight loss, cessation of smoking, and aerobic exercise. The spine surgeon should be keenly aware of the needs of the patient and the value to the patient provided by consultants and modalities. An open line of communication between the surgeon and the therapist is critical. A trial of membrane stabilizers (e.g., gabapentin or pregabalin) should also be considered in the appropriate patient (e.g., patients with neurogenic claudication or neuropathic pain) prior to operative management. Finally, the adverse consequences, whether they be short or long term, of spine surgery must not be taken lightly. Remember, "One can always do surgery, but one cannot undo surgery."

The best surgical candidates are those that meet the criteria for surgery and have demonstrated that they are medically compliant. Cessation of smoking, weight loss, and active participation in physical therapy programs are all means by which one can assess the patient's compliance. The absence of a commitment to such (i.e., noncompliance) correlates with poor outcome regardless of treatment employed. The patient and the surgeon must work together in order to achieve optimal results.

#### 52.2.2 Education

The surgeon must not underestimate the importance of the well-informed patient and family. Surgical success is measured not only by the physical outcome but also the patient's interpretation of the outcome. Patients and their families must be presented with realistic expectations and be well informed regarding the potential risks and complications of surgery so they can make well-informed decisions and be prepared prior to the day of surgery. If complications occur, the involved and informed patient and family will be much more likely to understand and work collaboratively with the surgeon to overcome the obstacles encountered. Additionally, it is important to establish a healthy and genuine relationship with patients. Expressions of empathy and active listening yield more accurate medical histories, as well as improved professional relationships between the patient and the surgeon. It is important for the surgeon to inform the patient of any conflicts of interests so the patient can take that information into account during the decision-making process. This serves as a means by which the surgeon can regulate his/ her own clinical bias and help avoid the perception of any impropriety [1].

When the aforementioned aspects are appropriately considered in the clinical decisionmaking process, an optimally tailored approach to care is easily achievable.

#### 52.3 The Intraoperative Period

From an operative perspective, a strategy that embodies the concept that "more is better" does not always hold true. Sometimes simplicity is best. More complex surgery does not ensure a decreased return to surgery rate. Doing less is not a sign of weakness, inefficiency, or incompetence, but may be a sign of maturity. The "older (mature) surgeon" adage of "the more I do, the less I do" applies here.

#### 52.3.1 The Surgical Team

The surgeon must work as a team member and as the leader of the team. The importance of teamwork cannot be overemphasized. This team approach is critical both in and outside the operating room. It establishes open lines of communication with all involved surgical team members. As the leader of the team, the surgeon must convey expectations, concerns, and overall operational strategies to the anesthesia and nursing teams in a timely and calculated manner. The surgeon must maintain composure regardless of the situation. Information should be exchanged in a nonthreatening and nonconfrontational manner. All members of the team look to the surgeon for guidance and leadership. The absence of composure sends a powerfully negative message to the remainder of the team that may induce panic and a suboptimal overall team effort. Communication with the patient's family is also important during surgery. Updating the family members during the operation reduces family stress and solidifies the doctor-patient relationship.

## 52.3.2 Unexpected Intraoperative Findings

It is not uncommon for intraoperative findings to be unexpected, based on preoperative imaging and clinical information. Surgeons are expected to be creative and to improvise in such situations, thus demonstrating their ability to adapt to adverse and unexpected circumstances.

It is important to think in terms of the "future" as well as the "present" when making intraoperative decisions. For example, fusing extra levels may be good for the "present," but be associated with excessive long-term consequences. Such long-term negative consequences may be preventable by appropriately considering extent of surgery. The operative decisionmaking process should be guided more by the principles of providing long-term relief of symptoms than by those symptoms associated with instant gratification. The surgeon should always consider the potential problems associated with an operation. Every attempt should be made to avoid creating a problem of equal or greater magnitude.

The surgeon should be mindful of how the intended operation will affect the patient's spine alignment and balance. Appropriate lumbar lordosis helps avoid adjacent segment disease. Fusions should be performed only after the effect on adjacent levels is appropriately considered. The surgeon should avoid extra or "prophylactic" surgery – making every attempt to perform the surgeon should also be mindful of the patient's wishes, culture, and religion preoperatively,

intraoperatively, and postoperatively. The surgeon's expectations should ALWAYS be aligned with those of the patient.

#### 52.3.3 Intraoperative Complications

If an intraoperative complication occurs, it is best that such be disclosed to the patient and their family, regardless of how trivial it may seem. The term complication should encompass unexpected outcomes. Such outcomes should be directly addressed. Honesty, forthrightness, and clarity of thought and action are key virtues in these situations.

Staging an operation is often prudent in cases with excessive operative time or blood loss. Staging surgical procedures is clearly a decision that is surgeon based, but made in collaboration with other members of the team (i.e., the anesthesiologist). The arguments for and against such must be carefully considered.

After every operation, it is imperative that the surgeon show gratitude and appreciation to the rest of the surgical team. A simple gesture such as a "thank you" or a "great job everyone" goes a long way. It makes the team members feel appreciated and it makes them look forward to the next interaction.

#### 52.4 The Postoperative Period

Patients and their family members universally look forward to seeing their surgeon postoperatively. They nearly always have questions and concerns and are nearly always anxious. Successful surgery is associated with a strategically planned postoperative course. The communication and enactment of a sound postoperative plan are essential. Wound care management, pain control, physical therapy plans, continued weight loss, and smoking cessation should all be part of the plan, when appropriate.

It is particularly important to establish and maintain an ongoing dialog with patients that have had suboptimal outcomes or who have terminal conditions. Avoidance of these patients often causes anger and leaves them with a sense abandonment. Always provide honest of responses to questions. Surgeons must provide comfort by objectively guiding the decision the patients must make. This simple act provides relief and security for patients and their family. As Rabbi Kushner [2] once said, "Caring about others, running the risk of feeling, and leaving an impact on people, brings happiness. Being kind to others is a way of being good to yourself." He emphasized the two major fears that terminal patients harbor: pain and abandonment. These can be addressed by adequately and appropriately treating the pain and by having an extra sitdown talk or two with the patient. Such talks are universally appreciated. They make the surgeon feel good, as well. Remember Kushner's words: "...Being kind to others is a way of being good to yourself."

One final comment regarding suboptimal outcomes: the patient who does not improve as expected from surgery deserves special consideration. It is extraordinarily important to reassess their problem. One must consider the notion that the patient did not improve because they did harbor the diagnosis that correlated with the surgery performed. The presence of a chronic pain syndrome is likely the most common reason for failure of lumbar spine surgery for pain. If the surgeon operates on a lumbar spondylolisthesis, but the patient's complaints are those of a chronic pain syndrome, the outcome is seldom positive. In this case, surgery is not the treatment of choice and is essentially doomed to failure.

#### 52.5 Summary

This chapter provides reflections into the lessons learned from a career in spine surgery. It is, however, important to note that these lessons can be applied not only to spine surgery but to all surgical specialties. Regardless of the contents of this chapter, as spine surgeons and as physicians, we must all remember that we must "first do no harm" and second to "do unto others as you would do unto yourself." It is important to be honest and to do what is right for our patients. We leave you with five important principles, outlined by Robert Grossman [3], which we believe all spine surgeons should live by:

- 1. There is no such thing as a simple neurosurgical spine operation. A neurosurgical spine is identical to an orthopedic spine.
- 2. It is easier to stay out of trouble than to get out of trouble.
- The time expended in avoiding complications will be more than compensated by the time saved in not having to treat them.
- 4. The patient's well-being is paramount. Therefore, the surgeon should never hesitate to request consultation, or assistance, during surgery.

5. Surgeons should always operate as if the patient were their family member.

## References

- Benzel EC. Spine surgery: techniques, complication avoidance, and management. 2nd ed. New York: Elsevier/Churchill Livingstone; 2005. p. 2167–72.
- Cook J, Deger S, Gibson LA. The book of positive quotations. Minneapolis: Fairview Press; 2007. p. 27.
- Grossman RG. Preoperative and surgical planning for avoiding complications. In: Apuzzo MLJ, editor. Brain surgery: complication avoidance and management. Supratentorial procedures. New York: Churchill Livingstone; 1994. p. 9.

## History of Lumbar Endoscopic Spinal Surgery and the Intradiskal Therapies

53

## Yong Ahn, Gun Choi, and Sang-Ho Lee

### 53.1 Introduction

In the field of spinal surgery, preservation of normal tissues has become an important issue with understanding human physiology more comprehensively. And the main goal of the latest spinal surgery is neural element decompression and stabilization of spinal motion segments without any normal tissue damage and its negative long-term consequences. To achieve these, many types of surgical technologies have been developed for minimizing their invasiveness. Especially, the introduction of endoscope to spinal surgery was an encouraging sign of achieving splendid development of minimalism. With recent advance and experience, several advantages such as reducing blood loss, operation time, recovery period, and

Y. Ahn, MD, PhD (⊠)
Department of Neurosurgery,
Nanoori Hospital,
731, Eonju-ro, Gangnam-gu,
Seoul 135-816, South Korea

Department of Neurosurgery, Wooridul Spine Hospital, 445, Hakdong-ro, Gangnam-gu, Seoul 135-951, South Korea e-mail: ns-ay@hanmail.net time to return to work have differentiated percutaneous endoscopic spinal surgery from conventional open surgery. However, conventional operations have been reported to be associated with good results and such has been accepted as a "golden standard" until now. Today, the spectrum of minimally invasive spinal surgery with a percutaneous endoscope is expanded from simple disk operations to include deformity operations. In this chapter, a historical account of percutaneous endoscopic lumbar disk operations is summarized in terms of their technical landmarks.

## 53.2 Era of Percutaneous Diskectomy (Fig. 53.1)

The concept of *posterolateral percutaneous diskectomy* was firstly introduced in 1973. Parvis Kambin conducted percutaneous indirect spinal canal decompression by nucleotomy using Craig's cannula in a nonvisualized posterolateral

G. Choi, MD, PhD Department of Neurosurgery, Wooridul Spine Hospital, Seoul, South Korea

S.-H. Lee, MD, PhD Department of Neurosurgery, Wooridul Spine Hospital, Seoul, South Korea

Financial support: This study was supported by a grant from the Wooridul Spine Foundation.

The authors have no personal financial or institutional interest in any of the drugs, materials, or devices in this chapter.



Fig. 53.1 Posterolateral percutaneous diskectomy

Fig. 53.2 Transforaminal endoscopic diskectomy for contained disk herniation

approach [1]. Hijikata firstly published the technique of nonvisualized posterolateral percutaneous diskectomy in 1975, which was a standalone procedure [2]. In 1983, the direct lateral approach for percutaneous nucleotomy was performed by William Friedman in which he reported the association between this procedure and a higher rate of bowel injury [3]. In the mid-1980s, Onik et al. introduced a motorized aspiration shaver to the percutaneous diskectomy technique [4], and then, Onik and Maroon reported the clinical outcome of "*automated percutaneous diskectomy*" using the Nucleotome [5, 6].

## 53.3 Era of Endoscopic Diskectomy (Contained Disk Herniation) (Fig. 53.2)

After the percutaneous diskectomy age, an endoscope termed the *diskoscope* was applied to the percutaneous diskectomy technique for direct visualization. Hausmann and Forst introduced a nucleoscope for viewing the intervertebral disk space [7]. Schreiber and Suezawa firstly reported a transdiskoscopic nucleotomy technique [8]. In the study, some of the patients underwent percutaneous diskectomy performed under the diskoscopic visualization. In 1988, Kambin et al. reported the first intraoperative diskoscopic views of herniated nucleus pulposus and he suggested that the visualization of the epidural space would be very important in his later articles [9]. In 1989, Schreiber et al. used a *biportal approach* with a diskoscope and injected indigocarmine dye into the disk to discriminate abnormal nucleus pulposus and annular fissure [10].

In 1990, Kambin also described an important anatomical feature regarding the transforaminal approach – the so-called *triangular working zone* in which there is neither vessel nor nerve [1]. The safe zone is bordered anteriorly by the exiting nerve root, inferiorly by the end plate of the lower vertebra, posteriorly by the superior articular process of the inferior vertebra, and medially by the traversing nerve root. The definitive description of the triangular working zone enabled the introduction of larger endoscope with larger instruments and more sophisticated decompression without exiting nerve root damage.

Mayer and Brock described the technique of *percutaneous endoscopic lumbar diskectomy* (*PELD*) for contained disk herniation using an angled lens scope allowing dorsal vision around the annular tear [11]. This was similar to Schreiber's biportal approach. They removed the herniated nucleus with rigid or flexible forceps, as well as with automated shaver system under

intermittent endoscopic control (diskoscopy). Since then, the so-called PELD is one of the representative terms of endoscopic lumbar diskectomy techniques.

## 53.4 Era of Endoscopic Diskectomy (Noncontained Disk Herniation) (Fig. 53.3)

Mathews 1996 [12] and Ditsworth 1998 [13] opened the era of real *transforaminal approach*. Ditsworth described the technique in which a working channel endoscopy passes completely through the foramen into the spinal canal and the surgeon directly removes free fragments and decompresses the nerve root and dural sac [13]. Since then, a truly transforaminal approach, as opposed to just going through part of the foramen and into the disk, has been developed and the target disk pathology is broaden from contained disk herniation.

In 1996, Kambin and Zhou published a technique of endoscopic decompression for the treatment of lateral recess syndrome. They decompressed the nerve roots, compromised by lateral recess stenosis, by annulectomy and osteophytectomy with forceps and trephines in using angled endoscopes (0- and 30-degree endoscopes) [14].

They described that in the case of large central disk herniation at the level of L5–S1, sequestered disk herniation, or migrated disk herniation, open surgery was required [14, 15].

## 53.5 Era of Endoscopic Selective Diskectomy (Noncontained Disk) (Fig. 53.4)

The basic concept of the percutaneous endoscopic surgery has been evolved from an indirect intradiskal decompression to a direct epidural or selective neural decompression. The working space has been extended from central nucleus to periannular and finally epidural space [16]. In the new millennium, various advanced endoscopic techniques have been



Fig. 53.3 Transforaminal endoscopic diskectomy for noncontained disk herniation



Fig. 53.4 Transforaminal endoscopic selective diskectomy

developed. *Selective endoscopic diskectomy* was named by Kambin et al. [17] and Yeung [18] independently. Yeung used the Yeung Endoscopic Spine System (YESS<sup>TM</sup>) which has a rigid rod lens and integrated, multichannel, wide-angled endoscope [19]. They described the selective endoscopic diskectomy technique

for extruded lumbar disk herniation. They also introduced a foraminoplastic approach at the L5–S1 level. In 2003, the endoscopic surgical system became more contemporary as launching YESS designed around the transforaminal endoscopic approach for intradiskal and epiduroscopic procedures. Yeung and Yeung also described the utility of provocative intraoperative diskography, thermal diskoplasty and annuloplasty, and annular resection for creation of an annular window to perform for aminoplasty using abrasive drills, burrs, and lasers [20]. Some authors reported the endoscopic technique for various situations such as recurrent disk herniation, disk herniation at L5–S1 level, migrated disk herniation, or upper lumbar disk herniation. Ahn and Lee described the endoscopic technique for recurrent disk herniation and upper lumbar disk herniation [21, 22]. Sometimes, the standard posterolateral approach might be associated with problems in reaching the epidural space due to anatomical peculiarities. This problem of poor visualization of the epidural space was solved with an extreme lateral approach described by Reutten et al. [23]. They pointed out that the usual transforaminal access is posterolateral and associated with problems in reaching the epidural space directly with unhindered vision, and then they described an extreme lateral access into the spinal canal using the full-endoscopic uniportal transforaminal approach. At the same time, Schubert and Hoogland described a foraminoplastic approach with bone reamer to remove the migrated and sequestrated disk herniation [24]. They used a bone reamer to undercut the part of superior facet to reach the epidurally extruded disk fragment. Choi and Lee introduced a technique of interlaminar approach to L5-S1 level or migrated disk herniation from L4 to 5 level [25, 26]. Lee et al. applied a classification system for the migrated disk herniation and demonstrated the clinical outcomes according to the disk migration zone [27].

In 2006, Lee et al. studied the limitation of endoscopic diskectomy in the aspect of the size and location of disk herniation [28]. They concluded that open surgery may be considered for herniations with high-canal compromise and high-grade migration. Lee et al. also described percutaneous endoscopic intraannular subligamentous *herniotomy* [29]. The concept of herniotomy is removal of whole iceberg, not just the tip of the iceberg. It is important to prevent recurrent disk herniation or incomplete removal which is one of the main concerns about minimally invasive endoscopic surgery [16].

## 53.6 Era of Endoscopic Foraminal Decompression (Fig. 53.5)

Although various foraminoplastic approaches were introduced, most of the techniques were not for true foraminal decompression, but for approach to the intracanal pathologies. Therefore, the history of foraminal endoscopic decompression started in the late 1990s. Knight et al. introduced endoscopic laser foraminoplasty for various foraminal nerve root entrapment syndromes [30-32]. The basic concept of foraminoplasty is reshaping foramen by ablating soft tissues such as foraminal ligaments and osteophyte using side-firing laser under endoscopic visualization. Ahn et al. described a percutaneous endoscopic lumbar foraminotomy technique using bone reamer and laser [33]. Schubert and Hoogland also reported the use of a bone reamer for foraminoplasty in the case of migrated disk herniation [24]. However, the previous techniques have limitations for definite foraminal decompression. The use of laser is effective for only neural entrapment caused by soft tissue or fragile osteophyte. However, it may be less effective for severe bony foraminal stenosis. Blind use of bone reamer also has inherent limitations such as bone bleeding and neural injury because it is a blind technique without any direct vision control. Nowadays, some authors reported more advanced endoscopic foraminal decompression techniques that can be performed for severe foraminal stenosis cases. The use of endoscopic burr and endopunches enables a safer and more effective fullscale foraminal decompression [16, 34].



Fig. 53.5 Percutaneous endoscopic lumbar foraminotomy

## 53.7 Era of Endoscopic Treatment for Diskogenic Back Pain (Fig. 53.6)

Most previous studies on percutaneous endoscopic surgeries have mainly described the technique which treats the lumbar radiculopathy caused by nerve root compromise. Several pioneers tried to treat diskogenic back pain with percutaneous endoscopic techniques. In 2004, the surgical technique of minimal access posterolateral transforaminal selective endoscopic diskectomy and bipolar radiofrequency thermal annuloplasty to interrupt the purported annular defect pain sensitization process was introduced to treat the patients with chronic lumbar diskogenic pain [35]. In this study, a total of 113 patients were recruited; however, the results showed lack of clinical benefit from this procedure. In 2010, Ahn and Lee reported the outcome predictors of percutaneous endoscopic lumbar diskectomy and thermal annuloplasty (PELDTA) for diskogenic low back pain, in which 83.5 % patients showed symptomatic improvement and the success rate was 70.9 % [36]. In their conclusion, PELDTA could be effective for the patients with chronic diskogenic low back pain. Lee and Kang used laser-assisted spinal endoscopy (LASE) kit for the percutaneous intradiskal



**Fig. 53.6** Percutaneous endoscopic lumbar annuloplasty for diskogenic back pain

decompression to evaporate and shrink the posterior and central nucleus for improvement of lumbar pain [37]. Percutaneous endoscopic laser annuloplasty, a new minimally invasive technique, used LASE for direct coagulation of the painful inflamed granulation tissue with new vessels and nerves in 30 patients with diskogenic low back pain. They reported favorable outcomes for carefully selected groups of patients with diskogenic low back pain. Although the theoretical background is fascinating, the clinical application of this technique is not, as of yet, established. The patient selection and surgical technique should be more standardized to produce a reliable clinical result in the future.

## 53.8 Era of Laser Diskectomy and Radiofrequency Ablation

The application of the laser to minimally invasive or percutaneous lumbar surgery is very attractive to spine surgeons because of the ability to deliver a large amount of energy through a small fiber to a focused spot area. The laser tissue interaction can be classified into three types: photochemical effects, photothermal effects, and photomechanical and photoionizing effects [38]. Until now, for percutaneous laser disk decompression (PLDD), lasers in the near-infrared region (Nd:YAG, Ho:YAG, and diode lasers) and with visible green radiation (frequency doubled Nd:YAG, called "KTP laser") were reported to be effective [38]. The basic technique of PLDD is similar for all trials. The procedure is conducted under local anesthesia of the skin and underlying muscles. After assessment of the correct disk level by using fluoroscopy, a hollow needle is inserted 10 cm from the midline, pointing toward the center of the disk. A laser fiber (0.4 mm) is inserted through the needle into the center of the nucleus pulposus. Laser energy is then delivered into the nucleus pulposus to vaporize its content and reduce intradiskal pressure [39].

In the mid-1980s, Ascher and Choy developed percutaneous laser diskectomy technique [40, 41]. Ascher firstly applied laser into the disk surgery using a neodymium:yttrium-alluminumgarnet (Nd:YAG) laser [40]. The procedure included fluoroscopically guided insertion of an 18-gauge needle into the disk space through which 400-nm laser fiber was advanced into the disk space. Approximately 1200 J of energy in short bursts to avoid heating the adjacent tissues was delivered to ablate a small intradiskal tissue. Through the spinal needle, the vaporized tissue was escaped. Finally, an adhesive bandage covered the needle site and patient was discharged. In 1990, Davis performed laser diskectomy with the potassium titanyl-phosphate (KTP) 532-nm laser in 40 patients and reported 85 % success rates [42]. Choy et al. introduced the new technique of percutaneous laser disk decompression (PLDD) in 1992 [43]. They reported the clinical outcomes of 333 patients with 78.4 % of good to fair results. Subsequently, Mayer and Brock suggested a combined technique between laser ablation and endoscopy in order to keep observing the amount of removing disk tissues during surgery [11]. There was a report on comparison study between Ho:YAG laser diskectomy and conservative treatment in 1993 [44]. Even though they could not identify a significant difference in terms of complications, they concluded that the laser diskectomy was a safe procedure to be effective in relieving symptoms in some patients. There was also a comparison study between the Ho:YAG and ND:YAG laser in 1994 [45]. They demonstrated that Ho:YAG laser was the best for compromising between efficacy of absorption and convenience of fiber-optic delivery. In 1995, Casper et al. reported an 84 % success rate at 1-year follow-up in the patients treated by the side-firing Ho:YAG laser [46]. In 1998, Knight et al. reported endoscopic laser foraminoplasty using a side-firing Ho: YAG laser for chronic low back pain and sciatica [30]. Then, Hellinger started to use the Ascher technique for Nd:YAG laser ablation in 1999, who had treated more than 2500 patients for 13 years and his overall success rate was approximately 80 % [47]. In 2000, Yeung reported an 84 % success rate in his experiences with more than 1000 patients whose herniated lumbar disks were treated by KTP laser [48]. Nowadays, the laser can be used as a supplementary role in ablation of bone and soft tissue. It can be also used for thermal laser annuloplasty in the treatment of diskogenic back pain.

#### 53.8.1 Radiofrequency Ablation

High-frequency radiofrequency (RF) ablation has progressed with developing several applications in spinal surgical field. To dissect and coagulate the tissue, high-frequency RF keeping low temperatures during surgical works was developed. Also on demand for targeted application and precise tissue ablation, Trigger-Flex Bipolar System (Elliquence, New York, USA) was developed, which is compatible with other endoscopic spinal surgery system.

RF ablation is considered a promising alternative to lasers in terms of their cost, safety, and convenience issues. In 2004, Tsou et al. reported the surgical outcomes of 113 consecutive patients with diskogenic low back pain, minimum 2-year follow-up, in which they used low-temperature RF tissue ablation [35]. There were no aborted procedures, unexpected hemorrhage, devicerelated complications, neurologic deficits, or instability.

Nowadays, the high-frequency RF with lowtemperature tissue ablation has been involved as a part of endoscopic spinal surgical system, and it is suggested that this RF ablation is very effective in controlling bleeding and shrinking tissue.

#### 53.9 Randomized Trials

Through the history of percutaneous endoscopic lumbar surgery, there are only four randomized controlled trials in which the effectiveness of percutaneous endoscopic diskectomy was compared to conventional open diskectomy [49–52]. In 1993, Mayer and Brock firstly reported a randomized trial evaluating the effectiveness of percutaneous endoscopic diskectomy for contained disk herniation compared to microdiskectomy [51]. They concluded that percutaneous endoscopic diskectomy appeared to offer an alternative to microdiskectomy for patients with "contained" and small subligamentous lumbar disk herniations. Hermantin et al. (n=60) demonstrated that the satisfaction rate between the endoscopy group and the open diskectomy group was the same [49]. However, the period of postoperative disability or narcotic use was significantly shorter in the endoscopy group. Hoogland et al. conducted a prospective randomized study recruiting 280 patients with lumbar disk herniation treated by endoscopy alone or endoscopic diskectomy with intradiskal injection of low-dose (1000-U) chymopapain [50]. Ruetten et al. (n=178) used full-endoscopic transforaminal diskectomy using 4.2-mm working channel endoscope, and they demonstrated that the results were comparable to those in conventional open-disk surgery [52]. However, the current level of evidence on the effectiveness of transforaminal endoscopic diskectomy is low because most studies have substantial design weaknesses in the randomization method or outcome measures with a high risk of bias. Only one adequately randomized controlled study has been identified thus far [49]. Therefore, high-quality randomized controlled trials with sufficiently large sample sizes are required to provide valid information on the effectiveness of endoscopic diskectomy in the future [16, 53].

#### 53.10 Future Perspective

The technical improvement of percutaneous endoscopic lumbar surgery has been splendid in surgical approaches, design of optics, and surgical instruments. Because of the rapid technical advancement, the paradigm of percutaneous endoscopic spine surgery is shifting. In earlier generations, the main topic of the percutaneous endoscopic surgery is soft disk herniation. Any stenotic component with lumbar disk herniation is regarded as a contraindication or a predictor of poor outcome for this technique. However, the indications are broadening to the degenerative lumbar stenosis. The percutaneous endoscopic decompression technique for various lumbar stenosis and even deformity correction will likely continue to evolve. Furthermore, this technique may be combined with various minimally invasive spine procedures such as motion-preservation technology, regenerative medicine, or biologic therapy for traumatic and degenerative spine disease.

Acknowledgments The authors would like to express a special acknowledgment to Ho Kim, B.S. and Jae Eun, Park, B.S. for preparing figures and manuscripts.

#### References

- Kambin P, editor. Arthroscopic microdiscectomy: minimal intervention spinal surgery. Baltimore: Urban & Schwarzenburg; 1990.
- Hijikata S. Percutaneous discectomy: a new treatment method for lumbar disc herniation. J Toden Hosp. 1975;5:5–13.
- Friedman WA. Percutaneous discectomy: an alternative to chemonucleolysis? Neurosurgery. 1983;13(5):542–7.
- Onik G, Helms CA, Ginsberg L, et al. Percutaneous lumbar diskectomy using a new aspiration probe: porcine and cadaver model. Radiology. 1985;155(1): 251–2.
- Maroon JC, Onik G. Percutaneous automated discectomy: a new method for lumbar disc removal. Technical note. J Neurosurg. 1987;66(1):143–6.
- Onik G, Maroon J, Helms C, et al. Automated percutaneous diskectomy: initial patient experience. Work in progress. Radiology. 1987;162(1 Pt 1):129–32.
- Hausmann B, Forst R. Nucleoscope. Instrumentarium for endoscopy of the intervertebral disc space. Arch Orthop Trauma Surg. 1983;102(1):57–9.
- Schreiber A, Suezawa Y. Transdiscoscopic percutaneous nucleotomy in disk herniation. Orthop Rev. 1986;15(1):35–8.
- Kambin P, Nixon JE, Chait A, et al. Annular protrusion: pathophysiology and roentgenographic appearance. Spine (Phila Pa 1976). 1988;13(6):671–5.
- Schreiber A, Suezawa Y, Leu H. Does percutaneous nucleotomy with discoscopy replace conventional discectomy? Eight years of experience and results in treatment of herniated lumbar disc. Clin Orthop Relat Res. 1989;238:35–42.
- Mayer HM, Brock M. Percutaneous endoscopic lumbar discectomy (PELD). Neurosurg Rev. 1993;16(2):115–20.
- Mathews HH. Transforaminal endoscopic microdiscectomy. Neurosurg Clin N Am. 1996;7(1):59–63.
- Ditsworth DA. Endoscopic transforaminal lumbar discectomy and reconfiguration: a postero-lateral approach into the spinal canal. Surg Neurol. 1998;49(6):588–97. discussion 597–8.
- Kambin P, Zhou L. History and current status of percutaneous arthroscopic disc surgery. Spine (Phila Pa 1976). 1996;21(24 Suppl):57S–61.
- Kambin P, Casey K, O'Brien E, et al. Transforaminal arthroscopic decompression of lateral recess stenosis. J Neurosurg. 1996;84(3):462–7.
- Ahn Y. Transforaminal percutaneous endoscopic lumbar discectomy: technical tips to prevent complications. Expert Rev Med Devices. 2012;9(4):361–6.
- Kambin P, O'Brien E, Zhou L, et al. Arthroscopic microdiscectomy and selective fragmentectomy. Clin Orthop Relat Res. 1998;347:150–67.
- Yeung AT, Tsou PM. Posterolateral endoscopic excision for lumbar disc herniation: surgical technique, outcome, and complications in 307 consecutive cases. Spine (Phila Pa 1976). 2002;27:722–31.

- Yeung AT. Minimally invasive disc surgery with the Yeung Endoscopic Spine System (YESS). Surg Technol Int. 1999;8:267–77.
- Yeung AT, Yeung CA. Advances in endoscopic disc and spine surgery: foraminal approach. Surg Technol Int. 2003;11:255–63.
- Ahn Y, Lee SH, Park WM, et al. Percutaneous endoscopic lumbar discectomy for recurrent disc herniation: surgical technique, outcome, and prognostic factors of 43 consecutive cases. Spine. 2004;29(16): E326.
- 22. Ahn Y, Lee SH, Lee JH, et al. Transforaminal percutaneous endoscopic lumbar discectomy for upper lumbar disc herniation: clinical outcome, prognostic factors, and technical consideration. Acta Neurochir (Wien). 2009;151(3):199–206.
- Ruetten S, Komp M, Godolias G. An extreme lateral access for the surgery of lumbar disc herniations inside the spinal canal using the full-endoscopic uniportal transforaminal approach-technique and prospective results of 463 patients. Spine (Phila Pa 1976). 2005;30(22):2570–8.
- Schubert M, Hoogland T. Endoscopic transforaminal nucleotomy with foraminoplasty for lumbar disk herniation. Oper Orthop Traumatol. 2005;17(6):641–61.
- 25. Choi G, Lee SH, Raiturker PP, et al. Percutaneous endoscopic interlaminar discectomy for intracanalicular disc herniations at L5–S1 using a rigid working channel endoscope. Neurosurgery. 2006;58(1 Suppl):ONS59–68. discussion ONS59-68.
- 26. Choi G, Lee SH, Lokhande P, et al. Percutaneous endoscopic approach for highly migrated intracanal disc herniations by foraminoplastic technique using rigid working channel endoscope. Spine (Phila Pa 1976). 2008;33(15):E508–15.
- Lee S, Kim SK, Lee SH, et al. Percutaneous endoscopic lumbar discectomy for migrated disc herniation: classification of disc migration and surgical approaches. Eur Spine J. 2007;16(3):431–7.
- Lee SH, Kang BU, Ahn Y, et al. Operative failure of percutaneous endoscopic lumbar discectomy: a radiologic analysis of 55 cases. Spine (Phila Pa 1976). 2006;31(10):E285–90.
- 29. Lee SH, Choi KC, Baek OK, et al. Percutaneous endoscopic intra-annular subligamentous herniotomy for large central disc herniation: a technical case report. Spine (Phila Pa 1976). 2014;39(7):E473–9. [Epub ahead of print].
- Knight MT, Vajda A, Jakab GV, et al. Endoscopic laser foraminoplasty on the lumbar spine – early experience. Minim Invasive Neurosurg. 1998;41(1):5–9.
- Knight MT, Goswami A, Patko JT, et al. Endoscopic foraminoplasty: a prospective study on 250 consecutive patients with independent evaluation. J Clin Laser Med Surg. 2001;19(2):73–81.
- Knight M, Goswami A. Management of isthmic spondylolisthesis with posterolateral endoscopic foraminal decompression. Spine (Phila Pa 1976). 2003;28(6): 573–81.

- Ahn Y, Lee SH, Park WM, et al. Posterolateral percutaneous endoscopic lumbar foraminotomy for L5–S1 foraminal or lateral exit zone stenosis. Technical note. J Neurosurg. 2003;99(3 Suppl):320–3.
- 34. Jasper GP, Francisco GM, Telfeian AE. A retrospective evaluation of the clinical success of transforaminal endoscopic discectomy with foraminotomy in geriatric patients. Pain Physician. 2013;16(3): 225–9.
- 35. Tsou PM, Alan Yeung C, Yeung AT. Posterolateral transforaminal selective endoscopic discectomy and thermal annuloplasty for chronic lumbar discogenic pain: a minimal access visualized intradiscal surgical procedure. Spine J. 2004;4(5):564–73.
- 36. Ahn Y, Lee SH. Outcome predictors of percutaneous endoscopic lumbar discectomy and thermal annuloplasty for discogenic low back pain. Acta Neurochir (Wien). 2010;152(10):1695–702.
- Lee SH, Kang HS. Percutaneous endoscopic laser annuloplasty for discogenic low back pain. World Neurosurg. 2010;73(3):198–206. discussion e33.
- Knappe V, Frank F, Rohde E. Principles of lasers and biophotonic effects. Photomed Laser Surg. 2004;22(5): 411–7.
- Schenk B, Brouwer PA, Peul WC, et al. Percutaneous laser disk decompression: a review of the literature. AJNR Am J Neuroradiol. 2006;27(1):232–5.
- Ascher PW. Status quo and new horizons of laser therapy in neurosurgery. Lasers Surg Med. 1985; 5(5):499–506.
- Choy DS, Case RB, Fielding W, et al. Percutaneous laser nucleolysis of lumbar disks. N Engl J Med. 1987;317(12):771–2.
- Davis JK. Percutaneous discectomy improved with KTP laser. Clin Laser Mon. 1990;8(7):105–6.
- Choy DS, Ascher PW, Ranu HS, et al. Percutaneous laser disc decompression. A new therapeutic modality. Spine (Phila Pa 1976). 1992;17(8):949–56.
- Sherk HH, Black JD, Prodoehl JA, et al. Laser diskectomy. Orthopedics. 1993;16(5):573–6.

- Quigley MR, Maroon JC, Shih T, et al. Laser discectomy. Comparison of systems. Spine (Phila Pa 1976). 1994;19(3):319–22.
- 46. Casper GD, Mullins LL, Hartman VL. Laser-assisted disc decompression: a clinical trial of the holmium: YAG laser with side-firing fiber. J Clin Laser Med Surg. 1995;13(1):27–32.
- Hellinger J. Technical aspects of the percutaneous cervical and lumbar laser-disc-decompression and -nucleotomy. Neurol Res. 1999;21(1):99–102.
- Yeung AT. The evolution of percutaneous spinal endoscopy and discectomy: state of the art. Mt Sinai J Med. 2000;67(4):327–32.
- 49. Hermantin FU, Peters T, Quartararo L, et al. A prospective, randomized study comparing the results of open discectomy with those of video-assisted arthroscopic microdiscectomy. J Bone Joint Surg Am. 1999;81(7):958–65.
- 50. Hoogland T, Schubert M, Miklitz B, et al. Transforaminal posterolateral endoscopic discectomy with or without the combination of a low-dose chymopapain: a prospective randomized study in 280 consecutive cases. Spine. 2006;31(24):E890–7.
- Mayer HM, Brock M. Percutaneous endoscopic discectomy: surgical technique and preliminary results compared to microsurgical discectomy. J Neurosurg. 1993;78(2):216–25.
- Ruetten S, Komp M, Merk H, et al. Full-endoscopic interlaminar and transforaminal lumbar discectomy versus conventional microsurgical technique: a prospective, randomized, controlled study. Spine (Phila Pa 1976). 2008;33(9):931–9.
- Nellensteijn J, Ostelo R, Bartels R, Peul W, van Royen B, van Tulder M. Transforaminal endoscopic surgery for symptomatic lumbar disc herniations: a systematic review of the literature. Eur Spine J. 2010;19(2): 181–204.

## The Importance of a Registry in Spinal Surgery

54

Emin Aghayev, Christoph Röder, Helton L.A. Defino, Carlos F. Herrero, and Max Aebi

## 54.1 Mainstays of Patient Care

The mainstays of patient care throughout the ages used to be intuition, psychology, and charisma. In this environment, which was characterized by trust on the part of the patients and society and self-confidence and dedication to the cause on the part of the clinician, considerable advances in medical therapy were made. Only a few play-

e-mail: emin.aghayev@memcenter.unibe.ch; christoph.roeder@memcenter.unibe.ch

Hirslanden-Salem Hospital, "Das Rückenzentrum", Bern, Switzerland e-mail: max.aebi@memcenter.unibe.ch ers in the medical arena made initiatives for a systematic assessment of what was done and what the result of those treatments were. Among them was Florence Nightingale, a nurse, who applied statistical methods for analyzing preventable deaths in the British military during the Crimean War as early as 1854. Ernest Codman, a US physician and the father of what is today considered as outcome management in patient care, became famous in the early 1900s for his "endresult system" which stated that every patient needs to be followed up to assess the benefits and complications of the received treatment. Finally, Maurice E. Müller, cofounder of AO/ASIF (Arbeitsgemeinschaft für Osteosynthesefragen/ Association for the Study of Internal Fixation), published his concept of a multi-site trauma registry with centralized database for assessment of surgeon performance, efficacy of surgical techniques, and postmarket surveillance of implants in 1963 [1].

## 54.2 Quality Control and Outcome Research

## 54.2.1 Definition of Quality in Health Care

To those not involved in quality improvement in a professional capacity, it might appear relatively simple to define quality; however, more than

E. Aghayev, MD, MSc (⊠) • C. Röder, MD, MPH Institute for Evaluative Research in Medicine, University of Bern, Stauffacherstrasse 78, CH-3014 Bern, Switzerland

H.L.A. Defino, MD • C.F. Herrero, MD Department of Biomechanics, Medicine and Rehabilitation of the Locomotor System, Ribeirão Preto Medical School, Ribeirão Preto, Brazil e-mail: hladefin@fmrp.usp.br

M. Aebi, MD, DHC

Institute for Evaluative Research in Medicine, University of Bern, Stauffacherstrasse 78, CH-3014 Bern, Switzerland

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016 J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_54

2000 years after Plato invented this term, there is still great debate regarding the meaning of the word [2]. The American Society for Quality (ASQ) defines quality as "a subjective term for which each person or sector has his or her own definition" [3]. According to a user-based approach, quality can be defined as "meeting or exceeding customer satisfaction" [4]. Quality is a multidimensional construct, and the dimensions are specific for each category. The US Agency for Healthcare Research and Quality defines quality in health care as "doing the right thing, at the right time, in the right way, for the right person, and having the best possible results [5]."

The quality measures in health care assess four components:

- Structures (resources such as staff or equipment)
- Indications (making the right therapeutic decision for a given pathology and its stage)
- Processes (therapeutic interventions, prescribing, interactions with patients)
- Outcomes (end results of health care such as mortality and patient satisfaction) [5, 6]

The measures used to obtain the patients' views can be classified into three categories:

- Preferences
- Evaluations
- Reports

Wensing and Elwyn defined preferences as patient ideas about what should occur in healthcare systems. Evaluations are patients' "reactions" to their experience of health care and reports are objective observations (e.g. how long the patients had to spend in the waiting room). The choice of the type of measure depends on the aspect being assessed and the purpose of the evaluation (educational, certification, accreditation, quality control, or quality improvement) [7]. One of the most widespread means of measuring processes and outcomes is the assessment of patient satisfaction (evaluation category). Outcome satisfaction is also one of the criteria for assessing the validity of process measures. Indeed, according to Chassin [8], a measure of process is valid when it is related to health outcomes (e.g., mortality, patient satisfaction, etc.). Hence, the responses to questions concerning satisfaction with treatment, typically used in treatment outcome studies, can also be seen as outcome measures in the quality control and improvement context.

### 54.2.2 Overlap of Outcome Research and Quality Control

Quality control is an important concept in all medical disciplines, but a factual implementation and application in a stringent and meaningful way is still lacking in many cases. The growing emphasis on an evidence-based approach in the medical setting has led to a corresponding increase in the number and quality of studies in the twenty-first century examining the efficacy of surgical and nonsurgical treatments. These studies are usually conducted in university hospitals and clinics that have an inhouse research staff or that cooperate with academic research institutions. The studies are not commonly perceived by the care provider (hospitals and clinics) as being something from which they can benefit, from an economical point of view; in contrast, carrying out such research can sometimes be seen as a drain on resources. The research activities on treatment outcomes are merely seen as something that may indirectly benefit the institution in terms of prestige and corporate social responsibility. However, the possibility of economic benefit from corporate social responsibility activities is not a sufficiently persuasive argument for increasing investment in research; otherwise, all the public and private hospital and clinics would likely have their own research departments or research staff. Significantly, in all of this, one important factor is typically overlooked: research projects in the field of treatment outcomes and their predictors can be useful to the provider in a much more direct way in terms of quality improvement and the control of service performance [9].

## 54.3 "Observational" Versus "Experimental"

The main goal of outcome research and quality control is the collection of distinct evidence, and an appropriate study design is one of its main prerequisites. The study designs can be divided into observational, such as a registry, and experimental ones, such as an RCT (Fig. 54.1).

Levels of evidence are defined by the study design and execution, and designs have been variously graded by their potential to eliminate bias [10, 11]. A hierarchy of study designs was first proposed by Campbell and Stanley [10]. The concept of evidence levels has been widely promoted since then to grade recommendations for clinical practice. The dogma of an RCT and RCT-based meta-analyses having superior evidence over an observational study has long been held as gospel despite numerous appeals to reconsider and to adapt the evidence levels [11– 16]. The reasons for this may be that the only available evidence pyramids focuses on efficacy studies, which allow for an accurate assessment of cause and effect. The terminology "observational" connotes that the data were collected without interference, so the decision to assign patients to an exposure is not random and may be self-determined by confounding variables that are highly correlated with the outcome. As a result, an observed significant difference in outcomes may not necessarily be attributable to the choice of treatment alone but to the confounding variable (or to some combination). Thus, according to the current evidence-based medicine practice, a small sample-sized RCT would bring undoubtful superior evidence than a large observational study. RCT study design has a high acceptance in all clinical disciplines despite a long list of its limitations such as ethical considerations, organizational burden, complexity of random allocation, complexity of blinding, time and personal resources consumption, compromised integrity of the clinical context, difficulty in studying rare events and outcomes in distant future, limitations in evaluation of populationbased interventions [11], sample size limitation for definitive results [17], patient refusal of participation, narrowing of the studied population, and limited external validity [18], as well as financial efforts. This is also despite the fact that so many efforts are required to answer only one main posed hypothesis in an RCT and the fact that the number of non-randomizable clinical questions is by far much higher than the number of randomizable ones.

In orthopedic surgery and traumatology, among other medical disciplines, it is often diffi-



Fig. 54.1 Anatomy of medical research

cult to provide an ethically justifiable patient randomization for differing and often insufficiently studied treatment methods. In medical disciplines other than surgical ones, like in conservative treatments or in pharmacology, the effect of a treatment is often assessed in comparison to a placebo. In surgical disciplines and particularly in orthopedic implantology, placebo comparisons are difficult to imagine. This is a nonnegligible and important organizational restriction of a randomized controlled study design. There are particular questions in implantology which may justify an RCT; however, the method of choice for implant comparisons is a registry with sufficient participation to ensure high external validity, i.e., generalizability of results to the routine clinical settings [19]. Particularly when introducing new implants and aiming at postmarket surveillance, an observational approach allowing for benchmarking of new and older implants appears to be the most reasonable one [19].

Moreover, one should be aware that a large observational cohort and an RCT are per se designed to answer different questions, and their different goals and setups result in their different characteristics (Table 54.1). Importantly, while an RCT answers the question "can it work?" an observational study addresses the question "does it work?" [20] In a large dataset, one could add "in whom does it work best?" as one of the key questions that the large observational comparisons can answer, since they include better and poorer indications for a certain therapy, as it is mostly the case when an innovation is adopted by the larger medical community. Table 54.1 summarizes characteristics of an RCT versus registry.

On the other hand, strong evidence exists on the similarity of the results in RCTs and in observational studies. Benson and Hartz reviewed RCTs and observational studies between 1985 and 1998 and found that in a vast majority of therapeutic comparisons, the estimates of the treatment effects from observational studies and randomized controlled trials were similar [13]. In the same time, Concato et al. reviewed literature between 1991 and 1995 and concluded that the results of well-designed observational studies with either a cohort or a case-control design do not systematically overestimate the magnitude of the effects of treatment as compared with those in randomized controlled trials on the same topic, as it is anticipated [12]. In orthopedics, Bhandari et al. have shown that, when adjusted for important risk factors, the results of observational studies (n=13) on revision and mortality rates after internal fixation of femoral neck fracture were similar to those in RCTs (n=14) [21]. Equality of the results in an RCT and in a cohort study was shown also in spinal surgery based on the results from SWISSspine registry [22] and the Spine Tango results of a single center [23].

The proponents of an observational study design are often opponents of an experimental approach and vice versa. In 1996, Black wrote that there is no perfect study design and that "the false conflict between those who advocate randomized trials in all situations and those who believe observational data provides sufficient evidence needs to be replaced with mutual recognition of the complementary roles of the two approaches" [24]. Spinal surgery is one of the youngest and, in the same time, one of the fastest growing clinical disciplines with high and further growing epidemiological and financial impacts on society. The evidence demand in spinal surgery is therefore high, and there is a compelling need for quality assurance, outcome research, and postmarket surveillance of spinal implants.

## 54.4 "Spine Tango": An International Spine Registry for Quality Assurance, Outcome Research, and Postmarket Surveillance of Implants

#### 54.4.1 History and Objectives

In the late 1990s, the founder of the first spine unit in Switzerland – Prof. Dieter Grob – was asked by the cantonal government of Zurich to conduct an outcome study of the results of surgical interventions on the spine. The investigation was completed and the data provided to the spon-

Characteristics	DCT	Pagistry
Type of evidence	Efficacy	
Principal question	Can it work? (the first step of evidence generation)	Does it work? (verification in clinical practice)
Internal validity (methodological quality)	+++	+/++ (expandable with, e.g., monitoring, audits, or validation with secondary data, etc.)
External validity (transferability/ generalizability)	-	+++
Bias	Low to very low	High to low depending on set-up (appropriate set-up and evaluation methodology reduces bias)
Levels of evidence	1a, 1b	2b-4, depending on methodology
Hypothesis-based approach	Yes	Usually no
Duration of observation period	Predefined	Predefined or open end
Focus of research/measurement	Sharp, narrow (hypothesis driven)	Broad
Quality assessment	Not intended (strictly defined indications, process quality at least derivable, outcome quality depends on effectiveness, a given indication, and process)	Structure, indication, process, outcome
Early warning system	Not possible	Possible
Long-term follow-up	Feasible	Feasible (depending on set-up only for a representative sample)
Coverage	Only among participants	From individual center/surgeon over representative clinic sample to full national/regional coverage
Benchmarking	Only benchmarking of RCT arms	Various (depending on the final composition of participants regional to nationally representative benchmark)
Type of quality assurance	Internal, external vs. benchmark of participants	Internal, external vs. representative regional or national benchmark
Efforts	Very high	Low
Cost per case	High to very high	Low
Cost per study	High to very high	Low-cost basis
Availability of potential patients	Usually low	Usually high
Patient compliance	Potentially low	High
Use of generated data	Only in the framework of the scientific goal/hypothesis	Open-hypothesis generation possible
Comparator	Given per definition	Ranges between none to numerous comparators, depending on registry set-up

Table 54.1 Characteristics of an RCT and a registry

<sup>a</sup>Unclear terminology, Cochrane called it "efficiency," better always specify what is meant (evidence derived from controlled experiment versus evidence derived from routine clinical practice)

sor. There was no further feedback or consequences as a result of this study. This triggered the idea of conducting a similar project on a larger scale. The Institute for Evaluative Research in Medicine in Bern, Switzerland, had by the time already an over 20-year experience with orthopedic registries, as the first detailed hip arthroplasty registry, the so-called IDES (International Documentation and Evaluation System), was set up in Bern in 1974 by Prof. institute lead in 2000, and who was a close friend and collaborator of Prof. Dieter Grob, put the vision of a first international spine registry to reality.

All over the world, efforts were being made to set up orthopedic registries on regional, state, or even national levels. Spine surgery represents a challenge for all registry endeavors. The variety of levels, pathologies, accesses, and surgical techniques confounds all attempts to invent a short comprehensive yet questionnaire. Consequently, the project launched by Grob and Aebi was first based on sketches and discussions on form content, large personal commitment and motivation, and by the most modern approach of the time - the internet-based approach. This effort was later introduced to SSE, the Spine Society of Europe as "Spine Tango," and the society adopted the registry in the early 2000s.

Goals of the Spine Tango are:

- Quality assurance and quality improvement
- Presentation of the *state-of-the-art* European spinal therapies, including all pathologies, levels, accesses, single as well as two-staged surgeries, and nonsurgical treatments
- Outcome research and prospective observational evaluation of different surgical and nonsurgical techniques as an alternative to randomized controlled trials
- *Benchmarking* on national and international levels
- Postmarket surveillance of implants

In the mid- and late 1990s, Swedish spine surgeons also implemented a spine registry based on a short questionnaire dedicated to low back pain [25]. This effort become a national one and included other spinal pathologies with various treatments in the early 2000s [26–28].

Spine Tango is probably the first spine registry initiative to face the challenge of developing a comprehensive questionnaire covering all major spine pathologies and treatments, as well as spanning all anatomical levels. To accomplish this task, a technically demanding computer application was a prerequisite. The consensus and piloting process for the Spine surgical questionnaires' "surgery," Tango "staged surgery," and "follow-up" took about 5 years and needed around 4000 completed forms. The result are two double-sided A4 questionnaires (surgery, staged surgery) and one single-sided questionnaire for follow-up that can all be completed online or using scannable paper questionnaires. At the same time that the physician-based content was finalized, a working group at the Schulthess hospital in Zurich, Switzerland, had developed and validated the COMI (Core Outcome Measures Index) instruments for neck and low back based on a proposal by Deyo et al. [29], which became the officially recommended patient-based outcome instruments in the framework of the Spine Tango registry [30]. Until today, the Spine Tango database has grown to over 70,000 cases and currently includes 70 hospitals from 16 countries in Europe, the USA [31], but even Latin America, Asia, and Australia [32].

The cumulated numbers of the main Tango forms increase exponentially (Fig. 54.2). During the first three active years about 10,000, the second 3 years about 25,000, and during the last 3 years about 40,000 primary cases were documented.

#### 54.4.2 Content

The refined set of questions still allows for a documentation of the broad spectrum of pathologies and treatments in spine surgery. This is made possible by means of a list of main pathologies and their specifications and the so-called surgical matrix, a terminology system reducing the interventions to their basic principles - decompression, fusion, stabilization rigid, stabilization motion preserving, percutaneous procedures, and others. The duplication and, hence, separation of these principles into anterior and posterior ones complete the matrix. The exact content of the surgeon- and patient-based Tango forms was already previously described [33-35], and the forms are available on the Eurospine Web page [36]. Kessler et al. have developed the first version of



documentation content for conservative spinal treatments for Spine Tango in 2011 [37].

#### 54.4.3 Technology

Spine Tango has long left the early stage of a simple Web page for data entry and has grown into an international project with a sophisticated IT (information technology) structure and a multitude of clinical and scientific experts serving the user community and developing the registry further. The central database is part of the powerful scientific MEMdoc documentation portal hosted by the University of Bern; it offers various methods for clinical, implant, and radiographic data collection and a multitude of possibilities for data downloads and online statistical queries. An important step was the implementation of so-called modules, national satellite servers that filter out sensitive data for protecting user and patient privacy in the respective country before sending the clinical dataset to the central server in Switzerland [33]. Such modules are meanwhile installed in Germany, Austria, Italy, Poland, Belgium, Switzerland, Great Britain, Australia, USA, Mexico, and Brazil. Users whose country does not yet have such a national filter server may use the international module [36].

#### 54.5 Benefits of Spine Tango

In 2004, Aebi and Grob wrote that it was increasingly understood that in any technology-driven surgical discipline, patients should be documented in a standardized way if they are recipients of novel inventions and implants. This to have a common language for reporting outcomes, complications, and unforeseen incidents which can be better recognized in a large, common data pool serving as an early warning system [38]. These words could be considered as the conception framework of the Spine Tango endeavor. The registry has now grown into one of the largest international clinical registries with one of the highest number of participating countries. Based on modern information technology, the participating spine surgeons have built a database in which numerous spinal treatments and, importantly, treatment indications are backed up by facts. Within the registry, each participating surgeon, department, and hospital has its own performance record. The analyses based on Spine Tango data reach from single center-based studies [39–41], overbenchmarking against the pooled data of other hospitals [34] to comparative studies across spine registries [42]. Also, from a methodological point of view, the analyses range between methodological papers [43] and registry-based technical notes [34], over case series [39–41] and subgroup analyses to propensity score-based weighted comparison of multiple treatment options [44].

The online tools such as data download, online statistics, follow-up calendar, and annual and benchmarking reports aim at facilitation of reallife quality assessment, outcome research, and performance monitoring for the users. Some European and non-European national societies recognizing the value of the registry are evaluating the possibility of introducing Spine Tango as their national quality assessment and outcome research tool and make it mandatory for certified spine centers.

In the perspective of quality control, the registry is able to provide evidence on indications, processes, and outcomes. Assessments of quality of structures (e.g., resources such as staff or equipment) are only a secondary goal but remain feasible for certain questions and to a certain extent.

In the societal perspective, the collection of state-of-the-art evidence with high external validity has a direct impact on clinical practice and treatment quality as well as outcome improvement. An example from the Swedish Hip Registry showed that serious complications and revision rates have declined significantly by making information from the registry available to the entire community of surgeons in Sweden [45]. The Tango dataset is reaching reasonably high numbers, which should now allow for systematic data analyses, evaluations, and dissemination of the results. This requires, however, time and resources. The Spine Tango committee of Eurospine is engaged in promoting publication of the observed results in the registry. Additional to studies based on one registry, cross-registry studies are possible and may help in generation of stronger evidence [42].

#### 54.6 Criticism of a Registry

The most relevant criticism to a registry in terms of the evidence level of findings is the unmonitored collection of heterogeneous data prone to selection bias. In an unmonitored voluntary registry, selection bias cannot be excluded. Audit and monitoring of data recordings are the main tools in reducing selection bias and controlling and improving the quality of the observational data. Aiming at collection of high-quality data, the Spine Tango committee plans to introduce the so-called pool of accredited participants. Those are participants who have a minimum 80 % case documentation rate. A case is defined as a combination of a preoperative COMI form, the surgical intervention form, and at least one postoperative COMI and physician follow-up form at an interval of 3 months after surgery or later. For the nonsurgical participants, a case is defined as a pretreatment COMI form, the conservative treatment form, and an end-of-treatment assessment with a COMI form. On-site audits by an independent party will then be performed for those participants. One should be aware of the enormous efforts required for making a large international registry a monitored study. A snap sample monitoring, however, seems to be a rather feasible endeavor.

Making documentation mandatory within hospital or national boarders is a further instrument against selection bias. As was mentioned above, some European and non-European national societies are evaluating the possibility of introducing Spine Tango as their national quality assessment and outcome research tool and making documentation mandatory. These policy changes are, however, lengthy and difficult negotiations, and their political dimensions and controversies should not be underestimated.

Many industry partners are interested in the Spine Tango registry as well. It is increasingly being recognized as a unique tool for postmarket surveillance of medical implants and therapies. Recent clinical disasters in plastic surgery with PIP implants and in hip surgery with DePuy metal-onmetal implants have impressively demonstrated the necessity of registration in implantology [46, 47]. Systematic analyses of registry data can result in early warning mechanisms much earlier than single surgeons or hospitals will raise their suspect.

The evaluation of registry data requires solid statistical methods. Comparative registry-based

studies need to be at least of multivariate-adjusted character to account for observed confounding factors [44, 48]. The analysis of a propensity score-matched sample can even mimic that of an RCT [49]. Furthermore, the representativeness of the selected study sample within Spine Tango should always be proven where appropriate. However, the representativeness of the documented and the treated patient populations in the participating clinics cannot be assessed, similarly as the representativeness of the international spine surgery can probably not be claimed yet.

One should remember that the primary goal of documentation in the registry is the user's own quality assurance and quality improvement. The data remains anonymized and belongs to the physician. An honest documentation is in the interest of the users themselves. Scientific data monitoring is regularly performed in Spine Tango to assess clinical indicators for honest documentation, such as proportions of complications per participating hospital, treatment type, diagnosis, etc. A good indicator of honest documentation is the proportion of dura lesions. The proportion of dura lesions in Spine Tango appears to be well comparable or even slightly higher than that in the Swedish Spine Registry [44], which is a national spine registry with a 90 % coverage rate [26].

#### 54.7 Financing of a Registry

Having a long experience in medical registries, the Institute for Evaluative Research in Medicine was and is hosting registries with various financing models. A certain evolution in the financing models of registries was observed over the last decades. In the oldest financial model of the IDES registry, documentation of total hip and knee arthroplasty was paid by the initiator, Prof. M. E. Müller. The money was coming from earnings of hip prostheses, which were supposed to be monitored in the registry. The registry collected 48,000 primary total hip arthroplasties, 12,000 revision arthroplasties, and 77,000 follow-up evaluations [50]. This type of registry was voluntary, but direct financial support of the documenting sites was attractive for participating

centers, and it caused such high documentation numbers at a time when quality assurance and postmarket surveillance were not hot topics yet.

A more complex model was used within the SWISSspine registry [22, 42, 51]. After the moratorium on total disk arthroplasty and kyphoplasty in Switzerland, the SWISSspine registry was launched according to the principles of coverage with evidence development. The registry is administered under the auspices of the Swiss Society for Spinal Surgery, but it was mandated by the Swiss Federal Office of Public Health for evaluating these specific spinal therapies. The implant producers operating in Switzerland became the financial supporters of the registry as this was the only way to bring their products to market and have them reimbursed by the basic health insurance. The sales figures and documentation numbers are compared by an independent trustee officer and reported to the Federal Office of Public Health. Also, an annual registry report is provided. Based on the observed results, balloon kyphoplasty was already released from the evaluation in 2009 [22, 52].

Meanwhile the so-called National Association for Quality Control in hospitals and clinics in Switzerland (ANQ) has been founded in Switzerland, whose main task is national populationbased observation of selected quality indicators. The association has its own budget out of which, among others, the large national registry for surgical site infections Swissnoso [53] is paid. Although hip and knee prosthesis do meanwhile belong to the list of quality assessment programs of ANQ, a new financial model had to be created due to the limited ANQ budget. According to this model, every participating hospital is charged 20 CHF (+ VAT) for each knee or hip implantation [54]. The money is used for organizing documentation, data collection, on-site and first level support, reporting, administration of the registry, and communication. The planned inclusion of other implant-based surgeries will reduce this charge. This model is beneficial from the perspective of patients who are indirectly participating in quality assurance and quality improvement and from the perspective of hospitals who are assuring their quality assessment as it is anchored in the Swiss healthcare law.

The Spine Tango registry with its abovementioned goals is running under the auspices of Eurospine, the Spine Society of Europe. The society and its members are financing the registry. This fact underlines the feasibility of large international registries embedded within a professional society. The type of registry remains voluntary. The mature user tools in the registry, such as annual and benchmarking reports, online statistics with benchmarking, data download, follow-up calendar, etc., bring meaningful benefits to the participating colleagues in their daily practice.

### 54.8 Outlook

The benefits of the collected data in terms of evidence generation range from epidemiological and descriptive analyses [33], subgroup analyses [48], and matched case-control studies to the development of clinical outcome prediction models [55]. A first model on prediction of complications after spine surgery was recently published by a research group from Seattle based on data from two hospitals [55]. Development of outcome prediction models represents an important step in utilization of the collected evidence and making it publicly available. Spine surgeons need to be able to make evidence-based predictions regarding the outcome of their surgical procedures, based on reliable prognostic information, and to share and discuss it with their patients. The risks and benefits of different treatment modalities must be adequately communicated to the patient. This is particularly important in view of recent developments toward shared decisionmaking, where physicians and patients both actively participate in selecting the therapeutic intervention. The use of validated risk prediction models to inform decisions for surgery will potentially improve treatment outcomes and save healthcare costs by providing information that can be used to avoid unnecessary treatments and to improve the selection of the most beneficial treatment for individual patients. These models will also provide societal benefits by providing the best available data about surgical outcomes in

an accessible format that can be individualized according to patient characteristics to support more detailed patient informed consent. The development of prediction models based on data from Spine Tango has started and should provide results soon.

The Spine Tango benchmarking project was recently initiated by the registry research group. The project aims at analyzing the most frequently encountered treatments for the most common degenerative diseases of the spine, in order to find out if and to what extent surgical spinal interventions "do" [20] work in day-to-day clinical settings. Disk herniation and spinal stenosis, making up about two thirds of all degenerative diseases recorded with Spine Tango, are the first pathologies the benchmarking project will assess.

#### Conclusions

Spine surgery is one of the fastest growing medical specialties with high epidemiological and financial societal impacts. There are compelling needs for evidence generation and quality assurance. Those aims require nothing other than a standardized documentation as one common language. An important scientific and methodological approach under such conditions is a registry, whereas objective data from care providers should be reasonably accompanied by subjective data from the patients. Not only high-quality data collection but also pragmatic methodological evaluation approaches are required for generation of sound evidence.

The European Spine Tango endeavor with over 10 years of technical and content development represents a unique international registry concept and data pool with large potential for evidence generation in spine surgery. The strengths lie in the large sample size, internationality, "common language" for conservative and surgical procedures, mature technical documentation solutions and tools, as well as a wide network of physicians. These aspects limit the potential of selection bias to a considerable extent. An introduction of accredited participants with (nearly) complete documentation coverage and follow-ups should help to
further increase the quality of the collected data and at the same time help to measure the influence of potential selection bias in nonconsecutively documenting hospitals. The fact that this large registry is financed by member fees of a professional society underlines the feasibility of such registries. Besides the dissemination of the collected evidence in scientific publication and conference proceedings, one of the next major steps should be the development of valid outcome prediction models. A joint endeavor in a "common language" is the current result of the registry and is the best prerequisite to assure its successful future.

# References

- Müller ME, Allgöwer M, Willenegger H. Die Gemeinschaftserhebung der Arbeitsgemeinschaft für Osteosynthesefragen. Arch klin Chir. 1963;304:808–17.
- Sower S, Fair F. There is more to quality than continuous improvement: listening to Plato. Qual Manage J. 2005;12:8–20.
- ASQ. American Society for Quality, Quality glossary. Web: http://asq.org/glossary/q.html. Accessed July 24, 2015.
- Garvin D. What does product quality really mean? Sloan Manage Rev. 1984;26:25–43.
- Varkey P, Reller MK, Resar RK. Basics of quality improvement in health care. Mayo Clin Proc. 2007;82:735–9.
- Campbell SM, Braspenning J, Hutchinson A, et al. Research methods used in developing and applying quality indicators in primary care. BMJ. 2003; 326:816–9.
- Wensing M, Elwyn G. Methods for incorporating patients' views in health care. BMJ. 2003;326:877–9.
- Chassin MR. Is health care ready for Six Sigma quality? Milbank Q. 1998;76:565–91. 10.
- Impellizzeri FM, Bizzini M, Leunig M, et al. Money matters: exploiting the data from outcomes research for quality improvement initiatives. Eur Spine J. 2009;18 Suppl 3:348–59.
- Campbell DT, Stanley JC. Experimental and quasiexperimental designs for research. Chicago: Rand McNally; 1963.
- Sanson-Fisher RW, Bonevski B, Green LW, et al. Limitations of the randomized controlled trial in evaluating population-based health interventions. Am J Prev Med. 2007;33:155–61.
- Concato J, Shah N, Horwitz RI. Randomized, controlled trials, observational studies, and the hierarchy of research designs. New Engl J Med. 2000;342:1887–92.

- Benson K, Hartz AJ. A comparison of observational studies and randomized, controlled trials. New Engl J Med. 2000;342:1878–86.
- Green LW, Glasgow RE. Evaluating the relevance, generalization, and applicability of research: issues in external validation and translation methodology. Eval Health Prof. 2006;29:126–53.
- Rosner A. Fables or foibles: inherent problems with RCTs. J Manipulative Physiol Ther. 2003;26:460–7.
- Rosner AL. Evidence-based medicine: revisiting the pyramid of priorities. J Bodyw Mov Ther. 2012; 16:42–9.
- 17. Freiman JA, Chalmers TC, Smith Jr H, et al. The importance of beta, the type II error and sample size in the design and interpretation of the randomized control trial. Survey of 71 "negative" trials. New Engl J Med. 1978;299:690–4.
- Rothwell PM. External validity of randomised controlled trials: "to whom do the results of this trial apply?". Lancet. 2005;365:82–93.
- Melloh M, Roder C, Staub LP, et al. Randomizedcontrolled trials for surgical implants: are registries an alternative? Orthopedics. 2011;34:161.
- Jarvinen TL, Sievanen H, Kannus P, et al. The true cost of pharmacological disease prevention. BMJ. 2011;342:d2175.
- Bhandari M, Richards RR, Sprague S, et al. The quality of reporting of randomized trials in the Journal of Bone and Joint Surgery from 1988 through 2000. JBJS Am. 2002;84-A:388–96.
- Hubschle L, Borgstrom F, Olafsson G, et al. Real-life results of balloon kyphoplasty for vertebral compression fractures from the SWISSspine registry. Spine J. 2013;14(9):2063–77.
- Grob D, Porchet F, Kleinstuck FS, et al. A comparison of outcomes of cervical disc arthroplasty and fusion in everyday clinical practice: surgical and methodological aspects. Eur Spine J. 2010;19:297–306.
- Black N. Why we need observational studies to evaluate the effectiveness of health care. BMJ. 1996; 312:1215–8.
- Stromqvist B, Jonsson B. Computerized follow-up after surgery for degenerative lumbar spine diseases. Acta Orthop Scand Suppl. 1993;251:138–42.
- Stromqvist B, Fritzell P, Hagg O, et al. Swespine: the Swedish spine register: the 2012 report. Eur Spine J. 2013;22:953–74.
- Stromqvist F, Jonsson B, Stromqvist B, et al. Dural lesions in decompression for lumbar spinal stenosis: incidence, risk factors and effect on outcome. Eur Spine J. 2012;21:825–8.
- Stromqvist F, Jonsson B, Stromqvist B, et al. Dural lesions in lumbar disc herniation surgery: incidence, risk factors, and outcome. Eur Spine J. 2010;19:439–42.
- Deyo RA, Battie M, Beurskens AJ, et al. Outcome measures for low back pain research. A proposal for standardized use. Spine. 1998;23:2003–13.
- Mannion AF, Elfering A, Staerkle R, et al. Outcome assessment in low back pain: how low can you go? Eur Spine J. 2005;14:1014–26.

- Roder C, Errico TJ, Spivak JM, et al. Hospital for joint diseases participates in international spine registry Spine Tango after successful pilot study. Bull NYU Hosp Jt Dis. 2012;70:254–8.
- Roder C, El-Kerdi A, Grob D, et al. A European spine registry. Eur Spine J. 2002;11:303–7.
- Melloh M, Staub L, Aghayev E, et al. The international spine registry SPINE TANGO: status quo and first results. Eur Spine J. 2008;17:1201–9.
- Roder C, Staub L, Dietrich D, et al. Benchmarking with Spine Tango: potentials and pitfalls. Eur Spine J. 2009;18 Suppl 3:305–11.
- Zweig T, Mannion AF, Grob D, et al. How to Tango: a manual for implementing Spine Tango. Eur Spine J. 2009;18 Suppl 3:312–20.
- Eurospine. Spine Tango. http://www.eurospine.org/ spine-tango.htm. Accessed Feb 2014.
- 37. Kessler JT, Melloh M, Zweig T, et al. Development of a documentation instrument for the conservative treatment of spinal disorders in the International Spine Registry, Spine Tango. Eur Spine J. 2011;20:369–79.
- Aebi M, Grob D. SSE Spine Tango: a European Spine Registry promoted by the Spine Society of Europe (SSE). Eur Spine J. 2004;13:661–2.
- 39. Burkhardt JK, Mannion AF, Marbacher S, et al. A comparative effectiveness study of patient-rated and radiographic outcome after 2 types of decompression with fusion for spondylotic myelopathy: anterior cervical discectomy versus corpectomy. Neurosurg Focus. 2013;35:E4.
- 40. Mannion AF, Fekete TF, O'Riordan D, et al. The assessment of complications after spine surgery: time for a paradigm shift? Spine J. 2013;13:615–24.
- 41. Kleinstueck FS, Fekete T, Jeszenszky D, et al. The outcome of decompression surgery for lumbar herniated disc is influenced by the level of concomitant preoperative low back pain. Eur Spine J. 2011;20: 1166–73.
- Aghayev E, Henning J, Munting E, et al. Comparative effectiveness research across two spine registries. Eur Spine J. 2012;21:1640–7.
- 43. Mannion AF, Boneschi M, Teli M, et al. Reliability and validity of the cross-culturally adapted Italian version of the Core Outcome Measures Index. Eur Spine J. 2012;21 Suppl 6:S737–49.

- 44. Munting E, Röder C, Sobottke R, et al. Patient outcomes after laminotomy, hemilaminectomy, laminectomy and laminectomy with instrumented fusion for spinal canal stenosis: a propensity score based study from the Spine Tango registry. Eur Spine J. 2015; 24(2):358–68.
- 45. Herberts P, Malchau H. Long-term registration has improved the quality of hip replacement: a review of the Swedish THR register comparing 160,000 cases. Acta Orthop Scand. 2000;71:111–21.
- Wikipedia. Poly implant prosthesis. 2014. http:// en.wikipedia.org/wiki/Poly\_Implant\_Proth%C3%A8se
- 47. Wikipedia. DePuy hip recall. 2014. http:// en.wikipedia.org/wiki/2010\_DePuy\_Hip\_Recall
- 48. Sobottke R, Aghayev E, Roder C, et al. Predictors of surgical, general and follow-up complications in lumbar spinal stenosis relative to patient age as emerged from the Spine Tango Registry. Eur Spine J. 2012;21: 411–7.
- Austin PC. An introduction to propensity score methods for reducing the effects of confounding in observational studies. Multivariate Behav Res. 2011;46: 399–424.
- 50. Aghayev E, Teuscher R, Neukamp M, et al. The course of radiographic loosening, pain and functional outcome around the first revision of a total hip arthroplasty. BMC Musculoskelet Disord. 2013;14:167.
- Roder C, Blozik E, Müller M, et al. SWISSspine: an outcome and quality registry of orthopaedic implants as a condition for reimbursement by basic health insurance. J Manag Mark Healthc. 2008;2:92–101.
- 52. Diel P, Reuss W, Aghayev E, et al. SWISSspine-a nationwide health technology assessment registry for balloon kyphoplasty: methodology and first results. Spine J. 2010;10:961–71.
- Swissnoso. Registry for surgical site infections in Switzerland. 2014. http://www.swissnoso.ch/. Accessed Feb 2014.
- SIRIS. The Swiss implant registry. 2014. http://sirisimplant.ch/. Accessed Feb 2014.
- 55. Lee MJ, Cizik AM, Hamilton D, et al. Predicting medical complications after spine surgery: a validated model using a prospective surgical registry. Spine S. 2014;14:291–9.

# Lessons from a Life: The Journey of Spinal Neurosurgery in the United States

55

Brian P. Curry, M. Yashar S. Kalani, and Volker K.H. Sonntag

## 55.1 Introduction

Given the rapid, continued development of increasingly sophisticated devices for internal spinal fixation, it is easy to overlook the fact that the history of spinal instrumentation spans only a short period. Although there is evidence that spinal disorders were recognized as a cause of significant morbidity from early antiquity, surgical attempts to address these disorders were extraordinarily rare [1]. Despite the great strides that occurred during the medieval and Renaissance periods in characterizing the anatomy and mechanical characteristics of the spine, surgical approaches to the spine were long considered daring, reckless endeavors, associated with unacceptably high mortality and morbidity, and ultimately doomed to failure. As a result, treatment for spinal disorders historically relied upon a variety of external braces and devices that were uncomfortable and unsuccessful in equal measure and avoided internal correction almost entirely.

M.Y.S. Kalani, MD, PhD • V.K.H. Sonntag, MD (⊠) Division of Neurological Surgery, Barrow Neurological Institute, St. Joseph's Hospital and Medical Center, 350 W. Thomas Road, Phoenix, AZ 85013, USA e-mail: Neuropub@dignityhealth.org

With the development and dissemination of anesthetic and antiseptic techniques in the nineteenth century, the practice of surgery transitioned from a frenzied, messy, and excruciatingly painful affair associated with significant mortality and postsurgical infection to a more careful and considered approach [2, 3]. No longer rushed to take decisive action with incomplete information, surgeons were free to innovate, and it was in this fertile bed that spinal instrumentation took root as a method to correct spinal instability and deformity. This chapter tells the story of spinal instrumentation from its humble beginnings in the late 1800s to the present day and includes a discussion of the crucial advances in construct design, materials, and techniques that have permitted the development of the wide array of technology available to today's spinal surgeon.

Unfortunately, we cannot hope to include mention of every incremental advance in instrumentation. This chapter is only a brief glimpse at the progression of spinal instrumentation since its birth, and thus we strive to describe those advances that substantially contributed to that progression. Likewise, there is unfortunately limited space in which to describe each innovation in detail. Nevertheless, the history of spinal instrumentation is a fascinating story that provides important context for understanding instrumentation technology today.

B.P. Curry, BS, MA, MD

Walter Reed Medical Center, Washington, DC, USA

<sup>©</sup> Springer-Verlag Berlin Heidelberg 2016

J.L. Pinheiro-Franco et al. (eds.), *Advanced Concepts in Lumbar Degenerative Disk Disease*, DOI 10.1007/978-3-662-47756-4\_55

# 55.2 The Birth of Spinal Instrumentation

The first recorded successful attempt at spinal instrumentation was performed in Ottawa, Kansas, in 1888 by W. F. Wilkins. Wilkins performed surgery on a newborn infant who had sustained a vertebral dislocation with traumatic rupture of a lumbar intervertebral disk. After reducing the disk herniation, Wilkins stabilized the spine by wiring the pedicles together with carbolized silver wire, and the infant recovered uneventfully [4].

А similar technique was subsequently employed by Berthold Ernest Hadra in Texas in 1891, though he was not aware of Wilkins' procedure at the time. Hadra was a Prussian-born surgeon who had immigrated to the United States after serving as an assistant surgeon in the Prussian army during the Austro-Prussian war [5]. He used silver wire to join the spinous processes of the sixth and seventh cervical vertebrae in a 30-year-old patient with an unstable cervical fracture-dislocation and neurological deterioration. The procedure was initially a success, although the surgery was repeated 3 weeks later when the wires loosened. Undaunted, Hadra posited that his interspinous wiring method was a success and could be adapted for use in the treatment of "any deviation of a vertebra" [6].

As it happened, the closing of the nineteenth century found physicians facing widespread outbreaks of tuberculosis worldwide. Progressive spinal deformity and neurological compromise as a result of extrapulmonary tuberculosis, first described by Percivall Pott in 1779, had long been treated with externally applied braces and devices, but remained a debilitating malady that surgeons of the day struggled to address. Pott himself had advocated surgical drainage of paraspinal tuberculosis abscesses as a solution, although this procedure did nothing to correct deformity and was associated with the development of secondary infections and draining sinus tracts [7].

Inspired by his relative success with interspinous wiring, Hadra suggested that his procedure might have applicability in the treatment of Pott's disease, although he had not attempted this himself [8]. Two years later, French neurosurgeon Antoine Chipault performed the first internal fixation for this very purpose. In 1898, Lovett published a report of his experience in five patients with tuberculosis spondylitis, describing a procedure in which he denuded adjacent spinous processes and then wired them together with silver wire [9]. However, despite achieving stability initially, this process resulted in short-lived success and patients frequently required a second surgery [10].

Inspired by the work of Hadra and Chipault, Fritz Lange of Munich was himself beginning to consider the problem of internal splinting for tuberculous spines. He envisioned an "artificial spinal column of steel," composed of bilateral tin-plated steel rods wired to the spinous processes. After his first attempt in 1902 resulted in postoperative complications from the sharp ends of the silver wire used to secure the rods, he set to work trying to characterize the optimum materials for use in his procedure. After several years of animal experiments, he settled on 5-mm-thick, 10-cm-long, tin-plated steel rods secured to the spinous processes with silk thread [11].

# 55.3 Electrolysis and the Development of New Materials

Ultimately, Lange abandoned metal rods for a celluloid material because of problems with corrosion. By the turn of the twentieth century, he and others recognized that a major limitation to the use of metals for internal spinal fixation was the tendency for metals to corrode when placed in the body. In addition, it was observed that such metals produced pathological changes in local tissues that impaired healing and subsequent stability. While many hypotheses were advanced to explain these changes, it was not until the 1930s that the true culprit was widely appreciated: electrolysis [12].

It was well known that materials such as silver, aluminum, nickel, and stainless steel were not only weakened by electrolysis but also produced local bone erosion that significantly hampered attempts to achieve fixation. In 1936, Venable and Stuck began investigating the causes for the breakdown of metal implants, and in 1938 Venable published a definitive description of the process of electrolysis as it related to fracture fixation [13]. Venable and Stuck experimented with several metals over a 3-year period and ultimately recommended Vitallium, a cobaltchromium-molybdenum alloy developed in 1932 that withstood the corrosive effects of electrolysis and produced no pathological changes in the bone [14].

Before that time, there had been little interest in discovering the optimal attributes of a material for implantation, despite the fact that the first use of a metal implant had occurred more than a century earlier. Once it became clear that the choice of materials had profound implications for the strength and durability of the implant, research began in earnest to understand the ideal chemical properties of implanted materials and to develop stronger materials that could resist corrosion. Stainless steel alloys were quickly developed that were better able to resist electrolysis, yet maintained much of the mechanical properties that made steel preferable for fracture stabilization and spinal instrumentation.

In 1951, Leventhal introduced the idea of using titanium in orthopedic procedures, noting its impressive strength and lack of pathological tissue changes [15]. Moreover, the mechanical properties of titanium and its alloys more closely approximated those of bone, making it an attractive candidate material for implantation in bone [16]. Surgeons also found that titanium produced less artifact on radiographic imaging than stainless steel or Vitallium [17]. Since that time, titanium alloys gradually became the primary material used in spinal fixation. The later widespread use of magnetic resonance (MR) spinal imaging solidified the place of titanium, which is compatible with MR imaging, in spinal instrumentation. Current researchers are investigating the role of ceramics and synthetic polymers in creating strong, lightweight constructs for orthopedic procedures [18]. One such polymer in wide use today is polyetheretherketone, the chemical and mechanical properties of which make it especially attractive in spinal and orthopedic surgery.

#### 55.4 Spinal Fusion

In 1911, Fred Albee and Robert Hibbs, both orthopedic surgeons in New York, independently struck upon the importance of arthrodesis in halting the progressive deformity of Pott's disease when they each developed procedures that used autologous bone grafts to achieve bony fusion of the spinous processes, thereby obviating the need for wires. In both procedures, this process corrected deformity and reinforced the dorsal spinal elements. The Albee procedure used tibial grafts as struts between the spinous processes of adjacent vertebrae [19], while Hibbs used bone from the spinous processes themselves to bridge vertebrae [20, 21]. The theoretical advantages of these procedures were immediately obvious, and their potential for halting spinal deformity from other causes was quickly recognized, such that these techniques were widely adopted soon after their invention. In 1914, Herbert Galloway became the first surgeon to use fusion techniques to correct paralytic scoliosis [22]. Nevertheless, these initial efforts at fusion resulted in unacceptably high rates of pseudarthrosis, and as a result, these procedures were often modified by others to include silver wiring to minimize graft movement and maximize arthrodesis [23].

The 1930s saw the development of more elegant techniques for spinal fusion, including interbody fusion of the lumbar spine. In 1933, Burns published his description of anterior lumbar interbody fusion for an L5–S1 spondylolisthesis [24]. Beginning in 1939, Cloward would go on to develop the posterior lumbar interbody fusion approach [25]. Such procedures were developed and refined over the course of the next 30 years. While all were promising methods for halting the progression of spinal deformity, all were also initially associated with fairly high rates of pseudarthrosis.

# 55.5 Paul Harrington and the Modern Era of Spinal Stabilization

Importantly, the foregoing sections illustrate the point that spinal instrumentation and fusion were developed more or less independently. These earliest methods for addressing deformities of the spine were limited in two ways: (1) They generally halted the progression of deformity, rather than correcting it, and (2) they failed to marry the techniques of spinal fusion, which provided the durable stabilization sought by these early pioneers, with spinal instrumentation, which helped to maintain the alignment and rigidity necessary to promote arthrodesis. It was not until the 1960s that the interdependence of these two parallel advances would become fully and widely recognized, and the modern era of spinal stabilization would begin. In the immediate aftermath of World War II, American physicians faced a recurrence of widespread poliomyelitis epidemics. In the late 1940s and early 1950s, polio crippled approximately 35,000 individuals each year, with the development of an effective vaccine still several years away. Aside from the immediate complications of polio, many sufferers subsequently developed progressive and debilitating thoracic scoliosis, often with attendant cardiorespiratory compromise that made treatment with braces and casts impossible. The treatment of these patients occupied spinal surgeons of the day.

One such individual was Paul Randall Harrington, an American orthopedic surgeon in Houston, Texas. After finishing his residency in orthopedic surgery in 1942, he joined the US Army, serving as the chief of orthopedic surgery for the 77th Evacuation Hospital. At the war's end in 1945, Harrington settled in Houston, where he worked at the Jefferson Davis City-County Hospital. City-County saw many polio patients and by 1953 had become the second Respiratory Center in the nation.

Recognizing that a large number of the polio patients seen at City-County often developed scoliosis, Harrington began in 1947 to develop a method to stop the progression of these patients' deformity and improve their cardiopulmonary function. Harrington's first attempts used screws to fix the facet joints in a corrected position, a technique introduced by Toumey [26], and King [27], as a means of providing rigid internal fixation to aid in arthrodesis. While Harrington's facet-screw procedure appeared promising at first, ultimately this method proved a failure, and he was forced to seek an alternative means for correcting the deformity [28]. The next iteration of Harrington's approach involved hooks connected to a threaded rod, by which he was able to apply distracting corrective forces. The hooks were placed at the spinous processes of the vertebrae at the superior and inferior extremes of a curve, with the rod spanning the concavity of the scoliotic curve.

Harrington continued to refine his surgical technique and instrumentation over the next 15 years, making the early instrumentation by hand. When his early handmade constructs failed, he partnered with engineers to develop sturdier instruments that could withstand the repetitive stress to which they were subjected. Ultimately, Harrington's work led to the creation of a system of stainless steel rods and distraction hooks. As he continued to develop an instrumentation system that could address the spinal deformity from polio, Harrington also began to apply it to cases of idiopathic scoliosis [29].

Despite the initially satisfactory correction of the deformity, Harrington was cognizant of the fact that hardware failure was inevitable, often as early as 6 months postprocedure. To achieve lasting deformity correction, he recognized that his instrumentation would have to incorporate a fusion within its extent. Rather than using hooks and rods as a dynamic means of establishing correction, he determined that instrumentation should serve as a means of permitting arthrodesis [28].

By demonstrating that instrumentation was merely a temporary means to the end of spinal fusion, rather than an end in itself, Harrington laid the foundation for future developments in spinal instrumentation. His experience revealed that instrumentation failure is inevitable, but that permitting arthrodesis can achieve long-term stability. By acknowledging the "race between instrumentation failure and the acquisition of spinal fusion," Harrington united the parallel advances in internal fixation and spinal fusion, thereby ushering in the modern era of spinal stabilization [28].

Almost immediately, Harrington's rod system became the state-of-the-art spinal instrumentation

through the 1970s. Indeed, use of Harrington instrumentation continues to the present day. The utility of the procedure was easily recognized, and the indications were soon extended beyond polio and idiopathic scoliosis to include trauma [30, 31], degenerative processes, and malignancy [32, 33]. With its widespread adoption, however, came the recognition of its limitations. Most notably, the effectiveness of distraction rods in correcting coronal curvature of the spine came at the expense of the natural sagittal curvature of the spine, resulting in loss of lumbar lordosis, the socalled flatback syndrome [34, 35]. Additionally, the procedure required that patients spend several months in plaster braces postoperatively. Lastly, repeated stresses occasionally resulted in hook dislodgement or rod breakage [36].

# 55.6 Segmental Spinal Fixation and Development of the Pedicle Screw

If Harrington provided the bedrock on which the modern era of spinal instrumentation was built, much of the foundation was laid by Eduardo Luque in the mid-1970s. Luque, an orthopedic surgeon in Mexico City, sought to address some of the limitations of the Harrington rod system. His work ultimately led to the introduction of segmental spinal fixation. Whereas Harrington used straight rods with hooks at either end of the deformity, Luque used a contoured steel rod, attached to the vertebrae at several points with sublaminar wiring. By doing so, Luque was able to more evenly distribute the forces across the construct, which not only increased the rigidity of the construct and reduced the potential for hardware failure but also offered the potential for greater correction and reduced the need for postoperative bracing. Most importantly, however, segmental spinal fixation permitted deformity correction without sacrificing sagittal curvature [37].

While spinal surgeons recognized the promise of Luque's segmental fixation technique, their enthusiasm was tempered by concern for potential neurological injury caused by the sublaminar wiring used to secure the rods. Indeed, reports of complications such as epidural hematoma and direct trauma caused by passing the sublaminar wire left surgeons searching for alternative means of achieving segmental fixation [38, 39]. To address these concerns, Drummond et al. published a description in 1984 of a technique using a button-wire implant threaded through a hole drilled at the base of the spinous process [40]. Although the construct was not as strong as Luque's sublaminar wiring construct, it had the advantage of reducing the potential for neurological injury, thereby making it a more attractive alternative to spinal surgeons of the time.

A major advance during the refinement of segmental fixation techniques was the introduction of pedicle screws as a means to secure the constructs to the vertebrae. As mentioned previously, the first accounts of the use of screws for internal fixation of the spine as an adjunct to fusion were published in the early 1940s by Toumey [26] and King [27]. Both used short bone screws that traversed the facet joints of the segment to be fused bilaterally. King reported a pseudarthrosis rate of only about 10 %, although Thompson and Ralston subsequently reported much higher rates of pseudarthrosis using a similar technique [41]. Thus, the technique did not gain widespread acceptance, and Bosworth suggested in 1957 that the benefits of screw fixation did not justify the difficulty in placing the screws [42].

H. H. Boucher, a Canadian spinal surgeon, believed that the concept of screw fixation was nevertheless sound. He developed and implemented a method of screw fixation using longer screws to traverse the facet joint and enter the pedicle and vertebral body below. In 1959, he published a paper describing his experience with this technique over a 12-year period for singleand multiple-level fusions and to treat spondylolisthesis [43]. Although he reported excellent results, he conceded that screws were a temporary solution and were likely to loosen over time, but that well-placed screws could offer superior fixation to minimize pseudarthrosis.

In Paris, Raymond Roy-Camille, under the guidance of Robert Judet, used pedicle screws as fixation points for a multiple-segment posterior plate system. Although he performed his first surgery in 1963, he did not publish the results until 1970 [44]. His system of plates and pedicle screws was applied to a variety of spinal disorders, and its success confirmed the superiority of pedicle screw segmental fixation [45]. He is largely credited with pioneering the use of pedicle screws in segmental instrumentation.

The advantages of pedicle screws over facet screws, hooks, and wire fixation are manifold. Biomechanically, pedicle screws gain better purchase and are capable of withstanding a wider range of forces across multiple vectors [46, 47]. Additionally, unlike hooks and wire fixation, which require posterior vertebral structures, they may be placed even after a laminectomy, a benefit that permitted the extension of instrumentation techniques to address various degenerative spine conditions.

In the United States, the first surgeon to popularize the use of pedicle screws for segmental fixation was Arthur Steffee, an orthopedic surgeon in Cleveland, Ohio. In the early 1980s, Steffee et al. noted that the pedicle acted as a "force nucleus" for the vertebra and was the conduit by which the forces acting upon the posterior vertebral elements could be transmitted anteriorly [48]. Accordingly, he posited that the pedicle was an ideal structure for placing a screw in a segmental fixation construct. Steffee placed headless screws in the pedicles and modified standard long bone fracture plates with slots that could slide onto the screws. He secured the plates in place with nuts, thereby pioneering "variable screw placement."

In Paris during roughly the same time, Cotrel and Dubousset were developing a segmental fixation system that used rods, as opposed to the plates that their contemporaries were using. Initially, their system used hooks at multiple segments, secured with bolts [49], and while this offered satisfactory correction, the bolt-locked hook system proved difficult to remove. Eventually, they adopted monoaxial pedicle screws as fixation points for their rods [50], allowing for easier adjustment or removal of the construct. Thus, Cotrel and Dubousset married the advantages of segmental rod fixation of the spine with those of pedicle screws. While plate constructs were preferred for their strength in spinal instrumentation, rods offered much greater flexibility and a lower profile, thus allowing surgeons to more precisely correct deformities in three dimensions and leaving more room for bone grafting.

B.P. Curry et al.

Since that time, further development of spinal instrumentation has largely centered around refinements of segmental rod fixation using pedicle screws as fixation points. Today's surgeon has a myriad of choices of instrumentation systems from multiple manufacturers. Advances in screw design, from the monoaxial screw used by Cotrel and Dubousset to uniaxial screws and the polyaxial screws in greater use today, have greatly aided spinal surgeons in creating even more complex constructs. Uniaxial screws, as the name implies, are mobile in one plane at the junction of the head and the shaft, making them ideal for derotation of the spine in the axial plane. Polyaxial screws have freely mobile heads, allowing surgeons to align the heads to the rod, providing greater ease of integrating fixation points into the construct. Additionally, minimally invasive techniques and instrumentation systems have been developed that allow multilevel instrumentation through small paraspinal incisions.

## 55.7 Dynamic Spinal Stabilization

Since the advent of the pedicle screw, spinal instrumentation with fusion has been increasingly used to treat degenerative spine conditions, based on the hypothesis that eliminating spinal motion across diseased segments could eliminate spinal pain and neurological symptoms. As use of this procedure has increased in frequency, however, surgeons have come to appreciate not only that the success of fusion does not guarantee clinical improvement but also that immobilizing spinal segments has an effect on the spinal levels adjacent to the fusion [51]. By altering the distribution of forces in the spinal column and increasing the relative mobility of the adjacent segments,

rigid spinal fusion has the untoward effect of subjecting adjacent spinal segments to abnormal mechanical conditions, causing pain and deformity, and often necessitating subsequent surgeries [52–55].

While rigid spinal fusion remains the surgical standard for addressing degenerative spinal conditions, currently there is widespread development of so-called dynamic stabilization systems that seek to redistribute the load away from diseased spinal levels while maintaining mobility at those levels. With these systems, adjacent levels are not subjected to increased relative motion, which reduces pathological changes and pain in the segments. At present, there are a number of approaches to dynamic stabilization, including anterior disk prostheses, such as the Prestige and Bryan (Medtronic plc, Dublin, Ireland) cervical disk prostheses and the Maverick (Medtronic plc) lumbar prosthesis, as well as posterior constructs utilizing flexible connections between pedicle screws, such as the Dynesys (Zimmer Spine, Zimmer Biomet Holdings, Inc., Warsaw, IN), the AccuFlex (Globus Medical Inc., Audubon, PA), and the CD Horizon Agile (Medtronic plc). It is not clear what role these systems will play in the future; currently, insufficient data exist to determine whether any or all are superior to spinal fusion in treating degenerative spine disease and reducing complications at adjacent segments.

Interestingly, the first use of dynamic stabilization of any sort occurred in the 1950s, when Paul Harmon began to perform disk arthroplasties, replacing diseased disks with spheres made of Vitallium [56]. These spheres, later called "Fernström balls" after Ulf Fernström, who used stainless steel spheres in the 1960s [57], did not gain widespread acceptance at that time, although Fernström and later McKenzie [56] reported excellent results. The technique was not commercialized further at that time, although the use of spheres for disk arthroplasty has reemerged in the form of Satellite Nucleus Replacement spheres (Medtronic plc), which are available in both cobalt chrome and polyetheretherketone.

## 55.8 Cervical Spine Instrumentation

Despite the fact that the history of spinal instrumentation truly begins with techniques for wiring cervical vertebrae, most of this chapter has thus far focused on the development of spinal instrumentation as it pertains to the thoracolumbar spine. It is, however, worth briefly mentioning some of the major advances in cervical instrumentation.

With regard to posterior cervical spine instrumentation, in 1939, Gallie developed a technique for fusion of C1-C2 for cases of trauma whereby an Albee or Hibbs' graft was secured and compressed between the C1 and C2 posterior arches by sublaminar wiring [58]. This technique, while associated with fairly high fusion rates, nevertheless suffered from the same potential for neurological injury that limited later techniques using sublaminar wiring, as mentioned above. In addition, the wiring did not fully eliminate rotational motion between C1 and C2, occasionally leading to a failure of bony fusion [59]. Brooks and Jenkins modified this technique in 1978 using a pair of ilium grafts wedged between the posterior arches of C1 and C2, again using sublaminar wire [60]. This technique resulted in better resistance to rotational motion, but passing wire under the laminae of multiple levels was associated with significant risk for neurological injury.

While the Gallie method was largely effective, sublaminar wiring required considerable skill. In 1975, Tucker reported the use of the Halifax clamp, a construct that used sublaminar hooks connected by a screw to compress the posterior elements [61]. The Halifax clamp could be rapidly applied and was comparable to the Gallie technique, although its bulk made it somewhat unwieldy and its sublaminar hooks encroached on the spinal canal [62].

Magerl in 1984 introduced transarticular screw placement for internal fixation of C1–C2. He passed sagittally oriented screws from the inferior articular facet of C2 through the facet joint to engage cortical bone of the anterior arch of C1 [63]. This technique avoided the need for passage of wires or clamps and provided immediate bilateral fixation. Combined with a Gallie-type bone graft, this technique resulted in stable fusion of C1–C2.

In 1989 Roy-Camille et al. reported on a novel, lateral, mass screw-and-plate system for stabilizing the cervical spine [64]. This original concept was later modified by An and Coppes [65], Anderson et al. [66], and Jeanneret et al. [67] to minimize the likelihood of injury to the nerve roots or the vertebral artery. The screwand-plate system had limitations, especially in the treatment of complex spinal disorders such as spondylosis. However, the late 1980s and 1990s ushered in the development and implementation of screw-rod constructs [68, 69]. These systems allowed for placement of polyaxial screws that accommodated a variety of anatomical abnormalities, and the connection of the screws via rod systems allowed for immobilization. Further refinements in technique, including introduction of cervical pedicle screws, have improved biomechanical stability of posterior cervical constructs **[70]**.

In 1991, Dickman and Sonntag described a modification of the Gallie technique that again used wire to secure an ilial graft between the posterior arches of C1 and C2. Instead of passing wire under the posterior arches of multiple levels, as Gallie and Brooks had, they used the spinous process of C2 to secure the wire inferiorly, thereby requiring sublaminar passage of wire at only one level [71]. In 2001, Harms and Melcher [72] described a posterior C1–C2 fixation technique, previously reported by Goel and Laheri [73]. The Goel-Harms construct consists of C1 lateral mass screws and C2 pedicle screws. A C2 translaminar modification of the Goel-Harms construct has been reported more recently [74, 75].

For anterior approaches, in 1967, Böhler reported using anterior cervical plating to treat cervical spine trauma, and, since that time, anterior plating has become a popular technique for treating many different conditions, including degenerative conditions, trauma, and infection [76]. Over time, refinements in plate design and screw placement, as well as graft design and construction, have increased the utility and versatility of anterior plating for cervical spine fixation and greatly reduced complications [77].

# 55.9 Fusion Grafts and Bone Morphogenetic Protein

Since 1911, when Albee [19], and Hibbs [20] described their techniques for achieving arthrodesis of the spine, surgeons have explored several different methods for using bone grafts in spinal surgery. Both autografts, obtained by harvesting bone from the patient, and allografts, which are obtained by harvesting bone from another donor, have classically been used to achieve spinal fusion. The most common sites are local to the site of surgery, such as the iliac crest and the rib. Autograft has proven superior in terms of its ability to stimulate bone formation [78] and is generally preferred over allograft. However, harvesting of autograft is invasive, requiring a surgical incision at the donor site, potentially leading to longstanding postoperative pain or numbness, increased risk of infection, and cosmetic defects at the donor site [79–81]. Such potential complications have led to a long-standing search for materials to replace or augment bone grafts.

The better results obtained by grafting autogenous bone are related to three properties of bone healing and growth. Osteogenicity-the capacity to form bone via the presence of osteoblasts-is a property of freshly harvested autologous bone or bone marrow. Osteoconductivity is a material's ability to serve as a scaffold for new bone formation and growth, a property shared not only by autogenous bone but also by allografts and even ceramics and synthetic polymers. Osteoinductivity refers to the presence of factors, the so-called bone morphogenetic proteins (BMPs) in the graft that stimulate osteogenesis in the recipient bed, as well as the differentiation of mesenchymal stem cells into osteoblasts [82]. Therefore, the ideal graft for spinal fusion would include all three characteristics, which are inherent in autologous bone, while minimizing or avoiding the potential complications of an autograft harvest.

To that end, researchers have developed bone graft extenders and replacements that aim to improve fusion results. Bone graft extenders are generally purely osteoconductive, lacking osteoinductivity or osteogenicity. These are typically used to augment autologous bone grafts and may be composed of materials such as calcium phosphate, calcium sulfate, or composites of calcium phosphate and collagen. Demineralized bone matrix, first described by Marshall Urist in 1965 [83], is obtained from a demineralized allograft and contains minute amounts of osteoinductive BMPs. Thus, it may be used to effectively extend and augment autologous bone grafts. Bone graft extenders have demonstrated promise in improving fusion rates while reducing the amount of autograft needed to perform the fusion [84].

Bone graft replacements seek to entirely eliminate the need for harvesting autografts, while improving fusion rates. This is achieved by using highly osteoinductive materials, namely, recombinant human BMPs (rhBMP), approved by the FDA in 2002 for fusion of the anterior lumbar spine. After animal trials revealed promising results using rhBMPs [85-87], human trials of rhBMP-2 and rhBMP-7 commenced in the early 2000s. At present, it is unclear what role rhBMPs will ultimately occupy in surgical treatment for spinal fusion, as studies have yielded inconsistent results [88–91], reports of higher complication rates [92], and recent evidence that the use of rhBMP may be associated with a higher incidence of benign tumors [93].

#### 55.10 Summary

From the earliest use of interspinous wiring in the cervical spine in the late 1800s through the intervening decades, spinal instrumentation has progressed dramatically. Today's surgeons have access to a myriad of options for internal fixation throughout the entire spinal column. As our understanding of the spine has increased and materials and techniques have improved, surgical treatment for a variety of debilitating and deforming spine conditions has continued evolving at an astounding rate.

Against such a backdrop, it is difficult to predict what the future of spinal instrumentation will look like and, furthermore, which of today's new technologies will contribute to shaping that future. Nevertheless, with the spinal instrumentation technology available today, and the promise of even more effective, elegant constructs in the future, this is truly an exciting time in spinal surgery.

#### References

- Goodrich JT. History of spine surgery in the ancient and medieval worlds. Neurosurg Focus. 2004;16(1):1–13.
- Bigelow HJ. Insensibility during surgical operations produced by inhalation. Boston Med Surg J. 1846;35(16):309–17.
- Lister J. On the antiseptic principle in the practice of surgery. Br Med J. 1867;2(351):246–8.
- Wilkins WF. Separation of the vertebrae with protrusion of hernia between the same. St Louis Med Surg J. 1888;54:340–1.
- 5. Eggers GWN. Berthold Ernest Hadra (1842–1903): a biography. Clin Orthop. 1961;21:32–9.
- Hadra BE. The classic: wiring of the vertebrae as a means of immobilization in fracture and Potts' disease. Berthold E. Hadra. Med Times and Register, Vol 22, May 23, 1891. Clin Orthop Relat Res. 1975;112:4–8.
- 7. Pott P. Remarks on that kind of palsy of the lower limbs, which is frequently found to accompany a curvature of the spine, and is supposed to be caused by it. Together with its method of cure. To which are added, observations on the necessity and propriety of amputation, in certain cases and under certain circumstances. London Johnson; 1779.
- Hadra BE. Wiring the spinous processes in Pott's disease. Trans Am Orthop Assoc. 1891;4:206–10.
- Lovett RW. The forcible correction of the deformity in Pott's disease: a review of the recent literature. Boston Med Surg J. 1898;138:228–30.
- Allison N. Pott's disease treated by ankylosing operations on the spinal column: a review of recent literature. Interstate Med J. 1912;19:456–7.
- Lange F. Support for the spondylitic spine by means of buried steel bars, attached to the vertebrae. Am J Orthop Surg. 1910;28:344–61.
- Venable CS, Stuck WG, Beach A. The effects on bone of the presence of metals; based upon electrolysis: an experimental study. Ann Surg. 1937;105(6):917–38.
- Venable CS, Stuck WG. Electrolysis controlling factor in the use of metals in treating fractures. JAMA. 1938;111(15):1349–52.
- Venable CS, Stuck WG. Three years' experience with vitallium in bone surgery. Ann Surg. 1941;114(2):309–15.
- Leventhal GS. Titanium, a metal for surgery. J Bone Joint Surg Am. 1951;33-A(2):473–4.
- Bannon BP, Mild EE. Titanium alloys for biomaterial application: an overview. In: Luckey HA, Kubli F Jr, eds. Titanium alloys in surgical implants. American society for testing and materials (ASTM STP796); 1983:7–15.

- Knott PT, Mardjetko SM, Kim RH, Cotter TM, Dunn MM, Patel ST, Spencer MJ, Wilson AS, Tager DS. A comparison of magnetic and radiographic imaging artifact after using three types of metal rods: stainless steel, titanium, and vitallium. Spine J. 2010;10(9):789–94.
- Bradley RJ, Chaoi EYS, Friedman J. Biomechanics and biomaterials. In: Dee R, Hurst L, Gruber M, Kottmaier S, eds. Principles of Orthopedic Practice (2nd ed). New York: McGraw-Hill; 1997:151–78.
- Albee FH. Transplantation of a portion of the tibia into the spine for Pott's disease: a preliminary report. JAMA. 1911;57(11):885–6.
- Hibbs RA. An operation for progressive spinal deformities: a preliminary report of three cases from the service of the orthopaedic hospital. N Y Med J. 1911;93:1013–6.
- Hibbs RA. A further consideration of an operation for Pott's disease of the spine: with report of cases from the service of the New York Orthopaedic Hospital. Ann Surg. 1912;55(5):682–8.
- Galloway HPH. The treatment of paralytic scoliosis by bone-grafting. Am J Orthop Surg. 1914;212(2):253–8.
- Smith EH. A consideration of the relative merits of the Albee operation and the Hibbs operation. Cal State J Med. 1915;13(5):194–5.
- 24. Burns BH. An operation for spondylolisthesis. Lancet. 1933;221:1233–5.
- Cloward RB. The treatment of ruptured lumbar intervertebral disc by vertebral body fusion. III. Method of use of banked bone. Ann Surg. 1952;136(6):987–92.
- Toumey JW. Internal fixation in fusion of the lumbosacral joints. Lahey Clin Bull. 1943;3:188–91.
- King D. Internal fixation for lumbosacral fusion. Am J Surg. 1944;66(3):357–61.
- Harrington PR. The history and development of Harrington instrumentation. Clin Orthop Relat Res. 1973;93:110–2.
- Harrington PR. Treatment of scoliosis: correction and internal fixation by spine instrumentation. J Bone Joint Surg Am. 1962;44-A:591–610.
- Andén U, Lake A, Nordwall A. The role of the anterior longitudinal ligament in Harrington rod fixation of unstable thoracolumbar spinal fractures. Spine (Phila Pa 1976). 1980;5(1):23–5.
- Wang GJ, Whitehill R, Stamp WG, Rosenberger R. The treatment of fracture dislocations of the thoracolumbar spine with halofemoral traction and Harrington rod instrumentation. Clin Orthop Relat Res. 1979;142:168–75.
- Livingston KE, Perrin RG. The neurosurgical management of spinal metastases causing cord and cauda equina compression. J Neurosurg. 1978;49(6):839–43.
- Sundaresan N, Galicich JH, Lane JM. Harrington rod stabilization for pathological fractures of the spine. J Neurosurg. 1984;60(2):282–6.
- Bridwell KH. Spinal instrumentation in the management of adolescent scoliosis. Clin Orthop Relat Res. 1997;335:64–72.

- Lagrone MO, Bradford DS, Moe JH, Lonstein JE, Winter RB, Ogilvie JW. Treatment of symptomatic flatback after spinal fusion. J Bone Joint Surg Am. 1988;70(4):569–80.
- Erwin WD, Dickson JH, Harrington PR. Clinical review of patients with broken Harrington rods. J Bone Joint Surg Am. 1980;62(8):1302–7.
- Luque ER. Segmental spinal instrumentation for correction of scoliosis. Clin Orthop Relat Res. 1982;163:192–8.
- Johnston CE II, Happel LT Jr, Norris R, Burke SW, King AG, Roberts JM. Delayed paraplegia complicating sublaminar segmental spinal instrumentation. J Bone Joint Surg Am. 1986;68(4):556–63.
- 39. Wilber RG, Thompson GH, Shaffer JW, Brown RH, Nash CL Jr. Postoperative neurological deficits in segmental spinal instrumentation: a study using spinal cord monitoring. J Bone Joint Surg Am. 1984;66(8):1178–87.
- Drummond D, Guadagni J, Keene JS, Breed A, Narechania R. Interspinous process segmental spinal instrumentation. J Pediatr Orthop. 1984;4(4):397–404.
- Thompson WA, Ralston EL. Pseudarthrosis following spine fusion. J Bone Joint Surg Am. 1949;31A(2):400–5.
- Bosworth DM. Surgery of the spine. American Academy of Orthopaedic Surgeons instructional course lectures. Ann Arbor: Edwards Brothers Incorporated; 1957:39–55.
- Boucher HH. A method of spinal fusion. J Bone Joint Surg. 1959;41B:248–59.
- 44. Roy-Camille R, Roy-Camille M, Demeulenaere C. Osteosynthesis of dorsal, lumbar, and lumbosacral spine with metallic plates screwed into vertebral pedicles and articular apophyses. Presse Med. 1970;78(32):1447–8.
- 45. Roy-Camille R, Benazet J-P, Desauge JP, Kuntz F. Lumbosacral fusion with pedicular screw plating instrumentation: a 10-year follow-up. Acta Orthop Scand Suppl. 1993;64(251 Suppl):100–4.
- 46. Abumi K, Panjabi MM, Duranceau J. Biomechanical evaluation of spinal fixation devices. Part III. Stability provided by six spinal fixation devices and interbody bone graft. Spine (Phila Pa 1976). 1989;14(11):1249–55.
- Krag MH. Biomechanics of thoracolumbar spinal fixation: a review. Spine (Phila Pa 1976). 1991;16(3 Suppl):S84–99.
- Steffee AD, Biscup RS, Sitkowski DJ. Segmental spine plates with pedicle screw fixation: a new internal fixation device for disorders of the lumbar and thoracolumbar spine. Clin Orthop Relat Res. 1986;203:45–53.
- Cotrel Y, Dubousset J. A new technic for segmental spinal osteosynthesis using the posterior approach. Rev Chir Orthop Reparatrice Appar Mot. 1984;70(6):489–94.
- Cotrel Y, Dubousset J, Guillaumat M. New universal instrumentation in spinal surgery. Clin Orthop Relat Res. 1988;227:10–23.

- 51. Cheh G, Bridwell KH, Lenke LG, Buchowski JM, Daubs MD, Kim Y, Baldus C. Adjacent segment disease following lumbar/thoracolumbar fusion with pedicle screw instrumentation: a minimum 5-year follow-up. Spine (Phila Pa 1976). 2007;32(20):2253–7.
- Dekutoski MB, Schendel MJ, Ogilvie JW, Olsewski JM, Wallace LJ, Lewis JL. Comparison of in vivo and in vitro adjacent segment motion after lumbar fusion. Spine (Phila Pa 1976). 1994;19(15):1745–51.
- Ha KY, Schendel MJ, Lewis JL, Ogilvie JW. Effect of immobilization and configuration on lumbar adjacent-segment biomechanics. J Spinal Disord. 1993;6(2):99–105.
- Hilibrand AS, Robbins M. Adjacent segment degeneration and adjacent segment disease: the consequences of spinal fusion? Spine J. 2004;4(6 Suppl):190S–4.
- Park P, Garton HJ, Gala VC, Hoff JT, McGillicuddy JE. Adjacent segment disease after lumbar or lumbosacral fusion: review of the literature. Spine (Phila Pa 1976). 2004;29(17):1938–44.
- McKenzie AH. Fernstrom intervertebral disc arthroplasty: a long-tern evaluation. Orthop Int Ed. 1995;3:313–24.
- Fernström U. Arthroplasty with intercorporal endoprosthesis in herniated disc and in painful disc. Acta Chir Scand Suppl. 1966;357:154–9.
- Gallie WE. Fractures and dislocations of the cervical spine. Am J Surg. 1939;46(3):495–9.
- Grob D, Crisco JJ III, Panjabi MM, Wang P, Dvorak J. Biomechanical evaluation of four different posterior atlantoaxial fixation techniques. Spine (Phila Pa 1976). 1992;17(5):480–90.
- Brooks AL, Jenkins EB. Atlanto-axial arthrodesis by the wedge compression method. J Bone Joint Surg Am. 1978;60(3):279–84.
- Tucker HH. Technical report: method of fixation of subluxed or dislocated cervical spine below C1-C2. Can J Neurol Sci. 1975;2(4):381–2.
- 62. Holness RO, Huestis WS, Howes WJ, Langille RA. Posterior stabilization with an interlaminar clamp in cervical injuries: technical note and review of the long term experience with the method. Neurosurgery. 1984;14(3):318–22.
- Magerl FP. Stabilization of the lower thoracic and lumbar spine with external skeletal fixation. Clin Orthop Relat Res. 1984;189:125–41.
- 64. Roy-Camille R, Saillant G, Mazel C. Internal fixation of the unstable cervical spine by posterior osteosynthesis with plates and screws. In: The Cervical Spine Research Society Editorial Committee, ed. The Cervical Spine. 2nd ed. Philadelphia: JB Lippincott; 1989:390–403.
- An HS, Coppes MA. Posterior cervical fixation for fracture and degenerative disc disease. Clin Orthop Relat Res. 1997;335:101–11.
- 66. Anderson PA, Henley MB, Grady MS, Montesano PX, Winn HR. Posterior cervical arthrodesis with AO reconstruction plates and bone graft. Spine (Phila Pa 1976). 1991;16(3 Suppl):S72–9.

- Jeanneret B, Gebhard JS, Magerl F. Transpedicular screw fixation of articular mass fracture-separation: results of an anatomical study and operative technique. J Spinal Disord. 1994;7(3):222–9.
- Fielding JW. The status of arthrodesis of the cervical spine. J Bone Joint Surg Am. 1988;70(10):1571–4.
- Jeanneret B. Posterior rod system of the cervical spine: a new implant allowing optimal screw insertion. Eur Spine J. 1996;5(5):350–6.
- Abumi K, Itoh H, Taneichi H, Kaneda K. Transpedicular screw fixation for traumatic lesions of the middle and lower cervical spine: description of the techniques and preliminary report. J Spinal Disord. 1994;7(1):19–28.
- Dickman CA, Sonntag VK, Papadopoulos SM, Hadley MN. The interspinous method of posterior atlantoaxial arthrodesis. J Neurosurg. 1991;74(2):190–8.
- Harms J, Melcher RP. Posterior C1-C2 fusion with polyaxial screw and rod fixation. Spine (Phila Pa 1976). 2001;26(22):2467–71.
- Goel A, Laheri V. Plate and screw fixation for atlanto-axial subluxation. Acta Neurochir (Wien). 1994;129(1–2):47–53.
- 74. Anderson RC, Ragel BT, Mocco J, Bohman LE, Brockmeyer DL. Selection of a rigid internal fixation construct for stabilization at the craniovertebral junction in pediatric patients. J Neurosurg. 2007;107(1 Suppl):36–42.
- Leonard JR, Wright NM. Pediatric atlantoaxial fixation with bilateral, crossing C-2 translaminar screws. Technical note. J Neurosurg. 2006;104(1 Suppl):59–63.
- Böhler J. Immediate and early treatment of traumatic paraplegias. Z Orthop Ihre Grenzgeb. 1967;103(4): 512–29.
- Gonugunta V, Krishnaney AA, Benzel EC. Anterior cervical plating. Neurol India. 2005;53(4):424–32.
- An HS, Simpson JM, Glover JM, Stephany J. Comparison between allograft plus demineralized bone matrix versus autograft in anterior cervical fusion: a prospective multicenter study. Spine (Phila Pa 1976). 1995;20(20):2211–6.
- Arrington ED, Smith WJ, Chambers HG, Bucknell AL, Davino NA. Complications of iliac crest bone graft harvesting. Clin Orthop Relat Res. 1996;8(329):300–9.
- Ebraheim NA, Elgafy H, Xu R. Bone-graft harvesting from iliac and fibular donor sites: techniques and complications. J Am Acad Orthop Surg. 2001;9(3):210–8.
- Kurz LT, Garfin SR, Booth REJr. Harvesting autogenous iliac bone grafts. A review of complications and techniques. Spine (Phila Pa 1976). 1989;14(12):1324–31.
- Kahnberg K-E. Biological principles of bone. In: Kahnberg K-E. Bone Grafting Techniques for Maxillary Implants. Oxford, UK: Blackwell Munksgaard; 2005:1–3.
- Urist MR. Bone: formation by autoinduction. Science. 1965;150(3698):893–9.
- 84. Girardi FP, Cammisa FP Jr. The effect of bone graft extenders to enhance the performance of iliac crest bone grafts in instrumented lumbar spine fusion. Orthopedics. 2003;26(5 Suppl):s545–8.

- 85. Cunningham BW, Kanayama M, Parker LM, Weis JC, Sefter JC, Fedder IL, McAfee PC. Osteogenic protein versus autologous interbody arthrodesis in the sheep thoracic spine: a comparative endoscopic study using the Bagby and Kuslich interbody fusion device. Spine (Phila Pa 1976). 1999;24(6):509–18.
- 86. David SM, Gruber HE, Meyer RA Jr, Murakami T, Tabor OB, Howard BA, Wozney JM, Hanley EN Jr. Lumbar spinal fusion using recombinant human bone morphogenetic protein in the canine: a comparison of three dosages and two carriers. Spine (Phila Pa 1976). 1999;24(19):1973–9.
- Fischgrund JS, James SB, Chabot MC, Hankin R, Herkowitz HN, Wozney JM, Shirkhoda A. Augmentation of autograft using rhBMP-2 and different carrier media in the canine spinal fusion model. J Spinal Disord. 1997;10(6):467–72.
- Carragee EJ, Hurwitz EL, Weiner BK. A critical review of recombinant human bone morphogenetic protein-2 trials in spinal surgery: emerging safety concerns and lessons learned. Spine J. 2011;11(6):471–91.

- Guppy KH, Paxton EW, Harris J, Alvarez J, Bernbeck J. Does bone morphogenetic protein change the operative nonunion rates in spine fusions? Spine (Phila Pa 1976). 2014;39(22):1831–9.
- Hodges SD, Eck JC, Newton D. Retrospective study of posterior cervical fusions with rhBMP-2. Orthopedics. 2012;35(6):e895–8.
- 91. Kim HJ, Buchowski JM, Zebala LP, Dickson DD, Koester L, Bridwell KH. RhBMP-2 is superior to iliac crest bone graft for long fusions to the sacrum in adult spinal deformity: 4- to 14-year follow-up. Spine (Phila Pa 1976). 2013;38(14):1209–15.
- Cahill KS, Chi JH, Day A, Claus EB. Prevalence, complications, and hospital charges associated with use of bone-morphogenetic proteins in spinal fusion procedures. JAMA. 2009;302(1):58–66.
- 93. Lad SP, Bagley JH, Karikari IO, Babu R, Ugiliweneza B, Kong M, Isaacs RE, Bagley CA, Gottfried ON, Patil CG, Boakye M. Cancer after spinal fusion: the role of bone morphogenetic protein. Neurosurgery. 2013;73(3):440–9.

# Index

## A

AAOS. See American Academy of Orthopedic Surgeons (AAOS) ACCUFLEX® system, 658 ACER. See Average cost-effectiveness ratio (ACER) Acetaminophen (paracetamol), 210, 300 Acquired degenerative intervertebral dysfunction (ADIVD), 423 conservative and surgical treatment, 437 loss of stability, 424 major dysfunction, 424-425, 437-438 maximal dysfunction, 425, 438 minimal dysfunction, 423-424, 436 minor dysfunction, 424, 437 AcroFlex, 605 Activ-L Artificial Disk, 618-620 Acupuncture, 212, 294, 758 Adipose-derived regenerative cell transplantation, 584-586 ADIVD. See Acquired degenerative intertebral dysfunction (ADIVD) Adjacent segment degeneration (ASD), 683, 685 Adjacent segment disease definitions, 311 motion-sparing surgery, 315-316 patient factor, 313 prevalence, 313 risk factor, 313-314 surgical intervention, 314-315 Adolescent idiopathic scoliosis (AIS), 695 ADS. See Adult degenerative scoliosis (ADS) Adult degenerative scoliosis (ADS), 695 AIS. See Adolescent idiopathic scoliosis (AIS) ALIF. See Anterior lumbar interbody fusion (ALIF) ALL. See Anterior longitudinal ligament (ALL) Allograft ACDF, 191 DBM, 333-334 femoral ring allograft, 370 mineralized allograft, 333 American Academy of Orthopedic Surgeons (AAOS), 181 American Society for Quality (ASQ), 7954

Anterior longitudinal ligament (ALL), 520, 620, 621, 627 Anterior lumbar interbody fusion (ALIF), 152, 281, 283, 315, 334, 409, 413, 615, 616, 617 anterior spinal fixation, 374 approach/exposure, 368 BERG technique, 369 contraindications, 368 femoral ring allograft, 370 graft material, 375-376 history and development, 365-366 indications debridement and fusion, 367 degenerative disk disease, 366-367 failed spine surgery, 367 spinal deformity, 367 spondylolisthesis, 367 laparoscopic/transperitoneal techniques, 368-369 mini-open retroperitoneal approach, 369-370 posterior spinal fixation, 372-373 preoperative planning/templating, 368 threaded cortical bone dowels, 370-371 threaded lordotic cages, 371-372 threaded titanium cylindrical cages, 371 Anterior transdiskal technique, 486 Aperius device, 682, 686-687 Apparent diffusion coefficient (ADC), 99, 100 ASD. See Adjacent segment degeneration (ASD) ASQ. See American Society for Quality (ASQ) Australopithecus A. afarensis, 10 A. anamensis, 10 Autologous stem cells, 338 Average cost-effectiveness ratio (ACER), 188, 189, 190.191 Axial low back pain, 299-300, 311, 534, 541, 548

## B

Baastrup disease, 98, 104, 732 Balloon-assisted endoscopic retroperitoneal gasless (BERG) technique, 369 BIOFLEX® system, 651, 659 Biportal approach, 784 BMI. *See* Body mass index (BMI) Body mass index (BMI), 204, 207, 211, 545, 580 Bone morphogenetic proteins (BMPs), 333–337, 375, 812, 813

#### С

Cauda equina syndrome (CES), 181 Caudal epidural steroid injections (CESI), 449, 453, 457-458 CEA. See Cost-effectiveness analysis (CEA) Cell transplantation, 575-576 Canine trial adipose-derived regenerative cell transplantation, 584–587 institutional guidelines, 576 observations emerged, 578 tissue analyses, 577, 579 Euro Disk Randomized Trial, 586 clinical evaluation, 581 descriptive analyses, 581-583 interim analysis, 581, 584 interventional surgery, 579 intraoperative setting, 580 MRI, 581, 583, 586 operative procedures, 580 Center sacral vertical line (CSVL), 699 Cervical Spine Outcomes Questionnaire (CSOQ), 198 CES. See Cauda equina syndrome (CES) CESI. See Caudal epidural steroid injections (CESI) Chronic low back pain, 41 Claudicatio spinalis, 240 Cobb method, 712–713 Coflex. 682 adjacent level, 688 adverse events, 687 complication, 673, 674 designing, 687 IDE study, 687 interspinous dynamic stabilization, 670 vs. laminectomy, 688 MRI compatible, 687 radiographic fusion, 688 spinous process fracture incidence, 687-688 spondylolisthesis, 688 surgical technique, 688 COMI. See Core Outcome Measures Index (COMI) Complimentary deoxyribonu-cleic acid (cDNA), 337 Core Outcome Measures Index (COMI), 348, 798, 800 Cost-effectiveness analysis (CEA), 187, 192-193 CEA, 187 comparison technologies, 188-190 CUA. 185 definition, 185 direct and indirect costs, 191 health care, true cost of, 188 HLD, 191 ICER, 186, 190, 191

vs. posterolateral fusion, 191 QALY direct cost, 191 health utilities, 185–188 vs. incremental costs, 186 retrospective analysis, 190–192 sensitivity analysis, 190, 192 treatment decisions, 192 Cost-utility analysis (CUA), 185, 186 Cryoanalgesia, 220, 237 Cryoneurolysis, 237 CSOQ. See Cervical Spine Outcomes Questionnaire (CSOQ) CSVL. See Center sacral tical line (CSVL) CUA. See Cost-utility analysis (CUA)

## D

Dascor<sup>™</sup>, 629 DBM. See Demineralized bone matrix (DBM) Degenerated disk disease (DDD), 203 action mechanisms, 426 ADIVD, 423 conservative and surgical treatment, 437 loss of stability, 424 major dysfunction, 424-425, 437-438 maximal dysfunction, 425, 438 minimal dysfunction, 423-424, 436 minor dysfunction, 424, 437 advantages, 426 anatomy and pathophysiology annulus fibrosis, 204 disk innervation, 204 nucleus pulposus, 204 biomechanical load, 52-53 clinical data, surgery FDA criteria, 296 MCID, 295 Oswestry Disability Index (ODI), 295 RCTs, 294 clinical implications, 57 conservative treatment acupuncture, 212 alternative treatments, 213 biological agents, 213 ESIs, 213 gene therapy, 214 intradiscal thermal procedures, 213 lifestyle modifications, 211-212 massage, 212 medications, 210-211 MPC, intradiscal injection, 213–214 physical activity, 209-210 physical therapy, 210 protein factors, 213 randomized placebo-controlled trial, 213 spinal manipulation therapy, 212 **TENs**, 212 traction. 212 definition, 31 degenerative changes, 203

diagnostic imaging bone scintigraphy, 293 computed tomography, 291-293 diskography, 293-294 MRI, 293-294 pain management, 294 radiographs, 291-292 spinal cord stimulation, 294 diskography, 208, 209 embryonal and foetal development, 44-45 end plate biomechanical and nutritional conditions, 50, 51 cadaveric human vertebrae, 50 cartilage end plate, 48-49 mechanical functions, 49-50 pathophysiologic role, 56-57 FJP, 226-227 genetic predisposition, 51-52 growth factors, 55-56 history age, 205, 206 aggravating and alleviating factors, 205-207 BMI, 207 diskogenic pain, 204 gender differences, 207 pain location, 205 psychosocial considerations, 207 red flag symptoms, 204-205 surgical, 207 infants and adolescents, 45 matrix-degrading enzymes, 54 metabolic and nutritional effects, 53 modic changes, 209 MRI, 208, 425 NF-KB Pathway, 56 nucleus pulposus memory-coiling spiral implant, 429 advantages, 433, 436 biomechanical testing, 436 design specifications, 430-431 dynamic flexion-extension X-rays, 434 implant geometry, 431 implant material, properties of, 431 mechanical properties, 431-432 pain and functional outcome, 433 patient selection, 432-433 postoperative MRI, 433, 435 preoperative MRI, 433, 434 twist CT scan, 435, 436 pathology, 289-290 patient evaluation, 290 percutaneous intradiskal radiofrequency techniques chymopapain enzyme, 425 classification, 426 clinical outcomes, 426 contained disk herniation, treatment of, 427-429 contraindications, 427 Discogel®, 429 history, 426

indications, 427 isolated disk degeneration, treatment of, 427, 428 minimally invasive transforaminal approach, 425 nucleolysis techniques, 425 patient selection, 427 principles of, 426 prophylactic antibiotics, 426 role of, 429 thin catheter, 425-426 physical examination, 207-208 physiopathological processes aging spine, 138 biochemical composition, 138 cellular phenotype and extracellular matrix, 139 chemical events, 139 disk nutrition disturbance, 138 hyperalgesia, 139 intervertebral disk, 138 marrow contact channels, 138 nerve ingrowth, 140 painful disks, 139 vertebral end plate, 140 pro-inflammatory mediators, 54-55 radiographs apparent diffusion coefficient, 99 asymmetrical disk space and osteophytes, 93 computed tomography, 93-96 EOS imaging, 100 facet arthrosis, 92 intervertebral disk calcifications, 92 L4/L5 disk space, 93 L5-S1 level, 92 MRI (see Magnetic resonance imaging (MRI)) MR neurography, 99 osteophytes, 91, 92 postoperative imaging, 99, 105-106 Schmorl nodes and disk calcification, 92 ultrashort TE (UTE), 100 vacuum disk phenomenon, 91 sagittal balance (see Sagittal balance) source of, 703 spinal degeneration cascade, 423 T2WI, 208, 209 twist CT scan, 425 X-rays, 425 young and medium-aged adults, 45-47 Degenerative marrow changes biomechanical stress, 164 cytokine propionic acid production, 167 diskography, 161 longitudinal study, 161 microbiologic analysis, 165-166 MR imaging, 159 pars fracture and pedicle hyperintensity, 163 PGP 9.5 immunoreactivity, 165 prevalence, 162 proinflammatory mediators, 165 TNF immunoreactive cells, 165 type I changes, 159

Degenerative marrow changes (cont.) type II changes, 159, 161, 162 type III changes, 159, 162, 164 vertebral end plates, 165 Wolff's law, 160 Degenerative scoliosis, 302 fusion level selection, 699-700 history, 695 level I treatment, 697 level II treatment, 697 level III treatment, 697-698 level IV treatment, 698 level VI treatment, 699 level V treatment, 698 Modic changes, 697-698 preoperative evaluation, 695-697 Degenerative spondylolisthesis, 95, 188, 226, 229, 240, 300, 301, 396, 408, 409, 410, 549, 654, 657, 672, 704, 717, 720 Demineralized bone matrix (DBM), 188, 333-334 DIAM device, 670, 671, 682, 684, 688-689 Diet, 84, 212 Discoplasty, 345, 349, 351 Disk arthroplasty ball joint mechanism, 602 biomechanical requirements, 601 development of, 602 lumbar fusion, 614 motion function, 602 nucleus replacements, 614, 630 Dascor<sup>™</sup>, 629 devices, 627-628 elastomeric, 628 mechanical, 629 NeuDisc<sup>™</sup>, 628 NuBac<sup>™</sup>, 629 NuCore<sup>™</sup>, 628 PDN. 628 Regain<sup>™</sup>, 629 nucleus replacement strategies, 602-603 optimal lifetime, 601 PDR, 613, 614 principles, 602 symptom-free biomechanical function, 601 TDR, 614 Activ-L Artificial Disk, 618-620 adjacent-segment degeneration, 622-624 advantages, 615 center of rotation, disk replacement, 620 complications, 627 FDA-approved artificial disk devices, 616 Fernstrom ball, 615 Flexicore artificial disk, 618, 619 implant-bone interface, stability of, 625 indications, 615 intervertebral disk, 620 Kineflex artificial disk, 618 load sharing, 620-621 Maverick artificial disk, 617 motion patterns, 621-622

motion preservation, 624-625 operative procedure, 615-616 ProDisc-L, 615-617 SB Charité III (DePuy Spine), 616 shock absorption capacity, 626 wear testing, 625-626 TDR system AcroFlex, 605 complications, 607 constrained implants, 606 Maverick artificial disk, 606 ProDisc, 605 SB Charité III, 603, 604 success rates, 606 Swedish prospective randomised controlled trial, 607 Swiss registry, 606-607 type I Charité artificial disks, 603, 604 type II Charité artificial disks, 603 type III Charité artificial disks, 603, 604 Diskectomy, 355 endoscopic diskectomy (see Endoscopic diskectomy) laser diskectomy, 427 lumbar microdiskectomy, 539-540 Disk injury compressive failure, 27 disk herniation anterolateral bending, 29 bending and compression, 29 cadaveric spine specimen, 27 mechanical consequences, 30 mechanism, 28 radioactive tracer study, 27 disk puncture, 128 endplate involvement, in disk prolapse, 29-30 genes and environmental influences, 34 genetic inheritance, 34 medicolegal considerations blame mechanical loading, 34 disk degeneration vs. herniation, 34 mechanical acceleration, 34 repetitive loading and fatigue failure, 33-34 torsion and bending injuries, 27 DISK nucleoplasty<sup>™</sup>, 428 Diskogenic pain, 110, 112, 128, 129, 130, 131, 161, 162, 203, 204, 207, 240, 294, 426, 427, 764-765 Diskography analgesic diskography, 129-130 caveats, 122 degenerative marrow changes, 161 diskography standards, 122-123 disk puncture AP and lateral projections, 115 diskogram needle insertion, 115 disk provocation, 118-120 down the beam/tunnel vision technique, 114 fluoroscope, 114 lateral/extra-pedicular approach, 113

lateral fluoroscopic view, 116 L5-S1 needle position, 117 Quinke tip spinal needle, 115 S1 superior articular process, 117 sterile gauze, 118 superior articular process (SAP), 113 two-needle technique, 114 false-positive rates, 126-128 imaging AP fluoroscopic view, 120-121 axial post-diskography CT scan, 122-123 contrast medium, 120 lateral fluoroscopic view, 120 Modified Dallas Discogram Scale, 121 post-diskography axial CT scanning, 121 indications and contraindications, 110-111 internal morphology and lateral protrusions, 109 lumbar diskography, 112 MRI and CT, 110 patient position, 113 patient preparation antibiotic prophylaxis, 111-112 sedation, 112 sterile technique, 112 post-procedure care, 123 potential risks and complications, 124-125 predictive value, 130-131 preprocedure evaluation, 111 presurgical planning test, 109 provocation, 125-126 validity and accuracy, 110 DSS® hybrid performance system (HPS), 660 DYNESYS® device, 637, 656-658

## Е

Endoscopic diskectomy advantages, 474, 484 anterior transdiskal technique, 486 complications and avoidance, 483-484 contraindications, 474 definitions, 471 disadvantages, 474 extraforaminal approach anatomical/pathological conditions, 476 bone resection, 478, 480 at caudal pedicle, 478-480 dilator, 476 herniation location, 478 intraforaminal/extraforaminal pathologies, 476 intraoperative view, 478, 480 operating sheath, 476, 478, 479 spinal cannula, 476, 479 indications disk herniation, 484-485 patient selection, 472-473 recurrent disk herniation, 485 spinal canal stenosis, 485-486 interlaminar approach

bone resection, 481, 482 dilator insertion, 479 indications, 474 intraoperative, 481 intraspinal cyst, 482 ligamentum flavum, 479-481 operating sheath, 480-482 over-the-top technique, 482 single-sided port, 482 skin incision, 479 spinal canal, 481 surgical principle, 471-472 lateral transforaminal approach, 486 indications, 473 surgical principle, 472 surgical technique, 476, 478, 479 minimally invasive spine procedures, 789 PELDTA, diskogenic back pain, 787-788 percutaneous endoscopic lumbar foraminotomy, 786–787 posterior foraminotomy techniques, 486 postoperative care, 483-484 preoperative planning anesthesia, 475 equipment, 475-477 examinations, 475 patients' informed consent, 475 positioning, 475 preparation, 475 radiofrequency ablation, 788-789 randomized trials, 789 rod-lens endoscopes, 484 terminology, 471 transforaminal endoscopic diskectomy contained disk herniation, 784-785 noncontained disk herniation, 785 selective diskectomy, 785-786 Epidemiology age and sex, 62 cigarette smoking, 63 genetic factors, 63 loading efforts, 62 postural factors, 62 psychological factors, 64 twin studies, 63-64 vibrations, 62-63 weight, 62 work/sports, 62 Epidural steroid injections (ESIs), 213 causes, 449 CESI, 449, 453, 457-458 contraindications, 449 corticosteroids, 449 histopathological findings, 449 ILESI, 453-454, 457-458 inflammatory mediators, 449 TFESI, 449-450 anesthetic test dose, 452 artery of Adamkiewicz, 452 bleeding, 451

Epidural steroid injections (ESIs) (cont.) common side effects, 450-451 digital subtraction angiography, 452 doses and injections number, 452-453 dural puncture, 452 efficacy of, 454-456 infection, 451 intermittent fluoroscopy, 452 local anesthetic, 450 needle placement, 450 predictors, 456-457 TF AP post-contrast injection, 450, 451 TF AP view, 450, 451 TF lateral view, 450, 451 transforaminal oblique view, 450 Epigenetics, 84 EQ-5D. See European Quality of Life-5 Dimensions (EQ-5D) ESIs. See Epidural steroid injections (ESIs) Euro Disk Randomized Trial cell transplantation clinical evaluation, 581 descriptive analyses, 581-583 interim analysis, 581, 584 interventional surgery, 579 intraoperative setting, 580 MRI, 581, 583 operative procedures, 580 European Quality of Life-5 Dimensions (EQ-5D), 187, 188, 197-199, 558 EXPEDIUM® PEEK rod, 647, 660 Extraforaminal approach, 271, 273, 473, 476, 478, 479, 480, 483

## F

Facet joint pain (FJP). See Zygapophyseal/zygapophysial joints Failed back surgery syndrome (FBSS) comparative studies, 531-532 CT myelography, 533 definition, 525 facet joint pain, 535 laminectomy, 534 lateral stenosis, 532 organic disease processes, 530 painful internal disk disruption, 535 quantitative studies, 534 solid fusion, 534 surgical complications, 533 Failed back syndrome, 761–762 Farcy's osteotomy, 323 Femoral ring allograft, 370, 373, 374, 375, 605 Finite element analysis (FEA), 642 Flexicore artificial disk, 618, 619 Foraminal disk herniation, 498-499 FSU. See Functional spinal unit (FSU) Functional spinal unit (FSU), 633, 635, 637, 641

## G

Gene therapy, 214, 337-338 Genetics and DNA sequencing technologies, 84 epigenetics, 84 familial aggregation and twin studies, 72 genetic association studies candidate gene approach, 76 case-control association studies, 75 genome-wide association and meta-analysis, 76-77 HapMap Project, 75 SNPs, 73 statistical analysis, 73 variable phenotypes, 74 genetic variations and markers, 68-70 LDD asporin gene, 80 genetic risk factor, 80-81 **GWAS**. 83 matrix metalloproteinases, 82 sickle tail (SKT) gene, 82 thrombospondin 2 (THBS2) gene, 82 Trp2 allele, 82 type I collagen, 82 variable number of tandem repeats (VNTR), 80 vitamin D receptor (VDR), 80 linkage and disequilibrium, 71–72 linkage studies, 73, 74 MC, 143 Mendelian genetics and complex disease, 70-71 mutations and polymorphisms, 70 phenotype, definitions, 77-78 rare variants, 83 statistical significance, 78 structure and function, 67-68 suspect genes, effects of, 78-79

## H

Healthcare Research and Quality, 794 Health-related quality of life (HRQOL), 295 clinical relevance, 198-199 cost-effectiveness, 199 disease-specific measures, 196-197 EQ-5D, 197-198 generic measures, 196 health utility scales, 197 health utility scores, 198 MCID. 199 pain scales, 197 PRO data, 195 SCB. 199 SF-6D, 198 Health Utilities Index (HUI), 197 Herniated umbar disks (HLD), 191 Hexenschuss, 229, 230 High intensity zones (HIZs), 208, 209 HIZs. See High intensity zones (HIZs)

HLD. See Herniated umbar disks (HLD) HORIZON BalanC<sup>™</sup>, 658–659 HRQOL. See Health-related quality of life (HRQOL) HUI. See Health Utilities Index (HUI) Human Genome Epidemiology Network (HuGENet), 79 Human immunodeficiency virus (HIV), 333

#### I

ICER. See Incremental cost-effectiveness ratio (ICER) ID. See Interspinous devices (ID) ILESI. See Interlaminar epidural steroid injection (ILESI) Iliac crest bone graft (ICBG), 188, 190 Impaction posterior osteotomy (IPO), 744, 745 Incremental cost-effectiveness ratio (ICER), 186, 188, 189.191 Interlaminar epidural steroid injection (ILESI), 453-454, 457-458 International Spine Intervention Society (ISIS) guidelines, 443, 445, 447, 460 International Spine Intervention Society (ISIS) Practice Guidelines, 528 Interspinous devices (ID) Aperius, 682, 686-687 biomechanical data cadaveric, 684 in vivo. 684-685 Coflex, 682 adjacent level, 688 adverse events, 687 designing, 687 IDE study, 687 vs. laminectomy, 688 MRI compatible, 687 radiographic fusion, 688 spinous process fracture incidence, 687-688 spondylolisthesis, 688 surgical technique, 688 complications, 684 descriptions of, 682, 683 DIAM, 682, 688-689 drawback, 684 indications, for placement, 682-683 laminectomy, 683-684 percutaneous/mini-open approach, 681 Wallis, 682 X-Stop, 682, 685-687 Interspinous dynamic stabilization device, 667 actuarial survivorship analysis, 666 clinical and radiological findings, 670-672 Coflex, 670 complication, 668, 672-674 DIAM device, 670 distraction. 668 floating interspinous device, 665 fusion and pedicle screw systems, 665 indications, 668

In-Space, 667 interspinous spacers, 668 ISPD, 670 LimiFlex, 670 Morse taper, 665 ROM, 669-670 VAS pain score, 667-668 Wallis system, 668-669 first-generation, 666 flexion-extension, 670 flexion-limiting bands, 670 indications and contraindications, 667 L4-L5, 669 UniWallis implant, 667 X-Stop, 667 Interspinous process distance (ISPD), 670 Intervertebral disks, 23 annulus fibrosus, 305 nucleus pulposus, 305 rostral and caudal endplates, 306 Sharpey's fibers, 306 Intervertebral disk space annulus fibrosus, 307 degenerative changes, 307-308 disk pressure, 306 extra-discal hydrostatic pressure, 307 fluid shifts, 307 nucleus pulposus, 307 oncotic pressure, 307 spinal instability, 308-310 vertebral column height, 306 Intra-articular joint injection hyaluronic acid, 446 intra-articular corticosteroid/saline, 446 ISIS guidelines, 443 8 mL of injectate, 445 nonionic contrast, 443, 444 SPECT imaging, 446 spinal needle insertion, 443, 444 Intradiskal electrothermal therapy (IDET), 213, 426 Investigational device exemption (IDE) study, 617, 659, 687 IPO. See Impaction posterior osteotomy (IPO) ISOBAR TTL<sup>™</sup> device dynamic instrumentation, 645-646 implant design, 643-645 ISOLOCK<sup>™</sup> device, 645 in vitro testing, 643-645 ISOLOCK® device, 639, 655-656

#### J

Joint Commission on Accreditation of Healthcare Organizations (JCAHO), 181

#### K

Kineflex artificial disk, 618

#### L

Lasègue maneuver, 13, 16 Laser-assisted spinal endoscopy (LASE), 787 Laser diskectomy, 427, 788-789 Lateral lumbar interbody fusion (LLIF), 519. See also Transpsoas approach LBG. See Local bone graft (LBG) LEGACY<sup>™</sup> PEEK rod implant, 658-659 Linkage disequilibrium (LD), 72 Local bone graft (LBG), 188, 190 Lower limbs, 7–8 ankle extension, 735 knee flexion, 735 Lumbar anal stenosis interspinous devices, 516-517 midline posterior ligaments, 514 open laminectomies, 514 radiographic and cadaver studies, 514 spondylosis, 514 SPORT, 514 technique description, 514-516 Lumbar degenerative disk disease legal aspects alternative management strategies, 180 CES, 181 civil law, 179 compensation status, 182 constitutional law, 179 direct and indirect costs, 182 legislative restrictions, 182 lumbar surgery, 179 neurosurgical malpractice claims, 180 poor records maintenance, 183 POVL, 180-181 Roman law-based system, 182 sharp instruments, 180 treatment plan, 180 wrong-level surgery, 181-182 wrong-patient surgery, 181 wrong-procedure errors, 181 wrong-site surgery, 181 psychosocial aspects disease model of illness, 172-174 epidemiology, 172 incidence, 171 low back pain and disability, 173 modern medical treatments, 171 risk factors, 172-173 work-related issues, 174-175 Lumbar disc disease (LDD), 67 asporin gene, 80 genetic risk factor, 80-81 **GWAS**. 83 matrix metalloproteinases, 82 sickle tail (SKT) gene, 82 thrombospondin 2 (THBS2) gene, 82 Trp2 allele, 82 type I collagen, 82 variable number of tandem repeats (VNTR), 80 vitamin D receptor (VDR), 80

Lumbar disk ageing and degeneration abnormal disk function, 31 back pain and sciatica, 32-33 bony endplate, 31 'endplate-driven' degeneration, 32 phenotypes, 32 Lumbar disk arthroplasty, 20, 316 biomechanics, 614, 620-626 success rates, 542-543 Lumbar disk herniation clinical assessment, 252 complications dural opening, 275-276 wrong-level surgery, 274-275 conservative interventions, 253-254 critical evaluation, 276 extraforaminal/far-lateral approach closure, 273 contraindication, 269 disk exposure, 273 indication, 269 positioning, 269, 271, 272 preoperative planning, 269 soft-tissue approach, 271-273 fragmentectomy vs.diskectomy, 263 interlaminar approach closure, 268 contraindications, 265 disk exposure, 267-268 indications, 265 positioning, 265-266 preoperative planning, 265 soft-tissue approach, 267 methodological considerations clinical equipoise, 258 randomized trials, 258 spine registries, 258-259 microsurgical vs.non-microsurgical surgical techniques, 263 MRI interpretation and consequences, 252-253 postoperative care CSF. 274 standard dural repair, 274 uncomplicated surgery, 274 prognostic factors, 253 recurrent disk herniation exposure, 274 positioning, 273 preoperative planning, 273 soft-tissue approach, 273-274 subperiosteal vs.transmuscular approach, 263-264 surgical techniques disk removal, 257 microscopic diskectomy, 254 minimally invasive techniques, 256 open vs. microscopic diskectomy, 255-256 spinal scarring, prevention, 256-257 translaminar approach closure, 269 contraindication, 268

disk exposure, 269 indications, 268 positioning, 268 preoperative planning, 268 soft-tissue approach, 268-271 tubular vs. conventional techniques, 264 Lumbar fusion, 185 disk arthroplasty, 614 non-instrumented lumbar fusion, 540-541 success rates, 541-542 Lumbar interbody fusion (LIF) autologous iliac crest graft, 357 complications, 360-361 PLIF. 356 surgical indications and technique biomechanical advantages, 358 indications and contraindications, 357 lumbar arthrodesis, 358 preoperative imaging, 358 Lumbar intervertebral disks nociceptive pain receptors, 458 provocative discography analgesic discography, 462-463 anatomic confirmation, 458 antibiotic prophylaxis, 459 asymptomatic disk, 458 contraindications, 459 CT/MRI discography, 462 disk pressure, 458 false positives, 460-461 fluoroscopic imaging, 458 immediate procedural complications, 459 indications, 459 internal controls, 459 intradiscal AP post-contrast, 460, 461 intradiscal AP pre-contrast, 460 intradiscal lateral post-contrast, 460 long-term complications, 459 potential intradiscal therapeutic agents, 463-464 pressure-controlled intradiscal injection, 460 therapeutic validity, 461-462 two-needle technique, 459 Lumbar lordosis bone graft anterior fusion, 327-328 CT and isotopic CT, 326 fusion material, 326 posterior fusion, 326 disk section, 321-322 facet joints opening, 321 instrumentation, 323-224 osteotomies, 322 patient positioning, 321-322 posterior compression hooks, 324 polyaxial screws, 324 total lumbar disk replacement, 328-329 translation, 325 Lumbar microdiskectomy, 257, 491, 539-540 Lumbar radiculopathy, 301, 454

Lumbar scoliosis correction, 513, 519–522 Lumbar spinal stenosis (LSS), 300–301 Lumbar spinal surgery ODI scoring system, 539 spine surgeons and spine interventionalists, 537–538 Lumbar spine bending moment, 26 compressive force, 25–26 gravitational loading, 26 inertial forces, 26 muscle tension, 26 shear force, 26 torsion, 26

#### М

Magnetic resonance imaging (MRI), 292 acute myositis, 98 Baastrup disease, 98, 104 canal stenosis, 94, 98 cartilaginous Schmorl node formation, 97 contraindications, 280 coronal slices, 265 end plate bone marrow, 96 FJP. 237 with gadolinium, 533 gadolinium-enhanced slices, 265 interpretation and consequences, 252 intradiscal gas, 96 isolated degeneration, of L4/L5 disk, 94, 97 ligament morphology, 98 L4-L5 facet joint osteoarthrosis, 98, 104 L4/L5 level and herniation, 94, 96 lumbar spine, 284 MC analysis, 151 Modic type 3, 97 Modic type 1 end plate, 96 Modic type 2 end plate, 97 multilevel disk degeneration, 94, 97 neovascularization, 95 nucleus pulposus dehydration, 425 nucleus pulposus implant, 435 paraspinal muscles, 98 Pfirrmann's classification, 77 reactive edematous changes, 98 respective disk height, 581 sagittal slices, 265 stenosis and cauda equina compression, 408 **VESC**, 145 zygapophysial joint, 227, 228 Matrix metalloproteinases (MMPs), 54–56, 76, 82, 312 Maverick artificial disk, 617 MCID. See Minimum clinically important difference (MCID) MCS. See Mental composite summary score (MCS) Medial branch blocks AP and lateral views, 444 false negatives, 448 false positives, 447-448

Medial branch blocks (cont.) ISIS guidelines, 445 local anesthetic, 444 pre-and post-procedure pain level, 445 RFA. 447 SAP. 444 technical complications, 445 venous flow, absence of, 444, 445 Medial branch neurotomy, 236 coagulation, 234 contraindications, 234 controlled medial branch blocks, 233 L5 dorsal ramus, 235 lesion size, 234 mamillo-accessory ligament, 235 optimal temperature, 234 oscillation heats, 234 patient selection, 234 radiofrequency electrode, 234-236 risks, 235 spinal needle insertion, 234 Medical Outcomes Study (MOS), 196 MEDS. See Microendoscopic decompression for stenosis (MEDS) Mental composite summary score (MCS), 196 Mesenchymal precursor cells (MPCs), 213 Microendoscopic decompression for stenosis (MEDS), 514, 515 Micro-endoscopic diskectomy (MED), 264 Minimal clinically important difference (MCID), 295 Minimally invasive spinal surgery (MISS) indications, 513 lumbar anal stenosis, decompression of interspinous devices, 516-517 midline posterior ligaments, 514 open laminectomies, 514 radiographic and cadaver studies, 514 spondylosis, 514 SPORT, 514 technique description, 514-516 lumbar scoliosis correction, 519-522 vs. open techniques, 513 spondylolisthesis, 517-518 Minimally invasive surgery (MIS) anesthesia and positioning, 493, 494 complications and avoidance, 499-500 disk herniations, 491, 492 equipment, 493 foraminal disk herniation, 498-499 obese patients, 492-493 paramedian disk herniation calcified annulus, 498 coagulation, 497 dilators, 493-494 fluoroscopy C-arm, 493 guide pin, 493 hemostasis, 498 incision, 493, 494 laminotomy, 496 ligamentum flavum removal, 496-497

medial foraminotomy, 496 monopolar cautery and pituitary rongeurs, 496, 497 nucleus fragment, 497 postoperative care, 499 Minimum clinically important difference (MCID), 199, 295 MIS. See Minimally invasive surgery (MIS) MMPs. See Matrix metalloproteinases (MMPs) Modic changes (MC) bone marrow structure, 145–146 clinical correlation and LBP, 149 L5-S1 level, 150 14-month longitudinal cohort study, 151 population-based study, 150 VESC, 150 diskcography, 151 epidemiology, 144-145 history, 140 mixed forms of, 147-148 modic 1 (M1) changes, 146-147, 544 modic 2 (M2) changes, 147 modic 3 (M3) changes, 147 natural history, 148-149 physiopathology animal models, 141 antibiotic protocol, 144 apoptosis, 141 cartilaginous end plate, 141 chymopapain and brutal acceleration, 142 - 143end plate fracture, 142 genetics, 143 inflammation, 143 lower lumbar levels, 141 mechanical component, 141-142 mechanical factors, 141 microtrauma, 141 programmed cell death, 141 Schmorls node, 141 surgical rupture, end plate, 142 vertebral end plates, 141 reproducibility, 145 and surgical series disk arthroplasty, 153 diskectomy, 153 dynamic fusion, 153 fusion, 152-153 intradiscal steroid injection, 153-154 Modified Dallas Discogram Scale, 121 Monitored anesthesia care (MAC), 687 MOS. See Medical Outcomes Study (MOS) Myelography, 109

## Ν

National Institute for Health and Care Excellence (NICE), 190, 192 NDI. See Neck Disability Index (NDI) Neck Disability Index (NDI), 196-198 NEL-like molecule-1 (NELL-1), 337 Neodymium:yttrium-alluminum-garnet (Nd:YAG) laser, 788 NeuDisc<sup>™</sup>. 628 NICE. See National Institute for Health and Care Excellence (NICE) Nonimage-guided injections, 441 Nonsteroidal antiinflammatory agents (NSAIDs), 210, 211, 437, 454 Normal disk structure annulus fibrosus (AF), 43 components, 42 nucleus pulposus (NP), 43 pathological condition, 42 physiological circumstance, 45 NuBac<sup>™</sup>, 629 NuCore<sup>™</sup>, 628

## 0

Obesity, 62, 211 Oblique lumbar interbody fusion advantage and disadvantage, 402 complications, 397, 400 diagnosis, 397 disk space exposure, 396 laparoscopic procedures, 393 lateral decubitus and radiograph, 394 L5-S1 disk, 395 minimally invasive anterior approach, 393 mini-open techniques, 393 muscle-splitting approach, 394 neurological injury, 400 operative time, 397 postoperative radiographs, 398, 399, 404 preoperative radiograph, 398, 399 psoas muscle, 394 retroperitoneal space, 394 segmental vessels, 395 "sliding window" technique, 395 stand-alone anterior fusion, 403 treatment strategies, 393 vertebral end-plates, 395 vertebra's pedicular vessels, 403 ODI. See Oswestry Disability Index (ODI) OPDQ. See Oswestry Low Back Pain Disability Questionnaire (OPDQ) Opioids, 112, 211, 565 Osteoporotic spine clinical outcomes, 348 degenerative scoliosis, 346-347 lateral interbody fusion, 345 mini-open lateral approach, 345 prophylactic cement reinforcement, 345 discoplasty procedure, 349, 351 fixation procedures, 343-344 iliac crest bone graft, 348 junctional kyphosis, 349 randomized controlled studies, 343

spinal stenosis, 344–345 T-score, 343 Oswestry Disability Index (ODI), 196–198, 295, 296, 427, 518, 538–539, 541, 542, 654, 659 Oswestry Low Back Pain Disability Questionnaire (OPDQ), 581, 582

## P

Paramedian disk herniation calcified annulus, 498 coagulation, 497 dilators, 493-494 fluoroscopy C-arm, 493 guide pin, 493 hemostasis, 498 incision, 493, 494 laminotomy, 496 ligamentum flavum removal, 496-497 medial foraminotomy, 496 monopolar cautery and pituitary rongeurs, 496, 497 nucleus fragment, 497 Partial disk replacement (PDR), 613, 614 Patient-reported outcomes (PRO), 195 PCS. See Physical composite summary score (PCS) PDN device. See Prosthetic disk nucleus (PDN) device PDN-SOLO®, 603 PDR. See Partial disk replacement (PDR) PDS. See Pedicle screw-based posterior dynamic systems (PDS) Pedicle screw-based posterior dynamic systems (PDS) ACCUFLEX® system, 658 advantage, 633-634 BIOFLEX® system, 659 classification, 633, 634 concept of, 647 DSS® and DSS® HPS, 660 dynamic fusion control Motion, 640, 641 dynamic instrumentation, 641-643, 647 intervertebral fusion, 640, 641 ISOBAR TTL<sup>™</sup> device (see ISOBAR TTL<sup>™</sup> device) PubMed search, 641 restore motion, 640, 641 dynamic stabilization, 634 alignment/posture, 638-639 implant longevity and adjacent level, 639-640 kinematics, 636-638 load distribution, 635-636 DYNESYS® system, 656-658 EXPEDIUM<sup>®</sup> PEEK rod, 660 graf ligament, 653-654 HORIZON BalanC<sup>™</sup>, 658–659 hybrid dynamic device, 653 implant designs, 647 ISOLOCK® device, 655-656 LEGACY<sup>™</sup> PEEK rod implant, 658-659 metallic rod, 650-652

Pedicle screw-based posterior dynamic systems (PDS) (cont.) soft device, 648-649 STABILIMAX® system, 659 TWINFLEX®, 654-655 Pelvis Australopithecus, 6 bipedalism, 7 gluteus maximus, 6 gluteus minimus, 6 iliac bones, 5 sacrum, 5 and spinal column, 7 static upright posture and bipedal gait, 6 ventrocranial iliac spine, 5 vertebrae morphology, 6 Percutaneous endoscopic lumbar diskectomy (PELD), 784, 785, 788 Percutaneous endoscopic lumbar diskectomy and thermal annuloplasty (PELDTA), 787-788 Percutaneous intradiskal radiofrequency techniques chymopapain enzyme, 425 classification, 426 clinical outcomes, 426 contained disk herniation, treatment of, 427-429 contraindications, 427 Discogel®, 429 history, 426 indications, 427 isolated disk degeneration, treatment of, 427, 428 minimally invasive transforaminal approach, 425 nucleolysis techniques, 425 patient selection, 427 principles of, 426 prophylactic antibiotics, 426 role of, 429 thin catheter, 425-426 Percutaneous intradiskal radiofrequency thermocoagulation (PIRFT) techniques, 427 Percutaneous laser disk decompression (PLDD), 788 Percutaneous nucleus replacement (PNR), 629 Physical composite summary score (PCS), 196 PIO. See Posterior impaction osteotomy (PIO) PIRFT. See Percutaneous intradiskal radiofrequency thermocoagulation (PIRFT) techniques PLIF. See Posterior lumbar interbody fusion (PLIF) PLSS. See Post-lumbar surgery syndrome (PLSS) Ponte-type osteotomy, 742, 746 Posterior foraminotomy techniques, 486 Posterior impaction osteotomy (PIO) bone impactors size, 744 bone resection, 743 cancellous bone, 743 corporeal osteotomy, 743-744 decancellation, 744 knife dissection, 743 mechanical stability, 744-745 partial pedicle resection, 745 patient, in radiolucent table, 743

pedicle-type distracting clamp, 744, 745 peroperative view, 744 posterior incision, 743 postoperative, 745 principles of, 743 root entrapment/impingement, 744 spinal fixation, 745 spreading meary-type clamp, 744, 745 stiff deformities, 746 wedge creation, 744 Posterior longitudinal ligament (PLL), 415, 615 Posterior lumbar interbody fusion (PLIF) advantages/limitations, 413 clinical cases, 416-418 indications, 409-410 late complications, 413-415 pedicle screw fixation, 407 perioperative complications, 413-414 principles, 408-409 technical aspects dorsal midline incision, 411 epidural bleeding, 412 facetectomy, 412 graft compression, 413 surgical steps, 411 wedge-shaped cages, 413 and TLIF, 415-416 Posterior subtraction osteotomy (PSO) complication, 742-743 flatback deformity, 749 levels, 747 localized deformity, 746, 747 spine shortening, 742 vs. SPO, 742 Posterolateral percutaneous diskectomy, 784-785 Post-lumbar surgery syndrome (PLSS) algorithmic approach, 543-544 anterior column appraoch intradiscal injections, 565 intradiscal thermal treatments, 564-565 definition, 526 etiologies, 533 imaging, 549-550 middle column appraoch adhesions, lysis of, 563-564 epidural steroid injections, 559-563 etiologies, 557-559 physical exam, 547-549 posterior column approach diagnostic blocks, 557 facet joint, 551, 552 myofascial pain, 556 radiofrequency ablation, 552-553 Revel's study, 551 sacroiliac joint pain, 553-556 superior cluneal neuropathy, 556-557 TDR, 552 Postoperative vision loss (POVL), 180-181 POVL. See Postoperative vision loss (POVL) PRO. See Patient-reported outcomes (PRO)

ProDisc-L, 613, 623 Prosthesis component motion patterns (PCMP), 621 Prosthetic disk nucleus (PDN) device, 436, 603, 628 Protein factors, 213 Provocative discography analgesic discography, 462-463 anatomic confirmation, 458 antibiotic prophylaxis, 459 asymptomatic disk, 458 contraindications, 459 CT/MRI discography, 462 disk pressure, 458 false positives, 460-461 fluoroscopic imaging, 458 immediate procedural complications, 459 indications, 459 internal controls, 459 intradiscal AP post-contrast, 460, 461 intradiscal AP pre-contrast, 460 intradiscal lateral post-contrast, 460 long-term complications, 459 potential intradiscal therapeutic agents, 463-464 pressure-controlled intradiscal injection, 460 therapeutic validity, 461-462 two-needle technique, 459 PSO. See Posterior subtraction osteotomy (PSO)

## Q

Quality-adjusted life years (QALYs) Cost/QALY, 198 direct cost, 191 health utilities, 185–188 ICER, 190 vs. incremental costs, 186 Quality of Well-Being Scale, 197 Quebec Back-Pain Disability Scale (QBPD), 581

#### R

Radiofrequency ablation (RFA), 447, 552-553 Range of motion (ROM), 207, 624, 637, 669, 671 Recombinant human bone morphogenetic protein (RhBMP), 188, 190, 335, 336, 813 Recombinant human growth and differentiation factor-5 (rhGDF-5), 335, 337 Recurrent disk herniation ALIF procedure, 283 diagnostic considerations, 280 incidence, 279 lumbar spine MRI, 284 prevention, 283 re-diskectomy and dorsoventral fusion, 285 surgical indication, 280-281 surgical techniques blunt dissection, 282 bony landmarks, 282 disk space height, 282 epidemiologic data, 279 incidence, 280

randomized controlled trials, 280 scar tissue, 282 subperiostal dissection, 282 Regain<sup>™</sup>, 629 Revision surgery strategy. See Recurrent disk herniation Rhizolysis, 220 RMDQ. See Roland-Morris Disability Questionnaire (RMDQ) Robotic spinal surgery applications, 592 computer-assisted systems, 598-599 da Vinci, 592 decision-making process, 598 financial investment, 598 fluoroscopic images, 595 Georgetown Robot, 592 image-guided surgery, 591-592, 595 instrumentation accuracy, 593-594 microsurgical dexterity, 592 microsurgical facetectomy, 595 minimally invasive surgery, 594-596 Miro system, 592 percutaneous procedures, 596 perform repetitive tasks, 592 postoperative CT scans, 595 preoperative planning, 595-597 radiation exposure, 593-594 RCT data, 596 Renaissance platform, 592 rod construct, 595 shared-control and hands-on robotic system, 598 small midline incision, 595 SpineAssist platform, 592, 597, 598 Roland-Morris Disability Questionnaire (RMDQ), 196, 197 ROM test. See Range of motion (ROM) Roussouly classification LL and PI relations, 710 LL and TK, 710 lumbar lordosis geometrical analysis, 707-708 SS values, 708-710

## S

Sacroiliac joint pain (SIJP), 205, 206, 553, 554 Sagittal balance, 725–726 in asymptomatic population, 726 bipedalism, 703–705 Cobb method, 712–713 compensatory mechanisms, 715–716 adjacent segments, hyperextension of, 731–733 algorithm, 731, 736 cervical hyperlordosis, 728–729 concept, 727–728 lower limbs, 735 pelvis, 734–735 positional parameters, 728, 730 retrolisthesis, 733, 734 sagittal imbalance, 727, 730 Sagittal balance (cont.) spinopelvic parameter changes, 726 theoretical pelvic tilt and lumbar lordosis, 728, 731 thoracic kyphosis reduction, 729, 730, 731 C7 plumb line maintenance, 703-704 global balance, 726-729 global balance, of spine, 710-712 gravity stress, on spine, 716 KEOPS software, 713 lumbar lordosis and bipedalism, 704-705 distribution values of, 713-714 geometrical analysis, 707-709 spine fusion, 721-722 pelvic parameters, 705-706, 713-714 Roussouly classification (see Roussouly classification) sagittal imbalance and clinical symptoms, 714-715 sagittal spinopelvic alignment, 704 spinal parameters, 706-708 spinopelvic morphology and MRI asymptomatic population, 720-721 contact forces, 716-717 modes, 717, 719 Roussouly hypothesis, 716 type 1 lordosis, 717-720 type 2 lordosis, 717-721 type 3 lordosis, 717, 718, 719 type 4 lordosis, 717, 718, 719 Sagittal imbalance anterior approach, 749 clinical assessment, 739, 740 closed wedge osteotomy, 739 complementary approach, 749 definition, 739 flatback posture, 739 flexion-extension films, 746 impaction bone grafting, 749 intraoperative incidents, 748 lateral full spine X-rays, 746 lower lumbar spine, 750 operative time and blood loss, 748 PIO bone impactors size, 744 bone resection, 743 cancellous bone, 743 corporeal osteotomy, 743-744 decancellation, 744 knife dissection, 743 levels, 747 localized deformity, 746 mechanical stability, 744-745 partial pedicle resection, 745 patient, in radiolucent table, 743 pedicle-type distracting clamp, 744, 745 peroperative view, 744 posterior incision, 743 postoperative, 745 principles of, 743

root entrapment/impingement, 744 spinal fixation, 745 spreading meary-type clamp, 744, 745 stiff deformities, 746 wedge creation, 744 Ponte osteotomy, 742, 746 postoperative complications, 748 postoperative radiological outcomes, 749 preoperative radiological data, 748 pre-op kyphotic deformity, 747, 748 PSO complication, 742-743 flatback deformity, 749 levels, 747 localized deformity, 746, 747 spine shortening, 742 vs. SPO, 742 radiological assessment, 739-741 sagittal alignment, 750 spinal fusion, 747 SPO, 739, 741, 750 X-ray, 750 Sagittal vertical axis (SVA), 519, 711–712, 726, 728 SCB. See Substantial clinical benefit (SCB) Sciatica acute back pain, 755 and back pain, 32-33 historical evolution ancient Hebrew medicine, 13 arthroplasty approach, 20 artificial implant, 20 chemical nucleolysis, 19 Cotugno, Domenico, 14 Cotugno's syndrome, 15 disk displacement, 15 Galen, 14 gas myelography, 18 Hippocrates, 14 hypesthesia/cutaneous anesthesia, 17 intervertebral disk, 15 landmark study, 19 Lasègue Forst sign, 16 Lasègue maneuver, 16 Medieval and Renaissance periods, 14 Mixter and Barr's diagnosis and laminectomy technique, 19 spinal conditions, 17 spinal fusion, 19-20 Valleix's points, 17 Vesalius, Andreas, 14 inflammatory irritation, 252 side-firing Ho: YAG laser, 788 Scoliosis Research Society-22R (SRS-22R), 196-198 Selective serotonin reuptake inhibitors (SSRIs), 211 Senescence accelerated mice (SAM), 56 Serotonin and norepinephrine reuptake inhibitors (SNRIs), 211 Sickle tail (SKT) gene, 82 SIJP. See Sacroiliac joint pain (SIJP)

Single-photon emission computed tomography (SPECT), 446, 550 Smith-Petersen osteotomy (SPO), 739, 741, 742, 750 SNRIs. See Serotonin and norepinephrine reuptake inhibitors (SNRIs) Somatic referred pain, 229 Spinal canal stenosis, 230-231, 240, 482, 484-486 Spinal cord stimulation (SCS), 294 Spinal fusion, 807 autologous bone, 331 biology, 331-332 definition, 331 iliac crest bone graft, 20 lumbar degenerative disk disease, 19 and spine injections analgesic-related deaths, 536 clinical outcomes, 535 costs, 535 policymakers and payors, 536 spinal interventionalist, 536-537 Sunshine Act, 536 Spinal instrumentation technology, 805 cervical, 811-812 electrolysis and new materials development, 806-807 fusion grafts and BMPs, 812-813 history, 806 pedicle screw development, 809-810 segmental spinal fixation, 809-810 spinal fusion, 807 spinal stabilization dynamic stabilization systems, 810-811 Harrington's approach, 807-809 Spinal manipulation therapy, 212 Spinal pain acute low back pain with/without sciatica, 757 causes, 756 clinical features, 758 conservative therapy, 757-758 diagnostic blocks facet blocks, 760 root blocks, 760 sinuvertebral block, 760-761 diskogenic pain and treatment, 764-765 failed back syndrome, 761-762 guidelines, 765-766 imaging correlations, 758-759 minimally invasive procedures, 764 pain recover, 758 pain-relieving procedures, 762 patient evaluation, 756-757 psychosocial features, chronic pain, 762-763 surgical decision, 761 Tarlov cyst, 763-764 Spinal stenosis non-instrumented lumbar fusion, 540-541 Spine Tango registry, 802 Spinal surgery anterior trans and retroperitoneal approaches, 771 artificial disk, 773

automated percutaneous lumbar diskectomy (APLD), 771 chemonucleolysis, 770 endoscopic procedure, 771 failed back surgery syndrome, 769 Harrington rods and Luque technique, 769 neurosurgical and orthopedic resident, 770 patient selection, 769 posterolateral percutaneous intradiscal injection, 770 randomized controlled trials (RCT), 774 registry criticism of, 800-801 financing models, 800-801 observational vs. experimental, 795-797 outcome research, 794 patient care, 793 prediction models, 802 quality control, 794 quality, in health care, 793-794 Spine Tango (see Spine Tango registry) spinal fusion techniques, 771 spinal tumors and metastases, 769 spine arthroplasty, 773 surgical goal, 771 surgical treatment, 769 total disk replacement, 773 Spine Arthroplasty Society (SAS), 773-774 Spine Outcomes Research Trial (SPORT), 263, 295, 315, 357, 514, 540, 545, 683 Spine registries, 258-259 Spine surgery intraoperative period complications, 780 surgical team, 779 unexpected intraoperative findings, 779-780 postoperative period, 780 preoperative period education, 778-779 nonsurgical management, 778 Spine Tango registry benefits of, 799-800 content, 798-799 disk herniation and spinal stenosis, 802 history and objectives, 796-798 technology, 799 Spondylolisthesis, 227–229, 240, 517–518 SRS-22R. See Scoliosis Research Society-22R (SRS-22R) STABILIMAX® system, 659 Substantial clinical benefit (SCB), 199 Superior articular process (SAP), 443, 444 Supraspinous ligament (SSL), 670 SWISSspine registry, 801 Synovial cyst, 227, 228 Synthetic bone graft, 334-335

#### Т

TCAs. See Tricyclic antidepressants (TCAs) TENs. See Transcutaneous electrical stimulation (TENs) Thrombospondin 2 (THBS2) gene, 82 Total disk replacement (TDR), 614 AcroFlex, 605 Activ-L Artificial Disk, 618-620 adjacent-segment degeneration, 622-624 advantages, 615 center of rotation, disk replacement, 620 complications, 607, 627 constrained implants, 606 FDA-approved artificial disk devices, 616 Fernstrom ball, 615 Flexicore artificial disk, 618, 619 implant-bone interface, stability of, 625 indications, 615 intervertebral disk, 620 Kineflex artificial disk, 618 load sharing, 620-621 Maverick artificial disk, 606, 617 motion patterns, 621-622 motion preservation, 624-625 operative procedure, 615-616 ProDisc, 605 ProDisc-L, 615-617 SB Charité III, 603, 604 SB Charité III (DePuy Spine), 616 shock absorption capacity, 626 success rates, 606 Swedish prospective randomised controlled trial, 607 Swiss registry, 606-607 type I Charité artificial disks, 603, 604 type II Charité artificial disks, 603 type III Charité artificial disks, 603, 604 wear testing, 625-626 Traditional lumbar fusion techniques, 613 Transcutaneous electrical stimulation (TENs), 212 Transforaminal epidural steroid injection (TFESI), 449-450 anesthetic test dose, 452 artery of Adamkiewicz, 452 bleeding, 451 common side effects, 450-451 digital subtraction angiography, 452 doses and injections number, 452-453 dural puncture, 452 efficacy of, 454-456 infection, 451 intermittent fluoroscopy, 452 local anesthetic, 450 needle placement, 450 predictors, 456-457 TF AP post-contrast injection, 450, 451 TF AP view, 450, 451 TF lateral view, 450, 451 transforaminal oblique view, 450 Transforaminal lumbar interbody fusion (TLIF), 359 advantages, 503 cage placement and grafting, 508-510 clinical outcomes, 510 complications, 360-361, 510 contraindications, 503-504

CT navigation system, 510-511 disadvantages, 503, 510 expendable retractor, 505-507 facet screws, 357 indications, 503-504 lumbar scoliosis correction, 519-521 minimally invasive variants, 356 mini-open and tubular variations, 360 mini-open technique, 356 osseous exposure, 358 patient positioning, 504 pedicle screws placement, 505 and PLIF, 415-416 rhBMP-2, 357 spondylolisthesis, 517-518 surgical robots, 594 transforaminal approach, 507-508 Transpsoas approach anatomy external and internal oblique, 384 L1-L2 disk space, 387 L4-L5 disk space, 387 lumbar plexus, 386 psoas muscle, 385 retroperitoneal fat, 384 zones, 386 complications, 390-391 indications, 384, 385, 386 minimally invasive transpsoas approach, 383 operative technique fluoroscopic localization, 388-389 neuromonitoring setup, 387 patient positioning, 388 preoperative planning, 387 surgical technique, 389-390 Tricyclic antidepressants (TCAs), 211 **TWINFLEX®**, 654-655

## U

Ultra-high-molecular-weight polyethylene (UHMWPE), 604 Ultrasound, 238–239 Upper instrumented vertebra (UIV), 696, 698, 699 Upper limbs, 9–10, 705

## v

Variable number of tandem repeats (VNTR), 64, 69, 79 Vertebral column, 5 Visual analogue scale (VAS), 129, 131, 197, 686

## W

Waddell signs, 208 Wallis device, 666, 670, 682, 688 Wallis system, 668–669 first-generation, 666 flexion-extension, 670 flexion-limiting bands, 670 indications and contraindications, 667 L4–L5, 669 UniWallis implant, 667 Wolff's law, 160

Х

X-Stop device, 682, 685-687

#### Y

Years lived with a disability (YLD), 527 Yeung Endoscopic Spine System (YESS<sup>™</sup>), 785 YODA Project, 337

## Z

Zurich Claudication Questionnaire (ZCQ), 686 Zygapophyseal/zygapophysial joints asymmetric load, 227, 228 cause of, 442 conservative treatment, 233 contraindications, 443 cryoneurolysis, 237 DDD, 226-227 definition, 219 degenerative changes, 225-227 diagnosis, 442 arthrography, 233 clinical findings, 231 intra-articular blocks, 233 medial branch blocks, 232 radiologic findings, 231-232 in elderly patient, 240 facet cyst injections, 448-449 failed back surgery, 240-241 history, 220 inflammatory arthritides, 229 injuries, 229 innervation, 223 intra-articular joint injection hyaluronic acid, 446 intra-articular corticosteroid/saline, 446 ISIS guidelines, 443 8 mL of injectate, 445

nonionic contrast, 443, 444 SPECT imaging, 446 spinal needle insertion, 443, 444 intra-articular steroids, 238 ioint articular facets, 221, 222 fibrous capsule, 221, 223 fluid, 220 functional spine unit/motion segment, 220 interbody joint, 220 intra-articular structures, 221-222 vertebral endplates, 220 medial branch blocks AP and lateral views, 444 false negatives, 448 false positives, 447-448 ISIS guidelines, 445 local anesthetic, 444 pre-and post-procedure pain level, 445 RFA, 447 SAP, 444 technical complications, 445 venous flow, absence of, 444, 445 physiology axial compression/distraction, 224 axial rotation, 225 extension, 224-225 flexion, 224, 225 lateral flexion, 225 prevalence of, 239 radiofrequency neurotomy historical development of, 235-236 medial branch (see Medial branch neurotomy) patient selection, 234, 236 percutaneous denervation procedures, 233 thermal radiofrequency neurotomy, 233-234 radiologic imaging, 238-239 relevant anatomy, 442-443 risk factors, 442 in spinal canal stenosis, 240 spondylolisthesis, 227-229, 240 surgery, 239 symptoms, 229-231 synovial cyst, 227, 228, 238 therapeutic joint blocks, 237